

SPECIFICATIONS

Warner Park Community Recreation Center Expansion

City of Madison Parks Division

Madison, Wisconsin

Exhibit C - Specifications

Contract #9502 Munis #17196

Engberg Anderson Project No. 223471.00

May 16, 2024

CIVIL CHARLE ъ JSD PROFESSIONAL SERVICES, INC. GROENIER 507 W VERONA AVE #500 E-50106-6 υ **VERONA**, WI 53593 LODI, PH 608-848-5060 WI h TTTTLANT LANT munnin S arner Par LANDSCAPE EVIN J. JSD PROFESSIONAL SERVICES, INC. YESKA 507 W VERONA AVE #500 LA-806 **VERONA, WI 53593** ADISON. V PH 608-848-5060 WILLIAM G ROBISC A7-Thinny Thunnan Kum ARCHITECTURAL **ENGBERG ANDERSON, INC. 305 W WASHINGTON AVE** MADISON, WI 53703 2024 PH 608-250-0100 ALL ARCHITECHINA THOMAS JUNGBLUTH E:100239-6 MADISON THE CER **STRUCTURAL ONEIDA TOTAL INTEGRATED ENTERPRISES** 1033 N MAYFAIR RD #200 MILWAUKEE, WI 53226 PH 414-257-4200 5/16/24 2hun

CITY OF MADISON SPECIFICATION May 16, 2024

FIRE PROTECTION JDR ENGINEERING, INC. 5525 NOBEL DR #110 MADISON, WI 53711 PH 608-277-1728 CARLOS F RUEF 11111 D 2305 P MOUNT HOREB PLUMBING JDR ENGINEERING, INC. 5525 NOBEL DR #110 11111 MADISON, WI 53711 05/16/2024 PH 608-277-1728 NICHOLAS MECHANICAL CLEAVER JDR ENGINEERING, INC. 5525 NOBEL DR #110 E-43972 MADISON, WI 53711 MADISON PH 608-277-1728 05/16/2024 THIN **ELECTRICAL** JDR ENGINEERING, INC. 5525 NOBEL DR #110 MADISON, WI 53711 PH 608-277-1728 05/16/2024

WARNER PARK COMM AND REC CENTER CONTRACT #9502 MUNIS #17196

WARNER PARK COMMUNITY RECREATION CENTER EXPANSION

TABLE OF CONTENTS

SECTION	TITLE	PAGES
TOC	Table of Contents	6
DIVISION 00	PROCUREMENT AND CONTRACT REQUIREMENTS	
00 31 46	Permits	1
00 43 25	Substitution Request Form (During Bidding)	3
00 43 43	Wage Rates Form	2
00 62 76.13	Sales Tax Form	1
DIVISION 01	GENERAL REQUIREMENTS	
01 25 13	Product Substitution Procedures	3
01 26 13	Request for Information (RFI)	2
01 26 46	Construction Bulletin (CB)	2
01 26 57	Change Order Requests (COR)	5
01 26 63	Change Order (CO)	2
01 29 73	Schedule of Values	3
01 29 76	Progress Payment Procedures	5
01 31 13	Project Coordination	3
01 31 19	Project Meetings	3
01 31 23	Project Management Web Site	3
01 32 16	Construction Progress Schedules	2
01 32 19	Submittals Schedule	3
01 32 23	Survey and Layout Data	3
01 32 26	Construction Progress Reporting	2
01 32 33	Photographic Documentation	2
01 33 23	Submittals	3
01 43 39	Mockups	3
01 43 50	Air Barrier Systems	4
01 45 16	Field Quality Control Procedures	3



SECTION	TITLE	PAGES
01 45 29	Testing Laboratory Services	4
01 50 00	Temporary Facilities and Controls	5
01 58 13	Temporary Project Signage	1
01 60 00	Product Requirements	4
01 71 23	Field Engineering	1
01 73 29	Cutting and Patching	3
01 74 13	Progress Cleaning	4
01 74 19	Construction Waste Management and Disposal	6
01 76 00	Protecting Installed Construction	6
01 77 00	Closeout Procedures	4
01 78 13	Completion and Correction List	1
01 78 23	Operation and Maintenance Data	4
01 78 36	Warranties	6
01 78 39	As-Built Drawings	5
01 78 43	Spare Parts and Extra Materials	3
01 79 00	Demonstration and Training	4
01 81 13	Sustainable Design Requirements - LEED for New Construction V4.0	13
01 91 00	Commissioning	8
01 95 00	Measurement and Verification	2
01 95 01	Monitoring-Based Commissioning	3
DIVISION 02	EXISTING CONDITIONS	
02 20 00	General Sitework Requirements	4
02 32 00	Geotechnical Investigation	32
02 41 19	Selective Demolition	3
DIVISION 03	CONCRETE	
03 10 00	Concrete Forming and Accessories	4
03 20 00	Concrete Reinforcing	4
03 30 00	Cast-In-Place Concrete	15
03 33 00	Architectural Concrete	7
DIVISION 04	MASONRY	
04 20 00	Unit Masonry	9
DIVISION 05	METALS	
05 12 00	Structural Steel Framing	5
05 31 00	Steel Decking	4
05 40 00	Cold-Formed Metal Framing	6
05 44 00	Cold-Formed Metal Trusses	5



SECTION	TITLE	PAGES
DIVISION 06	WOOD	
06 10 00	Rough Carpentry	3
06 20 23	Interior Finish Carpentry	2
DIVISION 07	THERMAL & MOISTURE PROTECTION	
07 11 13	Bituminous Dampproofing	2
07 21 00	Thermal Insulation	2
07 27 26	Fluid-Applied Membrane Air Barrier	2
07 41 13.16	Standing-Seam Metal Roof Panels	6
07 42 13.13	Formed Metal Wall Panels	3
07 42 93	Soffit Panels	3
07 53 23	Ethylene-Propylene-Diene-Monomer (EPDM) Roofing	6
07 62 00	Sheet Metal Flashing and Trim	8
07 72 53	Snow Guards	2
07 84 13	Penetration Firestopping	2
07 84 43	Joint Firestopping	3
07 92 00	Joint Sealants	5
07 92 19	Acoustical Joint Sealants	2
DIVISION 08	DOORS & WINDOWS	
08 11 13	Hollow Metal Doors and Frames	4
08 14 16	Flush Wood Doors	4
08 31 13	Access Doors and Frames	2
08 41 13	Aluminum-Framed Entrances and Storefronts	5
08 44 13	Glazed Aluminum Curtain Walls	4
08 71 00	Door Hardware	8
08 80 00	Glazing	5
08 83 00	Mirrors	2
08 91 19	Fixed Louvers	2
DIVISION 9	FINISHES	
09 22 16	Non-Structural Metal Framing	4
09 29 00	Gypsum Board	3
09 30 13	Ceramic Tiling	4
09 65 13	Resilient Base and Accessories	2
09 65 16	Resilient Sheet Flooring	2
09 66 23	Resinous Matrix Terrazzo Flooring	3
09 68 13	Tile Carpeting	2
09 91 23	Interior Painting	3
09 93 00	Staining and Transparent Finishing	2



SECTION	TITLE	PAGES
DIVISION 10	SPECIALTIES	
10 14 23	Panel Signage	3
10 21 13.19	1 13.19 Plastic Toilet Compartments	
10 28 00	Toilet, Bath, and Laundry Accessories	3
10 44 13	Fire Protection Cabinets	2
10 44 16	Fire Extinguishers	1
10 51 00	Plastic Lockers	3
DIVISION 11	EQUIPMENT	
11 66 23	Gymnasium Equipment	3
11 66 53	Gymnasium Dividers	2
DIVISION 12	FURNISHINGS	
12 36 61.16	Solid Surfacing Countertops	1
DIVISION 21	FIRE SUPPRESSION	
21 05 00	Common Work Results for Fire Suppression	9
21 05 29	Hangers and Supports for Fire-Suppression Piping and Equipment	5
21 10 00	Water-Based Fire-Suppression Systems	7
DIVISION 22	PLUMBING	
22 05 00	Common Work Results for Plumbing	13
22 05 14	Plumbing Specialties	2
22 05 29	Hangers and Supports for Plumbing Piping and Equipment	5
22 07 00	Plumbing Insulation	3
22 11 00	Facility Water Distribution	4
22 13 00	Facility Sanitary Sewerage	3
22 14 00	Facility Storm Drainage	3
22 14 05	Foundation Drainage	2
22 40 00	Plumbing Fixtures	3
22 42 23	Commercial Showers	2
DIVISION 23	HEATING VENTILATING AND AIR CONDITIONING	
23 05 00	Common Work Results for HVAC	9
23 05 13	Common Motor Requirements for HVAC Equipment	3
23 05 14	Variable Frequency Drives	5
23 05 15	Piping Specialties	5
23 05 23	General-Duty Valves for HVAC Piping	4
23 05 29	Hangers and Supports for HVAC Piping and Equipment	4
23 05 48	Vibration and Seismic Controls for HVAC Piping and Equipment	4
23 05 93	Testing, Adjusting, and Balancing for HVAC	5

	1	
	1	
	1	

SECTION	TITLE	PAGES
23 07 00	HVAC Insulation	8
23 09 14	Pneumatic and Electric Instrumentation and Control Devices for HVAC	10
23 09 24	Direct Digital Control System for HVAC	14
23 09 93	Sequence of Operations for HVAC Controls	4
23 21 13	Hydronic Piping	10
23 21 23	Hydronic Pumps	3
23 21 26	Ground Source Closed Circuit Vertical Exchanger Field	6
23 25 00	HVAC Water Treatment for Closed Loop Hydronic Systems	4
23 31 00	HVAC Ducts and Casings	7
23 33 00	Air Duct Accessories	6
23 34 00	HVAC Fans	3
23 36 00	Air Terminal Units	3
23 37 13	Diffusers, Registers & Grilles	3
23 41 00	Particulate Air Filtration	3
23 57 00	Heat Exchangers for HVAC	2
23 64 23	Modular Heat Recovery Chillers	8
23 73 12	Air Handling Unit Coils	2
23 73 13	Modular Indoor Central-Station Air-Handling Units	4
23 82 00	Heating and Cooling Terminal Units	2
DIVISION 26	ELECTRICAL	
26 05 00	Common Work Results for Electrical	7
26 05 02	Electrical Demolition for Remodeling	2
26 05 04	Cleaning, Inspection, and Testing of Electrical Equipment	2
26 05 19	Low-Voltage Electrical Power Conductors and Cables	4
26 05 23	Control-Voltage Electrical Power Cables	3
26 05 26	Grounding and Bonding for Electrical Systems	4
26 05 29	Hangers and Supports for Electrical Systems	3
26 05 33	Raceway and Boxes for Electrical Systems	8
26 05 53	Identification for Electrical Systems	3
26 24 16	Panelboards	2
26 27 02	Equipment Wiring Systems	3
26 27 26	Wiring Devices	6
26 27 28	Disconnect Switches	2
26 31 00	Photovoltaic System Performance Requirements	6
26 51 13	Interior Lighting Fixtures	6
DIVISION 27	COMMUNICATIONS	
27 00 05	Structured Communications Cabling	18
27 21 33	Wireless Access Points (WAP)	2

]	

SECTION	TITLE	PAGES
DIVISION 28	ELECTRONIC SAFETY AND SECURITY	
28 46 21	Addressable Fire-Alarm Systems	12
DIVISION 31	EARTHWORK	
31 05 00	Common Work Results for Earthwork	4
31 10 00	Site Clearing	2
31 20 00	Earthmoving	8
31 22 16.15	Subgrade Preparation	4
31 23 16.13	Trenching	4
31 25 00	Erosion Control	4
31 32 00	Soil Stabilization	2
DIVISION 32	EXTERIOR IMPROVEMENTS	
32 05 00	Common Work Results for Exterior Improvements	3
32 11 23	Aggregate Base Course	2
32 12 16.13	Plant-Mix Asphalt Paving	6
32 16 00	Concrete Pavement, Curb, and Sidewalks	5
32 17 23	Pavement Markings	2
32 91 13.50	Stormwater Rain Garden	4
32 91 19	Topsoil-Select Fill Materials and Application	2
32 92 00	Trees, Shrubs, and Other Plantings	5
32 92 19	Seeding and Sodding	4
DIVISION 33	UTILITIES	
33 30 00	Sanitary Sewer Utilities	7
33 40 00	Storm Sewer Construction	6
	END OF TABLE OF CONTENTS	



		SECTION 00 31 46 PERMITS
PART	1 – G	ENERAL
-	L.1.	
-	L.2.	
DADT	נ. 3. בי בי בי	GENERAL CONTRACTORS REQUIREMENTS
PART	2 – FI 3 – EX	(ECUTION – THIS SECTION NOT USED
PART	<u>1 – G</u>	ENERAL
1.1.	SUN	/MARY
	Α.	Each project has varying requirements for permits, inspections, and fees based on the scope, size, and locatic
		of the project.
	В.	The City of Madison (Owner) is subject to all permits, inspections and associated fees for construction,
		demolition, utility connection, storm water management, and other similar requirements that may be require
		to complete the scope of work associated with these contract documents.
	C.	The General Contractor (GC) shall be responsible for obtaining all permits, inspections and paying for all
		associated fees unless specifically identified within this specification.
1.2.	A KEF	EKENCES The following references are not intended to be all inclusive. It shall be the CC's responsibility to determine a
	А.	requirements based on the scope of work in the contract documents
	в	City of Madison Ordinances: Review all ordinances that may require a permit or fee that may be connected y
	υ.	a required permit. Contact the following City Agencies to determine the exact requirements during bidding
		1. Building Inspection
		2. Zoning
		3. Engineering
		4. Water Utility
		5. Traffic Engineering
		6. Others as may be specified by the contract documents.
	В.	State Statutes
	C.	Other Regulatory Regulations
	D.	Other Agencies or companies that may have related requirements
		1. Madison Metropolitan Sewerage District
		2. Local gas and electric utility companies
		3. Other utility companies
1.3.	GEN	IERAL CONTRACTORS REQUIREMENTS
	Α.	The GC shall be responsible for all of the following:
		1. Execute application for all required permits as may be required by the scope of work described within
		contract documents.
		 Scheduling all required inspections that may be conditions of any required permits. Device for other according to the second secon
		 Paying for other permits not explicitly stated as excluded in this section. The CC is not representible for neuring for the City Building, City Blackring, City Blackring, Medicen F
	в.	The GC is not responsible for paying for the City Building, City HVAC, City Electrical, City Plumbing, Madison F
	C	The GC shall provide high quality scanned images of all required permits and inspections and upload them to
	с.	Contract Documents-Regulatory Documents Library on the Project Management Web Site.
DADT	2 - P	
<u>ran i</u>	2-2	NODOCTS - THIS SECTION NOT USED
PART	3 – E	KECUTION – THIS SECTION NOT USED
		END OF SECTION

			SUBSTITUTION REQUEST FORM (DURING BIDDING)
	1 0		
PARI	1 1		
	1.1. 1.2		
	т.с. го п		
	2 - P		
PARI	3-E/ 31		
	5.1. 2.2		SING A SUBSTITUTION DURING BIDDING
•	2.2. 2.2		
	3.3. 3.4.	SUBSTIT	UTION REQUEST FORM
PART	1-6	ENERAL	
1.1.	SUI		
	Α.	The C	ity of Madison uses a specific list of preferred products for various specification items to establish
		stand	lards of quality, utility, and appearance required.
	В.	The C	ity of Madison will not allow substitutions for specified Products except as follows:
		1.	The Product is no longer produced or the product manufacturer is no longer in business.
		2.	The manufacturer has significantly changed performance data, product dimensions, or other such de
			criteria for the specified Product(s).
		3.	Products specified by naming one or more Products or manufacturer's and "or approved equal" or
	~		"approved equivalent."
	C.	The p	rocedures in this specification shall apply to all proposals by Contractors, Suppliers, Vendors, and
		Manı	ifacturers when the conditions in item 1.1.B. above have been met during the bidding phase.
1.2.	REL	ATED SPI	ECIFICATIONS
1.2. <u>PART</u>	REI A. 7 <u>2 - P</u>	ATED SPI 01 25 RODUCTS	ECIFICATIONS 13 Product Substitution Procedures 5 – THIS SECTION NOT USED
1.2. <u>PART</u> PART	REI A. 7 <u>2 – P</u> 7 <u>3 - E</u>	ATED SPI 01 25 RODUCTS	ECIFICATIONS 13 Product Substitution Procedures 5 – THIS SECTION NOT USED N
1.2. <u>PART</u> <u>PART</u> 3.1.	REI A. <u>2 – P</u> <u>3 - E</u> REC	ATED SPI 01 25 RODUCTS XECUTION	ECIFICATIONS 13 Product Substitution Procedures 5 – THIS SECTION NOT USED N S A SUBSTITUTION DURING BIDDING
1.2. <u>PART</u> PART 3.1.	REI A. <u>2 – P</u> <u>3 - E</u> REC A.	ATED SPI 01 25 RODUCTS XECUTION QUESTING In the	ECIFICATIONS 13 Product Substitution Procedures 5 – THIS SECTION NOT USED N S A SUBSTITUTION DURING BIDDING e event that a substitution is requested during the bidding phase the Contractor, Supplier, Vendor, or
1.2. <u>PART</u> PART 3.1.	REL A. <u>2 – P</u> <u>3 - E2</u> REC A.	ATED SPI 01 25 RODUCT EXECUTION QUESTING In the Manu	ECIFICATIONS 13 Product Substitution Procedures 5 – THIS SECTION NOT USED N S A SUBSTITUTION DURING BIDDING e event that a substitution is requested during the bidding phase the Contractor, Supplier, Vendor, or ufacturer shall do all of the following:
1.2. <u>PART</u> PART 3.1.	REL A. <u>2 – P</u> <u>3 - E</u> REC A.	ATED SPI 01 25 RODUCTS ERODUCTS EXECUTION QUESTING In the Manu 1.	ECIFICATIONS 13 Product Substitution Procedures 5 – THIS SECTION NOT USED S – SUBSTITUTION DURING BIDDING e event that a substitution is requested during the bidding phase the Contractor, Supplier, Vendor, or ifacturer shall do all of the following: Submit a Substitution Request Form for each different product. Use a printed/scanned copy of the following:
1.2. <u>PART</u> PART 3.1.	REL A. <u>2 – P</u> <u>3 - E2</u> REC A.	ATED SPI 01 25 RODUCTS ERODUCTS EXECUTION QUESTING In the Manu 1.	IS A SUBSTITUTION DURING BIDDING e event that a substitution is requested during the bidding phase the Contractor, Supplier, Vendor, or ufacturer shall do all of the following: Submit a Substitution Request Form for each different product. Use a printed/scanned copy of the for at the end of this specification as a cover sheet.
1.2. <u>PART</u> 9.1.	REL A. <u>2 – P</u> <u>3 - E2</u> REC A.	ATED SPI 01 25 RODUCTS XECUTION QUESTING In the Manu 1. 2.	IS Substitution Procedures Product Substitution Procedures S – THIS SECTION NOT USED Section NOT USED Section S
1.2. <u>PART</u> 9.1.	REL A. <u>2 – P</u> <u>3 - E2</u> REC A.	ATED SPI 01 25 RODUCTS XECUTION QUESTING In the Manu 1. 2.	THIS SECTION NOT USED Source Section NOT USED Source Section NOT USED Source Section NOT USED Source Section Sect
1.2. <u>PART</u> 3.1.	REL A. <u>2 – P</u> <u>3 - E2</u> REC A.	ATED SPI 01 25 RODUCTS XECUTION QUESTING In the Manu 1. 2.	THIS SECTION NOT USED Source Section NOT USED Source Section NOT USED Source Section NOT USED Source Section Sect
1.2. <u>PART</u> <u>PART</u> 3.1.	REL A. <u>2 – P</u> <u>3 - E</u> REC A.	ATED SPI 01 25 RODUCTS XECUTION QUESTING In the Manu 1. 2.	Product Substitution Procedures Product Substitution Procedures S – THIS SECTION NOT USED S – THIS SECTION NOT USED S – Substitution DURING BIDDING e event that a substitution is requested during the bidding phase the Contractor, Supplier, Vendor, or ufacturer shall do all of the following: Submit a Substitution Request Form for each different product. Use a printed/scanned copy of the for at the end of this specification as a cover sheet. Support your request with complete data, drawings, specifications, performance data and samples as appropriate. A complete submission shall include the following: a. Substitution Request Form as a cover sheet b Comparison of qualities of the proposed substitutions with that specified.
1.2. <u>PART</u> <u>PART</u> 3.1.	REL A. <u>7 2 - P</u> <u>7 3 - E)</u> REC A.	ATED SPI 01 25 RODUCTS XECUTION QUESTING In the Manu 1. 2.	 Product Substitution Procedures Product Substitution Procedures F - THIS SECTION NOT USED A SUBSTITUTION DURING BIDDING e event that a substitution is requested during the bidding phase the Contractor, Supplier, Vendor, or ufacturer shall do all of the following: Submit a Substitution Request Form for each different product. Use a printed/scanned copy of the for at the end of this specification as a cover sheet. Support your request with complete data, drawings, specifications, performance data and samples as appropriate. A complete submission shall include the following: a. Substitution Request Form as a cover sheet b. Comparison of qualities of the proposed substitutions with that specified. c. Changes required in other elements of the Work because of the substitution.
1.2. <u>PART</u> 9 <u>ART</u> 3.1.	REL A. <u>2 – P</u> <u>3 - E</u> REC A.	ATED SPI 01 25 RODUCTS XECUTION QUESTING In the Manu 1. 2.	 Product Substitution Procedures Product Substitution Procedures F - THIS SECTION NOT USED A SUBSTITUTION DURING BIDDING e event that a substitution is requested during the bidding phase the Contractor, Supplier, Vendor, or ufacturer shall do all of the following: Submit a Substitution Request Form for each different product. Use a printed/scanned copy of the for at the end of this specification as a cover sheet. Support your request with complete data, drawings, specifications, performance data and samples as appropriate. A complete submission shall include the following: a. Substitution Request Form as a cover sheet b Comparison of qualities of the proposed substitutions with that specified. c. Changes required in other elements of the Work because of the substitution. d. Effect on the construction schedule.
1.2. <u>PART</u> 9 <u>ART</u> 3.1.	REL A. <u>2 – P</u> <u>3 - E</u> REC A.	ATED SPI 01 25 RODUCTS XECUTION QUESTING In the Manu 1. 2.	ECIFICATIONS 13 Product Substitution Procedures S - THIS SECTION NOT USED N S - A SUBSTITUTION DURING BIDDING e event that a substitution is requested during the bidding phase the Contractor, Supplier, Vendor, or afacturer shall do all of the following: Submit a Substitution Request Form for each different product. Use a printed/scanned copy of the for at the end of this specification as a cover sheet. Support your request with complete data, drawings, specifications, performance data and samples as appropriate. A complete submission shall include the following: a. Substitution Request Form as a cover sheet b Comparison of qualities of the proposed substitutions with that specified. c. Changes required in other elements of the Work because of the substitution. d. Effect on the construction schedule. e. Cost data comparing the proposed substitution with the Product specified.
1.2. <u>PART</u> 9 <u>ART</u> 3.1.	REL A. <u>2 – P</u> <u>3 - E</u> REC A.	ATED SPI 01 25 RODUCTS XECUTION QUESTINC In the Manu 1. 2.	 Product Substitution Procedures 5 – THIS SECTION NOT USED 8 A SUBSTITUTION DURING BIDDING e event that a substitution is requested during the bidding phase the Contractor, Supplier, Vendor, or ufacturer shall do all of the following: Submit a Substitution Request Form for each different product. Use a printed/scanned copy of the for at the end of this specification as a cover sheet. Support your request with complete data, drawings, specifications, performance data and samples as appropriate. A complete submission shall include the following: a. Substitution Request Form as a cover sheet b. Comparison of qualities of the proposed substitutions with that specified. c. Changes required in other elements of the Work because of the substitution. d. Effect on the construction schedule. e. Cost data comparing the proposed substitution with the Product specified. f. Any required license fees or royalties.
1.2. <u>PART</u> 9 <u>ART</u> 3.1.	REL A. <u>3 - E7</u> REC A.	ATED SPI 01 25 RODUCTS XECUTION QUESTINC In the Manu 1. 2.	 Product Substitution Procedures FTHIS SECTION NOT USED A SUBSTITUTION DURING BIDDING e event that a substitution is requested during the bidding phase the Contractor, Supplier, Vendor, or ufacturer shall do all of the following: Submit a Substitution Request Form for each different product. Use a printed/scanned copy of the for at the end of this specification as a cover sheet. Support your request with complete data, drawings, specifications, performance data and samples as appropriate. A complete submission shall include the following: a. Substitution Request Form as a cover sheet b. Comparison of qualities of the proposed substitutions with that specified. c. Changes required in other elements of the Work because of the substitution. d. Effect on the construction schedule. e. Cost data comparing the proposed substitution with the Product specified. f. Any required license fees or royalties. g. Availability of maintenance service and source of replacement materials.
1.2. <u>PART</u> 9 <u>ART</u> 3.1.	REL A. <u>3 - E</u> REC A.	ATED SPI 01 25 RODUCTS XECUTION QUESTINC In the Manu 1. 2.	FOR Section Not Used For the construction of the proposed substitutions with that specified. Comparison of qualities of the proposed substitutions with that specified. Const data comparing the proposed substitutions with the Product specified. Cost data comparing the proposed substitution with the Product specified. Cost data comparing the proposed substitution with the Product specified. Any required license fees or royalties. g. Availability of maintenance service and source of replacement materials. Substitution Request Form and all required supporting documentation to the City Project
1.2. <u>PART</u> 9.1.	REL A. <u>3 - E</u> REC A.	ATED SPI 01 25 RODUCTS XECUTION QUESTINC In the Manu 1. 2.	 FIGURE CIFICATIONS Product Substitution Procedures FIGURE SECTION NOT USED FA SUBSTITUTION DURING BIDDING event that a substitution is requested during the bidding phase the Contractor, Supplier, Vendor, or ifacturer shall do all of the following: Submit a Substitution Request Form for each different product. Use a printed/scanned copy of the for at the end of this specification as a cover sheet. Support your request with complete data, drawings, specifications, performance data and samples as appropriate. A complete submission shall include the following: a. Substitution Request Form as a cover sheet b. Comparison of qualities of the proposed substitutions with that specified. c. Changes required in other elements of the Work because of the substitution. d. Effect on the construction schedule. e. Cost data comparing the proposed substitution with the Product specified. f. Any required license fees or royalties. g. Availability of maintenance service and source of replacement materials. Submit the Substitution Request Form and all required supporting documentation to the City Project Manager and Project Architect.
1.2. <u>PART</u> 9 <u>ART</u> 3.1.	REL A. <u>3 - E</u> REC A.	ATED SPI 01 25 RODUCTS RECUTION QUESTINC In the Manu 1. 2.	 Product Substitution Procedures THIS SECTION NOT USED A SUBSTITUTION DURING BIDDING event that a substitution is requested during the bidding phase the Contractor, Supplier, Vendor, or ifacturer shall do all of the following: Submit a Substitution Request Form for each different product. Use a printed/scanned copy of the for at the end of this specification as a cover sheet. Support your request with complete data, drawings, specifications, performance data and samples as appropriate. A complete submission shall include the following: a. Substitution Request Form as a cover sheet b. Comparison of qualities of the proposed substitutions with that specified. c. Changes required in other elements of the Work because of the substitution. d. Effect on the construction schedule. e. Cost data comparing the proposed substitution with the Product specified. f. Any required license fees or royalties. g. Availability of maintenance service and source of replacement materials. Submit the Substitution Request Form and all required supporting documentation to the City Project Manager and Project Architect. a. Submissions to be done as complete PDF files for each product, appropriately titled
1.2. <u>PART</u> 9ART 3.1.	REL A. <u>3 - E</u> REC A.	ATED SPI 01 25 RODUCTS RECUTION QUESTINC In the Manu 1. 2.	 FIGENERATIONS Product Substitution Procedures FIGENERATION NOT USED FASUBSTITUTION DURING BIDDING e event that a substitution is requested during the bidding phase the Contractor, Supplier, Vendor, or ifacturer shall do all of the following: Submit a Substitution Request Form for each different product. Use a printed/scanned copy of the for at the end of this specification as a cover sheet. Support your request with complete data, drawings, specifications, performance data and samples as appropriate. A complete submission shall include the following: a. Substitution Request Form as a cover sheet b. Comparison of qualities of the proposed substitutions with that specified. c. Changes required in other elements of the Work because of the substitution. d. Effect on the construction schedule. e. Cost data comparing the proposed substitution with the Product specified. f. Any required license fees or royalties. g. Availability of maintenance service and source of replacement materials. Submit the Substitution Request Form and all required supporting documentation to the City Project Manager and Project Architect. a. Submissions to be done as complete PDF files for each product, appropriately titled b. Email submissions to the Project Architect and City Project Manager at the email addresses
1.2. <u>PART</u> 9ART 3.1.	REL A. <u>3 - E</u> REC A.	ATED SPI 01 25 RODUCTS RECUTION QUESTINC In the Manu 1. 2.	 Product Substitution Procedures THIS SECTION NOT USED A SUBSTITUTION DURING BIDDING e event that a substitution is requested during the bidding phase the Contractor, Supplier, Vendor, or ifacturer shall do all of the following: Submit a Substitution Request Form for each different product. Use a printed/scanned copy of the for at the end of this specification as a cover sheet. Support your request with complete data, drawings, specifications, performance data and samples as appropriate. A complete submission shall include the following: a. Substitution Request Form as a cover sheet b. Comparison of qualities of the proposed substitutions with that specified. c. Changes required in other elements of the Work because of the substitution. d. Effect on the construction schedule. e. Cost data comparing the proposed substitution with the Product specified. f. Any required license fees or royalties. g. Availability of maintenance service and source of replacement materials. Submit the Substitution Request Form and all required supporting documentation to the City Project Manager and Project Architect. a. Submissions to be done as complete PDF files for each product, appropriately titled b. Email submissions to the Project Architect and City Project Manager at the email addresses provided on the last page of Section D of the contract documents.
1.2. <u>PART</u> 9ART 3.1.	REL A. <u>3 - E</u> REC A.	ATED SPI 01 25 RODUCTS RECUTION QUESTINC In the Manu 1. 2.	 Product Substitution Procedures THIS SECTION NOT USED A SUBSTITUTION DURING BIDDING event that a substitution is requested during the bidding phase the Contractor, Supplier, Vendor, or ufacturer shall do all of the following: Submit a Substitution Request Form for each different product. Use a printed/scanned copy of the for at the end of this specification as a cover sheet. Support your request with complete data, drawings, specifications, performance data and samples as appropriate. A complete submission shall include the following: a. Substitution Request Form as a cover sheet b. Comparison of qualities of the proposed substitutions with that specified. c. Changes required in other elements of the Work because of the substitution. d. Effect on the construction schedule. e. Cost data comparing the proposed substitution with the Product specified. f. Any required license fees or royalties. g. Availability of maintenance service and source of replacement materials. Submit the Substitution Request Form and all required supporting documentation to the City Project Manager and Project Architect. a. Submissions to be done as complete PDF files for each product, appropriately titled b. Email submissions to the Project Architect and City Project Manager at the email addresses provided on the last page of Section D of the contract documents. i. The subject line shall include the contract number and "Request for Substitution".
1.2. <u>PART</u> 9ART 3.1.	REL A. <u>3 - E</u> REC A.	ATED SPI 01 25 RODUCTS RECUTION QUESTINC In the Manu 1. 2.	 Product Substitution Procedures THIS SECTION NOT USED A SUBSTITUTION DURING BIDDING event that a substitution is requested during the bidding phase the Contractor, Supplier, Vendor, or ufacturer shall do all of the following: Submit a Substitution Request Form for each different product. Use a printed/scanned copy of the for at the end of this specification as a cover sheet. Support your request with complete data, drawings, specifications, performance data and samples as appropriate. A complete submission shall include the following: a. Substitution Request Form as a cover sheet b. Comparison of qualities of the proposed substitutions with that specified. c. Changes required in other elements of the Work because of the substitution. d. Effect on the construction schedule. e. Cost data comparing the proposed substitution with the Product specified. f. Any required license fees or royalties. g. Availability of maintenance service and source of replacement materials. Submit the Substitution Request Form and all required supporting documentation to the City Project Manager and Project Architect. a. Submissions to be done as complete PDF files for each product, appropriately titled b. Email submissions to the Project Architect and City Project Manager at the email addresses provided on the last page of Section D of the contract documents. i. The subject line shall include the contract number and "Request for Substitution". Example: Contract 1234 – Request for Substitution
1.2. <u>PART</u> 3.1.	REL A. <u>3 - E</u> REC A.	ATED SPI 01 25 RODUCTS RECUTION QUESTING In the Manu 1. 2. 3.	 Product Substitution Procedures THIS SECTION NOT USED A SUBSTITUTION DURING BIDDING e event that a substitution is requested during the bidding phase the Contractor, Supplier, Vendor, or ifacturer shall do all of the following: Submit a Substitution Request Form for each different product. Use a printed/scanned copy of the for at the end of this specification as a cover sheet. Support your request with complete data, drawings, specifications, performance data and samples as appropriate. A complete submission shall include the following: a. Substitution Request Form as a cover sheet. b. Comparison of qualities of the proposed substitutions with that specified. c. Changes required in other elements of the Work because of the substitution. d. Effect on the construction schedule. e. Cost data comparing the proposed substitution with the Product specified. f. Any required license fees or royalties. g. Availability of maintenance service and source of replacement materials. Submit the Substitution Request Form and all required supporting documentation to the City Project Manager and Project Architect. a. Submissions to be done as complete PDF files for each product, appropriately titled b. Email submissions to the Project Architect and City Project Manager at the email addresses provided on the last page of Section D of the contract documents. i. The subject line shall include the contract number and "Request for Substitution". Example: Contract 1234 – Request for Substitution
1.2. <u>PART</u> 3.1.	REL A. <u>3 - E</u> REC A.	ATED SPI 01 25 RODUCTS RECUTION QUESTING In the Manu 1. 2. 3.	 Product Substitution Procedures 5 - THIS SECTION NOT USED 6 A SUBSTITUTION DURING BIDDING event that a substitution is requested during the bidding phase the Contractor, Supplier, Vendor, or ifacturer shall do all of the following: Submit a Substitution Request Form for each different product. Use a printed/scanned copy of the for at the end of this specification as a cover sheet. Support your request with complete data, drawings, specifications, performance data and samples as appropriate. A complete submission shall include the following: a. Substitution Request Form as a cover sheet b. Comparison of qualities of the proposed substitutions with that specified. c. Changes required in other elements of the Work because of the substitution. d. Effect on the construction schedule. e. Cost data comparing the proposed substitution with the Product specified. f. Any required license fees or royalties. g. Availability of maintenance service and source of replacement materials. Submit the Substitution Request Form and all required supporting documentation to the City Project Manager and Project Architect. a. Submissions to be done as complete PDF files for each product, appropriately titled b. Email submissions to the Project Architect and City Project Manager at the email addresses provided on the last page of Section D of the contract documents. i. The subject line shall include the contract number and "Request for Substitution". Example: Contract 1234 – Request for Substitution

1	3.2.	SUBMISSION REVIEW
2		A. The Project Architect, City Project Manager, members of the design team, and the Owners staff shall review all
3		submissions for substitutions during the bidding phase.
4		
5	3.3.	SUBSTITUTION APPROVAL
6		A. All requests for substitutions that have been approved shall be published by Addenda to the bid documents.
7		· · · · · · · · · · · · · · · · · · ·
, 8		
0		NOTE SEE NEXT DAGE FOR SAMPLE SUBSTITUTION DECLIEST FORM
9 10		NOTE SEE NEXT FAGE FOR SAMPLE SUBSTITUTION REQUEST FORM.
11		
11		
12		
13		
14		
15		
16		
17		
18		
19		
20		
21		
22		
23		
24		
25		
26		
27		
28		
20		
29		
50 21		
31		
32		
33		
34		
35		
36		
37		
38		
39		
40		
41		
42		
43		
44		
45		
46		
47		
48		
10		
50		
50		
21		
J∠ ⊑2		
53 F4		
54		
55		
56		
57		
58		
	WARN	ER PARK COMMUNITY RECREATION CENTER
	EXPAN	ISION

3.4. SUBSTITUTION REQUEST FORM

For Pre-bid Substitution Requests all text boxes on this form are required information for a complete request.

Substitut	ion Request
Contract Number:	
eview the General Contractor aff	irms that all of the following statements are correct:
his request is in compliance with	the requirements described in Specification 01 25 13
of the proposed substitution are	a equal or superior to the specified item
ffect dimensions shown on the d	requirer superior to the specified item.
io adverse affects on other trades	s, the construction schedule, or any specified warranty
- In the second state of t	destablished and the second
e locally available for the propose	d substitution. (GC shall provide supporting documentation
onsible for any and all costs assoc	iated with this substitution request if approved. This
or building design, engineering d	esign tees, detailing tees, plan review tees, construction
GC Substitution Re	quest:
	Phone:
	Email:
	1
ER END OF SECTION	•
00 43 35 3	
T	END OF SECTION TER 00 43 25 - 3

		WAGE RATES FORM
PART	1 – GE	NERAL
-	1.1.	
	1.2.	RELATED SPECIFICATIONS
PART	2 – PR	ODUCTS – NOT USED
PART	3 - EXE	
	3.1.	GENERAL REQUIREMENTS
	3.2.	GENERAL CONTRACTORS RESPONSIBILITIES
PART	1 – GE	NERAL
1.1.	SUM	IMARY
	Α.	The Reimbursable Hourly Worksheet is a contractor provided document that indicates the basic rate of pay.
		fringe benefits, and each companies cost of required insurance for all Trades and Classifications that will be
		performing productive labor during the execution of this contract.
		1. Rates shall be similar to recognized rates published by the Bureau of Labor Statistics. Associated Gen
		Contractors (AGC), Associated Builders and Contractors (ABC), appropriate union contracts, and othe
		similar organizations or documents.
	B.	The Reimbursable Labor Rate Worksheet shall provide the basis for labor rates being used on Change Order
	Б.	Request forms
1.2.	RFLA	
	Δ	Section 01 26 57 Change Order Request
	R.	Section 01 29 76 Progress Payment Procedures
	C.	Section 01 31 23 Project Management Web Site (PMWS)
	с. Б	
	υ.	Section 01 32 19 Submittals Schedule
DADT	D.	Section 01 32 19 Submittals Schedule
PART	D. 2 – PR	Section 01 32 19 Submittals Schedule CODUCTS – NOT USED
<u>PART</u> PART	2 – PR 3 - EXI	Section 01 32 19 Submittals Schedule CODUCTS – NOT USED ECUTION
<u>PART</u> <u>PART</u> 3.1.	<u>2 – PR</u> <u>3 - EXI</u> GEN	Section 01 32 19 Submittals Schedule CODUCTS – NOT USED ECUTION ERAL REQUIREMENTS
<u>PART</u> <u>PART</u> 3.1.	D. 2 – PR 3 - EXI GEN A.	Section 01 32 19 Submittals Schedule CODUCTS – NOT USED ECUTION ERAL REQUIREMENTS Prior to the Pre-Construction Meeting the City Project Manager (CPM) or the City Construction Manager (CC
<u>PART</u> <u>PART</u> 3.1.	2 – PR 3 - EXI GEN A.	Section 01 32 19 Submittals Schedule CODUCTS – NOT USED ECUTION ERAL REQUIREMENTS Prior to the Pre-Construction Meeting the City Project Manager (CPM) or the City Construction Manager (CC shall provide the GC a copy of the <i>Reimbursable Labor Rate Worksheet.xls</i> .
<u>PART</u> PART 3.1.	D. <u>2 – PR</u> <u>3 - EXI</u> <u>GEN</u> A.	Section 01 32 19 Submittals Schedule CODUCTS – NOT USED ECUTION ERAL REQUIREMENTS Prior to the Pre-Construction Meeting the City Project Manager (CPM) or the City Construction Manager (CC shall provide the GC a copy of the <i>Reimbursable Labor Rate Worksheet.xls</i> . See the last page of this specification for an example of the worksheet
<u>PART</u> <u>PART</u> 3.1.	D. <u>2 – PR</u> <u>3 - EXI</u> <u>GEN</u> A. B.	Section 01 32 19 Submittals Schedule CODUCTS – NOT USED ECUTION ERAL REQUIREMENTS Prior to the Pre-Construction Meeting the City Project Manager (CPM) or the City Construction Manager (CC shall provide the GC a copy of the <i>Reimbursable Labor Rate Worksheet.xls.</i> 1. See the last page of this specification for an example of the worksheet. The GC shall provide all subcontractors that will be performing productive labor during the execution of this
<u>PART</u> <u>PART</u> 3.1.	D. <u>2 – PR</u> <u>3 - EXI</u> <u>GEN</u> A. B.	Section 01 32 19 Submittals Schedule
<u>PART</u> <u>PART</u> 3.1.	D. <u>2 – PR</u> <u>3 - EXI</u> <u>GEN</u> A. B.	Section 01 32 19 Submittals Schedule
<u>PART</u> <u>PART</u> 3.1.	D. <u>2 – PR</u> <u>3 - EXI</u> <u>GEN</u> A. B. C.	Section 01 32 19 Submittals Schedule
<u>PART</u> <u>PART</u> 3.1.	D. <u>2 – PR</u> <u>3 - EXI</u> <u>GEN</u> A. B. C.	Section 01 32 19 Submittals Schedule RODUCTS – NOT USED ECUTION ERAL REQUIREMENTS Prior to the Pre-Construction Meeting the City Project Manager (CPM) or the City Construction Manager (CC shall provide the GC a copy of the <i>Reimbursable Labor Rate Worksheet.xls</i> . 1. See the last page of this specification for an example of the worksheet. The GC shall provide all subcontractors that will be performing productive labor during the execution of this contract with additional copies of the worksheet as needed. All contractors shall be required to fill out and submit completed worksheets for all Trades and Classification labor that will be performing productive labor during the execution of this contract.
<u>PART</u> <u>9ART</u> 3.1. 3.2.	D. <u>2 – PR</u> <u>3 - EXI</u> <u>GEN</u> A. B. C. <u>GEN</u>	Section 01 32 19 Submittals Schedule
<u>PART</u> <u>PART</u> 3.1. 3.2.	D. <u>2 – PR</u> <u>3 - EXI</u> <u>GEN</u> A. B. C. <u>GEN</u> A.	Section 01 32 19 Submittals Schedule
<u>PART</u> <u>PART</u> 3.1. 3.2.	D. <u>2 – PR</u> <u>3 - EXI</u> GEN A. B. C. GEN A. B. B.	Section 01 32 19 Submittals Schedule SCODUCTS – NOT USED ECUTION ERAL REQUIREMENTS Prior to the Pre-Construction Meeting the City Project Manager (CPM) or the City Construction Manager (CC shall provide the GC a copy of the <i>Reimbursable Labor Rate Worksheet.xls</i> . 1. See the last page of this specification for an example of the worksheet. The GC shall provide all subcontractors that will be performing productive labor during the execution of this contract with additional copies of the worksheet as needed. All contractors shall be required to fill out and submit completed worksheets for all Trades and Classification labor that will be performing productive labor during the execution of this contract. ERAL CONTRACTORS RESPONSIBILITIES The GC shall provide the ITrades and Classifications into one master Excel Workbook of all trades. The GC shall provide the combined workbook as required by Section 1.6 of Specification 01 32 19 Submittals
<u>PART</u> <u>PART</u> 3.1. 3.2.	D. <u>2 – PR</u> <u>3 - EXI</u> GEN A. B. C. GEN A. B. B.	Section 01 32 19 Submittals Schedule SODUCTS – NOT USED ECUTION ERAL REQUIREMENTS Prior to the Pre-Construction Meeting the City Project Manager (CPM) or the City Construction Manager (CC shall provide the GC a copy of the <i>Reimbursable Labor Rate Worksheet.xls</i> . 1. See the last page of this specification for an example of the worksheet. The GC shall provide all subcontractors that will be performing productive labor during the execution of this contract with additional copies of the worksheet as needed. All contractors shall be required to fill out and submit completed worksheets for all Trades and Classification labor that will be performing productive labor during the execution of this contract. ERAL CONTRACTORS RESPONSIBILITIES The GC shall provide the ITrades and Classifications into one master Excel Workbook of all trades. The GC shall provide the combined workbook as required by Section 1.6 of Specification 01 32 19 Submittal: Schedule for review and approval by the Owners Representatives.
<u>PART</u> <u>9ART</u> 3.1. 3.2.	D. <u>2 – PR</u> <u>3 - EXI</u> GEN A. B. C. GEN A. B.	Section 01 32 19 Submittals Schedule SODUCTS – NOT USED ECUTION ERAL REQUIREMENTS Prior to the Pre-Construction Meeting the City Project Manager (CPM) or the City Construction Manager (CC shall provide the GC a copy of the <i>Reimbursable Labor Rate Worksheet.xls</i> . 1. See the last page of this specification for an example of the worksheet. The GC shall provide all subcontractors that will be performing productive labor during the execution of this contract with additional copies of the worksheet as needed. All contractors shall be required to fill out and submit completed worksheets for all Trades and Classification labor that will be performing the execution of this contract. ERAL CONTRACTORS RESPONSIBILITIES The GC shall provide the combined workbook as required by Section 1.6 of Specification 01 32 19 Submittals Schedule for review and approval by the Owners Representatives. 1. Submittal shall be an Exported PDF of the completed Excel Workbook.
<u>PART</u> <u>9ART</u> 3.1. 3.2.	D. <u>2 – PR</u> <u>3 - EXI</u> GEN A. B. C. GEN A. B.	Section 01 32 19 Submittals Schedule ECUTION ERAL REQUIREMENTS Prior to the Pre-Construction Meeting the City Project Manager (CPM) or the City Construction Manager (CC shall provide the GC a copy of the <i>Reimbursable Labor Rate Worksheet.xls</i> . 1. See the last page of this specification for an example of the worksheet. The GC shall provide all subcontractors that will be performing productive labor during the execution of this contract with additional copies of the worksheet as needed. All contractors shall be required to fill out and submit completed worksheets for all Trades and Classification labor that will be performing productive labor during the execution of this contract. ERAL CONTRACTORS RESPONSIBILITIES The GC shall provide the Combined workbook as required by Section 1.6 of Specification 01 32 19 Submittals Schedule for review and approval by the Owners Representatives. 1. Submittal shall be an Exported PDF of the completed Excel Workbook. a. As an Exported PDF the individual worksheets will be bookmarked and the document will be
<u>PART</u> <u>9ART</u> 3.1. 3.2.	D. <u>2 – PR</u> <u>3 - EXI</u> GEN A. B. C. GEN A. B.	Section 01 32 19 Submittals Schedule RODUCTS – NOT USED ECUTION ERAL REQUIREMENTS Prior to the Pre-Construction Meeting the City Project Manager (CPM) or the City Construction Manager (CC shall provide the GC a copy of the <i>Reimbursable Labor Rate Worksheet.xls</i> . 1. See the last page of this specification for an example of the worksheet. The GC shall provide all subcontractors that will be performing productive labor during the execution of this contract with additional copies of the worksheet as needed. All contractors shall be required to fill out and submit completed worksheets for all Trades and Classification labor that will be performing productive labor during the execution of this contract. ERAL CONTRACTORS RESPONSIBILITIES The GC shall provide the combined workbook as required by Section 1.6 of Specification 01 32 19 Submittal: Schedule for review and approval by the Owners Representatives. 1. Submittal shall be an Exported PDF of the completed Excel Workbook. a. As an Exported PDF the individual worksheets will be bookmarked and the document will be searchable for easy reference.
PART PART 3.1. 3.2.	D. <u>2 – PR</u> <u>3 - EXI</u> GEN A. B. C. GEN A. B. C.	 Section 01 32 19 Submittals Schedule Submittals Schedule Submittals Schedule Submittals Schedule Submittals Schedule Submittals Schedule Securion ERAL REQUIREMENTS Prior to the Pre-Construction Meeting the City Project Manager (CPM) or the City Construction Manager (CC shall provide the GC a copy of the <i>Reimbursable Labor Rate Worksheet.xls</i>. See the last page of this specification for an example of the worksheet. The GC shall provide all subcontractors that will be performing productive labor during the execution of this contract with additional copies of the worksheet as needed. All contractors shall be required to fill out and submit completed worksheets for all Trades and Classification labor that will be performing productive labor during the execution of this contract. ERAL CONTRACTORS RESPONSIBILITES The GC shall consolidate all Trades and Classifications into one master Excel Workbook of all trades. The GC shall provide the combined workbook as required by Section 1.6 of Specification 01 32 19 Submittal Schedule for review and approval by the Owners Representatives. Submittal shall be an Exported PDF of the completed Excel Workbook. As an Exported PDF the individual worksheets will be bookmarked and the document will be searchable for easy reference. The GC shall only use the rates posted in the approved submittal throughout the execution of this contract.
<u>PART</u> <u>9ART</u> 3.1. 3.2.	D. <u>2 – PR</u> <u>3 - EXI</u> GEN A. B. C. GEN A. B. C. C.	Section 01 32 19 Submittals Schedule EQDUCTS – NOT USED ECUTION ERAL REQUIREMENTS Prior to the Pre-Construction Meeting the City Project Manager (CPM) or the City Construction Manager (CC shall provide the GC a copy of the <i>Reimbursable Labor Rate Worksheet.xls</i> . 1. See the last page of this specification for an example of the worksheet. The GC shall provide all subcontractors that will be performing productive labor during the execution of this contract with additional copies of the worksheet as needed. All contractors shall be required to fill out and submit completed worksheets for all Trades and Classification labor that will be performing productive labor during the execution of this contract. ERAL CONTRACTORS RESPONSIBILITIES The GC shall provide the combined workbook as required by Section 1.6 of Specification 01 32 19 Submittal: Schedule for review and approval by the Owners Representatives. 1. Submittal shall be an Exported PDF of the completed Excel Workbook. a. As an Exported PDF the individual worksheets will be bookmarked and the document will be searchable for easy reference. The GC shall only use the rates posted in the approved submittal throughout the execution of this contract.
PART PART 3.1. 3.2.	D. <u>2 – PR</u> <u>3 - EXI</u> <u>6</u> EN A. B. C. <u>6</u> EN A. B. C.	Section 01 32 19 Submittals Schedule ECUTION ERAL REQUIREMENTS Prior to the Pre-Construction Meeting the City Project Manager (CPM) or the City Construction Manager (CC shall provide the GC a copy of the <i>Reimbursable Labor Rate Worksheet.xls.</i> 1. See the last page of this specification for an example of the worksheet. The GC shall provide all subcontractors that will be performing productive labor during the execution of this contract with additional copies of the worksheet as needed. All contractors shall be required to fill out and submit completed worksheets for all Trades and Classification labor that will be performing productive labor during the execution of this contract. ERAL CONTRACTORS RESPONSIBILITIES The GC shall consolidate all Trades and Classifications into one master Excel Workbook of all trades. The GC shall provide the combined workbook as required by Section 1.6 of Specification 01 32 19 Submittal Schedule for review and approval by the Owners Representatives. 1. Submittal shall be an Exported PDF of the completed Excel Workbook. a. As an Exported PDF the individual worksheets will be bookmarked and the document will be searchable for easy reference. The GC shall only use the rates posted in the approved submittal throughout the execution of this contract.
<u>PART</u> <u>9ART</u> 3.1.	D. <u>2 – PR</u> <u>3 - EXI</u> <u>6</u> EN A. B. C. <u>6</u> EN A. B. C. C.	 Section 01 32 19 Submittals Schedule SODUCTS – NOT USED ECUTION ERAL REQUIREMENTS Prior to the Pre-Construction Meeting the City Project Manager (CPM) or the City Construction Manager (CC shall provide the GC a copy of the <i>Reimbursable Labor Rate Worksheet.xls</i>. See the last page of this specification for an example of the worksheet. The GC shall provide all subcontractors that will be performing productive labor during the execution of this contract with additional copies of the worksheet as needed. All contractors shall be required to fill out and submit completed worksheets for all Trades and Classification labor that will be performing productive labor during the execution of this contract. ERAL CONTRACTORS RESPONSIBILITIES The GC shall provide the combined workbook as required by Section 1.6 of Specification 01 32 19 Submittal: Schedule for review and approval by the Owners Representatives. Submittal shall be an Exported PDF of the completed Excel Workbook. As an Exported PDF the individual worksheets will be bookmarked and the document will be searchable for easy reference. The GC shall only use the rates posted in the approved submittal throughout the execution of this contract.
<u>PART</u> <u>9ART</u> 3.1.	D. <u>2 – PR</u> <u>3 - EXI</u> <u>6</u> EN A. B. C. <u>6</u> EN A. B. C.	Section 01 32 19 Submittals Schedule SODUCTS – NOT USED ECUTION ERAL REQUIREMENTS Prior to the Pre-Construction Meeting the City Project Manager (CPM) or the City Construction Manager (CC shall provide the GC a copy of the <i>Reimbursable Labor Rate Worksheet.xls</i> . 1. See the last page of this specification for an example of the worksheet. The GC shall provide all subcontractors that will be performing productive labor during the execution of this contract with additional copies of the worksheet as needed. All contractors shall be required to fill out and submit completed worksheets for all Trades and Classification labor that will be performing productive labor during the execution of this contract. ERAL CONTRACTORS RESPONSIBILITIES The GC shall provide the combined workbook as required by Section 1.6 of Specification 01 32 19 Submittals Schedule for review and approval by the Owners Representatives. 1. Submittal shall be an Exported PDF of the completed Excel Workbook. a. As an Exported PDF the individual worksheets will be bookmarked and the document will be searchable for easy reference. The GC shall only use the rates posted in the approved submittal throughout the execution of this contract.
PART PART 3.1. 3.2.	D. 2 – PR 3 - EXI GEN A. B. C. GEN A. B. C. C.	Section 01 32 19 Submittals Schedule SODUCTS – NOT USED ECUTION ERAL REQUIREMENTS Prior to the Pre-Construction Meeting the City Project Manager (CPM) or the City Construction Manager (CC shall provide the GC a copy of the <i>Reimbursable Labor Rate Worksheet.xls.</i> 1. See the last page of this specification for an example of the worksheet. The GC shall provide all subcontractors that will be performing productive labor during the execution of this contract with additional copies of the worksheet as needed. All contractors shall be required to fill out and submit completed worksheets for all Trades and Classification labor that will be performing productive labor during the execution of this contract. ERAL CONTRACTORS RESPONSIBILITES The GC shall provide the combined workbook as required by Section 1.6 of Specification 01 32 19 Submittals Schedule for review and approval by the Owners Representatives. 1. Submittal shall be an Exported PDF of the completed Excel Workbook. a. As an Exported PDF the individual worksheets will be bookmarked and the document will be searchable for easy reference. The GC shall only use the rates posted in the approved submittal throughout the execution of this contract.
PART PART 3.1. 3.2.	D. 2 – PR 3 - EXI GEN A. B. C. GEN A. B. C. C.	Section 01 32 19 Submittals Schedule SODUCTS – NOT USED ECUTION ERAL REQUIREMENTS Prior to the Pre-Construction Meeting the City Project Manager (CPM) or the City Construction Manager (CC shall provide the GC a copy of the <i>Reimbursable Labor Rate Worksheet.xls</i> . 1. See the last page of this specification for an example of the worksheet. The GC shall provide all subcontractors that will be performing productive labor during the execution of this contract with additional copies of the worksheet as needed. All contractors shall be required to fill out and submit completed worksheets for all Trades and Classification labor that will be performing productive labor during the execution of this contract. ERAL CONTRACTORS RESPONSIBILITIES The GC shall provide the combined workbook as required by Section 1.6 of Specification 01 32 19 Submittals Schedule for review and approval by the Owners Representatives. 1. Submittal shall be an Exported PDF of the completed Excel Workbook. a. As an Exported PDF the individual worksheets will be bookmarked and the document will be v searchable for easy reference. The GC shall only use the rates posted in the approved submittal throughout the execution of this contract.
<u>PART</u> <u>9ART</u> 3.1.	D. <u>2 – PR</u> <u>3 - EXI</u> <u>6</u> EN A. B. C. GEN A. B. C.	Section 01 32 19 Submittals Schedule Schedule for review and approval by the Owners Representatives. Schedule for review and approval by the Owners Representatives. Schedule for review and approval by the Owners Representatives. Schedule for review and approval by the owners Representatives. The GC shall only use the rates posted in the approved submittal throughout the execution of this contract. Section 10 Submittal shall be an Exported PDF of the completed Excel Workbook. Schedule for easy reference. The GC shall only use the rates posted in the approved submittal throughout the execution of this contract. Section 01 Section 10 S

Reimbursable Hourly Rate Worksheet

(see bottm of page for instructions)

Project Name:						Enter	TRADE Here:	
Project Location						Car	penter	
Project Number	:				_			
Contractor: Rates are base following doc	ed on the umentaton:							
Classification:		Foreman	Journeyman	Laborer	Apprt 1	Other	Other	Other
Base Rate	e (BR)	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	Vacation	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Healt	th Insurance	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	Pension	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Арр	renticeship	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	Sub-total	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
BR Sub-	total	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Work. Comp	% of BR	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Gen Liability	% of BR	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
WI Unemploy	% of BR	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Fed Unemploy	% of BR	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
FICA	% of BR	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	Sub-total	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
TOTAL C	OST	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00

Enter YOUR percentage of base rate in the

column below

amn below.	
% of BR	
0	- Work. Comp
0	- Gen Liability
0	- WI Unemploy
0.6	- Fed Unemploy
7.65	- FICA

Form Instructions:

 Provide a work sheet for ALL Trade Classifications that will be performing on site productive labor during the execution of this project.

 Responsible contractor to complete only boxes that are shaded, all non-shaded boxes are formula driven.

 Contractor shall provide the name of the source used for these rates. (union contract, Bureau of Labor and Statistices, AGC, ABC, etc.) and be prepared to provide copies if so requested.

END OF SECTION

	SECTION 00 62 76.13 SALES TAX FORM
плрт	
PAR I 1	1 - GENERAL
1	1.2 RELATED SPECIFICATION SECTIONS
1	12 TAX EXEMPT FORM
PART	2 – PRODUCTS – THIS SECTION NOT USED
PART	3 – EXECUTION – THIS SECTION NOT USED
PART	1 – GENERAL
1.1.	SUMMARY
	A. The City of Madison is a qualifying tax exempt entity in the State of Wisconsin.
	B. The Contractor shall refer to Section 102.9 – Bidders Understanding of the City of Madison FACILITIES
	C. This project constructs or remodels facilities owned by the City of Madison in Madison, Wisconsin.
1.2.	RELATED SPECIFICATION SECTIONS
	A. Parts of this specification will reference articles within "The City of Madison FACILITIES MANAGEMENT
	SPECIFICATIONs for Public Works Construction".
	1. Use the following link to access the FACILITIES MANAGEMENT SPECIFICATIONs web page:
	http://www.cityofmadison.com/business/pw/specs.cfm
	a. Click on the "Part" chapter identified in the specification text. For example if the specification
	says "Refer to City of Madison FACILITIES MANAGEMENT SPECIFICATION 210.2" click the li
	Part II, the Part II PDF will open.
	b. Scroll through the index of Part II for specification 210.2 and click the text link which will ta
	to the referenced text.
1.3.	TAX EXEMPT FORM
	A. The Contractor can access Wisconsin Sales and Use Tax Exemption Certificates (form S-211, Wisconsin
	Department of Revenue) from the City of Madison Finance website.
	1. City of Madison tax exempt information and signature by Purchasing Supervisor is already comple
	2. Website: <u>http://www.cityofmadison.com/employeenet/finance/purchasing</u>
	a. Under the title <i>Purchasing Forms</i> , scroll down to the form link titled <i>Sales Tax Exempt Form</i>
PART	2 – PRODUCTS – THIS SECTION NOT USED
PART	3 – EXECUTION – THIS SECTION NOT USED
	END OF SECTION

			SECTION 01 25 13 PRODUCT SUBSTITUTION PROCEDURES
PART	1 – G	ENERAL	
	1.1.	SUMMARY	
-	1 2	RELATED SPECIFICATION	IS
PART	2 – PI	RODUCTS	
	2.1.	SUBSTITUTION REQUEST	[FORM
PART	3 - EX	ECUTION	
3	3.1.	REQUESTING A SUBSTITU	UTION DURING BIDDING
3	3.2.	REQUESTING A SUBSTITU	UTION AFTER AWARD OF CONTRACT
3	3.3.	UNAUTHORIZED SUBSTI	TUTIONS
PART	1 – G	ENERAL	
11	SUM	ΛΜΔΒΥ	
1.1.	۵ ۵	The City of Madison u	uses a specific list of preferred products for various specification items to establish
	7	standards of quality. L	utility, and appearance required.
	В.	The City of Madison w	vill not allow substitutions for specified Products except as follows:
	υ.	1. The Product is	no longer produced or the product manufacturer is no longer in business.
		2. The manufactu	urer has significantly changed performance data, product dimensions, or other such de
		3. Products speci	ified by naming one or more Products or manufacturer's and "or approved equal" or
		"approved equ	uivalent."
	C.	The City of Madison w	vill not allow substitutions for specified Products as follows:
		1. For Products s	pecified by naming only one Product and manufacturer, no substitute product will be
		considered.	
		2. For Products s	pecified by naming several Products or manufacturers select any one of the products of
	D.	manufacturers Request for substituti	s named, which complies with the specifications. No substitute product will be considered ons from any party other than the General Contractor (GC) will not be accepted.
1.2.	RFI	ATED SPECIFICATIONS	
	A.	Section 00 43 25	Substitution Request Form (During Bidding)
	В.	Section 01 26 13	Request for Information (RFI)
	C.	Section 01 31 23	Project Management Web Site (PMWS)
	D.	Section 01 33 23	Submittals
PART	2 – P	RODUCTS	
2 1	C1 10		
2.1.	<u>ع</u> رود ۲	During hidding all con	au tractors (General and Sub-contractors) and suppliers of materials or products shall
	Π.	reference Snecificatio	in Section 00.43.25 and provide a pdf_copy of the Substitution Request form located at
		end of that section wi	th all required attachments directly to the Project Architect
	в	After bidding only the	GC shall submit a request and shall use the form located at the end of this specification
	υ.	and submit the reque	st on the Project Management Web Site.
		and saying the reque	
PART	3 - E)	ECUTION	
3.1.	REC	UESTING A SUBSTITUTIO	ON DURING BIDDING
	Α.	In the event that a sul	bstitution is requested during the bidding phase the Contractor or Supplier shall meet
		substitution request d	leadline listed in the bidding documents. No substitution request will be considered d
		the bidding period aft	er the stated substitution request deadline.
	В.	See specification 00 4	3 25 Substitution Request Form (During Bidding).
			N AFTER AWARD OF CONTRACT
3.2.	REC	UESTING A SUBSTITUTIO	*** * ** * = ** * *** *** *** *** ******
3.2.	REC A.	QUESTING A SUBSTITUTIO	t will only be considered after award of contract if it meets the qualifying provisions as
3.2.	REC A.	A substitution request described in 1.1.B.1 a	t will only be considered after award of contract if it meets the qualifying provisions as nd .2 above.

1			1. Consulting Staff, Owner and Owners Representatives will review the request and provide the appropriate
2			approvals and feed back to the GC.
3			
4	3.3.	UNAU	HORIZED SUBSTITUTIONS
5		Α.	Any Contractor who substitutes products without proper authorization by the Owner and Architect will be
6			required to immediately remove and replace the product and all costs required to conform to the Contract
7			Documents shall be borne by the General Prime Contractor.
8			
9			
10			
11			
12			
13			NOTE SEE NEXT PAGE FOR SAMPLE SUBSTITUTION REQUEST FORM.
14			

1

Ć)	Substi	tutio	n Req	uest
'oday's Date:					
roject Title: roject Number:		Contract N	ımber:]]
y completing and 1 The Genera Product Su 2 The functio 3 The propos 4 The propos 4 The propos 5 Maintenan in the attac 6 The Genera includes bu costs, and i	submitting this form I Contractor affirms bstitution Procedure in, appearance, and red substitution doe: red substitution will rts. ce and service parts chments section bek al Contractor shall be it is not to limited to inspection fees.	a for review the General Contr that this request is in complia es. quality of the proposed substil s not affect dimensions shown have no adverse affects on oth will be locally available for the xw.) e responsible for any and all co fees for building design, engin	actor affirms th nee with the req tution are equal on the drawings er trades, the co proposed subst sts associated w leering design fe	at all of the followi uirements describe or superior to the s i. anstruction schedul itution. (GC shall p vith this substitutior es, detailing fees, p	ng statements are correct: d in Specification 01 25 13 pecified item. e, or any specified warranty rovide supporting documenta in request if approved. This lan review fees, construction
		GC Substituti	on Reques	t:	
5eneral Title:					
Related Specificat	tion:				
leason for Substi	tution:				
Proposed Substitu (inclu	<i>ition:</i> de Name, Model, etc.)				
ubmitted By:]	Phone	-	
Company:			Email:		

		SECTION 01 26 13 REQUEST FOR INFORMATION (RFI)
PART	1 – G	ENERAL
-	1.1.	SUMMARY
	1.2.	RELATED SPECIFICATIONS
-	1.3.	PERFORMANCE REQUIREMENTS
	1.4.	
PART	2 – Pl	
	2.1.	REQUEST FOR INFORMATION FORM
PARI	3 - EX	
	3.1.	
:	3.3.	
2	3.4.	COMMENCEMENT OF WORK RELATED TO AN RFI
PART	1 – G	ENERAL
1.1.	SUI	MMARY
	Α.	Contractors shall use the RFI form/process to request additional information or clarification regarding the
		construction documents.
	В.	All RFI documentation will be processed through the through the Project Management Web Site (PMWS).
1.2.	REL	ATED SPECIFICATIONS
	A.	Section 01 26 46 Construction Bulletin (CB)
	В.	Section 01 26 57 Change Order Request (COR)
	C.	Section 01 26 63 Change Order (CO)
	D.	Section 01 31 23 Project Management Web Site (PMWS)
	Ε.	Section 01 91 00 Commissioning
1 3	PFR	REORMANCE REQUIREMENTS
1.01	Α.	RELissues initiated by any contractor shall be done through the General Contractor (GC).
	7	1 REIs submitted by any Sub-contractor under the GCs control shall be returned with no response
	B.	Submit a new RFI for each issue. Only multiple questions that are of a similar nature may be combined into or
		RFI shall be allowed and responded to.
1.4.	QU	ALITY ASSURANCE
	A.	The GC shall be responsible for all of the following:
		1. Ensure that any request for additional information is valid and the information being requested is not
		addressed in the construction documents.
		2. Ensure that all requests are clearly stated and the RFI form is completely filled out.
		3. Ensure that all Work associated an RFI response is carried out as intended.
	В.	The Project Architect /Project Engineer (A/E PROJ MGR) shall be responsible for the following:
		1. Ensure that all responses to contractor initiated RFIs are properly responded to in a timely fashion.
		a. The CPM, Owner, consulting staff, and other City staff shall be responsible for the initial review
		the RFI. The A/E PROJ MGR shall be responsible for codifying all consultant and Owner/City sta
		comments into a unified RFI response.
PART	2 – P	RODUCTS
- 1	054	
2.1.		The PEL form is located on the Draiget Management Web Site
	А.	The KEI form is located on the Project Management web site.
PART	3 - E)	KECUTION
21	201	
J.1.	^	Immediately on discovery of the need for additional information or interpretation of the Contract Documents
	А.	any contractor may initiate an REI for additional information or clarification through the CC
	R	any contractor may initiate an AFT for additional mornialion of Clarification tinough the GC. The GC shall use the Project Management Web Site and completely fill out the form
	ь.	The de shan use the Project Management web site and completely fill out the form.

1 2 3 4 5 6 7			 Thoroughly explain the issue at hand, provide backup information (photographs, sketches, drawings, data, etc.) as necessary, and clearly state the question or problem that requires a resolution. Combine like or related issues but do not include multiple issues on one form. a. Example. If a duct interferes with other critical piping and electrical work include all issues into one RFI. b. Example. If you have a question regarding the chiller and another regarding toilet partitions create separate RFIs.
8			
9	3.3.	RFI RE	ONSES
10		Α.	esponses to simple RFI issues shall be completed within five (5) working days of the RFI form being submitted.
11		В.	esponses to more complex issues may require additional time or may require a Construction Bulletin to be
12			ublished. The initial RFI shall be responded to within five (5) working days stating that the RFI is being
13		c	eviewed and provide an estimated date for the response.
15		C.	Requests for approval of submittals
16			Requests for approval of substitutions
17			Requests for approval of Contractor's means and methods.
18			Requests for coordination information already indicated in the Contract Documents.
19			Requests for adjustments in the Contract Time or the Contract Sum.
20			Requests for interpretation of A/E's actions on submittals.
21			Incomplete RFI or inaccurately prepared RFI.
22			
23	3.4.	COMN	NCEMENT OF WORK RELATED TO AN RFI
24		Α.	ne GC shall only proceed with the Work of an RFI when additional information is not required.
25		В.	ne GC shall not proceed with any Work associated with an RFI while it is under review.
26		C.	ne GC shall not proceed with any Work associated with an RFI that clearly states a CB will be issued in response
27		_	o the RFI.
28		D.	he GC will be required to immediately remove and replace unauthorized Work and all costs required to
29			onform to the Contract Documents shall be borne by the GC.
21			
32			
33			END OF SECTION
34			
35			

1 2					SECTION 01 26 46 CONSTRUCTION BULLETIN (CB)
3					
4	PART	1 – G	ENERAL		
5	1	1.	SUMMA	RY	
6	1	2.	RELATED	SPECIFICATION	NS
7	1	3.	PERFORI		EMENTS
8	1	4.	QUALITY	ASSURANCE	
9	PART	2 – P	RODUCTS.		
10	2	2.1.	CONSTR	UCTION BULLET	IN FORM
11	PARI	3 - E>	LCUTION		
12	3	3.1.	WRITING	5 THE CONSTRU	CTION BULLETIN
13	5	3.2.	EXECUTI	NG THE CONST	RUCTION BULLETIN2
14 15	DADT	1 0			
15 16	PARI	1-0	DEINEKAL		
17	1.1.	SUI	MMARY		
18		A.	Const	ruction Bulletin	s (CB) are formal published construction documents that modify the original contract bid
19			docun	nents after cons	struction has commenced. CBs may be published for many reasons, including but not
20			limite	d to the followi	ng:
21			1.	Clarification o	f existing construction documents including specifications, plans, and details
22			2.	Change in pro	duct or equipment
23			3.	A response to	a Request for Information
24			4.	Change in sco	pe of the contract as either an add or a deduct of work
25		В.	CBs pr	rovide a higher	degree of detail in response to a Request for Information (RFI) through directives, revised
26			plans/	details, and spe	ecifications as necessary.
27		C.	The Cl	B may change tl	ne original contract documents through additions or deletions to the Work.
28		D.	Where	e the directives	of a CB are significant enough to warrant a Change Order Request (COR) the GC shall use all
29			inform	nation provided	in the CB to assemble all required back-up documentation for additions and deletions of
30			mater	ials, labor and o	other related contract costs for the COR.
31		Ε.	All CB	documentation	will be processed through the Project Management Web Site (PMWS).
32					
33	1.2.	REL	ATED SPE	CIFICATIONS	
34		Α.	Sectio	n 01 26 13	Request for Information (RFI)
35		В.	Sectio	n 01 26 57	Change Order Request (COR)
36		C.	Sectio	n 01 26 63	Change Order (CO)
37		D.	Sectio	n 01 31 23	Project Management Web Site (PMWS)
38		Ε.	Sectio	n 01 91 00	Commissioning
39					
40	1.3.	PEF	RFORMAN	CE REQUIREME	INTS
41		Α.	Projec	t Architect /Pro	ject Engineer (A/E PROJ MGR): The A/E PROJ MGR shall be the only person authorized to
42			publis	h a CB as neede	ed for any reason indicated in section 1.1.A above. The A/E PROJ MGR shall consult as
43			neces	sary with any of	the following while drafting the CB and shall confirm final direction with the CPM prior to
44			issuin	g a CB:	
45			1.	City Project m	anager (CPM)
46			2.	Owner	
47			3.	Members of t	he consulting staff
48			4.	Members of c	ity staff
49			5.	The General C	Contractor
50			6.	Sub-contracto	irs
51			7.	Commissionin	g Agent (CxA)
52		В.	Gener	al Contractor:	The GC shall be responsible for the following as needed:
53			1.	Executing the	directives of the CB when they believes that no changes in labor, materials, equipment, or
54				contract dura	tion will be required for additions or deletions.
55			2.	Submit a COR	when they believes that a change in labor, materials, equipment or contract duration will
56				be required fo	or additions or deletions.
57					

1	1.4.	QUA	LITY ASSURANCE
2		Α.	The A/E PROJ MGR shall be responsible for ensuring the final CB sufficiently provides direction, details,
3			specifications and other information as necessary for the GC to perform the intended Work.
4		В.	The A/E PROJ MGR shall be responsible for ensuring the final CB is published as expeditiously as practical based
5			on the complexity of the CB being written. CBs that may affect the GC critical path shall be given priority.
6			
7	PART	2 – PR	<u>ODUCTS</u>
8			
9	2.1.	CON	STRUCTION BULLETIN FORM
10		Α.	The CB form is located on the Project Management Web Site.
11			
12	PART	3 - EXE	ECUTION
13			
14	3.1.	WRI	TING THE CONSTRUCTION BULLETIN
15		Α.	The A/E PROJ MGR shall draft a CB as needed using the Construction Bulletin form on the Project Management
16			Web Site.
17			1. The A/E PROJ MGR and/or consulting staff as necessary shall provide specifications, model numbers and
18			performance data, details and other such information necessary to clearly state the intentions of the CB.
19			2. The consulting staff, CPM, Owner, CxA and other City Staff shall review the draft and recommend
20			changes as needed.
21			3. The A/E PROJ MGR shall amend the draft as necessary into a final CB for review.
22			4. Full plan sheets and entire specification sections referred to within a CB, shall be reissued with the CB.
23		В.	Once the final CB has been approved the A/E PROJ MGR shall "Submit" the CB through the Project Management
24			Web Site to the City Project Manager.
25		C.	The City Project Manager will close and distribute the CB.
26			
27	3.2.	EXEC	CUTING THE CONSTRUCTION BULLETIN
28		A.	The GC shall acknowledge receipt of the CB on the Project Management Web Site as instructed in the Tutorial
29			Manual provided to the awarded contractor.
30		В.	The GC shall notify all Sub-contractors of the CB and publish the CB to all field sets of drawings and specifications
31			as appropriate.
32		C.	The GC shall execute the directives of the CB or submit COR documentation as necessary during the execution
33			and implementation of the CB.
34			1. See Specification 01 26 57 Change Order Request (COR)
35			
36			
37			
38			END OF SECTION
39			

		SECTION 01 26 57
		CHANGE ORDER REQUESTS (COR)
плрт	T1 C	ENEDAL
PARI	11-0	
	1.1.	
	1.2.	RELATED SPECIFICATION SECTIONS
	1.5.	
	1.4.	
	1.5.	OVERHEAD AND PROFIL MARKUP
	1.6.	
	1./.	
PARI	12-P	
	2.1.	CHANGE ORDER REQUEST FORM
PART	Г З - EX	
	3.1.	ESTABLISHING A CHANGE ORDER REQUEST
	3.2.	SUBMIT A CHANGE ORDER REQUEST FORM
	3.3.	CHANGE ORDER REQUEST REVIEW, APPROVAL, AND PROCESSING
	3.4.	EMERGENCY CHANGE ORDER REQUEST
PAR	T 1 – G	iENERAL
1.1.	SUI	MMARY
	Α.	Except in cases of emergency, no changes in the Work required by the Contract Documents may be made
		by the General Contractor (GC) without having prior approval of the City Engineer or their representative.
	В.	The City may at any time, without invalidating the Contract and without Notice to Sureties, order changes in
		the Work by written Change Order (CO). Such changes may include additions and/or deletions.
	C.	Where the City desires to make changes in the Work through use of written Change Order Request (COR), the
		following procedures apply:
		1. If requested by the City, the GC shall prepare and submit a detailed proposal, including all cost and time
		adjustments to which the GC believes it will be entitled if the change proposed is incorporated into the
		Contract. The City shall be under no legal obligation to issue a Change Order for such proposal
		2 The partice child attempt in good faith to reach agreement on the adjustment needed to the Contract t
		2. The parties shall attempt in good abarrow for the adjustments the adjustment in each other contract to the contract to the contract of the adjustment in
		property incorporate the proposed change(s) into the work. In the event that the parties agree of such
		adjustments, the City may issue a Change Order and incorporate such changes and agreed to
		adjustments, if any.
		3. In some instances, it may be necessary for the City to authorize Work or direct changes in Work for whic
		no final and binding agreement has been reached and for which unit prices are not applicable. In such
		cases the following shall apply.
		a. Upon written request by the City, the GC shall perform proposed Work
		b. The cost of such change may be determined in accordance with this specification.
		c. In the event agreement cannot be accomplished as contemplated herein, the City may authorize
		the Work to be performed by City forces or to hire others to complete the Work. Such action on
		the part of the City shall not be the basis of a claim by the GC for failure to allow it to perform the
		changed Work.
	D.	Where changes in the Work are made by the City through use of a force account basis, the GC shall as soon as
		practicable, and in no case later than ten (10) working days from the receipt of such order, unless another time
		period has been agreed to by both parties, give the City written Notice, stating:
		1. The date, circumstances and source of the extra work; and,
		2. The cost of performing extra work described by such Order, if any: and.
		3. Effect of the order on the required completion date of the Project. if any.
	Ε.	The giving of each Notice by the GC as prescribed by this specification, shall be a requirement to liability of the
	_ .	City for payment of any additional costs incurred by the GC in implementing changes in the Work. Under this
		specification no order or statement of the City shall be treated as a Change Order, or shall entitle the GC to an
		equitable adjustment of the terms of this Contract or damages for costs incurred by the CC on any activity for
		which the Notice was not given
	-	which the Nuclet was hild given.
	⊢.	In the event work is required due to an emergency as described in this specification the GC must request an
		and table adjustment as an example below of the sector of
		equitable adjustment as soon as practicable, and in no case later than ten (10) working days of the

	G.	All GC requests for equitable adjustment shall be submitted to the CPM per the specifications below. Such		
		requests shall set forth with specificity the amount of and reason(s) for the proposed adjustment and shall be		
		accompanied by supporting information and documents.		
	Н.	No adjustment of any kind shall be made to this Contract, if asserted by the GC for the first time, after the date		
		of final payment.		
	I.	This specification shall be used by the GC when preparing documentation for any COR to ensure each has been		
		properly and completely filled out as required by the City of Madison.		
	J.	All COR documentation will be processed through the Project Management Web Site (PMWS).		
1.2.	RELA	TED SPECIFICATION SECTIONS		
	Α.	Section 01 26 13 Request for Information (RFI)		
	В.	Section 01 26 46 Construction Bulletins (CB)		
	C.	Section 01 26 63 Change Order (CO)		
	D.	Section 01 31 23 Project Management Web Site (PMWS)		
	Ε.	Section 01 91 00 Commissioning		
	F.	Parts of this specification will reference articles within "The City of Madison FACILITIES MANAGEMENT		
		SPECIFICATIONs for Public Works Construction".		
		1. Use the following link to access the FACILITIES MANAGEMENT SPECIFICATIONs web page:		
		http://www.cityofmadison.com/business/pw/specs.cfm		
		a. Click on the "Part" chapter identified in the specification text. For example if the specification		
		says "Refer to City of Madison FACILITIES MANAGEMENT SPECIFICATION 210.2" click the link for		
		Part II, the Part II PDF will open.		
		b. Scroll through the index of Part II for specification 210.2 and click the text link which will take you		
		to the referenced text		
1.3.	DEFI	VITIONS AND STANDARDS		
	Α.	LABOR: The amount of time and cost associated with the performance of human effort for a defined scope of		
		Work Labor is further defined as follows:		
		1 Jabor rate is the total hourly rate which includes the basic rate of nay fringe benefits plus each		
		company's cost of required insurance, also referred to as a reimbursable labor rate		
		2 Unit labor is the labor hours anticipated to install the corresponding unit of material		
		 abor cost is the labor hours multiplied by the bourly labor rates 		
	в	MATERIAL: Actual material cost is the amount naid, or to be naid, by the GC for materials, supplies and		
	5.	equipment entering nermanently into the Work including cost of transportation and annicable taxes. The cost		
		shall not exceed the usual and customary cost for such items available in the geographical read of the project		
	C	LARGE TOOLS AND MAIOR FOLIDEMENT, Large tools and major equipment are those with an initial cost greater		
	с.	that \$1.500 whather from the GC or other sources		
		1 Tool and equipment use and time allowed is only for extra work associated with change orders		
		Pontal Pate is the machine cost associated with operating a piece of equipment for a defined		
		a. Rental rate is the machine cost associated with operating a piece of equipment for a defined		
		\mathbf{C}		
		for such items available in the geographical area of the project		
		for such items available in the geographical area of the project.		
		for such items available in the geographical area of the project.b. Rental cost is the rental rate multiplied by the anticipated duration the equipment shall be required.		
		 for such items available in the geographical area of the project. b. Rental cost is the rental rate multiplied by the anticipated duration the equipment shall be required. 		
		 for such items available in the geographical area of the project. b. Rental cost is the rental rate multiplied by the anticipated duration the equipment shall be required. 2. The GC shall provide a breakdown of all rental rates to indicate what items and costs are associated with the standard standa		
		 for such items available in the geographical area of the project. b. Rental cost is the rental rate multiplied by the anticipated duration the equipment shall be required. 2. The GC shall provide a breakdown of all rental rates to indicate what items and costs are associated with the rate. Examples of items to include in the breakdown would be fuel consumption, lubrication, 		
		 for such items available in the geographical area of the project. b. Rental cost is the rental rate multiplied by the anticipated duration the equipment shall be required. 2. The GC shall provide a breakdown of all rental rates to indicate what items and costs are associated with the rate. Examples of items to include in the breakdown would be fuel consumption, lubrication, maintenance and other similar expenses but not including profit and overhead. 		
		 for such items available in the geographical area of the project. b. Rental cost is the rental rate multiplied by the anticipated duration the equipment shall be required. 2. The GC shall provide a breakdown of all rental rates to indicate what items and costs are associated with the rate. Examples of items to include in the breakdown would be fuel consumption, lubrication, maintenance and other similar expenses but not including profit and overhead. 3. When large tools and equipment needed for Change Order work are not already at the job site, the 		
		 for such items available in the geographical area of the project. b. Rental cost is the rental rate multiplied by the anticipated duration the equipment shall be required. 2. The GC shall provide a breakdown of all rental rates to indicate what items and costs are associated with the rate. Examples of items to include in the breakdown would be fuel consumption, lubrication, maintenance and other similar expenses but not including profit and overhead. 3. When large tools and equipment needed for Change Order work are not already at the job site, the actual cost to get the item there is also reimbursable. 		
	D.	 for such items available in the geographical area of the project. b. Rental cost is the rental rate multiplied by the anticipated duration the equipment shall be required. 2. The GC shall provide a breakdown of all rental rates to indicate what items and costs are associated with the rate. Examples of items to include in the breakdown would be fuel consumption, lubrication, maintenance and other similar expenses but not including profit and overhead. 3. When large tools and equipment needed for Change Order work are not already at the job site, the actual cost to get the item there is also reimbursable. BOND COST: The cost shall be calculated at 1% of the total proposed change order. 		
	D. E.	 for such items available in the geographical area of the project. b. Rental cost is the rental rate multiplied by the anticipated duration the equipment shall be required. 2. The GC shall provide a breakdown of all rental rates to indicate what items and costs are associated with the rate. Examples of items to include in the breakdown would be fuel consumption, lubrication, maintenance and other similar expenses but not including profit and overhead. 3. When large tools and equipment needed for Change Order work are not already at the job site, the actual cost to get the item there is also reimbursable. BOND COST: The cost shall be calculated at 1% of the total proposed change order. SUB-CONTRACTOR COSTS: Sub-contractor costs are for those labor, material, and equipment costs required by 		
	D. E.	 for such items available in the geographical area of the project. b. Rental cost is the rental rate multiplied by the anticipated duration the equipment shall be required. 2. The GC shall provide a breakdown of all rental rates to indicate what items and costs are associated with the rate. Examples of items to include in the breakdown would be fuel consumption, lubrication, maintenance and other similar expenses but not including profit and overhead. 3. When large tools and equipment needed for Change Order work are not already at the job site, the actual cost to get the item there is also reimbursable. BOND COST: The cost shall be calculated at 1% of the total proposed change order. SUB-CONTRACTOR COSTS: Sub-contractor costs are for those labor, material, and equipment costs required by subcontracted specialties to complete the Change Order work. 		
	D. E. F.	 for such items available in the geographical area of the project. b. Rental cost is the rental rate multiplied by the anticipated duration the equipment shall be required. 2. The GC shall provide a breakdown of all rental rates to indicate what items and costs are associated with the rate. Examples of items to include in the breakdown would be fuel consumption, lubrication, maintenance and other similar expenses but not including profit and overhead. 3. When large tools and equipment needed for Change Order work are not already at the job site, the actual cost to get the item there is also reimbursable. BOND COST: The cost shall be calculated at 1% of the total proposed change order. SUB-CONTRACTOR COSTS: Sub-contractor costs are for those labor, material, and equipment costs required by subcontracted specialties to complete the Change Order work. OVERHEAD AND PROFIT Markup: The allowable markup percentage to a COR by the GC and Sub-contractors for 		
	D. E. F.	 for such items available in the geographical area of the project. b. Rental cost is the rental rate multiplied by the anticipated duration the equipment shall be required. 2. The GC shall provide a breakdown of all rental rates to indicate what items and costs are associated with the rate. Examples of items to include in the breakdown would be fuel consumption, lubrication, maintenance and other similar expenses but not including profit and overhead. 3. When large tools and equipment needed for Change Order work are not already at the job site, the actual cost to get the item there is also reimbursable. BOND COST: The cost shall be calculated at 1% of the total proposed change order. SUB-CONTRACTOR COSTS: Sub-contractor costs are for those labor, material, and equipment costs required by subcontracted specialties to complete the Change Order work. OVERHEAD AND PROFIT Markup: The allowable markup percentage to a COR by the GC and Sub-contractors for overhead and profit. All of the following are expenses associated with overhead and profit and shall not be 		
	D. E. F.	 for such items available in the geographical area of the project. b. Rental cost is the rental rate multiplied by the anticipated duration the equipment shall be required. 2. The GC shall provide a breakdown of all rental rates to indicate what items and costs are associated with the rate. Examples of items to include in the breakdown would be fuel consumption, lubrication, maintenance and other similar expenses but not including profit and overhead. 3. When large tools and equipment needed for Change Order work are not already at the job site, the actual cost to get the item there is also reimbursable. BOND COST: The cost shall be calculated at 1% of the total proposed change order. SUB-CONTRACTOR COSTS: Sub-contractor costs are for those labor, material, and equipment costs required by subcontracted specialties to complete the Change Order work. OVERHEAD AND PROFIT Markup: The allowable markup percentage to a COR by the GC and Sub-contractors fo overhead and profit. All of the following are expenses associated with overhead and profit and shall not be reimbursable as individual items on any COR: 		
	D. E. F.	 for such items available in the geographical area of the project. b. Rental cost is the rental rate multiplied by the anticipated duration the equipment shall be required. 2. The GC shall provide a breakdown of all rental rates to indicate what items and costs are associated with the rate. Examples of items to include in the breakdown would be fuel consumption, lubrication, maintenance and other similar expenses but not including profit and overhead. 3. When large tools and equipment needed for Change Order work are not already at the job site, the actual cost to get the item there is also reimbursable. BOND COST: The cost shall be calculated at 1% of the total proposed change order. SUB-CONTRACTOR COSTS: Sub-contractor costs are for those labor, material, and equipment costs required by subcontracted specialties to complete the Change Order work. OVERHEAD AND PROFIT Markup: The allowable markup percentage to a COR by the GC and Sub-contractors for overhead and profit. All of the following are expenses associated with overhead and profit and shall not be reimbursable as individual items on any COR: 1. CHANGE ORDER PREPARATION: All costs associated with the preparing and processing of the change 		
	D. E. F.	 for such items available in the geographical area of the project. b. Rental cost is the rental rate multiplied by the anticipated duration the equipment shall be required. 2. The GC shall provide a breakdown of all rental rates to indicate what items and costs are associated with the rate. Examples of items to include in the breakdown would be fuel consumption, lubrication, maintenance and other similar expenses but not including profit and overhead. 3. When large tools and equipment needed for Change Order work are not already at the job site, the actual cost to get the item there is also reimbursable. BOND COST: The cost shall be calculated at 1% of the total proposed change order. SUB-CONTRACTOR COSTS: Sub-contractor costs are for those labor, material, and equipment costs required by subcontracted specialties to complete the Change Order work. OVERHEAD AND PROFIT Markup: The allowable markup percentage to a COR by the GC and Sub-contractors for overhead and profit. All of the following are expenses associated with overhead and profit and shall not be reimbursable as individual items on any COR: 1. CHANGE ORDER PREPARATION: All costs associated with the preparing and processing of the change order. 		
	D. E. F.	 for such items available in the geographical area of the project. b. Rental cost is the rental rate multiplied by the anticipated duration the equipment shall be required. The GC shall provide a breakdown of all rental rates to indicate what items and costs are associated with the rate. Examples of items to include in the breakdown would be fuel consumption, lubrication, maintenance and other similar expenses but not including profit and overhead. When large tools and equipment needed for Change Order work are not already at the job site, the actual cost to get the item there is also reimbursable. BOND COST: The cost shall be calculated at 1% of the total proposed change order. SUB-CONTRACTOR COSTS: Sub-contractor costs are for those labor, material, and equipment costs required by subcontracted specialties to complete the Change Order work. OVERHEAD AND PROFIT Markup: The allowable markup percentage to a COR by the GC and Sub-contractors for overhead and profit. All of the following are expenses associated with overhead and profit and shall not be reimbursable as individual items on any COR: CHANGE ORDER PREPARATION: All costs associated with the preparing and processing of the change order. DESIGN, ESTIMATING, AND SUPERVISION: All such efforts, unless specifically requested by Owner as 		

1 2 3 4 5 6 7 8 9 10 11 12			 INSTALLATION LAYOUT: The layout required for the installation of material and equipment, and the installation design, is the responsibility of the GC. SMALL TOOLS AND SUPPLIES: The cost of small hand tools with an initial cost of \$1,500 or less, along with consumable supplies and expendable items such as drill bits, saw blades, gasoline, lubricating or cutting oil, and similar items. GENERAL EXPENSE: The general expense, which is those items that are a specific job cost not associated with direct labor and material such as job trailers, foreman truck, and similar items. RECORD DRAWINGS: The preparation of record or as-built drawings. OTHER COSTS: Any miscellaneous cost not directly assessable to the execution of the Change Order including but not limited to the following: All association dues, assessments, and similar items. All education, training, and similar items.
13			c. All drafting and/or engineering, unless specifically requested by Owner as additional Work to be
14			documented as a Change Order proposal or portion thereof.
15			d. All other items including but not limited to review, coordination, estimating and expediting, field
16			and office supervision, administrative work, etc.
17		G.	Contract Extension: The necessary amount of time to be added to the contract deadlines for the completion of a
18			change order.
19			
20	1.4.	CONT	RACT EXTENSION
21		Α.	The GC shall not assume that every COR will require a Contract Extension. If the GC feels a contract extension is
22			warranted, they shall provide sufficient scheduling information that shows how the COR being requested
23			impacts the critical path of the project.
24		В.	The City of Madison strongly encourages the GC to explore alternative methods and practices prior to submitting
25			a COR with a request for contract extension.
26			
27	1.5.	OVER	HEAD AND PROFIT MARKUP
28		Α.	Pursuant to the City of Madison FACILITIES MANAGEMENT SPECIFICATIONs for Public Works Construction,
29			Section 104.7, Extra Work, the following maximum allowable markups shall be strictly enforced on all change
30			orders associated with the execution of this contract.
31			1. The total maximum overhead and profit shall not exceed fifteen percent (15%) of the total costs.
32			The total maximum overhead and profit shall be distributed as follows:
33			a. For work performed and materials provided solely by the General Contractor, fifteen percent
34			(15%) of the total costs.
35			b. For work performed and materials provided solely by Sub-contractors and supervised by the
36			General Contractor:
37			i. Supervision of the GC, five percent (5%) of the total Sub-contractor cost.
38			ii. Sub-contractors work and materials ten percent (10%) of the total Sub-contractor cost.
39			
40	1.6.	PERFC	DRMANCE REQUIREMENTS
41		Α.	The GC shall become thoroughly familiar with this specification as it will identify procedures and expenses that
42			are or are not allowed under the Change Order and Change Order Request process.
43		В.	The GC shall be responsible for all of the following:
44			1. Carefully reviewing the CB that is associated with the COR.
45			2. Collecting required supporting documentation from all contractors that quantify the need for a COR.
46			a. Labor hours and wage rates
47			b. Material costs
48			c. Equipment costs
49		C.	The following shall apply to establishing prices for labor, materials, and equipment costs:
50			1. Where Work to be completed has previously been established by individual bid items in the contract bid
51			proposal the GC shall use the unit bid prices previously established.
52			2. Where Work to be completed was bid as a Lump Sum without individual bid items the GC shall provide a
53			breakdown of all labor, materials, equipment including unit rates and quantities required.
54		D.	The completion date is determined by Owner. The schedule, however, is the responsibility of the GC. Time
55			extensions for extra Work will be considered when a schedule analysis of the critical path shows that the Change
56			Order Request places the Work beyond the completion date stated in the Contract.

1			
2	1.7.	QUAL	ITY ASSURANCE
3		Α.	The GC shall be responsible for ensuring that all COR supporting documentation meets the following
4			requirements prior to completing the COR form on the Project Management Web Site:
5			1. Sufficiently indicates labor, material, and other expenses related to completing the intent of the CB.
6 7			 No costs exceed the usual and customary amount for such items available in the geographical area of the project, and no costs exceed those established under the contract.
8		В.	The Project Architect /Project Engineer A/E PROJ MGR, Commissioning Agent (CxA), City Project Manager (CPM),
9			other members of the consulting staff, and city staff shall review all COR requests to ensure that the intent of the
10			CB will be met under the proposal of the COR or request additional information as necessary.
11			
12	PART	<u> 2 – PRC</u>	DDUCTS
13		~	
14	2.1.		IGE ORDER REQUEST FORM
15 16		А.	The COR form is located on the Project Management web Site.
10 17	DART	2 - EVE	
18	FALL	5 - EAE	
19	3.1.	ESTAR	BLISHING A CHANGE ORDER REQUEST
20	0.1	A.	Upon receipt of a Construction Bulletin (CB) where the GC believes a significant change in contract scope
21			warrants the submittal of a COR the GC shall do all of the following within ten (10) working days after receipt of
22			the CB:
23			1. Review the CB with all necessary trades and sub-contractors required by the change in scope.
24			a. Additions or deletions to the contract scope shall be as directed within the CB.
25			b. Additions or deletions of labor and materials shall be determined by the GC based on the
26			directives of the CB.
27			2. Assemble all required back-up documentation for additions and deletions of materials, labor and other
28			related contract costs as previously outlined in this specification.
29		_	3. Submit a COR request form on the Project Management Web Site.
30		В.	Submitting a COR does not obligate the GC to complete the work associated with the COR nor does it obligate
31			the Owner to approve the COR as a change to the contract.
32 33	3 2	SURM	ALT & CHANGE ORDER REQUEST FORM
34	5.2.	A.	This specification shall provide a subject overview only. In depth instructions shall be provided to the awarded
35			Contractor in a PDF Instructional Manual.
36		В.	The GC shall select the appropriate link on the Project Management Web Site.
37		C.	The software will open a new COR form and the GC shall provide all of the following information:
38			1. DO NOT perform any calculations on this worksheet, only provide the raw data as requested below. All
39			calculations, totals, and markups shall be computed as described within this specification.
40			2. Provide a summary description of the COR request, and justification for any requested time extension to
41			the contract, indicate the number of calendar days being requested for the extension and add any
42			attachments to the form as needed.
43			3. Provide all GC self-performance data including all of the following:
44			a. Materials description, quantities, and unit costs.
45			b. Labor hours and rates for all Foremen, Journeymen, and Apprentices by trade.
46			c. Equipment descriptions, quantities, unit costs and rates.
47			4. Provide all sub-contractor data including all of the following:
48			 Materials description, quantities, and unit costs. Labor bours and rates for all Foreman, Journauman, and Appropriates butrade
49 50			D. Labor nours and rates for all Poremen, Journeymen, and Apprentices by trade.
51			5 Ensure all calculations performed by the form have been completed correctly. Contact the CPM directly
52			if you suspect an error before hitting the save button
53		D.	When all data has been entered submit the COR form. This will kick off the COR Review and Approval process.
54			
55	3.3.	CHAN	IGE ORDER REQUEST REVIEW, APPROVAL, AND PROCESSING
56		Α.	The A/E PROJ MGR and CPM shall review all CORs submitted by the GC.

1			 Additional consulting staff and city staff having knowledge of the components of the COR shall review
2			and advise the A/E PROJ MGR and CPM as to the accuracy of the items, quantities, and associated costs
3			of the COR as directed by the CB.
4			2. The CPM shall review the COR with the Owner.
5		В.	If required the A/E PROJ MGR and CPM, shall in good faith, further negotiate the COR with the GC as necessary.
6			All amendments to any COR shall be documented within the Project Management Web Site software.
7		C.	After final review of the COR the CPM and Owner may accept the COR.
8		D.	The CPM shall prepare the COR in the form of an official Board of Public Works Change Order for final review and
9			approval as outlined in Section 01 26 63 Change Order (CO).
10		Ε.	The GC shall not act upon any accepted COR until it has received final approval through the Public Works process
11			as an official CO to the Work unless instructed to do so by the CPM. Proceeding without the final approval of a
12			fully authorized Change Order is at the GC's own risk.
13			
14	3.4.	EME	RGENCY CHANGE ORDER REQUEST
15		Α.	In the event Work is required due to an emergency as described in the Contract Documents, the GC must
16			request an equitable adjustment as soon as practicable, and in no case later than ten (10) working days of the
17			commencement of such emergency.
18		В.	The GC shall provide full documentation of all labor, materials and equipment used during the period of
19			emergency as part of the COR submittal.
20			
21			
22			
23			END OF SECTION
24			

1 2			SECTION 01 26 63 CHANGE ORDER (CO)	
3					
4	PART 1 – GENERAL				
5	5 1.1.			1	
5	1			1	
/			BUARD OF PUBLIC WORKS PROCEDURE	1	
8	PARI	2 – PK		2	
9	2	2.1. CHANGE ORDER FORM			
10					
11	3.1.			2	
12	3.2. EXECUTION OF THE CHANGE ORDER				
13	DADT	1 (1			
14 15	PARI	1-6	JENERAL		
15	1 1	SLIN	MMARY		
17	1.1.		Except in cases of emergency no changes in the Work re	equired by the Contract Documents may be made	
18		л.	by the General Contractor (GC) without having prior appro-	val of the City Project Manager (CPM).	
19		В.	The City may at any time, without invalidating the Contrac	t and without Notice to Sureties, order changes in	
20			the Work by written Change Order. Such changes may incl	ude additions and/or deletions.	
21		C.	The Change Order (CO) is a Board of Public Works (BPW) fo	rm that is reviewed and approved by a specific	
22			process.		
23		D.	The CO form is typically made up of multiple Change Order	Requests (CORs) and/or Bid Items as appropriate	
24			depending on the type of project and how the contract was	s bid.	
25		Ε.	All CO documentation shall be processed through the Proje	ct Management Web Site (PMWS).	
26					
27	1.2.	REL/	LATED SPECIFICATION SECTIONS		
28		Α.	Section 01 26 13 Request for Information (RFI)		
29		В.	Section 01 26 46 Construction Bulletin (CB)		
30		C.	Section 01 26 63 Change Order Request (COR)		
31		D.	Section 01 31 23 Project Management Web Site (PN	/WS)	
32		Ε.	Section 01 91 00 Commissioning		
33					
34	1.3.	BOA	ARD OF PUBLIC WORKS PROCEDURE		
35		Α.	The Board of Public Works has a very explicit procedure for	the review and approval of all change orders	
36			associated with any Public Works Contract as follows:		
37			1. The Supervisory Chain of the CPM shall review and a	approve any CO under \$20,000 provided it does not	
38			include either of the following:		
39			a. The CO does not request a time extension to	the contract.	
40			b. The CO does not cause the contract continge	ancy sum to be exceeded.	
41			2. The Board of Public Works shall review and approve	any CO that requires any of the following:	
42			a. Any CO over \$20,000.		
43			b. Any CO requesting a time extension to the c	Sontract regardless of the monetary value of the CO.	
44			c. Any CO that that causes the contract conting	gency sum to be exceeded.	
45		в.	and of Public works generally meets every other we	ak and only once in August and December. The GC is	
40			cautioned that, under normal scheduling, a CO requiring a l	3PW review will take a minimum of two (2) weeks to	
47 10			achieve IIIidi approval.	us to the Work caused by the scheduling constraints	
40 40			af the Reard of Public Works	is to the work caused by the scheduling constraints	
49 50		C	SPECIAL NOTE: The GC is cautioned to never proceed uplos	is told to do so by the CPM . Only in rare instances	
50		С.	may the CPM give a written notice to proceed on a COP wit	bout an approved CO Proceeding without the	
52			written notice of the CPM or an approved CO is at the CC's	own risk	
52			written notice of the crist of an approved co is at the dc s	OWITTISK.	
55					

1 PART 2 – PRODUCTS

2

4

5

6 7

3 2.1. CHANGE ORDER FORM

A. The CO form is located on the Project Management Web Site. The CPM shall click the link in the left margin of the project web site opening a new form. Project information is pre-loaded, the CPM only needs to enter information and make attachments as needed to complete the form.

8 PART 3 - EXECUTION

9					
10	3.1.	PREP	REPARATION OF THE CHANGE ORDER		
11		Α.	The CPM shall prepare the required CO forms in the Project Management Web Site as follows:		
12			1.	Provide information for all contract information.	
13			2.	Provide a general description of the items described within the change order.	
14			3.	Provide detailed information for each Item on the CO form. At the option of the CPM, they may include	
15				multiple Change Order Requests each as their own item.	
16			4.	Provide required pricing and accounting information as needed for the item.	
17			5.	Insert attachments of contractor/architect provided information that clarifies and quantifies the CO.	
18				Attachments may include but not be limited to material lists, estimated labor, revised details or	
19				specifications, and other documents that may be related to the requested change.	
20			6.	Save the final version of the completed CO.	
21					
22	3.2.	EXEC	CUTION OF THE CHANGE ORDER		
23		Α.	Upon saving the CO as described in section 3.1 above, the software associated with the Project Management		
24			Web	Site shall notify the GC that the CO has been drafted and is ready for review. The GC shall do the following:	
25			1.	Open the CO form using the link provided in the email notification and review all items on the form.	
26			2.	The GC shall notify the CPM immediately of any errors or discrepancies on the form and shall not sign or	
27				save it.	
28			2	a. The CPM shall make any corrections as needed, re-save the form, and notify the GC.	
29			3. After	If/When the GC concurs with the CO it shall be reacted the GC shall digitally sign the form and click SAVE.	
30		в.	After	The GC digitally signs/saves the CO it shall be routed through the Project Management web site for	
31			addit	ional review and/or approvals. The CPW shall do the following:	
32			1. 2	Monitor the review process to ensure the software is working property at each review step.	
21			Ζ.	Schedulo the CO on the payt available RDW agonda if required	
25				a. Schedule the CO of the next available by wagenda in required.	
36				ii The GC and/or the Project Architect /Project Engineer (A/E PROLMGR) may be required to	
30				attend the BPW meeting to address specific information as it relates to the Work and/or	
38				materials associated with the CO	
39			3	Monitor final approval and distribution of the CO	
40			4.	Notify the GC that the CO has been completed.	
41			5.	Ensure that the CO is posted to the next Public Works payment schedule.	
42			6.	Verify that the GC's next Progress Payment-Schedule of Values show the CO as part of the contract sum.	
43		C.	Upon	n final approval of the CO the GC may proceed with executing the Work associated with the CO.	
44			- 1	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	
45					
46					
47				END OF SECTION	
48					
		SCHEDULE OF VALUES			
------	--------	---			
	1 C				
PARI	1-G				
	1.1.				
·	1.2.				
	1.3.				
	1.4.				
PART	2 – Pl	ODUCTS – THIS SECTION NOT USED			
PART	3 - EX				
	3.1.				
	3.2.	PROJECT MANAGEMENT WEBSITE SOV SPREADSHEET			
	3.3.	INITIAL SCHEDULE OF VALUES SUBMITTAL			
1	3.4.	SOV FOR PROGRESS PAYMENT REQUESTS			
PART	1 – G	ENERAL			
1.1.	SUI	/MARY			
	Α.	The Schedule of Values (SOV) is a Contractor provided statement that allocates portions of the total contract			
		sum to various portions of the contracted work and shall be the basis for reviewing the Contractors Progress Payment Requests.			
	B.				
	C.	The General Contractor shall be responsible for filling out and updating the SOV in the Project Management			
		website with each Progress Payment Request.			
1.2.	REL	ATED SPECIFICATIONS			
	A.	Section 01 26 63 Change Order (CO)			
	В.	Section 01 29 76 Progress Payment Procedures			
	C.	Section 01 31 23 Project Management Web Site (PMWS)			
	D.	Section 01 32 26 Construction Progress Reporting			
	E.	Section 01 33 23 Submittals			
	F.	Parts of this specification will reference articles within "The City of Madison FACILITIES MANAGEMENT			
		SPECIFICATIONs for Public Works Construction".			
		1. Use the following link to access the FACILITIES MANAGEMENT SPECIFICATIONs web page:			
		http://www.cityofmadison.com/business/pw/specs.cfm			
		a. Click on the "Part" chapter identified in the specification text. For example, if the specification			
		says "Refer to City of Madison FACILITIES MANAGEMENT SPECIFICATION 210.2" click the link fo			
		Part II. the Part II PDF will open.			
		b. Scroll through the index of Part II for specification 210.2 and click the text link which will take vo			
		to the referenced text.			
1.3.	REL	ATED DOCUMENTS			
	Α.	The following documents shall be used as the basis for initiating and maintaining the SOV worksheets through			
		the execution of this contract.			
		1. Drawing documents and specifications (including general provisions) as provided with the bid set			
		documents and any published addendums.			
		2. Documents associated with revisions or clarifications to number 1 above after awarding of the contract			
		including but not limited to:			
		a. Construction Bulletins			
		b. Request for Information			
		c. Approved Change Orders			
		3. The latest daily/weekly Construction Progress Report			
		4. Other specifications as identified in Section 1.2 above			
1.4.	BAS	IS OF VALUES			
	Α.	The Contractor shall provide a breakdown of the Contract Sum in sufficient detail to assist the Architect and Ci			
		Project Manager in evaluating Progress Payment Requests. The breakdown detail may require a labor and			

Β. 1 The total sum of all items shall equal the Contract Sum. 2 3 PART 2 - PRODUCTS - THIS SECTION NOT USED 4 5 **PART 3 - EXECUTION** 6 7 **APPLICATION FOR PAYMENT** 3.1. 8 The Contractor shall use the Project Management website or Payment with each Progress Payment Request. Α. 9 Β. Completely fill out the Pay Application per the tutorial provided for the PMWS 10 1. Fill out to reflect the current status of the contract through the payment date being requested. 11 2. The City of Madison calculates retainage on Public Works Contracts as follows: In general, across the duration of the contract, 2.5% of the total contract sum, including change 12 a. 13 orders, is withheld for retainage as referenced from the City of Madison FACILITIES 14 **MANAGEMENT SPECIFICATION 110.2:** Beginning with Progress Payment 1, 5% retainage will be withheld until such time that 50% 15 i. 16 of the total contract sum has been paid out. 17 ii. No additional retainage will be withheld after 50% of the total contract sum has been paid, unless additional change orders have been approved after the 50% milestone has been 18 reached. Per City of Madison FACILITIES MANAGEMENT SPECIFICATION 110.2, additional 19 20 retainage up to 10%, may be held in the event there are holds placed by Affirmative Action 21 or liquidated damages by BPW. 22 iii. Retainage for additional change orders after the 50% milestone will be withheld at the rate 23 of 2.5% of the total cost of the change order. 24 iv. Retainage is based on the change orders posted to the City's contract worksheet at the 25 time the progress payment is processed. 26 C. Only change orders that have been finalized and posted to the City of Madison's Application for Partial Payment 27 worksheet may be itemized into the SOV documents. 28 D. The Contractor shall sign and date the application. 29 30 3.2. PROJECT MANAGEMENT WEBSITE SOV SPREADSHEET 31 The Contractor shall use the PMWS spreadsheet provided by the City to itemize their SOV for this contract. Α. 32 Provide additional sheets as necessary. 33 Β. Provide information by any method that allocates portions of the total contract sum to various portions of the 34 contracted work. Possible methods include combinations of the following: 35 1. By division of work 36 2. By contractor, sub-contractor, sub sub-contractor 37 3. By specialty item or group 38 4. Other methods of breakdown as may be requested by the City Project Manager or City Construction 39 Manager at the pre-construction meeting. 40 C. Provide total cost of the item/description of work including proportionate shares of profit and overhead related 41 to the item. 42 **INITIAL SCHEDULE OF VALUES SUBMITTAL** 43 3.3. 44 Α. The Contractor shall upload their initial SOV to the Project Management Web Site, no later than five (5) working 45 days after the Pre-construction Meeting. 46 1. The level of detail shall be as described in section 3.2 above. The Project Architect /Project Engineer (A/E PROJ MGR) and the City Project Manager (CPM) shall review the 47 Β. 48 SOV as any other submittal and may require modifications to reflect additional detail as necessary. 49 C. The Contractor shall resubmit the SOV as necessary until such time as the A/E PROJ MGR and CPM have 50 sufficient detail for assessing and approving future Progress Payment Applications. 51 D. Progress Payment Application 1 will not be processed until such time as the Contractor has met this requirement 52 regardless of the amount of work completed per the application. 53 54 3.4. SOV FOR PROGRESS PAYMENT REQUESTS The Contractor shall update the initial SOV with each Progress Payment Application as follows: 55 Α. 56 1. Initial items and values as part of Section 3.3 above will not be adjusted once the original Schedule of 57 Values submittal has been approved.

1		2. Change orders shall be added as additional items and values at the bottom of the SOV as they become
2		approved and posted to the City's contract worksheet. The value for each change order shall be the
3		value indicated on the SOV and shall stand alone. Values shall not be split out or combined with other
4		existing items with similar work descriptions on the original SOV.
5		3. Fill out columns to properly reflect the work completed and materials received since the last Progress
6		Payment Application.
7		4. Only materials delivered and stored on the project site may be reflected on SOV progress updates.
8	В.	Provide an updated project schedule with each Progress Payment application.
9	С.	See Specification 01 29 76 Progress Payment Procedures for additional information on submitting Progress
10		Payment Applications.
11		
12		
13		
14		END OF SECTION
15		

				SECTION 01 29 76 PROGRESS PAYMENT PROCEDURES
DACT				
PARI	1 – GE	NERAL		
1	.1.	SUMMAR	Υ	
1		RELATED	SPECIFICATIO	NS
1	.3.	RELATED		
1	.4.	PROGRESS	S PAYMENT N	AILESTONES
1	.5.	PROGRESS	5 PAYMENT S	
PART	2 - PR(THIS SECTION	NOT USED
PARI	3 - EXE	CUTION		
3	.1.	GENERAL		
3	.3.	CITYPROJ	ECT MANAG	EK PROCEDURE
PΔRT	1 – GF	NFRAI		
	1 01			
1.1.	SUM	IMARY		
	A.	The Ger request	neral Contrac	tor (GC) shall review this and all related specifications prior to submitting progress paym
	В.	Progres Manage	s payment re ement Web S	equests (Partial Payment-PP) for this contract shall be applied for by the GC in the Projectite (PMWS)
	C.	The City	y Project Mar	nager (CPM) shall review and amend or approve the PP on the Project Management Wel
		Site.		
	D.	After ap	proval of the	PP by the CPM, they shall forward the PP to the appropriate agencies for BPW contract
		review	and payment	processing.
1.2.	RELA	ATED SPEC	IFICATIONS	
	Α.	Section	01 26 63	Change Order (CO)
	В.	Section	01 29 73	Schedule of Values
	C.	Section	01 31 19	Progress Meetings
	D.	Section	01 31 23	Project Management Web Site (PMWS)
	Ε.	Section	01 32 16	Construction Progress Schedules
	F.	Section	01 32 26	Construction Progress Reporting
	G.	Section	01 33 23	Submittals
	Н.	Section	01 45 16	Field Quality Control Procedures
	I.	Section	01 77 00	Closeout Procedures
	J.	Section	01 78 13	Completion and Correction List
	К	Section	01 78 23	Operation and Maintenance Data
	L.	Section	01 78 36	Warranties
	М.	Section	01 78 39	As-Built Drawings
	N.	Section	01 78 43	Spare Parts and Extra Materials
	0.	Section	01 79 00	Demonstration and Training
1.3.	RELA	TED DOCI	JMENTS	
	Α.	The foll	owing docun	nents shall be used when evaluating PP requests.
		1.	Daily and we	ekly construction progress reports filed since the last payment request.
		2.	Contractors S	Schedule of Values as updated from the last payment request. See Specification 01 29 7
		3.	Any docume	nt that may be required to be submitted for review and approval, as noted by the
			specification	s listed in Section 1.2 above, or the Progress Payment Milestone Schedule in Section 1.4
			below, to acl	nieve a required bench mark of contract progression or contract requirement.
1.4.	PRO	GRESS PA		STONES
	А.	City Eng	gineering-Fac	lility Management has developed the Project Payment Milestone Schedule (Section 1.4
		below)	to assist the	GC in providing required construction specific documentation and general contractual
	_	docume	entation in a	timely manner.
	В.	The Pro	gress Payme	nt Milestone Schedule is not an all inclusive list. Multiple agencies review progress payn
		request	s and contra	ct closeout requests. Missing, incomplete, or incorrect documentation for any agency m

1		be a cause for not processing progress payments. It shall be the sole responsibility of the Contractor for
2		providing documentation as required or requested to the appropriate agencies.
3	С.	The milestone schedule is based on the contract total sum and shall be valid for most contracts. Milestone
4		submittals will be required with whatever progress payment hits the percentage of contract total indicated in
5		the schedule.
6	D.	The CPM shall review the milestone schedule with each progress payment request and at their option may elect
7		to hold processing the progress payment until such time as the contractor has met the requirements for
8		providing construction specific documentation.
9	Ε.	It shall be the General Contractors responsibility to comply with all BPW Contract Administration requirements
10		and related deadlines as outlined in the Award Letter, Award Checklist, and Start Work Letter.
11		

Progress Payn	nent (PP) Miles	tone Schedule
Milestone Description	Due Before	Remarks
 BPW Contract Administration Documentation Workforce profiles Best Value Contracting Documentation Sub-contractors prequalification approval & Affirmative Action plans Submittals Schedule Other as may be required 	PP-1, or start work as applicable	 For GC and Sub-contractors before PP- 1 regardless of scheduling Sub-contractors (if applicable), due 10 days before they may start work Sub-contractors (if applicable), due 10 days before they may start work Specification 01 32 19
Required Construction Submittals/Administrative Documents Contractors Project Directory Schedule of Values Waste Management Plan Closeout Requirement Checklist Warranty Checklist Time Lapse Construction Camera (camera installed and operational)	PP-1	References Specification 01 31 23 Specification 01 29 73 Specification 01 74 19 Specification 01 77 00 Specification 01 78 36 Specification 01 32 33
Early submittals, per submittal schedule Detailed Contract Schedules	PP-1	 Specifications for specific requirements Specification 01 32 19, Examples: concrete mix, structural steel, products with long lead times See Specification 01 32 16
General Construction Progress Requirements are all up to date Progress Schedules Submittals/Re-submittals (ongoing) Schedule of Values Progress Reporting LEED Documentation Waste Management documentation QMOs are being addressed and closed Progress Cleaning As-Built Drawings * All of the above are being update	Each future PP d on the Project	 Verified with each Progress Payment Request Specification 01 32 16 Specification 01 33 23 Specification 01 29 73 Specification 01 32 26 All specifications with LEED documentation requirements Specification 01 74 19 Specification 01 45 16 Specification 01 74 13 Specification 01 78 39
BPW Contract Administration Documentation	25% CT	See 1.4.E above. This progress payment will be

Progress Payn	nent (PP) Miles	tone Schedule		
Milestone Description	Due Before	Remarks		
 Weekly payroll reports 	or	with held by BPW for any missing contractual		
 Best Value Contracting Reports 	PP 2	documentation.		
SBE Reports				
Construction Progress Milestones				
Construction/Contract Closeout	50% CT	Specification 01 31 19		
Meeting #1				
Submittals/Re-submittals complete		Specification 01 33 23		
Operation and Maintenance (O.S.M) drafts	60% CT	Specification 01 78 22		
Operation and Maintenance (O & M) draits	00% C1	• Specification 01 78 25		
Construction/Contract Closeout Meeting #2		Specification 01 31 19		
Construction closeout checklist	70% CT	Specification 01 77 00		
RDW Contract Administration Desumantation		This is a recommendation to the GC and is not a		
Brivi Contract Auministration Documentation	80% CT	requirement of this PP.		
Request Finalization Review from BPW		Specification 01 77 00		
Construction Progress Milestones				
Operation and Maintenance (O & M)		Specification 01 78 23		
finals, accepted	80% CT	Creation 01 45 10, Howe that		
 All major QMO issues resolved 		 Specification 01 45 16; items that could provent occupancy. 		
As-Built Drawings Division Trades				
ready for GC review		Specification 01 78 39		
All of the following shall be completed for this		Contractor to determine the proper order of		
PP:		completion:		
Bogulatory Inspections completed		Coverning ordinances and statutes		
All OMO reports closed	00% CT	 Specification 01 45 16 		
Demonstration and Training	90%C1	• Specification 01 45 10		
benionstrution and framing				
completed		• Specification 01 79 00		
completedAttic Stock completed		Specification 01 79 00Specification 01 78 43		
completedAttic Stock completedFinal Cleaning		 Specification 01 79 00 Specification 01 78 43 Specification 01 74 13 		
completed Attic Stock completed Final Cleaning 		 Specification 01 79 00 Specification 01 78 43 Specification 01 74 13 		
completed Attic Stock completed Final Cleaning Construction Closeout Procedures:		 Specification 01 79 00 Specification 01 78 43 Specification 01 74 13 Specification 01 77 00 		
completed Attic Stock completed Final Cleaning Construction Closeout Procedures: Letter of Substantial Compliance sent		 Specification 01 79 00 Specification 01 78 43 Specification 01 74 13 Specification 01 77 00 Generated/Signed by the Architect 		
completed Attic Stock completed Final Cleaning Construction Closeout Procedures: Letter of Substantial Compliance sent to BI and DHS as needed		 Specification 01 79 00 Specification 01 78 43 Specification 01 74 13 Specification 01 77 00 Generated/Signed by the Architect 		
completed Attic Stock completed Final Cleaning Construction Closeout Procedures: Letter of Substantial Compliance sent to BI and DHS as needed Certificate of Occupancy issued 	100% CT	 Specification 01 79 00 Specification 01 78 43 Specification 01 74 13 Specification 01 77 00 Generated/Signed by the Architect Building Inspection 		
 completed Attic Stock completed Final Cleaning Construction Closeout Procedures: Letter of Substantial Compliance sent to BI and DHS as needed Certificate of Occupancy issued As-Built Drawings, finals, accepted 	100% CT	 Specification 01 79 00 Specification 01 78 43 Specification 01 74 13 Specification 01 77 00 Generated/Signed by the Architect Building Inspection Specification 01 78 39 		
 completed Attic Stock completed Final Cleaning Construction Closeout Procedures: Letter of Substantial Compliance sent to BI and DHS as needed Certificate of Occupancy issued As-Built Drawings, finals, accepted City Letter of Substantial Completion 	100% CT	 Specification 01 79 00 Specification 01 78 43 Specification 01 74 13 Specification 01 77 00 Generated/Signed by the Architect Building Inspection Specification 01 78 39 Signed by the City Engineer Signed by the City Engineer 		
completed Attic Stock completed Final Cleaning Construction Closeout Procedures: Letter of Substantial Compliance sent to BI and DHS as needed Certificate of Occupancy issued As-Built Drawings, finals, accepted City Letter of Substantial Completion Warranty letters dated and issued 	100% CT	 Specification 01 79 00 Specification 01 78 43 Specification 01 74 13 Specification 01 77 00 Generated/Signed by the Architect Building Inspection Specification 01 78 39 Signed by the City Engineer Specification 01 78 36 		
 completed Attic Stock completed Final Cleaning Construction Closeout Procedures: Letter of Substantial Compliance sent to BI and DHS as needed Certificate of Occupancy issued As-Built Drawings, finals, accepted City Letter of Substantial Completion Warranty letters dated and issued 	100% CT	 Specification 01 79 00 Specification 01 78 43 Specification 01 74 13 Specification 01 77 00 Generated/Signed by the Architect Building Inspection Specification 01 78 39 Signed by the City Engineer Specification 01 78 36 		
 completed Attic Stock completed Final Cleaning Construction Closeout Procedures: Letter of Substantial Compliance sent to BI and DHS as needed Certificate of Occupancy issued As-Built Drawings, finals, accepted City Letter of Substantial Completion Warranty letters dated and issued BPW Contract Administration Documentation 	100% CT	 Specification 01 79 00 Specification 01 78 43 Specification 01 74 13 Specification 01 77 00 Generated/Signed by the Architect Building Inspection Specification 01 78 39 Signed by the City Engineer Specification 01 78 36 		
completed Attic Stock completed Final Cleaning Construction Closeout Procedures: Letter of Substantial Compliance sent to BI and DHS as needed Certificate of Occupancy issued As-Built Drawings, finals, accepted City Letter of Substantial Completion Warranty letters dated and issued * Completion of t BPW Contract Administration Documentation Contract Closeout Procedures	100% CT	 Specification 01 79 00 Specification 01 78 43 Specification 01 74 13 Specification 01 77 00 Generated/Signed by the Architect Building Inspection Specification 01 78 39 Signed by the City Engineer Specification 01 78 36 year warranty. Specification 01 77 00 		
completed Attic Stock completed Final Cleaning Construction Closeout Procedures: Letter of Substantial Compliance sent to BI and DHS as needed Certificate of Occupancy issued As-Built Drawings, finals, accepted City Letter of Substantial Completion Warranty letters dated and issued * Completion of t BPW Contract Administration Documentation Contract Closeout Procedures Construction Closeout has been	100% CT	 Specification 01 79 00 Specification 01 78 43 Specification 01 74 13 Specification 01 77 00 Generated/Signed by the Architect Building Inspection Specification 01 78 39 Signed by the City Engineer Specification 01 78 36 		
completed Attic Stock completed Final Cleaning Construction Closeout Procedures: Letter of Substantial Compliance sent to BI and DHS as needed Certificate of Occupancy issued As-Built Drawings, finals, accepted City Letter of Substantial Completion Warranty letters dated and issued * Completion of t BPW Contract Administration Documentation Contract Closeout Procedures Construction Closeout has been completed	100% CT his begins the or Final	 Specification 01 79 00 Specification 01 78 43 Specification 01 74 13 Specification 01 77 00 Generated/Signed by the Architect Building Inspection Specification 01 78 39 Signed by the City Engineer Specification 01 78 36 year warranty. Specification 01 77 00 		
completed Attic Stock completed Final Cleaning Construction Closeout Procedures: Letter of Substantial Compliance sent to BI and DHS as needed Certificate of Occupancy issued As-Built Drawings, finals, accepted City Letter of Substantial Completion Warranty letters dated and issued * Completion of t BPW Contract Administration Documentation Contract Closeout Procedures Construction Closeout has been completed Contractor requests final payment of	100% CT his begins the or Final	 Specification 01 79 00 Specification 01 78 43 Specification 01 74 13 Specification 01 77 00 Generated/Signed by the Architect Building Inspection Specification 01 78 39 Signed by the City Engineer Specification 01 78 36 year warranty. Specification 01 77 00 		
completed Attic Stock completed Final Cleaning Construction Closeout Procedures: Letter of Substantial Compliance sent to BI and DHS as needed Certificate of Occupancy issued As-Built Drawings, finals, accepted City Letter of Substantial Completion Warranty letters dated and issued BPW Contract Administration Documentation Contract Closeout Procedures Construction Closeout has been completed Contractor requests final payment of retainage upon receiving City Letter of 	100% CT his begins the or Final	 Specification 01 79 00 Specification 01 78 43 Specification 01 74 13 Specification 01 77 00 Generated/Signed by the Architect Building Inspection Specification 01 78 39 Signed by the City Engineer Specification 01 78 36 year warranty. Specification 01 77 00 		

	-	Milestone Description Due Before Remarks
		All BPW contractual requirements are verified Section 2.1 Section
		* Completion of this closes the contract but not the warranty period/bond.
		NOTE: CT = Contract Total less held retainage
1.5.	PROC	GRESS PAYMENT SUBMITTAL
	Α.	Each progress payment submittal shall be completed in the Project Management Website. See guide on the
	В.	Submit all required construction progress documentation to the appropriate Project Management Web Site
		component as described in guides.
	C.	In general the following shall apply to all PP requests:
		1. Materials or products:
		a. On order, being shipped, etc. may not be invoiced.
		b. Received and stored on the project site may be invoiced.
		c. Being manufactured off site at any location may not be invoiced (example: cabinetry, ductwor
		etc.)
		d. Completed products stored off site locally waiting for delivery to the project site may be invoid
		with prior approval by the CPM. All of the following conditions must be met to be allowed:
		I. Items must be visually inspected by CPM to verify product is complete.
		ii. Item must be stored inside a compatible structure and the structure and contents mus
		insured.
		iii. Contractor is responsible for condition until installation is completed.
		2. All labor and equipment, including rental time for the current progress period may be invoiced.
		3. Only completed installations may be invoiced to 100% based on the Schedule of Values.
	D.	DO NOT submit BPW Contract Administration Documentation for review with Progress Payment Requests,
		submit them directly to the correct agency and in the correct format as instructed from information in your B
		Contract Award Packet instructions.
D	• • • •	
PART	2 - PRC	UDUCIS - THIS SECTION NOT USED
DADT	э гуг	
PARTS	3 - EXE	
3.1.	GENE	ERAL CONTRACTOR PROCEDURE
	A.	The GC shall use the Project Management Website for each PP request.
		1. The GC shall subtotal the work completed to date for all of the original Schedule of Value items.
		2. Ensure that any newly posted change orders have been entered.
		3. The GC shall submit the PP request in the Project Management Website. The username and date will
		automatically recorded.
		4. The GC shall provide the dates from and to for the PP being requested.
		5. The GC shall provide the list of all contractors/sub-contractors that were actively working during the
		dates indicated above. The guide details the appropriate location for this list.
		a. All contractors/sub-contractors named must be in compliance with all City requirements (Pre-
		qualified, Affirmative Action Plan on file, etc). The PP will be held and not processed by the Ci
		Madison until all contractors/sub-contractors are in compliance.
		b. Do not list the names of suppliers or manufacturers, doing so will slow down processing and
		require a re-submittal of the paperwork.
		6. The GC shall attach a copy of the current Project Schedule.
		PROJECT MANAGER PROCEDURE
2 2		
3.3.		The CPM shall review all documents submitted by the CC to oncure the schedule of values accurately reflects
3.3.	CITY A.	The CPM shall review all documents submitted by the GC to ensure the schedule of values accurately reflects work completed to date
3.3.	CITY A.	The CPM shall review all documents submitted by the GC to ensure the schedule of values accurately reflects work completed to date.

When verified, the CPM shall send the PP and required documentation to the appropriate City agencies for C. 1 2 further processing of the payment request. The PP processing will be completed and available for view within the PMWS. 3 D. 4 END OF SECTION

5

			SECTION 01 31 13 PROJECT COORDINATION
PART	1 – GE	NERAL	
1	.1.	SUMMARY	
1	.2.	RELATED SPECIE	FICATIONS
1	.3.	GENERAL REOU	IREMENTS
1	.4.	GENERAL CONT	RACTOR PERFORMANCE REQUIREMENTS
-	.5.	SUB-CONTRACT	OR PERFORMANCE REQUIREMENTS
PART	2 – PR	ODUCTS – THIS	SECTION NOT USED
PART	3 – EX	ECUTION – THIS	SECTION NOT USED
PART	1 – GI	NERAL	
1.1.	SUN	IMARY	
	Α.	Project Coorc of proper coc	Jination covers many areas within the execution of the Contract Documents and the requirements ordination are the applicable to all contractors executing the Work of this contract.
	В.	This specifica Sub-contracto	tion provides general information regarding project coordination for the General Contractor and a ors. All contractors shall be familiar with project coordination requirements and responsibilities
		that may be o	defined in other specification within these Contract Documents.
	C.	The General (Contractor shall at all times be responsible for the project, project site, and execution of the
		Contract Doc	uments.
1.2.	REL/	ATED SPECIFICAT	FIONS
	Α.	Section 01 29	976 Progress Payment Procedures
	В.	Section 01 31	19 Progress Meetings
	C.	Section 01 31	23 Project Management Web Site
	D.	Section 01 32	2 16 Construction Progress Schedules
	Ε.	Section 01 32	2 19 Submittals Schedule
	F.	Section 01 33	3 23 Submittals
	G.	Section 01 43	39 Mockups
	Н.	Section 01 45	5 16 Field Quality Control Procedures
	I.	Section 01 60) 00 Product Requirements
	J.	Section 01 77	'00 Closeout Procedures, including all specifications referenced therein
	К.	Section 01 91	. 00 Commissioning
1.3.	GEN	ERAL REQUIREN	ΛΕΝΤS
	A.	The following	general requirements shall applicable to all contractors:
		1. Coope	erate with the Owner, all authorized Owner Representatives, Project Architect and all consultants
		the O	wner.
		2. Mater	rials, products, and equipment shall be new, as specified and to industry standards except where
		other	wise noted.
		3. Labor	and workmanship shall be of a high quality and to industry standards.
	В.	Existing cond	itions:
		1. Verify	all existing conditions noted in the contract documents with actual filed locations. Verify
		dimer	nsions, sizes and locations, of structural, equipment, mechanical and utility components.
		2. Repor	rt any inconsistencies, errors, omissions, or code violations in writing to the General Contractor (G
		imme	diately.
		3. Annot	tate any inconsistencies, errors, omissions on the GC As-Built record drawings immediately for
		future	e reference.
	C.	Contract Doc	uments:
		1. The C	ontract Documents are intended to include everything necessary to perform the work. Every iten
		requir	red may not be specifically mentioned, shown, or detailed.
		a.	Except where specifically stated all systems and equipment shall be complete, installed, and full
			operable.
		L.	If a conflict exists within the contract documents the contractor shall furnish the item system of
		D.	In a contract exists within the contract documents the contractor shall runnish the item, system, c
		D.	workmanship of the highest quality, largest, largest quantity, or most closely fits the intent of th

1			c. Manufacturers recommended installation details shall be verified and used prior to installation of
2			products and equipment so as to not void warranties.
3		D.	Errors and Omissions
4			1. No Contractor shall take any advantage of any apparent error or omission in the construction documents.
5			2. The City of Madison shall be permitted to make such corrections and interpretations as may be deemed
6			necessary for the fulfillment of the intent of the construction documents.
7		Ε.	Owners Representatives
8			1. All contractors shall be familiar with various Owner Representatives having Quality Management
9			responsibilities for the duration of this project including but not limited to the following:
10			a. Project Architect, responsible for all decisions affecting the code compliance and design intent of
11			the construction documents.
12			b. Consulting Architects and Engineers, responsible for providing consulting services to the Project
13			Architect, Owner, and City Project Manager, also responsible for Quality Management of the
14			construction documents.
15			c. Owner, the designated representative of the City Agency that will occupy the project upon
16			completion.
17			d. City Project Manager, responsible for all day to day decisions regarding the execution and
18			performance of this Public Works Contract.
19			e. Consulting City Staff, responsible for providing consulting services to the Project Architect, Owner,
20			and City Project Manager, also responsible for Quality Management of the construction
21			documents.
22			f. Commissioning Agent (CxA), responsible for ensuring that the project is meeting the Owner's
23			Project Requirements and related quality assurance procedures.
24			2. Owner Representatives shall be attending progress meetings, pre-installation meetings, performing or
25			being present for final testing and acceptance and quality management reporting during the execution of
26			the contract documents as outlined in other specifications.
27			
28	1.4.	GENE	RAL CONTRACTOR PERFORMANCE REQUIREMENTS
29		Α.	Assume the responsibility for all Work specified in the Contract Documents except where specifically identified
30			to be performed by the Owner or other contractor separately hired by the Owner.
31			1. Coordinate all work by Owner, equipment provided Owner, or contractor hired by the Owner into the
32			project schedule.
33		В.	Provide all construction management responsibilities as specified in other Division 1 specifications including but
34			not limited to:
35			1. Scheduling of work
36			2. Coordination of work between other Trades and Sub-contractors
37			3. Construction administration and management
38			Site layout, cleanliness, and protection of completed work/stored materials
39			5. Waste Management
40			6. Quality Assurance and Quality Control
41		C.	Use Diggers Hotline and private utility locating companies to accurately locate all public and private utilities on
42			the property as needed. The GC is responsible for any repair or replacement to any public or private utility
43			damaged during the execution of the Work
44		D.	Report any inconsistencies, errors, omissions, or code violations in writing to the Project Architect immediately.
45			Failure to report inconsistencies prior to beginning work shall indicate that the GC accepted all existing
46			conditions.
47		Ε.	The GC shall be responsible for assigning work and related responsibilities where the Contract Documents may
48			not clearly state who is responsible for providing the work, material, or product.
49		F.	Provide construction management oversight of all items described in Section 1.5 below.
50		G.	Coordinate and assist CxA as outlined within 01 91 00 and as directed by Owner.
51		.	
52	1.5.	SUB-C	UNIKACIOR PERFORMANCE REQUIREMENTS
	-		De familieu with all af the continent de sources de la de
53	-	Α.	Be familiar with all of the contract documents as they pertain to your Work, adjacent work and the overall
53 54	-	A.	Be familiar with all of the contract documents as they pertain to your Work, adjacent work and the overall progress of the project.
53 54 55	-	A.	 Be familiar with all of the contract documents as they pertain to your Work, adjacent work and the overall progress of the project. 1. All Sub-contractors shall be familiar with all Division 1 specifications as they may apply to progress,
53 54 55 56	-	A.	 Be familiar with all of the contract documents as they pertain to your Work, adjacent work and the overall progress of the project. 1. All Sub-contractors shall be familiar with all Division 1 specifications as they may apply to progress, progress payments, quality control construction management, and closeout of the contract.

1		1.	Perform your work in proper sequence according to the GC's project schedule and in relation to the work
2			of other trades.
3		2.	Notify other sub-contractors and trades whose work may be connected to, combined with, or influenced
4			by your work and allow them reasonable time and access to complete their work.
5		3.	Join your work to the work of others in accordance with the intent of the Contract Documents.
6		4.	Order materials and schedule deliveries to facilitate the general progress of the Work.
7	С.	Cooper	rate with all other trades to facilitate the general progress of the work. This shall include providing every
8		reasona	able opportunity for the installation of work by others and the storage of their materials and equipment.
9		1.	In no case shall any contractor exclude from the premises or work any Sub-contractor or their employees.
10		2.	In no case shall any contractor interfere with the execution or installation of Work by any other Sub-
11			contractor or their employees.
12	D.	Arrange	e your work, equipment, and materials and dispose of your construction waste so as to not interfere with
13		the wo	rk or storage of materials of others.
14	Ε.	Coordir	nate all work as indicated during pre-installation meetings with Owner Representatives, the GC and other
15		trades.	Any work improperly coordinated shall be relocated as designated by the Owner Representative at no
16		additio	nal cost to the City.
17	F.	Coordir	nate and assist CxA as outlined within 01 91 00 and as directed by Owner.
18			
19	<u> PART 2 – PRC</u>	DUCTS -	- THIS SECTION NOT USED
20			
21	<u> PART 3 – EXE</u>	CUTION	– THIS SECTION NOT USED
22			
23			
24			
25			END OF SECTION
26			

			SECTION 01 31 19 PROJECT MEETINGS
PART 1	1 – GF	NFRAI	
1	1		RY
1	2		SPECIFICATIONS
1	.2. २		MEETING TYPES
1	.э. л	GENERAL	
	. 4 .)		
	2 - FN D EV		
2	2 - EA		
2	.1. ว		
2	.∠. ⊃		
2	.5. ⊿		TALLATION MEETINGS
2	.4. 6		
כ ר	.0		
3	./	UTHER S	PECIAL MEETINGS
PART :	1 – G	ENERAL	
1.1.	SUN	IMARY	
	Α.	The pu	urpose of this specification is to identify various project related meetings and the responsible parties fo
		schedi	uling, agendas, minutes, and required attendance.
	В.	This sp	pecification is not intended to be inclusive of all meeting types or a complete list of required meetings.
	C.	This sp	pecification is not intended to cover planning and execution meetings between the General Contractor
		(GC) a	nd their sub-contractors.
1.2.	REL	ATED SPE	CIFICATIONS
	Α.	01 31	23 Project Management Web Site
	В.	01 32	16 Construction Progress Schedules
	C.	01 43	39 Mockups
	D.	01 91	00 Commissioning
			-
1.3.	PRO	JECT MEE	TING TYPES
	A.	The fo	llowing project meeting types may be used but not limited to the following
		1.	Preconstruction Meeting
		2.	Project Management Web Site – Tutorial Meeting
		3.	Construction Progress Meetings
		4.	Pre-installation Meetings (including mock-up review meetings)
		5.	Weekly Trade Meetings
		6.	Special Meetings
		7.	Commissioning Meetings
1.4.	GEN	ERAL REC	DUIREMENTS
	Α.	Repres	sentatives of Contractors. Subcontractors. and suppliers attending meetings shall be qualified and
	-	autho	rized to act on behalf of the entity each represents.
			· ·
PART 2	2 – PF	RODUCTS	- NOT USED IN THIS SECTION
	2 <u>-</u> EV		
	J - EA		
3.1.	PRE	CONSTRU	ICTION MEETING
	Α.	After e	execution of the Contract the City Project Manager (CPM) shall schedule and conduct the Preconstruction
		Meeti	ng at the Owner's facilities. The CPM shall coordinate the meeting agenda with the Project Architect a
		the GC	Project Manager.
	В.	The CF	ንM shall be responsible for the final agenda.
	C.	The CF	PM and Project Architect shall take notes on the meeting and post completed meeting minutes.
	-	Attor	tance shall be required by all of the following:
	D.	Attent	ance shan be required by an or the renorming.

1			2.	Architect and applicable sub consultant(s)
2			3.	General Contractor and applicable subcontractors and suppliers
3			4.	City Quality Management Staff
4			5.	Commissioning Agent
5			6.	Others, as may be invited for particular agenda items.
6		E.	Topics	of the Preconstruction Meeting shall include but not be limited to the following:
7			1.	Staff and contractor introductions
8			2.	Completion Date
9			3.	BPW Administrative requirements and due outs
10			5.	a Small Business Enternrise (SRE) (if annlinable)
11				b Certified payroll forms
12				
12				d Bort Volue Contracting (BVC)
13			л	General Society Management Division 1 Specifications, including:
14 15			4.	General racinty Management Division 1 Specifications, Including,
15				a. Section 01 29 /6 Progress Payment Procedures
16				b. Section 01 31 23 Project Management Web Site (overview)
1/				c. Section 01 45 16 Field Quality Control Procedures
18				d. Section 01 77 00 Closeout Procedures
19				e. Section 01 91 00 Commissioning
20			5.	Project Meeting scheduling
21				a. Section 01 31 19 Project Meetings
22			6.	Construction Schedule
23			7.	Commissioning Process
24				
25	3.2.	PROJE	ECT MAN	NAGEMENT WEB SITE – TUTORIAL MEETING
26		Α.	The CP	PM shall schedule and conduct a virtual tutorial presentation of the PMWS prior to the beginning of
27			constru	uction.
28		В.	The CP	PM shall be responsible for the final agenda, there will be no minutes.
29		C.	The re	quired attendance list in 3.1.D. above shall apply except for City Staff in items 1 and 4 who are already
20			.	
50			familia	ar with the PMWS system.
30 31			familia	ar with the PMWS system.
30 31 32	3.3.	CONS	familia TRUCTIO	ar with the PMWS system.
30 31 32 33	3.3.	CONS A.	familia TRUCTIC	ar with the PMWS system. ON PROGRESS MEETINGS eral, all of the following shall apply:
31 32 33 34	3.3.	CONS A.	familia TRUCTIC In gene 1.	ar with the PMWS system. ON PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and
31 32 33 34 35	3.3.	CONS A.	familia TRUCTIC In gene 1.	ar with the PMWS system. ON PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents
31 32 33 34 35 36	3.3.	CONS A.	familia TRUCTIC In gene 1. 2	ar with the PMWS system. ON PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1 D, above
30 31 32 33 34 35 36 37	3.3.	CONS [®] A.	familia TRUCTIC In gene 1. 2. The Ge	ar with the PMWS system. ON PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. approximation of the shall of the shall in the sha
30 31 32 33 34 35 36 37 38	3.3.	CONS A. B.	familia TRUCTIC In gene 1. 2. The Ge	ar with the PMWS system. ON PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required
30 31 32 33 34 35 36 37 38 20	3.3.	CONS A. B.	familia TRUCTIC In gene 1. 2. The Ge 1. 2	ar with the PMWS system. ON PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Propage aganda for meetings but not limited to the following:
30 31 32 33 34 35 36 37 38 39	3.3.	CONS A. B.	familia TRUCTIC In gene 1. 2. The Ge 1. 2.	ar with the PMWS system. ON PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following:
30 31 32 33 34 35 36 37 38 39 40	3.3.	CONS [®] A. B.	familia TRUCTIC In gene 1. 2. The Ge 1. 2.	ar with the PMWS system. ON PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following: a. Safety
30 31 32 33 34 35 36 37 38 39 40 41	3.3.	CONS A. B.	familia TRUCTIC In gene 1. 2. The Ge 1. 2.	 ar with the PMWS system. ON PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following: a. Safety b. Current Schedule, including review of the critical path and 6-week look ahead schedule
30 31 32 33 34 35 36 37 38 39 40 41 42	3.3.	CONS A. B.	familia TRUCTIC In gene 1. 2. The Ge 1. 2.	 ar with the PMWS system. ON PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following: a. Safety b. Current Schedule, including review of the critical path and 6-week look ahead schedule c. Status of project related documentation (Submittals, RFIs, CBs, etc.)
30 31 32 33 34 35 36 37 38 39 40 41 42 43	3.3.	CONS A. B.	familia TRUCTIC In gene 1. 2. The Ge 1. 2.	 ar with the PMWS system. ON PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following: a. Safety b. Current Schedule, including review of the critical path and 6-week look ahead schedule c. Status of project related documentation (Submittals, RFIs, CBs, etc.) d. Quality Observation Log and status of correction of deficient items
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44	3.3.	CONS A. B.	familia In gene 1. 2. The Ge 1. 2.	 ar with the PMWS system. ON PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following: a. Safety b. Current Schedule, including review of the critical path and 6-week look ahead schedule c. Status of project related documentation (Submittals, RFIs, CBs, etc.) d. Quality Observation Log and status of correction of deficient items e. Project questions and issues from meeting attendees
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45	3.3.	CONS A. B.	familia TRUCTIC In gene 1. 2. The Ge 1. 2.	 ar with the PMWS system. ON PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following: a. Safety b. Current Schedule, including review of the critical path and 6-week look ahead schedule c. Status of project related documentation (Submittals, RFIs, CBs, etc.) d. Quality Observation Log and status of correction of deficient items e. Project questions and issues from meeting attendees f. BPW Administration Check
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46	3.3.	CONS A. B.	familia In gene 1. 2. The Ge 1. 2.	 ar with the PMWS system. ON PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following: a. Safety b. Current Schedule, including review of the critical path and 6-week look ahead schedule c. Status of project related documentation (Submittals, RFIs, CBs, etc.) d. Quality Observation Log and status of correction of deficient items e. Project questions and issues from meeting attendees f. BPW Administration Check g. Other as needed
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47	3.3.	CONS A. B.	familia TRUCTIC In gene 1. 2. The Ge 1. 2.	 ar with the PMWS system. ON PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following: a. Safety b. Current Schedule, including review of the critical path and 6-week look ahead schedule c. Status of project related documentation (Submittals, RFIs, CBs, etc.) d. Quality Observation Log and status of correction of deficient items e. Project questions and issues from meeting attendees f. BPW Administration Check g. Other as needed h. Status of CORs and COs to be reviewed outside the standard progress meeting time.
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48	3.3.	CONS A. B.	familia TRUCTIC In gene 1. 2. The Ge 1. 2. 3.	 ar with the PMWS system. DN PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following: a. Safety b. Current Schedule, including review of the critical path and 6-week look ahead schedule c. Status of project related documentation (Submittals, RFIs, CBs, etc.) d. Quality Observation Log and status of correction of deficient items e. Project questions and issues from meeting attendees f. BPW Administration Check g. Other as needed h. Status of CORs and COs to be reviewed outside the standard progress meeting time. Make physical arrangements for meetings.
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49	3.3.	CONS A. B.	familia TRUCTIC In gene 1. 2. The Ge 1. 2. 3. 4.	 ar with the PMWS system. DN PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following: a. Safety b. Current Schedule, including review of the critical path and 6-week look ahead schedule c. Status of project related documentation (Submittals, RFIs, CBs, etc.) d. Quality Observation Log and status of correction of deficient items e. Project questions and issues from meeting attendees f. BPW Administration Check g. Other as needed h. Status of CORs and COs to be reviewed outside the standard progress meeting time. Make physical arrangements for meetings. GCPM to post meeting agendas to the appropriate libraries on the Project Management Web Site
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50	3.3.	CONS A. B.	familia TRUCTIC In gene 1. 2. The Ge 1. 2. 3. 4.	 ar with the PMWS system. DN PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following: a. Safety b. Current Schedule, including review of the critical path and 6-week look ahead schedule c. Status of project related documentation (Submittals, RFIs, CBs, etc.) d. Quality Observation Log and status of correction of deficient items e. Project questions and issues from meeting attendees f. BPW Administration Check g. Other as needed h. Status of CORs and COs to be reviewed outside the standard progress meeting time. Make physical arrangements for meetings. GCPM to post meeting agendas to the appropriate libraries on the Project Management Web Site (PMWS) no less than two (2) working days prior to the scheduled meeting. Notify all required attendees,
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51	3.3.	CONS A. B.	familia TRUCTIC In gene 1. 2. The Ge 1. 2. 3. 4.	 ar with the PMWS system. DN PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following: a. Safety b. Current Schedule, including review of the critical path and 6-week look ahead schedule c. Status of project related documentation (Submittals, RFIs, CBs, etc.) d. Quality Observation Log and status of correction of deficient items e. Project questions and issues from meeting attendees f. BPW Administration Check g. Other as needed h. Status of CORs and COs to be reviewed outside the standard progress meeting time. Make physical arrangements for meetings. GCPM to post meeting agendas to the appropriate libraries on the Project Management Web Site (PMWS) no less than two (2) working days prior to the scheduled meeting. Notify all required attendees, applicable parties to the contract, and others affected of the posted meeting agenda.
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52	3.3.	CONS A. B.	familia TRUCTIC In gene 1. 2. The Ge 1. 2. 3. 4. 5.	 arr with the PMWS system. DN PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following: a. Safety b. Current Schedule, including review of the critical path and 6-week look ahead schedule c. Status of project related documentation (Submittals, RFIs, CBs, etc.) d. Quality Observation Log and status of correction of deficient items e. Project questions and issues from meeting attendees f. BPW Administration Check g. Other as needed h. Status of CORs and COs to be reviewed outside the standard progress meeting time. Make physical arrangements for meetings. GCPM to post meeting agendas to the appropriate libraries on the Project Management Web Site (PMWS) no less than two (2) working days prior to the scheduled meeting. Notify all required attendees, applicable parties to the contract, and others affected of the posted meeting agenda.
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53	3.3.	CONS A. B.	familia TRUCTIC In gene 1. 2. The Ge 1. 2. 3. 4. 5. 6.	 Any with the PMWS system. CON PROGRESS MEETINGS ereral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following: a. Safety b. Current Schedule, including review of the critical path and 6-week look ahead schedule c. Status of project related documentation (Submittals, RFIs, CBs, etc.) d. Quality Observation Log and status of correction of deficient items e. Project questions and issues from meeting attendees f. BPW Administration Check g. Other as needed h. Status of CORs and COs to be reviewed outside the standard progress meeting time. Make physical arrangements for meetings. GCPM to post meeting agendas to the appropriate libraries on the Project Management Web Site (PMWS) no less than two (2) working days prior to the scheduled meeting. Notify all required attendees, applicable parties to the contract, and others affected of the posted meeting agenda. Preside at meetings. Route a meeting attendance roster for attendees to sign-in on.
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54	3.3.	CONS A. B.	familia TRUCTIC In gene 1. 2. The Ge 1. 2. 3. 4. 5. 6. 7.	 ar with the PMWS system. DN PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following: a. Safety b. Current Schedule, including review of the critical path and 6-week look ahead schedule c. Status of project related documentation (Submittals, RFIs, CBs, etc.) d. Quality Observation Log and status of correction of deficient items e. Project questions and issues from meeting attendees f. BPW Administration Check g. Other as needed h. Status of CORs and COs to be reviewed outside the standard progress meeting time. Make physical arrangements for meetings. GCPM to post meeting agendas to the appropriate libraries on the Project Management Web Site (PMWS) no less than two (2) working days prior to the scheduled meeting. Notify all required attendees, applicable parties to the contract, and others affected of the posted meeting agenda. Preside at meetings. Route a meeting attendance roster for attendees to sign-in on. GCPM to record the minutes of the meeting; include significant proceedings and decisions. Post meeting
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54	3.3.	CONS A. B.	familia TRUCTIC In gene 1. 2. The Ge 1. 2. 3. 4. 5. 6. 7.	 arr with the PMWS system. CDN PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following: a. Safety b. Current Schedule, including review of the critical path and 6-week look ahead schedule c. Status of project related documentation (Submittals, RFIs, CBs, etc.) d. Quality Observation Log and status of correction of deficient items e. Project questions and issues from meeting attendees f. BPW Administration Check g. Other as needed h. Status of CORs and COs to be reviewed outside the standard progress meeting time. Make physical arrangements for meetings. GCPM to post meeting agendas to the appropriate libraries on the Project Management Web Site (PMWS) no less than two (2) working days prior to the scheduled meeting. Notify all required attendees, applicable parties to the contract, and others affected of the posted meeting agenda. Preside at meetings. ROPM to record the minutes of the meeting; include significant proceedings and decisions. Post meeting minutes to the PMWS no more than two (2) working days after the completed meeting. Meeting
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55 56	3.3.	CONS A. B.	familia TRUCTIC In gene 1. 2. The Ge 1. 2. 3. 4. 5. 6. 7.	 ar with the PMWS system. ON PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following: a. Safety b. Current Schedule, including review of the critical path and 6-week look ahead schedule c. Status of project related documentation (Submittals, RFIs, CBs, etc.) d. Quality Observation Log and status of correction of deficient items e. Project questions and issues from meeting attendees f. BPW Administration Check g. Other as needed h. Status of CORs and COs to be reviewed outside the standard progress meeting time. Make physical arrangements for meetings. GCPM to post meeting agendas to the appropriate libraries on the Project Management Web Site (PMWS) no less than two (2) working days prior to the scheduled meeting. Notify all required attendees, applicable parties to the contract, and others affected of the posted meeting agenda. Preside at meetings. Route a meeting attendance roster for attendees to sign-in on. GCPM to record the minutes of the meeting; include significant proceedings and decisions. Post meeting minutes to the PMWS no more than two (2) working days after the completed meeting. Meeting minutes shall include a scanned copy of the attendance sign-in sheet. Notify all required meeting
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55 56 57	3.3.	CONS A. B.	familia TRUCTIC In gene 1. 2. The Ge 1. 2. 3. 4. 5. 6. 7.	 ar with the PMWS system. DN PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following: a. Safety b. Current Schedule, including review of the critical path and 6-week look ahead schedule c. Status of project related documentation (Submittals, RFIs, CBs, etc.) d. Quality Observation Log and status of correction of deficient items e. Project questions and issues from meeting attendees f. BPW Administration Check g. Other as needed h. Status of CORs and COs to be reviewed outside the standard progress meeting time. Make physical arrangements for meetings. GCPM to post meeting agendas to the appropriate libraries on the Project Management Web Site (PMWS) no less than two (2) working days prior to the scheduled meeting. Notify all required attendees, applicable parties to the contract, and others affected of the posted meeting agenda. Preside at meetings. Route a meeting attendance roster for attendees to sign-in on. GCPM to record the minutes of the meeting; include significant proceedings and decisions. Post meeting minutes to the PMWS no more than two (2) working days after the completed meeting meeting minutes shall include a scanned copy of the attendance sign-in sheet Notify all required meeting attendees, applicable parties to the contract, and others affected by decisions made at the meeting
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55 56 57 58	3.3.	CONS A. B.	familia TRUCTIC In gene 1. 2. The Ge 1. 2. 3. 4. 5. 6. 7. 8.	 ar with the PMWS system. DN PROGRESS MEETINGS eral, all of the following shall apply: Representatives of Contractors, Subcontractors, and suppliers attending meetings shall be qualified and authorized to act on behalf of the entity each represents. The attendance shall be from the required attendance list in 3.1.D. above. eneral Contractor Project Manager (GCPM) shall: Schedule and conduct all construction progress meetings biweekly or more frequently as required. Prepare agenda for meetings including, but not limited to the following: a. Safety b. Current Schedule, including review of the critical path and 6-week look ahead schedule c. Status of project related documentation (Submittals, RFIs, CBs, etc.) d. Quality Observation Log and status of correction of deficient items e. Project questions and issues from meeting attendees f. BPW Administration Check g. Other as needed h. Status of CORs and COs to be reviewed outside the standard progress meeting time. Make physical arrangements for meetings. GCPM to post meeting agendas to the appropriate libraries on the Project Management Web Site (PMWS) no less than two (2) working days prior to the scheduled meeting. Notify all required attendees, applicable parties to the contract, and others affected of the posted meeting agenda. Preside at meetings. Route a meeting attendance roster for attendees to sign-in on. GCPM to record the minutes of the meeting; include significant proceedings and decisions. Post meeting minutes to the PMWS no more than two (2) working days after the completed meeting. Meeting minutes shall include a scanned copy of the attendance sign-in sheet. Notify all required meeting attendees, applicable parties to the contract, and others affected by decisions made at the meeting. The above requirements do not apply to GC/sub-contractor meetings.

1			
2	3.4.	PRE-II	NSTALLATION MEETINGS
3		Α.	The GCPM shall schedule and conduct all pre-installation meetings, including mockup reviews, before each
4			construction activity that requires coordination with other trades.
5		В.	The GCPM shall be responsible for the final agenda and meeting minutes.
6		C.	The GCPM will work with all concerned parties to resolve issues as needed and submit RFI's if necessary.
7		D.	Required attendance shall be from the list in 3.1.D. above and shall be personnel having a stake in the outcome
8			of the installation or knowledge of the system being installed.
9		Ε.	In the event the Contractor installs equipment or materials without a pre-installation meeting the Contractor
10			shall be solely responsible for removing, replacing, repositioning materials and equipment as instructed by the
11			Project Architect or City Project Manager at no additional cost to the City.
12			
13	3.6	PRE-C	ONTRACT CLOSEOUT MEETINGS
14		Α.	Two (2) Pre-contract Closeout Meetings shall be held to review the closeout procedures, requirements, and
15			contract deliverables.
16			1. Pre-contract Closeout Meeting #1 shall be scheduled prior to the 50% Progress Payment Request is being
17			requested. This meeting shall discuss items such as closing out QMO reports, providing O&M drafts and
18			finals, payroll and Affirmative Action documentation, and other contract deliverables.
19			2. Pre-contract Closeout Meeting #2 shall be scheduled prior to the 80% Progress Payment Request is being
20			requested. This meeting shall discuss, but not be limited to, the status of scheduling final regulatory
21			inspections, cleaning up outstanding QMO's, demonstration and training, attic stock; and finalization
22			review of payroll and other related documents.
23		В.	The GCPM shall schedule, coordinate, and make physical arrangements for both meetings.
24		C.	All of the following shall be required to attend both meetings:
25			1. The GCPM and the GC Field superintendent
26			2. All Subcontractor Project Managers regardless of the current status of their work.
27			a. The GCPM may excuse a Subcontractor PM if they are confident that all contractual requirements
28			for closeout by the subcontractor have been completed and/or delivered to the GCPIVI. The list of
29			attendees shall be reviewed and agreed upon with CPM anead of the meeting.
30			b. At the option of these project managers the field supervisors may also attend. The Desiret Auchitect and at least one desire consultant from each dissipling represented by the plane.
31			3. The Project Architect and a least one design consultant from each discipline represented by the plans
32 22			The Owner
27			4. The Owner
25 25			5. The Crivi
36			7 The Commissioning Agent
37		D	The CPM shall nublish an agenda and chair the meeting
38		υ.	
39	3.7	OTHE	R SPECIAL MEETINGS
40	0.7	A.	The Contractor shall schedule special meetings per the requirements of the LEED Specification, the Project
41			Quality Management Plan, the Commissioning Plan and as indicated by other specifications.
42		В.	Special meetings include but are not limited to the following:
43			1. Waste Management Conference
44			2. Equipment start up meetings
45			3. Testing and balancing meetings
46			4. Commissioning meetings
47			5. Other meetings as necessitated by the contract documents
48			- · ·
49			END OF SECTION

1					SECTION 01 31 23
2	DADT	1_6			PROJECT MANAGEMENT WEB SITE
5 /	PARI	1 – G 1 1	GENERAL .		۰۰۰۰۰۱ ۱
5		1 2			TION CLOUD PROCEDURE OVERVIEW
6		1.3.	RELATE		INS
7	PART	2 - PF	RODUCTS	S	2
8		2.1.	AUTOD	ESK CONSTRUC	TION CLOUD SYSTEM RELATED PRODUCTS
9	PART	3 - E>	KECUTION	۷	
10	:	3.1.	POST B	ID-OPENING	
11	:	3.2.	POST P	RE-CONSTRUCT	ION MEETING2
12					
13	PART	<u>1 – G</u>	SENERAL		
14 15	1.1.	GEI	NFRAL DI	ESCRIPTION	
16	1.1.	A.	The (City of Madison	(CoM) has established a cloud-based Project Management Tool (PMT) using an Autodesk
17			prod	uct called Auto	desk Construction Cloud (ACC).
18		В.	The s	software is used	I throughout the design, construction and warranty process of major remodels and new
19			cons	truction project	·S.
20		C.	Initia	Ily deployed in	mid-2023, the PMT software will be deployed on all projects. The PMT software is cloud-
21			base	d software and	d therefore will receive regular updates and enhancements.
22					
23	1.2.	AU	TODESK	CONSTRUCTION	N CLOUD PROCEDURE OVERVIEW
24		Α.	The (CoM PMT is 3 m	nain modules. The <u>Autodesk Docs (https://help.autodesk.com/view/DOCS/ENU/)</u> module is a
25			docu	ment managen	ent file system that is the foundation of ACC. The <u>Build</u>
26			https	s:/help.autodes	k.com/view/BUILD/ENU/ module has many sections that assist in performing day to day
27			funct	tions of design/	construction management while reducing the use of different software platforms, surface
28			mail,	email and ema	il attachments. Finally, the <u>Cost management</u>
29			(http	<u>s://help.autode</u>	<u>ssk.com/view/BUILD/ENU/?guid=Cost_Overview</u>) module is used to manage project finances.
30			1.	Files within A	Autodesk Docs can store a wide variety <u>file formats</u>
31				(https://neip	Adutodesk.com/view/DUCS/ENU/?guid=Supported_Files_Docs/including but not limited to
32			2	word, Excel,	PDF, photographs (all popular formats), etc.
33			2.	File Falder of	ction within the Build module is used for Punch Lists, Quality Control and Warranty Issues.
34 25		Б	3. ^ ++	File Folder an	To module section access are controlled by Permission Groups and Permission Level
35 26		в.	Alui	ract Additiona	I training will be provided as peopled for the GC and Sub Contractor (GC) will be GoM
30 27		c		DMT has prodof	in a mining will be provided as needed for the GC and Sub-Contractors (SC) by the Colvi.
20		C.	comr	nlatad Thasa y	vorkflows are designed for inhound information from the contractor as well as outhound
30			infor	mation from th	e Architectural/Engineer consultant and the Owner
40		D	The (GC will be requi	red to receive email notifications, access the internet to review related documentation and
41		υ.	he al	ble to unload/d	ownload documentation to the various project modules or folders
42		E.	The S	SC's will be reau	Jired (at a minimum) to receive email notifications and access the internet to review related
43			docu	mentation. Pri	or to setting up the final PMT the GC and CPM shall meet to review all ACC workflows, the
44			GC w	vill determine to	what level over the minimum requirements the SC's will be involved.
45		F.	At fir	nal project close	Pout with the GC, the CoM will provide the Project Architect/Project Engineer (A/E PROJ
46			MGR	t) and the GC, a	n exported version of the complete project in ACC.
47					
48	1.3.	REL	ATED SP	ECIFICATIONS	
49		Α.	The f	following specif	ication sections are directly related to the CoM PMT system.
50			1.	01 25 13	Product Substitution Procedures
51			2.	01 26 13	Request for Information (RFI)
52			3.	01 26 46	Construction Bulletins (CB)
53			4.	01 26 57	Change Order Request (COR)
54			5.	01 26 63	Change Order (CO)
55			6.	01 29 76	Progress Payment Procedures
56			7.	01 31 19	Project Meetings
57			8.	01 32 16	Construction Progress Schedules

1			9. 01 32 26 Construction Progress Reporting
2			10. 01 32 33 Photographic Documentation
3			11. 01 33 23 Submittals
4			12. 01 45 16 Field Quality Control Procedures (Owner)
6 7	<u>PART</u>	<u>2 - PRO</u>	<u>IDUCTS</u>
8	2.1.	AUTO	DESK CONSTRUCTION CLOUD SYSTEM RELATED PRODUCTS
9		Α.	Autodesk Construction Cloud is an Autodesk based software that requires no additional software installation,
10			hardware or other special requirements/applications for the users. There are no costs associated with the use of
11			this system.
12		В.	Please consult Autodesk's web site for the latest system requirements
13			<pre>(https://help.autodesk.com/view/BUILD/ENU/?guid=System_Requirements_ACC)</pre>
14 15	PART	3 - EXE	CUTION
16 17	3 1	POST	RID-OPENING
18	5.1.	A.	After bids have been opened, a successful bidder has been determined, and bid acceptance procedures have
19		<i>,</i>	been initiated the City Project Manager (CPM) will contact the GC to provide the following information.
20			1. Autodesk Construction Cloud Help (https://help.autodesk.com/view/BUILD/ENU/) and Learning Center
21			(https://learnacc.autodesk.com/) are kept up to date with latest ACC features.
22			2. For more customized workflows, Project Management Software Tutorials have been developed. These
23			tutorials are in a PDF printable format with screen shots and associated instructions on how to access and
24			use the PMT.
25			3. A blank Project Directory in an Excel spread sheet format. The contractor shall provide the following
26			information for GC and SC staffs as indicated on the spreadsheet. This will generally be the Project
27			Manager for the GC as well as the Sub-contractors and the GC Site Supervisor.
28			a. Last Name, First Name
29			D. Company Name
30 31			 Lindi duiless (value, work related) A Phone Contact number and professional name must be entered by each user themselves via
32			4. Inforce contract number and processional name must be entered by each user themselves via https://profile.autodesk.com/
33			5. The GC shall provide the above information for all SC's where the GC is not self-performing the work.
34			6. The GC may provide project foreperson information for work being self-performed if he/she so desires.
35			
36	3.2.	POST	PRE-CONSTRUCTION MEETING
37		А.	The GCPM will return the completed Project Directory spread sheet to the CPM no later than the Pre-
38		_	construction meeting.
39		В.	The City Project Admin is responsible for uploading all project directory data into ACC, adding users to project
40 41		c	and licenses to users for all non-city staff (GC/SC staffs).
41 42		C.	arcount if they do not already have one. It is the responsibility of each GC/SC to follow the instructions to setup
43			their own account
44		D.	Once the GCPM has received his/her project invitation, uploading of contract related documents can begin. This
45			would include but not be limited to project schedules, submittals, RFI's, and other documents as needed.
46		E.	All workflows, review of documentation, and general archiving of construction related documentation will be
47			conducted on the PMWS. These documents will generally not be emailed.
48		F.	The following documents related to the execution of the contract will not be part of the PMT:
49			 All documentation related to executing the contract, such as:
50			a. Sub Contractors list
51			b. Affirmative Action documentation
52			c. Bonding documentation
53			a. Documentation associated with placing out the contract
54 55			 Final documentation required/generated by ordinance, code or statute, such as:
56			 Any documentation required/generated by ordinance, code or statute, such as, Frosion Control inspections
57			b. Building Inspection Department inspections
58			

END OF SECTION

			SECTION 01 32 16 CONSTRUCTION PROGRESS SCHEDULES
PART 1	1 – GE	NERAL	
1	.1.	SCOPE	
1	.2.	RELATED SPECIFICATIO	NS
PART 2	2 – PR	ODUCTS – THIS SECTIO	N NOT USED
PART 3	3 - EX	ECUTION	
3	.1.	OVERALL PROJECT SCH	EDULE (OPS)
3	.2.	6 WEEK LOOK-OUT SCH	IEDULES (LOS)
3	.3.	PROJECT MANAGEMEN	IT WEB SITE (PMWS)
PART :	1 – GI	ENERAL	
1.1.	sco	PE	
	A.	This specification is t	o identify various project related schedules associated with indicating construction progress
		and outlook. The fol	llowing schedules are the responsibility of the General Contractor (GC).
		1. Overall Proje	ct Schedule
		2. 6 Week Look	-out Schedule
	В.	This specification is r	not intended to include internal schedules generated by the contractors during their
		planning and execut	ion of the contract.
1.2.	REL/	ATED SPECIFICATIONS	
	A.	Section 01 29 76	Progress Payment Procedures
	В.	Section 01 31 23	Project Management Web Site
	C.	Section 01 31 19	Progress Meetings
	D.	Section 01 74 13	Progress Cleaning
	E.	Section 01 77 00	Closeout Procedures
	F.	Section 01 78 23	Operation and Maintenance Data
	G.	Section 01 78 36	Warranties
	Н.	Section 01 78 39	As-Built Drawings
	I.	Section 01 78 43	Spare Parts and Extra Materials
	J.	Section 01 79 00	Demonstration and Training
	К.	Section 01 91 00	Commissioning
	L.	Other specification v Owner, Project Arch	vithin the construction documents that may indicate the need for scheduling any event with itect, Owner Representatives, including any owner provided equipment.
	2 _ DE		
	<u> </u>		
PART :	3 - EX	ECUTION	
3.1.	OVE	RALL PROJECT SCHEDU	LE (OPS)
	Α.	The GC shall prepare	an OPS that covers the duration of the contract from the pre-construction meeting through
		the end of construct	ion to final contract closeout.
		1. The GC shall	review Specification 01 77 00 Closeout Procedures to become familiar with definitions,
		differences, a	and requirements for closing out the construction and contract including the association with
	_	progress pay	ments.
	В.	The GC shall provide	copies and lead a discussion on the OPS during the pre-construction meeting.
	с. Б	The OPS shall indicat	te start and end dates of each task associated with the project.
	υ.	The GC shall undate	multate the childer path of the project.
	E		the OPS as often as necessary during the duration of the project. Opdates will be briefed as
	Ε.	noodod during bi wa	
	E.	needed during bi-we	ekiy progress meetings.
3.2.	E. 6 W	needed during bi-we	JLES (LOS)
3.2.	Е. 6 W А.	needed during bi-we EEK LOOK-OUT SCHEDU The GC shall prepare	JLES (LOS) e the initial LOS to include detail of daily tasks for the first six (6) weeks of construction in
3.2.	Е. 6 W А.	needed during bi-we EEK LOOK-OUT SCHEDL The GC shall prepare depth for the Pre-co	JLES (LOS) • the initial LOS to include detail of daily tasks for the first six (6) weeks of construction in nstruction meeting. The LOS shall be compatible and complimentary to the OPS.

1		C.	The LOS shall indicate start and end dates of each major task, associated related sub-tasks, and required parallel
2			or pre-requisite tasks required to complete the major task on time.
3		D.	The LOS shall also include identifying and scheduling such events as:
4			1. Pre-installation meetings and mock-up review meetings.
5			2. Quality management reviews of installations before they are covered.
6			3. Owner provided equipment as designated by the contract documents.
7			4. Work by others as designated by the contract documents.
8			5. Critical submittal dates.
9		Ε.	The GC shall update the LOS prior to each bi-weekly progress meeting to indicate the next 6 weeks of scheduled
10			work. Updates will be briefed during each bi-weekly progress meeting.
11			
12	3.3.	PROJI	ECT MANAGEMENT WEB SITE (PMWS)
13		Α.	The GC shall upload all project schedules and updates to the PMWS in an original PDF version of the scheduling
14			document. Scans will not be permitted.
15			
16			
17			END OF SECTION
18			

		SECTION 01 32 19 SUBMITTALS SCHEDULE
PAF	RII-G	
	1.1.	
	1.2.	RELATED DOCUMENTS
	1.3.	KELATED DUCUMENTS
	1.4.	
	1.5.	
	.0. ת בדת	
PAP	КI Z — Р рта гу	RODUCTS - THIS SECTION NOT USED
PAP	XI3-E/ 2.1	
	3.1. 2.2	
	3.Z. 2.2	
	5.5.	STAFF REVIEW RESPONSIBILITIES
PA	RT 1 – G	ENERAL
1.1	. SU	MMARY
	Α.	The General Contractor shall submit a complete and comprehensive list of all submittals anticipated during the
		execution of this contract.
	В.	The GC shall include the Administrative submittals identified in item 1.5 below and shall be required to up load
		them to the Project Management Web Site.
	C.	The initial Submittals Schedule shall be based on the original contract documents used at the time of bidding and
		any posted addenda through awarding of the contract.
	D.	The Submittal Schedule may be appended during the execution of the contract based on amendments to the
		contract in the form of Change Orders, Construction Bulletins, and other related documents that add, or change
		the scope of the work.
1.2	. REI	ATED SPECIFICATIONS
	Α.	Section 01 29 76 Progress Payment Procedures
	В.	Section 01 31 23 Project Management Web Site (PMWS)
	С.	Section 01 33 23 Submittals
	D.	Section 01 91 00 Commissioning
12	DEI	
1.5	. ΝΕΙ Δ	The following documents shall be used as the basis for initiating the original Submittals Schedule
	А.	1 Drawing documents and specifications (including general provisions) as provided with the hid set
		documents and any published addenda
	в	The following documents shall be used to amend the submittals schedule as needed during the execution of this
	D.	contract
		1 Documents associated with revisions or clarifications to number A 1 above after awarding of the
		contract including but not limited to:
		a Construction Bulletins
		h Annroved Change Orders
14	SUI	
1.4	. 301 A	Administrative Submittal: Any submittal that may be required by a Division 1 Specification and as noted in
	Π.	Section 1.5 below
	R	Critical Path Submittal: Any early submittal that needs a priority review due to early construction use or long
	υ.	lead times where a delay could affect the critical nath of the construction schedule
	ſ	Submittal: Any material product equipment or general requirement as outlined in this and other specifications
	с.	that require a favorable review or accentance prior to proceeding with procuring the item or proceeding with
		the Work.

1	1.5.	SUBM	IITTAL REQUIREMENTS
2		Α.	The GC and all Sub-contractors shall review the construction documents including the specifications of their
3			individual Division or Trade to compile a complete list of all materials, products, or equipment that will require a
4			positively reviewed submittal to be completed prior to procurement and installation.
5			1. Submittals shall include but not be limited to any of the following that may apply:
6			a. Shop Drawings
7			b. Product Data
8			c. Assembly Drawings
9			d. Engineered Drawings
10			e. Product Samples
11		В.	The following items will require an approved submittal, verify with specifications for specific needs and
12			requirements:
13			1. Contractor certifications for specialized work such as asbestos removal, well drilling, controls, AV, etc.
14			
15	1.6.		NISTRATIVE SUBMITTALS
16		А.	The GC shall upload the following submittals within 15 working days of receipt of the City of Madison Start Work
1/			Letter. All Administrative Submittals shall be approved prior to requesting Progress Payment Number 1.
18			1. Contractors Project Directory, see specification 01 31 23, discuss requirements with CPM
19			2. Schedule of Values, see Specification 01 29 73
20			3. Submittals Schedule, see Specification 01 32 19
21			4. Waste Management Plan, see Specification 01 /4 19
22			5. Closeout Requirement Checklist, see Specification 01 77 00
23			6. Warranty Checklist, see Specification 01 78 36
24			
25 26	PARIA	<u> – PRU</u>	DUCTS - THIS SECTION NOT USED
20	PART :	3 - FXF(
27	<u>. ANT</u>		
28			
28 29	3.1.	OVER	ALL RESPONSIBILITIES OF ALL CONTRACTORS
28 29 30	3.1.	OVER/ A.	ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work
28 29 30 31	3.1.	OVERA A.	ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor.
28 29 30 31 32	3.1.	OVERA A. B.	ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the
28 29 30 31 32 33	3.1.	OVERA A. B.	ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided
28 29 30 31 32 33 34	3.1.	OVER/ A. B.	ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved.
28 29 30 31 32 33 34 35	3.1.	OVERA A. B. C.	ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as
28 29 30 31 32 33 34 35 36	3.1.	OVERA A. B. C.	ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as follows:
28 29 30 31 32 33 34 35 36 37	3.1.	OVER/ A. B. C.	 ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as follows: 1. For items on the Critical Path as identified by the GC, five (5) working days
28 29 30 31 32 33 34 35 36 37 38	3.1.	OVER/ A. B. C.	 ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as follows: For items on the Critical Path as identified by the GC, five (5) working days For most other submittals ten (10) working days
28 29 30 31 32 33 34 35 36 37 38 39	3.1.	OVER/ A. B. C.	 ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as follows: For items on the Critical Path as identified by the GC, five (5) working days Additional time may be needed for complex submittals or if re-submittals are required.
28 29 30 31 32 33 34 35 36 37 38 39 40	3.1.	OVERA A. B. C.	 ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as follows: For items on the Critical Path as identified by the GC, five (5) working days For most other submittals ten (10) working days Additional time may be needed for complex submittals or if re-submittals are required.
28 29 30 31 32 33 34 35 36 37 38 39 40 41	3.1.	OVERA A. B. C. D.	 ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as follows: For items on the Critical Path as identified by the GC, five (5) working days For most other submittals ten (10) working days Additional time may be needed for complex submittals or if re-submittals are required. The City will provide a spreadsheet to provide the format of the Submittal Schedule as part of the first administrative submittals.
28 29 30 31 32 33 34 35 36 37 38 39 40 41 42	3.1.	OVERA A. B. C. D.	 ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as follows: For items on the Critical Path as identified by the GC, five (5) working days For most other submittals ten (10) working days Additional time may be needed for complex submittals or if re-submittals are required. The City will provide a spreadsheet to provide the format of the Submittal Schedule as part of the first administrative submittals.
28 29 30 31 32 33 34 35 36 37 38 39 40 41 42 43	3.1. 3.2.	A. B. C. D. GENEL	 ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as follows: For items on the Critical Path as identified by the GC, five (5) working days For most other submittals ten (10) working days Additional time may be needed for complex submittals or if re-submittals are required. The City will provide a spreadsheet to provide the format of the Submittal Schedule as part of the first administrative submittals. RAL CONTRACTORS RESPONSIBILITIES The General Contractor shall be responsible for all of the following:
28 29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44	3.1.	OVERA A. B. C. D. GENEI A.	 ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as follows: For items on the Critical Path as identified by the GC, five (5) working days For most other submittals ten (10) working days Additional time may be needed for complex submittals or if re-submittals are required. The City will provide a spreadsheet to provide the format of the Submittal Schedule as part of the first administrative submittals. RAL CONTRACTORS RESPONSIBILITIES The General Contractor shall be responsible for all of the following: Consolidating all submittal lists from individual contractors into one master list with the provided
28 29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45	3.1.	OVER. A. B. C. D. GENEI A.	 ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as follows: For items on the Critical Path as identified by the GC, five (5) working days For most other submittals ten (10) working days Additional time may be needed for complex submittals or if re-submittals are required. The City will provide a spreadsheet to provide the format of the Submittal Schedule as part of the first administrative submittals. RAL CONTRACTORS RESPONSIBILITIES The General Contractor shall be responsible for all of the following: Consolidating all submittal lists from individual contractors into one master list with the provided spreadsheet on the Project Management Web Site
28 29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46	3.1.	OVERA A. B. C. D. GENEI A.	 ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as follows: For items on the Critical Path as identified by the GC, five (5) working days For most other submittals ten (10) working days Additional time may be needed for complex submittals or if re-submittals are required. The City will provide a spreadsheet to provide the format of the Submittal Schedule as part of the first administrative submittals. RAL CONTRACTORS RESPONSIBILITIES The General Contractor shall be responsible for all of the following: Consolidating all submittal lists from individual contractors into one master list with the provided spreadsheet on the Project Management Web Site Reviewing all submitted lists for completeness, timing with the overall contract, etc. The GC shall meet
28 29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47	3.1.	OVER. A. B. C. D. GENEL A.	 ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as follows: For items on the Critical Path as identified by the GC, five (5) working days For most other submittals ten (10) working days Additional time may be needed for complex submittals or if re-submittals are required. The City will provide a spreadsheet to provide the format of the Submittal Schedule as part of the first administrative submittals. RAL CONTRACTORS RESPONSIBILITIES The General Contractor shall be responsible for all of the following: Consolidating all submittal lists from individual contractors into one master list with the provided spreadsheet on the Project Management Web Site Reviewing all submitted lists for completeness, timing with the overall contract, etc. The GC shall meet with individual contractors to make changes as necessary.
28 29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48	3.1.	OVERA A. B. C. D. GENEL A.	 ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as follows: For items on the Critical Path as identified by the GC, five (5) working days For most other submittals ten (10) working days Additional time may be needed for complex submittals or if re-submittals are required. The City will provide a spreadsheet to provide the format of the Submittal Schedule as part of the first administrative submittals. RAL CONTRACTORS RESPONSIBILITIES Consolidating all submittal lists from individual contractors into one master list with the provided spreadsheet on the Project Management Web Site Reviewing all submitted lists for completeness, timing with the overall contract, etc. The GC shall meet with individual contractors to make changes as necessary. Upload the completed Submittals Schedule to the Submittal Library on the Project Management Web Site
28 29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49	3.1.	OVERA A. B. C. D. GENEL A.	 ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as follows: For items on the Critical Path as identified by the GC, five (5) working days For most other submittals ten (10) working days Additional time may be needed for complex submittals or if re-submittals are required. The City will provide a spreadsheet to provide the format of the Submittal Schedule as part of the first administrative submittals. RAL CONTRACTORS RESPONSIBILITIES The General Contractor shall be responsible for all of the following: Consolidating all submittal lists from individual contractors into one master list with the provided spreadsheet on the Project Management Web Site Reviewing all submitted lists for completeness, timing with the overall contract, etc. The GC shall meet with individual contractors to make changes as necessary. Upload the completed Submittals Schedule to the Submittal Library on the Project Management Web Site See Specification 01 33 23 Submittals for more information on this procedure.
28 29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50	3.1.	OVERA A. B. C. D. GENEL A.	 ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as follows: For items on the Critical Path as identified by the GC, five (5) working days For most other submittals ten (10) working days Additional time may be needed for complex submittals or if re-submittals are required. The City will provide a spreadsheet to provide the format of the Submittal Schedule as part of the first administrative submittals. RAL CONTRACTORS RESPONSIBILITES The General Contractor shall be responsible for all of the following: Consolidating all submittal lists from individual contractors into one master list with the provided spreadsheet on the Project Management Web Site Reviewing all submittal lists for completeness, timing with the overall contract, etc. The GC shall meet with individual contractors to make changes as necessary. Upload the completed Submittals Schedule to the Submittal Library on the Project Management Web Site See Specification 01 33 23 Submittals for more information on this procedure. Resubmit the schedule as needed after initial reviews have been completed.
28 29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51	3.1.	A. B. C. D. GENEL A.	 ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as follows: For items on the Critical Path as identified by the GC, five (5) working days For most other submittals ten (10) working days Additional time may be needed for complex submittals or if re-submittals are required. The City will provide a spreadsheet to provide the format of the Submittal Schedule as part of the first administrative submittals. RAL CONTRACTORS RESPONSIBILITIES The General Contractor shall be responsible for all of the following: Consolidating all submittal lists from individual contractors into one master list with the provided spreadsheet on the Project Management Web Site Reviewing all submitted lists for completeness, timing with the overall contract, etc. The GC shall meet with individual contractors to make changes as necessary. Upload the completed Submittals Schedule to the Submittal Library on the Project Management Web Site See Specification 01 33 23 Submittals for more information on this procedure. Resubmit the schedule as needed after initial reviews have been completed.
28 29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52	3.1.	OVERA A. B. C. D. GENEL A. B.	 ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as follows: For items on the Critical Path as identified by the GC, five (5) working days For most other submittals ten (10) working days For most other submittals to provide the format of the Submittal Schedule as part of the first administrative submittals. RAL CONTRACTORS RESPONSIBILITIES The General Contractor shall be responsible for all of the following: Consolidating all submittal lists from individual contractors into one master list with the provided spreadsheet on the Project Management Web Site Reviewing all submittal Schedule to the Submittal Library on the Project Management Web Site See Specification 01 33 23 Submittals for more information on this procedure. Resubmit the schedule as needed after initial reviews have been completed.
28 29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 53	3.1.	A. B. C. D. GENEI A. B. C.	 ALL RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as follows: For items on the Critical Path as identified by the GC, five (5) working days For most other submittals ten (10) working days Additional time may be needed for complex submittals or if re-submittals are required. The City will provide a spreadsheet to provide the format of the Submittal Schedule as part of the first administrative submittals. RAL CONTRACTORS RESPONSIBILITES The General Contractor shall be responsible for all of the following: Consolidating all submittel lists from individual contractors into one master list with the provided spreadsheet on the Project Management Web Site Reviewing all submittel lists for completeness, timing with the overall contract, etc. The GC shall meet with individual contractors to make changes as necessary. Upload the completed Submittals for more information on this procedure. Resubmit the schedule as needed after initial reviews have been completed. The GC shall work with other contractors to amend the Submittal Schedule throughout the execution of the project based on changes and modifications as needed. The GC and Project Architect shall be responsible for reviewing and briefing the submittal schedule and execution of the project based on changes and modifications as needed.
28 29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 57	3.1.	A. B. C. D. GENEI A. B. C.	 All RESPONSIBILITIES OF ALL CONTRACTORS All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of submittals to the General Contractor. Each list shall indicate the title of the submittal, the associated specification of the submittal, whether the submittal can be considered an early/middle/late submittal, the anticipated date the submittal will be provided and the anticipated date the submittal needs to be approved. Contractors shall be aware that the goals for submittal review by the Architect staff and City staff will be as follows: For items on the Critical Path as identified by the GC, five (5) working days For most other submittals ten (10) working days Additional time may be needed for complex submittals or if re-submittals are required. The City will provide a spreadsheet to provide the format of the Submittal Schedule as part of the first administrative submittals. RAL CONTRACTORS RESPONSIBILITIES The General Contractor shall be responsible for all of the following: Consolidating all submittal lists from individual contractors into one master list with the provided spreadsheet on the Project Management Web Site Reviewing all submitted lists for completeness, timing with the overall contract, etc. The GC shall meet with individual contractors to make changes as necessary. Upload the completed Submittals Schedule to the Submittal Library on the Project Management Web Site See Specification 01 33 23 Submittals for more information on this procedure. Resubmit the schedule as needed after initial reviews have been completed. The GC shall work with other contractors to amend the Submittal Schedule throughout the execution of the project based on changes and modifications as needed. The GC shall work with other contractors to amend the Submittal

1	3.3.	STAF	F REVIE	W RESPONSIBILITIES
2		Α.	The F	Project Architect, consulting staff, Commissioning Agent (CxA), Owner, and city staff will review the
3			Subm	nittal Schedule for completeness per the plans and specifications within their divisions of work. The
4			revie	wing staff may provide comments as needed. Some examples might include the following:
5			1.	Submittal not required
6			2.	Provide photos of samples with digital submittal
7			3.	Insure one submittal for complete system
8			4.	Append the schedule to include
9			5.	See Specification section for additional requirements
10		В.	The F	Project Architect and City Project Manager will finalize review comments regarding the Submittal Schedule.
11			Re-su	ubmittal of the submittal schedule may be required.
12				
13				
14				
15				END OF SECTION
16				

		SURVEY AND LAYOUT DATA
DART	1 – G	
	11	SLIMMARY
	1.2.	RELATED SPECIFICATIONS
	1.3.	SURVEYOR QUALIFICATIONS
	1.4.	QUALITY ASSURANCE
	1.5.	SUBMITTALS
	1.6.	EXAMINATION
PART	2 – PF	ODUCTS – NOT USED
PART	3 - EX	ECUTION
	3.1.	PRE-CONSTRUCTION OWNER SUPPORT
	3.2.	UTILITY LOCATING
	3.3.	SURVEY CONTROL AND LAYOUT DATA
	3.4.	TOPOGRAPHIC SURVEYING
	3.5.	SITE SURVEY AS-BUILT
DART	1_6	ENERAL
	1-0	
1.1.	SUN	IMARY
	Α.	The purpose of this specification is to set forth the minimal required guidelines to be followed by the General
		Contractor (GC) and the Land Surveyor (Surveyor) including but not limited to the following:
		1. Surveyor Professional Requirements
		2. Horizontal and Vertical Datum Control
		3. Local Control (if any)
		4. Electronic File and Data Requirements
		5. As-Built Documentation Requirements
	В.	When working on any City of Madison project, OSHA standards must be complied with. The Surveyor shall
	~	provide appropriate traffic control in accordance to the Manual on Uniform Traffic Control Devices (MUTCD).
	C.	The Surveyor shall be responsible for notifying Diggers Hotline in advance of beginning the field work for this
		contract.
1 2	DEI	
1.2.		Section 01 29 76 Progress Payment Procedures
	В.	Section 01 31 23 Project Management Web Site (PMWS)
	С.	Section 01 33 23 Submittals
	D.	Section 01 78 39 As-Built Drawings
	E.	Section 105.9, Survey Points and Instructions, of the City of Madison FACILITIES MANAGEMENT SPECIFICATION
		for Public Works
1.3.	SUR	VEYOR QUALIFICATIONS
	Α.	The General Contractors, Land Surveyor Sub-Contractor shall meet or exceed the following:
		1. The Principal Land Surveyor (PLS) shall be licensed to practice in the State of Wisconsin.
		a. The PLS's license shall be current at the beginning of the contract and the PLS shall maintain an
		active license throughout the execution of this contract.
		2. The PLS shall have a minimum of minimum of ten (10) years of field experience on similar projects of
		scope and size.
		a. Land Surveyors working under the direction of the PLS shall have a minimum of five (5) years of field averaging and similar averaging a statistical states.
	D	experience on similar projects of scope and size. The DLS shall be responsible for checking and varifying all work being performed under the DLS's direction duri
	Б.	the execution of this contract. This shall include but not be limited to periodic field checks of equipment and
		survey data for accuracy and compliance with the contract documents
		salley wate for accuracy and compliance with the contract documents.
	QU	ALITY ASSURANCE
1.4.		The PLS shall do all surveying in City of Madicon Datum's as follows:
1.4.	Α.	The FLS shall do all surveying in city of Madison Datum s as follows.
1.4.	Α.	 All Horizontal Control shall be in the Dane County Coordinates (WISCRS), NAD 83(1997) datum, U

1			2. All Vertical Control shall be in NAVD88(1991).
2			3. Information on PLSS Section Corner Monuments and Tie Sheets can be found on the City Engineering
3			Mapping website http://gis.cityofmadison.com/Madison_PLSS/PLSS_TieSheets.html .
4			
5	1.5.	SUBM	ITTALS
6 7		A.	After initial project setup the PLS shall provide the following information as a Survey Data Submittal for review by the CPM/CCM, and Owner. See Specification 01 33 23 – Submittals for more information.
8			1. Copy of the PLS (and any supporting staff) current State of Wisconsin registration certificate/licenses.
9 10			 Digital survey submittal shall be uploaded to the project wanagement web site submittal survey shall be in Auto CAD format. Digital Submittal shall be of the project site sature showing all of the following:
10			be in Auto CAD format. Digital Submittal shall be of the project site setup showing all of the following:
12			a. Rey reactives not scheduled for demonstron, including but not infinited to building corners, root
12			h Location of construction limits fencing
1/			c Location of PLSS and/or project control points provided by the Owner
15			d Locations of project based control points provided by the owner.
16			 Printed Survey Submittal shall be the same as item 1 above in PDE format. PDE file shall be formatted to
17			5. Printed Survey Submittal shall be the same as item 1 above in PDF format. PDF the shall be formatted to r_{1} print to scale on 24° x36" sheets as required to show all features with text neatly organized for each item
18			identified When multiple sheets are used a match line and sheet references shall be required
19			4 PDE file of the complete level/laver scheme. Scheme shall be in tabular form formatted to 8.5 by 11
20			naner and shall include all of the following:
20			a Level/laver designation (abbreviation)
22			h Level/laver designation (full title)
23			c Feature attribute characteristics (line weight line style font etc.)
24			d. Cell attribute information
25			e. Samples of line styles and cells.
26			
27	1.6.	EXAM	INATION
28		Α.	The PLS shall be responsible for verifying all site data including the owner provided local control points (see
29			Section 3.1 below) prior to starting the Work.
30		В.	Notify the Project Architect and CPM/CCM immediately if any discrepancies are discovered.
31			
32	PART	2 – PRO	DUCTS – NOT USED
33			
34	PART	3 - EXEC	CUTION
35			
36	3.1.	PRE-C	ONSTRUCTION OWNER SUPPORT
37		Α.	The CPM/CCM shall provide the GC/PLS with a digital CAD seed file on or before the Pre-construction meeting.
38			1. Seed file shall be an Auto Cad seed file using the datum indicated above. Seed file shall be delivered as a
39			Auto Cad format as requested by the PLS.
40			a. Seed file shall be used as the PLS's initial base file for all future work on this contract.
41			
42	3.2.	UTILIT	YLOCATING
43		Α.	The GC and/or PLS shall be responsible for notifying Diggers Hotline for all utility locate requests.
44			
45	3.3.	SURVE	EY CONTROL AND LAYOUT DATA
46		А.	The GC and PLS are responsible for all other survey control and layout data required to perform the work in this
47			contract.
48			
49	3.4.	TOPO	GRAPHIC SURVEYING
50		Α.	The Surveyor may perform the topographic survey with properly calibrated equipment as follows:
51			1. Total station, achieving minimum accuracy for well-defined features of +/- 0.1 feet horizontal and +/-0.04
52			teet vertical at 95% confidence relative to control. "Well defined features" shall include but not be
53			limited to property irons, pavements, trees, landscaping features, buildings, utility locations, and other
54			permanent features.
55			2. RIK GPS shall be permitted in large open areas, along tree lines, and in brushy areas.
56			

1	3.5.	SITE SUF	RVEY AS-BUILT
2		A. S	ee Specification 01 78 39 As-Built Drawings, Section 3.2 for more information on required record site
3		i	nformation to be provided prior to contract closeout.
4		В. Т	he GC shall be responsible for scheduling the PLS to capture locations and depths of all buried utilities prior to
5		а	any contractor back filing trenches. The Owner may require missing information to be located and surveyed at
6		t	
7		c	
, Q			
0			
9			
10			END OF SECTION
11			
12			
13			
14			
15			
16			
17			
18			
19			
20			
21			
22			
23			
24			
25			
26			
27			
28			
29			
30			
31			
32			
33			
34			
35			
36			
37			
38			
30			
40			
40 //1			
41			
42			
45 11			
44 15			
45			
40			
47			
48			
49			
50			
51			
52			
53			
54			
55			
56			
57			
58			
	WAR	NER PARK	COMMUNITY RECREATION CENTER
	EXPA	NSION	

		SECTION 01 32 26
		CONSTRUCTION PROGRESS REPORTING
PART	1 – GE	ENERAL
	1.1.	SUMMARY
	1.2.	RELATED SPECIFICATION SECTIONS
-	1.3.	PERFORMANCE AND QUALITY ASSURANCE REQUIREMENTS
PART	2 – PF	RODUCTS - THIS SECTION NOT USED
PART	3 - EX	
3	3.1.	CONTRACTOR JOURNAL
3	3.2.	CONSTRUCTION PROGRESS MEETINGS
<u>PART</u>	1 – G	ENERAL
1.1.	SUN	MMARY
	Δ	Daily records of project activities resources used weather conditions and other information related to the
	7.	ongoing progress of the project activities, resources used, weather conditions, and other information related to the
	В.	Daily records provide the base for weekly progress reports and updating progress schedules.
	υ.	buily records provide the base for weekly progress reports and aparting progress schedules.
1.2.	REL	ATED SPECIFICATION SECTIONS
	Α.	Section 01 31 19 Project Meetings
	В.	Section 01 31 23 Project Management Web Site
	C.	Section 01 32 23 Photographic Documentation
1.3.	PER	RFORMANCE AND QUALITY ASSURANCE REQUIREMENTS
	Α.	The General Contractor (GC) shall be responsible for all Construction Progress Reporting as outlined in this
		other specifications as noted.
	В.	Ine GC shall maintain daily progress journals in a format of their choosing provided it is legible and contain
	-	information as outlined in Section3.1 below.
	C.	information as outlined in Section3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project
	C.	information as outlined in Section3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested.
DADT	С.	information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested.
<u>PART</u>	C. • 2 – Pi	information as outlined in Section3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED
<u>PART</u> PART	C. <u>2 – Pi</u> 3 - EX	information as outlined in Section3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED KECUTION
<u>PART</u> PART	C. • 2 – Pi • 3 - EX	information as outlined in Section3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED KECUTION
<u>PART</u> <u>PART</u> 3.1.	C. <u>2 – Pl</u> <u>3 - EX</u> CON	information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED KECUTION NTRACTOR JOURNAL
<u>PART</u> <u>PART</u> 3.1.	C. • <u>2 – PI</u> • <u>3 - EX</u> CON A.	information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED KECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for
<u>PART</u> <u>PART</u> 3.1.	C. <u>2 – Pi</u> <u>3 - EX</u> CON A.	information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED XECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for which the GC is responsible. Such reports shall include all relevant data concerning the progress of Work
<u>PART</u> <u>PART</u> 3.1.	C. <u>2 – PI</u> <u>3 - EX</u> CON A.	information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED XECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for which the GC is responsible. Such reports shall include all relevant data concerning the progress of Work activities the GC and Subcontractors are responsible for and the effect of that activity on the time of
<u>PART</u> <u>PART</u> 3.1.	C. <u>2 – PI</u> <u>3 - EX</u> CON A.	information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED XECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for which the GC is responsible. Such reports shall include all relevant data concerning the progress of Work activities the GC and Subcontractors are responsible for and the effect of that activity on the time of performance of the Contract.
<u>PART</u> <u>PART</u> 3.1.	C. <u>2 – PI</u> <u>3 - EX</u> CON A.	 information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED KECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for which the GC is responsible. Such reports shall include all relevant data concerning the progress of Work activities the GC and Subcontractors are responsible for and the effect of that activity on the time of performance of the Contract. 1. Some projects may not require weekly journals be kept instead of daily journals. This is at the sole
<u>PART</u> <u>PART</u> 3.1.	С. <u>2 – РІ</u> <u>3 - ЕХ</u> СОР А.	 information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED KECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for which the GC is responsible. Such reports shall include all relevant data concerning the progress of Work activities the GC and Subcontractors are responsible for and the effect of that activity on the time of performance of the Contract. 1. Some projects may not require weekly journals be kept instead of daily journals. This is at the sole discretion of the City Project Manager. A daily journal will generally be required when the contract
<u>PART</u> <u>PART</u> 3.1.	C. <u>2 – PI</u> <u>3 - EX</u> CON A.	 information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED KECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for which the GC is responsible. Such reports shall include all relevant data concerning the progress of Work activities the GC and Subcontractors are responsible for and the effect of that activity on the time of performance of the Contract. 1. Some projects may not require weekly journals be kept instead of daily journals. This is at the sole discretion of the City Project Manager. A daily journal will generally be required when the contract significant amount of site work. A weekly journal will generally be used when a contract is interior when the sole of the contract.
<u>PART</u> <u>PART</u> 3.1.	С. • <u>2 – РІ</u> • <u>3 - ЕХ</u> СОМ А.	 information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED RECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for which the GC is responsible. Such reports shall include all relevant data concerning the progress of Work activities the GC and Subcontractors are responsible for and the effect of that activity on the time of performance of the Contract. 1. Some projects may not require weekly journals be kept instead of daily journals. This is at the sole discretion of the City Project Manager. A daily journal will generally be required when the contract significant amount of site work. A weekly journal will generally be used when a contract is interior wonly.
<u>PART</u> <u>PART</u> 3.1.	С. <u>2 — РІ</u> <u>3 - ЕХ</u> СОМ А. В.	 information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED KECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for which the GC is responsible. Such reports shall include all relevant data concerning the progress of Work activities the GC and Subcontractors are responsible for and the effect of that activity on the time of performance of the Contract. 1. Some projects may not require weekly journals be kept instead of daily journals. This is at the sole discretion of the City Project Manager. A daily journal will generally be required when the contract significant amount of site work. A weekly journal will generally be used when a contract is interior wonly. Journal entries shall be made in the Project Management Web Site. The form consists of the following area
<u>PART</u> <u>PART</u> 3.1.	С. <u>2 – Pf</u> <u>3 - EX</u> <u>СОМ</u> А. В.	 information as outlined in Section3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED XECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for which the GC is responsible. Such reports shall include all relevant data concerning the progress of Work activities the GC and Subcontractors are responsible for and the effect of that activity on the time of performance of the Contract. 1. Some projects may not require weekly journals be kept instead of daily journals. This is at the sole discretion of the City Project Manager. A daily journal will generally be required when the contract significant amount of site work. A weekly journal will generally be used when a contract is interior wonly. Journal entries shall be made in the Project Management Web Site. The form consists of the following area 1. Weather; include temperature, humidity, precipitation, wind and other related information such as
<u>PART</u> <u>9ART</u> 3.1.	С. <u>2 – Р</u> <u>3 - ЕХ</u> СОМ А. В.	 information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED XECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for which the GC is responsible. Such reports shall include all relevant data concerning the progress of Work activities the GC and Subcontractors are responsible for and the effect of that activity on the time of performance of the Contract. 1. Some projects may not require weekly journals be kept instead of daily journals. This is at the sole discretion of the City Project Manager. A daily journal will generally be required when the contract significant amount of site work. A weekly journal will generally be used when a contract is interior wonly. Journal entries shall be made in the Project Management Web Site. The form consists of the following area 1. Weather; include temperature, humidity, precipitation, wind and other related information such as significant storm events, times, and details.
<u>PART</u> <u>9487</u> 3.1.	С. <u>2 – РІ</u> <u>3 - ЕХ</u> СОМ А.	 information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED XECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for which the GC is responsible. Such reports shall include all relevant data concerning the progress of Work activities the GC and Subcontractors are responsible for and the effect of that activity on the time of performance of the Contract. 1. Some projects may not require weekly journals be kept instead of daily journals. This is at the sole discretion of the City Project Manager. A daily journal will generally be required when the contract significant amount of site work. A weekly journal will generally be used when a contract is interior of only. Journal entries shall be made in the Project Management Web Site. The form consists of the following area 1. Weather; include temperature, humidity, precipitation, wind and other related information such as significant storm events, times, and details. 2. Work completed by trade
<u>PART</u> <u>9487</u> 3.1.	С. <u>2 — РІ</u> <u>3 - ЕХ</u> СОМ А.	 information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED KECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for which the GC is responsible. Such reports shall include all relevant data concerning the progress of Work activities the GC and Subcontractors are responsible for and the effect of that activity on the time of performance of the Contract. Some projects may not require weekly journals be kept instead of daily journals. This is at the sole discretion of the City Project Manager. A daily journal will generally be required when the contract significant amount of site work. A weekly journal will generally be used when a contract is interior work. Journal entries shall be made in the Project Management Web Site. The form consists of the following area Weather; include temperature, humidity, precipitation, wind and other related information such as significant storm events, times, and details. Work completed by trade Delays encountered
<u>PART</u> <u>PART</u> 3.1.	С. <u>2 — РІ</u> <u>3 - ЕХ</u> СОМ А.	 information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED XECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for which the GC is responsible. Such reports shall include all relevant data concerning the progress of Work activities the GC and Subcontractors are responsible for and the effect of that activity on the time of performance of the Contract. 1. Some projects may not require weekly journals be kept instead of daily journals. This is at the sole discretion of the City Project Manager. A daily journal will generally be required when the contract significant amount of site work. A weekly journal will generally be used when a contract is interior wonly. Journal entries shall be made in the Project Management Web Site. The form consists of the following area 1. Weather; include temperature, humidity, precipitation, wind and other related information such as significant storm events, times, and details. 2. Work completed by trade 3. Delays encountered 4. Deliveries received or delayed
<u>PART</u> <u>9ART</u> 3.1.	С. <u>2 — РІ</u> <u>3 - ЕХ</u> СОМ А.	 information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED XECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for which the GC is responsible. Such reports shall include all relevant data concerning the progress of Work activities the GC and Subcontractors are responsible for and the effect of that activity on the time of performance of the Contract. 1. Some projects may not require weekly journals be kept instead of daily journals. This is at the sole discretion of the City Project Manager. A daily journal will generally be required when the contract significant amount of site work. A weekly journal will generally be used when a contract is interior wonly. Journal entries shall be made in the Project Management Web Site. The form consists of the following area 1. Weather; include temperature, humidity, precipitation, wind and other related information such as significant storm events, times, and details. 2. Work completed by trade 3. Delays encountered 4. Deliveries received or delayed 5. Hot issues that need to be addressed
<u>PART</u> <u>9ART</u> 3.1.	С. <u>3 - ех</u> <u>со</u> м А.	 information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED KECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for which the GC is responsible. Such reports shall include all relevant data concerning the progress of Work activities the GC and Subcontractors are responsible for and the effect of that activity on the time of performance of the Contract. Some projects may not require weekly journals be kept instead of daily journals. This is at the sole discretion of the City Project Manager. A daily journal will generally be required when the contract significant amount of site work. A weekly journal will generally be used when a contract is interior wonly. Journal entries shall be made in the Project Management Web Site. The form consists of the following area Weather; include temperature, humidity, precipitation, wind and other related information such as significant storm events, times, and details. Work completed by trade Delays encountered Delays encountered Delays encountered Delays encountered Hot issues that need to be addressed Safety issues
<u>PART</u> <u>9487</u> 3.1.	С. <u>3 - ех</u> <u>со</u> м А.	 information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED XEECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for which the GC is responsible. Such reports shall include all relevant data concerning the progress of Work activities the GC and Subcontractors are responsible for and the effect of that activity on the time of performance of the Contract. Some projects may not require weekly journals be kept instead of daily journals. This is at the sole discretion of the City Project Manager. A daily journal will generally be required when the contract significant amount of site work. A weekly journal will generally be used when a contract is interior v only. Journal entries shall be made in the Project Management Web Site. The form consists of the following area Weather; include temperature, humidity, precipitation, wind and other related information such as significant storm events, times, and details. Work completed by trade Delays encountered Delays encountered Safety issues Photograph progress and upload to the Photo Library on the Project Management Web Site.
<u>PART</u> <u>9481</u> 3.1.	С. <u>3 - ех</u> <u>со</u> м А.	 information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED XECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for which the GC is responsible. Such reports shall include all relevant data concerning the progress of Work activities the GC and Subcontractors are responsible for and the effect of that activity on the time of performance of the Contract. Some projects may not require weekly journals be kept instead of daily journals. This is at the sole discretion of the City Project Manager. A daily journal will generally be required when the contract significant amount of site work. A weekly journal will generally be used when a contract is interior v only. Journal entries shall be made in the Project Management Web Site. The form consists of the following area 1. Weather; include temperature, humidity, precipitation, wind and other related information such as significant storm events, times, and details. Work completed by trade Delays encountered Delays encountered Safety issues Photograph progress and upload to the Photo Library on the Project Management Web Site. Other including inspections, testing, etc.
<u>PART</u> <u>9481</u> 3.1.	С. <u>3 - ех</u> <u>со</u> м А.	 information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED XECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for which the GC is responsible. Such reports shall include all relevant data concerning the progress of Work activities the GC and Subcontractors are responsible for and the effect of that activity on the time of performance of the Contract. 1. Some projects may not require weekly journals be kept instead of daily journals. This is at the sole discretion of the City Project Manager. A daily journal will generally be required when the contract significant amount of site work. A weekly journal will generally be used when a contract is interior or only. Journal entries shall be made in the Project Management Web Site. The form consists of the following area 1. Weather; include temperature, humidity, precipitation, wind and other related information such as significant storm events, times, and details. 2. Work completed by trade 3. Delays encountered 4. Deliveries received or delayed 5. Hot issues that need to be addressed 6. Safety issues 7. Photograph progress and upload to the Photo Library on the Project Management Web Site. 8. Other including inspections, testing, etc. 9. Space for attaching documents
<u>PART</u> <u>9481</u> 3.1.	С. <u>2 — РІ</u> <u>3 - ЕХ</u> СОМ А. В.	 information as outlined in Section 3.1 below. The journal shall be located in the job trailer and shall be reviewable by the Project Architect or City Project Manager if so requested. RODUCTS - THIS SECTION NOT USED XECUTION NTRACTOR JOURNAL The GC shall maintain a journal of daily progress on which Work is performed by any employee or entity for which the GC is responsible. Such reports shall include all relevant data concerning the progress of Work activities the GC and Subcontractors are responsible for and the effect of that activity on the time of performance of the Contract. 1. Some projects may not require weekly journals be kept instead of daily journals. This is at the sole discretion of the City Project Manager. A daily journal will generally be required when the contract significant amount of site work. A weekly journal will generally be used when a contract is interior of only. Journal entries shall be made in the Project Management Web Site. The form consists of the following area 1. Weather; include temperature, humidity, precipitation, wind and other related information such as significant storm events, times, and details. 2. Work completed by trade 3. Delays encountered 4. Deliveries received or delayed 5. Hot issues that need to be addressed 6. Safety issues 7. Photograph progress and upload to the Photo Library on the Project Management Web Site. 8. Other including inspections, testing, etc. 9. Space for attaching documents Contractor Daily/Weekly Report Forms shall be completed and signed by the GC's Job Superintendent or ot other contracts or project management Scontractor Daily/Weekly Report Forms shall be completed and signed by the GC's Job Superintendent or ot progress is a significant and progress and upload to the Photo Library on the GC's Job Superintendent or ot progress is a significant and progress and upload to the Photo Library on the GC's Job Superint

1		D.	If applicable the GC shall include schedules of quantities and costs, progress schedules, wage rates, reports,
2			estimates, invoices, records and other data as requested by the CPM concerning Work performed or to be
3			performed under this Contract if the CPM determines such information is needed to substantiate Change Order
4			proposals, claims, or to resolve disputes.
5			
6	3.2.	CONS	TRUCTION PROGRESS MEETINGS
7		Α.	The GC shall provide a verbal summary of the previous two (2) weeks progress reports at each bi-weekly
8			construction progress meeting.
9			
10			
11			END OF SECTION
12			

				PHOTOGRAPHIC DOCUMENTATION				
PART	1 – G	ENFRAI						
, , , , , ,	11	SCOPF						
-	1 2	RELATED SPECIFICATION SECTIONS						
1 3								
PART	2 – PI	RODUCTS	5					
	2.1.	DIGITAI	CAMFRA					
2	2.1.	TIME LA	APSE CONSTRUC	TION CAMERA (TLCC)				
PART	3 – E)		N					
3	3.1.	REOUIR	EMENTS FOR D	IGITAL PHOTOGRAPHS				
3	3.2.	REQUIR	EMENTS FOR T	ME LAPSE PHOTOGRAPHS				
PART	1 – G	ENERAL						
1.1.	SCOPE							
	Α.	The C	General Contrac	tor (GC) shall be required to take weekly digital photographs of interior and exterior				
	<i>,</i>	const	truction progres	as and upload the photos directly to the Project Management Web Site (PMWS)				
	Β.	The C	GC shall be requi	ired to provide digital time-lapse photo service of the project exterior -or interior when				
	5.	appli	cable- construct	tion progress. Exterior or interior location determination to be confirmed with City				
		Cons	truction Manag	er.				
		20113						
1.2.	REL	ATED SP	ECIFICATION SE	CTIONS				
	Α.	Section	on 01 29 76	Progress Payment Procedures				
	В.	Section	on 01 31 23	Project Management Web Site (PMWS)				
	C.	Section	on 01 32 19	Submittals Schedule				
	D.	Section	on 01 32 33	Submittals				
	Ε.	Section	on 01 77 00	Closeout Procedures				
1.3.	SUE	SUBMITTALS						
	Α.	The C	GC shall provide	general information on the type of camera being used for interior and exterior digital				
		phote	ographs.					
		1.	Information r	may be written on Contractor's transmittal sheet.				
			a. Includ	le camera name/type, aspect ratio setting, and average file size				
			b. Provid	de sample project pictures as part of PDF submittal.				
	В.	The C	GC shall provide	sufficient information on the type of time lapse system being used that meets the				
		requirements identified in section 2.2 below.						
PART	2 – P	RODUCT	s					
			-					
2.1.	DIG	ITAL CAN	VIERA					
	Α.	All di	gital photograp	hs shall be taken with a good quality digital camera, cell phone, tablet, and other such c				
	_	devic	.e.					
	В.	Digita	al photographs s	snail be formatted to achieve a good, clear, and detailed image where the final file size				
		betw	een 600 KB and	3.U IVIB (3UUUKB).				
22	TIN		CONSTRUCTION	Ν ΓΔΜΕΒΑ (ΤΙ ΓΓ)				
<u></u> .	Δ	The T		high quality weather proof camera owned and operated or leased, by the GC for the				
	л.	h ann an traite	tion of this cont	ract with the following minimum canabilities:				
		1	Pan-Tilt-700r	n (PT7) capable.				
		2	Wireless inte	rnet or built in cellular technology capable.				
		۷.	a Thom	se of memory cards will not be permitted				
		2	Widescreen	high resolution (5-30 MP rating)				
		۵. ۵	Powered hy '	120V AC				
		- T .	a Theu	se of battery packs will not be permitted				
		5	Web/cloud b	osted access to archived photos and video				
		с. С	Provides com	nlete time lanse video canability				

CONTRACT #9502 MUNIS #17196

	В.	 24/ / service and support for equipment, software, and hosting services. Approved equipment/services include but are not limited to the following: OxBlue Corporation www.oxblue.com 								
		2. EarthCam <u>www.earthcam.net</u>								
		4 Evercam www.evercam.com								
		- Everenni www.everennicom								
PART	3 – EXI	ECUTION								
3.1.	REQU	JIREMENTS FOR DIGITAL PHOTOGRAPHS								
	Α.	The GC shall take a minimum of two (2) exterior photographs each week. Exterior photographs will not be								
		required on projects that do not include any exterior work.								
		1. Exterior photos shall be taken from approximately the same location each week for the duration of project								
		project. When applicable this requirement shall begin prior to commencing any site work								
		 When applicable this requirement shall begin profit to commencing any site work. This requirement shall only be applicable when there is exterior work actively being conducted with 								
		project Periods of inactivity due to weather (winter conditions) do not require a photograph								
		 This requirement shall end when the exterior work has been substantially completed. 								
		5. This requirement may be suspended due to weather conditions or substantial delays in exterior pro-								
	В.	The GC shall take interior photographs each week that document interior construction progress.								
		1. This requirement will begin when exterior wall framing begins.								
		a. When an interior remodeling project includes demolition work interior photos shall be take								
		during the demolition process.								
		2. Pictures do not need to be taken from the same location each week.								
		3. This requirement shall end when the interior work has been substantially completed.								
	C.	Digital photographs shall be properly zoomed in/out, and flash used as needed, to capture a level of detail								
		required to properly show the progress being captured by the photograph.								
		1. Blurry and dark pictures will not be accepted.								
	D.	i ne camera default naming convention is acceptable. The GC does not need to rename or specifically ider								
	F	Pictures with a title. All digital photographs shall be saved in a IDEC / ing) format and unleaded directly to the Breiset Menager								
	L.	An aighar photographis shan be saved in a JFEG (Jpg) format and uploaded directly to the Project Manageri Web Site								
3.2.	REOL	JIREMENTS FOR TIME LAPSE PHOTOGRAPHS								
	Α.	The GC shall be responsible for all of the following:								
		1. Install an approved operation timelapse camera within 30 days after date fixed by Start Work Lette								
		and/or Notice to Proceed								
		2. Verify with the CPM/CCM a suitable place for mounting the camera and related equipment prior to								
		installation.								
		3. The complete installation, setup, maintenance, and removal of the camera and related equipment.								
		4. The hosting and access of all photographs and videos taken by the camera during the project.								
		5. Production of a final time lapse video (minimum of 3 minutes in length) of the project provided in a								
	-	viewable format to the Owner on a thumb drive or CD.								
	В.	I me lapse photos shall be taken from the same fixed position at approximately ten (10) minute intervals.								
		1. Inne lapse shall start before normal daily activities begin and end after normal daily activities have								
		COMPLETED.								
		a. The GC shall adjust the camera time lapse schedule as needed to accommodate any periods								
		h Time lanse shall not be taken during major periods of no activity including night hours hold								
		weather related (winter) inactivity, etc.								
	C.	All photos taken during the execution of this contract shall be accessible from a web-based service. Archiv								
		photos shall be organized by date and time so that they can be easily retrieved and viewed as needed.								
		1. If necessary, the GC shall coordinate usernames and passwords for access to the photos. The City of								
		Madison would prefer that the access be generic to accommodate a wide audience.								
		END OF SECTION								
1 2	SECTION 01 33 23 SUBMITTALS									
----------------	--------------------------------	--------------	--	--	--	--	--	--	--	--
3										
4	PART	1 – GI	ENERAL1							
5	:	1.1.	SUMMARY1							
6		1.2.	RELATED REFERENCES							
/		1.3.	SUBMITTAL REQUIREMENTS							
8	PARI	2 – Pł	RODUCTS – THIS SECTION NOT USED							
9 10	PARI	3 - EX								
10		3.1. วา	GENERAL CONTRACTOR S PROCEDURES							
11		5.Z. 5.2								
12		5.5.								
13 14 15	PART	<u>1 – G</u>	ENERAL							
16	1.1.	SUN	ΛΜΑRΥ							
17		Α.	The General Contractor (GC) shall be responsible for providing submittals for review of all contractors and sub-							
18			contractors as designated in the construction documents. Submittals shall include but not be limited to all of the							
19			following:							
20			1. Equipment specified and pre-approved in the specification; to ensure quality, construction, and							
21			performance specifications have not changed since final design.							
22			2. Equipment specified by performance in the specification; to ensure that the intended quality,							
23			construction, and performance specified is met by the selected material or product.							
24			3. Shop, piece, erection, and other such drawings as indicated in the specifications to ensure all structural,							
25			dimensional, and assembly requirements are being met.							
26			4. Submittals indicating installation sequencing							
27			5. Submittals indicating control sequencing							
28			 Contractor licensing, certification, and other such regulatory documentation when required by a specification. 							
29			specification.							
30 21		D	7. Other submittal process shall not be used to determine alternates to specified products or equipment. All							
33		ь.	considerations shall be reviewed during the hidding process and acceptable alternates shall be acknowledged by							
32			addendum prior to the closing of hidding. See hidding instructions for the information on submitting alternates							
34			for consideration							
35		D.	In the event that a manufacturer has significantly changed a product (discontinued a model, changed dimension							
36			or performance data changed available colors. etc.) since bid opening the GC shall submit a Request for							
37			Information (RFI) to the Project Architect requesting other approved alternates prior to uploading a digital							
38			submittal.							
39		Ε.	Contractors and sub-contractors shall be responsible for knowing the submittal requirements of ALL sections							
40			within their scope of work under the contract. The Owner reserves the right to request documentation on any							
41			materials, equipment, or product being installed where a submittal is not on file. If the material, equipment, or							
42			product installed is determined not to meet the intent of the specification the contractor/sub-contractor shall be							
43			required to remove and replace the items involved. The GC shall be solely responsible for all costs associated							
44			with the removal and replacement.							
45		F.	Doors, Frames + Hardware Submittals - After submission of all door/frame/hardware submittals (and related low							
46			voltage door hardware submittals) Contractor will organize a meeting(s) with Owner, Architect, General							
47			Contractor, Electrician, Door/Frame/Hardware Supplier(s)/Installer(s), Low-Voltage Supplier/Installer, and others							
48			as applicable to comprehensively review and explain each door opening's submitted hardware package							
49			operation. Prior to this meeting the low voltage contractor shall have completed a review with the Madison Fire							
50			Department for all access control doors and be prepared to explain any conflicts or concerns with all parties. No							
51			procurement of door hardware (and related low voltage components) shall be procured until this meeting is							
52 52	1 2	DEI	ATED BEEEBENCES							
53	1.2.	A NEL	Section 01 29 76 Progress Payment Procedures							
55		A. R	Section 01 21 23 Project Management Web Site (PMW/S)							
56		с.	Section 01 32 19 Submittals Schedule							
57		D.	Section 01 32 26 Construction Progress Reporting							
58		Е.	Section 01 91 00 Commissioning							
			~ ~							

	F.	All Technical Specifications, contract documents, construction drawings, and any published addendums during						
		the b	the bidding process.					
	G.	All co	All contract documents generated during the execution of the contract including but not limited to Requests for					
		Infor	Information (RFI) and Construction Bulletins (CB).					
1 2	CLIDA							
1.3.	SORI	VIIIIAL	REQUIREMENTS					
	А.	A COr	mpieted submittal shall meet the following requirements:					
		1.	Digital submittal shall be original PDF of manufacturer's data sheets or high quality color scan of the					
			same.					
			a. Submittals shall not include sales fliers or other similar documents that typically do not provide					
			complete manufacturers data.					
		2.	Documents within the PDF submittal shall be printable to a sized sheet no less than 8-1/2 by 11 inches					
			and no larger than 24 by 36 inches.					
		3.	At the beginning of each submittal the contractor shall identify the plan reference (WC-1, EF-3, etc.) in					
			RED block letters that the submittal is for.					
		4.	Where multiple model numbers appear in a table the contractor shall identify the specific model being					
			submitted by using a RED square, box, or other designation to distinguish the correct model from others					
			on the page.					
	В.	A cor	mplete submittal will include all information associated with the product or equipment as presented in					
		plans	s, equipment tables, and specifications. Information shall include but not be limited to the following:					
		1.	Dimensional data					
		2.	Performance data					
		3.	Resource requirements, power, water, waste, etc.					
		4.	Clearance and maintenance requirements					
		5.	Finish information, colors, textures, etc.					
		6.	Warranty information					
	C.	Whe	re a submittal includes material samples (carpet, tile, paint draw downs, etc.) the contractor shall do the					
		follo	wing:					
		1.	The Contractor shall submit the sample(s) as indicated in the specification.					
		2.	The Contractor shall include a quality photograph(s) of the product with the digital submittal.					
			Photographs shall meet the following requirements:					
			a. Formatted to be between 500Kb and 1.0 Mb in file size					
			b. Have no glare or flash reflection on the sample					
			c. Sample fills the frame of the photo and shows detail as needed. Include multiple photos from					
			other angles as needed.					
			d Scanned conject of products or photos are not accentable					
	р	Unlo	aded submittals should be relative and related to a specific written specification					
	υ.	1	Do not unload submittals under a broad category or division (LE_HVAC 23.00.00). Always unload by the					
		1.	<u>specific specification that identifies a required product or performance to be met</u>					
		2	Group related items together if the specification is written that way. (I.E. all of the plumbing fixtures and					
		۷.	trim relative teams together in the spectration is written that way. (i.i. all of the planning includes and trim relative teams together in the spectration should be submitted together).					
		2	in the relative to the specific specification should be submitted (openier).					
		5.	submittal shall be glouped and duriele to the divisions in the submittal schedule. Submittal schedule and/or specification divisions will be rejected for re-submittal					
			comorm to the submittal schedule and/or specification divisions will be rejected for re-submittal.					
PART	2 – PR	ODUCT	S – THIS SECTION NOT USED					
PART	3 - EXE	CUTIO	<u>N</u>					
3 1	GENI							
5.1.	Δ	All re	equired submittals will be unloaded to the Project Management Web Site (PMWS) by the GC					
	7	1	Fill in required information on the form that will be used for routing the review and comments					
		2	Attach all documentation as described in Section 1.3 above					
		۲.	a Submit samples under senarate cover to the Project Architect when necessary					
		Unio	a. Submit samples under separate cover to the Project Architect when heldssary.					
	P	opio	aung the submittal indicates that the GC has reviewed and approved the submittal against the contract					
	В.	d	locument requirements.					
	B.	docu	iment requirements.					
	В. С.	docu The (ment requirements. GC shall discuss submittal status at all progress meetings and shall monitor submittal review/approval/re-					
	В. С.	docu The (subm	ment requirements. GC shall discuss submittal status at all progress meetings and shall monitor submittal review/approval/re- nittal so as to not incur delays in the project schedule.					

1		E.	The GC and sub-contractors shall provide re-submittals as required.
2			
3	3.2.	SUBIV	IITTAL REVIEW
4		Α.	Upon completion of the submittal upload by the GC the PMWS automatically notifies the appropriate
5			Architect/Engineer and Owner Representative, including CxA, by Division/Specification number that there is a
6			submittal for review.
7		В.	The submittal shall be reviewed internally by the required Architect/Engineer and Owner Representative and
8			CxA in a timely fashion and provide commentary on missing items, incorrect information, or incomplete shop
9			drawings, etc as needed.
10		C.	When the internal review is completed the PMWS will notify the Project Architect the submittal is ready for final
11			review.
12			
13	3.3.	PROJE	ECT ARCHITECT'S REVIEW
14		Α.	Upon completion of the internal review the Project Architect shall review all internal review comments, confer
15			with the CPM and CxA as needed and determine the appropriate disposition status for the submittal (approved
16			or resubmit).
17		В.	The Project Architect shall summarize final internal review comments onto the submittal cover sheet, provide a
18			final disposition of the submittal and update the review status of the submittal to "Complete" (with or w/o
19			comments) or "Rejected".
20		C.	A completed Final Review status will be completed by the City Project or City Construction Manager and initiates
21			the PMWS to notify the GC and appropriate sub-contractor(s) that the review of the submittal has been
22			completed.
23			
24			
25			
26			END OF SECTION
27			

		SECTION 01 43 39
		MOCKUPS
PART	1 – G	ENERAL
	1.1.	SUMMARY
	1.2.	RELATED SPECIFICATIONS
	1.3.	RELATED DOCUMENTS
	1.4.	PERFORMANCE REQUIREMENTS
	1.5.	QUALITY ASSURANCE
PART	⁻ 2 - PF	RODUCTS
	2.1.	MATERIALS
PART	- 3 - Ελ	ECUTION
	3.1.	REVIEW THE PLANS AND SPECIFICATIONS
	3.2.	MOCKUP CONSTRUCTION
	3.3.	MOCKUP REVIEW
	3.4.	FINAL SUBMITTAL
PART	[1–G	ENERAL
1.1.	SUI	VIIVIAKY
	А.	Deminicum 1. Modeling are field complex constructed, applied, or accombined at the president site for restrict of the first
		1. Mockups are field samples constructed, applied, or assembled at the project site for review by the
		Owner, Owners Representative, Architect and Consultants.
		2. Mockups are three dimensional, true scale models that illustrate materials and methods, equipment,
		workmanship, or location; based on plans, details, and assemblies.
	В.	Approved mockups establish the standard of quality by which the final work will be judged.
	C.	Approved mockups shall be properly documented and entered into the Submittal Library on the Project
		Management Web Site like any other required submittal. See section 3.4 below for more information.
1 2	DEI	
1.2.		Sortion 01 26 12 Bequest for Information (PEI)
	д. Б	Section 01 26 16 Change Bulliotin (CP)
	ь. С	Section 01 26 40 Change Order (CO)
	С. D	Section 01 20 05 Change Order (CO)
	D. E	Section 01 32 16 Construction Programs Schodulos
	с. г	Section 01 32 10 Constituction Progress Schedules
	г. С	Section 01 33 23 Submittals
	G.	Section 01 45 00 Quality Control
1 2	DEI	
1.5.		The following documents shall be used for propering maskups
	А.	All plans, specifications, and datails including those derived as revisions (PELCP, CO)
		An plans, specifications, and details including those derived ds revisions (KFI, CB, CO). Construction Progress Schedules, Mackups shall be deep and completed in a timely factor.
		2. Construction Progress schedules. Workups shall be done and completed in a timely fashion for review
		and approval so as to not impact the contractors project schedule.
		ס. אויץ ואמוועומנוערפוט וווטנמוומנוטוו/מסטפוווטוץ וווטנו ענוטווט.
1.4	DEC	REORMANCE REQUIREMENTS
1.4.	ΓLΓ Δ	All Contractors shall be responsible for providing and constructing mackups as specified in their Division of M
	<i>.</i>	in the plans and specifications
	R	Materials to be used shall be as specified in the construction documents, full sized and properly assembled
	ь. С	Completed markups shall be as specified in the construction documents, run sized and property assembled.
	С.	completed motivaps shall be of sufficient size to provide visible detail of all components as needed for the
		sumpte.
1.5	011	
1.3.	∆ 2	The General Contractor (GC) shall be responsible for coordinating all of the following as needed.
	л.	1 Designating the location for the mockup construction
		2. Coordinating the work of all contractors and materials required to complete the medium
		2. Coordinating the work of an contractors and materials required to complete the mockup 2. Ensuring that the mockup meets the intent of the construction documents before scheduling the moc
		s. Ensuring that the mockup meets the intent of the construction documents before scheduling the mot

2 PART 2 - PRODUCTS

2.1. MATERIALS

- A. The materials used in mockups shall be only those materials indicated in the plans, specifications, and favorably reviewed submittals.
- B. Mockups shall be made of full scale materials as delivered to the project site.
- C. All materials associated with a particular detail, construction method, manufacturer's installation instructions shall be properly represented and visible in the mockup. This includes but is not limited to finished mortar joints, sealants, backer rods, tie bars, rebar, etc.

12 PART 3 - EXECUTION

13 14 **3.1**.

1

3 4

5

6

7

8

9

10

11

15

16

17 18

19 20

21

22

23

27

29

32

33

34

3.1. REVIEW THE PLANS AND SPECIFICATIONS

- A. The GC shall review the plans and specifications with all required contractors prior to constructing the mockup.
 - 1. Mockups that will be built and remain in place, if favorably reviewed, will be installed in an area easily accessible for review.
 - 2. Mockups that will not be built in place or will not remain will be constructed in a space on the project site protected from weather, construction traffic, and other such disturbances until such time as the associated work has been completed.
 - 3. Insure all products being represented in the mockup meet the plans, specifications, and any published changes.

24 **3.2. MOCKUP CONSTRUCTION**

- A. Mockups shall be of sufficient size to show various material adjacencies, connectivity, patterns, and other such
 related features.
 - B. Mockups shall be constructed in a layered fashion so that all products being used can be seen and evaluated.
- 28 C. The construction detail below is an example of a properly layered mockup.



30 31 D. REQUIRED MOCK UPS INCLUDE:

- WALL AT METAL PANEL
- WALL AT MASONRY

35 3.3. MOCKUP REVIEW

36	А.	The General Contractor and all associated Sub-contractors (Contracting Team) shall meet with the Owner,
37		Owners Representative, Architect and Consultants (Design Team) as necessary to review the mock-up.
38		Contractors shall be prepared to answer questions on materials and methods as necessary.
39	В.	The Contracting and Design Teams shall review the mockup in detail for materials, methods, and workmanship
40		with respect to the intent of the contract documents. Improvements or adjustments shall be discussed as
41		needed.
42	С.	If the mockup is incomplete or does not show sufficient detail of products and workmanship the General
43		Contractor shall resubmit a new mockup.

WARNER PARK COMMUNITY RECREATION CENTER EXPANSION CONTRACT #9502 MUNIS #17196

1		D.	Re-subr	mittal of mockups to meet the intent of the contract documents shall be the responsibility of the General
2			Contrac	ctor. No Change Orders will be processed for additional time or materials associated with re-submitting a
3			mockup	o for approval.
4			1.	In the event that a submitted mockup meets the criteria of the contract documents but does not meet
5				the expectations of the design team and alternative methods or materials are discussed the following
6				procedure shall be used:
7				a. Project Architect shall publish a Construction Bulletin (CB) to detail the required/recommended
8				changes.
9				b. The GC shall prepare and submit a new mockup.
10				
11	3.4.	FINAL	SUBMIT	TAL
12		A.	The fiel	d approved mockup shall be submitted by the General Contractor as any other submittal for project
13			docume	entation purposes. The mockup submittal shall consist of the following:
14			1.	Digitally photograph the field approved mockup. Take as many detailed photos as necessary to capture
15				the complexity of the mockup.
16			2.	Provide a written summary of the approved mockup. Include all recommended adjustments, level of
17				expected workmanship, and other such detail as discussed during the mockup review.
18			3.	Submit the mockup to the Project Management Web Site. See Specification 01 33 23 Submittals for
19				additional information.
20				
21				
22				
23				END OF SECTION
24				

1				SECTION 01 43 50	
2				AIR BARRIER SYSTEMS	
5 4	PΔRT	1 – H	FADING 1		1
5		1.1.	RELATE	D DOCUMENTS	1
6		1.2.	SUMMA	ABY	1
7	-	1.3.	DEFINIT	IONS	1
8		1.4.	PERFOR	MANCE REQUIREMENTS.	1
9		1.5.	SUBMIT	TAIS	2
10		1.6.	OUALIT	Y ASSURANCE	2
11		1.7.	PROIFC	T CONDITIONS	2
12	PART	2 – P		– NOT LISED	- २
13	PART	3 - FX	FCUTION		3
14		3.1.	FIFIDO		3
15		3.2	REPAIR	AND PROTECTION	4
16		J. <u>L</u> .	11217111		Ì
17	PART	1-н	FADING	1	
18	<u>. /</u>			<u>-</u>	
19	1.1.	RFI	ATED DO		
20		Α.	Draw	ings and general provisions of the Contract, including General and Supplementary Conditions and Division	
21		7	01 Sp	ecification Sections. Division 07 Specification Sections, apply to this Section.	
22			01 00		
22	12	SUI	MMARY		
23	1.2.	Δ	Contr	actor will engage a qualified consultant(s) to perform tests and inspections prior to the installation of air	
27		л.	barrie	actor win engage a quained consultant(s) to perform tests and inspections prior to the installation of an	
25		в	This s	ection includes administrative and procedural requirements for accomplishing an airtight building	
20		Б.	onclo	sure that controls infiltration or evfiltration of air	
27		c	Relat	ad Sactions:	
20		C.	1	Section 07 2E 00: Weather Parriers	
29			1. 2	Section 07 25 00. Wedner Barriers.	
50 21			۷.	siziant huilding anglesure, suctomized fabrication and installation procedures, not production of	
27				an tight building enclosure, customized rabilitation and installation procedures, not production of	
52 22				stanualu products.	
33 24	1 2	DE		N N N N N N N N N N N N N N N N N N N	
24 25	1.5.			> prrier System: The airtight components of the building enclosure and the joints, junctures and transitions	
35		А.	All Do	ane system. The antight components of the building enclosure and the joints, junctures and transitions	
30 27		р	Sorviv	cert indicide coordination between the trades, the proper scheduling and sequencing of the work, pro-	
27 20		ь.	const	ruction mostings, inspections, tests, and related actions, including reports performed by Contractor, by	
20			indor	ruction meetings, inspections, tests, and related actions, including reports performed by contractor, by	
40 29			norfo	rendent agencies, and by governing autionties. They do not include contract enforcement activities	
40			perio	Theo by Architect.	
41 42	1 4	DEE			
42 12	1.4.		Gono	ICE REQUIREMENTS	
45 11		А.	Gene	The perior mance. The contractor shall ensure that the intent of constituting the building enclosure with a pueue air barrier system to control air leakage into an out of the conditioned space is achieved. The air	
44 45			borrie	noous all barrier system to control all reakage into, of out of the conditioned space is achieved. The all	
45			Darrie	It shall be continuous, with all ininte cooled	
40			1. ว	It shall be continuous, with an joints seared.	
47 10			Ζ.	it shall be structurally supported to withstand positive and negative all pressures applied to the building	
40			2	Continuity of the air barrier materials and products with joints to provide complete accomplies	
49 E0			⊃. ⊿	Continuity of the an barrier materials and products with joints to provide complete assemblies.	-
50			4.	continuity of all the enclosure assemblies with joints and transition materials to provide a whole building	5
57 51		P	Conn	an parrier system.	
52		в.	Lonn 1	Ecundation and walk	
55			1.	roundation dhu Walls. Walls and windows as doors	
54 FF			2.	Walls and windows of doors.	
55			3. ¢	Different wan systems.	
50			4.	Wall and roof.	
5/			5.	wall and root over unconditioned space.	
58			ь.	wails, noor and root across construction, control and expansion joints.	
	\M/ A F		PARK COM	MUNITY RECREATION CENTER	

1			7. Walls, floors and roof to utility, pipe and duct penetrations.			
2		C.	Air Barrier Penetrations: All penetrations of the air barrier and paths of air infiltration / exfiltration shall be made			
3			air-tight.			
4		D.	Compliance Requirements:			
5			1. Assemblies: an air permeance not to exceed 0.03 cfm/ft2p under a pressure differential of 0.3 in. water			
6			(1.57psf) (0.15 L/s.m2 @ 75 Pa) when tested in accordance with ASTM E 1677.			
7			2. Materials: Materials used for the air barrier system in the opaque envelope shall have an air permeance			
8			not to exceed 0.004 cfm/ft2 under a pressure differential of 0.3 in. water (1.57psf) (0.02 L/s.m2 @ 75 Pa)			
9			when tested in accordance with ASTM E 2178. Or.			
10			3. Entire Building: The air leakage of the entire building shall not exceed 0.15 cfm/sf under a pressure			
11			differential of 0.3 in. water (1.57 nsf) (0.75 L/s m2 @ 75 Pa) when tested according to ASTM F 779.			
12						
13	1.5.	SUBM	μτται ς			
14		Δ	Field quality-control reports			
15		R.	Testing agency shall submit a certified written report in duplicate of each inspection test or similar service to			
16		Б.	the Architect of the Contractor is responsible for the service submit a certified written report in duplicate of			
17			and inspection that or similar service through the Contractor			
10			2. Submit additional conies of each written report directly to the governing authority, when the authority co			
10			1. Submit additional copies of each written report directly to the governing authority, when the authority so			
19		C	Ulleus. Depart Date: Written reports of each increation test or similar convice include, but are not limited to the			
20		C.	Report Data: written reports of each inspection, test, or similar service include, but are not limited to, the			
21			Tollowing:			
22			1. Date of issue.			
23			2. Project title and number.			
24			3. Name, address, and telephone number of testing agency.			
25			4. Dates and locations of samples and tests or inspections.			
26			5. Names of individuals making the inspection or test.			
27			6. Designation of the Work and test method.			
28			7. Identification of product and Specification Section.			
29			8. Complete inspection or test data.			
30			9. Test results and an interpretation of test results.			
31			Ambient conditions at the time of sample taking and testing.			
32			11. Comments or professional opinion on whether inspected or tested Work complies with Contract			
33			Document requirements.			
34			12. Name and signature of laboratory inspector.			
35			13. Recommendations on retesting.			
36						
37	1.6.	QUAL	ITY ASSURANCE			
38		Α.	General Performance: The Contractor shall ensure that the intent of constructing the building enclosure with a			
39			continuous air barrier system to control air leakage into, or out of the conditioned space is achieved. The air			
40			barrier system shall have the following characteristics:			
41		В.	Inspection and testing services are required to verify compliance with requirements specified or indicated. These			
42			services do not relieve Contractor of responsibility for compliance with Contract Document requirements.			
43			1. Qualifications for Air Barrier Testing and Inspection Agencies: Engage Air Barrier inspection and testing			
44			service agencies, including independent testing laboratories, that are prequalified and that specialize in			
45			the types of air barrier system inspections and tests to be performed.			
46		C.	Specific quality-control requirements for individual construction activities are specified in the sections of the			
47			specifications. Requirements in those sections may also cover production of standard products. It is the			
48			Contractor's responsibility to ensure that each subcontractor is adequately and satisfactorily performing the			
49			guality assurance documentation, tests and procedures required by each section.			
50		D	Specified inspections, tests, and related actions do not limit Contractor's quality-control procedures that			
51			facilitate compliance with Contract Document requirements.			
52						
53	1.7.	PROIF				
54		A.	Contractor Responsibilities: Unless otherwise indicated as the responsibility of another identified entity			
55			Contractor shall provide coordination of the trades, and the sequence of construction to ensure continuity of the			
56			air harrier system joints junctures and transitions between materials and assemblies of materials and products			
57			from substructure to walls to roof. Provide quality assurance procedures, testing and verification as specified			
58			herein. Facilitate inspections, tests, and other quality-control services specified elsewhere in the Contract			

1			Docur	ments a	nd required by authorities having jurisdiction or by the Owner. Costs for these services are included
2			in the	Contra	ict Sum.
3		В.	Organ	ize preo	construction meetings between the trades involved in the whole building's air barrier system to
4			discus	ss where	e each trade begins and ends and the responsibility and sequence of installation of all the air-tight
5			joints,	, junctu	res, and transitions between materials, products and assemblies of products specified in the
6			differe	ent sect	tions, to be installed by the different trades.
7		C.	Build a	a mock-	-up before proceeding with the work, satisfactory to the Architect, of each airtight joint type,
8			junctu	ure, and	transition between products, materials and assemblies.
9		D.	Associ	iated Se	ervices: Cooperate with agencies performing required inspections, tests, and similar services, and
10			provic	de reaso	onable auxiliary services as requested. Notify the agency sufficiently in advance of operations to
11			permi	it assign	iment of personnel. Auxiliary services required include, but are not limited to, the following:
12			1.	Provic	de access to the Work.
13			2.	Furnis	sh incidental labor and facilities necessary to facilitate inspections and tests.
14			3.	Take a	adequate quantities of representative samples of materials that require testing or assist the agency
15			_	in taki	ing samples.
16			4.	Delive	er samples to testing laboratories.
17		_	5.	Provid	de security and protection of samples and test equipment at the Project Site.
18		Ε.	Duties	s of the	Testing and Inspection Agency: The independent agency engaged to perform inspections, sampling,
19			and te	esting of	t air barrier materials, components and assemblies specified in individual Sections shall cooperate
20			with t	he Arch	nitect and the Contractor in performance of the agency's duties. The testing agency shall provide
21			qualifi	ied pers	sonnel to perform required inspections and tests.
22			1.	The a	gency shall notify the Architect and the Contractor promptly of irregularities or deficiencies
23			2	obser	ved in the Work during performance of its services.
24			2.	The ap	gency is not authorized to release, revoke, alter, or enlarge requirements of the Contract
25			2	Docur	ments or approve or accept any portion of the Work.
26		-	3. Canad	The ag	gency shall not perform any duties of the Contractor.
27		۲.	Coord	lination	.: Coordinate the sequence of activities to accommodate required services with a minimum of delay.
28			Coord	linate ad	ctivities to avoid the necessity of removing and replacing construction to accommodate inspections
29			and te	ests.	
30			1.	The Co	tion tractor is responsible for scheduling times for inspections, tests, taking samples, and similar
31 22				activit	ues.
33	PART	2 – PRC	DUCTS	S – NOT	USED
34	<u></u>				
35	PART	3 - EXE	CUTION	1	
36		-		-	
37	3.1.	FIELD	QUALIT	TY CON	TROL
38		Α.	Testin	ig Agen	cy: Contractor will engage a qualified testing agency to perform tests and inspections.
39		В.	Tests	and Ins	pections:
40			1.	Qualit	, tative Testing and Inspection:
41				a.	Daily reports of observations, with copies to the Owner, Contractor and Architect.
42				b.	Continuity of the air barrier system throughout the building enclosure with no gaps, holes.
43				c.	Structural support of the air barrier system to withstand design air pressures.
44				Ч	Masonry and concrete surfaces are smooth, clean and free of cavities, protrusions and mortar
45				u.	
46				u.	droppings, with mortar joints struck flush, or as required by the manufacturer of the air barrier
				u.	droppings, with mortar joints struck flush, or as required by the manufacturer of the air barrier material.
47				e.	droppings, with mortar joints struck flush, or as required by the manufacturer of the air barrier material. Site conditions for application temperature and dryness of substrates.
47 48				e. f.	droppings, with mortar joints struck flush, or as required by the manufacturer of the air barrier material. Site conditions for application temperature and dryness of substrates. Maximum length of exposure time of materials to ultra-violet deterioration.
47 48 49				e. f. g.	droppings, with mortar joints struck flush, or as required by the manufacturer of the air barrier material. Site conditions for application temperature and dryness of substrates. Maximum length of exposure time of materials to ultra-violet deterioration. Surfaces are properly primed.
47 48 49 50				e. f. g. h.	droppings, with mortar joints struck flush, or as required by the manufacturer of the air barrier material. Site conditions for application temperature and dryness of substrates. Maximum length of exposure time of materials to ultra-violet deterioration. Surfaces are properly primed. Laps in material are 2" minimum, shingled in the correct direction (or mastic applied on exposed
47 48 49 50 51				e. f. g. h.	 droppings, with mortar joints struck flush, or as required by the manufacturer of the air barrier material. Site conditions for application temperature and dryness of substrates. Maximum length of exposure time of materials to ultra-violet deterioration. Surfaces are properly primed. Laps in material are 2" minimum, shingled in the correct direction (or mastic applied on exposed edges), with no fishmouths.
47 48 49 50 51 52				e. f. g. h.	 droppings, with mortar joints struck flush, or as required by the manufacturer of the air barrier material. Site conditions for application temperature and dryness of substrates. Maximum length of exposure time of materials to ultra-violet deterioration. Surfaces are properly primed. Laps in material are 2" minimum, shingled in the correct direction (or mastic applied on exposed edges), with no fishmouths. Mastic applied on cut edges.
47 48 49 50 51 52 53				е. f. g. h. i. j.	 droppings, with mortar joints struck flush, or as required by the manufacturer of the air barrier material. Site conditions for application temperature and dryness of substrates. Maximum length of exposure time of materials to ultra-violet deterioration. Surfaces are properly primed. Laps in material are 2" minimum, shingled in the correct direction (or mastic applied on exposed edges), with no fishmouths. Mastic applied on cut edges. Roller has been used to enhance adhesion.
47 48 49 50 51 52 53 54				е. f. g. h. j. k.	 droppings, with mortar joints struck flush, or as required by the manufacturer of the air barrier material. Site conditions for application temperature and dryness of substrates. Maximum length of exposure time of materials to ultra-violet deterioration. Surfaces are properly primed. Laps in material are 2" minimum, shingled in the correct direction (or mastic applied on exposed edges), with no fishmouths. Mastic applied on cut edges. Roller has been used to enhance adhesion. Measure application thickness of liquid-applied materials to manufacturer's specifications for the
47 48 49 50 51 52 53 54 55				е. f. g. h. j. k.	 droppings, with mortar joints struck flush, or as required by the manufacturer of the air barrier material. Site conditions for application temperature and dryness of substrates. Maximum length of exposure time of materials to ultra-violet deterioration. Surfaces are properly primed. Laps in material are 2" minimum, shingled in the correct direction (or mastic applied on exposed edges), with no fishmouths. Mastic applied on cut edges. Roller has been used to enhance adhesion. Measure application thickness of liquid-applied materials to manufacturer's specifications for the specific substrate.
47 48 49 50 51 52 53 54 55 56				е. f. g. h. j. k.	 droppings, with mortar joints struck flush, or as required by the manufacturer of the air barrier material. Site conditions for application temperature and dryness of substrates. Maximum length of exposure time of materials to ultra-violet deterioration. Surfaces are properly primed. Laps in material are 2" minimum, shingled in the correct direction (or mastic applied on exposed edges), with no fishmouths. Mastic applied on cut edges. Roller has been used to enhance adhesion. Measure application thickness of liquid-applied materials to manufacturer's specifications for the specific substrate. Materials used for compatibility.

1				n.	Connections between assemblies (membrane and sealants) for cleaning, preparation and priming
2					of surfaces, structural support, integrity and continuity of seal.
3				о.	All penetrations sealed.
4			2.	ASTM	I E 1186/98 "Standard Practices for Air Leakage Site Detection in Building Envelopes and Air
5				Retai	uer systems.
0				a. L	Infrared scanning with pressurization/depressurization.
/				D.	Smoke pencil with pressurization/depressurization.
8				с.	Pressurization/depressurization with use or anemometer.
9				d.	Generated sound with sound detection.
10				e.	Tracer gas measurement of decay rate.
11				t.	Chamber pressurization/depressurization in conjunction with smoketracers.
12				g.	Chamber depressurization using detection liquids.
13			3.	Quan	titative Tests: Provide written test reports of all tests performed, with copies to the Owner,
14				Contr	actor and Architect.
15				a.	Material compliance for maximum air permeance, ASTM E 2178.
16				b.	ASTM E 283, Determining rate of Air Leakage Through Exterior Windows, Curtain Walls, and Doors
17					under Specified Pressure Differences Across the Specimen.
18				с.	Assemblies, ASTM E 1677, test pressure and allowable air leakage rate to be determined by design
19					professional for interior design conditions and location of project.
20				d.	CAN/CGSB 1986 Standard 149.10, Determination of the Airtightness of Building Envelopes by the
21					Fan Depressurization Method.
22				e.	CAN/CGSB 1996 Standard 149.15 Determination of the Overall Envelope Airtightness of Office
23					Buildings by the Fan Depressurization Method Using the Building's Air Handling System.
24				f.	Canadian National Master Specification Sections 07272 Air Barrier Systems for Exterior Walls of
25					Low-Rise Buildings.
26				σ	Canadian National Master Specification 07272 1 · Durability Assessment of Bead-Applied
27				ъ.	Lirethane-Based Sealant Foam for Air Barriers
28				h	Whole building floors or suites ASTM F779 Determining Airtightness of Buildings Air Leakage
20					Rate by Single Zone Air Pressurization
20				i	Windows and connections to adjacent onaque assemblies ASTM E783
21				i. i	Tracer gas testing ASTM E741
22				ј. И	Droccure test ASTM E220
32 22				к. I	Pressure lest, ASTIVIESSU.
33				1.	Bond to substrate, ASTNI D4341-95.
34 25				m.	winimum dry of wet him thickness for liquid-applied materials are per the manufacturer's
35					requirements.
36					
37	3.2.	REPAI	RANDI	PROTE	
38		А.	Upon o	comple	tion of inspection, testing, sample taking and similar services, repair damaged construction and
39			restore	e subst	rates and finishes. Comply with Contract Document requirements for Division 1 Section "Cutting
40			and Pa	itching	
41		В.	Protec	t const	ruction exposed by or for quality-control service activities, and protect repaired construction.
42		C.	Repair	and p	rotection is Contractor's responsibility, regardless of the assignment of responsibility for inspection,
43			testing	g, or sir	nilar services.
44					
45					
46					
47					
48					END OF SECTION

			SECTION 01	45 16	
			FIELD QUALITY CONTRO	OL PROCEDURES	
	CFN				
PARII					
1 1 '	1. 3 2 1			,	
1.4	2. г Эг				
1.3	з. г л (REIVIEIN I 3		
1.4	4. (5 (
PART 2	_ PR(
PART 3	- FXF(UTION			
3.	1. (IT RESPONSIBILITIES		
3.2	2. F	ESPONDING TO A QIV	0		
3.3	3. (SENERAL CONTRACTO	RS FOLLOW-UP		
3.4	4. (QMO CLOSEOUT PROC	EDURE		
3.5	5. (ONSTRUCTION CLOSE	OUT		
PART 1	– GEI	IERAL			
1.1.	SUMI	MARY			
	A.	The City of Madison	has developed a multi-faceted Q	uality Management F	Program that begins with contract
		signing and runs thre	ough contract closeout to ensure	the best quality mat	erials, workmanship, and product a
		delivered for the cor	tracted Work.		
		1. The Project M	Aanagement Web Site is a Const	uction Management	tool that provides contractors and
		staff a single	on-line location for the daily ope	rations and progress	ion of the Work.
		2. The Quality N	Nanagement Observation (QMO)	is an ongoing observ	vation of the construction process a
		progresses.	The City of Madison does not use	a "Punch List" or "Co	orrections List" as it is typically know
		throughout t	he construction industry. The QI	vio process acts as ar	n "in progress punch list".
		a. By us	ng the QMO process the City of	Madison's goal is to h	have a zero item punch list prior to t
	_	90% p	progress payment and owner occ	upancy.	
	В.	All contractors shall	be required to review the specifi	cations identified in S	Section 1.2 below, and other related
		specifications identi	ied therein to become familiar w	ith the terminology a	and expectations of this City of
	~	Madison Public Wor	ks contract.		
	C.	Contractor (CC) Pro	specification to outline the requ	nrements, expectatio	she items of Quality Assurance and
		Contractor (GC), Pro	ject Architect, and other represe	ntatives of the Owne	r for items of Quality Assurance and
		1 This specific:	tion is not intended to conflict w	ith Specification 01 4	10.00 Quality Paguiraments or other
		specification	requiring testing and inspecting	sonvicos	to be quality requirements of other
		2 This specific:	tion does not relieve the GC from	n any requirements a	associated with regulatory inspection
		2. Inits specification of the	the City of Madison Building Ing	spection Unit or insp	ectors from other agencies as requi
		by code	y the city of Mudison Bunding in		cettors from other agencies as requi
		3. Any testing r	erformed by an Owner's Represe	entative does not reli	eve the GC from performing any
		testing that r	nay required by the construction	documents.	
		C C			
1.2.	RELA	ED SPECIFICATION SE	CTIONS		
	A.	Section 01 26 13	Request for Information (RFI)	
	в.	Section 01 29 76	Progress Payment Procedure	25	
	ι. Γ	Section 01 31 13	Project Coordination		
	ט. ב	Section 01 31 23	Project Management Web S	te (PIVIVVS)	
	с. Г	Section 01 77 00	Closeout Procedures		
	r. G	Section 01 78 12	Completion and Correction I	ict	
	с. Н.	Section 01 91 00	Commissioning	.136	
			0		
1.3.	PERF	RMANCE REQUIREM	ENTS	-	
	A.	All contractors shall	be responsible for a proper quali	ty assurance/quality	control (QA/QC) program throughou
		the execution of the	work defined within the constru	ction documents, inc	cluding all recognized construction
		industry standards a	nd all applicable regulatory code	S	
WARNE	R PARI	COMMUNITY RECREAT	ION CENTER		
EXPANS	ION	00 MILINIC #47402			
CONTRA	ACT #9	UZ IVIUNIS #17196	01 45 16 -	1	FIELD QUALITY CONTROL PROCEDU

1		В.	The GC shall be responsible for all of the following:	
2			1. Monitor the quality of all workmanship, supplies, materials, and products being installed	by all
3			contractors and installers to ensure they meet or exceed the minimum requirements set	forth by the
4			construction documents.	,
5			2 Submit a Request for Information (REI) whenever manufacturers' instructions or referen	ced standards
6			conflict with the construction documents before proceeding with the Work	
7			Ensure that Work requiring special certifications or licensing is being performed by is being	ng performed
, Q			and supervised by personnal that meet the appropriate requirements	ng performed
0			and supervised by personner that meet the appropriate requirements.	ha project
9		c	a. Ensure that all certificates and incenses are current till oughout the execution of t	ne project.
10		C.	The coin and its representatives shall perform quality assurance and quality control activities in 2000 products the CC of maintaining an according to 0000 pr	
11			execution of this project. This in no way relieves the GC of maintaining an acceptable QA/QC pr	ogram. =
12		.		
13	1.4.	QUAL	Y ASSURANCE	
14		А.	The GC shall be responsible for the following:	
15			 All materials, equipment, and products shall be new, clean, undamaged, and meet the p 	erformance
16			specifications defined within the construction documents including favorably reviewed s	ubmittals.
17			a. Any material, equipment, or product that does not meet the requirements of the	construction
18			documents shall be removed and replaced, including any adjacent and related w	ork, at the GCs
19			expense.	
20			2. All Work shall be performed by persons properly trained and/or qualified to produce wo	rkmanship of the
21			guality specified in the construction documents.	•
22			 Providing access to updated as-builts, addenda, submittals, bulletins and other related c 	onstruction
23			documents at the project site	
23		B	The CoM and its representatives may be responsible for any of the following:	
24		Б.	Attend are installation meetings	
25			Attend construction programs received	
26			2. Attend construction progress meetings	
27			3. Review all submittals	
28			 Conduct field visits for QA/QC purposes, provide feedback to the GC and sub-contractor. 	s using Quality
29			Management Observation (QMO) reports.	
30			5. Review delivered equipment	
31			6. Witness equipment installations, startups, testing as specified in other specifications	
32				
33	1.5.	QUAL	Y MANAGEMENT OBSERVATION REPORT	
34		Α.	The Quality Management Observation report or QMO is used as a QA/QC tool by those entities	responsible for
35			QA/QC activities, including but not limited to, the GC, CoM, Project Architect /Project Engineer(A/E PROJ MGR),
36			CX agent. etc.	
37		В.	OMOs are designed to be an early observation of non-conforming construction work before it b	ecomes buried
38		5.	ny follow on work As such it is most often used as an "in progress punch list"	
30		C	TMO forms are part of the Quality Control Library on the Project Management Web Site	
10		С.		
40	DADT	2 000		
41	PARI	<u> </u>	UCIS - THIS SECTION NOT USED	
42				
43	PART	3 - EXE	IIION	
44				
45	3.1.	QUAL	Y MANAGEMENT RESPONSIBILITIES	
46		Α.	Nhile making routine progress visits to the construction project the GC, CPM, CxA and A/E PRO.	MGR, and
47			applicable others shall observe the details of the construction and installations to ensure that th	e intent of the
48			construction documents is being followed.	
49		В.	f during the progress visit there is a determination of contract non-conformance a QMO report	shall be initiated
50			o begin the documentation process.	
51			1. The GC field superintendent shall be informed immediately of any issue that may cause	harm, damage to
52			finished work, or be buried prior to properly filing a OMO report.	,
53		C	The following information when filing a OMO report:	
50		с.	Open a OMO report in the Project Management Web Site	
55			E. Open a give report in the roject management web site	
55			 Litter the date drive the original documents if any foregraphic and if institute the structure of the structure	ana dataila
00 57			 Provide references to construction documents if any (examples; specification, drawing p enserved submittele .PEL CPL stal) 	age, uetalls,
5/			approved submittals, KFI, CB, etc)	
58			4. Provide a short title for the observation being made	
-	WARN	ER PAR	OMMUNITY RECREATION CENTER	

1			5.	Provide a detailed description of the observation being made
2			6.	Select all categories (Sitework, Structure, Enclosure, Interior, etc) from the given list that may apply to
3				the observation being reported.
4 5				a. For each category selected additional boxes shall open with contractor names associated with each category.
6			7.	Select all contractors from the lists provided that may need to be aware of the observation.
7			8.	Provide any attachments that may help provide reference to the observation.
8		D.	The sof	tware for the Project Management Website will email notifications that a QMO report has been initiated.
9				
10	3.2.	RESP	ONDING 1	ΤΟ Α QMO
11		Α.	The GC	shall be responsible for determining the course of action required to remedy the non-conforming issue
12			and sha	all coordinate and direct the contractor(s) responsible for any work related to the observation.
13		в.	All cont	tractors assigned to remedy the observation by the GC shall provide follow-up responses
14			1.	Open the QMO report in the Project Management Web Site.
15			2.	Enter a description of your follow-up response in the box provided.
16			3.	Add attachments (pictures) if needed to show the work has been completed.
17				
18	3.3.	GENE	RAL CON	ITRACTORS FOLLOW-UP
19		А.	The GC	shall inspect the work to ensure that all assigned contractors have remedied the observation to the
20			intent c	of the construction documents.
21		В.	The GC	shall respond with any additional comments in their response box.
22				
23	3.4.	QMO	CLOSEOU	UT PROCEDURE
24		А.	The per	rson who initiated the QMO shall review the remedied work and if properly corrected shall close and date
25			the QM	IO form.
26			1.	In the event there are still issues the Quality Manager can add additional comments in the response area,
27		_	• • •	and re-issue the QMO for additional review as needed.
28		в.	Once th	he person who initiated the QMO has closed the item the CPM shall review and verify with the A/E PROJ
29			MGR th	hat the Observation has been properly remedied and provide final closure on the QMO.
30	25	CONG	TRUCTIO	
31	3.5.			IN CLOSEOUI
32 22		А.	1 Ine GC	Shall note that successful close out Qivios are required for construction closeout as follows.
33 24			1.	reports playments as identified in specification of 29.76 are contingent Qivio reports being
34 25			r	property closed out.
32			۷.	Specification of 77 of defines an construction closeout requirements.
27				
38				
30				
40				
-0				

			SECTION 01 45 29
			TESTING LABORATORY SERVICES
PART	1 – G	ENERAL .	
1	L.1.	REQUIE	
1	L.2.	RELATE	D REQUIREMENTS
1	L.3.	QUALIF	
1	L.4.	LABOR/	
1	L.5.	LIMITA	TIONS OF AUTHORITY OF TESTING LABORATORY
1	L.6.	CONTR	
1	L./.	SPECIFI	C LEST, INSPECTIONS, AND METHODS REQUIRED
PART	2 – PI	RODUCTS	5 – THIS SECTION NOT USED
PARI	3 – EX	KECUTIO	N – THIS SECTION NOT USED
PART	1 – G	ENERAL	
1 1	REC		
1.1.		Tho (INTO INCLUDED Contractor shall amploy and nay for the services of an independent testing laboratory to perform specific
	A.	convi	contractor shall employ and pay for the services of an independent testing laboratory to perform specific
	р	Tocti	Les difu lesting.
	Б.	1	Soction 02 20 00: Cact In Place Concrete
		1. 2	Section 05 30 00. Cast-III-Flace Coliciele
		2.	Section 05 12 00. Sci uccurat Steel Framing
		⊃. ⊿	Section 21 20 00: Earthwork
		4.	Section SI 20 00. Earthwork
12	RFI	ATED RE	OURFMENTS
1.2.		Conc	ditions of the Contract: Inspections and testing required by laws, ordinances, rules, regulations, orders or
	71.	annr	ovals of nublic authorities
	B	Relat	ted Beguirements Specified in Other Sections:
	υ.	1	Division 22 and 23: Testing of Mechanical Systems
		2	Division 26: Testing of Electrical Systems
		2.	
1.3.	QU	ALIFICAT	ION OF LABORATORY
	A.	Mee	t "Recommended Requirements of Independent Laboratory Qualification" published by American Counc
		Inde	pendent Laboratories.
	В.	Mee	t basic requirements of ASTM E 329, "Standards of Recommended Practice for Inspection and Testing
		Agen	icies for Concrete and Steel as Used in Construction."
	C.	Auth	orized to operate in State in which the Project is located.
		00470	
1.4.		GORATOR	(Y DUTIES
	A. D	Dorfe	per a consistent of the second section of the second section of the second section of the second sections of the second section of the section of the second section of the section
	р.	1	Comply with specified standards
		1. ว	Comply with specified standards.
	~	Z. Draw	Ascertain compliance of materials with requirements of contract Documents.
	С. Б	Prom	iptly notify the Owner, A/E and Contractor of observed irregularities or deficiencies of work or products.
	D.	Prom	iptly submit written report of each test and inspection; one copy each to A/E, Consulting Engineer, Owne
		and	Contractor. Each report shall include:
		1.	Date issued.
		2.	Project litle and number.
		3.	resting laboratory name, address and telephone number.
		4.	Name and signature of laboratory inspector.
		5.	Date and time of sampling or inspection.
		6.	Record of temperature and weather conditions.
		7.	Date of test.
		8.	Identification of product and specification section.
		9.	Location of sample or test in the Project.
		10.	Type of inspection or test.

1		-	12. Derfer	Interpretatio	n of test results, when requested by A/E or the Contractor.					
2		E.	Perfor	m additional t	ests as required by Owner, A/E or the Contractor.					
4	1.5.	LIMITATIONS OF AUTHORITY OF TESTING LABORATORY								
5		A.	Labora	atory is not au	thorized to:					
6			1.	Release. revo	oke, alter, or enlarge on requirements of Contract Documents.					
7			2.	Approve or a	ccept any portions of the Work other than those portions of the Work scheduled for testing.					
8			3.	Perform any	duties of the Contractor.					
9				,						
10	1.6.	CONT	RACTOF	R'S RESPONSIE	BILITIES					
11		Α.	Coope	erate with labo	ratory personnel, provide access to Work and to manufacturer's operations.					
12		В.	Secure	e and deliver to	o the laboratory, adequate quantities of representative samples of materials proposed to be					
13			used a	and which requ	ire testing. Submit concrete mix designs to A/E for approval prior to pouring concrete.					
14		C.	Provid	le to the labor	atory the preliminary design mix proposed to be used for concrete, and other material mixes					
15			that re	equire control	by the testing laboratory.					
16		D.	Furnis	h copies of Pro	oduct test reports as required.					
17		Ε.	Furnis	h incidental la	bor and facilities:					
18			1.	To provide a	ccess to Work to be tested.					
19			2.	To obtain an	d handle samples at the Project site or at the source of the product to be tested.					
20			3.	To facilitate i	nspections and tests.					
21			4.	For storage a	nd curing of test samples.					
22		F.	Notify	laboratory su	fficiently in advance of operations to allow for laboratory assignment of personnel and					
23			schedu	uling of tests.						
24		G.	Make	arrangements	with laboratory and pay for additional samples and tests required for Contractor's					
25			conve	nience.						
26		Н.	Emplo	y and pay for t	the services of a separate, equally qualified independent testing laboratory to perform					
27			additio	onal inspectior	ns, sampling and testing required when initial tests indicate work does not comply with					
28			Contra	act Documents						
29		Ι.	Tempo	orarily halt the	progress of the Work when tested materials do not comply with Contract Documents and					
30			promp	otly notify the	Owner or their designated representative and A/E.					
31		J.	Remov	ve and replace	at no cost to the Owner, all defective materials discovered upon testing not to comply with					
32			Contra	act Documents	, including cost for retesting and re-inspecting replaced Work that failed to comply with the					
33			Contra	act Documents						
34		CDECU								
35	1.7.	SPECII	FIC TEST	I, INSPECTION	S, AND METHODS REQUIRED					
36		А.	Sectio	n 03 30 00: Ca	ast-In-Place Concrete					
3/			1.	Secure samp	le of aggregates Contractor proposes to use and test for compliance with Specifications.					
38			2.	Certify comp	liance with Specifications of cement proposed for use by the Contractor.					
39			3.	Review and a	ipprove the Contractor's proposed concrete mix proportions for the required concrete					
40				strengths usi	ng materials Contractor proposed to use on the project. Incorporate specified admixtures					
41 42			4	Borform ann	than amounts of cement specified.					
42 12			4.	substantiato	mix decigns					
45 11			5	Inspect and t	nink designs.					
44 15			Э.	requirement						
46				a Testir	α.					
40 47				i i	5. Sample and test concrete in accordance with ASTM C 31 ASTM C 143 ASTM C 172 and					
48					ASTM C 231					
49				ii	Perform slump tests in accord with ASTM C 143 from same concrete batch used for test					
50					cylinders and record results and comments on compression test reports.					
51				iii.	Perform compression tests in accordance with ASTM C39.					
52				iv.	When air-entrained concrete is used, a minimum of one (1) air content test shall be					
53					performed in accordance with ASTM C 231 for each set of test cylinders taken.					
54				v.	Identify all test cylinders with symbols to indicate location on the job where concrete test					
55					was made. Record on project record drawings.					
56				vi.	Strength tests shall be made for: each day's pour; each class of concrete; each change of					
57					supplies or sources; and for each 100 cubic yards of concrete or fraction thereof.					

1			v	ii. One slump test shall be made for each set of test cylinders taken following the procedure
2				in ASTM C 143.
3			b. T	est Cylinders for all Concrete
4			i.	Each test shall consist of a minimum of four cylinders.
5			ii	. Make test cylinders in conformity with ASTM C 31.
6 7			ii	 After 24 hours three cylinders to be carefully transported to the testing laboratory for moisture curing and one cylinder to be field cured.
8			iv	v. One field cured cylinder to be tested at 7 days and two laboratory cured cylinders to be
9				tested at 28 days. Reserve one cylinder for further testing
10			v	The average of all strength tests representing each class of concrete as well as the average
11			•	of any three consecutive strength tests for each class of concrete, shall be equal to or
12				greater than the specified strength
12				i If the A/E bac reason to believe that cylinder strength tests are not representative of the
14			v	the strength of congrete in place. A/E shall require drilled cores to be suit and tested at the
14 1 F				Screngen of concrete in place, A/E shall require unlied cores to be cut and tested at the
15				Contractor's expense. Coring and testing shall be in accordance with ASTW C 42 standard
16		6		Wethod of Obtaining and Testing Drilled Cores and Sawed Beams of Concrete.
1/	В.	Sectio	on 05 12 0	J: Structural Steel Framing
18		1.	Welding	
19			a. P	Provide inspection of shop and field welding in accordance with Section 6 of AWS D1.1.
20			b. V	/isually inspect all welds, perform appropriate non-destructive tests on apparent defective welds.
21			V	/erify conformance with Specifications.
22			c. N	Ion-destructive testing shall be performed on 20 percent of the total length of all full penetration
23			v	velds. If a sufficient number of welds are deficient, additional testing may be performed at the
24			d	liscretion of the testing lab, at no cost to Owner.
25		2.	Bolting:	
26			a. V	/isually inspect all connections for proper number, size and type of bolt.
27			b. R	leview all bolted connections for compliance with "snug tight" requirements of AISC.
28			c. N	Io Slip-critical (SC) connections/bolts are required for this project.
29			d. S	hear Connectors, Headed/Deformed Bar Concrete Anchors:
30			i.	. Verify pre-production test records for installation of shear connectors, concrete anchors
31				and threaded studs.
32			ii	Shear connectors shall be struck with a hammer. Those not producing a "clean" pinging
33				sound indicative of a fully attached shear connector shall be bent 15 degrees off vertical
34				towards the nearest support by striking with a hammer. If shear connector does not
35				become loose and weld is not broken, it shall be considered acceptable, and shall be left in
36				the bent position. Replace failing shear connectors and test as before.
37			ii	i. A visual inspection shall be made of shear connectors and headed/deformed bar concrete
38				anchors after installation. If visual inspection reveals that a sound weld and a 360 degree
39				flash has not been obtained, the connector/anchor shall also be tested by bending a
40				minimum of 15 degrees off vertical opposite to the missing weld/flash, irrespective of the
41				results of the "ping" test required for shear connectors. If the connector/anchor does not
42				become loose it shall be considered acceptable and shall be left in this position. Replace
43				failing connector/anchors and inspect as before.
44	C.	Sectio	on 05 40 0	0: Cold Formed Steel Framing
45		1.	As direct	ted by A/E. Contractor's testing agency may inspect the maintenance of a quality control program
46			including	g spot checking weldments and welding procedures in accordance with AWS standards.
47	D.	Sectio	n 31 20 0	0: Soil Compaction Control and Trenching and Backfilling
48	2.	1	Soils Eng	vineer to be onsite during excavation operation
49		2	Visually	inspect test and certify that exposed undisturbed underlying soil is suitable for required footing
50		2.	hearing	capacity and placement of fills
51		З	Maximu	m and minimum density of fill soil for compaction percentage of relative density and moisture
52		э.	density	shall be determined in accordance with ASTM Designation D 1557 Testing agancy will test
52			compact	tion of soils in place according to ASTM D 1556 ASTM D 2167 ASTM D 2022 and ASTM D 2027
55			acapplic	non or sons in place according to Astriv D 2550, Astriv D 2107, ASTRI D 2522, dilu ASTRI D 2937, Ashla
54		4	as applic	,aure.
55		4.		ur icesis as runuws.
30 57			d. S	augraue, ondisturbed and Demonition Surfaces. Visual inspection and prope; test if required.
5/ E0			υ. II ο Γ	Interior Fills. One test per 2,500 sq. it for each two foot or less IIIT.
50			ι. Ε	גוניוטו דוווג. טוופ נפגר אפר 2,500 גע. וו וטר פמטו נאט דטסג סר ופגג וודג.
	WARNER PARI	ксомм		REATION CENTER

1		d.	Utility Trenches: One test per 50 lineal feet for each two foot or less lift.
2 3 4	PART 2 - PRODUCTS -	- THIS	SECTION NOT USED
5			
6 7	PART 3 – EXECUTION	– THIS	SECTION NOT USED
8			
9			END OF SECTION

1				SECTION 01 50 00
2				TEMPORARY FACILITIES AND CONTROLS
3				
4	PART	1 – G	NERAL	
5	1	1.		
6	1	2.	RELATED SPECIFIC	ATION SECTIONS
7	1	3.	QUALITY ASSURAN	VCE
8	1	4.	TEMPORARY UTILI	ITIES 2
9	1	5.	TELECOMMUNICA	ATIONS SERVICES AND WI-FI
10	1	6.	TEMPORARY SANI	ITARY FACILITIES
11	1	7.	BARRIERS	
12	1	8.	FENCING	
13	1	9.	EXTERIOR ENCLOS	SURES
14	1	10.	SECURITY	
15	1	11.	VEHICULAR ACCES	SS AND PARKING
16	1	12.	WASTE REMOVAL	
17	1	13.	PROJECT IDENTIFI	CATION
18	1	14.	FIELD OFFICES	
19	PART	2 - PR	DUCTS	
20	2	2.1.	TEMPORARY PART	TITIONS
21	2	2.2.	EQUIPMENT	
22	PART	3 - EX	CUTION	
23	3	8.1.	TEMPORARY FIRE	PROTECTION
24	3	3.2.	COLLECTION AND	DISPOSAL OF WASTE
25	Э	3.3.	ENVIRONMENTAL	. PROTECTION
26	Э	3.4.	REMOVAL OF TEM	/PORARY UTILITIES, FACILITIES, AND CONTROLS
27				
29 30	1.1.	SUN	MARY	
31		А.	I his Section incl	ludes general procedural requirements for temporary facilities and controls including, but not
32			limited to the fo	Silowing:
33			1. Tempora	ary utilities
34 25			2. Telecom	Imunications Services
35			3. Tempora	ary Sanitary Facilities
36			4. Barriers	
37			5. Fencing	
38			6. Exterior	Enclosures
39			7. Security	
40			8. Vehicula	ar Access and Parking
41			6. Waste R	temoval
12			7. Project l	Identification
13			8. Field Off	fices
14				
15	1.2.	REL	TED SPECIFICATIO	ON SECTIONS
46		Α.	Section 01 31 1	9 Progress Meetings
47		В.	Section 01 31 23	3 Project Management Web Site
48		C.	Section 01 74 19	9 Construction Waste Management and Disposal
49				
50	1.3.	QU	LITY ASSURANCE	
51		Α.	Regulations: Co	mply with industry standards and applicable laws and regulations if authorities having
52			jurisdiction, incl	luding but not limited to:
53			1. Building	; Code requirements
54			2. Health a	and safety regulations
55			3. Utility co	ompany regulations
56			4. Police. F	Fire Department and Rescue Squad rules
57			5. Environi	mental protection regulations
58			6. Joint Co	mmission - Hospital Accreditation Standards
-				

WARNER PARK COMMUNITY RECREATION CENTER EXPANSION

1 2 3 4 5		В. С.	Standards: Comply with NFPA 241 "Standard for Safeguarding Construction, Alterations, and Demolition Operations," ANSI A10 Series standards for "Safety Requirements for Construction and Demolition," and NECA Electrical Design Library "Temporary Electrical Facilities". Electrical Service: Comply with NEMA, NECA, and UL standards and regulations for temporary electric service. Install service in compliance with NFPA 70 "National Electric Code".
6			
7	1.4.	TEMP	ORARY UTILITIES
8		Α.	Contractor will provide the following:
9			 Electrical power and metering, consisting of existing facilities.
10			2. Water supply, consisting of existing facilities.
11		В.	General:
12			1. Existing facilities may be used.
13			2. New permanent facilities may be used.
14		C.	Water Service: water is available from existing building services.
15			 Use trigger-operated nozzles for water hoses, to avoid waste of water.
16		D.	Temporary Electric Power Service: Electrical Contractor shall extend temporary power from existing building
17			services.
18		Ε.	Temporary Lighting: Electrical Contractor shall provide temporary lighting with local switching
19 20 21			 Install and operate temporary lighting, minimum of 30 fc, to fulfill security and protection requirements, without operating the entire system, and will provide adequate illumination for all areas of work, including construction operations and traffic conditions.
21		F	Temporary Heat: General Contractor shall provide temporary heat required by construction activities for curing
22		1.	or drying of completed installations or protection of installed construction from adverse effects of low
23			temperatures or high humidity. Select safe equinment that will not have a harmful effect on completed
27			installations or elements being installed. Coordinate ventilation requirements to produce the ambient condition
25			required and minimize consumption of energy
27			1 Heating Facilities: Excent where use of the nermanent system is authorized inrovide vented self-
28			contained LP gas or fuel oil heaters with individual space thermostatic control
29			a Use of gasoline-hurning snace heaters, onen flame, or salamander type heating units is
30			nrohibited
30			promoteu.
32	1.5.	TELEC	OMMUNICATIONS SERVICES AND WI-FI
32	1.0.	Δ	Provide maintain and nay for telecommunications services to field office at time of project mobilization through
34		7.	construction closeout
35		в	Telecommunications services shall include:
36		5.	1 Windows-based personal computer dedicated to project telecommunications
37			 Shared access to the internet via WIFL or similar wireless connection
38			a Access must be canable to support minimum of 10 wireless devices
30			3 Email Account/address dedicated for GC Project Manager of GC Supervisor on site
10			
40 41	16	TEMP	ΟΡΑΡΥ ΣΑΝΙΤΑΡΥ ΕΔΟΙΙ ΙΤΙΕΣ
<u>1</u> 2	1.0.	Δ	Provide and maintain required facilities and enclosures. Provide at time of project mobilization
42 //3		R.	Temporary toilets: Comply with regulations and health codes for the type number location operation and
45 44		Б.	maintenance of fixtures and facilities. Install where facilities will best serve the Project's needs
77 15			1 Provide toilet tissue, namer towels, namer curs, and similar disposable materials foreach facility. Provide
45			covered waste containers for used material
40 //7			2 Toilets: Install self-contained toilet units. Shield toilets to ensure privacy
-, /8		C	An antain daily in clean and sanitary condition
40 //Q		с. D	Water: Provide notable water annroved by local health authorities
50		D.	water. Provide potable water approved by local health authorities
51	17	BARRI	FRS
52	±.,.	Δ	Provide barriers to prevent unauthorized entry to construction areas to prevent access to areas that could be
53			hazardous to workers or the public and to protect existing facilities and adjacent properties from damage from
54			construction operations and demolition
55			construction operations and actionation.
56	1 8	FENCI	NG
57	1.0.		Construction: Refer to Plan Documents and Specification Section 01 76 00: Fencing Materials and Barricades
58		<i>,</i>	construction of the rest of the booments and specification section of 70 out rending materials and barriedues

1	1.9.	EXTER	RIOR ENCLOSURES
2		Α.	Provide temporary weather tight closure of exterior openings to accommodate acceptable working conditions
3			and protection for Products, to allow for temporary heating and maintenance of required ambient temperatures
4			identified in individual specification sections, and to prevent entry of unauthorized persons. Provide access doors
5			with self-closing hardware and locks.
6			
7	1.10.	SECUF	RITY
8		Α.	Provide security and facilities to protect Work, existing facilities, and Owner's operations from unauthorized
9			entry, vandalism, or theft.
10			
11	1.11.	VEHIC	CULAR ACCESS AND PARKING
12		Α.	Comply with regulations relating to use of streets and sidewalks, access to emergency facilities, and access for
13			emergency vehicles.
14		В.	Coordinate access and haul routes with governing authorities and Owner.
15		C.	Provide and maintain access to fire hydrants, free of obstructions.
16		D.	Existing parking areas located in the lower area of the Warner Park Community Recreation Center as shown in
17			the Lands for Work may be used for construction parking for the duration of the construction work. Under no
18			circumstances shall construction vehicles be parked in the parking lot outside of the Lands for Work or in the
19			construction area.
20			
21	1.12.	WAST	'E REMOVAL
22		Α.	See Section 01 74 19 - Waste Management, for additional requirements.
23		В.	Provide waste removal facilities and services as required to maintain the site in clean and orderly condition.
24		C.	Provide containers with lids. Remove trash from site periodically.
25		D.	If materials to be recycled or re-used on the project must be stored on-site, provide suitable non-combustible
26			containers; locate containers holding flammable material outside the structure unless otherwise approved by the
27			authorities having jurisdiction.
28		E.	Open free-fall chutes are not permitted. Terminate closed chutes into appropriate containers with lids.
29			
30	1.13.	PROJE	ECT IDENTIFICATION
31		A.	Provide project identification sign of design and construction indicated in Section 01 58 13.
32		В.	Erect on site at location determined by Owner .
33		C.	No other signs are allowed without Owner permission except those required by law.
34			
35	1.14.	FIELD	OFFICES
36		А.	Office: Weather tight, with lighting, electrical outlets, heating, cooling equipment, and equipped with sturdy
37			furniture, drawing rack and drawing display table.
38		В.	Field Office shall be located in the Lands for Work.
39		C.	Provide space for Project Meetings with table and chairs to accommodate a minimum of ten (10) persons.
40		D.	Provide a minimum of a 40" LCD monitor or other digital projection device to be connected to the computer
41			identified in Section 1.4 Telecommunications Services (above), for use during progress meetings in connection
42			with reviewing construction progress information posted to the Project Management Web Site (Specification 01
43			31 23) hosted by the Owner.
44			
45	PART	2 - PRO	DUCTS
46			
47	2.1.	ТЕМР	ORARY PARTITIONS
48		Α.	Provide dustproof partitions to limit dust and dirt migration and to separate occupied areas from fumes and
49			noise.
50			1. Non-fire rated partitions, standard
51			a. Wood stud framing, 6-mil polyethylene
52			······································
53	2.2.	EQUIF	PMENT
54		Α.	Temporary Lifts and Hoists: Contractors requiring temporary lifts and hoists shall provide facilities for hoisting
55			materials and employees.
56		В.	Electrical Outlets: Electrical Contractor shall provide properly configured NEMA polarized outlets to prevent
57			insertion of 110-120 volt plugs into higher voltage outlets. Provide receptacle outlets equipped with ground-fault
58			circuit interrupters, reset button and pilot light, for connection of power tools and equipment.
	VV ARIN		

	C.	Electrical Power Cords: Contractors requiring power cords shall provide grounded extension cords; use "hard- service" cords where exposed to abrasion and traffic. Provide waterproof connectors to connect separate lengths of electric cords, if single lengths will not reach areas where construction activities are in progress. Do
	D.	not exceed safe length-voltage ratio. Lamps and Light Fixtures: Electrical Contractor shall provide general service incandescent lamps of wattage
		required for adequate illumination. Provide guard cages or tempered glass enclosures, where exposed to
	E	breakage. Provide exterior fixtures where exposed to moisture.
	с.	Heating Units: General Contractor shall provide temporary heating units that have been tested and labeled by
	F	Eist Aid Supplies: General Contractor shall provide first aid supplies complying with governing regulations
	G.	Fire Extinguishers: General Contractor shall provide hand-carried nortable III-rated fire extinguishers of NEPA
	0.	recommended classes for the exposures, extinguishing agent and size required by location and class of fire exposure.
PART	3 - EXE	CUTION
2 1	TEM	
5.1.		Until fire protection needs are supplied by permanent facilities. General Contractor shall install and maintain
	Λ.	temporary fire protection facilities of the types needed to protect against reasonably predictable and controllable fire losses.
	В.	Comply with NFPA 10 "Standard for Portable Fire Extinguishers," and NFPA 241 "Standard for Safeguarding Construction, Alterations and Demolition Operations".
	C.	Locate fire extinguishers where convenient and effective for their intended purpose.
	D.	Store combustible materials in containers in fire-safe locations.
	Ε.	Maintain unobstructed access to fire extinguishers, fire hydrants, temporary fire protection facilities, stairways
		and other access routes for fighting fires.
	F.	Prohibit smoking on the premises.
	G.	Supervise welding operations, combustion-type temporary heating units, and similar sources of fire ignition
		according to requirements of authorities having jurisdiction.
	Н.	Develop and supervise an overall fire-prevention and -protection program for personnel at Project site
	I.	Review needs with local fire department and establish procedures to be followed. Instruct personnel in methods and procedures. Post warnings and information.
3.2.	COLL	ECTION AND DISPOSAL OF WASTE
	Α.	Collect waste from construction areas and elsewhere daily
	В.	Comply with requirements of NFPA 241 for removal of combustible waste material and debris. Enforce requirements strictly.
	C.	Do not hold materials more than 7 days during normal weather or 3 days when the temperature is expected to rise above 80 deg F.
	D.	Handle hazardous, dangerous, or unsanitary waste materials separately from other waste by containerizing properly. Dispose of material in a lawful manner.
3.3.	ENVI	RONMENTAL PROTECTION
	Α.	Provide protection, operate temporary facilities and conduct construction in ways and by methods that comply
		with environmental regulations, and minimize the possibility that air, waterways and subsoil might be contaminated or polluted, or that other undesirable effects might result.
	В.	Avoid use of tools and equipment which produce harmful noise.
	C.	Restrict use of noise making tools and equipment to hours that will minimize complaints from persons or firms near the site.
2 /	REM	OVAL OF TEMPORARY LITH ITIES FACILITIES AND CONTROLS
J. 7 .	A.	Remove temporary utilities, equipment, facilities, and materials prior to Substantial Completion inspection
	в.	Remove underground installations to a minimum depth of 2 feet (600 mm). Grade site as indicated.
	С.	Clean and repair damage caused by installation or use of temporary work.
	D.	Restore existing facilities used during construction to original condition.
	Ε.	Restore new permanent facilities used during construction to specified condition.

1 2

END OF SECTION

		SECTION 01 58 13
		TEMPORARY PROJECT SIGNAGE
PART	1 – GI	ENERAL
1	1.	SECTION INCLUDES
1	2.	QUALITY ASSURANCE
1	3.	SUBMITTALS
PART	2 - PR	ODUCTS
2	.1.	SIGN MATERIALS
2	.2.	PROJECT IDENTIFICATION SIGN
PART	3 - EX	ECUTION
3	.1.	INSTALLATION
3	.2.	REMOVAL
PART	1 – G	ENERAL
1.1.	SEC	
	A.	Project identification sign.
	_	
1.2.	QU	ALITY ASSURANCE
	Α.	Design sign and structure to withstand 50 miles/hr wind velocity.
	В.	Sign Painter: Experienced as a professional sign painter for minimum three years.
	C.	Finishes, Painting: Adequate to withstand weathering, fading, and chipping for duration of construction.
1.3.	SUE	MITTALS
	Α.	See Section 01 30 00 – Administrative Requirements for submittal procedures.
	В.	Shop Drawing: Show content, layout, lettering, color, structure, sizes.
PART	2 - PF	COLICTS
	<u> </u>	
2.1.	SIG	N MATERIALS
	Α.	Structure and Framing: New, wood, structurally adequate.
	В.	Sign Surfaces: Exterior grade plywood with medium density overlay, minimum ¾" thick, standard large sizes
		minimize joints.
	C.	Rough Hardware: Galvanized
2.2.	PRC	JECT IDENTIFICATION SIGN
	Α.	One painted sign, 32 sq ft area, bottom 6 feet above ground.
	В.	Content:
		1. Project title, City of Madison, and Parks Division logo
		2. Names and title of Architect.
		3. Name of Prime Contractor.
		4. Full color project rendering from high resolution image as furnished by Architect.
PART	3 - EX	ECUTION
~ 4	1010	
3. 1 .	INS	IALLATION
	А. Р	install project identification sign within 30 days after date fixed by Notice to Proceed.
	в. С	ETECLIAL DESIGNATED IDCATION.
	L.	nistan sign surrate plumb and level, with butt joints. Anthor seturely.
3.2.	REN	IOVAL
	Α.	Remove sign, framing supports, and foundations at completion of Project and restore the area.

		PRODUCT REQUIREMENTS
DART	1_6	
1	1 - U	
-	1.1.	
_	1.2.	RELATED SPECIFICATIONS
	1.3.	
PARI	2 – PI	CODUCIS – THIS SECTION NOT USED
PARI	3 - EX	
1	3.1.	GENERAL CONTRACTOR REQUIREMENTS
3	3.2.	BULK MATERIAL
3	3.3.	DRY PACKAGED MATERIAL
3	3.4.	STRUCTURAL AND FRAMING MATERIAL
3	3.5.	EQUIPMENT
3	3.6.	FINISH PRODUCTS
3	3.7.	DUCTWORK, PIPING, AND CONDUIT
3	3.8.	OWNER PROVIDED, CONTRACTOR INSTALLED EQUIPMENT
PART	1 – G	ENERAL
1.1.	SUN	IMARY
	Α.	The purpose of this specification is to provide general guidelines and responsibilities related to the receiving,
		handling, and storage of all materials and products from arrival on the job site through installation.
		1. Immediate inspection of delivered goods means a timely replacement if damaged.
		2. Proper storage helps prevent damage and loss by weather, vandalism, theft, and job site accidents.
		3. Proper storage helps with job site performance and safety.
		2. Proper handling helps prevent damage and job site accidents.
	В.	Each Contractor shall be directly responsible for the receiving, handling, and storage of all materials and
		products associated with the Work of their Division or Trade.
	C.	Each Contractor responsible for Work associated with Owner provided materials or products shall be response
		for the receiving, handling and storage of the material/product as outlined in Section 3.8 below
12	RFI	ATED SPECIFICATIONS
1.2.		Parts of this specification will reference articles within "The City of Madison FACILITIES MANAGEMENT
	л.	SPECIFICATIONs for Public Works Construction"
		I have the following link to access the EACH TIES MANAGEMENT SPECIFICATIONS web page:
		1. Use the billowing link to access the FACIENTES MAINAGEMENT SPECIFICATIONS web page.
		<u>IIII.p.//www.cityoffidatisoff.com/business/pw/specs.cim</u>
		a. Click on the "Part" chapter identified in the specification text. For example if the specification
		says "Refer to City of Madison FACILITIES MANAGEMENT SPECIFICATION 2 10.2" click the link t
		Part II, the Part II PDF will open.
		b. Scroll through the index of Part II for specification 210.2 and click the text link which will take y scroll through the index of Part II for specification 210.2 and click the text link which will take y
		to the referenced text.
	_	c. City Standard Detail Drawings (SDD) may be located from the index in Part VIII.
	В.	Section 01 57 21 Indoor Air Quality
	C.	Section 01 74 13 Progress Cleaning
	D.	Section 01 76 00 Protecting Installed Construction
	Ε.	Other Divisions and Specifications that may address more specifically the requirements for the storage and
		handling of materials and products associated Work of other Divisions or Trades.
1.3.	QU	ALITY ASSURANCE
	Α.	The GC shall be responsible for ensuring that these minimum storage and handling requirements are met by
		contractors on the project site including but not limited to the following:
		1. Receiving deliveries of materials, products, and equipment.
		a. Inspect all deliveries upon arrival for damage, completeness, and compliance with the
		construction documents.
		i. Deliveries shall remain in original packaging or crates, shipping manifest shall be kept v
		the delivery and the packaging shall have visible identification of the items within the
		nackaging.
		DUCAUSIIS.

1				b. Immediately report any damaged products or equipment to the GC, begin arrangements for
2				Materials or equipment that have been damaged are incomplete, or do not comply with the
<u>л</u>				construction documents shall not be permitted to be installed
- 5			2	All materials and products shall be stored within the designated limits of the project site. Only store the
6			۷.	amount of material necessary for uncoming operations so as not to interfere with other construction
7				activities and access to Work by the Owner and Architect Any offsite storage shall be at the expense of
8				the contractor storing the material or product. All offsite storage requirements shall comply with this
9				specification All offsite storage of materials is subject to Owner Representative Quality Management
10				review at any time.
11			3.	Large storage containers may be used but shall be weather tight, securable, placed on concrete blocks.
12			•	timbers, or jack stands and shall be level.
13			4.	When lifting equipment is required the equipment rating shall be greater than the loading requirements
14				of the item being lifted. In addition all of the following shall apply as necessary:
15				a. Only designated and/or designed lift points shall be used.
16				b. Large items shall have tag lines and handlers at all times during lifting operations.
17				c. Lift at multiple points as needed to prevent bending.
18			5.	Materials and products stored inside of the structure shall comply with all of the following:
19				 Storage shall not be allowed to impede the flow of work in progress.
20				Storage shall not be allowed to hide completed work from review and inspections.
21				c. Storage shall not exceed the design loads of the structural components it is being stored upon.
22			6.	All materials and products shall be stored according the manufacturers minimum recommended
23				requirements. All of the following shall be considered before storing any product or material:
24				a. Dust and dirt
25				b. Moisture and humidity, including rain and snow
26				c. Excessive temperatures, direct sun, etc
27				d. Product or material weight and size
28				e. Potential for breakage
29				f. Product incompatibility with other products such as corrosiveness, chemical reactions,
30				fiammability, etc.
31			7	g. Product or material value and replacement cost
32			7.	The contractor shall be responsible for providing fully functional tarps or plastic wrap, to protect
33 24				tied, strapped, or weighted down to resist blowing
25 25			0	The Contractor shall be recognished for any temperary heating, cooling, or other utility requirement that
26			о.	may be according with the storage of a material or product
30			٩	The Contractor shall be recognished for securing materials and products of value such as copper Λ/V
38			5.	equinment atc. Such items shall be stored in securable shinning containers, job trailers or other such
39				storage devices. Container shall be kent secured when not in use
40		В.	The G	C shall inspect the job site daily to ensure that all products and materials stay weather tight and are
41		5.	secure	ed against vandalism or theft as required by this specification.
42		C.	The O	where Representative may at any time request improvements regarding storage of any material or product
43		-	being	provided under these construction documents.
44			U	
45	PART	2 – PRC	DUCTS	– THIS SECTION NOT USED
46				
47	PART	3 - EXE	CUTION	
48 10	21	GENE		
50	3.1.	A	Design	nate material storage and handling areas as needed including all of the following
51		71.	1	Designate specific areas of the site for delivery and storage of materials to be used during the execution
52				of the Work.
53			2.	Designated areas shall not be located so as to interfere with the installation of any Work including Work
54				by others such as the installation of utilities or the maintenance of existing utilities. This shall include not
55				storing items in active utility easements as designated by the site plan.
56		В.	Arran	ge for openings in the building as needed to allow delivery and installation of large items. Openings shall
57			be ap	propriately sized to include the use of booms, slings, and other such lifting devices that may be larger than
58			the ite	em being installed.
	WARN	IER PARI	сомм	UNITY RECREATION CENTER
	FXPAN			

1			1. When openings are required in completed Work (new or existing) the GC shall be responsible for				
2			providing an appropriate opening and for restoring the opening to the original or better condition upon				
3			completion. Restoration shall be weather tight and complete.				
4		C.	Repeated moving and handling of items being stored shall not be allowed. The GC shall be responsible for any				
5		0.	damage and replacement because of michandling or excessive bandling				
6			duringe and replacement because of misinarianing of excessive narianing.				
7	3.2.	BULK	MATERIAL				
8	0.2.	A	Bulk material such as sand, gravel, top soil and other types of fill shall be stored away from the construction area				
9		7	and shall be stock alled as follows:				
10			and shall be stock price as follows.				
11			1. An out in material shall be piece safety and efficiency in as shall an area as phatetian. Only store the				
12			amount of material necessary to upcoming operations so as not to interfere with other construction				
12			All stock rules should be used if for so (cosk property installed around the perimeter to provent erasion and				
13			2. All stock piles shall have shit rence/sock property installed around the perimeter to prevent erosion and loss of material. Before to City of Madican FACULTIES MANAGEMENT SPECIFICATION Section 210.1(f) and				
14			ioss of material. Refer to City of madison FACILITIES MANAGEMENT SPECIFICATION Section 210.1(1) and				
15			other related specification or details.				
16			3. Fine grained material shall be protected with tarps to prevent blowing. Tarps shall be weighted or staked				
1/		_	to stay in place.				
18		В.	Bulk material such as brick, concrete block, stone, and other palletized materials shall be stored on original				
19			shipping pallets until ready for use.				
20							
21	3.3.	DRY P	ACKAGED MATERIAL				
22		А.	Dry packaged material such as cement, mortar, etc shall be stored on pallets, on slightly elevated ground or clear				
23			stone pad to keep water away from the base of the material being stored. Protect from moisture.				
24							
25	3.4.	STRUG	CTURAL AND FRAMING MATERIAL				
26		Α.	All structural and framing material shall be stored in an organized manner arranged by type, size and dimension.				
27			Materials shall be stored on pallets or timbers as necessary and shall not be allowed to lie directly on the ground.				
28		В.	Long and heavy items shall be supported at several points to prevent bending and warping.				
29							
30	3.5.	EQUIF	MENT				
31		Α.	Equipment delivered to the site shall be stored away from all construction activities until the item can either be				
32			moved inside or properly installed.				
33		В.	Equipment shall be stored on slightly elevated ground or clear stone pad to keep water away from the base of				
34			the equipment.				
35							
36	3.6.	FINISH	I PRODUCTS				
37		Α.	Finish products such as flooring, tile, counters, lockers, toilets, partitions, lighting, and other similar items should				
38			not be delivered and stored until the structure has been enclosed, is weather tight, temperature controlled and				
39			the contractor is ready for such items to be installed.				
40			1. Storage of finished products outside for any length of time shall not be allowed.				
41		в	Products that cannot be stored inside the structure shall be stored in secured containers or job trailers until such				
42		В.	time as they are ready to be installed				
13		C	Products with a high potential for breakage such as glass mirrors tiles toilet fixtures etc. shall be stored with				
45 44		С.	additional protection as necessary such as but not limited to the following:				
 //5			1 Store in original shinning containers until ready for installation				
45			Store in original simpling containers with ready for instantation. Do not store in bight straffic areas				
40			2. Do not such a fingle traine acts.				
47			5. Sineid with other materials such as carubbard, prywood, or sinnar products.				
40	2.7	DUCT					
49	3.7.	DUCI	WORK, PIPING, AND CONDOIL				
50		А.	An piping and conduct shall be stored horizontally unless otherwise specified by the manufacturer of Division and				
51 51			Haue Specifications.				
52			1. Do not store directly on grade.				
53			2. Cover metal pipes and tubes to prevent rust and corrosion, allow ventilation to prevent condensation.				
54			3. whenever possible use pipe stands for storing pipe and conduit to prevent tripping and rolling hazards.				
55		В.	All ductwork shall be stored horizontally or vertically as necessary unless otherwise specified by the				
56			manufacturer or Division and Trade Specifications.				
57			1. During storage, both ends of each duct shall be protected with plastic sheathing to prevent dust and dirt				
58			trom getting inside the duct. Sheathing shall be sufficiently taped to the duct.				
	WARN EXPAN	ER PARK SION	COMMUNITY RECREATION CENTER				

1 2 3			2.	After installation, free/open ends shall remain protected with taped plastic sheathing and or temporary filters as specified by division or Trade specifications.
4	3.8.	OWN	ER PRO	VIDED, CONTRACTOR INSTALLED EQUIPMENT
5		Α.	Sectio	n 3.8.A. shall apply to all equipment being provided to any contractor directly from the Owner for
6			installa	ation under the contract.
7			1.	The Owner or Owners Representative shall do the following:
8				a. Inspect all deliveries upon receipt and notify manufacturer of any issues directly.
9				b. Review the received shipment with the contractor.
10 11				 Only provide products or materials to the contractor that were not damaged through shipping or handling.
12				ii. Confirm missing products or materials and anticipated delivery schedule if known.
13			2.	The Contractor responsible for the installation of Work associated with Owner provided materials or
14				products shall "take ownership" and provide safe and secure storage and handling as previously
15				described within this specification.
16				i. The Contractor shall be liable for the repair or replacement of any material or product
17				damaged after taking ownership of the product from receipt through final acceptance.
18		В.	Sectio	n 3.8.B. shall apply to all equipment being provided by the Owner but shipped directly to any sub-
19			contra	ictor or the project site for installation under the contract.
20			1.	The GC and/or Contractor responsible for the Work associated with the Owner provided materials or
21				products shall do the following:
22				a. Inspect all deliveries upon receipt and notify the Owner or Owners Representative of any issues
23				airectiy.
24 25				I. Owner or Owners Representative shall notify manufacturer of any issues directly.
25				b. Review the received shipment with the Owner of Owners Representative
20			2	The Contractor shall "take expression" and provide safe and secure storage and handling as provide the
27			Ζ.	described within this specification
20				i The Contractor shall be liable for the repair or replacement of any material or product
20				damaged after taking ownership of the product from receipt through final accentance
30				
32				
33				
34				END OF SECTION
35				

		SECTION 01 71 23 FIELD ENGINEERING
ρΔρτ 1	1 – 65	NERAL
1 rANT	1 GE	
1	.1. 2	
1	.2. 2	
1	.J. 1	
1	.4. c	
	.כ. מס נ	
	2 - PR	
PARTS	5 – EV	ECOTION – THIS SECTION NOT USED
	1 _ GI	
<u>ranı</u>	1-01	
11	RFO	
1.1.	^	The Contractor shall provide and pay for field angineering convices required for the Preject:
	A.	1 Land surviving convices required to execute the Work to include building addition location and layout
		1. Land surveying services required to execute the work, to include building addition location and layout
		and location and layout of pavements and an proposed site improvements.
		 vertification of existing building dimensions, elevations, and relationship to proposed additions. Professional Engineering convices to superior construction with the definition of the second s
		 Protessional Engineering services to execute Contractor's construction methods. Desistened Prefereigned Engineering the State of Wisconsistent determines the lead on the State of Wisconsistent determines the lead of Wisconsistent determines the lead of Wisconsistent determines the lead of Wisconsistent determines the State of Wisconsistent determines the lead of Wisconsistent determines the State of Wisconsistent determines the lead of Wisconsistent determines the State of Wisconsistent determ
		4. Registered Protessional Engineer in the State of Wisconsin to determine the load capacity of the exist
		structure for use of Contractors temporary facilities, equipment, lifts, machinery, material storage, et
1.2.	REL/	ATED REQUIREMENTS
	Α.	Conditions of the Contract
1.3.	PRO	CEDURES
	Α.	A property survey has been prepared for the Owner and has been bound with Contract Drawings. Surveys sh
		describe physical characteristics, legal limitations and utility locations for the site of the Project, and a legal
		description of the site. If information is incomplete, notify Owner to furnish additional information. Verify
		easement locations, front, side, and rear yard restrictions, if any; and property line locations. Verify control
		points, and establish bench marks. Locate and layout roads, walks, parking areas and all civil structures and a
		proposed site improvements.
	В.	Verify locations of underground services, utilities, structures, etc. which may be encountered or affected by t
		Work.
1.4.	PRO	JECT SURVEY REQUIREMENTS
	Α.	Using datum, the lot lines and present levels have been established as indicated on the Drawings. Other grac
		lines, levels and benchmarks, shall be established and maintained by the Contractor, who shall be responsible
		them. As work progresses, the Contractor shall layout on forms and floor, the locations of all partitions, walls
		and fix column centerlines as a guide to all trades. The Contractor shall make provision to preserve property
		stakes, benchmarks, or datum point. If any are lost, displaced or disturbed through neglect of any Contractor
		Contractor's agents or employee, the Contractor responsible shall pay the cost of restoration.
	В.	Establish lines and levels, locate and layout, by instrumentation and similar appropriate means. additions.
		column locations, floor levels, stakes for walks, etc.
	С	Provide data to all Subcontractors for their use as applicable
	с. D	From time to time, verify lavouts by same methods
	υ.	rion and to and, verify hybrid by sume methods.
15	REC	ORDS
1.3.		Maintain a complete accurate log of all control and survey work as it progresses
	А.	ואמוותמווי מ נטוואובוב, מננטומוב וטע טו מוו נטוונוטו מווט געו עצץ שטוא מג וג אוטעו בגבא.
PART	2 – PF	ODUCTS – THIS SECTION NOT USED
PART :	3 – EX	ECUTION – THIS SECTION NOT USED
		END OF SECTION

		SECTION 01 73 29 CUTTING AND PATCHING					
PART	1 – G	I 1					
1	1.1.	SUMMARY					
1	1.2.	RELATED SPECIFICATION SECTIONS					
1	1.3.	DEFINITIONS1					
1	1.4.	QUALITY ASSURANCE					
1	1.5.	WARRANTY2					
PART	2 - M	ATERIALS					
2	2.1.	GENERAL2					
PART	3 - EX	ECUTION2					
3	3.1.	EXAMINATION2					
3	3.2.	PREPARATION					
3	3.3.	PERFORMANCE					
3	3.4.	CLEANUP AND RESTORATION					
PART	1 – G	ENERAL					
1.1.	SUN	ΛΜΑRΥ					
	Α.	This Section includes general procedural requirements for cutting and patching including, but not limited to the					
		following:					
		1. Examination					
		2. Preparation					
		3. Performance					
		4. Cleanup and Restoration					
1.2.	REL	ATED SPECIFICATION SECTIONS					
	Α.	Divisions 02 through 32 Sections for specific requirements and limitations applicable to cutting and patching					
		individual parts of the Work.					
	В.	Division 07 Section "Penetration Fire Stopping" for patching fire-rated construction.					
1.3.	DEF	INITIONS					
	Α.	Cutting: Removal of in-place construction necessary to permit installation or performance of other Work.					
	В.	Patching: Fitting and repair work required to restore surfaces to original conditions after installation of other					
		Work.					
	C.	Level Alpha					
	~						
1.4.	QU	JALITY ASSURANCE					
	А.	structural Elements. Do not cut and patch structural elements in a manner that could change their load-carrying					
	B	Operational Elements: Do not cut and natch operating elements and related components in a manner that results					
	Б.	in reducing their capacity to perform as intended or that may result in increased maintenance or decreased					
		onerational life or safety					
	c	Miscellaneous Elements: Do not cut and natch miscellaneous elements or related components in a manner that					
	с.	could change their load-carrying capacity that results in reducing their capacity to perform as intended, or that					
		may result in increased maintenance or decreased operational life or safety. Some miscellaneous elements					
		include the following:					
		1 Water moisture or vanor barriers					
		2. Membranes and flashings					
		3. Exterior curtain-wall construction					
		4. Equipment supports					
		5. Piping, ductwork, vessels, and equipment					
		6. Noise and vibration control elements and systems					
	D.	Visual Requirements: Do not cut and patch construction in a manner that results in visual evidence of cutting and					
		patching. Do not cut and patch construction exposed on the exterior or in occupied spaces in a manner that					
		would, in Architect's opinion, reduce the building's aesthetic qualities. Remove and replace construction that has					
		been cut and patched in a visually unsatisfactory manner.					

1 **1.5. WARRANTY**

2

3

4

5

6 7

8

12 13

14

15

17

19

20

21

22

23

34

- A. Existing Warranties: Remove, replace, patch, and repair materials and surfaces cut or damaged during cutting and patching operations, by methods and with materials so as not to void existing warranties.
 - B. All cutting and patching work performed under this contract shall be warranted like new work as defined by the Specification governing the work.

PART 2 - MATERIALS

9 2.1. GENERAL

- A. Comply with requirements specified within other sections of the Specifications.
 B. In-Place Materials: Use materials identical to existing in-place materials. For ex
 - B. In-Place Materials: Use materials identical to existing in-place materials. For exposed surfaces use materials that visually match in-place adjacent surfaces to the fullest extent possible.
 - 1. If identical materials are unavailable or cannot be used, use materials that, when installed, will match the visual and functional performance of in-place materials.

16 PART 3 - EXECUTION

- 18 **3.1. EXAMINATION**
 - A. Examine surfaces to be cut and patched and conditions under which cutting and patching are to be performed.
 1. Compatibility: Before patching, verify compatibility with and suitability of substrates, including compatibility with in-place finishes or primers.
 - 2. Proceed with installation only after unsafe or unsatisfactory conditions have been corrected.

24 3.2. PREPARATION

- 25 A. Temporary Support: Provide temporary support of Work to be cut.
- B. Protection: Protect in-place construction and existing conditions during cutting and patching to prevent damage.
 Provide protection from adverse weather conditions for portions of Project that might be exposed during cutting
 and patching operations. If the failure to protect, or the lack of protection, of in-place construction and/or
 existing conditions results in damage, the contractor shall be responsible for repair to previous condition.
 Adjoining Areas: Avoid interference with use of adjoining areas or interruption of free passage to adjoining areas.
- 31D.Existing Utility Services and Mechanical/Electrical Systems: Where existing services/systems are required to be32removed, relocated, or abandoned, bypass such services/systems before cutting to eliminate interruption to33occupied areas.

35 3.3. PERFORMANCE

36 General: Employ skilled workers to perform cutting and patching. Proceed with cutting and patching at the A. 37 earliest feasible time, and complete without delay. 38 1. Cut in-place construction to provide for installation of other components or performance of other construction, and subsequently patch as required to restore surfaces to their original condition. 39 40 Β. Cutting: Cut in-place construction by sawing, drilling, breaking, chipping, grinding, and similar operations, 41 including excavation, using methods least likely to damage elements retained or adjoining construction. If 42 possible, review proposed procedures with original Installer; comply with original Installer's written 43 recommendations. 44 1. In general, use hand or small power tools designed for sawing and grinding, not hammering and 45 chopping. Cut holes and slots as small as possible, neatly to size required, and with minimum disturbance 46 of adjacent surfaces. Temporarily cover openings when not in use. 2. 47 Finished Surfaces: Cut or drill from the exposed or finished side into concealed surfaces. Concrete or Masonry: Cut using a cutting machine, such as an abrasive saw or a diamond-core drill. 48 3. 49 4. Excavating and Backfilling: Comply with requirements in applicable Division 3I Sections where required by 50 cutting and patching operations. 51 5. Mechanical and Electrical Services: Cut off pipe or conduit in walls or partitions to be removed. Cap, 52 valve, or plug and seal remaining portion of pipe or conduit to prevent entrance of moisture or other 53 foreign matter after cutting. 54 Proceed with patching after construction operations requiring cutting are complete. 6. 55 C. Patching: Patch construction by filling, repairing, refinishing, closing up, and similar operations following 56 performance of other Work. Patch with durable seams that are as invisible as possible. Provide materials and 57 comply with installation requirements specified in other Sections.
1 D. Inspection: Where feasible, test and inspect patched areas after completion to demonstrate integrity of 2 installation. 3 4 3.4. **CLEANUP AND RESTORATION** Restore exposed finishes of patched areas and extend finish restoration into retained adjoining construction in a 5 Α. 6 manner that will eliminate evidence of patching and refinishing. 7 Clean piping, conduit, and similar features before applying paint or other finishing materials. 1. 8 2. Restore damaged pipe covering to its original condition. 9 3. Floors and Walls: Where walls or partitions that are removed extend one finished area into another, 10 patch and repair floor and wall surfaces in the new space. Provide an even surface of uniform finish, 11 color, texture, and appearance. Remove in-place floor and wall coverings and replace with new materials, if necessary, to achieve uniform color and appearance. 12 13 4. Where patching occurs in a painted surface, apply primer and intermediate paint coats over the patch 14 and apply final paint coat over entire unbroken surface containing the patch. Provide additional coats until patch blends with adjacent surfaces. 15 16 5. Ceilings: Patch, repair, or re-hang in-place ceilings as necessary to provide an even-plane surface of 17 uniform appearance. 6. Exterior Building Enclosure: Patch components in a manner that restores enclosure to a weather tight 18 condition. 19 20 7. Cleaning: Clean areas and spaces where cutting and patching are performed. Completely remove paint, 21 mortar, oils, putty, and similar materials. 22 8. Any smoke and fire caulking that has been disturbed must be replaced by the Contractor as required by 23 code. 24 25 26 27 END OF SECTION 28

Page Intentionally Left Blank

		SECTION 01 74 13 PROGRESS CLEANING
PART	1 – GI	ENERAL
1	l.1.	SUMMARY
1	L.2.	RELATED SPECIFICAITONS
1	L.3.	QUALITY ASSURANCE
PART	2 - PR	
2	2.1.	CLEANING MATERIALS AND EQUIPMENT
PARI	3 - EX	
	5.1. N 2	
	5.Z.	PROJECT STE CLEANING
	2.7. 2.7	
	25	
PART	1 – G	ENERAL
1.1.	SUN	MMARY
	Α.	Throughout the execution of this contract all contractors shall be responsible for maintaining the project site
		standard of cleanliness as described in this specification.
	В.	All contractors shall also comply with the requirements for cleaning as described in other specifications.
	C.	Work included in this specification shall include but not be limited to:
		1. Safety Cleaning
		2. Project Site Cleaning
		3. Progress Cleaning
		4. Final Cleaning
1.2.	REL	ATED SPECIFICAITONS
	Α.	Section 01 35 00 Special Procedures
	В.	Section 01 60 00 Product Requirements
	C.	Section 01 74 19 Construction Waste Management and Disposal
	D.	Section 01 76 00 Protecting Installed Construction
1.3.	QU	ALITY ASSURANCE
	Α.	The General Contractor (GC) shall conduct daily inspections, more often if necessary, of the entire project sit
		ensure the requirements of cleanliness are being met as described within these specifications.
	В.	All contractors shall comply with other regulatory requirements as they apply to waste recycling, reuse, haul
		and disposal requirements of any governmental authority having jurisdiction.
	C.	The Owner reserves the right to have work done by others in the event any contractor fails to perform clean
		as described within these specifications. The cost of any Owner provided cleaning shall be charged to the
		contractor through a deduct change order.
PART	2 - PF	RODUCTS
2.1.	CLE	ANING MATERIALS AND EQUIPMENT
	Α.	The Contractor shall provide all required personnel, equipment, and materials necessary to maintain the
		required level of cleanliness as described in this specification.
	В.	Use only cleaning materials and equipment that are compatible with the surface being cleaned, as
		recommended by the manufacturer, or as approved by the A/E.
	C.	Use only cleaning materials, equipment, and methods as recommended in the manufacturers care and use g
		of the material, finish or equipment being cleaned.
PART	3 - EX	KECUTION
31	۶۸E	ETY CLEANING
3.1.	Δ	All Contractors shall be responsible for safety cleaning as required by OSHA and other regulatory requirement
	×	The second se

1 2		В.	Safety 1.	Cleaning shall include but not be limited to the following: All work areas, passageways, ramps, and stairs shall be kept free of debris, scrap materials, pallets, and			
3				other large items that would obstruct exiting routes. Small items such as tools, electrical cords, etc are			
4				picked up when not in use.			
5			2.	Form and scrap lumber shall have nails/screws removed or bent over. Lumber shall be neatly stacked in			
6				an area designated by the GC.			
7			3.	Spills of oil, grease, and other such liquids shall be cleaned immediately or sprinkled with sand/oil-dry			
8				first, then cleaned.			
9			4.	Oily, flammable, or hazardous items shall be stored in appropriate covered containers and storage			
10				devices unless actively being used.			
11			5.	Oily, or flammable rags, and other such waste shall only be disposed of in authorized covered containers.			
12			6.	Disposal by burning shall not be allowed at any time.			
13							
14	3.2.	PROJ	ECT SITE	CLEANING			
15		Α.	This se	ection applies to the general cleanliness of the project site as a whole for the duration of the execution of			
16			this co	ontract.			
17		В.	Exteri	or Project Site Areas			
18			1.	The GC and other Contractors as appropriate shall ensure the following levels of cleanliness are applied			
19				to the exterior project site areas.			
20				a. The overall appearance of the project site is neat and orderly. Defined areas for material storage.			
21				material waste, job trailers, and the project area are clean and well maintained.			
22				The construction force is maintained area with no gass and properly posted per all regulatory			
22				5. The construction reflect is maintained, creet with no gaps, and property posted per an regulatory			
23				All provide non-training and property maintained, cleaned, and repaired as peops and			
24				d. All loose materials (construction or waste) are properly trianguly tied or weighted down to resist blowing			
25				a. All construction materials are properly covered with fully functional targe or plastic wrap.			
20				e. An construction materials are property covered with fully during the days to plastic wrap,			
27				protected from the weather, coverings are tred, strapped, or weighted down to resist blowing.			
28		6	1	T. Dust control is applied as necessary or as required by any regulatory requirement.			
29		C.	Interio	or Project Site Areas			
30			1.	All Contractors shall ensure the following levels of cleanliness are applied to the interior project site			
31				areas.			
32				a. The overall appearance of the project site is neat and orderly. Defined areas for material storage,			
33				material waste, and project area are clean and well maintained.			
34				b. Stored materials are kept in original shipping containers whenever possible. Stored materials not			
35				in shipping containers are properly stored and protected according to other applicable			
36				specifications.			
37				c. All scraps and debris shall be properly disposed of as often as necessary to keep work areas,			
38				passageways, stairs, and ramps free of debris and clear for emergency exiting.			
39				d. Boxes, pallets, and other such shipping containers, are broken down, stored in a consolidated area			
40				or, disposed of as often as is necessary.			
41				e. Hand tools, supplies, materials, electrical cords not being used are picked up and sptored in gang			
42				boxes, not left as walking hazards in work areas, passageways, etc.			
43		D.	Job Tr	ailer			
44			1.	The interior of the job trailer shall be kept clean and available as a work space at all times. The GC shall			
45			-	ensure that the following is provided for within the job trailer:			
46				a Meeting snare including tables and chairs			
40 //7				 Meeting space including datasets the across the official construction documents, provide undates 			
47 18				b. Sufficient space for an contractors to access the orient construction documents, provide updates,			
10							
49				FANING			
50	5.5.	^	This e	Levine			
21		А.	rough	in-section shall apply to all Progress cleaning prior to the installation of infishes, fixtures, and trim (IE			
52			rough	Till). - For the numbers of this costion "clean" shall be defined as a lovel of clean basis for a fiduction built			
53 F (т.	For the purposes of this section, clean, shall be defined as a level of cleanliness free of dust and other			
54				material capable of being removed by use of reasonable effort using a good quality janitor broom and			
55			2	snop-vac.			
56			2.	Daily cleanings shall be conducted by all contractors at the end of the work day as follows:			
5/				a. Debris in excavated areas shall be removed prior to backfill and compaction.			
58				b. Debris in wall cavities, chase spaces, etc shall be removed prior to enclosing the spaces.			
	WARN	WARNER PARK COMMUNITY RECREATION CENTER EXPANSION					

1				c. Large items shall be properly stored, returned to designated areas, or disposed of as necessary.
2				d. Loose materials shall be properly secured.
3				e. Flammable or hazardous materials are properly stored or disposed of.
4			3.	Weekly cleaning shall be conducted by all contractors as designated by the GC. Weekly cleanings shall
5				include all the above for a daily cleaning and other necessary cleaning as designated by the GC.
6		В.	This su	ub-section shall apply to Progress Cleaning in preparation for the installation of finishes, fixtures, and trim.
7		2.		a Surfaces receiving finishes shall be thoroughly cleaned prior to contractors applying finish
, 8				materials. The GC shall be responsible for inspecting the area and surfaces being cleaned for
9				finish prior to the sub-contractor applying the finish. This shall include but not be limited to the
10				following:
10				ionowing.
11				I. Wait surfaces shall be wiped clean of dift and only residues, vacuumed nee of dust, and
12				shall be free of surface imperfections prior to painting or installing wall coverings.
13				ii. Metai surfaces shall be wiped clean of dirt and only residues, and be free of surface
14				imperfections prior to painting.
15				III. Flooring shall be broom swept of large and loose items then vacuumed clean of dust and
16				small particles, and damp mopped clean and dried prior to installing any flooring finish.
17				Additional cleaning may be required depending on the preparation requirements
18				recommended by the flooring material manufacturer.
19		C.	This su	ib-section shall apply to Progress Cleaning after the installation of finishes, fixtures, and trim.
20			1.	For the purposes of this section "clean" shall be defined as a level of cleanliness free of dust and other
21				material capable of damaging or visually disfiguring finished work, finishes, fixtures, and trim.
22			2.	Progress Cleaning at this point in the contract shall be conducted immediately as follows:
23				a. Dust, dirt, etc shall be swept and vacuumed off of finish flooring and trim.
24				b. Liquid spills shall be cleaned up according to the spill type. This shall include drips and spills
25				caused by paint, stain, sealants, and other such items.
26			3.	The Contractor(s) at no additional cost to the Owner shall be responsible for replacing any finished work.
27			-	finishes, fixtures, and trim damaged or disfigured because of inadequate or improper cleaning.
28				
29	3.4.	FINAI		NG
30	5	Δ	As not	ed in Specification 01 29 76 Progress Payment Procedures, Progress Payment Milestone Schedule, Final
31		<i>,</i>	Cleani	as shall not be conducted prior to requesting the 90% contract total progress payment and all of the
32			follow	ing shall he complete:
32			1	and shall be complete. All final regulatory inspections including but not limited to Building Inspection Department and Madison
24			1.	An inter regulatory inspections including but not initiate to building inspection Department and Madison
25			2	All Quality Management Observation (OMQ) reports have been closed out
35			2.	All Demonstration and Training to know completed
30			3 .	All Demonstration and training has been completed.
3/			4.	All Attic stock has been consolidated and located to its designated area
38			5.	All protection for installed construction shall be removed prior to final cleaning by the contractor
39				responsible for providing the protections. This shall include the removal of any adhesive residues left
40				behind from tapes. Contractors shall only use manufacturer authorized cleaning materials for removing
41				adhesives, etc.
42		В.	For th	e purposes of this section "clean" shall be defined as a level of cleanliness generally provided by skilled
43			cleane	rs using commercial quality building maintenance equipment and materials.
44		C.	The G	c shall be responsible for ensuring that all requirements under this section are being met.
45		D.	Gener	al Requirements
46			1.	Employ experienced personnel or professional cleaners for final cleaning as necessary for the areas or
47				equipment being cleaned.
48			2.	Cleaning equipment used shall be commercial grade equipment commonly used by professional cleaners.
49			3.	Cleaning equipment and materials shall be cleaned, rinsed, or replaced to ensure a uniform level of
50				cleanliness is being maintained during the final cleaning. This shall include but not be limited to the
51				following:
52				a. Vacuum cleaner bags and/or filters are changed and/or cleaned as often as necessary.
53				b. Dust & wine down rags are washed, rinsed, or replaced before starting each room
54				c Monning equinment
55				i Mon water for washing shall have cleaning solution added to the amount and temperature
56				ner manufacturer's recommendations. Mon washing water shall be replaced often to
57				per manufacturer si recommendations. Work washing water sharper required
58				ii Mon water for ringing shall remain clean, clear, and he replaced as often as persease.
- 50				יו. איסף אמנכי זטי וווזווק זומו וכוומוו נוכמו, נוכמו, מוע שב וכףומנכע מז טונכוו מז וופנפנזמוץ.
	WARN	ER PARK	COMM	JNITY RECREATION CENTER

1				iii. Mop heads shall be rinsed often and replaced as necessary.
2				iv. Mop heads and buckets shall be thoroughly rinsed with each change of water.
3				v. Only new mop heads shall be used for rinsing.
4		E.	Refer t	o all other specifications in this contract for specific requirements regarding final cleaning of finishes,
5			fixture	s, equipment, etc.
6		F.	Exterio	r Cleaning shall include but not be limited to the following:
7			1.	All exterior glazing surfaces have been professionally cleaned and are free of dust and streaking.
8			2.	Metal roofs, siding, and other surfaces shall be clean of dirt and free of splashed or excess materials such
9				as sealants, mortar, paint, etc.
10			3.	All exterior furnishings shall be clean, waste receptacles shall be empty.
11			4.	Paved areas shall be clean, free of dirt, oily stains and other such blemishes
12			5.	Exterior lights and diffusers are clean and free of dust.
13		G.	Interio	r Cleaning shall include but not be limited to the following:
14			1.	Remove all labels, stickers, tags, and other such items which are not required by code as permanent
15				labels.
16			2.	All interior glazing surfaces, including mirrors, have been professionally cleaned and are free of dust and
17				streaking.
18			3.	All interior surfaces have been cleaned of excess materials such as paint, sealants, etc and have been
19				wiped free of dust.
20			4.	Interior metals, fixtures, and trim have been cleaned free of dust and oily residues
21			5.	Carpet flooring has been thoroughly cleaned; vacuumed free of dust, excess glues and other stains
22				removed per manufacturers use and care instructions.
23			6.	Resilient flooring has been thoroughly cleaned; vacuumed free of dust, excess glues and other stains
24				removed, mopped and buffed per manufacturers use and care instructions.
25			7.	Interior non-occupied concrete floors shall be broom cleaned, vacuumed free of dust, excess glues and
26				other stains removed per manufacturers use and care instructions.
27			8.	Light fixtures, lamps, diffusers and other such items have been dusted and cleaned as necessary.
28				
29	3.5.	CALL B	BACK W	ORK
30		А.	The GC	shall be responsible for ensuring that any contractor returning to the project site for completion or
31			correct	ion work has re-cleaned and restored the area to the levels described in section 3.4 above upon
32			comple	etion of the work. This shall include but not be limited to the following:
33			1.	The immediate area(s) where work was completed.
34			2.	Adjacent areas where dust or debris may have traveled.
35			3.	Other areas occupied during the completion of the call back work.
36			4.	Path of entrance/exit, to/from the area(s) of work.
37				
38				
39				
40				END OF SECTION
41				

		SECTION 01 74 19
		CONSTRUCTION WASTE MANAGEMENT AND DISPOSAL
PART	1 – G	iENERAL
1	- 0 1.1.	SUMMARY
1	1.2.	RELATED SPECIFICAITONS
1	1.3.	CITY ORDINANCES
1	1.4.	DEFINITIONS
1	1.5.	PERFORMANCE REQUIREMENTS
1	1.6.	SUBMITTALS AND DELIVERABLES
1	1.7.	QUALITY ASSURANCE
1	1.8.	WASTE MANAGEMENT PLAN
PART	2 – PI	RODUCTS – THIS SECTION NOT USED
PART	3 - EX	KECUTION
3	3.1.	PLAN IMPLEMENTATION
3	3.2.	HAZARDOUS AND TOXIC WASTE
3	3.3.	GENERAL GUIDELINES FOR ALL WASTES
3	3.4.	GUIDELINES FOR RECYCLABLE, RE-USABLE, AND SALVAGEABLE WASTE
3	3.5.	GUIDELINES FOR DISPOSAL OF WASTES
PART	<u>1 – G</u>	<u>ieneral</u>
1.1.	SUM	MMARY
	A.	This specification includes administrative and procedural requirements for the recycling, re-use, salvaging, ar
		disposal of non-hazardous construction and demolition waste.
	В.	The General Contractor (GC) shall be fully responsible for complying with all applicable ordinances and other
	5.	such regulatory requirements during the execution of this contract.
1.2.	REL	LATED SPECIFICAITONS
	Α.	01 29 76 Progress Payment Procedures
	В.	01 31 23 Project Management Web site
	C.	01 32 19 Submittals Schedule
	D.	01 33 23 Submittals
	Ε.	01 77 00 Closeout Procedures
	F.	Other Divisions and Specifications that may address the proper disposal of construction or demolition waste
		pertains to work being conducted under that particular specification.
1.3.	CIT	Y ORDINANCES
	Α.	There are two (2) Madison General Ordinances (MGO) that the City of Madison has regarding construction ar
		demolition waste.
		1. MGO 10.185, Recycling and Reuse of Construction and Demolition Debris, describes the requirements
		associated with this ordinance including definitions, documentation requirements, and penalties.
		2. MGO 28.185, Approval of Demolition (Razing, Wrecking) and Removal, describes the requirements
		associated with applying for and receiving a demolition permit.
	В.	All City of Madison, Board of Public Works, contracts being conducted by City Engineering, Facility Manageme
		for construction, remodeling, or demolition shall comply with the above ordinances regardless of project type
		size.
1 4	D	
1.4.		Cleans Untreated and unpainted material free of contemination caused by allo calcents activity and other
	А.	clean: Untreated and unpainted material, free of contamination caused by oils, solvents, Caulks, and other
	P	Circliffication and Domalitian Dabria. Materials resulting from the construction remodeling resulting the
	в.	construction and Demonstron Debris: Waterials resulting from the construction, remodeling, repair, and
	~	demonition of utilities, structures, buildings, and roads.
	Ċ.	Disposal: UTT-site removal of construction and demolition debris and the subsequent sale, recycling, reuse, o
	-	deposit in authorized landfill or incinerator.
	D.	Hazardous: Exhibiting the characteristics of hazardous substance, i.e. ignitability, corrosiveness, toxicity, or
	-	reactivity and including but not limited to asbestos containing materials, lead, mercury and PCBs.
	E.	Non-nazardous: Exhibiting none of the characteristics of a hazardous substance.

1		E.	Nontoxic: Not immediately poisonous to humans or poisonous after a long period of exposure.
2		G.	Recyclable: The ability of a product or material to be recovered at the end of its life cycle and remanufactured
3		0.	into a new product.
4		Н.	Recycle: Any process by which construction or demolition debris is diverted from final disposal as solid waste at
5			a permitted landfill and instead is collected, separated, and/or processed into raw materials for new, reused, or
6			reconstituted products; or for the recovery of materials for energy production processes.
7		I.	Recycler: Any recycling facility, transfer station, or other waste handling facility which accepts construction and
8			demolition debris for recycling, or for other transferring to a recycling facility.
9		J.	Recycling: The process of sorting, cleaning, treating, or reconstituting solid waste and other discarded materials
10			for the purpose of preparing the material to be recyclable. Recycling does not include burning, incinerating or
11			thermally destroying waste.
12		К.	Return: To give back reusable items or unused products to vendors for credit.
13		L.	Reuse: Shall mean any of the following:
14			1. The on-site use of reprocessed construction and demolitions debris.
15			2. The off-site redistribution of a material, for use in the same manner or similar manner at another
16			location.
17			3. The use of non-toxic, clean wood as an alternative fuel source.
18		M.	Salvage: To remove a waste material from the project site for resale or reuse by the Owner or others.
19		N.	Toxic: Poisonous to humans either immediately or after a long period of exposure.
20		0.	Trash: Any product or material unable to be re-used, returned, recycled, or salvaged.
21		Ρ.	Waste: Extra materials or products that have reached the end of its useful life or its intended use. Waste
22			includes salvageable, returnable, recyclable and re-useable construction and demolition materials, and trash.
23			
24	1.5.	PERFC	RMANCE REQUIREMENTS
25		Α.	The GC shall develop a Waste Management Plan that results in end-of-project rates for salvage/recycling/reuse
26			of 95 percent (minimum) by weight of the total waste generated by the Work. Percentages may be adjusted on
27		-	a project by project basis depending on selected LEED goals associated with the project.
28		В.	The GC shall salvage or recycle 100 percent of all uncontaminated packaging materials including but not limited
29			to the following:
30			1. Paper
31			2. Cardboard
32			3. Beverage containers
22 21			4. BUXES 5. Blastic Shoot and film
24 25			5. Plastic Sheet and Initia
36			7. Wood crates and nallets
30			Noou craces and parets Plastic pails and buckets
38		C	Promote a resourceful use of supplies and materials through proper planning and handling. Generate the least
39		С.	amount of waste possible by minimizing errors, poor planning, breakage, mishandling, contamination or other
40			similar factors
41		D.	Use all reasonable means to divert construction waste from landfills and incinerators through recycling, reuse, or
42			salvage as appropriate.
43			
44	1.6.	SUBM	ITTALS AND DELIVERABLES
45		Α.	The GC shall provide their completed Waste Management Plan to the Project Management Web Site as a
46			submittal for review by the Project Architect and City Project Manager.
47			1. See item 1.8 below for Waste Management Plan submittal requirements.
48			2. The Waste Management Plan shall be completed, submitted, and approved as a pre-requisite for
49			Progress Payment number 1.
50			3. Copies of all documentation required by this specification shall be submitted to the appropriate Project
51			Management Web Site Library. Documentation shall be reviewed by the City Project Manager during all
52			Progress Payment reviews for compliance and accuracy.
53		В.	The Waste Management Coordinator shall provide copies of items 1 through 5 below to the appropriate Project
54			Management Web Site Library and shall update the Waste Management Summary Log to reflect the records
55			being submitted.
56			1. Records of Donations: Indicate receipt and acceptance of itemized salvageable waste donated to
57			individuals or organizations. Indicate if the organization is tax exempt.

		2.	Records of Sales: Indicate receipt and acceptance of itemized salvageable waste sold to individuals or
			organizations. Indicate if the organization is tax exempt.
		3.	Recycling and Processing Facility Records: Indicate receipt and acceptance of recyclable waste by
			recycling and processing facilities licensed to accept them. Include manifests, weight tickets, receipts ar
			invoices.
		4.	Landfill and Incinerator Disposal Records: Indicate receipt and acceptance of waste by landfills and
			incinerator facilities licensed to accept them. Include manifests, weight tickets, receipts and invoices.
		5.	Statement of Refrigerant Recovery: The Refrigerant Recovery Technician responsible for recovering
			refrigerant shall provide the GC with a statement indicating all of the following:
			 All recovery was performed according to EPA Regulations.
			b. All refrigerant present was recovered; indicate the total quantity recovered by unit.
			c. Date of Recovery.
			d. Name, address, company name, and phone number of technician performing the recovery.
			e. Technician shall sign and date the statement.
	C.	LEED S	Submittal: The GC shall provide the following information using the appropriate LEED letter template upo
	-	projec	t completion: indicating that the requirements of the credit have been met. NOTE: This requirement sh
		only a	noly to projects having a LFFD certification goal
		1	Total waste material generated
		2	Total waste material diverted by diversion method: recycling salvage re-use etc
		∠. 2	Which waste streams have been diverted: minimum four different streams required to achieve LEED
		э.	crodit
		л	ureur. Statement that the credit requirements have been met
		4. F	Statement that the treat requirements have been met.
		5.	oc shah sign the letter.
	~		
1.7.	QUA	LITY ASS	
	А.	Waste	Management Coordinator: The GC shall be responsible for designating a Waste Management
		Coord	inator. Coordinator may be the GC Supervisor, GC Project Manager or other member of the GC staff
		having	; knowledge of proper waste management procedures and all applicable regulations.
	В.	Regula	atory Requirements: comply with all hauling and disposal regulations of authorities having jurisdiction.
	C.	The W	aste Management Coordinator shall comply with Specification 01 31 19 Project Meetings, Section 3.7.B.1
		and co	onduct a Waste Management Conference at the job site. This conference shall be repeated as necessary a
		additio	onal trades are added to the Work. The conference shall include but not be limited to the following:
		1.	Identify the Waste Management Coordinator; provide trade contractors with name, phone, and email
			information.
		2.	Review and discuss the Waste Management Plan and the roles of the Coordinator.
		3.	Review the requirements for documenting and reporting procedures of each type of waste and its
			disposition.
		4.	Review procedures for material separation; indicate availability and locations of containers and bins.
		5.	Review procedures for periodic waste collection and transportation to recycling and disposal facilities.
		6.	Review waste management procedures specific to each trade.
	D	Refrig	reant Recovery Technician Qualifications: Certified by EPA-approved certification program
	υ.	nemb	
18	W/45		ΔGEMENT DI ΔΝ
1.0.	۵۳۸J		and a new consisting of waste identification, a waste reduction work plan, and cost/revenue analysis
	л.	Indicat	te quantities by weight or volume. Use the same units of measure throughout the waste management
		indica	te quantities by weight of volume. Use the same units of measure throughout the waste management
		pian.	
		1.	Waste identification: Indicate anticipated types and quantities of site clearing, demolition waste, and
			construction waste that will be generated during the execution of this contract. Include assumptions fo
			the estimates.
		2.	Waste Reduction Work Plan: The work plan shall consist of but not be limited to all of the following:
			a. Identify methods for reducing construction waste. Re-using, framing and forming materials, re-
			planning material cuts to minimize waste, etc.
			b. Identify what types of materials will be recycled. Provide lists of local companies that receive
			and/or process the materials. Include names, addresses, and phone numbers.
			c. Identify what types of materials will be disposed of and whether it will be disposed of in a landfi
			facility or by incineration facility. Provide lists of local companies that receive and/or process th
			materials. Include names, addresses, and phone numbers.
			d. Identify methods to be used on site for separating waste including all of the following

1 2 3			 i. Sizes of containers to be used. ii. Labels to be used on the containers to identify the type of waste allowed in the container. iii. Designated locations on the project site for waste material containers.
4 5		В.	If project requires demolition incorporate the ordinance required (MGO 28.185) Recycling and Reuse Plan into
5		C	Provide all of the following for the Waste Management Coordinator:
7		с.	1 Name employer employer address phone number and email address of the designated coordinator
8			a. The GC shall also provide this information with the required Project Directory Submittal at the
9			beginning of the project.
10		D.	If at the option of the GC, they choose to contract with a Waste Management Disposal Company that allows
11			comingled and unsorted waste materials, the GC shall include with their Waste Management Plan the following:
12			1. Name, address, phone number, state permitting information, and other pertinent information about the
13			disposal company.
14 15			 Documentation from the disposal company indicating company policies and procedures regarding comingled and unsorted waste materials to include:
16			a GC responsibilities on the project site
17			h Disposal company procedures for receiving sorting recycling and disposing of comingled and
18			unsorted waste material.
19			
20	PART	2 – PRO	DUCTS – THIS SECTION NOT USED
21			
22	PART	3 - EXEC	CUTION
23			
24	3.1.	PLAN	IMPLEMENTATION
25		Α.	Implement the approved waste management plan. Provide adequate containers, storage space, signage,
26			transportation and other items required to implement the plan during the execution of this contract.
27		В.	The GC and Waste Management Coordinator shall be responsible for monitoring and reporting the status of the
28			Waste Management Plan and shall monitor the waste management practices on site as frequently as needed.
29		C.	Train all workers, sub-contractors, and suppliers on proper waste management procedures as appropriate for
30			the work being conducted on the project site.
31			1. Distribute the waste management plan to everyone concerned within seven (7) days of submittal
32			approval.
33			2. Distribute the waste management plan to new workers, sub-contractors, and suppliers when they first
34			appear on the project site.
35			3. Conduct additional training as needed during the execution of the contract to keep a positive focus on
36			the waste management plan.
37 38		D.	and other adjacent and used facilities.
39			1. Designate and label specific areas on the project site necessary for separating materials to be salvaged,
40			recycled, reused, donated, and sold.
41			2. Comply with any specification or regulatory requirements pertaining to dust, dirt, environmental
42			protection, and noise control.
43			
44	3.2.	HAZAF	RDOUS AND TOXIC WASTE
45		А.	The Owner shall be responsible under separate contract for the removal of any asbestos related materials. All
46		_	other materials shall be removed by the GC.
47		В.	All hazardous and toxic waste shall be separated, stored, and disposed of according to all applicable regulations.
48		C.	All nazardous and toxic materials on site snall have a Material Safety and Data Sheet (MSDS) available that
49			indicates storage requirements, emergency information, and disposal requirements as necessary.
50		CENE	
51	3.3.	GENER	RAL GUIDELINES FOR ALL WASTES
52 53		А.	cite
55		R	All revenues savings rebates tax credits and other such incentives received from recycling reusing or
55		υ.	salvaging waste materials shall accrue to the GC unless specified otherwise in the contract documents
56		C.	Separate recyclable, reusable, and salvageable waste from other waste materials, trash, and dehris excent where
57			Waste Management Disposal Company allows comingled waste materials, see section 1.8.D above.

1			1.	Separate by type in appropriate containers or designated areas according to the approved waste
2				management plan away from the construction area. Do not store within the drip lines of existing trees.
3			2.	Inspect containers and bins frequently for contamination and inappropriately sorted materials. Remove
4				contaminated materials and resort as necessary.
5			3.	Stockoile bulk materials such as sand, topsoil, stone, etc., on site away from the construction area and
6			5.	without intermixing with other materials. Place grade and shape stockniles to drain surface water and
7				cover to prevent windflown dust. Do not store within the drin lines of existing trees
2 2			4	Whenever possible store items of the ground and/or protect them from the weather
0			4.	whenever possible store items on the ground and/or protect them nom the weather.
9	2.4			
10	5.4.	GOIDE	LINES F	'UK RECTLIADLE, RE-USADLE, AND SALVAGEADLE WASTE
11		А.	The to	nowing guidelines is not a complete or all inclusive list and shall be adjusted as needed by the methods
12		_	and pr	ocedures identified in the Waste Management Plan.
13		В.	Aspha	It Paving: Break-up into transportable pieces or grind, transport to an authorized recycling facility.
14		C.	Carpet	ι and Pad: Separate carpet and pad scraps, containerize and transport to an authorized recycling facility.
15		D.	Ceiling	g System Components: Suspended ceiling system components shall be sorted by material type as follows:
16			1.	Broken, cut, or damaged tiles shall be containerized, transport to an authorized recycling facility.
17			2.	Damaged, or cut tracks, trim and other metal grid system components shall be sorted with other metals
18				of similar types, palletize, transport to an authorized recycling facility.
19		E.	Clean I	Fill: When allowed by Division 31 Specifications; concrete, masonry, stone, asphalt pavement, sand and
20			other s	such materials may be used as clean fill on this project site. The GC shall verify with the Project Architect,
21			Structi	ural Engineer, or Civil Engineer as necessary prior to using any materials as clean fill. Materials shall be
22			proces	ssed, placed, and compacted as specified. If not being re-used on site, transport to an authorized recycling
23			facility	
24		F	Clean	Wood Materials: Including but not limited framing cutoffs wood sheathing or papeling materials
25			structi	ural or engineered wood products and nallets or crates. Clean Wood shall be free of naints stains oils
26			nrecer	varives and other such contaminates
20			1	Useable pieces shall be sorted by type and dimension, hundled and transported off site by the GC or
27			1.	oseable pieces shall be softed by type and dimension, bundled and transported on site by the GC of returned to the supplier
20			2	Required to the supplier.
29			2.	Non-useable pieces shall be panetized or containenzed, italisport to an autorized recycling facility.
30			3.	clean, uncontaminated sawdust and wood snavings shall be bagged, transport to an authorized recycling
31		C	C	facility.
32		G.	Concre	The Break-up into transportable pieces, remove all reinforcing and other metals, transport to an
33			author	rized recycling facility.
34		Н.	Glass F	Products: Shall be sorted by types, do not include light fixture lamps and bulbs. Products broken in
35			shipme	ent shall be returned to the supplier. Broken or cracked items still in frames shall be taped to prevent
36			furthe	r breakage and injury to workers. Transport to an authorized recycling facility.
37		I.	Gypsu	m Board: Stack large clean pieces on wooden pallets or container, store in a dry location, transport to an
38			author	rized recycling facility.
39		J.	Light F	ixture Lamps and Bulbs: Fluorescent tubes shall be containerized, transport to an authorized recycling
40			facility	1.
41		К.	Mason	ıry and CMU: Remove all metal reinforcing, anchors, and ties, clean undamaged pieces and neatly stack on
42			pallets	s, transport damaged pieces to an authorized recycling facility.
43		L.	Metals	s: Sort metals by type as follows, this does not include piping:
44			1.	Architectural metals including but not limited to siding, soffit, and roofing panels shall be sorted by
45				material, palletize or bundle as needed and transport to an authorized recycling facility.
46			2.	Structural steel, sort by size and type; palletize and transport to an authorized recycling facility.
47			3.	Miscellaneous metals such as aluminum, brass, bronze, etc shall be sorted by type, containerized or
48				palletized as necessary, transport to an authorized recycling facility.
49		M.	Packag	ging and shipping materials
50			1	Cardboard boxes and containers: Breakdown all cardboard boxes and containers into flat sheets. Bundle
51				and store in a dry location until transported for recycling
52			2	Pallets:
52			۷.	whenever nessible require deliveries using pallets to remove them from the project site
55				a. whenever possible require derivenes using parties to remove them to other companies for here to other companies for reusing them or providing them to other companies for
54				b. Ready stack panets in preparation for reusing them of providing them to other companies for salvage or relice
55				salvage ULLE-USE.
50 57				c. Break down pallets into component wood pieces that comply with the requirements for recycling
57				clean wood materials. Neatly stack or palletize pieces in preparation for transportation.

1 2 3 4 5 6 7 8 9 10 11 12 13 14		N. O. P.	 Crates: Break down crates into component wood pieces that comply with the requirements for recycling clean wood materials. Neatly stack or palletize pieces in preparation for transportation. Polystyrene Packaging: Separate and bag materials. Piping and conduit: Reduce all piping and conduit to straight lengths, sort and store by size, material and type. Remove supports, hangers, valves, boxes, sprinkler heads, and other such components, sort and store by size, material and type. Transport to authorized recycling facilities according to material types. Roofing: Roofing materials shall be sorted and containerized by type, transport to authorized recycling facilities according to material types. Site-Clearing Waste: Sort all site waste by type. Only stockpile soils types and quantities required for re-use on the project site. All remaining quantities shall be transported off site to an authorized facility that receives such materials. Brush, branches, and trees with no marketable re-use shall be transported to facilities for chipping into mulch. Trees with a marketable re-use shall be salvaged and transported to facilities that specialize in processing transported to facilities that sp
15 16			trees for future use as wood products.
10	3.5.	GUID	ELINES FOR DISPOSAL OF WASTES
18		A.	The following guidelines shall be adjusted as needed by the methods and procedures identified in the Waste
19			Management Plan.
20		В.	Any waste that is contaminated, organic, or cannot be recycled, re-used, or salvaged shall be legally disposed of
21			in an authorized landfill or incinerator. Disposal methods shall follow all applicable regulatory requirements.
22		C.	No waste material of any kind, except those types designated as clean fill in section 3.4 above, shall be allowed
23		_	to be buried on the project site at any time.
24		D.	No burning of any kind of waste material shall be permitted on this project site at any time.
25		E.	Paint and Stain: Paints, stains, and their containers shall be disposed of as follows:
26			1. Whenever possible containers should be thoroughly cleaned immediately after emptying and sorted with
27			as appropriate (metal or plastic) for recycling
20			2. Empty containers, regardless of type of base material, may be disposed of with lids of with general
29			gai Dage.
21			5. Eater paint may be placed with general gal bage in property solidined as follows.
32			harden Protect cans from rain and freezing
32			h large amounts (more than one inch). Mix paint with equal amounts of cat litter, stir and allow to
34			completely dry. Alternate method: mix with commercial paint hardener.
35			4. Oil-based or combustible paints and stains, regardless of liquid or solid, shall be transported to an
36			approved facility that takes such items such as Dane County Clean Sweep Sites.
37		F.	Treated Wood Materials: Treated wood materials including but not limited to wood that has been painted,
38			stained, or chemically treated shall not be recycled or incinerated.
39			
40			
41			
42			END OF SECTION
43			

1 2				SECTION 01 76 00 PROTECTING INSTALLED CONSTRUCTION
3				
4	PART	1-G	ENERAL	
5		1.1.	SUMMA	RY1
6		1.2.	QUALITY	ASSURANCE1
7		1.3.	RELATED	SPECIFICATIONS
8	PART	2 - PF	RODUCTS .	2
9		2.1.	FENCING	MATERIALS AND BARRICADES2
10		2.2.	EROSION	I CONTROL PROTECTION2
11		2.3.	INTERIO	REAL STREET A STREET
12	PART	3 - EX	ECUTION	
13		3.1.	GENERAI	L EXECUTION REQUIREMENTS
14		3.2.	PROTECT	ADJACENT PROPERTIES
15		3.3.	PROTECT	LANDSCAPING FEATURES4
16		3.4.	PROTECT	UTILITIES4
17		3.5.	PROTECT	PUBLIC RIGHT OF WAY4
18		3.6.	PROTECT	STORED MATERIALS
19		3.7.	PROTECT	WORK - EXTERIOR
20		3.8.	PROTECT	WORK - INTERIOR
21				
22	PART	1 – G	IENERAL	
23				
24	1.1.	SUI		
25		А.	ine pu	irpose of this specification is to provide clear responsibilities, guide lines, and requirements related to
26		_	provid	ing protection to already installed construction.
27		в.	Airead	y installed construction shall include but not be limited to the following:
28			1.	Any existing site reature such as pavement, curbs, drainage reatures, utilities, landscaping reatures (trees,
29				shrubbery, plantings, flagpoles, etc) and other such exterior items not associated with the building
30			2	whether on or adjacent to the project site.
31			2.	Any existing structure on or adjacent to the project site.
32			3.	Any existing interior work that may be adjacent to the new work including all paths of ingress/egress to
33 24			4	areas associated with accessing the work.
34 25			4.	Any existing feature of any kind within the public right-or-way that may be on the project site property,
35		c		adjacent to the project site of across the street from the project site.
30		C.	All COr	itractors shall be ramiliar with the specifications of their Division of work for specific requirements on
3/ 20		Р	Thoro	LION OF LINE WORK.
20		D.	i i i e i e	quitements noted within this specification do not relieve any contractor of the responsibility for
39			compi	rance with any code, statute, or dinance, or other such regulatory requirement having jurisdictional
40 11			autio	ity over these contract documents.
41 12	12	011	λι ιτν λςς	
42	1.2.	<u>ر</u> ن ۸	It chall	I ha the responsibility of every contractor and worker assigned to the project to be diligent in protecting all
43 AA		А.	ovictin	be the responsibility of every contractor and worker assigned to the project to be diligent in protecting an
44 15		в	lt chall	g work, and newly instance construction.
45		Ъ.	mothe	be the deneral contractors (GC) responsibility under the contract to provide an reasonable protection
40			within	this specification to the project as a whole
47 18			1	The GC shall be responsible to ensure any damaged new or existing construction is repaired or replaced
10			1.	at no additional cost to the Contract
			2	The GC at their discretion may direct other contractors to provide and maintain protection of completed
50			۷.	work associated with their Division of Work I E . The carnet installer may be required by the GC to
52				novide carnet protection along traveled naths ingress/egress etc after installation
52		C	It chall	be the responsibility of the GC to ensure that all materials being used to protect installed construction are
54		С.	compa	atible with and/or adjacent to the materials being protected. This shall include but not be limited to the
55			mater	ial used as covering, tapes used to fasten protective materials, etc.
				0,

1			
2	1.3.	RELAT	ED SPECIFICATIONS
3		Α.	Parts of this specification will reference articles within "The City of Madison FACILITIES MANAGEMENT
4			SPECIFICATIONs for Public Works Construction".
5			1. Use the following link to access the FACILITIES MANAGEMENT SPECIFICATIONs web page:
6			http://www.cityofmadison.com/business/pw/specs.cfm
7			a. Click on the "Part" chapter identified in the specification text. For example if the specification
8			says "Refer to City of Madison FACILITIES MANAGEMENT SPECIFICATION 21 0.2" click the link for
9			Part II, the Part II PDF will open.
10			b. Scroll through the index of Part II for specification 210.2 and click the text link which will take you
11			to the referenced text.
12			c. City Standard Detail Drawings (SDD) may be located from the index in Part VIII.
13		В.	Section 01 60 00 Product Requirements
14		C.	Section 01 74 13 Progress Cleaning
15		•.	
16	PART	2 - PRO	DUCTS
17	<u></u>		
18	2.1.	FENCI	NG MATERIALS AND BARRICADES
19		A.	Except where noted in other areas of the construction documents, the responsible contractor shall provide a six
20		<i>,</i>	foot galvanized chain link fence including full height mesh screen at the project lines as shown on the Civil
21			Drawings. For temporary barricade situations, the responsible contractor may provide one of the following that
22			sufficiently provide a sturdy physical barrier and/or visual barrier as necessary for the intended application
23			1. Standard orange construction barrels each with a standard rubber base ring and reflective tape
24			a Provide flashing amber lights as needed to increase night time visibility
25			2 Steel "T" style fence nosts
26			3 4/0" high standard grange construction fence
20			4 0 Ingli standard orange construction refice 4 Traffic harricades
27			5 larsey barriers
20			6 Other types of fencing or harricades typically used in the construction industry
20		R	The contractor responsible for providing the fencing materials and harricades shall also be responsible for
21		Б.	maintaining them. This shall include but not limited to fiving damaged fencing, standing up harrels that have
22			have known residentiation and an entries for him to the provide the first of the promotional at all times
52 22		c	The following foncing and harrisade designations, and their use descriptions shall be used throughout this
24		С.	specification to provide uniformity in describing protection requirements
24 25			Type A Jorsey Parriers, to be used as permanent blocking devices to deputacess to alternate project site
55 26			1. Type A, Jersey Barners, to be used as permanent blocking devices to deny access to alternate project site
50 27			Entrances of exits.
57 20			2. Type b, frame barnades, to be used as temporary blocking devices to deny access to atemate project
20			Site entrances of exits.
39			5. Type C, construction barrels without construction reficing shall be used for hare closures, temporary barrels without construction reficing leastions of C. identify the leastions of an
40			blocking devices to deriv access and the protection of single locations (i.e. identity the location of an
41			A Type D. Construction Parcels with construction foncing where it becomes necessary to surround an object
42			4. Type D, Construction Barrels with construction rencing where it becomes necessary to surround an object
43			with a complete visual barricade and it is impractical of unacceptable to install fence posts. The surround
44 45			shall be constructed in such a manner as to provide a burier zone around and access to the item being
45			protected.
46			5. Type E, steel T Fence Posts shall be used at the project lines, as indicated on the Civil Drawings, with six
47			toot galvanized chain link tencing to surround an object with a complete visual barricade and it is
48			practical to install fence posts. The surround shall be constructed in such a manner as to provide a buffer
49 50			zone around and access to the item being protected. All posts shall be driven installed. Surface mounted
5U F1			posis to only be used for temporary barricades.
51			 Type x, Other rending or barricade types that may be designated and detailed within the construction desuments shall use additional alpha numeric designations.
52			documents shall use additional alpha numeric designations.
53		FDO C:	
54	2.2.	ERUSI	UN CUNTRUL PROTECTION
55		А.	Refer to City of Madison FACILITIES MANAGEMENT SPECIFICATION 210.2 for authorized materials associated
50			with erosion control materials.
5/			

2.3.	INTEF	RIOR FINISH PROTECTION MATERIALS
	Α.	Except where noted in other areas of the construction documents or this specification the responsible
		contractor:
		1. Shall not provide the cheapest or least effective method as an effort to meet any protection requireme
		2. Shall provide materials of sufficient quality, and durability to provide adequate protection based on the
		seasonal conditions and the anticipated duration at the time the protection will be needed.
		3 Shall provide sufficient quantity of protection material to protect the construction as needed
	в	Prior to installing protective measures the responsible contractor shall propose to the GC Project Architect
	υ.	(DA)/Droject Engineer (DE) and City Project Manager (CDM) the proposed plan for protection, materials to be
		(rA)/rioject Engineer (rE) and City rioject Manager (Crivi) the proposed plan for protection, materials to be
		used and samples as necessary.
		alternate proposals.
PART	3 - EXE	ECUTION
3.1.	GENE	FRAL EXECUTION REQUIREMENTS
0.1	Δ	The GC shall be responsible for ensuring all of the following procedures and requirements are implemented as
	/	needed for the duration of the Work performed under this contract
	B	The GC shall also be responsible for the following:
	Б.	1 Reporting any incident of damage to existing property right of way, or utility to the CPM immediately
		1. Reporting any incident of damage to existing property, right-or-way, or damage to the error initiately upon rendering the incident safe, and notifying emergency response teams, and emergency utility cray
		apon rendering the incluent sale, and notifying emergency response teams, and emergency durity crev
		as needed.
		2. Conduct a site wark through phot to leaving at the end of each day to assess.
		a. Protection measures are property in place, provide correction actions as necessary.
		2. Ensure all contractors and workers are being diligent in protecting existing work, and nowly installed
		5. Ensure all contractors and workers are being unigent in protecting existing work, and newly installed
3.2.	PROT	FECT ADJACENT PROPERTIES
	Α.	Whenever possible through the design process the City of Madison shall have previously provided notice to
		adjacent property owners that work will be occurring on or near their property. The City of Madison shall also
		have obtained any permanent or temporary easements that may be necessary to complete any Work on
		adjacent properties.
	В.	It shall be the responsibility of the GC to do the following for all Work under this contract being performed on
		adjacent to the property line:
		1. Contact the adjacent property owner and provide them with information on the work to be done,
		equipment to be used, and estimated duration of the work. Information to be updated and
		communicated to property owner(s) as construction progresses and site conditions change.
		a. If any adjacent property is a rented or leased space the GC shall also make contact and provide
		the same information to the tenants.
		b. Determine from the owner and/or tenants if there are any concerns for children, pets, special
		plantings, or other concerns.
		2. Discuss the following with all contractors performing work on or near the property line.
		a. Work to be completed and timeline.
		b. Concerns of adjacent property owners/tenants from item 1 above.
		c. Which protective measures will be necessary to protect adjacent properties and address the
		concerns of adjacent property owners/tenants.
		3. Ensure all protective measures are placed and maintained during the execution of Work on or adjacent
		the property line. Interact with the adjacent property owners/tenants as needed.
	C.	Any contractor doing work on or adjacent to the property line shall install and maintain any protective measur
	-	identified in the contract documents, this specification, or as directed by the GC.
	D.	The GC shall be responsible for restoring any damage to structure and property located on or adjacent to the
	-	property line.
		property most
		its original condition or better.
		 Restoration shall include but not be inflited to repair or replacement using like materials and misnes to its original condition or better. Restoration of landscaping materials shall include watering of any seed, sod, or other planting of any ki
		 Restoration shall include but not be infitted to repair of replacement using like materials and misnes to its original condition or better. Restoration of landscaping materials shall include watering of any seed, sod, or other planting of any ki for a reasonable period of time to encourage germination and root development.

1				
2	3.3.	PRO	TECT LA	INDSCAPING FEATURES
3		Α.	Exce	pt where specifically stated in other areas of the construction documents the following minimal protection
4			requ	irements shall apply under this section.
5			1.	Whenever possible do not install new landscape features until exterior building construction has been
6				completed, equipment such as scatfolding and lifts are no longer needed and have been removed, and
/				heavy equipment operation is no longer required.
8			2.	Whenever possible remove and temporarily store all existing landscape features such as benches, waste
9				receptacles, signage, and other such features that will be within the area of Work that can be removed.
10			3.	Landscape features that cannot be removed such as flag poles, light poles, light bollards, etc. shall be
11				protected with Type D fencing for areas on pavement or Type E fencing for areas on soil.
12			4.	Planting beds shall be protected using Type E fencing around the exposed perimeter of the planting bed
13			-	as needed.
14			5.	The City of Madison FACILITIES MANAGEMENT SPECIFICATION 107.13 shall apply to all tree protection in
15				and around the project site at all times.
15	2.4	000		
10	3.4.	A		(ILITIES)
10		А.	nrote	contractor shall be responsible for nothying an utilities to determine emergency response procedures and
20			1	This includes requesting utility marking through Diagons Hotling
20			1.	Call 911 or 1 900 242 9511 to request a public utility locato
21				a. Call S11 of 1-600-242-6511 (0 request a public utility locate b. For americancy locate call (262) 422-7010 or (877) 500 0502
22			2	D. For energency locate call (202) 452-7910 of (877) 500-9592
23			۷.	available prior to calling a private utility locating company
24		R	Evce	available prior to carring a private drinky locating company.
25		υ.	requi	irements shall apply under this section
20			1	Hydrants Jamp nosts, electrical transformers, and other utility nedestals shall be protected with Type D
28			1.	fencing for areas on navement or Type E fencing for areas on soil. Eence nosts shall be located so as to
29				not he directly over the utility main
30			2	Storm sewer structures in navement shall have proper inlet protection according to City of Madison
31			2.	FACILITIES MANAGEMENT SPECIFICATION 210 1(g) and Type C Construction Barrels when necessary
32			3	Storm sewer structures in turf and other landscaped areas shall have proper inlet protection according to
33			5.	City of Madison FACII ITIES MANAGEMENT SPECIFICATION 210.1(g) and Type E fencing for areas on soil.
34			4.	Stormwater management features such as greenways, retention/detention ponds, bio-filtration ponds
35				and other such features shall be properly protected according to the appropriate erosion control
36				measure specified on the Erosion Control Plan. See multiple sections of City of Madison FACILITIES
37				MANAGEMENT SPECIFICATION 210.1
38				a. For the protection of hard to see items such as structures, castings, inlets, etc. in grassy areas
39				provide Type E fencing for areas on soil.
40				c. For the protection of storm water management features having special soils and plants such as
41				bio-filtration ponds provide Type E fencing for areas on soil.
42			5.	Other structures and covers including but not limited to cleanouts, wiring hand holes, valve boxes, access
43				structures, grease trap structures, etc shall be protected as follows:
44				a. Provide Type E fencing for areas on soil.
45				b. When paving operations are complete provide a construction barrel or cone near structures as
46				necessary depending on required heavy construction traffic.
47				
48	3.5.	PRO	ΓΕСΤ Ρι	JBLIC RIGHT OF WAY
49		A.	Exce	pt where specifically stated in other areas of the construction documents the following minimal protection
50			requ	irements shall apply under this section.
51			1.	All public right-of-way (area from behind the sidewalk to the centerline of the street) shall remain open
52				and accessible except during periods of active work. At such times the public right of way shall be
53				properly closed and signed as referenced in City of Madison FACILITIES MANAGEMENT SPECIFICATION
54				107.9.
55			2.	Bus stops and bus stop structures shall remain accessible at all times.
56			3.	Traffic signage and traffic signals, traffic control boxes shall be protected with Type D fencing for areas on
57				pavement or Type E fencing for areas on soil.

1			a. Protection at traffic signage/signals shall not obstruct the viewing of the sign/signal for its
2			intended purpose at any time.
3		В.	When additional protection for traffic control is required, the use of barricades, guardrails, lane closures and
4			other such procedures will be detailed within the construction documents.
5		C.	When additional protection for overhead sidewalk cover is required the contract documents shall indicate the
6			specific location and structural requirements of the protective structure.
/ 8	36		ECT STORED MATERIALS
Q	5.0.		All contractors shall refer to Specification 01.60.00 Product Requirements for all storage and protection
10		Α.	requirements of huilding materials and products delivered to the site
11			
12	3.7.	PROT	ECT WORK - EXTERIOR
13	-	A.	Provide all temporary services that may be required to protect the installed material from heat, cold, humidity,
14			etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing.
15		В.	Open trenches, pits, and other such excavations shall be properly covered, lined, or shored as needed during
16			periods of inclement weather to prevent the caving of soils onto existing work in progress. Refer to the
17			appropriate specifications and/or regulatory requirements governing this type of work as necessary.
18		C.	Provide adequate protection at all openings with heavy duty tarps, plastic sheathing, or wood framing and
19			sheathing as needed to protect interior work in progress from inclement weather as needed.
20		D.	Protect exterior finishes of all kinds with heavy duty tarps or plastic sheathing as needed while landscaping is
21			being installed through full germination of seeded areas or installation of filter fabric and mulches to keep dust,
22			dirt, and mud off of finished exterior surfaces.
23		Ε.	Designate specific curb mounting points and provide wood blocking where small vehicles, skid loaders and other
24		_	such equipment may need access to areas being landscaped.
25		F.	Provide plywood turning pads for skid loaders to turn on to prevent fire marking on new pavement.
26		G.	Do not permit the parking of vehicles with any kind of fluid leaks to park on new pavement.
27		н.	ine contractor shall be responsible for cleaning, repairing, or replacing any completed work or work in progress
20			under this specification as deemed necessary by the CPW without additional cost to the contract.
29			
29 30	3.8.	PROT	ECT WORK - INTERIOR
29 30 31	3.8.	PROT A.	ECT WORK - INTERIOR The GC shall do all of the following:
29 30 31 32	3.8.	PROT A.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold,
29 30 31 32 33	3.8.	PROT A.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing.
29 30 31 32 33 34	3.8.	PROT A.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work
29 30 31 32 33 34 35	3.8.	PROT A.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing.
29 30 31 32 33 34 35 36	3.8.	PROT A.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming
29 30 31 32 33 34 35 36 37	3.8.	PROT A.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming into the project site once finish work has begun.
29 30 31 32 33 34 35 36 37 38	3.8.	PROT A.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming into the project site once finish work has begun. Clean dirtied areas and repair/replace damaged areas immediately.
29 30 31 32 33 34 35 36 37 38 39	3.8.	PROT A. B.	 ECT WORK - INTERIOR Provide all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming into the project site once finish work has begun. Clean dirtied areas and repair/replace damaged areas immediately. The contractors responsible for interior work shall be responsible for protecting their work and finishes from dirt,
29 30 31 32 33 34 35 36 37 38 39 40	3.8.	PROT A. B.	 The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming into the project site once finish work has begun. Clean dirtied areas and repair/replace damaged areas immediately. The contractors responsible for interior work shall be responsible for protecting their work and finishes from dirt, mud, snow, spills, splatters, and physical damage after installation as follows:
29 30 31 32 33 34 35 36 37 38 39 40 41	3.8.	PROT A. B.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming into the project site once finish work has begun. Clean dirtied areas and repair/replace damaged areas immediately. The contractors responsible for interior work shall be responsible for protecting their work and finishes from dirt, mud, snow, spills, splatters, and physical damage after installation as follows: Protect vinyl composite, rubber composite, painted/stained concrete, and tiled flooring as follows:
29 30 31 32 33 34 35 36 37 38 39 40 41 42	3.8.	PROT A. B.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming into the project site once finish work has begun. Clean dirtied areas and repair/replace damaged areas immediately. The contractors responsible for interior work shall be responsible for protecting their work and finishes from dirt, mud, snow, spills, splatters, and physical damage after installation as follows: Protect vinyl composite, rubber composite, painted/stained concrete, and tiled flooring as follows: Define foot traffic areas and protect with Ramboard Temporary Floor Protection products as a
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43	3.8.	PROT A. B.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming into the project site once finish work has begun. Clean dirtied areas and repair/replace damaged areas immediately. The contractors responsible for interior work shall be responsible for protecting their work and finishes from dirt, mud, snow, spills, splatters, and physical damage after installation as follows: Protect vinyl composite, rubber composite, painted/stained concrete, and tiled flooring as follows: Define foot traffic areas and protect with Ramboard Temporary Floor Protection products as a minimum basis of design or other protection product(s) compatible with installed flooring product
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44	3.8.	PROT A. B.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming into the project site once finish work has begun. Clean dirtied areas and repair/replace damaged areas immediately. The contractors responsible for interior work shall be responsible for protecting their work and finishes from dirt, mud, snow, spills, splatters, and physical damage after installation as follows: Define foot traffic areas and protect with Ramboard Temporary Floor Protection products as a minimum basis of design or other protection product(s) compatible with installed flooring product if Ramboard is not compatible. Products to be used shall be new.
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45	3.8.	PROT A. B.	 ECT WORK - INTERIOR Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming into the project site once finish work has begun. Clean dirtied areas and repair/replace damaged areas immediately. The contractors responsible for interior work shall be responsible for protecting their work and finishes from dirt, mud, snow, spills, splatters, and physical damage after installation as follows: Define foot traffic areas and protect with Ramboard Temporary Floor Protection products as a minimum basis of design or other protection product(s) compatible with installed flooring product if Ramboard is not compatible. Products to be used shall be new. Tape all edges, seams, etc with a good quality tape that does not leave sticky residue. Do not allow and physical or and the areas the protection product is and the protection product so and the protection product so and the protection protection areas the material between the installed flooring and blooring product is a compatible. Products to be used shall be new.
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47	3.8.	PROT A.	 ECT WORK - INTERIOR Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming into the project site once finish work has begun. Clean dirtied areas and repair/replace damaged areas immediately. The contractors responsible for interior work shall be responsible for protecting their work and finishes from dirt, mud, snow, spills, splatters, and physical damage after installation as follows: Define foot traffic areas and protect with Ramboard Temporary Floor Protection products as a minimum basis of design or other protection product(s) compatible with installed flooring product if Ramboard is not compatible. Products to be used shall be new. Tape all edges, seams, etc with a good quality tape that does not leave sticky residue. Do not allow any debris or other material between the installed flooring and the protection material
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48	3.8.	PROT A.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming into the project site once finish work has begun. Clean dirtied areas and repair/replace damaged areas immediately. The contractors responsible for interior work shall be responsible for protecting their work and finishes from dirt, mud, snow, spills, splatters, and physical damage after installation as follows: Protect vinyl composite, rubber composite, painted/stained concrete, and tiled flooring as follows: Define foot traffic areas and protect with Ramboard Temporary Floor Protection products as a minimum basis of design or other protection product(s) compatible with installed flooring product if Ramboard is not compatible. Products to be used shall be new. Tape all edges, seams, etc with a good quality tape that does not leave sticky residue. Do not allow any debris or other material between the installed flooring and the protection material.
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49	3.8.	PROT A.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming into the project site once finish work has begun. Clean dirtied areas and repair/replace damaged areas immediately. The contractors responsible for interior work shall be responsible for protecting their work and finishes from dirt, mud, snow, spills, splatters, and physical damage after installation as follows: Protect vinyl composite, rubber composite, painted/stained concrete, and tiled flooring as follows: Tape all edges, seams, etc with a good quality tape that does not leave sticky residue. Do not allow any debris or other material between the installed flooring and the protection material. Repair tears immediately, replace worn areas with like material as necessary.
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50	3.8.	PROT A.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming into the project site once finish work has begun. Clean dirtied areas and repair/replace damaged areas immediately. The contractors responsible for interior work shall be responsible for protecting their work and finishes from dirt, mud, snow, spills, splatters, and physical damage after installation as follows: Define foot traffic areas and protect with Ramboard Temporary Floor Protection products as a minimum basis of design or other protects to be used shall be new. Tape all edges, seams, etc with a good quality tape that does not leave sticky residue. Do not allow any debris or other material between the installed flooring and the protection material. Repair tears immediately, replace worn areas with like material as necessary.
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51	3.8.	PROT A.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming into the project site once finish work has begun. Clean dirtied areas and repair/replace damaged areas immediately. The contractors responsible for interior work shall be responsible for protecting their work and finishes from dirt, mud, snow, spills, splatters, and physical damage after installation as follows: Protect vinyl composite, rubber composite, painted/stained concrete, and tiled flooring products as a minimum basis of design or other protection product(s) compatible with installed flooring product if Ramboard is not compatible. Products to be used shall be new. Tape all edges, seams, etc with a good quality tape that does not leave sticky residue. Do not allow any debris or other material between the installed flooring and the protection material. Repair tears immediately, replace worn areas with like material as necessary. Protect carpeted areas as follows: Define foot traffic areas and protect with a minimum of 6mil, clear, polyethylene sheeting 3 feet wide. Products to be used shall be new.
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52	3.8.	PROT A.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming into the project site once finish work has begun. Clean dirtied areas and repair/replace damaged areas immediately. The contractors responsible for interior work shall be responsible for protecting their work and finishes from dirt, mud, snow, spills, splatters, and physical damage after installation as follows: Define foot traffic areas and protect with Ramboard Temporary Floor Protection products as a minimum basis of design or other protection product(s) compatible with installed flooring product if Ramboard is not compatible. Products to be used shall be new. Tape all edges, seams, etc with a good quality tape that does not leave sticky residue. Do not allow any debris or other material between the installed flooring and the protection material. Repair tears immediately, replace worn areas with like material as necessary. Protect carpeted areas as follows: Tape all edges, seams, etc with a good quality tape that does not leave sticky residue. Do not allow any debris or other material between the installed flooring and the protection material. Tape all edges, seams, etc with a pood quality tape that does not leave sticky residue. Do not allow any debris or other material between the installed flooring and the protection material. Tape all edges, seams, etc with a pood quality tape that does not leave sticky res
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53	3.8.	PROT A.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming into the project site once finish work has begun. Clean dirtied areas and repair/replace damaged areas immediately. The contractors responsible for interior work shall be responsible for protecting their work and finishes from dirt, mud, snow, spills, splatters, and physical damage after installation as follows: Protect vinyl composite, rubber composite, painted/stained concrete, and tiled flooring as follows: a. Define foot traffic areas and protect with Ramboard Temporary Floor Protection products as a minimum basis of design or other protection product(s) compatible with installed flooring product if Ramboard is not compatible. Products to be used shall be new. i. Tape all edges, seams, etc with a good quality tape that does not leave sticky residue. Do not allow any debris or other material between the installed flooring and the protection material. Protect carpeted areas as follows: a. Define foot traffic areas and protect with a minimum of 6mil, clear, polyethylene sheeting 3 feet wide. Products to be used shall be new. i. Tape all edges, seams, etc with a good quality tape that does not leave sticky residue. Do not allow any debris or other material between the installed flooring and the protection material.
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54	3.8.	PROT A.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming into the project site once finish work has begun. Clean dirtied areas and repair/replace damaged areas immediately. The contractors responsible for interior work shall be responsible for protecting their work and finishes from dirt, mud, snow, spills, splatters, and physical damage after installation as follows: Protect vinyl composite, rubber composite, painted/stained concrete, and tiled flooring as follows: Protect vinyl composite, rubber composite, painted/stained concrete, and tiled flooring product if Ramboard is not compatible. Products to be used shall be new. Tape all edges, seams, etc with a good quality tape that does not leave sticky residue. Do not allow any debris or other material between the installed flooring and the protection material. Protect carpeted areas as follows: Define foot traffic areas and protect with a minimum of 6mil, clear, polyethylene sheeting 3 feet wide. Products to be used shall be new. Tape all edges, seams, etc with a good quality tape that does not leave sticky residue. Do not allow any debris or other material between the installed flooring and the protection material.
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55	3.8.	PROT A.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming into the project site once finish work has begun. Clean dirtied areas and repair/replace damaged areas immediately. The contractors responsible for interior work shall be responsible for protecting their work and finishes from dirt, mud, snow, spills, splatters, and physical damage after installation as follows: Protect vinyl composite, rubber composite, painted/stained concrete, and tiled flooring as follows: Define foot traffic areas and protect with Ramboard Temporary Floor Protection products as a minimum basis of design or other protection product(s) compatible with installed flooring product if Ramboard is not compatible. Products to be used shall be new. Tape all edges, seams, etc with a good quality tape that does not leave sticky residue. Do not allow any debris or other material between the installed flooring and the protection material. Repair tears immediately, replace worn areas with like material as necessary. Tape all edges, seams, etc with a good quality tape that does not leave sticky residue. Do not allow any debris or other material between the installed flooring and the protection material. Repair tears inducted with a good quality tape that does not leave sticky residue. Do not allow any debris or other material between the installed flooring and the protection material.
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55 56	3.8.	PROT A.	 ECT WORK - INTERIOR The GC shall do all of the following: Provide all temporary services that may be required to protect the installed material from heat, cold, humidity, etc, while materials such as concrete, mortar, sealants, paints, etc, are drying and/or curing. Provide adequate visual and/or physical protection as needed to protect newly completed interior work such as paint, flooring material, sealants, grouts, etc that may be drying and/or curing. Provide adequate space and materials for cleaning boots, tool boxes, supplies, and other items coming into the project site once finish work has begun. Clean dirtied areas and repair/replace damaged areas immediately. The contractors responsible for interior work shall be responsible for protecting their work and finishes from dirt, mud, snow, spills, splatters, and physical damage after installation as follows: Define foot traffic areas and protect with Ramboard Temporary Floor Protection products as a minimum basis of design or other protection product(s) compatible with installed flooring product if Ramboard is not compatible. Products to be used shall be new. Tape all edges, seams, etc with a good quality tape that does not leave sticky residue. Do not allow any debris or other material between the installed flooring and the protection material. Repair tears and protect with a minimum of Gmil, clear, polyethylene sheeting 3 feet wide. Products to be used shall be new. Tape all edges, seams, etc with a good quality tape that does not leave sticky residue. Do not allow any debris or other material between the installed flooring and the protection material. Repair tears and protect with a minimum of Gmil, clear, polyethylene sheeting 3 feet wide. Products to be used shall be new. Tape all edges, seams, etc with a good quality tape that does not leave sticky residue. Do not allow any debris or other material between the installed floo

1		. The elledges some standthe and such that the table to be the large stick and the De
1		i. Tape an edges, seams, etc with a good quality tape that does not leave sticky residue. Do
2		not allow any debris or other material between the installed flooring and the protection
3		material.
4		ii. Repair tears immediately, replace worn areas with like materials as necessary.
5		3. Protect counter tops, cabinets, and other finished surfaces with large sheets of thick cardboard or
6		Ramboard products. Do not allow toolboxes, finish materials, parts and other such items to be placed on
7		finished materials.
8	С.	All protection shall stay in place until the CPM, PA/PE, and GC mutually deem the project is ready for Final
9		Cleaning. The contractors responsible for protecting the work shall be responsible for removing the protection
10		and removing any adhesive residue at that time. Contractors shall only use manufacturer authorized cleaning
11		materials for removing adhesives, etc.
12	D.	Contractors doing work in un-protected areas of finished work shall be required to provide drop cloths and other
13		protection as noted within this specification for the duration of their work.
14		1. Finished areas shall be sufficiently covered to accommodate all equipment, and materials being used to
15		complete the work being done.
16		2. Finished areas shall be sufficiently covered to prevent splatters, over spray, etc when doing touch-up
17		work.
18		3. Contractors who do not provide sufficient protection under this sub-section shall be responsible for any
19		costs associated with cleaning, repairing or replacing already finished construction at no additional cost
20		to the contract.
21		
22		
23		
24		END OF SECTION
25		

				SECTION 01 77 00
				CLOSEOUT PROCEDURES
PART	1 – G	ENERAL .		,
	1.1.	SUMMA	ARY	,
	1.2.	RELATE	D SPECIFICATIO	NS
	1.3.	DEFINIT	IONS	
	1.4.	QUALIT	Y ASSURANCE –	CONSTRUCTION CLOSEOUT
	1.5.	QUALIT	Y ASSURANCE –	CONTRACT CLOSEOUT
PARI	2 – P	RODUCIS	- THIS SECTION	I NOT USED
PARI	3-EX	CONCT		
	3.1. วา	CONSTR		
	3.Z. วว	CONSTR		
	5.5. 21	CONTR		
	3.4. 3.5.	CONTR	ACT CLOSEOUT F	PROCEDURE
	1 0			
PARI	1-0	IEINERAL		
1.1.	SUI	MMARY	urnasa of this s	nocification is to clearly define and quantify the requirements associated with clearing a City
	А.	of M:	adison Public W	pecification is to clearly define and quantify the requirements associated with closing a City orks Contract for facility related work
	в		ontracts have tw	o distinct but related naths. Each nath needs to be properly closed independently in order.
	υ.	to clo	ose the contract	as a whole.
		1.	Construction	closeout is related to closing out all of the Work associated with the construction
			documents.	
			a. It shal	l be the responsibility of all contractors to be fully aware of the required Work and closeour
			requir	ements involved in their individual trades.
		2.	Contract close	eout is related to closing out all of the administrative aspects of the contract in general.
			a. It shal	I be the responsibility of all contractors to be fully aware of the administrative requirement
			requir	ed by the contract and to provide the supporting documentation required.
		3.	Construction	Closeout must be completed before Contract Closeout can begin.
	C.	This s	specification will	I provide general knowledge associated with the following areas:
		1.	Construction	Closeout Requirements
		2.	Construction	Closeout Procedure
		3.	Contract Clos	eout Requirements
		4.	Contract Clos	eout Procedure
		5.	Final Paymen	t and Certificate of Completion
1.2.	REL	ATED SP	ECIFICATIONS	
	Α.	Cont	ractors shall revi	iew all references to other specifications including specifications relating to the execution o
	_	the V	Vork associated	with their Division or Trade.
	В.	Section	on 01 29 76	Progress Payment Procedures
	C.	Section	on 01 31 23	Project Management Web Site (PMWS)
	D.	Section	on 01 32 26	Construction Progress Reporting
	E.	Section	on 01 45 16	Field Quality Control Procedures
	F.	Section	on 01 74 13	Progress Cleaning
	G.	Section	on 01 45 16	Construction waste Management and Disposal
	п.	Section	on 01 70 00	Completion and Correction List
	т. Т	Section	on 01 78 72	Operation and Maintenance Data
	ĸ	Section	on 01 78 36	Warranties
	к. Т	Section	on 01 78 39	As-Built Drawings
	<u>с</u> . М	Section	on 01 78 43	Spare Parts and Extra Materials
	N.	Section	on 01 79 00	Demonstration and Training
	0	Sectio	on 01 91 00	Commissioning
	P	Othe	r requirements a	as noted in the contract documents signed by the General Contractor

1	1.3.	DEFIN	ITIONS	
2		Α.	Substa	Intial Compliance: A letter provided to the City of Madison Building Inspection and signed by the Project
3			Archite	ect indicating that all Work has been completed to a level that would allow Owner Occupancy and that all
4			constr	uction is in compliance with the construction documents. A copy of this letter is also provided to the
5			State o	of Wisconsin Department of Health and Safety as necessary to clear plan review requirements. This letter
6			<u>does n</u>	ot represent construction closeout.
7		В.	Certifi	cate of Occupancy: The Regulatory letter from the City of Madison Building Inspection Department
8			indicat	ing that all regulatory requirements and inspections have been completed and the building may now be
9			occupi	ed for its intended use. This letter does not represent construction closeout.
10		C.	Certifi	cate of Substantial Completion: A letter provided by the Department of Public Works, signed by the City
11			Engine	er indicating that Construction activities are substantially complete. This letter does represent
12			constr	uction closeout and the date of this letter begins the date of the Warranty Period
13		D	Constr	<i>intring Closequit</i> . The point in the contract where all contractual requirements associated the execution of
14		υ.	the Wa	ork as described in the plans specifications, and other documents have been successfully met and the
15			items	described in 1.3 A B and Cabove have been completed
16		E	Einal D	Disperses Deventer . The progress payment associated with achieving Construction closeout as described in
10		с.		Progress Payment. The progress payment associated with achieving construction crossoul as described in
10			1.5.0 a	bove. At this point the contractor may request an nomes associated with the contract be paid with the
10		-	except	
19		F.	Contro	ict Closeout: The point in the contract where all contractual requirements associated with the City of
20		~	iviadise	on, Board of Public Works contract has been successfully met.
21		G.	Final P	<i>ayment</i> : The final contract payment submittal that may be approved by the City of Madison after all
22			contra	ctual requirements of the Public Works Contract have been met and any remaining monies (retainage)
23			due to	the contractor may be released for the Final Payment.
24		_		
25	1.4.	QUALI	ITY ASS	URANCE – CONSTRUCTION CLOSEOUT
26		Α.	All con	tractors shall be responsible for properly executing the construction closeout requirements associated
27			with th	neir Work as described in the specifications governing their Work.
28		В.	The GO	C shall be responsible for all of the following:
29			1.	Ensuring that all contractors have met the construction closeout requirements associated with their
30				Work.
31			2.	Coordinate the collection of all construction closeout deliverables from all contractors, provide the
32				deliverables to the Project Architect and City Project Manager for review as necessary, and ensure all
33				contractors correct deficiencies of deliverables and resubmit as needed for final acceptance.
34			3.	Ensure all closeout requirements identified in the Construction Closeout Checklist below have been
35				completed as intended by the construction documents.
36				
37	1.5.	QUALI	ITY ASS	URANCE – CONTRACT CLOSEOUT
38		Α.	The Cit	ty of Madison, Department of Civil Rights (DCR) monitors contract compliance for construction and
39			procur	ement contracts to ensure that local, state and federal regulations are followed by contractors working on
40			City of	Madison Public Works (PW) projects. DCR will monitor all PW projects from contract award through the
41			final pa	ayment at the close of the project. Contractors will be required to submit reporting paperwork
42			throug	hout the PW project process.
43			1.	Contractors are encouraged to visit the web site identified below for additional information, checklists,
44				forms, and other information provided by DCR as it relates to Contract Compliance.
45				http://www.cityofmadison.com/Business/PW/contractCompliance.cfm
46			2.	Questions regarding the process should be directed to parties and offices as identified on the various
47				forms. documents. and instructions or contact:
48				City of Madison, Department of Civil Rights
49				210 Martin Luther King Jr. Blvd., Room 523
50				Madison. WI 53703
51				(608) 266-4910
52		В.	All Sub	-Contractors have submitted the applicable required documents described in item 1.5 D below to the
53		2.	Gener	al Contractor (GC) for Contract Closeout
54		C		That submitted the required applicable documents described in item 1.5.D below for all contractors to the
55		с.	annror	riate City of Madison Agency per instructions associated with each submittal
55		р		share city of maching for submittal to the City of Madican for Contract Classout may include any/all of the
50		υ.	itome	licted below depending on contract type. It is the cole responsibility of all contractors to know and submit
5/			the re-	instea below depending on contract type. It is the sole responsibility of all contractors to know and submit
- 00			the rec	קטורכט מווע כטוווטוובווכוונמנוטוו ווו מ נווופוץ ומצוווטוו.
	WARN	ER PARK	COMMU	JNITY RECREATION CENTER

		1. 2	Week Emplo	ly Payroll Re	eports tion Reno	rts		
		2. 3	Docu	mentation r	equired fo	or Small Business Enterprise (SBE) goals		
		3. 4.	Other	r documents	as mavb	e required or requested through the Fin	alization Review Proc	ess
			01.101	uooumente				
PART	2 – PR	ODUCTS	6 – THIS	SECTION N	<u>OT USED</u>			
PART	3 - EXI		<u>1</u>					
3.1.	CON	STRUCT		DSEOUT CHE	CKLIST			
	A.	All co	ntracto	rs shall be re	esponsible	e for reviewing the drawings and specifi	cations within their D	visions of Work
		to pro	ovide a d	complete an	d compre	hensive list of all Construction Closeout	Requirements to the	GC.
		1.	The c	hecklist shal	l include	all items identified within the constructi	on documents that re	quire any of the
			follow	ving (and ex	amples) p	rior to moving into Contract Closeout P	rocedures:	
			a.	Document	ts indicati	ng a specified level of performance has	been achieved, such a	as:
				i. Te	st reports	of all types		
				ii. Sta	artup repo	orts		
			b.	Required	document	tation, such as:		
				i. As	-builts and	d record drawings		
				ii. Op	eration a	nd maintenance data		
			с.	Physical it	ems to be	e turned over to the owner, such as:		
				i. Att	tic stock			
				ii. Ke	ys			
			d.	Required	maintena	nce completed, such as:		
				i. Du	cts cleane	ed		
				ii. Filt	ters repla	ced		
			e.	Commissi	oning and	LEED related items and submittals		
			f	Owner an	d Mainter	nance Training		
	В.	Each	list shall	l indicate th	e title of t	he closeout requirement, the associate	d specification of the	requirement, the
		requi	red resu	ult or deliver	able, the	responsible contractor(s), and a column	n to verify the item ha	s been turned in
	•	and c	omplete	ed.				
	C.	The G	C shall	be responsil	ole for all	of the following:	constant Charal Hist	
		1.	Consc	The sheet	the closed	but lists into one master construction ci	oseout Checklist.	
		2	a.	The check	list shall t	De in a tabular data format similar to the	Site for review	
		2. 2	Docuk	a the comp	election of the	const to the Project Management webs	Sile for review.	
	П	J. The G	Resur	work with a	ll contract	tors to amond the Construction Closeou	t Checklist throughou	t the execution o
	D.	the n	roject h	ased on cha	ngos and	modifications as necessary	it checklist throughou	
		the p	l oject b		inges anu	mouncations as necessary.		
		Tit	le	Spec	ification	Description	Responsibility	Completed
	Qu	ality Ma	nageme	ent 01	45 16	All QMO reports have been properly	/ All, GC	
	Ob	servatio	n Repor	rts		responded to, reviewed and closed b the CPM.	Ŷ	
	A	s-Built D	rawing	s 01	78 39	As-Built drawings have been reviewe	d All, GC	
	Tor	ting and	Balanci	ing 72	09.23	Provide final TnB reports indicating	Ηνας	
	103	of H\		23	0.7 2.3	design performance has been achieve	ad horse	
	1	0111						

- A. The timely submittal or completion of closeout requirements shall go hand in hand with the Progress Payment Milestone Schedule that can be found in Specification 01 29 76 Progress Payments. No payments shall be made until all requirements for that payment have been met.
 - 1. The GC and all major Subcontractors, Project Architect /Project EngineerA/E PROJ MGR, and CPM, shall review all requirements for Construction/Contract Closeout during two (2) special meetings.
 - a. The first meeting shall be held at the 50% Contract Total Payment milestone. This meeting shall discuss the requirements associated with various construction/contract closeout documentation and events when they are due with respect to progress payments.

44

45

46

47

48

49

1 2				b.	The second m shall review th	eeting shall be held ne contractors prog	at the 70% Contra ress regarding the	ct Total Payment miles closeout checklist, begi	tone. This meeting n making plans for
3 4					with respect t	o progress paymen	duling training, wh ts.	ere to put attic stock, a	nd when they are due
5 6 7			2.	The G const	3C, A/E PROJ MG truction closeout	R, and CPM, shall ut requirements have	tilize the Construct been met.	tion Closeout checklist t	to ensure that all
/ 8 3	3.3.	CONS	TRUCT		DSEOUT PROCE	DURE			
9		A.	Upon	succes	sful completion :	and final acceptanc	e of all Constructio	n Closeout Requiremer	nts the GC may submit
0			to the	e CPM a	and A/E PROJ Me	GR the request for F	inal Progress Paym	ent (100% contract tota	al, less retainage).
1		В.	The A	/E PRO	J MGR will confi	rm with the design	consultants, CPM, a	and other City of Madis	on staff that all
2			requi	rement	s of the Work ha	ive been completed	l and will do the fol	lowing:	
.3			1.	Appr	ove the final pro	gress payment app	lication		
4			2.	Provi	de the required	signed payment do	cuments to the CPI	N	
5			3.	Provi	de the required	Letter of Substantia	I Compliance to th	e following as required	:
6				a.	State Safety a	nd Building Division			
7				b.	Local Building	Inspection office			
.8				с.	GC				
.9				d.	CPM				
.0 1		C.	The C state	PM sha any of t	III draft the City I the following tha	Letter of Substantia at may still be tied t	I Completion for sign of the contract and,	gnature by the City Eng /or warranty:	ineer. This letter shall
2			1.	Indica	ate that the date	e of the letter shall a	also be the beginni	ng of the Warranty peri	od.
.3			2.	Indica	ate any allowed	due outs, reasons f	or them, and antici	pated dates of finalizat	ion.
24				a.	QMO issues su	uch as off season te	sting of equipment		
5			The	D.	Off season tra	ining of equipment			
b 7		D.	Ine G	c and a	all subcontractor	s shall finalize all w	arranty letters asso	clated with their work	using the date noted
0			On the	e City L fication	etter of Substant	tial Completion, and	a provide the CPIVI	with all warranties as d	CDM may initiate final
0			speci		fthe Einel Drogr	acc Baymont (100%)	contract total loss	rotainaga)	CPIVI IIIdy IIIIIdle IIIIdi
.9			proce	ssing o	i the final Plogre	ess Payment (100%	contract total, less	retailiage).	
1 :	24	CONT				NTS			
2		Δ	The G	C and a	all sub-contracto	rs shall follow all re	quirements associa	ated with documenting	contract compliance
3		74.	and n	rovide	documentation a	as required or requi	ested by DCR or PV	V staff. All contractors	are encouraged to stav
4			curre	nt with	submissions of t	he following docun	nentation:		
5			1.	Weel	kly Payroll Repor	ts no later than the	Progress Payment	equal to 50% of the co	ntract total.
6			2.	Empl	ovee Utilization	Reports	0 ,	•	
57			3.	Agen	t or Subcontract	or Affidavit of Com	pliance with Prevai	ling Wage Rate Determ	ination
8			4.	Prime	e Contractor Affi	davit of Compliance	e with Prevailing W	age Rate Determination	า
9			5.	Docu	mentation requi	red for Small Busin	ess Enterprise (SBE) goals	
0			6.	Othe	r documents as r	maybe required or i	requested through	the Finalization Review	Process
1		В.	Near	the Pro	gress Payment e	equal to 80% of the	contract total the (GC shall request in writi	ng a Finalization
2			Revie	w.Att	hat time DCR or	PW staff shall prep	are a report of all c	ontract documentation	submitted to date. A
3			list of	missin	g items or outsta	anding issues will be	e emailed to the GO	2. No additional follow-	<u>up will be generated</u>
4			by DC	R or PV	<u>N Staff</u> .				
5									
6 3	3.5.	CONT	RACT C	LOSEO	UT PROCEDURE				
7		Α.	The C	ontract	Closeout Proce	dure will not begin	until the Construct	ion Closeout Procedure	has been completed.
8		В.	Wher	the GO	teels they have	successfully met a	I of the Contract Cl	oseout Requirements a	issociated with Section
19		<u> </u>	3.3 at	ove th	e GC may submit	t to the request for	Final Payment to t	he CPM.	
0		C.	The C	PM sha	III sign and subm	it the Final Paymen	t request for proce	essing.	2. 4 1
) 1		ט. ר	DCR a	ind PW	statt shall do a c	complete review of	an accumentation	associated with item 3.	3.A above.
2		E.	ine G		be notified direc	ther outstary line	Lart of any docume	nuation that may still be	e missing, nave
5			incom	ipiete I	and DW staff	til all documentation	sues. It shall be the	e responsibility of the G	c to continue follow-
4 5		c	up wi		and PVV staff UN	ation accorded and	th Contract Classes	signing submitted and activity bas been successfully	Lepieu.
5		г.	wher	tod by		ff the City of Media	on contract Closeo	at has been successfully	romaining monies
7			includ	ling rot		The City of Madis	on shall process the	e i mai r'ayment ui any	remaining monies
8			menue	ing ret	aniage.		SECTION		
ں 						ENDU	JECTION		
١	WARN	ER PAR		IUNITY F	RECREATION CENT	ER			
l		ISION	502 841 11	NIC #171	196	01 7	7 00 - 4		
	CONT	VACI #9	JU∠ IVIUI	413 #1/1		U1 /	, uu - 4		CLUDLOUI PROCEDUKES

			SECTION 01 78 13 COMPLETION AND CORRECTION LIST
PART	1 – GE	NERAL	
1	.1.		с
	1.2. 2. pp/		
PARI	2 – PK		
PARI	3 – EXI	CUTION - THIS SECTION	INOT USED
<u>PART</u>	<u> 1 – GE</u>	NERAL	
1.1.	SUM	MARY	
	A.	The City of Madison h	as developed a multi-faceted Quality Management Program that begins with contract
		signing and runs throu delivered for the cont	igh contract closeout to ensure the best quality materials, workmanship, and product a racted Work.
		1. The Project Ma consultants. ar	anagement Web Site is a Construction Management tool that provides contractors, nd staff a single on-line location for the daily operations and progression of the Work.
		2. The Quality Ma	anagement Observation (QMO) is an ongoing observation of the construction process a
		progresses. Th	ne City of Madison does not use a "Punch List" or "Corrections List" as it is typically kno
		throughout the	e construction industry. The QMO process acts as an "in progress punch list". Work
		identified as no	ot in compliance with the contract documents by the Owner, Owner Representatives,
		Owner Consult	tants, etc. shall be resolved immediately at the Contractor's expense. Unresolved issue
		will be subject	to withholding of progress payment(s) until completed.
		3. Very stringent	expectations are tied to Construction Closeout and Contract Closeout procedures. Spe
		milestones thr	oughout the project need to be met and the milestones are tied to the Progress Payme
	_	Schedule.	
	В.	All contractors shall be	e required to review the specifications identified in Section 1.2 below, and other relate
		specifications identifie	ed therein to become familiar with the terminology and expectations of this Lity of
		Madison Public Works	s contract.
1.2.	RELA	TED SPECIFICATIONS	
	Α.	Section 01 29 76	Progress Payment Procedures
	В.	Section 01 31 23	Project Management Web Site (PMWS)
	C.	Section 01 45 16	Field Quality Control Procedures
	D.	Section 01 77 00	Closeout Procedures
DADT	י ם נ		
PARI	<u>2 - PK</u>		
PART	3 – EX	ECUTION – THIS SECTION	N NOT USED
			END OF SECTION

Page Intentionally Left Blank

			SECTION 01 78 23
			OPERATION AND MAINTENANCE DATA
PART	1 – G	ENERAL	
:	1.1.	SUMMARY	
	1.2.	RELATED SPE	ECIFICATIONS
	1.3.	QUALITY ASS	SURANCE
	1.4.	O&M DATA F	REQUIREMENTS
	1.5.	O&M DATA S	SUBMITTALS
PART	2 – Pl	RODUCTS – TH	HIS SECTION NOT USED
PART	3 - EX	ECUTION	
3	3.1.	O&M DATA F	PREPARATION - GENERAL
-	3.2.	O&M DATA [DRAFT SUBMITTAL
3	3.3.	O&M DATA F	FINAL SUBMITTAL
	3.4.	CONSTRUCT	ION CLOSEOUT
PART	1 – G	ENERAL	
1.1.	SUN	MMARY	
	Α.	The purpo	ose of this specification is to provide clear responsibilities and guide lines related to providing well
		document	ted and complete Operation and Maintenance (O&M) Data related to general facility use, equipment
		systems, fi	inishes, and materials to City of Madison Staff (Owner, Owner Representatives, Maintenance, and
		Custodial I	Personnel) as needed.
	В.	Operation	and Maintenance Data shall apply to both of the following categories except where specific
		requireme	ents are noted under their separate titles as follows:
		1. Op	peration and Maintenance Data: Generally shall mean the owner manual that provides information of
		sta	art-up, shut-down, operation, troubleshooting, maintenance, parts, and other such documentation a
		pei	rtains to all equipment and systems installed under the Work.
		2. Use	e and Care instructions: Where applicable use and care instructions shall also be considered O&M for the second state of the s
		SUC	ch things as flooring, tile, partitions, and other such finishes and trim related items, installed under ti
		Wo	ork.
1 2	DEI		CATIONS
1.2.		Section 01	1 29 76 Progress Payment Procedures
	л. В	Section 01	1 31 23 Project Management Web Site
	C.	Section 01	1 77 00 Closeout Procedures
	D.	Section 01	1 78 13 Completion and Correction List
	F.	Section 01	1 78 19 Maintenance Contracts
	Е.	Section 01	1 78 36 Warranties
	G.	Section 01	1 79 00 Demonstration and Training
	H.	Section 01	1 91 00 Commissioning
	Ι.	Other Divi	isions and Specifications that may address more specifically the requirements for O&M Data.
1.3.	QU	ALITY ASSURA	ANCE
	Α.	All O&M D	Data shall meet the requirements identified in Section 1.4 below.
	В.	All contrac	ctors snall provide U&M Data for each piece of equipment, system, or finish installed during the
		installation	n of the work. U&IVI Data shall be provided to the General Contractor (GC) for verification and
	c	supmittal.	In the responsible for reasing all required OOM Date files from all contractors for world the that all
	L.	files sub-	an be responsible for receiving an required U&IVI Data files from all contractors for verifying that all sitted most the requirements in Section 1.4 below.
		mes submi	ווונפט חופפר נחפ דפקטורפוזופוונג זוז גפטנוטח 1.4 גפוטש.
1.4.	0&	M DATA REQU	UIREMENTS
	Α.	O&M Data	a shall be provided in digital PDF format as follows:
		1. PD)F files shall be complete first generation consumer useable editions of PDF documents as provided b
		any	y of the following:
		a.	Product manufacturer
		b.	Supplier of product

1			2. Acceptable PDF files shall have the following functionality:
2			a. Word searchable
3			b. Key areas are bookmarked
4			c. Table of Contents and/or Index linked to content is preferred whenever possible.
5			3. Scanned printed material, with word searchable capabilities, saved as a PDF, is not acceptable and will be
6		_	rejected without further review.
7		В.	O&M Data shall include but not be limited to the following manufacturers' published information as appropriate
8			for the equipment, system, material, or finish:
9			1. Installation instructions
10			2. Parts lists, assembly diagrams, explosion diagrams
11			3. Wiring diagrams
12			4. Start-up, shut-down, troubleshooting and other related operation procedures
13			5. Lubrication, testing, parts replacement, and other such maintenance procedures
14			6. General use, care, and cleaning instructions
15			7. Special precautions and safety requirements
16			8. A list of certified equipment vendors, service companies, parts suppliers including company name,
17			address, and phone number
18			9. A list of the recommended spare parts to have on hand at all times
19			10. A list by type of all recommended lubes, oils, packing material, and other maintenance supplies
20			11. Copies of final test reports, balance reports, and other related documentation
21			12. Warranty information for equipment and systems
22			
23	1.5.	0&M I	DATA SUBMITTALS
24		А.	O&M Data shall be prepared as identified in this specification and shall be submitted for review as per the
25		_	schedule identified in Specification Section 01 29 76, Progress Payment Procedures.
26		В.	O&M Data Draft submittals will be reviewed for content, procedure, and compliance only. A general critique
27			with recommendations for improvement will be made but re-submittals will not be required.
28		С.	O&M Data Final submittals will be reviewed for content, procedure, and compliance. Re-submittals will be
29			required until such time as each submittal is accepted.
30			
31		<u>NOTE:</u>	Acceptance of O&M Data Final submittals is required to be complete prior to scheduling and conducting owner
32			related training and construction closeout.
33			
34	PARI	<u> – PRO</u>	DUCTS - THIS SECTION NOT USED
35	DADT		
30	PARI	3 - EXEC	
3/	2.1	00.04	
38	3.1.		DATA PREPARATION - GENERAL
39		А.	All contractors shall prepare O&IVI Data for draft and final submission as follows:
40			1. Obtain digital PDF files for each piece of equipment, system, material or finish as described in Sections
41 42			1.4.A.1 dHU 1.4.A.2 dUUVE.
42			2. Verify that all information as described in Section 1.4.8 above is included with the PDF file. Obtain missing information as pagagant for a complete submitted
43			missing information as necessary for a complete submittal.
44		в.	Rename each individual PDF file as follows.
45			1. Do not use special characters such as #, %, &, /, etc. These characters are reserved by the Project
46			Management web Site software the City of Madison uses; however the under-score (or under-bar) '_' is
47			an allowed character.
48			2. Use the following format and examples for renaming your file:
49			a. Format: Equipment name_what_wARNER PARK COMMUNITY RECREATION CENTER
50			EXPANSION_Contract number_year
51			I. Equipment ivame represents the name of any equipment, system, material or finish as
52			designated in the Contract Documents.
53			II. What represents what the file is about
54			III. WAKNEK PAKK COMMUNITY RECREATION CENTER EXPANSION represents the title of the
55			project or contract. A shortened version of the title may be identified by the City Project
56			Manager to be used by all contractors.
5/			iv. <i>Contract number</i> is the specific identification number the Work was bid under and appears
58			on the plan set title sheet and in each sheet title block
	WARN	ER PARK	COMMUNITY RECREATION CENTER

C. D. 3.2. O&M A. B. C. Over Air H Watu 3.3. O&M A. B. C.	All cc O&M O&M W DATA All cc 1. 2. The C and s 1. The F O&M 1. 2. rhead Do handling er Heate M DATA All cc 1.	ii. CPT 2_ ontractors shall submit to 1 Data submission dead 1 Data shall be submitte DRAFT SUBMITTAL ontractors shall prepare Prepare three (3) cor Review all specificati all equipment, syster below and shall indic specification, and a c GC shall be required to r shall return any to the o When acceptable to library on the Project Project Architect, City Pr 1 Data draft submittals a Provide general critic provide all contracto a. Re-submittal Review in detail the 0 a. Re-submittal <u>Title</u> oor Operator (Unit (AHU-3) er (WH-1) FINAL SUBMITTAL prepare complete O8	Use and Care_MPD V the completed digital I lines as described in Sp ed and reviewed as des and submit the follow mplete O&M Data file ons within their Divisi- ms, materials, or finish cate the title (and plan column to verify the ite review all contractors' originating contractor t the GC, they shall uplo t Management Web Si roject Manager, CxA, O and checklist within fif que comments by Divi- rs with information or of the O&M Data sam O&M Data Checklist Specifica 08 36 00 23 00 00 22 30 00	Vest_9876_20 PDF files to the pecification Se scribed in sect ving for an O& samples as de on of Work an identifier whe em has been t samples and that are insuff pad each O&M te. Consulting Sta teen 15 worki sion on O&M in strengths an ples will not b or completene will be require	2011 he GC in sufficient Section 01 29 76, F ctions 3.2 and 3.3 &M Data Draft rev lescribed in sectio ind prepare a com t shall be in tabula hen applicable) of turned in and con d checklists for cor fficient for re-subr M Data draft subn affs and Owner Re king days as follow 1 Data samples suf nd weaknesses of be required. iess. Provide comi red until accepted	t time for the GC to meet the Progress Payment Procedures below. view submittal: in 3.1 above. oplete O&M Data checklist listi ar form similar to the example the O&M Data, the associated mpleted. mpliance with this specificatio mittal. nittal file to the O&M Draft epresentatives shall review th vs: bmitted. Critique is intended their submittals. ments as needed. d.
D. 3.2. O&N A. B. C. Over Air H Watu 3.3. O&N A. B.	O&M O&M VI DATA All cc 1. 2. The C and s 1. The F O&M 1. 2. rhead Dc Heate VI DATA All cc 1. 2.	A Data shall be submitted DRAFT SUBMITTAL Drive of the second shall prepare Prepare three (3) cor Review all specificati all equipment, syster below and shall indice specification, and a conditional system below and shall indice specification, and a conditional system to a second system provide general critice provide general critice provide general critice provide all contractoonal the conditional Review in detail the conditional system a. Re-submittal Review in detail the conditional system a. Re-submittal <u>Title</u> oor Operator <u>submit (AHU-3)</u> er (WH-1) FINAL SUBMITTAL prepare complete OS	and submit the follow mplete O&M Data file ons within their Divisions, materials, or finish cate the title (and plan column to verify the ite review all contractors' originating contractor t the GC, they shall uplot t Management Web Si roject Manager, CxA, (and checklist within fif que comments by Divi rs with information or of the O&M Data sam O&M Data Checklist Specifica 08 36 00 23 00 00 22 30 00	ving for an O& samples as de on of Work an identifier whe em has been t samples and d that are insuffi bad each O&W te. Consulting Sta teen 15 worki sion on O&M in strengths an- ples will not b or completene will be require	&M Data Draft rev described in sectio and prepare a com t shall be in tabula hen applicable) of turned in and con d checklists for cor fficient for re-subr M Data draft subn affs and Owner Re king days as follow 1 Data samples sub nd weaknesses of be required. ness. Provide com red until accepted	view submittal: on 3.1 above. oplete O&M Data checklist listi ar form similar to the example i the O&M Data, the associated opleted. mpliance with this specificatio mittal. nittal file to the O&M Draft epresentatives shall review th ws: bmitted. Critique is intended i their submittals. ments as needed. d.
3.2. O&N A. B. C. Over Air H Watu 3.3. O&N A. B.	M DATA All cc 1. 2. The C and s 1. The F O&M 1. 2. <u>rhead Dc</u> Handling er Heate M DATA All cc 1.	DRAFT SUBMITTAL ontractors shall prepare Prepare three (3) cor Review all specificati all equipment, syster below and shall indic specification, and a c GC shall be required to r shall return any to the o When acceptable to library on the Project Project Architect, City Pr 1 Data draft submittals a Provide general critic provide all contracto a. Re-submittal Review in detail the 0 a. Re-submittal <u>Title</u> oor Operator (Unit (AHU-3) er (WH-1) FINAL SUBMITTAL prepare complete O8	and submit the follow mplete O&M Data file ons within their Divisions, materials, or finish cate the title (and plan column to verify the ite review all contractors' originating contractor t the GC, they shall uplot t Management Web Si roject Manager, CxA, (and checklist within fif que comments by Divi rs with information or of the O&M Data sam O&M Data Checklist Specifica 08 36 00 23 00 00 22 30 00	ving for an O& samples as de on of Work an ies. Checklist identifier whe em has been t samples and that are insuffi bad each O&W te. Consulting Sta teen 15 worki sion on O&M n strengths an ples will not b or completene will be require	&M Data Draft rev described in sectio and prepare a com t shall be in tabula hen applicable) of turned in and con d checklists for cor fficient for re-subr M Data draft subn affs and Owner Re king days as follow 1 Data samples sub nd weaknesses of be required. hess. Provide comp red until accepted	view submittal: on 3.1 above. oplete O&M Data checklist listi ar form similar to the example the O&M Data, the associated opleted. mpliance with this specification mittal. nittal file to the O&M Draft epresentatives shall review th ws: bmitted. Critique is intended their submittals. ments as needed. d.
A. B. C. Over Air H Watı 3.3. O&N A. B.	All cc 1. 2. The C and s 1. The F O&M 1. 2. rhead Do handling rer Heate M DATA All cc 1.	ontractors shall prepare Prepare three (3) cor Review all specificati all equipment, syster below and shall indic specification, and a c GC shall be required to r shall return any to the o When acceptable to library on the Project Project Architect, City Pr 1 Data draft submittals a Provide general critic provide all contracto a. Re-submittal Review in detail the G a. Re-submittal <u>Title</u> oor Operator g Unit (AHU-3) er (WH-1) FINAL SUBMITTAL prepare complete O8	and submit the follow mplete O&M Data file ons within their Divisi- ms, materials, or finish cate the title (and plan column to verify the ite review all contractors' originating contractor t the GC, they shall uplo t Management Web Si roject Manager, CxA, C and checklist within fif que comments by Divi- rs with information or of the O&M Data sam O&M Data Checklist Specifica 08 36 00 23 00 00 22 30 00	ving for an O& samples as de on of Work an des. Checklist identifier whe em has been t samples and that are insuffi bad each O&M te. Consulting Sta teen 15 worki sion on O&M n strengths an ples will not b or completene will be require	&M Data Draft rev lescribed in sectio and prepare a com t shall be in tabula hen applicable) of turned in and con d checklists for cor fficient for re-subr M Data draft subn affs and Owner Re king days as follow 1 Data samples sub nd weaknesses of be required. hess. Provide comp red until accepted	view submittal: on 3.1 above. applete O&M Data checklist listi ar form similar to the example the O&M Data, the associated appleted. mpliance with this specification mittal. mittal file to the O&M Draft epresentatives shall review th ws: bmitted. Critique is intended their submittals. ments as needed. d.
B. C. Over Air H Watı 3.3. O&N A. B.	1. 2. The G and s 1. The F O&M 1. 2. rhead Do Handling er Heate M DATA All co 1.	Prepare three (3) cor Review all specificati all equipment, syster below and shall indic specification, and a c GC shall be required to r shall return any to the o When acceptable to library on the Project Project Architect, City Pr 1 Data draft submittals a Provide general critic provide all contracto a. Re-submittal Review in detail the 0 a. Re-submittal <u>Title</u> oor Operator (Unit (AHU-3) er (WH-1) FINAL SUBMITTAL prepare complete O8	mplete O&M Data file ons within their Divisions, materials, or finish cate the title (and plan column to verify the ite review all contractors' originating contractor t the GC, they shall uple t Management Web Si roject Manager, CxA, (and checklist within fif que comments by Divi rs with information or of the O&M Data sam O&M Data Checklist Specifica 08 36 00 23 00 00 22 30 00	samples as de on of Work an ies. Checklist identifier whe em has been t samples and that are insuff boad each O&M te. Consulting Stai teen 15 worki sion on O&M in strengths an ples will not b or completene will be require	described in section and prepare a com t shall be in tabula hen applicable) of turned in and con d checklists for cor fficient for re-subr M Data draft subn affs and Owner Re king days as follow 1 Data samples sub nd weaknesses of be required. hess. Provide completed	In 3.1 above. Inplete O&M Data checklist listi ar form similar to the example the O&M Data, the associated inpleted. Impliance with this specification mittal. mittal file to the O&M Draft epresentatives shall review the vs: bmitted. Critique is intended their submittals. Iments as needed. d.
B. C. Over Air H Watu 3.3. O&N A. B.	2. The C and s 1. The F O&M 1. 2. rhead DC Handling er Heate M DATA All cc 1.	Review all specificati all equipment, syster below and shall indic specification, and a c GC shall be required to a shall return any to the o When acceptable to library on the Project Project Architect, City Pr 1 Data draft submittals a Provide general critic provide all contracto a. Re-submittal Review in detail the 0 a. Re-submittal <u>Title</u> oor Operator (Unit (AHU-3) er (WH-1) FINAL SUBMITTAL prepare complete O8	ons within their Divisions, materials, or finish cate the title (and plan column to verify the ite review all contractors' originating contractor t the GC, they shall uplot t Management Web Si roject Manager, CxA, (and checklist within fif que comments by Divi rs with information or of the O&M Data sam O&M Data Checklist Specifica 08 36 00 23 00 00 22 30 00	on of Work an les. Checklist identifier whe em has been t samples and that are insuff bad each O&M te. Consulting Sta teen 15 worki sion on O&M n strengths an ples will not b or completene will be require	Ind prepare a com t shall be in tabula hen applicable) of turned in and con d checklists for cor fficient for re-subr M Data draft subn affs and Owner Re king days as follow 1 Data samples sub nd weaknesses of be required. hess. Provide comp red until accepted	Inplete O&M Data checklist list ar form similar to the example the O&M Data, the associated npleted. mpliance with this specificatio mittal. nittal file to the O&M Draft epresentatives shall review th ws: bmitted. Critique is intended their submittals. ments as needed. d.
B. C. Over Air H Watu 3.3. O&N A. B.	The C and s 1. The F O&M 1. 2. rhead Do Handling er Heate M DATA All co 1.	GC shall be required to reshall return any to the oreganized of the shall return any to the shall retu	review all contractors' riginating contractors' the GC, they shall uplo t Management Web Si roject Manager, CxA, G and checklist within fif que comments by Divi rs with information or of the O&M Data sam O&M Data Checklist fc of the O&M Checklist Specifica 08 36 00 23 00 00 22 30 00	samples and other insufficial are insufficial each O&M te. Consulting Static teen 15 working ison on O&M there is the strengths and ples will not bor completene will be required will be required to the strength of the stre	d checklists for cor fficient for re-subr M Data draft subn affs and Owner Re king days as follow 1 Data samples sub nd weaknesses of be required. iess. Provide comi red until accepted	mpliance with this specificatio mittal. nittal file to the O&M Draft epresentatives shall review th vs: bmitted. Critique is intended their submittals. ments as needed.
C. C. Over Air F Wate 3.3. O&N A. B.	rhead Do Handling er Heate M DATA All co 1.	shall be required to r shall return any to the o When acceptable to library on the Project Project Architect, City Pr 1 Data draft submittals a Provide general critic provide all contracto a. Re-submittal Review in detail the o a. Re-submittal <u>Title</u> oor Operator (Unit (AHU-3) er (WH-1) FINAL SUBMITTAL prepare complete O8	riginating contractors originating contractor t the GC, they shall uplo t Management Web Si roject Manager, CxA, (and checklist within fif que comments by Divi- rs with information or of the O&M Data sam O&M Data Checklist for of the O&M Checklist Specifica 08 36 00 23 00 00 22 30 00	ten are insuff bad each O&M te. Consulting Sta teen 15 worki sion on O&M of strengths and ples will not b bor completene will be require	fficient for re-subr M Data draft subn affs and Owner Re king days as follow 1 Data samples subn d weaknesses of be required. red until accepted <u>Completed</u>	mittal. mittal file to the O&M Draft epresentatives shall review th vs: bmitted. Critique is intended their submittals. ments as needed.
C. Over Air F Watr 3.3. O&N A. B. C.	The F O&M 1. 2. rhead Do Handling er Heate M DATA All co 1.	When acceptable to library on the Project Project Architect, City Pi 1 Data draft submittals a Provide general critic provide all contracto a. Re-submittal Review in detail the 0 a. Re-submittal <u>Title</u> oor Operator ; Unit (AHU-3) er (WH-1) FINAL SUBMITTAL prepare complete O8	the GC, they shall uplo the GC, they shall uplo t Management Web Si roject Manager, CxA, C and checklist within fif que comments by Divi rs with information or of the O&M Data sam O&M Data Checklist fo of the O&M Checklist Specifica 08 36 00 23 00 00 22 30 00	and each O&M te. Consulting Sta teen 15 worki sion on O&M n strengths and ples will not b or completene will be require	M Data draft subn affs and Owner Re king days as follow 1 Data samples subn d weaknesses of be required. ess. Provide com red until accepted	nittal file to the O&M Draft epresentatives shall review th vs: bmitted. Critique is intended their submittals. ments as needed. d.
C. Over Air F Watr 3.3. O&M A. B. C.	The F O&M 1. 2. rhead Do Handling er Heate M DATA All co 1.	library on the Project Project Architect, City Pr 1 Data draft submittals a Provide general critic provide all contracto a. Re-submittal Review in detail the (a. Re-submittal <u>Title</u> oor Operator (Unit (AHU-3) er (WH-1) FINAL SUBMITTAL prepare complete O8	thanagement Web Si roject Manager, CxA, (and checklist within fif que comments by Divi- rs with information or of the O&M Data sam O&M Data Checklist for of the O&M Checklist Specifica 08 36 00 23 00 00 22 30 00	te. Consulting Sta teen 15 worki sion on O&M n strengths an ples will not b or completene will be require	affs and Owner Re king days as follow 1 Data samples sub nd weaknesses of be required. ness. Provide com red until accepted	epresentatives shall review th ws: bmitted. Critique is intended their submittals. ments as needed.
C. Over Air H Wat 3.3. O&N A. B.	The F O&M 1. 2. rhead Do Handling er Heate M DATA All co 1.	Project Architect, City Pr 1 Data draft submittals a Provide general critic provide all contracto a. Re-submittal Review in detail the (a. Re-submittal <u>Title</u> oor Operator (Unit (AHU-3) er (WH-1) FINAL SUBMITTAL prepare complete O8	roject Manager, CxA, (and checklist within fif que comments by Divi. rs with information or of the O&M Data sam O&M Data Checklist fc of the O&M Checklist Specifica 08 36 00 23 00 00 22 30 00	Consulting Sta Consulting Sta iteen 15 worki sion on O&M on strengths and ples will not b por completene will be required will be required will be required tion	affs and Owner Re king days as follow 1 Data samples sub nd weaknesses of be required. ness. Provide comp red until accepted <u>Completed</u>	epresentatives shall review th ws: bmitted. Critique is intended their submittals. ments as needed. d.
Over Air H Wat 3.3. O&N A. B.	O&M 1. 2. rhead Do landling er Heate M DATA All co 1.	1 Data draft submittals a Provide general critic provide all contracto a. Re-submittal Review in detail the G a. Re-submittal <u>Title</u> oor Operator Unit (AHU-3) er (WH-1) FINAL SUBMITTAL pontractors shall prepare Prepare complete O8	and checklist within fif que comments by Divis rs with information or of the O&M Data sam O&M Data Checklist fc of the O&M Checklist <u>Specifica</u> 08 36 00 23 00 00 22 30 00	teen 15 worki sion on O&M n strengths an ples will not b or completene will be require	king days as follow 1 Data samples sub nd weaknesses of be required. ness. Provide com red until accepted <u>Completed</u>	vs: bmitted. Critique is intended their submittals. ments as needed. d.
Over Air H Watı 3.3. O&N A. B. C.	1. 2. rhead Do tandling er Heate M DATA All co 1.	Provide general critic provide all contracto a. Re-submittal Review in detail the G a. Re-submittal <u>Title</u> oor Operator (Unit (AHU-3) er (WH-1) FINAL SUBMITTAL pontractors shall prepare Prepare complete O8	que comments by Divis rs with information or of the O&M Data sam O&M Data Checklist fc of the O&M Checklist	sion on O&M n strengths and ples will not b or completene will be require	1 Data samples sul nd weaknesses of be required. ness. Provide comi red until accepted	bmitted. Critique is intended their submittals. ments as needed. d.
Over Air F Watu 3.3. O&N A. B.	2. rhead Do tandling er Heate M DATA All co 1.	provide all contracto a. Re-submittal Review in detail the (a. Re-submittal <u>Title</u> oor Operator (Unit (AHU-3) er (WH-1) FINAL SUBMITTAL pontractors shall prepare Prepare complete O8	rs with information or of the O&M Data sam O&M Data Checklist fc of the O&M Checklist 08 36 00 23 00 00 22 30 00	n strengths an ples will not b or completene will be require tion	nd weaknesses of be required. less. Provide comi red until accepted	their submittals. ments as needed. d.
Over Air F Watı 3.3. O&N A. B. C.	2. rhead Do Handling er Heate M DATA All co 1.	a. Re-submittal Review in detail the (a. Re-submittal <u>Title</u> oor Operator (Unit (AHU-3) er (WH-1) FINAL SUBMITTAL pontractors shall prepare Prepare complete O8	of the O&M Data sam O&M Data Checklist for of the O&M Checklist 08 36 00 23 00 00 22 30 00	ples will not b or completene will be require tion	be required. less. Provide comi red until accepted <u>Completed</u>	ments as needed. d.
Over Air F Wate 3.3. O&N A. B.	2. rhead Do Handling er Heate VI DATA All co 1.	Review in detail the 0 a. Re-submittal <u>Title</u> oor Operator ; Unit (AHU-3) er (WH-1) FINAL SUBMITTAL pontractors shall prepare Prepare complete O8	O&M Data Checklist for of the O&M Checklist 08 36 00 23 00 00 22 30 00	or completene will be require tion (ness. Provide com red until accepted Completed	ments as needed. d.
Over Air F Watr 3.3. O&N A. B.	rhead Do Handling er Heate VI DATA All co 1.	a. Re-submittal <u>Title</u> oor Operator (Unit (AHU-3)) er (WH-1) FINAL SUBMITTAL pontractors shall prepare Prepare complete O8	of the O&M Checklist Specifica 08 36 00 23 00 00 22 30 00 and submit the follow	will be require	red until accepted	d.
Over Air F Wate 3.3. O&N A. B.	rhead Do Handling eer Heate M DATA All co 1.	<u>Title</u> oor Operator ; Unit (AHU-3) er (WH-1) FINAL SUBMITTAL ontractors shall prepare Prepare complete O8	Specifica 08 36 00 23 00 00 22 30 00 and submit the follow	ition (<u>Completed</u>	
Over Air F Wat 3.3. O&N A. B. C.	rhead Do Handling Her Heate VI DATA All co 1.	oor Operator <u>;</u> Unit (AHU-3) er (WH-1) FINAL SUBMITTAL pontractors shall prepare Prepare complete O8	08 36 00 23 00 00 22 30 00 and submit the follow			
Air H Watu 3.3. O&N A. B. C.	Handling Per Heate M DATA All co 1.	; Unit (AHU-3) er (WH-1) FINAL SUBMITTAL potractors shall prepare Prepare complete O8	23 00 00 22 30 00 and submit the follow			
Watı 3.3. O&N A. B. C.	<u>er Heate</u> VI DATA All co 1.	er (WH-1) FINAL SUBMITTAL ontractors shall prepare Prepare complete O8	and submit the follow			
3.3. O&N A. B. C.	M DATA All co 1.	FINAL SUBMITTAL ontractors shall prepare Prepare complete O8	and submit the follow			
В. С.	2.	as described in Section Submit completed ch	on 3.2 above. necklist and all final O8	ving for an O& ibed in Sectior &M Data files t	&M Data Final rev on 3.1 above acco s to the GC for fina	view submittal: rding to their approved check al submittal review.
C.	The G for co re-su 1.	GC shall be required to s ompliance with this spe Ibmittal. When acceptable to on the Project Mana	spot check all contract cification and shall ret the GC, they shall uplo gement Web Site.	ors' submittal curn any to the bad each O&N	als for completene ne originating cont M Data final subm	ess against their checklists and tractor that are insufficient fo nittal file to the O&M Final libr
	The F O&M	Project Architect, City Pi 1 Data final submittals a	roject Manager, CxA, (Ind checklist within fift	Consulting Sta teen (15) work	affs and Owner Re rking days as follo	epresentatives shall review thows:
	1.	Review the files subr	nitted against the che	cklist and requ	quest any missing	files through the GC.
	2.	Review in detail all o	f the O&M Data files f	or completene	ness.	
		a. Submittals sh	all be accepted or reje	ected as indivi	vidual PDF files.	
		b. Contractors s	hall re-submit entire (0&M submitta	tal if any portion is	s rejected or incomplete.
3.4 CON	ISTRUCT					
Δ		ontractors shall review ^q	Specification 01 77 00	Closeout Pro	ocedures and Spec	cification 01 79 00
70	7 11 00	Demonstration and Tra	ining.	closcout i i o	occurres and spec	
	1.	Acceptance of all fina	al O&M Data submitta	ls is required i	d prior to scheduli	ng Demonstration and Trainin
		Sessions.				
	2.	Completion of all De	monstration and Train	ing Sessions is	is required to rece	eive the Substantial Complian
		for Occupancy Certif	icate, and to begin Co	nstruction Clo	oseout procedure	PS.

END OF SECTION

1 2			SECTION 01 78 36 WARRANTIES							
3 1	ΡΔ RT	1 – GI	NFRAI 1							
5		11	ιε ινα ε ΝΙΜΔΩΡΥ							
6		1.2.	FLATED SPECIFICATIONS							
7		1.3.	DEFINITIONS							
8		1.4.	GENERAL CONTRACTORS RESPONSIBILITIES							
9	PART	2 – PF	RODUCTS - THIS SECTION NOT USED							
10	PART	3 - EX	CUTION							
11	:	3.1.	WARRANTY CHECKLIST							
12	1	3.2.	LETTERS OF WARRANTY							
13	:	3.3.	STANDARD PRODUCT WARRANTY							
14	1	3.4.	FINAL WARRANTY SUBMITTAL4							
15 16	3	3.5.	WARRANTY NOTIFICATION, RESPONSE, EXECUTION AND FOLLOW-UP4							
17 18	PART	<u>1 – G</u>	ENERAL							
19	1.1.	SUN	ΙΜΔRΥ							
20		A.	The purpose of this specification is to provide clear responsibilities and guide lines related to providing all							
21			Warranties and Guarantees related to the Work, workmanship, materials, equipment, and other such items							
22			required by the Construction Documents.							
23		В.	Manufacturers' disclaimers and limitations on product warranties do not relieve any contractor of the warranty							
24			on the Work that includes the product.							
25		C.	Manufacturers' disclaimers and limitations on product warranties do not relieve suppliers, manufacturers and							
26			any contractor required to provide special warranties under the contract documents.							
27										
28	1.2.	KEL	ATED SPECIFICATIONS							
29 20		A. D	Section 01 21 22 Progress Payment Mob Site							
30		C.	Section 01 77 00 Closeout Procedures							
32		D.	Section 01 78 23 Operation and Maintenance Data							
33		E.	Section 01 91 00 Commissioning							
34		F.	Other Divisions and Specifications that may address more specifically the requirements for Warranties related to							
35			the installation of all items and equipment installed under the execution of the Work.							
36										
37	1.3.	DEF	INITIONS							
38		Α.	See specification 01 77 00 for the definitions of the following terms that may also be used in this specification:							
39			1. Substantial Compliance							
40			2. Certificate of Occupancy							
41 12			Centruction Clospout							
42 12			5 Contract Closeout							
44		в	Emergency Renair: The Owner or Owner Representative reserves the right to make emergency renairs as							
45		υ.	required to keep equipment or materials in operation or to prevent damage to property and injury to persons							
46			without voiding the contractors warranty or bond or relieving the contractor of their responsibilities during the							
47			warranty period.							
48		C.	Installer: The company or contractor hired to install a finished product that was manufactured and supplied							
49			specifically for the Work within this contract. The Installer may or may not be the same company that supplied							
50			the product. See the definition for supplier.							
51		D.	Supplier: Any company that makes a specific finished product for the Work from information within the Contract							
52			Documents. Examples of suppliers would include custom cabinets, steel stairs and railings, etc. A supplier would							
53		-	not be a company that distributes items manufactured by others such as an electrical or plumbing supplier.							
54 FF		E.	warranty: A written guarantee from the manufacturer to the owner on the integrity of a product and its							
55			installation, and the manufacturers' responsibility to repair or replace the defective product or components							
50 57			within a specified time from the date of ownership. Warranty may also be used interchangeably with Guarantee. The following warranty types may be part of any specification within the Work associated with the							
58			Construction Documents:							
50			construction bottoments.							

1			1.	Expressed Warranty: A warranty that provides specific repair or replacement for covered components of
2				a product over a specified length of time.
3			2.	Implied Warranty: A warranty that is not stated explicitly by a seller or manufacturer that the product is
4			-	merchantable and fit for the intended purpose.
5			3.	Standard Product Warranty: Preprinted written warranties published by individual manufacturers for
6				particular products and are specifically endorsed by the manufacturer to the Owner. Standard warranties
/				may be for any amount of time but shall not be for anything less than one (1) year from the warranty
0 0			4	Udle.
9 10			4.	Special warranty. A written warranty required by the contract Documents either to extend the time
11		F	Warrar	The provided didei a standard warrantly of to provide greater rights to the Owner.
12		1.	work_n	hy bate. The effective date that begins all warranty periods required for products, installations, and
13			the CPI	M
14		G.	Related	d Damages and Losses: When correcting failed or damaged Warranted Work, remove and reinstall (or
15		С.	replace	e if necessary) the construction that has been damaged as a result of the failure or the construction that
16			must b	e removed and replaced to obtain access for the correction of Warranted Work.
17		Н.	Reinsta	atement of Warranty: When Work covered by a warranty has failed and been corrected reinstate the
18			warran	ty by a new written endorsement. The reinstated warranty shall be equal to the original warranty with an
19			equitat	ble adjustment for depreciation unless specifically noted otherwise in a specification.
20		I.	Replace	ement Cost: All costs that may be associated with Work being replaced under warranty including but not
21			limited	to the following:
22			1.	Related damages and losses
23			2.	Labor, material and equipment
24			3.	Permits and inspection fees
25			4.	This shall be regardless of any benefit the Owner may have had from the Work through any portion of its
26				anticipated useful service life.
27		J.	Replace	ement Work: All materials, products, required labor, and equipment necessary to replace failed or
28			damag	ed warranted to an acceptable condition that complies with the requirements of the original Construction
29			Docum	ients.
30		К.	Owner	s Recourse: Expressed warranties made to the Owner are in addition to implied warranties and shall not
31			limit th	e duties, obligations, rights, and remedies otherwise available under the law. Expressed warranty periods
32			shall no	ot be interpreted as limitations on the time in which the Owner can enforce such other duties, obligations,
33			rights,	and remedies.
34			1.	Rejection of Warranties: The Owner reserves the right to reject any warranty and to limit the selection of
35			2	products with warranties not in conflict with the requirements of the contract documents.
30			2.	where the Contract Documents require a Special Warranty or similar commitment on the Work or
37 20				product, the Owner reserves the right to refuse acceptance of the work until the contractor presents
30				evidence the entities required to countersign such required commitments have done so.
40	14	GENER		ITRACTORS RESPONSIBILITIES
41	1.4.	A.	The Ge	neral Contractor (GC) shall be responsible to remedy, at their expense, any defect in the Work and any
42			damag	e to City owned or controlled real or personal property when the damage is a result of:
43			1.	The GC's failure to conform to Contract Document requirements.
44				a. Any substitutions not properly approved and authorized may be considered defective.
45			2.	Any defect in workmanship, materials, equipment, or design furnished by the GC or Sub-contractors.
46		В.	All war	ranties as described in this specification and these Contract Documents shall take effect on the date
47			establis	shed by the CPM, as noted in Section 1.3F above.
48			1.	All warranties shall remain in effect for one (1) year thereafter unless specifically stated otherwise in the
49				Contract Documents or where standard manufacturer warranties are greater.
50		C.	The GC	's warranty with respect to Work repaired or replaced, including restored or replaced Work due to
51			damag	e, will run for one (1) year from the date of Owner Acceptance of said repair or replacement.
52			1.	This shall be regardless of any benefit the Owner may have had from the Work through any portion of its
53				anticipated useful service life.
54		D.	Warrar	nty Response
55			1.	See Section 3.5 of this specification.

4 5

6

7

8

9

10

11

12

13

14

15

16

17

PART 2 – PRODUCTS - THIS SECTION NOT USED

3 PART 3 - EXECUTION

3.1. WARRANTY CHECKLIST

- A. All contractors shall be responsible for reviewing the drawings and specifications within their Divisions of Work to provide a complete and comprehensive list of all Warranty Requirements to the GC.
- B. Each list shall indicate the title (and plan identifier when applicable) of the warranted item, the associated specification of the warranted item, the terms of the warranty (years), and a column to verify the item has been turned in and completed.
 - C. The GC shall be responsible for all of the following:
 - 1. Consolidating all the warranty lists into one master Warranty Checklist.
 - a. The checklist shall be in a tabular data format similar to the sample below.
 - 2. Upload the completed checklist to the Submittal Library on the Project Management Web Site for review. See Specification 01 33 23 Submittals for more information on this procedure.
 - 3. Resubmit the schedule as needed after initial reviews have been completed.
 - D. The GC shall work with all contractors to amend the Warranty Checklist throughout the execution of the project based on changes and modifications as necessary.

18 19

<u>Title</u>	Specification	<u>Terms</u>	Completed
Overhead Door Operator	08 36 00	MFR 2yr	
Exterior Bench and Trash	12 93 00	MFR 3 year warranty on finish	
Receptacles			
Kitchen Sink (SK-1)	22 42 00	MFR 5 year	
Disposal (D-1)	22 42 00	MFR 7 year parts and in-home service	
Toilet (WC-1)	22 42 00	MFR 1 year limited	

20

21 3.2. LETTERS OF WARRANTY 22 A. All letters of warranty shall be in a typed letter format and provide the following information: 23 1. The letter shall be on official company stationary including company name, address, and phone number. 2. Indicate WARNER PARK COMMUNITY RECREATION CENTER EXPANSION, contract number, and contract 24 25 address the warranty is for on the reference line. 26 3. Provide a description of the warranty(ies) being provided. 27 Include Division, Trade, or Specification information as necessary. a. 28 b. Only combine warranties of related Divisional Work together. Create new letters for additional 29 Divisions as necessary. 30 4. Indicate the effective Warranty Date. As noted in Section 1.3.F above, the Warranty Date shall be the 31 date the Certificate of Substantial Completion was signed by the City Engineer. 32 5. Contractor Letters of Warranty shall only be signed by a principal officer of the company. 33 6. After signing the letter provide the GC with a high quality color scanned image in PDF format and the 34 original signed letter. 35 Β. The GC shall be responsible for the Final Warranty submittal as identified in Section 3.4 below. 36 C. The GC shall obtain letters of warranty from all of the following: 37 The General Contractor shall provide warranty letters for all Work that was self performed under the 1. 38 contract documents, identify all trades or Divisions of Work. 2. All Sub-contractors shall provide warranty letters for Work performed under the contract documents; 39 40 identify all trades or Divisions of Work. 3. Suppliers, as required by other specifications within the Construction Documents where the manufacture 41 42 of a specific product unique to the Work of this contract was required. The terms and conditions of the Supplier Letter of Warranty shall be as defined by the 43 a. 44 specifications associated with the Work but shall not be less than the industry standard of repair, 45 or replace defective materials and workmanship within one (1) year of the warranty date. b. When the supplier is also the installer a single written letter may be submitted identifying both 46 47 the warranty for the manufacture of the product and the warranty for the installation of the 48 product. 49 4. Installers as required by other specifications within the Construction Documents where the installation of a specific product unique to the Work of this contract was required. 50

1 2 3			1.	The terms and conditions of the Installer Letter of Warranty shall be as defined by the specifications associated with the Work but shall not be less than the industry standard of repair, or replace defective materials and workmanship associated with the installation of the product within one (1) year of the warranty date.
5			5. Spec	ial Letters of Warranty shall be required from any contractor, supplier, installer or manufacturer who
6 7			agree Prod	es to provide warranty services required by any Division Specification in excess of their standard uct Warranty.
8 9	3.3.	STAN	IDARD PRODU	CT WARRANTY
10		A.	All contracto	ors shall be responsible for collecting and providing copies of all standard product warranties for
11			commerciall	y available products purchased and installed under this contract.
12		В.	Only one co	py of the manufacturers' standard warranty needs to be submitted as representative for all
13			quantities of	f the same model number used throughout the Work.
14 15		C.	Provide the Warranty su	manufacturers certificate, letter, or other standard documentation for each Standard Product bmitted as follows:
16			1. Whe	never possible a PDF version of the document shall be used.
17 18			а.	If a PDF version is used all additional information shall be completed using simple PDF editing tools such as text boxes, highlight, etc.
19			b.	If a PDF version is not available and an original document is furnished the additional information
20 21				shall be neatly hand written and highlighted on the document in such a fashion so that it does not obscure any part of the written warranty.
22			2. Provi	ide the following additional information on each warranty document:
23			a.	Contract warranty date.
24			b.	Provide the manufacturer name and model number of the product if not specified within the
25				warranty.
26				i. Where the manufacturer name and model number is specified within the warranty it shall
27				be highlighted for visibility.
28			с.	Provide the plan identifier (LAV-1, WC-2, etc) when applicable.
29		D.	Each comple	eted warranty shall be saved as a digital PDF. The file shall be named using the specification number
30			and item des	scription. I.E. 22 42 00 Tollet (WC-1).pdf
31			a.	where an original certificate was furnished provide a high quality colored scan of the completed
32 22				document with the additional information. Save the scanned image in PDF format and use the same naming convention as indicated above
33		F	Provide all P	DE files and any original documents to the GC for final consolidation to be provided to the Owner
35		_ .		
36	3.4.	FINA	L WARRANTY	SUBMITTAL
37 38		Α.	The GC shall suppliers, in:	receive all required warranties (digital PDF and any original documents) from all contractors, stallers and manufacturers.
39		в.	The GC shall	inventory all received warranties with the Warranty Submittal List to ensure all required warranties
40			have been re	eceived and all warranty periods are correct according to the specifications.
41		C.	Provide with	each Operation and Maintenance Manual a complete copy of any associated warranty.
42		D.	Scan all war	ranties into a single organized electronic PDF file as follows:
43			1. Orga	nize the PDF file into an orderly sequence based on the table of contents of the Specifications.
44 15			2. PIOVI	ide bookmarks and links to each individual DDE to enable quick navigation through the DDE
45			docu	ine booking is and hins to each individual PDP to enable quick havigation through the PDP
47		F	Unload the y	warranty submittal to the appropriate document library on the Project Management Web Site for
48		_ .	review by th	e Project Architect (PA)/Project Engineer (PE) and CPM.
49		F.	Correct any	deficiencies or omissions and resubmit as necessary.
50				
51	3.5.	WAR	RANTY NOTIFI	CATION, RESPONSE, EXECUTION AND FOLLOW-UP
52		Α.	Warranty No	otification:
53			1. The (Lity of Madison, Project Management Web Site, uses an email notification system for all warranty
54 55			relation	eu issues. The GC will be required to provide, and keep current during the warranty period, a mum of two (2) ombil addrossos and abone numbers of surrent ombilises to receive ambility
55			notif	inations and provide response regarding Work associated with those construction documents
57			1000	In the event a Warranty Issue is deemed by the City of Madicon to be an emergency, the CC shall
58			α.	first receive a phone call with a follow-up email from the Project Management Web Site
	\A/A DA			
	WARN	LIN FAR		

1			b.	The Contract Closeout-Warranty Issue Library on the Project Management Web Site uses a form
2				for each warranty issue that is logged into the system.
3				I. The GC shall open each warranty issue form, review the issue description and any attached
4				documentation or photos.
5				II. The GC shall also notify any other sub-contractor, supplier, or installer that may be
0	в	Marra	atu Bocr	required to review the warranty issue.
/	в.	vvarrar 1	The CC	JOINSE. Caball upon notification by the City of Medican provide warranty recences as follows:
8		1.	The GC	, shall upon notification by the City of Madison provide warranty response as follows:
9			a.	critical systems or equipment: where damage to equipment and other building components, or
10				injury to personnel is probable provide immediate emergency shut-down information and an on-
11				site response team as soon as possible but in no case shall on-site response exceed 24 nours.
12			b.	For non-critical responses where damage or injury is unlikely provide on-site response no later
13				than the next business day.
14			с.	where Technical Assistance support is part of the written warranty provide all assistance
15				necessary via phone, text, or internet systems as indicated by the warranty. If issues cannot be
16				resolved provide on-site response no later than the next business day.
17			d.	If the request cannot be supported in sufficient time as outlined above the Owner (or Owner
18				Representative) reserves the right to contact other contractors or service companies having
19				similar capability to expedite the repair or replacement and shall invoice all associated costs to
20				the Owner back to the GC.
21	C.	Warrar	nty Exec	cution:
22		1.	The GC	shall provide all repairs or replacements as necessary to restore broken or damaged Work to the
23			origina	I level of acceptance as intended by the Contract Documents.
24			a.	Provide all materials, equipment, products, and labor necessary to complete the repair or
25				replacement associated with the Warranty Issue.
26			b.	Provide all cleaning services as may be required before, during, and after the repair or
27				replacement as per Specification 01 74 13 Progress Cleaning.
28			с.	Provide any protection necessary for existing construction as per Specification 01 76 00 Protecting
29				Installed Construction
30			d.	Provide new letters of warranty when required.
31	D.	Warrar	nty Follo	ow-up:
32		1.	Logged	d Warranty Issues:
33			a.	The GC shall provide complete documented responses of all logged Warranty Issues. Responses
34				shall provide a description of work completed, by who, inclusive dates, and photos of completed
35				or repaired work.
36				i. Provide call back response if work is not acceptable.
37			b.	The City Project Manager shall review the submitted response documentation and do a field
38				inspection if necessary.
39				i. If work is not acceptable, contact GC to review details and expectations of the repair as
40				needed.
41				ii. If work is acceptable close the Warranty Issue.
42		2.	Quarte	erly Warranty Reviews:
43			a.	The GC shall be responsible for scheduling quarterly on-site review with all of the following:
44				i. City Project Manager, and other City staff as needed
45				ii Owner and Owner Tenant Representative
15				iii Commissioning Agent (CvA)
40				iv Plumbing Heating Electrical Sub-contractors
47				v. Other Sub-contractors that may be responsible for open Warranty issues
40			h	Quartarly ravious shall be scheduled at 2 months 6 months and 11 months after the affective
4J 50			ы.	data of the warranty. The review meetings shall:
50				i Poviow the status of all open Warranty Issues, determine course of action and estimated
51				date of completion
52				Udie of completion.
55				ii. In the appropriate quarter, provide shut-down, start-up, testing, and training of on-season
54 EE				Equipment as required by the contract documents.
55				m. The iii month review shan review an open warranty issues, final plan for resolution, and
50 57				all warrancy issues where a new letter of warranty may have been issued.
5/ E0				
JØ				
WARN	ER PARK	COMMU	JNITY RE	CREATION CENTER

END OF SECTION

1 2				SECTION 01 78 39 AS-BUILT DRAWINGS							
3 4	PART	1 – G	FNFRAL	NFRAI 1							
5		1.1.	SUMMAR	γ							
6		1.2.	RELATED	SPECIFICAITONS							
7		1.3.	RELATED	DOCUMENTS							
8		1.4.	PERFORM	IANCE REQUIREMENTS							
9		1.5.	QUALITY A	ASSURANCE							
0	PART	2 – P	RODUCTS	2							
1		2.1.	OFFICE SU	JPPLIES							
2	PART	3 - E>	KECUTION	2							
3	1	3.1.	FIELD DOO	CUMENT AS-BUILTS2							
4	3	3.2.	SITE SURV	/EY AS-BUILT							
5	1	3.3.	MASTER A	AS-BUILT DOCUMENT SET							
5	1	3.4.	AS-BUILT	REVIEW AND ACCEPTANCE4							
	3	3.5.	CHANGES	AFTER ACCEPTANCE							
	PART	1-6	<u>SENERAL</u>								
	11	suu	MMARV								
		Α.	This spe	ecification is intended to provide clear guidelines and identify the responsibilities of all contractors as they							
			pertain	to City of Madison contract procedures regarding the accurate recording of the Work associated with the							
			executi	on of this contract. This shall include but not be limited to work that will be hidden, concealed, or buried.							
		В.	Each co	ontractor shall be responsible for maintaining an accurate record of all installations, locations, and							
			change	s to the contract documents during the execution of this contract as it may relate to their specific division							
			or trade	e.							
		C.	The Ge	neral Contractor (GC) shall be responsible for ensuring all contractors provide as-built record information							
			to the N	Master As-Built Document Set as described in this specification.							
	1.2.	REL	LATED SPEC	IFICAITONS							
		Α.	00 31 2	1 Survey Information							
		В.	01 26 1	3 Request for Information							
		C.	01 31 2	3 Construction Bulletin							
		D.	01 32 3	3 Photographic Documentation							
		E.	01 26 6	3 Change Orders							
		F.	01 29 7	6 Progress Payment Procedures							
		G.	01 31 2	3 Project Management Web Site							
		Н.	01 33 2	3 Submittals							
		١.	01 77 0	0 Closeout Procedures							
		J	01 91 0	0 Commissioning							
		К.	Other D	Divisions and Specifications that may address more specifically the requirements for field recording the							
			installa	tion of all items associated with the execution of this contract by Division or Trade.							
	1.3.	REL	LATED DOCI	UMENTS							
		Α.	Other r	elated documents shall include but not be limited to the following:							
			1.	Bidding documents including drawings, specifications, and addenda.							
			2.	Required regulatory documents of conditional approval.							
			3.	Field orders, verbal or written by inspectors having regulatory jurisdiction.							
			4.	Shop drawings and installation drawings.							
	1.4.	PEF	RFORMANC	E REQUIREMENTS							
		Α.	The GC	shall be responsible for maintaining the "Master As-Built Document Set" in the job trailer at all times							
			during	the execution of this contract. This document set shall include all of the following:							
			1.	Master As-Built Plan Set							
			2.	Master As-Built Specification Set							
			3.	Other Document Sets							

1 2		В.	The GC shall designate one person of the GC staff to be responsible for maintaining the Master As-Built Document Set at the job trailer. This shall include, posting updates, revisions, deletions and the monitoring of all							
3			contractors posting as-built information as described in this specification.							
4		C.	All contractors shall use this specification as a general guideline regarding the requirements for documenting							
5			their completed Work. Contractors shall explicitly follow additional specification requirements within their own							
6 7			Division of Trade as it may apply to this specification.							
8	1.5.	QUA	LITY ASSURANCE							
9		Α.	The GC shall be responsible for all of the following:							
10			a. Spot checking all sub-contractors field documents to insure daily information is being re-	ecorded as						
11			work progresses.							
12			 Discuss as-built recording to the plan set at weekly iob meetings with all sub-contractor 	rs on site.						
13			c. Schedule time with sub-contractors in the job trailer for recording as-built information	to the plan						
14			set.							
15			d. Insure that all sub-contractors are providing clear and accurate information to the plan	set in a						
16			neat and organized manner.							
17			e Insure sub-contractors who have completed work have finalized recording all as-built in	oformation						
18			to the plan set before releasing them from the project site	lionnation						
10		в	The Project Architect the City Project Manager. Commissioning Agent and other design team staff will	Inerform						
20		υ.	random checks of the Master As-Built Document Set during the execution of this contract to ensure a	s-built						
20			information is being recorded in a timely faction as the Work progresses. An undated and current Ma	stor As-						
21			Built Document Set is a cliquiding for annously fashion as the work progresses. An appared and current we							
22			built becument set is a supulation for approval of the progress payment.							
23	DADT	2 _ DP								
24 25	FANI	<u>2 - FN</u>								
26	2.1.	OFFI	CE SUPPLIES							
27		Α.	The GC shall provide a sufficient supply of office products in the job trailer at all times for all contractor	ors to use in						
28			recording as-built information into the plan set. This shall include but not be limited to the following:							
29			a. Red ink pens, medium point. Pens that bleed through paper, markers, and felt tips will	not be						
30			accepted.							
31			b. The use of highlighters is acceptable. Assign colors to various trades for consistency in	recording						
32			information.	U						
33			c. Straight edges of various lengths for drawing dimension, extension and other lines.							
34			d. Civil and Architectural scales							
35			e. Clear transparent, non-vellowing, single sided tape.							
36			f. Correction tape or correction fluid for correcting small errors.							
37										
38	PART	3 - EXE	ECUTION							
39										
40	3.1.	FIELD	D DOCUMENT AS-BUILTS							
41		Α.	The GC and all Sub-contractors shall be responsible for keeping their own field set of as-built documer	its						
42			including plans, specifications and published changes.							
43		В.	Field sets shall be kept dry and in good condition at all times.							
44		C.	No Work shall be buried, covered, or hidden, by any additional Work, regardless of Contractor or Trad	<u>e, until</u>						
45			locations of all materials and equipment has been properly documented as described below.							
46		D.	All contractors shall be required to record the following as-built information:							
47			a. Notes on the daily installation of materials and equipment.							
48			b. Sketches, corrections, and markups indicating final location, positioning, and arrangem	ent of						
49			materials and equipment such as pipes, conduits, valves, cleanouts, pull boxes and other	er such						
50			items. Note all final locations on plan sheets, indicate dimension off identifiable buildir	ng features.						
51			Riser diagrams need only be corrected for significant changes in locations, routing or							
52			configuration.							
53			i. The use of photographs in lieu of hand drawn sketches is acceptable.							
54			ii. Photos shall be taken according to Specification 01 32 33 Photographic Docume	ntation						
55			iii. Print photo and markup with dimensions or notes as necessary.							
56			c. Identify by the use of existing plan symbology and notes the size, type, quantity, and us	se as						
57			applicable of materials such as pipes, valves, conduits, etc.							
-										
1			d	l. No fin	te whether horizontal runs are below slab or above ceiling, include dimensions above or below iched floor elevation					
---------	------	--------	-------------	-------------------------------	---					
2		E	All contr	nn actors sh	all he responsible for transferring the information from their field set of documents to the					
5 /		с.	Master A	ACLOIS SII	an be responsible for transferring the information from their field set of documents to the					
-		E	All contr	actors sh	all undate the GC Master Blan Set as often as necessary, but not less than once per work week					
5		г.	All contra		an update the GC Master Plan Set as often as necessary, but not less than once per work week.					
7	32	SITE S	I IRVEV AS							
, 8	5.2.		The Land		or Sub-Contractor shall provide digital as-built information including but not be limited to the					
9		л.	following	а Эйг үс уо л .						
9 10			10110WIII18	<u></u> . Б.	r underground huried utility laterals and convices of all types legate all of the following that may					
10			a	i. 10 20	nuce ground buried utility laterals and services of an types locate an of the following that may					
12				ap ;	pry. Connection points at all mains					
12				ı. ;;	Storm discharge points to open air					
13					All corners and hands recordless of angle large radius success shall have multiple point.					
14					All conters and benus regardless of angle, large radius sweeps shall have multiple point					
15					All verticel deepe					
10				IV.	All vertical drops					
17				V.	All Wells Drivete huried at little each achuried destrict eshe initestication each ante					
18				VI.	Private buried utilities such as buried electrical cables, irrigation systems, etc.					
19				v.	Other information that may need to be located in the future by the owner prior to digging					
20			b	o. Re	cord all surface features including but not limited to the following:					
21				i.	Building corners, pavement edges, and other permanent structural features.					
22				ii.	All surface covers for inlets, catch basins, cleanouts, access structures, curb stops and					
23					other such devices.					
24				iii.	Other permanent surface features such as hydrants, lamp posts, and other permanent site					
25					amenities.					
26			C	. Th	e following data shall be recorded while locating items in sub-sections 3.2.a and 3.2.b above:					
27				i.	Flow lines at both ends of pipes					
28				ii.	Pipe sizes and material types					
29				iii.	Rim elevations for all covers					
30				iv.	Sump elevations and invert elevations of all structures					
31				۷.	Spot elevations for all pads, driveways, walks, stoops, and floors					
32		В.	The Surv	eyor sha	Il provide the final digital as-built on a media and in a format specified in Specification 00 31 21					
33			Survey Ir	nformatic	on to the GC for turn in to the Project Architect and the Civil Engineer.					
34		C.	The Surv	eyor sha	Il provide two printed as-built site plans to the GC for inclusion in the Master As-Built Plan Set					
35			as follow	/S:						
36			1. C	One sheet	to show all features (but not contour information) with text neatly organized for each item					
37			ic	dentified.						
38			2. C	One sheet	showing contours, contour labels, and features from item 1 above, but with no additional text.					
39										
40	3.3.	MAST	ER AS-BUI	ILT DOCU	IMENT SET					
41		Α.	The GC s	hall be re	esponsible for maintaining the Master As-Built Document Set in the job trailer at all times.					
42			1. T	he Maste	er As-Built Plan Set (Plan Set) shall begin with one complete bid set of drawings and any					
43			а	dditional	sheets that were supplied by published addenda during the bidding process. The cover sheet					
44			s	hall be tit	tled as the "Master As-Built Plan Set" in large bold red letters approximately 2" in height and					
45			s	hall not b	be used for any other purpose.					
46			а	. Th	e Plan Set shall be kept dry, legible, and in good condition at all times.					
47			b	. Th	e Plan Set shall be kept up to date with new revisions within two (2) working days of					
48				su	pplemental drawings being issued. Revisions shall be posted as follows:					
49				i.	Insert new, revised sheets into the plan set. Void old sheets but do not remove them from					
50					the plan set. Indicate date received and what document (RFI, CB, CO, etc) caused the					
51					change.					
52				ii.	Insert new, revised individual details into the plan set. Void old details, tape new details					
53					over the old details with a "tape hinge" to allow them to be viewed. Indicate date					
54					received and what document (RFI, CB, CO, etc) caused the change.					
55				iii.	Add new details in appropriate white space on relevant sheets. If no space is available use					
56					the back side of the previous sheet or insert a new sheet. Indicate date received and what					
57					document (RFI, CB, CO, etc) caused the change.					

1				с.	The Plan Set shall be available at anytime for easy reference during progress meetings and for
2					emergency location information of new work already completed.
3			2.	The M	aster As-Built Specification Set (Spec Set) shall begin with one complete bid set of specifications
4				and an	y additional specifications that were supplied by published addenda during the bidding process.
5				The Sp	ec Set shall be provided in three "D" ring type binders of sufficient thickness to accommodate the
6				specifi	cation set. Multiple binders are allowed as necessary. Label the front cover and binding edge with
7				"Maste	er As-Built Specifications" in bold red letters. Provide other information as necessary to distinguish
8				the co	ntents of multi-volume sets.
9				a.	The Spec Set shall be kept dry, legible, and in good condition at all times.
10				b.	The Spec Set shall be kept up to date with new revisions within two (2) working days of
11					supplemental drawings being issued.
12				с.	The Spec Set shall be available at anytime for easy reference during progress meetings.
13			3.	Other	Document Sets may be kept at the GCs option in three "D" ring type binders of sufficient thickness
14				to acco	ommodate the documentation. Other documentation sets may include but not be limited to RFIs,
15				CBs, C	Os, etc.
16		C.	The La	nd Surv	eyor Sub-Contractor shall be required to use digital surveying for all exterior site surveying, and
17			provid	e delive	rable digital as-builts as specified in Specification 00 31 21 Survey Information. As soon as practical
18			the su	rveyor s	hall provide the GC with a preliminary copy of installed buried utilities for inclusion with the plan
19			set in t	the job t	trailer. The surveyor shall provide final digital as builts as per section 3.2 above.
20		D.	All con	tractors	s shall be responsible for updating the Plan Set from their field sets at least once per work week.
21			Update	es shall	include but not be limited to the following procedures:
22				a.	All updates shall be done only in red ink. Place a "cloud" around small areas of correction to call
23					attention to the change.
24				b.	Whenever possible place general work notes, field sketches, supplemental details, photos, and
25					other such information on the reverse side of the preceding sheet. Installation notes including
26					dates shall be kept neatly organized in chronological order as necessary.
27				с.	Accurately locate items on the plan set as follows:
28					i. For items that are located as dimensioned provide a check mark or circle indicating the
29					dimension was verified.
30					ii. For items that are within 5 feet of the location indicated on the plans leave as shown and:
31					Provide correct dimensions to existing dimension strings or.
32					Accurately locate with new dimension strings
33					iii For items that are more than 5 feet from the location indicated on the plans
31					Accurately draw the items in the new location as installed and
25					Accurately locate with new dimension strings and
35					Accurately locate with new dimension strings and,
30 27				ч	 Note that the existing location is volu.
20				u.	under fleere, in wells er eheve seilings
30					Under Hoors, in wais of above centrigs.
39					 Dimensions shall be pulled from identifiable building realures, not from centers of columns or other buried features.
40					or other buried real ures.
41					ii. When necessary pull more dimensions as needed from opposing directions to properly
42					locate single items.
43	2.4	AC DU			
44 4E	5.4.	A3-DU			ID ACCEPTANCE
45		А.	Dreied	- Snan p	Tovide the Master As-Built Plan set to the Project Architect (PA)/Project Engineer (PE), the City
40			the Dr	l Widfidg	ger (CPM), the commissioning Agent (CXA) and other design team stan for content review prior to
47			the Pro	ugress P	ayment milestone indicated in Specification 01 2976 Progress Payment Procedures. The
48			Submit	tted pla	n set shall include the digital survey information produced under Section 3.2 above.
49			1.	n the p	Jan Set is not approved: The DA/DE and CDM shall only be required to generalize deficiencies by trade there shall be an
50				d.	The PAPE and CPN shall only be required to generalize deficiencies by trade there shall be no
51 21				h	The CC and Sub-contractors as necessary shall be reasonable for installed in the installed in and
52 52				D.	The GC and Sub-contractors as necessary shall be responsible for inspecting the installation and
JJ ⊑4					correcting the drawings as needed.
54 FF			2	C.	The GC shall re-subfill the plan set for review.
55			۷.	ii the p	Jan set is approved the PA/PE shall take possession of the plan set to be used in providing the
50				owner	with digital CAD record drawings. Upon completion of transferring the information to CAD the
57				PA/PE	shall provide the Owner with CAD record drawings, record PDFS, and the Master As-Bullt Plan Set.
20	14/4 201				
	WARN	EK PARK		JINITY RE	CREATION CENTER

1	3.5.	CHANGES AFTER ACCEPTANCE				
2		A.	No Contractor shall be responsible for making changes to the As-Built record documents after acceptance by the			
3			PA/PE and CPM except when necessitated by changes resulting from any Work made by the Contractor as part			
4			of their guarantee.			
5						
6						
7						
8			END OF SECTION			
9						

Page Intentionally Left Blank

		SECTION 01 78 43 SPARE PARTS AND EXTRA MATERIALS
DADT	1 – 6	
PARI	1 - 0	
-	1.1. 1.2	
-	1.2. 1.2	
-	1.3. 1 4	
-	1.4.	
DADT	1.5. 2 D	
PARI	2 - Pi	(UDUCTS – THIS SECTION NOT USED
PARI	3 - EX	
:	3.1.	
:	3.Z.	LABELING
:	3.3.	
:	3.4.	
:	3.5.	CLOSEOUT PROCEDURE
PART	1 – G	ENERAL
1.1.	SUN	ЛМАRY
	Α.	This specification is intended to provide clear guidelines and identify the responsibilities of all contractors as t
		pertain to City of Madison contract procedures regarding spare parts, special tools, special materials, and extr
		materials.
	В.	Each contractor shall be responsible for knowing the specific requirements of their Division Specifications as t
		may relate to the general information provided in this specification.
	C.	The General Contractor (GC) shall be responsible for ensuring all contractors provide spare parts and extra
		materials as described in this specification.
1 2		
1.2.	REL	ATED SPECIFICATIONS
	A.	01 29 76 Progress Payment Procedures
	В.	01 31 23 Project Management Web Site
	С.	01 // 00 Closeout Procedures
	D.	Other Divisions and Specifications that may address more specifically how to proceed with spare parts, specia
		tools, special materials, and extra materials.
1.3.	DEF	INITIONS
	Α.	Spare Parts: Any component of a product or assembly that comes pre-packaged or was specially ordered for
		explicit use of the product or assembly. This shall include but not be limited to fastening devices, mounting
		brackets, replacement parts, wheels, pulleys, wiring, alternate assembly pieces, etc.
	В.	Special Tools: Any tool of any kind that was pre-packaged or specially ordered, and is required to be used for
		installation or maintenance of an installed product or assembly as part of this contract.
	C.	Special Materials: Any oil, lubricant, glue, touch-up paint, or other such material that comes pre-packaged or
		was specially ordered and is required to be used for the installation or maintenance of an installed product or
		assembly as part of this contract.
	D.	Extra Materials (Attic Stock): Any surplus materials in new and useable condition that was installed a part of t
		contract. Attic Stock shall include but not be limited to the following: ceiling tiles, paint, stain, floor coverings
		ceramic tiles, light bulbs/lamps, filters, strainers, etc. Attic Stock shall include partially opened bulk items and
		additional unopened quantities as directed by other specifications.
1 4		
1.4.	PER	FURIVIAINCE REQUIREIVIENID
	А.	an contractors shall be responsible for consolidating spare parts, special tools, special materials, and attic stor
	п	as it pertains to the specific work within their Division of Hdue.
	в.	An contractors shall use this specification as a general guideline regarding the requirements for turning spare
		parts, special tools, special materials, and attic stock over to the owner. Contractors shall explicitly follow
		specification requirements within their own Division of Trade.
	QU	ALITY ASSURANCE
1.5.		
1.5.	Α.	The General Contractor (GC) shall be responsible for all of the following:

		1. Coordinate the location for and the delivery of all spare parts, special tools, special materials, and attic stock being provided by all contractors under this contract to one centralized location as designated by
		the Owner.
		2. Verify that all items being delivered are:
		a. Clean, new, and in a usable condition.
		b. Properly sealed, protected, and labeled
		c. Properly documented
PART	2 – PR	ODUCTS – THIS SECTION NOT USED
PART	3 - FXF	
<u>1 AN1</u>		
3.1.	PAC	AGING
	А. В.	Whenever possible all surplus items should remain in their original packaging such as parts envelopes. Package small parts in re-sealable plastic bags (Ziploc) or envelopes with clasp fasteners. Do not use envelopes
	C	that seal with glue or tape envelopes closed. Do not leave packaging unsealed.
	с. п	Many small nackages may be grouped together into a larger container by trade
	F.	Do not use unrelated hoves or containers for nackaging snare items. LE do not use a light fixture hov for snare
	L.	breakers, or flushometers parts.
3.2.	LABE	LING
	Α.	Whenever possible the original labeling indicating part numbers and other pertinent information shall remain on
		the original packaging.
	В.	If original labeling is not available the contractor shall label all parts and packages using tape or labels and
		permanent black markers. Tape or labels being used shall absorb the permanent marker without bleeding or
		allowing ink to be smeared or rubbed off.
	C.	Labels shall include the name of the product or equipment the item belongs to, part number and/or name, and
		any other information that would assist maintenance personnel in identifying the piece and related product.
	D.	Labels shall include plan or specification designations (WC-1, LAV-3, DF-2, CPT-1, etc) that identify the particular
		product or finish material it represents.
	E.	Labels for parts stored in clear re-sealable plastic bags may be placed inside the bag. Label shall face out and be
	-	able to be read from one side. Multiple bags shall be numbered individually for identification.
	г.	Laber the outside of large containers with the trade name (Plumbing, Electrical, etc).
3.3.	INVE	NTORY
	A.	All contractors shall provide the GC with complete inventories of all spare parts, special tools, special materials,
		and attic stock that they are providing at the end of the contract. The inventories shall be organized as follows:
		1. The cover sheet shall indicate the Contractors name, address, phone number, identify that the document
		is the "Spare Parts and Extra Materials Inventory", and identify the Division or Trade the inventory is for.
		2. Provide an inventory in a tabular format of all items being provided under this and other specifications.
		The minimum information to be provided for each item on the inventory shall be as follows:
		a. Bag or container number, all items of one bag or container shall be grouped together on the
		inventory
		b. Item description
		c. Item size (if applicable)
		a. Total quantity provided
	R	The GC shall consolidate inventories from all sub-contractors into one tabular data sheet organized by Division or
	Б.	Trade of Work
		1 Upon completing the consolidated list the GC shall upload the completed inventory to the Contract
		Closeout-Attic Stock Library on the Project Management Web Site.
		2. The GC shall notify the Project Architect and City Project Manager that the scans have been unloaded.
		3. Consulting Staff and Owner Staff shall review the inventories prior to Final Review to verify that minimum
		required quantities have been met. Deficiencies shall be noted and returned back to the GC for
		corrective action.
WARN	ER PAR	K COMMUNITY RECREATION CENTER

1							
2	3.4.	STOP	RAGE				
3		Α.	Prior to the 80% Progress Payment milestone the GC shall coordinate with the City Project Manager and				
4			Maintenance Personnel where spare parts, special tools, special materials, and attic stock shall be stored.				
5		В.	The GC shall instruct all contractors as to the location and proper storage procedures.				
6		C.	The GC shall be responsible for ensuring the storage area is kept neat and orderly as follows:				
7			1. Like items are stored together by material, product, or trade as necessary.				
8 9			 Liquids are stored in sealable containers and the lids have been properly installed to prevent drying out, spillage, etc. 				
10			 All labels are clearly visible and provide the required information. 				
11		D.	Large items shall be stored so as not to damage other items. Do not stack heavy items or items with distinct				
12			shapes/outlines on softer items that may get crushed or imprinted.				
13			· · · · · · · · · · · · · · · · · · ·				
14	3.5.	CLOS	EOUT PROCEDURE				
15		A.	Prior to the 90% Progress Payment milestone the GC shall review all attic stock already stored by the contractors				
16			to ensure the following:				
17			1. Materials are stored in the proper location(s).				
18			2. All boxes, containers and items are properly labeled according to the submitted/approved inventory.				
19			 Quantities are correct according to the submitted/approved inventory. 				
20		В.	The GC shall ensure that all deficiencies are corrected prior to conducting Demonstration and Training Sessions.				
21		C.	The GC shall review with Maintenance Staff all inventories and labeling during the scheduled Demonstration and				
22			Training Sessions.				
23		D.	Any discrepancies associated with Attic Stock shall be resolved and verified prior to the CPM releasing the 90%				
24			CT progress payment.				
25							
26							
27			END OF SECTION				
28							

Page Intentionally Left Blank

			SECTION 01 79 00 DEMONSTRATION AND TRAINING			
PART 1 – GENERAL						
1	. 0.	SUMMARY				
1	.2.	RELATED SPECIFICATIONS	· · · · · · · · · · · · · · · · · · ·			
1	3.	OUALITY ASSURANCE				
PART 2	2 – PF	ODUCTS – THIS SECTION I	NOT USED			
PART 3	3 - EX	ECUTION				
3	1.	GENERAL REQUIREMENT	S			
3	2.	COORDINATING AND SCH	IEDULING THE TRAINING			
3	3.	TRAINING OBJECTIVES				
3	4.	DEMONSTRATION AND T	RAINING PROGRAM PREPARATION			
3	5.	CONDUCTING A DEMONS	TRATION AND TRAINING SESSION			
3	.6.	CLOSEOUT PROCEDURE .				
PART :	L – G	ENERAL				
1.1.	SUN	IMARY				
	Α.	The purpose of this spe	cification is to provide clear responsibilities and guidelines related to providing			
		Demonstration and Tra	ining (D&T) Sessions related to general facility use, equipment, systems, finishes, and			
		materials to City of Ma	dison Staff (Owner, Owner Representatives, Maintenance, and Custodial Personnel) as			
	_	needed.				
	В.	All D&T shall be coordi	nated through the General Contractor (GC), Project Architect (PA)/Project Engineer (PE			
		and City Project Manag	ger (CPIVI), and Will be based on or customized to the needs of City of Madison Staff bei			
		trained. New equipme	nt and systems may have complete D&I sessions as described in this specification while the first feasible reaction while the first feasible reaction of the session of the			
		equipment or systems	start is raminar with may have sessions more focused on maintenance only.			
1.2	DEL					
1.2.		Section 01 29 76	Progress Payment Procedures			
	R.	Section 01 78 13	Completion and Correction List			
	С.	Section 01 78 19	Maintenance Contracts			
	D.	Section 01 78 23	Operation and Maintenance Data			
	E.	Section 01 78 36	Warranties			
	F.	Section 01 78 39	As-Built Drawings			
	G.	Section 01 78 43	Spare Parts and Extra Materials			
	н	Section 01 91 00	Commissioning			
	I.	Other Divisions and Sp	ecifications that may address more specifically the requirements for D&T sessions relat			
		to the installation of al	l items and equipment installed under the execution of the Work.			
1.3.	QUA	ALITY ASSURANCE				
	Α.	All contractors shall ha	ve the responsibility of preparing for and conducting D&T sessions as determined by th			
		and other Division or I	rade related specifications, Owner Operation and Maintenance Manuals, and other suc			
		documentation related	to the Work.			
	в.	Ine GC shall have resp	onsibility for:			
		1. Ensuring that a	il contractors required to conduct a D&T session have successfully completed all of the			
		tollowing:	in all required documentation for review and documentation has been approved/accor			
		a. Tullieu	schoduling D&T sossions			
		h Other r	scried in the D&T sessions.			
		c All syste	ins have been started, tested, and running as per appropriate specification and/or			
		manufa	cturers recommendations prior to scheduling D&T sessions.			
		d. All cont	ractors are sufficiently prepared for their D&T session			
		e. Docume	ents the D&T session including date, time, contractor and company name, attendees an			
		other in	formation regarding the session			
		2. Organizing the	coordination and scheduling of all D&T sessions between all contractors and the			
		appropriate rer	presentatives of the Owner. These representatives may include any of the following			
		depending on t	he Work of the Contract:			

		a.	Owner – end users
		b.	Facility Maintenance personnel
			i. Facility general operation procedures including custodial services
			ii. Electrical
			iii. Mechanical
			iv. Plumbing
			v. Site
		с.	Information Technology (IT) Department
		d.	Traffic Engineering – Radio Shop
		e.	Architects, Engineers and Facility Management staff as project completion overview
PART	2 – PR	ODUCTS – THIS	SECTION NOT USED
PART	3 - EXE	CUTION	
3 1	GENI	FRAI REOLIIREN	MENTS
5.1.		The GC shall	develop a specific D&T plan to be scheduled and conducted as described below but no sooper than
	А.	the meeting	discussed in 3.2.4.2 helow
	C	The GC shall	not schedule D&T sessions to preclude required personnel from attending multiple sessions
	С.	The Ge shall	not schedule ber sessions to preclude required personner non attending multiple sessions.
3.2.	coo	RDINATING AN	D SCHEDULING THE TRAINING
-	Α.	The GC. PA/P	2E. CxA and CPM, shall review all Training and Demonstration requirements during two (2) special
		meetings.	-,
		1. The fi	irst meeting shall be held at the 50% Contract Total Payment. During this meeting the following
		shall l	be discussed:
		a.	Preliminary schedule of training dates to be completed prior to beginning construction closeout.
		b.	List of documentation and items that need to be completed and available before and during the
			training session.
		с.	Who (Owner, Maintenance, etc) will be attending what training session(s).
		2. The s	econd meeting shall be held at the 80% Contract Total Payment. This meeting shall review due outs
		that h	nave not yet been completed for the 90% Contract Total Payment and the requirements necessary
		for Co	onstruction Closeout. All Demonstration and Training sessions shall be completed prior to receiving
		the 90	0% progress payment and beginning Construction Closeout Procedures (see Specification 01 77 00).
		a.	This does not include any requirement associated with off season equipment preparation and/or
			demonstration and Training Sessions.
	В.	All of the Cor	nstruction Work shall be operationally ready prior to conducting training as follows:
		1. All co	ntractors shall have their As-Built Drawing Records available for reviewing locations of system
		comp	ionents during training.
		2. All <u>fin</u>	al and approved Operations and Maintenance Data shall be completed no less than two (2) full
		week	s prior to the scheduled training.
		3. All sys	stems shall have been started, functionally tested, balanced, and fully operational, and all piping
		and e	quipment labeling complete at least two (2) days prior to the scheduled training.
		a.	Seasonal equipment shall not be trained out of season. Contractors having seasonal equipment
			shall work with the GC and CPM for coordinating additional training sessions as appropriate for
			seasonal equipment.
	C.	Correction lis	t items that prevent a piece of equipment or system from being fully operational for training shall
		be corrected	prior to conducting the training.
3.3.	TRAI	NING OBJECTIV	/ES
	Α.	For each piec	ce of equipment or system installed train on the following objectives/topics as applicable:
		1. Syste	m design, concept, and capabilities
		2. Revie	w of related contractor as-built drawings
		3. Facilit	ty walkthrough to identify key components of the system
		4. Syste	m operation and programming including weekly, monthly, annual test procedures
		5. Syste	m maintenance requirements
		b. Syste	m troublesnooting procedures
		7. Testir	ig, inspection, and reporting requirements associated with any regulatory requirements
		o	

		9. Review of system documentation including the following:
		a. Operation and maintenance data
		b. Warranties
		c. Valve charts, tags, and pipe identification markers
	В.	For each piece of specialty equipment train on the following objectives/topics as applicable:
		1. Manufacturers operations instructions
		2. Manufacturers use and care instructions
		3. Manufacturers maintenance and troubleshooting instructions
		4. System operation and programming including weekly, monthly, annual test procedures
		5. Identification of any correction list items still outstanding
		6. Review of system documentation including the following:
		a. Operation and maintenance data
	6	b. Warranties
	C.	End User Orientation
		1. Facility walkthrough
		2. Security and emergency features
	-	3. General facility operation procedures
	D.	Facility General Use and Custodial Services – if requested
		1. Facility walkthrough
		2. Security and emergency features
		3. General facility operation procedures
		4. Care and maintenance of specialty items, finishes, etc as requested
		5. Attic stock inventory and material designations
~ ~		
3.4.	DEIVI	UNSTRATION AND TRAINING PROGRAM PREPARATION
	А.	Each contractor having a responsibility for providing D&T sessions shall meet with the GC, CPM, and other City
		Start as needed to review the extent of the Training Objectives in section 3.3 above needed for each piece of
		equipment, system, finish, etc. This meeting shall occur no less than four (4) weeks prior to the anticipated
	-	training session.
	В.	The contractor shall use the information from item 3.4.A above to prepare a formal training program for each
		piece of equipment or system based on the Training Objectives in 3.3 above.
		1. The formal training program shall include the following information:
		a. Session title
		b. List of systems, equipment, use, care, etc to be covered during the session
		c. Provide the following for each systems, equipment, use, care, etc to be covered during the sess
		i. Name and affiliation of each instructor to be used. As needed and discretion of the Own
		the GC to require attendance by the installing technician, installing Contractor and the
		appropriate trade or manufacturer's representative.
		ii. Qualifications of each instructor to be used. Practical building operation expertise as we
		as in-depth knowledge of all modes of operation of the specific piece of equipment as
		installed in this project is required by the training personnel. If Owner determines train
		was not adequate, the training shall be repeated until acceptable to Owner.
		iii. A checklist of all documentation and system/equipment requirements necessary to
		complete a successful training session and the current status of each
		iv. Any additional documents, training aids, video or other items to be used to complete th
		training
		v. Any special requirements or needs associated with item iv above to complete the training
		d. The intended audience for the training
		e. The approximate duration of each objective or topic to be covered
		2. Submit the completed training program to the GC for review and approval by the PA/PE and CPM.
	C.	The PA/PE and CPM shall work with staff as necessary to ensure all points of anticipated training needs have
		been met. The PA/PE and CPM will approve the program as submitted or recommend changes for re-submitted
		as necessary.
3.5.	CON	DUCTING A DEMONSTRATION AND TRAINING SESSION
	Α.	All contractors shall conduct their required D&T Sessions as follows:
		1. Begin with a classroom session

1				b. Provide an overview of the training to be conducted including the approximate schedule.		
2			2.	Conduct a general walk-through of the site.		
3				a. Point out locations of various equipment, valves, charts, and other related items.		
4				b. Use the Division or Trade As-Built record drawings to indicate locations of hidden or buried items.		
5			3.	Provide a demonstration of general equipment/system operation including using the O&M manual.		
6				a. Startup and shutdown procedures.		
7				b. Normal operational levels as depicted by any gauges, software, etc.		
8				c. Indicate warning devices, signs etc. and demonstrate emergency shut-down procedures.		
9			4.	Provide a demonstration of all owner level maintenance using the O&M manual.		
10				a. Indicate frequency of maintenance.		
11				b. Provide and review all spare parts, special tools, and special materials.		
12			5.	Provide and review all spare parts, special tools, special materials, or attic stock as applicable.		
13			6.	While conducting D&T sessions:		
14				a. Allow hands on training whenever practical.		
15				b. Answer questions promptly		
16				 Repeat demonstrations and procedures as necessary. 		
17		В.	Within	two (2) working days of completing the D&T session the contractor responsible for the session shall turn-		
18			in any	documentation generated including the sign in roster to the GC.		
19		C.	The G	C shall turn over all training documentation to the PA/PE and CPM upon completion of D&T sessions.		
20		D.	Re-sch	iedule any training that has been determined to be inadequate or inappropriate for any reason including		
21			but no	of limited to any of the following;		
22			1.	Unqualified instructor		
23			2.	System installation incomplete or untested to the specifications		
24			3.	Equipment failure during demonstration		
25			4.	Un-expected cancellation		
26	2.6	<u> </u>				
27	3.6.	CLOSE				
28		А.	Prior t	o receiving the 90% Progress payment the GC shall:		
29			1.	verify with the PA/PE and CPW that each Demonstration and Training Session was conducted properly		
3U 21			n	Any required "Off Season" equipment tecting belancing and Demonstration and Training Sessions have		
27			Ζ.	Any required Off Season equipment testing, balancing, and Demonstration and Training Sessions have		
32 22				Representatives as necessary		
32				אבטיבאבוונמנועבא מא וובנבאאמו אי		
35						
36				END OF SECTION		
37						

1 2		SECTION 01 81 13 SUSTAINABLE DESIGN REQUIREMENTS – LEED FOR NEW CONSTRUCTION V4.0
3		
4	PARI 1 - 1 1	
5	1.1	
7	1.2	
י צ	1.5	ADMINISTRATIVE REGUIREMENTS
9	1.4	
10	1.5	INFORMATIONAL SUBMITTALS
11	1.0	OLIALITY ASSURANCE
12	1.8	CONTRACTOR RESPONSIBILITIES 5
13	PART 2 -	PRODUCTS
14	2.1	MATERIALS. GENERAL
15	2.2	BUILDING PRODUCT DISCLOSURE AND OPTIMIZATION
16	2.3	LOW-EMITTING MATERIALS
17	PART 3 -	EXECUTION
18	3.1	NONSMOKING BUILDING
19	3.2	CONSTRUCTION ACTIVITIES POLLUTION PREVENTION
20	3.3	BUILDING PRODUCT DISCLOSURE AND OPTIMIZATION7
21	3.4	CONSTRUCTION AND DEMOLITION WASTE MANAGEMENT PLANNING7
22	3.5	LOW EMITTING MATERIALS
23	3.6	CONSTRUCTION INDOOR-AIR-QUALITY MANAGEMENT PLAN
24	3.7	SUPPLEMENT9
20 27 28 29	1.1 R	ELATED DOCUMENTS Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division
30		01 Specification Sections, apply to thisSection.
31	В	. Comply with Wisconsin Commercial Building Codes/International Building Code (IBC).
32	C	Comply with Americans with Disabilities Architectural Guidelines, and ICC/ANSI A117.1-Latest Edition.
33	C	Comply with USGBC LEED prerequisites and credits shown in the attached checklist for Project to obtain
34		certification based on USGBC's LEEDv4.0 BD&C: New Construction and MajorRenovations" Process.
35	E	. Refer to attached LEED v4.0 for BD+C: New Construction and Major Renovations checklist, with LEED credits
36		clearly marked yes or no.
37		
38	1.2 S	UMMARY
39	A	Project registration and review fees associated with GBCI and leedonline.com are paid by the City.
40	В	. Section includes general requirements and procedures for compliance with certain USGBC LEED prerequisites
41		and credits needed for Project to obtain certification based on USGBC's LEED BD&C: New Construction and
42		Major Renovations" Version4.0.
43		1. Other LEED prerequisites and credits needed to obtain LEED certification depend on product selections
44		and may not be specifically identified as LEED requirements. Compliance with requirements needed to
45		obtain LEED prerequisites and credits may be used as one criterion to evaluate substitution requests and
46		comparable product requests.
47		2. Additional LEED prerequisites and credits needed to obtain the indicated LEED certification depend on Architectle design and other equates of Preject that are not part of the Work of the Contract
48		Architect s design and other aspects of Project that are not part of the Work of the Contract.
49 50		5. A copy of the LEED Project thetalls is allathed at the end of this Section.
50	ſ	4. Specific requirements for LEED are included in greater detail in other Sections. A significant partian of the credits required for certification are the responsibility of the A/E and Owner (design
52	Ľ	credits) These credits are not explicitly outlined in this specification section, however many aspects of the
52		construction documents reflect intent to document and achieve the design credits. This section documents
55		requirements of the contractor to meet the requirements for documenting the construction credits
55	г	Related Sections: Divisions 01 through 32 Sections for LEED requirements specific to the work of each of these
56	L	Sections Requirements may or may not include reference to LEED
57		

1 1.3	DEFIN	ITIONS
2	Α.	Albedo (a.k.a. solar reflectance): The ratio of the reflected electromagnetic energy to the incoming
3		electromagnetic energy.
4	В.	Chain-of-Custody Certificates: Certificates signed by manufacturers certifying that wood used to make products
5		was obtained from forests certified by an FSC-accredited certification body to comply with FSC STD-01-001, "FSC
6		Principles and Criteria for Forest Stewardship." Certificates shall include evidence that manufacturer is certified
7	<u> </u>	for chain of custody by an FSC-accredited certification body.
8	C.	Emissivity (a.k.a. infrared emittance): A parameter between 0 and 1 that indicates the ability of a material to
9		shed infrared radiation.
10	D.	Environmental Product Declarations: (EPD) is a transparent, objective report that communicates what a product
11	-	is made of and now it impacts the environment across its entire life cycle.
12	Ε.	Health Product Declaration (HPD) is a material ingredient reporting standard developed under the guidance of
13	E.	Life HPD Collaborative.
14	г.	hydronuorocarbons (HFCS): Reingerants used in building equipment that do not depiete the stratospheric ozone
15	G	Idyer. IEED: Loadorshin in Energy and Environmental Decign, Green Building Pating System representing the US Green.
10	О.	Building Council's effort to provide a national standard for what constitutes a "green building". The standard
18		requires quantitative and technical documentation to demonstrate compliance with goals described in the LIS
10		Green Ruilding Council's Green Ruilding Rating System, Version 3.0
20	н	I FED Project Administrator: I FED Certified Professional bired by the project owner to review I FED submittals
20	1	I ocally-Manufactured: Refers to the final assembly of components into the building product that is furnished and
22		installed by the trades people. For example, if the hardware comes from Seoul, South Korea, the lumber from
23		Vancouver, British Columbia, and the joist is assembled in Kent Washington, then the location of the final
24		assembly is Kent, Washington.
25	J.	Post-Consumer Recycled Content: The percentage of waste material by weight available from consumer use
26		incorporated into a building material.
27	к.	Pre-consumer (aka Post-Industrial Recycled) Content: The percentage of waste material by weight available from
28		industrial use incorporated into a building material. Post-industrial recyclable materials are different from
29		industrial scrap, a by-product of industrial processes that can easily be reused as a feedstock.
30	L.	Potable Water: Water that is suitable for drinking and is supplied from wells or municipal water systems.
31	M.	Recycling: The collection, reprocessing, marketing and use of materials that were recovered or diverted from the
32		solid waste stream. Note that LEED uses the term "pre-consumer" rather than "post-industrial." Also note that
33		when manufacturers and trade associations use the term "post- industrial" it often includes spills, scraps, and
34		damaged and surplus materials that are fed back into the same manufacturing process and that these materials
35		are not considered recycled content by the LEED rating systems.
36	N.	Recycled Content: The recycled content value of a material assembly shall be determined by weight. The
37		recycled fraction of the assembly is then multiplied by the cost of assembly to determine the recycled content
38		value. "Post-consumer" material is defined as waste material generated by households or by commercial,
39		industrial, and institutional facilities in their role as end users of the product, which can no longer be used for its
40		intended purpose. "Pre-consumer" material is defined as material diverted from the waste stream during the
41		manufacturing process. Excluded is reutilization of materials such as rework, regrind, or scrap generated in a
42	0	process and capable of being reclaimed within the same process that generated it.
45	0.	500 miles of Project cite. If only a fraction of a product or material is extracted/harvested/recovered and
44		manufactured locally then only that percentage (by weight) shall contribute to the regional value
45	D	Regionally Manufactured Materials: Materials that are manufactured within a radius of 500 miles from Project
40	г.	site Manufacturing refers to the final assembly of components into the building product that is installed at
48		Project site
49	0	Regionally Extracted and Manufactured Materials: Regionally manufactured materials made from raw materials
50	ц.	that are extracted, harvested, or recovered within a radius of 500 miles from Project site.
51	R.	Solar Reflectance: See "Albedo."
52	S.	Sustainable Forestry: The practice of managing forest resources to meet the long-term product needs of humans
53		while maintaining the biodiversity of forested landscapes. The primary goal is to restore, enhance, and sustain a
54		full range of forest values, both economic andecological.
55	т.	Type A Finishes: Material and finishes with potential for short-term levels of off gassing from chemicals inherent
56		in their manufacturing process, or which are applied in form requiring vehicles or carriers for spreading which
57		release high level of particulate matter in process of installation and/or curing. Including, but not limited to:

1 2 3 4 5 6 7 8 9 10 11 12 13 14 15 16		U.	 Composite wood products, specifically including particleboard from which millwork, wood paneling, doors, or furniture may befabricated. Adhesives, sealants, and glazing compounds, specifically those with petrochemical vehicles or carriers. Wood preservatives, finishes, and paint. Control and/or expansion jointfillers. Hard finishes requiring adhesive installation. Gypsum board and associated finish processes. Type B Finishes: Fuzzy material and finishes which are woven, fibrous, or porous in nature and tend to adsorb chemicals off-gassed by Type A finishes or may be adversely affected by particulates. These materials become "sink" for deleterious substances which may be released much later, or collectors of contaminants that may promote subsequent bacterial growth. Including, but not limited to: Carpeting and padding. Fabric wallcovering. Insulation exposed to air stream. Acoustic ceiling materials. Fabric covered acoustic wall panels.
17			6. Upholstered furnishings.
18			Materials that can be categorized as both Type A and TypeB.
19 20		V.	Ventilation: The process of supplying and removing air to and from interior spaces by natural or mechanical means.
21		W.	Volatile organic compounds (VOCs): Chemical compounds based on carbon and hydrogen structures that are
22			vaporized at room temperatures. VOCs are one type of indoor aircontaminant.
23		Х.	Waste Materials: Large and small pieces of materials indicated which are excess to contract requirements and
24			generally include materials salvaged from existing construction and items of trimmings, cuttings, and damaged
25			goods resulting from new installations which cannot be effectively used in Work.
26			
27	1.4	ADMI	NISTRATIVE REQUIREMENTS
28		Α.	Respond to questions and requests from Architect and the Green Building Certification Institute (GBCI; an agent
29			of USGBC that handles the review process) regarding LEED credits that are the responsibility of the Contractor,
30			that depend on product selection or product qualities, or that depend on Contractor's procedures until GBCI has
31			made its determination on the project's LEED certification application. Document responses as informational
32			submittals.
33			
34	1.5	ΑΟΤΙΟ	N SUBMITTALS
35		A.	General: Submit additional LEED submittals required by other Specification Sections.
36		В.	LEED submittals are in addition to other submittals. If submitted item is identical to that submitted to comply
37			with other requirements, submit duplicate copies as a separate submittal to verify compliance with indicated
38			LEEDrequirements.
39		C.	LEED Submittals: Submit LEED related information under a separate Tab within each product submittal. The LEED
40		0.	submittal shallinclude:
41			1. Summary Sheet: A summary, on General Contractors letterhead, of all LEED information requested in
42			specifications shall include:
43			a. Project name.
44			b. LEED Submittal List: A list of all materials being submitted. For products com- posed of multiple
45			materials the submittal shall include a list of all materials composing the product
46			c For Products in Divisions 2 - 10 include the following information:
47			i. Material costs, for each material on the LEED submittal list, excluding labor costs, delivery
48			cost, cost of installation, as well as profit and overhead.
49			ii. The preconsumer and post-consumer recycled content of each material on the LEED
50			submittal list.
51			iii. List of all material manufacturing locations.
52			iv. Provide distance between manufacturing and construction site.
53			d. All other LEED information required in specification.
54			2. Manufacturer's literature with information highlighted that confirm the figures used in the summary
55			report.
56			a. If a range is used in the manufacturer's literature, the summary report shall use the lowest
57			number in the range.
<i></i>			

1				b. For VOC Submissions: Submit MSDS sheets or manufacturer's literature with VOC figure
2				highlighted.
3		D.	Proje	ect Material Costs Data: Provide a statement, on Contractor's letterhead, documenting the total material for
4			the p	project. Include a spreadsheet tallying the material cost for all materials specified in Divisions 2 - 32. The
5			total	in the material cost data will be used in the LEED Online template to be completed by the Contractor as the
6			actua	al material cost of the project.
7		Ε.	LEED	Action Plan: Provide preliminary submittal within 30 days of Notice to Proceed that contains:
8			1.	Example spreadsheets for each construction credit identified in this section.
9			2.	Contact information for Contractor's LEED coordinators.
10			3.	Brief description of how the following requirements will be met.
11				a. SS Prerequisite: Construction Activities Pollution Prevention complying with Section 31 25 00,
12				Erosion Control.
13				b. MR Prerequisite: Construction and Demolition Waste Management Reporting
14				c. MR Credit: Building Product Disclosure – Environmental Product Disclosures
15				d. MR Credit: Building Product Disclosure – Source Materials
16				e. MR Credit: Building Product Disclosure – Material Ingredients
17				f. MR Credit: Construction and Demo Waste Management complying with Section 01 74 19
18				Construction Waste Management and Disposal. Include a sample spreadsheet showing how the
19				tipping information is going to be recorded to comply with LEED requirements.
20				g. IEQ Credit: Low-Emitting Materials
21				h. IEQ Credit: Construction IAQ Management Plan
22			4.	After CPM approval of the Preliminary Action Plan the Contractor shall update the plan monthly with
23				LEED information collected to date and be submitted as part of a monthly progress report.
24		F.	LEED	Progress Reports: Concurrent with each Application for Payment, submit reports comparing the actual
25			const	truction and purchasing activities with LEED requirements for the following:
26			1.	SS Prerequisite: Construction Activities Pollution Prevention
27			2.	MR Prerequisite: Construction and Demolition Waste Management Reporting
28			3.	MR Credit: Building Product Disclosure – Environmental Product Disclosures
29			4.	MR Credit: Building Product Disclosure – Source Materials
30			5.	MR Credit: Building Product Disclosure – Material Ingredients
31			6.	MR Credit: Construction and Demo Waste Management
32			7.	IEQ Credit: Low-Emitting Materials
33		G.	LEED	Documentation Online Submittals: The Contractor shall be responsible for completing the following LEED
34			subm	hissions using the LEED online tool for credit submission to USGBC. The LEED Project Administrator will
35			detei	rmine if the information prepared by the Contractor is satisfactory for USGBC submission.
36			1.	SS Prerequisite: Construction Activities Pollution Prevention
37			2.	MR Prerequisite: Construction and Demolition Waste Management Reporting
38			3.	MR Credit: Building Product Disclosure – Environmental Product Disclosures
39			4.	MR Credit: Building Product Disclosure – Source Materials
40			5.	MR Credit: Building Product Disclosure – Material Ingredients
41			6.	MR Credit: Construction and Demo Waste Management
42			7.	IEQ Credit: Low-Emitting Materials
43				
44	1.6		RIVIATI	UNAL SUBMITTALS
45		A.	Qual	Iffication Data: For LEED coordinator.
46		в.	Proje	ect Materials Cost Data: Provide statement indicating total cost for materials used for Project. Costs exclude
47			labor	, overhead, and profit. Include breakout of costs for the following categories of items:
48			1.	Furniture.
49			2.	Plumping.
5U E1			ე. ⊿	Ivieulaliudi.
21			4. E	Electrical. Specialty items such as elevators and equipment
52			э. с	Specially items such as elevators and equipment.
53			ь.	wood-based construction materials.
54 55	17			SUPANCE
55	1./			JURANUE Coordinator: The Contractor is to opeage an experienced LEED Associated Disfersional to socializate LEED.
50 57		А.	LEED	iconominator. The contractor is to engage an experienced LEED-Accredited Professional to coordinate LEED
57 58			requi	nements. LLLD Coordinator may also serve as waste management coordinator.
20				
	WARN	IER PAR		/UNITY RECREATION CENTER

1	1.8	CONT	'RACTOR RESPONSIBILITIES
2 3		A.	This project has been registered with USGBC via LEED Online. The Contractor shall provide all necessary documentation for LEED BD&C v4.0 certification in accordance with the specifications. Format and content of all construction documentation must be in accordance with the LEED Peference Guide requirements for supporting
4 5			data required in event of USGPC audit of the particular credit. Contractor is required to coordinate all
5			requirements for credits stated in this section to assure assembled data is acceptable to USGBC and respond to
7			USGBC requests for additional construction data in the course of preparing the project for certification.
8 9 10	<u>PART</u>	<u>2 – PRC</u>	<u>DDUCTS</u>
10	2.1	МАТЕ	FRIALS, GENERAL
12		A.	Provide products and procedures necessary to obtain LEED credits required in this Section. Although other
13 14			Sections may specify some requirements that contribute to LEED credits, the Contractor shall determine additional materials and procedures necessary to obtain LEED credits indicated.
15		В.	Refer to LEED Guidebook for further information.
16 17	2.2	BUILD	DING PRODUCT DISCLOSURE AND OPTIMIZATION
18		Α.	MR Credit Product Disclosure and Optimization - Environmental Product Declarations (EPD)
19 20			1. At least 20 different products from at least five different manufacturers shall have Environmental Product Declarations that comply with LEED requirements. Industry-wide (generic) Environmental Product
21			Declarations shall be valued as one-half of a product.
22		В.	MR Credit Product Disclosure and Optimization – Sourcing of Raw Materials
23			1. At least 20 different products from at least five different manufacturers shall have publically released
24			reports that comply with LEED requirements for raw material source and extraction reporting. Self-
25			declared reports by manufacturers shall be valued as one-half of a product.
26		C.	MR Credit Product Disclosure and Optimization – Material Ingredients
27			1. At least 20 different products from at least five different manufacturers shall comply with LEED
28			requirements for material ingredient reporting.
29 30	2.3	IOW-	FMITTING MATERIALS
31		A.	Paints and Coatings
32			1. For field applications that are inside the weatherproofing system, paints and coatings shall comply with
33			VOC content limits of authorities having jurisdiction and the following VOC content limits:
34			i. Flat Paints and Coatings: 50 g/L.
35			ii. Non-flat Paints and Coatings: 50 g/L.
36			iii. Dry-Fog Coatings: 150 g/L.
37			iv. Primers, Sealers, and Undercoaters: 100 g/L.
38			v. Rust-Preventive Coatings: 100 g/L.
39			vi. Zinc-Rich Industrial Maintenance Primers: 100 g/L.
40			vii. Pretreatment Wash Primers: 420 g/L.
41			viii. Clear Wood Finishes, Varnishes: 275 g/L.
42			ix. Clear Wood Finishes, Lacquers: 275 g/L.
43			x. Floor Coatings: 50 g/L.
44			xi. Shellacs, Clear: 730 g/L.
45			xii. Shellacs, Pigmented: 550 g/L.
46			xiii. Stains: 100 g/L.
47			2. For field applications that are inside the weatherproofing system, 90 percent of paints and coatings shall
48			comply with the requirements of the California Department of Public Health's "Standard Method for the
49			Testing and Evaluation of Volatile Organic Chemical Emissions from Indoor Sources Using Environmental
50			Chambers."
51		В.	Adhesives and Sealants
52			1. For field applications that are inside the weatherproofing system, adhesives and sealants shall comply
53			with VOC content limits of authorities having jurisdiction and the following VOC content limits:
54			i. Wood Glues: 30 g/L.
55			ii. Metal-to-Metal Adhesives: 30 g/L.
56			iii. Adhesives for Porous Materials (Except Wood): 50 g/L.
57			iv. Subfloor Adhesives: 50 g/L.
58			v. Plastic Foam Adhesives: 50 g/L.
	WARN EXPAN	ier pari Ision	K COMMUNITY RECREATION CENTER

1			vi.	Carpet Adhesives: 50 g/L.
2			vii.	Carpet Pad Adhesives: 50 g/L.
3			viii.	VCT and Asphalt Tile Adhesives: 50 g/L.
4			ix.	Cove Base Adhesives: 50 g/L.
5			х.	Gypsum Board and Panel Adhesives: 50 g/L.
6			xi.	Rubber Floor Adhesives: 60 g/L.
7			xii.	Ceramic Tile Adhesives: 65 g/L.
8			xiii.	Multipurpose Construction Adhesives: 70 g/L.
9			xiv.	Fiberglass Adhesives: 80 g/L.
10			xv.	Contact Adhesives: 80 g/L.
11			xvi.	Structural Glazing Adhesives: 100 g/L.
12			xvii.	Wood Flooring Adhesives: 100 g/L.
13			xviii.	Structural Wood Member Adhesives: 140 g/L.
14			xix.	Single-Ply Roof Membrane Adhesives: 250 g/L.
15			XX.	Special-Purpose Contact Adhesives (That Are Used to Bond Melamine-Covered Board,
16				Metal, Unsupported Vinyl, Rubber, or Wood Veneer 1/16 Inch or Less in Thickness to Any
17				Surface): 250 g/L.
18			xxi.	Top and Trim Adhesives: 250 g/L.
19			xxii.	Plastic Cement Welding Compounds: 250 g/L.
20			xxiii.	ABS Welding Compounds: 325 g/L.
21			xxiv.	CPVC Welding Compounds: 490 g/L.
22			xxv.	PVC Welding Compounds: 510 g/L.
23			xxvi.	Adhesive Primer for Plastic: 550 g/L.
24			xxvii.	Sheet-Applied Rubber Lining Adhesives: 850 g/L.
25			xxviii.	Aerosol Adhesive, General-Purpose Mist Spray; 65 percent by weight.
26			xxix.	Aerosol Adhesive, General-Purpose Web Spray: 55 percent by weight.
27			xxx.	Special-Purpose Aerosol Adhesives (All Types): 70 percent by weight.
28			xxxi.	Other Adhesives: 250 g/L.
29			xxxii.	Architectural Sealants: 250 g/L.
30			xxxiii.	Non-membrane Roof Sealants: 300 g/L.
31			xxxiv.	Single-Plv Roof Membrane Sealants: 450 g/L.
32			xxxv.	Other Sealants: 420 g/L.
33			xxxvi.	Sealant Primers for Nonporous Substrates: 250 g/L.
34			xxxvii.	Sealant Primers for Porous Substrates: 775 g/L.
35			xxxviii.	Modified Bituminous Sealant Primers: 500 g/L.
36			xxxix.	Other Sealant Primers: 750 g/L
37		2 For	field annli	cations that are inside the weatherproofing system 90 percent of adhesives and sealants
38		sha	all comply y	vith the requirements of the California Department of Public Health's "Standard Method for
39		the	Testing	and Evaluation of Volatile Organic Chemical Emissions from Indoor Sources Using
40		Env	/ironmenta	I Chambers "
41	F.	Flooring	, nonnenta	
42		1 Flo	oring shall	comply with the requirements of the California Department of Public Health's "Standard
43		1. Ne	othod for th	e Testing and Evaluation of Volatile Organic Chemical Emissions from Indoor Sources Using
44		Env	/ironmenta	I Chambers "
45	F	Composite	Wood	
46	••	1 Co	mnosite wa	and agrifiber products, and adhesives shall be made using ultra-low emitting formal debyde
40		res	ins as defi	ned in the California Air Resources Board's "Airborne Toxic Control Measure to Reduce
48		For	maldehvde	Emissions from Composite Wood Products" or shall be made with no added
40 //Q		for	maldehvde	Emissions from composite wood froducts of shall be made with no added
50	G	Ceilings M	alls and Tl	nermal Insulation
50	0.		lings walls	and thermal insulation shall comply with the requirements of the California Department of
52		I. Cer Pul	nic Health'	s "Standard Method for the Testing and Evaluation of Volatile Organic Chemical Emissions
52		fro	m Indoor S	ources Using Environmental Chambers "
54		110		
55	PART 3 – FYF			
56		2011014		
50				

-	3.1			
2		Α.	Smol	king is not permitted within the building or within 25 feet (8 m) of entrances, operable windows, or outdoor-
3			air in	takes.
4				
5	3.2	CONS	STRUCT	ION ACTIVITIES POLLUTION PREVENTION
6		Α.	SS Pr	erequisite - Construction Activities Pollution Prevention:
7			1.	Follow LEED instructions in LEED NCv4.0 Reference Guide and complying with Section 31 25 00, Erosion
8				Control. Comply with EPA Construction General Permit (CGP) standard 2012.
9			2.	Contractor is responsible for completing the LEED online credit template and attaching the following
10				information to the template:
11				a. Provide record of compliance with Erosion and Sediment Control Plan:
12				i. Monthly photographs of barriers and containment.
13				ii. Monthly photographs of dust control measures
14				iii. Records of inspections by agency in charge of overseeing compliance.
15				iv. Include dust control measures
16				b. Several early, a middle and several near end prevention plan checks and reports will be required
17				as an upload to LEED Online – assume 6 checks and reports over the duration of the project.
18			3.	The LEED Project Administrator will determine if the information prepared by the Contractor is
19				satisfactory for GBCI submission.
20				
21	3.3	BUILI	DING P	RODUCT DISCLOSURE AND OPTIMIZATION
22		Α.	MR C	redits Building Product Disclosure Optimization – EPDs, Sourcing and Ingredients
23			1.	Environmental Product Declarations – comply with one or both of the following Options:
24				a. Option 1: Environmental Product Declarations (1 point)
25				b. Option 2: Multi-Attribute Optimization (1 point) including products that demonstrate impact
26				reduction below industry average in global warming potential, ozone depletion, acidification of
27				land and water, eutrophication, tropospheric ozone, or other USGBC approved program.
28			2.	Sourcing of Raw Materials – comply with one or both of the following Options:
29				a. Option 1: Raw Material Source and Extraction Reporting (1 point)
30				b. Option 2: Leadership Extraction Practices (1 point) including producer responsibility, bio-based
31				materials, wood products, material reuse, recycle content or other approved USGBC program
32			3.	Material Ingredients - comply with one or two of the following Options:
33				a. Option 1: Material Ingredient Reporting (1 point)
34				b. Option 2: Material Ingredient Optimization (1 point) including GreenScreen v1.2 Benchmark,
35				Cradle to Cradle Certification, REACH Optimization or other approved USGBC program.
36				c. Option 3: Product Manufacturer Supply Chain Optimization (1 point) including products from
37				manufacturers with validated and robust safety, health, hazard and risk programs that document
38				99% by weight of the ingredients used to make the product.
39			4.	Contractor to complete and submit the MR building product disclosure and optimization calculator.
40				available with the project in LEED Online
41			5.	Contractor to submit supporting documentation including EPD and LCA reports, corporate sustainability
42				reports, product declarations, labels, REACH, GreenScreen Benchmark, LT scores or other compliance
43				summary documents. LEED project administrator and/or GBCI may require revisions and additions to this
44				documentation and Contractor should plan accordingly.
45			6.	The LEED Project Administrator will determine if the information prepared by the Contractor is
46			-	satisfactory for GBCI submission.
47				
48	3.4	CONS	STRUCT	ION AND DEMOLITION WASTE MANAGEMENT PLANNING
49		Α.	MR P	Prerequisite and Credit: Comply with Division 1 Section "Construction Waste Management and Disposal".
50		-	1.	Contractor is required to create a Construction Waste Management Plan that includes:
51			•	a. Establishing waste diversion goals for the project by identifying at least five material streams
52				targeted for diversion. Approximate a percentage of the overall project waste that these materials
53				represent.
54				b. Specifying whether materials will be separated or commingled and describe the diversion
55				strategies planned for the project. Describe where the material will be taken and how the
56				recycling facility will process the material.
57				c. A final report detailing all major waste streams generated including disposal and diversion rates
			2	Contractory is required to most the following minimum goals
58				CONTACIOUS LEONIED TO WEEL THE TONOMINE WINNING BOAT.

1				a. Option 1 Path 2 – Divert 75% and four material streams, or
2				i. A material stream can be a specific material category that is diverted in a specific way or a
3				mixture of several material categories that are diverted in a specific way.
4				ii. Best practice is that a material stream should constitute at least 5% (by weight or volume)
5				of total diverted materials.
6				iii Examples of material streams include Plastic Carnet Paper/Cardboard Wood metal
7				Sheetrock, Brick, Concrete, Shingles, deconstructed materials, commingled waste, reuse of
, 8				deconstructed materials onsite source separation where each material is sent to a specific
q				facility or suppliers take back of materials
10				h Ontion 2 – Do not generate more than 2.5 nounds of construction waste per square foot of the
11				buildings floor area (2 points)
12			2	Contractor is responsible for completing the LEED online credit template. Attached documentation in
12			5.	contractor is responsible for completing the LLLD online credit template. Attached documentation in
13				Monthly photographs of waste recycling serting area including:
14				a. Monthly photographs of waste recycling softing area including.
15				i. Debits control rending.
10				ii. Signage cleany identifying the containers content.
10				b. Spreadsheet containing the following monitation.
18				i. Diverted materials description.
19				II. Diverted materials/waste nauler name.
20				III. Date of each haul
21				iv. Quantity of material in each naul.
22			4	c. Copies of recycling vender and waste nauler tipping receipts.
23			4.	The LEED Project Administrator will determine if the information prepared by the Contractor is
24				satisfactory for GBCI submission.
25	Э Е			
20	3.5			radit Low Emitting Materials: Intent is to reduce concentrations of chemical contaminants that can
27		А.	damag	real - Low Emitting Materials. Interior is to reduce concentrations of chemical contaminants that can
20				E all quality, numar reach, productivity and the environment.
29			1. 2	Contractor is required to complete and uplead the following documentation to LEED Online:
21			Ζ.	LISGBC low-emitting materials calculator (available at the project resources in LEED Online)
32				a. OSOBE low-emitting materials calculator (available at the project resources in EEED online) b. Product information (e.g. MSDS, third party certifications, testing reports, etc.) for relevant
32				materials
37			3	Contractor is responsible for one of the following point options:
35			5.	a Ontion 1: Product Category threshold compliance in 2 of the following categories (1 point)
36				i Interior naints and coatings annlied onsite: 90% by volume for emissions 100% VOC
37				content
38				ii Interior adhesives and sealants applied onsite (including flooring adhesive): 90% by volume
39				for emissions and 100% for VOC content
40				iii. Flooring: 100% emissions
41				iv. Composite Wood: 100% emissions (separate Composite Wood Evaluation)
42				v. Ceilings, walls, thermal and acoustic insulation: 100% emissions
43				vi. Furniture: 90% by cost (separate Furniture Evaluation)
44				b. Option 2: If some products in a category do not meet the criteria, use the Budget Calculation
45				Method meeting << >=50% and <70% (1 point) in any of the following categories:
46				i. flooring.
47				ii. ceilings,
48				iii. walls,
49				iv. thermal and acoustic insulation or
50				v. furniture
51			4.	Composite Wood Evaluation - Composite wood, agrifiber products, and adhesives shall be made using
52				ultra-low emitting formaldehyde resins as defined in the California Air Resources Board's "Airborne Toxic
53				Control Measure to Reduce Formaldehyde Emissions from Composite Wood Products" or shall be made
54				with no added formaldehyde. Salvaged and reused architectural millwork more than one year old at the
55				time of occupancy is considered compliant, provided it meets the requirements for any site-applied
56				paints, coatings, adhesives, and sealants.
57			5.	<u>Furniture Evaluation</u> - New furniture and furnishing items must be tested in accordance with ANSI/BIFMA
58				Standard Method M7.1-2011. Comply with ANSI/BIFMA e3-2011 Furniture Sustainability Standard,
-	WARN	ER PARK	сомм	UNITY RECREATION CENTER

1 2 3 4 5 6 7 8 9			6.	sections 7.6.1 and 7.6.2, using either the concentration modeling approach or the emissions factor approach. Model the test results using the open plan, private office, or seating scenario in ANSI/BIFMA M7.1, as appropriate. USGBC-approved equivalent testing methodologies and contaminant thresholds are also acceptable. Salvaged and reused furniture more than one year old at the time of use is considered compliant, provided it meets the requirements for any site-applied paints, coatings, adhesives, and sealants. The LEED Project Administrator will determine if the information prepared by the Contractor is satisfactory for GBCI submission. Revisions and time to answer review questions should be assumed.
10	3.6	CONST	TRUCTIC	NN INDOOR-AIR-OUALITY MANAGEMENT PLAN
11	0.0	A.	IFO Cre	edit Construction IAO Management Plan: Intent is to promote the well-being of construction workers and
12			buildin	g occupants by minimizing indoor air quality problems associated with construction and renovation.
13			Contra	ctor to include at a minimum the following elements into the plan:
14			1.	Comply with SMACNA's "SMACNA IAO Guideline for Occupied Buildings under Construction."
15			2.	Prohibit the use of tobacco products inside the building and within 25 feet of the building entrances
16				during construction.
17			3.	Protect absorptive materials stored on-site and installed from moisture damage.
18			4.	If Owner authorizes use of permanent heating, cooling, and ventilating systems during construction
19				period as specified in Division 1 Section "Temporary Facilities and Controls", install filter media having a
20				MERV 8 according to ASHRAE 52.2 at each return-air inlet for the air-handling system used during
21				construction.
22			5.	Replace all air filters immediately prior tooccupancy.
23		В.	Provide	e record of compliance with Indoor Air Quality Management Plan:
24				a. Monthly photographs of equipment and ductwork protection.
25				b. Monthly photographs of filters used to protect air distribution and equipment.
26				c. Contractor's report documenting that MERV 8 filters were used to protect equipment during
27				construction and filters meeting final design requirements were installed prior to occupancy.
28				
29	3.7	SUPPL	EMENT	
30		Α.	The su	pplement listed below, up to "End of Section," is a part of this Specification:
31			1.	LEED BD&C v4.0 Project Checklist.
32				a. All credits listed for reference
33				b. Only Bold , <i>Italic</i> credits or prerequisites listed with a "C" are in the Scope of the Contractor
34				c. All identified construction Prerequisites are required to be achieved to complete the certification
35				process and are the responsibility of the Contractor. Care needs to be taken to ensure all
36				prerequisites are awarded to the project.
3/				a. All identified construction Credits are required to be achieved and are the responsibility of the
38				contractor. Given certain point totais and project specific circumstances as the project
39				progresses, with proper notice to the CPIVI, certain credits or credit point thresholds can be
4U 41			2 Tare	emminated from the project. Written notice and approval is required.
41 40			z. rarg	et CREL VOCS, TABLE 4-1 TOF INCOOL AIT QUAITY TESTING
42				

Warner Park Community Recreation Center Expansion LEED v4.0 for BD+C: New Construction and Major Renovations Project Checklist 05-16-2024 1 0 0 0 **Integrative Process** Possible Points: 1 Y ?Υ ?N Ν D/C D Credit 1 1 Integrative Process 6 0 0 10 **Location and Transportation** Possible Points: 16 D/C Υ ?Y ?N Ν 0 D Credit LEED for Neighborhood Development Location 16 or Credit 1 Sensitive Land Protection 1 D 2 D Credit **High Priority Site** 1-2 Credit 3 1-5 Surrounding Density and Diverse Uses 2 D Credit 4 1-5 1 D Access to Quality Transit Credit 1 0 D **Bicycle Facilities** 1 Credit 1 0 **Reduced Parking Footprint** 1 D **Green Vehicles** 1 D Credit 1 2 8 10 0 0 **Sustainable Sites** Possible Points: Υ ?Υ ?N Ν D/C **Construction Activity Pollution Prevention** Υ -С Prereq ---Credit 1 1 D Site Assessment 2 1-2 D Credit Site Development - Protect or Restore Habitat 1 D Credit Open Space 1 2-3 3 D Credit Rainwater Management 2 D Credit Heat Island Reduction 1-2 1 D Credit **Light Pollution Reduction** 1 2 9 Possible Points: 11 ٥ 0 Water Efficiency Y ?Y ?N Ν D/C Υ _ -_ D Prereq Water Use Reduction—20% Reduction _ Υ ---Prereg Water Efficient Landscaping -D Υ --_ D Prereq Innovative Wastewater Technologies 2 D Credit **Outdoor Water Use Reduction** 1-2 6 Credit 1-6 D Indoor Water Use Reduction 1-2 2 D Credit Cooling Tower Water Use 1 D Credit 1 Water Metering

1

21	0	0	12		Energy and A	tmosphere	Possible Points:	33
Y	?Y	?N	Ν	D/C				
Y	-	-	-	С	Prereq	Fundamental Commissioning and Verifice	ation	-
Y	-	-	-	D	Prereq	rereq Minimum Energy Performance		-
Y	-	-	-	D	Prereq	rereq Building Level Energy Metering		-
Y	-	-	-	D	Prereq	Fundamental Refrigerant Management		-
6				С	Credit	Enhanced Commissioning		2-6
9			9	D	Credit	Optimize Energy Performance		1-18
			1	D	Credit	Advanced Energy Metering		1
			2	D	Credit	Demand Response		2
3				D	Credit	Renewable Energy Production		1-3
1				D	Credit	Enhanced Refrigerant Management		1
2				D	Credit	Green Power and Carbon Offsets		1-2
		•						
5	0	0	8		Materials and	d Resources	Possible Points:	13
Y	?Y	?N	Ν	D/C				
Y	-	-	-	D	Prereq	Storage and Collection of Recyclables		-
Ŷ	-	-	-	С	Prereq	Construction and Demolition Waste Man	agement Reporting	-
			5	D	Credit	Building Life-Cycle Impact Reduction		2-5
1			1	С	Credit	redit Building Product Disclosure - EPD 1		1-2 pts
1			1	С	Credit	edit Building Product Disclosure – Source Materials 1-		1-2 pts
1			1	С	Credit	Credit Building Product Disclosure – Material Ingredients 1		1-2 pts
2				С	Credit	Credit Construction and Demo Waste Management 1		1-2 pts
10	0	0	6		Indoor Enviro	onmental Quality	Possible Points:	16
Y	?Y	?N	Ν	D/C				
Y	-	-	-	D	Prereq	Minimum Indoor Air Quality Performance		-
Y	-	-	-	D	Prereq	Environmental Tobacco Smoke (ETS) Cont	rol	-
2				D	Credit	Enhanced Indoor Air Quality Strategies		1-2
3				С	Credit	Low-Emitting Materials		1-3 pts
1				С	Credit	Construction IAQ Management Plan		1
			2	С	Credit	Indoor Air Quality Assessment		1-2 pts
1				D	Credit	Thermal Comfort		1
1			1	D	Credit	Interior Lighting		1-2
1			2	D	Credit	Daylight		1-3
			1	D	Credit	Quality Views		1
1				D	Credit	Acoustic Performance		1

1	0	0	4		Innovation a	nd Design Process	Possible Points:	6
Y	?Υ	?N	Ν	D/C				
			0	D/C	Credit 1.1	Innovation in Design: X		1
1				D	Credit 2	LEED Accredited Professional		1
3	0	0	0		Regional Prio	rity Credits	Possible Points:	4
Y	?Υ	?N	Ν	D/C				
1				D/C	Credit 1.1	Regional Priority: Bicycle Facilities		1
1				D/C	Credit 1.2	Regional Priority: Sensitive Land		1
1				D/C	Credit 1.3	Regional Priority: Energy Performance		1
	•	•	•	•	•			
51	0	0	59		Total		Possible Points:	110
Y	?Υ	?N	Ν					

1

Table 4-1 Target CREL VOCs and their maximum allowable concentrations

No.	Compound Name	CAS No.	Allowable Conc. ^a (μg/m ³)
1	Acetaldehyde	75-07-0	70
2	Benzene	71-43-2	30
3	Carbon disulfide	75-15-0	400
4	Carbon tetrachloride	56-23-5	20
5	Chlorobenzene	108-90-7	500
6	Chloroform	67-66-3	150
7	Dichlorobenzene (1,4-)	106-46-7	400
8	Dichloroethylene (1,1)	75-35-4	35
9	Dimethylformamide (N,N-)	68-12-2	40
10	Dioxane (1,4-)	123-91-1	1,500
11	Epichlorohydrin	106-89-8	1.5
12	Ethylbenzene	100-41-4	1,000
13	Ethylene glycol	107-21-1	200
14	Ethylene glycol monoethyl ether	110-80-5	35
15	Ethylene glycol monoethyl ether acetate	111-15-9	150
16	Ethylene glycol monomethyl ether	109-86-4	30
17	Ethylene glycol monomethyl ether acetate	110-49-6	45
18	n/a	n/a	n/a
19	Hexane (n-)	110-54-3	3,500
20	Isophorone	78-59-1	1,000
21	Isopropanol	67-63-0	3,500
22	Methyl chloroform	71-55-6	500
23	Methylene chloride	75-09-2	200
24	Methyl <i>t</i> -butyl ether	1634-04-4	4,000
25	Naphthalene	91-20-3	4.5
26	Phenol	108-95-2	100
27	Propylene glycol monomethyl ether	107-98-2	3,500
28	Styrene	100-42-5	450
29	Tetrachloroethylene	127-18-4	17.5
30	Toluene	108-88-3	150
31	Trichloroethylene	79-01-6	300
32	Vinyl acetate	108-05-4	100
33-35	5 Xylenes, technical mixture (m-,	108-38-3,	350
	o-, p-xylene combined)	95-47-6,	
		106-42-3	

3

8

a) Refer to http://www.oehha.ca.gov/air/chronic_rels/AllChrels.html. All maximum allowable concentrations are one-half the corresponding CREL adopted by Cal/EPA OEHHA with the exception of formaldehyde. For any future changes in the CREL list by OEHHA, values in Table 4.1 shall continue to apply until these changes are published in the Standard Method.

END OF SECTION

Page Intentionally Left Blank

			COMMISSIONING
DVDT	1_0	ENERAL	
FANI	1 1		
	1.1.		NC
	1.2.		
	1.5 1 /		
	1.4		
	1.5		
	1.0	SVSTEMS TO BE COMM	
рлрт	т./ ГЭ_ DI		
FANI	2 - FI 2 1		
DART	2.1 . 3 - EX		
	2 1		Λ
	2.1 2.7		TINGS
	22		
	5.5 ว⊿		
	э.4 э с		
	5.5 2 C		11001011100 FRUCEDURES
	3.0 ว 7		IK LALIDKATIUN
	3./	NON-CONFORMANCE.	
	1 0		
	~		
1.1.	SUN		and a second state of the second second second at the second second second as the second second second second s
	А.	Purpose: Define the	responsibilities of the parties involved and the procedures related to the commissioning
		process	
_		ATED ODEOLEICATIONS	
	KFI	ATED SPECIFICATIONS	
1.2.	^		Duringt Management and Consultantian
1.2.	A.	Section 01 31 13	Project Management and Coordination
1.2.	A. B.	Section 01 31 13 Section 01 31 19	Project Management and Coordination Project Meetings
1.2.	А. В. С.	Section 01 31 13 Section 01 31 19 Section 01 31 23	Project Management and Coordination Project Meetings Project Management
1.2.	А. В. С. D.	Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting
1.2.	A. B. C. D. E.	Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 33 23	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals
1.2.	A. B. C. D. E. F.	Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 33 23 Section 01 45 16	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control
1.2.	A. B. C. D. E. F. G.	Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 33 23 Section 01 45 16 Section 01 77 00	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures
1.2.	A. B. C. D. E. F. G. H.	Section 01 31 13 Section 01 31 13 Section 01 31 23 Section 01 32 26 Section 01 33 23 Section 01 45 16 Section 01 77 00 Section 01 78 23	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data
1.2.	A. B. C. D. E. F. G. H.	Section 01 31 13 Section 01 31 13 Section 01 31 23 Section 01 32 26 Section 01 32 23 Section 01 45 16 Section 01 77 00 Section 01 78 23 Section 01 78 39	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings
1.2.	A. B. C. D. E. F. G. H. I. J.	Section 01 31 13 Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 33 23 Section 01 45 16 Section 01 77 00 Section 01 78 23 Section 01 78 39 Section 01 79 00	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training
1.2.	A. B. C. D. E. F. G. H. I. J. K.	Section 01 31 13 Section 01 31 13 Section 01 31 23 Section 01 32 26 Section 01 32 23 Section 01 45 16 Section 01 77 00 Section 01 78 23 Section 01 78 39 Section 01 79 00 Section 01 81 13	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training Sustainable Design Requirements
1.2.	A. B. C. D. E. F. G. H. I. J. K. L.	Section 01 31 13 Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 33 23 Section 01 45 16 Section 01 77 00 Section 01 78 23 Section 01 78 39 Section 01 79 00 Section 01 81 13 Section 01 95 00	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training Sustainable Design Requirements Measurement & Verification
1.2.	A. B. C. D. E. F. G. H. I. J. K. L.	Section 01 31 13 Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 32 23 Section 01 45 16 Section 01 77 00 Section 01 78 23 Section 01 78 39 Section 01 79 00 Section 01 81 13 Section 01 95 00 Section 23 05 93	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training Sustainable Design Requirements Measurement & Verification Testing, Adjusting, and Balancing for HVAC
1.2.	A. B. C. D. E. F. G. H. I. J. K. L. N.	Section 01 31 13 Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 32 23 Section 01 45 16 Section 01 77 00 Section 01 78 23 Section 01 78 39 Section 01 79 00 Section 01 79 00 Section 01 81 13 Section 01 95 00 Section 23 05 93 Section 23 09 00	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training Sustainable Design Requirements Measurement & Verification Testing, Adjusting, and Balancing for HVAC Instrumentation and Control for HVAC
1.2.	A. B. C. D. E. F. G. H. J. K. L. N. O.	Section 01 31 13 Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 32 23 Section 01 45 16 Section 01 77 00 Section 01 78 23 Section 01 78 39 Section 01 79 00 Section 01 79 00 Section 01 81 13 Section 01 95 00 Section 23 05 93 Section 23 09 00 Section 23 09 23	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training Sustainable Design Requirements Measurement & Verification Testing, Adjusting, and Balancing for HVAC Instrumentation and Control for HVAC Direct Digital Control (DDC) System for HVAC
1.2.	A. B. C. D. E. F. G. H. I. J. K. L. N. O. P.	Section 01 31 13 Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 32 23 Section 01 45 16 Section 01 45 16 Section 01 77 00 Section 01 78 23 Section 01 78 39 Section 01 79 00 Section 01 79 00 Section 01 81 13 Section 01 95 00 Section 23 05 93 Section 23 09 00 Section 23 09 23 Section 23 09 93	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training Sustainable Design Requirements Measurement & Verification Testing, Adjusting, and Balancing for HVAC Instrumentation and Control for HVAC Direct Digital Control (DDC) System for HVAC Sequence of Operations for HVAC DDC
1.2.	A. B. C. D. E. F. G. H. I. J. K. L. N. O. P.	Section 01 31 13 Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 32 23 Section 01 45 16 Section 01 77 00 Section 01 78 23 Section 01 78 39 Section 01 79 00 Section 01 79 00 Section 01 81 13 Section 01 95 00 Section 23 05 93 Section 23 09 00 Section 23 09 23 Section 23 09 93	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training Sustainable Design Requirements Measurement & Verification Testing, Adjusting, and Balancing for HVAC Instrumentation and Control for HVAC Direct Digital Control (DDC) System for HVAC Sequence of Operations for HVAC DDC
1.2.	A. B. C. E. F. G. H. I. J. K. L. N. O. P.	Section 01 31 13 Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 32 23 Section 01 45 16 Section 01 77 00 Section 01 78 23 Section 01 78 39 Section 01 79 00 Section 01 79 00 Section 01 81 13 Section 01 95 00 Section 23 05 93 Section 23 09 00 Section 23 09 93	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training Sustainable Design Requirements Measurement & Verification Testing, Adjusting, and Balancing for HVAC Instrumentation and Control for HVAC Direct Digital Control (DDC) System for HVAC Sequence of Operations for HVAC DDC
1.2.	A. B. C. D. E. F. G. H. I. J. K. L. N. O. P. REF	Section 01 31 13 Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 32 26 Section 01 32 23 Section 01 45 16 Section 01 77 00 Section 01 78 23 Section 01 78 39 Section 01 79 00 Section 01 79 00 Section 01 81 13 Section 01 95 00 Section 23 05 93 Section 23 09 00 Section 23 09 93 Section 23 09 93 Section 23 09 93	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training Sustainable Design Requirements Measurement & Verification Testing, Adjusting, and Balancing for HVAC Instrumentation and Control for HVAC Direct Digital Control (DDC) System for HVAC Sequence of Operations for HVAC DDC
1.2.	A. B. C. D. E. F. G. H. I. J. K. L. N. O. P. REF A. R	Section 01 31 13 Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 32 23 Section 01 45 16 Section 01 45 16 Section 01 77 00 Section 01 78 23 Section 01 78 39 Section 01 79 00 Section 01 79 00 Section 01 81 13 Section 01 95 00 Section 23 05 93 Section 23 09 00 Section 23 09 93 Section 23 09 93 Section 23 09 93 Section 23 09 93	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training Sustainable Design Requirements Measurement & Verification Testing, Adjusting, and Balancing for HVAC Instrumentation and Control for HVAC Direct Digital Control (DDC) System for HVAC Sequence of Operations for HVAC DDC
1.2.	A. B. C. D. E. F. G. H. I. J. K. L. N. O. P. REF A. B. C	Section 01 31 13 Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 32 26 Section 01 32 23 Section 01 45 16 Section 01 77 00 Section 01 78 23 Section 01 78 39 Section 01 78 00 Section 01 79 00 Section 01 79 00 Section 01 81 13 Section 01 95 00 Section 23 05 93 Section 23 09 00 Section 23 09 93 Section 23 09 93 Section 23 09 93 Section 23 09 93 SECTION 23 09 93	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training Sustainable Design Requirements Measurement & Verification Testing, Adjusting, and Balancing for HVAC Instrumentation and Control for HVAC Direct Digital Control (DDC) System for HVAC Sequence of Operations for HVAC DDC
1.2.	A. B. C. D. E. F. G. H. I. J. K. L. N. O. P. REF A. B. C. D	Section 01 31 13 Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 32 26 Section 01 32 23 Section 01 45 16 Section 01 45 16 Section 01 77 00 Section 01 78 23 Section 01 78 39 Section 01 78 00 Section 01 79 00 Section 01 79 00 Section 01 79 00 Section 01 81 13 Section 01 95 00 Section 23 05 93 Section 23 09 00 Section 23 09 93 Section 23 09 93	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training Sustainable Design Requirements Measurement & Verification Testing, Adjusting, and Balancing for HVAC Instrumentation and Control for HVAC Direct Digital Control (DDC) System for HVAC Sequence of Operations for HVAC DDC
1.2.	A. B. C. D. E. F. G. H. I. J. K. L. N. O. P. REF A. B. C. D. F.	Section 01 31 13 Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 32 23 Section 01 45 16 Section 01 45 16 Section 01 77 00 Section 01 78 23 Section 01 78 39 Section 01 78 00 Section 01 79 00 Section 01 79 00 Section 01 79 00 Section 01 79 00 Section 01 81 13 Section 01 95 00 Section 23 05 93 Section 23 09 00 Section 23 09 93 Section 23 09 93 Section 23 09 93 Section 23 09 93 SERENCES ASHRAE Guideline 1. ASTM E2947-16: Star ASTM E2813-12: Star	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training Sustainable Design Requirements Measurement & Verification Testing, Adjusting, and Balancing for HVAC Instrumentation and Control for HVAC Direct Digital Control (DDC) System for HVAC Sequence of Operations for HVAC DDC
1.2.	A. B. C. D. E. F. G. H. I. J. K. L. N. O. P. REF A. B. C. D. E.	Section 01 31 13 Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 32 23 Section 01 45 16 Section 01 45 16 Section 01 77 00 Section 01 78 23 Section 01 78 39 Section 01 79 00 Section 01 79 00 Section 01 79 00 Section 01 81 13 Section 01 95 00 Section 23 05 93 Section 23 09 00 Section 23 09 93 Section 24 00 93 Section 25 00 93 Sectio	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training Sustainable Design Requirements Measurement & Verification Testing, Adjusting, and Balancing for HVAC Instrumentation and Control for HVAC Direct Digital Control (DDC) System for HVAC Sequence of Operations for HVAC DDC
1.2.	A. B. C. D. E. F. G. H. I. J. K. L. N. O. P. REF A. B. C. D. E.	Section 01 31 13 Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 32 23 Section 01 45 16 Section 01 77 00 Section 01 77 00 Section 01 78 23 Section 01 78 23 Section 01 78 00 Section 01 79 00 Section 23 09 93 Section 23 09 93 SERENCES ASHRAE Guideline 1. ASHRAE Guideline 0- ASTM E2947-16: Stat ASTM E2813-12: Stat NEBB – Procedural Stat	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training Sustainable Design Requirements Measurement & Verification Testing, Adjusting, and Balancing for HVAC Instrumentation and Control for HVAC Direct Digital Control (DDC) System for HVAC Sequence of Operations for HVAC DDC
1.2. 1.3	A. B. C. D. E. F. G. H. I. J. K. L. N. O. P. REF A. B. C. D. E. DEF	Section 01 31 13 Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 32 23 Section 01 45 16 Section 01 77 00 Section 01 77 00 Section 01 78 23 Section 01 78 23 Section 01 78 00 Section 01 79 00 Section 01 95 00 Section 23 09 90 Section 23 09 00 Section 23 09 93 Section 24 00 93 Section 25 00 93 Section 25 00 93 Section 26 00 93 Sectio	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training Sustainable Design Requirements Measurement & Verification Testing, Adjusting, and Balancing for HVAC Instrumentation and Control for HVAC Direct Digital Control (DDC) System for HVAC Sequence of Operations for HVAC DDC 1-2007, "HVAC&R Technical Requirements for The Commissioning Process". 2013, "The Commissioning Process". ndard Guide for Building Enclosure Commissioning ndard Practice for Building Enclosure Commissioning tandards for Building Systems Commissioning.
1.2. 1.3	A. B. C. D. E. F. G. H. I. J. K. L. N. O. P. REF A. B. C. D. E. DEF	Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 77 00 Section 01 78 23 Section 01 78 39 Section 01 79 00 Section 01 95 00 Section 23 05 93 Section 23 09 00 Section 23 09 03 Section 23 09 93 Section 23 09 23 Section 23 09 24 ASHRAE Guideline 1. ASHRAE Guideline 0- ASTM E2813-12: Stat NEBB – Procedural Stat Sectortions Acceptance Phase. P	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training Sustainable Design Requirements Measurement & Verification Testing, Adjusting, and Balancing for HVAC Instrumentation and Control for HVAC Direct Digital Control (DDC) System for HVAC Sequence of Operations for HVAC DDC 1-2007, "HVAC&R Technical Requirements for The Commissioning Process". 2013, "The Commissioning Process". ndard Guide for Building Enclosure Commissioning ndard Practice for Building Enclosure Commissioning tandards for Building Systems Commissioning.
1.2. 1.3	A. B. C. D. E. F. G. H. I. J. K. L. N. O. P. REF A. B. C. D. E. DEF A.	Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 77 00 Section 01 77 00 Section 01 78 23 Section 01 78 00 Section 01 79 00 Section 01 95 00 Section 23 05 93 Section 23 09 00 Section 23 09 03 Section 23 09 93 Section 23 09 12 ASHRAE Guideline 1. ASHRAE Guideline 0- ASTM E2813-12: Stat NEBB – Procedural Si Settiontons Acceptance Phase. P are performed.	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training Sustainable Design Requirements Measurement & Verification Testing, Adjusting, and Balancing for HVAC Instrumentation and Control for HVAC Direct Digital Control (DDC) System for HVAC Sequence of Operations for HVAC DDC 1-2007, "HVAC&R Technical Requirements for The Commissioning Process". 2013, "The Commissioning Process". Indard Guide for Building Enclosure Commissioning Indard For Building Enclosure Commissioning Indard Source for Building Enclosure Commissioning Indards for Building Systems Commissioning.
1.2. 1.3	A. B. C. D. E. F. G. H. I. J. K. L. N. O. P. REF A. B. C. D. E. DEF A.	Section 01 31 13 Section 01 31 19 Section 01 31 23 Section 01 32 26 Section 01 32 23 Section 01 32 23 Section 01 77 00 Section 01 78 23 Section 01 78 23 Section 01 78 00 Section 01 79 00 Section 01 79 00 Section 01 95 00 Section 23 09 93 Section 23 09 00 Section 23 09 23 Section 23 09 93 Section 2	Project Management and Coordination Project Meetings Project Management Construction Progress Reporting Submittals Field Quality Control Closeout Procedures Operation and Maintenance Data As-Built Drawings Demonstration and Training Sustainable Design Requirements Measurement & Verification Testing, Adjusting, and Balancing for HVAC Instrumentation and Control for HVAC Direct Digital Control (DDC) System for HVAC Direct Digital Control (DDC) System for HVAC Sequence of Operations for HVAC DDC 1-2007, "HVAC&R Technical Requirements for The Commissioning Process". 2013, "The Commissioning Process". ndard Guide for Building Enclosure Commissioning indard Practice for Building Enclosure Commissioning tandards for Building Systems Commissioning.

1		C.	Commissioning Plan (Cx Plan). An overall plan, developed before or after bidding, that provides the structure,
2			schedule and coordination planning for the commissioning process. The Cx Plan is included in the bid documents
3			and is to be reviewed by all contractors before submitting their bid.
4		D.	<u>Contract Documents.</u> The documents binding on parties involved in the construction of this project (drawings,
5			specifications, change orders, amendments, contracts, Cx Plan, etc.).
6 7		E.	<u>Construction Checklist (CC)</u> . a list of items to inspect and test equipment and components to verify proper installation of equipment. The CCs are provided by the CxA to the Sub.
8		F.	Datalogging Monitoring flows, currents, status, pressures, etc. of equipment using stand-alone dataloggers
9			separate from the control system.
10		G.	Deferred System Performance Tests. SPT's that are performed later, after substantial completion, due to partial
11			occupancy, equipment, seasonal requirements, design or other site conditions that prevent the tests from being
12			performed earlier.
13		Н.	Deficiency. A condition in the installation or function of a component, piece of equipment or system that is not in
14			compliance with the Contract Documents (that is, does not perform properly or is not complying with the
15			Owner's Project Requirements).
16		I.	Factory Testing. Testing of equipment on-site or at the factory by factory personnel with an Owner's
17			representative present.
18		J.	Indirect Indicators. Indicators of a response or condition, such as a reading from a control system screen
19			reporting a damper to be 100% closed.
20		К.	Manual Test. Using hand-held instruments, immediate control system readouts or direct observation to verify
21			performance (contrasted to analyzing monitored data taken over time to make the "observation").
22		L.	Monitoring. Recording parameters (flow, current, status, pressure, etc.) of equipment operation using
23			dataloggers or the trending capabilities of control systems.
24		M.	Over-written Value. Writing over a sensor value in the control system to see the response of a system (e.g.,
25			changing the outside air temperature value from 75F to 50F to verify economizer operation). See also "Simulated
26			Signal."
27		N.	<u>Owner's Project Requirements (OPR).</u> A document that describes what the Owner and stakeholders want to
28			achieve with this project and what expectations they have of the completed project.
29		0.	<u>Sampling.</u> Reviewing or testing only a fraction of the total number of identical or near identical pieces of
30		Б	equipment.
31 22		Ρ.	<u>Seasonal Performance Tests.</u> SPT's that are deferred until the system(s) will experience conditions closer to their design conditions
22		0	Simulated Condition Condition that is created for the nurnese of testing the response of a system (e.g. applying
37		Q.	$\frac{1}{2}$ have to a space sensor to see the response in a VAV box)
35		R	Simulated Signal Disconnecting a sensor and using a signal generator to send an amperage resistance or
36			pressure to the transducer and DDC system to simulate a sensor value
37		S.	System Performance Test (SPT). Dynamic testing of entire systems (rather than just components of the system)
38		0.	under full operation.
39		т.	Trending. Monitoring of control points using the building automation system.
40			
41	1.5	DESCR	IPTION
42		Α.	General: Commissioning (Cx) is a systematic process of verifying that all building systems perform interactively to
43			meet the Owner's Project Requirements (OPR). This is achieved by beginning in the planning phase with
44			documenting the OPR and continuing through design, construction, acceptance, and the warranty period with
45			verification of performance. The Cx process shall encompass and coordinate the traditionally separate functions
46			of system documentation, equipment startup, control system calibration, testing and balancing, performance
47			testing and training. Cx during the construction phase is intended to achieve the following specific objectives
48			according to the Contract Documents:
49			1. Verify that applicable equipment and systems are installed according to the manufacturer's
50			recommendations and to industry accepted minimum standards and that they receive adequate
51			operational checkout by installing contractors.
52			2. Verify and document proper performance of equipment and systems.
53			3. Verify that O&M documentation is complete.
54		_	4. Verity that the Owner's operating personnel are adequately trained.
55		В.	The Cx process does not take away from or reduce the responsibility of the system designers or installing
56		C	contractors to provide a finished and fully functioning product.
5/ E0		ι.	The commissioning authority (CXA) has no authority to change, modify or direct any work. The CXA can only provide commonte and suggestions
50			provide comments and suggestions.
	WARN	ER PARK	COMMUNITY RECREATION CENTER

1 2 3		D.	Commissionir Cx Plan regula Plan.	ng Plan arly as t	. The Cx Plan provides guidance in the execution of the Cx process. The CxA will update the the project progresses. The Drawings and Specifications will take precedence over the Cx
4	_				
5	1.6	RESP	ONSIBILITIES		
6		Α.	General Cont	ractor	(GC) and Subcontractors (Subs)
7			1.	Const	ruction and Acceptance Phase
8				a.	Provide assistance to the Construction Manager CM in the coordination of the Cx work by
9					the CxA, and with the CM and CxA ensure that Cx activities are being scheduled into the
10					master schedule.
11				b.	Provide an updated construction schedule to the CxA any time the schedule changes.
12				с.	Include the Cx activities in the contract.
13				d.	Furnish a copy of all submittals and shop drawings pertaining to the commissioned
14					systems for review concurrently with the Architect and Engineers.
15				e.	Furnish a copy of all construction meeting agendas and minutes to the CxA.
16				f.	In each purchase order or subcontract written, include requirements for submittal data,
17					O&M data, Cx tasks and training.
18				g.	GC will ensure that all Subs execute their Cx responsibilities according to the Contract
19				-	Documents and schedule.
20				h.	A representative from the GC and each sub associated with the Cx process shall attend the
21					Cx pre- construction meeting and the regular Cx meetings scheduled by the CxA to
22					facilitate the Cx process.
23				i.	Coordinate and execute the training of Owner personnel.
24				i.	Prepare O&M manuals, according to the Contract Documents, including clarifying and
25				,	updating the original sequences of operation to as-built conditions.
26				k.	Prepare and submit draft forms, including but not limited to start-up procedures. Testing
27					and Balancing (TAB) forms, calibration forms, etc. for review by the CxA before execution.
28				I.	Submit test reports to the CxA of all tests performed on components and equipment to be
20					commissioned that are not included as part of the Construction Checklist and SPT
20					nrocedures
21				m	Complete all construction checklist and functional performance test forms as required by
22					the Cy process
22				n	Support the CVA with verification of the completion of construction checklist and
22 24					support the CXA with vernication of the completion of construction checklist and functional performance tests as outlined in DAPT 2
54 25				•	Complete and increase all installations. Cortify that all components and systems are
35				0.	complete and inspect an installations. Certify that an components and systems are
30					operating as intended per Contract Documents.
3/				р.	Remedy all deficiencies immediately as they are identified throughout construction.
38				q.	Demonstrate functionality of all systems and equipment.
39				r.	Maintain an updated set of record drawings (on a daily basis) on the construction site.
40				s.	Provide support and instrumentation to verify TAB reports, start-up reports, calibration
41					reports, and any other report pertinent to the commissioned equipment and systems.
42				t.	Notify the CXA no less than 21 days before all testing, start-up, and training.
43				u.	Update the CxA on a weekly basis on the progress of the Cx activities.
44				v.	Submit trend data in electronic format or allow access to trending data by internet
45					connection as requested by the CxA.
46				w.	Install access points by every sensor such that the sensor can be calibrated without
47					removal (P/T plugs, plugged holes in ducts etc.).
48			2.	Warr	anty Period
49				a.	Execute seasonal or deferred functional performance testing, witnessed by the CxA,
50					according to the specifications.
51				b.	Correct deficiencies and make necessary adjustments to O&M manuals and record
52					drawings for applicable issues identified in any seasonal testing.
53		В.	Equipment Su	uppliers	5
54			1.	Provi	de all requested submittal data, including detailed start-up procedures and specific
55				respo	nsibilities of the Owner to keep warranties in force.
56			2.	Assist	in equipment testing per agreements with Subs.
57			3.	Inclue	de all special tools and instruments (only available from vendor, specific to a piece of
58				equip	ment) required for testing equipment according to these Contract Documents in the base
	WAR	IER PAR		ECREAT	ION CENTER

IVIAY 1	16, 2024	
		bid price to the Contractor, except for stand-alone data logging equipment that may be used by
		the CxA.4. Provide information requested by CxA regarding equipment sequence of operation and testing
		procedures. 5. Review test procedures for equipment installed by factory representatives.
.7	SYST	EMS TO BE COMMISSIONED
	A.	The entire Heating, Ventilation and Air Conditioning (HVAC) system (chillers, pumps, piping, and air distribution systems)
	В.	Building Automation System (BAS) for the HVAC system
	C.	Domestic Hot Water
	D.	Building envelope and rooting system
	F.	Solar electric (PV) System
ART	2 – PR	<u>ODUCTS</u>
2.1	TEST	INFORMATION
	Α.	All instruments needed to verify sensor readings, component performance, and system performance will be
		provided by GC and Subs and be available to the CxA. These instruments will not be beyond what the contractors need to complete the work specified in these construction documents. Any data logging equipment required in
	R	addition to the BAS will be provided by the CXA. All instruments shall be of sufficient quality and accuracy to test and/or measure system performance with the
	υ.	tolerances specified in the Contract Documents. Refer to specification section 23 05 93- Testing, Adjusting, and
		Balancing for required instrument tolerances.
PART	3 - EXE	CUTION
3.1	сом	MISSIONING TEAM
	Α.	The members of the commissioning team consist of the Commissioning Authority (CxA), the Owner's Project
		Manager (PM), the designated representative of the Owner's Construction Management team (CM), the General
		Contractor, the TAB Contractor, the Controls Contractor, any other installing subcontractors or suppliers of
		equipment.
	В.	Each Cx Team member shall designate one person who is responsible for coordinating the commissioning efforts
		with the CxA.
3.2	SCHE	DUI ING AND MEETINGS
	A.	Scheduling. The CxA will work with the other members of the Cx Team according to established protocols to
		schedule the Cx activities. The CxA will provide sufficient notice to the Cx Team for scheduling Cx activities. The
		GC will integrate all Cx activities into the master schedule. All parties will address scheduling problems and make
	_	necessary notifications in a timely manner in order to expedite the Cx process.
	В.	The CxA will provide the initial schedule of primary Cx events at the Cx pre-construction meeting. The Cx Plan
		provides a format for this schedule. As construction progresses more detailed schedules are developed by the
	C.	Pre-Construction Meeting. Within 60 days of selection of the GC, the CxA will schedule, plan, and conduct a Cx
	с.	pre-construction meeting with the entire Cx team in attendance. Meeting minutes will be distributed to all
		parties by the CxA. Information gathered from this meeting will allow the CxA to revise the Cx Plan which will
		also be distributed to all parties.
	D.	Meetings. The Cx meetings will be scheduled approximately once a month during construction. These meetings
		will be scheduled directly before or after the regular construction meetings if practical. These meetings will cover
		coordination, denciency resolution and planning issues with particular SUBS. The CXA Will plan these meetings and will minimize unnecessary time being spent by Subs
		and with minimize diffecessary time being spent by Subs
.3	REPC	RTING
	Α.	The CxA will provide regular reports to the Owner as construction and Cx progresses. Standard forms are
		provided and referenced in the Cx Plan.

1		В.	The CxA will	regularly communicate with all members of the Cx team, keeping them apprised of Cx progress and
2		C	scheduling c	hanges through memos, progress reports, etc.
3		C.	Testing or re	view approvals and non-conformance and deficiency reports are made regularly with the review and
4			testing as de	scribed in later sections.
6	3.4	RECO		S
7		A.	The CxA will	verify that the record drawings are updated throughout the construction. If a discrepancy is found
8			between the	e record drawings and the installations, the CxA will notify the GC immediately. It is the GC and
9			subcontracto	ors responsibility to then inspect the installations and immediately and completely update the record
10			drawings suc	ch that they accurately reflect the installation.
11			U	
12	3.5	CON	STRUCTION CO	MMISSIONING PROCEDURES
13		Α.	The followin	g procedures apply to all equipment to be commissioned.
14		В.	<u>General.</u> Cor	struction checklists are important to ensure that the equipment and systems are hooked up and
15			operational.	It ensures that system performance testing (in-depth system checkout) may proceed without
16			unnecessary	delays. Each piece of equipment receives full checkout. No sampling strategies are used. All
17			construction	checklists for a given system must be successfully completed prior to formal system performance
18			testing of eq	uipment or subsystems of the given system.
19		C.	Construction	i Checklists.
20			1.	I ne primary purpose of the construction checklists is to provide the individual workers with the
21				key criteria for a successful installation. The secondary purpose is to track the progress of the
22			2	delivery and installation.
23			2.	The CXA will develop construction checklists for all commissioned equipment and distribute these
24				to the responsible contractor. The GC and Subs will review the construction checklists for each
25				equipment type and provide comments to the CXA. The CXA will then print and distribute the
20			2	The GC and Subs are responsible for all requirements in the specification, not only the
27 78			5.	requirements listed on the checklists
20			1	The checklists answer format will be to circle ves /no or provide a brief answer such as providing
20			4.	the model or serial numbers
21			5	These checklists are provided by the CVA to the GC. The GC determines which trade is responsible
32			5.	for executing and documenting each of the line item tasks and notes that trade on the form. Each
33				form may have more than one trade responsible for its execution
34			6.	The construction checklists shall be completed as delivery is completed and the installation
35			0.	progresses.
36			7.	Only individuals who have direct knowledge and witnessed that a line item task on the
37				construction checklist was actually performed shall initial or check that item off. It is not
38				acceptable for supervisors without direct knowledge or who have not witnessed the line item task
39				on the construction checklist to fill out these forms.
40			8.	Any negative response shall immediately be brought to the attention of the CxA. All negative
41				replies shall be explained in detail on the construction checklist.
42			9.	The GC and Subs are responsible for recording the completion of the checklists. Checklists shall be
43				submitted electronically to SharePoint in .pdf format in separate files by Division. Each file shall be
44				bookmarked by checklist tag.
45			10.	Non-itemized installations such as wiring, ductwork, piping etc. will not have checklists to be
46				completed, but the GC and Subs will be provided the key criteria for successful installation.
47			11.	The CxA will verify the construction checklist completion by a sampling of the delivered and
48				installed equipment. The sampling process will be described in the Cx Plan.
49		D.	Sensor Calib	ration. Calibration of all sensors shall be included as part of the construction checklists performed by
50			the Contract	ors. Calibration information is provided in specification Section 23 09 23 - Direct Digital Control
51		_	System for H	VAC
52		Ε.	Deticiencies,	Non-Conformance and Approval in Checklists and Startup.
53			1.	I he Subs shall clearly list any outstanding items of the construction checklist that were not
54				completed successfully, at the bottom of the procedures form or on an attached sheet. The
55				procedures form and any outstanding deficiencies are provided to the CXA within two days of task
50			2	completion.
5/ 50			۷.	the CXA reviews the report and submits either a non-compliance report or an approval form to
ō				the sub of Civil the CXA shall work with the subs and vendors to correct deficiencies of
58	WARI	NER PAR		the Sub or CM. The CxA shall work with the Subs and vendors to correct deficiencies or RECREATION CENTER

1		uncompleted items. The CxA will involve the CM and ot	hers as necessary. The installing Subs or
2		vendors shall correct all areas that are deficient or incor	nplete in the checklists and tests in a
3		timely manner, and shall notify the CxA as soon as outst	anding items have been corrected and
4		include a Statement of Correction on the original non- c	ompliance report. When satisfactorily
5		completed, the CxA recommends approval of the comp	letion of the checklists to the CM using a
6		standard form.	
7		3. Items left incomplete, which later cause deficiencies or	delays during functional testing may result
8		in back charges to the responsible party.	
9	F.	System Performance Tests (SPT). SPTs shall be performed to demonstr	ate that each system is operating
10		according to the documented OPR and Contract Documents. System te	esting differs to the tests required in the
11		Construction Checklist in that they facilitate bringing all the individual	components together to verify that they
12		operate collectively on a system level to provide the required design co	onditions.
13		 Development of Test Procedures. The CxA shall prepare 	the SPT forms and procedures in
14		accordance with the criteria defined in the Cx Plan. The	GC and Subs shall assist the CxA in the
15		preparation of these procedures by answering queries a	and forwarding site-specific information. A
16		sample System Performance Test form is provided at th	e end of this specification section.
17		Participation: The GC and the Subs are responsible for t	esting all systems to be commissioned
18		such that they function as described in the contract doc	uments. The CxA will verify the
19		performance of the systems. The CxA will direct, witnes	s and document the SPT verification and
20		GC and Subs will execute the verification tests.	
21	G.	Problem Solving. The CxA will recommend solutions to problems found	d, however the burden of responsibility to
22		solve, correct and retest problems is with the GC, Subs and A/E.	
23	H.	<u>Seasonal Testing</u> . During the warranty period, seasonal testing (tests d	elayed until weather conditions are closer
24		to the system's design) shall be completed as part of this contract. The	CxA shall coordinate this activity. Tests
25		will be executed, documented and deficiencies corrected by the appro	priate Subs, with facilities staff and the
26		CxA witnessing. Any final adjustments to the O&M manuals and record	I documents due to the testing will be
27		made.	
28	١.	<u>Unforeseen Deferred Tests.</u> If any check or test cannot be completed d	lue to the building structure, required
29		occupancy condition or other deficiency, execution of checklists and fu	inctional testing may be delayed upon
30		approval of the PM. These tests will be conducted in the same manner	as the seasonal tests.
31			
22	26 65		
32	3.6 SEN	NSOR AND ACTUATOR CALIBRATION	avide eacher diavide and process
32 33	3.6 SEN A.	NSOR AND ACTUATOR CALIBRATION Calibrate all field-installed temperature, relative humidity, carbon mor	noxide, carbon dioxide, and pressure
32 33 34 25	3.6 SEN A.	ENSOR AND ACTUATOR CALIBRATION Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece	noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors
32 33 34 35	3.6 SEM A.	ENSOR AND ACTUATOR CALIBRATION Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below: alternate methods may	noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated.
32 33 34 35 36 37	3.6 SEN A. B.	ENSOR AND ACTUATOR CALIBRATION Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met	noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner bods used on the relevant Construction
32 33 34 35 36 37 28	3.6 SEN A. B.	ENSOR AND ACTUATOR CALIBRATION Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklict or other cuitable forms, documenting initial intermediate an	noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results.
32 33 34 35 36 37 38 29	3.6 SEN A. B.	ENSOR AND ACTUATOR CALIBRATION Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Soncorc:	noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results.
32 33 34 35 36 37 38 39 40	3.6 SEN A. B. C.	ENSOR AND ACTUATOR CALIBRATION Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors:	noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results.
32 33 34 35 36 37 38 39 40	3.6 SEN A. B. C.	 ENSOR AND ACTUATOR CALIBRATION Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors: Verify that sensor location is appropriate and away from 2 Verify that sensor with chielded cable are grounded on 	noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results.
32 33 34 35 36 37 38 39 40 41	3.6 SEN А. В. С.	 ENSOR AND ACTUATOR CALIBRATION Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors: Verify that sensor location is appropriate and away from 2. Verify that sensors with shielded cable are grounded on 2. 	noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results. In potential causes of erratic operation. ly at one end.
32 33 34 35 36 37 38 39 40 41 42 42	3.6 SEN А. В. С.	 ENSOR AND ACTUATOR CALIBRATION Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors: Verify that sensor location is appropriate and away from 2. Verify that sensors with shielded cable are grounded on 3. For sensor pairs that are used to determine a temperature temperature make sure they are reading within 0.2 dage 	noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results. n potential causes of erratic operation. ly at one end. ure or pressure difference, for
32 33 34 35 36 37 38 39 40 41 42 43	3.6 SEN А. В. С.	 ENSOR AND ACTUATOR CALIBRATION Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors: Verify that sensor location is appropriate and away from 2. Verify that sensors with shielded cable are grounded on 3. For sensor pairs that are used to determine a temperature temperature make sure they are reading within 0.2 deg proceure, within tolerance equal to 2 percent of the reading 	noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results. n potential causes of erratic operation. ly at one end. ure or pressure difference, for ree F (0.1 degree C) of each other, and for ding, of each other.
32 33 34 35 36 37 38 39 40 41 42 43 44	3.6 SEN А. В. С.	 Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors: Verify that sensor location is appropriate and away from 2. Verify that sensors with shielded cable are grounded on 3. For sensor pairs that are used to determine a temperatur temperature make sure they are reading within 0.2 deg pressure, within tolerance equal to 2 percent of the reading temperature may be tighter. 	noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results. In potential causes of erratic operation. Ily at one end. ure or pressure difference, for ree F (0.1 degree C) of each other, and for ding, of each other.
32 33 34 35 36 37 38 39 40 41 42 43 44 45 46	3.6 SEP А. В. С.	 Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors: Verify that sensor location is appropriate and away from 2. Verify that sensors with shielded cable are grounded on 3. For sensor pairs that are used to determine a temperature make sure they are reading within 0.2 deg pressure, within tolerance equal to 2 percent of the rea Tolerances for critical applications may be tighter. 	noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results. In potential causes of erratic operation. If a one end. ure or pressure difference, for ree F (0.1 degree C) of each other, and for ding, of each other.
32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47	3.6 SEP А. В. С. D.	 Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors: Verify that sensor location is appropriate and away from 2. Verify that sensors with shielded cable are grounded on 3. For sensor pairs that are used to determine a temperature make sure they are reading within 0.2 deg pressure, within tolerance equal to 2 percent of the rea Tolerances for critical applications may be tighter. 	noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results. In potential causes of erratic operation. ly at one end. ure or pressure difference, for ree F (0.1 degree C) of each other, and for ding, of each other.
32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48	3.6 SEP А. В. С. D.	 Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors: Verify that sensor location is appropriate and away from 2. Verify that sensors with shielded cable are grounded on 3. For sensor pairs that are used to determine a temperature make sure they are reading within 0.2 deg pressure, within tolerance equal to 2 percent of the rea Tolerances for critical applications may be tighter. Sensors without Transmitters - Standard Application: Make a reading with a calibrated test instrument within Verify that be sensor reading, via the permanent them 	noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results. In potential causes of erratic operation. If y at one end. ure or pressure difference, for ree F (0.1 degree C) of each other, and for ding, of each other.
32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49	3.6 SEP А. В. С. D.	 Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors: Verify that sensor location is appropriate and away from 2. Verify that sensors with shielded cable are grounded on 3. For sensor pairs that are used to determine a temperature make sure they are reading within 0.2 deg pressure, within tolerance equal to 2 percent of the rea Tolerances for critical applications: Make a reading with a calibrated test instrument within 2. Verify that the sensor reading, via the permanent therm protom is within the colorances in the table below of the 	noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results. In potential causes of erratic operation. If y at one end. ure or pressure difference, for ree F (0.1 degree C) of each other, and for ding, of each other.
32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50	3.6 SE Р А. В. С. D.	 Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors: Verify that sensor location is appropriate and away from Verify that sensor swith shielded cable are grounded on For sensor pairs that are used to determine a temperature make sure they are reading within 0.2 deg pressure, within tolerance equal to 2 percent of the rea Tolerances for critical applications: Make a reading with a calibrated test instrument within Verify that the sensor reading, via the permanent therm system, is within the tolerances in the table below of th 	noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results. In potential causes of erratic operation. ly at one end. ure or pressure difference, for ree F (0.1 degree C) of each other, and for ding, of each other.
32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51	3.6 SEN А. В. С. D.	 Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors: Verify that sensor location is appropriate and away from Verify that sensor location is appropriate and away from For sensor pairs that are used to determine a temperatur temperature make sure they are reading within 0.2 deg pressure, within tolerance equal to 2 percent of the rea Tolerances for critical applications: Make a reading with a calibrated test instrument within Verify that the sensor reading, via the permanent therm system, is within the tolerances in the table below of th 	 noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results. n potential causes of erratic operation. ly at one end. ure or pressure difference, for ree F (0.1 degree C) of each other, and for ding, of each other. 6 inches (150 mm) of the site sensor. nostat, gage or building automation e instrument-measured value.
32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52	3.6 SEN А. В. С. D. Е.	 Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors: Verify that sensor location is appropriate and away from Verify that sensor location is appropriate and away from For sensor pairs that are used to determine a temperatur temperature make sure they are reading within 0.2 deg pressure, within tolerance equal to 2 percent of the rea Tolerances for critical applications: Make a reading with a calibrated test instrument therm system, is within the tolerances in the table below of th If not, install offset, calibrate or replace sensor. 	 noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results. n potential causes of erratic operation. ly at one end. ure or pressure difference, for ree F (0.1 degree C) of each other, and for ding, of each other. 6 inches (150 mm) of the site sensor. nostat, gage or building automation e instrument-measured value.
32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53	 3.6 SEN A. В. С. D. Е. 	 Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors: Verify that sensor location is appropriate and away from Verify that sensor location is appropriate and away from For sensor pairs that are used to determine a temperatur temperature make sure they are reading within 0.2 deg pressure, within tolerance equal to 2 percent of the rea Tolerances for critical applications may be tighter. Sensors without Transmitters - Standard Application: Make a reading with a calibrated test instrument therm system, is within the tolerances in the table below of th If not, install offset, calibrate or replace sensor. 	 noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results. n potential causes of erratic operation. ly at one end. ure or pressure difference, for ree F (0.1 degree C) of each other, and for ding, of each other. 6 inches (150 mm) of the site sensor. nostat, gage or building automation e instrument-measured value.
32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54	 3.6 SEN A. В. С. D. Е. 	 Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors: Verify that sensor location is appropriate and away from Verify that sensor location is appropriate and away from Verify that sensor swith shielded cable are grounded on For sensor pairs that are used to determine a temperatur temperature make sure they are reading within 0.2 deg pressure, within tolerance equal to 2 percent of the rea Tolerances for critical applications may be tighter. Sensors without Transmitters - Standard Application: Make a reading with a calibrated test instrument therm system, is within the tolerances in the table below of th If not, install offset, calibrate or replace sensor. Sensors with Transmitters - Standard Application. Disconnect sensor. Connect a signal generator in place of sensor. 	 noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results. n potential causes of erratic operation. ly at one end. ure or pressure difference, for ree F (0.1 degree C) of each other, and for ding, of each other. 6 inches (150 mm) of the site sensor. nostat, gage or building automation e instrument-measured value.
32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55	 3.6 SEN A. В. С. D. Е. 	 Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors: Verify that sensor location is appropriate and away from Verify that sensor location is appropriate and away from Verify that sensor swith shielded cable are grounded on For sensor pairs that are used to determine a temperatur temperature make sure they are reading within 0.2 deg pressure, within tolerance equal to 2 percent of the rea Tolerances for critical applications may be tighter. Sensors without Transmitters - Standard Application: Make a reading with a calibrated test instrument therm system, is within the tolerances in the table below of th If not, install offset, calibrate or replace sensor. Sensors with Transmitters - Standard Application. Disconnect sensor. Connect a signal generator in place of sensor. Connect ammeter in series between transmitter and bu Using manufacturer's resistance-temperature data simplication. 	 noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results. n potential causes of erratic operation. ly at one end. ure or pressure difference, for ree F (0.1 degree C) of each other, and for ding, of each other. 6 inches (150 mm) of the site sensor. nostat, gage or building automation e instrument-measured value.
32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55 55 55	 3.6 SEN A. В. С. D. Е. 	 INSOR AND ACTUATOR CALIBRATION Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors: Verify that sensor location is appropriate and away from 2. Verify that sensor location is appropriate and away from 2. Verify that sensor swith shielded cable are grounded on 3. For sensor pairs that are used to determine a temperatur temperature make sure they are reading within 0.2 deg pressure, within tolerance equal to 2 percent of the rea Tolerances for critical applications may be tighter. Sensors without Transmitters - Standard Application: Make a reading with a calibrated test instrument within Verify that the sensor reading, via the permanent therm system, is within the tolerances in the table below of th If not, install offset, calibrate or replace sensor. Sensors with Transmitters - Standard Application. Disconnect asignal generator in place of sensor. Connect a signal generator in place of sensor. Make transmitter perture in series between transmitter and bu Using manufacturer's resistance-temperature data, sim 	 noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results. n potential causes of erratic operation. ly at one end. ure or pressure difference, for ree F (0.1 degree C) of each other, and for ding, of each other. 6 inches (150 mm) of the site sensor. nostat, gage or building automation e instrument-measured value. ilding automation system control panel. ulate minimum desired temperature. d by the ammeter
32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 951 52 53 54 55 55 57	 3.6 SEP A. В. С. D. Е. 	 INSOR AND ACTUATOR CALIBRATION Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors: Verify that sensor location is appropriate and away from 2. Verify that sensor location is appropriate and away from 2. Verify that sensor swith shielded cable are grounded on 3. For sensor pairs that are used to determine a temperatur temperature make sure they are reading within 0.2 deg pressure, within tolerance equal to 2 percent of the rea Tolerances for critical applications may be tighter. Sensors without Transmitters - Standard Application: Make a reading with a calibrated test instrument within Verify that the sensor reading, via the permanent therm system, is within the tolerances in the table below of th If not, install offset, calibrate or replace sensor. Sensors with Transmitters - Standard Application. Disconnect asignal generator in place of sensor. Connect a signal generator in place of sensor. Adjust transmitter potentiometer zero until 4 mA is rear 	 noxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results. n potential causes of erratic operation. ly at one end. ure or pressure difference, for ree F (0.1 degree C) of each other, and for ding, of each other. 6 inches (150 mm) of the site sensor. nostat, gage or building automation e instrument-measured value. ilding automation system control panel. ulate minimum desired temperature. d by the ammeter. to the potentiometer span or maximum
32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 9 51 52 53 54 55 57 58	 3.6 SEP A. В. С. D. Е. 	 INSOR AND ACTUATOR CALIBRATION Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors: Verify that sensor location is appropriate and away from Verify that sensors with shielded cable are grounded on For sensor pairs that are used to determine a temperatu temperature make sure they are reading within 0.2 deg pressure, within tolerance equal to 2 percent of the rea Tolerances for critical applications may be tighter. Sensors without Transmitters - Standard Application: Make a reading with a calibrated test instrument therm system, is within the tolerances in the table below of th If not, install offset, calibrate or replace sensor. Sensors with Transmitters - Standard Application. Disconnect sensor. Connect a signal generator in place of sensor. Connect a signal generator in place of sensor. Adjust transmitter potentiometer zero until 4 mA is reading. 	 hoxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results. n potential causes of erratic operation. ly at one end. ure or pressure difference, for ree F (0.1 degree C) of each other, and for ding, of each other. 6 inches (150 mm) of the site sensor. hostat, gage or building automation e instrument-measured value. ilding automation system control panel. ulate minimum desired temperature. d by the ammeter. to the potentiometer span or maximum
32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 9 50 51 52 53 54 55 55 58	3.6 SEN А. В. С. D. Е.	 Calibrate all field-installed temperature, relative humidity, carbon mor sensors and gages, and all actuators (dampers and valves) on this piece installed in the unit at the factory with calibration certification provide Calibrate using the methods described below; alternate methods may beforehand. See PART 2 for test instrument requirements. Record met Checklist or other suitable forms, documenting initial, intermediate an All Sensors: Verify that sensor location is appropriate and away from Verify that sensor location is appropriate and away from Verify that sensor pairs that are used to determine a temperatur temperature make sure they are reading within 0.2 deg pressure, within tolerance equal to 2 percent of the rea Tolerances for critical applications may be tighter. Sensors without Transmitters - Standard Application: Make a reading with a calibrate dest instrument therm system, is within the tolerances in the table below of th If not, install offset, calibrate or replace sensor. Sensors with Transmitters - Standard Application. Disconnect sensor. Connect a signal generator in place of sensor. Connect a signal generator in place of sensor. Adjust transmitter potentiometer zero until 4 mA is reading and verify at the building automation system. 	 hoxide, carbon dioxide, and pressure e of equipment shall be calibrated. Sensors d need not be field calibrated. be used, if approved by Owner hods used on the relevant Construction d final results. n potential causes of erratic operation. ly at one end. ure or pressure difference, for ree F (0.1 degree C) of each other, and for ding, of each other. 6 inches (150 mm) of the site sensor. hostat, gage or building automation e instrument-measured value. ilding automation system control panel. ulate minimum desired temperature. d by the ammeter. to the potentiometer span or maximum

1			7	Record all values and recalibrate controller as necessary to conform with specified control ramps
1 2			7.	reset schedules, propertianal relationship, reset relationship and D/L reaction
2			0	Peser schedules, proportional relationship, reset relationship and P/Treaction.
3			ð. 0	Reconnect sensor. Make a reading with a calibrated test instrument within 6 inches (150 mm) of the site sensor.
4			9.	Make a reduing with a calibrated test instrument within 6 inches (150 mm) of the site sensor.
5			10.	verify that the sensor reading, via the permanent thermostat, gage or building automation
6				system, is within the tolerances in the table below of the instrument-measured value.
/			11.	If not, replace sensor and repeat.
8			12.	For pressure sensors, perform a similar process with a suitable signal generator.
9		F.	Sensor Tolera	ances for Standard Applications: Plus/minus the following maximums:
10			1.	Watthour, Voltage, Amperage: 1 percent of design.
11			2.	Pressure, Air, Water, Gas: 3 percent of design.
12			3.	Air Temperatures (Outside Air, Space Air, and Duct Air): 0.4 degrees F (0.2 degree C).
13			4.	Relative Humidity: 4 percent of design.
14			5.	Barometric Pressure: 0.1 inch of Hg (340 Pa).
15			6.	Flow Rate, Air: 10 percent of design.
16			7.	Flow Rate, Water: 4 percent of design.
17			8.	Flow Rate, Steam: 3 percent of design.
18			9.	AHU Wet Bulb and Dew Point: 2.0 degrees F (1.1 degrees C).
19			10.	Hot Water Coil and Boiler Water Temperature: 1.5 degrees F (0.8 degrees C).
20			11.	Cooling Coil. Chilled and Condenser Water Temperatures: 0.4 degrees E (0.2 degree C).
21			12.	Combustion Flue Temperature: 5.0 degrees E (2.8 degrees C).
22			13	Ovugen and CO2 Monitors: O 1 nercentage points
22			17.	CO Monitor: 0.01 percentage points
23			14.	Natural Gas and Oil Flow Rate: 1 percent of design
24 25		c	LJ. Critical Appli	rational Gas and Oil Flow Rate. I percent of design.
25		G.		cations. For some applications more rigorous calibration techniques may be required for selected
20			Sensors. Des	The any such methods used on an attached sheet.
27		н.	valve/Dampe	er stroke setup and Check:
28			1.	For all valve/damper actuator positions checked, verify the actual position against the control
29			_	system readout.
30			2.	Set pump/fan to normal operating mode.
31			3.	Command valve/damper closed; visually verify that valve/damper is closed and adjust output zero
32				signal as required.
33			4.	Command valve/damper to open; verify position is full open and adjust output signal as required.
34			5.	Command valve/damper to a few intermediate positions.
35			6.	If actual valve/damper position does not reasonably correspond, replace actuator
36		١.	Isolation Valv	ve or System Valve Leak Check: For valves not associated with coils.
37			1.	With full pressure in the system, command valve closed.
38			2.	Use an ultra-sonic flow meter to detect flow or leakage.
39				
40	3.7	NON-	CONFORMAN	CE
41		Α.	All deficienci	es or non-conformance issues shall be noted and reported by the GC to the CM on a standard non-
42			compliance f	orm.
43		В.	Corrections o	of minor deficiencies identified may be made during the tests at the discretion of the CxA. In such
44			cases the def	ficiency and resolution will be documented on the procedure form.
45		C.	Every effort v	will be made to expedite the testing process and minimize unnecessary delays, while not
46		с.	compromisin	in the integrity of the procedures. However, the CyA will not be pressured into overlooking deficient
40			work or loos	ening accentance criteria to satisfy scheduling or cost issues unless there is an overriding reason to
-, 18			do so at the	request of the CM and the Owner
10		D	Ac tosts prog	request of the civitation of the CVA discusses the issue with the executing contractor
+9 50		υ.	As lesis prog	When there is no dispute on the deficiency and the Cub accents responsibility to correct it:
50			1.	when there is no dispute on the deficiency and the Sub accepts responsibility to correct it.
5T 2T				a. The CXA documents the denciency and the Sub s response and intentions and they go on
52				to another test of sequence. After the day's work, the CXA submits the non-compliance
こ ろ				reports to the Civitor signature, if required. A copy is provided to the Sub and CXA. The
54				Sub corrects the deficiency, signs the statement of correction at the bottom of the non-
55				compliance form certifying that the equipment is ready to be retested and sends it back to
56				the CxA.
57				b. The CxA reschedules the test and the test is repeated.
58			2. I	t there is a dispute about a deficiency, regarding whether it is a deficiency or who is responsible:
-	WARN	ER PARK		RECREATION CENTER

1		a.	The deficiency shall be documented on the non-compliance form with the Sub's response
2			and a copy given to the CM and to the Sub representative assumed to be responsible.
3		b.	Resolutions are made at the lowest management level possible. Other parties are brought
4			into the discussions as needed. Final interpretive authority is with the A/E. Final
5			acceptance authority is with the Project Manager.
6		С.	The CxA documents the resolution process.
/		d.	Once the interpretation and resolution have been decided, the appropriate party corrects
8			the deficiency, signs the statement of correction on the non-compliance form and provides
9 10			It to the CXA. The CXA rescribedules the test and the test is repeated until satisfactory
10		2 Cost o	f Potosting
12		5. COSLO	The cost incurred by the Subs to retest a construction checklist item or functional test if
12		a.	they are responsible for the deficiency shall be theirs. If they are not responsible any cost
14			recovery for retesting costs shall be negotiated with the GC
15		b	For a deficiency identified not related to any construction checklist or start-up fault, the
16			following shall apply: The CxA and CM will direct the retesting of the equipment once at no
17			"charge" to the GC for their time. However, the CxA's and CM's time for a second retest
18			will be charged to the GC, who may choose to recover costs from the responsible Sub.
19		с.	The time for the CxA and CM to direct any retesting required because a specific
20			construction checklist or start-up test item, reported to have been successfully completed,
21			but determined during functional testing to be faulty, will be backcharged to the GC, who
22			may choose to recover costs from the party responsible for executing the faulty
23			installation or test.
24		d.	The Contractor shall respond in writing to the CxA and CM at least as often as Cx meetings
25			are being scheduled concerning the status of each apparent outstanding discrepancy
26			identified during Cx. Discussion shall cover explanations of any disagreements and
27			proposals for their resolution.
28		e.	The CxA retains the original non-conformance forms until the end of the project.
29		f.	Failure Due to Manufacturer Defect. If 10%, or three, whichever is greater, of identical
30			pieces (size alone does not constitute a difference) of equipment fail to perform to the
31			Contract Documents (mechanically or substantively) due to manufacturing defect, not
32			allowing it to meet its submitted performance spec, all identical units may be considered
33			the following:
34 25		a	the following: Within one week of notification from the CM or DM, the Contractor or manufacturor's
35		g.	within one week of notification from the CW of PW, the Contractor of manufacturers
20 27			findings shall be provided to the CM or PM within two works of the original potice
32		h	Within two weeks of the original potification the Contractor or manufacturer shall provide
30			a signed and dated written explanation of the problem cause of failures etc. and all
40			proposed solutions which shall include full equipment submittals. The proposed solutions
40			shall not significantly exceed the specification requirements of the original installation. The
42			CM or PM will determine whether a replacement of all identical units or a repair is
43			acceptable.
44		i.	Two examples of the proposed solution will be installed by the Contractor and the CM will
45			be allowed to test the installations for up to one week, upon which the CM or PM will
46			decide whether to accept the solution.
47		ј.	Upon acceptance, the Contractor and/or manufacturer shall replace or repair all identical
48			items, at their expense and extend the warranty accordingly, if the original equipment
49			warranty had begun. The replacement/repair work shall proceed with reasonable speed
50			beginning within one week from when parts can be obtained.
51	E.	Approval. The CxA no	tes each satisfactorily demonstrated function on the test form. Formal approval of the
52		functional test is mad	le later after review by the CxA and by the CM, if necessary. The CxA recommends
53		acceptance of each te	est to the CM using a standard form. The CM gives final approval on each test using the
54		same form, providing	a signed copy to the CxA and the Contractor.
55			
50			END OF SECTION
57			

			SECTION 01 95 00			
			MEASUREMENT AND VERIFICATION			
PART	1-G	ENERAL				
	1.1	SUMMARY				
	1.2	DEFINITIONS				
	1.3	MECHANICAL	CONTRACTOR RESPONSIBILITIES			
	1.4	ELECTRICAL CO	ONTRACTOR RESPONSIBILITIES			
	1.5	CONTROLS CO	INTRACTOR RESPONSIBILITIES			
	1.6	M&V PROVIDE	ERS RESPONSIBILITIES			
PART	2 – P	RODUCTS – THIS	S SECTION NOT USED			
	2.1	METERS AND S	SUB-METERS			
PART	3 - EX	(ECUTION				
	3.1	METER				
	3.2	NATURAL GAS)			
	3.3	TEMPORARY N	MONITORING			
	3.4	DDC TRENDS				
PART	[1-G	ENERAL				
11	SUI	ΜΜΔΒΥ				
	A	Purnose Th	his section includes general requirements that apply to implementation of measurement and			
		verification				
	В.	RELATED W	/ORK AND REOUIREMENTS			
	5.	1.	Section 01 31 13 Project Coordination			
		2.	Section 01 31 19 Project Meetings			
		3.	Section 01 31 23 Project Management Web Site			
		4.	Section 01 91 00 Commissioning			
		5	Section 23.09.00 Instrumentation and Control for HVAC			
		6	Section 23 09 23 Direct Digital Control (DDC) System for HVAC			
		7.	Section 23 09 93 Sequence of Operations for HVAC DDC			
		8.	Section 26 24 13 Switchboards			
		9.	Section 26 24 16 Panelboards			
1 7	БГГ					
1.2			Duilding Automation System			
	А. Р	BAS -	Building Automation System			
	В.	DHW -	Domestic Hot Water			
	C.	IVI&V -	Neasurement and verification			
	D.	KVV -	Electric power read from utility meter			
	E.	KWN -	Electric energy consumption read from utility meter			
	F.	Plug Loads -	- Electric power and consumption from Wall receptacles			
1.3	ME	CHANICAL CON	ITRACTOR RESPONSIBILITIES			
	Α.	Contractor s	shall assign representatives with expertise and authority to act on its behalf and shall schedule the			
		to participat	te in and perform M&V activities including, but not limited to, the following:			
		1.	Follow activities identified in the M&V Plan.			
		2.	Coordinate connection of gas and DHW monitoring equipment with BAS.			
		3.	Cooperate with the M&V Provider and Controls Contractor for resolution of issues related to da			
			collection.			
		4.	Attend team meetings during construction and post-construction M&V period (1 year).			
	ELF					
1 /		LECTRICAL CONTRACTOR RESPONSIBILITIES				
1.4	Δ	to participat	shall assign representatives with expertise and authority to act on its benair and shall schedule the			
1.4	71.	to participat	te in and perform M&V activities including, but not limited to, the following:			
1.4	71.	4				
1.4	7.	1.	Consultation of a later in a state in a state in a state in a state in the BAC			
1.4	<i>,</i>	1. 2.	Coordinate connection of electrical monitoring equipment with BAS			
1.4	7.	1. 2. 3.	Coordinate connection of electrical monitoring equipment with BAS Cooperate with the M&V Provider and Controls Contractor for resolution of issues related to da			

 CONTROLS CONTRACTOR RESPONSIBILITIES Contractor shall assign representatives with expertise and authority to act on its behalf and shall schedule the to participate in and perform MAX valicities including, but not limited to, the following:		4.	Attend team meetings during construction and	d post-construction M&V period (1 year).			
 A. Contractor shall assign representatives with expertise and authority to act on its behalf and shall schedule the to participate in and perform M&V activities including, but not limited to, the following: Follow activities identified in the M&V Plan. Coordinate connection of electrical, gas, and DIW monitoring equipment with BAS Cooperate with the M&V Provide TheAnnical Contractor and Electrical Contractor for resolution of successful to establishing connection between BAS and monitoring meters and equipmer Attend team meetings during construction and post-construction M&V period (1 year). May provide The Section MAV period (1 year). Provide Tesponsibilities include: Orgonize and lead the M&V team. Provide Tesponsibilities include: Cooperate with the Machanical Contractor, Electrical Contractor, and Controls Contractor for resolution of Issues related to establishing connection between BAS and monitoring meters and equipment. Cooperate with the Machanical Contractor, Electrical Contractor, and Controls Contractor for resolution of Issues related to establishing connection between BAS and monitoring meters and equipment. Provide not M&V report at 1 year post construction. METER AND SUB-METES Monitoring meters and sub-meters, both gas and electric, to have the ability to connect to the BAS and provid data to BAS at a minimum of 15 minute intervals. It is acceptable to use the utility for this purpose if allowable utility company. METER Monitoring meters and sub-meters, both gas and electric, to have the ability wand kWh data. Data is to collected 15 minute intervals. Strange of at least 3 months of	1.5	CONTROLS CONTR	ACTOR RESPONSIBILITIES				
 to participate in and perform M&V activities including, but not limited to, the following: Follow activities identified in the M&V Pan. Cooperate with the M&V Provider Mechanical Contractor and Electrical Contractor for resolution of issues related to estabilishing connection between BAS and monitoring meters and equipment Attend team meetings during construction and post-construction M&V period (1 year). M&V PROVIDER RESPONSIBILITIES Organize and lead the M&V team. Provides responsibilities include: 	-	A Contractor shall assign representatives with expertise and authority to act on its behalf and shall schedule them					
 Follow activities identified in the M&V Plan. Coordinate connection of electrical, gas, and DHW monitoring equipment with BAS Cooperate with the M&V Provide Nechanical Contractor and Electrical Contractor for resolution of issues related to establishing connection between BAS and monitoring meters and equipmer Attend team meetings during construction and post-construction M&V period (1 year). M&V PROVIDERS RESPONSIBILITIES Providers responsibilities include: Organize and lead the M&V team. Provide M&V meetings as needed. Cooperate with the McChanical Contractor, Electrical Contractor, and Controls Contractor for resolution of issues related to establishing connection between BAS and monitoring meters and equipment. Provide an M&V report at 1 year post construction. ART 2 - PRODUCTS - THIS SECTION NOT USED MCTERS AND SUB-METERS Monitoring meters and sub-meters, both gas and electric, to have the ability to connect to the BAS and provid data to BAS at a minimum of 15 minute intervals. It is acceptable to use the utility for this purpose if allowable utility company. MRTER Provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building utility meter serving the HVAC, lighting, and plug loads and provide the data input to healuiding thromation synet: an a bub-meters. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 3 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate the work. NATURAL GAS		to participate in and perform M&V activities including, but not limited to, the following:					
 Coordinate connection of electrical, gas, and DHW monitoring equipment with BAS Cooperate with the M&V Provider Mechanical Contractor and Electrical Contractor for resolution of issues related to establishing connection between BAS and monitoring meters and equipment Attend team meetings during construction and post-construction M&V period (1 year). MeV PROVIDERS RESPONSIBILITIES Organize and lead the M&V team. Provider responsibilities include: Provide M&V plan. Convene M&V meetings as needed. Convene M&V meetings as needed. Convene M&V report at 1 year post construction. CMETERS AND SUB-METERS A Monitoring meters and sub-meters, both gas and electric, to have the ability to connect to the BAS and monitoring meters and equipment. Convide an M&V report at 1 year post construction. CMETER AND SUB-METERS A Monitoring meters and sub-meters, both gas and electric, to have the ability to connect to the BAS and provide dra ta bas as a minimum of 15 minute intervals. It is acceptable to use the utility for this purpose if allowable utility company. CMET 3 - EXECUTION CANT 4 - EXECUTION CANT 4 - EXECUTION CANT 4 - EXECUTION CANT 5 - EXECUTION CANT 5 - EXECUTION CANT 5 - EXECUTION CANT 5 - EXECUTION <		1	Follow activities identified in the M&V Plan				
 Cooperate with the M&V Provider Mechanical Contractor and Electrical Contractor for resolution of issues related to establishing connection between BAS and monitoring meters and equipmert Antend team meetings during construction and post-construction M&V period (1 year). M&V PROVIDERS RESPONSIBILITIES Provide RESPONSIBILITIES Provide M&V plan. Cooperate with the Mechanical Contractor, Electrical Contractor, and Controls Contractor for resolution of issues related to establishing connection between BAS and monitoring meters and equipment. Cooperate with the Mechanical Contractor, Electrical Contractor, and Controls Contractor for resolution of issues related to establishing connection between BAS and monitoring meters and equipment. Provide an M&V report at 1 year post construction. ART 2 – PRODUCTS – This SECTION NOT USED Antiontoring meters and sub-meters, both gas and electric, to have the ability to connect to the BAS and provide data to BAS at a minimum of 15 minute intervals. It is acceptable to use the utility for this purpose if allowable utility company. ART 3 – EXECUTION A Provide real-time monitoring of the whole building electricity KW and KWh use by using a signal from the building utility meter serving the HVAC, lighting, and plug loads and provide the data input to the BAS must be capable of trending this KW and KWh and. Data is to be collected 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data is to collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer wit		2.	Coordinate connection of electrical, gas, and C	OHW monitoring equipment with BAS			
A Attend team meetings during construction between BAS and monitoring meters and equipmer A Attend team meetings during construction and post-construction M&V period (1 year). A Providers responsibilities include: 1. Organize and lead the M&V team. 2. Providers responsibilities include: 1. Organize and lead the M&V team. 3. Convene M&V meetings as needed. 4. Cooperate with the Mechanical Contractor, Electrical Contractor, and Controls Contractor for resolution of issues related to establishing connection between BAS and monitoring meters and equipment. 5. Provide an M&V report at 1 year post construction. AMAT 2 - PRODUCTS - THIS SECTION NOT USED A Monitoring meters and sub-meters, both gas and electric, to have the ability to connect to the BAS and provid data to BAS at a minimum of 15 minute intervals. It is acceptable to use the utility for this purpose if allowable utility company. AMAT 3 - EXECUTION A Provide real-time monitoring of the whole building electricity KW and kWh use by using a signal from the building utility company. Anterse A. Provide real-time monitoring of the whole building electricity KW and kWh use by using a signal from the building utility company. Automation System (BAS). The BAS must be capable of trending this KW and kWh act. Data is to be collected 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS computer without being overwritten. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 wears can be overwritten. It is the responsibility of the mechanical contractor to coordinate this work. ANTURAL GAS A Provide real-time monitoring of whole building natural gas consumption by using a signal from the building util mater is to be automatically saved and archived on the BAS computer without being overwritten. Data is to collected in 15 minute intervals. Storage of at least		<u>-</u> . 3	Cooperate with the M&V Provider Mechanical	Contractor and Electrical Contractor for resolutio			
 A tend team meetings during construction and post-construction M&V period (1 year). M&V PROVIDERS RESPONSIBILITIES Provider responsibilities include: Congranze and lead the M&V team. Provide M&V meetings during construction construction M&V period (1 year). Congranze and lead the M&V team. Provide M&V meetings as needed. Cooperate with the Mechanical Contractor, Electrical Contractor, and Controls Contractor for resolution of issues related to establishing connection between BAS and monitoring meters and equipment. Provide an M&V report at 1 year post construction. CMTETES AND SUB-METERS Monitoring meters and sub-meters, both gas and electric, to have the ability to connect to the BAS and provide data to BAS at a minimum of 15 minute intervals. It is acceptable to use the utility for this purpose if allowable utility company. METER Provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building utility meter serving the HVAC, lighting, and plug loads and provide the data input to the Building Automation System (BAS). The BAS must be capable of trending this KW and kWh data. Data is be collected 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. A Provide real-time monitoring of whole building natural gas consumption by using a signal from the building utility meter serving the HVAC, lighting, and plug loads and provide the dAS. Data older than 5 years can be overwritten. It is the responsibility of the electrical contra		5.	of issues related to establishing connection be	tween BAS and monitoring meters and equinment			
 A. Providers RESPONSIBILITIES Provide real-time monitoring of the WAY team. Convene M&V meetings as needed. Convene M&V report at 1 year post construction. Provide an M&V report at 1 year post construction. AMETERS AND SUB-METERS A. Monitoring meters and sub-meters, both gas and electric, to have the ability to connect to the BAS and provid data to BAS at a minimum of 15 minute intervals. It is acceptable to use the utility for this purpose if allowable utility company. AMETER A. Provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building utility meter serving the HVAC, lighting, and plug bads and provide the data input to the BAI and brow the capable of trending this KW and kWh data. Data is to be collected 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data alder than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. AUTURAL GAS A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building utility meter servers. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. CONTINCE CONTINUE A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building utility meter servers and be automatically saved and archived on the BAS. Data older than 5 wears can be overwritten. It is the		4.	Attend team meetings during construction and	d post-construction M&V period (1 year).			
 A. Providers responsibilities include: Organize and lead the M&V team. Provide M&V plan. Convene M&V meetings as needed. Coperate with the Mechanical Contractor, Electrical Contractor, and Controls Contractor for resolution of issues related to establishing connection between BAS and monitoring meters and equipment. Provide an M&V report at 1 year post construction. ART 2 - PRODUCTS - THIS SECTION NOT USED METERS AND SUB-METERS A. Monitoring meters and sub-meters, both gas and electric, to have the ability to connect to the BAS and provid data to BAS at a minimum of 15 minute intervals. It is acceptable to use the utility for this purpose if allowable utility company. METER A. Provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building ultity meter serving the HVAC, lighting, and plug leads and provide the data input to the Building Automation System (BAS). The BAS must be capable of trending this W and kWh data. Data is to be collected 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building ut meter to provide the data input to the BAS. The BAS must be capable of trending as required on the BAS. Data is tol collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data is col collected in 15 minute intervals. Storage of at least 3 mo	16	M&V PROVIDERS F	ESPONSIBILITIES				
 Organize and lead the M&V team. Provide M&V plan. Cooperate with the Mechanical Contractor, Electrical Contractor, and Controls Contractor for resolution of issues related to establishing connection between BAS and monitoring meters and equipment. Provide an M&V report at 1 year post construction. ART 2 – PRODUCTS – THIS SECTION NOT USED INTERTS AND SUB-METERS A Monitoring meters and sub-meters, both gas and electric, to have the ability to connect to the BAS and provid data to BAS at a minimum of 15 minute intervals. It is acceptable to use the utility for this purpose if allowable utility company. ART 3 - EXECUTION INTERTS A Provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building utility meter serving the HVAC, lighting, and plug loads and provide the data input to the BAS. Data older than 3 months is to be automatically saved and archived on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 3 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. INTURAL GAS A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building utility meter scrang of whole building natural gas consumption by using a signal from the building utility meter as can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. INTURAL GAS A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building utility meter scrang to the BAS. The BAS. The BAS must be capable of trending gas consumption. Data is to lo collected in 15 minute thats. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the	1.0	A Providers re	sponsibilities include:				
 Provide M&V plan. Convene M&V meetings as needed. Cooperate with the Mechanical Contractor, Electrical Contractor, and Controls Contractor for resolution of issues related to establishing connection between BAS and monitoring meters and equipment. Provide an M&V report at 1 year post construction. XART 2 – PRODUCTS – THIS SECTION NOT USED 1 METERS AND SUB-METERS A Monitoring meters and sub-meters, both gas and electric, to have the ability to connect to the BAS and provid data to BAS at a minimum of 15 minute intervals. It is acceptable to use the utility for this purpose if allowable utility company. ART 3 - EXECUTION 1 METER A Provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building Automation System (BAS). The BAS must be capable of trending this kW and kWh data. Data is to be collected 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be overwritten. It is the responsibility of the electrical contractor to coordinate this work. 2 NATURAL GAS A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building utimeter to provide the data input to the BAS. The BAS. The BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 3 months or to the use of at least 3 months of 15 minute data is required on the BAS. Data older than 3 under the any storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 unotths is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 3 wears can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. 3 C PAPORARY MONITORING A. Provide easy access to allow for the temporary installat		1.	Organize and lead the M&V team.				
 Convene M&W meetings as needed. Cooverne M&W meetings as needed. Provide an M&V report at 1 year post construction. 		2	Provide M&V plan				
 4. Cooperate with the Mechanical Contractor, Electrical Contractor, and Controls Contractor for resolution of issues related to establishing connection between BAS and monitoring meters and equipment. 5. Provide an M&V report at 1 year post construction. XART 2 – PRODUCTS – THIS SECTION NOT USED METERS AND SUB-METERS A. Monitoring meters and sub-meters, both gas and electric, to have the ability to connect to the BAS and provid data to BAS at a minimum of 15 minute intervals. It is acceptable to use the utility for this purpose if allowable utility company. VART 3 - EXECUTION A. Provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building utility meter serving the HVAC, lighting, and plug loads and provide the data input to the Building Automation System (BAS). The BAS must be capable of trending this kW and kWh data. Data is to be collected 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS to deler than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 3 months is to be automatically saved and archived on the EAS computer without being overwritten. Data older than 3 months is to be automatically saved and archived on the BAS. Computer without being overwritten. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 3 consult to the BAS. The BAS must be capable of trending gas consumption. Data is to collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 worths is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 3 worths is to be automatically saved and archived on the BAS. Computer without being overwritten building ut		<u>-</u> . 3	Convene M&V meetings as needed				
 A provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building utility or the solution of the solution of the solution of the solution the solution the solution the solution of the solution of the solution of the solution of the solution the solutis of the solution of the solution the solution the solution the		3. 4	Cooperate with the Mechanical Contractor, Fl	ectrical Contractor, and Controls Contractor for			
 equipment. Provide an M&V report at 1 year post construction. PRODUCTS – THIS SECTION NOT USED METERS AND SUB-METERS Monitoring meters and sub-meters, both gas and electric, to have the ability to connect to the BAS and provid data to BAS at a minimum of 15 minute intervals. It is acceptable to use the utility for this purpose if allowable utility company. METER Provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building utility meter serving the HVAC, lighting, and plug loads and provide the data input to the Building Automation System (BAS). The BAS must be capable of trending this kW and kWh data. Data is to be collected 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building util meter to provide the data input to the BAS. The BAS must be capable of trending gas consumption. Data is to 1 collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate this work. FEMPORARY MONTORING FemPORARY MONTORING Chipting Pick plast Ughting Pick plast MUS cequipment including chillers, fans, circulation pumps, and air handling units to download trendable points. END OF SECTION 			resolution of issues related to establishing con	nection between BAS and monitoring meters and			
 5. Provide an M&V report at 1 year post construction. 2. PRODUCTS – THIS SECTION NOT USED 2.1 METERS AND SUB-METERS A. Monitoring meters and sub-meters, both gas and electric, to have the ability to connect to the BAS and provide data to BAS at a minimum of 15 minute intervals. It is acceptable to use the utility for this purpose if allowable utility company. 2.4 METER A. Provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building utility meter serving the HVAC, lighting, and plug loads and provide the data input to the Building Automation System (BAS). The BAS must be capable of trending this kW and kWh data. Data is to be collected 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. 2. NATURAL GAS A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building util meter to provide the data input to the BAS. The BAS must be capable of trending gas consupption. Data is to 1 collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 worths is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 3 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate to work. 3. TEMPORARY MONITORING			equinment				
ART 2 – PRODUCTS – THIS SECTION NOT USED A MONITORING WHETERS A. Monitoring meters and sub-meters, both gas and electric, to have the ability to connect to the BAS and provid data to BAS at a minimum of 15 minute intervals. It is acceptable to use the utility for this purpose if allowable utility company. ART 3 - EXECUTION A Provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building utility meter serving the HVAC, lighting, and plug loads and provide the data input to the Building Automation System (BAS). The BAS must be capable of trending this kW and kWh data. Data is to be collected 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. A Provide real-time monitoring of whole building natural gas consumption by using a signal from the building ut meter to provide the data input to the BAS. The BAS must be capable of trending as consumption. Data is to 1 collected in 15 minute intervals. Storage of a least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten Data older than 3 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate 1 work. A Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors for the electrical measurement and datalogging on the following systems: Lighting Plug loads HVAC equipment including chillers, fans, circulation pumps, and air handling units A DDC TRENDS A. The Contractor is to provide provision for remote access to BAS to view status of building and the abil to download trendable points. END OF SECTION VARNER PARK COMMUNITY RECREATION CE		5.	Provide an M&V report at 1 year post construct	ction.			
 INITE TROUCLES THE DECLEMENTED TO CONTROL TO CONTROL TO CONTROL TO THE PROVIDENT OF THE PROVIDE			SECTION NOT LISED				
 A. METERS AND SUB-METERS A. Monitoring meters and sub-meters, both gas and electric, to have the ability to connect to the BAS and provid data to BAS at a minimum of 15 minute intervals. It is acceptable to use the utility for this purpose if allowable utility company. XART 3 - EXECUTION A. Provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building utility meter serving the HVAC, lighting, and plug loads and provide the data input to the Building Automation System (BAS). The BAS must be capable of trending this kW and kWh data. Data is to be collected 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. ANTURAL GAS A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building utimeter to provide the data input to the BAS. The BAS must be capable of trending gas consumption. Data is to 1 collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 3 works is to be automatically saved and archived on the BAS computer without being overwritten. Data is to 1 collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 works is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 3 works is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 3 w	<u>r ANI</u>	<u>2 - FRODUCIS - III</u>	<u>-Stenowior Ostb</u>				
 A. Monitoring meters and sub-meters, both gas and electric, to have the ability to connect to the BAS and provid data to BAS at a minimum of 15 minute intervals. It is acceptable to use the utility for this purpose if allowable utility company. XART 3 - EXECUTION A. Provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building utility meter serving the HVAC, lighting, and plug loads and provide the data input to the Building Automation System (BAS). The BAS must be capable of trending this kW and kWh data. Data is to be collected 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwrithen. Data older than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. ANTURAL GAS A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building util meter to provide the data input to the BAS. The BAS must be capable of trending gas consumption. Data is to 1 collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Computer without being overwritten bats older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten bats older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate i work. TEMPORARY MONITORING	2.1	METERS AND SUB-	ИETERS				
data to BAS at a minimum of 15 minute intervals. It is acceptable to use the utility for this purpose if allowable utility company. IMETER A. Provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building utility meter serving the HVAC, lighting, and plug loads and provide the data input to the Building Automation System (BAS). The BAS must be capable of trending this kW and kWh data. Data is to be collected 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. 2. NATURAL GAS A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building util meter to provide the data input to the BAS. The BAS must be capable of trending gas consumption. Data is to 1 collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritte. Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate the work. 3. TEMPORARY MONITORING A. Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors for the electrical measurement and datalogging on the following systems: Lighting Plug loads HVAC equipment including chillers, fans, circulation pumps, and air handling units IMD C TRENDS A		A. Monitoring	neters and sub-meters, both gas and electric, to	have the ability to connect to the BAS and provide			
VART 3 - EXECUTION *.1 METER A. Provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building utility meter serving the HVAC, lighting, and plug loads and provide the data input to the Building Automation System (BAS). The BAS must be capable of trending this kW and kWh data. Data is to be collected 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. 2. NATURAL GAS A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building util meter to provide the data input to the BAS. The BAS must be capable of trending gas consumption. Data is to 1 collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritte. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritte. Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate 1 work. 3. TEMPORARY MONITORING A. Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors fo the electrical measurement and datalogging on the following systems: 1. Lighting 2. Plug loads 3. HVAC equipment including chillers, fans, circulation p		data to BAS	at a minimum of 15 minute intervals. It is accepta	able to use the utility for this purpose if allowable			
ART 3 - EXECUTION A. Provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building utility meter serving the HVAC, lighting, and plug loads and provide the data input to the Building Automation System (BAS). The BAS must be capable of trending this kW and kWh data. Data is to be collected 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building uti meter to provide the data input to the BAS. The BAS must be capable of trending gas consumption. Data is to I collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwrite. Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate i work. A. Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors for the electrical measurement and datalogging on the following systems: 1. Lighting 2. Plug loads 3. HVAC equipment including chillers, fans, circulation pumps, and air handling units A. The Controls Contractor is to provide provision for remote access to BAS to view status of building and the abi to download trendable points. END OF SECTION VARNER PARK COMMUNITY RECREATION CENTER XPANSION		utility comp	iny.				
A. Provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building utility meter serving the HVAC, lighting, and plug loads and provide the data input to the Building Automation System (BAS). The BAS must be capable of trending this kW and kWh data. Data is to be collected 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building utimeter to provide the data input to the BAS. The BAS must be capable of trending gas consumption. Data is to l collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwrittee. Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate to work. 3 TEMPORARY MONITORING A. Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors for the electrical measurement and datalogging on the following systems: 1. Lighting 2. Plug loads 3. HVAC equipment including chillers, fans, circulation pumps, and air handling units 4 DOC TRENDS A. The Controls Contractor is to provide provision for remote access to BAS to view status of building and the abi to download trendable points. END OF SECTION VARNER PARK COMMUNITY RECREATION CENTER XPANSION ONTRACT #9592 MUNIS #17196	PART	3 - EXECUTION					
 METER Provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building utility meter serving the HVAC, lighting, and plug loads and provide the data input to the Building Automation System (BAS). The BAS must be capable of trending this kW and kWh data. Data is to be collected 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. NATURAL GAS Provide real-time monitoring of whole building natural gas consumption by using a signal from the building util meter to provide the data input to the BAS. The BAS must be capable of trending gas consumption. Data is to 1 collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritte. Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate it work. TEMPORARY MONITORING Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors fo the electrical measurement and datalogging on the following systems:							
 A. Provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building utility meter serving the HVAC, lighting, and plug loads and provide the data input to the Building Automation System (BAS). The BAS must be capable of trending this kW and kWh data. Data is to be collected 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. 2 NATURAL GAS A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building utimeter to provide the data input to the BAS. The BAS must be capable of trending gas consumption. Data is to 1 collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwrittee. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwrittee. Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate i work. 3 TEMPORARY MONITORING A. Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors for the electrical measurement and datalogging on the following systems: Lighting	3.1	METER					
 building utility meter serving the HVAC, lighting, and plug loads and provide the data input to the Building Automation System (BAS). The BAS must be capable of trending this kW and kWh data. Data is to be collected 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. 2. NATURAL GAS A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building utimeter to provide the data input to the BAS. The BAS must be capable of trending gas consumption. Data is to a collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritte. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritte. Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate to work. 3. TEMPORARY MONITORING A. Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors fo the electrical measurement and datalogging on the following systems:		A. Provide rea	time monitoring of the whole building electricity	/ kW and kWh use by using a signal from the			
Automation System (BAS). The BAS must be capable of trending this kW and kWh data. Data is to be collected 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. 2 NATURAL GAS A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building uti meter to provide the data input to the BAS. The BAS must be capable of trending gas consumption. Data is to 1 collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritter. Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate the work. 3 TEMPORARY MONITORING A. Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors for the electrical measurement and datalogging on the following systems: 1. Lighting 2. Plug loads 3. HVAC equipment including chillers, fans, circulation pumps, and air handling units 4 DDC TRENDS A. The Controls Contractor is to provide provision for remote access to BAS to view status of building and the abi to download trendable points. END OF SECTION VARNER PARK COMMUNITY RECREATION CENTER XPANSION ONTRACT #5502 MUNIS #17196 01 95 00 - 2 MEASUREMENT AND VERIFICAT		building util	ty meter serving the HVAC, lighting, and plug loa-	ds and provide the data input to the Building			
 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building utimeter to provide the data input to the BAS. The BAS must be capable of trending gas consumption. Data is to I collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwrittee. Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate twork. 3 TEMPORARY MONITORING A. Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors for the electrical measurement and datalogging on the following systems:		Automation	System (BAS). The BAS must be capable of trend	ing this kW and kWh data. Data is to be collected i			
 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. NATURAL GAS Provide real-time monitoring of whole building natural gas consumption by using a signal from the building utimeter to provide the data input to the BAS. The BAS must be capable of trending gas consumption. Data is to a collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwrittee. Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate twork. TEMPORARY MONITORING Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors for the electrical measurement and datalogging on the following systems:		15 minute i	tervals. Storage of at least 3 months of 15 minut	e data is required on the BAS. Data older than 3			
 than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work. 1.2 NATURAL GAS Provide real-time monitoring of whole building natural gas consumption by using a signal from the building utimeter to provide the data input to the BAS. The BAS must be capable of trending gas consumption. Data is to a collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwrittee Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate twork. 3 TEMPORARY MONITORING		months is to	be automatically saved and archived on the BAS	computer without being overwritten. Data older			
 NATURAL GAS Provide real-time monitoring of whole building natural gas consumption by using a signal from the building utimeter to provide the data input to the BAS. The BAS must be capable of trending gas consumption. Data is to l collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwrittee Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate twork. TEMPORARY MONITORING Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors for the electrical measurement and datalogging on the following systems: Lighting Plug loads HVAC equipment including chillers, fans, circulation pumps, and air handling units A DDC TRENDS A. The Controls Contractor is to provide provision for remote access to BAS to view status of building and the abit to download trendable points. WARNER PARK COMMUNITY RECREATION CENTER XPANSION OUTRACT #9502 MUNIS #17196 01 95 00 - 2 MEASUREMENT AND VERIFICAT 		than 5 years	can be overwritten. It is the responsibility of the	electrical contractor to coordinate this work.			
 A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building utimeter to provide the data input to the BAS. The BAS must be capable of trending gas consumption. Data is to l collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwrittee Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate i work. 3 TEMPORARY MONITORING A. Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors for the electrical measurement and datalogging on the following systems:	3.2	NATURAI GAS					
 meter to provide the data input to the BAS. The BAS must be capable of trending gas consumption. Data is to 1 collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwrittee. Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate twork. 3. TEMPORARY MONITORING A. Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors for the electrical measurement and datalogging on the following systems:	0.2	A. Provide real	time monitoring of whole building natural gas cc	onsumption by using a signal from the building util			
 collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwrittee. Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate i work. 3 TEMPORARY MONITORING A. Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors for the electrical measurement and datalogging on the following systems:		meter to pr	wide the data input to the BAS. The BAS must be	capable of trending gas consumption. Data is to b			
 older than 3 months is to be automatically saved and archived on the BAS computer without being overwritter. Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate twork. 3 TEMPORARY MONITORING A. Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors for the electrical measurement and datalogging on the following systems:		collected in	15 minute intervals. Storage of at least 3 months	of 15 minute data is required on the BAS. Data			
Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate i work. 3. TEMPORARY MONITORING A. Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors for the electrical measurement and datalogging on the following systems: 1. Lighting 2. Plug loads 3. HVAC equipment including chillers, fans, circulation pumps, and air handling units 3. HVAC equipment including chillers, fans, circulation pumps, and air handling units 4. DDC TRENDS A. The Controls Contractor is to provide provision for remote access to BAS to view status of building and the abi to download trendable points. END OF SECTION VARNER PARK COMMUNITY RECREATION CENTER XPANSION ONTRACT #9502 MUNIS #17196 01 95 00 - 2 MEASUREMENT AND VERIFICAT		older than 3	months is to be automatically saved and archive	d on the BAS computer without being overwritter			
work. 3. TEMPORARY MONITORING A. Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors for the electrical measurement and datalogging on the following systems: 1. Lighting 2. Plug loads 3. HVAC equipment including chillers, fans, circulation pumps, and air handling units 3. HVAC equipment including chillers, fans, circulation pumps, and air handling units 3. HVAC equipment including chillers for remote access to BAS to view status of building and the abit to download trendable points. END OF SECTION VARNER PARK COMMUNITY RECREATION CENTER XPANSION ONTRACT #9502 MUNIS #17196 01 95 00 - 2 MEASUREMENT AND VERIFICAT		Data older t	nan 5 years can be overwritten. It is the responsil	bility of the mechanical contractor to coordinate t			
 5.3 TEMPORARY MONITORING A. Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors for the electrical measurement and datalogging on the following systems: Lighting Plug loads HVAC equipment including chillers, fans, circulation pumps, and air handling units A. The Controls Contractor is to provide provision for remote access to BAS to view status of building and the abit to download trendable points. END OF SECTION VARNER PARK COMMUNITY RECREATION CENTER XPANSION ONTRACT #9502 MUNIS #17196		work.	,				
 TEMPORARY MONITORING A. Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors fo the electrical measurement and datalogging on the following systems: Lighting Plug loads HVAC equipment including chillers, fans, circulation pumps, and air handling units A. DDC TRENDS A. The Controls Contractor is to provide provision for remote access to BAS to view status of building and the abi to download trendable points. END OF SECTION VARNER PARK COMMUNITY RECREATION CENTER XPANSION							
 A. Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors to the electrical measurement and datalogging on the following systems: Lighting Plug loads HVAC equipment including chillers, fans, circulation pumps, and air handling units A. The Controls Contractor is to provide provision for remote access to BAS to view status of building and the abi to download trendable points. END OF SECTION VARNER PARK COMMUNITY RECREATION CENTER XPANSION ONTRACT #9502 MUNIS #17196 01 95 00 - 2 MEASUREMENT AND VERIFICAT	3.3						
		A. Provide eas	access to allow for the temporary installation of	i spin-core current sensors and voltage sensors for			
		the electrica	i measurement and datalogging on the following	systems:			
2. Plug loads 3. HVAC equipment including chillers, fans, circulation pumps, and air handling units 4. DDC TRENDS A. The Controls Contractor is to provide provision for remote access to BAS to view status of building and the abi to download trendable points. END OF SECTION VARNER PARK COMMUNITY RECREATION CENTER XPANSION ONTRACT #9502 MUNIS #17196 01 95 00 - 2		1.	Lignung				
A. The Controls Contractor is to provide provision for remote access to BAS to view status of building and the abi to download trendable points. END OF SECTION VARNER PARK COMMUNITY RECREATION CENTER XPANSION ONTRACT #9502 MUNIS #17196 O1 95 00 - 2 MEASUREMENT AND VERIFICAT		2.	riug 1080s	lation number and air handling write			
A. The Controls Contractor is to provide provision for remote access to BAS to view status of building and the abi to download trendable points. END OF SECTION VARNER PARK COMMUNITY RECREATION CENTER XPANSION ONTRACT #9502 MUNIS #17196 01 95 00 - 2 MEASUREMENT AND VERIFICAT		3.	HVAC equipment including chillers, fans, circul	lation pumps, and air nandling units			
A. The Controls Contractor is to provide provision for remote access to BAS to view status of building and the abi to download trendable points. END OF SECTION VARNER PARK COMMUNITY RECREATION CENTER XPANSION ONTRACT #9502 MUNIS #17196 01 95 00 - 2 MEASUREMENT AND VERIFICAT	3.4	DDC TRENDS					
END OF SECTION VARNER PARK COMMUNITY RECREATION CENTER XPANSION ONTRACT #9502 MUNIS #17196 01 95 00 - 2 MEASUREMENT AND VERIFICAT	 .	A. The Control	Contractor is to provide provision for remote ac	cess to BAS to view status of building and the abili			
END OF SECTION VARNER PARK COMMUNITY RECREATION CENTER XPANSION ONTRACT #9502 MUNIS #17196 01 95 00 - 2 MEASUREMENT AND VERIFICAT		to download trendable points.					
VARNER PARK COMMUNITY RECREATION CENTER XPANSION ONTRACT #9502 MUNIS #17196 01 95 00 - 2 MEASUREMENT AND VERIFICAT							
VAKNER PARK COMMUNITY RECREATION CENTER XPANSION ONTRACT #9502 MUNIS #17196 01 95 00 - 2 MEASUREMENT AND VERIFICAT							
ONTRACT #9502 MUNIS #17196 01 95 00 - 2 MEASUREMENT AND VERIFICAT	WARN FXPA	NER PARK COMMUNITY	RECREATION CENTER				
	CONT	RACT #9502 MUNIS #17	196 01 95 00 - 2	MEASUREMENT AND VERIFICATI			
1				SECTION 01 95 01			
-----------	--------	---------	-----------	--			
2				MONITORING-BASED COMMISSIONING			
3							
4	PART :	1 – GE	NERAL				
5	1	.1	SUMMAR	Υ1			
6	1	.2	DEFINITIC	INS			
7	1	.3	MECHANI	CAL CONTRACTOR RESPONSIBILITIES			
8	1	.4	ELECTRIC	AL CONTRACTOR RESPONSIBILITIES			
9	1	.5	CONTROL	S CONTRACTOR RESPONSIBILITIES			
10	1	.6	MBCX PR	2 DVIDERS RESPONSIBILITIES			
11	PART	2 – PR	ODUCTS				
12	2	.1	METERS A	ND SUB-METERS			
13	PART 3	3 - EXE	CUTION				
14	3	.1	ELECTRIC	METER			
15	3	.3	NATURAL	GAS			
16	3	.4	HEATING	HOT WATER			
17	3	.5	CHILLED V	VATER			
18	3	.6	TEMPORA	IRY MONITORING			
19	3	.7	DDC TREN	IDS			
20							
21	PART	1 – GE	NERAL				
22							
23	1.1	SUM	MARY				
24		А.	Purpos	e: This section includes general requirements that apply to implementation of monitoring-based			
25			commis	ssioning (MbCx). MbCx is a component of the LEED v4.0 Rating System and the Commissioning Process.			
26				Deess replaces the Measurement and Verification process that was used in the LEED V3 Rating System.			
27		в.	RELATE	D WORK AND REQUIREMENTS			
28			1.	Section 01 31 13 Project Coordination			
29			2.	Section 01 31 19 Project Meetings			
30			J.	Section 01 31 23 Project Management web Site			
31			4. r	Section 01 91 00 Commissioning			
32			5.	Section 23 09 00 Instrumentation and Control for HVAC			
33			о. 7	Section 23 09 23 Direct Digital Control (DDC) System for HVAC			
34 25			7. o	Section 25 09 93 Sequence of Operations for HVAC DDC			
35			ð. 0	Section 26 24 16 Danalhaarda			
30			9.	Section 26 24 16 Panelboards			
37 20	1 2	DEEL					
38	1.2			Duilding Automation System			
39		А. D	BAS -	Commissioning			
40		в. С		Commissioning Demostic Liet Water			
41		с. Б		Domestic Hot Water			
42		D. г		Monitoring-Based Commissioning			
43		E. r	KVV -	Electric power read from utility meter			
44		г. С	KVVII -	Electric energy consumption read from utility meter			
45		G.	Plug Lo	ads – Electric power and consumption from wall receptacles			
40	1 2						
47	1.5		Control	LUNI RACI UR RESPOnsibilities with expertice and authority to act on its hehelf and shall eshedule them			
48		А.	Contrac	Lor shall assign representatives with expertise and authority to act on its behall and shall schedule them			
49			to parti	Cipate in and perform MDCX activities including, but not innited to, the following:			
50			1.	Follow activities identified in the CX Plan.			
21 21			2.	Coordinate connection of gas and Driv monitoring equipment with BAS.			
52 E 2			э.	issues related to data collection			
55 E /			л	Issues related to udid collection.			
54			4.	Attend team meetings during construction and post-construction MDCX period (1 year). Attend quarterly montings			
55			E	niceungs. Followen training or repairs needed to maintain performance			
50			э.	Followup training of repairs needed to maintain performance.			
57							

1	1.4	ELECT	RICAL C	ONTRAC	TOR RESPONSIBILITIES
2		Α.	Contra	actor sha	Il assign representatives with expertise and authority to act on its behalf and shall schedule them
3			to par	ticipate i	n and perform MbCx activities including, but not limited to, the following:
4			1.	Follow	activities identified in the Cx Plan.
5			2.	Coordir	nate connection of electrical monitoring equipment with BAS
6			3.	Cooper	ate with the Cx Agent, owner, Mechanical Contractor and Controls Contractor for resolution of
7				issues r	elated to data collection.
8			4.	Attend	team meetings during construction and post-construction MbCx period (1 year). Attend quarterly
9				meeting	gs.
10			5.	Followu	up training or repairs needed to maintain performance.
11					
12	1.5	CONT	ROLS CO	ONTRACT	TOR RESPONSIBILITIES
13		Α.	Contra	actor sha	Il assign representatives with expertise and authority to act on its behalf and shall schedule them
14			to par	ticipate i	n and perform MbCx activities including, but not limited to, the following:
15			1.	Follow	activities identified in the Cx Plan.
16			2.	Coordir	nate connection of electrical monitoring equipment with BAS
17			3.	Coordir	nate connection of gas and DHW monitoring equipment with BAS.
18			4.	Coordir	nate connection of measurement requirements (points, data access) with BAS.
19			5.	Cooper	ate with the Cx Agent, owner, Mechanical Contractor and Electrical Contractor for resolution of
20				issues r	related to data collection.
21			6.	Attend	team meetings during construction and post-construction MbCx period (1 year). Attend guarterly
22			0.	meetin	ps
22			5	Followi	un training or renairs needed to maintain nerformance
23			5.	10110110	ap danning of repairs needed to maintain performance.
25	16	MBC			
25	1.0	Δ	The C	V Agents	responsibilities related to MbCy include:
20		л.	THE C	1	Organize and lead the MbCy team
27				1. 2	Provide a Cy plan that includes the following procedures and information:
20				۷.	rolote a cx plan that includes the following procedures and information.
29					a. Totes and responsibilities as they relate to Mixex,
50 21					b. Inedsurement requirements (meters, points, metering systems, data access),
31 22					 the points to be tracked, with frequency and duration for trend monitoring; the limits of accentable values for tracked points and meterod values (where appropriate
32					u. The limits of acceptable values for tracked points and metered values (where appropriate,
33					predictive algorithms may be used to compare ideal values with actual values);
34					e. the elements used to evaluate performance, including conflict between systems, out-of
35					sequence systems components, and energy and water usage profiles;
36					t. an action plan for identifying and correcting operational errors and deficiencies;
37					g. training to prevent errors;
38					h. planning for repairs needed to maintain performance; and
39					 the frequency of analyses in the first year of occupancy (at least quarterly).
40				3.	Convene MbCx meetings as needed, but at least quarterly for 1 year post construction.
41				4.	Cooperate with the Mechanical Contractor, Electrical Contractor, and Controls Contractor for
42					resolution of issues related to establishing connection between BAS and monitoring meters and
43					equipment.
44				5.	Provide a final MbCx report at 1 year post construction.
45				6.	Update the systems manual with any modifications or new settings, and give the reason for any
46					modifications from the original design.
47					
48	PART	2 – PRC	DUCTS		
49					
50	2.1	METE	RS AND	SUB-ME	ITERS
51		Α.	Monit	oring me	eters and sub-meters, both gas and electric, to have the ability to connect to the BAS and provide
52			data to	o BAS at	a minimum of 15 minute intervals. It is acceptable to use the utility for this purpose if allowable by
53			utility	company	y.
54			,		
55	<u>PAR</u> T	<u>3 - EX</u> E	<u>CUTION</u>	<u>I</u>	

56

1 3.1 ELECTRIC METER

2

3

4

5

6

7 8

10

11

12 13

14

15

16

18

19 20

21

22

23

24

32

34

35

36

37

38

39

40

41

46

47

A. Provide real-time monitoring of the whole building electricity kW and kWh use by using a signal from the building utility meter serving the HVAC, lighting, and plug loads and provide the data input to the Building Automation System (BAS). The BAS must be capable of trending this kW and kWh data. Data is to be collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 years can be overwritten. It is the responsibility of the electrical contractor to coordinate this work.

9 3.3 NATURAL GAS

A. Provide real-time monitoring of whole building natural gas consumption by using a signal from the building utility meter to provide the data input to the BAS. The BAS must be capable of trending gas consumption. Data is to be collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate this work.

17 3.4 HEATING HOT WATER

A. Provide real-time monitoring of the heating hot water (HW) system by measuring water flow to the boiler(s) and HW supply and return temperatures and providing data input to the BAS. The BAS must be capable of trending. Data is to be collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS. Data older than 3 months is to be automatically saved and archived on the BAS computer without being overwritten. Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to coordinate this work.

25 3.5 CHILLED WATER

26A.Provide real-time monitoring of the chilled water (CW) system by measuring water flow to the chillers(s) and CW27supply and return temperatures and providing data input to the BAS. The BAS must be capable of trending. Data28is to be collected in 15 minute intervals. Storage of at least 3 months of 15 minute data is required on the BAS.29Data older than 3 months is to be automatically saved and archived on the BAS computer without being30overwritten. Data older than 5 years can be overwritten. It is the responsibility of the mechanical contractor to31coordinate this work.

33 3.6 TEMPORARY MONITORING

- Provide easy access to allow for the temporary installation of split-core current sensors and voltage sensors for the electrical measurement and datalogging on the following systems:
 - 1. Lighting

2.

- Plug loads
- 3. HVAC equipment including chillers, fans, circulation pumps, and air handling units
- 4. DHW equipment
- B. Temporary monitoring equipment will be provided by the Cx Agent.

42 **3.7 DDC TRENDS**

A. The Controls Contractor is to provide provision for remote access to BAS to view status of building and the ability
 to download trendable points per the MbCx requirements in the Cx Plan.

END OF SECTION

Page Intentionally Left Blank

			SECTION 02 20 00
			GENERAL SITEWORK REQUIREMENTS
Р	ART 1 - GEI	NERAL	
1	.1 SCOF	PE	
	Α.	The work under	r this section shall consist of providing all work, materials, labor, equipment, and supervision
	р	necessary to pr	ovide the work on the site as provided for in the technical specifications and on the Drawings.
	ь.	Madison Const	ruction Standards stated below
	C	All work shall h	e in accordance with applicable manufacturer's instructions
	С.		
1	.2 RELA	TED WORK AND F	PROVISIONS
	Α.	This section rela	ates to all Division 2 sections as they pertain to "sitework" to be performed.
	В.	This specification	on shall apply to all site work unless otherwise specified.
	С.	Applicable prov	visions of Division 1 shall govern all work under Division 2 and all Division 2 specifications.
1	.3 REFE	RENCE STANDARI	DS
	Α.	Abbreviations of	of standards or organizations referenced in this specification are as follows:
		AASHTO	American Association of State Highway and Transportation Officials
		ABMA	American Boiler Manufacturers Association
		ACPA	American Concrete Pipe Association
			American Gas Association
			American National Standards Institute
			Air Conditioning and Refrigeration Institute
		ASME	American Society of Mechanical Engineers
		ASPE	American society of Plumbing Engineers
		ASSE	American Society of Sanitary Engineering
		ASTM	American Society for Testing and Materials
		AWWA	American Water Works Association
		AWS	American Welding Society
		CISPI	Cast Iron Soil Pipe Institute
		CS	Commercial Standards, Products Standards Sections, Office of Eng. Standards Service, NBS
		EPA	Environmental Protection Agency
		FS	Federal Specifications, Superintendent of Documents, U.S. Government Printing Office
		IAPMO	International Association of Plumbing & Mechanical Officials
		IEEE	Institute of Electrical and Electronics Engineers
		ISA	Instrument Society of America
		MSS	Manufacturer's Standardization Society of the Valve & Fitting Industry, Inc.
		NBS	National Bureau of Standards
			National Electrical Manufacturors Association
			National Fire Protection Association
		NSE	National Sanitation Foundation
		PDI	Plumbing and Drainage Institute
		STI	Steel Tank Institute
		UL	Underwriters Laboratories Inc.
	В.	Where reference	ce is made to the "Construction Standards," it shall be construed to mean the pertinent section of
		the City of Mad	lison's Construction Standards.
1	.4 SUBS	TITUTIONS	
	Α.	Substitution of	Materials: Refer to the General Conditions of the Contract.
	В.	Where equipm	ent, accessories, or materials are used which differ in arrangement, configuration, dimensions,
		ratings, or engi	neering parameters from those indicated in the contract documents, the Contractor is
		responsible for	all costs involved in integrating the equipment or accessories into the system and for obtaining
		the intended pe	errormance from the system into which these items are placed.

1	1.5	CONTI	NUITY OF EXISTING TRAFFIC, PARKING, AND UTILITIES
2		Α.	Refer also to Division 1 - Contract General Requirements.
3		В.	Do not interrupt or change existing traffic, delivery, parking, or utility services without prior written approval
4			from the Construction Representative. When interruption is required, coordinate schedule with the Owner
5			agency to minimize disruptions. Unless specifically stated, all work involved in interrupting or changing existing
6			services is to be done during normal working hours.
7		C.	Verify the locations of any water, drainage, gas, sewer, electric, drainage, gas, sewer, electric, telephone, fuel,
8			steam lines or other utilities, and site features which may be encountered in any excavations or other sitework.
9			All lines shall be properly underpinned and supported to avoid disruption of service. Any service connections
10			encountered which are to be removed shall be cut off at the limits of the excavation and capped in accordance
11			with the requirements of permits governing such removals. Any permits required for this work will be obtained
12			by the Owner upon request of the Contractor.
13		D.	The Contractor shall comply with Wisconsin Statutes 62.15(11) Street Obstructions specifically that doing any
14			work which shall in any manner obstruct the streets or sidewalks shall put up and maintain barriers conforming
15			to the standards for traffic control devices in the manual adopted by the Department of Transportation under
16			s. 84.02 (4) (e) to prevent accidents, and be liable for all damages caused by failure so to do. All contracts shall
17			contain a provision covering this liability, and also a provision making the contractor liable for all damages
18			caused by the negligent digging up of streets, alleys or public grounds, or which may result from the Contractor's
19			carelessness in the prosecution of such work.
20			
21	1.6	PROTE	CTION OF EXISTING WORK AND FACILITIES
22		A.	Verify the locations of, and protect, any signs, paved surfaces, buildings, structures, landscaping, streetlights,
23			utilities, and all other such facilities that may be encountered or interfered with during the progress of the work.
24			Take all measures necessary to safeguard all existing work and facilities which are outside the limits of the work
25			or items which are within the construction limits but are intended to remain.
26		В.	Protect all paved, turfed, and landscaped surfaces to remain. Protect all areas outside of the construction limits
27			from the effects of erosion in accordance with the Erosion Control specification section.
28			·
29	1.7	CONST	RUCTION LIMITS
30		Α.	Construction Limits are indicated on the Drawings. In the absence of such a designation on the Drawings,
31			confine work to the minimum area reasonably necessary to undertake the work as determined by the Engineer.
32			All areas disturbed by excavation and grading, plus such additional areas as are disturbed by construction related
33			activities including construction access and storage and installation of materials shall be considered the
34			"Construction Area."
35			
36	1.8	EQUIPI	MENT AND MATERIALS FURNISHED BY OTHERS
37		Α.	None.
38			
39	1.9	SUBMI	TTALS
40		Α.	Refer also to Division 1 - Contract General Requirements.
41		В.	Submit manufacturer's preproduction (shop) drawings for any off-site constructed sitework items for approval
42			prior to the start of manufacturing and any electrically powered equipment.
43			
44	1.10	CERTIF	ICATIONS AND INSPECTIONS
45		A.	Refer also to Division 1 - General Conditions of the Contract.
46		В.	Obtain and pay for all required sampling, testing, inspections, and certifications except those provided by the
47			Architect/Engineer (A/E). Deliver originals of certificates and documents to the Owner's Project Representative.
48			Include copies of the certifications and documents in the Operating and Maintenance instructions.
49			
50	1.11	OPERA	TING AND MAINTENANCE INSTRUCTIONS
51		A.	Refer also to Division 1 - Contract General Requirements.
52		В.	Assemble material in an operating and maintenance manual composed of three-ring or post binders, using and
53			index at the front of each volume and tabs for each system or type of equipment installed. In addition to the
54			data indicated in the General Requirements, include the following information:
55			1. Copies of all approved shop drawings;
56			 Ivianutacturer's wiring diagrams for electrically powered equipment; Describes fitteste use formulate settific equivalence with a settific equiv
5/			 Records of tests performed to certify compliance with system requirements; Contification of transmission in the system requirements;
58			 Certificates of inspection by regulatory agencies;

1			5. Parts lists for manufactured equipment;
2			6. Lubrication instructions, including lists of frequency of lubrication during construction;
3			7. Warranties and/or guarantees; and
4			8. Additional information as indicated in the technical specification sections.
5			
6	1.12	TRAIN	IING OF OWNER PERSONNEL
7		Α.	Instruct Owner's personnel or their designee in the proper operation and maintenance of systems and
8			equipment provided as part of this project.
9			
10	1.13	RECO	RD DRAWINGS
11		Α.	Refer also to Division 1 - Contract General Requirements.
12		В.	Include copies of record drawings with the Operating and Maintenance instructions.
13			
14	PART	<u> - PRO</u>	<u>DUCTS</u>
15			
16	2.1	TRAFF	FIC CONTROL - BARRICADES, SIGNS, AND WARNING DEVICES
17		Α.	Provide traffic barricades, traffic signs, and warning devices in accordance with governing codes and regulations
18			and the Manual of Uniform Traffic Control Devices (MUTCD).
19		В.	Provide excavation barrier fencing in plastic, blaze orange color together with all supports and braces necessary
20			to provide an adequate safety barrier to unattended excavations.
21			
22	2.2	WARM	NING SIGNS
23		Α.	Provide all necessary warning signing as required by OSHA, these specifications, as directed by the City of
24			Madison's Construction Representative and as shown on the Drawings. Payment for providing, placing,
25			maintaining, and removing traffic control devices will be paid under the traffic control bid item as a lump sum
26			item for all project areas.
27			
28	PART	3 - EXEC	CUTION
29			
30	3.1	PROJE	ECT SITE CONDITIONS
31		Α.	Maintain a clean, safe, and orderly site.
32		В.	Provide adequate barricades, guards, warning lights, and other protection required at excavation and hazards
33			created by work.
34		C.	Control access to the site by only authorized personnel and vehicles.
35		D.	Maintain site housekeeping to provide for a safe and orderly project site. Collect and dispose of debris as it
36			accumulates.
37		F.	Provide shoring, bracing, sheet piling, planking, and forming required by the work.
38		Е.	Locate and protect overhead and underground utilities, sidewalks, drains, curbs, trees (including roots) shrubs,
39		••	ground cover, bench marks, monuments, other reference points, adjacent buildings, materials, and property
40			owned by others that are to remain
40 41		G	Protect items bearing responsibility for and replacement cost of damage arising from all operations connected
42		0.	with the work. If items are disturbed or destroyed, replace as directed by the Owner's representative
43		н	Fence and/or hox in all trees and plant material which are to remain at the drin line before work is started. Do
44			not nermit heavy equinment or stockniles within branch spread. Remove interfering branches without injury to
45			trunks and cover scars with tree naint
46		1	Control grading around structures: nitch ground to prevent water running into excavated areas
40 17		 I	Dits tranches within huilding lines and other excavations shall be maintained free of water
47 18		к 1.	Provide transform punding mices, dia other excernations shall be maintained ince of water.
40 /0		к. Т	Notify City and Owner's Project Representatives if springs or running water are encountered in evcayation:
50		с.	notify end owner a roject representatives in springs of running water are encountered in exceded 1011,
50			Dwner's Project Representative of points and areas that water will be discharged. Control discharge with
52			methods accentable to Wisconsin Department of Natural Pescurses (WDNP), the City's Project Peprosentative
52			and Local Municipal Degulations. At the Engineer's option, the Contractor shall drain the spring to the storm
55 54			and Local Municipal Regulations. At the Engineer's option, the Contractor shall drain the spring to the storm
54 EE		N.4	Sewer system by the use of held the. Be responsible for control managings to provent demogra from floading, exercise, and codimentation to an eiter
22		IVI.	be responsible for control measures to prevent damage from hooding, erosion, and sedimentation to on-site
EC			

24			END OF SECTION
23			
22		C.	Thoroughly clean all sewers and structures and remove and dispose of all debris and mud.
21		В.	Burning is not permitted.
20			deposit of excavated earth. Remove all surplus material, tools and equipment.
19		A.	Level off/shape all waste disposal areas and clean up areas used for the storage of materials or the temporary
18	3.4	CLEA	N UP
17			
16		В.	See applicable sections for specific restoration requirements.
15			and neat lines and repair with like materials to the full depth of the pavement as existed prior to the work.
14			Sawcut and remove all damaged pavements to the nearest existing joints, or with prior permission, to straight
13			prior to the work. Restore the surface of all disturbed areas to a like condition of the surface prior to the work.
12	3.3 SITE RESTORATION A. Unless otherwise specified or noted on the Drawings, fully ar		Unless otherwise specified or noted on the Drawings, fully and completely restore the site to a condition present
11	3.3 SITE RESTORATION		RESTORATION
9 10			
ō Q			state specifications. Water for base compaction shall be incidental to the base aggregate items and will not be
/			tor dust control. Water shall be provided by the contractor and placed in accordance with Section 624 of the
6		A.	Contractor shall apply water to the subgrade as directed by the City of Madison's Construction Representative
5	3.2	WAT	ER (DUST CONTROL)
4			
3			(WPDES) and City of Madison permits obtained for the project to prevent damage.
2			measures as shown on Drawings and as described in the Wisconsin Pollutant Discharge Elimination System
1		Ν.	Install and maintain temporary desilting basins, terraces, contour furrows, channel linings, waterways, and other

		SECTION 02 32 00 GEOTECHNICAL INVESTIGATION				
PART	1 - GFN	VERAL				
1.1	SCOP	É				
	A.	The work under this section shall consist of providing all work, materials, labor, section provides informat resulting from subsurface investigations completed at the site as part of this project. This section may co information applicable to ALL sitework and other technical specification sections as well. All contractors a expected to review this information as part of their duties to familiarize themselves with the site. Results of the gentechnical investigation apply only to the locations at which data was collected and at the				
	Б.	specific time it was collected. Gentechnical conditions may differ elsewhere on the site				
	C.	Supplementary Geotechnical Exploration Report (boring report and report logs) for the proposed Warner Community Recreation Center building in Madison, Wisconsin, prepared by Soils & Engineering Services, I				
	D.	Prior to making additional investigations of his/her own using test pits, borings, or other methods, Bidder				
		first gain permission from property owner and Owner's Project Manager.				
	E.	E. Geotechnical investigations completed by Bidder shall comply with all applicable requirements of Division 1 through Division 16 of this project.				
12	RFLA					
1.2	Δ	Applicable provisions of Division 1 govern work under this Section Related sections include:				
		1 Section 31 05 00 - Common Work Results for Earthwork				
		2. Section 31 20 00 - Earthmoving				
		3. Section 31 22 16.15 - Subgrade Prenaration				
		4. Section 32 11 23 - Aggregate Base Course				
		5. Section 32 17 23 - Pavement Markings				
PART	2 - PRC	<u>iducts</u>				
21	REDO	IRTS				
L	Δ	The Gentechnical Investigation Report at the end of this section is included for the Contractor's informati				
	В.	This report is also available by contacting the Geotechnical Consultant.				
	5.	1. Soils & Engineering Services, Inc., 1102 Stewart Street, Madison, WI 53713 (Phone: 608-274-7600				
	2 . EVE					
	J-LAL					
Not u	sed.					
		END OF SECTION				



Construction • Geotechnical Consulting Engineering/Testing

March 24, 2023 C23051-6

Ms. Amy Loewenstein Scanlon, AIA Department of Public Works – Engineering Division City County Building, Room 115 210 Martin Luther King, Jr. Blvd. Madison WI 53703-3342

Re: Geotechnical Exploration Report Proposed Warner Park Rec Center Gym Addition 1625 Northport Drive Madison, WI

Dear Ms. Lowenstein Scanlon:

Construction • Geotechnical Consultants, Inc. (CGC) has completed the subsurface exploration program for the above-referenced project. The purpose of this program was to evaluate the subsurface conditions within the proposed construction area and to provide geotechnical recommendations regarding site preparation, foundation, floor slab, below-grade wall, retaining wall and pavement design/construction. A determination of the site class for seismic design and a discussion regarding stormwater infiltration potential are also included. We are sending you an electronic copy of this report, and we can provide a paper copy upon request.

SITE AND PROJECT DESCRIPTION

We understand that a multi-purpose gym space addition is proposed for the Warner Park Community Rec Center in Madison, WI. The existing building is a single-story, slab-on-grade building with parking lots to the south and southeast. The site of the planned addition is bounded by a public access road to the east, parking to the south, the existing rec center to the west, and undeveloped land/lawn to the north. Existing site grades within the construction area slope down from the northeast towards the southwest at elevations ranging from about EL 873 ft to EL 861 ft based on a provided topographic site plan (JSD Professional Services; 1-ft contours).

The addition is planned approximately on the southeast corner of the existing rec center building, which is planned to include a multi-purpose gym space and a storage/pre-function area. New handicapped parking is planned directly south of the addition, as well as a new stormwater management facility to the north of the rec center. We anticipate that finished floor elevation of the new addition will be at about the same elevation as the existing building, which appears to be about EL 862 ft based on the provided topographic site plan, and that parking grades will generally follow current site topography. Considering the existing site topography, the north and/or east walls or the planned addition may be partially backfilled following construction.



SUBSURFACE CONDITIONS

Subsurface conditions for this study were explored by drilling four (4) Standard Penetration Test (SPT) soil borings near the planned corners of the gym addition to depths of 30 ft below current site grades. The soil borings were conducted by Americas Drilling Company (ADC; under subcontract to CGC) on March 2 and 8, 2023, using a truck-mounted CME-55 drill rig equipped with hollow stem augers and an automatic SPT hammer. The specific procedures used for drilling and sampling are described in Appendix A.

In addition to the soil borings, one test pit was performed within the planned stormwater management area. The test pit was excavated by Hellenbrand Brothers Excavating (HBE; under subcontract to CGC) and logged in the field by CGC on February 14, 2023. The test pit was extended to 10 feet below the ground surface.

The soil boring and test pit locations and depths were selected by JSD and field-staked by CGC and are shown in plan on the Soil Boring & Test Pit Location Exhibit attached in Appendix B. Ground surface elevations were surveyed by JSD and provided to CGC.

The subsurface profiles at the boring and test pit locations were fairly consistent, and the following strata were typically encountered (in descending order):

- About 4 to 15 in. of *topsoil*; over
- Loose to very dense *sand* soils with varying silt and gravel contents as well as zones including scattered cobbles and boulders to the maximum depths explored.

Groundwater was encountered in the borings at depths between 13 and 23.5 ft below the ground surface during and up to approximately 1 hour after the completion of drilling. Groundwater was not encountered in the test pit during or upon the completion of excavating. Groundwater levels are expected to fluctuate with seasonal variations in precipitation, infiltration, evapotranspiration, the water level in nearby waterbodies as well as other factors.

A more detailed description of the site soil and groundwater conditions is presented on the Soil Boring Logs attached in Appendix B, and on the WDSPS *Soil and Site Evaluation – Storm* form for the test pit, which is attached in Appendix E. A representative sample taken from the test pit was analyzed with regard to its particle size distribution (gradation) to aid in classification, and the lab report is also attached in Appendix E.

DISCUSSION AND RECOMMENDATIONS

Subject to the limitations discussed below and based on the subsurface exploration, it is our opinion that the site is generally suitable for construction and that the proposed addition can be supported by a conventional spread footing foundation system bearing on medium dense to very dense native granular



soil. Our recommendations for site preparation, foundation, floor slab, below-grade wall, retaining wall and pavement design/construction, along with our assessment of the site class for seismic design and discussion of stormwater infiltration potential, are presented in the following subsections. Additional information regarding the conclusions and recommendations presented in this report is discussed in Appendix C.

1. <u>Site Preparation</u>

We recommend that topsoil be stripped at least 10 ft beyond the proposed construction area, including areas requiring fill beyond the building footprint and new pavement limits. The topsoil can be stockpiled on-site and later re-used as fill in landscaped areas. As noted previously, topsoil thicknesses ranged from about 4 to 15 in. in the borings and the test pit, but variable topsoil thicknesses should be expected between and beyond boring/test pit locations due to previous grading activities.

After topsoil stripping, we envision that the next step in site development will include excavation up to about 11 ft below current site grades to the slab elevation, with additional excavation to footing grades (and potential undercuts). Based on the current site layout, we anticipate that excavation sidewalls can generally be sloped back according to OSHA requirements. The sand soils with generally high amounts of fines (denoted as SM on the boring logs) are expected to control excavation slopes, and these soils are generally classified as OSHA "Type B" soils, with excavation slopes of 1.0H:1.0V expected to be at least temporarily stable. Note that flatter side slopes may be required where perched or seeping water is present that destabilizes the side slopes. *The appropriate excavation side slopes should be determined by a competent person completing the earthwork in accordance with OSHA slope guidelines.* Where adequate sloping is not possible, temporary shoring (earth retention) will be required. We recommend shoring systems be designed by an appropriately qualified professional engineer.

Following excavation to grade, where required, exposed subgrades are largely expected to consist of loose to dense native sand soils. In areas remaining at-grade or requiring fill, we recommend that cohesive and fine-grained subgrades (i.e., clay and silt), if present, be statically recompacted (i.e., without vibration) and subsequently proof-rolled with a piece of heavy rubber-tire construction equipment, such as a loaded tri-axle dump truck, to check for soft/yielding areas. Where soft/yielding areas are observed, these soils should be undercut and replaced with granular backfill compacted to at least 95% compaction based on modified Proctor methods (ASTM D1557) in accordance with our Recommended Compacted Fill Specifications presented in Appendix D. Alternatively, 3-in. dense graded base (DGB) that is placed in loose 10-in. lifts and compacted until deflection ceases can also be used to restore grades in undercut areas. Granular subgrades (i.e., sand and gravel) should be thoroughly recompacted with a vibratory smooth-drum roller, and zones that remain loose after recompaction should be undercut and replaced as described above.



2. <u>Foundation Design</u>

We assume that the finished floor elevation of the addition will match the existing building at about 862 ft, and we anticipate perimeter footings to extend at least 4 ft below the floor elevation (or finished exterior site grades) for frost protection. Interior footings or perimeter footings where the building walls are partially backfilled may also bear at shallower elevations. Accordingly, we expect footing subgrades to largely consist of medium dense to very dense sand soils. Isolated zones of loose sand soils could occur beneath footings and may need to be subsequently undercut and replaced if/where encountered.

Provided that the recommendations outlined above and in further detail below are followed, we recommend the following parameters be used for foundation design:

•	Maximum net allowable bearing pressure:	5,000 psf
•	 <u>Minimum foundation widths:</u> Continuous wall footings: Column pad footings: 	18 in. 30 in.
•	 <u>Minimum footing depths below finish site grades:</u> Exterior/perimeter footings: Interior footings: 	4 ft no minimum requirement

Where new footings are planned adjacent to existing building foundations, the effects of overlapping soil stresses must be considered, and the recommended maximum net allowable bearing pressure must not be exceeded. If the existing building footings are designed for a lower allowable bearing pressure, the lower bearing pressure will control the maximum allowable overlapping soil stresses. Care must also be exercised not to undermine the existing building foundations during new footing and undercut excavations.

As a variety of subsurface conditions may be encountered across the building footprint, foundation subgrades should be checked by a CGC field representative to document that the subgrade soils are suitable for footing support or otherwise advise on corrective measures, such as undercutting. We recommend using a smooth-edged backhoe bucket for footing and undercut excavations. Granular soils exposed at footing grade should be thoroughly recompacted with a large vibratory plate compactor or an excavator-mounted hoe-pack prior formwork/concrete placement. Granular soils which remain loose following recompaction will require undercutting and replacement. Soils potentially susceptible to disturbance from vibratory compaction (e.g., cohesive/fine-grained soils or sands with elevated moisture contents) should be hand-trimmed. OSHA slope guidelines should be followed if workers need to enter the excavations.



Undercutting will generally be required where loose or disturbed sand soils that cannot be recompacted satisfactorily remain at and slightly below footing grades. *The base of undercut excavations should be widened beyond the footing edges at least 0.5 ft in each direction for each foot of undercut depth for stress distribution purposes.* Where undercutting of loose sand is required, we recommend that the base of the undercut be thoroughly recompacted, followed by replacement of the excavated sand to restore footing grades. The backfill should be compacted to at least 95% compaction based on modified Proctor methods (ASTM D1557), in accordance with the Recommended Compacted Fill Specifications presented in Appendix D. Alternatively, imported granular fill soils compacted to 95% modified Proctor or 3-in. DGB that is placed in loose 10-in. lifts and compacted until deflection ceases can also be used to restore foundation grades.

Provided the foundation design/construction recommendations discussed above are followed, we estimate that total and differential settlements should be on the order of 1.0 and 0.5 in., respectively.

3. <u>Seismic Site Class</u>

In our opinion, the average soil properties in the upper 100 ft of the site [based on SPT blow counts (N-values) projected to be between 15 and 50 blows/ft, on average, in the granular soils underlying the site] may be characterized as a stiff soil profile. This characterization would place the site in Site Class D for seismic design according to the International Building Code and ASCE 7.

4. Floor Slab

Based on a presumed finished floor elevation of about EL 862 ft, floor slab subgrades are generally anticipated to consist of native sand soils, with minor placement of structural fill possibly required near the southwest building corner. We recommend that granular floor slab subgrades be thoroughly recompacted with a vibratory smooth-drum roller prior to concrete placement. Cohesive and fine-grained floor slab subgrades (if present) should be statically recompacted (i.e., without vibration) and subsequently proof-rolled. Areas that remain loose after recompaction or where soft/yielding zones are observed should be undercut and replaced with well-compacted compacted 3-in. DGB or granular backfill. *Some subgrade improvement should be expected where loose sands are present at floor slab grades, possibly near borings B-1 and B-4 (based on the assumed FFE), and we recommend that the project budget include a contingency for such operations.*

To act as a capillary break below the slab, we recommend including a minimum 6-in. thick layer of well-graded sand/gravel with less than 5% by weight passing the No. 200 U.S. standard sieve. Note, however, that some structural engineers require a layer of dense graded base, such as 1¹/₄-in. DGB, rather than sand/gravel below floor slabs to increase the subgrade modulus immediately below the slab. To further reduce the potential for moisture migration through the slab, a plastic vapor barrier can also be utilized. Fill and base layer material below the floor slab should be placed as described in the Site Preparation section of this report. Slabs constructed on a minimum 6-in. thick dense graded base layer may be designed utilizing a subgrade modulus of 150 pci, and a subgrade modulus of 100 pci should



be used for the design of slabs that are constructed on a sand/gravel layer. The design subgrade moduli are based on a firm or adequately stabilized, recompacted subgrade such that non-yielding conditions are developed. The slab should be structurally separated from the footings with a compressible filler and have construction joints and reinforcement for crack control.

5. <u>Below-Grade Walls</u>

If exterior building walls will be (partially) backfilled with soil, such as potentially at the north and east sides of the addition, we assume these walls will be laterally restrained against rotating by structural means. Therefore, *at-rest* lateral earth pressures should be used during design of these walls. To reduce the buildup of such pressures, high-quality backfill should be placed within 4 to 6 ft of the walls. We recommend that a perimeter drainage system be installed to intercept potential surface water infiltration, and that the granular backfill be continuously connected to the drainage system, which discharges water to one or more sumps. The granular backfill should be well-graded sand or gravel having no more than 12% by weight passing the No. 200 U.S. standard sieve (i.e., USCS designations SP, SP-SM, GP or GP-GM). Native sand soils with higher fines-content (denoted SM on the borings logs) may potentially also be used as below-grade wall backfill if a three-dimensional drainage board is included in the wall design. Soils containing cobbles/boulders should not be used in direct contact with below-grade walls. To impede the inflow of surface moisture, the final 2 ft of backfill in unpaved areas should consist of a clayey fill cap. The clayey cap (or pavement) should be graded to promote positive drainage away from the walls.

Before placing the wall backfill, the exterior walls should be damp-proofed with spray-applied or mopped-on rubber or bituminous sealer. Compaction of the backfill within 3 to 5 ft of the walls should be performed with lightweight equipment to avoid the development of excessive lateral earth pressures. The backfill should generally be compacted to a minimum compaction level of 93% modified Proctor following Appendix D guidelines. Below-grade walls constructed in accordance with the above recommendations may be designed for an equivalent fluid pressure of 55 psf per ft of depth (*at-rest* conditions). Additionally, the wall design should also account for surcharge effects that could be applied during or after construction.

6. <u>Retaining Walls</u>

Site retaining walls that are not laterally restrained from rotating can be designed for *active* earth pressures behind the walls and *passive* pressures in front of the walls. Lateral earth pressures behind the retaining walls can be reduced by backfilling with sand with less than 12% by weight passing the No. 200 U.S. standard sieve, as described in the preceding section. In addition, weepholes should be placed near the base of these walls on 10-ft centers to provide drainage of the wall backfill. The weepholes should be hydraulically connected with the backfill and should be protected with a non-woven geotextile fabric to minimize soil loss through the weepholes. The wall designer may have other and/or additional drainage requirements.



Assuming the inclusion of free-draining wall backfill, retaining walls that are free to rotate and constructed in accordance with the above recommendations may be designed for an *active* equivalent fluid pressure of 35 psf per ft of depth. *Passive* pressures are expected to be on the order of 200 psf per ft of depth. The passive pressure value includes a safety factor of 2 to prevent excessive wall deflection. The retaining wall design should also take into account surcharge effects which could be applied during or after construction.

We recommend using an *ultimate* concrete-to-soil friction factor of 0.3 for retaining wall footings bearing on at least medium stiff clay and silt soils. For footings bearing on sand, an *ultimate* concrete-to-soil friction factor of 0.4 may be implemented.

7. <u>Pavement Design</u>

We anticipate that pavement design will be controlled by the shallow sand soils, and subgrades should be prepared as described in the Site Preparation section of this report, with recompaction/proof-rolling completed prior to base course placement. Areas requiring undercutting/stabilization and the depth of undercutting should be determined in the field by proof-rolling prior to installing the base course layer. We anticipate that asphalt pavement on this site will primarily be exposed to automobile traffic with less than one 18-kip equivalent single axle load (ESAL) per day. In view of this, we have assumed Traffic Class I following Wisconsin Asphalt Pavement Association (WAPA) recommendations for smaller parking areas and driveways that are mainly used by light passenger vehicles. However, main sections of the driveways are likely to experience heavier traffic loads (e.g., due to large trucks). For pavement areas where trucks will routinely travel, as well as parking lots with 50 or more stalls, we have assumed a traffic load of up to 5 ESALs per day and Traffic Class II according to WAPA. The pavement sections summarized in Table 1 below were selected assuming a Soil Support Value "SSV" of about 4.0 for a firm or adequately stabilized silty sand subgrade and a design life of 20 years.

	Thickne	sses (in.)			
Material	Traffic Class I (Light Duty)	Traffic Class II (Medium Duty)	WDOT Specification ⁽¹⁾		
Bituminous Upper Layer ^(2,3)	1.75	1.75	Section 460, Table 460-1		
Bituminous Lower Layer ^(2,3)	1.75	2.25	Section 460, Table 460-1		
Dense Graded Base Course ^(2,4)	8.0	10.0	Sections 301 and 305		
Total Thickness	11.5	14.0			

TABLE 1Recommended Pavement Sections



Notes:

- 1) Wisconsin DOT Standard Specifications for Highway and Structure Construction, latest edition, including supplemental specifications, and Wisconsin Asphalt Pavement Association 2020 Asphalt Pavement Design Guide.
- 2) Compaction requirements:
 - Bituminous concrete: Refer to Section 460-3.
 - Base course: Refer to Section 301.3.4.2, Standard Compaction
- 3) Mixture Type LT (or E-0.3) bituminous; refer to Section 460, Table 460-2 of the *Standard Specifications*.
- 4) The upper 4 in. should consist of 1¹/₄-in. DGB; the bottom part of the layer can consist of 3-in. DGB.

The medium-duty pavement section may be considered across the entire area for constructability purposes. The recommended pavement sections assume regular maintenance (crack sealing, etc.) will occur, as needed. Note that if traffic volumes are greater than those assumed, CGC should be allowed to review the recommended pavement sections and adjust them accordingly. Alternative pavement designs may prove acceptable and should be reviewed by CGC. If there is a delay between subgrade preparation and placing the base course, the subgrade should be recompacted.

Where concrete pavement may be used, such as in pavement areas subjected to concentrated wheel loads (e.g., entrance slabs, etc.), we recommend that the concrete be at least 6 in. thick and contain adequate reinforcement for crack control. Concrete slabs underlain by a minimum 6-in. thick dense graded base layer over a firm or stabilized subgrade can be designed utilizing a subgrade modulus of 150 pci.

8. <u>Stormwater Infiltration Potential</u>

We understand that a stormwater management feature is planned to the northwest of the proposed building addition, and test pit TP-1 was performed in this area to evaluate the subsurface conditions with regard to their stormwater infiltration potential. The profile in TP-1 was comprised of 15 in. of surficial topsoil (silt loam), underlain by sandy loam to gravelly sandy loam strata to the maximum depth explored.

Provided the infiltration system extends through the shallow silt loam horizon and into the coarsergrained layers (or lower-permeability soils are undercut below the bottom of the infiltration system and replaced with appropriate sandier soils), we anticipate that some infiltration will likely be possible. Note, however, that the granular soils were found to contain occasional seams of lower-permeability soil (e.g., *silty clay loam*), which are expected to control the vertical infiltration rate. In addition, the



granular soils are apparent overconsolidated glacial till, and the relatively high density may also reduce the infiltration potential. In an effort to improve the infiltration potential, we recommend the soils be excavated and blended (or deep-tilled, ripped, etc.) in order to break up the existing soil structure and fairly thin lower-permeability seams. *Thicker silt and clay layers will require excavation and removal*. After removal of the overlying lower-permeability strata, we recommend that the deep-tilling process extend at least 5 ft (potentially deeper pending field observations) below the bottom of the infiltration system. Samples of the mixed soils should be collected during construction to document that the gradations of the mixed samples are consistent with the soil texture that the design infiltration rate is based upon (per Table 2 of WDNR Tech. Std. 1002). *Variability in the soil conditions should be expected across the site that could result in a wide range of undercut depths to reach soil suitable for the design infiltration rate*.

Infiltration Potential: The following is a summary of the estimated infiltration rates for the soils encountered in Test Pit TP-1, per Table 2 of the WDNR Conservation Practice Standard 1002, *Site Evaluation for Storm Water Infiltration. Where lower-permeability soil (e.g., silty clay loam, silt loam, etc.) seams/layers exist within otherwise more permeable soils (e.g., granular, coarse-grained soils), the infiltration rate of the lower-permeability seams/layers will control the overall vertical infiltration rate, unless the lower-permeability seams are removed or the layer (with scattered seams) is excavated and blended, as previously discussed. The estimated infiltration rates are as follows:*

•	Silty clay loam (SiCL)	0.04 in./hr
•	Silt loam (SiL)	0.13 in./hr
•	Sandy loam (SL)	0.50 in./hr
•	Gravelly sandy loam (FSL)	0.50 in./hr

Note that the infiltration rates should be considered approximate since they are merely based on soil texture and do not account for in-place soil density and other factors, which will affect the infiltration rate. We recommend that, at the time of construction, the soils at and several feet below the bottom of stormwater management system be checked by a certified soil tester *in conjunction with the basin designer* to document that the soils are appropriate for the design infiltration rate or recommend remedial measures, if necessary. Please refer to the WDSPS Soil and Site Evaluation – Storm form for the test pit, contained in Appendix E, for a more detailed description of the encountered subsurface conditions.

Groundwater: Groundwater was not encountered in the test pit During or upon the completion of excavating. Based on the soil borings performed within the planned building area, the depth to groundwater ranges between about 13 and 23.5 ft below existing site grades, although seasonally higher groundwater levels are also likely.



Bedrock: Bedrock was not encountered in the test pit or soil borings performed for this study. The depth of bedrock should be expected to vary across the site.

During construction, appropriate erosion control should be provided to prevent eroded soil from contaminating the stormwater management area. Where appropriate, the stormwater system design should include pretreatment to remove fine-grained soils (silt/clay) and clogging materials (oils/greases) from stormwater prior to entering the infiltration area. Additionally, a regular maintenance plan should be developed to remove silt/clay soils and clogging materials that may accumulate in the bottom of the stormwater management area over time. Failure to adequately control fine-grained soils and clogging materials from entering the infiltration area or failure to regularly remove fine-grained soils and clogging materials that accumulate at the base of the stormwater infiltration system will likely cause the stormwater management system to fail. Additionally, it is important that the soils in the bottom of the infiltration system do not become compacted during construction. Refer to WDNR *Conservation Practice Standards 1002, 1003 and 1004*, as well as *NR151* for additional information.

CONSTRUCTION CONSIDERATIONS

Due to variations in weather, construction methods and other factors, specific construction problems are difficult to predict. Soil related difficulties which could be encountered on the site are discussed below:

- Due to the potentially sensitive nature of some of the on-site soils, we recommend that final site grading activities be completed during dry weather, if possible. Construction traffic should be avoided on prepared subgrades to minimize potential disturbance.
- Contingencies in the project budget for subgrade stabilization with coarse aggregate in pavement and floor slab areas should be increased if the project schedule requires that work proceed during adverse weather conditions.
- Earthwork construction during the late fall through early spring could be complicated as a result of wet weather and freezing temperatures. During cold weather, exposed subgrades should be protected from freezing before and after footing construction. Fill should never be placed while frozen or on frozen ground.
- Excavations extending greater than 4 ft in depth below the existing ground surface should be sloped or braced in accordance with current OSHA standards. Earth retention systems, if needed, should be designed by an appropriately qualified professional engineer.



- Care must be exercised not to undermine the existing building foundations during new foundation and potential undercut excavations. The need for underpinning or working in discrete sections should be evaluated by the contractor.
- Based on the observations made during our field exploration, we generally do not anticipate groundwater to be encountered during construction. However, water accumulating at the bottom of excavations as a result of precipitation or seepage should be quickly removed, with dewatering means and methods the contractor's responsibility.

RECOMMENDED CONSTRUCTION MONITORING

The quality of the foundation, floor slab and pavement subgrades will be largely determined by the level of care exercised during site development. To check that earthwork and foundation construction proceed in accordance with our recommendations, the following operations should be monitored by CGC:

- Topsoil stripping and subgrade proof-rolling/compaction;
- Fill/backfill placement and compaction;
- Foundation excavation/subgrade preparation; and
- Concrete placement.

* * * * *



It has been a pleasure to serve you on this project. If you have any questions or need additional consultation, please contact us.

Sincerely,

CGC, Inc.

Caren mma

Emma L. Carew, EIT Staff Engineer

Tim F. Gassenheimer, PE, CST Senior Staff Engineer

Encl:	Appendix A -	Field Exploration			
	Appendix B -	Soil Boring & Test Pit Location Exhibit			
		Logs of Test Borings (4)			
		Log of Test Boring-General Notes			
		Unified Soil Classification System			
	Appendix C -	Document Qualifications			
	Appendix D -	Recommended Compacted Fill Specifications			

Appendix E - WDSPS *Soil and Site Evaluation – Storm* Form Particle Size Distribution Test Report

APPENDIX A

FIELD EXPLORATION

APPENDIX A

FIELD EXPLORATION

Subsurface conditions for this study were explored by drilling four (4) Standard Penetration Test (SPT) soil borings to depths of 30 ft below current site grades, which were sampled at 2.5-ft intervals to a depth of 10 ft and at 5-ft intervals thereafter. The samples were obtained in general accordance with specifications for standard penetration testing, ASTM D 1586. The specific procedures used for drilling and sampling are described below.

1. Boring Procedures between Samples

The boring is extended downward, between samples, by a hollow-stem auger.

2. <u>Standard Penetration Test and Split-Barrel Sampling of Soils</u> (ASTM Designation: D 1586)

This method consists of driving a 2-inch outside diameter split-barrel sampler using a 140-pound weight falling freely through a distance of 30 inches. The sampler is first seated 6 inches into the material to be sampled and then driven 12 inches. The number of blows required to drive the sampler the final 12 inches is recorded on the log of borings and is known as the Standard Penetration Resistance.

During the field exploration, the driller visually classified the soil and prepared a field log. *Field* screening of the soil samples for possible environmental contaminants was not conducted by the driller as these services were not part of CGC's work scope. Water level observations were made in each boring during and after drilling and are shown at the bottom of each boring log. Upon completion of drilling, the borings were backfilled with bentonite to satisfy WDNR regulations, and the soil samples were delivered to our laboratory for visual classification. The soils were visually classified by a geotechnical engineer using the Unified Soil Classification System (USCS).

In addition to the soil borings, one test pit was excavated within the planned stormwater management area. The excavation was logged (per the USDA classification system) in the field by a certified soil tester (CST) and subsequently backfilled with excavation spoils to reach original site grades.

The final boring logs prepared by the engineer, along with a Soil Boring & Test Pit Location Exhibit and a description of the Unified Soil Classification System are presented in Appendix B. The WDSPS *Soil and Site Evaluation – Storm* form for the test pit is contained in Appendix E.

APPENDIX B

SOIL BORING LOCATION EXHIBIT LOGS OF TEST BORINGS (4) LOG OF TEST BORING-GENERAL NOTES UNIFIED SOIL CLASSIFICATION SYSTEM



				_ 29	Location Madison, WI	Job No. Sheet	1	of	1-6 1	·····
	SA	MPL	E		VISUAL CLASSIFICATION	SOIL	PRC	PEF	RTIE	S
No.	F Rec (in.)	Moist	N	Depth (ft)	and Remarks	qu (qa) (tsf)	W	LL	PL	LO:
1	16	М	8		$\begin{array}{c} 4 \pm \text{ in. TOPSOIL} \\ \text{Loose, Brown Fine to Medium SAND, Some Silt,} \\ \text{Trace Clay, Little to Some Gravel (SM; Possible} \\ \end{array}$					
2	18	М	21	 └ ┝ ┝ ┝	Medium Dense, Tan Fine to Medium SAND, Trace to Little Silt, Some Gravel (SP/SP-SM)					
3	18	М	51		Very Dense, Brown Fine to Medium SAND, Some Silt and Gravel (SM)					
4	12	М	64		Very Dense, Tan Silty Fine SAND, Some Gravel (SM)					
5	6	M	50/3"		Very Dense, Brown Fine to Medium SAND, Some Silt and Gravel, Scattered Cobbles and Boulders (SM)					
6	18	W	73		Very Dense, Brown Fine to Coarse SAND, Trace to Little Silt, Some Gravel (SP/SP-SM)					
7	12	W	73							
8	14	W	50/4"	└ └ ┝ 30─	Very Dense, Tan Silty Fine SAND, Some Gravel, Scattered Cobbles and Boulders (SM)					
					End of Boring at 30 ft Borehole Backfilled with Bentonite Chips and Soil Cuttings					
			W		LEVEL OBSERVATIONS	GENERA		TES	5	

1				
The stratification	lines represent	the appro	ximate boundary	betweer
soil types and the	transition may	be gradual.	_	

Drill Method 2.25" HSA; Autohammer

5

	G	C	Inc		Pr Lc	LOG OF TEST BORING oject Warner Park Rec Center ocation Madison, WI	Boring No Surface E Job No. Sheet	o. levation (B n (ft) 2 2305 of	- 2 873. 1-6 1	2
				- 29	921 Pe	rry Street, Madison, WI 53713 (608) 288-4100, FAX (608)	288-7887 —				
	SA	MPL	.E			VISUAL CLASSIFICATION	SOIL	PRC	PEF	RTIE	S
No.	T Rec P (in.)	Moist	N	Depth (ft)		and Remarks	qu (qa) (tsf)	W	LL	PL	LOI
					////	8 ± in. TOPSOIL					
1	12	M	7			Loose, Brown Fine to Medium Clayey SAND, Trace Gravel (SC)					
2	12	M	9	└ └ ┼── 5─		Loose to Very Dense, Brown Fine to Medium SAND, Some Silt and Gravel, Scattered Cobbles					
3	16	M	19			and Boulders (SM)					
4	14	М	41								
4	14		41	└ ┝ 10- ┝ └							
5	8	M	50/4"								
6	18	M	69								
7	18	W	74								
8	14	W	75	⊢ ├─ └ ↓ 30-		Very Dense, Tan Fine SAND, Trace to Little Silt, Trace Gravel (SP/SP-SM)					
						End of Boring at 30 ft					
						Borehole Backfilled with Bentonite Chips and Soil Cuttings					
			W		R LE	EVEL OBSERVATIONS	GENERA		TES	5	
Whil Time Dept Dept	e Dril e After h to W h to C	ling Drilli ater ave in	$\underline{\nabla}$ 2 ng	23.5'	eprese	Jpon Completion of Drilling Start 3. 22 ↓ Cogger Drill Methors approximate boundary between Drill Methors approximate boundary between Drill Methors and the second start approximate boundary between Drill Methors and the second start approximate boundary between Drill Methors and the second start approximate boundary between Drill Methors and the second start approximate boundary between Drill Methors and the second start approximate boundary between Drill Methors and the second start approximate boundary between Drill Methors and the second start approximate boundary between Drill Methors and the second start approximate boundary between Drill Methors and the second start approximate boundary between Drill Methors and the second start approximate boundary between Drill Methors and the second start approximate boundary between Drill Methors and the second start approximate boundary between Drill Methors and the second start approximate boundary between Drill Methors and the second start approximate boundary between Drill Methors and the second start approximate boundary between Drill Methors approximate bounda	/8/23 End ADC Chief DB Edito od 2.25" 1	3/8/ F K r EL HSA; A	/23 D F .C Autoh:	Rig C	ME-55 2r

	G	СІ	nc		LOG OF TEST BORING Project Warner Park Rec Center Location Madison, WI 21 Perry Street Madison, WI	Boring No Surface E Job No. Sheet	o. levation C	B • n (ft) 2 2305 of	- 3 862 1-6 1	3
	SA	MPL	E			SOIL	PRC	PEF	RTIE	S
No.	T Rec P (in.)	Moist	N	Depth (ft)	and Remarks	qu (qa)	w	LL	PL	LOI
	E				$10 \pm in. TOPSOIL$	(tsi)				
1	6	М	14	⊥ └- ┝ ┍	Medium Dense to Dense, Brown Fine to Medium SAND, Some Silt and Gravel, Scattered Cobbles and Boulders (SM)					
2	6	М	32	└ └ ┼── 5─						
3	12	М	24		Occasional Pockets of Topsoil in Sample 1. Medium Dense, Tan Fine to Medium SAND, Trace to Little Silt, Little Gravel (SP/SP-SM)					
4	16	М	51	† └ ┝ └ └	Very Dense, Brown Fine to Medium SAND, Some Silt and Gravel, Scattered Cobbles and Boulders (SM)					
5	18	W	61		Very Dense, Tan Silty Fine SAND, Trace Gravel (SM)					
6	14	M/W	50/4"		Very Dense, Brown Fine to Medium SAND, Some Silt and Gravel, Scattered Cobbles and Boulders (SM)					
7	18	W	57		Very Dense, Brown Fine to Coarse SAND, Trace Silt, Some Gravel (SP)					
8	14	W	81	⊥ └ ↓ 30−	Very Dense, Brown Fine to Medium SAND, Little Silt, Some Gravel (SP-SM)					
					End of Boring at 30 ft Borehole Backfilled with Bentonite Chips and Soil Cuttings					
	1	1	W	ATER	LEVEL OBSERVATIONS	GENERA	LNC	TES	5	
While Time Dept Dept	e Dril After h to W h to C	ling Drilli ater ave in	∏ 1 ng	13.5'	Upon Completion of Drilling 30 min Driller A 	8/23 End DC Chief DB Edito d 2.25" I	3/8/ F Kl r EL HSA; A	23 D F C Autoh:	Rig Cl	ME-55 er

	G	СІ	nc		LOG OF TEST BORING Project Warner Park Rec Center Location Madison, WI 21 Perry Street, Madison, WI 53713 (608) 288-4100, FAX (608)	Boring No Surface E Job No. Sheet	o. levation C	B n (ft) 2 2305 of	- 4 866.7 1-6 1	2
	SA	MPL	E			SOIL	PRO	PEF	RTIE	S
No.	T Rec	Moist	N	Depth (ft)	and Remarks	qu (qa)	w	LL	PL	LOI
1	E ()			Г Г	$12 \pm in.$ TOPSOIL	(tsf)				
1	10	M	8	└── └── └── └──	Loose to Medium Dense, Brown Fine to Medium SAND, Some Silt and Gravel (SM)					
2	14	M	12	 └ └─ └─ └─						
3	16	M	25							
4	16	M	34	⊢ └ ⊢ └ └ └	Dense, Tan Fine to Coarse SAND, Little Silt, Some Gravel (SP-SM)	-				
5	5	M	50/5"		Dense to Very Dense, Brown Silty Fine SAND, Trace Gravel, Scattered Cobbles (SM)					
6	18	M/W	49							
7	12	W	31		Dense, Brown Fine to Coarse SAND, Trace to Little Silt, Little Gravel (SP/SP-SM)					
8	14	W	34	T	Occasional Silt Seams in Sample 7.					
				∔ 30— ⊢	End of Boring at 30 ft					
					Borehole Backfilled with Bentonite Chips and Soil Cuttings					
			VV/	AIER	LEVEL UBSERVATIONS	GENERA				
While Time Deptl Deptl	e Drill After h to W h to C	ling Drilli Vater ave in	∏ 1 ng	8.5'	Upon Completion of Drilling Start Driller Upon Completion of Drilling Toriller Driller Driller Drill Methem Drill Meth	3/2/23 End ADC Chief DB Edito nod 2.25" 1	3/2/ Kl r EL HSA; A	23 D F C Autoha	Rig Cl	ME-55 er

LOG OF TEST BORING

General Notes

DESCRIPTIVE SOIL CLASSIFICATION

Grain Size Terminology

Soil Fraction	Particle Size	U.S. Standard Sieve Size
Boulders	Larger than 12"	Larger than 12"
Cobbles	3" to 12"	3" to 12"
Gravel: Coarse	³ ⁄ ₄ " to 3"	³ ⁄4" to 3"
Fine	4.76 mm to 3/4"	#4 to ¾"
Sand: Coarse	2.00 mm to 4.76 mm	#10 to #4
Medium	0.42 to mm to 2.00 mm	#40 to #10
Fine	0.074 mm to 0.42 mm	#200 to #40
Silt	0.005 mm to 0.074 mm	Smaller than #200
Clay	Smaller than 0.005 mm	Smaller than #200

Plasticity characteristics differentiate between silt and clay.

General Terminology

CGC, Inc.

_			-
Re	lative	Den	sit

"N" Value

Physical Characteristics	Term	"N" Value
Color, moisture, grain shape, fineness, etc.	Very Loose	0 - 4
Major Constituents	Loose	4 - 10
Clay, silt, sand, gravel	Medium Den	se10 - 30
Structure	Dense	30 - 50
Laminated, varved, fibrous, stratified, cemented, fissured, etc.	Very Dense	Over 50
Geologic Origin		
Glacial, alluvial, eolian, residual, etc.		

Relative Proportions Of Cohesionless Soils

Proportional	Defining Range by	Term
Term	Percentage of Weight	Very Soft
		Soft
Trace	0% - 5%	Medium.
Little	5% - 12%	Stiff
Some	12% - 35%	Very Stiff
And	35% - 50%	Hard

Organic Content by Combustion Method

Soil Description	Loss on Ignition
Non Organic	Less than 4%
Organic Silt/Clay	4 – 12%
Sedimentary Peat	12% - 50%
Fibrous and Woody	Peat More than 50%

Term	q _u -tons/sq. ft
Very Soft	0.0 to 0.25
Soft	0.25 to 0.50
Medium	0.50 to 1.0
Stiff	1.0 to 2.0
Very Stiff	2.0 to 4.0
Hard	Over 4.0

Consistency

Plasticity

<u>Term</u>	Plastic Index
None to Slight	0 - 4
Slight	5 - 7
Medium	8 - 22
High to Very High	Over 22

The penetration resistance, N, is the summation of the number of blows required to effect two successive 6" penetrations of the 2" split-barrel sampler. The sampler is driven with a 140 lb. weight falling 30" and is seated to a depth of 6" before commencing the standard penetration test.

SYMBOLS

Drilling and Sampling

CS – Continuous Sampling RC - Rock Coring: Size AW, BW, NW, 2"W RQD - Rock Quality Designation **RB – Rock Bit/Roller Bit** FT – Fish Tail DC – Drove Casing C - Casing: Size 2 1/2", NW, 4", HW CW – Clear Water DM – Drilling Mud HSA – Hollow Stem Auger FA – Flight Auger HA – Hand Auger COA – Clean-Out Auger SS - 2" Dia. Split-Barrel Sample 2ST – 2" Dia. Thin-Walled Tube Sample 3ST – 3" Dia. Thin-Walled Tube Sample PT – 3" Dia. Piston Tube Sample AS – Auger Sample WS - Wash Sample PTS – Peat Sample PS – Pitcher Sample NR – No Recovery S – Sounding PMT – Borehole Pressuremeter Test VS – Vane Shear Test WPT – Water Pressure Test

Laboratory Tests

qa - Penetrometer Reading, tons/sq ft q_a – Unconfined Strength, tons/sq ft W – Moisture Content, % LL – Liquid Limit, % PL - Plastic Limit, % SL – Shrinkage Limit, % LI – Loss on Ignition D – Dry Unit Weight, Ibs/cu ft

- pH Measure of Soil Alkalinity or Acidity
- FS Free Swell, %

Water Level Measurement

abla- Water Level at Time Shown NW – No Water Encountered WD – While Drilling BCR – Before Casing Removal ACR – After Casing Removal CW - Cave and Wet CM – Caved and Moist

Note: Water level measurements shown on the boring logs represent conditions at the time indicated and may not reflect static levels, especially in cohesive soils.

CGC, Inc.

Madison - Milwaukee

UNIFIED SO		ASSIF	ICATION AND SYMBOL CHART
	(COARSE	E-GRAINED SOILS
(more thar	n 50%	of mater	ial is larger than No. 200 sieve size)
		Clean G	ravels (Less than 5% fines)
		GW	Well-graded gravels, gravel-sand mixtures, little or no fines
GRAVELS More than 50% of		GP	Poorly-graded gravels, gravel-sand mixtures, little or no fines
coarse fraction		Gravels	with fines (More than 12% fines)
sieve size		GM	Silty gravels, gravel-sand-silt mixtures
		GC	Clayey gravels, gravel-sand-clay mixtures
		Clean S	ands (Less than 5% fines)
		SW	Well-graded sands, gravelly sands, little or no fines
SANDS 50% or more of		SP	Poorly graded sands, gravelly sands, little or no fines
smaller than No. 4		Sands v	vith fines (More than 12% fines)
sieve size		SM	Silty sands, sand-silt mixtures
		SC	Clayey sands, sand-clay mixtures
(50% or m	ore of	FINE-0 material	GRAINED SOILS is smaller than No. 200 sieve size.)
SILTS AND		ML	Inorganic silts and very fine sands, rock flour, silty or clayey fine sands or clayey silts with slight plasticity
CLAYS Liquid limit less than 50%		CL	Inorganic clays of low to medium plasticity, gravelly clays, sandy clays, silty clays, lean clays
		OL	Organic silts and organic silty clays of low plasticity
SILTS AND		MH	Inorganic silts, micaceous or diatomaceous fine sandy or silty soils, elastic silts
CLAYS Liquid limit 50% or		СН	Inorganic clays of high plasticity, fat clays
greater	*****	OH	Organic clays of medium to high plasticity, organic silts
HIGHLY ORGANIC SOILS	24 24	PT	Peat and other highly organic soils

Unified Soil Classification System

LABORATORY CLASSIFICATION CRITERIA

C	θW	$C_u = \frac{D_{60}}{D_{10}}$ greater than 4; $C_C = \frac{D_{30}}{D_{10} \times D_{60}}$ between 1 and 3								
GP		Not meeting all gradation requirements for GW								
GM		Atterberg limts below "A" line or P.I. less than 4				Above "A" line with P.I. between 4 and 7 are borderline cases requiring use of dual symbols				
GC		Atterberg limts above "A" line or P.I. greater than 7								
SW $C_u = \frac{D_{60}}{D_{10}}$ greater than 4; $C_C = \frac{D_{30}}{D_{10} \times D_{60}}$ between 1 and 3										d 3
SP Not meeting all gradation requirements for GW										
SM		Atterberg limits below "A" line or P.I. less than 4				Limits plotting in shaded zone with P.I. between 4 and 7 are borderline cases requiring use of dual symbols				
SC		Atterberg limits above "A" line with P.I. greater than 7								
Determine percentages of sand and gravel from grain-size curve. Depending on percentage of fines (fraction smaller than No. 200 sieve size), coarse- grained soils are classified as follows:										
Less than 5 percent GW, GP, SW, SP More than 12 percent GM, GC, SM, SC 5 to 12 percent Borderline cases requiring dual symbols										
PLASTICITY CHART										
60										
3 (%) S	-			<u>,</u>			СН			
NDEX (PI	-							\nearrow	A LINE	:
STICITY I	-						\sim	P	l=0.73(L	20)
PLA:				CL		\checkmark				

(CL-ML)

ML&OL

LIQUID LIMIT (LL) (%)

APPENDIX C

DOCUMENT QUALIFICATIONS

APPENDIX C DOCUMENT QUALIFICATIONS

I. GENERAL RECOMMENDATIONS/LIMITATIONS

CGC, Inc. should be provided the opportunity for a general review of the final design and specifications to confirm that earthwork and foundation requirements have been properly interpreted in the design and specifications. CGC should be retained to provide soil engineering services during excavation and subgrade preparation. This will allow us to observe that construction proceeds in compliance with the design concepts, specifications and recommendations, and also will allow design changes to be made in the event that subsurface conditions differ from those anticipated prior to the start of construction. CGC does not assume responsibility for compliance with the recommendations in this report unless we are retained to provide construction testing and observation services. This report has been prepared in accordance with generally accepted soil and foundation engineering practices and no other warranties are expressed or implied. The opinions and recommendations submitted in this report are based on interpretation of the subsurface information revealed by the test borings indicated on the location plan. The report does not reflect potential variations in subsurface conditions between or beyond these borings. Therefore, variations in soil conditions can be expected between the boring locations and fluctuations of groundwater levels may occur with time. The nature and extent of the variations may not become evident until construction.

II. IMPORTANT INFORMATION ABOUT YOUR GEOTECHNICAL ENGINEERING REPORT

Subsurface problems are a principal cause of construction delays, cost overruns, claims, and disputes. While you cannot eliminate all such risks, you can manage them. The following information is provided to help.

Geotechnical engineers structure their services to meet the specific needs of their clients. A geotechnical engineering study conducted for a civil engineer may not fulfill the needs of a construction contractor or even another civil engineer. Because each geotechnical engineering study is unique, each geotechnical engineering report is unique, prepared *solely* for the client. *No one except you* should rely on your geotechnical engineering report without first conferring with the geotechnical engineer who prepared it. *And no one - not even you* - should apply the report for any purpose or project except the one originally contemplated.

READ THE FULL REPORT

Serious problems have occurred because those relying on a geotechnical engineering report did not read it all. Do not rely on an executive summary. Do not read selected elements only.

A GEOTECHNICAL ENGINEERING REPORT IS BASED ON A UNIQUE SET OF PROJECT-SPECIFIC FACTORS

Geotechnical engineers consider a number of unique, project-specific factors when establishing the scope of a study. Typical factors include: the client's goals, objectives, and risk management preferences; the general nature of the structure involved, its size, and configuration; the location of the structure on the site; and other planned or existing site improvements, such as access roads, parking lots, and underground utilities. Unless the geotechnical engineer who conducted the study specifically indicates otherwise, *do not rely on a geotechnical engineering report* that was:

- not prepared for you,
- not prepared for your project,
- not prepared for the specific site explored, or
- completed before important project changes were made.

Typical changes that can erode the reliability of an existing geotechnical report include those that affect:

- the function of the proposed structure, as when it's changed from a parking garage to an office building, or from a light industrial plant to a refrigerated warehouse,
- elevation, configuration, location, orientation, or weight of the proposed structure,
- composition of the design team, or project ownership.

As a general rule, *always* inform your geotechnical engineer of project changes - even minor ones - and request an assessment of their impact. *CGC cannot accept responsibility or liability for problems that occur because our reports do not consider developments of which we were not informed.*

SUBSURFACE CONDITIONS CAN CHANGE

A geotechnical engineering report is based on conditions that existed at the time the geotechnical engineer performed the study. *Do not rely on a geotechnical engineering report* whose adequacy may have been affected by: the passage of time; by man-made events, such as construction on or adjacent to the site; or by natural events, such as floods, earthquakes, or groundwater fluctuations. *Always* contact the geotechnical engineer before applying the report to determine if it is still reliable. A minor amount of additional testing or analysis could prevent major problems.

MOST GEOTECHNICAL FINDINGS ARE PROFESSIONAL OPINION

Site exploration identifies subsurface conditions only at those points where subsurface tests are conducted or samples are taken. Geotechnical engineers review field and laboratory data and then apply their professional judgement to render an opinion about subsurface conditions throughout the site. Actual subsurface conditions may differ - sometimes significantly - from those indicated in your report. Retaining the geotechnical engineer who developed your report to provide construction observation is the most effective method of managing the risks associated with unanticipated conditions.

A REPORT'S RECOMMENDATIONS ARE NOT FINAL

Do not over-rely on the confirmation-dependent recommendations included in your report. *Those confirmation-dependent recommendations are not final*, because geotechnical engineers develop them principally from judgement and opinion. Geotechnical engineers can finalize their recommendations *only* by observing actual subsurface conditions revealed during construction. *CGC cannot assume responsibility or liability for the report's confirmation-dependent recommendations if we do not perform the geotechnical-construction observation required to confirm the recommendations' applicability.*

A GEOTECHNICAL ENGINEERING REPORT IS SUBJECT TO MISINTERPRETATION

Other design team members' misinterpretation of geotechnical engineering reports has resulted in costly problems. Confront that risk by having your geotechnical engineer confer with appropriate members of the design team after submitting the report. Also retain your geotechnical engineer to review pertinent elements of the design team's plans and specifications. Constructors can also misinterpret a geotechnical engineering report. Confront that risk by having CGC participate in prebid and preconstruction conferences, and by providing geotechnical construction observation.

DO NOT REDRAW THE ENGINEER'S LOGS

Geotechnical engineers prepare final boring and testing logs based upon their interpretation of field logs and laboratory data. To prevent errors or omissions, the logs included in a geotechnical engineering report should *never* be redrawn for inclusion in architectural or other design drawings. Only photographic or electronic reproduction is acceptable, *but recognize that separating logs from the report can elevate risk.*

GIVE CONSTRUCTORS A COMPLETE REPORT AND GUIDANCE

Some owners and design professionals mistakenly believe they can make constructors liable for unanticipated subsurface conditions by limiting what they provide for bid preparation. To help prevent costly problems, give constructors the complete geotechnical engineering report. but preface it with a clearly written letter of transmittal. In that letter, advise constructors that the report was not prepared for purposes of bid development and that the report's accuracy is limited; encourage them to confer with the geotechnical engineer who prepared the report (a modest fee may be required) and/or to conduct additional study to obtain the specific types of information they need or prefer. A prebid conference can also be valuable. Be sure constructors have sufficient time to perform additional study. Only then might you be in a position to give constructors the best information available to you, while requiring them to at least share some of the financial responsibilities stemming from unanticipated conditions.

READ RESPONSIBILITY PROVISIONS CLOSELY

Some clients, design professionals, and constructors do not recognize that geotechnical engineering is far less exact than other engineering disciplines. This lack of understanding has created unrealistic expectations that have led to disappointments, claims, and disputes. To help reduce the risk of such outcomes, geotechnical engineers commonly include a variety of explanatory provisions in their reports. Sometimes labeled "limitations," many of these provisions indicate where geotechnical engineer's responsibilities begin and end, to help others recognize their own responsibilities and risks. *Read these provisions closely*. Ask questions. Your geotechnical engineer should respond fully and frankly.

ENVIRONMENTAL CONCERNS ARE NOT COVERED

The equipment, techniques, and personnel used to perform an *environmental* study differ significantly from those used to perform a *geotechnical* study. For that reason, a geotechnical engineering report does not usually relate any environmental findings, conclusions, or recommendations; e.g., about the likelihood of encountering underground storage tanks or regulated contaminants. *Unanticipated environmental problems have led to numerous project failures.* If you have not yet obtained your own environmental information, ask your geotechnical consultant for risk management guidance. *Do not rely on an environmental report prepared for someone else.*

OBTAIN PROFESSIONAL ASSISTANCE TO DEAL WITH MOLD

Diverse strategies can be applied during building design, construction, operation, and maintenance to prevent significant amounts of mold from growing on indoor surfaces. To be effective, all such strategies should be devised for the express purpose of mold prevention, integrated into a comprehensive plan, and executed with diligent oversight by a professional mold prevention consultant. Because just a small amount of water or moisture can lead to the development of severe mold infestations, many mold prevention strategies focus on keeping building surfaces dry. While groundwater, water infiltration, and similar issues may have been addressed as part of the geotechnical engineering study whose findings are conveyed in this report, the geotechnical engineer in charge of this project is not a mold prevention consultant; none of the services performed in connection with the geotechnical engineer's study were designed or conducted for the purpose of mold Proper implementation of the recommendations prevention. conveyed in this report will not of itself be sufficient to prevent mold from growing in or on the structure involved.

RELY ON YOUR GEOTECHNICAL ENGINEER FOR ADDITIONAL ASSISTANCE

Membership in the Geotechnical Business Council (GBC) of Geoprofessional Business Association exposes geotechnical engineers to a wide array of risk confrontation techniques that can be of genuine benefit for everyone involved with a construction project. Confer with CGC, a member of GBC, for more information.

Modified and reprinted with permission from:

Geotechnical Business Council of the Geoprofessional Business Association 8811 Colesville Road, Suite G 106 Silver Spring, MD 20910

APPENDIX D

RECOMMENDED COMPACTED FILL SPECIFICATIONS

APPENDIX D

CGC, INC.

RECOMMENDED COMPACTED FILL SPECIFICATIONS

General Fill Materials

Proposed fill shall contain no vegetation, roots, topsoil, peat, ash, wood or any other non-soil material which by decomposition might cause settlement. Also, fill shall never be placed while frozen or on frozen surfaces. Rock, stone or broken concrete greater than 6 in. in the largest dimension shall not be placed within 10 ft of the building area. Fill used greater than 10 ft beyond the building limits shall not contain rock, boulders or concrete pieces greater than a 2 sq ft area and shall not be placed within the final 2 ft of finish subgrade or in designated utility construction areas. Fill containing rock, boulders or concrete pieces should include sufficient finer material to fill voids among the larger fragments.

Special Fill Materials

In certain cases, special fill materials may be required for specific purposes, such as stabilizing subgrades, backfilling undercut excavations or filling behind retaining walls. For reference, WisDOT gradation specifications for various types of granular fill are attached in Table 1.

Placement Method

The approved fill shall be placed, spread and leveled in layers generally not exceeding 10 in. in thickness before compaction. The fill shall be placed at moisture content capable of achieving the desired compaction level. For clay soils or granular soils containing an appreciable amount of cohesive fines, moisture conditioning will likely be required.

It is the Contractor's responsibility to provide all necessary compaction equipment and other grading equipment that may be required to attain the specified compaction. Hand-guided vibratory or tamping compactors will be required whenever fill is placed adjacent to walls, footings, columns or in confined areas.

Compaction Specifications

Maximum dry density and optimum moisture content of the fill soil shall be determined in accordance with modified Proctor methods (ASTM D1557). The recommended field compaction as a percentage of the maximum dry density is shown in Table 2. Note that these compaction guidelines would generally not apply to coarse gravel/stone fill. Instead, a method specification would apply (e.g., compact in thin lifts with a vibratory compactor until no further consolidation is evident).

Testing Procedures

Representative samples of proposed fill shall be submitted to CGC, Inc. for optimum moisture-maximum density determination (ASTM D1557) prior to the start of fill placement. The sample size should be approximately 50 lb.

CGC, Inc. shall be retained to perform field density tests to determine the level of compaction being achieved in the fill. The tests shall generally be conducted on each lift at the beginning of fill placement and at a frequency mutually agreed upon by the project team for the remainder of the project.
Table 1Gradation of Special Fill Materials

Material	WisDOT Section 311	WisDOT Section 312	W	isDOT Section 3	05	WisDOT S	Section 209	WisDOT Section 210
Material	Breaker Run	Select Crushed Material	3-in. Dense Graded Base	1 1/4-in. Dense Graded Base	3/4-in. Dense Graded Base	Grade 1 Granular Backfill	Grade 2 Granular Backfill	Structure Backfill
Sieve Size				Percent Pa	ssing by Weigh	ıt		
6 in.	100							
5 in.		90-100						
3 in.			90-100					100
1 1/2 in.		20-50	60-85					
1 1/4 in.				95-100				
1 in.					100			
3/4 in.			40-65	70-93	95-100			
3/8 in.				42-80	50-90			
No. 4			15-40	25-63	35-70	100 (2)	100 (2)	25-100
No. 10		0-10	10-30	16-48	15-55			
No. 40			5-20	8-28	10-35	75 (2)		
No. 100						15 (2)	30 (2)	
No. 200			2-12	2-12	5-15	8 (2)	15 (2)	15 (2)

Notes:

1. Reference: Wisconsin Department of Transportation Standard Specifications for Highway and Structure Construction.

2. Percentage applies to the material passing the No. 4 sieve, not the entire sample.

3. Per WisDOT specifications, both breaker run and select crushed material can include concrete that is 'substantially free of steel, building materials and other deleterious material'.

Table 2Compaction Guidelines

	I	Percent Compaction (1)
Area	Clay/Silt	Sand/Gravel
Within 10 ft of building lines		
Footing bearing soils	93 - 95	95
Under floors, steps and walks		
- Lightly loaded floor slab	90	90
- Heavily loaded floor slab and thicker fill zones	92	95
Beyond 10 ft of building lines		
Under walks and pavements		
- Less than 2 ft below subgrade	92	95
- Greater than 2 ft below subgrade	90	90
Landscaping	85	90

Notes:

1. Based on Modified Proctor Dry Density (ASTM D 1557)

APPENDIX E

WISCONSIN DEPARTMENT OF SAFETY & PROFESSIONAL SERVICES SOIL AND SITE EVALUATION – STORM FORM PARTICLE SIZE DISTRIBUTION TEST REPORT

1002-CPS-23

ST	DEPAI	TMEN	TON	6
VISCON	Ds		5.00	TET
AN	7	PS		3)
Pr	OFESSI	ONAL	SERVI	

Address

129 Milky Way, Madison, WI 53718

Attachment 2:

Division of Industry Services P.O. Box 2658 Madison, Wisconsin 53701

SOIL AND SITE EVALUATION - STORM

In accordance with SPS 382.365, 385, Wis. Adm. Code, and WDNR Standard 1002

200	SIONAL			, -	,			Page	1	of 1
Attach a	complete s	site plan on paper not	less than 8 1/2 x 11 inche	s in size.	Plan must include	e, but not limited	County		Dane	;
to: vert	ical and ho	arrow, a	nt (BM), direction and pe and BM referenced to nea	rcent of s arest road	lope, scale or dim I	ensions, north	Parcel I.D	. 251	/0809-3	861-0096-8
Pers	onal inform	F Antion you provide ma	Please print all informat	ion nurnoses	Privacy Law s	15 ()4(1)(m)]	Reviewed by Date:	<u>/:</u>		
1 613		lation you provide me	ly be used for secondary	puiposes		13.04(1)(11)]	Date.			
Property (Owner	City Warne	of Madison Parks er Playfield Stadium		Property Location Govt. Lot N	n W 1/4 NE 1/4	S	36 T 8	N R	9 E
Property (Owner's Ma	ail Address 330 E Lakesi	de St		Lot # Block#	Subd.	Name or CSI	VI #		
City Mad	dison	State Zip Code WI 53	Phone Numbe	er	XCity	Village To Madison	wn No	earest Ro	ad Jorthport [Drive
Drainage	area		sq ft acr	es	Hydraulic App	blication Test Met	hod Soil M Date	loisture of soil bor	ings:	
Test site s Bic Re	suitable for pretention; suse;	(check all that apply)	: Site not su isperal System; Other	uitable;	X Morpholog Double Rin Other: (sp	gical Evaluation ng Infiltrometer ecify)	USDA		VETS Val ry = 1; ormal = 2; /et = 3.	ue:
TP-1 #O	BS.	X Pit Boring	Ground surface eleva	ation	866.5 ft.	Elevation of li	miting factor	<856	6.5 ft.	
Horizon	Approx. Depth in.	Dominant Color Munsell	Redox Description Qu. Sz. Cont. Color	Textur	e Structure Gr. Sz. Sh.	Consistence	Boundary	% Rock Frags.	% Fines (P200)	Hydraulic App Rate Inches/Hr
1	0-15	10YR 4/1	none	SiL	2mgr	mfr	CS	<5		0.13
2	15-43	10YR 4/4	none	SL, SiC Seams	L 1fsbk	mfr	gs	<5		0.04-0.50 ⁽¹⁾
3	43-120	10YR 5/4	none	GRSL	. Osg	ml	n/a	15 ⁽²⁾	23 (2)	0.50
Comment ⁽¹⁾ Vertical granular c consistent ⁽²⁾ Based o	<u>s:</u> Ground I infiltration leposit to b t with the d on a sampl	water was not encour rate is expected to bu reak up the lower-per eisgn infiltration rate. e taken at about 7 ft t	tered during or upon the e controlled by occasiona meability seams; sample pelow grade.	completion I <i>silty cla</i> Is should	on of excavating. y <i>loam seams</i> , bu be collected durin	t can potentially b g construction to	e improved b check that the	y excavati e texture o	ing/turning f the blen	g over the ded soil is
	<u>Overall</u>	Site Comments: See	Comments above and S (CGC Project)	Stormwate No. C230	er Infiltration Poter 51-6; dated Marcl	ntial section in our h 24, 2023).	Geotechnica	l Explorat	tion Repor	ť
Name (Ple	ease Print)	Tim F.	Gassenheimer	Signature	° Carr	**		Credentia	al Numbe SP-01190	r 00004

Date Evaluation Conducted

February 14, 2023

Telephone Number (608) 288-4100



Tested By: JFS

Checked By: KJS

1

SECTION 02 41 19

2		SELECTIVE DEMOLITION
3	PART 1 - (GENERAL
4	1.1	SUMMARY
5	Α.	Section Includes (all items might not apply – see below and drawings):
6		 Demolition and removal of selected portions of building or structure.
7		2. Demolition and removal of selected site elements.
8		3. Salvage of existing items to be reused or recycled.
9	1.2	MATERIALS OWNERSHIP
10	Α.	Unless otherwise indicated, demolition waste becomes property of Contractor.
11	В.	Historic items, relics, antiques, and similar objects including, but not limited to, cornerstones and their contents,
12		commemorative plaques and tablets, and other items of interest or value to Owner that may be uncovered during
13		demolition remain the property of Owner.
14		1. Carefully salvage in a manner to prevent damage and promptly return to Owner.
15	1.3	PREINSTALLATION MEETINGS
16	Α.	Predemolition Conference: Conduct conference at Project site.
17	1.4	INFORMATIONAL SUBMITTALS
18	Α.	Engineering Survey: Submit engineering survey of condition of building.
19	В.	Proposed Protection Measures: Submit report, including Drawings, that indicates the measures proposed for
20		protecting individuals and property, for environmental protection, for dust control and, for noise control. Indicate
21		proposed locations and construction of barriers.
22	С.	Schedule of selective demolition activities with starting and ending dates for each activity.
23	D.	Predemolition photographs or video.
24	E.	Statement of Refrigerant Recovery if any noted to be removed: Signed by refrigerant recovery technician.
25	1.5	CLOSEOUT SUBMITTALS
26	Α.	Inventory of items that have been removed and salvaged.
27	1.6	QUALITY ASSURANCE
28	Α.	Refrigerant Recovery Technician Qualifications: Certified by an EPA-approved certification program.
29	1.7	FIELD CONDITIONS
30	Α.	Owner will occupy portions of building immediately adjacent to selective demolition area. Conduct selective
31		demolition so Owner's operations will not be disrupted.
32	В.	Conditions existing at time of inspection for bidding purpose will be maintained by Owner as far as practical.
33		1. Before selective demolition, Owner will remove the following items:
34		a. All furniture, fixtures, and equipment (i.e., items not mounted or anchored to building) within
35		project work area.
36	С.	Notify Architect of discrepancies between existing conditions and Drawings before proceeding with selective
37		demolition.
38	D.	Hazardous Materials: It is not expected that hazardous materials will be encountered in the Work because the
39		building was constructed in 1999.
40		1. If suspected hazardous materials are encountered, do not disturb; immediately notify Architect and Owner.
41		Hazardous materials will be removed by Owner under a separate contract.
42	Ε.	Storage or sale of removed items or materials on-site is not permitted.
43	F.	Utility Service: Maintain existing utilities indicated to remain in service and protect them against damage during
44		selective demolition operations.
45		1. Maintain fire-protection facilities in service during selective demolition operations.
46	G.	Arrange selective demolition schedule so as not to interfere with Owner's operations.
47	1.8	WARRANTY
48	Α.	Existing Warranties: Remove, replace, patch, and repair materials and surfaces cut or damaged during selective
49		demolition, by methods and with materials and using approved contractors so as not to void existing warranties.
50	PART 2 - I	PRODUCTS
51	2.1	PERFORMANCE REQUIREMENTS
52	Α.	Regulatory Requirements: Comply with governing EPA notification regulations before beginning selective
53		demolition. Comply with hauling and disposal regulations of authorities having jurisdiction.
54	В.	Standards: Comply with ANSI/ASSP A10.6 and NFPA 241.
55	PART 3 - 1	EXECUTION
56	3.1	EXAMINATION
57	Α.	Verify that utilities have been disconnected and capped before starting selective demolition operations.

May 16	i, 2024
В.	Perform an engineering survey of condition of building to determine whether removing any element might result in structural deficiency or unplanned collapse of any portion of structure or adjacent structures during selective building demolition operations.
	1. Engage a professional engineer to perform the survey if beyond capability of General Contractor or
	Demolition Contractor. Cost of Engineering Services to be included in bid.
C.	Inventory and record the condition of items to be removed and salvaged.
3.2	PREPARATION
Α.	Refrigerant: Before starting demolition, remove refrigerant from mechanical equipment according to 40 CFR 82 and
	regulations of authorities having jurisdiction.
3.3	UTILITY SERVICES AND MECHANICAL/ELECTRICAL SYSTEMS
Α.	Existing Services/Systems to Remain: Maintain services/systems indicated to remain and protect them against
~ -	damage.
3.4	PROTECTION
А.	demose to adjacent buildings and facilities to remain
Р	damage to adjacent buildings and facilities to remain.
в.	stability and prevent movement settlement or collapse of construction and finishes to remain, and to prevent
	unevpected or uncontrolled movement or collapse of construction being demoliched
C	Remove temporary barricades and protections where bazards no longer exist
35	SELECTIVE DEMOLITION
A.	General: Demolish and remove existing construction only to the extent required by new construction and as
	indicated. Use methods required to complete the Work within limitations of governing regulations and as follows:
	1. Neatly cut openings and holes plumb, square, and true to dimensions required. Use cutting methods least
	likely to damage construction to remain or adjoining construction. Use hand tools or small power tools
	designed for sawing or grinding, not hammering and chopping. Temporarily cover openings to remain.
	2. Cut or drill from the exposed or finished side into concealed surfaces to avoid marring existing finished
	surfaces.
	3. Do not use cutting torches until work area is cleared of flammable materials. At concealed spaces, such as
	duct and pipe interiors, verify condition and contents of hidden space before starting flame-cutting
	operations. Maintain portable fire-suppression devices during flame-cutting operations.
	4. Maintain fire watch during and for at least two hours after flame-cutting operations.
	5. Locate selective demolition equipment and remove debris and materials so as not to impose excessive loads
	on supporting walls, floors, or framing.
	Dispose of demolished items and materials promptly.
В.	Site Access and Temporary Controls: Conduct selective demolition and debris-removal operations to ensure
	minimum interference with roads, streets, walks, walkways, and other adjacent occupied and used facilities.
С.	Removed and Salvaged Items:
	1. Clean salvaged items.
	2. Pack or crate items after cleaning. Identify contents of containers.
	3. Store items in a secure area until delivery to Owner.
	4. I ransport items to Owner's storage area on-site unless directed otherwise in drawings.
P	5. Protect items from damage during transport and storage.
D.	Removed and Remistaned items to functional condition adequate for intended rays
	Clean and repair items to functional condition adequate for intended reuse. Dark or create items after cleaning and repairing. Identify contents of containers
	2. Fack of crate items after cleaning and repairing, identify contents of containers.
	A Reinstall items in locations indicated Comply with installation requirements for new materials and
	equipment Provide connections supports and miscellaneous materials necessary to make item functional
	for use indicated
E.	Existing Items to Remain: Protect construction indicated to remain against damage and soiling during selective
	demolition. When permitted by Architect, items may be removed to a suitable, protected storage location during
	selective demolition and cleaned and reinstalled in their original locations after selective demolition operations are
	complete.
F.	The following items are to be salvaged or reinstalled:
	1. See drawings.
3.6	CLEANING
Α.	Remove demolition waste materials from Project site and dispose of them in an EPA-approved construction and
	demolition waste landfill acceptable to authorities having jurisdiction.

	iviay 16,	2024
1		1. Do not allow demolished materials to accumulate on-site.
2		2. Remove and transport debris in a manner that will prevent spillage on adjacent surfaces and areas.
3		3. Remove debris from elevated portions of building by chute, hoist, or other device that will convey debris to
4		grade level in a controlled descent.
5		4. Comply with requirements specified in Section 01 74 19 "Construction Waste Management and Disposal."
6	В.	Burning: Do not burn demolished materials.
7	С.	Clean adjacent structures and improvements of dust, dirt, and debris caused by selective demolition operations.
8		Return adjacent areas to condition existing before selective demolition operations began.
9		END OF SECTION

Page Intentionally Left Blank

											c	СС	oı	N	ICI	RE	ET	٢E	S F	5E :0	:C	R	N	(1)	10 11	N 10	0 3	3 A	1 .N	1	0 C))	0 A	0	20	CE	ES	55	60	DF	RI	E	S																													
ΡΔRT 1 _	GENERAL																																																																							
1 1	RELATE			 MFI	JTS			•••		•••				•••	••••	••••		••••		•••	••	•	•		•••	•••		•••	••	••	•••	•••	•••	•••	•••	•••	••	••		•••	•••	•••	•••		••••	•••	••••	•••	•••	•••	••••	••••	•••			•••	••••	•••	••	••	••	•••	•••		•••	••	•	•••	•••	•••	•••	•••
1 2	SUMMA	ARY	001					•••								••••		••••		•••	••				•••	•••	••	••	••	•••		••	••	•••	•••		•••	•••	•••	•••	•••	•••	•••			•••	•••		•••	••	••••	•••				•••	•••	•••	••	••		••	•••		•••	•			•••		•••	•••
13	DEFINIT		 S											•••		••••		••••		•••	••				•••	•••		••	••	••	•••	••	••	•••	•••		•••	•••	•••	•••	•••	•••	•••			•••				•••	•••	•••	•••			•••	•••		••	••	••	••	•••		•••				•••	•••	•••	•••
1.3.	PRFINST		ΔΤΙC	 ז מכ	ЛFF	TIN	 NG	 GS	 ς	••••	•••		•••	•••	••••	••••		••••		•••	••				•••	•••		•••	••	••	•••		••	•••	••		••	•••		•••	•••	•••	•••		••••	•••			•••	•••	••••	••••	•••		•••	•••	•••	•••	••	••	••	••	•••		•••		•		•••	•••	•••	••
15	ACTION	SUB	мп	ΓΤΑΙ	S		10	55	۶.		•••		•••	•••		••••		••••		•••	••				•••	•••	••	•••		••	•••		••	•••	•••		•••	•••	•••	•••	•••	•••	•••		••••	•••	••••		•••	•••	•••	•••				•••	•••	•••	••	••		•••	•••		•••	••			•••	•••	•••	••
1.5.	INFORM		JNA		JRIV	/IT7	 ТА	Δ1	 ו ר	 ς			•••	•••		••••		••••		•••	••				•••	•••	••	•••	••	••	•••	••	••	•••	••		•••	•••	•••	•••	•••	•••	•••			•••			•••	•••	••••	••••	•••		•••	•••	•••	•••	••	••		••	•••		•••	•			•••	•••	•••	••
17	OUALIT	Y ASS	SUR		`F		.,			0.	••••			•••		••••		••••		•••	••				•••	•••	••	••	••	•••	•••		••	•••			•••	•••	•••	•••	•••	•••	•••			•••	•••		•••	••		Fı	r	۰۰۰ ۲	•	B	80	0	 sk	٢n	 n:	 aı	rl	k	r	 16			d	ei	fi	in
1.8		iγiss ≀γ st	FOR	AGI	 	ND) Н	-ι	۰ ۸۱		םו	וכ	 11		 G	••••		••••		•••	••				•••	•••	••	••	••	•••		••	••	•••	••		•••	•••	•••	•••	•••	•••	•••			•••			•••	•••	•••				•									Ì						<u> </u>	•••	
PART	2 – PRODU	CTS	Ön		., ,																					•••															•••		•••													•••										•••						
2.1	PFRFOR	MAN	JCF	RFC)UIF	₹FN	MF	F٨	N7	IT:	rs.	5																																																												
2.2	FORM-F	ACIN			FRI	ALS	S						•••																												•••		•••																							•••						
2.3	INSULA	TING	CO	NCF	ETE	EFC	OR	R٨	M	٨S	s.																																									Eı	rc	or	٠ı	B	ßc	0)k	'n	n	aı	rl	k	r	10	D	t i	d	ef	fi	'n
2.4	WATER	STOP	S																																																																					
2.5	RELATE	D MA	ATE	RIAI	S																																																																			
PART	3 – EXECUT	10N																																																																						
3.1	INSTALL	ATIO)N (DF F	ORN	лw	NC	ЭF	Rk	к	<u>.</u>																																																													
3.2	INSTALL	ATIO)N (DF E	MB	EDI	DE	EC	D) ľ	IT	TE	ΞN	MS	S.																																																									
3.3	INSTALL	ATIO)N (DF V	VAT	FRS	ST	ΓC	_ DF	P	s	5																																																												
3.4	INSTALL	ATIO)N (DF II	VSU	LA7	TIL	IN ⁽	١G	G	i C	СС	0	N	CF	RE	ET	Έ	E!	0)F	R	v	15	5.																											Eı	rc	or	·I	E	Sc	0)k	'n	n	aı	rl	k	r	10	D1	t i	d	ef	fi	'n
3.5	REMOV	ING A	ANE	D RE	USII	NG	5 F(-0	DR	R۱	١N	15	S.																																																											
3.6	SHORIN	G AN	JD F	RESH	IOR	ING	G١	IN	NS	S	sΤ.	ΓA	ΑL!	L	ΑT	τιс	or	N																																		Eı	rc	or	·I	В	8c	0	γk	m	n	aı	rl	k	r	10	Dt	t i	d	eí	fi	'n
37	FIFLD O		TY (TRC	וו				-																																										Fi	r	٦r	÷	B	30	0	k	'n	n	aı	rl	k	r	10	1	•	d	e	fi	in
Α.	Drawing Specifica	s and ation	d ge Ser	ene ctio	ิal p าร, a)ro∖ 3pp	vis ply	isio y t	io to	on o	ns • t	s th	o is	of s S	th Se	he ect	e C tic	Co on	วท ำ.	iti	ra	a	C	t,	i	n	cl	u	d	li	n	١Ę	3	G	ie	er	1e	er	al	;	ar	n	d	S	Jp	p	le	n	ne	er	ita	ar	/ (Co)r	n	di	ti	0	'n	iS	ć	16	n	d		D	i١	/i:	si	0	'n
1.2. SUN	/IMARY																																																																							
Α.	Section	Inclu	des	::						_																																																														
	1.	Form	ı-fao	cing	ma	teri	rial	al f	fc	0	۶r	́С	ca	ast	t-i	in-	-p	งla	ac	e	: (С	0	n	C	re	:te	э.	•																																											
1.3. DEF	INITIONS				_																						_	_																																												
А.	Form-Fa	cing	Ma	ter	al: I	len	mp	ро	or	ra	ar	ry	y :	st	tru	uc	ctu	ur	e	. C	O	r	r	n	0	10	11	í C)	٢	t	tr	٦e	e	S	u	p	p	0	rt	t (0	t	СС	on	cr	e	te	9	w	h	le	t	he	e	C	CC	n	iC	r	et	te	9	13	S	S	66	et	tı	n	g	; ;
_	gaining	suttici	cien	it sti	eng	;th i	to	o t	b	e Je	es	se	el	lt-	-Sl	up	pp	00	rt	tir	n	g																																																		
В.	Formwo	rk: 11 e, as v	he t wel	tota Il as	l sys sup	ster por	.m orti	۱٥ in:	ot ng	t s g	sı n	su me	ip	m	or าb	rt er	o1 rs,	†† ;,	fre ha	es ar	s C	h ∦∖	v	/ a	p re	la e,	a	n	d IC) k	c r	:0 ni	e e		es	e ss	te a	e, rγ	и у	no b	ra ra	u ac	d cii	ng	g ' ;.	th	e	r	n	ol	d	0	· s	h	e	a	tł	าเ	n	g	t	h	a	ıt	. (20	וכ	n	ta	C	ts	S
1.4. PRE	INSTALLAT		ME	ETIN	IGS																																																																			
Α.	Preinsta	llatio	on C	Conf	erer	nce	e: (Сс	o	on	nd	dι	uc	ct	: C(or	nf	er	re	en	10	ce	è	a	t	Ρ	rc	οj	e	c	ct	t	si	ite	e.																																					
	1.	Revie	ew t	the	ollo) wi	ing	g:	:																																																															
		a.	(Con	stru	ctic	on	n,	, n	m	nc	٥v	ve	en	ne	en	۱t,	, c	:0	n	ıt	r	a	ct	ti	0	n,	ā	31	n	IC	t	is	50	bla	a	ti	o	n	j¢	зi	in	Its	5																												
		b.	I	Forr	ns a	nd	l fc	or	rn	m	n-i	-re	er	m	יסו	va	al	lir	m	it	ta	at	i	b	n	s.																																														
		с.	/	Anc	10r I	rod	d a	an	nc	d	l a	ar	nc	ch	າວ	ra	age	;e	d	e	؛ ۷	/i	C	e	ii	ns	ta	зl	li	a	۱t	:ic	01	n	t	0	le	er	а	n	ce	es	s.																													
1.5. ACT	ION SUBM	ITTAL	LS																																																																					
Α.	Product	Data	a: Fo	or e	ach (of t	th	۱e	٩f	fc	ol	llo	٥١	w	/in	۱g:																																																								
	1.	Expos	sed	l sur	face	e fo	orn	m	1-f	-fa	iac	ıci	in	١g	; m	na	at€	er	ria	эl,																																																				
	2.	Conce	eal	ed s	urfa	ice	e fc	or	rn	m	n-f	-fa	ac	cir	ng	g n	ma	at	ter	ri	iə	l																																																		
	3.	Form	۱ tie	es.																																																																				
	4.	Wate	erst	ops																																																																				
	5.	Form	ı-re'	leas	e ag	zen	nt.																																																																	

1	В.	Shop Drawings: Prepared by, and signed and sealed by, a qualified professional engineer responsible for their
2		preparation, detailing fabrication, assembly, and support of forms.
3		1. For exposed vertical concrete walls, indicate dimensions and form tie locations.
4		2. Indicate dimension and locations of construction and movement joints required to construct the structure
5		in accordance with ACI 301.
6		a. Location of construction joints is subject to approval of the Architect.
7		3. Indicate proposed schedule and sequence of stripping of forms.
8	C.	Samples:
9		1. For waterstops.
10	1.6 INFO	NRMATIONAL SUBMITTALS
12	1.0. INIX	Field quality-control reports.
13	В.	Minutes of preinstallation conference.
14		
15		
16	1.7. DEL	VERY, STORAGE, AND HANDLING
17	Α.	Waterstops: Store waterstops under cover to protect from moisture, sunlight, dirt, oil, and other contaminants.
18	DADT 2	
20	<u>FANI 2 -</u>	
21	2.1 PER	FORMANCE REQUIREMENTS
22	А.	Concrete Formwork: Design, engineer, erect, shore, brace, and maintain formwork in accordance with ACI 301, to
23		support vertical, lateral, static, and dynamic loads, and construction loads that might be applied, until structure can
24		support such loads, so that resulting concrete conforms to the required shapes, lines, and dimensions.
25		1. Design wood panel forms in accordance with APA's "Concrete Forming Design/Construction Guide."
26		2. Design formwork to limit deflection of form-facing material to 1/240 of center-to-center spacing of
27		supports.
28		a. For architectural concrete specified in Section 033300 "Architectural Concrete," limit deflection of
29		form-facing material, studs, and walers to 0.0025 times their respective clear spans (L/400).
30	2.2 500	
33	2.2 FUR	As-Cast Surface Form-Facing Material:
32	Α.	1 Provide continuous true and smooth concrete surfaces
34		 Fronde continuous, and, and smooth contracts surfaces. Furnish in largest practicable sizes to minimize number of joints.
35		3. Acceptable Materials: As required to comply with Surface Finish designations specified in Section 033000
36		"Cast-In-Place Concrete, and as follows:
37		a. Plywood, metal, or other approved panel materials.
38		b. Exterior-grade plywood panels, suitable for concrete forms, complying with DOC PS 1, and as
39		follows:
40		1) APA Structural 1 Plyform, B-B or better; mill oiled and edge sealed.
41	В.	Concealed Surface Form-Facing Material: Lumber, plywood, metal, plastic, or another approved material.
42		1. Provide lumber dressed on at least two edges and one side for tight fit.
43	22 14/4-	
44 15	2.5 VVA	Elevible Rubber Waterstons: U.S. Army Corns of Engineers CRD-C 513 with factory-installed metal evelets for
46	Α.	embedding in concrete to prevent passage of fluids through joints with factory fabricated corners intersections
47		and directional changes.
48		1. Profile: Ribbed with center bulb.
49		2. Dimensions: 4 inches by 3/16 inch thick; nontapered.
50	В.	Self-Expanding Rubber Strip Waterstops: Manufactured rectangular or trapezoidal strip, bentonite-free hydrophilic
51		polymer-modified chloroprene rubber, for adhesive bonding to concrete, 3/8 by 3/4 inch.
52		
53	2.4 REL/	ATED MATERIALS
54	С.	Dovetail Anchor Slots: Hot-dip galvanized-steel sheet, not less than 0.034 inch thick, with bent tab anchors.
55	~	I emporarily fill or cover face opening of slots to prevent intrusion of concrete or debris.
50 57	ש. ר	Chamter Strips: Wood, metal, PVC, or rubber strips, 3/4 by 3/4 inch, minimum.
5/ 50	E.	ronn-release Agent: commercially formulated form-release agent that does not bond with, stain, or adversely
20		anect concrete surfaces and does not impail subsequent frediments of concrete surfaces.

1		1. Formulate form-release agent with rust inhibitor for steel form-facing materials.
2		2. Form release agent for form liners shall be acceptable to form liner manufacturer.
3	F.	Form Ties: Factory-fabricated, removable or snap-off, glass-fiber-reinforced plastic or metal form ties designed to
4		resist lateral pressure of fresh concrete on forms and to prevent spalling of concrete on removal.
5		1. Furnish units that leave no corrodible metal closer than 1 inch to the plane of exposed concrete surface.
6		2. Furnish ties that, when removed, leave holes no larger than 1 inch in diameter in concrete surface.
7		
8	<u> PART 3 – </u>	EXECUTION
9 10	3.1 INST	ALLATION OF FORMWORK
11	Α.	Comply with ACI 301.
12	В.	Construct formwork, so concrete members and structures are of size, shape, alignment, elevation, and position
13		indicated, within tolerance limits of ACI 117 and to comply with the Surface Finish designations specified in Section
14		033000 "Cast-In-Place Concrete" for as-cast finishes.
15	C.	Limit concrete surface irregularities as follows:
16		1. Surface Finish-2.0: ACI 117 Class B, 1/4 inch.
17	D.	Construct forms tight enough to prevent loss of concrete mortar.
18		1. Minimize joints.
19		2. Exposed Concrete: Symmetrically align joints in forms.
20	E.	Construct removable forms for easy removal without hammering or prying against concrete surfaces.
21		1. Provide crush or wrecking plates where stripping may damage cast-concrete surfaces.
22		2. Provide top forms for inclined surfaces steeper than 1.5 horizontal to 1 vertical.
23		 Install keyways, reglets, recesses, and other accessories, for easy removal.
24	F.	Do not use rust-stained, steel, form-facing material.
25	G.	Set edge forms, bulkheads, and intermediate screed strips for slabs to achieve required elevations and slopes in
26		finished concrete surfaces.
27		1. Provide and secure units to support screed strips
28		 Use strike-off templates or compacting-type screeds.
29	H.	Provide temporary openings for cleanouts and inspection ports where interior area of formwork is inaccessible.
30		1 Close openings with papels tightly fitted to forms and securely braced to prevent loss of concrete mortar
31		 Locate temporary openings in forms at inconsnicuous locations
32	1	2. Change exterior corners and edges of normanently exposed concrete
32	1.	At construction joints, overlap forms onto previously placed concrete not less than 12 inches
34	к 1.	Construction and Movement Joints:
25	к.	Construction and wovement joints.
36		 Construct joints true to line with faces perpendicular to surface plane of concrete. Install so strength and appearance of concrete are not impaired at locations indicated or as approved by
30		2. Install so strength and appearance of concrete are not imparied, at locations indicated of as approved by Architect
20		Place joints perpendicular to main reinforcement
20		5. Flace joints perpendicular to main remotement.
<u>40</u>		4. Excluse joints for bearing, stabs, joists, and griders in the initial entration of spans.
40 //1		a. Onset joints in gruers a minimum distance of twice the beam width norm a beam-gruer
41 12		5 Locate horizontal joints in walk and columns at underside of floors, slabs, hears, and girders and at the top
12		of footings or floor clabs
45 AA		6 Space vertical joints in walls as indicated on Drawings
77 15		a locate joints headed niers integral with walk near corners and in concealed locations where
45		a. Elecate joints beside piers integral with wails, hear corners, and in concealed locations where
40		Provide temperary parts or energings in formwork where required to facilitate cleaning and inspection
47 10	L.	1 Locate parts and energings in bottom of vertical forms, in inconstitute cleaning and inspection.
40		1. Excluse points and openings in bottom of ventical forms, in inconspicuous location, to allow hushing water to
49 E0		Uldill.
50		2. Close temporary ports and openings with tight-fitting panels, hush with histor face of form, and heatry
51	N.4	Clean forms and adjacent surfaces to receive concrete. Demous shins, wood, soundust, dist, and other debric just
52 52	IVI.	clean forms and adjacent surfaces to receive concrete. Remove chips, wood, sawdust, dirt, and other debris just
JJ ⊑4	N.	Detroite placing confidete.
54 FF	N.	Relignment
55	~	alignment.
50	0.	coat contact surfaces of forms with form-release agent, according to manufacturer's written instructions, before
5/		placing remorcement.
58		

1	3.2 INST	ALLATION OF EMBEDDED ITEMS
2	Α.	Place and secure anchorage devices and other embedded items required for adjoining work that is attached to or
3		supported by cast-in-place concrete.
4 5		1. Use setting drawings, templates, diagrams, instructions, and directions furnished with items to be embedded.
6		2. Install anchor rods, accurately located, to elevations required and complying with tolerances in Section 7.5
7		of AISC 303.
8		3. Clean embedded items immediately prior to concrete placement.
9 10	3.3 INST	TALLATION OF WATERSTOPS
11	A.	Elexible Waterstops: Install in construction joints and at other joints indicated to form a continuous diaphragm.
12		1. Install in longest lengths practicable.
13		 Locate waterstops in center of joint unless otherwise indicated on Drawings.
14 15		3. Allow clearance between waterstop and reinforcing steel of not less than 2 times the largest concrete
15		A Socure waterstops in correct position at 12 inches on conter
17		 Secure waterstops in correct position at 12 inches on center. Field fabricate joints in accordance with manufacturer's instructions using heat welding
10		3. Miter corpers intersections and directional changes in waterstons
10		h Align center bulls
20		6 Clean waterstops immediately prior to placement of concrete
20		 Support and protect exposed waterstops during progress of the Work
22	R	Self-Expanding Strip Waterstops: Install in construction joints and at other locations indicated on Drawings
23	υ.	according to manufacturer's written instructions, by adhesive bonding, mechanically fastening, and firmly pressing
24		into place.
25		1. Install in longest lengths practicable.
26		 Locate waterstops in center of joint unless otherwise indicated on Drawings.
27		 Protect exposed waterstops during progress of the Work.
28		
29	3.4 REM	IOVING AND REUSING FORMS
30	Α.	Formwork for sides of walls and similar parts of the Work that does not support weight of concrete may be
31		removed after cumulatively curing at not less than 50 deg F for 24 hours after placing concrete. Concrete has to be
32		hard enough to not be damaged by form-removal operations, and curing and protection operations need to be
33		maintained.
34	В.	Clean and repair surfaces of forms to be reused in the Work.
35		1. Split, frayed, delaminated, or otherwise damaged form-facing material are unacceptable for exposed
36		surfaces.
37		2. Apply new form-release agent.
38	С.	When forms are reused, clean surfaces, remove fins and laitance, and tighten to close joints.
39		 Align and secure joints to avoid offsets.
40		Do not use patched forms for exposed concrete surfaces unless approved by Architect.
41		
42		
43		
44		END OF SECTION

	SECTION 03 20 00 CONCRETE REINFORCING
PART 1	- GENERAL
1.1	SUMMARY
1.2	ACTION SUBMITTALS 1
1.3	INFORMATIONAL SUBMITTALS1
1.4	QUALITY ASSURANCE
1.5	DELIVERY, STORAGE, AND HANDLING
PART 2	- PRODUCTS
2.1	PERFORMANCE REQUIREMENTS
2.2	STEEL REINFORCEMENT
2.3	REINFORCEMENT ACCESSORIES
	FABRICATING REINFORCEIVIENT
PARI 3	
2.1	
3.2	
3.5	INSTALLATION TOLERANCES
3.5	FIELD QUALITY CONTROL
0.0	
PART 1	- GENERAL
1.1 SI	JMMARY
Α.	Section Includes:
	1. Steel reinforcement bars.
	2. Welded-wire reinforcement.
В.	Related Requirements:
	1. Section 034100 "Precast Structural Concrete" for reinforcing used in precast structural concrete.
	2. Section 034500 "Precast Architectural Concrete" for reinforcing used in precast architectural concrete.
	3. Section 321313 "Concrete Paving" for reinforcing related to concrete pavement and walks.
1.2 A	CTION SUBMITTALS
А.	Product Data: For the following:
	Each type of steel reinforcement. Energy appearance continent.
	2. Epoxy repair coating.
	3. Zinc repair indicidi.
	4. Bai supports.
	5. Mechanical splice couplets. 6. Structural thermal break insulated connection system
в	Shon Drawings: Comply with ACI SP-066.
Б.	1 Include placing drawings that detail fabrication bending and placement
	2 Include har sizes lengths materials grades har schedules stirrun spacing hent har diagrams har
	arrangement location of splices lengths of lan splices details of mechanical splice couplers details of
	welding splices, tie spacing, hoop spacing, and supports for concrete reinforcement.
	3. For structural thermal break insulated connection system, indicate general configuration, insulation
	dimensions, tension bars, compression pads, shear bars, and dimensions.
С.	Construction Joint Layout: Indicate proposed construction joints required to build the structure.
	1. Location of construction joints is subject to approval of Architect.
1.3 IN	FORMATIONAL SUBMITTALS
Α.	Qualification Statements: For testing and inspection agency.
В.	Delegated Design Engineer Qualifications: Include the following:
	1. Experience providing delegated design engineering services of the type indicated.
	2. Documentation that delegated design engineer is licensed in the state in which Project is located.
С.	Welding certificates.
	1. Reinforcement To Be Welded: Welding procedure specification in accordance with AWS D1.4/D1.4M.

D.	Material Certificates: For each of the following, signed by manufacturers:
	1. Epoxy-Coated Reinforcement: CRSI's "Epoxy Coating Plant Certification."
	2. Dual-Coated Reinforcement: CRSI's "Epoxy Coating Plant Certification."
E.	Material Test Reports: For the following, from a qualified testing agency:
	1. Steel Reinforcement:
	a. For reinforcement to be welded, mill test analysis for chemical composition and carbon equivalen
	of the steel in accordance with ASTM A706/A706M
	2 Mechanical solice counters
c	2. With an and the couplet st
г.	
1.4 QU	ALITY ASSURANCE
Α.	Testing Agency Qualifications: An independent agency qualified in accordance with ASTM C1077 and ASTM E329 fo
	testing indicated.
В.	Welding Qualifications: Qualify procedures and personnel in accordance with AWS D1.4/D 1.4M.
	IVENT, STORAGE, AND HANDLING Steel Reinforcement: Deliver, store, and handle steel reinforcement to provent handing and damage
А.	Steer reinforcement, benver, store, and indude steer reinforcement to prevent behands dhu udmage.
	Store reinforcement to dvoid contact with editin.
	2. Do not allow epoxy-coated reinforcement to be stored outdoors for more than 60 days without being
	stored under an opaque covering.
	3. Do not allow dual-coated reinforcement to be stored outdoors for more than 60 days without being stored
	under an opaque covering.
	4. Do not allow stainless steel reinforcement to come into contact with uncoated reinforcement.
2.1 PER	FORMANCE REQUIREMENTS
А.	Structural Performance of Structural mermal break insulating connection system. Structural thermal break
	insulateu connection system to withstand the following loads and stresses.
	1. Dedu Lodus: As indicated on Drawings.
	d. Shedi Lodu: As indicated on Drawings.
	b. Benang Moment. As indicated on Drawings.
	2. Live Loads: As indicated on Drawings.
	a. Shear Load: As indicated on Drawings.
	b. Bending Moment: As indicated on Drawings.
2.2 STE	EL REINFORCEMENT
Α.	Reinforcing Bars: ASTM A615/A615M, Grade 60deformed.
В.	Headed-Steel Reinforcing Bars: ASTM A970/A970M.
C.	Epoxy-Coated Reinforcing Bars:
	1. Steel Bars: ASTM A615/A615M, Grade 60, deformed bars.
	2. Epoxy Coating: ASTM A775/A775M or ASTM A934/A934M with less than 2 percent damaged coating in eacl
	12-inch bar length.
D.	Plain-Steel Welded-Wire Reinforcement: ASTM A1064/A1064M. plain. fabricated from as-drawn steel wire into fla
υ.	sharts
E.	Deformed-Steel Welded-Wire Reinforcement: ASTM A1064/A1064M, flat sheet.
2.3 REI	NFORCEMENT ACCESSORIES
A.	Joint Dowei Bars: ASIM A615/A615M, Grade 60, plain-steel bars, cut true to length with ends square and free o
_	burrs.
В.	Epoxy-Coated Joint Dowel Bars: ASTM A615/A615M, Grade 60, plain-steel bars, ASTM A775/A775M epoxy coated.
C.	Bar Supports: Bolsters, chairs, spacers, and other devices for spacing, supporting, and fastening reinforcing bars and
	welded-wire reinforcement in place.
	1. Manufacture bar supports from steel wire, plastic, or precast concrete in accordance with CRSI's "Manual c
	Standard Practice," of greater compressive strength than concrete and as follows:
	a. For concrete surfaces exposed to view, where legs of wire bar supports contact forms, use CRS
	Class 1 plastic-protected steel wire, all-plastic bar supports, or CRSI Class 2 stainless steel ba
	supports.

	b. For epoxy-coated reinforcement, use CRSI Class 1A epoxy-coated or other dielectric-polymer-coated
	wire bar supports. c. For dual-coated reinforcement, use CRSI Class 1A epoxy-coated or other dielectric-polymer-coated
	wire bar supports.
	d. For zinc-coated reinforcement, use galvanized wire or dielectric-polymer-coated wire bar supports.
	e. For stainless steel reinforcement, use CRSI Class 1 plastic-protected steel wire, all-plastic bar supports, or CRSI Class 2 stainless steel bar supports.
D.	Mechanical Splice Couplers: ACI 318 [Type 1] [Type 2], same material of reinforcing bar being spliced;
-	[compression-only type] [tension-compression type] [dowel-bar type] [mechanical-lap type].
с.	1. Finish: Plain.
2.4 FAB	RICATING REINFORCEMENT
A.	Fabricate steel reinforcement according to CRSI's "Manual of Standard Practice."
PART 3 -	EXECUTION
<u></u>	
3.1 PRE	PARATION
А.	Protection of in-Place Conditions:
	Do not cut of puncture vapor retarder before placing concrete
в	Clean reinforcement of loose rust and mill scale earth ice and other foreign materials that reduce hand to
Б.	concrete.
3.2 INS	TALLATION OF STEEL REINFORCEMENT
Α.	Comply with CRSI's "Manual of Standard Practice" for placing and supporting reinforcement.
В.	Accurately position, support, and secure reinforcement against displacement.
	1. Locate and support reinforcement with bar supports to maintain minimum concrete cover.
6	2. Do not tack weld crossing reinforcing bars.
С.	Preserve clearance between bars of not less than 1 inch, not less than one bar diameter, or not less than 1-1/3
Р	times size of large aggregate, whichever is greater.
D. F	Provide concrete coverage in accordance with ACI 516.
F.	Splices: Lan splices as indicated on Drawings
	1 Bars indicated to be continuous, and all vertical bars to be lanned not less than 36 bar diameters at solices
	or 24 inches whichever is greater
	2. Stagger splices in accordance with ACI 318.
	 Mechanical Splice Couplers: Install in accordance with manufacturer's instructions.
	4. Weld reinforcing bars in accordance with AWS D1.4/D 1.4M, where indicated on Drawings.
G.	Install welded-wire reinforcement in longest practicable lengths.
	1. Support welded-wire reinforcement in accordance with CRSI "Manual of Standard Practice."
	a. For reinforcement less than W4.0 or D4.0, continuous support spacing to not exceed 12 inches.
	2. Lap edges and ends of adjoining sheets at least one wire spacing plus 2 inches for plain wire and 8 inches for
	deformed wire.
	3. Offset laps of adjoining sheet widths to prevent continuous laps in either direction.
	4. Lace overlaps with wire.
Н.	
3.3 JOII	NTS
Α.	Construction Joints: Install so strength and appearance of concrete are not impaired, at locations indicated or as
	approved by Architect.
	1. Place joints perpendicular to main reinforcement.
	2. Continue reinforcement across construction joints unless otherwise indicated.
	3. Do not continue reinforcement through sides of strip placements of floors and slabs.
В.	Doweled Joints: Install dowel bars and support assemblies at joints where indicated. Lubricate or asphalt coat one-
	half of dowel length, to prevent concrete bonding to one side of joint.

1	3.4 INST	ALLATION TOLERANCES
2	Α.	Comply with ACI 117.
3		
4	3.5 FIELD	QUALITY CONTROL
5	Α.	Special Inspections: Owner will engage a special inspector to perform field tests and inspections and prepare test
6		reports.
7		
8	В.	Inspections:
9		1. Steel-reinforcement placement.
10		2. Steel-reinforcement mechanical splice couplers.
11		3. Steel-reinforcement welding.
12		
13		END OF SECTION

1		SECTION 03 30 00	
2		CAST-IN-PLACE CONCRETE	
3			
4	PART 1 - 0	GENERAL	1
5	1.1	SUMMARY	1
6	1.2	DEFINITIONS	2
7	1.3	PREINSTALLATION MEETINGS	2
8	1.4	ACTION SUBMITTALS	2
9	1.5	INFORMATIONAL SUBMITTALS	3
10	1.6	QUALITY ASSURANCE	3
11	1.7	PRECONSTRUCTION TESTING	4
12	1.8	DELIVERY, STORAGE, AND HANDLING	4
13	1.9	FIELD CONDITIONS	4
14	1.10	WARRANTY	4
15	PART 2 - F	PRODUCTS	4
16	2.1	CONCRETE, GENERAL	5
17	2.2	CONCRETE MATERIALS	5
18	2.3	VAPOR RETARDERS	6
19	2.4	LIQUID FLOOR TREATMENTS	6
20	2.5	CURING MATERIALS	6
21	2.6	RELATED MATERIALS	6
22	2.7	REPAIR MATERIALS	6
23	2.8	CONCRETE MIXTURES, GENERAL	6
24	2.9	CONCRETE MIXTURES	7
25	2.10	CONCRETE MIXING	7
26	PART 3 - E	EXECUTION	7
27	3.1	EXAMINATION	7
28	3.2	PREPARATION	7
29	3.3	INSTALLATION OF EMBEDDED ITEMS	8
30	3.4	INSTALLATION OF VAPOR RETARDER	8
31	3.5	IOINTS	8
32	3.6	CONCRETE PLACEMENT	8
33	3.7	EINISHING EORMED SUBFACES	9
34	3.8	EINISHING FLOORS AND SLABS	10
35	3.9	INSTALLATION OF MISCELLANFOLIS CONCRETE ITEMS	10
36	3 10	CONCRETE CURING	11
37	3.10	TOLERANCES	12
38	3 12	ΔΡΡΙΙΔΑΤΙΩΝ ΩΕΙ ΙΩΙ ΙΙΩ ΕΙ ΩΩR TREATMENTS	12
39	3 13	IOINT FILLING	12
40	2 1 <i>1</i>	CONCRETE SURFACE REPAIRS	12
40 41	2 15		13 1/
42 42	2 16	PROTECTION	15
12	5.10		13
43			

44 PART 1 - GENERAL

46 **1.1 SUMMARY**

45

47	Α.	In gene	ral, the work includes the following:
48		1.	1. Footings
49		2.	Stoops (at doors)
50		3.	Aprons (at overhead doors, including concrete infill between stoops and aprons)
51		4.	Cast-in-place concrete work
52		5.	Interior slabs on grade
53		6.	Vapor retarder (barrier)
54		7.	Recessed slab, below-slab drain containment and other concrete work below the floor slabs.
55		8.	Floor trenches or utility trenches
56		9.	Cast-in-place concrete, including concrete materials, mixture design, placement procedures, and finishes.
57	В.	Related	Requirements:

1 2 3 4 5 6		 Section Section Section Section Section Section 	ion 031000 "Concrete Forming and Accessories" for form-facing materials, form liners, insulating crete forms, and waterstops. ion 032000 "Concrete Reinforcing" for steel reinforcing bars and welded-wire reinforcement. ion 079200 "Joint Sealant" ion 321313 "Concrete Paving" for concrete pavement and walks.
7	1.2 DEFI	ITIONS	
8 9	A.	Cementitiou hydraulic ce	Is Materials: Portland cement alone or in combination with one or more of the following: blended ement, fly ash, slag cement, other pozzolans, and silica fume; materials subject to compliance with
10	В.	Water/Ceme	ent Ratio (w/cm): The ratio by weight of water to cementitious materials.
12	1.2 0050		METHOD
13	1.3 PKEII	Broinstallati	MEETINGS on Conference: Conduct conference at Braiget site
14	А.		Un conference: conduct conference at Project site .
15		I. Requ	uire representatives of each entity directly concerned with cast-in-place concrete to attend, including
16		the f	ronowing:
1/		a.	Contractor's superintendent.
18		b.	Independent testing agency responsible for concrete design mixtures.
19		с.	Ready-mix concrete manufacturer.
20		d.	Concrete Subcontractor.
21		2. Revi	ew the following:
22		a.	Construction joints, control joints, isolation joints, and joint-filler strips.
23		b.	Semirigid joint fillers.
24		с.	Vapor-retarder installation.
25		d.	Anchor rod and anchorage device installation tolerances.
26		e.	Cold and hot weather concreting procedures.
27		f.	Concrete finishes and finishing.
28		g.	Curing procedures.
29		h	Forms and form-removal limitations
30		i	Methods for achieving specified floor and slab flatness and levelness
21		i.	Floor and slab flatness and levelopes measurements
22		j. k	Concrete repair precedures
32 22		K.	Concrete repair procedures.
33		I.	Concrete protection.
34		m.	Initial curing and field curing of field test cylinders (ASTM C31/C31ML)
35		n.	Protection of field cured field test cylinders.
36	_		
37	1.4 ACTI0	ON SUBMITTA	
38	А.	Product Data	a: For each of the following.
39		1. Port	land cement.
40		2. Fly a	ish.
41		3. Slag	cement.
42		4. Aggr	regates.
43		5. Adm	nixtures:
44		a.	Include limitations of use, including restrictions on cementitious materials, supplementary
45			cementitious materials, air entrainment, aggregates, temperature at time of concrete placement,
46			relative humidity at time of concrete placement, curing conditions, and use of other admixtures.
47		6. Fibe	r reinforcement.
48		7. Vapo	or retarders.
49		8. Floo	r and slab treatments.
50		9. Liqui	id floor treatments.
51		10 Curi	ng materials
52		cum	Include documentation from color nigment manufacturer indicating that proposed mothods of
52		а.	curing are recommended by color nigment manufacturer, indicating that proposed methods of
55		11 10:	t filore
54 FF			L IIIIEIS.
35		та кера	dii ilidiciidis.
50	В.	Design Mixti	ures: For each concrete mixture, include the following:
5/		I. Mixt	cure identification.
58		2. Mini	imum 28-day compressive strength.

1		3. Durability exposure class.
2		4. Maximum w/cm.
3		5. Slump limit.
4		6. Air content.
5		7. Nominal maximum aggregate size.
6		8. Indicate amounts of mixing water to be withheld for later addition at Project site if permitted.
7		9. Intended placement method.
8		10. Submit alternate design mixtures when characteristics of materials. Project conditions, weather, test
9		results, or other circumstances warrant adjustments.
10	C.	Shop Drawings:
11	-	1. Construction Joint Layout: Indicate proposed construction joints required to construct the structure.
12		a. Location of construction joints is subject to approval of the Architect.
13	D.	Samples: For vapor retarder.
14	Ε.	Concrete Schedule: For each location of each Class of concrete indicated in "Concrete Mixtures" Article, including
15		the following:
16		1. Concrete Class designation.
17		2. Location within Project.
18		3. Exposure Class designation.
19		4. Formed Surface Finish designation and final finish.
20		5. Final finish for floors.
21		6. Curing process.
22		7. Floor treatment if any.
23		,
24	1.5 INFO	RMATIONAL SUBMITTALS
25	Α.	Material Certificates: For each of the following, signed by manufacturers:
26		1. Cementitious materials.
27		2. Admixtures.
28		3. Fiber reinforcement.
29		4. Curing compounds.
30		5. Floor and slab treatments.
31		6. Bonding agents.
32		7. Adhesives.
33		8. Vapor retarders.
34		9. Semirigid joint filler.
35		10. Joint-filler strips.
36		11. Repair materials.
37	В.	Material Test Reports: For the following, from a qualified testing agency:
38		1. Portland cement.
39		2. Fly ash.
40		3. Slag cement.
41		4. Aggregates.
42		5. Admixtures:
43	С.	Floor surface flatness and levelness measurements report, indicating compliance with specified tolerances.
44	D.	Research Reports:
45		1. For concrete admixtures in accordance with ICC's Acceptance Criteria AC198.
46		2. For sheet vapor retarder/termite barrier, showing compliance with ICC AC380.
47	Ε.	Preconstruction Test Reports: For each mix design.
48	F.	Field quality-control reports.
49	G.	Minutes of preinstallation conference.
50		
51	1.6 QUA	LITY ASSURANCE
52	Α.	Laboratory Testing Agency Qualifications: A testing agency qualified in accordance with ASTM C1077 and
53		ASTM E329 for testing indicated and employing an ACI-certified Concrete Quality Control Technical Manager.
54		1. Personnel performing laboratory tests to be an ACI-certified Concrete Strength Testing Technician and
55		Concrete Laboratory Testing Technician, Grade I. Testing agency laboratory supervisor to be an ACI-certified
56		Concrete Laboratory Testing Technician, Grade II.
57	В.	Field Quality-Control Testing Agency Qualifications: An independent agency, acceptable to authorities having
58		jurisdiction, qualified in accordance with ASTM C1077 and ASTM E329 for testing indicated.

1 2		1.	Personnel conducting field tests to be qualified as an ACI Concrete Field-Testing Technician, Grade 1, in accordance with ACI CPP 610.1 or an equivalent certification program.
3 4 5	C.	Mock finish selec	sups: Before casting architectural concrete, build mockups, using the same procedures, equipment, materials, ing procedures, and curing procedures that will be used for producing architectural concrete, to verify tions made under Sample submittals and to demonstrate typical joints, surface finish, color, texture,
6		tolera	ances, and standard of workmanship. Build mockups to comply with the following requirements, using
7		mate	rials indicated for the completed work:
8		1.	Build mockups in the location and of the size indicated or, if not indicated, as directed by architect.
9		2.	Build mockups of typical wall of cast-in-place architectural concrete as shown on drawings, including vertical
10			and horizontal rustication joints, and any sculptured features.
11		3.	Construct mockups to include at least two lifts having heights equal to those anticipated for construction.
12 13		4.	Demonstrate curing, cleaning, and protecting of cast-in-place architectural concrete, finishes, and contraction joints, as applicable.
14		5.	In presence of architect, damage part of the exposed-face surface for each finish, color, and texture, and
15 16		6	demonstrate materials and techniques proposed for repair to match adjacent undamaged surfaces.
17		-	blemishes to match adjacent undamaged surfaces.
18		7.	Obtain architect's approval of mockups before casting architectural concrete.
20 19	17 DRFC		
20	A.	Preco	postruction Testing Service: Engage a qualified testing agency to perform preconstruction testing on each
22		concr	rete mixture.
23		1.	Include the following information in each test report:
24			a. Admixture dosage rates.
25			b. Slump.
26			c. Air content.
27			d. Seven-day compressive strength.
28			e. 28-day compressive strength.
29			f. Permeability.
30		/FDV C	
37 31		Comr	NURAGE, AND HANDLING
32 33	А.	Com	
34	1.9 FIELD	COND	ITIONS
35	Α.	Cold-	Weather Placement: Comply with ACI 301 and ACI 306.1 and as follows.
36 37		1.	Protect concrete work from physical damage or reduced strength that could be caused by frost, freezing
38		2.	When average high and low temperature is expected to fall below 40 deg E for three successive days.
39			maintain delivered concrete mixture temperature within the temperature range required by ACI 301.
40		3.	Do not use frozen materials or materials containing ice or snow.
41		4.	Do not place concrete in contact with surfaces less than 35 deg F, other than reinforcing steel.
42		5.	Do not use calcium chloride, sait, or other materials containing antifreeze agents or chemical accelerators
43 44	в	Hot \	unless otherwise specified and approved in mixture designs.
44 15	Б.	1	Maintain concrete temperature at time of discharge to not exceed 95 deg E
45 46		1. 2	Fog-spray forms steel reinforcement and subgrade just before placing concrete. Keen subgrade uniformly
40		2.	moist without standing water soft snots or dry areas
48			
49	1.10 WAR	RANTY	
50	Α.	Manu	ufacturer's Warranty: Manufacturer agrees to furnish replacement sheet vapor retarder/termite barrier
51		mate	rial and accessories for sheet vapor retarder/ termite barrier and accessories that do not comply with
52		requi	rements or that fail to resist penetration by termites within specified warranty period.
53		1.	Warranty Period: 10 years from date of Substantial Completion.
54	BAR		
ככ	PAKIZ-P	KUUU	

56

1	2.1 CON	ICRETE, GENERAL
2	Α.	ACI Publications: Comply with ACI 301 unless modified by requirements in the Contract Documents.
3	2.2	FORM-FACING MATERIALS
Л	Δ	Form-Facing wood forms:
-	11.	
5		
6		a. Exterior-grade rough sawn board form planks, that will provide continuous, true, and pattern
/		architectural concrete surfaces, Douglas Fir, construction grade, or better, mili-applied release agent
٥ ٥		and edge sealed, complying with DOC PS 1.
9		2. Election. Where holed on drawings.
10		5. Size. 4-incit nonlinal neight
12	23 001	ΙCRETE ΜΑΤΕRΙΔΙ S
13	A.	Source Limitations:
14	7	1. Obtain all concrete mixtures from a single ready-mixed concrete manufacturer for entire Project.
15		 Obtain each type or class of cementitious material of the same brand from the same manufacturer's plant.
16		3. Obtain aggregate from single source.
L7		4. Obtain each type of admixture from a single source from single manufacturer.
18	В.	Cementitious Materials:
19		 Portland Cement: ASTM C150/C150M, Type I/II, gray.
20		2. Fly Ash: ASTM C618, Class C or F.
21	С.	Normal-Weight Aggregates: ASTM C33/C33M, Class 3S coarse aggregate or better, graded. Provide aggregates from
22		a single source.
23		 Maximum Coarse-Aggregate Size: 1-1/2 inches nominal.
24		2. Fine Aggregate: Free of materials with deleterious reactivity to alkali in cement.
25	D.	Air-Entraining Admixture: ASTM C260/C260M.
26	Ε.	Chemical Admixtures: Certified by manufacturer to be compatible with other admixtures that do not contribute
27		water-soluble chloride ions exceeding those permitted in hardened concrete. Do not use calcium chloride or
28		admixtures containing calcium chloride in steel-reinforced concrete.
29	•	1 Mater Deducing Admintures ACTNA CADA/CADANA Tures A
3U 01		Water-Reducing Admixture: ASTM C494/C494W, Type A. Betarding Admixture: ASTM C494/C494W, Type A.
51 51		2. Relations Admixture: ASTM C494/C494W, Type B.
22		A High-Range Water-Reducing Admixture: ASTM C494/C494M, Type D.
33 84		 High-Range, Water-Reducing Admixture: ASTM C494/C494M, Type T. High-Range, Water-Reducing and -Retarding Admixture: ASTM C494/C494M. Type G.
35		6 Plasticizing and Retarding Admixture: ASTM C1017/C1017M Type II
36		7. Set-Accelerating Corrosion-Inhibiting Admixture: Commercially formulated, anodic inhibitor or mixed
37		cathodic and anodic inhibitor: capable of forming a protective barrier and minimizing chloride reactions
88		with steel reinforcement in concrete and complying with ASTM C494/C494M, Type C.
39		8. Non-Set-Accelerating Corrosion-Inhibiting Admixture: Commercially formulated, non-set-accelerating,
10		anodic inhibitor or mixed cathodic and anodic inhibitor; capable of forming a protective barrier and
11		minimizing chloride reactions with steel reinforcement in concrete.
	-	
+Z 10	г.	Superplasticizer shall be used in all interior natwork and any pumped concrete unless otherwise approved in writing by the Architect. All concrete with W/C of 0.48 or loss and where numping equipment is used, requires this
+5 17		by the Architect. All concrete with W/C of 0.46 of less, and where pumping equipment is used, requires this admixture. In general, concrete shall be delivered to the site with a slump of 2" to 2 1/". Admixture shall be delivered to the site with a slump of 2" to 2 1/".
+4 15		added in lieu of any additional water
+J 16	G	Water and Water Used to Make Ice: ASTM C94/C94M, potable or complying with ASTM C1602/C1602M, including
17	0.	all limits listed in Table 2 and the requirements of paragraph 5.4
18		
19	2.4 FIBE	R REINFORCEMENT
50	А.	Synthetic Macro-Fiber: Synthetic macro-fibers engineered and designed for use in concrete, complying with
51 - 2		ASIM C1116/C1116M, Type III, [1 to 2-1/4 inches] < Insert dimensions > long.
52		

1	2.5 VAP	POR RETARDERS
2	Α.	Sheet Vapor Retarder, Class A: ASTM E1745, Class A; not less than 15 mils thick. Include manufacturer's
3		recommended adhesive or pressure-sensitive tape.
4		
5		
6	26 110	UD ELOOR TREATMENTS (SC-1)
7	2.0 LIQ	Benetrating Liquid Floor Treatment: Clear, chemically reactive, waterborne solution of inorganic silicate or siliconate
0	Α.	materials and analysis and an analysis is a second second second densities concerned suite of since the second sec
0		naterials and proprietary components, outpriess, that penetrates, nardens, and densines concrete surfaces.
9		1. <u>Provide</u> Prosoco SLX100 water and Oli Repellant or equal.
10		
11	2.7 CUF	RING MATERIALS (SC-2)
12	А.	Evaporation Retarder: Waterborne, monomolecular film forming, manufactured for application to fresh concrete.
13	В.	Absorptive Cover: AASHTO M 182, Class 2, burlap cloth made from jute or kenaf, weighing approximately 9 oz./sq.
14		yd. when dry.
15	С.	Moisture-Retaining Cover: ASTM C171, polyethylene film burlap-polyethylene sheet.
16		1. Color:
17		a. Ambient Temperature Below 50 deg F: Black.
18		b. Ambient Temperature between 50 deg F and 85 deg F: Any color.
19		c. Ambient Temperature Above 85 deg F: White.
20	D.	Water: Potable or complying with ASTM C1602/C1602M.
21	Ε.	Clear, Waterborne, Membrane-Forming, Curing and Sealing Compound: ASTM C1315, Type 1, Class A.
22		
23	2.8 REL	ATED MATERIALS
24	Δ	Expansion- and Isolation-Joint-Filler Strips: ASTM D1751, asphalt-saturated cellulosic fiber
25	л В	Enoxy Bonding Adhesive: ASTM C881 two-component enoxy resin capable of humid curing and bonding to damp
26	Б.	surfaces of class suitable for annication temporture and of grade and class to suit requirements and as follows:
20		Types L and L nonload bearing for bonding bardened or freshly mixed concrete to bardened concrete
27		1. Types rate it, nonoau bearing, for bonding nardened of restry mixed concrete to nardened concrete.
28		
29	2.9 REP	
30	Α.	Repair Underlayment: Cement-based, polymer-modified, self-leveling product that can be applied in thicknesses
31		from 1/8 inch and that can be feathered at edges to match adjacent floor elevations.
32		1. Cement Binder: ASTM C150/C150M portland cement or hydraulic or blended hydraulic cement, as defined
33		in ASTM C219.
34		2. Primer: Product of underlayment manufacturer recommended for substrate, conditions, and application.
35		3. Aggregate: Well-graded, washed gravel, 1/8 to 1/4 inch or coarse sand, as recommended by underlayment
36		manufacturer.
37		4. Compressive Strength: Not less than 5000 psi at 28 days when tested in accordance with
38		ASTM C109/C109M.
39	В.	Repair Overlayment: Cement-based, polymer-modified, self-leveling product that can be applied in thicknesses
40		from 1/4 inch and that can be filled in over a scarified surface to match adjacent floor elevations.
41		1. Cement Binder: ASTM C150/C150M portland cement or hydraulic or blended hydraulic cement, as defined
42		in ASTM C219.
43		2. Primer: Product of topping manufacturer recommended for substrate, conditions, and application.
44		3. Aggregate: Well-graded, washed gravel, 1/8 to 1/4 inch or coarse sand as recommended by topping
45		manufacturer
46		4 Compressive Strength: Not less than 5000 nsi at 28 days when tested in accordance with
40 47		4. Compressive strength. Not less than 5000 psi at 20 days when rested in accordance with
47 10		ASTIVI C105/C105IVI.
40 10	2 10 00	
49	2.10 CON	Note le Mixi UNES, General
50	А.	Prepare design mixtures for each type and strength of concrete, proportioned on the basis of laboratory trial
51		mixture or field test data, or both, in accordance with ACI 301.
52		1. Use a qualified testing agency for preparing and reporting proposed mixture designs, based on laboratory
53		trial mixtures.
54		
55	В.	Cementitious Materials: Limit percentage, by weight, of cementitious materials other than portland cement in
56		concrete as follows:
57		1. Fly Ash or Other Pozzolans: 20 percent by mass.
58		2. Slag Cement: 35 percent by mass.

1	C	Admixtures: Use admixtures in accordance with manufacturer's written instructions
2	с.	1 Use water-reducing admixture in concrete as required for placement and workability
2		 Use water-reducing and -retarding admixture when required by high temperatures low humidity or other
1		adverse placement conditions
5		3 Use water-reducing admixture in numned concrete , concrete for heavy-use industrial slabs
6		
7	2.11 CON	CRETE MIXTURES
8	Α.	Class A: Normal-weight concrete used for footings.
9		1. Exposure Class: ACI 318 F1-S0-W0-C0.
10		2. Minimum Compressive Strength 3500 psi at 28 days.
11		3. Maximum w/cm: 0.55.
12		4. Slump Limit: 4 inches, plus or minus 1 inch.
13		5. Air Content:
14		a. Exposure Class F1: 4.5 percent, plus or minus 1.5 percent at point of delivery for concrete
15		containing 1-1/2-inch nominal maximum aggregate size.
16	В.	Class B: Normal-weight concrete used for foundation walls.
17		1. Exposure Class: ACI 318 F1-S0-W0-C0.
18		2. Minimum Compressive Strength: 4000 psi at 28 days.
19		3. Maximum w/cm: 0.45
20		4. Slump Limit: 4 inches, plus or minus 1 inch.
21		5. Air Content:
22		a. Exposure Class F1: 5.0 percent, plus or minus 1.5 percent at point of delivery for concrete
23		containing 3/4-inch nominal maximum aggregate size
24 25	c	6. Limit water-soluble, chloride-ion content in hardened concrete to 1.00 percent by weight or cement. Class C: Normal weight concrete used for interior slobs on ground.
25	C.	Class C. Normal-weight concrete used for interior slabs-on-ground.
20 27		Exposure Class: ACI 318 F0-S0-W0-C0. Minimum Compressive Strength: /E00 nsi at 28 days
27		2. Withinfull compressive strength. 4500 psi at 26 days. 3. Maximum w/cm: 0.40
20		A Slump Limit: 5 inches plus or minus 1 inch
30		5 Air Content:
31		a. Do not use an air-entraining admixture or allow total air content to exceed 3 percent for concrete
32		used in trowel-finished floors.
33		6. Synthetic Macro-Fiber: Where indicated on plans, uniformly disperse in concrete mixture at manufacturer's
34		recommended rate, but not less than a rate of 5 lb/cu. vd. .
35		
36	2.12 CON	CRETE MIXING
37	Α.	Ready-Mixed Concrete: Measure, batch, mix, and deliver concrete in accordance with ASTM C94/C94M, and furnish
38		batch ticket information.
39		
40	<u> Part 3 - E</u>	XECUTION
41		
42	3.1 EXAN	AINATION Marification of Constitutions
43	А.	Verification of Conditions:
44 15		1. Before placing concrete, verify that installation of concrete forms, accessories, and remorcement, and
45		2 Do not proceed until upsatisfactory conditions have been corrected
40		2. Do not proceed until disatisfactory conditions have been corrected.
47	3.2 PRFP	ΔΡΑΤΙΟΝ
49	A.	Provide reasonable auxiliary services to accommodate field testing and inspections, acceptable to testing agency,
50		including the following:
51		1. Daily access to the Work.
52		2. Incidental labor and facilities necessary to facilitate tests and inspections.
53		3. Secure space for storage, initial curing, and field curing of test samples, including source of water and
54		continuous electrical power at Project site during site curing period for test samples.
55		4. Security and protection for test samples and for testing and inspection equipment at Project site.
56		

1	3.3 INST	LATION OF EMBEDDED ITEMS
2	Α.	Place and secure anchorage devices and other embedded items required for adjoining Work that is attached to or
3		supported by cast-in-place concrete.
4 5		 Use setting drawings, templates, diagrams, instructions, and directions furnished with items to be embedded.
6 7 8		2. Install anchor rods, accurately located, to elevations required and complying with tolerances in Section 7.5 of ANSI/AISC 303.
9	3.4 INST	LATION OF VAPOR RETARDER
10	A.	Sheet Vapor Retarders: Place, protect, and repair sheet vapor retarder in accordance with ASTM E1643 and
11		manufacturer's written instructions.
12		1. Install vapor retarder with longest dimension parallel with direction of concrete pour.
13		2. Face laps away from exposed direction of concrete pour.
14		3. Lap vapor retarder over footings and grade beams not less than 6 inches, sealing vapor retarder to concrete.
15		4. Lap joints 6 inches and seal with manufacturer's recommended tape.
16 17		5. Terminate vapor retarder at the top of floor slabs, grade beams, and pile caps, sealing entire perimeter to floor slabs, grade beams, foundation walks, or pile caps
18		Seal nenetrations in accordance with vanor retarder manufacturer's instructions
19		All overlapped conditions and terminations shall be sealed off and taped in a watertight manner
20		Protect vanor retarder during placement of reinforcement and concrete
21		a. Repair damaged areas by patching with vapor retarder material, overlapping damages area by 6
22		inches on all sides, and sealing to vapor retarder.
23	В.	Bituminous Vapor Retarders: Place, protect, and repair bituminous vapor retarder in accordance with
24		manufacturer's written instructions.
25		
26	3.5 JOIN	
27	Α.	Construct joints true to line, with faces perpendicular to surface plane of concrete.
28	В.	Construction Joints: Coordinate with floor slab pattern and concrete placement sequence.
29		1. Install so strength and appearance of concrete are not impaired, at locations indicated on Drawings or as
30		approved by Architect.
31		Place joints perpendicular to main reinforcement.
32		a. Continue reinforcement across construction joints unless otherwise indicated.
33		b. Do not continue reinforcement through sides of strip placements of floors and slabs.
34		3. Locate horizontal joints in walls and columns at underside of floors, slabs, beams, and girders and at the top
35		of footings or floor slabs.
36		4. Use a bonding agent at locations where fresh concrete is placed against hardened or partially hardened
37		concrete surfaces.
38 20		5. Use epoxy-bonding adhesive at locations where fresh concrete is placed against hardened or partially hardened constrate surfaces
<u>40</u>	C	Talueneu concrete surfaces. Control Jointe in Slabe on Ground: Form workened along control jointe, sostioning concrete integrades of indicated
40 //1	C.	Control Joints in Slabs-on-Ground. Form weakened-plane control joints, sectioning concrete thickness as follows:
42		Sawed Joints: Form control joints with nower saws equipped with shatterproof abrasive or diamond-
43		rimmed blades. Cut 1/8-inch- wide joints into concrete when cutting action does not tear abrade or
44		otherwise damage surface and before concrete develops random cracks.
45	D.	Isolation Joints in Slabs-on-Ground: After removing formwork, install joint-filler strips at slab junctions with vertical
46		surfaces, such as column pedestals, foundation walls, grade beams, and other locations, as indicated.
47		1. Extend joint-filler strips full width and depth of joint, terminating flush with finished concrete surface unless
48		otherwise indicated on Drawings.
49		2. Terminate full-width joint-filler strips not less than 1/2 inch or more than 1 inch below finished concrete
50		surface, where joint sealants, specified in Section 079200 "Joint Sealants," are indicated.
51		
52	3.6 CON	ETE PLACEMENT
53	Α.	Before placing concrete, verify that installation of formwork, reinforcement, embedded items, and vapor retarder is
54		complete and that required inspections are completed.
55		1. Immediately prior to concrete placement, inspect vapor retarder for damage and deficient installation, and
56		repair defective areas.
57		2. Provide continuous inspection of vapor retarder during concrete placement and make necessary repairs to
58		damaged areas as Work progresses.

1	В.	Notify Architect and testing and inspection agencies 24 hours prior to commencement of concrete placement.
2	С.	Do not add water to concrete during delivery, at Project site, or during placement unless approved by Architect in
3		writing, but not to exceed the amount indicated on the concrete delivery ticket.
4	D.	Before test sampling and placing concrete, water may be added at Project site, subject to limitations of ACI 301, but
5		not to exceed the amount indicated on the concrete delivery ticket.
6	Ε.	Deposit concrete continuously in one layer or in horizontal layers of such thickness that no new concrete is placed
7		on concrete that has hardened enough to cause seams or planes of weakness.
8		1. If a section cannot be placed continuously, provide construction joints as indicated.
9		2. Deposit concrete to avoid segregation.
10		3. Deposit concrete in horizontal layers of depth not to exceed formwork design pressures and in a manner to
11		avoid inclined construction ioints.
12		4. Consolidate placed concrete with mechanical vibrating equipment in accordance with ACI 301.
13		a Do not use vibrators to transport concrete inside forms
14		h Insert and withdraw vibrators vertically at uniformly spaced locations to rapidly penetrate placed
15		laver and at least 6 inches into preceding laver
16		c Do not insert vibrators into lower layers of concrete that have begun to lose plasticity
17		d At each insertion limit duration of vibration to time necessary to consolidate concrete and
18		complete embedment of reinforcement and other embedded items without causing mixture
10		constituents to segregate
20	F	Denosit and consolidate concrete for floors and slabs in a continuous operation, within limits of construction joints
20	1.	until placement of a papel or section is complete
22		1 Do not place concrete floors and slabs in a checkerboard sequence
23		 2 Consolidate concrete during placement operations, so concrete is thoroughly worked around reinforcement
24		and other embedded items and into corners
25		3. Maintain reinforcement in position on chairs during concrete placement.
26		 Screed slab surfaces with a straightedge and strike off to correct elevations.
27		5. Level concrete, cut high areas, and fill low areas.
28		6. Slope surfaces uniformly to drains where required.
29		7. Begin initial floating using bull floats or darbies to form a uniform and open-textured surface plane, before
30		excess bleedwater appears on the surface.
31		8. Do not further disturb slab surfaces before starting finishing operations.
32	G.	After floating, begin first trowel finish operation using a power-driven trowel. Begin final troweling when surface
33		produces a ringing sound as trowel is moved over surface. Consolidate concrete surface by final hang-troweling
34		operation, free of trowel marks, uniform in texture and appearance, and with a level surface plane so that
35		depressions between high spots do not exceed tolerances listed below. Grind smoot defects which would telegraph
36		through applied floor covering system only in acceptable to Architect.
37		1. Finish tolerances:
38		a. All locations except as noted:
39		1) specified overall $F_F = 50$
40		2) specified overall $F_L = 35$
41		3) specified local $F_F = 30$
42		4) specified local $F_L = 20$
43		
44	3.7 FINIS	HING FORMED SURFACES
45	А.	As-Cast Surface Finishes:
40		1. ACI SUI SUITACE FINISH SF-1.0: AS-Cast concrete texture imparted by form-facing material.
47		a. Patch voids larger than 1-1/2 inches wide or 1/2 inch deep.
48		b. Remove projections larger than 1 inch.
49		c. The holes do hot require patching.
		 Surface roleratice: Act 117 class D. Apply to concrete surfaces not expected to public view.
52		 Approximation contracts and exposed to public view. ACI 201 Surface Einish SE-2.0: Access concrete toyture imported by form facing material arranged in an
52		2. Act soft sufface transmistical manner with a minimum of seams
54		a Patch voids larger than 3/4 inch wide or 1/2 inch deen
55		h Remove projections larger than $1/4$ inch
56		c Patch tie holes
57		d. Surface Tolerance: ACI 117 Class B.

1			e. Locations: Apply to concrete surfaces exposed to public view .
2		3.	ACI 301 Surface Finish SF-3.0:
3			a. Patch voids larger than 3/4 inch wide or 1/2 inch deep.
4			b. Remove projections larger than 1/8 inch.
5			c. Patch tie holes.
6			d. Surface Tolerance: ACI 117 Class A.
7			e. Locations: Apply to concrete surfaces exposed to public view .
ð Q	3.8 FINIS		LOORS AND SLARS
10	A.	Comp	ly with ACI 302.1R recommendations for screeding, restraightening, and finishing operations for concrete
11		surfac	es. Do not wet concrete surfaces.
12	В.	Float I	Finish:
13		1.	When bleedwater sheen has disappeared and concrete surface has stiffened sufficiently to permit
14			operation of specific float apparatus, consolidate concrete surface with power-driven floats or by hand
15			floating if area is small or inaccessible to power-driven floats.
16		2.	Repeat float passes and restraightening until surface is left with a uniform, smooth, granular texture and
17			complies with ACI 117 tolerances for conventional concrete.
18		3.	Apply float finish to surfaces to receive trowel finish.
19	С.	Trowe	۱ Finish:
20		1.	After applying float finish, apply first troweling and consolidate concrete by hand or power-driven trowel.
21		2.	Continue troweling passes and restraighten until surface is free of trowel marks and uniform in texture and
22			appearance.
23		3.	Grind smooth any surface defects that would telegraph through applied coatings or floor coverings.
24		4.	Do not add water to concrete surface.
25		5.	Do not apply hard-troweled finish to concrete, which has a total air content greater than 3 percent.
26		6. 7	Apply a trowel finish to surfaces exposed to view .
27		7.	Finish surfaces to the following tolerances, in accordance with ASTIVIE1155, for a randomly trafficked floor
20			Sullace.
29			 a. Slabs off diound. 1) Einish and measure surface so gan at any point between concrete surface and an unleveled
30			freestanding 10-ft - long straightedge resting on two high spots and placed anywhere on
32			the surface does not exceed 1/8 inch
33			
34	3.9 INST	ALLATIC	IN OF MISCELLANEOUS CONCRETE ITEMS
35	Α.	Filling	In:
36		1.	Fill in holes and openings left in concrete structures after Work of other trades is in place unless otherwise
37			indicated.
38		2.	Mix, place, and cure concrete, as specified, to blend with in-place construction.
39		3.	Provide other miscellaneous concrete filling indicated or required to complete the Work.
40	В.	Curbs	: Provide monolithic finish to interior curbs by stripping forms while concrete is still green and by steel-
41		trowe	ling surfaces to a hard, dense finish with corners, intersections, and terminations slightly rounded.
42	C.	Equip	ment Bases and Foundations:
43		1.	Coordinate sizes and locations of concrete bases with actual equipment provided.
44		2.	Construct concrete bases 4 inches high unless otherwise indicated on Drawings, and extend base not less
45			than 6 inches in each direction beyond the maximum dimensions of supported equipment unless otherwise
46		2	Indicated on Drawings, or unless required for seismic anchor support.
47		3. ⊿	Install devide to connect concrete base to concrete floor. Unless otherwise indicated, install devide reds
40 10		4.	on 18-inch centers around the full perimeter of concrete base
50		5	For supported equipment install enorgy-coated anchor holts that extend through concrete base and anchor
51		5.	into structural concrete substrate.
52		6.	Prior to pouring concrete, place and secure anchorage devices.
53			a. Use setting drawings, templates, diagrams, instructions, and directions furnished with items to be
54			embedded.
55			b. Cast anchor-bolt insert into bases.
56			c. Install anchor bolts to elevations required for proper attachment to supported equipment.
57	D.	Steel I	Pan Stairs: Provide concrete fill for steel pan stair treads, landings, and associated items.
58		1.	Cast-in inserts and accessories, as shown on Drawings.

1		2.	Screed,	, tamp, a	and trowel finish concrete surfaces.
2					
3	3.10 CON	CRETE CL	JRING		
4	Α.	Protect	t freshly	placed of	concrete from premature drying and excessive cold or hot temperatures.
5		1.	Comply	with A	CI 301 and ACI 306.1 for cold weather protection during curing.
6		2.	Comply	with A	CI 301 and ACI 305.1 for hot-weather protection during curing.
7		3.	Mainta	in moist	ture loss no more than 0.2 lb/sq. ft. x h. calculated in accordance with ACI 305.1, before and
8		-	during	finishing	operations.
9	в	Curing	Formed	Surface	s: Comply with ACI 308 1 as follows:
10	5.	1	Cure fo	ormed co	oncrete surfaces including underside of beams supported slabs and other similar surfaces
11		2	Cure co	ncrete	containing color nigments in accordance with color nigment manufacturer's instructions
12		2.	If form	s romair	during curing pariod, maist cure after loosaning forms
12		J.	Ifromo	s i eiliail	routing curing period, moist cure after loosening forms.
10		4.	n remu	Continu	This before end of curing period, continue curing for remainder of curing period, as follows.
14			d. ⊾	Continu	Jous Fogging: Maintain standing water on concrete surface until final setting of concrete.
15			D.	Continu	uous sprinkling: Maintain concrete surface continuously wet.
16			с.	Absorp	tive Cover: Pre-dampen absorptive material before application; apply additional water to
1/				absorp	tive material to maintain concrete surface continuously wet.
18			d.	Water-	Retention Sheeting Materials: Cover exposed concrete surfaces with sheeting material,
19				taping,	or lapping seams.
20			e.	Membr	rane-Forming Curing Compound: Apply uniformly in continuous operation by power spray or
21				roller ir	n accordance with manufacturer's written instructions.
22				1)	Recoat areas subject to heavy rainfall within three hours after initial application.
23				2)	Maintain continuity of coating and repair damage during curing period.
24	С.	Curing	Unform	ed Surfa	ces: Comply with ACI 308.1 as follows:
25		1.	Begin c	uring im	mediately after finishing concrete.
26		2.	Interio	r Concre	te Floors:
27			a.	Floors	to Receive Floor Coverings Specified in Other Sections: Contractor has option of the
28				followi	ng:
29				1)	Absorptive Cover: As soon as concrete has sufficient set to permit application without
30					marring concrete surface, install prewetted absorptive cover over entire area of floor.
31					a) Lap edges and ends of absorptive cover not less than 12 inches.
32					b) Maintain absorptive cover water saturated, and in place, for duration of curing
33					period, but not less than seven days.
34				2)	Moisture-Retaining-Cover Curing: Cover concrete surfaces with moisture-retaining cover for
35				-,	curing concrete, placed in widest practicable width, with sides and ends lapped at least 12
36					inches and sealed by waterproof tape or adhesive
37					a) Immediately renair any holes or tears during curing period using cover material and
38					waterproof tane
30					b) Cure for not less than seven days
10				2)	Donding or Continuous Sprinkling of Water: Maintain concrete surfaces continuously wet for
40				5)	not loss than solven days, utilizing one or a combination of the following:
41					not less than seven days, utilizing one, or a combination of, the following.
42					d) Water.
43				F I	b) Continuous water-tog spray.
44			D.	FIOORS	to Receive Penetrating Liquid Floor Treatments: Contractor has option of the following:
45				1)	Absorptive Cover: As soon as concrete has sufficient set to permit application without
46					marring concrete surface, install prewetted absorptive cover over entire area of floor.
47					a) Lap edges and ends of absorptive cover not less than 12 inches.
48					b) Maintain absorptive cover water saturated, and in place, for duration of curing
49					period, but not less than seven days.
50				2)	Moisture-Retaining-Cover Curing: Cover concrete surfaces with moisture-retaining cover for
51					curing concrete, placed in widest practicable width, with sides and ends lapped at least 12
52					inches, and sealed by waterproof tape or adhesive.
53					a) Immediately repair any holes or tears during curing period, using cover material and
54					waterproof tape.
55					b) Cure for not less than seven days.
56				3)	Ponding or Continuous Sprinkling of Water: Maintain concrete surfaces continuously wet for
57					not less than seven days, utilizing one, or a combination of, the following:
58					a) Water.

1					b) Continuous water-fog spray.
2			с.	Floors	to Receive Polished Finish: Contractor has option of the following:
3				1)	Absorptive Cover: As soon as concrete has sufficient set to permit application without
4					marring concrete surface, install prewetted absorptive cover over entire area of floor.
5					a) Lap edges and ends of absorptive cover not less than 12 inches.
6					b) Maintain absorptive cover water saturated, and in place, for duration of curing
7					period, but not less than seven days.
8				2)	Ponding or Continuous Sprinkling of Water: Maintain concrete surfaces continuously wet for
9					not less than seven days, utilizing one, or a combination of, the following:
10					a) Water.
11					b) Continuous water-fog spray.
12			d.	Floors	to Receive Chemical Stain:
13				1)	As soon as concrete has sufficient set to permit application without marring concrete
14					surface, install curing paper over entire area of floor.
15				2)	Install curing paper square to building lines, without wrinkles, and in a single length without
16					end joints.
17				3)	Butt sides of curing paper tight; do not overlap sides of curing paper.
18				4)	Leave curing paper in place for duration of curing period, but not less than 28 days.
19			e.	Floors	to Receive Urethane Flooring:
20				1)	As soon as concrete has sufficient set to permit application without marring concrete
21					surface, install prewetted absorptive cover over entire area of floor.
22				2)	Rewet absorptive cover and cover immediately with polyethylene moisture-retaining cover
23				- 1	with edges lapped 6 inches and sealed in place.
24				3)	Secure polyethylene moisture-retaining cover in place to prohibit air from circulating under
25				- 1	polyethylene moisture-retaining cover.
26				4)	Leave absorptive cover and polyethylene moisture-retaining cover in place for duration of
27			<i>c</i>	-	curing period, but not less than 28 days.
28			t.	Floors	to Receive Curing Compound:
29				1)	Apply uniformly in continuous operation by power spray or roller in accordance with
30				2)	manufacturer's written instructions.
31				2)	Recoat areas subjected to heavy rainfall within three hours after initial application.
32				3)	Maintain continuity of coating, and repair damage during curing period.
33				4)	Removal: After curing period has elapsed, remove curing compound without damaging
34 25					concrete surfaces by method recommended by curing compound manufacturer unless
35					manufacturer certifies curing compound does not interfere with bonding of floor covering
30			_	Fleens	used on Project.
3/			g.	FIOORS	to Receive Curing and Sealing Compound:
38 20				I)	Apply uniformity to noors and slabs indicated in a continuous operation by power spray of roller in accordance with manufacturer's written instructions.
39				2)	Protect areas subjected to begin rainfall within three bours ofter initial application
40 41				2) 2)	Recoal areas subjected to neavy rainian within three hours after initial application.
+⊥ 42				J	and renair damage during curing period
42					מווים רבאמו ממווומשב ממוווש במרווש אבווסמי
44	3.11 TOLE	RANCES			
45	A.	Confor	m to AC	117.	
46		2211011		/·	
47	3.12 APPI	ICATION			OR TREATMENTS
48	A.	Penetra	ating Lic	auid Flo	or Treatment: Prepare, apply, and finish penetrating liquid floor treatment in accordance with
49		manufa	cturer'	s writte	n instructions.
50		1.	Remov	e curine	g compounds, sealers, oil, dirt, laitance, and other contaminants and complete surface repairs.
51		2.	Do not	apply t	o concrete that is less than 14 days' old.
52		3.	Apply	liquid ur	ntil surface is saturated, scrubbing into surface until a gel forms; rewet; and repeat brooming
53			or scru	bbing.	
54		4.	Rinse v	vith wat	er; remove excess material until surface is dry.
55		5.	Apply a	a secono	coat in a similar manner if the surface is rough or porous.
56					
57	3.13 JOIN	T FILLING			
58	Α.	Prepare	e, clean	, and ins	stall joint filler in accordance with manufacturer's written instructions.

1		1.	efer joint filling until concrete has aged at least six month(s).	
2		2.	o not fill joints until construction traffic has permanently ceas	sed.
3	В.	Remo	irt, debris, saw cuttings, curing compounds, and sealers from	n joints; leave contact faces of joints clean and
4		dry.		
5	С.	Install	nirigid joint filler full depth in saw-cut joints and at least 2 inc	ches deep in formed joints.
6	D.	Overfi	int, and trim joint filler flush with top of joint after hardening	7
7				·
8	3.14 CON	CRETE S	ACE REPAIRS	
9	Α.	Defec	Concrete:	
10		1.	epair and patch defective areas when approved by Architect.	
11		2.	emove and replace concrete that cannot be repaired and pat	ched to Architect's approval.
12	В.	Patch	Mortar: Mix dry pack patching mortar, consisting of 1 part r	portland cement to 2-1/2 parts fine aggregate
13		passir	No. 16 sieve, using only enough water for handling and placing	ng.
14	C.	Repai	Formed Surfaces: Surface defects include color and text	ure irregularities, cracks, spalls, air bubbles,
15		honey	ubs, rock pockets, fins and other projections on the surfa	ace, and stains and other discolorations that
16		canno	removed by cleaning.	
17		1.	mediately after form removal, cut out honeycombs, rock r	pockets, and voids more than $1/2$ inch in any
18			mension to solid concrete.	
19			Limit cut depth to 3/4 inch.	
20			Make edges of cuts perpendicular to concrete surface	
21			Clean dampen with water and brush-coat holes and y	roids with bonding agent
22			Fill and compact with patching mortar before bonding	agent has dried
23			Fill form-tie voids with patching mortar or cone plugs s	secured in place with bonding agent
24		2	enair defects on surfaces exposed to view by blending wh	hite portland cement and standard portland
25		2.	ment so that when dry natching mortar matches surround	ling color
26			Patch a test area at inconsnicuous locations to verify	misture and color match before proceeding
20			with natching	mixture and color materi before proceeding
27			Compact mortar in place and strike off slightly higher t	than surrounding surface
20		3	anair defects on concepted formed surfaces that will	affect concrete's durability and structural
20		э.	arformance as determined by Architect	anect concrete's durability and structural
30	П	Ronai	Informed Surfaces:	
32	D.	1	est unformed surfaces such as floors and slabs for finish ar	nd verify surface tolerances specified for each
22		1.		in verify surface tolerances specified for each
37			Correct low and high areas	
25			Tost surfaces sloped to drain for trueposs of slope and	smoothness: use a sloped template
35		n	rest surfaces sloped to drain for trueness of slope and	sinootimess, use a sloped template.
27		Ζ.	azing and cracks in excess of 0.01 inch wide or that ponet	s spalls, populis, noneycomos, rock pockets,
20			azing, and cracks in excess of 0.01 inch wide of that pener	value conditions
20		2	ter concrete has sured at least 14 days, correct high areas h	able conditions.
10		J. 1	prost localized low areas during, or immediately after com	r grinning.
40		4.	It low areas and replacing them with patching mortar	pleting surface-missing operations by cutting
41			Finish renaired areas to blend into adjacent concrete	
42		5	arrest other low areas scheduled to remain exposed with ren	airtanning
45		Э.	Cut out low areas to onsure a minimum repair toppi	ng donth of 1/4 inch to match adjacent floor
44			elevations	
45			Prenare mix and apply repair topping and primer	in accordance with manufacturer's written
40			instructions to produce a smooth uniform plane and	lovel surface
47 10		6	instructions to produce a smooth, uniform, plane, and	1 inch or loss in diamotor, by sutting out and
40		0.	placing with frosh concrete	I men of less in diameter, by cutting out and
49 E0			Pomovo defective areas with clean, square suts, and a	where steel reinforcement with at least a 2/4
50			inch clearance all around	spose steer remotement with at least a 3/4-
52			Dampan concrete surfaces in contact with patching co	ncrete and apply bonding agent
52			Mix patching concrete of same materials and mixture	a as original concrete, except without secret
55				e as original concrete, except without coalse
54			aggregate. Diace compact and finish to blond with adjacent finish	hed concrete
55			Cure in same manner as adjacent concrete	
50		7	cure in same manner as dujdtent tuntiete.	er with natching mortar
57		1.	-pair random cracks and single notes I men of less in aldifield	ci with patching mortal.

1		a. Groove top of cracks and cut out holes to sound concrete, and clean off dust, dirt, and loose
2		particles.
3		 Dampen cleaned concrete surfaces and apply bonding agent.
4		c. Place patching mortar before bonding agent has dried.
5		d. Compact patching mortar and finish to match adjacent concrete.
6	_	e. Keep patched area continuously moist for at least 72 hours.
7	Ε.	Perform structural repairs of concrete, subject to Architect's approval, using epoxy adhesive and patching mortar.
8	F.	Repair materials and installation not specified above may be used, subject to Architect's approval.
9		
10	3.15 FIELD	D QUALITY CONTROL
11	Α.	Testing Agency: Engage a qualified testing and inspecting agency to perform tests and inspections and to submit
12		reports.
13		1. Testing agency to be responsible for providing curing container for composite samples on Site and verifying
14		that field-cured composite samples are cured in accordance with ASTM C31/C31M.
15		2. Testing agency to immediately report to Architect, Contractor, and concrete manufacturer any failure of
16		Work to comply with Contract Documents.
17		3. Testing agency to report results of tests and inspections, in writing, to Owner, Architect, Contractor, and
18		concrete manufacturer within 48 hours of inspections and tests.
19		a. Test reports to include reporting requirements of ASTM C31/C31M, ASTM C39/C39M, and ACI 301
20		including the following as applicable to each test and inspection:
21		1) Project name.
22		2) Name of testing agency.
23		3) Names and certification numbers of field and laboratory technicians performing inspections
24		and testing.
25		4) Name of concrete manufacturer.
26		5) Date and time of inspection, sampling, and field testing.
27		6) Date and time of concrete placement.
28		7) Location in Work of concrete represented by samples.
29		8) Date and time sample was obtained.
30		9) Truck and batch ticket numbers.
31		10) Design compressive strength at 28 days.
32		11) Concrete mixture designation, proportions, and materials.
33		12) Field test results.
34		13) Information on storage and curing of samples before testing, including curing method and
35		maximum and minimum temperatures during initial curing period.
36		14) Type of fracture and compressive break strengths at seven days and 28 days.
37	В.	Batch Tickets: For each load delivered, submit three copies of batch delivery ticket to testing agency, indicating
38		quantity, mix identification, admixtures, design strength, aggregate size, design air content, design slump at time of
39		batching, and amount of water that can be added at Project site.
40	C.	Concrete Tests: Testing of composite samples of fresh concrete obtained in accordance with ASTM C 172/C 172M to
41		be performed in accordance with the following requirements:
42		1. Testing Frequency: Obtain one composite sample for each day's pour of each concrete mixture exceeding 5
43		cu. yd., but less than 25 cu. yd., plus one set for each additional 50 cu. yd. or fraction thereof.
44		a. When frequency of testing provides fewer than five compressive-strength tests for each concrete
45		mixture, testing to be conducted from at least five randomly selected batches or from each batch in
46		fewer than five are used.
47		2. Slump: ASTM C143/C143M:
48		a. One test at point of placement for each composite sample, but not less than one test for each day's
49		pour of each concrete mixture.
50		b. Perform additional tests when concrete consistency appears to change.
51		3. Air Content: ASTM C231/C231M pressure method. for normal-weight concrete.
52		a. One test for each composite sample, but not less than one test for each day's pour of each concrete
53		mixture.
54		4. Concrete Temperature: ASTM C1064/C1064M:
55		a. One test hourly when air temperature is 40 deg F and below or 80 deg F and above and one test for
56		each composite sample.
57		5. Compression Test Specimens: ASTM C31/C31M:

1			a. Cast and laboratory cure two sets of two 6-inch by 12-inch or 4-inch by 8-inch cylinder specimens
2			for each composite sample.
3		6.	Compressive-Strength Tests: ASTM C39/C39M.
4			a. Test one set of two laboratory-cured specimens at seven days and one set of two specimens at 28
5			days.
6			b. A compressive-strength test to be the average compressive strength from a set of two specimens
7			obtained from same composite sample and tested at age indicated.
8		7.	When strength of field-cured cylinders is less than 85 percent of companion laboratory-cured cylinders,
9			Contractor to evaluate operations and provide corrective procedures for protecting and curing in-place
10			concrete.
11		8.	Strength of each concrete mixture will be satisfactory if every average of any three consecutive
12			compressive-strength tests equals or exceeds specified compressive strength, and no compressive-strength
13			test value falls below specified compressive strength by more than 500 psi if specified compressive strength
14			is 5000 psi, or no compressive strength test value is less than 10 percent of specified compressive strength if
15			specified compressive strength is greater than 5000 psi.
16		9.	Nondestructive Testing: Impact hammer, sonoscope, or other nondestructive device may be permitted by
17			Architect but will not be used as sole basis for approval or rejection of concrete.
18		10.	Additional Tests:
19			a. Testing and inspecting agency to make additional tests of concrete when test results indicate that
20			slump, air entrainment, compressive strengths, or other requirements have not been met, as
21			directed by Architect.
22			b. Testing and inspecting agency may conduct tests to determine adequacy of concrete by cored
23			cylinders complying with ASTM C42/C42M or by other methods as directed by Architect.
24			1) Acceptance criteria for concrete strength to be in accordance with ACI 301, Section 1.6.6.3.
25		11.	Additional testing and inspecting, at Contractor's expense, will be performed to determine compliance of
26			replaced or additional work with specified requirements.
27		12.	Correct deficiencies in the Work that test reports and inspections indicate do not comply with the Contract
28			Documents.
29	D.	Measu	are floor and slab flatness and levelness in accordance with ASTM E1155 within 24 hours of completion of
30		floor fi	inishing and promptly report test results to Architect.
31			
32	3.16 PRO	FECTION	
33	Α.	Protec	t concrete surfaces as follows:
34		1.	Protect from petroleum stains.
35		2.	Diaper hydraulic equipment used over concrete surfaces.
36		3.	Prohibit vehicles from interior concrete slabs.
37		4.	Prohibit use of pipe-cutting machinery over concrete surfaces.
38		5.	Prohibit placement of steel items on concrete surfaces.
39		6.	Prohibit use of acids or acidic detergents over concrete surfaces.
40		7.	Protect liquid floor treatment from damage and wear during the remainder of construction period. Use
41			protective methods and materials, including temporary covering, recommended in writing by liquid floor
42		0	treatments installer.
43		8.	Protect concrete surfaces scheduled to receive surface hardener or polished concrete finish using Floor Slab
44			Protective Covering.
45			
46			END OF SECTION

Page Intentionally Left Blank

1

2

SECTION 03 33 00 ARCHITECTURAL CONCRETE

PART 1 -	GENERAL
1.1	SUMMARY
1.2	DEFINITIONS
1.3	PREINSTALLATION MEETINGS
1.4	ACTION SUBMITTALS
1.5	INFORMATIONAL SUBMITTALS
1.6	QUALITY ASSURANCE
1.7	PRECONSTRUCTION TESTING
1.8	DELIVERY, STORAGE, AND HANDLING
1.9	FIELD CONDITIONS
PART 2 -	PRODUCTS
2.1	CONCRETE, GENERAL
2.2	FORM-FACING MATERIALS
2.3	REINFORCEMENT ACCESSORIES
2.4	CONCRETE MATERIALS
2.5	CURING MATERIALS
2.6	REPAIR MATERIALS
2.7	CONCRETE MIXTURES, GENERAL
2.8	CONCRETE MIXTURES
2.9	CONCRETE MIXING
PART 3 -	EXECUTION
3.1	INSTALLATION OF FORMWORK
3.2	INSTALLATION OF REINFORCEMENT AND ACCESSORIES
3.3	JOINTS
3.4	CONCRETE PLACEMENT
3.5	FINISHING FORMED SURFACES
3.6	CONCRETE CURING
3.7	REPAIR
3.8	FIELD QUALITY CONTROL
3.9	CLEANING
3.10	PROTECTION
0.20	FINAL ACCEPTANCE
3 11	7
0.11	
<u> PART 1 -</u>	GENERAL
1.1	SUMMARY
Α.	Section Includes:
	1. Cast-in-place architectural concrete, including form facings, reinforcement accessories. concrete materia
	concrete mixtures, concrete placement, and concrete finishes.
	2. Requirements in Section 033000 "Cast-in-Place Concrete" apply to this Section.
1.2	
A.	Last-In-Place Architectural Concrete: Concrete that is exposed to view, is designated as architectural concrete, a
	that requires special concrete materials, formwork, placement, or finishes to obtain specified architectu appearance.
R	Cementitions Materials: Portland cement alone or in combination with one or more of the following: blend
υ.	hydraulic cement fly ash slag cement other nozzolans and silica fume materials subject to compliance w
	requirements
C	Water/Coment Ratio (w/cm): The ratio by weight of water to comentitious materials
ι.	water/cement hatio (w/tin). The fatio by weight of water to tementitious indicidits.
1.3	PREINSTALLATION MEETINGS
-	Desirate lletion. Conferences Conduct and forman at Desirate the

- 56 A. Preinstallation Conference: Conduct conference at **Project site**.
- 571.Require representatives of each entity directly concerned with cast-in-place architectural concrete to
attend, including the following:

1			a. Contractor's superintendent.
2			 Independent testing agency responsible for concrete design mixtures.
3			c. Ready-mix concrete manufacturer.
4			d. Cast-in-place architectural concrete Subcontractor.
5		2.	Review the following:
6			 Construction joints, control joints, isolation joints, and joint-filler strips.
7			b. Reinforcement accessory installation.
8			c. Cold- and hot-weather concreting procedures.
9			d. Concrete finishes and finishing.
10			e. Curing procedures.
11			f. Forms and form-removal limitations.
12			g. Shoring and reshoring procedures.
13			h Concrete repair procedures
14			i Protection of cast-in-place architectural concrete
15			i Initial curing and field curing of field test cylinders (ASTM C31/C31M)
16			J. Initial caring and field caring of field test cylinders k Distriction of field-cured field test cylinders
10			k. Protection of held-caled held test cylinders.
10 10	1 /	ΛΟΤΙΟ	
10	1.4	Brodu	rt Data: For each of the following:
20	А.	1	
20		1.	Form-facing panels.
21		2.	Form liners.
22		3.	Form Joint Tape.
23		4.	Form Joint sealant.
24		5.	Wood sealer.
25		6.	Form-release agent.
26		7.	Surface retarder.
27		8.	Form ties.
28		9.	Bar supports.
29		10.	Portland cement.
30		11.	Fly ash.
31		12.	Slag cement.
32		13.	Aggregates.
33		14.	Admixtures:
34			a. Include limitations of use, including restrictions on cementitious materials, supplementary
35			cementitious materials, air entrainment, aggregates, temperature at time of concrete placement.
36			relative humidity at time of concrete placement, curing conditions, and use of other admixtures.
37		15.	Repair materials.
38	в	Design	Mixtures: For each concrete mixture, include the following:
39	υ.	1	Mixture identification
40		2	Minimum 28-day compressive strength
4 0 Л1		2.	Nurability exposure class
41 12		J. ⊿	Maximum w/cm
42 40		4. r	
43		5. C	Sump innit.
44		ь. ¬	Air content.
45		7.	Nominal maximum aggregate size.
46		8.	Amounts of mixing water to be withheld for later addition at Project site if permitted.
47		9.	Intended placement method.
48		10.	Alternative design mixtures when characteristics of materials, Project conditions, weather, test results, or
49			other circumstances warrant adjustments.
50	С.	Shop [Drawings:
51		1.	Formwork: Prepared by, and signed and sealed by, a qualified professional engineer responsible for their
52			preparation, detailing fabrication, assembly, and support of forms.
53			a. Show formwork construction, including form-liner layout, form-liner termination details,
54			dimensioned locations of form-facing material joints, rustications, construction and contraction
55			joints, form joint-sealant details, form-tie locations and patterns, inserts and embedments, cutouts,
56			cleanout panels, and other items that visually affect cast-in-place architectural concrete.
57			b. ct cast-in-place architectural concrete.

1 2		1) Indicate proposed schedule and sequence of stripping of forms, shoring removal, and reshoring installation and removal.
3		2) Location of construction ionists is subject to approval of Architect
1	П	Samples for Verifications, Architectural concrete Samples cast vertically, approximately 18 by 18 by 2 inches (450 by
5	D.	450 by 50 mm) of finisher colors and textures to match design reference sample. Jourday ample statistic showing the
5		full range of variations expression in these characteristics
о 7		Tui range of variations expected in these characteristics.
8	1.5	INFORMATIONAL SUBMITTALS
9	Α.	Qualification Data: For the following:
10		1. Installer: Include copies of applicable ACI certificates.
11		2. Ready-mixed concrete manufacturer.
12	В.	Material Certificates: For each of the following:
13		1. Cementitious materials.
14		2. Admixtures.
15		3. Form materials and form-release agents.
16		4. Renair materials.
17	C	Material Test Reports: For the following, by a qualified testing agency:
18	с.	1 Portland cement
19		2 Fly ash
20		2. Sing company
20		A Aggregator
21	D	4. Aggregates.
22	D. F	Research reports, For Concrete administures in accordance with ICC AC136.
25	с. г	Preconstruction rest reports, nor each mix design.
24	F.	Concrete Repair: Submit a written, detailed description of materials, methods, equipment, and sequence of
25		operations to be used for repairing architectural concrete, including protection of surrounding materials and Project
26		site.
27		1. If materials and methods other than those indicated are proposed for any repairs to architectural concrete,
28		add a written description of such materials and methods, including evidence of successful use on
29		comparable projects, and demonstrations to show their effectiveness for this Project and Installer's ability
30		to use such materials and methods properly.
31		Project and Installer's ability to use such materials and methods properly.
32	G.	Minutes of preinstallation conference.
33		
34	1.6	QUALITY ASSURANCE
35	Α.	Ready-Mixed Concrete Manufacturer Qualifications: A firm experienced in manufacturing ready-mixed concrete
36		products and that complies with ASTM C94/C94M requirements for production facilities and equipment.
37		1. Manufacturer certified in accordance with NRMCA's "Certification of Ready Mixed Concrete Production
38		Facilities."
39	В.	Installer Qualifications: An experienced cast-in-place architectural concrete installer, as evidenced by not less than
40		five consecutive years' experience, specializing in installing cast-in-place architectural concrete similar in material,
41		design, and extent to that indicated for this Project and whose work has resulted in construction with a record of
42		successful in-service performance.
43		1. Provide written evidence of qualifications and experience.
44		2. Include locations, descriptions, and photographs of completed projects, including name of architect,
45		substantiating the quality of the installer's experience.
46	С.	Laboratory Testing Agency Qualifications: A testing agency qualified in accordance with ASTM C1077 and
47		ASTM E329 for testing indicated and employing an ACI-certified Concrete Quality Technical Manager.
48		1. Personnel performing laboratory tests to be an ACI-certified Concrete Strength Testing Technician and
49		Concrete Laboratory Testing Technician, Level I.
50		2 Testing agency laboratory supervisor to be an ACI-certified Concrete Laboratory Testing Technician Level II
51	П	Field Sample Panels: After approval of verification sample and before casting architectural concrete produce field
52	5.	sample panels to demonstrate the approved range of selections made under Sample submittals. Produce a
53		minimum of three sets of full-scale nanels, cast vertically, approximately 48 by 48 by 6 inches (1200 by 1200 by 150
54		mm) minimum to demonstrate the expected range of finish color, and texture variations
55		1 I ocate papels as indicated or if not indicated as directed by Architect
56		Demonstrate methods of curing aggregate exposure wood scalars, and coatings, as applicable
20		2. Demonstrate methods of carme, aggregate exposure, wood searchs, and coatings, as applicable.

1 2		3. In presence of Architect, damage part of an exposed-face surface for each finish, color, and texture, and demonstrate materials and techniques proposed for repair of tie holes and surface blemishes to match
3		adjacent undamaged surfaces.
4 5		4. Maintain field sample panels during construction in an undisturbed condition as a standard for judging the completed Work.
6		5. Demolish and remove field sample panels when directed.
7		
8	1.7	PRECONSTRUCTION TESTING
9	Α.	Preconstruction Testing Service: Engage a qualified testing agency to perform preconstruction testing on concrete
10		mixtures.
11		1. Include the following information in each test report:
12		a. Admixture dosage rates.
13		b. Slump.
14		c. Air content.
15		d. Seven-day compressive strength.
16		e. 28-day compressive strength.
1/		f. Permeability.
18	10	
19	1.8	DELIVERT, STORAGE, AND HANDLING
20	А.	Comply with ASTM C94/C94IVI and ACI 301 (ACI 301IVI).
21	10	
22	1.9	FIELD CONDITIONS
23	А. D	Hot Weather Placement: Comply with Section 033000 "Cast-in-Flace Concrete."
24	Б.	Hot-Weather Placement. Comply with Section 055000 Cast-III-Place Concrete.
25	PART 2 -	PRODUCTS
27	<u></u>	
28	2.1	CONCRETE, GENERAL
29	Α.	ACI Publications: Comply with ACI 301 (ACI 301M) unless modified by requirements in the Contract Documents.
30		
31	2.2	FORM-FACING MATERIALS
32	Α.	Comply with Section 031000 "Concrete Forming and Accessories" for formwork and other form-facing material
33		requirements, and as specified in this Section.
34	В.	Source Limitations: Obtain each type of form-facing material from single source from single manufacturer.
35		
36	2.3	REINFORCEMENT ACCESSORIES
37	Α.	Bar Supports: Bolsters, chairs, spacers, and other devices for spacing, supporting, and fastening reinforcing bars and
38		welded-wire fabric in place.
39		1. Manufacture bar supports in accordance with CRSI's "Manual of Standard Practice."
40		2. Where legs of wire bar supports contact forms, use gray, all-plastic bar supports.
41		
42	2.4	CONCRETE MATERIALS
43	Α.	Cementitious Materials:
44		 Portland Cement: ASTM C150/C150M, Type I/IIgray.
45		2. Fly Ash: ASTM C618, Class C or Class F.
46	В.	Normal-Weight Aggregates: ASTM C33/C33M, Class 5S coarse aggregate or better, graded. Provide aggregates from
47		single source from single manufacturer.
48		1. Maximum Coarse-Aggregate Size: 1 inch (25 mm).
49		2. Gradation: Uniformly graded.
50	С.	Normal-Weight Fine Aggregate: ASTM C33/C33M, manufactured or natural sand, free of materials with deleterious
51		reactivity to alkali in cement, from same source for entire Project.
52	D.	Air-Entraining Admixture: As specified in Section 033000 "Cast-in-Place Concrete."
53	Ε.	Chemical Admixtures: As specified in Section 033000 "Cast-in-Place Concrete," and certified by manufacturer to be
54		compatible with other admixtures and that do not contribute water-soluble chloride ions exceeding those
55		permitted in hardened concrete. Do not use calcium chloride or admixtures containing calcium chloride.
56	F.	Water and Water Used to Make Ice: ASTM C94/C94M, potable.
57		
1	2.5	CURING MATERIALS
-----------	------------------	--
2	Α.	Comply with Section 03 30 00 "Cast-in-Place Concrete."
3		1. For integrally colored concrete, curing materials to be approved by color pigment manufacturer.
4		2. For concrete indicated to be sealed, curing materials to be compatible with sealer.
5		
6	2.6	REPAIR MATERIALS
7	Α.	Bonding Agent: ASTM C1059/C1059M, Type II, nonredispersible, acrylic emulsion or styrene butadiene.
8		
9	2.7	CONCRETE MIXTURES, GENERAL
10	Α.	Obtain each color, size, type, and variety of concrete mixture from single manufacturer with resources to provide
11		cast-in-place architectural concrete of consistent quality in appearance and physical properties.
12	В.	Prepare design mixtures for each type and strength of cast-in-place architectural concrete proportioned on basis of
13		laboratory trial mixture or field test data, or both, in accordance with ACI 301 (ACI 301M).
14		1. Use a qualified independent testing agency for preparing and reporting proposed concrete mixture designs,
15		based on laboratory trial mixtures.
16	С.	Admixtures: Use admixtures in accordance with manufacturer's written instructions.
1/ 10	28	
19	2.0	Class K: Normal-weight concrete
20	Π.	$1 \qquad \text{Exposure Class: ACI 318 (ACI 318M) F1}$
20 21		 Exposure class. Act 516 (Act 5160) 11. Minimum Compressive Strength: 4000 nci (27.6 MPa) at 28 days
22		3 Maximum w/cm: 0.45
23		4 Slump Limit: 4 inches (100 mm) , plus or minus 1 inch (25 mm)
24		5 Air Content:
25		a. Exposure Class F1: 5.0 percent. plus or minus 1.5 percent at point of delivery for concrete
26		containing 3/4-inch (19-mm) nominal maximum aggregate size.
27		 Limit water-soluble, chloride-ion content in hardened concrete to 1.00 percent by weight of cement.
28		
29	2.9	CONCRETE MIXING
30	Α.	Ready-Mixed Architectural Concrete: Measure, batch, mix, and deliver concrete in accordance with
31		ASTM C94/C94M, and furnish batch ticket information.
32		1. Clean equipment used to mix and deliver cast-in-place architectural concrete to prevent contamination
33		from other concrete.
34		2. For mixer capacity of 1 cu. yd. (0.76 cu. m) or smaller, continue mixing at least 1-1/2 minutes, but not more
35		than five minutes after ingredients are in mixer, before any part of batch is released.
36		3. For mixer capacity larger than 1 cu. yd. (0.76 cu. m), increase mixing time by 15 seconds for each additional
37		1 cu. yd. (0.76 cu. m).
38		4. Provide batch ticket for each batch discharged and used in the Work, indicating Project identification name
39		and number, date, mixture type, mixture time, quantity, and amount of water added. Record approximate
10 11		location of final deposit in structure.
12	<u> PART 3 -</u>	EXECUTION
13	• •	
14	3.1	INSTALLATION OF FORMWORK
15	Α.	Comply with Section 031000 "Concrete Forming and Accessories" for formwork, embedded items, and shoring and
46		reshoring, and as specified in this Section.
47	В.	Limit deflection of form-facing panels to not exceed ACI 301 (ACI 301M) requirements.
48	C.	Limit cast-in-place architectural concrete surface irregularities, as follows:
49		1. Surface Finish-1.0: ACI 117 (ACI 117M) Class D, 1 inch (25 mm).
5U - 1		2. Surface Finish-2.0: ACI 117 (ACI 117M) Class B, 1/4 inch (6 mm).
51 - 2	_	3. Surrace Finish-3.0: ACI 117 (ACI 117M) Class A, 1/8 inch (3.0 mm).
52 - 2	D.	Construct forms to result in cast-in-place architectural concrete that complies with ACI 11/ (ACI 11/M).
od - ∕	E.	Seal form joints, chamfers, rustication joints, and penetrations at form ties with form joint tape or form joint sealant
54 		to prevent cement paste leakage.
5		1. Provide closure backing materials if indented rustication is used over a ribbed form line, and seal joint
	-	between rustication strip and form with joint sealant.
ו כ	F.	Chamier exterior corners and edges of cast-in-place architectural concrete.

1 2	G.	Coat contact surfaces of wood rustications and chamfer strips with wood sealer before placing reinforcement, anchoring devices, and embedded items.
3 4 5	H.	Coat contact surfaces of forms with form-release agent, in accordance with manufacturer's written instructions, before placing reinforcement, anchoring devices, and embedded items.
6 7 8 9	3.2 A.	INSTALLATION OF REINFORCEMENT AND ACCESSORIES Comply with Section 032000 "Concrete Reinforcing" for fabricating and installing steel reinforcement and accessories.
10	3.3	JOINTS
11 12 13	Α.	Construction Joints: Install construction joints true to line, with faces perpendicular to surface plane of cast-in-place architectural concrete, so strength and appearance of concrete are not impaired, at locations indicated or as approved by Architect.
14 15		1. Place joints perpendicular to main reinforcement. Continue reinforcement across construction joints unless otherwise indicated.
16 17		2. Form keyed joints as indicated. Embed keys at least 1-1/2 inches (38 mm) into concrete. Align construction joint within rustications attached to form-facing material.
18 19		 Locate joints for beams, slabs, joists, and girders at third points of spans. Offset joints in girders a minimum distance of twice the beam width from a beam-girder intersection.
20 21 22		 Locate horizontal joints in walls and columns at underside of floors, slabs, beams, and girders and at top of footings or floor slabs. Space vertical joints in walls as indicated on Drawings, Unless otherwise indicated on Drawings, locate
22 23 24		 Space vertical joints in wais as indicated on Drawings. Onless otherwise indicated on Drawings, locate joints beside piers integral with walls, near corners, and in concealed locations where possible. Use bonding agent at locations where fresh concrete is placed against hardened or partially hardened concrete surfaces.
26 27 28	В.	Contraction Joints: Form weakened-plane contraction joints true to line, with faces perpendicular to surface plane of cast-in-place architectural concrete, so strength and appearance of concrete are not impaired, at locations indicated on Drawings or as approved by Architect.
29	2.4	
50 21	5.4	Concrete Placetvieni
32	А.	Comply with Section 033000 Cast-in-Place Concrete.
31 32 33	A. 3.5	FINISHING FORMED SURFACES
31 32 33 34	A. 3.5 A.	FINISHING FORMED SURFACES Comply with Section 033000 "Cast-in-Place Concrete."
31 32 33 34 35 36	А. 3.5 А. В.	FINISHING FORMED SURFACES Comply with Section 033000 "Cast-in-Place Concrete." Architectural Concrete Finish: Match Architect's design reference sample, identified and described as indicated, to satisfaction of Architect.
31 32 33 34 35 36 37 38 39	А. 3.5 А. В. С.	 FINISHING FORMED SURFACES Comply with Section 033000 "Cast-in-Place Concrete." Architectural Concrete Finish: Match Architect's design reference sample, identified and described as indicated, to satisfaction of Architect. As-Cast Surface Finishes: Comply with Section 033000 "Cast-in-Place Concrete" for the following: ACI 301 (ACI 301M) Surface Finish-1.0 (SF-1.0). ACI 301 (ACI 301M) Surface Finish-2.0 (SF-2.0).
31 32 33 34 35 36 37 38 39 40 41 42	A. 3.5 A. B. C. D.	 FINISHING FORMED SURFACES Comply with Section 033000 "Cast-in-Place Concrete." Architectural Concrete Finish: Match Architect's design reference sample, identified and described as indicated, to satisfaction of Architect. As-Cast Surface Finishes: Comply with Section 033000 "Cast-in-Place Concrete" for the following: ACI 301 (ACI 301M) Surface Finish-1.0 (SF-1.0). ACI 301 (ACI 301M) Surface Finish-2.0 (SF-2.0). ACI 301 (ACI 301M) Surface Finish-3.0 (SF-3.0). Final Concrete Finish: Comply with Section 033000 "Cast-in-Place Concrete" for the following: Smooth-rubbed finish.
31 32 33 34 35 36 37 38 39 40 41 42 43 44	A. 3.5 B. C. D. E.	 FINISHING FORMED SURFACES Comply with Section 033000 "Cast-in-Place Concrete." Architectural Concrete Finish: Match Architect's design reference sample, identified and described as indicated, to satisfaction of Architect. As-Cast Surface Finishes: Comply with Section 033000 "Cast-in-Place Concrete" for the following: ACI 301 (ACI 301M) Surface Finish-1.0 (SF-1.0). ACI 301 (ACI 301M) Surface Finish-2.0 (SF-2.0). ACI 301 (ACI 301M) Surface Finish-3.0 (SF-3.0). Final Concrete Finish: Comply with Section 033000 "Cast-in-Place Concrete" for the following: Smooth-rubbed finish. Form-Liner Finish: Produce a textured surface free of pockets, streaks, and honeycombs, and of uniform appearance, color, and texture.
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46	A. 3.5 B. C. D. E. F.	 FINISHING FORMED SURFACES Comply with Section 033000 "Cast-in-Place Concrete." Architectural Concrete Finish: Match Architect's design reference sample, identified and described as indicated, to satisfaction of Architect. As-Cast Surface Finishes: Comply with Section 033000 "Cast-in-Place Concrete" for the following: ACI 301 (ACI 301M) Surface Finish-1.0 (SF-1.0). ACI 301 (ACI 301M) Surface Finish-2.0 (SF-2.0). ACI 301 (ACI 301M) Surface Finish-3.0 (SF-3.0). Final Concrete Finish: Comply with Section 033000 "Cast-in-Place Concrete" for the following: Smooth-rubbed finish. Form-Liner Finish: Produce a textured surface free of pockets, streaks, and honeycombs, and of uniform appearance, color, and texture. Maintain uniformity of architectural concrete finishes over construction joints unless otherwise indicated.
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47	A. 3.5 A. B. C. D. E. F. 3.6	 FINISHING FORMED SURFACES Comply with Section 033000 "Cast-in-Place Concrete." Architectural Concrete Finish: Match Architect's design reference sample, identified and described as indicated, to satisfaction of Architect. As-Cast Surface Finishes: Comply with Section 033000 "Cast-in-Place Concrete" for the following: ACI 301 (ACI 301M) Surface Finish-1.0 (SF-1.0). ACI 301 (ACI 301M) Surface Finish-2.0 (SF-2.0). ACI 301 (ACI 301M) Surface Finish-3.0 (SF-3.0). Final Concrete Finish: Comply with Section 033000 "Cast-in-Place Concrete" for the following: Smooth-rubbed finish. Form-Liner Finish: Produce a textured surface free of pockets, streaks, and honeycombs, and of uniform appearance, color, and texture. Maintain uniformity of architectural concrete finishes over construction joints unless otherwise indicated.
31 32 33 34 35 36 37 38 39 40 41 42 43 44 50	A. 3.5 A. B. C. D. E. F. 3.6 A.	 FINISHING FORMED SURFACES Comply with Section 033000 "Cast-in-Place Concrete." Architectural Concrete Finish: Match Architect's design reference sample, identified and described as indicated, to satisfaction of Architect. As-Cast Surface Finishes: Comply with Section 033000 "Cast-in-Place Concrete" for the following: ACI 301 (ACI 301M) Surface Finish-1.0 (SF-1.0). ACI 301 (ACI 301M) Surface Finish-2.0 (SF-2.0). ACI 301 (ACI 301M) Surface Finish-3.0 (SF-3.0). Final Concrete Finish: Comply with Section 033000 "Cast-in-Place Concrete" for the following: Smooth-rubbed finish. Form-Liner Finish: Produce a textured surface free of pockets, streaks, and honeycombs, and of uniform appearance, color, and texture. Maintain uniformity of architectural concrete finishes over construction joints unless otherwise indicated. CONCRETE CURING Comply with Section 033000 "Cast-in-Place Concrete" using identical curing procedures to that used for field sample panels.
31 32 33 34 35 36 37 38 39 40 41 42 43 44 50 51	A. 3.5 A. B. C. D. E. F. 3.6 A. 3.7	 Comply with Section 033000 "Cast-in-Place Concrete." FINISHING FORMED SURFACES Comply with Section 033000 "Cast-in-Place Concrete." Architectural Concrete Finish: Match Architect's design reference sample, identified and described as indicated, to satisfaction of Architect. As-Cast Surface Finishes: Comply with Section 033000 "Cast-in-Place Concrete" for the following: ACI 301 (ACI 301M) Surface Finish-1.0 (SF-1.0). ACI 301 (ACI 301M) Surface Finish-2.0 (SF-2.0). ACI 301 (ACI 301M) Surface Finish-3.0 (SF-3.0). Final Concrete Finish: Comply with Section 033000 "Cast-in-Place Concrete" for the following: Smooth-rubbed finish. Form-Liner Finish: Produce a textured surface free of pockets, streaks, and honeycombs, and of uniform appearance, color, and texture. Maintain uniformity of architectural concrete finishes over construction joints unless otherwise indicated. CONCRETE CURING Comply with Section 033000 "Cast-in-Place Concrete" using identical curing procedures to that used for field sample panels. REPAIR
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 7 89 50 51 52	A. 3.5 A. B. C. D. E. F. 3.6 A. 3.7 A.	 Comply with Section 033000 "Cast-in-Place Concrete." FINISHING FORMED SURFACES Comply with Section 033000 "Cast-in-Place Concrete." Architectural Concrete Finish: Match Architect's design reference sample, identified and described as indicated, to satisfaction of Architect. As-Cast Surface Finishes: Comply with Section 033000 "Cast-in-Place Concrete" for the following: ACI 301 (ACI 301M) Surface Finish-1.0 (SF-1.0). ACI 301 (ACI 301M) Surface Finish-2.0 (SF-2.0). ACI 301 (ACI 301M) Surface Finish-3.0 (SF-3.0). Final Concrete Finish: Comply with Section 033000 "Cast-in-Place Concrete" for the following: Smooth-rubbed finish. Form-Liner Finish: Produce a textured surface free of pockets, streaks, and honeycombs, and of uniform appearance, color, and texture. Maintain uniformity of architectural concrete finishes over construction joints unless otherwise indicated. CONCRETE CURING Comply with Section 033000 "Cast-in-Place Concrete" using identical curing procedures to that used for field sample panels. REPAIR Comply with ACI 301 (ACI 301M).
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 89 50 51 52 53	A. 3.5 A. B. C. D. E. F. 3.6 A. 3.7 A. B.	 Comply with Section 033000 "Cast-in-Place Concrete." FINISHING FORMED SURFACES Comply with Section 033000 "Cast-in-Place Concrete." Architectural Concrete Finish: Match Architect's design reference sample, identified and described as indicated, to satisfaction of Architect. As-Cast Surface Finishes: Comply with Section 033000 "Cast-in-Place Concrete" for the following:
31 32 33 34 35 36 37 38 39 40 41 42 43 44 50 51 52 53 55 55	A. 3.5 A. B. C. D. E. F. 3.6 A. 3.7 A. B. C.	 Comply with Section 033000 'Cast-in-Place Concrete." FINISHING FORMED SURFACES Comply with Section 033000 "Cast-in-Place Concrete." Architectural Concrete Finish: Match Architect's design reference sample, identified and described as indicated, to satisfaction of Architect. As-Cast Surface Finishes: Comply with Section 033000 "Cast-in-Place Concrete" for the following: ACI 301 (ACI 301M) Surface Finish-1.0 (SF-1.0). ACI 301 (ACI 301M) Surface Finish-2.0 (SF-2.0). ACI 301 (ACI 301M) Surface Finish-3.0 (SF-3.0). Final Concrete Finish: Comply with Section 033000 "Cast-in-Place Concrete" for the following: Smooth-rubbed finish. Form-Liner Finish: Produce a textured surface free of pockets, streaks, and honeycombs, and of uniform appearance, color, and texture. Maintain uniformity of architectural concrete finishes over construction joints unless otherwise indicated. CONCRETE CURING Comply with Section 033000 "Cast-in-Place Concrete" using identical curing procedures to that used for field sample panels. REPAIR Comply with ACI 301 (ACI 301M). Repair damaged finished surfaces of cast-in-place architectural concrete when repairing is approved by Architect Match repairs to color, texture, and uniformity of surrounding surfaces and to repairs on approved field sample panels.

1	3.8	FIELD QUALITY CONTROL
2	Α.	Comply with Section 033000 "Cast-in-Place Concrete."
3		
4	3.9	CLEANING
5	Α.	Clean cast-in-place architectural concrete surfaces after finish treatment to remove stains, markings, dust, and
6		debris.
7	В.	Wash and rinse surfaces in accordance with concrete finish applicator's written instructions.
8		1. Protect other Work from staining or damage due to cleaning operations.
9		2. Do not use cleaning materials or processes that could change the appearance of cast-in-place architectural
10		concrete finishes.
11		
12	3.10	PROTECTION
13	Α.	Protect corners, edges, and surfaces of cast-in-place architectural concrete from damage; use guards and
14		barricades.
15	В.	Protect cast-in-place architectural concrete from staining, laitance, and contamination during remainder of
16		construction period.
17		
18	3.11	FINAL ACCEPTANCE
19	Α.	Final acceptance of completed architectural concrete Work will be determined by Architect by comparing approved
20		design reference sample with installed Work, when viewed at a distance of 20 feet (6 m).
21		
22		END OF SECTION

1		SECTION 04 20 00
2		UNIT MASONRY
3	PART 1 -	GENERAL
4	1.1	SUMMARY1
5	1.2	DEFINITIONS1
6	1.3	PREINSTALLATION MEETINGS
7	1.4	ACTION SUBMITTALS
8	1.5	INFORMATIONAL SUBMITTALS
9	1.6	QUALITY ASSURANCE
10	1.7	DELIVERY, STORAGE, AND HANDLING
11	1.8	FIELD CONDITIONS
12	PART 2 -	PRODUCTS
13	2.1	SOURCE LIMITATIONS
14	2.2	UNIT MASONRY, GENERAL
15	2.3	CONCRETE MASONRY UNITS
16	2.4	LINTELS
17	2.5	MORTAR AND GROUT MATERIALS
18	2.6	REINFORCEMENT
19	2.7	TIES AND ANCHORS
20	2.8	MORTAR AND GROUT MIXES
21	PART 3 -	EXECUTION
22	3.1	EXAMINATION
23	3.2	INSTALLATION, GENERAL
24	3.3	TOLERANCES
25	3.4	LAYING MASONRY WALLS
26	3.5	MORTAR BEDDING AND JOINTING
27	3.6	MASONRY-JOINT REINFORCEMENT
28	3.7	CONTROL AND EXPANSION JOINTS
29	3.8	LINTELS
30	3.9	REINFORCED UNIT MASONRY
31	3.10	REPAIRING, POINTING, AND CLEANING
32	3.11	MASONRY WASTE DISPOSAL
33		
34	PART 1 -	GENERAL
35		
36	11	SUMMARY
37	 	Section Includes:
38	Π.	1 Concrete masonry units
20		2 lintals
10		2. Entris. 3. Mortar and grout materials
40 //1		A Poinforcoment
41		4. Remotement.
42 40		5. Mortar and grout mixed
45 11	D	0. Mortal and grout mixes. Broducts Installed but not Furnished under This Section:
44 15	Б.	1 Steel lintals in unit mesonny
45		 Steel inities in unit mason y. Steel chelf angles for supporting unit mason y.
40	c	2. Sieer sien angles for supporting unit masonily.
47 10	С.	An Accessories" for installing devetail or shannel slots for masonny
40 40		1. Section 051000 Concrete Forms and Accessories for instailing doverall of chaliner slots for masonry-
49 50		vencer antinors. 2 Socian 051200 "Structural Staal Framing" for installing anchor castions of adjustable massamy anchors for
		2. Section 051200 Structural steel framing for installing anchor sections of adjustable masonry anchors for sonnosting to structural steel frame.
E.J DT		connecting to structural steel frame.
52 E 2	1 2	DEFINITIONS
55 E /	1.2	DEFINITIONS
54 E E	А. Р	Civio(s). Concrete Indsonny Macanny containing rainforcing stack in growted calls
55	в.	הפווויטי גבע ואומצטוויץ. ואומצטוויץ גטווגמוווווצ ופוווטיגווצ גופפו ווו צוטעופע גפווג.
50		

1	1.3	PREINSTALLATION MEETINGS
2	Α.	Preinstallation Conference: Conduct conference at Project site.
3		
4	1.4	ACTION SUBMITTALS
5	Α.	Product Data: For each type of product.
6	В.	Shop Drawings: For the following:
7		1. Masonry Units: Indicate sizes, profiles, coursing, and locations of special shapes.
8		2. Reinforcing Steel: Indicate bending, lap lengths, and placement of unit masonry reinforcing bars. Comply
9		with ACI 315R
10		3 Eabricated Elashing: Detail corner units, end-dam units, and other special applications
11		
12	15	INFORMATIONAL SURMITTALS
13	Δ	Material Certificates: For each type of the following:
14	73.	1 Masonry units
15		a Include data on material properties
16		b. For brick include size-variation data varifying that actual range of sizes falls within specified
17		b. For block, include size variation data verifying that detail range of sizes rais within specificat
10		Ear exposed brick include test report for efflorescence in accordance with ASTM C67/C67M
10		d For surface-coated brick, include test report for durability of surface appearance after 50 cycles of
20		d. For surface-coaled blick, include test report for datability of surface appearance after 50 cycles of
20		For maconny units include data and calculations establishing average not area compressive strength
21		e. For masonry units include data and calculations establishing average net-area compressive strength
22		Of units.
23		2. Mortar admixtures
24		5. World dumixtures.
25		4. Preblended, dry montal mixes. Include description of type and proportions of ingredients.
20		5. Grout mixes, include description of type and proportions of ingredients.
27		o. Relificiting dats.
20		7. Joint remorcement.
29		8. Anchors, ties, and metal accessories.
30	В.	Mix Designs: For each type of mortar and grout . Include description of type and proportions of ingredients.
31		1. include test reports for mortar mixes required to comply with property specification. Test in accordance
32		with ASIM C109/C109M for compressive strength, ASIM C1506 for water retention, and ASIM C91/C91M
33		for air content.
34		2. Include test reports, in accordance with ASTM C1019, for grout mixes required to comply with compressive
35		strength requirement.
36	C.	Statement of Compressive Strength of Masonry: For each combination of masonry unit type and mortar type,
37		provide statement of average net-area compressive strength of masonry units, mortar type, and resulting net-area
38	_	compressive strength of masonry determined in accordance with TMS 602.
39	D.	Cold-Weather and Hot-Weather Procedures: Detailed description of methods, materials, and equipment to be used
40		to comply with requirements.
41		
42	1.6	QUALITY ASSURANCE
43	Α.	Qualifications:
44		1. Installers: All masonry flashing installers must complete the International Masonry Institute Flashing
45		Upgrade training course.
46		2. Testing Agency Qualifications: Qualified in accordance with ASTM C1093 for testing indicated.
47		
48	1.7	DELIVERY, STORAGE, AND HANDLING
49	Α.	Store masonry units on elevated platforms in a dry location. If units are not stored in an enclosed location, cover
50		tops and sides of stacks with waterproof sheeting, securely tied. If units become wet, do not install until they are
51		dry.
52	В.	Store cementitious materials on elevated platforms, under cover, and in a dry location. Do not use cementitious
53		materials that have become damp.
54	С.	Store aggregates where grading and other required characteristics can be maintained and contamination avoided.
55	D.	Deliver preblended, dry mortar mix in moisture-resistant containers. Store preblended, dry mortar mix in delivery
56		containers on elevated platforms in a dry location or in covered weatherproof dispensing silos.
57	Ε.	Store masonry accessories, including metal items, to prevent corrosion and accumulation of dirt and oil.

2	1.8	FIELD CONDITIONS
3	Α.	Protection of Masonry: During construction, cover tops of walls, projections, and sills with waterproof sheeting at
4		end of each day's work. Cover partially completed masonry when construction is not in progress.
5		1. Extend cover a minimum of 24 inches (610 mm) down both sides of walls, and hold cover securely in place.
6		2. Where one wythe of multi-wythe masonry walls is completed in advance of other wythes, secure cover a
7		minimum of 24 inches (610 mm) down face next to unconstructed wythe, and hold cover in place.
8	В.	Do not apply uniform floor or roof loads for at least 12 hours and concentrated loads for at least three days after
9		building masonry walls or columns.
10	C.	Stain Prevention: Prevent grout, mortar, and soil from staining the face of masonry to be left exposed or painted.
11		Immediately remove grout, mortar, and soil that come in contact with such masonry.
12		1. Protect base of walls from rain-splashed mud and from mortar splatter by spreading coverings on ground
13		and over wall surface.
14		2. Protect sills, ledges, and projections from mortar droppings.
15		3. Protect surfaces of window and door frames, as well as similar products with painted and integral finishes.
16		from mortar droppings.
17		4. Turn scaffold boards near the wall on edge at the end of each day to prevent rain from splashing mortar and
18		dirt onto completed masonry.
19	D.	Cold-Weather Requirements: Do not use frozen materials or materials mixed or coated with ice or frost. Do not
20	υ.	build on frozen substrates. Remove and replace unit masonry damaged by frost or by freezing conditions. Comply
20		with cold-weather construction requirements contained in TMS 602
22		1 Cold-Weather Cleaning: Use liquid cleaning methods only when air temperature is 40 deg E (4 deg C) and
22		higher and will remain so until masonry has dried, but not less than seven days after completing cleaning
20	F	Hot-Weather Requirements: Comply with hot-weather construction requirements contained in TMS 602
25	۲.	not weather requirements, comply with not weather construction requirements contained in two obz.
25	PART 2 -	PRODUCTS
20		
27	2.4	
28	2.1	SOURCE LIMITATIONS
29	А.	For exposed masonry units and cementitious mortar components, obtain each color and grade from single source
30		with resources to provide materials of consistent quality in appearance and physical properties.
31		
32	2.2	UNIT MASUNKT, GENERAL Maconny Standardy Comply with TMS 602, avaant as madified by requirements in the Contrast Desumants
33 24	A.	Masonry Standard. Comply with TWS 602, except as modified by requirements in the contract Documents.
34 25	В.	Delective Units: Referenced masonry unit standards may allow a certain percentage of units to contain chips,
35		cracks, or other defects exceeding limits stated. Do not use units where such defects are exposed in the completed
36	6	WORK Size Designed Designed Consultantiations and for fire and states and a second by designed in disected.
37	C.	Fire-Resistance Ratings: Comply with requirements for fire-resistance-rated assembly designs indicated.
38		1. Where fire-resistance-rated construction is indicated, units are listed by UL or a qualified testing agency
39		acceptable to authorities having jurisdiction.
40		
41	2.3	
42	А.	Verify CMU's are manufactured within 100 minies of Project site from aggregates and cement that have been
43	_	extracted, harvested, or recovered, as well as manufactured, within 100 miles of Project site
44	В.	Shapes: Provide shapes indicated and as follows, with exposed surfaces matching exposed faces of adjacent units
45		unless otherwise indicated.
46		1. Provide special shapes for lintels, corners, jambs, sashes, movement joints, headers, bonding, and other
47		special conditions.
48		2. Provide [bullnose units for outside corners unless otherwise indicated.
49	С.	CMUs: ASTM C90, normal weight.
50		1. Unit Compressive Strength: Provide units with minimum average net-area compressive strength of 2800 psi
51		(19.3 MPa).
52		2. Size (Width): Manufactured to dimensions 3/8 inch (10 mm) less than nominal dimensions.
53		3. Exposed Faces: Provide color and texture matching the range represented by Architect's sample.
54		4. Faces to Receive Plaster: Where units are indicated to receive a direct application of plaster, provide
55		textured-face units made with gap-graded aggregates.
56	D.	Concrete Building Brick: ASTM C55, normal weight.].

1 2		1. Unit Compressive Strength: Provide units with minimum average net-area compressive strength of 2800 psi (19.3 MPa)
3	_	 Size (Actual Dimensions): Manufactured to dimensions 3/8 inch less-than-nominal dimensions.
4	E.	Decorative CMUS: ASTM C90, normal weight.
5		Onli Compressive Strength: Provide units with minimum average net-area compressive strength of 2800 psi .
0 7		2. Size (width): Manufactured to dimensions specified in Civitis Paragraph above.
/		5. Pattern and resture:
8		a. MISIN-1: County Materials Decorative Concrete Masonry, Spintace texture, 8 x
9		16 running bond pattern.
10		b. MSN-2: Arriscraft Thin-Clad Renaissance Units, Smooth texture, 12" x 24"
11		running bond pattern.
12		4. Colors:
13		a. MSN-1: 18-121C Bisque (Matched to Existing)
14		b. MSN-2: Oak Ridge.
15		5. Unit Compressive Strength: Provide units with minimum average net-area compressive strength of 2800 psi
16		(19.3 MPa).
1/		6. Size: Manufactured to dimensions specified in "CMUs" Paragraph above but with pre-faced surfaces having
18		1/16-inch- (1.6-mm-) wide returns of facing to create 1/4-inch- (6.2-mm-) wide mortar joints with modular
19		coursing.
20	24	LINTELS
22	 -	Solid Concrete Masonry Lintels: ASTM C1623, matching CMUs in color, texture, and density classification; and with
23		reinforcing bars indicated.
24	В.	Concrete Lintels: Precast or formed-in-place concrete lintels complying with requirements in Section 032000
25		"Concrete Reinforcing," and with reinforcing bars indicated.
26	С.	Masonry Lintels: Prefabricated or built-in-place masonry lintels made from bond beam CMUs matching adjacent
27		CMUs in color, texture, and density classification, with reinforcing bars placed as indicated and filled with coarse
28		grout. Cure precast lintels before handling and installing. Temporarily support built-in-place lintels until cured.
29	D.	Offset Angle Supports: Steel plate brackets anchored to structure, allowing continuous insulation behind shelf angle
30		supporting veneer. Component and anchor size and spacing engineered by manufacturer.
31	25	
32 22	2.5	MURIAR AND GROUT MATERIALS
33	А.	Provide natural color or white cement as required to produce mortar color indicated
35		1. Alkali content will not be more than 0.1 percent when tested in accordance with ASTM C114.
36	В.	Hydrated Lime: ASTM C207. Type S.
37	С.	Portland Cement-Lime Mix: Packaged blend of portland cement and hydrated lime containing no other ingredients.
38	D.	Masonry Cement: ASTM C91/C91M.
39	Ε.	Mortar Cement: ASTM C1329/C1329M.
40	F.	Aggregate for Mortar: ASTM C144.
41		1. For mortar that is exposed to view, use washed aggregate consisting of natural sand or crushed stone.
42		2. For joints less than 1/4 inch (6.4 mm) thick, use aggregate graded with 100 percent passing the No. 16
43		(1.18-mm) sieve.
44		3. White-Mortar Aggregates: Natural white sand or crushed white stone.
45		4. Colored-Mortar Aggregates: Natural sand or crushed stone of color necessary to produce required mortar
46 47	G	COLOF.
47 18	ц	Aggregate for Grout. As init C404. Water: Potable
49	11.	
50	2.6	REINFORCEMENT
51	Α.	Uncoated-Steel Reinforcing Bars: ASTM A615/A615M or ASTM A996/A996M, Grade 60 (Grade 420).
52	В.	Reinforcing Bar Positioners: Wire units designed to fit into mortar bed joints spanning masonry unit cells and to
53		hold reinforcing bars in center of cells. Units are formed from 0.148-inch (3.77-mm) steel wire, hot-dip galvanized
54		after fabrication. Provide units designed for number of bars indicated.
55	С.	Masonry-Joint Reinforcement, General: ASTM A951/A951M.
56		1. Interior Walls: Galvanized carbon steel.

	3. Wire Size for Side Rods: [0.148-inch (3.77-mm)] diameter.
	4. Wire Size for Cross Rods: [U.148-inch (3.77-mm)] diameter.
	5. Spacing of Cross Rods, Tabs, and Cross Ties: Not more than 16 inches (406 mm) o.c.
D	 Provide in lengths of not less than 10 it. (3 in). Maconny Joint Boinforcement for Single Wythe Maconny: Ladder type with single pair of side rods.
D. E	Masonry-Joint Reinforcement for Single-Wythe Masonry: Ladder type with single pair of side rods.
с.	
27	TIES AND ANCHORS
Δ	General: Ties and anchors extend at least 1-1/2 inches (38 mm) into veneer but with at least a 5/8-inch (16-mm)
7.0	cover on outside face.
В.	Materials: Provide ties and anchors specified in this article that are made from materials that comply with the
	following unless otherwise indicated:
	1. Mill-Galvanized, Carbon-Steel Wire: ASTM A1064/A1064M, with ASTM A641/A641M, Class 1 coating.
	2. Steel Plates, Shapes, and Bars: ASTM A36/A36M.
C.	Adjustable Anchors for Connecting to Structural Steel Framing: Provide anchors that allow vertical or horizontal
	adjustment but resist tension and compression forces perpendicular to plane of wall.
	1. Anchor Section for Welding to Steel Frame: Crimped 1/4-inch- (6.4-mm-) diameter, hot-dip galvanized steel
	wire.
	2. Tie Section: Triangular-shaped wire tie made from [0.187-inch- (4.76-mm-)] diameter, [hot-dip galvanized
	steel wire.
D.	Rigid Anchors: Fabricate from steel bars 1-1/2 inches (38 mm) wide by 1/4 inch (6.4 mm) thick by 24 inches (610
	mm) long, with ends turned up 2 inches (51 mm) or with cross pins unless otherwise indicated.
	1. Corrosion Protection: Hot-dip galvanized to comply with ASTM A153/A153M .
2.8	MORTAR AND GROUT MIXES
Α.	General: Do not use admixtures, including pigments, air-entraining agents, accelerators, retarders, water-repellent
	agents, antifreeze compounds, or other admixtures unless otherwise indicated.
	1. Do not use calcium chloride in mortar or grout.
	2. Use portland cement-lime mortar unless otherwise indicated
	3. For exterior masonry, use portland cement-lime mortar.
	4. For reinforced masonry, use portland cement-lime mortar.
	5. Add cold-weather admixture (if used) at same rate for all mortar that will be exposed to view, regardless of
	weather conditions, to ensure that mortar color is consistent.
В.	Preblended, Dry Mortar Mix: Furnish dry mortar ingredients in form of a preblended mix. Measure quantities by
	weight to ensure accurate proportions, and thoroughly blend ingredients before delivering to Project site.
С.	Mortar for Unit Masonry: Comply with ASTM C270, Property Specification. Provide the following types of mortar for
	applications stated unless another type is indicated.
	1. For masonry below grade or in contact with earth, use Type M.
	 For reinforced masonry, use Type S. For exterior above grade load bearing walls and paramet walls for interior load bearing.
	3. For exterior, above-grade, load-bearing, nonioad-bearing wails, and parapet wails; for interior load-bearing partitions; and for other applications where another type is not
	waits, for interior nonioad-bearing partitions, and for other applications where another type is not indicated use Type N
	For interior popload-bearing partitions. Type O may be used instead of Type N
D	Grout for Unit Masonry: Comply with ASTM (476
D.	1 Use grout of type indicated or if not otherwise indicated of type (fine or coarse) that will comply with
	TMS 602 for dimensions of grout spaces and nour height
	2. Proportion grout in accordance with ASTM C476.
	3. Provide grout with a slump of [8 to 11 inches (203 to 279 mm)] [10 to 11 inches (254 to 279 mm)] as
	measured in accordance with ASTM C143/C143M.
<u> PART 3 - E</u>	XECUTION

55 A. Examine conditions, with Installer present, for compliance with requirements for installation tolerances and other 56 conditions affecting performance of the Work.

1 2		1. For the record, prepare written report, endorsed by Installer, listing conditions detrimental to performance of the Work.
3		2. Verify that foundations are within tolerances specified.
4		3. Verify that reinforcing dowels are properly placed.
5		 Verify that substrates are free of substances that impair mortar bond.
6	в	Refore installation, examine rough-in and built-in construction for nining systems to verify actual locations of nining
7	5.	connections.
, 8 9	C.	Proceed with installation only after unsatisfactory conditions have been corrected.
10	3.2	INSTALLATION, GENERAL
11	Α.	Build chases and recesses to accommodate items specified in this and other Sections.
12	В.	Leave openings for equipment to be installed before completing masonry. After installing equipment, complete
13		masonry to match construction immediately adjacent to opening.
14	C.	Use full-size units without cutting if possible. If cutting is required to provide a continuous pattern or to fit adjoining
15	-	construction, cut units with motor-driven saws; provide clean, sharp, unchipped edges. Allow units to dry before
16		laving unless wetting of units is specified. Install cut units with cut surfaces and, where possible, cut edges
17		concealed.
18		
19	3.3	TOLERANCES
20	Α.	Dimensions and Locations of Elements:
21		1. For dimensions in cross section or elevation, do not vary by more than plus 1/2 inch (13 mm) or minus 1/4
22		inch (6.4 mm).
23		2. For location of elements in plan, do not vary from that indicated by more than plus or minus 1/2 inch (13
24		mm).
25		3. For location of elements in elevation, do not vary from that indicated by more than plus or minus 1/4 inch
26		(6.4 mm) in a story height or 1/2 inch (13 mm) total.
27	B.	Lines and Levels:
28		1. For bed joints and top surfaces of bearing walls, do not vary from level by more than 1/4 inch in 10 ft. (6.4
29		mm in 3 m), or $1/2$ -inch (13-mm) maximum.
30		 For conspicuous horizontal lines, such as lintels, sills, parapets, and reveals, do not vary from level by more.
31		than 1/8 inch in 10 ft. (3.2 mm in 3 m). 1/4 inch in 20 ft. (6.4 mm in 6 m). or 1/2-inch (13-mm) maximum.
32		3. For vertical lines and surfaces, do not vary from plumb by more than 1/4 inch in 10 ft. (6.4 mm in 3 m), 3/8
33		inch in 20 ft. (10 mm in 6 m). or 1/2-inch (13-mm) maximum.
34		4. For conspicuous vertical lines, such as external corners, door jambs, reveals, and expansion and control
35		ioints, do not vary from plumb by more than 1/8 inch in 10 ft. (3.2 mm in 3 m), 1/4 inch in 20 ft. (6.4 mm in
36		6 m), or 1/2-inch (13-mm) maximum.
37		5. For lines and surfaces, do not vary from straight by more than 1/4 inch in 10 ft. (6.4 mm in 3 m), 3/8 inch in
38		20 ft. (10 mm in 6 m). or 1/2-inch (13-mm) maximum.
39		6. For vertical alignment of exposed head joints, do not vary from plumb by more than 1/4 inch in 10 ft. (6.4
40		mm in 3 m), or 1/2-inch (13-mm) maximum.
41	C.	Joints:
42		1. For bed joints, do not vary from thickness indicated by more than plus or minus 1/8 inch (3.2 mm), with a
43		maximum thickness limited to 1/2 inch (13 mm).
44		2. For exposed bed joints, do not vary from bed-joint thickness of adjacent courses by more than 1/8 inch (3.2
45		mm).
46		3. For head and collar joints, do not vary from thickness indicated by more than plus 3/8 inch (10 mm) or
47		minus 1/4 inch (6.4 mm).
48		4. For exposed head joints, do not vary from thickness indicated by more than plus or minus 1/8 inch (3.2
49		mm).
50		5. For exposed bed joints and head joints of stacked bond, do not vary from a straight line by more than 1/16
51		inch (1.6 mm) from one masonry unit to the next.
52		· · · · ·
53	3.4	LAYING MASONRY WALLS
54	Α.	Lay out walls in advance for accurate spacing of surface bond patterns with uniform joint thicknesses and for
55		accurate location of openings, movement-type joints, returns, and offsets. Avoid using less-than-half-size units,
56		particularly at corners, jambs, and, where possible, at other locations.

1 2	В.	Bond Pattern for Exposed Masonry: Unless otherwise indicated, lay exposed masonry in running bond ; do not use units with less-than-nominal 4-inch (102-mm) horizontal face dimensions at corners or jambs.
2	C	Law concealed maconry with all units in a wythe in running bond or bonded by lanning not less than 4 inches (102)
1	С.	mm) Bond and interlock each course of each withe at corners. Do not use units with less than a mines (102
5		(102-mm) horizontal face dimensions at corners or jambs
5	Р	(102-1111) horizontal face dimensions at corners of jamos.
0	D.	stopping and resuming work. Stop work by stepping back units in each course from those in course below; do not
/		tooth. when resuming work, clean masonry surfaces that are to receive mortar, remove loose masonry units and
8	-	mortar, and wet brick if required before laying fresh masonry.
9	E.	Built-in Work: As construction progresses, build in items specified in this and other sections. Fill in solidly with
10	_	masonry around built-in items.
11	F.	Fill space between steel frames and masonry solidly with mortar unless otherwise indicated.
12	G.	Where built-in items are to be embedded in cores of hollow masonry units, place a layer of metal lath, wire mesh,
13		or plastic mesh in the joint below, and rod mortar or grout into core.
14	Н.	Fill cores in hollow CMUs with grout 24 inches (610 mm) under bearing plates, beams, lintels, posts, and similar
15	_	items unless otherwise indicated.
16	Ι.	Build nonload-bearing interior partitions full height of story to underside of solid floor or roof structure above
17		unless otherwise indicated.
18		1. Install compressible filler in joint between top of partition and underside of structure above. Fasten partition
19		top anchors to structure above and build into top of partition. Grout cells of CMUs solidly around plastic
20		tubes of anchors, and push tubes down into grout to provide 1/2-inch (13-mm) clearance between end of
21		anchor rod and end of tube. Space anchors 48 inches (1219 mm) o.c. unless otherwise indicated.
22		2. Wedge nonload-bearing partitions against structure above with small pieces of tile, slate, or metal. Fill joint
23		with mortar after dead-load deflection of structure above approaches final position.
24		3. At fire-rated partitions, treat joint between top of partition and underside of structure above to comply
25		with Section 078443 "Joint Firestopping."
26		
27	3.5	MORTAR BEDDING AND JOINTING
28	Α.	Lay CMUs as follows:
29		 Bed face shells in mortar and make head joints of depth equal to bed joints.
30		2. Bed webs in mortar in all courses of piers, columns, and pilasters.
31		 Bed webs in mortar in grouted masonry, including starting course on footings.
32		4. Fully bed entire units, including areas under cells, at starting course on footings where cells are not grouted.
33		5. Fully bed units and fill cells with mortar at anchors and ties as needed to fully embed anchors and ties in
34		mortar.
35	В.	Lay solid masonry units with completely filled bed and head joints; butter ends with sufficient mortar to fill head
36		joints and shove into place. Do not deeply furrow bed joints or slush head joints.
37	С.	Tool exposed joints slightly concave when thumbprint hard, using a jointer larger than joint thickness unless
38		otherwise indicated.
39		1. For glazed masonry units, use a nonmetallic jointer 3/4 inch (19 mm) or more in width.
40	D.	Cut joints flush for masonry walls to receive plaster or other direct-applied finishes (other than paint) unless
41		otherwise indicated.
42	Ε.	Cut joints flush where indicated to receive waterproofing unless otherwise indicated.
43		
44	3.6	MASONRY-JOINT REINFORCEMENT
45	Α.	General: Install entire length of longitudinal side rods in mortar with a minimum cover of 5/8 inch (16 mm) on
46		exterior side of walls, 1/2 inch (13 mm) elsewhere. Lap reinforcement a minimum of 6 inches (152 mm).
47		1. Space reinforcement not more than 16 inches (406 mm) o.c.
48		2. Space reinforcement not more than 8 inches (203 mm) o.c. in foundation walls and parapet walls.
49		3. Provide reinforcement not more than 8 inches (203 mm) above and below wall openings and extending 12
50		inches (305 mm) beyond openings in addition to continuous reinforcement.
51	В.	Interrupt joint reinforcement at control and expansion joints unless otherwise indicated.
52	С.	Cut and bend reinforcing units as directed by manufacturer for continuity at[corners,] returns, offsets, column
53		fireproofing, pipe enclosures, and other special conditions.
54		
55	3.7	CONTROL AND EXPANSION JOINTS
56	Α.	General: Install control- and expansion-joint materials in unit masonry as masonry progresses. Do not allow
57		materials to span control and expansion joints without provision to allow for in-plane wall or partition movement.

1	В.	Form control joints in concrete masonry using one of the following methods:
2		1. Fit bond-breaker strips into hollow contour in ends of CMUs on one side of control joint. Fill resultant core
3		with grout, and rake out joints in exposed faces for application of sealant.
4		2. Install preformed control-joint gaskets designed to fit standard sash block.
5		3. Install interlocking units designed for control joints. Install bond-breaker strips at joint. Keep head joints
6 7		Tree and clear of mortar, or rake out joint for application of sealant.
/ 8		4. Install temporary roam-plastic liner in nead joints, and remove liner when unit masonry is complete for application of sealant
q		
10	3.8	LINTELS
11	A.	Install steel lintels where indicated.
12	В.	Provide masonrylintels where indicated and where openings of more than 12 inches (305 mm) for brick-size units
13		and 24 inches (610 mm) for block-size units are indicated without structural steel or other supporting lintels.
14	С.	Provide minimum bearing of 8 inches (203 mm) at each jamb unless otherwise indicated.
15		
16	3.9	REINFORCED UNIT MASONRY
17	Α.	Temporary Formwork and Shores: Construct formwork and shores as needed to support reinforced masonry
18		elements during construction.
19		1. Construct formwork to provide shape, line, and dimensions of completed masonry as indicated. Make forms
20		sufficiently tight to prevent leakage of mortar and grout. Brace, tie, and support forms to maintain position
21		and shape during construction and curing of reinforced masonry.
22		2. Do not remove forms and shores until remorced masonry members have hardened sunctiently to carry their own weight and that of other leads that may be placed on them during construction
23	R	Placing Reinforcement: Comply with requirements in TMS 602
24 25	ь. С	Grouting. Do not place grout until entire height of masonry to be grouted has attained enough strength to resist
26	с.	grout pressure
27		1. Comply with requirements in TMS 602 for cleanouts and for grout placement, including minimum grout
28		space and maximum pour height.
29		2. Limit height of vertical grout pours to not more than 60 inches (1524 mm)
30		
31	3.10	REPAIRING, POINTING, AND CLEANING
32	Α.	Remove and replace masonry units that are loose, chipped, broken, stained, or otherwise damaged or that do not
33		match adjoining units. Install new units to match adjoining units; install in fresh mortar, pointed to eliminate
34		evidence of replacement.
35	В.	Pointing: During the tooling of joints, enlarge voids and holes, except weep holes, and completely fill with mortar.
36		Point up joints, including corners, openings, and adjacent construction, to provide a neat, uniform appearance.
3/ 20	c	Prepare joints for sealant application, where indicated.
30	с. D	lion, where multided.
40	D.	hefore tooling joints
41	E.	Final Cleaning: After mortar is thoroughly set and cured, clean exposed masonry as follows:
42		1. Remove large mortar particles by hand with wooden paddles and nonmetallic scrape hoes or chisels.
43		2. Test cleaning methods on sample wall panel; leave one-half of panel uncleaned for comparison purposes.
44		Obtain Architect's approval of sample cleaning before proceeding with cleaning of masonry.
45		3. Clean concrete masonry by applicable cleaning methods indicated in NCMA TEK 8-4A.
46		4. Clean masonry with a proprietary acidic masonry cleaner applied according to manufacturer's written
47		instructions.
48		
49	3.11	MASONRY WASTE DISPOSAL
50	Α.	Salvageable Materials: Unless otherwise indicated, excess masonry materials are Contractor's property. At
51		completion of unit masonry work, remove from Project site.
52 E 2	в.	waste Disposal as Fill Material: Dispose of clean masonry Waste, including excess or soll-contaminated sand, Waste
53 54		norrar, and proven masonry units, by crushing and mixing with fill material as fill is placed.
54		 Crush mason y waste to less than 4 mores (102 mill) in edition dimension. Do not dispose of masonry waste as fill within 18 inches (457 mm) of finished grade
56	ſ	Masonry Waste Recycling: Return broken CMUs not used as fill to manufacturer for recycling
	. .	,

D. Excess Masonry Waste: Remove excess clean masonry waste that cannot be used as fill, as described above or recycled, and other masonry waste, and legally dispose of off Owner's property.

2 3 4

1

END OF SECTION

1		SECTION 05 12 00
2		STRUCTURAL STEEL FRAMING
з Л	PART 1 - (SENERAL 1
5	11	SUMMARY
6	1 2	
7	1 3	
, 8	1.5	PREINSTALLATION MEETINGS
q	15	
10	1.5	
11	1.0	
12	1.8	DELIVERY STORAGE AND HANDLING 2
13	PART 2 - F	PRODUCTS
14	21	PERFORMANCE REQUIREMENTS 2
15	2.1	STRUCTURAL-STEEL MATERIALS
16	23	BOITS AND CONNECTORS
17	2.3	RODS
18	2.5	FORGED-STEEL STRUCTURAL HARDWARE
19	2.6	SLIDE BEARINGS.
20	2.7	PRIMER
21	2.8	SHRINKAGE-RESISTANT GROUT
22	2.9	EABRICATION
23	2.10	SHOP CONNECTIONS
24	2.11	PREFABRICATED BUILDING COLUMNS
25	2.12	GALVANIZING Error! Bookmark not defined.
26	2.13	SHOP PRIMING
27	2.14	SOURCE QUALITY CONTROL Error! Bookmark not defined.
28	PART 3 - E	EXECUTION
29	3.1	EXAMINATION
30	3.2	PREPARATION
31	3.3	ERECTION
32	3.4	FIELD CONNECTIONS
33	3.5	INSTALLATION OF PREFABRICATED BUILDING COLUMNSError! Bookmark not defined.
34	3.6	REPAIR
35	3.7	FIELD QUALITY CONTROL
36		
37	PART 1 - 0	GENERAL
38		
39	1.1 SUM	MARY
40	Α.	Section Includes:
41		1. Structural-steel materials.
42		2. Shrinkage-resistant grout.
43		
44	1.2 DEFI	NITIONS
45	Α.	Structural Steel: Elements of the structural frame indicated on Drawings and as described in ANSI/AISC 303.
46		
47	1.3 COO	RDINATION
48	Α.	Coordinate selection of shop primers with topcoats to be applied over them. Comply with paint and coating
49		manufacturers' written recommendations to ensure that shop primers and topcoats are compatible with one
50		another.
51	В.	Coordinate installation of anchorage items to be embedded in or attached to other construction without delaying
52		the Work. Provide setting diagrams, sheet metal templates, instructions, and directions for installation.
53		

54 **1.4 PREINSTALLATION MEETINGS**

55 A. Preinstallation Conference: Conduct conference at Project site.

1	1.5 ACTION SUBMITTALS			
2	A		Product Data:	
3			1. Structural-steel materials.	
4			2. High-strength, bolt-nut-washer assemblies.	
5			3. Anchor rods.	
6			4. Threaded rods.	
7			5. Forged-steel hardware.	
8			6. Shop primer.	
9			7. Shrinkage-resistant grout.	
10	B	•	Shop Drawings: Show fabrication of structural-steel components.	
11			1. Include details of cuts, connections, splices, camber, holes, and other pertinent data.	
12			2. Include embedment Drawings.	
13			3. Indicate welds by standard AWS symbols, distinguishing between shop and field welds, and show size,	
14			length, and type of each weld. Show backing bars that are to be removed and supplemental fillet welds	
15			where backing bars are to remain.	
16			4. Indicate type, size, and length of bolts, distinguishing between shop and field bolts. Identify pretensioned	
1/			and slip-critical, nigh-strength bolted connections.	
18	c		5. Identify members not to be shop primed.	
19	Ľ.	•	AWS D1 1/D1 1M for each welded joint including the following:	
20			Aws D1.1/D1.1M for each weided joint, including the following:	
21			 Power source (constant current of constant voltage). Electrode manufacturer and trade name, for demand-critical welds. 	
22				
23	16 1		ΜΑΤΙΩΝΑΙ SUBMITTAI S	
25	1.0 I		Welding certificates.	
26	B		Paint Compatibility Certificates: From manufacturers of topcoats applied over shop primers, certifying that shop	
27			primers are compatible with topcoats.	
28	C.		Mill test reports for structural-steel materials, including chemical and physical properties.	
29	D		Product Test Reports: For the following:	
30			1. Bolts, nuts, and washers, including mechanical properties and chemical analysis.	
31			2. Direct-tension indicators.	
32			3. Tension-control, high-strength, bolt-nut-washer assemblies.	
33	E.		Survey of existing conditions.	
34	F.		Source quality-control reports.	
35	G	i.	Field quality-control reports.	
36				
37	1.7 (QUALI	TY ASSURANCE	
38 39	A		Welding Qualifications: Qualify procedures and personnel in accordance with AWS D1.1/D1.1M.	
40	1.8 [DELIVE	ERY. STORAGE. AND HANDLING	
41	A		Store materials to permit easy access for inspection and identification. Keep steel members off ground and spaced	
42			by using pallets, dunnage, or other supports and spacers. Protect steel members and packaged materials from	
43			corrosion and deterioration.	
44			1. Do not store materials on structure in a manner that might cause distortion, damage, or overload to	
45			members or supporting structures. Repair or replace damaged materials or structures as directed.	
46	B		Store fasteners in a protected place in sealed containers with manufacturer's labels intact.	
47			1. Fasteners may be repackaged provided Owner's testing and inspecting agency observes repackaging and	
48			seals containers.	
49			2. Clean and relubricate bolts and nuts that become dry or rusty before use.	
50			3. Comply with manufacturers' written recommendations for cleaning and lubricating ASTM F3125/F3125M,	
51			Grade F1852 bolt assemblies and for retesting bolt assemblies after lubrication.	
52				
53	PART	2 - PR	<u>IODUCTS</u>	
54				
55	2.1 F	PERFO	RMANCE REQUIREMENTS	
56	A	•	Comply with applicable provisions of the following specifications and documents:	
57	_		1. ANSI/AISC 303.	
58	B		Connection Design Information:	

1 2 3			 Option 1: Connection designs have been completed and connections indicated on the Drawings. Option 2: Fabricator's experienced steel detailer selects or completes connections in accordance with ANSI/AISC 303.
4 5			 a. Select and complete connections using schematic details indicated. b. Use Allowable Stress Design; data are given at service-load level.
6		C.	Moment Connections: Type PR, partially restrained.
8	2.2	STRUG	CTURAL-STEEL MATERIALS
9		Α.	W-Shapes: ASTM A992/A992M.
10		В.	Channels, Angles, M-Shapes: ASTM A36/A36M.
11		C.	Channels, Angles, S-Shapes: ASTM A36/A36M.
12		D.	Plate and Bar: ASTM A36/A36M.
13		Ε.	Cold-Formed Hollow Structural Sections: ASTM A500/A500M, Grade B structural tubing.
14		F.	Steel Forgings: ASTM A668/A668M.
15		G.	Welding Electrodes: Comply with AWS requirements.
16 17	2.3	BOLTS	S AND CONNECTORS
18		Α.	High-Strength A325 Bolts, Nuts, and Washers: ASTM F3125/F3125M, Grade A325, Type 1, heavy-hex steel structural
19			bolts; ASTM A563, Grade DH, heavy-hex carbon-steel nuts; and ASTM F436/F436M, Type 1, hardened carbon-steel
20			washers; all with plain finish.
21 22			1. Direct-Tension Indicators: ASTM F959/F959M, Type 325-1, compressible-washer type with plain finish.
23	2.4	RODS	
24		Α.	Headed Anchor Rods: ASTM F1554, Grade 36, straight.
25			1. Nuts: ASTM A563 hex carbon steel.
26			2. Plate Washers: ASTM A36/A36M carbon steel.
27			3. Washers: ASTM F436, Type 1, hardened carbon steel.
28			4. Finish: Plain.
29 30			
31	2.5	PRIMI	ER
32		A.	Steel Primer:
33			1. Fabricator's standard lead- and chromate-free, nonasphaltic, rust-inhibiting primer complying with MPI#79
34			and compatible with topcoat.
35	26		
30 27	2.0		Nonmetallic Shrinkage Registrant Grout: ASTM C1107/C1107M factory packaged, permetallic aggregate grout
38		А.	noncorrosive and nonstaining, mixed with water to consistency suitable for application and a 30-minute working
39			time.
40	~ -		
41	2.7	FABRI	CATION
42 43		А.	ANSI/AISC 303 and to ANSI/AISC 360.
44			1. Camber structural-steel members where indicated.
45			2. Fabricate beams with rolling camber up.
46			3. Identify high-strength structural steel in accordance with ASTM A6/A6M and maintain markings until
47			structural-steel framing has been erected.
48			4. Mark and match-mark materials for field assembly.
49			5. Complete structural-steel assemblies, including welding of units, before starting shop-priming operations.
50		В.	Thermal Cutting: Perform thermal cutting by machine to greatest extent possible.
51			1. Plane thermally cut edges to be welded to comply with requirements in AWS D1.1/D1.1M.
52		C.	Bolt Holes: Cut, drill, mechanically thermal cut, or punch standard bolt holes perpendicular to metal surfaces.
53		D.	Finishing: Accurately finish ends of columns and other members transmitting bearing loads.
54		E.	Cleaning: Clean and prepare steel surfaces that are to remain unpainted in accordance with SSPC-SP 1.
55		F.	Holes: Provide holes required for securing other work to structural steel and for other work to pass through steel
56			members.
57			1. Cut, drill, or punch holes perpendicular to steel surfaces. Do not thermally cut bolt holes or enlarge holes by
			burning.

1 2			 Baseplate Holes: Cut, drill, mechanically thermal cut, or punch holes perpendicular to steel surfaces. Weld threaded nuts to framing and other specialty items indicated to receive other work.
3	20		
4 5 6	2.0	A.	High-Strength Bolts: Shop install high-strength bolts in accordance with RCSC's "Specification for Structural Joints Using High-Strength Bolts" for type of bolt and type of joint specified.
7			1. Joint Type: Snug tightened.
8		В.	Weld Connections: Comply with AWS D1.1/D1.1M and AWS D1.8/D1.8M for tolerances, appearances, welding
9			procedure specifications, weld quality, and methods used in correcting welding work.
10			1. Assemble and weld built-up sections by methods that maintain true alignment of axes without exceeding
11			tolerances in ANSI/AISC 303 for mill material.
12			
14	2.9	SHOP	PRIMING
15		A.	Shop prime steel surfaces, except the following:
16 17			1. Surfaces embedded in concrete or mortar. Extend priming of partially embedded members to a depth of 2 inches.
18			2. Surfaces to be field welded.
19 20		В.	Surface Preparation of Steel: Clean surfaces to be painted. Remove loose rust and mill scale and spatter, slag, or flux deposits. Prepare surfaces in accordance with the following specifications and standards:
21			1. SSPC-SP 2.
22		c	2. SSPC-SP 3. Driming: Immediately after surface proparation apply primer in accordance with manufacturar's written.
25 24		C.	instructions and at rate recommended by SSPC to provide a minimum dry film thickness of 1.5 mile. Use priming
24 25			methods that result in full coverage of joints corners edges and exposed surfaces
26			1. Stripe paint corners, crevices, bolts, welds, and sharp edges.
27			2. Apply two coats of shop paint to surfaces that are inaccessible after assembly or erection. Change color of
28			second coat to distinguish it from first.
29			
30			
31	PA	<u> RT 3 - E</u>	XECUTION
32 22	2 1		
33	3.1		Verify with certified steel erector present elevations of concrete- and masonry-bearing surfaces and locations of
35		/	anchor rods, bearing plates, and other embedments for compliance with requirements.
36			1. Prepare a certified survey of existing conditions. Include bearing surfaces, anchor rods, bearing plates, and
37			other embedments showing dimensions, locations, angles, and elevations.
38		В.	Proceed with installation only after unsatisfactory conditions have been corrected.
39			
40	3.2	PREP	ARATION
41		A.	Provide temporary shores, guys, braces, and other supports during erection to keep structural steel secure, plumb,
42			and in alignment against temporary construction loads and loads equal in intensity to design loads. Remove
43 11			indicated on Drawings
44 45			indicated on Drawings.
46	3.3	EREC	ΓΙΟΝ
47	0.0	A.	Set structural steel accurately in locations and to elevations indicated and in accordance with ANSI/AISC 303 and
48			ANSI/AISC 360.
49		В.	Baseplates, Bearing Plates, and Leveling Plates: Clean concrete- and masonry-bearing surfaces of bond-reducing
50			materials, and roughen surfaces prior to setting plates. Clean bottom surface of plates.
51			1. Set plates for structural members on wedges, shims, or setting nuts as required.
52			2. Weld plate washers to top of baseplate.
53			3. Snug-tighten anchor rods after supported members have been positioned and plumbed. Do not remove
54			wedges or shims but, it protruding, cut off flush with edge of plate before packing with grout.
55			 Promptly pack snrinkage-resistant grout solidly between bearing surfaces and plates, so no voids remain. Neath, finish supposed surfaces, protect grout and allow to sure. Comply with manufacture la without and allow to sure.
50 57			installation instructions for grouting
58		C	Maintain erection tolerances of structural steel within ANSI/AISC 303
20			

1	D.	Align and adjust various members that form part of complete frame or structure before permanently fastening.
2		Before assembly, clean bearing surfaces and other surfaces that are in permanent contact with members. Perform
3		necessary adjustments to compensate for discrepancies in elevations and alignment.
4		1. Level and plumb individual members of structure. Slope roof framing members to slopes indicated on
5		Drawings.
6	E.	Splice members only where indicated.
7	F.	Do not use thermal cutting during erection unless approved by Architect. Finish thermally cut sections within
8		smoothness limits in AWS D1.1/D1.1M.
9	G.	Do not enlarge unfair holes in members by burning or using drift pins. Ream holes that must be enlarged to admit
10		bolts.
11		
12	3.4 FIELD	D CONNECTIONS
13	Α.	High-Strength Bolts: Install high-strength bolts in accordance with RCSC's "Specification for Structural Joints Using
14		High-Strength Bolts" for bolt and joint type specified.
15		1. Joint Type: Snug tightened.
16	В.	Weld Connections: Comply with AWS D1.1/D1.1M for tolerances, appearances, welding procedure specifications,
17		weld quality, and methods used in correcting welding work.
18		1. Comply with ANSI/AISC 303 and ANSI/AISC 360 for bearing, alignment, adequacy of temporary connections,
19		and removal of paint on surfaces adjacent to field welds.
20		2. Remove backing bars or runoff tabs where indicated, back gouge, and grind steel smooth.
21		3. Assemble and weld built-up sections by methods that maintain true alignment of axes without exceeding
22		tolerances in ANSI/AISC 303 for mill material.
23		
24	3.5 REP/	AIR
25	Α.	Touchup Painting:
26		1. Immediately after erection, clean exposed areas where primer is damaged or missing, and paint with the
27		same material as used for shop painting to comply with SSPC-PA 1 for touching up shop-painted surfaces.
28		a. Clean and prepare surfaces by SSPC-SP 2 hand-tool cleaning or SSPC-SP 3 power-tool cleaning.
29		
30	3.6 FIELI	D QUALITY CONTROL
31	-	
32	Α.	lesting Agency: Engage a qualified testing agency to perform tests and inspections.
33		1. Bolted Connections: Inspect and test bolted connections in accordance with RCSC's "Specification for
34		Structural Joints Using High-Strength Bolts."
35		2. Welded Connections: Visually inspect field welds in accordance with AWS D1.1/D1.1M.
36		a. In addition to visual inspection, test and inspect field welds in accordance with AWS D1.1/D1.1M
37		and the following inspection procedures, at testing agency's option:
38		1) Liquid Penetrant Inspection: ASTM E165/E165M.
39		2) Magnetic Particle Inspection: ASTM E709; performed on root pass and on finished weld.
40		Cracks or zones of incomplete fusion or penetration are not accepted.
41		3) UITRASONIC INSPECTION: ASTNI E164.
4Z		4) Radiographic inspection: ASTIVE 94/E94M.
43 11		
44		

SECTION 053100 STEEL DECKING

3					
4	PAR	T 1 - GE	NERAL1		
5	1	1	SUMMARY		
6	1	.2	ACTION SUBMITTALS		
7	1	3	INFORMATIONAL SUBMITTALS 1		
8	1	.4	QUALITY ASSURANCE		
9	1	.5	DELIVERY, STORAGE, AND HANDLING		
10	PAR	T 2 - PF	ODUCTS		
11	2	.1	PERFORMANCE REQUIREMENTS		
12	2	.2	ROOF DECK		
13	2	.3	ACOUSTICAL ROOF DECK		
14	2	.4	ACCESSORIES		
15	PAR	Т 3 - ЕХ	ECUTION		
16	3	1	FXAMINATION 3		
17	3	2	INSTALLATION GENERAL 3		
18	2	3	INSTALLATION OF ROOF DECK		
10	2				
20	J				
20		T 1 C			
21	PAR	1 I - GI			
22		C1 18 48			
23	1.1	SUIVIN	IART		
24		А.	Section Includes:		
25			1. ROOT deck.		
26		_	2. Acoustical root deck.		
27		в.	Related Requirements:		
28			1. Section 055000 "Metal Fabrications" for framing deck openings with miscellaneous steel shapes.		
29					
30	1.2	ACTIO	N SUBMITTALS		
31		А.	Product Data:		
32			1. ROOT deck.		
33		_	2. Acoustical roof deck.		
34		В.	Shop Drawings:		
35			1. Include layout and types of deck panels, anchorage details, reinforcing channels, pans, cut deck openings,		
36			special jointing, accessories, and attachments to other construction.		
37					
38	1.3	INFOR	MATIONAL SUBMITTALS		
39		Α.	Welding certificates.		
40			1. Product Test Reports: For tests performed by a qualified testing agency, indicating that each of the		
41			following complies with requirements:		
42			a. Acoustical roof deck.		
43					
44	1.4	QUAL	TY ASSURANCE		
45		Α.	Welding Qualifications: Qualify procedures and personnel in accordance with SDI QA/QC and the following welding		
46			codes:		
47			1. AWS D1.1/D1.1M.		
48			2. AWS D1.3/D1.3M.		
49					
50	1.5	DELIV	ERY, STORAGE, AND HANDLING		
51		A.	Protect steel deck from corrosion, deformation, and other damage during delivery, storage, and handling.		
52		В.	Store products in accordance with SDI MOC3. Stack steel deck on platforms or pallets and slope to provide		
53			drainage. Protect with a waterproof covering and ventilate to avoid condensation.		

1. Protect and ventilate acoustical cellular roof deck with factory-installed insulation to maintain insulation free of moisture.

1 PART 2 - PRODUCTS 2 3 2.1 PERFORMANCE REQUIREMENTS 4 AISI Specifications: Comply with calculated structural characteristics of steel deck in accordance with AISI S100. Α. 5 6 2.2 ROOF DECK 7 Fabrication of Roof Deck: Fabricate panels, without top-flange stiffening grooves, to comply with SDI RD and with Α. 8 the following: 9 Galvanized-Steel Sheet: ASTM A653/A653M, Structural Steel (SS), Grade 33, G60 zinc coating. 1. 10 2. Galvanized- and Shop-Primed Steel Sheet: ASTM A653/A653M, Structural Steel (SS), Grade 33, G60 zinc coating; cleaned, pretreated, and primed with manufacturer's standard baked-on, rust-inhibitive primer. 11 12 Color: [Manufacturer's standard] [Gray] [White] [Gray top surface with white underside]. a. 13 3. Deck Profile: As indicated. 14 4. Profile Depth: As indicated. 15 5. Design Uncoated-Steel Thickness: As indicated. Design Uncoated-Steel Thicknesses; Deck Unit/Bottom Plate: As indicated. 16 6. 17 7. Span Condition: Double span minimum. 18 8. Side Laps: Interlocking seam. 19 2.3 ACOUSTICAL ROOF DECK 20 21 Fabrication of Acoustical Roof Deck: Fabricate panels, without top-flange stiffening grooves, to comply with SDI RD Α. 22 and with the following: 23 Galvanized- and Shop-Primed Steel Sheet: ASTM A653/A653M, Structural Steel (SS), Grade 33, G60 zinc 1. 24 coating; cleaned, pretreated, and primed with manufacturer's standard baked-on, rust-inhibitive primer. 25 Color: [Manufacturer's standard] [Gray] [White] [Gray top surface with white underside]. а. 26 2. Deck Profile: As indicated. 27 Cellular Deck Profile: As indicated, with bottom plate. 3. 28 4. Profile Depth: As indicated. 29 5. Design Uncoated-Steel Thickness: As indicated. 30 6. Design Uncoated-Steel Thicknesses; Deck Unit/Bottom Plate: As indicated. 31 7. Span Condition: Double span minimum. 32 8. Side Laps: Interlocking seam. 33 9. Acoustical Perforations: [Deck units with manufacturer's standard perforated vertical webs] [Cellular deck 34 units with manufacturer's standard perforated flat-bottom plate welded to ribbed deck]. 35 10. Sound-Absorbing Insulation: Manufacturer's standard premolded roll or strip of glass or mineral fiber. 36 <Insert actual physical properties and thicknesses of insulation>. 37 Factory install sound-absorbing insulation into cells of cellular deck. a. Installation of sound-absorbing insulation is specified in <Insert Section number and title>. 38 b. 39 11. Acoustical Performance: NRC [0.65] [0.75] [0.80] [0.85] [0.90], tested in accordance with ASTM C423. 40 41 2.4 ACCESSORIES 42 Α. Provide manufacturer's standard accessory materials for deck that comply with requirements indicated. 43 Β. Mechanical Fasteners: Corrosion-resistant, low-velocity, power-actuated or pneumatically driven carbon-steel 44 fasteners; or self-drilling, self-threading screws. C. 45 Side-Lap Fasteners: Corrosion-resistant, hexagonal washer head; self-drilling, carbon-steel screws, No. 10 minimum diameter. 46 47 D. Flexible Closure Strips: Vulcanized, closed-cell, synthetic rubber. 48 Miscellaneous Sheet Metal Deck Accessories: Steel sheet, minimum yield strength of 33,000 psi, not less than F. 49 0.0359-inch design uncoated thickness, of same material and finish as deck; of profile indicated or required for 50 application. F. 51 Column Closures, End Closures, Z-Closures, and Cover Plates: Steel sheet, of same material, finish, and thickness as 52 deck unless otherwise indicated. 53 G. Weld Washers: Uncoated steel sheet, shaped to fit deck rib, 0.0598 inch thick, with factory-punched hole of 3/8-54 inch minimum diameter. 55 н. Flat Sump Plates: Single-piece steel sheet, 0.0747 inch thick, of same material and finish as deck. For drains, cut 56 holes in the field. 57 I. Recessed Sump Pans: Single-piece steel sheet, 0.0747 inch thick, of same material and finish as deck, with 3-inch-58 wide flanges and sloped recessed pans of 1-1/2-inch minimum depth. For drains, cut holes in the field.

2

3 4

5 6

7

8

9

10 11

12

13

14

15

16

17

18

19

24

25

26

27

28

29

33

34

35

36

39

40

41

45

49

51

52

57

- J. Galvanizing Repair Paint: ASTM A780/A780M.
 - K. Repair Paint: Manufacturer's standard rust-inhibitive primer of same color as primer.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine supporting frame and field conditions for compliance with requirements for installation tolerances and other conditions affecting performance of the Work.
- B. Proceed with installation only after unsatisfactory conditions have been corrected.

3.2 INSTALLATION, GENERAL

- A. Install deck panels and accessories in accordance with SDI C, SDI NC, and SDI RD, as applicable; manufacturer's written instructions; and requirements in this Section.
- B. Install temporary shoring before placing deck panels if required to meet deflection limitations.
- C. Locate deck bundles to prevent overloading of supporting members.
- D. Place deck panels on supporting frame and adjust to final position with ends accurately aligned and bearing on supporting frame before being permanently fastened. Do not stretch or contract side-lap interlocks.
 - 1. Align cellular deck panels over full length of cell runs and align cells at ends of abutting panels.
 - E. Place deck panels flat and square and fasten to supporting frame without warp or deflection.
- F. Cut and neatly fit deck panels and accessories around openings and other work projecting through or adjacent to deck.
- G. Provide additional reinforcement and closure pieces at openings as required for strength, continuity of deck, and
 support of other work.
 - H. Comply with AWS requirements and procedures for manual shielded metal arc welding, appearance and quality of welds, and methods used for correcting welding work.
 - I. Retain paragraph below if mechanical fastening is permitted.
 - J. Mechanical fasteners may be used in lieu of welding to fasten deck. Locate mechanical fasteners and install in accordance with deck manufacturer's written instructions.

30 3.3 INSTALLATION OF ROOF DECK

- 31A.Fasten roof-deck panels to steel supporting members by arc spot (puddle) welds of the surface diameter indicated32or arc seam welds with an equal perimeter that is not less than 1-1/2 inches long, and as follows:
 - 1. Weld Diameter: 5/8 inch, nominal.
 - 2. Weld Spacing: Weld edge and interior ribs of deck units with a minimum of two welds per deck unit at each support. Space welds as indicated.
 - 3. Weld Washers: Install weld washers at each weld location.
- B. Side-Lap and Perimeter Edge Fastening: Fasten side laps and perimeter edges of panels between supports, at
 intervals not exceeding the lesser of one-half of the span or 18 inches, and as follows:
 - 1. Mechanically fasten with self-drilling, No. 10 diameter or larger, carbon-steel screws.
 - 2. Mechanically clinch or button punch.
 - 3. End Joints: Lapped 2 inches minimum or butted at Contractor's option.
- 42 C. Roof Sump Pans and Sump Plates: Install over openings provided in roof deck and weld or mechanically fasten
 43 flanges to top of deck. Space welds or mechanical fasteners not more than 12 inches apart with at least one weld
 44 or fastener at each corner.
 - 1. Install reinforcing channels or zees in ribs to span between supports and weld or mechanically fasten.
- D. Miscellaneous Roof-Deck Accessories: Install ridge and valley plates, finish strips, end closures, and reinforcing
 channels in accordance with deck manufacturer's written instructions. Weld or mechanically fasten to substrate to
 provide a complete deck installation.
 - 1. Weld cover plates at changes in direction of roof-deck panels unless otherwise indicated.
- 50 E. Sound-Absorbing Insulation: Installation into topside ribs of deck as specified in **<Insert Section number and title>**.

3.4 REPAIR

- 53A.Galvanizing Repairs: Prepare and repair damaged galvanized coatings on both surfaces of deck with galvanized54repair paint in accordance with ASTM A780/A780M and manufacturer's written instructions.
- 55 B. Repair Painting: 56 1. Wire brus
 - 1. Wire brush and clean rust spots, welds, and abraded areas on both surfaces of prime-painted deck immediately after installation and apply repair paint.

2.	Apply repair paint, of same color as adjacent shop-primed deck, to bottom surfaces of deck exposed to view.
3.	Wire brushing, cleaning, and repair painting of bottom deck surfaces are included in Section 099113
	"Exterior Painting" and Section 099123 "Interior Painting."
	END OF SECTION
	2. 3.

1		SECTION 05 40 00	
2		COLD-FORMED METAL FRAMING	
3			
4 5	1 1	SUMMΔRV 1	
6	1.1	PRFINSTALLATION MEETINGS	
7	1.3	ACTION SUBMITTALS	
8	1.4	INFORMATIONAL SUBMITTALS	
9	1.5	OUALITY ASSURANCE	
10	1.6	DELIVERY, STORAGE, AND HANDLING	
11	PART 2 - F	2 PRODUCTS	
12	2.1	PERFORMANCE REQUIREMENTS	
13	2.2	COLD-FORMED STEEL FRAMING MATERIALS	
14	2.3	EXTERIOR NON-LOAD-BEARING WALL FRAMING	
15	2.4	FRAMING ACCESSORIES	
16	2.5	ANCHORS, CLIPS, AND FASTENERS	
17	2.6	MISCELLANEOUS MATERIALS	
18	2.7	FABRICATION 4	
19	PART 3 - E	XECUTION	
20	3.1	EXAMINATION	
21	3.2	PREPARATION	
22	3.3	INSTALLATION, GENERAL	
23	3.4	INSTALLATION OF EXTERIOR NONLOADBEARING WALL FRAMING	
24	3.5	INSTALLATION TOLERANCES	
25	3.6	REPAIR	
26	3.7	FIELD QUALITY CONTROL	
27	3.8	PROTECTION	
29 30	<u> PART 1 - (</u>	GENERAL	
31	1.1 SUM	MARY	
32	Α.	Section Includes:	
33		1. Exterior non-load-bearing wall framing.	
34			
35	1.2 PREI	NSTALLATION MEETINGS	
36	А.	Preinstallation Conference: Conduct conference at Project site .	
37			
38	1.3 ACTI		
39	А.	Product Data: For the following:	
40		1. Cold-formed steel framing materials.	
41		2. Exterior non-load-bearing wall framing.	
42	В.	Shop Drawings:	
43		1. Include layout, spacings, sizes, thicknesses, and types of cold-formed sized framing; labrication, and factoring and anchorage details including mechanical factories.	
44 15		Idicate reinforcing channels, encluding mechanical idicities.	
45		2. Indicate remoticing channels, opening framming, suppremental framming, strapping, bracing, bridging, splices,	
40	C	Delegated Design Submittal: For cold-formed steel framing	
47	С.	Delegated Design Submittal. For cold-formed steer manning.	
40 49	1.4 INFO	RMATIONAL SURMITTALS	
50	Δ	Welding certificates	
51	B.	Product Certificates: For each type of code-compliance certification for studs and tracks	
52	С.	Product Test Reports: For each listed product, for tests performed by manufacturer and witnessed by a qualific	
53	2.	testing agency.	
54		1. Steel sheet.	
55		2. Expansion anchors.	
56		3. Power-actuated anchors.	
57		4. Mechanical fasteners.	
58		5. Vertical deflection clips.	

		6. Horizontal drift deflection clips
		 Miscellaneous structural clips and accessories.
	D.	Research Reports:
		1. For nonstandard cold-formed steel framing post-installed anchors and power-actuated fasteners , from
		ICC-ES or other qualified testing agency acceptable to authorities having jurisdiction.
		2. For sill sealer gasket/termite barrier, showing compliance with ICC-ES AC380.
1.5	OUAL	ITY ASSURANCE
	A.	Product Tests: Mill certificates or data from a qualified independent testing agency, or in-house testing with
		calibrated test equipment, indicating steel sheet complies with requirements, including base-metal thickness, yield
		strength, tensile strength, total elongation, chemical requirements, and metallic-coating thickness.
	В.	Code-Compliance Certification of Studs and Tracks: Provide documentation that framing members are certified
		according to the product-certification program of the Certified Steel Stud Association.
	C.	Welding Qualifications: Qualify procedures and personnel according to the following:
		1. AWS D1.1/D1.1M, "Structural Welding Code - Steel."
		 AWS D1.3/D1.3M, "Structural Welding Code - Sheet Steel."
16		
1.0		Protect and store cold-formed steel framing from corrosion moisture staining deformation and other damage
	/	during delivery, storage, and handling as required in AISI \$202
PAF	RT 2 - P	<u>RODUCTS</u>
~ 4		
2.1	A	DRIMANCE REQUIREMENTS Delegated Decign: Engage a gualified professional engineer as defined in Section 014000 "Quality Requirements "
	А.	to design cold-formed steel framing
	в	Structural Performance: Provide cold-formed steel framing canable of withstanding design loads within limits and
	Б.	under conditions indicated.
		1. Design Loads: As indicated on Drawings.
		2. Deflection Limits: Design framing systems to withstand design loads without deflections greater than the
		following:
		a. Exterior Non-Load-Bearing Framing: Horizontal deflection of 1/360 of the wall height.
		3. Design framing systems to provide for movement of framing members located outside the insulated
		building envelope without damage or overstressing, sheathing failure, connection failure, undue strain on
		fasteners and anchors, or other detrimental effects when subject to a maximum ambient temperature
		change of 120 deg F.
		4. Design framing system to maintain clearances at openings, to allow for construction tolerances, and to
		accommodate live load deflection of primary building structure as follows:
		a. Upward and downward movement of 1/2 inch.
		E Design autorian non-load bearing wall framing to accommodate begingered deflection without accord for
		contribution of sheathing materials
	C	Cold-Formed Steel Framing Standards: Unless more stringent requirements are indicated framing complies with
	с.	AISI \$100 and ASTM C955.
	D.	Fire-Resistance Ratings: Comply with ASTM E119: testing by a qualified testing agency. Identify products with
		appropriate markings of applicable testing agency.
		1. Indicate design designations from UL's "Fire Resistance Directory" or from the listings of another qualified
		testing agency acceptable to authorities having jurisdiction.
		· · · - ·
2.2	COLD	-FORMED STEEL FRAMING MATERIALS
	Α.	Framing Members, General: Comply with ASTM C955 for conditions indicated.
	В.	Steel Sheet: ASTM A1003/A1003M, Structural Grade, Type H, metallic coated, of grade and coating designation as
		follows:
		1. Grade: As required by structural performance
		2. Coating: G60, A60, AZ50, or GF30
	C.	Steel Sheet for Vertical Deflection Clips: ASTM A653/A653M, structural steel, zinc coated, of grade and coating as
		TOIIOWS:

1		1. Grade: As required by structural performance			
2		2. Coating: G60			
3					
4	2.3 EXTE	ERIOR NON-LOAD-BEARING WALL FRAMING			
5	Α.	Steel Studs: Manufacturer's standard C-shaped steel studs, of web depths indicated, punched, with stiffened			
6		flanges, and as follows:			
7		1. Minimum Base-Metal Thickness: 0.0538 inch.			
8		2. Flange Width: 1-5/8 inches.			
9	В.	Steel Track: Manufacturer's standard U-shaped steel track, of web depths indicated, unpunched, with unstiffened			
10		flanges, and as follows:			
11		1. Minimum Base-Metal Thickness: 0.0329 inch.			
12		2. Flange Width: 1-1/4 inches.			
13	С.	Vertical Deflection Clips, Exterior: Manufacturer's standard head clips, capable of accommodating upward and			
14		downward vertical displacement of primary structure through positive mechanical attachment to stud web.			
15	D.	Single Deflection Track: Manufacturer's single, deep-leg, U-shaped steel track; unpunched, with unstiffened flanges,			
16		of web depth to contain studs while allowing free vertical movement, with flanges designed to support horizontal			
17		loads and transfer them to the primary structure, and as follows:			
18		1. Minimum Base-Metal Thickness: 0.0428 inch.			
19		2. Flange Width: 1 inch plus the design gap for one-story structures.			
20	Ε.	Drift Clips: Manufacturer's standard bypass or head clips, capable of isolating wall stud from upward and downward			
21		vertical displacement and lateral drift of primary structure through positive mechanical attachment to stud web and			
22		structure.			
23					
24	2.4 FRAM	IING ACCESSORIES			
25	Α.	Fabricate steel-framing accessories from ASTM A1003/A1003M, Structural Grade, Type H, metallic coated steel			
26		sheet, of same grade and coating designation used for framing members.			
27	В.	Provide accessories of manufacturer's standard thickness and configuration, unless otherwise indicated, as follows:			
28		1. Supplementary framing.			
29		2. Bracing, bridging, and solid blocking.			
30		3. Web stiffeners.			
31		4. Anchor clips.			
32		5. End clips.			
33		6. Foundation clips.			
34		7. Gusset plates.			
35		8. Stud kickers and knee braces.			
36		9. Joist hangers and end closures.			
37		10. Hole-reinforcing plates.			
38		11. Backer plates.			
39					
40	2.5 ANC	ORS, CLIPS, AND FASTENERS			
41	Α.	Steel Shapes and Clips: ASTM A36/A36M, zinc coated by hot-dip process according to ASTM A123/A123M.			
42	В.	Anchor Bolts: ASTM F1554, Grade 36, threaded carbon-steel hex-headed bolts, carbon-steel nuts, and flat,			
43		hardened-steel washers; zinc coated by hot-dip process according to ASTM A153/A153M, Class C.			
44	С.	Post-Installed Anchors: Fastener systems with bolts of same basic metal as fastened metal, if visible, unless			
45		otherwise indicated; with working capacity greater than or equal to the design load, according to an evaluation			
46		report acceptable to authorities having jurisdiction, based on ICC-ES AC01 as appropriate for the substrate.			
47		 Uses: Securing cold-formed steel framing to structure. 			
48		2. Type: Torque-controlled expansion anchor.			
49		3. Material for Interior Locations: Carbon-steel components zinc plated to comply with ASTM B633 or			
50		ASTM F1941, Class Fe/Zn 5, unless otherwise indicated.			
51		4. Material for Exterior or Interior Locations and Where Stainless Steel Is Indicated: Alloy Group 1 stainless			
52		steel bolts, ASTM F593, and nuts, ASTM F594.			
53	D.	Power-Actuated Anchors: Fastener systems with working capacity greater than or equal to the design load,			
54		according to an evaluation report acceptable to authorities having jurisdiction, based on ICC-ES AC70.			
55	Ε.	Mechanical Fasteners: ASTM C1513, corrosion-resistant-coated, self-drilling, self-tapping, steel drill screws.			
56		1. Head Type: Low-profile head beneath sheathing; manufacturer's standard elsewhere.			
57	F.	Welding Electrodes: Comply with AWS standards.			
58					

1	2.6 M	ISCELLANEOUS MATERIALS
2	Α.	Galvanizing Repair Paint: ASTM A780/A780M.
3	В.	Cement Grout: Portland cement, ASTM C150/C150M, Type I; and clean, natural sand, ASTM C404. Mix at ratio of
4		1 part cement to 2-1/2 parts sand, by volume, with minimum water required for placement and hydration.
5	С.	Nonmetallic, Nonshrink Grout: Factory-packaged, nonmetallic, noncorrosive, nonstaining grout, complying with
6		ASTM C1107/C1107M, and with a fluid consistency and 30-minute working time.
7	D.	Shims: Load-bearing, high-density, multimonomer, nonleaching plastic; or cold-formed steel of same grade and
8		metallic coating as framing members supported by shims.
9	Ε.	Sill Sealer Gasket: Closed-cell neoprene foam, 1/4 inch thick, selected from manufacturer's standard widths to
10		match width of bottom track or rim track members as required.
11	F.	Sill Sealer Gasket/Termite Barrier: Minimum 68-mil nominal thickness, self-adhering sheet consisting of 64 mils of
12		rubberized asphalt laminated on one side to a 4-mil- thick, polyethylene-film reinforcement, and with release liner
13		on adhesive side.
14		1. Physical Properties:
15		a. Peel Adhesion: 17.0 lb/in of width when tested in accordance with ASTM D412.
16 17		 Low-Temperature Flexibility: Pass at minus 25 deg F when tested in accordance with ASTM D146/D146M.
18 10		c. Water Vapor Permeance: 0.05 perm maximum when tested in accordance with ASTM E96/E96M,
19		Melliou D. d Basistance to Termite Depotration: Comply with ICC ES AC290
20		a. Resistance to remite renetration. Comply with ICC-ES ACS60.
21	27 FA	BRICATION
22	Δ	Fabricate cold-formed steel framing and accessories plumb, square, and true to line, and with connections securely
24		fastened, according to referenced AISI's specifications and standards, manufacturer's written instructions, and
25		requirements in this Section.
26		1. Fabricate framing assemblies using jigs or templates.
27		 Cut framing members by sawing or shearing: do not torch cut.
28		3. Fasten cold-formed steel framing members by welding, screw fastening, clinch fastening, pneumatic pin
29		fastening, or riveting as standard with fabricator. Wire tying of framing members is not permitted.
30		a. Comply with AWS D1.3/D1.3M requirements and procedures for welding, appearance and guality of
31		welds, and methods used in correcting welding work.
32		b. Locate mechanical fasteners and install according to Shop Drawings, with screws penetrating joined
33		members by no fewer than three exposed screw threads.
34		4. Fasten other materials to cold-formed steel framing by welding, bolting, pneumatic pin fastening, or screw
35		fastening, according to Shop Drawings.
36	В.	Reinforce, stiffen, and brace framing assemblies to withstand handling, delivery, and erection stresses. Lift
37		fabricated assemblies by means that prevent damage or permanent distortion.
38	С.	Tolerances: Fabricate assemblies level, plumb, and true to line to a maximum allowable variation of 1/8 inch in 10
39		feet and as follows:
40		1. Spacing: Space individual framing members no more than plus or minus 1/8 inch from plan location.
41		Cumulative error are not to exceed minimum fastening requirements of sheathing or other finishing
42		materials.
43		2. Squareness: Fabricate each cold-formed steel framing assembly to a maximum out-of-square tolerance of
44		1/8 inch.
45 46	PART 3	- EXECUTION
47		
48	3.1 EX	AMINATION
49	Α.	Examine substrates, areas, conditions, and abutting structural framing for compliance with requirements for
50		installation tolerances and other conditions affecting performance of the Work.
51	В.	Proceed with installation only after unsatisfactory conditions have been corrected.
52 52	22 00	
55	J.Z PR	Refore snraved fire-resistive materials are applied attach continuous angles supplementary framing or tracks to
55	Π.	structural members indicated to receive sprayed fire-resistive materials
56	R	Install sill sealer gasket/termite harrier in accordance with manufacturer's written instructions at the underside of
57	υ.	wall bottom track or rim track and at the top of foundation wall or slab at stud or joist locations
58		

1	3.3 INST	ALLATION, GENERAL
2	Α.	Cold-formed steel framing may be shop or field fabricated for installation, or it may be field assembled.
3	В.	Install cold-formed steel framing according to AISI S200, AISI S202, and manufacturer's written instructions unless
4		more stringent requirements are indicated.
5	C.	Install shop- or field-fabricated, cold-formed framing and securely anchor to supporting structure.
6		1. Screw, bolt, or weld wall panels at horizontal and vertical junctures to produce flush, even, true-to-line
7		ioints with maximum variation in plane and true position between fabricated panels not exceeding 1/16
8		inch.
9	П	Install cold-formed steel framing and accessories plumb square and true to line, and with connections securely
10	υ.	fastened
11		1 Cut framing members by sawing or shearing: do not torch cut
12		2. Easten cold-formed steal framing members by welding screw fastening clinch fastening or riveting. Wire
12		2. Instantion of framing members is not normitted
17		comply with AWS D1 2/D1 2M requirements and procedures for welding, appearance and quality of
14		a. Comply with AWS D1.5/D1.5W requirements and procedures for weiging, appearance and quality of weights and mothods used in correcting weights and procedures for weights, appearance and quality of
15		weids, and methods used in correcting weiding work.
16		b. Locate mechanical fasteners, install according to Shop Drawings, and comply with requirements for
1/	-	spacing, edge distances, and screw penetration.
18	E.	Install framing members in one-piece lengths unless splice connections are indicated for track or tension members.
19	F.	install temporary bracing and supports to secure framing and support loads equal to those for which structure was
20		designed. Maintain braces and supports in place, undisturbed, until entire integrated supporting structure has been
21		completed and permanent connections to framing are secured.
22	G.	Fasten hole-reinforcing plate over web penetrations that exceed size of manufacturer's approved or standard
23		punched openings.
24		
25		
26	3.4 INST	ALLATION OF EXTERIOR NONLOADBEARING WALL FRAMING
27	Α.	Install continuous tracks sized to match studs. Align tracks accurately and securely anchor to supporting structure.
28	В.	Fasten both flanges of studs to top and bottom track unless otherwise indicated. Space studs as follows:
29		1. Stud Spacing: 16 inches.
30	С.	Set studs plumb, except as needed for diagonal bracing or required for nonplumb walls or warped surfaces and
31		similar requirements.
32	D.	Isolate non-load-bearing steel framing from building structure to prevent transfer of vertical loads while providing
33		lateral support.
34		 Install single deep-leg deflection tracks and anchor to building structure.
35		2. Install double deep-leg deflection tracks and anchor outer track to building structure.
36		3. Connect vertical deflection clips to infill studs and anchor to building structure.
37		4. Connect drift clips to cold-formed steel framing and anchor to building structure.
38	Ε.	Install horizontal bridging in wall studs, spaced vertically in rows indicated on Shop Drawings but not more than 48
39		inches apart. Fasten at each stud intersection.
40		1. Channel Bridging: Cold-rolled steel channel, welded or mechanically fastened to webs of punched studs.
41		2. Strap Bridging: Combination of flat, taut, steel sheet straps of width and thickness indicated and stud-track
42		solid blocking of width and thickness to match studs. Fasten flat straps to stud flanges and secure solid
43		blocking to stud webs or flanges.
44		3. Bar Bridging: Proprietary bridging bars installed according to manufacturer's written instructions.
45	F.	Top Bridging for Single Deflection Track: Install row of horizontal bridging within 12 inche of single deflection track.
46		Install a combination of bridging and stud or stud-track solid blocking of width and thickness matching studs,
47		secured to stud webs or flanges.
48		1. Install solid blocking at centers indicated on Shop Drawings.
49	G.	Install miscellaneous framing and connections, including stud kickers, web stiffeners, clip angles, continuous angles,
50		anchors, and fasteners, to provide a complete and stable wall-framing system.
51		,
52	3.5 INST	ALLATION TOLERANCES
53	A.	Install cold-formed steel framing level, plumb, and true to line to a maximum allowable tolerance variation of 1/8
54		inch in 10 feet and as follows:
55		1. Space individual framing members no more than plus or minus 1/8 inch from plan location. Cumulative
56		error are not to exceed minimum fastening requirements of sheathing or other finishing materials
57		

1 3.6 REPAIR

4

5

13

18

A. Galvanizing Repairs: Prepare and repair damaged galvanized coatings on fabricated and installed cold-formed steel
 framing with galvanized repair paint according to ASTM A780/A780M and manufacturer's written instructions.

3.7 FIELD QUALITY CONTROL

- 6 A. Testing: Owner will engage a qualified independent testing and inspecting agency to perform field tests and 7 inspections and prepare test reports.
- 8 B. Field and shop welds will be subject to testing and inspecting.
- 9 C. Testing agency will report test results promptly and in writing to Contractor and Architect.
- 10 D. Cold-formed steel framing will be considered defective if it does not pass tests and inspections.
- 11E.Additional testing and inspecting, at Contractor's expense, will be performed to determine compliance of replaced12or additional work with specified requirements.

14 3.8 PROTECTION

A. Provide final protection and maintain conditions, in a manner acceptable to manufacturer and Installer, that ensure
 that cold-formed steel framing is without damage or deterioration at time of Substantial Completion.

END OF SECTION

1 2				SECTION 05 44 00 COLD-FORMED METAL TRUSSES
3	PAR	Т1-	GENER	AL
4	1	.1	RELAT	TED DOCUMENTS
5	1	.2	SUMN	/IARY
6	1	.3	PREIN	ISTALLATION MEETINGS
7	1	.4	ACTIC	IN SUBMITTALS
8	1	.5	INFOF	MATIONAL SUBMITTALS
9	1	.6	QUAL	ITY ASSURANCE
10	PAR	Т2-	PRODU	JCTS
11	2	.1	PERFC	DRMANCE REQUIREMENTS
12	2	.2	COLD	-FORMED STEEL TRUSS MATERIALS
13	2	.3	ROOF	TRUSSES
14	2	.4	TRUSS	S ACCESSORIES
15	2	.5	ANCH	ORS, CLIPS, AND FASTENERS
16	2	.6	MISCE	LLANEOUS MATERIALS
17	2	.7	FABRI	CATION
18	PAR	тз-	EXECU	TION
19	3	.1	EXAM	INATION
20	3	.2	PRFPA	ARATION 4
21	3	3	INSTA	4 ATION
22	3	 ۵	FRECT	
22	2	5	REDAI	R A
23	2	6		
25	2	.0	PROT	ECTION 5
25	5	.,	T NOT	
20		т1.	GENER	
27	<u>r An</u>	1 4 -	ULINLI	
28				
29	1.1	REL	ATED D	JOCUMENTS
30		Α.	Drawir	igs and general provisions of the Contract, including General and Supplementary Conditions and Divi-
31			sion 01	. Specification Sections, apply to this Section.
32				
33	1.2	SUN	MMAR	1
34		Α.	SECTI	ON INCLUDES: Roof trusses.
35		В.	RELAT	ED REQUIREMENTS:
36			1.	Section 05 12 00 "Structural Steel Framing"
37			2.	Section 05 40 00 "Cold-Formed Metal Framing"
38				
39	1.3	PRE	INSTAL	LATION MEETINGS
40		Α.	Preins	stallation Conference: Conduct conference at Project site with the following attendees:
41			1. C	Contractor's Superintendent
42			2. A	Architect
43			3. C	OFD Construction Representative
44			4. C	cold Formed Metal Truss Supplier
45			5. C	cold Formed Metal Truss Subcontractor
46				
47	1.4	АСТ		JBMITTALS
48		Α.	PROD	UCT DATA:
49			1. C	Cold-formed steel truss materials.
50			2. 0	Cold-formed steel truss clips and connectors.
51			3. A	Anchor bolts.
52			4. F	ost-installed anchors.
53			5. F	Power-actuated fasteners.
54			6 1	Aechanical fasteners
55		в	SHOP	DRAWINGS
56		5.	1	Include layout spacings sizes thicknesses and types of cold-formed steel trusses: fabrication:
57			1.	and fastaning and anchorage details including mechanical fastaners
57				and restering and anonorage actails, including methanical resteries.

1			2. Indicate reinforcing channels, opening framing, supplemental framing, strapping, bracing, bridg-
2			ing, spices, accessories, connection details, and attactiment to aujoining work.
3			3. Delegated-Design Submittal: For Cold-formed steel trusses.
4			
5	1.5	INFO	
6		Α.	Qualification Data: For testing agency.
7		В.	Welding certificates.
8		C.	Product Test Reports: For each listed product, for tests performed by manufacturer and witnessed by a
9			qualified testing agency.
10			1. Steel sheet.
11			2. Expansion anchors.
12			3. Power-actuated anchors.
13			4. Mechanical fasteners.
14			5. Miscellaneous structural clips and accessories.
15			6. Field quality-control reports.
16			
17	1.6	QUA	ALITY ASSURANCE
18		Α.	Testing Agency Qualifications: Qualified according to ASTM E329 for testing indicated.
19		В.	Product Tests: Mill certificates or data from a qualified independent testing agency, or in-house testing with
20			calibrated test equipment, indicating steel sheet complies with requirements, including base-metal thick-
21			ness, yield strength, tensile strength, total elongation, chemical requirements, and metallic-coating thick-
22			ness.
23		C	Welding Qualifications: Qualify procedures and personnel according to the following:
24		с.	1 AWS D1 1/D1 1M "Structural Welding Code - Steel "
25			2 AWS D1 3/D1 3M "Structural Welding Code - Sheet Steel "
26			
27	PAR	Т2-	PRODUCTS
28			
29	2.1	PER	FORMANCE REQUIREMENTS
30		Α.	Delegated Design: Engage a qualified professional engineer, as defined in Section 014000 "Quality Require-
31			ments," to design cold-formed steel trusses.
32		В.	Structural Performance: Provide cold-formed steel trusses capable of withstanding design loads within lim-
33			its and under conditions indicated.
34			1. Design Loads: As indicated on Drawings
35			 Deflection Limits: Design trusses to withstand design loads without deflections greater than the follow-
36			ing:
37			i Boof Trusses: Vertical deflection of 1/240 of the span
38			3 Design trusses to provide for movement of truss members located outside the insulated building enve-
30			Iong without damage or overstressing, sheathing failure, connection failure, undue strain on fasteners
40			and anchore or other datamental officies when subject to a maximum ambient temperature changes of
40			and ancients, of other detrimental effects when subject to a maximum ambient temperature change of 120 dog E
41 42			120 UEB F. A Cold Formed Steel Truce Standards: Unless more stringent requirements are indicated trucses shall
42			4. Concromed steer muss standards. Othess more stringent requirements are multicated, trusses shall
43			comply with the following:
44			I. Floor and Root Systems: AISI 5210.
45			II. Lateral Design: AISI 5213.
46		~	III. Root Irusses: AISI S214.
47		C.	Fire-Resistance Ratings: Comply with ASTM E119; testing by a qualified testing agency. Identify products
48		_	with appropriate markings of applicable testing agency.
49		D.	Indicate design designations from UL or from the listings of another qualified testing agency acceptable to
50			authorities having jurisdiction.
51			
52	2.2	COL	D-FORMED STEEL TRUSS MATERIALS
53		A. S	teel Sheet: ASTM A1003/A1003M, Structural Grade, Type H, metallic coated, of grade and coating designa-
54		tio	on as follows:
55		1.	Grade: As required by structural performance.
56		2.	Coating: G60.

1	2.3	ROOF TRUSSES			
2		A. Roof Truss Members: Manufacturer's standard steel sections.			
3		1. Connecting Flange Width: 1-5/8 inches, minimum at top and bottom chords connecting to sheath			
4		other directly fastened construction.			
5		2. Minimum Base-Metal Thickness: 0.0428 inch.			
6		3. Section Properties: As required by design.			
7					
8	2.4	TRUSS ACCESSORIES			
9		A. Fabricate steel-truss accessories from steel sheet, ASTM A1003/A1003M, Structural Grade, Type H, metallic			
10		coated steel sheet, of same grade and coating designation used for truss members.			
11		B. Provide accessories of manufacturer's standard thickness and configuration unless otherwise indicated.			
12					
13	2.5	ANCHORS, CLIPS, AND FASTENERS			
14		C. Steel Shapes and Clips: ASTM A36/A36M, zinc coated by hot-dip process according to ASTM A123/A123M.			
15		D. Anchor Bolts: ASTM F1554, Grade 36, threaded carbon-steel hex-headed bolts, carbon-steel nuts, and flat,			
16		hardened-steel washers; zinc coated by hot-dip process according to ASTM A153/A153M, Class C.			
17		E. Post-Installed Anchors: Fastener systems with bolts of same basic metal as fastened metal, if visible, unless			
18		otherwise indicated; with working capacity greater than or equal to the design load, according to an evalua-			
19		tion report acceptable to authorities having jurisdiction, based on ICC-ES AC01, ICC-ES AC193, ICC-ES AC58			
20		or ICC-ES AC308 as appropriate for the substrate.			
21		1. Uses: Securing cold-formed steel trusses to structure.			
22		2. Type: Torque-controlled expansion anchor, Torque-controlled adhesive anchor or adhesive an-			
23		chor.			
24		3. Material for Interior Locations: Carbon-steel components zinc plated to comply with ASTM B633			
25		or ASTM F1941, Class Fe/Zn 5, unless otherwise indicated.			
26		4. Material for Exterior or Interior Locations and Where Stainless Steel Is Indicated: Alloy Group 1, or			
27		Group 2 stainless-steel bolts, ASTM F593, and nuts, ASTM F594.			
28		F. Power-Actuated Fasteriers: Fasterier systems with working capacity greater than or equal to the design			
29		Nachanical Eastenard, ACTM (1512, correction resistant coated, solf drilling, solf tenning steel drill corous			
50 21		 Michaillan Fasteriers. As finit CLS15, conosion-resistant-coaled, sen-unning, sen-tapping steel unit screws. Head Type: Low profile head honeath cheathing: manufacturer's standard elsewhere. 			
37		H Welding Electrodes: Comply with AWS standards			
32		The weighing Electrodes. Comply with Aws standards.			
34	26	MISCELLANEQUIS MATERIALS			
35	2.0	Galvanizing Renair Paint: ASTM A780/A780M MII-P-21035B or SSPC-Paint 20			
36		B. Shims: Load-bearing, high-density multimonomer, nonleaching plastic: or cold-formed steel of same grade			
37		and metallic coating as truss members supported by shims.			
38					
39	2.7	FABRICATION			
40		A. Fabricate cold-formed steel trusses and accessories plumb, square, and true to line, and with connections			
41		securely fastened, according to referenced AISI's specifications and standards, manufacturer's written in-			
42		structions, and requirements in this Section.			
43		1. Fabricate trusses using jigs or templates.			
44		2. Cut truss members by sawing or shearing; do not torch cut.			
45		3. Fasten cold-formed steel truss members by welding, screw fastening, clinch fastening, pneumatic pin			
46		fastening, or riveting as standard with fabricator.			
47		a. Comply with AWS D1.3/D1.3M requirements and procedures for welding, appearance and quality			
48		of welds, and methods used in correcting welding work.			
49		4. Fasten other materials to cold-formed steel trusses by welding, bolting, pneumatic pin fastening, or			
50		screw fastening, according to Shop Drawings.			
51		B. Reinforce, stiffen, and brace trusses to withstand handling, delivery, and erection stresses. Lift fabricated			
52		trusses by means that prevent damage or permanent distortion.			
53		C. Tolerances: Fabricate assemblies level, plumb, and true to line to a maximum allowable variation of 1/8 inch			
54		in 10 feet (1:960) and as follows:			
55		a. Spacing: Space individual truss members no more than plus or minus 1/8 inch from plan location.			
56		Cumulative error shall not exceed minimum fastening requirements of sheathing or other finish-			
57		ing materials.			

1			b. Squareness: Fabricate each cold-formed steel truss to a maximum out-of-square tolerance of 1/8					
2			inch.					
3								
4	<u>PAR</u>	PART 3 - EXECUTION						
5	3.1	EXAMI	INATION					
7	5.1	A. F	xamine substrates, areas, conditions, and abutting trusses and framing for compliance with requirements					
8		fo	or installation tolerances and other conditions affecting performance of the Work.					
9		B. Pr	roceed with installation only after unsatisfactory conditions have been corrected.					
10								
11	3.2	PREPA	RATION					
12		A. Be	efore sprayed fire-resistive materials are applied, attach continuous angles, supplementary framing, or					
13		tr	acks to structural members indicated to receive sprayed fire-resistive materials.					
14		B. At	Iter applying sprayed fire-resistive materials, remove only as much of these materials as needed to com-					
15		pi +h	ete installation of cold-formed steel trusses without reducing thickness of fire-resistive materials below					
10		d:						
18		u	anage.					
19	3.3	INSTAL	LATION					
20		A. In	stall bridge, and brace cold-formed steel trusses according to AISI S200, AISI S202, AISI S214, and manu-					
21		fa	cturer's written instructions unless more stringent requirements are indicated.					
22		1.	Coordinate with wall framing to align webs of bottom chords and load-bearing studs or continuously					
23			reinforce track to transfer loads to structure.					
24		2.	Anchor trusses securely at all bearing points.					
25		3.	Install continuous bridging and permanently brace trusses as indicated on Shop Drawings and designed					
26			according to CFSEI's Technical Note 551e, "Design Guide: Permanent Bracing of Cold-Formed Steel					
27		P In	I russes Istall cold formed stool trusses and accessories true to line and location, and with connections securely					
20 29		р. III fa	staned					
30		1	Frect trusses with plane of truss webs plumb and parallel to each other. Align and accurately position					
31			trusses at required spacings.					
32		2.	Erect trusses without damaging truss members or connections.					
33		3.	Fasten cold-formed steel trusses by welding or mechanical fasteners.					
34			i. Comply with AWS D1.3/D1.3M requirements and procedures for welding, appearance and					
35			quality of welds, and methods used in correcting welding work.					
36			ii. Locate mechanical fasteners, install according to Shop Drawings, and comply with require-					
37		- ·	ments for spacing, edge distances, and screw penetration.					
38		C. In	stall temporary bracing and supports to secure trusses and support loads equal to those for which struc-					
39		tu	ire was designed. Maintain braces and supports in place, undisturbed, until entire integrated supporting					
40 //1			ructure has been completed and permanent connections to trusses are secured.					
42		F D	o not alter cut or remove truss members or connections of trusses					
43		2. 0						
44	3.4	ERECT	ION TOLERANCES					
45		A. In	stall cold-formed steel trusses level, plumb, and true to line to a maximum allowable tolerance variation					
46		of 1/8	inch in 10 feet (1:960) and as follows:					
47		1.	Space individual trusses no more than plus or minus 1/8 inch from plan location. Cumulative error shall					
48			not exceed minimum fastening requirements of sheathing or other finishing materials.					
49	<u> </u>							
50	3.5	REPAIR	(
51 52		A. G	aivanizing Repairs: Prepare and repair damaged gaivanized coatings on fabricated and installed cold-					
52 53		T .	written instructions					
55 54		v						
55	3.6	FIELD (QUALITY CONTROL					
56	-	A. Te	esting Agency: Owner will engage a qualified testing agency to perform tests and inspections.					

- В. Cold-formed metal trusses will be considered defective if they do not pass tests and inspections. 2
 - Prepare test and inspection reports. C.
- 3 4

1

3.7 PROTECTION

- 5 Provide final protection and maintain conditions, in a manner acceptable to manufacturer and Installer, Α. 6 which ensure that cold-formed steel trusses are without damage or deterioration at time of Substantial 7 Completion. 8
 - END OF SECTION
SECTION 06 10 00 ROUGH CARPENTRY

2		ROUGH CARPENTRY
3	PART 1 -	GENERAL
4	1.1	SUMMARY
5	Α.	Section Includes:
6		1. Rooftop equipment bases and support curbs.
7		2. Wood blocking, cants, and nailers.
8		3. Plywood backing panels.
9	1.2	ACTION SUBMITTALS
10	А.	Product Data:
11		1. For each type of process and factory-fabricated product.
12		2. For preservative-treated wood products.
13	1.3	INFORMATIONAL SUBMITTALS
14	А.	Material Certificates:
15		1. For dimension lumber specified to comply with minimum allowable unit stresses. Indicate species and grade
16		selected for each use and design values approved by the ALSC Board of Review.
17		2. For preservative-treated wood products. Indicate type of preservative used and net amount of preservative
18		retained.
19	В.	Evaluation Reports: For the following, from ICC-ES:
20		1. Wood-preservative-treated wood.
21		2. Fire-retardant-treated wood.
22		3. Engineered wood products.
23		4. Shear panels.
24		5. Power-driven fasteners.
25		6. Post-installed anchors.
26		7. Metal framing anchors.
27	PART 2 -	PRODUCTS
28	2.1	WOOD PRODUCTS, GENERAL
29	 A.	Lumber: Comply with DOC PS 20 and applicable rules of grading agencies indicated. If no grading agency is
30		indicated, comply with the applicable rules of any rules-writing agency certified by the ALSC Board of Review, Grade
31		lumber by an agency certified by the ALSC Board of Review to inspect and grade lumber under the rules indicated.
32		1. Factory mark each piece of lumber with grade stamp of grading agency.
33		 Dress lumber, S4S, unless otherwise indicated.
34	B.	Maximum Moisture Content:
35	5.	1. Boards: 15 percent.
36		 Dimension Lumber: 19 percent unless otherwise indicated.
37		3. Timber: 19 percent.
38	C.	Engineered Wood Products: Acceptable to authorities having jurisdiction and for which current model code
39	0.	research or evaluation reports exist that show compliance with building code in effect for Project.
40		1. Allowable design stresses, as published by manufacturer, are to meet or exceed those indicated.
41		Manufacturer's published values are to be determined from empirical data or by rational engineering
42		analysis and demonstrated by comprehensive testing performed by a qualified independent testing agency.
43	2.2	PRESERVATIVE TREATMENT
44	Δ	Preservative Treatment by Pressure Process: AWPA 111. Use Category 11C2
45	7	1 Preservative Chemicals: Accentable to authorities having jurisdiction and containing no arsenic or
46		chromium Do not use inorganic horon (SBX) for sill plates
47	в	Kiln-dry lumber after treatment to a maximum moisture content of 19 percent. Do not use material that is warned
48	5.	or that does not comply with requirements for untreated material
49	C	Mark lumber with treatment quality mark of an inspection agency approved by the ALSC Board of Review
50	D.	Application: Treat items indicated on Drawings, and the following:
51	υ.	1. Wood cants, nailers, curbs, equipment support bases blocking stripping and similar members in
52		connection with roofing, flashing, vapor barriers, and waterproofing
53		2. Wood sills, sleepers, blocking, furring, stripping, and similar concealed members in contact with masonry or
54		concrete
55		3. Wood framing and furring attached directly to the interior of helow-grade exterior masonry or concrete
56		walls.

May 16, 2024

2.3		FIRE-RETARDANT-TREATMENT
A	۹.	General: Where fire-retardant-treated materials are indicated, materials are to comply with requirements in this
		article, that are acceptable to authorities having jurisdiction, and with fire-test-response characteristics specified as
		determined by testing identical products per test method indicated by a qualified testing agency.
E	3.	Fire-Retardant-Treated Lumber and Plywood by Pressure Process: Products with a flame-spread index of 25 or less
		when tested according to ASTM E84, and with no evidence of significant progressive combustion when the test is
		extended an additional 20 minutes, and with the flame front not extending more than 10.5 feet beyond the
		centerline of the burners at any time during the test.
		1. Exterior Type: Treated materials are to comply with requirements specified above for fire-retardant-treated
		lumber and plywood by pressure process after being subjected to accelerated weathering according to
		ASTM D2898. Use for exterior locations and where indicated.
		2. Interior Type A. Treated indefinits are to have a moisture content of 26 percent of less when tested
		indicated
C		Kiln-dry lumber after treatment to maximum moisture content of 19 percent. Kiln-dry plywood after treatment to
		maximum moisture content of 15 percent.
C).	Identify fire-retardant-treated wood with appropriate classification marking of qualified testing agency.
E		Application: Treat
		1. Framing for raised platforms.
		2. Framing for stages.
		3. Concealed blocking.
		4. Plywood backing panels.
2.4		MISCELLANEOUS LUMBER
A	۹.	Provide miscellaneous lumber indicated and lumber for support or attachment of other construction, including the
		following:
		1. Blocking.
		2. Nailers.
		3. Roottop equipment bases and support curbs.
		4. Cdills.
F	2	J. Fulling. Dimension Lymber Items: Construction or No. 2 grade lymber of any species
). `	Concealed Boards: 15 percent maximum moisture content and any of the following species and grades:
		1. Mixed southern pine or southern pine: No. 2 grade: SPIB.
		 Eastern softwoods: No. 2 Common grade: NeLMA.
		3. Northern species; No. 2 Common grade; NLGA.
		4. Western woods; Construction or No. 2 Common grade; WCLIB or WWPA.
2.5		PLYWOOD BACKING PANELS
A	۹.	Equipment Backing Panels: Plywood, DOC PS 1, Exterior, A-C, fire-retardant treated, in thickness indicated or, if not
		indicated, not less than 3/4-inch nominal thickness.
2.6		FASTENERS
A	۹.	General: Fasteners are to be of size and type indicated and comply with requirements specified in this article for
		material and manufacture. Provide nails or screws, in sufficient length, to penetrate not less than 1-1/2 inches into
		wood substrate.
		1. where rough carpentry is exposed to weather, in ground contact, pressure-preservative treated, or in area of high relative humidity, provide factories with het dia zine coating complying with ASTM A1E2/A1E2/A
	,	Of high relative humidity, provide fasteners with hot-dip zinc coating complying with ASTM A153/A153M.
	.	based on ICC-ES AC70
C	•	Post-Installed Anchors: Fastener systems with an evaluation report acceptable to authorities having jurisdiction
		hased on ICC-FS AC01_ICC-FS AC58_ICC-FS AC193_or ICC-FS AC308 as appropriate for the substrate
PART	Г 3 - EX	
3.1	-	INSTALLATION
A	۹.	Framing Standard: Comply with AF&PA's WCD 1, "Details for Conventional Wood Frame Construction," unless
		otherwise indicated.
	3.	Framing with Engineered Wood Products: Install engineered wood products to comply with manufacturer's written
6		instructions.
-		
C	2.	Set work to required levels and lines, with members plumb, true to line, cut, and fitted. Fit rough carpentry
C		Set work to required levels and lines, with members plumb, true to line, cut, and fitted. Fit rough carpentry accurately to other construction. Locate furring, nailers, blocking, grounds, and similar supports to comply with

1	D.	Secure	ely attach rough carpentry work to substrate by anchoring and fastening as indicated, complying with the
2		follow	ing:
3		1.	Table 2304.9.1, "Fastening Schedule," in ICC's International Building Code (IBC).
4		2.	Table R602.3(1), "Fastener Schedule for Structural Members," and Table R602.3(2), "Alternate
5			Attachments," in ICC's International Residential Code for One- and Two-Family Dwellings.
6		3.	ICC-ES evaluation report for fastener.
7			END OF SECTION

Page Intentionally Left Blank

SECTION 06 20 23

2		INTERIOR FINISH CARPENTRY
3	PART 1 - 0	GENERAL
4	1.1	SUMMARY
5	Α.	Section Includes:
6		1. Interior trim and <u>plywood construction</u> .
7		2. Fire-retardant-treated wood material.
8	1.2	DEFINITIONS
9	Α.	MDF: Medium-density fiberboard.
10	В.	MDO: Plywood with a medium-density overlay on the face.
11	C.	PVC: Polyvinyl chloride.
12	1.3	ACTION SUBMITTALS
13	Α.	Product Data: For each type of process and factory-fabricated product.
14	В.	Samples: For each exposed product and for each color and texture specified.
15	PART 2 - F	PRODUCTS
16	2.1	INTERIOR TRIM AND SHEET MATERIAL
17	Α.	Hardwood Lumber Trim for Transparent Finish (Stain or Clear Finish):
18		1. Species and Grade: Birch.
19		a. <u>Field verify to match existing plywood as noted in drawings</u> .
20		2. Maximum Moisture Content: 13 percent.
21		3. Finger Jointing: Not allowed.
22		4. Gluing for Width: Allowed - Use for lumber trim wider than 6 inches.
23		5. Veneered Trim Material: Not allowed.
24		6. Plywood Core: Wood veneer.
25		a. <u>Field verify core quantity to match existing plywood as noted in drawings</u> .
26		7. Face Surface: Surfaced (smooth) and sanded.
27		8. Matching: Selected for compatible grain and color.
28	2.2	FIRE-RETARDANT-TREATED WOOD MATERIALS
29	Α.	Fire-Retardant-Treated Wood Materials: Where fire-retardant-treated materials are indicated – generally in
30		buildings with "non-combustible" construction type - use materials complying with requirements that are
31		acceptable to authorities having jurisdiction and with fire-test-response characteristics specified as determined by
32		testing identical products in accordance with test method indicated by a qualified testing agency.
33		1. Use treated materials that comply with requirements of the Architectural Woodwork Standards. Do not use
34		materials that are warped, discolored, or otherwise defective.
35		2. Use fire-retardant-treatment formulations that do not bleed through or otherwise adversely affect finishes.
36		Do not use colorants to distinguish treated materials from untreated materials.
37		3. Identify fire-retardant-treated materials with appropriate classification marking of qualified testing agency
38		in the form of removable paper label or imprint on surfaces that will be concealed from view after
39		installation.
40	В.	Fire-Retardant-Treated Lumber and Plywood: Products with a flame-spread index of 25 or less when tested in
41		accordance with ASTM E84, with no evidence of significant progressive combustion when the test is extended an
42		additional 20 minutes, and with the flame front not extending more than 10.5 feet beyond the centerline of the
43		burners at any time during the test.
44		1. Kiln-dry lumber and plywood after treatment to a maximum moisture content of 19 and 15 percent,
45		respectively.
46		2. For items indicated to receive a stained, transparent, or natural finish, use organic resin chemical
47		formulation.
48		3. Mill lumber after treatment within limits set for wood removal that do not affect listed fire-test-response
49		characteristics, using a woodworking shop certified by testing and inspecting agency.
50		4. Mill lumber before treatment, and implement procedures during treatment and drying processes that
51		prevent lumber from warping and developing discolorations from drying sticks or other causes, marring,
52		and other defects affecting appearance of treated woodwork.
53	2.3	MISCELLANEOUS MATERIALS
54	Α.	Furring, Blocking, Shims, and Nailers: Softwood or hardwood lumber, kiln-dried to less than 15 percent moisture
55		content.
56		1. Fire-Retardant Treatment: Complying with requirements; provide where indicated.
57	В.	Fasteners for Interior Finish Carpentry: Nails, screws, and other anchoring devices of type, size, material, and finish
58		required for application indicated to provide secure attachment, concealed where possible.

SPECIFICATION

	May 16	5, 2024
1	С.	Glue: Aliphatic-resin, polyurethane, or resorcinol wood glue recommended by manufacturer for general carpentry
2		use.
3	D.	Multipurpose Construction Adhesive: Formulation, complying with ASTM D3498, that is recommended for indicated
4		use by adhesive manufacturer.
5	PART 3 -	EXECUTION
6	3.1	PREPARATION
7	Α.	Clean substrates of projections and substances detrimental to application.
8	В.	Before installing interior finish carpentry, condition materials to average prevailing humidity in installation areas for
9		a minimum of 24 hours unless longer conditioning is recommended by manufacturer.
10	3.2	INSTALLATION, GENERAL
11	Α.	Install interior finish carpentry level, plumb, true, and aligned with adjacent materials.
12		1. Use concealed shims where necessary for alignment.
13		2. Scribe and cut interior finish carpentry to fit adjoining work. Refinish and seal cuts as recommended by
14		manufacturer.
15		3. Where face fastening is unavoidable, countersink fasteners, fill surface flush, and sand unless otherwise
16		indicated.
17		4. Install to tolerance of 1/8 inch in 96 inches for level and plumb. Install adjoining interior finish carpentry
18		with 1/32-inch maximum offset for flush installation and 1/16-inch maximum offset for reveal installation.
19		5. Coordinate interior finish carpentry with materials and systems in or adjacent to it. Provide cutouts for
20		mechanical and electrical items that penetrate interior finish carpentry.
21	3.3	INSTALLATION OF STANDING AND RUNNING TRIM
22	Α.	Install trim with minimum number of joints as is practical, using full-length pieces from maximum lengths of lumber
23		available.
24		 Do not use pieces less than 24 inches long, except where necessary.
25		Stagger joints in adjacent and related standing and running trim.
26		3. Cope at returns, miter at outside corners, and cope at inside corners to produce tight-fitting joints with full-
27		surface contact throughout length of joint.
28		Use scarf joints for end-to-end joints.
29		5. Plane backs of casings to provide uniform thickness across joints where necessary for alignment.
30		6. Match color and grain pattern of trim for transparent finish (stain or clear finish) across joints.
31		7. Install trim after gypsum-board joint finishing operations are completed.
32		8. Install without splitting; drill pilot holes before fastening where necessary to prevent splitting.
33		9. Fasten to prevent movement or warping.
34		10. Countersink fastener heads on exposed carpentry work and fill holes.
35	3.4	INSTALLATION OF ITEMS MADE OF PLYWOOD AS NOTED IN DRAWINGS
36	Α.	Match detailing of existing.
37	В.	Prepare for stain and clear coat as identified in Section 09 93 00 - Staining and Transparent Finishing.
38		END OF SECTION

SECTION 07 11 13 • A1817

1		SECTION 07 11 13
2		BITUMINOUS DAMPPROOFING
3	PART 1 - (GENERAL
4	1.1	SUMMARY
5	А.	Section Includes:
6		1. Cold-applied. cut-back-asphalt dampproofing.
7	В.	Related Section: 072100 - Thermal Insulation.
8	1.2	ACTION SUBMITTALS
9	Δ	Product Data: For each type of product
10	PART 2 - 1	PRODUCTS
11	21	
12	Δ	VOC Content: Products are to comply with VOC content limits of authorities having jurisdiction unless otherwise
12	η.	indicated
1/	22	
15	2.2	Manufacturers: Subject to compliance with requirements, provide products by one of the following:
16	Α.	Henry Company: a Carlicle company
17		2. W. R. Meadows, Inc.
10	Р	2. W. N. Medudwa, Inc.
10	Б.	Prush and Spray Coate: ASTM D4300/D4300M, Type I, Class 1, Ilbered.
19	1 2	bi ush anu spray coats. ASTNI 04479/04479/04, Type I, Ibbred.
20	2.5	AUXILIANT WATERIALS
21	А.	Furnish auxiliary materials recommended in writing by dampproofing manufacturer for intended use and
22		Compatible with bituminous dampproofing.
23	В.	Cut-Back-Aspnalt Primer: ASTM D41/D41M.
24	C.	Asphalt-Coated Glass Fabric: ASTM D1668/D1668/M, Type I.
25	D.	Protection Course: Fan folded, with a core of extruded-polystyrene board insulation faced on one side or both sides
26		with plastic film, nominal thickness 1/4 inch, with a compressive strength of not less than 8 psi per ASTM D1621,
27		and maximum water absorption by volume of 0.6 percent per ASTM C272/C272M.
28	PART 3 - I	
29	3.1	APPLICATION, GENERAL
30	А.	Comply with manufacturer's written instructions for dampproofing application, cure time between coats, and
31		drying time before backfilling unless otherwise indicated.
32		1. Apply dampproofing to provide continuous plane of protection.
33		2. Apply additional coats if recommended in writing by manufacturer or to achieve a smooth surface and
34		uninterrupted coverage.
35	В.	Where dampproofing footings and foundation walls, apply from finished-grade line to top of footing; extend over
36		top of footing and down a minimum of 6 inches over outside face of footing.
37		1. Extend dampproofing 12 inches onto intersecting walls and footings, but do not extend onto surfaces
38		exposed to view when Project is completed.
39		2. Install flashings and corner protection stripping at internal and external corners, changes in plane,
40		construction joints, cracks, and where indicated as "reinforced," by embedding an 8-inch-wide strip of
41		asphalt-coated glass fabric in a heavy coat of dampproofing. Dampproofing coat for embedding fabric is in
42		addition to other coats required.
43	С.	Where dampproofing exterior face of inner wythe of exterior masonry cavity walls, lap dampproofing at least 1/4
44		inch onto flashing, masonry reinforcement, veneer ties, and other items that penetrate inner wythe.
45		1. Extend dampproofing over outer face of structural members and concrete slabs that interrupt inner wythe.
46		2. Lap dampproofing at least 1/4 inch onto shelf angles supporting veneer.
47	D.	Where dampproofing interior face of above-grade, exterior concrete and masonry single-wythe masonry walls,
48		continue dampproofing through intersecting walls by keeping vertical mortar joints at intersection temporarily
49		open or by dampproofing wall before constructing intersecting walls.
50	3.2	COLD-APPLIED, CUT-BACK-ASPHALT DAMPPROOFING
51	Α.	Concrete Foundations and Parged Masonry Foundation Walls: Apply two brush or spray coats at not less than 1.25
52		gal./100 sq. ft. for first coat and 1 gal./100 sq. ft. for second coat.
53	В.	Unparged Masonry Foundation Walls: Apply primer and two brush or spray coats at not less than 1.25 gal./100 sq.
54		ft. for first coat and 1 gal./100 sq. ft. for second coat.
55	C.	Unexposed Face of Concrete Retaining Walls: Apply one brush or spray coat at not less than 1.25 gal./100 sq. ft
56	D.	Unexposed Face of Masonry Retaining Walls: Apply primer and one brush or spray coat at not less than 1.25
57		gal./100 sq. ft

58	3.3	PROTECTION COURSE INSTALLATION
59	Α.	Install protection course over completed-and-cured dampproofing. Comply with dampproofing-material and
60		protection-course manufacturers' written instructions for attaching protection course.
61	В.	Protection course is not required if drawings call for foundation insulation on exterior side of foundation wall.
62		END OF SECTION

WARNER PARK COMM AND REC CENTER CONTRACT #9502 MUNIS #17196

1 **SECTION 07 21 00** 2 THERMAL INSULATION 3 PART 1 - GENERAL 4 1.1 SUMMARY 5 Α. Section Includes: 6 Extruded polystyrene foam-plastic board insulation. 1. 7 Polyisocyanurate foam-plastic board insulation. 2. 8 В. **Related Section:** 9 See roofing membrane specification for roofing insulation. 1 10 See Section 09 29 00 SF - Gypsum Board for Sound Attenuating Blankets. 2. 11 1.2 ACTION SUBMITTALS 12 Product Data: For the following: Α. 13 Extruded polystyrene foam-plastic board insulation. 1. 14 2. Polyisocyanurate foam-plastic board insulation. 15 1.3 **INFORMATIONAL SUBMITTALS** 16 Installer's Certification: Listing type, manufacturer, and R-value of insulation installed in each element of the A. 17 building thermal envelope. 18 1. Sign, date, and post the certification in a conspicuous location on Project site. 19 В. Product test reports. 20 C. Research reports. PART 2 - PRODUCTS 21 22 PERFORMANCE REQUIREMENTS 2.1 23 Surface-Burning Characteristics: Maximum flame-spread and smoke-developed indexes less than 25 and 450 when Α. 24 tested in accordance with ASTM E84. 25 В. Fire-Resistance Ratings: Comply with ASTM E119 or UL 263; testing by a qualified testing agency. Identify products 26 with appropriate markings of applicable testing agency. 27 Indicate design designations from UL's "Fire Resistance Directory" or from listings of another qualified 1. 28 testing agency. 29 C. Fire Propagation Characteristics: Passes NFPA 285 testing as part of an approved assembly. 30 D. Labeling: Provide identification of mark indicating R-value of each piece of insulation 12 inches and wider in width. 31 Ε. Thermal-Resistance Value (R-Value) in accordance with ASTM C518: Walls (above grade): R-20 min. 32 1. 33 2. Foundation Walls: R-20 min. 34 Underslab: Not applicable. 3. EXTRUDED POLYSTYRENE FOAM-PLASTIC BOARD INSULATION 35 2.2 36 Α. Manufacturers: Subject to compliance with requirements, provide products by one of the following: 37 Atlas Polyiso Roof and Wall Insulation. a. 38 b. Carlisle Coatings & Waterproofing Inc. 39 c. DuPont de Nemours, Inc. 40 d. Johns Manville; a Berkshire Hathaway company. 41 Rmax, Inc. e. 42 f. The Dow Chemical Company. 43 Β. Extruded Polystyrene Board Insulation, Type IV: ASTM C578, Type IV, 25-psi minimum compressive strength; 44 unfaced - for foundation condition. 45 2.3 POLYISOCYANURATE FOAM-PLASTIC BOARD INSULATION 46 For exterior walls (not foundation). Α. 47 Β. Polyisocyanurate Board Insulation, Foil Faced: ASTM C1289, foil faced, Type I, Class 1 or 2. 48 1. Manufacturers: Subject to compliance with requirements, provide products by one of the following: 49 Atlas Polyiso Roof and Wall Insulation. a. 50 b. Carlisle Coatings & Waterproofing Inc. 51 c. DuPont de Nemours, Inc. d. 52 Johns Manville; a Berkshire Hathaway company. 53 Rmax, Inc. e. 54 f. The Dow Chemical Company. 55 2.4 ACCESSORIES 56 Insulation for Miscellaneous Voids: Α. 57 1. Glass-Fiber Insulation: ASTM C764, Type II, loose fill; with maximum flame-spread and smoke-developed

	IVIAY 10	, 2024
1 2		2. Polyurethane Pour-In-Place Insulation: Closed cell, with maximum flame-spread and smoke-developed indexes of 75 and 450, respectively, per ASTM E84, specifically formulated for pour-in-place applications.
3	В.	Insulation Anchors. Spindles, and Standoffs: As recommended by manufacturer.
4		a. Adhesive for Bonding Insulation: Product compatible with insulation and air and water barrier
5		materials, and with demonstrated capability to bond insulation securely to substrates without
6		damaging insulation and substrates.
7	PART 3 -	EXECUTION
8	3.1	INSTALLATION, GENERAL
9	Α.	Comply with insulation manufacturer's written instructions applicable to products and applications.
10	В.	Install insulation that is undamaged, dry, and unsoiled and that has not been left exposed to ice, rain, or snow at
11		any time.
12	С.	Install insulation with manufacturer's R-value label exposed after insulation is installed.
13	D.	Extend insulation to envelop entire area to be insulated. Fit tightly around obstructions and fill voids with insulation.
14		Remove projections that interfere with placement.
15	Ε.	Provide sizes to fit applications and selected from manufacturer's standard thicknesses, widths, and lengths. Apply
16		single layer of insulation units unless multiple layers are otherwise shown or required to make up total thickness or
17		to achieve R-value.
18	3.2	INSTALLATION OF FOUNDATION WALL INSULATION
19	А.	Butt panels together for tight fit.
20	В.	Adhesive Installation: Install with adhesive or press into tacky waterproofing or dampproofing according to
21		manufacturer's written instructions.
22	С.	If not otherwise indicated, extend insulation a minimum of 48 inches in from exterior walls.
23	3.3	INSTALLATION OF CAVITY-WALL INSULATION
24	Α.	Foam-Plastic Board Insulation: Install pads of adhesive spaced approximately 24 inches o.c. both ways on inside face
25		and as recommended by manufacturer.
26		1. Fit courses of insulation between wall ties and other obstructions, with edges butted tightly in both
27		directions, and with faces flush.
28		2. Press units firmly against inside substrates.
29		3. Supplement adhesive attachment of insulation by securing boards with two-piece wall ties designed for this
30		purpose.
31		END OF SECTION

SECTION 07 27 26 FLUID-APPLIED MEMBRANE AIR BARRIERS

2		FLUID-APPLIED MEMBRANE AIR BARRIERS
3	PART 1 -	GENERAL
4	1.1	SUMMARY
5	Α.	Section Includes:
6		1. Vapor-retarding, fluid-applied air barriers.
7	1.2	PREINSTALLATION MEETINGS
8	Α.	Preinstallation Conference: Conduct conference at Project site.
9	1.3	ACTION SUBMITTALS
10	Α.	Product Data: For each type of product.
11	В.	Shop Drawings: For air-barrier assemblies.
12		1. Include details for substrate joints and cracks, counterflashing strips, penetrations, inside and outside
13		corners, terminations, and tie-ins with adjoining construction.
14	1.4	INFORMATIONAL SUBMITTALS
15	Α.	Product certificates.
16	В.	Product test reports.
17	С.	Field quality-control reports.
18	1.5	QUALITY ASSURANCE
19	Α.	Installer Qualifications: An entity that employs installers and supervisors who are trained and approved by
20		manufacturer.
21	В.	Mockups: Build mockups to set quality standards for materials and execution.
22		1. Build integrated mockups of exterior wall assembly, 150 sq. ft., incorporating backup wall construction,
23		external cladding, window, storefront, door frame and sill, insulation, ties and other penetrations, and
24		flashing to demonstrate surface preparation, crack and joint treatment, application of air barriers, and
25		sealing of gaps, terminations, and penetrations of air-barrier assembly.
26		a. Coordinate construction of mockups to permit inspection and testing of air barrier before external
27		insulation and cladding are installed.
28		b. Include junction with roofing membrane, building corner condition, and foundation wall
29		intersection.
30	PART 2 -	PRODUCTS
31	2.1	PERFORMANCE REQUIREMENTS
32	А.	Air-Barrier Performance: Air-barrier assembly and seals with adjacent construction to be capable of performing as a
33		continuous air barrier and as a liquid-water drainage plane flashed to discharge to the exterior incidental
34		condensation or water penetration. Air-barrier assemblies to be capable of accommodating substrate movement and
35		of sealing substrate expansion and control joints, construction material changes, penetrations, and transitions at
36		perimeter conditions without deterioration and air leakage exceeding specified limits.
37	В.	Air-Barrier Assembly Air Leakage: Maximum 0.04 cfm/sq. ft. of surface area at 1.57 lbf/sq. ft., when tested in
38		accordance with ASTM E2357.
39	C.	Air Permeance: Maximum 0.004 cfm/sq. ft. of surface area at 1.57 lbf/sq. ft. pressure difference; ASTM E2178.
40	D.	Ultimate Elongation: Minimum 200 percent; ASTM D412, Die C.
41	Ε.	Adhesion to Substrate: Minimum 16 lbf/sq. in. when tested in accordance with ASTM D4541.
42	F.	Fire Propagation Characteristics: Passes NFPA 285 testing as part of an approved assembly.
43	G.	UV Resistance: Can be exposed to sunlight for 60 days in accordance with manufacturer's written instructions.
44	2.1	HIGH-BUILD AIR BARRIERS, VAPOR RETARDING
45	Α.	High-Build, Vapor-Retarding Air Barrier Modified Bituminous Type: Modified bituminous membrane with an installed
46		dry film thickness, according to manufacturer's written instructions, of 35 mils or thicker over smooth, void-free
47		substrates.
48		1. <u>Manufacturers</u> : Subject to compliance with requirements, provide products by one of the following:
49		a. Carlisle Coatings & Waterproofing Inc.
50		b. <u>Henry Company; a Carlisle company</u> .
51		c. <u>Tremco Incorporated</u> .
52		d. W. R. Meadows, Inc.
53	В.	Vapor Permeance: Maximum 0.1 perm; ASTM E96/E96M, Procedure A, Desiccant Method.
54	PART 3 -	EXECUTION
55	3.1	SURFACE PREPARATION
56	Α.	Clean, prepare, treat, fill, and seal substrate and joints and cracks in substrate in accordance with manufacturer's
57		written instructions and details. Provide clean, dust-free, and dry substrate for air-barrier application.

WARNER PARK COMM AND REC CENTER
CONTRACT #9502 MUNIS #17196

	May 16	, 2024
1 2	В.	Mask off adjoining surfaces not covered by air barrier to prevent spillage and overspray affecting other construction.
3 4	C.	Remove fins, ridges, mortar, and other projections and fill honeycomb, aggregate pockets, holes, and other voids in concrete with substrate-patching material.
5	D.	Remove excess mortar from masonry ties, shelf angles, and other obstructions.
6	E.	At changes in substrate plane, apply sealant or termination mastic beads at sharp corners and edges to form a
/	_	smooth transition from one plane to another.
8 9	F.	Bridge isolation joints, expansion joints, and discontinuous wall-to-wall, deck-to-wall, and deck-to-deck joints with air-barrier accessory material that accommodates joint movement in accordance with manufacturer's written
10		instructions and details.
11	3.2	INSTALLATION
12 13	Α.	Install materials in accordance with air-barrier manufacturer's written instructions and details to form a seal with adjacent construction and ensure continuity of air and water barrier
1/		1 Coordinate the installation of air barrier with installation of roofing membrane and base flashing to ensure
15		continuity of air barrier with roofing membrane
16		 Install transition strip on roofing membrane or base flashing so that a minimum of 3 inches of coverage is
17		achieved over each substrate.
18		3. Unless manufacturer recommends in writing against priming, apply primer to substrates at required rate
19		and allow it to dry.
20		4. Apply primer to substrates at required rate and allow it to dry. Limit priming to areas that will be covered by
21		air-barrier material on same day. Reprime areas exposed for more than 24 hours.
22	В.	Connect and seal exterior wall air-barrier material continuously to roofing-membrane air barrier, concrete below-
23		grade structures, floor-to-floor construction, exterior glazing and window systems, glazed curtain-wall systems,
24		storefront systems, exterior louvers, exterior door framing, and other construction used in exterior wall openings.
25		using accessory materials.
26	C.	Wall Openings: Prime concealed, perimeter frame surfaces of windows, curtain walls, storefronts, and doors. Apply
27		transition strip so that a minimum of 3 inches of coverage is achieved over each substrate. Maintain 3 inches of full
28		contact over firm bearing to perimeter frames.
29	D.	Repair punctures, voids, and deficient lapped seams in strips and transition strips. Slit and flatten fishmouths and
30		blisters. Patch with transition strips extending 6 inches beyond repaired areas in strip direction.
31	E.	High-Build Air Barriers: Apply continuous unbroken air-barrier material to substrates according to the following
32		thickness. Apply air-barrier material in full contact around protrusions such as masonry ties.
33		1. Vapor-Retarding, High-Build Air Barrier: Total dry film thickness as recommended in writing by
34		manufacturer to comply with performance requirements, but not less than 40 mils, applied in one or more
35		equal coats.
36	F.	Do not cover air barrier until it has been tested and inspected by testing agency.
37	G.	Correct deficiencies in or remove air barrier that does not comply with requirements; repair substrates and reapply
38		air-barrier components.
39	3.3	FIELD QUALITY CONTROL
40	Α.	Testing Agency: Engage a qualified testing agency to perform tests and inspections.
41	В.	Tests: As determined by testing agency from among the following tests:
42		1. Air-barrier dry film thickness.
43		2. Air-Leakage-Location Testing: Air-barrier assemblies will be tested for evidence of air leakage in accordance
44		with ASTM E1186, chamber pressurization or depressurization with smoke tracers.
45		3. Air-Leakage-Volume Testing: Air-barrier assemblies will be tested for air-leakage rate in accordance with
46		ASTM E783 or ASTM E2357.
47	С.	Air barriers will be considered defective if they do not pass tests and inspections.
48		1. Apply additional air-barrier material, in accordance with manufacturer's written instructions, where
49		inspection results indicate insufficient thickness.
50		2. Remove and replace deficient air-barrier components for retesting as specified above.
51	D.	Repair damage to air barriers caused by testing; follow manufacturer's written instructions.
52	E.	Prepare test and inspection reports.
53	3.4	CLEANING AND PROTECTION
54	Α.	Protect air-barrier system from damage during application and remainder of construction period, in accordance
55		with manufacturer's written instructions.
56	В.	Remove masking materials after installation.

masking materials after installatio

57

END OF SECTION

SECTION 07 41 13.16 STANDING-SEAM METAL ROOF PANELS

2		STANDING-SEAM METAL ROOF PANELS
3	PART 1 - 0	GENERAL
4	1.1	SUMMARY
5	А.	Section Includes:
6		1. Vertical-rib, seamed-joint, standing-seam metal roof panels.
7		2. Substrate board.
8		3. Vapor retarder.
9		4. Roof insulation.
10		5. Cover board.
11		6 Underlayment
12	12	PREINSTALLATION MEETINGS
13	 	Preinstallation Conference: Conduct conference at Project site
1/	12	
15	1.5	Product data
16	А. В	Shop Drawinge:
17	Б.	Situp Drawings.
10		1. Include fabilitation and installation adjusts of metal panets, details of edge conditions, panet promes,
18		corners, anchorages, attachment system, trim, trasnings, closures, and accessories; and special details.
19		2. Accessories: include details of the flashing, trim, and anchorage systems, at a scale of not less than 1-
20		1/2 inches per 12 inches.
21	1.4	
22	А.	Certificates for portable roll-forming equipment.
23	В.	Product test reports.
24	С.	Sample warranties.
25	1.5	CLOSEOUT SUBMITTALS
26	А.	Maintenance data.
27	1.6	QUALITY ASSURANCE
28	Α.	Roof Installer qualifications.
29	В.	Portable Roll-Forming Equipment Certification: UL-certified, portable roll-forming equipment capable of producing
30		metal panels warranted by manufacturer to be the same as factory-formed products. Maintain UL certification of
31		portable roll-forming equipment for duration of Work.
32	1.7	WARRANTY
33	Α.	Special Warranty: Manufacturer agrees to repair or replace components of metal panel systems that fail in
34		materials or workmanship within specified warranty period.
35		1. Warranty Period: Two years from date of Substantial Completion.
36	В.	Special Warranty on Panel Finishes: Manufacturer agrees to repair finish or replace metal panels that show
37		evidence of deterioration of factory-applied finishes within specified warranty period.
38		1. Finish Warranty Period: 20 years from date of Substantial Completion.
39	PART 2 - 1	PRODUCTS
40	2.1	PERFORMANCE REQUIREMENTS
41	А.	Structural Performance: Provide metal panel systems capable of withstanding the effects of the following loads.
42		based on testing in accordance with ASTM E1592:
43		1. Deflection Limits: For wind loads, no greater than 1/180 of the span.
44	В.	Air Infiltration: Air leakage of not more than 0.06 cfm/sg. ft. when tested in accordance with ASTM E1680 or
45	21	ASTM F283/F283M at the following test-pressure difference:
46		1 Test-Pressure Difference: 1 57 lbf/sg ft
17	C	Water penetration under Static Pressure: No water penetration when tested in accordance with ASTM F1646 or
48	с.	ASTM F331 at the following test-pressure difference:
10		1 Tat-Dressure Difference: 2.86 hf/ca ft
50	D	Watertightness: No water negotiation when tested in accordance with ASTM E2140 for hydrostatic-head
50	D.	valet lightess. No water penetration when tested in accordance with ASTM 22140 for hydrostatic-head
51	E	Teststatile. Wind Unlift Pasistance: Brouide metal reaf namel accomplies that comply with ULE 500 for wind unlift resistance
52	Е.	vinte-Opint Resistance. Provide metal root panel assemblies that comply with OL 580 for wind-upint-resistance
55		Lidos IIIuliditeu.
54 FF	-	1. Upini Kalifigi UL 90.
55	F.	Five Approvals Listing: Provide metal roof panels and component materials that comply with requirements in
50		rivi Approvais 44/1 as part of a panel rooting system and that are listed in FM's "Approval Guide" for Class 1 or
5/		noncompustible construction, as applicable. Identify materials with FM Approvals markings.
58		1. Fire/Windstorm Classification: Class 1A- 90.

	May 16	, 2024
59		2. Hail Resistance: SH.
60	G.	Thermal Movements: Allow for thermal movements from ambient and surface temperature changes.
61		1. Temperature Change: 120 deg F, ambient; 180 deg F, material surfaces.
62	Н.	Energy Performance:
63		1. Provide roof panels in accordance with one of the following when tested in accordance with CRRC-1:
64		a. Three-year, aged solar reflectance of not less than 0.55 and emissivity of not less than 0.75.
65		b. Three-year, aged Solar Reflectance Index (SRI) of not less than 64 when calculated in accordance
66		with ASTM E1980.
67	2.2	STANDING-SEAM METAL ROOF PANELS, GENERAL
68	Α.	Provide factory-formed metal roof panels designed to be installed by lapping and interconnecting raised side edges
69		of adjacent panels with joint type indicated and mechanically attaching panels to supports using concealed clips in
70		side laps. Include clips, cleats, pressure plates, and accessories required for weathertight installation.
71		1. Steel Panel Systems: Unless more stringent requirements are indicated, comply with ASTM E1514.
72	2.3	VERTICAL-RIB, SEAMED-JOINT, STANDING-SEAM METAL ROOF PANELS
73	Α.	Manufacturers: Subject to compliance with requirements, provide products by one of the following:
74		1. ATAS International, Inc.
75		2. Berridge Manufacturing Company.
76		3. CENTRIA, a Nucor Brand.
77		4. PAC-CLAD; Petersen; a Carlisle company.
78	В.	Basis-of-Design: Atas, 2 3/8" Field-Lok, 18" coverage with stiffening ribs.
79	С.	Panels: Formed with vertical ribs at panel edges; designed for sequential installation by mechanically attaching
80		panels to supports using concealed clips located under one side of panels, engaging opposite edge of adjacent
81		panels, and mechanically seaming panels together.
82		1. Structural Support: Over solid deck.
83		2. Material: Metallic-coated steel.
84		3. Seam Type: Double folded.
85		4. Panel Profile: Intermediate stiffening ribs symmetrically spaced between ribs.
86		5. Panel Coverage: 18 Inches.
87		6. Panel Height: 1.5 Inches nominal.
88		7. Clips: Two piece, floating, designed to accommodate thermal movement.
89		a. Steel Clips: 0.028-inch-nominal thickness, zinc-coated (gaivanized) or aluminum-zinc alloy-coated
90 01		Steel Sheet.
02	24	
92	2.4	Glass-Mat Gynsum Boof Substrate Board: ASTM C1177/C1177M, water-resistant gynsum board
93 94	А.	1 Manufacturers: Subject to compliance with requirements, provide products by one of the following:
95		a CertainTeed: SAINT-GOBAIN
96		h Georgia-Pacific Gynsum II C
97		c. USG Corporation.
98		2. Thickness: Type X. 5/8 inch.
99		3. Surface Finish: Factory primed.
100	В.	Fasteners: Factory-coated steel fasteners and metal or plastic plates complying with corrosion-resistance provisions
101		for fastening substrate panel to roof deck.
102	2.5	VAPOR RETARDER
103	Α.	IF PROJECT HAS A SINGLE-PLY ROOFING MEMBRANE SPECIFICATION, use vapor retarder specified for that system.
104		Otherwise, use
105	В.	Rubberized-Asphalt-Sheet Vapor Retarder, Self-Adhering: ASTM D1970/D1970M polyethylene film laminated to
106		layer of rubberized asphalt adhesive, minimum 40-mil total thickness; maximum permeance rating of 0.1 perm; cold
107		applied, with slip-resisting surface and release paper backing. Provide primer when recommended by vapor-
108		retarder manufacturer.
109	2.6	ROOF INSULATION
110	Α.	Roof insulation assembly must provide an overall average of R-35.
111	В.	Insulation over Solid Deck:
112		1. Polyisocyanurate Board Insulation: ASTM C1289, Type II, Class 1, Grade 2, felt or glass-fiber mat facer on
113		both major surfaces.
114		a. Manufacturers: Subject to compliance with requirements, available manufacturers offering
115		products that may be incorporated into the Work include, but are not limited to the following:
116		1) Atlas Polyiso Roof and Wall Insulation.

Ividy 1	0, 2024
	2) Carlisle Syntec Systems.
	3) CertainTeed; SAINT-GOBAIN.
	4) GAF.
	5) Johns Manville; a Berkshire Hathaway company.
	6) Rmax, A Business Unit of Sika Corporation.
	b. Compressive Strength: 25 psi.
	c. Size: 48 by 96 inches.
	d. Thickness:
	1) Base Layer: 1-1/2 inches.
	2) Upper Layer: As needed to meet required R-value.
2.7	COVER BOARD
Α.	Oriented Strand Board or Plywood – <u>as recommended by root panel manufacturer</u> : DOC PS 2, Exposure 1, 3/4 inch
	thick.
2.8	UNDERLAYMENT Self Adhesing - Mich Tennessterne Hunderleumente Describe - self adhesing - seld anglisch - shart - underleument
А.	Self-Adhering, High-Temperature Underlayment: Provide self-adhering, cold-applied, sheet underlayment, a
	minimum of 30 mills thick, consisting of slip-resistant, polyethylene-film top surface laminated to a layer of butyl or
	SBS-modified asphalt adnesive, with release-paper backing. Provide primer when recommended by underlayment
	manufacturer.
	Inermal Stability: Stable after testing at 240 deg F; ASTM D1970/D1970M.
	2. Low-reinperature riexibility. Passes after testing at minus 20 deg F, ASTW D1970/D1970W.
	5. Manuacturers. Subject to compliance with requirements, provide products by one of the following.
	 ATAS international, inc. b Carlisle WIP Products: a brand of Carlisle Construction Materials
	c Henry Company: a Carlisle company
	d Owens Corning
В.	Slip Sheet: Manufacturer's recommended slip sheet, of type required for application.
2.9	PANEL MATERIALS
 A.	Metallic-Coated Steel Sheet: Zinc-coated (galvanized) steel sheet complying with minimum ASTM A653/A653M.
	G90 coating designation, or aluminum-zinc alloy-coated steel sheet complying with minimum ASTM A792/A792M,
	Class AZ50 coating designation; structural quality. Prepainted by the coil-coating process to comply with
	ASTM A755/A755M.
	1. Nominal Thickness: 0.028 inch.
	2. Surface: Smooth, flat finish.
2.10	MISCELLANEOUS MATERIALS
Α.	Miscellaneous Metal Subframing and Furring: ASTM C645; cold-formed, metallic-coated steel sheet, minimum
	ASTM A653/A653M, G90 hot-dip galvanized coating designation or ASTM A792/A792M, Class AZ50 coating
	designation. Provide manufacturer's standard sections as required for support and alignment of metal panel
	system.
В.	Panel Accessories: Provide components required for a complete, weathertight panel system including trim, copings,
	fasciae, mullions, sills, corner units, clips, flashings, sealants, gaskets, fillers, closure strips, and similar items. Match
	material and finish of metal panels unless otherwise indicated.
	 Closures: Provide closures at eaves and ridges, fabricated of same metal as metal panels.
	2. Backing Plates: Provide metal backing plates at panel end splices, fabricated from material recommended
	by manufacturer.
	3. Closure Strips: Closed-cell, expanded, cellular, rubber or crosslinked, polyolefin-foam or closed-cell
	laminated polyethylene; minimum 1-inch-thick, flexible closure strips; cut or premolded to match metal
	panel profile. Provide closure strips where indicated or necessary to ensure weathertight construction.
C.	Flashing and Trim: Provide flashing and trim formed from same material as metal panels as required to seal against
	weather and to provide finished appearance. Locations include, but are not limited to, eaves, rakes, corners, bases,
	framed openings, ridges, fasciae, and fillers. Finish flashing and trim with same finish system as adjacent meta
	panels. Cottage and Decompositor Constinue 07 C2 C2 C2 Chart Mattel Flocking and Trice
D.	Gutters and Downspouts: See Section 07.62.00 - Sheet Metal Flashing and Trim.
E. r	Partier Fasteriers. Self-tapping screws designed to withstand design 1080s.
F.	ranei sealants: Provide sealant type recommended by manufacturer that are compatible with panel materials, are
	nonstanting, and up not uping patient initial. 1 Sealant Tane: Pressure-sensitive, 100 nercent solids, gray polyisphytylang, compound solant tang with
	release-namer backing Provide permanently elastic nonsag nontovic nonstaining tane 1/2 inch wide and
	1/8 inch thick

Iviay	10, 2024
	2. Joint Sealant: ASTM C920; elastomeric polyurethane or silicone sealant; of type, grade, class, and use classifications required to seal joints in metal panels and remain weathertight; and as recommended in writing by metal panel manufacturer.
	2 Putul Pubber Paced Solvent Poleace Scalant: ASTM C1211
2 1 1	FARRICATION
 A.	Fabricate and finish metal panels and accessories at the factory, by manufacturer's standard procedures and
	processes, as necessary to fulfill indicated performance requirements demonstrated by laboratory testing. Comply
	with indicated profiles and with dimensional and structural requirements.
В.	On-site Fabrication: Subject to compliance with requirements of this Section, metal panels may be fabricated on-
	site using UL-certified, portable roll-forming equipment if panels are of same profile and warranted by
	manufacturer to be equal to factory-formed panels. Fabricate in accordance with equipment manufacturer's written instructions and to comply with details shown
C.	Provide panel profile, including major ribs and intermediate stiffening ribs, if any, for full length of panel.
D.	Fabricate metal panel joints with factory-installed captive gaskets or separator strips that provide a weathertight
	seal and prevent metal-to-metal contact, and that minimize noise from movements.
E.	Sheet Metal Flashing and Trim: Fabricate flashing and trim to comply with manufacturer's recommendations that
	apply to design, dimensions, metal, and other characteristics of item indicated.
	1. Form exposed sheet metal accessories that are without excessive oil canning, buckling, and tool marks and
	that are true to line and levels indicated, with exposed edges folded back to form hems.
	2. Seams for Aluminum: Fabricate nonmoving seams with flat-lock seams. Form seams and seal with epoxy
	seam sealer. Rivet joints for additional strength.
	3. Sealed Joints: Form nonexpansion, but movable, joints in metal to accommodate sealant and to comply
	with manufacturer's recommendations.
	4. Conceal fasteners and expansion provisions where possible. Exposed fasteners are not permitted on faces
	of accessories exposed to view.
	5. Fabilitate clears and attachment devices from same material as accessoly being anchored of from compatible popearrosive metal recommanded in writing by metal papel manufacturer.
	a Size: As recommended by metal nanel manufacturer for application, but not less than thickness of
	metal being secured.
2.12	FINISHES
Α.	Protect mechanical and painted finishes on exposed surfaces from damage by applying a strippable, temporary
	protective covering before shipping.
В.	Appearance of Finished Work: Variations in appearance of abutting or adjacent pieces are acceptable if they are
	within one-half of the range of approved Samples. Noticeable variations in same piece are unacceptable. Variations
	in appearance of other components are acceptable if they are within the range of approved Samples and are
	assembled or installed to minimize contrast.
C.	Steel and Aluminum Panels and Accessories:
	1. Three-Coat Fluoropolymer: Fluoropolymer finish containing not less than 70 percent PVDF resin by weight
	In both color coat and clear topcoat. Prepare, pretreat, and apply coating to exposed metal surfaces to
	Comply with coating and resin manufacturers written instructions.
	2. Concealed Finish: Apply pretreatment and manufacturer's standard white or light-colored acrylic or nelvester backer finish consisting of prime cost and wash cost with a minimum total dry film thickness of
	D 5 mil
PART	
3.1	PREPARATION
о А.	Miscellaneous Supports: Install subframing, furring, and other miscellaneous panel support members and
	anchorages in accordance with ASTM C754 and metal panel manufacturer's written installation instructions.
3.2	INSTALLATION OF SUBSTRATE BOARD
A.	Install substrate board with long joints in continuous straight lines, with end joints staggered not less than 24 inches
	in adjacent rows.
	1. At steel roof decks, install substrate board at right angle to flutes of deck.
	a. Locate end joints over crests of steel roof deck.
	2. Tightly butt substrate boards together.
	3. Cut substrate board to fit tight around penetrations and projections, and to fit tight to intersecting sloping
	roof decks.
	4. Fasten substrate board in accordance with roofing system manufacturers' written installation instructions.

SPECIFICATION

	May 16, 2024		
231	3.3	INSTALLATION OF VAPOR RETARDER	
232	Α.	Loosely lay vapor retarder in a single layer over area to receive vapor retarder, side and end lapping each sheet a	
233		minimum of 2 and 6 inches, respectively.	
234		1. Extend vertically up parapet walls and projections to a minimum height equal to height of insulation and	
235		cover board.	
236		2 Continuously seal side and end lans with tane	
237	34	INSTALLATION OF ROOF INSULATION	
237	Δ	General: Install insulation concurrently with metal nanel installation in thickness indicated to cover entire surface	
230	7	in accordance with manufacturer's written installation instructions	
235		1 Set vanor-retarder-faced units with vanor retarder toward warm side of construction unless otherwise	
240		indicated. Do not obstruct ventilation spaces except for firestonning	
241		2 Tane joints and runtures in vanor retarder and seal each continuous area of insulation to the surrounding	
242		construction to ensure airtight installation	
243	2 5		
244	3.5	Install cover board over insulation in accordance with manufacturer's written installation instructions. Install with	
245	А.	long joints in continuous straight lines with and joints staggered between rows. Offset joints of insulation helpwice	
240		minimum of 6 inchos in each direction	
247	26		
248	5.0	INSTALLATION OF UNDERLATIVIENT	
249	А.	self-Adhering Sheet Underlayment. Apply primer in required by manufacturer. Comply with temperature	
250		restrictions of underlayment manufacturer for installation. Apply at locations indicated below, whilkle free, in	
251		shingle rashion to shed water, and with end laps of not less than 6 inches staggered 24 inches between courses.	
252		Overlap side edges not less than 3-1/2 inches. Extend underlayment into gutter trough. Roll laps with roller. Cover	
253		underlayment within 14 days.	
254	_	1. Apply over the entire root surface.	
255	В.	Slip Sheet: Apply slip sheet over underlayment before installing metal root panels.	
256	C.	Flashings: Install flashings to cover underlayment to comply with requirements specified in Section 07.62.00 "Sheet	
257		Metal Flashing and Trim."	
258	3.7	INSTALLATION OF STANDING-SEAM METAL ROOF PANELS	
259	А.	Install metal panels in accordance with manufacturer's written installation instructions and approved Shop	
260		Drawings in orientation, sizes, and locations indicated. Anchor metal panels and other components of the Work	
261		securely in place, with provisions for thermal and structural movement.	
262		1. Shim or otherwise plumb substrates receiving metal panels.	
263		2. Flash and seal metal panels at perimeter of all openings. Fasten with self-tapping screws. Do not begin	
264		installation until air- or water-resistive barriers and flashings that will be concealed by metal panels are	
265		installed.	
266		3. Install screw fasteners in predrilled holes.	
267		4. Locate and space fastenings in uniform vertical and horizontal alignment.	
268		Install flashing and trim as metal panel work proceeds.	
269		6. Locate panel splices over, but not attached to, structural supports. Stagger panel splices and end laps to	
270		avoid a four-panel lap splice condition.	
271		7. Align bottoms of metal panels and fasten with blind rivets, bolts, or self-tapping screws. Fasten flashings	
272		and trim around openings and similar elements with self-tapping screws.	
273		Provide weathertight escutcheons for pipe- and conduit-penetrating panels.	
274	В.	Fasteners:	
275		1. Steel Panels: Use stainless steel fasteners for surfaces exposed to the exterior; use galvanized-steel	
276		fasteners for surfaces exposed to the interior.	
277	С.	Anchor Clips: Anchor metal roof panels and other components of the Work securely in place, using manufacturer's	
278		approved fasteners in accordance with manufacturers' written installation instructions.	
279	D.	Metal Protection: Where dissimilar metals contact each other or corrosive substrates, protect against galvanic	
280		action as recommended in writing by metal panel manufacturer.	
281	E.	Standing-Seam Metal Roof Panel Installation: Fasten metal roof panels to supports with concealed clips at each	
282		standing-seam joint at location, spacing, and with fasteners recommended in writing by manufacturer.	
283		1. Install clips to supports with self-tapping fasteners.	
284		2. Install pressure plates at locations indicated in manufacturer's written installation instructions.	
285		3. Snap Joint: Nest standing seams and fasten together by interlocking and completely engaging factory-	
286		applied sealant.	
287		4. Seamed Joint: Crimp standing seams with manufacturer-approved, motorized seamer tool so clip. metal	
288		roof panel, and factory-applied sealant are completely engaged.	

	May 16	, 2024	
89		5.	Watertight Installation:
90			a. Apply a continuous ribbon of sealant or tape to seal joints of metal panels, using sealant or tape as
91			recommended in writing by manufacturer as needed to make panels watertight.
92			b. Provide sealant or tape between panels and protruding equipment, vents, and accessories.
93 94			 At panel splices, nest panels with minimum 6-inch end lap, sealed with sealant and fastened together by interlocking clamping plates.
95	F.	Access	ory Installation: Install accessories with positive anchorage to building and weathertight mounting and
96		provid	e for thermal expansion. Coordinate installation with flashings and other components.
97		1.	Install components required for a complete metal panel system including trim, copings, corners, seam
.98 99			covers, flashings, sealants, gaskets, fillers, closure strips, and similar items. Provide types indicated by metal
200	G	Flashin	and Trim: Comply with performance requirements and manufacturer's written installation instructions
100	0.	Provid	e concealed fasteners where possible and set units true to line and level as indicated. Install work with lans
202		inints	and seams that will be permanently watertight and weather resistant
102		1	Install exposed flashing and trim that is without buckling and tool marks and that is true to line and levels
04			indicated with exposed edges folded back to form hems. Install sheet metal flashing and trim to fit
05			substrates and achieve waterproof and weather-resistant performance.
06		2.	Expansion Provisions: Provide for thermal expansion of exposed flashing and trim. Space movement joints
07			at a maximum of 10 ft, with no joints allowed within 24 inches of corner or intersection. Where lapped
808			expansion provisions cannot be used or would not be sufficiently weather resistant and waterproof, form
09			expansion joints of intermeshing hooked flanges, not less than 1 inch deep, filled with mastic sealant
10			(concealed within ioints).
11	Н.	Pipe ar	nd Conduit Penetrations: Fasten and seal to metal roof panels as recommended by manufacturer.
12	3.8	CLEAN	ING AND PROTECTION
13	Α.	Remov	ve temporary protective coverings and strippable films, if any, as metal panels are installed, unless otherwise
14		indicat	ed in manufacturer's written installation instructions. On completion of metal panel installation, clean
15		finishe	d surfaces as recommended by metal panel manufacturer. Maintain in a clean condition during construction.
16	В.	Replac	e metal panels that have been damaged or have deteriorated beyond successful repair by finish touchup or
17		similar	minor repair procedures.
10			

С

SECTION 07 42 13.13 FORMED METAL WALL PANELS

3	PART 1 - (GENERAL
4	1.1	SUMMARY
5	Α.	Section Includes:
6		 Concealed-fastener, lap-seam metal wall panels.
7	1.2	PREINSTALLATION MEETINGS
8	А.	Preinstallation Conference: Conduct conference at Project site.
9	1.3	ACTION SUBMITTALS
10	Α.	Product Data: For each type of product.
11	В.	Shop Drawings: Include fabrication and installation layouts of metal panels; details of edge conditions, joints, panel
12		profiles, corners, anchorages, attachment system, trim, flashings, closures, and accessories; and special details.
13	C.	Samples: For each type of metal panel indicated.
14	1.4	INFORMATIONAL SUBMITTALS
15	А.	Product test reports.
16	В.	Warranties: Samples of special warranties.
17	1.5	CLOSEOUT SUBMITTALS
18	Α.	Maintenance data.
19	1.6	OUALITY ASSURANCE
20		Installer Qualifications: An entity that employs installers and supervisors who are trained and approved by
21		manufacturer.
22	B.	UI-Certified. Portable Roll-Forming Equipment: UI-certified, portable roll-forming equipment capable of producing
23	5.	metal nanels warranted by manufacturer to be the same as factory-formed products. Maintain III certification of
24		nortable roll-forming equipment for duration of work
25	17	WARRANTY
26	Δ	Special Warranty: Manufacturer's standard form in which manufacturer agrees to repair or replace components of
27	<i>,</i>	metal nanel systems that fail in materials or workmanshin within snerified warranty period
27		1 Warranty Period: Two years from date of Substantial Completion
20	в	Special Warranty on Panel Einishes: Manufacturer's standard form in which manufacturer agrees to renair finish or
20	D.	replace metal papels that show evidence of deterioration of factory-applied finishes within specified warranty
21		ported
27 27		periou. 1 Einich Warranty Poriod: 20 years from date of Substantial Completion
32 22		
27	2 1	
34 25	2.1	Structural Deformance: Dravide metal nanel systems canable of withstanding the offects of the following loads
25	А.	based on testing according to ACTM E1502:
30 27		Mind Loads: As indicated on Drawings
37 20		Wind Loads. As indicated on Drawings. Deflection Limits: For wind loads, no greater than 1/190 of the snan
20	р	2. Deflection Limits. For white loads, no greater than 1/100 of the span. Air infiltration: Air loakage of not more than 0.06 cfm/cg. ft. when tested according to ASTM E282 at the following
40 29	Б.	An initiation. An reakage of not more than 0.06 cm/sq. it. when tested according to Astron 2265 at the following
40 41		test-pressure unrerence.
41 42	C	1. Test-Pressure Difference: 1.57 ID/Sq. It
42 12	C.	following test proceure difference:
45		1 Toto Prosoure Difference: 2.86 lbf/cm.ft
44 45	D	1. Test-Pressure Dirierence. 2.80 iD/34, 10.
45 46	D.	hudding energing of joints exerctions of components foilure of joint coalents foilure of connections and other
40		bucking, opening of joints, overstressing of components, failure of joint searants, failure of connections, and other
47		detrimental effects. Base calculations on surface temperatures of materials due to both solar heat gain and
48		nightime-ský neat loss.
49 50	-	1. Temperature Change (Range): 120 deg F, ambient; 180 deg F, material surfaces.
5U F1	E.	FITE-RESISTANCE Ratings: Comply with ASTIVIELLY; testing by a qualified testing agency. Identify products with
51		appropriate markings of applicable testing agency.
52		1. Indicate design designations from UL's "Fire Resistance Directory" or from the listings of another qualified
53		testing agency.
54	2.2	CONCEALED-FASTENER, LAP-SEAM METAL WALL PANELS
55	Α.	Provide factory-formed metal panels designed to be field assembled by lapping and interconnecting side edges of
56		adjacent panels and mechanically attaching through panel to supports using concealed fasteners and factory-
57		applied sealant in side laps. Include accessories required for weathertight installation.

Iviay 16	5, 2024
В.	Flush-Profile, Concealed-Fastener Metal Wall Panels MP-1-2-3: Formed with vertical panel edges and a flat pan
	between panel edges; with flush joint between panels.
	1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
	a. ATAS International, Inc.
	b. Berridge Manufacturing Company.
	c. CENTRIA, a Nucor Brand.
	d. Dri-Design.
	e. PAC-CLAD; Petersen; a Carlisle company.
	2. <u>Basis-of-Design</u> : Dri-Design, En-V, A80 Series.
	3. Aluminum Sheet: Coil-coated sheet, ASTM B209, alloy as standard with manufacturer, with temper as
	required to suit forming operations and structural performance required.
	a. Thickness: 0.040 inch.
	b. Surface: Smooth, flat finish.
	c. Exterior Finish: Three-coat fluoropolymer.
	d. Color: As indicated by manufacturer's designations in drawings.
	4. Panel Coverage: See drawings.
	5. Panel Height: 1.25 inch.
2.3	MISCELLANEOUS MATERIALS
Α.	Miscellaneous Metal Subframing and Furring: ASTM C645. cold-formed. metallic-coated steel sheet.
	ASTM A653/A653M, G90 hot-dip galvanized coating designation or ASTM A792/A792M. Class AZ50 aluminum-zinc-
	alloy coating designation unless otherwise indicated. Provide manufacturer's standard sections as required for
	support and alignment of metal panel system.
в	Panel Accessories: Provide components required for a complete weathertight panel system including trim conings
υ.	fasciae multions sills corner units clins flashings sealants gaskets fillers closure strins and similar items Match
	material and finish of metal nanals unless otherwise indicated
	1 Closures: Provide closures at eaves and rakes fabricated of same metal as metal nanels
	 Backing Plates: Provide metal backing plates at papel and splices, fabricated from material recommended
	2. Dacking rates, rowide metal backing plates at panel end splices, labitated nom material recommended
	3 Closure Strips: Closed-cell expanded cellular rubber or crosslipked polyolefin-foam or closed-cell
	5. Closure strips. Closed-cell, expanded, cellular, rubber of clossifiked, polydellification of closed-cell
	annihilateu polyetinyene, minimum 1-inter-trike, nexule closule surps, cut of preniotee to match metal
C	Flashing and Trim. Provide flashing and trim formed from same material as motal papels as required to scal against
C.	Flashing and Thin. Provide flashing and thin formed from same include, but are not limited to bases, dring sills, isome
	weather and to provide ministed appearance. Locations include, but are not minited to, bases, drips, sins, jamps,
	corners, endwalls, framed openings, rakes, fasciae, parapet caps, somits, reveals, and fillers. Finish fashing and trim
D	With some minish system as adjatent metal panels.
D.	Panel Fasteners: Self-tapping screws designed to withstand design loads. Provide exposed fasteners with neads
	matching color of metal panels by means of plastic caps of factory-applied coating. Provide EPDIVI of PVC sealing
-	wasners for exposed fasteners.
E.	Panel Sealants: Provide sealant type recommended by manufacturer that are compatible with panel materials, are
	nonstaining, and do not damage panel finish.
	1. Sealant Tape: Pressure-sensitive, 100 percent solids, gray polyisobutylene compound sealant tape with
	release-paper backing; 1/2 inch wide and 1/8 inch thick.
	2. Joint Sealant: ASTM C920; as recommended in writing by metal panel manufacturer.
	Butyl-Rubber-Based, Solvent-Release Sealant: ASTM C1311.
2.4	FABRICATION
Α.	Fabricate and finish metal panels and accessories at the factory, by manufacturer's standard procedures and
	processes, as necessary to fulfill indicated performance requirements demonstrated by laboratory testing. Comply
	with indicated profiles and with dimensional and structural requirements.
В.	On-Site Fabrication: Subject to compliance with requirements of this Section, metal panels may be fabricated on-
	site using UL-certified, portable roll-forming equipment if panels are of same profile and warranted by
	manufacturer to be equal to factory-formed panels. Fabricate according to equipment manufacturer's written
	instructions and to comply with details shown.
С.	Provide panel profile, including major ribs and intermediate stiffening ribs, if any, for full length of panel.
D.	Fabricate metal panel joints with factory-installed captive gaskets or separator strips that provide a weathertight
	seal and prevent metal-to-metal contact, and that minimize noise from movements.
E.	Sheet Metal Flashing and Trim: Fabricate flashing and trim to comply with manufacturer's recommendations and
	recommendations in SMACNA's "Architectural Sheet Metal Manual" that apply to design, dimensions, metal, and
	other characteristics of item indicated

	IVIAY 10, 2024		
1	2.5	FINISHES	
2	Α.	Panels and Accessories:	
3		1. Three-Coat Fluoropolymer: AAMA 621. Fluoropolymer finish containing not less than 70 percent	
4		polyvinylidene fluoride (PVDF) resin by weight in both color coat and clear topcoat.	
5		2. Concealed Finish: White or light-colored acrylic or polyester backer finish.	
6	PART 3 -	EXECUTION	
7	3.1	PREPARATION	
8	Α.	Miscellaneous Supports: Install subframing, furring, and other miscellaneous panel support members and	
9		anchorages according to ASTM C754 and metal panel manufacturer's written recommendations.	
0	3.2	INSTALLATION	
1	Α.	Lap-Seam Metal Panels: Fasten metal panels to supports with fasteners at each lapped joint at location and spacing	
2		recommended by manufacturer.	
3		1. Lap ribbed or fluted sheets one full rib. Apply panels and associated items true to line for neat and	
4		weathertight enclosure.	
5		2. Provide metal-backed washers under heads of exposed fasteners bearing on weather side of metal panels.	
6		3. Locate and space exposed fasteners in uniform vertical and horizontal alignment. Use proper tools to obtain	
/		controlled uniform compression for positive seal without rupture of washer.	
8		4. Install screw fasteners with power tools having controlled torque adjusted to compress washer tightly	
9		without damage to washer, screw threads, or panels. Install screws in predrilled noies.	
1	P	5. Flash and seal panels with weather closures at perimeter of all openings.	
1	В.	watertight installation:	
2		1. Apply a continuous ribbon of sealant of tape to seal lapped joints of metal panels, using sealant of tape as	
3		recommend by manufacturer on side laps of nesting-type panels; and elsewhere as needed to make panels	
4 r		Water light.	
5 C		 Provide sediant of tape between panels and protifuting equipment, vents, and accessories. At papel splices, past papels with minimum 6 inch and lan, scaled with scalapt and fastened together by 	
0		5. At panel splices, nest panels with minimum o-inch end lap, sealed with sealaht and fastened together by	
/ 0	C	Accessory Installation: Install accessories with positive anchorage to building and weathertight mounting, and	
٥ ۵	C.	provide for thermal expansion. Coordinate installation with flashings and other components	
0	П	Flashing and Trim: Comply with performance requirements, manufacturer's written installation instructions, and	
1	D.	SMACNA's "Architectural Sheet Metal Manual " Provide concealed fasteners where possible and set units true to	
2		line and level as indicated Install work with lans joints and seams that are normanently watertight	
2 2	33	cife and level as indicated. Install work with laps, joints, and seams that are permanently watertight.	
۵ ۵	Δ	Remove temporary protective coverings and strippable films if any as metal papels are installed unless otherwise	
5	/ \.	indicated in manufacturer's written installation instructions. On completion of metal panel installation clear	
6		finished surfaces as recommended by metal panel manufacturer. Maintain in a clean condition during construction	
-			

Page Intentionally Left Blank

SECTION 07 42 93

1		SECTION 07 42 93
2		SOFFIT PANELS
3	PART 1 -	GENERAL
4	1.1	SUMMARY
5	Α.	Section Includes:
6		1. Metal soffit panels.
7	1.2	PREINSTALLATION MEETINGS
8	Α.	Preinstallation Conference: Conduct conference at Project site.
9	1.3	ACTION SUBMITTALS
10	Α.	Product Data: For each type of product.
11	В.	Shop Drawings: Include fabrication and installation layouts of metal panels; details of edge conditions, joints, panel
12		profiles, corners, anchorages, attachment system, trim, flashings, closures, and accessories; and special details.
13	C.	Samples: For each type of metal panel indicated.
14	1.4	INFORMATIONAL SUBMITTALS
15	Α.	Product test reports.
16	В.	Warranties: Samples of special warranties.
17	1.5	CLOSEOUT SUBMITTALS
18	А.	Maintenance data.
19	1.6	WARRANTY
20	Α.	Special Warranty: Manufacturer's standard form in which manufacturer agrees to repair or replace components of
21		metal panel systems that fail in materials or workmanship within specified warranty period.
22		1. Warranty Period: Two years from date of Substantial Completion.
23	В.	Special Warranty on Panel Finishes: Manufacturer's standard form in which manufacturer agrees to repair finish or
24		replace metal panels that show evidence of deterioration of factory-applied finishes within specified warranty
25		period.
26		1. Finish Warranty Period: 20 years from date of Substantial Completion.
27	PART 2 -	PRODUCTS
28	2.1	PERFORMANCE REQUIREMENTS
29	А.	Structural Performance: Provide metal panel systems capable of withstanding the effects of the following loads.
30		based on testing according to ASTM E1592:
31		1. Deflection Limits: For wind loads, no greater than 1/180 of the span.
32	В.	Air Infiltration: Air leakage of not more than 0.06 cfm/sq. ft. when tested according to ASTM E283 at the following
33		test-pressure difference:
34		1. Test-Pressure Difference: 1.57 lbf/sg. ft.
35	C.	Water Penetration under Static Pressure: No water penetration when tested according to ASTM E331 at the
36	-	following test-pressure difference:
37		1. Test-Pressure Difference: 2.86 lbf/sg. ft
38	D.	Thermal Movements: Allow for thermal movements from ambient and surface temperature changes by preventing
39		buckling, opening of joints, overstressing of components, failure of joint sealants, failure of connections, and other
40		detrimental effects. Base calculations on surface temperatures of materials due to both solar heat gain and
41		nighttime-sky heat loss.
42		1. Temperature Change (Range): 120 deg F. ambient: 180 deg F. material surfaces.
43	2.2	METAL SOFFIT PANELS
44	Α.	Provide metal soffit panels designed to be installed by lapping and interconnecting side edges of adjacent panels
45		and mechanically attaching through panel to supports using concealed fasteners in side laps. Include accessories
46		required for weathertight installation.
47	В.	Flush-Profile Metal Soffit Panels: Solid panels formed with vertical panel edges and a flat pan between panel edges:
48		with flush joint between panels.
49		1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
50		a. ATAS International. Inc.
51		b. Berridge Manufacturing Company.
52		c. CENTRIA. a Nucor Brand.
53		d. PAC-CLAD: Petersen: a Carlisle company.
54		2 Basis-of-Design: Pac-Clad Flush Soffit
55		 Metallic-Coated Steel Sheet: Zinc-coated (galvanized) steel sheet complying with ΔSTM Δ653/Δ653M_GQ0
56		coating designation, or aluminum-zinc allov-coated steel sheet complying with ASTM A797/A797M
57		Class A750 coating designation: structural guality. Prenainted by the coil-coating process to comply with
58		ASTM A755/A755M
50		

	a. Nominal Thickness: 0.022 inch.
	c. Color: As selected by Architect from manufacturer's full range.
	4. Panel Coverage: 12 inches.
	5. Panel Height: 1.0 inch.
2.3	MISCELLANEOUS MATERIALS
Α.	Miscellaneous Metal Subframing and Furring: ASTM C645, cold-formed, metallic-coated steel sheet,
	ASTM A653/A653M, G90 hot-dip galvanized coating designation or ASTM A792/A792M, Class AZ50 aluminum-zinc-
	alloy coating designation unless otherwise indicated. Provide manufacturer's standard sections as required for
_	support and alignment of metal panel system.
В.	Panel Accessories: Provide components required for a complete, weathertight panel system including trim, clips,
	otherwise indicated
	1 Closure Strins: Closed-cell expanded cellular rubber or crosslinked polyolefin-foam or closed-cell
	laminated polyethylene: minimum 1-inch-thick, flexible closure strips: cut or premolded to match metal
	panel profile. Provide closure strips where indicated or necessary to ensure weathertight construction.
C.	Flashing and Trim: Provide flashing and trim formed from same material as metal panels as required to seal against
	weather and to provide finished appearance. Finish flashing and trim with same finish system as adjacent metal
	panels.
D.	Panel Fasteners: Self-tapping screws designed to withstand design loads. Provide exposed fasteners with heads
	matching color of metal panels by means of plastic caps or factory-applied coating. Provide EPDM or PVC sealing
	washers for exposed fasteners.
Ε.	Panel Sealants: Provide sealant types recommended by manufacturer that are compatible with panel materials, are
	nonstaining, and do not damage panel finish.
	1. Sealant Tape: Pressure-sensitive, 100 percent solids, gray polyisobutylene compound sealant tape with release paper backing, 1/8 inch thick
	leiedse-paper backing; 1/8 inch thick.
	 Joint Sealant: ASTM C520, as recommended in writing by metal parter manufacturer. Butyl-Rubher-Based Solvent-Release Sealant: ASTM C1311
2.4	FABRICATION
Α.	Fabricate and finish metal panels and accessories at the factory, by manufacturer's standard procedures and
	processes, as necessary to fulfill indicated performance requirements demonstrated by laboratory testing. Comply
	with indicated profiles and with dimensional and structural requirements.
В.	On-Site Fabrication: Subject to compliance with requirements of this Section, metal panels may be fabricated on-
	site using UL-certified, portable roll-forming equipment if panels are of same profile and warranted by
	manufacturer to be equal to factory-formed panels. Fabricate according to equipment manufacturer's written
c	Instructions and to comply with details shown.
С.	Provide panel profile, including major ribs and intermediate stiffening ribs, it any, for full length of panel.
D.	Fabricate metal panel joints with factory-installed captive gaskets or separator strips that provide a weathertight
F	Seal and prevent metal-to-metal contact, and that minimize holds from movements.
L.	recommendations in SMACNA's "Architectural Sheet Metal Manual" that apply to design dimensions metal and
	other characteristics of item indicated.
2.5	FINISHES
А.	Panels and Accessories:
	1. Two-Coat Fluoropolymer: AAMA 621. Fluoropolymer finish containing not less than 70 percent
	polyvinylidene fluoride (PVDF) resin by weight in color coat. Prepare, pretreat, and apply coating to exposed
	metal surfaces to comply with coating and resin manufacturers' written instructions.
	2. Concealed Finish: White or light-colored acrylic or polyester backer finish.
PART 3 -	EXECUTION
3.1	PREPARATION
А.	iviscellaneous supports: Install subtraming, turring, and other miscellaneous panel support members and
	anciorages according to ASTRI C754 and metal panel manufacturer's written recommendations.
3.2	
А .	Metal Soffit Panels: Fasten metal panels to supports with fasteners at each lapped joint at location and spacing
71.	recommended by manufacturer.
	1. Apply panels and associated items true to line for neat and weathertight enclosure.
	2. Provide metal-backed washers under heads of exposed fasteners bearing on weather side of metal panels.

	IVIAY 10	, 2024
1		3. Locate and space exposed fasteners in uniform vertical and horizontal alignment. Use proper tools to obtain
2		controlled uniform compression for positive seal without rupture of washer.
3		4. Install screw fasteners with power tools having controlled torque adjusted to compress washer tightly
4		without damage to washer, screw threads, or panels. Install screws in predrilled holes.
5	В.	Watertight Installation:
6		1. Apply a continuous ribbon of sealant or tape to seal lapped joints of metal panels, using sealant or tape as
7		recommend by manufacturer on side laps of nesting-type panels and elsewhere as needed to make panels
8		watertight.
9		2. Provide sealant or tape between panels and protruding equipment, vents, and accessories.
10		3. At panel splices, nest panels with minimum 6-inch end lap, sealed with sealant and fastened together by
11		interlocking clamping plates.
12	С.	Accessory Installation: Install accessories with positive anchorage to building and weathertight mounting, and
13		provide for thermal expansion. Coordinate installation with flashings and other components.
14	D.	Flashing and Trim: Comply with performance requirements, manufacturer's written installation instructions, and
15		SMACNA's "Architectural Sheet Metal Manual." Provide concealed fasteners where possible, and set units true to
16		line and level as indicated. Install work with laps, joints, and seams that are permanently watertight.
17	3.3	CLEANING
18	Α.	Remove temporary protective coverings and strippable films, if any, as metal panels are installed unless otherwise
19		indicated in manufacturer's written installation instructions. On completion of metal panel installation, clean
20		finished surfaces as recommended by metal panel manufacturer. Maintain in a clean condition during construction.
21		END OF SECTION

Page Intentionally Left Blank

SECTION 07 53 23 ETHYLENE-PROPYLENE-DIENE-MONOMER (EPDM) ROOFING

2	ETHYLENE-PROPYLENE-DIENE-MONOMER (EPDM) ROOFING			
3	PART 1 - GENERAL			
4	1.1	SUMMARY		
5	Α.	Section Includes:		
6		1. Adhered ethylene-propylene-diene-terpolymer (EPDM) roofing system.		
7		2. Vapor barrier.		
8		3. Accessory roofing materials.		
9		4. Substrate board.		
10		5. Roof insulation.		
11		6. Insulation accessories.		
12		7. Asphalt materials.		
13		8. Walkways.		
14	1.2	PREINSTALLATION MEETINGS		
15	А.	Preliminary Conference: Conduct conference at Project site.		
16	1.3	ACTION SUBMITTALS		
17	Α.	Product Data: For each type of product.		
18		1. For insulation and roof system component fasteners, include copy of FM Approvals' RoofNav listing.		
19	В.	Shop Drawings: Include roof plans, sections, details, and attachments to other work, including the following:		
20		1. Layout and thickness if insulation.		
21		2. Base flashings and membrane terminations.		
22		3. Flashing details at penetrations.		
23		4. Tapered insulation, thickness, and slopes.		
24		5. Roof plan showing orientation of steel roof deck and orientation of roof membrane and fastening spacings		
25		and patterns for mechanically fastened roofing system.		
26		6. Insulation fastening patterns for corner, perimeter, and field-of-roof locations.		
27		7. Tie-in with air barrier.		
28	C.	Samples: For the following products:		
29		1. Boof membrane and flashings of color required		
30		2. Walkway pads or rolls, of color required.		
31	D	Wind Unlift Resistance Submittal: For roofing system indicating compliance with wind unlift performance		
32		requirements		
33	1.4	INFORMATIONAL SUBMITTALS		
34	Δ	Manufacturer Certificates:		
35	73.	1 Performance Requirement Certificate: Signed by roof membrane manufacturer, certifying that roofing		
36		system complies with requirements specified in "Performance Requirements" Article		
37		a Submit evidence of complying with performance requirements		
38		2 Special Warranty Certificate: Signed by roof membrane manufacturer, certifying that all materials supplied		
39		under this Section are acceptable for special warranty		
40	в	Product Test Reports: For components of roof membrane and insulation, for tests performed by a qualified testing		
41	5.	agency indicating compliance with specified requirements		
42	C	Research renorts		
42	с. D	Field Test Reports:		
ч 3 ЛЛ	υ.	1 Concrete internal relative humidity test reports		
45		 Eastener-nullout test results and manufacturer's revised requirements for fastener natterns 		
46	F	Field quality-control reports		
40 17	E.	Sample warranties		
47	15			
40 //Q	1.5	Maintenance data		
50	д. В	Cartified statement from existing roof membrane manufacturer stating that existing roof warranty has not been		
50	Б.	offected by Work performed under this Section		
27	16			
52	7.0 V	QUALITEASSURANUE		
33 E 4	А.	Qualifications.		
54 FF		 Information of the second secon		
55 E 6		to trial used for this project.		
50 57		install manufacturer's product and that is eligible to receive manufacturer's special warranty.		

May 16	, 2024
1.7	WARRANTY
Α.	Special Warranty: Manufacturer agrees to repair or replace components of roofing system that fail in materials or workmanship within specified warranty period.
	1. Warranty Period: 20 years from Date of Substantial Completion.
PART 2 -	PRODUCTS
2.1	PERFORMANCE REQUIREMENTS
Α.	Accelerated Weathering: Roof membrane to withstand 2000 hours of exposure when tested in accordance with ASTM G152, ASTM G154, or ASTM G155.
В.	Impact Resistance: Roof membrane to resist impact damage when tested in accordance with ASTM D3746 ASTM D4272, or the "Resistance to Foot Traffic Test" in FM Approvals 4470.
C.	FM Approvals' RoofNav Listing: Roof membrane, base flashings, and component materials comply with requirements in FM Approvals 4450 or FM Approvals 4470 as part of a roofing system, and are listed in FM Approvals' RoofNav for Class 1 or noncombustible construction, as applicable. Identify materials with FM Approvals
	Certification markings.
	1. Fire/Windstorm Classification: Class 1A-90.
	2. Hail-Resistance Rating: FM Global Property Loss Prevention Data Sheet 1-34 SH.
D.	Exterior Fire-Test Exposure: ASTM E108 or UL 790, Class A; for application and roof slopes indicated; testing by a qualified testing agency. Identify products with appropriate markings of applicable testing agency.
Ε.	Fire-Resistance Ratings: Comply with fire-resistance-rated assembly designs indicated. Identify products with
	appropriate markings of applicable testing agency.
2.2	ETHYLENE-PROPYLENE-DIENE-TERPOLYMER (EPDM) ROOFING
Α.	EPDM Sheet: ASTM D4637/D4637M, Type I, nonreinforced, EPDM sheet with factory-applied seam tape.
	1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
	a. Carlisle Syntec Systems.
	b. Johns Manville; a Berkshire Hathaway company.
	c. Versico Roofing Systems; Carlisle Construction Materials.
	2. Thickness: 90 mils, nominal.
	3. Exposed Face Color: Black.
2.3	VAPOR BARRIER
Α.	Composite Sheet for wood or glass-mat sheathing or concrete: Self-adhering 35-mil rubberized asphalt laminated to
	5-mil polypropylene film.
	1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
	a. Carlisle Syntec Systems.
	1) Basis-of-Design: VapAir Seal 725TR.
	b. Johns Manville; a Berkshire Hathaway company.
	c. Versico Roofing Systems; Carlisle Construction Materials.
	2. Thickness: 40 mils, nominal.
	3. Permeability: .015 perms per ASTM D1970 tested to E96 standards.
	4. Air Permeance: .000 L*m ² @75 Pa per ASTM E2148.
В.	Composite Sheet for metal deck: Reinforced aluminum foil with self-adhering SBS; 15-mil overall thickness.
	1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
	a. Carlisle Syntec Systems.
	1) Basis-of-Design: VapAir Seal MD.
	b. Johns Manville; a Berkshire Hathaway company.
	c. Versico Roofing Systems; Carlisle Construction Materials.
	2. Thickness: 40 mils, nominal.
	3. Permeability: .030 perms per ASTM D1970 tested to E96 standards.
	 Air Permeance: .000 L*m²@75 Pa per ASTM E2148.
2.4	ACCESSORY ROOFING MATERIALS
Α.	General: Accessory materials recommended by roofing system manufacturer for intended use and compatible with
	other roofing components.
	 Adhesive and Sealants: Comply with VOC limits of authorities having jurisdiction.
В.	Sheet Flashing: 60-mil-thick EPDM, partially cured or cured, according to application.
C.	Protection Sheet: Epichlorohydrin or neoprene nonreinforced flexible sheet, 55 to 60 mils thick, recommended by
	EPDM manufacturer for resistance to hydrocarbons, non-aromatic solvents, grease, and oil. Install within 8'-0" of
	perimeter of kitchen exhaust duct.
D.	Prefabricated Pipe Flashings: As recommended by roof membrane manufacturer.
Ε.	Bonding Adhesive: Manufacturer's standard.

IVIAY 1	b, 2024
F.	Seaming Material: Use factory-applied seam tape, width as recommended by manufacturer to extent possible. Then use manufacturer's standard, synthetic-rubber polymer primer and 3-inch-wide minimum, butyl splice tape
G	with release film Factory-applied seam tape, width as recommended by manufacturer.
С. Н.	Water Cutoff Mastic: Manufacturer's standard butyl mastic sealant.
I.	Metal Termination Bars: Manufacturer's standard, predrilled stainless steel or aluminum bars, approximately 1 by
	1/8 Inch Thick; With anchors. Easteners: Eactory-coated steel fasteners and metal or plastic plates complying with corrosion-resistance provisions
5.	in FM Approvals 4470, designed for fastening components to substrate, and acceptable to roofing system manufacturer.
К.	Miscellaneous Accessories: Provide pourable sealers, preformed cone and vent sheet flashings, molded pipe boot flashings, preformed inside and outside corner sheet flashings, reinforced EPDM securement strips, T-joint covers, inseem sealants, termination reglets, cover strips, and other accessories
	1. Provide white flashing accessories for white EPDM membrane roofing.
2.5	SUBSTRATE BOARD
Α.	Glass-Mat Gypsum Roof Substrate Board: ASTM C1177/C1177M, water-resistant gypsum board.
	1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
	a. CertainTeed; SAINT-GOBAIN.
	b. Georgia-Pacific Gypsum LLC.
	c. USG Corporation.
	2. Inickness: Type X, 5/8 inch.
р	5. Surface Finish. Factory primeu.
Б.	in EM Approvals 4470, designed for factoring substrate papel to roof dock
26	
Δ	Roof insulation assembly must provide an overall average of R-35
Л.	Polyisocyanurate Board Insulation: ASTM C1289 Type II Class 1 felt facer on both major surfaces
υ.	1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
	a. Atlas Polyiso Roof and Wall Insulation.
	b. Carlisle Syntec Systems.
	c. CertainTeed; SAINT-GOBAIN.
	d. GAF.
	e. Hunter Panels; a Carlisle company.
	f. Insulfoam; a Carlisle Company.
	g. Johns Manville; a Berkshire Hathaway company.
	h. Rmax, A Business Unit of Sika Corporation.
	2. Size: 48 by 96 inches.
	3. Thickness:
	a. Base Layer: 1-1/2 inches.
_	b. Upper Layer: As needed per tapered insulation plan to meet required R-value.
С.	Tapered Insulation (if applicable): Provide factory-tapered insulation boards.
	1. Material: Match root insulation.
	2. Minimum Thickness: 1/4 Inch.
	3. Slope:
	 a. Root Fleta: 1/4 first per tool unless otherwise indicated on Drawings. b. Saddles and Crickets: 1/2 inch per foot unless otherwise indicated on Drawings.
27	INSTITUTION ACCESSORIES
Δ.,	Easteners: Factory-coated steel fasteners and metal or plastic plates complying with corrosion-resistance provisions
	in FM Approvals 4470, designed for fastening roof insulation to substrate, and acceptable to roofing system
	manufacturer.
В.	Insulation Adhesive: Insulation manufacturer's recommended adhesive formulated to attach roof insulation to
	substrate or to another insulation layer as follows:
	1. Bead-applied, low-rise, one-component or multicomponent urethane adhesive.
С.	Protection Mat: Woven or nonwoven polypropylene, polyolefin, or polyester fabric; water permeable and resistant
	to UV degradation; type and weight as recommended by roofing system manufacturer for application.
2.8	ASPHALT MATERIALS
Α.	Roofing Asphalt: ASTM D312/D312M, Type III or Type IV.
В.	Asphalt Primer: ASTM D41/D41M.

iviay 10	, 2027
2.9	WALKWAYS
Α.	Flexible Walkways (if called for in drawings): Factory-formed, nonporous, heavy-duty, slip-resisting, surface
	textured walkway pads or rolls, approximately 3/16 inch thick and acceptable to roofing system manufacturer.
	1. Size: Approximately 36 by 60 inches.
	2. Color: Contrasting with roof membrane.
PART 3 -	EXECUTION
3.1	EXAMINATION
Α.	Examine substrates, areas, and conditions, with Installer present, for compliance with requirements and other
	conditions affecting performance of the Work.
	1. Verify that minimum concrete drying period recommended by roofing system manufacturer has passed.
	2. Verify that concrete substrate is visibly dry and free of moisture, and that minimum concrete interna
	relative humidity is not more than 75 percent, or as recommended by roofing system manufacturer wher
	tested in accordance with ASTM F2170.
	a. Test Frequency: One test probe per each 1000 sq. ft., or portion thereof, of roof deck, with not less
	than three test probes.
	b. Submit test reports within 24 hours of performing tests.
	3. Verify that concrete-curing compounds that will impair adhesion of roofing components to roof deck have
	been removed.
	4. Verify that joints in precast concrete roof decks have been grouted flush with top of concrete.
3.2	PREPARATION
Α.	Perform fastener-pullout tests in accordance with roof system manufacturer's written instructions.
	1. Submit test result within 24 hours of performing tests.
	a. Include manufacturer's requirements for any revision to previously submitted fastener patterns
	required to achieve specified wind uplift requirements.
3.3	INSTALLATION OF ROOFING, GENERAL
Α.	Install roofing system in accordance with roofing system manufacturer's written instructions, FM Approvals
	RoofNav assembly requirements, and FM Global Property Loss Prevention Data Sheet 1-29.
В.	Complete terminations and base flashings and provide temporary seals to prevent water from entering completed
	sections of roofing system at end of workday or when rain is forecast. Remove and discard temporary seals before
	beginning work on adjoining roofing.
С.	Install roof membrane and auxiliary materials to tie in to existing roofing to maintain weathertightness of
	transition (and to not void warranty for existing roofing system when applicable).
D.	Coordinate installation and transition of roofing system component serving as an air barrier with air barrier
	specified under Section 07 27 26 "Fluid-Applied Membrane Air Barriers."
3.4	INSTALLATION OF SUBSTRATE BOARD
Α.	Install where identified in roofing details and/or roof assembly descriptions in the drawings.
В.	Install substrate board with long joints in continuous straight lines, with end joints staggered not less than 24 inches
	in adjacent rows.
	 At steel roof decks, install substrate board at right angle to flutes of deck.
	a. Locate end joints over crests of steel roof deck.
	At wood sheathing, install substrate board parallel to long axis of wood panels.
	a. Locate end joints over structural framing below but not over wood sheathing joints.
	3. Tightly butt substrate boards together.
	4. Cut substrate board to fit tight around penetrations and projections, and to fit tight to intersecting sloping
	roof decks.
	5. Fasten substrate board to top flanges of steel deck (or wood sheathing) in accordance with
	recommendations in FM Approvals' RoofNav listed roof assembly requirements for specified Windstorm
	Resistance Classification and FM Global Property Loss Prevention Data Sheet 1-29.
	6. Fasten substrate board to top flanges of steel deck (or wood sheathing) to resist uplift pressure at corners
	perimeter, and field of roof in accordance with roofing system manufacturers' written instructions.
3.5	INSTALLATION OF VAPOR BARRIER
Α.	See product description above for correct product for application.
В.	Install self-adhering membrane per manufacturer's written instructions.
3.6	INSTALLATION OF INSULATION
Α.	Coordinate installing roofing system components so insulation is not exposed to precipitation or left exposed at enc
	of workday.
В.	Comply with roofing system and insulation manufacturer's written instructions for installing roof insulation.
C.	Insulation installation intent:

	1. Top-most course of insulation is to be adhered to course below with lower courses screwed to deck as a stack (i.e., lower courses are not screwed per course).
П	Installation Over Metal Decking (applies to decks with or without substrate board):
υ.	1. Install base and subsequent layers of insulation with joints staggered not less than 24 inches in adjacent
	rows, end joints staggered not less than 12 inches in adjacent rows, and with long joints continuous at right
	angle to flutes of decking.
	a. Locate end joints over crests of decking.
	b. Where installing composite and noncomposite insulation in two or more layers, install
	noncomposite board insulation for bottom layer and intermediate layers, if applicable, and install
	composite board insulation for top layer.
	c. Trim insulation neatly to fit around penetrations and projections, and to fit tight to intersecting
	sloping roof decks.
	d. Make joints between adjacent insulation boards not more than 1/4 inch in width.
	e. At internal roof drains, slope insulation to create a square drain sump with each side equal to the
	diameter of the drain bowl plus 24 inches.
	1) Trim insulation so that water flow is unrestricted.
	f. Fill gaps exceeding 1/4 inch with insulation.
	g. Cut and fit insulation within 1/4 inch of nailers, projections, and penetrations.
	h. Loosely lay base layer of insulation units over substrate.
	i. Mechanically attach base layer of insulation (and substrate board when noted) using mechanical
	fasteners specifically designed and sized for fastening specified board-type roof insulation to metal
	decks.
	1) Fasten insulation in accordance with requirements in FM Approvals' RootNav for specified
	Windstorm Resistance Classification.
	2) Fasten insulation to resist specified uplift pressure at corners, perimeter, and field of roor.
	2. Install top layer of tapered insulation with joints of each layer offset not less than 12 inches non previous
	layer of illisulation.
	a. Staggered end joints within each ager not less than 24 niches in adjacent rows.
	rows
	Trim insulation neatly to fit around penetrations and projections and to fit tight to intersecting
	sloping roof decks.
	d. Make joints between adjacent insulation boards not more than 1/4 inch in width.
	e. At internal roof drains, slope insulation to create a square drain sump with each side equal to the
	diameter of the drain bowl plus 24 inches.
	1) Trim insulation so that water flow is unrestricted.
	f. Fill gaps exceeding 1/4 inch with insulation.
	g. Cut and fit insulation within 1/4 inch of nailers, projections, and penetrations.
	h. Adhere top layer of tapered insulation to layer below using adhesive in accordance with FM
	Approvals' RoofNav listed roof assembly requirements for specified Windstorm Resistance
	Classification and FM Global Property Loss Prevention Data Sheet 1-29, as follows:
	1) Set each layer of insulation in ribbons of bead-applied insulation adhesive, firmly pressing
	and maintaining insulation in place.
E.	Installation Over Wood Sheathing (applies to decks with or without substrate board):
	1. Follow direction for "Installation Over Metal Decking" above, except that ends of insulation to be
	mechanically attached are to align with wood framing below.
3.7	ADHERED ROOFING INSTALLATION
Α.	Adhere roof membrane over area to receive roofing in accordance with roofing system manufacturer's written
	instructions.
В.	Unroll membrane roof membrane and allow to relax before installing.
С.	Start installation of roofing in presence of roofing system manufacturer's technical personnel.
D.	Accurately align roof membrane, and maintain uniform side and end laps of minimum dimensions required by
_	manufacturer. Stagger end laps.
Ε.	Bonding Adhesive: Apply to substrate and underside of roof membrane at rate required by manufacturer, and allow
-	to partially dry before installing root membrane. Do not apply to splice area of root membrane.
F.	in addition to adhering, mechanically fasten roof membrane securely at terminations, penetrations, and perimeters.
G.	Apply root memorate with side laps simgled with slope of root deck where possible.
н.	rape search installation. Clean and prime both faces of splice areas, apply splice tape.

	May 16	, 2024
1		1. Firmly roll side and end laps of overlapping roof membrane to ensure a watertight seam installation.
2		2. Apply lap sealant and seal exposed edges of roofing terminations.
3	Ι.	Factory-Applied Seam Tape Installation: Clean and prime surface to receive tape.
4		1. Firmly roll side and end laps of overlapping roof membrane to ensure a watertight seam installation.
5		2. Apply lap sealant and seal exposed edges of roofing terminations.
6	J.	Spread sealant or mastic bed over deck-drain flange at roof drains, and securely seal roof membrane in place with
7		clamping ring.
8	К.	Adhere protection sheet over roof membrane at locations indicated.
9	3.8	INSTALLATION OF BASE FLASHING
10	Α.	Install sheet flashings and preformed flashing accessories, and adhere to substrates in accordance with roofing
11		system manufacturer's written instructions.
12	В.	Apply bonding adhesive to substrate and underside of sheet flashing at required rate, and allow to partially dry. Do
13		not apply to seam area of flashing.
14	C.	Flash penetrations and field-formed inside and outside corners with cured or uncured sheet flashing.
15	D.	Clean splice areas, apply splicing cement, and firmly roll side and end laps of overlapping sheets to ensure a
16		watertight seam installation. Apply lap sealant and seal exposed edges of sheet flashing terminations.
17	E.	Terminate and seal top of sheet flashings and mechanically anchor to substrate through termination bars.
18	3.9	INSTALLATION OF WALKWAYS
19	Α.	Flexible Walkways: Install walkway products in accordance with manufacturer's written instructions.
20		1. Install flexible walkways at the following locations:
21		a. Perimeter of each rooftop unit.
22		b. Between each rooftop unit location, creating a continuous path connecting rooftop unit locations.
23		c. Between each roof hatch and each rooftop unit location or path connecting rooftop unit locations.
24		d. Top and bottom of each roof access ladder.
25		e. Between each roof access ladder and each rooftop unit location or path connecting rooftop unit
26		locations.
27		f. Locations indicated on Drawings.
28		g. As required by roof membrane manufacturer's warranty requirements.
29		2. Provide 6-inch clearance between adjoining pads.
30		3. Adhere walkway products to substrate with compatible adhesive in accordance with roofing system
31		manufacturer's written instructions.
32	3.10	FIELD QUALITY CONTROL
33	Α.	Final Roof Inspection: Arrange for roofing system manufacturer's technical personnel to inspect roofing installation
34		on completion, in presence of Architect, and to prepare inspection report.
35	В.	Repair or remove and replace components of roofing system where inspections indicate that they do not comply
36		with specified requirements.
37	3.11	PROTECTING AND CLEANING
38	Α.	Protect roofing system from damage and wear during remainder of construction period. When remaining
39		construction does not affect or endanger roofing system, inspect roofing system for deterioration and damage,
40		describing its nature and extent in a written report, with copies to Architect and Owner.
41	В.	Correct deficiencies in or remove roofing system that does not comply with requirements, repair substrates, and
42		repair or reinstall roofing system to a condition free of damage and deterioration at time of Substantial Completion
43		and in accordance with warranty requirements.
44	С.	Clean overspray and spillage from adjacent construction using cleaning agents and procedures recommended by
45		manufacturer of affected construction.
46		END OF SECTION

SECTION 07 62 00 SHEET METAL FLASHING AND TRIM

2		SHEET METAL FLASHING AND TRIM
3	PART 1 - (GENERAL
4	1.1	SUMMARY
5	Α.	Section Includes: Custom units for the following applications:
6		1. Sheet metal materials.
7		 Wall sheet metal fabrications – <u>for through-wall flashing and drip edge</u>.
8		3. Underlayment.
9		4. Miscellaneous materials.
10	В.	Related Section: See Section 07 71 00 - Roof Specialties for manufactured units if required.
11	1.2	PREINSTALLATION MEETINGS
12	Α.	Preinstallation Conference: Conduct conference at Project site.
13	1.3	ACTION SUBMITTALS
14	Α.	Product data.
15	В.	Shop Drawings: For sheet metal flashing and trim.
16		1. Plans, elevations, sections, and attachment details.
17		2. Fabrication and installation layouts, expansion-joint locations, and keyed details. Distinguish between shop-
18		and field-assembled Work.
19		 Identification of material, thickness, weight, and finish for each item and location in Project. Details for forming including and files shares around elementing.
20		4. Details for forming, including profiles, snapes, seams, and dimensions.
21		5. Details for joining, supporting, and securing, including layout and spacing of fasteners, cleats, clips, and
22		Other attachments. Include pattern of seams.
25		 Details of expansion joints and expansion joint covers including showing direction of expansion and
24		contraction from fixed points
25		8 Details of roof-nenetration flashing
20		9 Details of edge conditions including eaves ridges valleys rakes crickets flashings and counterflashings
28		10 Details of special conditions
29		11. Details of connections to adjoining work.
30		12. Formed flashing and trim at scale of not less than 1-1/2 inches per 12 inches.
31	C.	Samples: For each exposed product and for each color and texture specified. 12 inches long by actual width.
32	1.4	INFORMATIONAL SUBMITTALS
33	Α.	Certificates: For each type of coping and roof edge flashing that is FM Approvals approved.
34	В.	Product test reports.
35	С.	Research reports.
36	D.	Sample warranties.
37	1.5	CLOSEOUT SUBMITTALS
38	Α.	Maintenance data.
39	1.6	QUALITY ASSURANCE
40	Α.	Installer Qualifications: Entity that employs a supervisor who is an NRCA ProCertified Roofing Foreman or installers
41		who are NRCA ProCertified Architectural Metal Flashings and Accessories Installers.
42	В.	For roof edge flashings and copings that are ANSI/SPRI/FM 4435/ES-1 tested and FM Approvals approved, shop is to
43		be listed as able to fabricate required details as tested and approved.
44	1.7	WARRANTY
45	Α.	Special Warranty on Finishes: Manufacturer agrees to repair finish or replace sheet metal flashing and trim that
46		shows evidence of deterioration of factory-applied finishes within specified warranty period.
4/		1. Exposed Panel Finish: Deterioration includes, but is not limited to, the following:
48		a. Color fading more than 5 Delta E units when tested in accordance with ASTM D2244.
49 F0		D. Chaiking in excess of a NO. 8 rating when tested in accordance with ASTM D4214. Cracking chacking chacking chailing or failure of paint to adhere to have matal.
50		C. Cracking, checking, peeling, or failure of paint to adhere to bare metal.
21	ם אחד א	2. Finish warranty Period: 20 years from date of Substantial Completion.
52 52	7 ANIZ-1	
55	۲.۲	FEM ONWARDE REQUIREMENTS Sheet metal flaching and trim assemblies, including cleats, anchors, and factoners, are to withstand wind loads
55	А.	structural movement thermally induced movement and exposure to weather without failure due to defective
56		manufacture fabrication installation or other defects in construction. Completed sheet metal flashing and trim are
57		not to rattle leak or loosen and are to remain watertight
5,		not to ratio leavy or loosely and are to remain water abits

	May 16	2024
1 2	В.	Sheet Metal Standard for Flashing and Trim: Comply with NRCA's "The NRCA Roofing Manual: Architectural Metal Flashing, Condensation and Air Leakage Control, and Reroofing" and SMACNA's "Architectural Sheet Metal Manual"
3	_	requirements for dimensions and profiles shown unless more stringent requirements are indicated.
4	С.	FM Approvals Listing: Manufacture and install roof edge flashings and copings that comply with requirements in
5		FM Approvals 4471 as part of a roofing system and that are listed in FM Approvals' "Approval Guide" and approved
6		for windstorm classification, Class 1A-90. Identify materials with name of fabricator and design approved by FM
7	_	Approvals.
8	D.	Thermal Movements: Allow for thermal movements from ambient and surface temperature changes.
9		1. Temperature Change: 120 deg F, ambient; 180 deg F, material surfaces.
10	2.2	SHEET METAL MATERIALS
11	Α.	Protect mechanical and other finishes on exposed surfaces from damage by applying strippable, temporary
12		protective film before shipping.
13	В.	Metallic-Coated Steel Sneet: Zinc-coated (galvanized) steel sneet complying with minimum ASTM A653/A653M,
14		G90 coating designation, or aluminum-zinc alloy-coated steel sheet complying with minimum ASTM A/92/A/92M,
15		class AZ50 coating designation; structural quality. Prepainted by the coll-coating process to comply with
10		ASTIMA/SS/A/SSIM.
10		1. Nominal Inickness: 0.028 inch.
10		2. Surface: Smooth, flat.
19		3. EXPOSED COIl-Codled Finish:
20		a. Two-coat Fluoropolymer. Fluoropolymer filmsh containing not less than 70 percent PVDF resin by
21		weight in color coat. Prepare, precreat, and apply coating to exposed metal surfaces to comply with
22		Color: As selected by Architect from manufacturer's full range
23		 Concepted Einich: Pretreat with manufacturer's standard white or light-colored acrulic or polyester backer
24		finish consisting of nrime cost and wash cost with minimum total dry film thickness of 0.5 mil
26	C	Stainless Steel Sheet: ASTM A240/A240M Type 304 dead soft fully annealed
20	с.	1 Nominal Thickness:
27		a Unless noted otherwise: 0.0188 inch
29		b Drin Edge: 0.0250 inch
30		c. Built-In Gutters: 0.0250 inch.
31		2. Surface: Smooth. flat.
32		 Exterior Finish: ASTM A480/A480M. No. 2D (dull. cold rolled).
33		a. Surface Preparation: Remove tool and die marks and stretch lines, or blend into finish.
34	2.3	UNDERLAYMENT
35	Α.	Self-Adhering, High-Temperature Sheet Underlayment: Provide self-adhering, cold-applied, sheet underlayment, a
36		minimum of 30 mils thick, specifically designed to withstand high metal temperatures beneath metal roofing.
37		Provide primer when recommended by underlayment manufacturer.
38		1. Thermal Stability: Stable after testing at 240 deg F; ASTM D1970/D1970M.
39		2. Low-Temperature Flexibility: Passes after testing at minus 20 deg F or lower; ASTM D1970/D1970M.
40		3. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
41		a. ATAS International, Inc.
42		b. Carlisle WIP Products; a brand of Carlisle Construction Materials.
43		c. Henry Company; a Carlisle company.
44		d. Owens Corning.
45	В.	Slip Sheet: Rosin-sized building paper, 3 lb/100 sq. ft. minimum, of type required for application.
46	2.4	MISCELLANEOUS MATERIALS
47	Α.	Provide materials and types of fasteners, solder, protective coatings, sealants, and other miscellaneous items as
48		required for complete sheet metal flashing and trim installation and as recommended by manufacturer of primary
49		sheet metal or manufactured item unless otherwise indicated.
50	В.	Fasteners: Wood screws, annular threaded nails, self-tapping screws, self-locking rivets and bolts, and other
51		suitable fasteners designed to withstand design loads and recommended by manufacturer of primary sheet
52		metal or manufactured item.
53		1. General: Blind fasteners or self-drilling screws, gasketed, with hex-washer head.
54		a. Exposed Fasteners: Heads matching color of sheet metal using plastic caps or factory-applied
55		coating. Provide metal-backed EPDM or PVC sealing washers under heads of exposed fasteners
56		bearing on weather side of metal.
5/		D. Blind Fasteners: High-strength aluminum or stainless steel rivets suitable for metal being fastened.
58		c. Spikes and Ferrules: Same material as gutter; with spike with ferrule matching internal gutter width.

IVIAY 10	, 2024
	2. Fasteners for Zinc-Coated (Galvanized) or Aluminum-Zinc Alloy-Coated Steel Sheet: Series 300 stainless steel or hot-dip galvanized steel in accordance with ASTM A153/A153M or ASTM F2329/F2329M.
	3. Fasteners for Aluminum Sheet: Aluminum or Series 300 stainless steel.
	4. Fasteners for Stainless Steel Sheet: Series 300 stainless steel.
	5. Fasteners for Copper Sheet: Copper, hardware bronze or passivated Series 300 stainless steel.
С.	Solder:
	1. For Zinc-Coated (Galvanized) Steel: ASTM B32, Grade Sn50, 50 percent tin and 50 percent lead or
	Grade Sn60, 60 percent tin and 40 percent lead with maximum lead content of 0.2 percent.
	2. For Stainless Steel: ASTM B32, Grade Sn60, with acid flux of type recommended by stainless steel sheet
	manufacturer.
D.	Sealant Tape: Pressure-sensitive, 100 percent solids, polyisobutylene compound sealant tape with release-paper
	backing. Provide permanently elastic, nonsag, nontoxic, nonstaining tape 1/2 inch wide and 1/8 inch thick.
Ε.	Elastomeric Sealant: ASTM C920, elastomeric polyurethane polymer sealant; of type, grade, class, and use
	classifications required to seal joints in sheet metal flashing and trim and remain watertight.
F.	Butyl Sealant: ASTM C1311, single-component, solvent-release butyl rubber sealant; polyisobutylene plasticized;
	heavy bodied for hooked-type expansion joints with limited movement.
G.	Bituminous Coating: Cold-applied asphalt emulsion in accordance with ASTM D1187/D1187M.
Н.	Asphalt Roofing Cement: ASTM D4586/D4586M, asbestos free, of consistency required for application.
Ι.	Reglets: Units of type, material, and profile required, formed to provide secure interlocking of separate reglet and
	counterflashing pieces, and compatible with flashing indicated with factory-mitered and -welded corners and
	junctions and with interlocking counterflashing on exterior face, of same metal as reglet.
	1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
	a. Fry Reglet Corporation.
	b. Heckmann Building Products, Inc.
	c. Hohmann & Barnard, Inc.
	d. Metal-Era, Inc.
	2. Material: Stainless steel, 0.0188 inch thick.
	3. Surface-Mounted Type: Provide with slotted holes for fastening to substrate, with neoprene or other
	suitable weatherproofing washers, and with channel for sealant at top edge.
	4. Stucco Type: Provide with upturned fastening flange and extension leg of length to match thickness of
	applied finish materials.
	5. Concrete Type: Provide temporary closure tape to keep reglet free of concrete materials, special fasteners
	for attaching reglet to concrete forms, and guides to ensure alignment of reglet section ends.
	6. Masonry Type: Provide with offset top flange for embedment in masonry mortar joint.
	7. Accessories:
	a. Flexible-Flashing Retainer: Provide resilient plastic or rubber accessory to secure flexible flashing in
	reglet where clearance does not permit use of standard metal counterflashing or where Drawings
	show reglet without metal counterflashing.
	b. Counterflashing Wind-Restraint Clips: Provide clips to be installed before counterflashing to prevent
	wind uplift of counterflashing's lower edge.
	8. Finish: With manufacturer's standard color coating.
2.5	FABRICATION, GENERAL
Α.	Custom fabricate sheet metal flashing and trim to comply with details indicated and recommendations in cited
	sheet metal standard that apply to design, dimensions, geometry, metal thickness, and other characteristics of item
	required.
	 Fabricate sheet metal flashing and trim in shop to greatest extent possible.
	2. Fabricate sheet metal flashing and trim in thickness or weight needed to comply with performance
	requirements, but not less than that specified for each application and metal.
	3. Verify shapes and dimensions of surfaces to be covered and obtain field measurements for accurate fit
	before shop fabrication.
	4. Form sheet metal flashing and trim to fit substrates without excessive oil-canning, buckling, and tool marks;
	true to line, levels, and slopes; and with exposed edges folded back to form hems.
	5. Conceal fasteners and expansion provisions where possible. Do not use exposed fasteners on faces exposed
	to view.
В.	Fabrication Tolerances:
	1. Fabricate sheet metal flashing and trim that is capable of installation to a tolerance of 1/4 inch in 20 ft. on
	slope and location lines indicated on Drawings and within 1/8-inch offset of adjoining faces and of
	alignment of matching profiles.

CITY OF MADISON SPECIFICATION

	May 16	, 2024
1		2. Fabricate sheet metal flashing and trim that is capable of installation to tolerances specified.
2	С.	Expansion Provisions: Form metal for thermal expansion of exposed flashing and trim.
3		1. Form expansion joints of intermeshing hooked flanges, not less than 1 inch deep, filled with butyl sealant
4		concealed within joints.
5		Use lapped expansion joints only where indicated on Drawings.
6	D.	Sealant Joints: Where movable, nonexpansion-type joints are required, form metal in accordance with cited sheet
7		metal standard to provide for proper installation of elastomeric sealant.
8	Ε.	Fabricate cleats and attachment devices from same material as accessory being anchored or from compatible,
9		noncorrosive metal.
10	F.	Fabricate cleats and attachment devices of sizes as recommended by cited sheet metal standard for application, but
11		not less than thickness of metal being secured.
12	G.	Seams - Typical:
13		1. Fabricate nonmoving seams with flat-lock seams. Form seams and seal with elastomeric sealant unless
14		otherwise recommended by sealant manufacturer for intended use. Rivet joints where necessary for
15		strength.
16		2. Seams for Aluminum: Fabricate nonmoving seams with flat-lock seams. Form seams and seal with epoxy
17		seam sealer. Rivet joints where necessary for strength.
18	Н.	Seams – Concealed Gutters and Thru-Wall Scuppers:
19		1. Fabricate nonmoving seams with flat-lock seams. Tin edges to be seamed, form seams, and solder.
20	2.6	ROOF-DRAINAGE SHEET METAL FABRICATIONS
21	Α.	Hanging Gutters:
22		1. Fabricate to cross section required, complete with end pieces, outlet tubes, and other accessories as
23		required.
24		2. Fabricate in minimum 96-inch-long sections.
25		3. Furnish flat-stock gutter brackets and flat-stock gutter spacers and straps fabricated from same metal as
26		gutters, of size recommended by cited sheet metal standard, but with thickness not less than twice the
27		gutter thickness.
28		 Fabricate expansion joints, expansion-joint covers, gutter bead reinforcing bars, and gutter accessories from
29		same metal as gutters. Shop fabricate interior and exterior corners.
30		5. Gutters with Girth up to 15 Inches: Fabricate from the following materials:
31		a Aluminum-Zinc Allov-Coated Steel: See above
32	В	Built-in Gutters:
33	υ.	1 Eabricate to cross section required with riveted and soldered joints complete with end pieces outlet
34		tubes and other special accessories as required
35		 Fabricate in minimum 96-inch-long sections. Fabricate expansion joints and accessories from same metal as
36		gutters unless otherwise indicated
37		3 Fabricate gutters with built-in expansion joints and gutter-end expansion joints at walls
38		4 Fabricate from the following materials:
30		a Stainless Staal See above
40	C	Downshouts: Fabricate rectangular downshouts to dimensions indicated on Drawings complete with mitered
4 0 Л1	С.	elhows Furnish with metal hangers from same material as downshouts and anchors. Shon fabricate elhows
42		1 Fabricate from the following materials:
42 43		a Aluminum-7inc Alloy-Coated Steel: See above
45 AA	П	Splach Pane: Eabricate to dimensions and shape required and from the following materials:
44	D.	1 Stainless Steel: See above
45	27	
40	2.7	Boof Edge Electing (Cravel Step) and Eastin Car: Easticate in minimum 06 inch long, but not exceeding 12 ft long
47 10	А.	continue furnish with 6 inch wide loats cover plates. Shon fabricate in training in solution and outside rear and outside interview compare
40		Sections. Furnish with 6-inch-wide, joint cover plates. Shop fabricate interior and exterior corners.
49		1. Fabricate from the following materials:
50	P	a. Juliilless Jueen abuve.
51	В.	Copings: Fabricate in minimum 96-inch-iong, but not exceeding 12 ft iong, sections. Fabricate joint plates of same
52		thickness as copings. Furnish with continuous cleats to support edge of external leg and drill elongated noies for
55 F 4		asteriers on interior leg. while corners, lasten and seal watertight. Shop radricate interior and exterior corners.
54		1. Fabricate from the following materials:
55	~	a. Aluminum-Zinc Alloy-Coated Steel: 0.040 inch thick .
50	ί.	Base Hashing: Shop tabricate interior and exterior corners. Fabricate from the following materials:
5/	_	1. Aluminum-Zinc Alloy-Coated Steel: See above.
58	D.	Counterflashing: Shop fabricate interior and exterior corners. Fabricate from the following materials:
11109 10) 2021	
----------	-------------	--
F	1. Boof-	Aluminum-Zinc Alloy-Coated Steel: See above.
L.	1	Stainless Steel: See above
F	Boof.	Drain Flashing: Fahricate from the following materials:
1.	1	Stainless Steel: See above
28	Δ. ΜΔΠ	Stanness Steel. See above.
2.0 A	Throu	ugh-Wall Flashing: Fabricate continuous flashings in minimum 96-inch-long, but not exceeding 12 ft, long
А.	sectio	and similar flashings and at shelf angles. Fabricate discontinuous linter sill, and similar flashings to extend
	6 incl	hes beyond each side of wall openings: and form with 2-inch-high end dams. Fabricate from the following
	mate	rials.
	1	Stainless Steel Elevible Elashing (self-adhering): 0.012 inch thick including butyl adhesive
в	Onen	stances steel <u>reasone rasming</u> (sen adhering). 0.012 men thet melading bady adhesive.
2.	wall	ppenings. Form head and sill flashing with 2-inch-high, end dams. Fabricate from the following materials:
	1.	Stainless Steel Elexible Elashing (self-adhering): 0.012 inch thick including butyl adhesive.
C.	Basis	-of-Design: York 304 SA.
D.	Hemi	med Drip-Edge: Manufactured from stainless steel sheet.
	1.	Thickness: See above.
Ε.	Wall	Expansion-Joint Cover: Fabricate from the following materials:
	1.	Aluminum-Zinc Alloy-Coated Steel: See above.
PART 3 -	EXECUT	ION
3.1	EXAN	ΛΙΝΑΤΙΟΝ
Α.	Exam	ine substrates, areas, and conditions, with Installer present, for compliance with requirements for installation
	tolera	ances, substrates, and other conditions affecting performance of the Work.
	1.	Verify compliance with requirements for installation tolerances of substrates.
	2.	Verify that substrate is sound, dry, smooth, clean, sloped for drainage, and securely anchored.
	3.	Verify that air- or water-resistant barriers have been installed over substrate to prevent air infiltration or
		water penetration.
В.	Proce	eed with installation only after unsatisfactory conditions have been corrected.
.2	INST	ALLATION OF UNDERLAYMENT
Α.	Self-A	Adhering, High-Temperature Sheet Underlayment:
	1.	Install self-adhering, high-temperature sheet underlayment; wrinkle free.
	2.	Prime substrate if recommended by underlayment manufacturer.
	3.	Comply with temperature restrictions of underlayment manufacturer for installation; use primer for
		installing underlayment at low temperatures.
	4.	Apply in shingle fashion to shed water, with end laps of not less than 6 inches staggered 24 inches between
		courses.
	5.	Overlap side edges not less than 3-1/2 inches. Roll laps and edges with roller.
	6.	Roll laps and edges with roller.
	7.	Cover underlayment within 14 days.
В.	Insta	ll slip sheet, wrinkle free, over underlayment before installing sheet metal flashing and trim.
	1.	Install in shingle fashion to shed water.
	2.	Lapp joints not less than 4 inches.
3.3	INST	ALLATION OF SHEET METAL FLASHING AND TRIM, GENERAL
Α.	Insta	Il sheet metal flashing and trim to comply with details indicated and recommendations of cited sheet metal
	stand	lard that apply to installation characteristics required unless otherwise indicated on Drawings.
	1.	Install fasteners, solder, protective coatings, separators, sealants, and other miscellaneous items as
	-	required to complete sheet metal flashing and trim system.
	2.	Install sheet metal flashing and trim true to line, levels, and slopes. Provide uniform, neat seams with
	2	minimum exposure of solder and/orsealant.
	3.	Anchor sheet metal flashing and trim and other components of the Work securely in place, with provisions
		for thermal and structural movement.
	4.	install sneet metal flashing and trim to fit substrates and to result in watertight performance.
	5.	Install continuous cleats with fasteners spaced not more than 12 inches o.c.
	ь.	space individual cleats not more than 12 inches apart. Attach each cleat with at least two fasteners. Bend
	7	tabs over tasteriers.
	/.	install exposed sneet metal flacking and trim with limited oil-canning, and free of buckling and tool marks.
	ð. O	Do not use graphite papelle to mark metal surfaces
	9.	Do not use graphite pencils to mark metal surfaces.

	May 16	, 2024
1	В.	Metal Protection: Where dissimilar metals contact each other, or where metal contacts pressure-treated wood or
2		other corrosive substrates, protect against galvanic action or corrosion by painting contact surfaces with bituminous
3		coating or by other permanent separation as recommended by sheet metal manufacturer or cited sheet metal
4		standard.
5		1. Coat concealed side of uncoated-aluminum and stainless steel sheet metal flashing and trim with
6		bituminous coating where flashing and trim contact wood, ferrous metal, or cementitious construction.
7		2. Underlayment: Where installing sheet metal flashing and trim directly on cementitious or wood substrates,
8		install underlayment and cover with slip sheet.
)	C.	Expansion Provisions: Provide for thermal expansion of exposed flashing and trim.
)		1. Space movement joints at maximum of 10 ft. with no joints within 24 inches of corner or intersection.
		2. Form expansion joints of intermeshing hooked flanges, not less than 1 inch deep, filled with sealant
2		concealed within joints.
		 Use Japped expansion joints only where indicated on Drawings.
	D	Easteners: Use fastener sizes that penetrate substrate not less than recommended by fastener manufacturer to
	υ.	achieve maximum null-out resistance
	F	Conceal fasteners and expansion provisions where possible in exposed work and locate to minimize possibility of
,	L.	leakage Cover and seal facteners and anchors as required for a tight installation
,	с	Sol joints as required for watertight construction
))	г.	1 Use sealant filled joints unless otherwise indicated
י ר		Ose sealantement joints unless other wise montace. Embod booked flagares of joint members not loss than 1 inch into sealant
, 1		a. Embed house hanges of joint members not less than I inch into sealant.
L)		b. Form joints to completely concean search.
2		c. when another temperature at time of installation is between 40 and 70 deg F, set joint members
•		for 50 percent movement each way.
ŀ		d. Adjust setting proportionately for instantation at higher ambient temperatures.
) -		1) Do not install sealant-type joints at temperatures below 40 deg F.
	~	2. Prepare joints and apply sealants to comply with requirements in Section 07.92.00 "Joint Sealants."
	G.	Soldered Joints: Clean surfaces to be soldered, removing oils and foreign matter.
5		1. Pretin edges of sheets with solder to width of 1-1/2 inches; however, reduce pretinning where pretinned
)		surface would show in completed Work.
)		2. Do not solder metallic-coated steel and aluminum sheet.
L		3. Do not use torches for soldering.
2		4. Heat surfaces to receive solder, and flow solder into joint.
3		a. Fill joint completely.
1		 Completely remove flux and spatter from exposed surfaces.
5		5. Stainless Steel Soldering:
6		a. Tin edges of uncoated sheets, using solder for stainless steel and acid flux.
7		b. Promptly remove acid-flux residue from metal after tinning and soldering.
3		c. Comply with solder manufacturer's recommended methods for cleaning and neutralization.
Э	Н.	Rivets: Rivet joints in uncoated aluminum where necessary for strength.
)	3.4	INSTALLATION OF ROOF-DRAINAGE SHEET METAL FABRICATIONS
L	Α.	Install sheet metal roof-drainage items to produce complete roof-drainage system in accordance with cited sheet
2		metal standard unless otherwise indicated. Coordinate installation of roof perimeter flashing with installation of
3		roof-drainage system.
4	В.	Hanging Gutters:
5		1. Join sections with riveted joints sealed with sealant.
6		2. Provide for thermal expansion.
7		3. Attach gutters at eave or fascia to firmly anchor them in position.
8		4. Provide end closures and seal watertight with sealant.
9		5. Slope to downspouts.
0		6. Install gutter with expansion joints at locations indicated on Drawings, but not exceeding, 50 ft. apart. Install
1		expansion-joint caps.
2	C.	Built-in Gutters:
3		1. Join sections with riveted and soldered joints.
4		2. Provide for thermal expansion.
5		3. Slope to downspouts.
6		4. Provide end closures and seal watertight with sealant.
7		5. Install underlayment layer in built-in gutter trough and extend to drip edge at eaves and under
8		underlayment on roof sheathing.

	May 16	, 2024
1		a. Lap sides minimum of 2 inches over underlying course.
2		b. Lap ends minimum of 4 inches.
3		c. Stagger end laps between succeeding courses at least 72 inches.
4		d. Fasten with roofing nails.
5		e. Install slip sheet over underlayment.
6 7		6. Install gutter with expansion joints at locations indicated on Drawings, but not exceeding, 50 ft. apart. Install expansion-joint caps.
8	D.	Downspouts:
9		1. Join sections with 1-1/2-inch telescoping joints.
10		Provide hangers with fasteners designed to hold downspouts securely to walls.
11		3. Locate hangers at top and bottom and at approximately 60 inches o.c.
12		4. Provide elbows at base of downspout to direct water away from building.
13	_	5. Connect downspouts to underground drainage system.
14	E.	Splash Pans:
15		1. Install where downspouts discharge on low-slope roots.
10	25	2. Set in asphalt rooting cement or elastomeric sealant compatible with the substrate.
10	3.5	INSTALLATION OF SLOPED ROOF SHEET WETAL FADRICATIONS
10	А.	1 Provide concealed fasteners where possible and set units true to line levels and clones
20		 Install work with lans joints and seams that are permanently watertight and weather resistant
21	В.	Roof Edge Elashing:
22	5.	1. Install roof edge flashings in accordance with ANSI/SPRI/FM 4435/ES-1.
23		2. Anchor to resist uplift and outward forces in accordance with recommendations in cited sheet metal
24		standard unless otherwise indicated. Interlock bottom edge of roof edge flashing with continuous cleat
25		anchored to substrate at staggered 3-inch centers.
26		3. Anchor to resist uplift and outward forces in accordance with recommendations in FM Global Property Loss
27		Prevention Data Sheet 1-49 for FM Approvals' listing for required windstorm classification.
28	С.	Copings:
29		1. Install copings in accordance with ANSI/SPRI/FM 4435/ES-1.
30		2. Anchor to resist uplift and outward forces in accordance with recommendations in cited sheet metal
31		standard unless otherwise indicated.
32 22		a. Interlock exterior bottom edge of coping with continuous cleat anchored to substrate at 24-inch
24		Lenders.
34 35		b. Anchor interior leg or coping with washers and screw fasteners through slotted holes at 24-inch
36	D.	Pipe or Post Counterflashing: Install counterflashing umbrella with close-fitting collar with top edge flared for
37		elastomeric sealant, extending minimum of 4 inches over base flashing. Install stainless steel draw band and
38		tighten.
39	E.	Counterflashing: Coordinate installation of counterflashing with installation of base flashing.
40		1. Insert counterflashing in reglets or receivers and fit tightly to base flashing.
41		2. Extend counterflashing 4 inches over base flashing.
42		3. Lap counterflashing joints minimum of 4 inches.
43	F.	Roof-Penetration Flashing: Coordinate installation of roof-penetration flashing with installation of roofing and other
44		items penetrating roof. Seal with elastomeric sealant and clamp flashing to pipes that penetrate roof.
45	3.6	INSTALLATION OF WALL SHEET METAL FABRICATIONS
46	А.	Install sheet metal wall flashing to intercept and exclude penetrating moisture in accordance with cited sheet metal
47		standard unless otherwise indicated. Coordinate installation of wall flashing with installation of wall-opening
48	р	Components such as windows, doors, and louvers.
49 50	Б.	beyond wall openings
50	C	Reglets: Installation of reglets is specified in Section 03.30.00 "Cast-in-Place Concrete" or Section 04.20.00 "Linit
52	С.	Masonry."
53	3.7	INSTALLATION TOLERANCES
54	А.	Shim and align sheet metal flashing and trim within installed tolerance of 1/4 inch in 20 ft. on slope and location
55		lines indicated on Drawings and within 1/8-inch offset of adjoining faces and of alignment of matching profiles.
56	3.8	CLEANING
57	Α.	Clean exposed metal surfaces of substances that interfere with uniform oxidation and weathering.
58	В.	Clean and neutralize flux materials. Clean off excess solder.

1	C.	Clean off excess sealants.
2	3.9	PROTECTION
3	Α.	Remove temporary protective coverings and strippable films as sheet metal flashing and trim are installed unless
4		otherwise indicated in manufacturer's written installation instructions.
5	В.	Replace sheet metal flashing and trim that have been damaged or that have deteriorated beyond successful repair
6		by finish touchup or similar minor repair procedures, as determined by Architect.
7		END OF SECTION

SECTION 07 72 53 SNOW GUARDS

3	PART 1 - 0	GENERAL
4	1.1	SUMMARY
5	Α.	Section Includes:
6		1. Pad-type, flat-mounted metal snow guards.
7	1.2	ACTION SUBMITTALS
8	Α.	Product Data: For each type of product.
9	В.	Shop Drawings: Include roof plans showing layouts and attachment details of snow guards.
10		1. Include details of rail-type snow guards.
11	C.	Samples:
12		1. Pad-Type Snow Guards: Full-size unit with installation hardware.
13		a. For units with factory-applied finishes, submit manufacturer's standard color selections.
14	D	Delegated Design Submittals: For snow guards include analysis reports signed and sealed by the gualified
15	υ.	professional angineer responsible for the prenaration
16		1 Include calculation of number and location of snow guards
17	13	
10	1.5	Qualification Data: For professional engineer's experience with providing delegated design engineering services of
10	А.	the kind indicated including documentation that the engineer is licensed in the state in which the Project is located
19	D	Breduct Test Benerics: For each type of snow guard, for tests performed by a gualified testing agency, indicating
20	Б.	Product Test Reports. For each type of show guard, for tests performed by a qualined testing agency, indicating
21		load at failure of attachment to root system dentical to root system used on this project.
22	PARI 2 - F	
23	2.1	PERFORMANCE REQUIREMENTS
24	А.	Delegated Design: Engage a qualified professional engineer, as defined in Section 01 40 00 "Quality Requirements,"
25		to design show guards, including attachment to rooting material and root deck, applicable for attachment method,
26		based on the following:
27		1. Roof snow load.
28		2. Snow drifting
29		3. Roof slope.
30		4. Roof type.
31		5. Roof dimensions.
32		6. Roofing substrate type and thickness.
33		7. Snow guard type.
34		8. Snow guard fastening method and strength.
35		9. Snow guard spacing.
36		10. Coefficient of Friction Between Snow and Roof Surface: 0.
37		11. Factor of Safety: 2.
38	В.	Performance Requirements: Provide snow guards that withstand exposure to weather and resist thermally induced
39		movement without failure, rattling, or fastener disengagement due to defective manufacture, fabrication,
40		installation, or other defects in construction.
41		1. Temperature Change: 120 deg F, ambient; 180 deg F, material surfaces.
42	С.	Structural Performance: Snow guards shall withstand the effects of gravity loads and the following loads and
43		stresses within limits and under conditions indicated.
44		1. Snow Loads: As indicated on Drawings.
45	2.2	PAD-TYPE SNOW GUARDS
46	Α.	Pad-Type, Flat-Mounted Metal Snow Guards:
47		1. Material:
48		a. ASTM A792/A792M, Class AZ50 aluminum-zinc alloy-coated steel sheet, Grade 40 but not less than
49		0.022 inch thick.
50		1) Finish: High-performance organic two-coat fluoropolymer finish complying with
51		AAMA 2605 and containing not less than 70 percent PVDF resin by weight in color coat.
52		a) Color: As selected by Architect from manufacturer's full range.
53		 Attachment: Manufacturer's tested system, capable of resisting design loads.
54	PART 3 - F	
55	3.1	INSTALLATION
56	Δ	Install snow guards according to manufacturer's written instructions
57	A .	1 Share rows as indicated on Shon Drawings
58	R	Attachment for Standing-Seam Metal Roofing
50	ь.	

SECTION 07 84 13 PENETRATION FIRESTOPPING

1		SECTION 07 84 13
2		PENETRATION FIRESTOPPING
3	PART 1 -	GENERAL
4	1.1	SUMMARY
5	Α.	Section Includes:
6		1. Penetration firestopping systems.
7		2. Penetrations in fire-resistance-rated walls.
8		3. Penetrations in horizontal assemblies.
9		4. Penetrations in smoke barriers.
10		5. Exposed penetration firestopping systems.
11	1.2	PREINSTALLATION MEETINGS
12	Α.	Preinstallation Conference: Conduct conference at Project site.
13	1.3	ACTION SUBMITTALS
14	Α.	Product data.
15	Р.	Product Schedule: For each nenetration firestonning system. Include location, illustration of firestonning system.
16	5.	and design designation of qualified testing and inspecting agency
17		1 Engineering Judgments: Where Project conditions require modification to a qualified testing and inspecting
10		1. Engineering subgritters where i to be consistent and the model and the second states and the modifications
10		agency's must allow the a particular percentagence in the stopping system, submit mestation, with momentagence as an
20		angine independent or equivalent fire-resistance-rated assembly developed in accordance with current
20		International Exector Council (EC) quidelines. Obtain approval of authorities baying inright when the
21		submittal
22	1 4	
25	1.4	listed system designs
24 25	A.	LISTED SYSTEM DESIGNS.
25	1.5	CLOSEOUT SUBMITTALS
20	А.	installer Certificates: From installer indicating that penetration firestopping systems have been installed in
27		compliance with requirements and manufacturer's written instructions.
28	1.6	QUALITY ASSURANCE
29	А.	Installer Qualifications: Entity that has been approved by FM Approvals in accordance with FM Approvals 4991 or
30		been evaluated by UL and found to comply with UL's "UL Solutions Qualified Firestop Contractor Program."
31	В.	Manufacturer Qualifications: Entity that has received UL's "Firestop Movement Certification," which demonstrates
32		that manufacturer's firestopping products designated with M-Ratings are based on exposure to cyclic movement
33		and UL 1479 fire test evaluation when tested in accordance with ASTM E3037.
34	PART 2 -	PRODUCTS
35	2.1	PERFORMANCE REQUIREMENTS
36	Α.	Fire-Test-Response Characteristics:
37		1. Perform penetration firestopping system tests by a qualified testing agency acceptable to authorities having
38		jurisdiction.
39		2. Test in accordance with testing standards referenced in "Penetration Firestopping Systems" Article. Provide
40		rated systems complying with the following requirements:
41		a. Penetration firestop systems installed with products bearing the classification marking of a qualified
42		testing agency.
43		1) UL in its online directory "Product iQ."
44		2) Intertek Group in its "Directory of Building Products."
45		3) FM Approvals in its "Approval Guide."
46	2.2	PENETRATION FIRESTOPPING SYSTEMS
47	Α.	Penetration Firestopping Systems: Systems that resist spread of fire, passage of smoke and other gases, and
48		maintain original fire-resistance rating of construction penetrated. Penetration firestopping systems are to be
49		compatible with one another, with the substrates forming openings, and with penetrating items if any.
50		1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
51		a. 3M Building and Construction.
52		b. Hilti Inc.
53		c. Tremco Incorporated.
54	R	Penetrations in Fire-Resistance-Rated Walls. Penetration firestonning systems with ratings determined in
55	υ.	accordance with ASTM F814 or III 1479
56		1 F-Rating: Not less than the fire-resistance rating of the wall penetrated
57		 Membrane Penetrations: Install recessed fixtures such that the required fire resistance will not be reduced
		2. Mensioner energations, instan recessed intraces such that the required me resistance will not be reduced.

Iviay 10,	2024
	3. M-Rating: Provide penetration firestopping systems meeting specified F-Rating after being tested in accordance with ASTM E3037.
C.	Penetrations in Horizontal Assemblies: Penetration firestopping systems with ratings determined in accordance with ASTM 5814 or UL 1470
	With ASTME E014 UI UL 1479.
	 F-Rating: At least one hour, but not less than the fire-resistance rating of the floor. The following floor T-Rating: At least one hour, but not less than the fire-resistance rating of the floor. The following floor
	penetrations do not require a T-rating:
	a. Those within the cavity of a wall.
	b. Floor, tub, or shower drains within a concealed space.
	c. 4-inch or smaller metal conduit penetrating directly into metal-enclosed electrical switchgear.
	3. W-Rating: Provide penetration firestopping systems with a Class 1 W-rating in accordance with UL 1479.
	4. M-Rating: Provide penetration firestopping systems meeting specified F-Rating, T-Rating, and W-Rating
	after being tested in accordance with ASTM E3037.
D.	Penetrations in Smoke Barriers: Penetration firestopping systems with ratings determined in accordance with
	UL 1479.
	1. L-Rating: Not exceeding 5.0 cfm/sq. ft. of penetration opening and no more than 50-cfm cumulative total
	for any 100 sq. ft. at both ambient and elevated temperatures.
	2. M-Rating: Provide penetration firestopping systems meeting specified L-Rating after being tested in
	accordance with ASTM E3037.
E.	Exposed Penetration Firestopping Systems: Flame-spread and smoke-developed indexes of less than 25 and 450,
	respectively, when tested in accordance with ASTM E84 or UL 723.
F.	Accessories: Provide components for each penetration firestopping system that are needed to install fill materials
	and to maintain ratings required. Use only those components specified by penetration firestopping system
	manufacturer and approved by qualified testing and inspecting agency for conditions indicated.
PART 3 - I	XECUTION
3.1	INSTALLATION OF PENETRATION FIRESTOPPING SYSTEMS
Α.	Examine substrates and conditions, with Installer present, for compliance with requirements for opening
_	configurations, penetrating items, substrates, and other conditions affecting performance of the Work.
В.	General: Install penetration firestopping systems to comply with manufacturer's written installation instructions
-	and published drawings for products and applications.
C.	Install forming materials and other accessories of types required to support fill materials during their application
	and in the position needed to produce cross-sectional shapes and depths required to achieve fire ratings.
	1. After installing fill materials and allowing them to fully cure, remove combustible forming materials and
_	other accessories not forming permanent components of firestopping.
D.	Install fill materials by proven techniques to produce the following results:
	1. Fill voids and cavities formed by openings, forming materials, accessories and penetrating items to achieve
	required fire-resistance ratings.
	2. Apply materials so they contact and adhere to substrates formed by openings and penetrating items.
	3. For fill materials that will remain exposed after completing the Work, finish to produce smooth, uniform
	surfaces that are flush with adjoining finishes.
3.2	IDENTIFICATION
А.	waii identification: Permanently label waiis containing penetration firestopping systems with the words "FIRE
	AND/OK SIVIOKE BAKKIEK - PROTECT ALL OPEININGS, USING JETTERING NOT JESS THAN 3 INCHES HIGH AND WITH MINIMUM
	U.3/5-INCN STROKES.
	1. Locate in accessible concealed noor, noor-ceiling, or attic space at 15 ft. from end of wall and at intervals

1 **SECTION 07 84 43** 2 JOINT FIRESTOPPING 3 PART 1 - GENERAL 4 1.1 SUMMARY 5 Α. Section Includes: 6 Joints in or between fire-resistance-rated construction. 1. 7 2. Joints at exterior curtain-wall/floor intersections. 8 3. Joints in smoke barriers. 9 PREINSTALLATION MEETINGS 1.2 10 Preinstallation Conference: Conduct conference at Project site. Α. **ACTION SUBMITTALS** 11 1.3 12 Product data. Α. 13 В. Product Schedule: For each joint firestopping system. Include location, illustration of firestopping system, and 14 design designation of qualified testing agency. 15 Engineering Judgments: Where Project conditions require modification to a qualified testing agency's 1. 16 illustration for a particular joint firestopping system condition, submit illustration, with modifications 17 marked, approved by joint firestopping system manufacturer's fire-protection engineer as an EJ or 18 equivalent fire-resistance-rated assembly developed in accordance with current IFC guidelines. 19 **INFORMATIONAL SUBMITTALS** 1.4 20 Α. Qualification Data: For Installer. 21 Listed System Designs: For each joint firestopping system, for tests performed by a qualified testing agency. Β. 22 **CLOSEOUT SUBMITTALS** 1.5 23 Installer Certificates: From Installer indicating that joint firestopping systems have been installed in compliance with Α. 24 requirements and manufacturer's written installation instructions. 25 QUALITY ASSURANCE 1.6 Installer Qualifications: A firm that has been approved by FM Approvals in accordance with FM Approvals 4991 or 26 Α. 27 been evaluated by UL and found to comply with UL's "UL Solutions Qualified Firestop Contractor Program." 28 PART 2 - PRODUCTS 29 PERFORMANCE REQUIREMENTS 2.1 30 Α. Fire-Test-Response Characteristics: 31 1. A qualified testing agency, acceptable to authorities having jurisdiction, will perform joint firestopping 32 system tests. 33 2. Test in accordance with testing standards referenced in "Joint Firestopping Systems" Article. Provide rated 34 systems complying with the following requirements: Joint firestop systems installed with products bearing the classification marking of a qualified 35 а. product certification agency in accordance with listed system designs published by a qualified 36 37 testing agency. UL in its online directory "Product iQ." 38 1) 39 2) Intertek Group in its "Directory of Building Products." 40 Rain/Water Resistance: For perimeter fire-barrier system applications, where inclement weather or greater-than-Β. 41 transient water exposure is expected, use products that dry rapidly and cure in the presence of atmospheric moisture sufficient to pass ASTM D6904 early rain-resistance test (24-hour exposure). 42 43 2.2 JOINT FIRESTOPPING SYSTEM TYPES 44 Α. General: Systems that resist spread of fire, passage of smoke and other gases, and maintain original fire-resistance 45 rating of assemblies in or between which joint firestopping systems are installed. Joint firestopping systems must 46 accommodate building movements without impairing their ability to resist the passage of fire and hot gases. 47 1. Joint firestopping systems that are compatible with one another, with the substrates forming openings, and 48 with penetrating items, if any. 49 2. Provide products that, upon curing, do not re-emulsify, dissolve, leach, break down, or otherwise 50 deteriorate over time from exposure to atmospheric moisture, sweating pipes, ponding water or other 51 forms of moisture. Provide firestop products that do not contain ethylene glycol. 52 3. 53 Β. Joints in or between Fire-Resistance-Rated Construction: Provide joint firestopping systems with ratings determined in accordance with ASTM E1966 or UL 2079, with published L-Ratings for ambient and elevated temperatures as 54 55 evidence of the ability of the fire-resistive joint system to restrict the movement of smoke. 56 Manufacturers: Subject to compliance with requirements, provide products by one of the following: 1. 57 a. 3M Building and Construction. 58 b. Hilti, Inc.

	May 16	, 2024
59		c. Tremco Incorporated.
60		2. Fire-Resistance Rating: Equal to or exceeding the fire-resistance rating of the wall. floor, or roof in or
61		between which it is installed.
62	C.	Joints at Exterior Curtain-Wall/Floor Intersections: Provide joint firestopping systems with rating determined in
63	0.	accordance with ASTM E2307.
64		1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
65		a 3M Building and Construction
66		h Hilti Inc
67		c Tremco Incorporated
68		2 E-Rating: Figual to or exceeding the fire-resistance rating of the floor assembly
69	П	Joints in Smoke Barriers: Provide joint firestonning systems with ratings determined in accordance with 111 2029
70	D.	hased on testing at a nositive pressure differential of 0.30 inch wg
70		Manufacturers: Subject to compliance with requirements, provide products by one of the following:
72		a a 3M Building and Construction
72		h Hilti Inc
73		c. Tremco Incorporated
74		L_R Label and the second polarized.
75	E	2. E-Maing, Not exceeding 3.0 cm/nt. or joint at both amplent and elevated temperatures.
70	с.	exposed joint filestopping systems. Frame-spread and smoke-developed indexes of less than 25 and 450,
77	, ,	
70	2.5	ACCESSORIES Drouide components of joint firestanning systems, including primers and forming materials, that are needed to
79	А.	Fronce components of joint mestopping systems, including primers and forming materials, that are needed to
80		firstenning system manufacturer and approved by the qualified testing against for components specified by joint
02		mestopping system manufacturer and approved by the quaimed testing agency for conditions multated.
82 82	PARI 5 -	
83 84	5.1	PREPARATION Surface Cleaning: Defere installing joint firestenning surtance clean joints in accordance with fire resistive joint
84 05	А.	Surface Cleaning. Before installing joint mestopping systems, clean joints in accordance with me-resistive joint
85		system manufacturer's written installation instructions and the following requirements:
86		1. Remove foreign materials from substrate surfaces that could interfere with adhesion of elastomeric fill
87		materials or compromise fire-resistive rating.
88		2. Clean joint substrates to produce clean, sound surfaces capable of developing optimum bond with
89		elastomeric fill materials. Remove loose particles remaining from cleaning operation.
90	-	3. Remove laitance and form-release agents from concrete.
91	В.	Prime substrates in accordance with joint firestopping system manufacturer's written installation instructions, using
92		that manufacturer's recommended products and methods. Confine primers to areas of bond; do not allow spillage
93	<u> </u>	and migration onto exposed surfaces.
94	C.	Apply a suitable bond breaker to prevent three-sided adhesion in applications where condition occurs.
95	3.2	INSTALLATION
96	Α.	General: Install joint firestopping systems in accordance with manufacturer's written installation instructions and
97	_	published drawings for products and applications indicated.
98	В.	Install forming materials and other accessories of types required to support elastomeric fill materials during their
99		application and in position needed to produce cross-sectional shapes and depths required to achieve fire ratings
100		indicated.
101		1. After installing elastomeric fill materials and allowing them to fully cure, remove combustible forming
102		materials and other accessories not indicated as permanent components of fire-resistive joint system.
103	C.	Install elastomeric fill materials for joint firestopping systems by proven techniques to produce the following
104		results:
105		1. Apply elastomeric fill in voids and cavities formed by joints and forming materials as required to achieve
106		fire-resistance ratings indicated.
107		2. Apply elastomeric fill materials so they contact and adhere to substrates formed by joints.
108		3. For elastomeric fill materials that will remain exposed after completing the Work, finish to produce smooth,
109		uniform surfaces that are flush with adjoining finishes.
110	3.3	IDENTIFICATION
111	Α.	Wall Identification: Permanently label walls containing firestopping systems with the words "FIRE AND/OR SMOKE
112		BARRIER - PROTECT ALL OPENINGS," using lettering not less than 3 inches high and with minimum 0.375-inch
113		strokes.
114		1. Locate in accessible concealed floor, floor-ceiling, or attic space at 15 ft. from end of wall and at intervals
115		not exceeding 30 ft

116	3.4	CLEANING AND PROTECTION
117	Α.	Clean off excess elastomeric fill materials adjacent to joints as the Work progresses by methods and with cleaning
118		materials that are approved in writing by joint firestopping system manufacturers and that do not damage materials
119		in which joints occur.
120	В.	Provide final protection and maintain conditions during and after installation that ensure joint firestopping systems
121		are without damage or deterioration at time of Substantial Completion. If damage or deterioration occurs despite
122		such protection, cut out and remove damaged or deteriorated joint firestopping systems immediately and install
123		new materials to produce joint firestopping systems complying with specified requirements.
124		END OF SECTION

Page Intentionally Left Blank

SECTION 07 92 00

2		
л Л	1 1	
4 C	1.1	
5	А.	Section includes.
0		1. Silicolle joint sealants.
/		2. Nonstalling sincore joint sealants.
8		3. Uretnane joint sealants.
9		4. Immersible joint sealants.
10		5. Mildew-resistant joint sealants.
11		6. Latex joint sealants.
12	1.2	PREINSTALLATION MEETINGS
13	Α.	Preinstallation Conference: Conduct conference at Project site.
14	1.3	ACTION SUBMITTALS
15	Α.	Product data.
16	В.	Samples: Manufacturer's standard color charts consisting of strips of cured sealants showing the full range of colors
17		available for each product exposed to view.
18	С.	Joint-sealant schedule.
19	1.4	INFORMATIONAL SUBMITTALS
20	Α.	Field Quality-Control Reports: For field-adhesion-test reports, for each sealant application tested.
21	В.	Sample warranties.
22	1.5	CLOSEOUT SUBMITTALS
23	Α.	Warranty Documentation:
24		1. Manufacturers' special warranties.
25		2. Installer's special warranties.
26	1.6	QUALITY ASSURANCE
27	Α.	Installer Qualifications: Authorized representative who is trained and approved by manufacturer.
28	В.	Testing Agency Qualifications: Qualified in accordance with ASTM C1021 to conduct the testing indicated.
29	1.7	WARRANTY
30	Α.	Special Installer's Warranty: Installer agrees to repair or replace joint sealants that do not comply with performance
31		and other requirements specified in this Section within specified warranty period.
32		1. Warranty Period: Two years from date of Substantial Completion.
33	В.	Special Manufacturer's Warranty: Manufacturer agrees to furnish joint sealants to repair or replace those joint
34		sealants that do not comply with performance and other requirements specified in this Section within specified
35		warranty period.
36		1. Warranty Period: Five years from date of Substantial Completion.
37	С.	Special warranties specified in this article exclude deterioration or failure of joint sealants from the following:
38		1. Movement of the structure caused by stresses on the sealant exceeding sealant manufacturer's written
39		specifications for sealant elongation and compression.
40		2. Disintegration of joint substrates from causes exceeding design specifications.
41		3. Mechanical damage caused by individuals, tools, or other outside agents.
42		4. Changes in sealant appearance caused by accumulation of dirt or other atmospheric contaminants.
43	PART 2 -	PRODUCTS
44	2.1	JOINT SEALANTS, GENERAL
45	Α.	Compatibility: Provide joint sealants, backings, and other related materials that are compatible with one another
46		and with joint substrates under conditions of service and application, as demonstrated by joint-sealant
47		manufacturer, based on testing and field experience.
48	В.	Colors of Exposed Joint Sealants: As selected by Architect from manufacturer's full range.
49	2.2	NONSTAINING SILICONE JOINT SEALANTS
50	Α.	Nonstaining Joint Sealants: No staining of substrates when tested in accordance with ASTM C1248.
51	В.	Silicone, Nonstaining, S, NS, 50, NT: Nonstaining, single-component, nonsag, plus 50 percent and minus 50 percent
52		movement capability, nontraffic-use, neutral-curing silicone joint sealant; ASTM C920, Type S, Grade NS, Class 50,
53		Use NT.
54		1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
55		a. GE Construction Sealants; Momentive Performance Materials Inc.
56		b. Sika Corporation - Building Components.
57		c. The Dow Chemical Company.
58		d. Tremco Incorporated.
		·

2.3	UKEIMANE JUINI SEALANIS
А.	nontraffic-use, urethane joint sealant; ASTM C920, Type S, Grade NS, Class 25, Use NT.
	1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
	a. PPG Paints; PPG Industries, Inc.
	b. Sherwin-Williams Company (The).
	c. Sika Corporation - Building Components.
	d. Tremco Incorporated.
В.	Urethane, S, P, 25, T, NT: Single-component, pourable, plus 25 percent and minus 25 percent movement capability,
	traffic- and nontraffic-use, urethane joint sealant; ASTM C920, Type S, Grade P, Class 25, Uses T and NT.
	1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
	a. Master Builders Solutions; brand of MBCC Group.
	b. Pecora Corporation.
	c. Sherwin-Williams Company (The).
С.	Urethane, M, P, 50, T, NT: Multicomponent, pourable, plus 50 percent and minus 50 percent movement capability,
	traffic- and nontraffic-use, urethane joint sealant; ASTM C920, Type M, Grade P, Class 50, Uses T and NT.
	1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that
	may be incorporated into the Work include, but are not limited to the following:
2.4	a. Lymiai international, inc.
2.4	IVIVIERSIBLE JUINT SEALANTS
А.	atherwise indicated
р	United wise inducated Urothang Immercials S. R. 25, T. N.T. I: Immercials single component neurable plus 25 percent and minus 25
Б.	nercent movement capability traffic, and nontraffic-use urethane joint sealant: ASTM C020. Type S. Grade P.
	Class 25 Lises T NT and I
	1 Manufacturers: Subject to compliance with requirements, provide products by one of the following:
	a Tremco Incorporated
	b W.R. Meadows Inc
2.5	MILDEW-RESISTANT JOINT SEALANTS
	Mildew-Resistant Joint Sealants: Formulated for prolonged exposure to humidity with fungicide to prevent mold
	and mildew growth.
В.	Silicone, Mildew Resistant, Acid Curing, S, NS, 25, NT: Mildew-resistant, single-component, nonsag, plus 25 percent
	and minus 25 percent movement capability, nontraffic-use, acid-curing silicone joint sealant; ASTM C920, Type S,
	Grade NS, Class 25, Use NT.
	1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
	a. GE Construction Sealants; Momentive Performance Materials Inc.
	b. PPG Paints; PPG Industries, Inc.
	c. Sika Corporation - Building Components.
	d. The Dow Chemical Company.
	e. Tremco Incorporated.
2.6	LATEX JOINT SEALANTS
Α.	Acrylic Latex: Acrylic latex or siliconized acrylic latex, ASTM C834, Type OP, Grade NF.
	1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
	a. PPG Paints; PPG Industries, Inc.
	b. Sherwin-Williams Company (The).
	c. Tremco Incorporated.
2.7	JOINT-SEALANT BACKING
Α.	Sealant Backing Material, General: Nonstaining; compatible with joint substrates, sealants, primers, and other joint
	fillers; and approved for applications indicated by sealant manufacturer based on field experience and laboratory
	testing.
В.	Cylindrical Sealant Backings: ASTM C1330, Type C (closed-cell material with a surface skin), Type O (open-cell
	material), Type B (bicellular material with a surface skin), or any of the preceding types, as approved in writing by
	joint-sealant manufacturer for joint application indicated, and of size and density to control sealant depth and
	otherwise contribute to producing optimum sealant performance.
C.	Bond-Breaker Tape: Polyethylene tape or other plastic tape recommended by sealant manufacturer for preventing
	sealant from adhering to rigid, inflexible joint-filler materials or joint surfaces at back of joint. Provide self-adhesive

tape where applicable.

57

May 16 2024

	iviay 16,	2024
1	2.8	MISCELLANEOUS MATERIALS
2	Α.	Primer: Material recommended by joint-sealant manufacturer where required for adhesion of sealant to joint
3		substrates indicated, as determined from preconstruction joint-sealant-substrate tests and field tests.
4	В.	Cleaners for Nonporous Surfaces: Chemical cleaners acceptable to manufacturers of sealants and sealant backing
5		materials, free of oily residues or other substances capable of staining or harming joint substrates and adjacent
6		nonporous surfaces in any way, and formulated to promote optimum adhesion of sealants to joint substrates.
7	С.	Masking Tape: Nonstaining, nonabsorbent material compatible with joint sealants and surfaces adjacent to joints.
8	PART 3 - I	EXECUTION
9	3.1	PREPARATION
10	Α.	Surface Cleaning of Joints: Clean out joints immediately before installing joint sealants to comply with joint-sealant
11		manufacturer's written instructions and the following requirements:
12		1. Remove all foreign material from joint substrates that could interfere with adhesion of joint sealant,
13		including dust, paints (except for permanent, protective coatings tested and approved for sealant adhesion
14		and compatibility by sealant manufacturer), old joint sealants, oil, grease, waterproofing, water repellents,
15		water, surface dirt, and frost.
16		2. Clean porous joint substrate surfaces by brushing, grinding, mechanical abrading, or a combination of these
17		methods to produce a clean, sound substrate capable of developing optimum bond with joint sealants.
18		Remove loose particles remaining after cleaning operations above by vacuuming or blowing out joints with
19		oil-free compressed air. Porous joint substrates include the following:
20		a. Concrete.
21		b. Masonry.
22		c. Unglazed surfaces of ceramic tile.
23		d. Exterior insulation and finish systems.
24		3. Remove laitance and form-release agents from concrete.
25		4. Clean nonporous joint substrate surfaces with chemical cleaners or other means that do not stain, harm
26		substrates, or leave residues capable of interfering with adhesion of joint sealants. Nonporous joint
27		substrates include the following:
28		a. Metal.
29		b. Glass.
30		c. Porcelain enamel.
31		d. Glazed surfaces of ceramic tile.
32	В.	Joint Priming: Prime joint substrates where recommended by joint-sealant manufacturer or as indicated by
33		preconstruction joint-sealant-substrate tests or prior experience. Apply primer to comply with joint-sealant
34		manufacturer's written instructions. Confine primers to areas of joint-sealant bond; do not allow spillage or
35		migration onto adjoining surfaces.
36	С.	Masking Tape: Use masking tape where required to prevent contact of sealant or primer with adjoining surfaces
37		that otherwise would be permanently stained or damaged by such contact or by cleaning methods required to
38		remove sealant smears. Remove tape immediately after tooling without disturbing joint seal.
39	3.2	INSTALLATION OF JOINT SEALANTS
40	Α.	General: Comply with joint-sealant manufacturer's written installation instructions for products and applications
41		indicated, unless more stringent requirements apply.
42	В.	Sealant Installation Standard: Comply with recommendations in ASTM C1193 for use of joint sealants as applicable
43		to materials, applications, and conditions indicated.
44	С.	Install sealant backings of type indicated to support sealants during application and at position required to produce
45		cross-sectional shapes and depths of installed sealants relative to joint widths that allow optimum sealant
46		movement capability.
47		1. Do not leave gaps between ends of sealant backings.
48		2. Do not stretch, twist, puncture, or tear sealant backings.
49		3. Remove absorbent sealant backings that have become wet before sealant application, and replace them
50		with dry materials.
51	D.	Install bond-breaker tape behind sealants where sealant backings are not used between sealants and backs of
52		joints.
53	E.	Install sealants using proven techniques that comply with the following and at the same time backings are installed:
54		1. Place sealants so they directly contact and fully wet joint substrates.
55		2. Completely fill recesses in each joint configuration.
56		3. Produce uniform, cross-sectional shapes and depths relative to joint widths that allow optimum sealant
57		movement capability.

	May 16	, 2024						
-	F.	Tooling of Nonsag Sealants: Immediately after sealant application and before skinning or curing begins, tool sealants in accordance with requirements specified in subnaragraphs below to form smooth uniform beads of						
		configuration indicated; to eliminate air pockets; and to ensure contact and adhesion of sealant with sides of joint.						
		1. Remove excess sealant from surfaces adjacent to joints.						
		 Use tooling agents that are approved in writing by sealant manufacturer and that do not discolor sealants or adjacent surfaces. 						
		3. Provide concave joint profile in accordance with Figure 8A in ASTM C1193 unless otherwise indicated.						
	G.	Clean off excess sealant or sealant smears adjacent to joints as the Work progresses by methods and with cleaning						
		materials approved in writing by manufacturers of joint sealants and of products in which joints occur.						
	Н.	Protect joint sealants during and after curing period from contact with contaminating substances and from damage						
		resulting from construction operations or other causes so sealants are without deterioration or damage at time of						
		Substantial Completion. If, despite such protection, damage or deterioration occurs, cut out, remove, and repair						
		damaged or deteriorated joint sealants immediately so installations with repaired areas are indistinguishable from						
		original work.						
3	.3	JOINT-SEALANT SCHEDULE						
	Α.	Exterior joints in horizontal traffic surfaces:						
		1. Joint Locations:						
		a. Control and expansion joints in brick pavers.						
		b. Isolation and contraction joints in cast-in-place concrete slabs.						
		c. Joints between plant-precast architectural concrete paving units.						
		d. Joints in stone paving units, including steps.						
		e. Tile control and expansion joints.						
		f. Joints between different materials listed above.						
		2. Joint Sealant: Urethane, M, P, 50, T, NT.						
		3. Joint-Sealant Color: As selected by Architect from manufacturer's full range of colors.						
	В.	Exterior joints in horizontal traffic surfaces subject to water immersion:						
		1. Joint Locations:						
		a. Joints in pedestrian plazas.						
		b. Joints in swimming pool decks.						
		Joint Sealant: Urethane, immersible, S, P, 25, T, NT, I.						
		3. Joint-Sealant Color: As selected by Architect from manufacturer's full range of colors.						
	C.	Exterior joints in vertical surfaces and horizontal nontraffic surfaces:						
		1. Joint Locations:						
		a. Construction joints in cast-in-place concrete.						
		b. Joints between plant-precast architectural concrete units.						
		c. Control and expansion joints in unit masonry.						
		d. Joints in dimension stone cladding.						
		e. Joints in glass unit masonry assemblies.						
		f. Joints in exterior insulation and finish systems.						
		g. Joints between metal panels.						
		n. Joints between different materials listed above.						
		 Perimeter joints between materials listed above and trames of doors, windows, and louvers. Control and expansion isints in collings and other everband surfaces. 						
		J. Control and expansion joints in cenings and other overhead surfaces.						
		2. Joint Sediant: <u>Sincone, nonstaining, 5, N5, 50, N1</u> .						
	P	3. Joint-Sealant Color: As selected by Architect from manufacturer's full range of colors.						
	D.	Interior joints in horizontal trainc surfaces.						
		1. Julii Lucationis.						
		a. Isolation joints in cast-in-place concrete stabs.						
		c. Control and expansion joints in score nooring.						
		d Control and expansion joints in the flooring						
		2 Joint Sealant: Urethane S P 25 T NT						
		 Joint Sediant. Oleriane, S, F, ZS, T, NT. Joint Sediant Color: As selected by Architect from manufacturer's full range of colors. 						
	F	Interior joints in vertical surfaces and horizontal pontraffic surfaces.						
	E.	1 Ioint Locations:						
		Journe Locations Control and expansion joints on exposed interior surfaces of exterior walls						
		b Tile control and expansion joints						
		 Mercial joints on expansion joints. Vertical joints on exposed surfaces of unit masonny concrete walls and partitions. 						
		c. Vertical joints on exposed surfaces of unit masonry, concrete walls, and partitions.						

	IVIAY 10	2024	
1		d. Joints o	n underside of plant-precast structural concrete beams and planks.
2		2. Joint Sealant: L	Jrethane, S, NS, 25, NT.
3		3. Joint-Sealant Co	olor: As selected by Architect from manufacturer's full range of colors.
4	F.	Interior joints in vertica	I surfaces and horizontal nontraffic surfaces not subject to significant movement:
5		1. Joint Locations:	
6		a. Control	joints on exposed interior surfaces of exterior walls.
7		b. Perimet	ter joints between interior wall surfaces and frames of interior doors, windows, and elevator
8		entranc	ces.
9		2. Joint Sealant: <u></u>	Acrylic latex or siliconized acrylic latex.
10		3. Joint-Sealant Co	olor: As selected by Architect from manufacturer's full range of colors.
11	G.	Mildew-resistant interio	or joints in vertical surfaces and horizontal nontraffic surfaces:
12		1. Joint Locations:	
13		a. Joints b	etween plumbing fixtures and adjoining walls, floors, and counters.
14		b. Tile con	trol and expansion joints where indicated.
15		2. Joint Sealant: Sea	ilicone, mildew resistant, acid curing, S, NS, 25, NT.
16		3. Joint-Sealant Co	olor: As selected by Architect from manufacturer's full range of colors.
17	Н.	Concealed mastics:	
18		1. Joint Locations:	
19		a. Alumin	um thresholds.
20		b. Sill plat	es.
21		2. Joint Sealant: E	Butyl-rubber based.
22			END OF SECTION

Page Intentionally Left Blank

SECTION 07 92 19 ACOUSTICAL JOINT SEALANTS

2		ACOUSTICAL JOINT SEALANTS
3	PART 1 - 6	GENERAL
4	1.1	SUMMARY
5	Α.	Section Includes: Acoustical joint sealants.
6	1.2	ACTION SUBMITTALS
7	Α.	Product data.
8	B.	Samples: Manufacturer's color charts consisting of strips of cured sealants, showing full range of available colors for
9		each product exposed to view.
10	C.	Acoustical joint-sealant schedule
11	13	INFORMATIONAL SUBMITTALS
12	Δ	Test and Evaluation Reports:
13	7	1 Product test reports
1/	в	Sample warranties
15	14	
16	Δ	Warranty Documentation:
17	А.	Manufacturers' special warranties
10		Installar's special warranties.
10	1 5	2. Installer's special warranties.
20	1.5	WARNAN I
20	А.	installer's special warranty, installer agrees to repair of replace acoustical joint sealants that do not comply with
21		performance and other requirements specified in this section within specified warranty period.
22		
25	PARI 2 - P	
24 25	2.1	Accussingly is interested and the second transmission through norimeter is interest.
25	А.	Accusical joint-search products that enectively reduce an borne sound transmission through perimeter joints and
20		ACTM EQO
27	D	ASTIVIESU.
20	D.	Acoustical sealant for Exposed and Concealed Joints. Manuacturer's standard nonsag, paintable, nonstaining latex
29		Acoustical searant complying with ASTIN Cos4. Manufacturars: Subject to compliance with requirements, provide products by one of the following:
21		1. Manufacturers. Subject to compliance with requirements, provide products by one of the following.
31		d. milli, mc.
32		b. OSI Sealants; Henkel Corporation.
33		C. Pecora Corporation.
34 25		2. Colors of Exposed Acoustical Joint Sealants: As selected by Architect from manufacturer's full range of
35		
20	2.2	MISCELLANEOUS MATERIALS
3/	А.	Primer: Material recommended by acoustical joint-sealant manufacturer where required for aunesion of sealant to
38		Joint substrates.
39	В.	cleaners for Nonporous Surfaces. Chemical cleaners acceptable to manufacturers of sediants and sediant backing
40		materials, nee of only residues of other substances capable of staming of narming joint substrates and adjacent
41	6	nonporous surfaces in any way, and formulated to promote optimum adnesion of sealants to joint substrates.
42		Masking Tape: Nonstaining, nonabsorbent material compatible with joint sealants and surfaces adjacent to joints.
43	PARI 3 - E	
44	3.1	EXAMINATION
45	А.	examine joints indicated to receive acoustical joint sealants, with installer present, for compliance with
46		requirements for joint configuration, installation tolerances, and other conditions affecting performance of the
47		WORK.
48	В.	Proceed with installation only after unsatisfactory conditions have been corrected.
49	3.2	PREPARATION
50	А.	surface cleaning of joints: clean out joints immediately before installing acoustical joint sealants to comply with
51	_	Joint-sealant manufacturer's written instructions.
52 52	в.	Joint Priming: Prime joint substrates where recommended by acoustical joint-sealant manufacturer. Apply primer to
53		comply with joint-sealant manufacturer's written instructions. Confine primers to areas of joint-sealant bond; do
54	<u> </u>	not allow spillage or migration onto adjoining surfaces.
55	C.	iviasking Tape: Use masking tape where required to prevent contact of sealant or primer with adjoining surfaces
56		that otherwise would be permanently stained or damaged by such contact or by cleaning methods required to
5/		remove sealant smears. Remove tape immediately after tooling without disturbing joint seal.

1	3.3	INSTALLATION OF ACOUSTICAL JOINT SEALANTS
2	Α.	Comply with acoustical joint-sealant manufacturer's written installation instructions unless more stringent
3		requirements apply.
4	В.	STC-Rated Assemblies: Seal construction at perimeters, behind control joints, and at openings and penetrations
5		with a continuous bead of acoustical joint sealant. Install acoustical joint sealants at both faces of partitions, at
6		perimeters, and through penetrations. Comply with ASTM C919, ASTM C1193, and manufacturer's written
7		instructions for closing off sound-flanking paths around or through assemblies, including sealing partitions to
8		underside of floor slabs above acoustical ceilings.
9	С.	Acoustical Ceiling Areas: Apply acoustical joint sealant at perimeter edge moldings of acoustical ceiling areas in a
10		continuous ribbon concealed on back of vertical legs of moldings before they are installed.
11		END OF SECTION

2

SECTION 08 11 13 HOLLOW METAL DOORS AND FRAMES

3	PART 1 -	GENERAL				
4	1.1	SUMMARY				
5	Α.	Section Includes:				
6		1. Interior standard steel doors and frames.				
7		2. Exterior standard steel doors and frames.				
8	1.2	ACTION SUBMITTALS				
9	Α.	Product Data: For each type of product.				
10	В.	Shop Drawings: Include the following:				
11		1. Elevations of each door type.				
12		Details of doors, including vertical- and horizontal-edge details and metal thicknesses.				
13		 Frame details for each frame type, including dimensioned profiles and metal thicknesses. 				
14	C	Product Schedule: For hollow-metal doors and frames, prenared by or under the supervision of supplier, using same				
15	с.	reference numbers for details and openings as those on Drawings. Coordinate with final door hardware schedule				
16	13	INFORMATIONAL SUBMITTALS				
17	Δ	Product test renorts				
10	д. В	Field quality-control reports				
10	р. 1 л					
20	1.4	Record Documents: For fire rated doors, list of door numbers and applicable room name and number to which door				
20	А.	Record Documents. For me-rated doors, list of door numbers and applicable room name and number to which door				
21	4 5					
22	1.5	QUALITY ASSURANCE Fire Dated Dass Inserates Qualifications, lass atom for field quality, control inserations of fire roted dass according				
23	А.	Fire-Rated Door inspector Qualifications: inspector for field quality-control inspections of fire-rated door assemblies				
24		is to meet the qualifications set forth in NFPA 80, Section 5.2.3.1 and the following:				
25		1. Door and Hardware institute Fire and Egress Door Assembly inspector (FDA) certification.				
26	В.	Egress Door inspector Qualifications: inspector for field quality-control inspections of egress door assemblies is to				
27		meet the qualifications set forth in NFPA 101, Section 7.2.1.15.4 and the following:				
28		1. Door and Hardware Institute Fire and Egress Door Assembly Inspector (FDAI) certification.				
29	PARI 2 -					
30	2.1	MANUFACIUKERS				
31	А.	Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be				
32		incorporated into the Work include, but are not limited to the following:				
33		1. Ceco Door; AADG, Inc.; ASSA ABLOY.				
34		2. Curries, AADG, Inc.; ASSA ABLOY Group.				
35		3. LaForce, LLC.				
36	2.2	PERFORMANCE REQUIREMENTS				
37	Α.	Fire-Rated Door Assemblies: Assemblies complying with NFPA 80 that are listed and labeled by a qualified testing				
38		agency acceptable to authorities having jurisdiction for fire-protection ratings and temperature-rise limits indicated				
39		on Drawings, based on testing at positive pressure in accordance with NFPA 252 or UL 10C.				
40		1. Smoke- and Draft-Control Door Assemblies: Listed and labeled for smoke and draft control by a qualified				
41		testing agency acceptable to authorities having jurisdiction, based on testing in accordance with UL 1784				
42		and installed in compliance with NFPA 105.				
43		2. Temperature-Rise Limit: At vertical exit enclosures and exit passageways, provide doors that have a				
44		maximum transmitted temperature end point of not more than 450 deg F above ambient after 30 minutes				
45		of standard fire-test exposure.				
46	В.	Fire-Rated, Borrowed-Lite Assemblies: Assemblies complying with NFPA 80 and listed and labeled by a qualified				
47		testing agency acceptable to authorities having jurisdiction, for fire-protection ratings indicated, based on testing in				
48		accordance with NFPA 257 or UL 9.				
49	С.	Thermally Rated Door Assemblies: Provide door assemblies with U-factor of not more than 0.50 deg Btu/F x h x sq.				
50		ft. when tested in accordance with ASTM C1363 or ASTM E1423.				
51	2.3	INTERIOR STANDARD STEEL DOORS AND FRAMES				
52	Α.	Construct hollow-metal doors and frames to comply with standards indicated for materials, fabrication, hardware				
53		locations, hardware reinforcement, tolerances, and clearances, and as specified.				
54	В.	Heavy-Duty Doors and Frames: ANSI/SDI A250.8, Level 2; ANSI/SDI A250.4, Level B				
55		1. Doors:				
56		a. Type: As indicated in the Door and Frame Schedule on Drawings.				
57		b. Thickness: 1-3/4 inches.				
58		c. Face: Uncoated steel sheet, minimum thickness of 0.042 inch.				

	May 16	, 2024					
1		d. Edge Construction: Model 1, Full Flush.					
2		e. Core: Polystyrene.					
3		f. Fire-Rated Core: Manufacturer's standard vertical steel stiffener core for fire-rated and					
4		temperature-rise-rated doors.					
5		2. Frames:					
6		a. Materials: Metallic-coated steel sheet, minimum thickness of 0.053 inch.					
7		b. Sidelite and Transom Frames: Fabricated from same thickness material as adjacent door frame.					
8		c. Construction: Full profile welded.					
9	2.4	EXTERIOR STANDARD STEEL DOORS AND FRAMES					
10	А.	Construct hollow-metal doors and frames to comply with standards indicated for materials, fabrication, hardware					
11		locations, hardware reinforcement, tolerances, and clearances, and as specified.					
12	В.	Heavy-Duty Doors and Frames: ANSI/SDI A250.8, Level 2; ANSI/SDI A250.4, Level B.					
13		1. Doors:					
14		a. Type: As indicated in the Door and Frame Schedule on Drawings.					
15		b. Thickness: 1-3/4 inches.					
16		c. Face: Metallic-coated steel sheet, minimum thickness of 0.042 inch, with minimum A40 coating.					
17		d. Edge Construction: Model 1, Full Flush.					
18		e. Edge Bevel: Provide manufacturer's standard beveled or square edges.					
19		f. Top Edge Closures: Close top edges of doors with flush closures of same material as face sheets.					
20		Seal joints against water penetration.					
21		g. Bottom Edges: Close bottom edges of doors with end closures or channels of same material as face					
22		sheets. Provide weep-hole openings in bottoms of exterior doors to permit moisture to escape.					
23		h. Core: Polyisocyanurate.					
24		i. Fire-Rated Core: Manufacturer's standard vertical steel stiffener with insulation core for fire-rated					
25		doors.					
26		2. Frames:					
27		a. Materials: Metallic-coated steel sheet, minimum thickness of 0.053 inch, with minimum A40					
28		Coating.					
29	25	D. Construction: Full profile weided.					
3U 21	2.5	BURROWED LITES					
31 วา	A. D	Fabricate of uncoaled steel sheet, minimum thickness of 0.053 mch.					
52 33	ь. С	Construction. Face welded.					
55 24	C.	fabricated in sections due to shipping or handling limitations, provide alignment plates or angles at each joint					
54 25		radicated in sections due to shipping or handling limitations, provide alignment plates or angles at each joint,					
22 26	р	rapricated of metal of same or greater thickness as metal as frames.					
30 27	26	Provide countersunk, flat- or oval-nead exposed screws and bolts for exposed fasteners unless otherwise indicated.					
28 28	2.0						
20 20	А.	Tune: Anchors of minimum size and type required by applicable door and frame standard, and suitable for					
40 40		nerformance level indicated					
40 //1		2 Quantity: Minimum of three anchors per jamb, with one additional anchor for frames with no floor anchor					
41 //2		2. Quality. Willing of three anchors per jamb, with one additional anchor for names with no noor anchor. Provide one additional anchor for each 24 inches of frame height above 7 feet					
72 // 2		2 Postinstalled Expansion Anchor: Minimum 3/8-inch-diameter holts with expansion shields or inserts with					
ч <u>э</u> ЛЛ		manufacturer's standard nine snacer					
44 45	в	Floor Anchors: Provide floor anchors for each jamb and mullion that extends to floor					
45 46	C.	FIGUR ANCHORS, PROVIDE HOUR ANCHORS FOR EACH JAMP AND MULTION THAT EXCENDS TO FIGUR.					
40 47	с.	less than 2-inch height adjustment. Terminate bottom of frames at top of underlayment					
47 48	D	Material: ASTM A879/A879M Commercial Steel (CS) 047 coating designation: mill phosphatized					
49 49	υ.	1 For anchors built into exterior walls steel sheet complying with ASTM A1008/A1008M or					
50		ASTM A1011/A1011M: hot-dip galvanized in accordance with ASTM A153/A153M. Class B.					
51	2.7						
52		Cold-Rolled Steel Sheet: ASTM A1008/A1008M. Commercial Steel (CS). Type R: suitable for exposed applications					
53	В.	Hot-Rolled Steel Sheet: ASTM A1011/A1011M. Commercial Steel (CS). Type B: free of scale, nitting or surface					
54	2.	defects: nickled and oiled					
55	С.	Metallic-Coated Steel Sheet: ASTM A653/A653M. Commercial Steel (CS). Type B.					
56	D.	Inserts, Bolts, and Fasteners: Hot-dip galvanized in accordance with ASTM A153/A153M.					

	<u>May</u> 16	, 2024					
1 2 3	E.	Power-Actuated Fasteners in Concrete: Fastener system of type suitable for application indicated, fabricated from corrosion-resistant materials, with clips or other accessory devices for attaching hollow-metal frames of type indicated.					
4 5 6	F.	Mineral-Fiber Insulation: ASTM C665, Type I (blankets without membrane facing); consisting of fibers manufactured from slag or rock wool; with maximum flame-spread and smoke-developed indexes of 25 and 50, respectively; passing ASTM E136 for combustion characteristics.					
7 8	G. 28	Glazing: Comply with requirements in Section 08 80 00 "Glazing."					
9		Door Astragals: Provide overlapping astragal on one leaf of pairs of doors where required by NEPA 80 for fire-					
10 11		performance rating or where indicated. Extend minimum 3/4 inch beyond edge of door on which astragal is mounted or as required to comply with published listing of qualified testing agency.					
12	В.	Hollow-Metal Frames: Fabricate in one piece except where handling and shipping limitations require multiple					
15 14		metal of same or greater thickness as frames.					
15 16		1. Sidelite and Transom Bar Frames: Provide closed tubular members with no visible face seams or joints, fabricated from same material as door frame. Fasten members at crossings and to jambs by welding.					
17 18		 Provide countersunk, flat- or oval-head exposed screws and bolts for exposed fasteners unless otherwise indicated. 					
19 20		3. Door Silencers: Except on weather-stripped frames, drill stops to receive door silencers as follows. Keep holes clear during construction					
21		a. Single-Door Frames: Drill stop in strike jamb to receive three door silencers.					
22	C	b. Double-Door Frames: Drill stop in head jamb to receive two door silencers.					
23 24 25	L.	and electrical wiring; include cutouts, reinforcement, mortising, drilling, and tapping in accordance with					
25		ANSI/SULAZSU.6, the Door Hardware Schedule on Drawings, and templates.					
20		2. Comply with BHMA A156 115 for preparing hollow-metal doors and frames for hardware.					
27	D	Glazed Lites: Provide stops and moldings around glazed lites where indicated. Form corners of stops and moldings					
29	5.	with mitered hairline joints.					
30		1. Provide stops and moldings flush with face of door, and with beveled stops unless otherwise indicated.					
31		2. Multiple Glazed Lites: Provide fixed and removable stops and moldings so that each glazed lite is capable of					
32		being removed independently.					
33 24		3. Provide fixed frame moldings on outside of exterior and on secure side of interior doors and frames. Provide					
34 25		Coordinate rabbet width between fixed and removable stops with glazing and installation types indicated					
36		5. Provide stops for installation with countersunk flat- or oval-head machine screws spaced uniformly not					
37		more than 9 inches o.c. and not more than 2 inches o.c. from each corner.					
38	2.9	STEEL FINISHES					
39	А.	Prime Finish: Clean, pretreat, and apply manufacturer's standard primer.					
40		1. Shop Primer: Manufacturer's standard, fast-curing, lead- and chromate-free primer complying with					
41		ANSI/SDI A250.10; recommended by primer manufacturer for substrate; compatible with substrate and					
42		field-applied coatings despite prolonged exposure.					
43	PART 3 -	EXECUTION					
44	3.1	PREPARATION					
45 46	А.	Remove welded-in shipping spreaders installed at factory. Restore exposed finish by grinding, filling, and dressing, as required to make repaired area smooth. flush, and invisible on exposed faces. Touch up factory-applied finishes					
47		where spreaders are removed.					
48	В.	Drill and tap doors and frames to receive nontemplated, mortised, and surface-mounted door hardware.					
49	3.2	INSTALLATION					
50	Α.	Hollow-Metal Frames: Comply with ANSI/SDI A250.11.					
51		1. Set frames accurately in position; plumbed, aligned, and braced securely until permanent anchors are set.					
52		After wall construction is complete, remove temporary braces without damage to completed Work.					
53		a. Where frames are fabricated in sections, field splice at approved locations by welding face joint					
54		continuously; grind, fill, dress, and make splice smooth, flush, and invisible on exposed faces. Touch-					
55		up finishes.					
56		b. Install frames with removable stops located on secure side of opening.					
57		2. Fire-Rated Openings: Install frames in accordance with NFPA 80.					
58		3. Floor Anchors: Secure with postinstalled expansion anchors.					

	IVIAY 10	, 2024
1		a. Floor anchors may be set with power-actuated fasteners instead of postinstalled expansion anchors
2		if so indicated and approved on Shop Drawings.
3		4. Solidly pack mineral-fiber insulation inside frames.
4 5		5. Masonry Walls: Coordinate installation of frames to allow for solidly filling space between frames and masonry with grout or mortar
6		In-Diara Concrete or Masonny Construction: Secure frames in place with postinstalled expansion
7		o. In the contract of analogy of the and make most fluxes in place with positionated expansion
8		 Installation Tolerances: Adjust hollow-metal frames to the following tolerances:
9 10		a. Squareness: Plus or minus 1/16 inch, measured at door rabbet on a line 90 degrees from jamb perpendicular to frame head.
11		b. Alignment: Plus or minus 1/16 inch, measured at jambs on a horizontal line parallel to plane of wall.
12 13		c. Twist: Plus or minus 1/16 inch, measured at opposite face corners of jambs on parallel lines, and perpendicular to plane of wall
-0 14		 Plumbness: Plus or minus 1/16 inch measured at jambs at floor
15	в	Hollow-Metal Doors: Fits and adjust hollow-metal doors accurately in frames, within clearances specified below
16	Б.	1. Non-Fire-Rated Steel Doors: Comply with ANSI/SDI A250.8.
17		2. Fire-Rated Doors: Install doors with clearances in accordance with NFPA 80.
18		3. Smoke-Control Doors: Install doors in accordance with NFPA 105.
19	C.	Glazing: Comply with installation requirements in Section 08 80 00 "Glazing" and with hollow-metal manufacturer's
20		written instructions.
21	3.3	REPAIR
22	Α.	Prime-Coat Touchup: Immediately after erection, sand smooth rusted or damaged areas of prime coat and apply
23		touchup of compatible air-drying, rust-inhibitive primer.
24	В.	Metallic-Coated Surface Touchup: Clean abraded areas and repair with galvanizing repair paint in accordance with
25		manufacturer's written instructions.
26		END OF SECTION

SECTION 08 14 16

1		SECTION 08 14 16	
2		FLUSH WOOD DOORS	
3	PART 1 -	GENERAL	
4	1.1	SUMMARY	
5	А.	Section Includes:	
6		1. Five-ply flush wood veneer-faced doors for transparent finish.	
7		2. Fire-rated wood door frames.	
8		3. Factory priming and/or finishing flush wood doors and frames (if indicated in drawings).	
9		Factory fitting flush wood doors to frames and factory machining for hardware.	
LO	1.2	ACTION SUBMITTALS	
1	А.	Product Data: For each type of product, including the following:	
12		1. Door core materials and construction.	
13		2. Door edge construction	
14		3. Door face type and characteristics.	
L5		4. Door louvers.	
16		5. Door trim for openings.	
L7		6. Door frame construction.	
18		7. Factory-machining criteria.	
19	_	8. Factory-priming and/or finishing specifications.	
20	В.	Shop Drawings: Indicate location, size, and hand of each door; elevation of each type of door; construction details	
21		not covered in Product Data; and the following:	
22		1. Door schedule indicating door location, type, size, fire protection rating, and swing.	
23		2. Door elevations, dimension and locations of nardware, lite and louver cutouts, and glazing thicknesses.	
24		 Details of frame for each frame type, including dimensions and profile. Details of clustrical excession of a clustering for clustering for clustering and profile. 	
25		4. Details of electrical raceway and preparation for electrified hardware, access control systems, and security	
20		Systems.	
<u>//</u>		5. Dimensions and locations of blocking for nardware attachment.	
20		 Clearances and undercuts. Dequirements for vances matching. 	
29	C	7. Requirements for veneer matching.	
50	L.		
51 51	1.5	INFORMATIONAL SUDIVITIALS	
52 52	А.	Qualification Data. For about inspector.	
)) /		File-Rated Dool Inspector: Submit documentation of compliance with NEDA 101 Section 7.2.1.15 4	
94 95		2. Egress Door Inspector. Submit documentation of compliance with NFPA 101, Section 7.2.1.15.4.	
26	в	Field quality-control reports	
87	14	CLOSEDUIT SUBMITTALS	
28	Δ	CLUSEUUT SUDIVITITALS Record Documents: For fire rated doors, list of door numbers and applicable room name and number to which door	
20	Π.		
.0	1.5	QUALITY ASSURANCE	
.1	<u></u>	Manufacturer's Certification: Licensed participant in AWI's Quality Certification Program	
2	Р.	Fire-Rated Door Inspector Qualifications: Inspector for field quality-control inspections of fire-rated door assemblies	
3	5.	complies with gualifications set forth in NEPA 80. Section 5.2.3.1 and the following:	
4		1. DHI's Fire and Egress Door Assembly Inspector (FDAI) certification.	
5	C.	Egress Door Inspector Qualifications: Inspector for field quality-control inspections of egress door assemblies	
6	с.	complies with gualifications set forth in NEPA 101. Section 7.2.1.15.4 and the following:	
7		1 DHI's Fire and Egress Door Assembly Inspector (EDAI) certification	
.8	PART 2 -	PRODUCTS	
9	2.1	PERFORMANCE REQUIREMENTS	
50	A.	Fire-Rated Wood Door Assemblies: Assemblies complying with NFPA 80 that are listed and labeled by a gualified	
51		testing agency acceptable to authorities having jurisdiction, for fire-protection ratings indicated on Drawings, based	
52		on testing at positive pressure in accordance with UL 10C or NFPA 252.	
53		1. Oversize Fire-Rated Door Assemblies: For units exceeding sizes of tested assemblies. provide certification by	
54		a gualified testing agency that doors comply with standard construction requirements for tested and	
55		labeled fire-rated door assemblies except for size.	
56	В.	Smoke- and Draft-Control Door Assemblies: Listed and labeled for smoke and draft control by a qualified testing	
57		agency acceptable to authorities having jurisdiction, based on testing in accordance with UL 1784 and installed in	
58		compliance with NFPA 105.	
		•	

2.2	FLUS	н woo	D DOOR	S AND FRAMES, GENERAL
Α.	Quality Standard: In addition to requirements specified, comply with "Architectural Woodwork Standards."			
2.3	MANUFACTURERS			
Α.	Manufacturers: Subject to compliance with requirements, available manufacturers offering products that ma			
	incor	porated	l into the	e Work include, but are not limited to the following:
		a.	Orego	on Door.
		b.	Oshk	osh Door Company.
		с.	VT In	dustries, Inc.
2.4	SOLII	D-CORE,	, FIVE-PL	Y FLUSH WOOD VENEER-FACED DOORS FOR TRANSPARENT FINISH
Α.	Inter	ior Door	s, Solid-	Core Five-Ply Veneer-Faced:
	1.	Perfo	ormance	Grade by Location:
		a.	ANSI/	WDMA I.S. 1A Extra Heavy Duty: All locations but as noted for Standard Duty.
		b.	ANSI/	WDMA I.S. 1A Standard Duty: Closets (not including janitor's closets) and private toilets.
	2.	Archi	itectural	Woodwork Standards Grade: Premium.
	3.	Faces	s: Single-	ply wood veneer not less than 1/50 inch thick.
		a.	Speci	es: Select white birch.
		b.	Cut:	Plain sliced (flat sliced).
		с.	Matc	h between Veneer Leaves: Book match.
		d.	Asser	nbly of Veneer Leaves on Door Faces: Center-balance match.
		e.	Pair a	and Set Match: Provide for doors hung in same opening or separated only by mullions.
		f.	Room	n Match:
			1)	Match door faces within each separate room or area of building. Corridor-door faces do not
				need to match where they are separated by 10 feet or more.
			2)	Provide door faces of compatible color and grain within each separate room or area of
				building.
		g.	Bluep	print Match: Where indicated, provide doors with faces produced from same flitches as
			adjac	ent wood paneling and arranged to provide blueprint match with wood paneling. Comply with
			requi	rements in Section 06 42 16 "Flush Wood Paneling."
	4.	Ехро	sed Vert	ical Edges: Same species as faces or a compatible species - Architectural Woodwork Standards
		edge	Type A.	
		a.	Fire-F	Rated Single Doors: Provide edge construction with intumescent seals concealed by outer stile.
			Comp	ply with specified requirements for exposed vertical edges.
		b.	Fire-F	Rated Pairs of Doors:
			1)	Provide fire-retardant stiles that are listed and labeled for applications indicated without
				formed-steel edges and astragals. Provide stiles with concealed intumescent seals. Comply
				with specified requirements for exposed edges.
		с.	Mine	ral-Core Doors: At hinge stiles, provide laminated-edge construction with improved screw-
			holdi	ng capability and split resistance. Comply with specified requirements for exposed edges.
			1)	Screw-Holding Capability: 550 lbf in accordance with WDMA T.M. 10.
	5.	Core	for Non-	-Fire-Rated Doors:
		a.	ANSI	A208.1, Grade LD-1 particleboard.
			1)	Blocking: Provide wood blocking in particleboard-core doors as follows:
				a) 5-inch top-rail blocking, in doors indicated to have closers.
				b) 5-inch bottom-rail blocking, in exterior doors and doors indicated to have kick, mop,
				or armor plates.
				c) 5-inch midrail blocking, in doors indicated to have exit devices.
			2)	Provide doors with glued-wood-stave or WDMA I.S. 10 structural-composite-lumber cores
				instead of particleboard cores for doors scheduled to receive exit devices in
				Section 08 71 00 "Door Hardware."
	6.	Core	for Fire-	Rated Doors: As required to achieve fire-protection rating indicated on Drawings.
		a.	Block	ing for Mineral-Core Doors: Provide composite blocking with improved screw-holding
			capat	pility approved for use in doors of fire-protection ratings indicated on Drawings as follows:
			1)	5-inch top-rail blocking.
			2)	5-inch bottom-rail blocking, in doors indicated to have protection plates.
			3)	5-inch midrail blocking, in doors indicated to have armor plates.
	_	_	4)	5-inch midrail blocking, in doors indicated to have exit devices.
	7.	Cons	truction	: Five plies, hot-pressed bonded (vertical and horizontal edging is bonded to core), with entire
		unit a	abrasive	planed before veneering.

SPECIFICATION

	May 16	5, 2024		
1	2.5	LIGHT FRAMES AND LOUVERS		
2	Α.	Wood Beads for Light Openings in Wood Doors: Provide manufacturer's standard wood beads unless otherw		
3		indicated.		
4		1. Wood Species: Same species as door faces.		
5		2. Profile: Manufacturer's standard shape.		
6		3. At wood-core doors with 20-minute fire-protection ratings, provide wood beads and metal glazing clips		
7		approved for such use.		
8	В.	Wood-Veneered Beads for Light Openings in Fire-Rated Doors: Manufacturer's standard wood-veneered		
9		noncombustible beads matching veneer species of door faces and approved for use in doors of fire-protection		
10		rating indicated on Drawings. Include concealed metal glazing clips where required for opening size and fire-		
11		protection rating indicated.		
12	С.	Wood Louvers: Door manufacturer's standard solid-wood louvers unless otherwise indicated.		
13		1. Wood Species: Same species as door faces.		
14		2. Profile: Flat.		
15	D.	Louvers for Fire-Rated Doors: Metal louvers with fusible link and closing device, listed and labeled for use in doors		
16		with fire-protection rating of 1-1/2 hours and less.		
17		1. Metal and Finish: Hot-dip galvanized steel, 0.040 inch thick, factory primed for paint finish.		
18	2.6	FABRICATION		
19	Α.	Factory fit doors to suit frame-opening sizes indicated.		
20		1. Comply with clearance requirements of referenced quality standard for fitting unless otherwise indicated.		
21		Comply with NFPA 80 requirements for fire-rated doors.		
22	В.	Factory machine doors for hardware that is not surface applied.		
23		1. Locate hardware to comply with DHI-WDHS-3.		
24		2. Comply with final hardware schedules, door frame Shop Drawings, ANSI/BHMA-156.115-W, and hardware		
25		templates.		
26		3. Coordinate with hardware mortises in metal frames, to verify dimensions and alignment before factory		
27		machining.		
28		4. For doors scheduled to receive electrified locksets, provide factory-installed raceway and wiring to		
29		accommodate specified hardware.		
30		5. Metal Astragals: Factory machine astragals and formed-steel edges for hardware for pairs of fire-rated		
31	6	doors.		
32	C.	Openings: Factory cut and trim openings through doors.		
33		Light Openings: Trim openings with molarings of material and profile indicated.		
34 25		2. Glazing: Factory install glazing in doors indicated to be factory finished. Comply with applicable		
35 26		Pequirements in Section 08 80 00 Glazing.		
20 27	р	5. Louvers. Factory treat exterior deers with water repellent after fabrication has been completed but before		
20	D.	factory priming or finishing		
30		1 Elach top of outswinging doors with manufacturer's standard metal flaching		
40	27			
40 41	Δ	Doors for Onaque Finish: Factory prime faces all four edges edges of cutouts and mortises with one coat of wood		
42	7	nrimer specified in Section 09 91 23" Interior Painting "		
43	2.8	FACTORY FINISHING		
44		Comply with referenced quality standard for factory finishing.		
45		1. Complete fabrication, including fitting doors for openings and machining for hardware that is not surface		
46		applied, before finishing.		
47		2. Finish faces, all four edges, edges of cutouts, and mortises.		
48		3. Stains and fillers may be omitted on top and bottom edges, edges of cutouts, and mortises.		
49	В.	Factory finish doors where indicated in schedules or on Drawings as factory finished.		
50	С.	Transparent Finish:		
51		1. Architectural Woodwork Standards System-5, Varnish, Conversion.		
52		2. Staining: Match Architect's sample (existing doors).		
53		3. Sheen: Satin.		
54	PART 3 -	EXECUTION		
55	3.1	INSTALLATION		
56	Α.	Hardware: For installation, see Section 08 71 00 "Door Hardware."		
57	В.	Install doors (and frames where noted) to comply with manufacturer's written instructions and referenced quality		
58		standard, and as indicated.		

	May 16,	2024					
1	C.	Install frames level, plumb, true, and straight.					
2		1. Shim as required with concealed shims. Install level and plumb to a tolerance of 1/8 inch in 96 inches.					
3		Anchor frames to anchors or blocking built in or directly attached to substrates.					
4		a. Secure with countersunk, concealed fasteners and blind nailing.					
5		b. Use fine finishing nails for exposed fastening, countersunk and filled flush with woodwork.					
6		 For factory-finished items, use filler matching finish of items being installed. 					
7		Install fire-rated doors and frames in accordance with NFPA 80.					
8		4. Install smoke- and draft-control doors in accordance with NFPA 105.					
9	D.	Job-Fitted Doors:					
10		 Align and fit doors in frames with uniform clearances and bevels as indicated below. 					
11		a. Do not trim stiles and rails in excess of limits set by manufacturer or permitted for fire-rated doors.					
12		2. Machine doors for hardware.					
13		3. Seal edges of doors, edges of cutouts, and mortises after fitting and machining.					
14		Clearances:					
15		a. Provide 1/8 inch at heads, jambs, and between pairs of doors.					
16		b. Provide 1/8 inch from bottom of door to top of decorative floor finish or covering unless otherwise					
17		indicated on Drawings.					
18		c. Where threshold is shown or scheduled, provide 1/4 inch from bottom of door to top of threshold					
19		uniess otherwise indicated.					
20		a. Comply with NFPA 80 for fire-rated doors.					
21		5. Bevel non-fire-rated doors 1/8 inch in 2 inches at lock and ninge edges.					
22		b. Bevel fire-rated doors 1/8 inch in 2 inches at lock edge; trim stiles and rails only to extent permitted by					
23	E	Eactory Eithed Doors: Align in frames for uniform clearance at each edge					
24	L. F	Factory-Fitted Doors: Align in trames for Uniform clearance at each edge.					
25	32	ractory-rinished boors: Restore finish before installation if fitting or machining is required at Project site.					
27	Δ	Operation: Rehang or replace doors that do not swing or operate freely					
28	л. В	Einiched Doors: Replace doors that are damaged or that do not comply with requirements. Doors may be repaired					
29	υ.	or refinished if Work complies with requirements and shows no evidence of renair or refinishing					
30		END OF SECTION					

1 **SECTION 08 31 13** 2 ACCESS DOORS AND FRAMES 3 PART 1 - GENERAL 4 1.1 SUMMARY 5 Α. Section Includes: 6 Access doors and frames. 1. 7 Fire-rated access doors and frames. 2. 8 1.2 ACTION SUBMITTALS 9 Product Data: For each type of product. Α. 10 В. Samples: For each type of access door and frame and for each finish specified. 11 C. Product Schedule: For access doors and frames. 12 1.3 **CLOSEOUT SUBMITTALS** 13 Record Documents: For fire-rated doors, list of applicable room name and number in which access door is located. Α. 14 1.4 QUALITY ASSURANCE 15 Fire-Rated Door Inspector Qualifications: Inspector for field quality control inspections of fire-rated door assemblies Α. 16 meets the qualifications set forth in NFPA 80, Section 5.2.3.1 and the following: 17 1. Door and Hardware Institute Fire and Egress Door Assembly Inspector (FDAI) certification. 18 **PART 2 - PRODUCTS** 19 PERFORMANCE REQUIREMENTS 2.1 Fire-Rated Access Doors and Frames: Assemblies complying with NFPA 80 that are listed and labeled by a qualified 20 A. 21 testing agency, for fire-protection ratings indicated, according to NFPA 252 or UL 10B. 22 ACCESS DOORS AND FRAMES 2.2 23 Flush Access Doors with Concealed Flanges: Α. 24 1. Description: Face of door flush with frame, with exposed flange and concealed hinge. 25 2. Optional Features: Gasketing, piano hinges, and masonry anchors (where relevant). 26 3. Locations: Wall and ceiling. Metallic-Coated Steel Sheet for Door: Nominal 0.064 inch, 16 gage, factory primed. 27 4. 28 Frame Material: Same material, thickness, and finish as door. 5. 29 Latch and Lock: Cam latch, screwdriver operated for walls in non-public spaces and ceilings, and cam latch, 6. 30 hex-head wrench operated for walls with public access. 31 2.3 FIRE-RATED ACCESS DOORS AND FRAMES 32 A. Fire-Rated, Flush Access Doors with Concealed Flanges: 33 Description: Door face flush with frame, with a core of mineral-fiber insulation enclosed in sheet metal; with 1. 34 exposed flange, self-closing door, and concealed hinge. 35 2. Optional Features: Gasketing, piano hinges, and masonry anchors (where relevant). 36 3. Locations: Wall and ceiling. 37 4. Fire-Resistance Rating: Not less than that of adjacent construction. Temperature-Rise Rating: 450 deg F at the end of 30 minutes. 38 5. 39 6. Metallic-Coated Steel Sheet for Door: Nominal 0.040 inch, 20 gage, factory primed. 40 7. Frame Material: Same material, thickness, and finish as door. 41 Latch and Lock: Self-closing and self-latching door hardware, operated by key. 8. MATERIALS 42 2.4 43 Steel Plates, Shapes, and Bars: ASTM A36/A36M. Α. 44 Β. Steel Sheet: Uncoated or electrolytic zinc coated, ASTM A879/A879M, with cold-rolled steel sheet substrate 45 complying with ASTM A1008/A1008M, Commercial Steel (CS), exposed. 46 C. Metallic-Coated Steel Sheet: ASTM A653/A653M, Commercial Steel (CS), Type B; with minimum G60 or A60 metallic 47 coating. D. 48 Frame Anchors: Same material as door face. 49 Inserts, Bolts, and Anchor Fasteners: Hot-dip galvanized steel according to ASTM A153/A153M or ASTM F2329. Ε. 50 2.5 FABRICATION 51 Metal Surfaces: For metal surfaces exposed to view in the completed Work, provide materials with smooth, flat Α. 52 surfaces without blemishes. Do not use materials with exposed pitting, seam marks, roller marks, rolled trade 53 names, or roughness. Doors and Frames: Grind exposed welds smooth and flush with adjacent surfaces. Furnish mounting holes, 54 Β. 55 attachment devices and fasteners of type required to secure access doors to types of supports indicated. 56 C. Latch and Lock Hardware: 57 1. Quantity: Furnish number of latches and locks required to hold doors tightly closed. 58 2. Keys: Furnish two keys per lock and key all locks alike.

May 16, 2024		
1	2.6	FINISHES
2	Α.	Painted Finishes: Comply with coating manufacturer's written instructions for cleaning, conversion coating, and
3		applying and baking finish.
4		1. Factory Primed: Apply manufacturer's standard, lead- and chromate-free, universal primer immediately
5		after surface preparation and pretreatment.
6	PART 3 -	EXECUTION
7	3.1	INSTALLATION
8	Α.	Comply with manufacturer's written instructions for installing access doors and frames.
9	В.	Adjust doors and hardware, after installation, for proper operation.
10	3.2	FIELD QUALITY CONTROL
11	Α.	Inspection Agency: Engage a qualified inspector to perform inspections and to furnish reports to Architect.
12	В.	Inspections:
13		1. Fire-Rated Door Inspections: Inspect each fire-rated access door in accordance with NFPA 80, Section 5.2.
14	С.	Repair or remove and replace installations where inspections indicate that they do not comply with specified
15		requirements.
16	D.	Reinspect repaired or replaced installations to determine if replaced or repaired door assembly installations comply
17		with specified requirements.
18	Ε.	Prepare and submit separate inspection report for each fire-rated access door indicating compliance with each item
19		listed in NFPA 80 and NFPA 101.
20		END OF SECTION

С

SECTION 08 41 13 ALUMINUM-FRAMED ENTRANCES AND STOREFRONTS

3	PART 1 - (GENERAL
4	1.1	SUMMARY
5	Α.	Section Includes:
6		1. Aluminum-framed entrance and storefront systems.
7	1.2	PREINSTALLATION MEETINGS
8	Α.	Preinstallation Conference: Conduct conference at Project site.
9	1.3	ACTION SUBMITTALS
10	Α.	Product data.
11	В.	Shop Drawings:
12		1. Plans, elevations, sections, full-size details, and attachments to other work.
13		2. Connection to and continuity with adjacent thermal, weather, air, and vapor barriers.
14		3. Point-to-point wiring diagrams.
15	С.	Samples: Manufacturer's standard color sheets, showing full range of available colors for each type of exposed
16		finish.
17	D.	Entrance Door Hardware Schedule: Prepared by or under supervision of supplier, detailing fabrication and assembly
18		of entrance door hardware, as well as procedures and diagrams.
19	Ε.	Delegated Design Submittals: For aluminum-framed entrance and storefront systems, including analysis data signed
20		and sealed by the qualified professional engineer responsible for their preparation.
21	1.4	INFORMATIONAL SUBMITTALS
22	Α.	Energy Performance Certificates: NFRC-certified energy performance values from manufacturer.
23	В.	Product test reports.
24	С.	Source quality-control reports.
25	D.	Field quality-control reports.
26	Ε.	Sample warranties.
27	1.5	CLOSEOUT SUBMITTALS
28	Α.	Operation and maintenance data.
29	1.6	QUALITY ASSURANCE
30	А.	Installer Qualifications:
31		1. Fabricator of products.
32		2. Entity that employs installers and supervisors who are trained and approved by manufacturer.
33		3. Authorized representative who is trained and approved by manufacturer.
34		4. Entity that is certified under the North American Contractor Certification Program (NACC) and that employs
35		installers and supervisors who are trained and approved by manufacturer.
36	В.	Delegated Design Engineer Qualifications: A professional engineer who is legally qualified to practice in state where
37		Project is located and who is experienced in providing engineering services of the type indicated.
38	С.	Testing Agency Qualifications: Qualified in accordance with ASTM E699 for testing indicated and acceptable to
39		Owner and Architect.
40	D.	Egress Door Inspector Qualifications:
41		1. Inspector for field quality-control inspections of egress door assemblies to comply with qualifications set
42		forth in NFPA 101, Ch. / "Means of Egress," Section "Means of Egress Components," Article "Inspection of
43		Door Openings."
44		2. Inspector for field quality-control inspections of egress door assemblies to be certified under DHI's
45		certification program as a Fire and Egress Door Assembly Inspector (FDAI) or a Certified Fire and Egress Door
46	-	Assembly Inspector (CFDAI).
4/	E.	Product Options: Information on Drawings and in Specifications establishes requirements for aesthetic effects and
48		performance characteristics of assemblies. Aesthetic effects are indicated by dimensions, arrangements, alignment,
49		and profiles of components and assemblies as they relate to signifines, to one another, and to adjoining
50		construction.
51 51		1. Do not change intended aestnetic effects, as judged solely by Architect, except with Architect's approval. If
52	17	changes are proposed, submit comprehensive explanatory data to Architect for review.
55	1.7	WARRAINT
54 55	А.	and storefront systems that fail in materials or workmanship within specified warranty period
56		Marranty Period: Five years from date of Substantial Completion
50		1. Warranty renou, rive years non date of substantial completion.

May 1	b, 2024
В.	Special Finish Warranty, Factory-Applied Finishes: Standard form in which manufacturer agrees to repair finishes or replace aluminum that shows evidence of deterioration of baked-enamel, powder-coat, or organic finishes within
	specified warranty period.
C.	 Warranty Period: 10 years from date of Substantial Completion. Special Finish Warranty, Anodized Finishes: Standard form in which manufacturer agrees to repair finishes or
	replace aluminum that shows evidence of deterioration of anodized finishes within specified warranty period. Warranty Period: 10 years from date of Substantial Completion.
PART 2	PRODUCTS
2.1	PERFORMANCE REQUIREMENTS
Α.	Delegated Design: Engage a qualified professional engineer, as defined in Section 01 40 00 "Quality Requirements,"
R	General Performance: Comply with performance requirements specified as determined by testing of aluminum-
D.	framed entrance and storefront systems representing those indicated for this Project without failure due to
	defective manufacture, fabrication, installation, or other defects in construction.
	 Aluminum-framed entrance and storefront systems to withstand movements of supporting structure, including, but not limited to, twist, column shortening, long-term creep, and deflection from uniformly
	distributed and concentrated live loads.
	2. Failure also includes the following:
	a. Thermal stresses transferring to building structure.
	b. Glass breakage.
	c. Noise or vibration created by wind and thermal and structural movements.
	d. Loosening or weakening of fasteners, attachments, and other components.
	e. Failure of operating units.
С.	Structural Loads:
	1. Wind Loads: As indicated on Drawings.
	2. Other Design Loads: As indicated on Drawings.
D.	Deflection of Framing Members Supporting Glass: At design wind load, as follows:
	1. Deflection Normal to Wall Plane: Limited to 1/175 of clear span for spans of up to 13 feet 6 inches and to
	1/240 of clear span plus 1/4 inch for spans greater than 13 feet 6 inches.
	2. Deflection Parallel to Glazing Plane: Limited to amount not exceeding that which reduces glazing bite to less
	than 75 percent of design dimension and that which reduces edge clearance between framing members
	and glazing or other fixed components to less than 1/8 inch.
	a. Operable Units: Provide a minimum 1/16-inch clearance between framing members and operable
	Units.
E	5. Cantilevel Deflection. Limited to 2L/175 at unsupported cantilevels.
L.	1 When tested at positive and negative wind-load design pressures storefront assemblies including entrance
	doors do not evidence deflection exceeding specified limits
	2 When tested at 150 percent of positive and pegative wind-load design pressures, storefront assemblies
	including entrance doors and anchorage, do not evidence material failures, structural distress, or
	permanent deformation of main framing members exceeding 0.2 percent of span.
	 Test Durations: As required by design wind velocity, but not less than 10 seconds.
F.	Water Penetration under Static Pressure: Test in accordance with ASTM E331 as follows:
	1. No evidence of water penetration through fixed glazing and framing areas, including entrance doors, when
	tested in accordance with a minimum static-air-pressure differential of 20 percent of positive wind-load
	design pressure, but not less than 10 lbf/sq. ft
G.	Energy Performance: Certified and labeled by manufacturer for energy performance as follows:
	1. Thermal Transmittance (U-factor):
	a. Fixed Glazing and Framing Areas: U-factor for the system of not more than 0.41 Btu/sq. ft. x h x
	deg F as determined in accordance with NFRC 100.
	b. Entrance Doors: U-factor of not more than 0.68 Btu/sq. ft. x h x deg F as determined in accordance
	with NFRC 100.
	2. Solar Heat-Gain Coefficient (SHGC):
	a. Fixed Glazing and Framing Areas: SHGC for the system of not more than 0.26 as determined in
	accordance with NFRC 200.
	b. Entrance Doors: SHGC of not more than 0.22 as determined in accordance with NFRC 200.
	3. Air Leakage:

May 16	5, 2024
	a. Fixed Glazing and Framing Areas: Air leakage for the system of not more than 0.06 cfm/sq. ft. at a
	static-air-pressure differential of 6.24 lbf/sq. ft. when tested in accordance with ASTM E283.
	b. Entrance Doors: Air leakage of not more than 1.0 cfm/sq. ft. at a static-air-pressure differential of
	1.57 lbf/sq. ft
	4. Condensation Resistance Factor (CRF):
	a. Fixed Glazing and Framing Areas: CRF for the system of not less than 55 as determined in
	accordance with AAMA 1503.
	b. Entrance Doors: CRE of not less than 57 as determined in accordance with AAMA 1503.
H.	Windborne-Debris Impact Resistance: Passes ASTM F1886 missile-impact and cyclic-pressure tests in accordance
	with ASTM F1996 for Wind Zone 1 for basic protection
	1 Jarge-Missile Test: For glazing located within 30 feet of grade
	2 Small-Missile Test: For glazing located between 30 feet and 60 feet above grade
I.	Thermal Movements: Allow for thermal movements resulting from ambient and surface temperature changes
1.	1 Temperature Change: 120 deg E ambient: 180 deg E material surfaces
, ,	1. Infiniting change for an of the constraint of
2.2	ALOWINOW-FRANCE AND STOREFRONT STSTEWS
А.	The second s
	Free Corporation. Free Areasis Conservation
	2. Kawneer Company, Inc.; Arconic Corporation.
	3. Oldcastie BuildingEnvelope (OBE).
	4. Iruite Glass & Aluminum Solutions, LLC.
_	5. Iubelite inc.
В.	Basis-of-Design: Kawneer, Iri-Fab Versagiaze 60101 (2x6 framing).
С.	Framing Members: Manufacturer's extruded- or formed-aluminum framing members of thickness required and
	reinforced as required to support imposed loads.
	1. Exterior Framing Construction: Thermally broken.
	2. Interior Vestibule Framing Construction: Nonthermal.
	3. Glazing System: Retained mechanically with gaskets on four sides.
	4. Finish: Color anodic finish.
	5. Fabrication Method: Field-fabricated stick system.
	6. Aluminum: Alloy and temper recommended by manufacturer for type of use and finish indicated.
	7. Steel Reinforcement: As required by manufacturer.
D.	Backer Plates: Manufacturer's standard, continuous backer plates for framing members, if not integral, where
	framing abuts adjacent construction.
Ε.	Brackets and Reinforcements: Manufacturer's standard high-strength aluminum with nonstaining, nonferrous shims
	for aligning system components.
F.	Entrance Doors: Manufacturer's standard glazed entrance doors for manual-swing or automatic operation.
	1. Door Construction: 1-3/4-inch overall thickness, with minimum 0.125-inch-thick, extruded-aluminum
	tubular rail and stile members. Mechanically fasten corners with reinforcing brackets that are deeply
	penetrated and fillet welded or that incorporate concealed tie rods.
	a. Thermal Construction: High-performance plastic connectors separate aluminum members exposed
	to the exterior from members exposed to the interior.
	2. Door Design: As indicated.
	3. Glazing Stops and Gaskets: Beveled, snap-on, extruded-aluminum stops and preformed gaskets.
	a. Provide nonremovable glazing stops on outside of door.
2.3	ENTRANCE DOOR HARDWARE
Α.	Entrance Door Hardware: Hardware not specified in this Section is specified in Section 08 71 00 "Door Hardware."
В.	General: Provide entrance door hardware and entrance door hardware sets indicated in door and frame schedule
	for each entrance door, to comply with requirements in this Section.
	1. Opening-Force Requirements:
	a. Egress Doors: Not more than 15 lbf to release the latch and not more than 30 lbf to set the door in
	motion and not more than 15 lbf to open the door to its minimum required width
	h Accessible Interior Doors: Not more than 5 lbf to fully open door
2.4	GI AZING
-	Glazing: Comply with Section 08 80 00 "Glazing"
д. В	Glazing Gaskats: Manufacturer's standard sealed-corner pressure-dazing system of black resilient electomeric
D.	diazing daskets setting blocks and shims or snapers
C	Biazing Bashels, setting biolos, and simils UI spaces. Glazing Saslante: As recommended by manufacturer
L.	טומבוווד שבמומונש. אש דפרטווווופוועפע שע וומוועומרנערפו.

SPECIFICATION May 16, 2024

2.5	MATERIALS
Α.	Sheet and Plate: ASTM B209.
В.	Extruded Bars, Rods, Profiles, and Tubes: ASTM B221.
C.	Structural Profiles: ASTM B308/B308M.
D.	Steel Reinforcement:
	1. Structural Shapes, Plates, and Bars: ASTM A36/A36M.
	2. Cold-Rolled Sheet and Strip: ASTM A1008/A1008M.
	3. Hot-Rolled Sheet and Strip: ASTM A1011/A1011M.
E.	Steel Reinforcement Primer: Manufacturer's standard zinc-rich, corrosion-resistant primer complying with SSPC-
	PS Guide No. 12.00; applied immediately after surface preparation and pretreatment. Select surface preparation
	methods in accordance with recommendations in SSPC-SP COM, and prepare surfaces in accordance with applicable
	SSPC standard.
2.6	FABRICATION
Α.	Form or extrude aluminum shapes before finishing.
В.	Weld in concealed locations to greatest extent possible to minimize distortion or discoloration of finish. Remove
	weld spatter and welding oxides from exposed surfaces by descaling or grinding.
C.	Fabricate components that, when assembled, have the following characteristics:
	1. Profiles that are sharp, straight, and free of defects or deformations.
	2. Accurately fitted joints with ends coped or mitered.
	3. Physical and thermal isolation of glazing from framing members.
	4. Accommodations for thermal and mechanical movements of glazing and framing to maintain required
	glazing edge clearances.
	5. Provisions for field replacement of glazing from exterior.
	6. Fasteners, anchors, and connection devices that are concealed from view to greatest extent possible.
D.	Mechanically Glazed Framing Members: Fabricate for flush glazing without projecting stops.
Ε.	Entrance Door Frames: Reinforce as required to support loads imposed by door operation and for installing
	entrance door hardware.
F.	Entrance Doors: Reinforce doors as required for installing entrance door hardware.
G.	Entrance Door Hardware Installation: Factory install entrance door hardware to the greatest extent possible. Cut,
	drill, and tap for factory-installed entrance door hardware before applying finishes.
Н.	After fabrication, clearly mark components to identify their locations in Project in accordance with Shop Drawings.
2.7	ALUMINUM FINISHES
Α.	Color Anodic Finish: AAMA 611, AA-M12C22A42/A44, Class I, 0.018 mm or thicker.
	1. Color: As selected by Architect from full range of industry colors and color densities.
PART 3 - I	EXECUTION
3.1	INSTALLATION OF ALUMINUM-FRAMED ENTRANCE AND STOREFRONT SYSTEMS
Α.	Comply with manufacturer's written instructions.
В.	Do not install damaged components.
С.	Fit joints to produce hairline joints free of burrs and distortion.
D.	Rigidly secure nonmovement joints.
E.	Install anchors with separators and isolators to prevent metal corrosion and electrolytic deterioration and to
	prevent impeding movement of moving joints.
F.	Seal perimeter and other joints watertight unless otherwise indicated.
G.	Metal Protection:
	1. Where aluminum is in contact with dissimilar metals, protect against galvanic action by painting contact
	surfaces with materials recommended by manufacturer for this purpose or by installing nonconductive
	spacers.
	spacers. 2. Where aluminum is in contact with concrete or masonry, protect against corrosion by painting contact
	spacers.Where aluminum is in contact with concrete or masonry, protect against corrosion by painting contact surfaces with bituminous paint.
Н.	 spacers. Where aluminum is in contact with concrete or masonry, protect against corrosion by painting contact surfaces with bituminous paint. Set continuous sill members and flashing in full sealant bed, as specified in Section 07 92 00 "Joint Sealants," to
Н.	 spacers. Where aluminum is in contact with concrete or masonry, protect against corrosion by painting contact surfaces with bituminous paint. Set continuous sill members and flashing in full sealant bed, as specified in Section 07 92 00 "Joint Sealants," to produce weathertight installation.
Н. I.	 spacers. Where aluminum is in contact with concrete or masonry, protect against corrosion by painting contact surfaces with bituminous paint. Set continuous sill members and flashing in full sealant bed, as specified in Section 07 92 00 "Joint Sealants," to produce weathertight installation. Install joint filler behind sealant as recommended by sealant manufacturer.
Н. I. J.	 spacers. Where aluminum is in contact with concrete or masonry, protect against corrosion by painting contact surfaces with bituminous paint. Set continuous sill members and flashing in full sealant bed, as specified in Section 07 92 00 "Joint Sealants," to produce weathertight installation. Install joint filler behind sealant as recommended by sealant manufacturer. Install components plumb and true in alignment with established lines and grades.
H. I. J. K.	 spacers. Where aluminum is in contact with concrete or masonry, protect against corrosion by painting contact surfaces with bituminous paint. Set continuous sill members and flashing in full sealant bed, as specified in Section 07 92 00 "Joint Sealants," to produce weathertight installation. Install joint filler behind sealant as recommended by sealant manufacturer. Install components plumb and true in alignment with established lines and grades. Install entrance doors to produce smooth operation and tight fit at contact points.
Н. І. Ј. К.	 spacers. Where aluminum is in contact with concrete or masonry, protect against corrosion by painting contact surfaces with bituminous paint. Set continuous sill members and flashing in full sealant bed, as specified in Section 07 92 00 "Joint Sealants," to produce weathertight installation. Install joint filler behind sealant as recommended by sealant manufacturer. Install components plumb and true in alignment with established lines and grades. Install entrance doors to produce smooth operation and tight fit at contact points. Exterior Doors: Install to produce weathertight enclosure and tight fit at weather stripping.
Н. І. Ј. К.	 spacers. Where aluminum is in contact with concrete or masonry, protect against corrosion by painting contact surfaces with bituminous paint. Set continuous sill members and flashing in full sealant bed, as specified in Section 07 92 00 "Joint Sealants," to produce weathertight installation. Install joint filler behind sealant as recommended by sealant manufacturer. Install components plumb and true in alignment with established lines and grades. Install entrance doors to produce smooth operation and tight fit at contact points. Exterior Doors: Install to produce weathertight enclosure and tight fit at weather stripping. Field-Installed Entrance Door Hardware: Install surface-mounted entrance door hardware in accordance
Н. І. Ј. К.	 spacers. Where aluminum is in contact with concrete or masonry, protect against corrosion by painting contact surfaces with bituminous paint. Set continuous sill members and flashing in full sealant bed, as specified in Section 07 92 00 "Joint Sealants," to produce weathertight installation. Install joint filler behind sealant as recommended by sealant manufacturer. Install components plumb and true in alignment with established lines and grades. Install entrance doors to produce smooth operation and tight fit at contact points. Exterior Doors: Install to produce weathertight enclosure and tight fit at weather stripping. Field-Installed Entrance Door Hardware: Install surface-mounted entrance door hardware in accordance with entrance door hardware manufacturers' written instructions using concealed fasteners to greatest

L.	Install glazing as specified in Section 08 80 00 "Glazing."
3.2	FIELD QUALITY CONTROL
Α.	Testing Agency: Engage a qualified testing agency to perform tests.
В.	Tests: Perform the following test on representative areas of aluminum-framed entrance and storefront systems.
	1. Water-Spray Test: Before installation of interior finishes has begun, areas designated by Architect to be
	tested in accordance with AAMA 501.2 and to not evidence any water penetration.
	a. Perform a minimum of two tests in areas as directed by Architect.
	b. Ensure that spray area includes perimeter condition.
	2. Air Leakage: Follow AAMA 503 per ASTM E783 at 1.5 times the rate specified for laboratory testing in
	"Performance Requirements" Article but not more than 0.09 cfm/sq. ft. at a static-air-pressure differential
	of 1.57 lbf/sq. ft
	a. Perform a minimum of two tests in areas as directed by Architect.
	3. Water Penetration: Follow AAMA 503 per ASTM E1105 at a minimum uniform and cyclic static-air-pressure
	differential of 1.00 times the static-air-pressure differential specified for laboratory testing in "Performance
	Requirements" Article, but not less than 6.24 lbf/sq. ft., and to not evidence any water penetration.
С.	Duration of Tests: Run all timed tests to end, and do not stop upon first evidence of minor failure. Intent is to
	understand if failure is due to window system or intersection of window system with adjacent wall (i.e., perimeter
	condition).
D.	Aluminum-framed entrance and storefront systems will be considered defective if they do not pass tests and
	inspections.
Ε.	Prepare test and inspection reports.
	END OF SECTION
	L. 3.2 A. B. C. D. E.

Page Intentionally Left Blank
1

SECTION 08 44 13 GLAZED ALUMINUM CURTAIN WALLS

2		GLAZED ALUMINUM CURTAIN WALLS
3	PART 1 - 0	GENERAL
4	1.1	SUMMARY
5	Α.	Section Includes:
6		1. Glazed aluminum curtain wall systems.
7	1.2	PREINSTALLATION MEETINGS
8	Α.	Preinstallation Conference: Conduct conference at Project site.
9	1.3	ACTION SUBMITTALS
10	A.	Product Data: For each type of product.
11	В.	Shop Drawings: Include plans, elevations, sections, full-size details, and attachments to other work.
12		1. Show connection to and continuity with adjacent thermal, weather, air, and vapor barriers.
13	C.	Samples: For each type of exposed finish required
14	D.	Delegated-Design Submittal: For glazed aluminum curtain walls including analysis data signed and sealed by the
15	0.	qualified professional engineer responsible for their preparation
16	14	
17	1. 4 Δ	Energy Performance Certificates: NERC-certified energy performance values for each glazed aluminum curtain wall
10	R.	Droduct test reports
10	Б. С	Source quality control reports
20	с. D	Field quality control reports
20	D. F	Comple warrenties
21	L. 1 F	
22	1.5	CLOSEOUT SUBMITTALS
23	A.	
24	1.6	QUALITY ASSURANCE
25	А.	installer Qualifications: An entity that employs installers and supervisors who are trained and approved by
26		manufacturer and that employs a qualified glazing contractor for this Project who is certified under the North
27	_	American Contractor Certification Program (NACC) for Architectural Glass & Metal (AGIV) contractors.
28	В.	Testing Agency Qualifications: Qualified in accordance with ASTM E699 for testing indicated and acceptable to
29		Owner and Architect.
30	C.	Product Options: Information on Drawings and in Specifications establishes requirements for aesthetic effects and
31		performance characteristics of assemblies. Aesthetic effects are indicated by dimensions, arrangements, alignment,
32		and profiles of components and assemblies as they relate to sightlines, to one another, and to adjoining
33		construction.
34		1. Do not change intended aesthetic effects, as judged solely by Architect, except with Architect's approval. If
35		changes are proposed, submit comprehensive explanatory data to Architect for review.
36	1.7	WARRANTY
37	Α.	Special Assembly Warranty: Manufacturer agrees to repair or replace components of glazed aluminum curtain wall
38		that do not comply with requirements or that fail in materials or workmanship within specified warranty period.
39		 Warranty Period: Five years from date of Substantial Completion.
40	В.	Special Finish Warranty, Factory-Applied Finishes: Standard form in which manufacturer agrees to repair finishes or
41		replace aluminum that shows evidence of deterioration of baked enamel, powder coat, or organic finishes within
42		specified warranty period.
43		1. Warranty Period: 10 years from date of Substantial Completion.
44	С.	Special Finish Warranty, Anodized Finishes: Standard form in which manufacturer agrees to repair finishes or
45		replace aluminum that shows evidence of deterioration of anodized finishes within specified warranty period.
46		 Warranty Period: 10 years from date of Substantial Completion.
47	PART 2 - F	PRODUCTS
48	2.1	PERFORMANCE REQUIREMENTS
49	Α.	Delegated Design: Engage a qualified professional engineer, as defined in Section 01 40 00 "Quality Requirements,"
50		to design glazed aluminum curtain walls.
51	В.	General Performance: Comply with performance requirements specified, as determined by testing of glazed
52		aluminum curtain walls representing those indicated for this Project without failure due to defective manufacture,
53		fabrication, installation, or other defects in construction.
54		1. Glazed aluminum curtain walls shall withstand movements of supporting structure, including, but not
55		limited to, story drift, twist, column shortening, long-term creep, and deflection from uniformly distributed
56		and concentrated live loads.
57		2. Failure also includes the following:
58		a. Thermal stresses transferring to building structure.

	b. Glass breakage.	
	c Noise or vibration created by wind and thermal and structural movements	
	d Loosening or weakening of fasteners, attachments, and other components	
	e Failure of operating units	
C	Structural Loade:	
С.	1 Wind Loads: As indicated on Drawings	
	2 Other Design Loads: As indicated on Drawings.	
р	2. Other Design Loads. As indicated on Drawings. Deflection of Eraming Members Supporting Class: At design wind load as follows:	
D.	Deflection for Framing Members Supporting Glass. At design wind Ioda, as follows.	
	1/240 of cloar span plus 1/4 inch for spans of greater than 12 feet 6 inches	
	2 Deflection Parallel to Glazing Diane: Limited to amount not exceeding that which reduces also in the to be	
	2. Deficition randier to Glazing Flane. Limited to amount not exceeding that which reduces glazing bite to lest	
	and glazing or other fixed components to less than 1/8 inch	
	and glazing of other fixed components to less than 1/8 men.	
	a. Operable offics. Provide a minimum 1/10-inch clearance between maning members and operable	
	Captilover Deflection: Limited to 21/175 at uncurnerted captilovers	
F	S. Calcillevel Defection. Limited to 2/173 at disupported calcilevels.	
L.	When tested at positive and negative wind lead design proceures assemblies do not evidence deflection	
	1. When residual positive and negative wind-load design pressures, assemblies do not evidence denectio	
	2 When tested at 150 nercent of positive and negative wind-load design pressures assemblies includin	
	2. When resided at 150 percent of positive and negative wind-load design pressures, assembles, including approximation of mai	
	framing members exceeding 0.2 percent of span	
	Test Durations: As required by design wind velocity, but not less than 10 seconds	
F	Water Depatration under Static Pressure: Test in accordance with ASTM F331 as follows:	
1.	1 No evidence of water penetration through fixed glazing and framing areas when tested in accordance with	
	minimum static-air-pressure differential of 20 percent of positive wind-load design pressure but not les	
	than 10 lbf/cg ft	
G	Energy Derformance: Certified and labelled by manufacturer for energy performance as follows:	
О.	1 Thermal Transmittance (Il-factor):	
	a Fixed Glazing and Framing Areas: IL-factor for the system of not more than 0.29 Btu/sq. ft. x h	
	deg E as determined in accordance with NERC 100	
	2 Solar Heat Gain Coefficient (SHGC):	
	a Fixed Glazing and Framing Areas: SHGC for the system of not more than 0.26 as determined i	
	accordance with NERC 200	
	3 Air Leakage:	
	a Fixed Glazing and Framing Areas: Air leakage for the system of not more than 0.06 cfm/sg. ft. at	
	static-air-pressure differential of 1.57 lbf/sq. ft. when tested in accordance with ASTM F283	
	A Condensation Resistance Factor (CRE):	
	a Fixed Glazing and Framing Areas: CRE for the system of not less than 55 as determined i	
	a. The dialing and training Areas. ent for the system of hot less than 55 as determined f	
н	Windhorne-Dehris Impact Resistance: Pass ASTM F1886 missile-impact and cyclic-pressure tests in accordance wit	
	ASTM F1996 for Wind Zone 1 for basic protection	
	1. Large-Missile Test: For glazing located within 30 feet of grade.	
	 Small-Missile Test: For glazing located between 30 feet and 60 feet above grade 	
1	Thermal Movements: Allow for thermal movements resulting from amhient and surface temperature changes:	
	1 Temperature Change: 120 deg E ambient: 180 deg E material surfaces	
22	GLAZED ALLIMINI M CHRTAIN WALL SYSTEMS	
Δ	Manufacturers: Subject to compliance with requirements, provide products by one of the following:	
7	1 EFCO Corporation	
	2. Kawneer Company, Inc.: Arconic Corporation.	
	3 OldCastle BuildingEnvelope (OBE)	
	4 TRACO a division of Kawneer	
	5 Trulite Glass & Aluminum Solutions 11C	
	6. Tubelite Inc.	
	b. I UDEIITE INC. Pasis of Design: Kowpoor 1600 Wall System 1 (3):6 framing)	
в	Basis-of-Design: Kawneer 1600 Wall System 1 (2x6 framing)	
B. C	Basis-of-Design: Kawneer 1600 Wall System 1 (2x6 framing). Framing Members: Manufacturer's extruded, or formed-aluminum framing members of thickness required an	
В. С.	Basis-of-Design: Kawneer 1600 Wall System 1 (2x6 framing). Framing Members: Manufacturer's extruded- or formed-aluminum framing members of thickness required an reinforced as required to support imposed loads	

SPECIFICATION

May 16, 2024 1 2. Glazing System: Retained mechanically with gaskets on four sides. 2 3. Glazing Plane: Front. 3 4. Finish: Color anodic finish. 4 5. System: Either stick or unitized system. 5 6. Aluminum: Alloy and temper recommended by manufacturer for type of use and finish indicated. 6 Steel Reinforcement: As required by manufacturer. 7. 7 D. Pressure Caps: Manufacturer's standard aluminum components that mechanically retain glazing. 8 Include snap-on aluminum trim that conceals fasteners. 1. 9 E. Brackets and Reinforcements: Manufacturer's standard high-strength aluminum with nonstaining, nonferrous shims 10 for aligning system components. GLAZING 11 2.3 12 Glazing: Comply with Section 08 80 00 "Glazing." Α. 13 В. Glazing Gaskets: ASTM C509 or ASTM C864. Compression-type, replaceable EPDM. 14 Color: Black. 1. 15 C. Glazing Sealants: As recommended by manufacturer. 16 2.4 MATERIALS 17 Α. Sheet and Plate: ASTM B209. 18 Β. Extruded Bars, Rods, Profiles, and Tubes: ASTM B221. 19 Structural Profiles: ASTM B308/B308M. C. 20 D. Steel Reinforcement: 21 1. Structural Shapes, Plates, and Bars: ASTM A36/A36M. 22 2. Cold-Rolled Sheet and Strip: ASTM A1008/A1008M. 23 3. Hot-Rolled Sheet and Strip: ASTM A1011/A1011M. 24 Ε. Steel Reinforcement Primer: Manufacturer's standard zinc-rich, corrosion-resistant primer complying with SSPC-25 PS Guide No. 12.00; applied immediately after surface preparation and pretreatment. Select surface preparation methods in accordance with recommendations in SSPC-SP COM, and prepare surfaces in accordance with applicable 26 27 SSPC standard. 28 2.5 FABRICATION 29 Form or extrude aluminum shapes before finishing. Α. 30 Β. Weld in concealed locations to greatest extent possible to minimize distortion or discoloration of finish. Remove 31 weld spatter and welding oxides from exposed surfaces by descaling or grinding. 32 C. Fabricate components that, when assembled, have the following characteristics: 33 Profiles that are sharp, straight, and free of defects or deformations. 1. 34 2. Accurately fitted joints with ends coped or mitered. 35 3. Physical and thermal isolation of glazing from framing members. 36 4. Accommodations for thermal and mechanical movements of glazing and framing to maintain required 37 glazing edge clearances. 38 5. Provisions for field replacement of glazing from exterior. 39 6. Provisions for safety railings mounted between mullions at interior. 7. 40 Fasteners, anchors, and connection devices that are concealed from view to greatest extent possible. 41 Components curved to indicated radii. 8. 42 D. Fabricate components to resist water penetration as follows: 43 1. Internal guttering system or other means to drain water passing joints, condensation occurring within 44 framing members, and moisture migrating within glazed aluminum curtain wall to exterior. 45 2. Pressure-equalized system or double barrier design with primary air and vapor barrier at interior side of 46 glazed aluminum curtain wall and secondary seal weeped and vented to exterior. 47 Ε. Curtain-Wall Framing: Fabricate components for assembly using manufacturer's standard assembly method. 48 F. Factory-Assembled Frame Units: 49 Rigidly secure nonmovement joints. 1. 50 2. Prepare surfaces that are in contact with structural sealant in accordance with sealant manufacturer's 51 written instructions, to ensure compatibility and adhesion. 3. 52 Preparation includes, but is not limited to, cleaning and priming surfaces. 53 4. Seal joints watertight unless otherwise indicated. 54 Install glazing to comply with requirements in Section 08 80 00 "Glazing." 5. 55 G. After fabrication, clearly mark components to identify their locations in Project in accordance with Shop Drawings. 56 **ALUMINUM FINISHES** 2.6 57 Α. Color Anodic Finish: AAMA 611, AA-M12C22A42/A44, Class I, 0.018 mm or thicker. 58 Color: As selected by Architect from full range of industry colors and color densities. 1.

SPECIFICATION

	May 16, 2024			
1	PART 3 -	PART 3 - EXECUTION		
2	3.1	INSTALLATION, GENERAL		
3	Α.	Comply with manufacturer's written instructions.		
4	В.	Do not install damaged components.		
5	С.	Fit joints to produce hairline joints free of burrs and distortion.		
6	D.	Rigidly secure nonmovement joints.		
7	Ε.	Install anchors with separators and isolators to prevent metal corrosion and electrolytic deterioration and to		
;		prevent impeding movement of moving joints.		
)	F.	Where welding is required, weld components in concealed locations to minimize distortion or discoloration of		
)		finish. Protect glazing surfaces from welding.		
	G.	Seal joints watertight unless otherwise indicated.		
	Н.	Metal Protection:		
5		1. Where aluminum is in contact with dissimilar metals, protect against galvanic action by painting contact		
		surfaces with primer, applying sealant or tape, or installing nonconductive spacers as recommended by		
		manufacturer for this purpose.		
		2. Where aluminum is in contact with concrete or masonry, protect against corrosion by painting contact		
, ,		surfaces with bituminous paint.		
3	Ι.	Install components to drain water passing joints, condensation occurring within framing members, and moisture		
)		migrating within glazed aluminum curtain wall to exterior.		
)	J.	Install components plumb and true in alignment with established lines and grades.		
L	3.2	INSTALLATION OF GLAZING		
-	Α.	Install glazing as specified in Section 08 80 00 "Glazing."		
5	3.3	FIELD QUALITY CONTROL		
ł	Α.	Testing Agency: Engage a qualified testing agency to perform tests and inspections.		
5	В.	Test Area: Perform tests on representative areas of glazed aluminum curtain walls.		
5	С.	Field Quality-Control Testing: Perform the following test on representative areas of glazed aluminum curtain walls.		
/		1. Water-Spray Test: Before installation of interior finishes has begun, areas designated by Architect shall be		
3		tested in accordance with AAMA 501.2 and shall not evidence any water penetration.		
)		a. Perform a minimum of two tests in areas as directed by Architect.		
)		b. Ensure that spray area includes perimeter condition.		
-	D.	Duration of Tests: Run all timed tests to end, and do not stop upon first evidence of minor failure. Intent is to		
2		understand if failure is due to window system or intersection of window system with adjacent wall (i.e., perimeter		
3		<u>condition)</u> .		
1	Ε.	Glazed aluminum curtain walls will be considered defective if they do not pass tests and inspections.		
<i>.</i>	F.	Prepare test and inspection reports.		
ô		END OF SECTION		

1		SECTION 08 71 00	
2		DOOR HARDWARE	
3			
4	PART 1 – (GENERAL	1
5	1.1	SUMMARY	1
6	1.2	WORK FURNISHED BUT INSTALLED UNDER OTHER SECTIONS	1
7	1.3	RELATED WORK	1
8	1.4	REFERENCES	1
9	1.5	COORDINATION	2
10	1.6	QUALITY ASSURANCE	2
11	1.7	REGULATORY REQUIREMENTS	2
12	1.8	CERTIFICATIONS	2
13	1.9	SUBMITTALS	2
14	1.10	OPERATION AND MAINTENANCE DATA	2
15	1.11	DELIVERY, STORAGE, AND HANDLING	2
16	1.12	MAINTENANCE MATERIALS	3
17	1.13	WARRANTY	3
18	PART 2 - P	PRODUCTS	3
19	2.1	SOURCE LIMITATIONS	3
20	2.2	PERFORMANCE REQUIREMENTS	3
21	2.3	MATERIALS	3
22	2.4	KEYING	4
23	2.5	FABRICATION	4
24	2.6	FINISHES	4
25	PART 3 - E	EXECUTION	5
26	3.1	EXAMINATION	5
27	3.2	PREPARATION	5
28	3.3	INSTALLATION	5
29	3.4	INSPECTION AND ADJUSTMENTS	5
30	3.5	CLEANING AND PROTECTION	5
31	3.6	SCHEDULE	6
32			
33	PART 1 – 0	GENERAL	
34			
35	1.1	SUMMARY	
36	Α.	Section Includes:	
37		1. Commercial door hardware for aluminum entrances, wood and steel doors	
38		2. Coordination and interface with Owner's Security System.	
39		3. Templates	
40		4. Keying System	
41	1.2	WORK FURNISHED BUT INSTALLED UNDER OTHER SECTIONS	
42	Α.	Furnish all hardware to Section 06200 for installation.	
43	1.3	RELATED WORK	
44	Α.	Section 08 11 00 - Metal Doors and Frames: Doors and Frames prepared for finish hardware.	
45	В.	Section 08 21 00 - Wood Doors: Wood doors: Doors and Frames prepared for finish hardware.	
46	C.	Division 26 - Electrical Power supply to electric hardware devices and low voltage control	
47	1.4	REFERENCES	
48	Δ	ANSI A117.1 - Specifications for Making Buildings and Facilities Accessible to and Usable by Physically Handicanner	
49		Peonle	
50	в	ANSI/NEPA 80 - Fire Doors and Windows	
51	C.	BHMA - Builders' Hardware Manufacturers Association	
52	с. п	DHI - Door and Hardware Institute	
52	F.	NAAMM - National Association of Architectural Metal Manufacturers	
54	L. F	NFPA 101 - Life Safety Code	
55	۰. د	SDL- Steel Door Institute	
56	о. ц	ADA - The Americans with Disabilities Act	
50	п.	ADA - ME AMERICANS WITH DISADIMIES ACL.	

1	1.5	COORDINATION
2 A. Coordinate work of this Section with other directly affected Sections involving manufactu		Coordinate work of this Section with other directly affected Sections involving manufacturer of any internal
3		reinforcement for hardware.
4	В.	Furnish template for application to doors and jambs with machine or wood screws or through-bolts as required.
5		Templates or hardware or both shall be delivered to factory or building as required by those furnishing items to which
6		hardware is to be applied. Refer to other Sections of Specifications for this information. Locksets for metal doors
7		shall have beveled faces to correspond with bevel or doors. Strikes shall be ASA Standard dimension. Locks having
8		bolts or latches engaging with mullions or jambs of hollow metal shall have box type strikes with curved lips.
9	C.	Electrical System Roughing-In: Coordinate layout and installation of electrified door hardware with connections to
10		power supplies and building safety and security systems.
11	D.	Cooperate with installing contractor and others with regard to application of hardware. Make occasional inspections
12		to verify that items are properly used, installed, in correct location and master-key system is maintained. Report
13		improper application of hardware.
14	E.	Prior to preparing shop drawing submittal, contact Architect to set up a meeting with the Owner to review door
15		functions.
16	1.6	QUALITY ASSURANCE
17	А.	Manufacturers: Companies specializing in manufacturing door hardware with minimum three years experience.
18	В.	Hardware Supplier: Company specializing in supplying commercial and institutional door hardware with five years
19		experience.
20	C.	Each type of hardware shall be furnished from only one manufacturer.
21	D.	No hardware shall be attached to metal frames with self-tapping or sheet metal screws.
22	E.	Furnish thru bolts for fastening overhead holders and closers on composite core doors.
23	1.7	REGULATORY REQUIREMENTS
24	А.	Conform to NFPA Standard No. 80 requirements applicable to fire rated doors and frames.
25	В.	Provide only hardware which has been tested and listed by U.L. for types and sizes of doors required and complies
26		with requirements of door and door frame labels.
27	1.8	CERTIFICATIONS
28	Α.	Architectural Hardware Vendor shall inspect complete installation and certify that hardware and installation has been
29		furnished and installed in accordance with manufacturer's instructions and as specified herein.
30	В.	Provide two copies of certifications to Architect.
31	1.9	SUBMITTALS
32	Α.	Before ordering material, prepare and submit complete vertical hardware schedule for all hardware materials to
33		Architect for review. Approval of schedule does not relieve Contractor of any responsibility for furnishing material in
34		accordance with requirements of work.
35	В.	Schedule shall be specific and conclusive with respect to catalog numbers, finishes, template requirements, brackets,
36		type of fasteners and locations. Incomplete schedule will not be checked.
37	С.	Include in schedule, installation dimensions, hardware locations and mounting heights of each type of hardware.
38	D.	Prepare detailed keying schedule after obtaining Owner's instructions and requirements, and submit for approval.
39	E.	Samples, if requested shall be submitted to Architect for approval. Approved samples, if of proper finish, will be
40		delivered to job for ultimate use; otherwise samples will be returned to Contractor upon completion.
41	F.	Submit certificate that hardware to be furnished meets or exceed specified requirements.
42	G.	Submit catalog cut sheets describing all hardware items.
43	Н.	Coordinate with Electrical Contractor and Owner's Security Design Vendor installation requirements for locksets that
44		have electric strikes, door position switches and other line or low voltage requirements.
45	١.	Operation and Maintenance data.
46	1.10	OPERATION AND MAINTENANCE DATA
47	А.	Submit operation and maintenance data for all hardware items provided.
48	В.	Include data on operating hardware, lubrication requirements, and inspection procedures related to preventative
49		maintenance.
50	C.	Include one copy of final approved hardware and keying schedule.
51	1.11	DELIVERY, STORAGE, AND HANDLING
52	А.	Send duplicate list of hardware in each shipment to Contractor. Original list shall accompany shipment. Hardware
53		vendor shall pay shipping and delivery charges.
54	В.	Deliver hardware to Carpenter (Section 06200) or to respective shops of other Contractors as required. Consult with
55		named Contractors and follow their directions regarding manner, sequence and time of delivery and obtain receipt.
56	C.	Responsibility for safekeeping after delivery rests with trade to whom hardware was delivered.

1 2 2	D.	Hardware shall be sorted and delivered to jobsite plainly marked to correspond with item numbers of vendor's approved schedule and be specific as to exact openings and other locations for which items are packaged. Each door opening shall receive constraints item number on Hardware Schedule	
3 4 5	E.	Plainly mark packages of hardware so that locations of their use may be ascertained without breaking package. Where several packages are needed to complete schedule for one location, securely tie together or box.	
6	F.	Pack hardware by building area unless Contractor receiving hardware orders otherwise.	
7	G.	Hardware Supplier shall check all shipments to insure proper accessories and templates.	
8	Н.	Deliver hardware only after detailed schedule, keying diagrams, and samples have been approved.	
9	I.	Provide secure lock-up for hardware delivered to the project. Control handling and installation of hardware items	
10		which are not immediately replaceable so that the completion of the work will not be delayed by hardware losses.	
11	1.12	MAINTENANCE MATERIALS	
12	Α.	Provide three each special wrenches and tools applicable to each difference or special hardware component.	
13	В.	Provide three each maintenance tools and accessories supplied by hardware component manufacturer.	
14	1.13	WARRANTY Deside many factors developments	
15	А.	Provide manufacturer's standard warranty.	
15			
10	PARIZ-P	RODUCIS	
10			
20	2.1		
20	Δ	Obtain each type of door hardware from single manufacturer	
22	22	PERFORMANCE REGUIREMENTS	
23	<u> А</u> .	Fire-Rated Door Assemblies: Where fire-rated doors are indicated, provide door hardware complying with NEPA 80	
24		that is listed and labeled by a gualified testing agency, for fire-protection ratings indicated, based on testing at positive	
25		pressure in accordance with NFPA 252 or UL 10C.	
26	В.	Electrified Door Hardware: Listed and labeled as defined in NFPA 70, by a gualified testing agency, and marked for	
27		intended location and application.	
28	C.	Means of Egress Doors: Latches do not require more than 15 lbf (67 N) to release the latch. Locks do not require use	
29		of a key, tool, or special knowledge for operation.	
30	D.	Accessibility Requirements: For door hardware on doors in an accessible route, comply with ICC A117.1.	
31		1. Provide operating devices that do not require tight grasping, pinching, or twisting of the wrist and that	
32		operate with a force of not more than 5 lbf (22.2 N).	
33		2. Comply with the following maximum opening-force requirements:	
34		a. Interior, Non-Fire-Rated Hinged Doors: 5 lbf (22.2 N) applied perpendicular to door.	
35		b. Fire Doors: Minimum opening force allowable by authorities having jurisdiction.	
36		3. Bevel raised thresholds with a slope of not more than 1:2. Provide thresholds not more than 1/2 inch (13 mm)	
37		high.	
38		4. Adjust door closer sweep periods so that, from an open position of 90 degrees, the door will take at least 5	
39		seconds to move to a position of 12 degrees from the latch.	
40 41		5. Adjust spring ninges so that, from an open position of 70 degrees, the door will take at least 1.5 seconds to	
41 42	22		
42 // 2	2.5	Incks Latches Cylinders and Dead Locks	
44	74.	1 Shall be Schlage L series as noted in hardware sets. Backset shall be 2-3/4" for all locks, latches and dead	
45		locks unless otherwise required to match existing door pren's. No substitutions	
46		2. All locks and latches shall be of one manufacturer.	
47		3. Included construction cores for all cylinders.	
48		4. All cylinder cores shall be Schlage Primus brand cores. No substitutions. Primus cores are to be supplied by	
49		owner.	
50	В.	Exit Devices	
51		1. Devices shall be as manufactured by Von Duprin as noted in hardware sets. No substitutions.	
52		2. Function shall be as noted in schedule.	
53		3. Mount exit devices to match mullions at adjacent doors and frames wherever possible.	
54	C.	Butt Hinges	
55		1. Shall be Best, Hager, or Ives ball bearing, nonrising loose pin, flat button tip, unless specified to the contrary.	
56		2. Number of butts required:	
57		a. Doors up to 7' – 4" - 3 butts	
58		b. Doors $7-4^{\circ}$ up to $10^{\circ}-4$ butts	

1			3 Butt size requirements
2			5. But size requirements a Interior doors up to 36" wide $A_1/2 \times A_1/2$ standard weight
2			h Interior doors over 36° wide $4 \cdot 1/2 \times 4 \cdot 1/2$ standard weight
4			c Exterior doors $4-1/2 \times 4-1/2$ heavy weight
5			4. Door butt legend: (unless otherwise noted in Schedule)
6			a Exterior doors and interior doors exposed to garage/wet areas –stainless steel-NRP
7			h All other doors steel based
8			5 Eurnish III approved butts on labeled doors
9			6 Provide non-removable nins (NRP) on all exterior and lockable outswinging doors
10			 Continuous Gear Hinges: Hager, Ives, or Select.
11		D.	Door Closers
12			 Shall be LCN 4040XP series unless otherwise specified in hardware groups. No substitutions.
13			 Closers shall have key adjusting device.
14			3. Mount to provide maximum opening permitted by building construction or equipment. Note on schedule
15			the maximum swing per location for other trades involved in reinforcement or installation.
16			4. All door closers shall be similar in design and appearance to those listed in the schedule, so far as possible, of
17			one manufacturer. Furnish special arms and applications as indicated in hardware schedule or as dictated by
18			structural conditions or local code requirements. Provide appropriate brackets for doors with transoms.
19			5. Door closers at labeled fire doors shall bear UL approval. Provide thru-bolts for mineral core doors unless
20			otherwise specified in door specifications.
21		E.	Pushes, Pulls, and Kickplates
22			1. Shall be as manufactured by Hager, Ives, Rockwood, or Trimco
23			2. All plates shall be 16 gauge (0.50), beveled sides and with countersunk screw holes at intervals of not over 6"
24			on all four sides. Screws shall be stainless steel oval head, finish to match plates.
25	l	F.	Stops and Bumpers
26			1. Shall be Hager, Ives, Rixson, Rockwood, or Trimco
27			2. Install bumper behind each door.
28			3. Overhead stops and holder types as specified in hardware groups.
29			4. Apply with expansion shield and machine screws only.
30			5. Provide special templates as required for proper coordination of door closers and overhead door holders.
31		G.	Manual and Automatic flushbolts.
32			1. Shall be Hager, Ives, Rockwood, or Trimco.
33		Н.	Low Energy Openers
34			1. ADA Automatic Openers: Stanley Magic Force, Horton 4000LE or Motion Access Condor Swing as specified in
35			hardware groups. Automatic operators are to be included as supply and install. Coordinate installation and
36			operation of opener and switches with electrical contractor.
37	I	I.	Thresholds, Weatherstripping, Sound Seals
38			1. National Guard, Hager, or Pemko.
39		J.	Door Silencers or Mutes
40			1. Furnish three for each pressed steel frame for single doors, two for each pressed steel frame for pairs of
41			doors.
42	l	К.	Other Materials
43			1. Provide other materials not specifically described but required for a complete and proper installation, as
44			selected by the Contractor and approved by the Architect.
45	2.4		KEYING
46		A.	Keying is by owner.
47	2.5	_	FABRICATION
48		Α.	Base Metals: Produce door hardware units of base metal indicated, fabricated by forming method indicated, using
49			manufacturer's standard metal alloy, composition, temper, and hardness. Furnish metals of a quality equal to or
50		_	greater than that of specified door hardware units and ANSI/BHMA A156.18.
51	l	В.	Fasteners: Provide door hardware manufactured to comply with published templates prepared for machine, wood,
52			and sneet metal screws. Provide screws that comply with commercially recognized industry standards for application
ン こ こ			intended; nowever, auminum fasteners are not permitted. Provide Phillips flat-nead screws with finished heads to
54 EE	26		match surface of door nardware unless otherwise indicated.
55 E.C	2.6	^	FINISHES
50		н. D	Provide ministics complying with ANSI/BEINA ALSO. To as indicated in door nardware schedule.
52	I	ט .	covering before shipping
50			covering before simpping.

1 2 3 4	1 C. Appearance of Finished Work: Variations in appearance of abutting or adjacent pieces are acceptable 2 within one-half of the range of approved Samples. Noticeable variations in the same piece are not a 3 Variations in appearance of other components are acceptable if they are within the range of approved Sa 4 are assembled or installed to minimize contrast.	
6 7	PART 3 - E	EXECUTION
8		
9	3.1	EXAMINATION
10	Α.	Examine doors and frames, with Installer present, for compliance with requirements for installation tolerances,
11		labeled fire-rated door assembly construction, wall and floor construction, and other conditions affecting
12	_	performance of the Work.
13	В.	Examine roughing-in for electrical power systems to verify actual locations of wiring connections before electrified
14	C	door nardware installation.
15	ر. م	
10	3.Z A	Steel Deers and Frames: For surface applied deer bardware, drill and tap deers and frames in accordance with
10	А.	
19	3 3	
20	Δ	Mounting Heights: Mount door hardware units at heights to comply with the following unless otherwise indicated
21	7	or required to comply with governing regulations.
22		1. Standard Steel Doors and Frames: ANSI/SDI A250.8.
23		2. Custom Steel Doors and Frames: HMMA 831.
24	В.	Install each door hardware item to comply with manufacturer's written instructions. Where cutting and fitting are
25		required to install door hardware onto or into surfaces that are later to be painted or finished in another way,
26		coordinate removal, storage, and reinstallation of surface protective trim units with finishing work. Do not install
27		surface-mounted items until finishes have been completed on substrates involved.
28		1. Set units level, plumb, and true to line and location. Adjust and reinforce attachment substrates as necessary
29		for proper installation and operation.
30		2. Drill and countersink units that are not factory prepared for anchorage fasteners. Space fasteners and anchors
31		in accordance with industry standards.
32	С.	Hinges: Install types and in quantities indicated in door hardware schedule, but not fewer than the number
33		recommended by manufacturer for application indicated or one hinge for every 30 inches (760 mm) of door height,
34		whichever is more stringent, unless other equivalent means of support for door, such as spring hinges or pivots, are
35	_	provided.
36	D.	Boxed Power Supplies: Locate power supplies as indicated or, if not indicated, above accessible ceilings or in
3/		equipment room. Verify location with Architect.
38 20		Configuration: Provide one power supply for each door opening of least number of power supplies required to adoquately corve doors with electrified door bardware
<u>40</u>	F	Thresholds: Set thresholds for exterior doors and other doors indicated in full hed of sealant complying with
40 //1	L.	requirements specified in Section 079200 "Joint Sealants "
42	F	Perimeter Gasketing: Apply to head and jamb. forming seal between door and frame
43		1 Do not notch perimeter gasketing to install other surface-applied bardware
44	3.4	INSPECTION AND ADJUSTMENTS
45	Α.	Hardware vendor shall, before substantial completion of project, and/or as directed by Architect, inspect all locks
46		that are part of this project to see that keys pass proper locks as required.
47	В.	Check that all locks and latches are properly lubricated, as recommended by manufacturer, with lock lubricant and
48		that all moving parts are adjusted correctly to insure free, smooth operation.
49	С.	Door closers and holders shall be checked for proper lubrication. After building is in use, arrange with factory
50		representative of closer manufacturer to make final adjustments to closers to meet building conditions.
51		
52	3.5	CLEANING AND PROTECTION
53	Α.	Clean adjacent surfaces soiled by door hardware installation.
54	В.	Clean operating items as necessary to restore proper function and finish.
55	С.	Provide final protection and maintain conditions that ensure that door hardware is without damage or deterioration
56		at time of Substantial Completion.

1					
2					
3					
4	SCH	HEDULE			
5	Cho	ock spor	ified schedule against latest revised plans	when making up schedule for approval. Schedule	oach door congratoly and
7	whe	ere prac	tical, item numbers shall be same as door	numbers and in consecutive sequence.	each door separately and,
8 9 10	lf h	ardware	e is not scheduled for a particular door, fu	rnish hardware of types specified for similar locati	ons as far as practical.
10	Incl	lude acc	essories required to fully equip in satisfac	tory manner, doors and the like as shown and sp	ecified. Include necessary
12	scre	ews, nut	s, bolts, expansion shields and other device	ces necessary for proper application.]	
13	Iter	ns of ha	ardware herein described shall be conside	red as standard and, unless otherwise specifically	mentioned, all hardware
15	use	d throu	ghout work shall be equal thereto in size,	weight, material and workmanship. Revision of	standard hardware, which
16	ma	y be neo	essary to conform to details shall be provi	ded. Items not specifically mentioned but necessa	ary for completion of work
17	sha	ll be of	most suitable type matching in quality and	finish items which are described.	
18					
20	SET	01			
21	<u>.</u>	EA	HINGES	AS SPECIFIED	640 IVES
22	1	EA	CLASSROOM LOCK	L9070T X 03A	613 SCHLAGE
23	1	EA	PRIMUS CORE	AS REQUIRED	613 SCHLAGE
24	2	EA	CLOSER	4040XP-H	695 LCN
25	2	EA	OVERHEAD STOP	GJ100 SERIES	613 GLYNN-JOHNSON
26	1	SET	AUTO FLUSHBOLTS	FB31P/FB41P	613 IVES
27	1	EA	DUSTPROOF STRIKE		613 IVES
28	1	EA			711 IVES
29	2	EA	KICKPLATE	IO XI LDW	613 IVES
31	**P	ROVIDE	LCN SPECIAL TEMPLATE ST1630 AND 404	0XP-18TI PLATE AS REQUIRED FOR CLOSER AND (OVERHEAD STOP
32		INS	TALLATION.		
33					
34					
35	<u>SET</u>	02			
36		EA	HINGES	AS SPECIFIED	640 IVES
37	1	EA	CLASSROOM LOCK	L9070T X 03A	613 SCHLAGE
38	1	EA	PRIMUS CORE	AS REQUIRED	613 SCHLAGE
39	1	EA			695 LCN
40 11	1	EA			613 GLYNN-JUHNSUN
41 //2	т	LA	NERFLATE	10 X 2 LDW	013 1013
43					
44	SET	03			
45	-	EA	HINGES	AS SPECIFIED	640 IVES
46	1	EA	PRIVACY	L9040 X 03A X L283-722	613 SCHLAGE
47	1	EA	CLOSER	4040XP	695 LCN
48	1	EA	WALL STOP	409	613 ROCKWOOD
49	1	EA	KICKPLATE	10" X 2" LDW	613 IVES
50					
51 52	677	. 04			
52 53	<u>3E I</u>	<u>04</u> FA	HINGES		640 IVES
55 54	1	EA FA			
55	- 1	FΔ	EXIT DEVICE	9927DT X I BR	613 VON DUPRIN
56	1	EA	I/C CYLINDER	AS REQUIRED	613 SCHLAGE
57	1	EA	PRIMUS CORE	AS REQUIRED	613 SCHLAGE
58	2	EA	CLOSER	4040XP	695 LCN

1	2	EA	OVERHEAD HOLDER	GJ100H SERIES	613 GLYNN-JOHNSON
2	2	EA	KICKPLATE	10" X 1" LDW	613 IVES
4 5 6	**P	REP DO ACC	ORS WITH RACEWAYS FOR FUTURE ELECT ESS.	RIC LATCH RETRACTION KITS AND ELECTRIC HING	ES FOR FUTURE CARD
7 8	SET	05			
9		EA	HINGES	AS SPECIFIED	640 IVES
10	2	EA	EXIT DEVICE	9927DT X LBR	613 VON DUPRIN
11	2	EA	CLOSER	4040XP	695 LCN
12	2	EA	OVERHEAD HOLDER	GJ100H SERIES	613 GLYNN-JOHNSON
13	2	EA	KICKPLATE	10" X 1" LDW	613 IVES
14 15 16 17	**P	REP DO ACC	ORS WITH RACEWAYS FOR FUTURE ELECT ESS.	RIC LATCH RETRACTION KITS AND ELECTRIC HING	ES FOR FUTURE CARD
18	СЕТ	06			
20	361	<u>60</u> FΔ	HINGES	224HD	640 IVES
20	1	FA	FXIT DEVICE	99FO	613 VON DUPRIN
22	1	EA	CLOSER	4040XP	695 LCN
23	1	EA	OVERHEAD STOP	GJ100 SERIES	613 GLYNN-JOHNSON
24	1	SET	WEATHERSTRIP	700SDKB	DKB NGP
25	1	EA	SWEEP	200NDKB	DKB NGP
26 27	1	EA	THRESHOLD	8425	AL NGP
28 29	SFT	07			
30	1	FA	CONTINUOUS HINGE	112HD	DKB IVES
31	1	EA	EXIT DEVICE	99EO	613 VON DUPRIN
32	1	EA	CLOSER	4040XP	695 LCN
33	1	EA	OVERHEAD STOP	GJ100 SERIES	613 GLYNN-JOHNSON
34	1	EA	SWEEP	200NDKB	DKB NGP
35	1	EA	THRESHOLD	8425	AL NGP
36	***				
37	**S	EALS BY	DOOR SUPPLIER.		
30 39	P	ROVIDE	DROP PLATE AND BLADE STOP SPACER AS	S REQUIRED FOR CLOSER INSTALLATION.	
40					
41	SET	08			
42		EA	HINGES	AS SPECIFIED	640 IVES
43	1	EA	STOREROOM LOCK	L9080T X 03A	613 SCHLAGE
44	1	EA	PRIMUS CORE	AS REQUIRED	613 SCHLAGE
45	2	EA	CLOSER	4040XP-SCUSH	695 LCN
46	1	SET	AUTO FLUSHBOLTS	FB31P/FB41P	613 IVES
47	1	EA			613 IVES
48	1	EA			711 IVES
49 50	2				BLK IVES
50	2 1	SET	SEALS	5050C	BIK NGP
52	1	SET	ASTRAGAL	115N	DKB NGP
53 54 55	2	EA	AUTO DR BOTTOM	320N	AL NGP
56	SET	09			
57		EA	HINGES	AS SPECIFIED	640 IVES
58	1	EA	STOREROOM LOCK	L9080T X 03A	613 SCHLAGE

	SI N	PECIFIC/ Aay 16, 2	ATION 2024		
1	1	EA	PRIMUS CORE	AS REQUIRED	613 SCHLAGE
2	2	EA	CLOSER	4040XP	695 LCN
3	2	EA	OVERHEAD HOLDER	GJ100H SERIES	613 GLYNN-JOHNSON
4	1	SET	AUTO FLUSHBOLTS	FB31P/FB41P	613 IVES
5	1	EA	DUSTPROOF STRIKE	DP2	613 IVES
6	1	EA	COORDINATOR	COR X FL	711 IVES
7	2	EA	MNTG BRACKETS	AS REQUIRED	BLK IVES
8	2	EA	KICKPLATE	10" X 1" LDW	613 IVES
9	1	SET	SEALS	5050C	BLK NGP
10	1	SET	ASTRAGAL	115N	DKB NGP
11	2	EA	AUTO DR BOTTOM	320N	AL NGP
12					
13					
14	SE	<u>T 10</u>			
15		EA	HINGES	AS SPECIFIED	640 IVES
16	1	EA	STOREROOM LOCK	L9080T X 03A	613 SCHLAGE
17	1	EA	PRIMUS CORE	AS REQUIRED	613 SCHLAGE
18	1	EA	CLOSER	4040XP	695 LCN
19	1	EA	WALL STOP	409	613E ROCKWOOD
20	1	EA	KICKPLATE	10" X 2" LDW	613 IVES
21	1	EA	DEADBOLT FILLER PLATE	AS REQUIRED	
22					
23	**(GC TO	CUT NEW 4 7/8" ANSI STRIKE PREP I	N FRAME FOR NEW LOCK.	
24	**	PROVIE	DE NEW FILLER FOR EXISTING DEADE	BOLT STRIKE. FIELD VERIFY.	
25					
26					
27	SE	T 11			
28		EA	HINGES	AS SPECIFIED	640 IVES
29	1	EA	PUSH	70C	613 ROCKWOOD
30	1	EA	PULL	BF111 X 70C	613 ROCKWOOD
31	1	EA	CLOSER	4040XP	695 LCN
32	1	EA	WALL STOP	409	613E ROCKWOOD
33	1	EA	KICKPLATE	10" X 2" LDW	613 IVES
34	1	EA	FOOT PULL	FP1230	613E ROCKWOOD
35					
36					
37					
38					
39					
40					
41					
42					
43					
44					
45				END OF SECTION	

CITY OF MADISON

1

SECTION 08 80 00

2		GLAZING		
3	PART 1 -	GENERAL		
4	1.1	SUMMARY		
5	Α.	Section Includes:		
6		1. Glass products.		
7		2. Laminated glass.		
8		3. Insulating glass.		
9		4. Glazing sealants.		
10		5. Glazing tapes.		
11		6. Miscellaneous glazing materials.		
12	1.2	COORDINATION		
13	А.	Coordinate glazing channel dimensions to provide necessary bite on glass, minimum edge and face clearances, and		
14		adequate sealant thicknesses, with reasonable tolerances to achieve proper safety margins for glazing retention		
15		under each design load case, load case combination, and service condition.		
16	1.3	PREINSTALLATION MEETINGS		
17	Α.	Preinstallation Conference: Conduct conference at Project site.		
18	1.4	ACTION SUBMITTALS		
19	А.	Product Data: For each type of product.		
20	B.	Glass Samples: For each type of glass product other than clear monolithic vision glass: 12 inches square.		
21	C.	Delegated Design Submittal: For glass indicated to comply with performance requirements and design criteria.		
 22	0.	including analysis data signed and sealed by qualified professional engineer responsible for their preparation		
23	1.5	INFORMATIONAL SUBMITTALS		
24	Δ	Product Certificates: For glass		
25	B	Product test renorts		
26	C	Preconstruction adhesion and compatibility test report		
27	D.	Sample warranties		
27 28	16			
20	Δ	Sealant Testing Agency Qualifications: An independent testing agency qualified according to ASTM (1021 to		
20	Π.	conduct the testing indicated		
21	17			
37	1.7	Manufacturer's Special Warranty for Coated-Glass Products: Manufacturer agrees to replace coated-glass units that		
22	Α.	deteriorate within specified warranty period. Deterioration of coated glass is defined as defects developed from		
27		normal use that are not attributed to glass breakage or to maintaining and cleaning coated glass contrary to		
24 25		manufacturar's written instructions. Defects include peoling, cracking, and other indications of deterioration in		
25		costing		
30 27		Cudillig.		
57 20	Р	Wandhity Period. 10 years from date of Substantial Completion.		
20	Б.	deteriorate within specified warranty period. Deterioration of laminated glass is defined as defects developed from		
39		deteriorate within specified warranty period. Deterioration of faminated glass is defined as defects developed from		
40		normal use that are not attributed to glass breakage or to maintaining and cleaning laminated glass contrary to		
41 42		manufacturer's written instructions. Defects include edge separation, delamination materially obstructing vision		
42		through glass, and blemisnes exceeding those allowed by reterenced laminated-glass standard.		
43	6	1. Warranty Period: 10 years from date of Substantial Completion.		
44	C.	Manufacturer's Special Warranty for Insulating Glass: Manufacturer agrees to replace insulating-glass units that		
45		deteriorate within specified warranty period. Deterioration of insulating glass is defined as failure of hermetic seal		
46		under normal use that is not attributed to glass breakage or to maintaining and cleaning insulating glass contrary to		
47		manufacturer's written instructions. Evidence of failure is obstruction of vision by dust, moisture, or film on interior		
48		surfaces of glass.		
49		1. Warranty Period: 10 years from date of Substantial Completion.		
50	PART 2 -	- PRODUCTS		
51	2.1	PERFORMANCE REQUIREMENTS		
52	Α.	Delegated Design: Engage a qualified professional engineer, as defined in Section 01 40 00 "Quality Requirements,"		
53		to design glazing.		
54	В.	Structural Performance: Glazing shall withstand the following design loads within limits and under conditions		
55		indicated determined in accordance with the IBC and ASTM E1300:		
56		1. Design Wind Pressures: As indicated on Drawings.		
57		2. Design Snow Loads: As indicated on Drawings.		

CITY OF MADISON SPECIFICATION May 16, 2024

Tridy 10	, 2024
	3. Thermal Loads: Design glazing to resist thermal stress breakage induced by differential temperature
c	Windhorno Dobrie Impact Registance: Exterior glazing chall page ACTM 51996 missile impact and unlis another sectors.
C.	windborne-Debris-Impact Resistance: Exterior glazing shall pass ASTM E1886 missile-Impact and cyclic-pressure
	tests in accordance with ASTM E1996 for Wind Zone 1 for basic protection.
	Large-Missile Test: For glazing located within 30 feet of grade. Small Missile Test: For glazing located between 20 feet and 60 feet above stude
	2. Small-Missile Test: For glazing located between 30 feet and 60 feet above grade.
D.	Satety Glazing: where satety glazing is indicated, provide glazing that compiles with 16 CFR 1201, Category II.
E.	Inermal and Optical Performance Properties: Provide glass with performance properties specified, as indicated in
	manufacturer's published test data, based on procedures indicated below:
	1. O-Factors: Center-of-glazing values, in accordance with NFRC 100 and based on most current non-beta
	Version of LBL's window computer program, expressed as Btu/sq. ft. x n x deg F.
	2. Singe and visible transmittance: center-or-glazing values, in accordance with NFRC 200 and based on most
	Current non-beta version of LBL's windows computer program.
F	3. Visible Reflectance: Center-of-glazing values, in accordance with NFRC 300.
г.	Acoustic Performance:
	1. Exterior Glazing, 28 OFFC min
. .	
2.2	GLASS PRODUCTS, GENERAL
А.	Glazing Publications: Comply with published recommendations of glass product manufacturers and organizations helew upless more stringent requirements are indicated. See these publications for glasing terms not athenuise
	defined in this Costion on in referenced standards
	defined in this section of inferenced standards.
	NGA Publications. Laminated Glazing Reference Manual and Glazing Manual. AAMA Publications: AAMA CDSC 1 "Class Design for Sloped Clasing" and AAMA TIP A7 "Sloped Clasing"
	2. AAMA Publications: AAMA GDSG-1, Glass Design for Sloped Glazing, and AAMA TR A7, Sloped Glazing
	Guidelines.
	3. IGMA Publication for Inculating Class: SIGMA TA 2000, "North American Clasing Cuidelines for Scaled
	4. IdiviA Publication for Insulating Glass. SidiviA TVI-5000, North American Glazing Guidelines for Sealed
р	Insulating Glass Units for Commercial and Residential Ose.
р.	Salety Glazing Labeling. Where salety glazing is indicated, permanently mark glazing with tertification label of the
	socc of another certification agency acceptable to authornies having jurisdiction. Laber shall indicate
c	Inditude ture sindifie, type of glass, thickness, and safety glazing standard with which glass completes.
C.	units with appropriate contification label of the ICCC
р	Thickness: Where glass thickness is indicated, it is a minimum Provide glass that complies with performance
D.	requirements and is not loss than this loss indicated, it is a minimum. From the glass that complete with performance
E	requirements and is not less than tinckness mulcated, arouide appealed fleat glass, heat strengthened fleat glass, or
с.	fully tempered float glass as needed to comply with "Performance Requirements" Article. Where heat strengthenes
	float glass is indicated, provide heat-strengthened float glass or fully tempered float glass as needed to comply with
	"Performance Requirements" Article. Where fully tempered float glass is indicated, provide fully tempered float
	alass
23	
Δ	Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be
	incorporated into the Work include, but are not limited to the following.
	a Cardinal Glass Industries Inc
	h Guardian Glass II C
	c Pilkington North America: NSG Group
	d Saint-Gobain Glass Corn
	e Viracon
R	Clear Annealed Float Glass: ASTM C1036 Type L Class 1 (clear) Quality-O3
Б. С	Low-Iron Annealed Float Glass: ASTM C1036 Type I, Class I (clear), Quality-Q3.
С.	not less than 91 nercent
П	Tinted Annealed Float Glass: ASTM (1036 Tyne I Class 2 /tinted) Quality-02
D. F	Fully Tempered Float Glass: ASTM C1030, Type I, Class 2 (Linced), Quality-Q3.
L.	indicated Type I Class 1 (clear) or Class 2 (tinted) as indicated Quality_Q3
F	Heat-Strengthened Float Glass: ASTM C1048 Kind HS (heat strengthened) Type I Condition A (uncepted) unloss
г.	otherwise indicated Type I Class 1 (clear) or Class 2 (tinted) as indicated Ouality_O2
G	Reflective- and Low-E-Costed Vicion Glass ACTM C1376
о. ц	Ceramic Coated Vision Glass: ASTM (1018 Condition C Type I Class 1 (clear) or Class 2 (tintod) as indicated
11.	Ouality-O3: and complying with Specification No. 95-1-31 in NGA's "Engineering Standards Manual."
	quality que compression with specification no. 35 1 51 in NOA 5 Engineering standards Mandal.

May 1	6, 2024		
I.	Ceramic-Coated Spandrel Glass: ASTM C1048, Type I, Condition B, Quality-Q3.		
J.	Silicone-Coated Spandrel Glass: ASTM C1048, Type I, Condition C, Quality-Q3.		
К.	Reflective- and Low-E-Coated Spandrel Glass: ASTM C1376, Kind CS.		
2.4	LAMINATED GLASS		
Α.	Laminated Glass: ASTM C1172. Use materials that have a proven record of no tendency to bubble, discolor, or lose		
	1 Construction: Laminate glass with polygingl butgral interlayer ionoplast interlayer or cast-in-place and		
	cured-transparent-resin interlayer to comply with interlayer manufacturer's written instructions		
	2 Interlayer Thickness: Provide thickness not less than that indicated and as needed to comply with		
	requirements.		
	3. Interlayer Color: Clear unless otherwise indicated.		
2.5	INSULATING GLASS		
	Insulating-Glass Units: Factory-assembled units consisting of sealed lites of glass separated by a dehydrated		
	interspace, gualified in accordance with ASTM E2190.		
	1. Sealing System: Dual seal, with manufacturer's standard primary and secondary sealants.		
	2. Perimeter Spacer: Manufacturer's standard spacer material and construction.		
	3. Desiccant: Molecular sieve or silica gel, or a blend of both.		
2.6	GLAZING SEALANTS		
Α.	General:		
	1. Compatibility: Compatible with one another and with other materials they contact, including glass products		
	seals of insulating-glass units, and glazing channel substrates, under conditions of service and application, as		
	demonstrated by sealant manufacturer based on testing and field experience.		
	2. Suitability: Comply with sealant and glass manufacturers' written instructions for selecting glazing sealants		
	suitable for applications indicated and for conditions existing at time of installation.		
	3. Colors of Exposed Glazing Sealants: As selected by Architect from manufacturer's full range of industry		
	colors.		
В.	Neutral-Curing Silicone Glazing Sealant, Class 100/50: Complying with ASTM C920, Type S, Grade NS, Use NT.		
2.7	GLAZING TAPES		
Α.	Back-Bedding Mastic Glazing Tapes: Preformed, butyl-based, 100 percent solids elastomeric tape; nonstaining and		
	nonmigrating in contact with nonporous surfaces; with or without spacer rod as recommended in writing by tape		
	and glass manufacturers for application indicated; and complying with ASTM C1281 and AAMA 800 for products		
	indicated below:		
	1. AAMA 804.3 tape, where indicated.		
	2. AAMA 806.3 tape, for glazing applications in which tape is subject to continuous pressure.		
	3. AAMA 807.3 tape, for glazing applications in which tape is not subject to continuous pressure.		
2.8	MISCELLANEOUS GLAZING MATERIALS		
Α.	Cleaners, Primers, and Sealers: Types recommended by sealant or gasket manufacturer.		
В.	Setting Blocks:		
	1. Type recommended in writing by sealant or glass manufacturer.		
C.	Spacers:		
_	1. Type recommended in writing by sealant or glass manufacturer.		
D.	Edge Blocks:		
_	1. Type recommended in writing by sealant or glass manufacturer.		
E.	Cylindrical Glazing Sealant Backing: ASTM C1330, Type O (open-cell material), of size and density to control a		
	sealant depth and otherwise produce optimum glazing sealant performance.		
PARI 3	- EXECUTION		
3.1	GLAZING, GENEKAL		
А.	Comply with complete written instructions of manufacturers of glass, sealants, gaskets, and other glazing materials,		
Б	unless more sumgent requirements are mutated, including those in referenced glazing publications.		
в.	Frotect glass edges from damage during nationing and installation. Kemove damaged glass from Project site and		
	iegally dispose of off Project site. Damaged glass includes glass with edge damage or other imperfections that, wher		
c	Installed, could weaken glass, impair performance, or impair appearance.		
C.	Apply primers to joint surfaces where required for agnesion of searants, as determined by preconstruction testing.		
D.	install setting blocks in sill rabbets, sized and located to comply with referenced glazing publications, unless		
E	Other wise required by glass manufacturer. Set blocks in thin course of compatible sealant suitable for neel bead.		
с. г	Do not exceed edge pressures supurated by glass manufacturers for installing glass lites.		
г.	FI ONIGE SPACETS TOT BIASS THES WHETE TENBLIT PLUS WHUTH IS IALBET THAT SO THATES.		

G. Provide edge blocking where indicated or needed to prevent glass lites from moving sideways in glass recommended in writing by glass manufacturer and in accordance with requirements in republications.	
2 2	
Δ	Position tapes on fixed stops so that when compressed by glass their exposed edges are flush with or protrude
7.0	slightly above sightline of stops.
В.	Install tapes continuously, but not necessarily in one continuous length. Do not stretch tapes to make them fit
	opening.
C.	Cover vertical framing joints by applying tapes to heads and sills first, then to jambs. Cover horizontal framing joints
	by applying tapes to jambs, then to heads and sills.
D.	Place joints in tapes at corners of opening with adjoining lengths butted together, not lapped. Seal joints in tapes
	with compatible sealant approved by tape manufacturer.
Ε.	Apply heel bead of elastomeric sealant.
F.	Center glass lites in openings on setting blocks, and press firmly against tape by inserting dense compression
	gaskets formed and installed to lock in place against faces of removable stops. Start gasket applications at corners
	and work toward centers of openings.
G.	Apply cap bead of elastomeric sealant over exposed edge of tape.
3.3	GASKET GLAZING (DRY)
А.	Cut compression gaskets to lengths recommended by gasket manufacturer to fit openings exactly, with allowance
	for stretch during installation.
В.	Insert soft compression gasket between glass and frame or fixed stop so it is securely in place with joints miter cut
C	and bonded together at corners.
C.	soft compression gasket by inserting donce compression gaskets formed and installed to lock in place against faces
	of removable stons. Start gasket applications at corners and work toward centers of openings. Compress gaskets to
	nroduce a weathertight seal without developing bending stresses in glass. Seal gasket joints with sealant
	recommended in writing by gasket manufacturer
D.	Installation with Pressure-Glazing Stops: Center glass lites in openings on setting blocks, and press firmly against
5.	soft compression gasket. Install dense compression gaskets and pressure-glazing stops, applying pressure uniformly
	to compression gaskets. Compress gaskets to produce a weathertight seal without developing bending stresses in
	glass. Seal gasket joints with sealant recommended in writing by gasket manufacturer.
Ε.	Install gaskets so they protrude past face of glazing stops.
3.4	SEALANT GLAZING (WET)
Α.	Install continuous spacers, or spacers combined with cylindrical sealant backing, between glass lites and glazing
	stops to maintain glass face clearances and to prevent sealant from extruding into glass channel and blocking weep
	systems until sealants cure. Secure spacers or spacers and backings in place and in position to control depth of
	installed sealant relative to edge clearance for optimum sealant performance.
В.	Force sealants into glazing channels to eliminate voids and to ensure complete wetting or bond of sealant to glass
	and channel surfaces.
C.	Tool exposed surfaces of sealants to provide a substantial wash away from glass.
3.5	CLEANING AND PROTECTION
A.	Immediately after installation, remove nonpermanent labels and clean surfaces.
В.	Protect glass from contact with contaminating substances resulting from construction operations. Examine glass
	surfaces adjacent to or below exterior concrete and other masonry surfaces at frequent intervals during
	construction, but not less than once a month, for buildup of dirt, scum, alkaline deposits, or stains.
	in mediately as recommended in writing by glass manufacturer. Remove and replace glass that cannot be
	cleaned without damage to coatings
C	Remove and replace glass that is damaged during construction period
3.6	MONOLITHIC GLASS SCHEDULE
A.	Clear Glass Type GL-4 :
	1. Annealed float glass. Provide fully tempered where safety glazing is required as noted in drawings.
	2. Minimum Thickness: 6 mm.
3.7	LAMINATED GLASS SCHEDULE
Α.	Clear Laminated Glass Type GL-3 :
	1. Two plies of annealed float glass.
	2. Minimum Thickness of Each Glass Ply: 3 mm.
	3. Interlayer Thickness: 0.030 inch min.

CITY OF MADISON

SPECIFICATION

May 16, 2024

1		4.	Safety glazing required.
2	3.8	INSUL	ATING GLASS SCHEDULE
3	Α.	Clear \	With Applied Membrane Insulating Glass Type GL-2:
4		1.	Provide fully tempered where safety glazing is required as noted in drawings.
5		2.	Basis-of-Design Product: Solera, Solera S R5+Aerogel.
6		3.	Applied Membrane: Light Diffusing Veil (by Solera).
7		4.	Overall Unit Thickness: 1 inch.
8		5.	Minimum Thickness of Each Glass Lite: 6 mm.
9		6.	Outdoor Lite: Annealed float glass.
10		7.	Interspace Content: Aerogel.
11		8.	Indoor Lite: Annealed float glass.
12		9.	Safety glazing required.
13		10.	Panel Characteristics:
14			a. SHGC: 0.37 max.
15			b. VLT: 0.40 min.
16			c. U-factor: 0.20 max.
17	В.	Low-E	&Ceramic-Coated, Insulating Vision Glass Type GL-1 :
18		1.	Provide fully tempered where safety glazing is required as noted in drawings.
19		2.	Basis-of-Design Product: Viracon, 51767 Bird Friendly glass.
20		3.	Ceramic Coating Color and Pattern: 1% coverage, 1/4" dot, 2x2, staggered in warm grey.
21		4.	Overall Unit Thickness: 1 inch.
22		5.	Minimum Thickness of Each Glass Lite: 6 mm.
23		6.	Outdoor Lite: Clear heat-strengthened float glass.
24		7.	Interspace Content: Argon.
25		8.	Indoor Lite: Clear heat-strengthened float glass.
26		9.	Ceramic Coating Location: Second surface.
27		10.	Low-E Coating: Pyrolytic or sputtered on third surface.
28		11.	Panel Characteristics:
29			a. SHGC: 0.25 max.
30			b. VLT: 0.60 min.
31			c. U-factor: 0.20 max.
32			END OF SECTION

Page Intentionally Left Blank

SECTION 08 83 00

1		SECTION 08 83 00			
2	MIRRORS				
3	PART 1 - GENERAL				
4	1.1	SUMMARY			
5	Α.	This section does NOT apply to those tagged as Toilet Accessories in drawings.			
6	В.	Section Includes:			
7		1. Silvered flat glass mirrors.			
8	1.2	ACTION SUBMITTALS			
9	Α.	Product Data: For each type of product.			
10	В.	Shop Drawings: Include mirror elevations, edge details, mirror hardware, and attachment details.			
11	С.	Samples: For each type of the following:			
12		1. Mirrors: 12 inches square, including edge treatment on two adjoining edges.			
13		2. Mirror Clips: Full size.			
14		3. Mirror Trim: 12 inches long.			
15	1.3	INFORMATIONAL SUBVITTALS			
10	А.	preconstruction Test Reports: From mirror manufacturer indicating that mirror mastic was tested for compatibility			
10	р	and addesion with mirror backing and substrates on which mirrors are installed.			
10	D. 1 /				
19	1.4	CLOSECUT SUBIVITIALS Maintenance Data: For mirrors to include in maintenance manuals			
20	4. 1 E				
21	1.5	Installer Qualifications: A qualified Installer, who employs glass installers for this Project who are certified under the			
22	Π.	National Glass Association's Certified Glass Installer Program			
23	16	PRECONSTRUCTION TESTING			
25	1.0 А.	Preconstruction Mirror Mastic Compatibility Test: Submit mirror mastic products to mirror manufacturer for testing			
26	7	to determine compatibility of mastic with mirror backing.			
27		 Testing is not required if data are submitted based on previous testing of mirror mastic products and mirror 			
28		backing matching those submitted.			
29	1.7	WARRANTY			
30	Α.	Special Warranty: Manufacturer agrees to replace mirrors that deteriorate within specified warranty period.			
31		Deterioration of mirrors is defined as defects developed from normal use that are not attributed to mirror breakage			
32		or to maintaining and cleaning mirrors contrary to manufacturer's written instructions. Defects include			
33		discoloration, black spots, and clouding of the silver film.			
34		1. Warranty Period: Five years from date of Substantial Completion.			
35	PART 2 -	PRODUCTS			
36	2.1	SILVERED FLAT GLASS MIRRORS			
37	Α.	Mirrors, General: ASTM C1503.			
38	В.	Annealed Monolithic Glass Mirrors: Mirror Select Quality, clear.			
39		1. Nominal Thickness: 4.0 mm.			
40	2.2	MISCELLANEOUS MATERIALS			
41	Α.	Setting Blocks: Elastomeric material with a Shore, Type A durometer hardness of 85, plus or minus 5.			
42	В.	Edge Sealer: Coating compatible with glass coating and approved by mirror manufacturer for use in protecting			
43		against silver deterioration at mirrored glass edges.			
44	С.	Mirror Mastic: An adhesive setting compound, asbestos-free, produced specifically for setting mirrors.			
45	D.	Film Backing for Safety Mirrors: Film backing and pressure-sensitive adhesive; both compatible with mirror backing			
46		paint as certified by mirror manufacturer.			
4/	2.3				
48	А.	Aluminum J-Channels: Aluminum extrusions with a return deep enough to produce a glazing channel to			
49 F0		accommodate mirrors of thickness indicated and in lengths required to cover edges of mirrors in a single piece.			
		1. Automound 3-channel boltom min. 3-channels formed with front leg and back leg not less than 3/8 and 7/8 inch in height, respectively, and a thickness of not less than 0.04 inch			
57		mentin neight, respectively, and a thickness of not less than 0.04 men.			
52		2. Automation recontinent op minit, recontinets formed with from leg and back leg for less tridit 5/8 dfld 1 filling for height respectively, and a thickness of not less than 0.04 inch			
57		Finish: Clear bright anodized			
55	R	Easteners: Fabricated of same basic metal and alloy as fastened metal and matching it in finished color and texture			
56	υ.	where fasteners are exposed.			

SPECIFICATION

May 16	, 2024			
2.4	FABRICATION			
Α.	Fabricate cutouts for notches and holes in mirrors without marring visible surfaces. Locate and size cutouts, so they			
	fit closely around penetrations in mirrors.			
В.	Mirror Edge Treatment: Flat polished.			
	1. Seal edges of mirrors with edge sealer after edge treatment to prevent chemical or atmospheric			
	penetration of glass coating.			
С.	Film-Backed Safety Mirrors: Apply film backing with adhesive coating over mirror backing paint, as recommended in			
	writing by film-backing manufacturer, to produce a surface free of bubbles, blisters, and other imperfections.			
PART 3 -	EXECUTION			
3.1	EXAMINATION			
Α.	Examine substrates, over which mirrors are to be mounted, with Installer present, for compliance with installation			
	tolerances, substrate preparation, and other conditions affecting performance of the Work.			
В.	Verify compatibility with and suitability of substrates, including compatibility of existing finishes or primers with			
	mirror mastic.			
С.	Proceed with installation only after unsatisfactory conditions have been corrected and surfaces are dry.			
3.2	PREPARATION			
Α.	Comply with mastic manufacturer's written installation instructions for preparation of substrates, including coating			
	substrates with mastic manufacturer's special bond coating where applicable.			
3.3	INSTALLATION			
Α.	General: Install mirrors to comply with mirror manufacturer's written instructions and with referenced National			
	Glass Association (NGA) publications. Mount mirrors accurately in place in a manner that avoids distorting reflected			
	images.			
В.	Install mirrors with mastic and mirror hardware. Attach mirror hardware securely to mounting surfaces with			
	mechanical fasteners installed with anchors or inserts as applicable. Install fasteners so heads do not impose point			
	loads on backs of mirrors.			
	1. Aluminum J-Channels: Provide setting blocks 1/8 inch thick by 4 inches long at quarter points. To prevent			
	trapping water, provide, between setting blocks, two slotted weeps not less than 1/4 inch wide by 3/8 inch			
	long at bottom channel.			
	2. Install mastic as follows:			
	a. Apply barrier coat to mirror backing where approved in writing by manufacturers of mirrors and			
	backing material.			
	b. Apply mastic to comply with mastic manufacturer's written instructions for coverage and to allow			
	air circulation between back of mirrors and face of mounting surface.			
	c. After mastic is applied, align mirrors and press into place while maintaining a minimum airspace of			
	1/8 inch between back of mirrors and mounting surface.			
С.	Clean exposed surface of mirrors not more than four days before date scheduled for inspections that establish date			
	of Substantial Completion. Clean mirrors as recommended in writing by mirror manufacturer and NGA's publication			
	"Proper Procedures for Cleaning Flat Glass Mirrors."			
	END OF SECTION			

SECTION 08 91 19 FIXED LOUVERS

1	SECTION 08 91 19				
2	FIXED LOUVERS				
3	PART 1 - GENERAL				
4	1.1 SUMMARY				
5	A. Section includes fixed extruded-aluminum louvers.				
6	1.2	ACTION SUBMITTALS			
7	Α.	Product Data: For each type of product.			
8		1. For louvers specified to bear AMCA seal, include printed catalog pages showing specified models with			
9		appropriate AMCA Certified Ratings Seals.			
10	В.	Shop Drawings: For louvers and accessories. Include plans, elevations, sections, details, and attachments to other			
11		work. Show frame profiles and blade profiles, angles, and spacing.			
12	C.	Samples: For each type of metal finish required.			
13	D.	Delegated Design Submittal: For louvers indicated to comply with structural performance requirements and design			
14		criteria, including analysis data signed and sealed by the qualified professional engineer responsible for their			
15		preparation.			
16	1.3	INFORMATIONAL SUBMITTALS			
17	Α.	Product Test Reports: Based on tests performed according to AMCA 500-L.			
18	В.	Sample warranties.			
19	1.4	QUALITY ASSURANCE			
20	Α.	Welding Qualifications: Qualify procedures and personnel according to the following:			
21		1. AWS D1.2/D1.2M, "Structural Welding Code - Aluminum."			
22		2. AWS D1.3/D1.3M, "Structural Welding Code - Sheet Steel."			
23		3. AWS D1.6/D1.6M, "Structural Welding Code - Stainless Steel."			
24	1.5	WARRANTY			
25	Α.	Special Finish Warranty, Anodized Finishes: Standard form in which manufacturer agrees to repair finishes or			
26		replace aluminum that shows evidence of deterioration of anodized finishes within specified warranty period.			
27		1. Warranty Period: 10 years from date of Substantial Completion.			
28	PART 2 - I	PRODUCTS			
29	2.1	PERFORMANCE REQUIREMENTS			
30	Α.	Delegated Design: Design louvers, including comprehensive engineering analysis by a qualified professional			
31		engineer, using structural performance requirements and design criteria indicated.			
32	В.	Structural Performance: Louvers withstand the effects of gravity loads and the following loads and stresses within			
33		limits and under conditions indicated without permanent deformation of louver components, noise or metal fatigue			
34		caused by louver-blade rattle or flutter, or permanent damage to fasteners and anchors. Wind pressures are			
35		considered to act normal to the face of the building.			
36		1. Wind Loads:			
37		a. Determine loads based on a uniform pressure of 20 lbf/sq. ft., acting inward or outward.			
38	С.	Louver Performance Ratings: Provide louvers complying with requirements specified, as demonstrated by testing			
39		manufacturer's stock units identical to those provided, except for length and width according to AMCA 500-L.			
40	2.2	FIXED EXTRUDED-ALUMINUM LOUVERS			
41	Α.	Horizontal, Wind-Driven-Rain-Resistant Louver, Extruded Aluminum:			
42		1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:			
43		a. Greenheck Fan Corporation.			
44		b. Ruskin; Air Distribution Technologies, Inc.; Johnson Controls, Inc.			
45		2. Louver Depth: 6 inches.			
46		3. Frame and Blade Nominal Thickness: Not less than 0.080 inch.			
47		4. Louver Performance Ratings:			
48		a. Free Area: See Mechanical for requirement.			
49		b. Air Performance: Not more than X-inch wg static pressure drop at Y-fpm free-area exhaust and/or			
50		intake velocity. See Mechanical for values.			
51		c. Wind-Driven Rain Performance: Not less than 99 percent effectiveness when subjected to a rainfall			
52		rate of 3 inches per hour and a wind speed of 29 mph at a core-area intake velocity of 300 fpm.			
53		5. AMCA Seal: Mark units with AMCA Certified Ratings Seal.			
54	В.	Vertical, Wind-Driven-Rain-Resistant Louver, Extruded Aluminum:			
55		1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:			
56		a. Greenheck Fan Corporation.			
57		b. Ruskin; Air Distribution Technologies, Inc.; Johnson Controls, Inc.			
58		2. Louver Depth: 6 inches.			

May 1	5, 2024
	3. Frame and Blade Nominal Thickness: Not less than 0.080 inch.
	4. Louver Performance Ratings:
	a. Free Area: See Mechanical for requirement.
	b. Air Performance: Not more than X-inch wg static pressure drop at Y-fpm free-area exhaust and/or
	intake velocity. See Mechanical for values.
	c. Wind-Driven Rain Performance: Not less than 99 percent effectiveness when subjected to a rainfall
	rate of 3 inches per hour and a wind speed of 29 mph at a core-area intake velocity of 300 fpm.
	5. AMCA Seal: Mark units with AMCA Certified Ratings Seal.
2.3	LOUVER SCREENS
Α.	General: Provide screen at each exterior louver.
	1. Screen Location for Fixed Louvers: Interior face.
	2. Screening Type: Bird screening.
В.	Louver Screen Frames: Same type and form of metal as indicated for louver to which screens are attached.
С.	Louver Screening for Aluminum Louvers:
	1. Bird Screening, Stainless Steel: 1/2-inch-square mesh, 0.047-inch wire.
2.4	MATERIALS
Α.	Aluminum Extrusions: ASTM B221, Alloy 6063-T5, T-52, or T6.
В.	Aluminum Sheet: ASTM B209, Alloy 3003 or 5005, with temper as required for forming, or as otherwise
	recommended by metal producer for required finish.
С.	Galvanized-Steel Sheet: ASTM A653/A653M, G60 zinc coating, mill phosphatized.
D.	Stainless Steel Sheet: ASTM A240/A240M, Type 304, No. 2B finish.
Ε.	Fasteners: Use types and sizes to suit unit installation conditions.
	 Use Phillips flat-head screws for exposed fasteners unless otherwise indicated.
	2. For fastening aluminum, use aluminum or 300 series stainless steel fasteners.
	3. For fastening galvanized steel, use hot-dip-galvanized-steel or 300 series stainless steel fasteners.
	4. For fastening stainless steel, use 300 series stainless steel fasteners.
	5. For color-finished louvers, use fasteners with heads that match color of louvers.
F.	Postinstalled Fasteners for Concrete and Masonry: Torque-controlled expansion anchors, fabricated from stainless
	steel components, with allowable load or strength design capacities calculated according to ICC-ES AC193 and
	ACI 318 greater than or equal to the design load, as determined by testing according to ASTM E488/E488M
	conducted by a qualified testing agency.
G.	Bituminous Paint: Cold-applied asphalt emulsion complying with ASTM D1187/D1187M.
2.5	FABRICATION
Α.	Fabricate frames, including integral sills, to fit in openings of sizes indicated, with allowances made for fabrication
	and installation tolerances, adjoining material tolerances, and perimeter sealant joints.
В.	Join frame members to each other and to fixed louver blades with fillet welds concealed from view, threaded
	fasteners, or both, as standard with louver manufacturer unless otherwise indicated or size of louver assembly
	makes bolted connections between frame members necessary.
2.6	ALUMINUM FINISHES
Α.	Color Anodic Finish: AAMA 611, AA-M12C22A42/A44, Class I, 0.018 mm or thicker.
	1. Color: As selected by Architect from full range of industry colors and color densities.
PART 3	EXECUTION
3.1	INSTALLATION
А.	Locate and place louvers level, plumb, and at indicated alignment with adjacent work.
В.	Use concealed anchorages where possible. Provide brass or lead washers fitted to screws where required to protect
6	metal surfaces and to make a weathertight connection.
С. Г	Provide perimeter reveals and openings of uniform width for sealants and joint fillers, as indicated.
D.	Protect unpainted gaivanized- and nonrerrous-metal surfaces that are in contact with concrete, masonry, or discipling metals from correction and columnia action by architect a basis contact with concrete, masonry, or
	ussimilar metals from corrosion and galvanic action by applying a neavy coating of bituminous paint or by
	separating surfaces with waterproof gaskets or nonmetallic hasning.
3.2	AUJUSTING Desters laware demograd during installation and construction, or an avidence remains of corrective work. If would
А.	Restore louvers damaged during installation and construction, so no evidence remains of corrective work. If results
	or restoration are unsuccessful, as determined by Architect, remove damaged units and replace with new units.
	END OF SECTION

1

2

SECTION 09 22 16 NON-STRUCTURAL METAL FRAMING

3	PART 1 - (- GENERAL		
4	1.1	SUMMARY		
5	Α.	Section Includes:		
6		1. Framing systems.		
7		2. Suspension systems.		
8	1.2	ACTION SUBMITTALS		
9	Α.	Product Data: For each product.		
10	1.3	INFORMATIONAL SUBMITTALS		
11	Α.	Product Certificates: For each type of code-compliance certification for studs and tracks.		
12	В.	Evaluation reports for high-strength steel studs and tracks, firestop tracks, post-installed anchors, and power-		
13		actuated fasteners.		
14	1.4	QUALITY ASSURANCE		
15	Α.	Code-Compliance Certification of Studs and Tracks: Provide documentation that framing members are certified		
16		according to the product-certification program of the Certified Steel Stud Association, the Steel Framing Industry		
17		Association, the Steel Stud Manufacturers Association, or the Supreme Steel Framing System Association.		
18	PART 2 - 1	PRODUCTS		
19	2.1	PERFORMANCE REQUIREMENTS		
20	Α.	Fire-Test-Response Characteristics: For fire-resistance-rated assemblies that incorporate non-load-bearing steel		
21		framing, provide materials and construction identical to those tested in assembly indicated, in accordance with		
22		ASTM E119 by an independent testing agency.		
23	В.	STC-Rated Assemblies: For STC-rated assemblies, provide materials and construction identical to those tested in		
24		assembly indicated on Drawings, in accordance with ASTM E90 and classified in accordance with ASTM E413 by an		
25		independent testing agency.		
26	2.2	FRAMING SYSTEMS		
27	Α.	Framing Members, General: Comply with ASTM C645 for conditions indicated.		
28		1. Steel Sheet Components: Comply with ASTM C645 requirements for metal unless otherwise indicated		
29		2. Protective Coating: Comply with ASTM C645; ASTM A653/A653M, G40; or coating with equivalent corrosion		
30		resistance. Galvannealed products are unacceptable.		
31		a. Coating demonstrates equivalent corrosion resistance with an evaluation report acceptable to		
32		authorities having jurisdiction.		
33	В.	Studs and Track: ASTM C645.		
34		1. Minimum Base-Steel Thickness: As required by performance requirements for horizontal deflection.		
35		2. Depth: As indicated on Drawings.		
36	С.	High-Strength Steel Studs and Tracks: Roll-formed with surface deformations to stiffen the framing members.		
37		1. Minimum Base-Steel Thickness: As required by horizontal deflection performance requirements.		
38		2. Depth: As indicated on Drawings.		
39	D.	Slip-Type Head Joints: Where indicated, provide one of the following:		
40		1. Clip System: Clips designed for use in head-of-wall deflection conditions that provide a positive attachment		
41		of studs to tracks while allowing 2-inch minimum vertical movement.		
42		2. Single Long-Leg Track System: Top track with 2-inch-deep flanges in thickness not less than indicated for		
43		studs, installed with studs friction fit into top track and with continuous bridging located within 12 inches of		
44		the top of studs to provide lateral bracing.		
45		3. Double-Track System: Top outer tracks, inside track with 2-inch-deep flanges in thickness not less than		
46		indicated for studs and fastened to studs, and outer track sized to friction-fit over inner track.		
47		4. Deflection Track: Steel sheet top track manufactured to prevent cracking of finishes applied to interior		
48		partition framing resulting from deflection of structure above; in thickness not less than indicated for studs		
49		and in width to accommodate depth of studs.		
50	Ε.	Firestop Tracks: Top track manufactured to allow partition heads to expand and contract with movement of		
51		structure while maintaining continuity of fire-resistance-rated assembly indicated; in thickness not less than		
52		indicated for studs and in width to accommodate depth of studs.		
53	F.	Flat Strap and Backing Plate: Steel sheet for blocking and bracing in length and width indicated.		
54		1. Minimum Base-Steel Thickness: 0.0296 inch.		
55	G.	Cold-Rolled Channel Bridging: Steel, 0.0538-inch minimum base-steel thickness, with minimum 1/2-inch-wide		
56		flanges.		
57		1. Depth: 1-1/2 inches.		
58		2. Clip Angle: Not less than 1-1/2 by 1-1/2 inches, 0.068-inch-thick, galvanized steel.		
		-		

	May 16, 2024				
1	Н.	Hat-Shaped, Rigid Furring Channels:			
2		1. Minimum Base-Steel Thickness: 0.0296 inch.			
3		2. Depth: As indicated on Drawings.			
4	Ι.	Resilient Furring Channels: 1/2-inch-deep, steel sheet members designed to reduce sound transmission.			
5		1. Configuration: Asymmetrical or hat shaped.			
6	J.	Cold-Rolled Furring Channels: 0.053-inch uncoated-steel thickness. with minimum 1/2-inch-wide flanges.			
7		1. Depth: As indicated on Drawings.			
8		2. Furring Brackets: Adjustable, corrugated-edge-type steel sheet with minimum uncoated-steel thickness of			
9		0.0329 inch.			
10		3. Tie Wire: ASTM A641/A641M, Class 1 zinc coating, soft temper, 0.062-inch-diameter wire, or double strand			
11		of 0.048-inch-diameter wire.			
12	К.	Z-Shaped Furring: With slotted or nonslotted web, face flange of 1-1/4 inches, wall attachment flange of 3/4 inch,			
13		minimum uncoated-steel thickness of 0.0179 inch, and depth required to fit insulation thickness indicated.			
14	2.3	SUSPENSION SYSTEMS			
15	Α.	Tie Wire: ASTM A641/A641M, Class 1 zinc coating, soft temper, 0.062-inch-diameter wire, or double strand of			
16		0.048-inch-diameter wire.			
17	В.	Hanger Attachments to Concrete:			
18		1. Post-Installed Anchors: Fastener systems with an evaluation report acceptable to authorities having			
19		jurisdiction, based on ICC-ES AC01, AC193, AC58, or AC308 as appropriate for the substrate.			
20		a. Uses: Securing hangers to structure.			
21		b. Type: Torque-controlled, expansion anchor torque-controlled, adhesive.			
22		c. Material for Interior Locations: Carbon-steel components zinc-plated to comply with ASTM B633 or			
23		ASTM F1941, Class Fe/Zn 5, unless otherwise indicated.			
24		d. Material for Exterior or Interior Locations and Where Stainless Steel Is Indicated: Alloy Group 1			
25		stainless steel bolts, ASTM F593, and nuts, ASTM F594.			
26	С.	Wire Hangers: ASTM A641/A641M, Class 1 zinc coating, soft temper, 0.16 inch in diameter.			
27	D.	Flat Hangers: Steel sheet, 1 by 3/16 inch by length indicated.			
28	E.	Carrying Channels (Main Runners): Cold-rolled, commercial-steel sheet with a base-steel thickness of 0.0538 inch			
29		and minimum 1/2-inch-wide flanges.			
30		1. Depth: 2-1/2 inches.			
31	F.	Furring Channels (Furring Members):			
32		1. Cold-Rolled Channels: 0.0538-inch uncoated-steel thickness, with minimum 1/2-inch-wide flanges, 3/4 inch			
33		deep.			
34		2. Steel Studs and Tracks: Use either conventional steel studs and tracks or embossed, high-strength steel			
35		studs and tracks.			
36		a. Minimum Base-Steel Thickness: 0.0269 inch.			
37		b. Depth: As indicated on Drawings.			
38		3. High-Strength Steel Studs and Tracks:			
39		a. Minimum Base-Steel Thickness: 0.0180 inch.			
40		b. Depth: As indicated on Drawings.			
41		4. Hat-Shaped, Rigid Furring Channels: 7/8 inch deep.			
42		a. Minimum Base-Steel Thickness: 0.0296 inch.			
43		5. Resilient Furring Channels: 1/2-inch-deep members designed to reduce sound transmission.			
44		a. Configuration: Asymmetrical or hat shaped.			
45	2.4	AUXILIARY MATERIALS			
46	Α.	General: Provide auxiliary materials that comply with referenced installation standards.			
47		1. Fasteners for Steel Framing: Of type, material, size, corrosion resistance, holding power, and other			
48		properties required to fasten steel members to substrates.			
49	В.	Isolation Strip at Exterior Walls: Provide one of the following:			
50		1. Asphalt-Saturated Organic Felt: ASTM D226/D226M, Type I (No. 15 asphalt felt), nonperforated.			
51		2. Foam Gasket: Adhesive-backed, closed-cell vinyl foam strips that allow fastener penetration without foam			
52		displacement, 1/8 inch thick, in width to suit steel stud size.			
53	PART 3 - EXECUTION				
54	3.1	INSTALLATION, GENERAL			
55	Α.	Installation Standard: ASTM C754.			
56		1. Gypsum Plaster Assemblies: Also comply with requirements in ASTM C841 that apply to framing installation.			
57		2. Portland Cement Plaster Assemblies: Also comply with requirements in ASTM C1063 that apply to framing			
58		installation.			

	IVIAY 10	, 2024					
1 2		3. Gypsum Veneer Plaster Assemblies: Also comply with requirements in ASTM C844 that apply to framing installation.					
3		 Gypsum Board Assemblies: Also comply with requirements in ASTM C840 that apply to framing installation. 					
4	В.	Install framing and accessories plumb, square, and true to line, with connections securely fastened.					
5	C.	Install supplementary framing, and blocking to support fixtures, equipment services, heavy trim, grab bars, toilet					
6		accessories, furnishings, or similar construction.					
7	D.	Install bracing at terminations in assemblies.					
8	Ε.	Do not bridge building control and expansion joints with non-load-bearing steel framing members. Frame both sides					
9		of joints independently.					
10	3.2	INSTALLATION OF FRAMING SYSTEMS					
11	Α.	Install framing system components according to spacings indicated, but not greater than spacings required by					
12		referenced installation standards for assembly types.					
13	В.	Where studs are installed directly against exterior masonry walls or dissimilar metals at exterior walls, install					
14		isolation strip between studs and exterior wall.					
15	С.	Install studs so flanges within framing system point in same direction.					
16	D.	Install tracks at floors and overhead supports. Extend framing full height to structural supports or substrates above					
17		suspended ceilings except where partitions are indicated to terminate at suspended ceilings. Continue framing					
18		around ducts that penetrate partitions above ceiling.					
19		1. Slip-Type Head Joints: Where framing extends to overhead structural supports, install to produce joints at					
20		tops of framing systems that prevent axial loading of finished assemblies.					
21		2. Door Openings: Screw vertical study at jambs to jamb anchor clips on door frames; install track section (for					
22		cripple studs) at nead and secure to jamb studs.					
23		a. Install two stude at each jaind unless otherwise indicated.					
24		b. Install cripple study at field adjacent to each jamb stud, with a minimum 1/2-min clearance from iomediate and the allow for installation of control ioint in finished assembly.					
25		Jamp studies through susponded collings and attach to underside of overhead structure.					
20		2 Other Framed Openings: Frame openings other than door openings the same as required for door openings					
27		unless otherwise indicated Install framing below sills of openings to match framing required above door					
20		heads					
30		4. Fire-Resistance-Rated Partitions: Install framing to comply with fire-resistance-rated assembly indicated and					
31		support closures and to make partitions continuous from floor to underside of solid structure.					
32		a. Firestop Track: Where indicated, install to maintain continuity of fire-resistance-rated assembly					
33		indicated.					
34		5. Sound-Rated Partitions: Install framing to comply with sound-rated assembly indicated.					
35		6. Curved Partitions:					
36		a. Bend track to uniform curve and locate straight lengths so they are tangent to arcs.					
37		b. Begin and end each arc with a stud, and space intermediate studs equally along arcs. On straight					
38		lengths of no fewer than two studs at ends of arcs, place studs 6 inches o.c.					
39	Ε.	Direct Furring:					
40		1. Screw to wood framing.					
41		2. Attach to concrete or masonry with stub nails, screws designed for masonry attachment, or powder-driven					
42		fasteners spaced 24 inches o.c.					
43	F.	Z-Shaped Furring Members:					
44		1. Erect insulation, specified in Section 07 21 00 "Thermal Insulation," vertically and hold in place with Z-					
45		shaped furring members spaced 24 inches o.c.					
46		2. Except at exterior corners, securely attach narrow flanges of furring members to wall with concrete stub					
47		nails, screws designed for masonry attachment, or powder-driven fasteners spaced 24 inches o.c.					
48		3. At exterior corners, attach wide flange of furring members to wall with short flange extending beyond					
49		corner; on adjacent wall surface, screw-attach short flange of furring channel to web of attached channel.					
50	6	At interior corners, space second member no more than 12 inches from corner and cut insulation to fit.					
51	G.	installation i olerance: install each framing member so fastening surfaces vary not more than 1/8 inch from the					
52		plane formed by faces of adjacent framing.					
55	3.3	INSTALLATION OF SUSPENSION SYSTEMS					
54 55	А.	install suspension system components according to spacings indicated, but not greater than spacings required by					
56	R	Increased installation statual us for assentibly types. Isolate suspension systems from huilding structure where they shut or are penetrated by huilding structure to					
57	ם.	nevent transfer of loading imposed by structural movement					
58	ſ	Suspend hangers from building structure as follows:					
50	с.	Suspensi nanjens nom pullum Structure us follows.					

	1VIU 10,	2024	
1		1.	Install hangers plumb and free from contact with insulation or other objects within ceiling plenum that are
2			not part of supporting structural or suspension system.
3			a. Splay hangers only where required to miss obstructions and offset resulting horizontal forces by
4			bracing, countersplaying, or other equally effective means.
5		2.	Where width of ducts and other construction within ceiling plenum produces hanger spacings that interfere
6			with locations of hangers required to support standard suspension system members, install supplemental
7			suspension members and hangers in the form of trapezes or equivalent devices.
8 9			 Size supplemental suspension members and hangers to support ceiling loads within performance limits established by referenced installation standards.
10		3.	Wire Hangers: Secure by looping and wire tying, either directly to structures or to inserts, eye screws, or
11			other devices and fasteners that are secure and appropriate for substrate, and in a manner that will not
12			cause hangers to deteriorate or otherwise fail.
13		4.	Flat Hangers: Secure to structure, including intermediate framing members, by attaching to inserts, eye
14			screws, or other devices and fasteners that are secure and appropriate for structure and hanger, and in a
15			manner that will not cause hangers to deteriorate or otherwise fail.
16		5.	Do not attach hangers to steel roof deck.
17		6.	Do not attach hangers to permanent metal forms. Furnish cast-in-place hanger inserts that extend through
18		-	forms.
19		7.	Do not attach hangers to rolled-in hanger tabs of composite steel floor deck.
20	_	8.	Do not connect or suspend steel framing from ducts, pipes, or conduit.
21	D.	Fire-R	esistance-Rated Assemblies: Wire tie furring channels to supports.
22	Ε.	Seismi	ic Bracing: Sway-brace suspension systems with hangers used for support.
23	3.4	FIELD	QUALITY CONTROL
24	Α.	Install	ation Tolerances: Install suspension systems that are level to within 1/8 inch in 12 feet measured lengthwise
25		on ead	ch member that will receive finishes and transversely between parallel members that will receive finishes.
26			END OF SECTION

SECTION 09 29 00

1		SECTION 09 29 00
2		GYPSUM BOARD
3	PART 1 - 0	GENERAL
4	1.1	SUMMARY
5	Α.	Section Includes:
6		1. Interior gypsum board.
7		2. Tile backing panels.
8		3. Sound attenuating blankets.
9 10	1.2	ACTION SUBMITTALS
10	A.	Product data.
11	В.	Shop Drawings: Show locations and installation of control and expansion joints, including plans, elevations, sections,
12	c	details of components, and attachments to other work.
13	U.	
14 15	PARI 2 - F	
15	2.1	FERFORINANCE REQUIREMENTS
17	А.	to those tested in assembly indicated in accordance with ASTM 5110 by an independent testing agency.
10	B	STC_Rated Assemblies: For STC_rated assemblies, provide materials and construction identical to those tested in
10	D.	assembly indicated in accordance with ASTM E90 and classified in accordance with ASTM E413 by an independent
20		testing agency
20	22	GYPSIIM BOARD GENERAL
22	Δ	Size: Provide maximum lengths and widths available that will minimize joints in each area and that correspond with
23	7.	support system indicated.
24	2.3	INTERIOR GYPSUM BOARD
25	 A.	Gypsum Board, Type X: ASTM C1396/C1396M.
26		1. Thickness: 5/8 inch.
27		2. Long Edges: Tapered.
28		3. Typical wall and ceiling board unless noted otherwise.
29	В.	Abuse-Resistant Gypsum Board: ASTM C1396/C1396M gypsum board, tested in accordance with
30		ASTM C1629/C1629M.
31		1. Core: 5/8 inch, Type X.
32		2. Surface Abrasion: ASTM C1629/C1629M, meets or exceeds Level 1 requirements.
33		3. Indentation: ASTM C1629/C1629M, meets or exceeds Level 1 requirements.
34		4. Soft-Body Impact: ASTM C1629/C1629M, meets or exceeds Level 1 requirements.
35		5. Long Edges: Tapered.
36		6. Mold Resistance: ASTM D3273, score of 10 as rated in accordance with ASTM D3274.
37		7. Used where called for in drawings.
38	С.	Mold-Resistant Gypsum Board: ASTM C1396/C1396M. With moisture- and mold-resistant core and paper surfaces.
39		1. Core: 5/8 inch, Type X.
40		2. Long Edges: Tapered.
41		3. Mold Resistance: ASTM D3273, score of 10 as rated in accordance with ASTM D3274.
42		4. Except surfaces to receive tile, used at wet walls and all walls/ceilings of restrooms and shower/locker
43		rooms.
44	2.4	SPECIALTY GYPSUM BOARD
45	Α.	Gypsum Board, Type C: ASTM C1396/C1396M. Manufactured to have increased fire-resistive capability.
46		 Thickness: As required by fire-resistance-rated assembly indicated on Drawings.
47		2. Long Edges: Tapered.
48		3. Used where required by UL design. See wall types.
49	2.5	TILE BACKING PANELS
50	Α.	Cementitious Backer Units: ANSI A118.9 and ASTM C1288 or ASTM C1325, with manufacturer's standard edges.
51		1. Thickness: 1/2 inch unless noted/detailed otherwise in drawings.
52		 Mold Resistance: ASTM D3273, score of 10 as rated in accordance with ASTM D3274. Used behind encoded as the state
53		3. Used behind any wall or ceiling tile.
54	2.6	I RIM ACCESSORIES
55	Α.	Interior Frim: ASTM C1047.
50		1. Inviaterial: Galvanized or aluminum-coated steel sheet, rolled zinc, plastic, or paper-faced galvanized-steel
5/		sneet.
58		2. Snapes:

	May 16	, 2024
9		a. Cornerbead.
0		b. Bullnose bead.
1		c. LC-Bead: J-shaped; exposed long flange receives joint compound.
2		d. L-Bead: L-shaped; exposed long flange receives joint compound.
		e. U-Bead: J-shaped; exposed short flange does not receive joint compound.
		f. Expansion (control) joint.
		g. Curved-Edge Cornerbead: With notched or flexible flanges.
	2.7	JOINT TREATMENT MATERIALS
	Α.	General: Comply with ASTM C475/C475M.
	В.	Joint Tape:
		1. Interior Gypsum Board: Paper.
		2. Exterior Gypsum Soffit Board: Paper.
		3. Glass-Mat Gypsum Sheathing Board: 10-by-10 glass mesh.
		4. Tile Backing Panels: As recommended by panel manufacturer.
	C.	Joint Compound for Interior Gypsum Board: For each coat, use formulation that is compatible with other
	0.	compounds applied on previous or for successive coats.
		1. Prefilling: At open joints, rounded or beveled panel edges, and damaged surface areas, use setting-type
		taping compound.
		2 Embedding and First Coat: For embedding tane and first coat on joints fasteners and trim flanges use
		setting-type taning compound
		a Use setting-type compound for installing naner-faced metal trim accessories
		3 Fill Coat: For second coat, use setting-type, sandable tonning compound
		4 Finish Coat: For third coat, use setting-type, sandable topping compound
		5 Skim Coat: For final coat of Level 5 finish use setting-type, sandable topping compound
	П	Joint Compound for Tile Backing Panels:
	υ.	1 Cementitious Backer Units: As recommended by backer unit manufacturer
	28	
	Δ	Provide auxiliary materials that comply with referenced installation standards and manufacturer's written
	73.	instructions
	в	Laminating Adhesive: Adhesive or joint compound recommended for directly adhering gynsum nanels to
	5.	continuous substrate
	C	Steel Drill Screws: ASTM C1002 unless otherwise indicated
	с.	1 Use screws complying with ASTM C954 for fastening papels to steel members from 0.033 to 0.112 inch
		thick
		2 For fastening comentitious backer units use screws of type and size recommended by panel manufacturer
	П	Sound-Attenuation Blankets: ASTM C665 Type I (blankets without membrane facing) produced by combining
	υ.	thermosetting resins with mineral fibers manufactured from glass slag wool or rock wool
		1 Fire-Resistance-Rated Assemblies: Comply with mineral-fiber requirements of assembly
	F	Acoustical Sealant: As specified in Section 07.92.19 "Acoustical Joint Sealants."
	с. F	Thermal Insulation: As specified in Section 07 32 10 "Thermal Insulation "
	г. С	Vapor Potardor: As specified in Section 07.26.00 "Vapor Potardore" if used
	DAPT 2	
	21	INSTALLATION OF DANFLS
	5.1	Evamine namely before installation. Reject namely that are wat maisture damaged, and mold damaged
	A. R	Comply with ASTM CRAD
	в. С	Comply with ASTM Co40.
	C.	1/2 inch wide spaces at these locations and trim addres with addre trim where addres of papels are expected. Scal
		inite between edges and abutting structural surfaces with acoustical scalant
	П	Joints between edges and abutting structural surfaces with acoustical sediant.
	D.	For this with back hanges intended for lasteners, attach to manning with same fasteners used for panels. Otherwise,
	5.2	Profill open joints, rounded or bougled edges, and damaged surface areas
	A.	Prefili open joints, rounded or beveled edges, and damaged surface areas.
	в.	Apply joint take over gypsum board joints, except for thim products specifically indicated as not intended to receive
	~	tape. Computer Record Finish Levels, Finish namels to levels indicate difference difference with ACTA1 CO10.
	C.	Gypsum Board Finish Levels: Finish panels to levels indicated below and in accordance with ASTM C840:
		Level 1: Celling plenum areas, concealed areas, and where indicated.
		2. Level 2: Parleis that are substrate for the of for acoustical the.
		S. LEVELS. N/A.

	Tridy 10	, 2027
117		4. Level 4: At panel surfaces that will be exposed to view unless otherwise indicated - most common
118		situation.
119		a. Primer and its application to surfaces are specified in Section 09 91 23 "Interior Painting."
120		5. Level 5: At panel surfaces that will be exposed to view with gloss or high gloss painted finish, or where
121		indicated on Drawings.
122		a. Primer and its application to surfaces are specified in Section 09 91 23 "Interior Painting."
123	D.	Glass-Mat Gypsum Sheathing Board: Finish according to manufacturer's written instructions for use as exposed
124		soffit board.
125	Ε.	Glass-Mat Faced Panels: Finish according to manufacturer's written instructions.
126	F.	Cementitious Backer Units: Finish according to manufacturer's written instructions.
127	3.3	PROTECTION
128	Α.	Protect installed products from damage from weather, condensation, direct sunlight, construction, and other
129		causes during remainder of the construction period.
130	В.	Remove and replace panels that are wet, moisture damaged, and mold damaged.
131		END OF SECTION

Page Intentionally Left Blank

SECTION 09 30 13 CEDANAIC THUNG

1		SECTION 09 30 13
2		CERAMIC TILING
3	PART 1 -	GENERAL
4	1.1	SUMMARY
5	Α.	Section Includes:
6		1. Porcelain tile.
7		2. Glazed wall tile.
8		3. Waterproof membranes.
9		4. Crack isolation membranes.
10		5. Setting material.
11		6. Grout materials.
12	В.	See Section 09 29 00 SF - Gypsum Board for tile backing panels.
13	1.2	ACTION SUBMITTALS
14	Α.	Product data.
15	В.	Shop Drawings: Show locations, plans, and elevations, of each type of tile and tile pattern. Show widths, details, and
16	2.	locations of movement joints in tile substrates and finished tile surfaces.
17	C.	Samples:
18	-	1. Full-size units of each type and composition of tile and for each color and finish required. For ceramic
19		mosaic tile in color blend patterns, provide full sheets of each color blend. For tile with aesthetic
20		classification V3 or V4, provide 12 tiles from same production run.
21		2. Assembled samples mounted on a rigid panel, with grouted joints, for each type and composition of tile and
22		for each color and finish required. Make samples at least 12 inches square, but not fewer than four tiles.
23		Use grout of type and in color or colors approved for completed Work.
24		3. Full-size units of each type of trim and accessory for each color and finish required.
25		4. Stone thresholds in 6-inch lengths.
26	1.3	INFORMATIONAL SUBMITTALS
27	Α.	Qualification Data: For Installer.
28	1.4	MAINTENANCE MATERIAL SUBMITTALS
29	Α.	Extra Stock Material: Furnish extra materials, from the same production run, to Owner that match products
30		installed and that are packaged with protective covering for storage and identified with labels describing contents.
31		1. Tile and Trim Units: Furnish quantity of full-size units equal to ONE box for each type, composition, color,
32		pattern, and size indicated.
33	1.5	QUALITY ASSURANCE
34	Α.	Installer Qualifications:
35		1. Installer employs at least one installer for Project that has completed the Advanced Certification for Tile
36		Installers (ACT) certification for installation of mud floors, mud walls, membranes, shower receptors, and
37		large format tile.
38	PART 2 -	PRODUCTS
39	2.1	PRODUCTS, GENERAL
40	Α.	ANSI Ceramic Tile Standard: Provide tile that complies with ANSI A137.1 for types, compositions, and other
41		characteristics indicated.
42	В.	ANSI Standards for Tile Installation Materials: Provide materials complying with ANSI A108.02, ANSI standards
43		referenced in other Part 2 articles, ANSI standards referenced by TCNA installation methods specified in tile
44		installation schedules, and other requirements specified.
45	2.2	PORCELAIN TILE
46	Α.	Porcelain Tile Type:
47		1. See Finish Schedule and Finish Key in drawings for selected tiles.
48		2. Certification: Tile certified by the Porcelain Tile Certification Agency.
49		3. Face Size Variation: Rectified.
50		4. Tile Color, Glaze, and Pattern: Per Finish Key.
51		5. Grout Color: As selected by Architect from manufacturer's full range if not identified in Finish Key.
52		6. Precoat with temporary protective coating.
53		7. Trim Units: Coordinated with sizes and coursing of adjoining flat tile where applicable and matching
54		characteristics of adjoining flat tile. Provide shapes as follows, selected from manufacturer's standard
55		shapes:
56		a. Base Cap: Surface bullnose, module size same as adjoining flat tile.
57		b. Wainscot Cap: Surface bullnose, module size same as adjoining flat tile.

May 16	, 2024
	c. Wainscot Cap for Flush Conditions: Regular flat tile for conditions where tile wainscot is shown flush
	with wall surface above it; same size as adjoining flat tile.
	d. External Corners: Surface bullnose, module size same as adjoining flat tile.
	e. Internal Corners: Field-butted square corners.
	f. Tapered Transition Tile: Shape designed to effect transition between thickness of tile floor an
	adjoining floor finishes of different thickness, tapered to provide reduction in thickness from
	1/2 to 1/4 inch across nominal 4-inch dimension.
2.3	GLAZED WALL TILE
Α.	Glazed Wall Tile Type:
	1. See Finish Schedule and Finish Key in drawings for selected tiles.
	2. Face Size Variation: Rectified.
	3. Life Color and Pattern: Per Finish Key.
	4. Grout Color: As selected by Architect from manufacturer's full range if not identified in Finish Key.
	5. Trim Units: Coordinated with sizes and coursing of adjoining hat the where applicable and matching characteristics of adjoining flat tile. Browide change as follows, colorted from manufacturer's standay
	shapes:
	a. Base: Straight, module size per Finish Key.
	b. Wainscot Cap: Surface bullnose, module size per Finish Key.
	c. External Corners: Surface bullnose; same size as adjoining flat tile.
	d. Internal Corners: Field-butted square corners. For coved base and cap, use angle pieces designed t
	fit with stretcher shapes.
2.4	WATERPROOF MEMBRANES
Α.	General: Manufacturer's standard product that complies with ANSI A118.10 and ANSI A118.12 and is recommended
_	by manufacturer for application indicated. Include reinforcement and accessories recommended by manufacturer.
В.	Waterproof Membrane, Sheet: Polyethylene sheet faced on one or both sides with polyester fabric.
	1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
	a. ARDEX AMERICAS.
	c MAPEL Corporation
	d Schluter Systems L P
	2. Nominal Thickness: 0.008 inch.
2.5	CRACK ISOLATION MEMBRANES
Α.	General: Manufacturer's standard product that complies with ANSI A118.12 for standard performance and
	recommended by manufacturer for application indicated. Include reinforcement and accessories recommended by
	manufacturer.
В.	Crack Isolation Membrane, Polyethylene Sheet: Polyethylene faced on both sides with polyester fabric.
	1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
	a. Custom Building Products.
	b. Laticrete International, Inc.
	c. MAPEI Corporation.
• •	d. Schluter Systems L.P.
2.6	SETTING MATERIALS
А.	Portiand Cement Montar (Thickset) Installation Materials: ANSI A108.02.
	Cleavage Membrane: Installer's option of material that complete with ANSI A108.02, paragraph 3.8. Painforcing Wire Enbric: Calvanized, wolded wire fabric, 2 by 2 inches by 0.062 inch diameter: comply wire
	ASTM A1064/A1064M excent for minimum wire size
	3 Expanded Metal Lath: Diamond-mesh lath complying with ASTM C847
	4. Latex Additive: Manufacturer's standard water emulsion, serving as replacement for part or all of gagin
	water, of type specifically recommended by latex-additive manufacturer for use with field-mixed portlar
	cement and aggregate mortar bed.
В.	Improved Modified Dry-Set Mortar (Thinset): ANSI A118.15.
	1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
	a. ARDEX Americas.
	b. Custom Building Products.
	c. H.B. Fuller Construction Products Inc. / TEC.
	d. Laticrete International, Inc.
	e. MAPEI Corporation.

	May 16	i, 2024
1 2		2. Provide prepackaged, dry-mortar mix containing dry, redispersible, vinyl acetate or acrylic additive to which only water must be added at Project site.
2 3 4		 Provide prepackaged, dry-mortar mix combined with acrylic resin or styrene-butadiene-rubber liquid-latex additive at Project site.
5 6		4. For wall applications, provide mortar that complies with requirements for nonsagging mortar in addition to other requirements in ANSI A118.15.
7	2.7	GROUT MATERIALS
8	Α.	Sand-Portland Cement Grout: ANSI A108.10, consisting of white or gray cement and white or colored aggregate as
9		required to produce color indicated.
10	В.	High-Performance Tile Grout: ANSI A118.7.
11		1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
12		a. ARDEX Americas.
13		D. Custom Building Products.
14 1 F		c. H.B. Fuller Construction Products Inc. / TEC.
15 1 <i>C</i>		a. Laticrete international, inc.
10 17	20	
17 10	2.0	Trowolable Linderlayments and Patching Compounds: Latex modified, portland coment based formulation provided
10 19 20	A.	or approved by manufacturer of tile-setting and adhesive materials for installations indicated.
20 21	ь. С	Vapor-Related Membrane. Polyethyrene sneeting, ASTM D4597, 4.0 mis tinck.
21 22	C.	hed thickness
22		1 Manufacturers: Subject to compliance with requirements, provide products by the following:
24		a. Schluter Systems L.P.
25		 Description: L-shaped unless noted otherwise in the drawings.
26		3. Material and Finish: Metallic; exposed-edge material.
27		a. Color: See finish schedule and key.
28	D.	Temporary Protective Coating: Formulated to protect exposed surfaces of tile against adherence of mortar and
29		grout; compatible with tile, mortar, and grout products and easily removable after grouting is completed without
30		damaging grout or tile.
31	Ε.	Tile Cleaner: A neutral cleaner capable of removing soil and residue without harming tile and grout surfaces,
32		specifically approved for materials and installations indicated by tile and grout manufacturers.
33	F.	Grout Sealer: Grout manufacturer's standard product for sealing grout joints that does not change color or
34		appearance of grout.
35	PART 3 -	EXECUTION
36	3.1	EXAMINATION
3/	А.	Examine substrates, areas, and conditions where the will be installed, with installer present, for compliance with
38 20		requirements for installation tolerances and other conditions affecting performance of the work.
59 40		1. Verify that substrates for setting the are firm, dry, clean, free of coatings that are incompatible with the
40 /11		and comply with flatness tolerances required by ANSI A108 01 for installations indicated
42		 Verify that concrete substrates for tile floors installed with adhesives, bonded mortar bed, or thinset mortar
43		comply with surface finish requirements in ANSI A108.01 for installations indicated.
44	B.	Proceed with installation only after unsatisfactory conditions have been corrected.
45	3.2	PREPARATION
46	Α.	Remove coatings, including curing compounds or other coatings, that are incompatible with tile-setting materials.
47	В.	Fill cracks, holes, and depressions in concrete substrates for tile floors installed with adhesives or thinset mortar
48		with trowelable leveling and patching compound specifically recommended by tile-setting material manufacturer.
49	С.	Where indicated, prepare substrates to receive waterproof membrane by applying a reinforced mortar bed that
50		complies with ANSI A108.1 and is sloped 1/4 inch per foot toward drains.
51	D.	Blending: For tile exhibiting color variations, verify that tile has been factory blended and packaged so tile units
52		taken from one package show same range of colors as those taken from other packages and match approved
53		Samples. If not factory blended, either return to manufacturer or blend tiles at Project site before installing.
54	Ε.	Substrate Flatness:
55		1. For tile shorter than 15 inches, confirm that structure or substrate is limited to variation of 1/4 inch in 10 ft.
56		trom the required plane, and no more than 1/16 inch in 12 inches when measured from tile surface high
57		points.

	Iviay 10	, 2024
1		2. For large format tile, tile with at least one edge 15 inches or longer, confirm that structure or substrate is
2		limited to 1/8 inch in 10 ft. from the required plane, and no more than 1/16 inch in 24 inches when
3		measured from tile surface high points.
4	F.	Field-Applied Temporary Protective Coating: If indicated under tile type or needed to prevent grout from staining or
5		adhering to exposed tile surfaces, precoat them with continuous film of temporary protective coating, taking care
6		not to coat unexposed tile surfaces.
7	3.3	INSTALLATION OF CERAMIC TILE SYSTEM
8	Α.	Install tile backing panels and treat joints in accordance with ANSI A108.11 and manufacturer's written instructions
9		for type of application indicated.
10	В.	Install waterproof membrane to comply with ANSI A108.13 and manufacturer's written instructions to produce
11		waterproof membrane of uniform thickness that is bonded securely to substrate.
12 13		 Allow waterproof membrane to cure and verify by testing that it is watertight before installing tile or setting materials over it.
14	С.	Install crack isolation membrane to comply with ANSI A108.17 and manufacturer's written instructions to produce
15		membrane of uniform thickness that is bonded securely to substrate.
16		1. Allow crack isolation membrane to cure before installing tile or setting materials over it.
17	D.	Install tile in accordance with TCNA's "Handbook for Ceramic, Glass, and Stone Tile Installation" for TCNA
18		installation methods specified in tile installation schedules. Comply with parts of ANSI A108 series that are
19		referenced in TCNA installation methods and specified in tile installation schedules, and apply to types of setting
20		and grouting materials used.
21	Ε.	Extend tile work into recesses and under or behind equipment and fixtures to form complete covering without
22		interruptions unless otherwise indicated. Terminate work neatly at obstructions, edges, and corners without
23		disrupting pattern or joint alignments.
24	F.	Accurately form intersections and returns. Perform cutting and drilling of tile without marring visible surfaces.
25		Carefully grind cut edges of tile abutting trim, finish, or built-in items for straight aligned joints. Fit tile closely to
26		electrical outlets, piping, fixtures, and other penetrations so plates, collars, or covers overlap tile.
27	G.	Provide manufacturer's standard trim shapes where necessary to eliminate exposed tile edges.
28	Н.	Where accent tile differs in thickness from field tile, vary setting-bed thickness so that tiles are flush.
29	I.	Jointing Pattern: Lay tile in grid pattern unless otherwise indicated. Lay out tile work and center tile fields in both
30		directions in each space or on each wall area. Lay out tile work to minimize use of pieces that are less than half of a
31		tile. Provide uniform joint widths unless otherwise indicated.
32	J.	Lay out tile wainscots to dimensions indicated or to next full tile beyond dimensions indicated.
33	К.	Movement Joints: Provide movement joints and other sealant-filled joints, including control, contraction, and
34		isolation joints, where indicated on Drawings. Form joints during installation of setting materials, mortar beds, and
35		tile. Keep joints free of dirt, debris, and setting materials prior to filling with sealants. Do not saw-cut joints after
36		installing tiles.
37		1. Where joints occur in concrete substrates, locate joints in tile surfaces directly above them.
38	L.	Metal Flooring Transitions: Install where exposed edge of tile flooring meets carpet, wood, or other flooring that
39		finishes flush with or below top of tile and no threshold is indicated.
40	M.	Grout Sealer: Apply grout sealer to cementitious grout joints in tile floors in accordance with manufacturer's written
41		instructions. As soon as sealer has penetrated grout joints, remove excess sealer and sealer from tile faces by
42		wiping with soft cloth.
43		END OF SECTION

END OF SECTION

SECTION 09 65 13

1		SECTION 09 65 13
2		RESILIENT BASE AND ACCESSORIES
3	PART 1 - 0	GENERAL
4	1.1	SUMMARY
5	Α.	Section Includes:
6		1. Thermoset-rubber base.
7		2. Rubber stair accessories.
8		3. Rubber molding accessories.
9	1.2	ACTION SUBMITTALS
10	Α.	Product Data: For each type of product.
11	В.	Samples: For each exposed product and for each color and texture specified.
12	PART 2 - F	PRODUCTS
13	2.1	THERMOSET-RUBBER BASE
14	Α.	Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be
15		incorporated into the Work include, but are not limited to the following:
16		1. Johnsonite; a Tarkett company.
17		2. Nora by Interface.
18		3. Roppe Corporation; Roppe Holding Company.
19	В.	Product Standard: ASTM F1861, Type TS (rubber, vulcanized thermoset), Group I (solid, homogeneous).
20		 Style and Location: As selected in the Finish Key in the drawings.
21	С.	Thickness: 0.125 inch.
22	D.	Height: As selected in the Finish Key in the drawings.
23	E.	Lengths: Coils in manufacturer's standard length.
24	F.	Outside Corners: Preformed.
25	G.	Inside Corners: Job formed.
26	Н.	Colors: As selected in the Finish Key in the drawings.
27	2.2	
28	Α.	Manufacturer: To match adjacent resilient floor finish or stair accessory or as noted in Finish Key.
29	В.	Profile and Dimensions: Provide rubber molding accessories at all transitions between floor finishes and where
30		otherwise noted in drawings.
31	C.	Colors and Patterns: See Finish Key for selections.
32	2.3	
33	А.	rowelable Leveling and Patching Compounds: Latex-modified, portland-cement-based or blended hydraulic-
34 25		cement-based formulation provided or approved by resilient-product manufacturer for applications indicated.
35		1. Addresives: water-resistant type recommended by resilient-product manufacturer for resilient products and
30 27	DADT 2 0	
20	2 1	
30	Δ.1	Prenare substrates according to manufacturer's written instructions to ensure adhesion of resilient products
10	R.	Concrete Substrates for Resilient Stair Accessories: Prenare horizontal surfaces according to ASTM F710
ч о //1	υ.	1 Verify that substrates are dry and free of curing compounds sealers, and hardeners
42		 Verify that substrates are dry and rice of curring compounds, scalers, and hardeners. Remove substrate coatings and other substances that are incompatible with adhesives and that contain
43		soan wax oil or silicone using mechanical methods recommended by manufacturer. Do not use solvents
44		3 Alkalinity and Adhesion Testing: Perform tests recommended by manufacturer. Proceed with installation
45		only after substrate alkalinity falls within range on pH scale recommended by manufacturer in writing, but
46		not less than 5 or more than 9 pH.
47		4. Moisture Testing: Perform tests so that each test area does not exceed 200 sq. ft., and perform no fewer
48		than three tests in each installation area and with test areas evenly spaced in installation areas.
49		a. Anhydrous Calcium Chloride Test: ASTM F1869. Proceed with installation only after substrates have
50		maximum moisture-vapor-emission rate of 3 lb of water/1000 sq. ft. in 24 hours.
51		b. Relative Humidity Test: Using in-situ probes. ASTM F2170. Proceed with installation only after
52		substrates have a maximum 75 percent relative humidity level measurement.
53	C.	Fill cracks, holes, and depressions in substrates with trowelable leveling and patching compound: remove bumps
54		and ridges to produce a uniform and smooth substrate.
55	D.	Do not install resilient products until materials are the same temperature as space where they are to be installed.
56	E.	Immediately before installation, sweep and vacuum clean substrates to be covered by resilient products.
57	3.2	RESILIENT BASE INSTALLATION
58	Α.	Comply with manufacturer's written instructions for installing resilient base.

	IVIUY 10	, 2024
1	В.	Apply resilient base to walls, columns, pilasters, casework and cabinets in toe spaces, and other permanent fixtures
2	C	Install resilient base in lengths as long as practical without gaps at seams and with tops of adjacent pieces aligned
1	с. П	Tightly adhere resilient base to substrate throughout length of each piece, with base in continuous contact with
5	D.	horizontal and vertical substrates
6	F	Nonzontal and ventical substrates.
7	E.	On masonry surfaces or other similar irregular substrates fill yoids along ton edge of resilient base with
8	1.	manufacturer's recommended adhesive filler material
9	G	Preformed Corners: Install preformed corners before installing straight pieces
10	H.	Joh-Formed Corners:
11		1. Outside Corners: Use straight pieces of maximum lengths possible and form with returns not less than 18
12		inches in length.
13		a. Form without producing discoloration (whitening) at bends.
14		2. Inside Corners: Use straight pieces of maximum lengths possible and form with returns not less than 18
15		inches in length.
16		a. Cope corners to minimize open joints.
17	3.3	RESILIENT ACCESSORY INSTALLATION
18	Α.	Comply with manufacturer's written instructions for installing resilient accessories.
19	В.	Resilient Molding Accessories: Butt to adjacent materials and tightly adhere to substrates throughout length of each
20		piece. Install reducer strips at edges of floor covering that would otherwise be exposed.
21	3.4	CLEANING AND PROTECTION
22	Α.	Comply with manufacturer's written instructions for cleaning and protecting resilient products.
23	В.	Floor Polish (when required by flooring manufacturer): Remove soil, adhesive, and blemishes from resilient stair
24		treads before applying liquid floor polish.
25		1. Apply two coat(s).
26	С.	Cover resilient products subject to wear and foot traffic until Substantial Completion.
27		END OF SECTION
2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

29

30

31

34

35

36

42 43

46

50

51

54

55

56

57

1.1

1.2

1.3

1.4

2.1

Α.

A.

Β.

Α.

Α.

Α.

Β.

C.

D.

RESILIENT SHEET FLOORING PART 1 - GENERAL SUMMARY Section Includes: Vinyl sheet flooring with backing. 1. 2. Rubber sheet flooring with backing. 3. Resilient sports flooring. **ACTION SUBMITTALS** Product Data: For each type of product. Samples: For each exposed product and for each color, texture, and pattern specified. **CLOSEOUT SUBMITTALS** Maintenance data. QUALITY ASSURANCE Installer Qualifications: An entity that employs installers and supervisors who are competent in techniques required by manufacturer for resilient sheet flooring installation and seaming method indicated. 1. Engage an installer who employs workers for this Project who are trained or certified by resilient sheet flooring manufacturer for installation techniques required. PART 2 - PRODUCTS PERFORMANCE REQUIREMENTS Fire-Test-Response Characteristics: For resilient sheet flooring, as determined by testing identical products according to ASTM E648 or NFPA 253 by a qualified testing agency. Critical Radiant Flux Classification: Class I, not less than 0.45 W/sq. cm. 1. Product Standard: ASTM F1303. Type (Binder Content): Type I, minimum binder content of 90 percent. 1. 2. Wear-Layer Thickness: Grade 1. 3. Overall Thickness: As standard with manufacturer. Backing Class: Class A (fibrous). 4. Sheet Width: As standard with manufacturer. Seamless-Installation Method: Heat welded.

SECTION 09 65 16

Ε. Other Requirements: See Finish Schedule and Finish Key for selected products.

RUBBER SHEET FLOORING WITH BACKING 32 2.2 33

- Α. Product Standard: ASTM F1860. Type: Type I, homogeneous rubber sheet floor covering with backing. 1.
 - 2. Backing: Fibrous.
 - 3. Hardness: Manufacturer's standard hardness, measured using Shore, Type A durometer per ASTM D2240.
- 37 Β. Sheet Width: As standard with manufacturer.
- 38 C. Seamless-Installation Method: Heat welded. 39
 - D. Other Requirements: See Finish Schedule and Finish Key for selected products.

40 **RESILIENT SPORTS FLOOR** 2.3 41

Provide flooring material and all associated and installation materials by or recommended by the 1. manufacturer as identified in the Finish Schedule and Finish Key. This includes all sports/field lines as identified in Finish Plan.

44 2.4 INSTALLATION MATERIALS 45

Α. Trowelable Leveling and Patching Compounds: Latex-modified, portland-cement-based or blended hydrauliccement-based formulation provided or approved by resilient sheet flooring manufacturer for applications indicated. Β. Adhesives: Water-resistant type recommended by flooring and adhesive manufacturers to suit resilient sheet

47 48 flooring and substrate conditions indicated. 49

C. Seamless-Installation Accessories:

Heat-Welding Bead: Manufacturer's solid-strand product for heat welding seams. 1. a.

Colors: As selected by Architect from manufacturer's full range.

52 D. Integral-Flash-Cove-Base Accessories: 53

- Cove Strip: 1-inch radius provided or approved by resilient sheet flooring manufacturer. 1.
- 2. Cap Strip: Square metal cap provided or approved by resilient sheet flooring manufacturer.
- 3. Corners: Metal inside and outside corners and end stops provided or approved by resilient sheet flooring manufacturer.
- Ε. Floor Polish: Provide protective, liquid floor-polish products recommended by resilient sheet flooring manufacturer.

May 16 2024

IVIAY 1	5, 2024	
PART 3	EXECUTION	
3.1	PREPARATION	
Α.	Prepare substrates according to resilient sheet flooring manufacturer's written instructions to ensure adhesion	
	resilient sneet flooring.	
в.	Concrete Substrates: Prepare according to ASTM F710.	
	Verify that substrates are dry and free of curing compounds, sealers, and hardeners. Beneve substrate costings and other substances that are incompatible with adhesives and that contain	
	2. Remove substrate coalings and other substances that are incompatible with adhesives and that contain soon way oil or silicono using mechanical methods recommended by resilient sheet flooring	
	manufacturer. Do not use solvents	
	3 Alkalinity and Adhesion Testing: Perform tests recommended by resilient sheet flooring manufacturer	
	Proceed with installation only after substrate alkalinity falls within range on pH scale recommended by	
	manufacturer in writing, but not less than 5 or more than 9 pH.	
	4. Moisture Testing: Perform tests so that each test area does not exceed 200 sq. ft., and perform no fewer	
	than three tests in each installation area and with test areas evenly spaced in installation areas.	
	a. Anhydrous Calcium Chloride Test: ASTM F1869. Proceed with installation only after substrates have	
	maximum moisture-vapor-emission rate of 3 lb of water/1000 sq. ft. in 24 hours.	
	b. Relative Humidity Test: Using in-situ probes, ASTM F2170. Proceed with installation only after	
	substrates have a maximum 75 percent relative humidity level measurement.	
С.	Fill cracks, holes, and depressions in substrates with trowelable leveling and patching compound; remove bumps	
	and ridges to produce a uniform and smooth substrate.	
D.	Do not install resilient sheet flooring until materials are the same temperature as space where they are to be	
	installed.	
	1. At least 48 hours in advance of installation, move flooring and installation materials into spaces where they	
-	will be installed.	
E.	Immediately before installation, sweep and vacuum clean substrates to be covered by resilient sheet nooring.	
·-Z	Comply with manufacturar's written instructions for installing resilient sheet flooring	
R.	Lincoll resilient sheet flooring and allow it to stabilize before cutting and fitting	
C.	Lay out resilient sheet flooring as follows:	
С.	1. Maintain uniformity of flooring direction.	
	 Minimize number of seams: place seams in inconspicuous and low-traffic areas, at least 6 inches away from 	
	parallel joints in flooring substrates.	
	3. Match edges of flooring for color shading at seams.	
	4. Avoid cross seams.	
D.	Scribe and cut resilient sheet flooring to butt neatly and tightly to vertical surfaces and permanent fixtures including	
	built-in furniture, cabinets, pipes, outlets, and door frames.	
Ε.	Extend resilient sheet flooring into toe spaces, door reveals, closets, and similar openings.	
F.	Maintain reference markers, holes, and openings that are in place or marked for future cutting by repeating on	
	resilient sheet flooring as marked on substrates. Use chalk or other nonpermanent marking device.	
G.	Install resilient sheet flooring on covers for telephone and electrical ducts and similar items in installation areas.	
	Maintain overall continuity of color and pattern between pieces of flooring installed on covers and adjoining	
	flooring. Lightly adhere flooring edges to substrates that abut covers and to cover perimeters.	
н.	Adhere resilient sheet mooring to substrates using a full spread of adhesive applied to substrate to produce a	
	completed installation without open cracks, voids, raising and puckering at joints, telegraphing of adhesive spreader	
1	Seamless Installation:	
1.	1 Heat-Welded Seams: Comply with ASTM F1516 Rout joints and heat weld with welding head to fuse	
	sections permanently into a seamless flooring installation. Prepare weld and finish seams to produce	
	surfaces flush with adioining flooring surfaces.	
J.	Integral-Flash-Cove Base: Cove resilient sheet flooring 6 inches (unless indicated otherwise in drawings) up vertical	
	surfaces. Support flooring at horizontal and vertical junction with cove strip. Butt at top against cap strip.	
	1. Install metal corners at inside and outside corners.	
К.	Floor Polish: Remove soil, adhesive, and blemishes from flooring surfaces before applying liquid floor polish.	
	1. Apply two coats if recommended by manufacturer.	
3.3	RESILIENT SPORTS FLOORING INSTALLATION	
Α.	Comply with manufacturer's written instructions for installing resilient sheet flooring.	
	END OF SECTION	

1		SECTION 09 66 23				
2		RESINOUS MATRIX TERRAZZO FLOORING				
3	PART 1 -	GENERAL				
4	1.1	SUMMARY				
5	Α.	Section includes thin-set, epoxy-resin terrazzo flooring and base (if noted in drawings) – a system that is to be				
6		ground.				
7	1.2	PREINSTALLATION MEETINGS				
8	Α.	Preinstallation Conference: Conduct conference at Project site.				
9	1.3	ACTION SUBMITTALS				
10	А.	Product Data: For each type of product.				
11	В.	Shop Drawings: Include terrazzo installation requirements. Include plans, sections, component details, and				
12	6	relationship to other work.				
13	C.	Samples: For each exposed product and for each color and texture specified.				
14	1.4	INFORMATIONAL SUBMITTALS				
15	A.	Qualification Data: For Installer.				
10	В.	Material certificates.				
10	ر. ۱۳					
10	1.5	CLOSEOUT SUBIVITTALS				
19	A. 1 C					
20	1.0	QUALITY ASSURANCE				
21	А.	1 Engage an installer who is a contractor member of NTMA				
22		 Engage an installer who is a contractor member of NTMA. Engage an installer who is certified in writing by terrazzo manufacturer as qualified to install manufacturer's 				
20		2. Engage an instance who is certified in writing by terrazzo manufacturer as qualified to instan manufacturer s				
24 25	PART 2 -					
26	2.1	PERFORMANCE REQUIREMENTS				
27	Δ	NTMA Standards: Comply with NTMA's written recommendations for terrazzo type indicated unless more stringent				
28	7	requirements are specified.				
29	2.2	FPOXY-RESIN TERRA77O				
30	 A.	Epoxy-Resin Terrazzo: Comply with manufacturer's written instructions for matrix and aggregate proportions and				
31		mixing.				
32		1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:				
33		a. Key Resin Company.				
34		b. Terrazzo & Marble Supply Companies.				
35	В.	Mix Color and Pattern: Match Architect's sample. See Finish Key for basis-of-design.				
36		a. Note: Bidder is responsible for determining aggregate type and mix of sample prior to bid. A				
37		change order for aggregate will not be entertained unless Architect changes sample.				
38	C.	Materials:				
39		1. Moisture-Vapor-Emission-Control Membrane: Two-component, high-solids, high-density, low-odor, epoxy-				
40		based membrane-forming product produced by epoxy terrazzo manufacturer that reduces moisture				
41		emission from concrete substrate to not more than 3 lb of water/1000 sq. ft. in 24 hours.				
42		2. Substrate-Crack-Suppression Membrane: Product of terrazzo-resin manufacturer, having minimum 120				
43		percent elongation potential according to ASTM D412.				
44		a. Reinforcement: Fiberglass scrim.				
45		b. Note: Bidder to assume membrane will cover 20 percent of terrazzo floor finish area.				
46		3. Primer: Manufacturer's product recommended for substrate and use indicated.				
47		4. Epoxy-Resin Matrix: Manufacturer's standard recommended for use indicated and in color required for mix				
48		indicated.				
49		5. Finishing Grout: Resin based.				
50	2.3	STRIP MATERIALS				
51	Α.	Thin-Set Divider Strips: L-type angle in depth required for topping thickness indicated.				
52		1. Material: Match Heavy-Top strips.				
53		2. Top Width: 1/8 inch.				
54		3. Application: Curved design.				
55	В.	Heavy-Top Divider Strips: L-type angle in depth required for topping thickness indicated.				
56		1. Bottom-Section Material: Matching top-section material.				
57		2. Top-Section Material: Brass.				
58		3. Top-Section Width: 1/8 inch.				

May 10	5, 2024
	4. Application: All straight lines.
С.	Control-Joint Strips: Separate, double L-type angles, positioned back to back, that match material and color of
	divider strips and in depth required for topping thickness indicated.
D.	Accessory Strips: Match divider-strip width, material, and color unless otherwise indicated. Use the following types
	of accessory strips as required to provide a complete installation:
	1. Base-bead strips for exposed top edge of terrazzo base.
	2. Edge-bead strips for exposed edges of terrazzo.
	3. Nosings for terrazzo stair treads and landings.
Ε.	Abrasive Strips: Three-line abrasive inserts at nosings. Silicon carbide or aluminum oxide, or combination of both, in
	epoxy-resin binder and set in channel.
	1. Width: 1/2 inch.
	2. Depth: As required by terrazzo thickness.
	3. Length: 4 inches less than stair width.
	4. Color: As selected by Architect from full range of industry colors.
2.4	MISCELLANEOUS ACCESSORIES
Α.	Strip Adhesive: Epoxy-resin adhesive recommended by adhesive manufacturer for this use.
В.	Anchoring Devices: Provide mechanical anchoring devices or adhesives for strip materials as recommended by
	manufacturer and as required for secure attachment to substrate.
С.	Patching and Fill Material: Terrazzo manufacturer's resinous product approved and recommended by manufacturer
	for application indicated.
D.	Joint Compound: Terrazzo manufacturer's resinous product approved and recommended by manufacturer for
	application indicated.
Ε.	Resinous Matrix Terrazzo Cleaner: Chemically neutral cleaner with pH factor between 7 and 10 that is
	biodegradable, phosphate free, and recommended by sealer manufacturer for use on terrazzo type indicated.
F.	Sealer: Slip- and stain-resistant, penetrating-type sealer that is chemically neutral; does not affect terrazzo color or
	physical properties; and is recommended by sealer manufacturer for application.
	1. Surface Friction: Not less than 0.6 according to ASTM D2047.
	2. Acid-Base Properties: With pH factor between 7 and 10.
	3. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
	a. Johnson Diversey, Plaza Plus.
PART 3	EXECUTION
3.1	PREPARATION
А.	Clean substrates of substances, including oil, grease, and curing compounds, that might impair terrazzo bond.
_	Provide clean, dry, and neutral substrate for terrazzo application.
В.	Concrete Slabs:
	1. Provide sound concrete surfaces free of laitance, glaze, efflorescence, curing compounds, form-release
	agents, dust, dirt, grease, oil, and other contaminants incompatible with terrazzo.
	2. Prepare existing concrete slabs as follows:
	a. Shot-blast surfaces with an apparatus that abrades the concrete surface, contains the dispensed
	shot within the apparatus, and recirculates the shot by vacuum pickup. Surface preparation results
	should achieve a CSP3- CSP5 profile according to International Concrete Repair Institute Guideline
	No. 03732. Grinding is preferred if acceptable in writing from manufacturer.
	b. Repair damaged and deteriorated concrete according to terrazzo manufacturer's written
	instructions.
	c. Use patching and fill material to fill holes and depressions in substrates according to terrazzo
	manufacturer's written instructions.
С.	Verify that concrete substrates are dry and moisture-vapor emissions are within acceptable levels according to
_	manufacturer's written instructions.
D.	Preinstallation Moisture Lesting:
	1. Iesting Agency: Engage a qualified testing agency to perform tests.
	2. Moisture Testing: Perform tests so that each test area does not exceed 1000 sq. ft., and perform no fewer
	than three tests in each installation area and with test areas evenly spaced in installation areas.
	a. Relative Humidity Test: Maximum 80 percent relative humidity measurement when tested
	according to ASTM F2170 using in-situ probes.
	3. Proceed with terrazzo installation only after concrete substrates pass moisture testing or after installation
-	of moisture-vapor-emission-control membrane on substrate areas that fail testing.
Ε.	Moisture-Vapor-Emission-Control Membrane: Install according to manufacturer's written instructions.
	 Install on concrete substrates that incorporate lightweight aggregates.

SPECIFICATION

1 install concrete substrates that fail preinstallation moisture testing. 2 F. Substrate-Crack-Suppression Membrane: Install to isolate and suppress substrate cracks according to manufacturer's written instructions. 3 Install membrane as ubstrate cracks with membrane material. 4 Install membrane as ubstrate cracks with membrane material. 5 Reinforce membrane with fiberglass scrim. 6 Protect other work from water and dust generated by grinding operations. Control water and dust to comply with environmental protection regulations. 10 Install control-joint strips enclosures and other suitable methods to limit water damage and dust migration and to ensure adequate ambient temperatures and ventilation conditions during installation. 11 B. EPOYR-RESIN TERRAZZO INSTALLATION 2. Protect other and control-joint Strips: 3. a. Locate divider strips in locations indicated. 4. b. Install control-joint Strips is an location strips, and install sealant in gap. 3. Accessory Strips: install with surface of abrasive strip positioned J/16 inch higher than terrazzo surface. 2. Accessory Strips: install as required to provide a complete installation. 3. Abrasive Strips: install with surface of abrasive strip positioned J/16 inch higher than terrazzo surface. 3.		May 16,	2024
2 F. Substrate-Crack-Suppression Membrane: Install to isolate and suppress substrate cracks according to manufacturer's written instructions. 3 Install membrane at substrate cracks with membrane material. 5 Install membrane at substrate cracks with membrane material. 6 Protect other work from water and dust generated by grinding operations. Control water and dust to comply with environmental protection regulations. 7 G. Protect other work from water and dust generated by grinding operations. Control water and dust to comply with environmental protection regulations. 9 I. Erect and maintain temporary enclosures and other suitable methods to limit water damage and dust metrication and to ensure adequate ambient temporatures and ventilation conditions during installation. 11 3.2 EPOXY-RESIN TERRAZZO INSTALLATION 12 A. Comply with NTMA's written recommendations for terrazzo and accessory installation. 13 B. Strip Materials: 14 1. Divider and Control-Joint Strips in locations indicated. 15 a. Locate divider strips in locations indicated. 16 b. Install strips in adhesive setting bed without voids below strips, or mechanically anchor strips as required to attach strips to substrate, as recommended by strip manufacturer. 16 Install strips in adhesive setting bed	1		2. Install concrete substrates that fail preinstallation moisture testing.
 manufacturer's written instructions. Prepare and prefil substrate cracks with membrane material. Install membrane at substrate cracks with membrane material. Install membrane with fiberglas scrim. Protect other work from water and dust generated by grinding operations. Control water and dust to comply with environmental protection regulations. Erect and maintain temporary enclosures and other suitable methods to limit water damage and dust migration and to ensure adequate ambient temperatures and ventilation conditions during installation. EPOY-RESIN TERRAZ20 INSTALLATION A. Comply with NTMA's written recommendations for terrazzo and accessory installation. B. Strip Materials: L. Ecct and maintain temporary enclosures and other suitable methods to limit water damage and dust migration and to ensure adequate ambient temperatures and ventilation conditions during installation. EPOY-RESIN TERRAZ20 INSTALLATION A. Comply with NTMA's written recommendations for terrazzo and accessory installation. B. Strip Materials:	2	F.	Substrate-Crack-Suppression Membrane: Install to isolate and suppress substrate cracks according to
4 1. Prepare and prefil substrate cracks with membrane material. 5 2. Install membrane as substrate cracks in areas to receive terrazzo. 6 9 Protect other work from water and dust generated by grinding operations. Control water and dust to comply with environmental protection regulations. 7 6. Protect other work from water and dust generated by grinding operations. Control water and dust to comply with environmental protection regulations. 9 1. Erect and maintain temporary enclosures and other suitable methods to limit water damage and dust migration and to ensure adequate ambient temporatures and ventilation conditions during installation. 11 3.2 EPOXY-RESIN TERRAZZO INSTALLATION 12 A. Comply with NTMA's written recommendations indicated. 13 B. Strip Materials: 14 1. Install control-joint strips back to back and directly above concrete-slab control joints. 15 a. Locate divider strips in locations indicated. 16 b. Install control-joint strips with 1/4-inch gap between strips, and install sealant in gap. 17 c. Accessory Strips: install as required to provide a complete installation. 18 d. Install strips in adhesive setting between strips. 19 c.	3		manufacturer's written instructions.
 Install membrane at substrate cracks in areas to receive terrazzo. Install membrane at substrate cracks in areas to receive terrazzo. Install membrane at substrate cracks in areas to receive terrazzo. Protect other work from water and dust generated by grinding operations. Control water and dust to comply with environmental protection regulations. Protect and maintain temporary enclosures and other suitable methods to limit water damage and dust migration and to ensure adequate ambient temperatures and ventilation conditions during installation. JO EPOV-RESIN TERRAZZO INSTALLATION A Comply with NTMA's written recommendations for terrazzo and accessory installation. B. Strip Materials: Divider and Control-Joint Strips: Divider and Control-Joint Strips back to back and directly above concrete-slab control joints. C. Install control-Joint strips back to back and directly above concrete-slab control joints. C. Install control-Joint strips back to back and directly above concrete-slab control joints. Accessory Strips: Install as required to provide a complete installation. Jo abraise Strips. Install as required to provide a complete installation. Accessory Strips: Install as required of abravier strip positioned J/16 inch higher than terrazzo surface. Place, rough grind, grout, cure grout, fine grind, and finish terrazzo acording to manufacturer's written instructions. Installed Thickness: 3/8 inch nominal. T. Installed Toleraace Finishing: Ensure that matrix components and fluids from grinding operations do not stain terrazzo by reacting with divider and control-joint strips. G. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction	4		 Prepare and prefill substrate cracks with membrane material.
 A Reinforce membrane with fiberglass scrim. Protect other work from water and dust generated by grinding operations. Control water and dust to comply with environmental protection regulations. I. Frect and maintain temporary enclosures and other suitable methods to limit water damage and dust migration and to ensure adequate ambient temperatures and ventilation conditions during installation. 3.2 PPOXY-RESIN TERRAZZO INSTALLATION A. Comply with NTMA's written recommendations for terrazzo and accessory installation. 3.4 Comply with NTMA's written recommendations for terrazzo and accessory installation. 3.5 Comply with NTMA's written recommendations for terrazzo and accessory installation. 4. 1. Divider and Control-joint Strips back to back and directly above concrete-sila control joints. 5. 1 Install control-joint strips back to back and directly above concrete-sila control joints. 6. Install control-joint strips with 1/4-inch gap between strips, and install sealant in gap. 6. Install control-joint strips to substrate, as recommended by strip manufacturer. 7. Accessory Strips: Install with surface of abraive strip positioned 1/16 inch higher than terrazzo surface. 7. Accessory Strips: Install with surface of abraive strip positioned 1/16 inch higher than terrazzo surface. 7. Apply primer to terrazzo substrates according to manufacturer's written instructions. 7. Intralled Thickness: 3/8 inch nominal. 7. Terrazzo Finishing: Ensure that matrix components and fluids from grinding operations do not stain terrazzo by reacting with divider and control-joint strips. 8. Brough Grinding: Grinding With 2/4-grin of finer stones or with comparable diamond abrasives. Follow initial grind with 6/280-grit stones or with comparable diamond abrasives. Follow initial grind with 	5		2. Install membrane at substrate cracks in areas to receive terrazzo.
G. Protect other work from water and dust generated by grinding operations. Control water and dust to comply with environmental protection regulations. 9 1. Erect and maintain temporary enclosures and other suitable methods to limit water damage and dust migration and to ensure adequate ambient temperatures and ventilation conditions during installation. 10 3.2 EPOY-RESIN TERRAZ2O INSTALLATION 12 A. Comply with NTMA's written recommendations for terrazzo and accessory installation. 13 B. Strip Materials: 14 1. Divider and Control-Joint Strips: in locations indicated. 15 a. Locate divider strips in adcatos setting bed without voids bed wes strips, and innancially anchor strips as required to attach strips to substrate, as recommended by strip manufacturer. 16 . Install control-joint strips with 1/4-inch gap between strips, and install sealant in gap. 17 . . Accessory Strips: Install as required to provide a complete installation. 18 . . Accessory Strips: Install with water ce of abrasive strip positioned 1/16 inch higher than terrazzo surface. 16 . . Natavise Strips. Install with Mider and control-joint strips. 17 . . Accessory Strips. Install with Withot Acce of abrasive strip positioned 1/16 inch higher than terrazz	6		3. Reinforce membrane with fiberglass scrim.
 environmental protection regulations. Erect and maintain temporary enclosures and other suitable methods to limit water damage and dust migration and to ensure adequate ambient temperatures and ventilation conditions during installation. 3.2 FPOXY-RESIN TERRAZCO INSTALLATION A. Comply with NTMA's written recommendations for terrazzo and accessory installation. B. Strip Materials: I. Divider and Control-Joint Strips: a. Locate divider strips in locations indicated. b. Install control-Joint Strips but 1/4 honk pag between strips, and install sealant in gap. d. Install scientori-Joint Strips but 1/4 honk pag between strips, or mechanically anchor strips as required to attach strips to substrate, as recommended by strip nanufacturer. C. Apply prime to terrazzo substrate saccording to manufacturer's written instructions. Place, rough grind, grout, cure grout, fine grind, and finish terrazzo according to manufacturer's written instructions. I. Installed Thickness: 3/8 inch nominal. C. Terrazzo Finishing: Ensure that matrix components and fluids from grinding operations do not stain terrazzo by reacting with 60/80-grit stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. Grouting: Before grouting, clean terrazzo withwater, rinse, and allow to dry. Apply and cure epoxy grout. C. Hinstall attrino in diraction circuid with 24-grit or fine grinding out terrazzo flooring is installed. Grouting: Before grouting, clean terrazzo with water, rinse, and allow to dry. Apply and cure epoxy grout. Grouting: Before grouting, clean terrazzo with accessor with comparable diamond abrasives until grout is removed from surface.	7	G.	Protect other work from water and dust generated by grinding operations. Control water and dust to comply with
9 1. Erect and maintain temporary enclosures and other suitable methods to limit water damage and dust migration and to ensure adequate ambient temperatures and ventilation conditions during installation. 10 3.2 EPOXY-RESIN TERRAZIO INSTALLATION 12 A. Comply with NTMA's written recommendations for terrazzo and accessory installation. 13 B. Strip Materials: 14 1. Divider and Control-Joint Strips: 15 a. Locate divider strips in locations indicated. 16 b. Install control-Joint strips back to back and directly above concrete-slab control joints. 17 c. Install control-Joint strips back to back and directly above concrete-slab control joints. 18 d. Install control-Joint strips to substrate, as recommended by strip manufacturer. 19 3. Abraive Strips: Install as required to provide a complete installation. 10 3. Abraive Strips: Install as required to provide a complete installation. 11 1. Installed Thicknees: 3/8 inch nominal. 12 2. Terazzo Finishing: Ensure that matrix components and fluids from grinding operations do not stain terrazzo initial grind with 6/30 cyrit stones or with comparable diamond abrasives. Follow initial grind with 6/30 cyrit stones or with comparable diamond abrasives. 13 Abraive Strips: Install as restricted. Grind with 20-grit to finer stones or with comparable diamond abrasives.	8		environmental protection regulations.
10 migration and to ensure adequate ambient temperatures and ventilation conditions during installation. 11 8.2 FOXV-RESINTERRAZZO INSTALLATION 13 8. Strip Materials: 14 1. Divider and control-Joint Strips: 15 1. Divider and control-Joint Strips: 16 1. Divider and control-Joint strips back to back and directly above concrete-slab control Joints. 17 2. Install control-Joint strips with 1/4-Inch gap between strips, and install sealant in gap. 18 3. Install control-Joint strips with 1/4-Inch gap between strips, or mechanically anchor strips as required to attack strips to substrate, as recommended by strip manufacturer. 20 2. Accessory Strips: Install as required to provide a complete installation. 21 2. Accessory Strips: Install as required to provide a complete installation. 23 D. Place, rough grind, grout, cure grout, fine grind, and finish terrazzo according to manufacturer's written instructions. 23 D. Installed Thickness: 3/8 inch nominal. 24 1. Installed Thickness: 3/8 inch nominal. 25 2. Terrazzo Finishing: Ensure that matrix components and fluids from grinding operations do not stain terrazzo upractin with divider and control-joint strips. <td>9</td> <td></td> <td>1. Erect and maintain temporary enclosures and other suitable methods to limit water damage and dust</td>	9		1. Erect and maintain temporary enclosures and other suitable methods to limit water damage and dust
11 3.2 FPOXY-RESIN TERRAZZO INSTALLATION 12 A. Comply with NTMA's written recommendations for terrazzo and accessory installation. 12 A. Strip Materials: 14 1. Divider and Control-Joint Strips: 15 a. Locate divider strips in locations indicated. 16 b. Install control-joint strips with 1/4-inch gap between strips, and install sealant in gap. 17 c. Install control-joint strips with 1/4-inch gap between strips, and install sealant in gap. 18 d. Install control-joint strips with 1/4-inch gap between strips, and install sealant in gap. 19 - Accessory Strips: Install as required to provide a complete installation. 20 2. Accessory Strips: Install as required to provide a complete installation. 21 3. Abrasive Strips: Install as required to provide a complete installation. 22 C. Apply primer to terrazzo substrates according to manufacturer's written instructions. 23 D. Place, rough grind, grout, cure grout, fine grind, and finish terrazzo according to manufacturer's written instructions. 24 Install ed Thickness: 3/8 inch nominal. Install ed Thickness: 3/8 inch nominal. 25 Install actin Thishigr:	10		migration and to ensure adequate ambient temperatures and ventilation conditions during installation.
12 A. Comply with NTMA's written recommendations for terrazzo and accessory installation. 13 B. Strip Materials: 14 1. Divider and Control-Joint Strips: a. Locate divider strips in locations indicated. b. Install control-joint strips back to back and directly above concrete-slab control joints. c. Install control-joint strips with 1/4-inch gap between strips, and install sealant in gap. 1. Install strips in adhesive setting bed without voids below strips, or mechanically anchor strips as required to attach strips to substrates, as recommended by strip manufacturer. 2. Accessory Strips: Install as required to provide a complete installation. 3. Abrasive Strips: Install with surface of abrasive strip positioned 1/16 inch higher than terrazzo surface. Apply primer to terrazzo substrates according to manufacturer's written instructions. 1. Installed Thickness: 3/8 inch nominal. 2. Terrazzo Finishing: Ensure that matrix components and fluids from grinding operations do not stain terrazzo by reacting with floi/Go.grit stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. Follow initial grind with	11	3.2	EPOXY-RESIN TERRAZZO INSTALLATION
13 B. Strip Materials: 14 1. Divider and Control-Joint Strips: 15 a. Locate divider strips in locations indicated. 16 b. Install control-Joint strips back to back and directly above concrete-slab control joints. 17 c. Install control-Joint strips back to back and directly above concrete-slab control joints. 18 d. Install strips in adhesive setting bed without voids below strips, or mechanically anchor strips as required to attach strips to substrate, as recommended by strip manufacturer. 20 2. Accessory Strips: Install as required to provide a complete installation. 21 3. Abrasive Strips: Install are required to provide a complete installation. 22 C. Apply primer to terrazzo substrates according to manufacturer's written instructions. 23 D. Place, rough grind, grout, cure grout, fine grind, and finish terrazzo according to manufacturer's written instructions. 24 installed Thickness: 3/8 inch nominal. 25 1. Installed Thickness: 3/8 inch nominal. 26 C. Fine Grinding: Grind with 24-grit or finer stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. 26 C. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction traffic through are is restricted. Grind with 20-grit stones or with comparable diamond abrasives until grout is moved from s	12	Α.	Comply with NTMA's written recommendations for terrazzo and accessory installation.
14 1. Divider and Control-Joint Strips: 15 a. Locate divider strips in locations indicated. 16 b. Install control-joint strips back to back and directly above concrete-slab control joints. 17 c. Install control-joint strips with 1/4-inch gap between strips, and install sealant in gap. 18 d. Install control-joint strips with 1/4-inch gap between strips, or mechanically anchor strips as required to attach strips to substrate, as recommended by strip manufacturer. 20 2. Accessory Strips: Install as required to provide a complete installation. 21 3. Abrasive Strips: Install with surface of abrasive strip positioned 1/16 inch higher than terrazzo surface. 22 C. Apply primer to terrazzo substrates according to manufacturer's written instructions. 23 D. Place, rough grind, grout, cure grout, fine grind, and finish terrazzo according to manufacturer's written instructions. 24 install control-joint strips. a. Rough Grinding: Grind with 24-grit or finer stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. 26 . Fine Grinding/Polishin: Delay fine grinding until heavy trade work is complete and construction traffic through area is restricted. Grind with	13	В.	Strip Materials:
 a. Locate divider strips in locations indicated. b. Install control-joint strips back to back and directly above concte-slab control joints. c. Install control-joint strips back to back and directly above concte-slab control joints. d. Install control-joint strips back to back and directly above concte-slab control joints. d. Install strips in adhesive setting bed without voids below strips, or mechanically anchor strips as required to attach strips to substrate, as recommended by strip manufacturer. 2. Accessor strips: Install as required to provide a complete installation. 2. Accessor strips: Install are required to provide a complete installation. 2. Accessor strips: Install are required to provide a complete installation. 2. Accessor strips: Install are required to provide a complete installation. 2. Accessor strips: Install are required to provide a complete installation. 2. Accessor strips: Install are required to provide a complete installation. 3. Abrasive Strips: Install are required to provide a complete installation. 2. Accessor strips: Install are required to provide a complete installation. 2. Terrazzo finishing: Ensure that matrix components and fluids from grinding operations do not stain terrazzo up reacting with divider and control-joint strips. 3. Rough Grinding: Grind with 24-grit or finer stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. 3. Grouting: Before grouting, clean terrazzo with water, rinse, and allow to dry. Apply and cure epoxy grout. 4. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction traffic through area is restricted. Grind with 200-grit stones or with comparable diamond abrasives. 4. Installatin finish poured-in-place terraz	14		1. Divider and Control-Joint Strips:
 b. Install control-joint strips back to back and directly above concrete-slab control joints. c. Install control-joint strips with 1/4-inch gap between strips, and install selant in gap. d. Install strips in andhesive setting bed without voids below strips, or mechanically anchor strips as required to attach strips to substrate, as recommended by strip manufacturer. Accessory Strips: Install as required to provide a complete installation. Accessory Strips: Install with surface of abrasive strip positioned 1/16 inch higher than terrazzo surface. Apply primer to terrazzo substrates according to manufacturer's written instructions. Place, rough grind, grout, cure grout, fine grind, and finish terrazzo according to manufacturer's written instructions. Installed Thickness: 3/8 inch nominal. Terrazzo Finishing: Ensure that matrix components and fluids from grinding operations do not stain terrazzo by reacting with 60/80-grit stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. Grouting: Before grouting, clean terrazzo with water, rinse, and allow to dry. Apply and cure epoxy grout. C. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction traffic through area is restricted. Grind with 200-grit stones or with comparable diamond abrasives until grout is removed from surface. Install and finish poured-in-place terrazzo stars at the same time the adjacent terrazzo flooring is installed. F. Install and finish poured-in-place terrazzo as at the same time the adjacent terrazzo flooring is installed. G. Cut out and replace to match adjacent terrazzo, or repair panels according to NTMA's written recommendations and manufacturer's written instructions.<!--</td--><td>15</td><td></td><td>a. Locate divider strips in locations indicated.</td>	15		a. Locate divider strips in locations indicated.
17 c. Install control-joint strips with 1/4-inch gap between strips, and install sealant in gap. 18 d. Install strips in adhesive setting bed without voids below strips, or mechanically anchor strips as required to attach strips to substrate, as recommended by strips anufacturer. 20 2. Accessory Strips: Install as required to provide a complete installation. 21 3. Abrasive Strips: Install with surface of abrasive strip positioned 1/16 inch higher than terrazo subface. 22 C. Apply primer to terrazo substrates according to manufacturer's written instructions. 23 D. Place, rough grind, grout, cure grout, fine grind, and finish terrazzo according to manufacturer's written instructions. 24 instructions. a. Rough Grinding: Grind with 24-grit or finer stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. 28 a. Rough Grinding: Crind with 24-grit or finer stones or with comparable diamond abrasives. 29 . Grouting: Before grouting, clean terrazzo substrate. 24 . . Install and finish poured-in-place terrazzo according to find with 200-grit stones or with comparable diamond abrasives. 20 21 	16		b. Install control-joint strips back to back and directly above concrete-slab control joints.
 Install strips in adhesive setting bed without voids below strips, or mechanically anchor strips as required to attach strips to substrate, as recommended by strip manufacturer. 2. Accessory Strips: Install as required to provide a complete installation. 3. Abrasive Strips: Install with surface of abrasive strip positioned 1/16 inch higher than terrazzo surface. 2. Apply primer to terrazzo substrates according to manufacturer's written instructions. D. Place, rough grind, grout, cure grout, fine grind, and finish terrazzo according to manufacturer's written instructions. 2. Terrazzo Finishing: Ensure that matrix components and fluids from grinding operations do not stain terrazzo by reacting with divider and control-joint strips. a. Rough Grinding: Grind with 24-grit or finer stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. b. Grouting: Before grouting, clean terrazzo with water, rinse, and allow to dry. Apply and cure epoxy grout. c. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction traffic through area is restricted. Grind with 200 grit stones or with comparable diamond abrasives. a. Install and finish poured-in-place terrazzo suase at the same time the adjacent terrazzo flooring is installed. E. Install and finish poured-in-place terrazzo sate at the same time the adjacent terrazzo flooring is installed. G. Cut out and replace tor match adjacent terrazzo, or repair panels according to NTMA's written recommendations and manufacturer's written instructions; maturet adjacent terrazzo flooring is installed. G. Cut out and replace terrazzo areas that evidence lack of bond with substrate. Cut o	17		c. Install control-joint strips with 1/4-inch gap between strips, and install sealant in gap.
19 required to attach strips to substrate, as recommended by strip manufacturer. 20 2. Accessory Strips: Install as required to provide a complete installation. 21 3. Abrasive Strips: Install with surface of abrasive strip positioned 1/16 inch higher than terrazzo surface. 23 D. Place, rough grind, grout, cure grout, fine grind, and finish terrazzo according to manufacturer's written instructions. 24 Installed Thickness: 3/8 inch nominal. 25 1. Installed Thickness: 3/8 inch nominal. 26 2. Terrazzo Finishing: Ensure that matrix components and fluids from grinding operations do not stain terrazzo by reacting with divider and control-joint strips. 28 a. Rough Grinding: Grind with 24-grit or finer stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. 30 b. Grouting: Before grouting, clean terrazzo with water, rinse, and allow to dry. Apply and cure epoxy grout. 31 Installation Tolerance: Limit variation in terrazzo surface from level to 1/4 inch in 10 feet; noncumulative. 34 1. Installation Tolerance: Limit variation in terrazzo surface from level to 1/4 inch in 10 feet; noncumulative. 35 3. Installation Tolerance: Limit variation in terrazzo surface from level to 1/4 inch in 10 feet; noncumulative. 35 G. Cut out and replace terrazzo areas that evidence lack of bond with substrate. Cut out terrazzo areas in panels defined	18		d. Install strips in adhesive setting bed without voids below strips, or mechanically anchor strips as
 Accessory Strips: Install as required to provide a complete installation. Abrasive Strips: Install with surface of abrasive strip positioned 1/16 inch higher than terrazzo surface. Apply primer to terrazzo substrates according to manufacturer's written instructions. Place, rough grind, grout, cure grout, fine grind, and finish terrazzo according to manufacturer's written instructions. Installed Thickness: 3/8 inch nominal. Terrazzo Finishing: Ensure that matrix components and fluids from grinding operations do not stain terrazzo by reacting with divider and control-joint strips. Rough Grinding: Grind with 24-grit or finer stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. Follow grout. Grouting: Before grouting, clean terrazzo with water, rinse, and allow to dry. Apply and cure epoxy grout. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction traffic through area is restricted. Grind with 200-grit stones or with comparable diamond abrasives until grout is removed from surface. Install and finish poured-in-place terrazzo baris at the same time the adjacent terrazzo flooring is installed. Cut out and replace terrazzo areas that evidence lack of bond with substrate. Cut out terrazzo areas in panels defined by strips and replace to match adjacent terrazzo, foroning is installed. Remove grinding dust from installation and adjacent areas. Wash surfaces with water and allow them to dry thoroughly. Sealing: Sealing: Sealing: Sealing: Sealing: Sealing: Sealing: Sealing: Sealing: Sealing: Sealing:	19		required to attach strips to substrate, as recommended by strip manufacturer.
 Abrasive Strips: Install with surface of abrasive strip positioned 1/16 inch higher than terrazzo surface. Apply primer to terrazzo substrates according to manufacturer's written instructions. Place, rough grind, grout, cure grout, fine grind, and finish terrazzo according to manufacturer's written instructions. Installed Thickness: 3/8 inch nominal. Terrazzo Finishing: Ensure that matrix components and fluids from grinding operations do not stain terrazzo by reacting with divider and control-joint strips. Rough Grinding: Grind with 24-grit or finer stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. Follow grout. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction traffic through area is restricted. Grind with 200-grit stones or with comparable diamond abrasives until grout is removed from surface. Installation Tolerance: Limit variation in terrazzo stars at the same time the adjacent terrazzo flooring is installed. Lu out and replace terrazzo areas that evidence lack of bond with substrate. Cut out terrazzo areas in panels defined by strips and replace to match adjacent terrazzo, or repair panels according to NTMA's written instructions; and proved surfaces with cleaner according to NTMA's written recommendations, as approved by Architect. H. Cleaning: Sealing: Seal surfaces according to NTMA's written recommendations. Sealing: Apply sealer according to NTMA's written recommendations. 	20		2. Accessory Strips: Install as required to provide a complete installation.
22 C. Apply primer to terrazzo substrates according to manufacturer's written instructions. 23 D. Place, rough grind, grout, cure grout, fine grind, and finish terrazzo according to manufacturer's written instructions. 24 Installed Thickness: 3/8 inch nominal. Installed Thickness: 3/8 inch nominal. 26 2. Terrazzo Finishing: Ensure that matrix components and fluids from grinding operations do not stain terrazzo by reacting with divider and control-joint strips. 28 a. Rough Grinding: Grind with 24-grit or finer stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. 30 b. Grouting: Before grouting, clean terrazzo with water, rinse, and allow to dry. Apply and cure epoxy grout. 31 c. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction traffic through area is restricted. Grind with 200-grit stones or with comparable diamond abrasives until grout is removed from surface. 34 Installation Tolerance: Limit variation in terrazzo surface from level to 1/4 inch in 10 feet; noncumulative. 35 3. Installation Tolerance: Limit variation in terrazzo flooring is installed. 37 F. Install and finish poured-in-place terrazzo base at the same time the adjacent terrazzo flooring is installed. 38 G. Cu t out and replace terrazzo areas that evidenc	21		3. Abrasive Strips: Install with surface of abrasive strip positioned 1/16 inch higher than terrazzo surface.
 D. Place, rough grind, grout, cure grout, fine grind, and finish terrazzo according to manufacturer's written instructions. Installed Thickness: 3/8 inch nominal. Terrazzo Finishing: Ensure that matrix components and fluids from grinding operations do not stain terrazzo by reacting with divider and control-joint strips. Rough Grinding: Grind with 24-grit or finer stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. Follow grout. Grouting: Before grouting, clean terrazzo with water, rinse, and allow to dry. Apply and cure epoxy grout. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction traffic through area is restricted. Grind with 200-grit stones or with comparable diamond abrasives. Installation Tolerance: Limit variation in terrazzo surface from level to 1/4 inch in 10 feet; noncumulative. Install and finish poured-in-place terrazzo stairs at the same time the adjacent terrazzo flooring is installed. Install and finish poured-in-place terrazzo stairs at the same time the adjacent terrazzo flooring to NTMA's written recommendations, as approved by Architect. H. Cleaning: Remove grinding dust from installation and adjacent areas. Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rinse surfaces with water and allow them to dry thoroughly. Sealing: Apply sealer according to NTMA's written recommendations. Apply sealer according to NTMA's written instructions. 	22	C.	Apply primer to terrazzo substrates according to manufacturer's written instructions.
 instructions. Installed Thickness: 3/8 inch nominal. Terrazzo Finishing: Ensure that matrix components and fluids from grinding operations do not stain terrazzo by reacting with divider and control-joint strips. Rough Grinding: Grind with 24-grit or finer stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. Grouting: Before grouting, clean terrazzo with water, rinse, and allow to dry. Apply and cure epoxy grout. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction traffic through area is restricted. Grind with 200-grit stones or with comparable diamond abrasives. Install and finish poured-in-place terrazzo stairs at the same time the adjacent terrazzo flooring is installed. Install and finish poured-in-place terrazzo base at the same time the adjacent terrazzo flooring is installed. Gut out and replace to match adjacent terrazzo, or repair panels according to NTMA's written recommendations, as approved by Architect. Remove grinding dust from installation and adjacent areas. Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rinse surfaces with water and allow them to dry thoroughly. Sealing: Apply sealer according to NTMA's written instructions. END OF SECTION 	23	D.	Place, rough grind, grout, cure grout, fine grind, and finish terrazzo according to manufacturer's written
 1. Installed Thickness: 3/8 inch nominal. 2. Terrazzo Finishing: Ensure that matrix components and fluids from grinding operations do not stain terrazzo by reacting with divider and control-joint strips. a. Rough Grinding: Grind with 24-grit or finer stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. b. Grouting: Before grouting, clean terrazzo with water, rinse, and allow to dry. Apply and cure epoxy grout. c. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction traffic through area is restricted. Grind with 200-grit stones or with comparable diamond abrasives until grout is removed from surface. 3. Installation Tolerance: Limit variation in terrazzo surface from level to 1/4 inch in 10 feet; noncumulative. F. Install and finish poured-in-place terrazzo stairs at the same time the adjacent terrazzo flooring is installed. G. Cut out and replace terrazzo areas that evidence lack of bond with substrate. Cut out terrazzo areas in panels defined by strips and replace to match adjacent terrazzo, or repair panels according to NTMA's written recommendations, as approved by Architect. H. Cleaning: Remove grinding dust from installation and adjacent areas. Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rinse surfaces with water and allow them to dry thoroughly. I. Sealing: Apply sealer according to NTMA's written recommendations. Apply sealer according to SECTION 	24		instructions.
 26 2. Terrazzo Finishing: Ensure that matrix components and fluids from grinding operations do not stain terrazzo by reacting with divider and control-joint strips. 28 a. Rough Grinding: Grind with 24-grit or finer stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. 30 b. Grouting: Before grouting, clean terrazzo with water, rinse, and allow to dry. Apply and cure epoxy grout. 32 c. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction traffic through area is restricted. Grind with 200-grit stones or with comparable diamond abrasives until grout is removed from surface. 33 3. Installation Tolerance: Limit variation in terrazzo surface from level to 1/4 inch in 10 feet; noncumulative. 36 E. Install and finish poured-in-place terrazzo tairs at the same time the adjacent terrazzo flooring is installed. 37 F. Install and finish poured-in-place terrazzo base at the same time the adjacent terrazzo flooring is installed. 38 G. Cut out and replace terrazzo areas that evidence lack of bond with substrate. Cut out terrazzo areas in panels defined by strips and replace to match adjacent terrazzo, or repair panels according to NTMA's written recommendations, as approved by Architect. 41 H. Cleaning: 1. Remove grinding dust from installation and adjacent areas. 2. Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rinse surfaces according to NTMA's written recommendations. 43 44 45 1. Sealing: 46 1. Seal surfaces according to NTMA's written recommendations. 48 49 40 41 41 42 43 44 44 45 45 46 47 <li< td=""><td>25</td><td></td><td>1. Installed Thickness: 3/8 inch nominal.</td></li<>	25		1. Installed Thickness: 3/8 inch nominal.
 by reacting with divider and control-joint strips. a. Rough Grinding: Grind with 24-grit or finer stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. b. Grouting: Before grouting, clean terrazzo with water, rinse, and allow to dry. Apply and cure epoxy grout. c. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction traffic through area is restricted. Grind with 200-grit stones or with comparable diamond abrasives until grout is removed from surface. 3. Installation Tolerance: Limit variation in terrazzo surface from level to 1/4 inch in 10 feet; noncumulative. E. Install and finish poured-in-place terrazzo stairs at the same time the adjacent terrazzo flooring is installed. F. Install and finish poured-in-place terrazzo base at the same time the adjacent terrazzo flooring is installed. G. Cut out and replace terrazzo areas that evidence lack of bond with substrate. Cut out terrazzo areas in panels defined by strips and replace to match adjacent terrazzo, or repair panels according to NTMA's written recommendations, as approved by Architect. H. Cleaning: Kaenove grinding dust from installation and adjacent areas. Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rinse surfaces with water and allow them to dry thoroughly. I. Sealing: Kael surfaces according to NTMA's written recommendations. Apply sealer according to SECTION 	26		2. Terrazzo Finishing: Ensure that matrix components and fluids from grinding operations do not stain terrazzo
 a. Rough Grinding: Grind with 24-grit or finer stones or with comparable diamond abrasives. Follow initial grind with 60/80-grit stones or with comparable diamond abrasives. b. Grouting: Before grouting, clean terrazzo with water, rinse, and allow to dry. Apply and cure epoxy grout. c. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction traffic through area is restricted. Grind with 200-grit stones or with comparable diamond abrasives. c. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction traffic through area is restricted. Grind with 200-grit stones or with comparable diamond abrasives until grout is removed from surface. 3. Installation Tolerance: Limit variation in terrazzo surface from level to 1/4 inch in 10 feet; noncumulative. E. Install and finish poured-in-place terrazzo base at the same time the adjacent terrazzo flooring is installed. F. Install and finish poured-in-place terrazzo base at the same time the adjacent terrazzo flooring is installed. G. Cut out and replace terrazzo areas that evidence lack of bond with substrate. Cut out terrazzo areas in panels defined by strips and replace to match adjacent terrazzo, or repair panels according to NTMA's written recommendations, as approved by Architect. H. Cleaning: Remove grinding dust from installation and adjacent areas. Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rines urfaces according to NTMA's written recommendations. Seal surfaces according to NTMA's written recommendations. Seal surfaces according to STMA's written recommendations. 48 UPO FSECTION 	27		by reacting with divider and control-joint strips.
 initial grind with 60/80-grit stones or with comparable diamond abrasives. Grouting: Before grouting, clean terrazzo with water, rinse, and allow to dry. Apply and cure epoxy grout. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction traffic through area is restricted. Grind with 200-grit stones or with comparable diamond abrasives until grout is removed from surface. Installation Tolerance: Limit variation in terrazzo surface from level to 1/4 inch in 10 feet; noncumulative. Install and finish poured-in-place terrazzo stairs at the same time the adjacent terrazzo flooring is installed. Install and finish poured-in-place terrazzo base at the same time the adjacent terrazzo flooring is installed. Install and finish poured-in-place terrazzo base at the same time the adjacent terrazzo flooring is installed. Cut out and replace terrazzo areas that evidence lack of bond with substrate. Cut out terrazzo areas in panels defined by strips and replace to match adjacent terrazzo, or repair panels according to NTMA's written recommendations, as approved by Architect. H. Cleaning: Remove grinding dust from installation and adjacent areas. Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rinse surfaces with water and allow them to dry thoroughly. Sealing: Seal surfaces according to NTMA's written recommendations. Apply sealer according to sealer manufacturer's written instructions. Apply sealer according to sealer manufacturer's written instructions. END OF SECTION 	28		a. Rough Grinding: Grind with 24-grit or finer stones or with comparable diamond abrasives. Follow
 b. Grouting: Before grouting, clean terrazzo with water, rinse, and allow to dry. Apply and cure epoxy grout. c. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction traffic through area is restricted. Grind with 200-grit stones or with comparable diamond abrasives until grout is removed from surface. 3. Installation Tolerance: Limit variation in terrazzo surface from level to 1/4 inch in 10 feet; noncumulative. Install and finish poured-in-place terrazzo tairs at the same time the adjacent terrazzo flooring is installed. F. Install and finish poured-in-place terrazzo base at the same time the adjacent terrazzo flooring is installed. G. Cut out and replace terrazzo areas that evidence lack of bond with substrate. Cut out terrazzo areas in panels defined by strips and replace to match adjacent terrazzo, or repair panels according to NTMA's written recommendations, as approved by Architect. H. Cleaning: Remove grinding dust from installation and adjacent areas. Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rinse surfaces with water and allow them to dry thoroughly. Seal surfaces according to NTMA's written recommendations. Apply sealer according to SECTION 	29		initial grind with 60/80-grit stones or with comparable diamond abrasives.
 grout. c. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction traffic through area is restricted. Grind with 200-grit stones or with comparable diamond abrasives until grout is removed from surface. 3. Installation Tolerance: Limit variation in terrazzo surface from level to 1/4 inch in 10 feet; noncumulative. Install and finish poured-in-place terrazzo stairs at the same time the adjacent terrazzo flooring is installed. F. Install and finish poured-in-place terrazzo base at the same time the adjacent terrazzo flooring is installed. G. Cut out and replace terrazzo areas that evidence lack of bond with substrate. Cut out terrazzo areas in panels defined by strips and replace to match adjacent terrazzo, or repair panels according to NTMA's written recommendations, as approved by Architect. H. Cleaning: 2. Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rinse surfaces with water and allow them to dry thoroughly. I. Sealing: I. Sealing: Apply sealer according to NTMA's written recommendations. 2. Apply sealer according to STMA's written instructions. 48 END OF SECTION 	30		b. Grouting: Before grouting, clean terrazzo with water, rinse, and allow to dry. Apply and cure epoxy
 c. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction traffic through area is restricted. Grind with 200-grit stones or with comparable diamond abrasives until grout is removed from surface. 3. Installation Tolerance: Limit variation in terrazzo surface from level to 1/4 inch in 10 feet; noncumulative. Install and finish poured-in-place terrazzo stairs at the same time the adjacent terrazzo flooring is installed. Install and finish poured-in-place terrazzo base at the same time the adjacent terrazzo flooring is installed. G. Cut out and replace terrazzo areas that evidence lack of bond with substrate. Cut out terrazzo areas in panels defined by strips and replace to match adjacent terrazzo, or repair panels according to NTMA's written recommendations, as approved by Architect. H. Cleaning: Remove grinding dust from installation and adjacent areas. Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rinse surfaces with water and allow them to dry thoroughly. Seal surfaces according to NTMA's written recommendations. Apply sealer according to SECTION 	31		grout.
 traffic through area is restricted. Grind with 200-grit stones or with comparable diamond abrasives until grout is removed from surface. Installation Tolerance: Limit variation in terrazzo surface from level to 1/4 inch in 10 feet; noncumulative. Install and finish poured-in-place terrazzo stairs at the same time the adjacent terrazzo flooring is installed. Install and finish poured-in-place terrazzo base at the same time the adjacent terrazzo flooring is installed. Install and finish poured-in-place terrazzo base at the same time the adjacent terrazzo flooring is installed. Cut out and replace terrazzo areas that evidence lack of bond with substrate. Cut out terrazzo areas in panels defined by strips and replace to match adjacent terrazzo, or repair panels according to NTMA's written recommendations, as approved by Architect. H. Cleaning: Remove grinding dust from installation and adjacent areas. Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rinse surfaces with water and allow them to dry thoroughly. Sealing: Seal surfaces according to NTMA's written recommendations. Apply sealer according to sealer manufacturer's written instructions. Apply sealer according to sealer manufacturer's written instructions.	32		c. Fine Grinding/Polishing: Delay fine grinding until heavy trade work is complete and construction
34until grout is removed from surface.353.Installation Tolerance: Limit variation in terrazzo surface from level to 1/4 inch in 10 feet; noncumulative.36E.Install and finish poured-in-place terrazzo stairs at the same time the adjacent terrazzo flooring is installed.37F.Install and finish poured-in-place terrazzo base at the same time the adjacent terrazzo flooring is installed.38G.Cut out and replace terrazzo areas that evidence lack of bond with substrate. Cut out terrazzo areas in panels39defined by strips and replace to match adjacent terrazzo, or repair panels according to NTMA's written40recommendations, as approved by Architect.41H.Cleaning:421.Remove grinding dust from installation and adjacent areas.432.Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rinse surfaces with water and allow them to dry thoroughly.45I.Sealing:461.Seal surfaces according to NTMA's written recommendations.472.Apply sealer according to sealer manufacturer's written instructions.48END OF SECTION	33		traffic through area is restricted. Grind with 200-grit stones or with comparable diamond abrasives
 3. Installation Tolerance: Limit variation in terrazzo surface from level to 1/4 inch in 10 feet; noncumulative. 36 E. Install and finish poured-in-place terrazzo stairs at the same time the adjacent terrazzo flooring is installed. 37 F. Install and finish poured-in-place terrazzo base at the same time the adjacent terrazzo flooring is installed. 38 G. Cut out and replace terrazzo areas that evidence lack of bond with substrate. Cut out terrazzo areas in panels defined by strips and replace to match adjacent terrazzo, or repair panels according to NTMA's written recommendations, as approved by Architect. 41 H. Cleaning: 42 1. Remove grinding dust from installation and adjacent areas. 43 2. Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rinse surfaces with water and allow them to dry thoroughly. 45 I. Sealing: 46 1. Seal surfaces according to NTMA's written recommendations. 47 2. Apply sealer according to sealer manufacturer's written instructions. 48 END OF SECTION 	34		until grout is removed from surface.
36E.Install and finish poured-in-place terrazzo stairs at the same time the adjacent terrazzo flooring is installed.37F.Install and finish poured-in-place terrazzo base at the same time the adjacent terrazzo flooring is installed.38G.Cut out and replace terrazzo areas that evidence lack of bond with substrate. Cut out terrazzo areas in panels defined by strips and replace to match adjacent terrazzo, or repair panels according to NTMA's written recommendations, as approved by Architect.41H.Cleaning:421.Remove grinding dust from installation and adjacent areas.432.Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rinse surfaces with water and allow them to dry thoroughly.45I.Sealing:461.Seal surfaces according to NTMA's written recommendations.472.Apply sealer according to sealer manufacturer's written instructions.48END OF SECTION	35		3. Installation Tolerance: Limit variation in terrazzo surface from level to 1/4 inch in 10 feet; noncumulative.
 F. Install and finish poured-in-place terrazzo base at the same time the adjacent terrazzo flooring is installed. G. Cut out and replace terrazzo areas that evidence lack of bond with substrate. Cut out terrazzo areas in panels defined by strips and replace to match adjacent terrazzo, or repair panels according to NTMA's written recommendations, as approved by Architect. H. Cleaning: Remove grinding dust from installation and adjacent areas. Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rinse surfaces with water and allow them to dry thoroughly. I. Sealing: Sealing: Apply sealer according to NTMA's written instructions. 	36	Ε.	Install and finish poured-in-place terrazzo stairs at the same time the adjacent terrazzo flooring is installed.
 G. Cut out and replace terrazzo areas that evidence lack of bond with substrate. Cut out terrazzo areas in panels defined by strips and replace to match adjacent terrazzo, or repair panels according to NTMA's written recommendations, as approved by Architect. H. Cleaning: Remove grinding dust from installation and adjacent areas. Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rinse surfaces with water and allow them to dry thoroughly. I. Sealing: Sealing: Seal surfaces according to NTMA's written recommendations. Apply sealer according to sealer manufacturer's written instructions. END OF SECTION 	37	F.	Install and finish poured-in-place terrazzo base at the same time the adjacent terrazzo flooring is installed.
 defined by strips and replace to match adjacent terrazzo, or repair panels according to NTMA's written recommendations, as approved by Architect. H. Cleaning: Remove grinding dust from installation and adjacent areas. Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rinse surfaces with water and allow them to dry thoroughly. I. Sealing: Sealing: Sealing: Seal surfaces according to NTMA's written recommendations. Apply sealer according to sealer manufacturer's written instructions. END OF SECTION 	38	G.	Cut out and replace terrazzo areas that evidence lack of bond with substrate. Cut out terrazzo areas in panels
40 recommendations, as approved by Architect. 41 H. Cleaning: 42 1. Remove grinding dust from installation and adjacent areas. 43 2. Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rinse surfaces with water and allow them to dry thoroughly. 44	39		defined by strips and replace to match adjacent terrazzo, or repair panels according to NTMA's written
41 H. Cleaning: 42 1. Remove grinding dust from installation and adjacent areas. 43 2. Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rinse surfaces with water and allow them to dry thoroughly. 44 instructions; rinse surfaces with water and allow them to dry thoroughly. 45 I. Sealing: 46 1. Seal surfaces according to NTMA's written recommendations. 47 2. Apply sealer according to sealer manufacturer's written instructions. 48 END OF SECTION	40		recommendations, as approved by Architect.
42 1. Remove grinding dust from installation and adjacent areas. 43 2. Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written instructions; rinse surfaces with water and allow them to dry thoroughly. 44 instructions; rinse surfaces with water and allow them to dry thoroughly. 45 I. 46 1. 47 2. 48 END OF SECTION	41	Н.	Cleaning:
 43 43 44 44 45 46 47 48 48 43 44 44 45 44 45 46 47 46 47 46 47 48 47 48 49 4	42		1. Remove grinding dust from installation and adjacent areas.
44 instructions; rinse surfaces with water and allow them to dry thoroughly. 45 I. 46 1. 47 2. 48 END OF SECTION	43		2. Wash surfaces with cleaner according to NTMA's written recommendations and manufacturer's written
 Sealing: Seal surfaces according to NTMA's written recommendations. Apply sealer according to sealer manufacturer's written instructions. BND OF SECTION 	44		instructions; rinse surfaces with water and allow them to dry thoroughly.
 Seal surfaces according to NTMA's written recommendations. Apply sealer according to sealer manufacturer's written instructions. END OF SECTION 	45	I.	Sealing:
 Apply sealer according to sealer manufacturer's written instructions. END OF SECTION 	46		1. Seal surfaces according to NTMA's written recommendations.
48 END OF SECTION	47		2. Apply sealer according to sealer manufacturer's written instructions.
	48		END OF SECTION

SECTION 09 68 13 THE CARDETING

1		SECTION 09 68 13			
2		TILE CARPETING			
3	PART 1 -	JENERAL			
4	1.1	SUMMARY			
5	А.	Section Includes:			
6		1. Modular carpet tile.			
7	1.2	PREINSTALLATION MEETINGS			
8	Α.	Preinstallation Conference: Conduct conference at Project site.			
9	1.3	ACTION SUBMITTALS			
10	А.	Product Data: For each type of product.			
11	В.	Shop Drawings: For carpet tile installation, plans showing the following:			
12		1. Columns, doorways, enclosing walls or partitions, built-in cabinets, and locations where cutouts	are		
13		required in carpet files.			
14		2. Carpet file type, color, and dye lot.			
15		3. Type of subfloor.			
10		4. Type of installation.			
10		5. Pattern of installation.			
10		 Pattern type, location, and direction. Dile direction 			
19		7. Pile difection.			
20		 Type, color, and location of edge transition and other accessory string 			
21		9. Type, coor, and location of edge, transition, and other accessory strips.			
22 72	C	Samples: For each exposed product and for each color and texture required			
23 24	14				
24 25	Δ	Product test reports			
26	R.	Sample warranty			
27	15	MAINTENANCE MATERIAL SUBMITTALS			
28	1.5 A	Extra Stock Material: Eurnish extra materials from the same production run to Owner that match production	ucts		
29	7	installed and that are nackaged with protective covering for storage and identified with labels describing contents			
30		1. Full-size units equal to ONE box for each type indicated.			
31	1.6	CLOSEOUT SUBMITTALS			
32	 A.	Maintenance data.			
33	1.7	OUALITY ASSURANCE			
34	А.	Installer Qualifications: Certified by the International Certified Floorcovering Installers Association at	the		
35		Commercial II certification level.			
36	1.8	WARRANTY			
37	А.	Special Warranty for Carpet Tiles: Manufacturer agrees to repair or replace components of carpet tile installa	tion		
38		that fail in materials or workmanship within specified warranty period.			
39		1. Warranty Period: 10 years from date of Substantial Completion.			
40	PART 2 -	PRODUCTS			
41	2.1	CARPET TILE			
42	Α.	Applied Treatments:			
43		1. Soil-Resistance Treatment: Manufacturer's standard treatment.			
44		2. Antimicrobial Treatment: Manufacturer's standard treatment that protects carpet tiles as follows:			
45		a. Antimicrobial Activity: Not less than 2-mm halo of inhibition for gram-positive bacteria, not less t	than		
46		1-mm halo of inhibition for gram-negative bacteria, and no fungal growth, according to AATCC 1	74.		
47	В.	Performance Characteristics:			
48		1. Appearance Retention Rating: Heavy traffic, 3.0 minimum according to ASTM D7330.			
49		2. Critical Radiant Flux Classification: Not less than 0.45 W/sq. cm according to NFPA 253.			
50		3. Dry Breaking Strength: Not less than 100 lbf according to ASTM D2646.			
51		4. Tuft Bind: Not less than 5 lbf according to ASTM D1335.			
52		5. Delamination: Not less than 4 lbf/in. according to ASTM D3936.			
53		6. Dimensional Tolerance: Within 1/32 inch of specified size dimensions, as determined by phys	sical		
54		measurement.			
55		7. Dimensional Stability: 0.2 percent or less according to ISO 2551 (Aachen Test).			
56		8. Colorfastness to Crocking: Not less than 4, wet and dry, according to AATCC 165.			
57		9. Colorfastness to Light: Not less than 4 after 40 AFU (AATCC fading units) according to AATCC 16, Option F	Ε.		
58		10. Electrostatic Propensity: Less than 3.5 kV according to AATCC 134.			

SPECIFICATION

May 16 2024

	iviay 16	, 2024				
1	2.2	INSTALLATION ACCESSORIES				
2	Α.	Trowelable Leveling and Patching Compounds: Latex-modified, hydraulic-cement-based formulation provided or				
3		recommended by carpet tile manufacturer.				
4	В.	Adhesives: Water-resistant, mildew-resistant, nonstaining, pressure-sensitive type to suit products and subflo				
5		conditions indicated, that comply with flammability requirements for installed carpet tile, and are recommended by				
6		carpet tile manufacturer for releasable installation.				
7	PART 3 -	EXECUTION				
8	3.1	EXAMINATION				
9	Α.	Concrete Slabs:				
10		1. Moisture Testing: Perform tests so that each test area does not exceed 200 sq. ft., and perform no fewer				
11		than three tests in each installation area and with test areas evenly spaced in installation areas.				
12		a. Anhydrous Calcium Chloride Test: ASTM F1869. Proceed with installation only after substrates have				
13		maximum moisture-vapor-emission rate of 3 lb of water/1000 sq. ft. in 24 hours.				
14		b. Relative Humidity Test: Using in situ probes, ASTM F2170. Proceed with installation only after				
15		substrates have a maximum 75 percent relative humidity level measurement.				
16		c. Perform additional moisture tests recommended in writing by adhesive and carpet tile				
17		manufacturers. Proceed with installation only after substrates pass testing.				
18	В.	Wood Subfloors: Verify that underlayment surface is free of irregularities and substances that may interfere with				
19		adhesive bond or show through surface.				
20	С.	Metal Subfloors: Verify that underlayment surface is free of irregularities and substances that may interfere with				
21		adhesive bond or show through surface.				
22	D.	Painted Subfloors: Perform bond test recommended in writing by adhesive manufacturer.				
23		1. Access Flooring Systems: Verify access floor substrate is compatible with carpet tile and adhesive, if any,				
24		and underlayment surface is gaps greater than 1/8 inch and protrusions more than 1/32 inch.				
25	3.2	PREPARATION				
26	Α.	General: Comply with the Carpet and Rug Institute's CRI 104 and with carpet tile manufacturer's written installation				
27		instructions for preparing substrates indicated to receive carpet tile.				
28	В.	Use trowelable leveling and patching compounds, according to manufacturer's written instructions, to fill cracks,				
29		holes, depressions, and protrusions in substrates. Fill or level cracks, holes and depressions 1/8 inch wide or wider,				
30		and protrusions more than 1/32 inch unless more stringent requirements are required by manufacturer's written				
31		instructions.				
32	С.	Concrete Substrates: Remove coatings, including curing compounds, and other substances that are incompatible				
33		with adhesives and that contain soap, wax, oil, or silicone, without using solvents. Use mechanical methods				
34		recommended in writing by adhesive and carpet tile manufacturers.				
35	D.	Metal Substrates: Clean grease, oil, soil and rust, and prime if recommended in writing by adhesive manufacturer.				
36		Rough sand painted metal surfaces and remove loose paint. Sand aluminum surfaces, to remove metal oxides,				
37	_	immediately before applying adhesive.				
38	E.	Broom and vacuum clean substrates to be covered immediately before installing carpet tile.				
39	3.3	INSTALLATION				
40	А.	General: Comply with the Carpet and Rug Institute's CRI 104, Section 10, Carpet Tile, and with carpet tile				
41	п	manufacturer's written installation instructions.				
4Z 42	в.	Maintain due let integrity. De net mix due lete in some area				
45 44	С. р	Maintain dye-lot integrity. Do not mix dye lots in same area.				
44 15	D. E	Cut and fit carnet tile to butt tightly to vertical surfaces, permanent fixtures, and built in furniture including				
45 46	с.	cohinets pines outlets advings thresholds and pesings. Bind or soal out adves as recommended by carrect tile				
40		cabilities, pipes, outlets, edgings, tillesholds, and hosings. Billo of sear cut edges as recommended by carpet the				
47 10	F	Individualer.				
40 40	г.	exterio carpet the into toe spaces, door reveals, closets, open-bottomed obstructions, removable hanges, alcoves,				
49 50	G	And similar openings. Maintain reference markers, holes, and openings that are in place or marked for future cutting by repeating on				
50	0.	cornet tile as marked on subfloor. Use nonpormanent, nonstaining marking device				
52	ц	Install nattern narallel to walk and borders unless indicated otherwise in drawings				
52 52	11.	Access Flooring: Stagger joints of carnet tiles so carnet tile grid is offset from access flooring namel grid. Do not fill				
50	1.	seams of access flooring namels with carnet adhesive: keen seams free of adhesive				
54	I	Protect carnet tile against damage from construction operations and placement of equipment and fivtures during				
56	J.	the remainder of construction period. Use protection methods indicated or recommended in writing by carpet tile				
57		manufacturer				
58		FND OF SECTION				

SECTION 09 91 23 INTERIOR PAINTING

1		SECTION 09 91 23						
2		INTERIOR PAINTING						
3	PART 1 -	ART 1 - GENERAL						
4	1.1	SUMMARY						
5	Α.	Section Includes:						
6		1. Primers.						
7		2. Water-based finish coatings.						
8		3. Solvent-based finish coatings.						
9		4. Floor sealers and paints.						
10		5. Dry fall coatings.						
11	1.2	ACTION SUBMITTALS						
12	Α.	Product Data: For each type of product.						
13	В.	Samples: For each type of topcoat product.						
14	С.	Product Schedule: Use same designations indicated on Drawings and in the Interior Painting Schedule to cross-						
15		reference paint systems specified in this Section. Include color designations.						
16	PART 2 -	PRODUCTS						
17	2.1	MANUFACTURERS						
18	Α.	Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be						
19		incorporated into the Work include, but are not limited to the following:						
20		1. Benjamin Moore & Co.						
21		2. PPG Paints; PPG Industries, Inc.						
22		3. Sherwin-Williams Company (The).						
23	2.2	PAINT PRODUCTS, GENERAL						
24	Α.	Material Compatibility:						
25		1. Materials for use within each paint system shall be compatible with one another and substrates indicated,						
26		under conditions of service and application as demonstrated by manufacturer, based on testing and field						
27		experience.						
28		2. For each coat in a paint system, products shall be recommended in writing by topcoat manufacturers for						
29		use in paint system and on substrate indicated.						
30	В.	Colors: As indicated in a color schedule.						
31	2.3	PRIMERS – see 'Interior Paint Schedule' article below for direction for each substrate						
32	Α.	Items to be shipped "primed" are to use applicable primer from this list.						
33	В.	Interior/Exterior Latex Block Filler: Water-based, high-solids, emulsion coating formulated to bridge and fill porous						
34		surfaces of exterior concrete masonry units in preparation for specified subsequent coatings.						
35	С.	Alkali-Resistant, Water-Based Primer: Water-based primer formulated for use on alkaline surfaces, such as plaster,						
36		vertical concrete, and masonry.						
37	D.	Interior, Institutional Low-Odor/VOC Primer Sealer: Water-based primer sealer with low-odor characteristics and a						
38		VOC of less than 10 grams per liter for use on new interior plaster, concrete, and gypsum wallboard surfaces that						
39		are subsequently to be painted with latex finish coats.						
40	Ε.	Interior Alkyd Primer Sealer: Solvent-based, alkyd-type, primer/sealer for new interior wood, plaster, and porous						
41		surfaces,						
42	F.	Alkyd Quick-Dry Primer for Metal: Corrosion-resistant, solvent-based, modified-alkyd primer; lead and chromate						
43		free; formulated for quick-drying capabilities and for use on cleaned, interior steel surfaces.						
44	G.	Surface-Tolerant Metal Primer: Corrosion-resistant, solvent-based metal primer formulated for use on structural						
45		steel and metal fabrications that have been minimally prepared.						
46	Н.	Cementitious Galvanized Primer: Solvent-based primer composed of linseed oil/alkyd resin and portland cement for						
47		cleaned galvanized metal prior to finish coating.						
48	Ι.	Quick-Drying Aluminum Primer: Corrosion-resistant, solvent-based, alkyd or modified-alkyd primer formulated for						
49		quick-drying capabilities and for use on prepared exterior aluminum.						
50	2.4	WATER-BASED FINISH COATS - see 'Interior Paint Schedule' article below for direction for each substrate						
51	Α.	Interior, Latex, Institutional Low Odor/VOC: White or colored latex paint with low-odor characteristics and a VOC of						
52		less than 10 grams per liter, for use in areas, such as hospitals and other occupied buildings, where the odor and						
53		VOC levels of conventional latex products would preclude their use.						
54	В.	Interior, Latex, Institutional Low Odor/VOC: White or colored latex paint with low-odor characteristics and a VOC of						
55		less than 10 grams per liter, for use in areas, such as hospitals and other occupied buildings, where the odor and						
56		VOC levels of conventional latex products would preclude their use.						

25	SOLVENT_BASED EINISH COATS - see 'Interior Daint Schedule' article below for direction for each substrate
A.	Interior, Alkyd: Pigmented, solvent-based alkyd paint for use on primed/sealed interior plaster, gypsum, wood, and
	metal walls primarily in residential and moderate traffic commercial environments.
2.6	FLOOR SEALERS AND PAINTS
А.	Water-Based Concrete Floor Sealer: Clear, water-based, acrylic-copolymer-emulsion sealer formulated for oil,
	gasoline, alkali, and water resistance and for use on concrete traffic surfaces.
	1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
	a. H&C Decorative Concrete Products; a brand of Sherwin-Williams Co.
	D. Hempel (USA), Inc.
	d Rust-Oleum Corneration: a subsidiary of RDM International Inc
	e Sherwin-Williams Company (The)
2.7	DRY FALL COATINGS
A.	Dry Fall, Latex. Flat: Pigmented, water-based, emulsion-type, fast-drying coating for use on interior plaster.
	concrete, gypsum board, primed wood, and metal ceilings.
	1. Gloss and Sheen Level: Manufacturer's standard flat finish.
В.	Water Based, Dry Fall for Galvanized Steel, Flat: Pigmented, water-based coating for direct application to cleaned,
	interior galvanized-metal ceiling surfaces and adjacent primed metals.
	1. Gloss and Sheen Level: Manufacturer's standard flat finish.
PART 3 -	EXECUTION
3.1	EXAMINATION
А.	Verify suitability of substrates, including surface conditions and compatibility, with existing finishes and primers.
В.	Proceed with coating application only after unsatisfactory conditions have been corrected.
	1. Application of coating indicates acceptance of surfaces and conditions.
3.2	PREPARATION
А.	comply with manufacturer's written instructions and recommendations applicable to substrates and paint systems
в	Illuicateu. Remove hardware, covers, plates, and similar items already in place that are removable and are not to be painted. If
Б.	removal is impractical or impossible because of size or weight of item provide surface-applied protection before
	surface preparation and painting.
C.	After completing painting operations, use workers skilled in the trades involved to reinstall items that were
-	removed. Remove surface-applied protection if any.
3.3	INSTALLATION
Α.	Apply paints according to manufacturer's written instructions.
В.	Apply paints to produce surface films without cloudiness, spotting, holidays, laps, brush marks, roller tracking, runs,
	sags, ropiness, or other surface imperfections. Cut in sharp lines and color breaks.
C.	Painting Fire-Suppression, Plumbing, HVAC, Electrical, Communication, and Electronic Safety and Security Work:
	1. Paint the following work where exposed in all <u>but</u> mechanical or electrical rooms:
	a. Equipment, including panelboards and switch gear.
	b. Uninsulated metal piping.
	c. Uninsulated plastic piping.
	u. Pipe hangers and supports.
	f Plastic conduit
	g. Tanks that do not have factory-applied final finishes.
	h. Duct, equipment, and pipe insulation having cotton or canvas insulation covering or other paintable
	iacket material.
	 Paint portions of internal surfaces of metal ducts, without liner, behind air inlets and outlets that are visible
	from occupied spaces.
3.4	CLEANING AND PROTECTION
Α.	After completing paint application, clean spattered surfaces. Remove spattered paints by washing, scraping, or
	other methods. Do not scratch or damage adjacent finished surfaces.
В.	Protect work of other trades against damage from paint application. Correct damage to work of other trades by
	cleaning, repairing, replacing, and refinishing, as approved by Architect, and leave in an undamaged condition.
С.	At completion of construction activities of other trades, touch up and restore damaged or defaced painted surfaces.
3.5	INTERIOR PAINTING SCHEDULE – all systems might not apply
Α.	Concrete Substrates, Nontraffic Surfaces:
	1. Institutional Low-Odor/VOC Latex System:

	IVIAY 10	, 2024		
1			a.	Prime Coat: Interior, institutional low-odor/VOC primer sealer.
2			b.	Intermediate Coat: Matching topcoat.
3			с.	Topcoat: Interior latex, institutional low odor/VOC, low sheen.
4	В.	Concr	ete Sub	ostrates, Traffic Surfaces:
5		1.	See C	133543 - Polished Concrete Finishing if called for in Finish Schedule.
6		2.	See C	171800 - Traffic Coatings if called for in Finish Schedule.
7		3.	Wate	er-Based Concrete Floor Sealer System if called for in Finish Schedule:
8			a.	First Coat: Matching topcoat.
9			b.	Topcoat: Water-based concrete floor sealer.
10	С.	CMU	Substra	tes:
11		1.	Instit	utional Low-Odor/VOC Latex System:
12			a.	Block Filler: Interior/exterior latex block filler.
13			b.	Intermediate Coat: Matching topcoat.
14			с.	Topcoat: Interior, latex, institutional low odor/VOC, low sheen.
15	D.	Steel S	Substra	tes:
16		1.	Instit	utional Low-Odor/VOC Latex over Shop-Applied Quick-Drying Shop Primer System:
17			a.	Prime Coat: Pre-primed.
18			b.	Intermediate Coat: Matching topcoat.
19			с.	Topcoat: Interior, latex, institutional low odor/VOC, satin.
20			d.	For pre-primed elements like hollow metal doors/frames.
21		2.	Instit	utional Low-Odor/VOC Latex over Alkyd Primer System:
22			a.	Prime Coat: Alkyd Quick-Dry Primer for Metal.
23			b.	Intermediate Coat: Matching topcoat.
24			с.	Topcoat: Interior, latex, institutional low odor/VOC, satin.
25			d.	For unprimed/uncoated elements.
26		3.	Wate	r-Based Dry Fall over Shop-Applied Shop Primer System:
27			a.	Prime Coat: Shop primer specified in Section where substrate is specified.
28			b.	Topcoat: Dry fall, latex, flat.
29			с.	For non-galvanized ceiling elements.
30	Ε.	Galva	nized-N	1etal Substrates:
31		1.	Instit	utional Low-Odor/VOC Latex System:
32			a.	Prime Coat: Water-based galvanized primer.
33			b.	Intermediate Coat: Matching topcoat.
34			с.	Topcoat: Interior, latex, institutional low odor/VOC, satin.
35			d.	For galvanized elements.
36		2.	Wate	r-based Dry Fall for Galvanized Steel System:
37			a.	Prime Coat: Water-based dry fall for galvanized steel.
38			b.	Topcoat: Dry fall, latex, flat.
39			с.	For galvanized ceiling elements.
40	F.	Finish	Carpe	htry & Architectural Woodwork: Wood trim, Doors, Windows, and Wood board paneling.
41		1.	Instit	utional Low-Odor/VOC Latex over Alkyd Primer System:
42			a.	Prime Coat: Interior alkyd primer sealer.
43			b.	Intermediate Coat: Matching topcoat.
44			с.	Topcoat: Interior, latex, institutional low odor/VOC, satin.
45			d.	For wood trim, doors, windows, wood paneling, and casework.
46	G.	Gypsi	um Boa	rd Substrates:
47		1.	Instit	utional Low-Odor/VOC Latex System:
48			a.	Prime Coat: Interior, institutional low-odor/VOC primer sealer.
49			b.	Intermediate Coat: Matching topcoat.
50			с.	Topcoat: Interior, latex, institutional low odor/VOC, flat (ceilings/soffits) and eggshell (walls).
51			d.	For walls, ceilings, and soffits.
52				END OF SECTION

1 **SECTION 09 93 00** 2 STAINING AND TRANSPARENT FINISHING 3 PART 1 - GENERAL 4 1.1 SUMMARY 5 Α. Section Includes: 6 Wood stains. 1. 7 Transparent finishes. 2. 8 1.2 **ACTION SUBMITTALS** 9 Product Data: For each type of product. Α. 10 Samples: For each type of finish system and in each color and gloss of finish required. B. 11 C. Product List: Cross-reference to finish system and locations of application areas. Use same designations indicated 12 on Drawings and in schedules. Include color designations. 13 PART 2 - PRODUCTS 14 MANUFACTURERS 2.1 15 A. Manufacturers: Subject to compliance with requirements, provide products by one of the following: 16 Benjamin Moore & Co. 1. 17 2. Penofin. 18 3. PPG Paints; PPG Industries, Inc. 19 Sherwin-Williams Company (The). 4. 20 Sikkens. 5. **MATERIALS, GENERAL** 21 2.2 22 Material Compatibility: A. 23 Provide materials for use within each coating system that are compatible with one another and substrates 1. 24 indicated, under conditions of service and application as demonstrated by manufacturer, based on testing 25 and field experience. 26 2.3 WOOD STAINS 27 Stain, Interior, Semitransparent, for Interior Wood: Solvent-based, oil or oil/alkyd, semitransparent, pigmented Α. 28 stain for new interior wood surfaces that are to be finished with a clear varnish. 29 2.4 TRANSPARENT FINISHES 30 Α. Varnish, Interior, Polyurethane, Oil Modified: Solvent-based, one-component, oil-modified polyurethane clear satin 31 varnish for new or previously varnished or stained interior wood surfaces. 32 **PART 3 - EXECUTION** 33 **EXAMINATION** 3.1 34 Α. Maximum Moisture Content of Exterior Wood Substrates: 15 percent, when measured with an electronic moisture 35 meter. 36 B. Maximum Moisture Content of Interior Wood Substrates: 15 percent, when measured with an electronic moisture 37 meter. 38 3.2 PREPARATION 39 Remove hardware, covers, plates, and similar items already in place that are removable. If removal is impractical or Α. impossible because of size or weight of item, provide surface-applied protection before surface preparation and 40 41 finishing. 42 1. After completing finishing operations, use workers skilled in the trades involved to reinstall items that were 43 removed. Remove surface-applied protection if any. 44 Β. Clean and prepare surfaces to be finished according to manufacturer's written instructions for each substrate 45 condition and as specified. 46 1. Remove dust, dirt, oil, and grease by washing with a detergent solution; rinse thoroughly with clean water 47 and allow to dry. Remove grade stamps and pencil marks by sanding lightly. Remove loose wood fibers by 48 brushing. 49 2. Remove mildew by scrubbing with a commercial wash formulated for mildew removal and as 50 recommended by stain manufacturer. 51 3.3 APPLICATION 52 Apply finishes according to manufacturer's written instructions. A. 53 Apply finishes to produce surface films without cloudiness, holidays, lap marks, brush marks, runs, ropiness, or Β. other surface imperfections. 54 55 3.4 **CLEANING AND PROTECTION** 56 Protect work of other trades against damage from finish application. Correct damage by cleaning, repairing, Α. 57 replacing, and refinishing, as approved by Architect, and leave in an undamaged condition.

1	В.	At completion of construction activities of other trades, touch up and restore damaged or defaced finished wood
2		surfaces.
3	3.5	INTERIOR WOOD-FINISH-SYSTEM SCHEDULE
4	Α.	Wood Substrates, Wood Trim, Architectural Woodwork, Doors, Windows, and/or Wood Board Paneling:
5		1. Polyurethane Varnish over Stain System:
5		a. Stain Coat: Stain, semitransparent, for interior wood – color to match existing.
7		b. First Intermediate Coat: Polyurethane varnish matching topcoat.
3		c. Second Intermediate Coat: Polyurethane varnish matching topcoat.
Э		d. Topcoat: Varnish, interior, polyurethane, oil modified, satin.
)		END OF SECTION

SECTION 10 14 23 PANEL SIGNAGE

2		PANEL SIGNAGE				
3	PART 1 - (GENERAL				
4	1.1	SUMMARY				
5	Α.	Section Includes:				
6		1. Panel signs – <u>for interior, wall-mounted applications</u> .				
7	1.2	ACTION SUBMITTALS				
8	Α.	Product Data: For panel signs.				
9	В.	Shop Drawings: For panel signs.				
10		 Include fabrication and installation details and attachments to other work. 				
11		2. Show sign mounting heights, locations of supplementary supports to be provided by other installers, and				
12		accessories.				
13		3. Show message list, typestyles, graphic elements, including raised characters and Braille, and layout for each				
14		sign at least half size.				
15		4. Show locations of electrical service connections.				
16	-	5. Include diagrams for power, signal, and control wiring.				
17	С.	Samples: For each exposed product and for each color and texture specified.				
18	1.3	CLOSEOUT SUBMITTALS				
19	Α.	Maintenance data.				
20	1.4					
21	А.	Special Warranty: Manufacturer agrees to repair or replace components of signs that fail in materials or				
22		workmanship within specified warranty period.				
23		1. Warranty Period: Five years from date of Substantial Completion.				
24 25	2 1					
25	2.1	Accessibility Standard: Comply with applicable provisions in the USDOI's "2010 ADA Standards for Accessible				
20	А.	Design " the ARA standards of the Federal agency baying jurisdiction and ICC A117.1				
27 28	22					
20 29	Δ	Panel Sign: Sign with smooth uniform surfaces: with message and characters having uniform faces, sharp corpers				
30	Α.	and precisely formed lines and profiles: and as follows:				
31		1 Integral Tactile Plaque: Granhics and Plaque materials are one niece. Tactile Photonolymer Inserts are 080"				
32		nhenolic photopolymer with raised copy and fully domed Grade 2 Braille dots etched to 1/32" Background				
33		color is painted in acrylic lacguer in the specified Insert color. Top surface of copy characters is then added				
34		by roller printing in specified copy color using Silkscreen inks.				
35		2. Sign-Panel Perimeter: Finish edges smooth.				
36		a. Edge Condition:				
37		1) Vertical Edges: Square cut.				
38		2) Horizontal Edges: Square cut.				
39		b. Corner Condition in Elevation: Square.				
40		3. Mounting: Surface mounted to wall with adhesive and/or two-face tape.				
41		4. Color as selected by Architect from full range of industry colors.				
42		a. Painted Finish and Graphics: Manufacturer's standard, factory-applied acrylic lacquer, in color as				
43		selected by Architect from manufacturer's full range.				
44		5. Text and Typeface: Accessible raised characters and Braille. Typeface matching Architect's schedule. Finish				
45		raised characters to contrast with background color, and finish Braille to match background color.				
46		6. Flatness Tolerance: Sign is to remain flat under installed conditions within a tolerance of plus or minus 1/16				
47		inch measured diagonally from corner to corner.				
48	2.3	PANEL-SIGN MATERIALS				
49	Α.	Paints and Coatings for Sheet Materials: Inks, dyes, and paints that are recommended by manufacturer for optimum				
50		adherence to surface and are UV and water resistant for colors and exposure indicated. Fade resistant to 5 years.				
51	2.4	ACCESSORIES				
52	Α.	Fasteners and Anchors: Manufacturer's standard as required for secure anchorage of signs, noncorrosive and				
53		compatible with each material joined, and complying with the following unless otherwise indicated:				
54		1. Use concealed fasteners and anchors unless indicated to be exposed.				
55	В.	Adhesive: As recommended by sign manufacturer.				
56	C.	Two-Face Tape: Manufacturer's standard high-bond, foam-core tape, 0.045 inch thick, with adhesive on both sides.				
57	2.5	FABRICATION				
58	Α.	General: Provide manufacturer's standard sign assemblies according to requirements indicated.				

IVIAY 1	0, 2024					
В.	Desig F. wi	on components allow for expansion and contraction for a minimum material temperature range of 100 degrees thout causing buckling or over stressing of adhesives and fasteners.				
C.	Form	Form work to required shapes and sizes, with true lines and angles. Provide necessary rebates, lugs, and brackets				
0.	for a	for assembly of units.				
D.	Cont	Contact surfaces of connected members must be true. Assembled so joints will be tight and practically				
	unno	pticeable, without use of filling compound.				
Ε.	Signs	s shall have fine, even texture and be flat and sound. Lines and miters sharp, arises unbroken, profiles accurate				
	and	ornament true to pattern. Plane surfaces should be smooth, flat and without oil-canning, free of rack and				
	twist	. Maximum variation from plane of surface plus or minus .032". Restore texture to filled or cut areas.				
F.	Leve	l or straighten wrought work. Members shall have sharp lines and angels and smooth surfaces.				
G.	All p	ainted surfaces properly primed. Finish coating of paint to have complete coverage with no light or thin				
	appli	cations allowing substrate or primer to show. Parts are checked for approval against the color match master				
	chip.	Finished surface smooth, free of scratches, gouges, drips, bubbles, thickness variations, foreign matter, and				
	othe	r imperfections.				
2.6	GEN	ERAL FINISH REQUIREMENTS				
Α.	Appe	arance of Finished Work: Noticeable variations in same piece are not acceptable. Variations in appearance of				
	adjoi	ning components are acceptable if they are within the range of approved Samples and are assembled or				
	insta	lled to minimize contrast.				
PART 3	- EXECUT	10N				
3.1	INSI					
А.	Gene	ral: Install signs using mounting methods indicated and according to manufacturer's written instructions.				
	1.	install signs level, plumb, true to line, and at locations and heights indicated, with sign surfaces free of				
	2	ustortion and other defects in appearance.				
	2.	Install signs so they do not protrude or obstruct according to the accessibility standard.				
	5.	before installation, verify that sign surfaces are clean and free of materials of debris that would impair				
в	A.c.c.o	listaliation.				
Б.	stand	ssible signage. Instan in locations on wans as indicated on brawings and according to the accessibility				
C	Mou	nting Methods:				
с.	1000	Concealed Studs: Using a template drill holes in substrate aligning with studs on back of sign. Remove loose				
		debris from hole and substrate surface				
		a. Masonry Substrates: Fill holes with adhesive. Leave recess space in hole for displaced adhesive.				
		Place sign in position and push until flush to surface, embedding studs in holes. Temporarily support				
		sign in position until adhesive fully sets.				
		b. Thin or Hollow Surfaces: Place sign in position and flush to surface, install washers and nuts on studs				
		projecting through opposite side of surface, and tighten.				
	2.	Projecting Studs: Using a template, drill holes in substrate aligning with studs on back of sign. Remove loose				
		debris from hole and substrate surface.				
		a. Masonry Substrates: Fill holes with adhesive. Leave recess space in hole for displaced adhesive.				
		Place spacers on studs, place sign in position, and push until spacers are pinched between sign and				
		substrate, embedding the stud ends in holes. Temporarily support sign in position until adhesive				
		fully sets.				
		b. Thin or Hollow Surfaces: Place spacers on studs, place sign in position with spacers pinched between				
		sign and substrate, and install washers and nuts on stud ends projecting through opposite side of				
		surface, and tighten.				
	3.	Through Fasteners: Drill holes in substrate using predrilled holes in sign as template. Countersink holes in				
		sign if required. Place sign in position and flush to surface. Install through fasteners and tighten.				
	4.	Adhesive: Clean bond-breaking materials from substrate surface and remove loose debris. Apply linear				
		beads or spots of adhesive symmetrically to back of sign and of suitable quantity to support weight of sign				
		after cure without slippage. Keep adhesive away from edges to prevent adhesive extrusion as sign is applied				
		and to prevent visibility of cured adhesive at sign edges. Place sign in position, and push to engage adhesive.				
		Temporarily support sign in position until adhesive fully sets.				
	5.	Two-Face Tape: Clean bond-breaking materials from substrate surface and remove loose debris. Apply tape				
		strips symmetrically to back of sign and of suitable quantity to support weight of sign without slippage.				
		Keep strips away from edges to prevent visibility at sign edges. Place sign in position, and push to engage				
		tape adhesive.				

1	3.2	ADJUSTING AND CLEANING
2	Α.	Remove and replace damaged or deformed signs and signs that do not comply with specified requirements. Replace
3		signs with damaged or deteriorated finishes or components that cannot be successfully repaired by finish touchup
4		or similar minor repair procedures.
5	В.	Remove temporary protective coverings and strippable films as signs are installed.
6	С.	On completion of installation, clean exposed surfaces of signs according to manufacturer's written instructions, and
7		touch up minor nicks and abrasions in finish. Maintain signs in a clean condition during construction and protect
8		from damage until acceptance by Owner.

END OF SECTION

1 SECTION 10 21 13.19 2 PLASTIC TOILET COMPARTMENTS 3 PART 1 - GENERAL 4 1.1 SUMMARY 5 Α. Section Includes: 6 Solid-plastic toilet compartments. 1. 7 Β. **Related Requirements:** 8 Section 06 10 00 "Rough Carpentry" for blocking. 1. 9 Section 09 22 16 "Non-Structural Metal Framing" for blocking. 2. 10 Section 10 28 00 "Toilet, Bath, and Laundry Accessories" for accessories mounted on toilet compartments. 3. **ACTION SUBMITTALS** 11 1.2 12 Product data. Α. 13 Shop Drawings: Plans, elevations, sections, details, and attachment details. B. 14 C. Samples: Manufacturer's standard color sheets, showing full range of available colors for each type of toilet 15 compartment. D. 16 Delegated Design Submittals: For grab bars mounted on toilet compartment panels, including analysis data signed 17 and sealed by the qualified professional engineer responsible for their preparation. 18 1. Include structural design calculations indicating compliance with specified structural-performance 19 requirements. **CLOSEOUT SUBMITTALS** 20 1.3 21 Operation and maintenance data. Α. PART 2 - PRODUCTS 22 23 PERFORMANCE REQUIREMENTS 2.1 24 Fire Performance: Tested in accordance with, and pass the acceptance criteria of, NFPA 286. Α. 25 Structural Performance: Where grab bars are mounted on toilet compartments, design panels to comply with the B. 26 following requirements: 27 Panels are able to withstand a concentrated load on grab bar of at least 250 lbf applied at any direction and 1 28 at any point, without deformation of panel. 29 C. Regulatory Requirements: Comply with applicable provisions in the USDOJ's "2010 ADA Standards for Accessible 30 Design" and ICC A117.1 for toilet compartments designated as accessible. 31 2.2 SOLID-PLASTIC TOILET COMPARTMENTS Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be 32 Α. 33 incorporated into the Work include, but are not limited to the following: 34 ASI Global Partitions. 1. 35 Scranton Products. 2. 36 Β. Basis-of-Design: Scranton Products. 37 C. Toilet-Enclosure Style: Overhead braced. 38 D. Entrance-Screen Style: Overhead braced. 39 Ε. Urinal-Screen Style: Wall hung. F. 40 Door, Panel, and Pilaster Construction: Solid, high-density polyethylene (HDPE) material, not less than 1 inch thick, 41 seamless, with eased edges, and with homogenous color throughout thickness of material. Provide with no-42 sightline system consisting of door and pilaster lapped edges on strike side of door and door and pilaster lapped 43 edges on hinge side of door (unless continuous hinge is used). 44 1. Heat-Sink Strip: Manufacturer's continuous, extruded-aluminum or stainless steel strip fastened to exposed 45 bottom edges of solid-plastic components to hinder malicious combustion. 2. 46 Color: One color in each room as selected by Architect from manufacturer's full range. 47 G. Entrance-Screen Construction: Matching panel construction. 48 Η. Urinal-Screen Construction: Matching panel construction. 49 Pilaster Sleeves (Caps): Manufacturer's standard design; solid plastic or stainless steel. I. 50 1. Plastic Color: Matching pilaster. 51 J. Urinal-Screen Post: 1-3/4-inch-square aluminum tube with satin finish; with sleeve (cap) matching that on the 52 pilaster (i.e., not to floor) 53 К. Brackets (Fittings): 54 Full-Height (Continuous) Type: Manufacturer's standard design; stainless steel. 1. 55 2.3 HARDWARE AND ACCESSORIES 56 Door Hardware and Accessories: Manufacturer's operating hardware and accessories. Mount with through bolts. Α. 57 1. Hinges:

IVIAY 16,	, 2024
	 Manufacturer's continuous, cam type that swings to a closed or partially open position, allowing emergency access by lifting door. Material Continuous Hinge: Stainless steel
	 Latch and Keeper: Manufacturer's surface-mounted latch unit, designed for emergency access, and with combination rubber-faced door strike and keeper. Provide units that comply with regulatory requirements for accessibility at toilet englosures designated as accessible.
	Tor accessionity at tonet enclosures designated as accessible.
	d. Midlefidi. Stalliess steel.
	 Coat nook. As identified in Toilet Accessory section. Door Bumper: Manufacturer's rubber-tipped humper at outswinging doors UNLESS specified in Toilet
	Accessory section
	a. Material: Stainless steel.
	5. Door Pull: Manufacturer's unit at outswinging doors that complies with regulatory requirements for
	accessibility. Provide units on both sides of doors at toilet enclosures designated as accessible. a. Material: Stainless steel.
В.	Overhead Bracing: Manufacturer's standard continuous, extruded-aluminum head rail with antigrip profile and in
	manufacturer's standard finish.
С.	Anchorages and Fasteners: Manufacturer's standard exposed fasteners of stainless steel, finished to match items
	they are securing, with theft-resistant-type heads. Provide sex-type bolts for through-bolt applications. For
	concealed anchors, use stainless steel, hot-dip galvanized steel, or other rust-resistant, protective-coated steel
	compatible with related materials.
2.4	MATERIALS
Α.	Aluminum Castings: ASTM B26/B26M.
В.	Aluminum Extrusions: ASTM B221.
C.	Stainless Steel Sheet: ASTM AZ40/AZ40M or ASTM A666, Type 304, stretcher-leveled standard of flatness.
D. E	Staliness Steel Castings: ASTM A/43/A/43M.
с. Э г	Zamac. ASTW B80, commercial zinc-alloy die castings.
2.5	Expricate toilet compartment components to sizes indicated. Coordinate requirements and provide cutouts for
А.	through-partition toilet accessories where required for attachment of toilet accessories
в	Overhead-Braced Units: Manufacturer's standard corrosion-resistant supports leveling mechanism and anchors at
5.	pilasters and walls to suit floor and wall conditions. Provide shoes at pilasters to conceal supports and leveling
	mechanism.
C.	Urinal-Screen Posts: Manufacturer's standard corrosion-resistant anchoring assemblies at posts and walls, with
	leveling adjustment nuts at tops and bottoms of posts. Provide sleeves (caps) at posts to conceal anchorage.
D.	Door Size and Swings: Unless otherwise indicated, provide 24-inch-wide, inswinging doors for standard toilet
	enclosures and 36-inch-wide, outswinging doors with a minimum 32-inch-wide, clear opening for toilet enclosures
	designated as accessible.
PART 3 -	EXECUTION
3.1	INSTALLATION
Α.	General: Comply with manufacturer's written installation instructions. Install units rigid, straight, level, and plumb.
	Secure units in position with manufacturer's recommended anchoring devices.
	1. Maximum Clearances:
	a. Phasters and Panels of Screens: 1/2 Inch.
	 D. Panels of screens and Walls: 1 Inch. 2. Full Height (Continuous) Practets: Socium panels or screens to walls and to pilasters with full height.
	2. Full-Height (Continuous) blackets. Secure panels of screens to waits and to phasters with full-height
	a locate bracket fasteners, so boles for wall anchors occur in masonry or tile joints
	 Align brackets at nilasters with brackets at walls
В.	Overhead-Braced Units: Secure pilasters to floor and level, plumb, and tighten. Set pilasters with anchors
5.	penetrating not less than 1-3/4 inches into structural floor unless otherwise indicated in manufacturer's written
	instructions. Secure continuous head rail to each pilaster with no fewer than two fasteners. Hang doors to align
	tops of doors with tops of panels and adjust, so tops of doors are parallel with overhead brace when doors are in
	closed position.
C.	Urinal Screens: Attach with anchoring devices to suit supporting structure. Set units level and plumb, rigid, and secured to resist lateral impact.
	B. C. 2.4 A. B. C. D. E. 2.5 A. B. C. D. PART 3 - 1 3.1 A. B. C.

1	3.2	ADJUSTING
2	Α.	Hardware Adjustment: Adjust and lubricate hardware in accordance with hardware manufacturer's written
3		instructions for proper operation. Set hinges on inswinging doors to hold doors open approximately 30 degrees
4		from closed position when unlatched. Set hinges on outswinging doors to return doors to fully closed position.
5		END OF SECTION

SECTION 10 28 00 TOILET, BATH, AND LAUNDRY ACCESSORIES

2	TOILET, BATH, AND LAUNDRY ACCESSORIES			
3	PART 1 - 0	GENERAL		
4	1.1	SUMMARY		
5	Α.	Section Includes:		
6		1. Public&Private-use washroom accessories.		
7		2. Public-use shower room accessories.		
8		3. Childcare accessories.		
9		4. Needle disposal cabinet.		
10		5. Sanitary Napkin Dispenser.		
11	1.2	ACTION SUBMITTALS		
12	Α.	Product data.		
13	В.	Samples: For each exposed product and for each finish specified, full size.		
14		 Approved full-size Samples will be returned and may be used in the Work. 		
15	С.	Delegated Design Submittals: For grab bars and shower seats.		
16		1. Include structural design calculations indicating compliance with specified structural-performance		
17		requirements.		
18	1.3	INFORMATIONAL SUBMITTALS		
19	Α.	Sample warranties.		
20	1.4	CLOSEOUT SUBMITTALS		
21	Α.	Maintenance data.		
22	1.5	WARRANTY		
23	Α.	Manufacturer's Special Warranty for Mirrors: Manufacturer agrees to repair or replace mirrors that fail in materials		
24		or workmanship within specified warranty period.		
25		1. Warranty Period: 10 years from date of Substantial Completion.		
26	В.	Manufacturer's Special Warranty for Hand Dryers: Manufacturer agrees to repair or replace hand dryers that fail in		
27		materials or workmanship within specified warranty period.		
28		 Warranty Period: Two years from date of Substantial Completion. 		
29	PART 2 - F	PRODUCTS		
30	2.1	PERFORMANCE REQUIREMENTS		
31	Α.	Electrical Components, Devices, and Accessories: Listed and labeled as defined in NFPA 70, by a qualified testing		
32		agency, and marked for intended location and application.		
33	В.	Structural Performance: Design accessories and fasteners to comply with the following requirements:		
34		1. Grab Bars: Installed units are able to resist 250 lbf concentrated load applied in any direction and at any		
35		point.		
36		2. Shower Seats: Installed units are able to resist 250 lbf concentrated load applied in any direction and at any		
37		point.		
38	2.2	MANUFACTURER		
39		1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:		
40		a. ASI-American Specialties, Inc.		
41		b. Bobrick Washroom Equipment, Inc.		
42		c. Bradley Corporation.		
43		d. Gamco Commercial Restroom Accessories.		
44		2. Style: Unless noted otherwise, all accessories to be the same style from the same manufacturer.		
45		3. Basis-of-Design unless noted otherwise: Bobrick Washroom Equipment.		
46	2.3			
47	А.	Toilet Tissue (Roll) Dispenser TA-03 :		
48	_	1. Owner furnished; Contractor installed.		
49	В.	Grab Bar IA-UZ :		
50		1. Basis-of-Design: B-b806.		
51		2. Mounting: Flanges with concealed fasteners.		
52 52		5. IVIALENTIAL STAILLESS SLEET, U.US INCH THICK.		
55 E/		a. rinish. shiouth, Ashivi A460/A46000 No. 4 Tifish (Satin).		
54 E E		4. UD: 1-1/4 INCNES.		
22 56	C	5. Comiguration and Length: As indicated on Drawings.		
50	L.	Janilary-Ivaphili Dispusal Ullik IA-04. 1 Owner furniched: Contractor installed		
50	Л			
50	υ.			

SPECIFICATION

May 16	, 2024
	1. Basis-of-Design: B-672.
	2. Description: Coat hook.
	3. Mounting: Concealed.
	4. Material and Finish: Stainless steel, ASTM A480/A480M No. 4 finish (satin).
E.	Shelf TA-07 :
	1. Basis-of-Design: B-298.
	2 Description: 5"x24" shelf with integral end brackets
	3 Mounting: Surface mounted
	4 Material and Finish: Stainless steel ASTM A480/A480M No. 4 finish (satin)
24	PLIBLIC-LISE SHOWER ROOM ACCESSORIES
 ∓ ∆	Shower Curtain Rod TA-10:
Α.	1 Paris of Docing: P. 6047
	2 Description: 1.25-inch-OD straight rod
	2. Configuration: Ac indicated on Drawings
	Comparing Elanges: Consequent factories in material and finish matching red
	4. Mounting Flanges: Concealed Tasteners; in material and finish matching rou.
D	5. Kou Material and Finish. Stamess steel, ASTM A480/A480M No. 4 finish (satin).
В.	Shower Curtain TA-IU :
	1. Size: Minimum 6 inches wider than opening by 72 inches high.
	2. Material: Nylon-reinforced vinyl, minimum 9 oz. or 0.008-inch-thick vinyl, with integral antibacterial and
	flame-retardant agents.
	3. Color: As selected from manufacturer's full range.
	4. Grommets: Corrosion resistant at minimum 6 inches o.c. through top hem.
	5. Shower Curtain Hooks: Stainless steel, spring wire curtain hooks with snap fasteners, sized to
	accommodate specified curtain rod. Provide one hook per curtain grommet.
С.	Folding Shower Seat TA-06 :
	1. Basis-of-Design: B-5181.
	2. Configuration: As indicated on Drawings.
	3. Seat: Per basis-of-design.
	4. Mounting Mechanism: Stainless steel, ASTM A480/A480M No. 4 finish (satin).
	5. Dimensions: As indicated on Drawings.
2.5	CHILDCARE ACCESSORIES
Α.	Diaper-Changing Station TA-05 :
	1. Basis-of-Design: KB300-SS.
	2. Description: Horizontal and/or vertical (as indicated in drawings) unit that opens by folding down from
	stored position and with child-protection strap.
	a. Engineered to support minimum of 250 lb static load when opened.
	3. Mounting: Surface mounted, with unit projecting not more than 4 inches from wall when closed.
	4. Operation: By pneumatic shock-absorbing mechanism.
	5. Material and Finish: Stainless steel cladding over steel and polypropylene frame.
	6. Liner Dispenser: Provide built-in dispenser for disposable sanitary liners.
	7. Bag Hook: Independently mounted, stainless steel bag hook.
2.6	NEEDLE DISPOSAL CABINET
Α.	Needle Disposal Cabinet TA-11:
	1. Owner furnished; Contractor installed.
2.7	SANITARY NAPKIN DISPENSER
Α.	Sanitary Napkin Dispenser TA-08 :
	1. Owner furnished: Contractor installed
2.8	FABRICATION
Δ	Keys: Provide universal keys for internal access to accessories for servicing and resupplying. Provide minimum of si
7.0	keys to Owner's representative
PART 3 -	EXECUTION
2 1	
J.1 ^	Install accessories in accordance with manufacturars' written instructions, using fastoners appropriate to substrate
А.	indicated and recommended by unit manufacturer. Install units loud, alumb, and firmly analyzed in leasting and
	indicated and recommended by unit manufacturer. Install units level, plumb, and firmly anchored in locations and at beights indicated
	at neights multilated.
_	Kemove temporary labels and protective coatings. Crob Para Install to comply with encodied structure in a ferrogeneous requirements
	LETTE MORE INCOMING COMPINI WITH COORTING STRUCTURAL PORTORMANCO COMUNAMONTO
B.	Grab Bars, install to comply with specified structural-performance requirements.

END OF SECTION

SECTION 10 44 13

2		FIRE PROTECTION CABINETS
3	PART 1 - 0	GENERAL
4	1.1	SUMMARY
5	Α.	Section Includes:
6		1. Fire-protection cabinets for portable fire extinguishers.
7		a. FEC-1 & FEC-2.
8	1.2	PREINSTALLATION CONFERENCE
9	А.	Preinstallation Conference: Conduct conference at Project site.
10	1.3	ACTION SUBMITTALS
11	Α.	Product Data: For each type of product.
12	B.	Shop Drawings: For fire-protection cabinets.
13	C.	Samples: For each type of exposed finish required.
14	1.4	CLOSEQUT SUBMITTALS
15	 A.	Maintenance data.
16	1.5	COORDINATION
17	 A.	Coordinate size of fire-protection cabinets to ensure that type and capacity of fire extinguishers indicated are
18		accommodated
19	в	Coordinate sizes and locations of fire-protection cabinets with wall denths
20	PART 2 - P	PRODUCTS
21	2.1	PERFORMANCE REQUIREMENTS
22	Δ	Fire-Rated Fire-Protection Cabinets: Listed and labeled to comply with requirements in ASTM F814 for fire-
23		resistance rating of walls where they are installed.
24	2.2	FIRE-PROTECTION CABINET
25	 A	Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be
26	7	incorporated into the Work include, but are not limited to, the following:
27		1 Larsen's Manufacturing Company
28		a Models listed below are basis-of-design
29	в	Cabinet Type: Suitable for fire extinguisher
30	C.	Cabinet Construction: Rated according to wall cabinet is recessed within
31	с.	1 Fire-Rated Cabinets: Construct fire-rated cabinets with double walls fabricated from 0.043-inch-thick cold-
32		rolled steel sheet lined with minimum 5/8-inch-thick fire-harrier material Provide factory-drilled mounting
32		holes
3/	р	Cabinet Material: Cold-rolled steel sheet
35	F.	Fully-recessed or Semi-recessed Cabinet – FEC-1 : One-piece combination trim and perimeter door frame
36	L.	overlapping surrounding wall surface with exposed trim face and wall return at outer edge (backbend) Model
37		2409 series.
38		1 Intent is for all public locations to have fully-recessed cabinets: semi-recessed only when wall denth does not
39		nermit fully-recessed
40		2 Square-Edge Trim: 1-1/4- to 1-1/2-inch backbend depth
41		3 Rolled-Edge Trim: 2-1/2-inch backbend denth
42	F	Surface-Mounted Cabinet – FFC-2 : Cabinet box fully exposed and mounted directly on wall with no trim Model
43		2409 series.
44	G	Cabinet Trim Material: Same material and finish as door
45	о. Н	Door Material: Stainless steel sheet
46	1	Door Style: Solid
47		Door Glazing: N/A
48	к У.	Door Hardware: Manufacturer's standard door-operating bardware of proper type for cabinet type trim style and
40	κ.	door material and style indicated
50	I.	
51	L.	1 Mounting Bracket: Manufacturer's standard steel, designed to secure fire extinguisher to fire-protection
52		cabinet of sizes required for types and capacities of fire extinguishers indicated with plated or baked-
52		enamel finish
54		2 Identification: Lettering complying with authorities having jurisdiction for letter style size spacing and
54		2. Incritication. Lettering comprying with authorities having jurisdiction for letter style, size, spacing, and
55		Identify fire extinguisher in fire-protection cabinet with the words "EIDE EVTINGUISHED "
57		a. Incenting the excitigation in the protection cabinet with the words FIRE EXTINGUISHER.
50		2) Application Process: Silk scrooped
20		2) איר

	1010 y 10,	2027
		3) Lettering Color: Red.
		4) Orientation: Vertical.
	M.	Materials:
		 Cold-Rolled Steel: ASTM A1008/A1008M, Commercial Steel (CS), Type B.
		a. Finish: Baked enamel, TGIC polyester powder coat, HAA polyester powder coat, epoxy powder coat
		or polyester/epoxy hybrid powder coat, complying with AAMA 2603.
		b. Color: As selected by Architect from manufacturer's full range.
		2. Aluminum: ASTM B221 for extruded shapes and aluminum sheet, with strength and durability
		characteristics of not less than Alloy 6063-T5 for aluminum sheet.
		a. Finish: Clear anodic.
		3. Stainless Steel: ASTM A240/A240M or ASTM A666, Type 304.
		a. Finish: ASTM A480/A480M No. 4 directional satin finish,.
		4. Tempered Break Glass: ASTM C1048, Kind FT, Condition A, Type I, Quality q3, 1.5 mm thick.
2	2.3	FABRICATION
	Α.	Fire-Protection Cabinets: Provide manufacturer's standard box (tub) with trim, frame, door, and hardware to suit
		cabinet type, trim style, and door style indicated.
F	PART 3 - E	EXECUTION
3	3.1	INSTALLATION
	Α.	Prepare recesses for recessed and semirecessed fire-protection cabinets as required by type and size of cabinet and
		trim style.
	В.	Install fire-protection cabinets in locations and at mounting heights indicated or, if not indicated, at heights
		acceptable to authorities having jurisdiction.
	C.	Fire-Protection Cabinets: Fasten cabinets to structure, square and plumb.
	D.	Identification: Apply decals and/or vinyl lettering at locations indicated.
	Ε.	Adjust fire-protection cabinet doors to operate easily without binding. Verify that integral locking devices operate
		properly.
		END OF SECTION

SECTION 10 44 16

1	SECTION 10 44 16					
2		FIRE EXTINGUISHERS				
3	PART 1 -	ART 1 - GENERAL				
4	1.1	SUMMARY				
5	Α.	Section includes portable, hand-carried fire extinguishers.				
6		1. FE-1 & FE-2.				
7	1.2	PREINSTALLATION MEETINGS				
8	А.	Preinstallation Conference: Conduct conference at Project site.				
9	1.3	ACTION SUBMITTALS				
10	А.	Product Data: For each type of product.				
11	1.4	INFORMATIONAL SUBMITTALS				
12	Α.	Warranty: Sample of special warranty.				
13	1.5	CLOSEOUT SUBMITTALS				
14	Α.	Operation and maintenance data.				
15	1.6	COORDINATION				
16	Α.	Coordinate type and capacity of fire extinguishers with fire-protection cabinets to ensure fit and function.				
17	1.7	WARRANTY				
18	Α.	Special Warranty: Manufacturer's standard form in which manufacturer agrees to repair or replace fire				
19		extinguishers that fail in materials or workmanship within specified warranty period.				
20		1. Warranty Period: Six years from date of Substantial Completion.				
21	PART 2 -	PRODUCTS				
22	2.1	PERFORMANCE REQUIREMENTS				
23	Α.	NFPA Compliance: Fabricate and label fire extinguishers to comply with NFPA 10, "Portable Fire Extinguishers."				
24	В.	Fire Extinguishers: Listed and labeled for type, rating, and classification by an independent testing agency				
25		acceptable to authorities having jurisdiction.				
26	2.2	PORTABLE, HAND-CARRIED FIRE EXTINGUISHERS				
27	Α.	Fire Extinguishers: Type, size, and capacity for each fire-protection cabinet and mounting bracket indicated.				
28		1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that				
29		may be incorporated into the Work include, but are not limited to the following:				
30		a. Larsen's Manufacturing Company.				
31		 Models listed below are basis-of-design. 				
32		2. Instruction Labels: Include pictorial marking system complying with NFPA 10, Appendix B.				
33	В.	Multipurpose/ABC Dry-Chemical Type – FE-1: UL-rated UL-rated 4A-80B:C, 10 pound nominal capacity, with				
34		monoammonium phosphate-based dry chemical in manufacturer's standard red enameled container. Model MP-				
35		10.				
36	С.	Wet chemical class K Type (for use in commercial kitchen) – FE-2: UL-rated 2A:K, 2.5 gallon nominal capacity, with				
37		pressure-indicating gage. Model WC2 1/2.				
38	PART 3 -	EXECUTION				
39	3.1	INSTALLATION				
40	Α.	Examine fire extinguishers for proper charging and tagging.				
41		 Remove and replace damaged, defective, or undercharged fire extinguishers. 				
42	В.	Install fire extinguishers in locations indicated and in compliance with requirements of authorities having				
43		jurisdiction.				
44	С.	Mounting Brackets (where noted in drawings): Fasten mounting brackets to surfaces, square and plumb, at				
45		locations indicated.				
46		1. Mounting Height: Top of fire extinguisher to be at 42 inches above finished floor.				
47		END OF SECTION				

May 16, 2024

1			
2			SECTION 105100
3			PLASTIC LOCKERS
4			
5	PARI	1 GE	
6	1.1	SECT	
1		A.	Lockers.
8	1.2	REFE	RENCES
9		Α.	NFPA 286 - Standard Methods of Fire Tests for Evaluating Contribution of Wall and Ceiling
10			Interior Finish to Room Fire Growth.
11	1.3	SUBM	ITALS
12		Α.	Product Data: Manufacturer's data sheets on each product to be used, including:
13			1. Preparation instructions and recommendations.
14			2. Storage and handling requirements and recommendations.
15		-	3. Installation methods.
16		В.	Shop Drawings: Drawings shall be submitted showing individual locker size and overall
17		-	dimensions.
18		С.	Selection Samples: For each finish product specified, two complete sets of color chips
19		<u></u>	representing manufacturer's full range of available colors and patterns.
20	1.4	QUALI	ITY ASSURANCE
21		Α.	Manufacturer Qualifications: A company regularly engaged in manufacture of products
22		:	specified in this section, and whose products have been in satisfactory use under similar
23		- · ·	service conditions for not less than 5 years.
24		В.	Installer Qualifications: A company regularly engaged in Installation of products specified in
25	4 5		this Section, with a minimum of 5 years experience.
20	1.5		ERY, STORAGE, AND HANDLING
21		A	Store products in manufacturer's unopened packaging until ready for installation.
20		Б.	Locker components shall be stored hat until assembly. All finishes shall be protected from
29	1.0		solling and damage during handling.
3U 21	1.0	A	EUT UUNDITIONS Maintain anvironmental conditions (temperature, humidity, and ventilation) within limits
ວ I ວ ວ		А.	recommended by manufacturer for entingum reculte. Do not install products under
ວ∠ ວວ			anvironmentel conditions outside manufacturer's recommonded limits
20 24	17		
34 25	1.7		Mini T Warranty: Manufacturar guarantoos its plastic against broakago, corrector, and
30		А.	delamination under normal conditions for 15 years from the data of receipt by the sustemer
37			If materials are found to be defective during that period for reasons listed above, the
38			materials will be replaced free of charge. (Labor not included in warranty)
30	DADT	2 00	ODUCTS
40	2.1		ODUCIS ORMANCE REQUIREMENTS
40 //1	2.1		Fire Performance: Tested in accordance with and pass the acceptance criteria of NEPA
42 42		Π.	286
42 43		B	Regulatory Regulirements: Comply with applicable provisions in the USDO I's "2010 ADA
40 44		D.	Standards for Accessible Design" and ICC A117.1 for toilet compartments designated as
45			arressible
46	22	MANU	FACTURERS
47	2.2	Δ	Manufacturers: Subject to compliance with requirements, available manufacturers offering
48		<i>,</i>	products that may be incorporated into the Work include, but are not limited to the following:
49			1 Scranton Products
50		B.	Basis-of-Design: Scranton Products, DuraLife Plastic School Lockers.
51	2.3	LOCK	ERS
52		A.	Design: Solid plastic storage locker.
53			1. See drawings for configuration.
54		В.	Size: Individual and stack height as indicated on drawings.
55			1. Locker Depth: 18 inches.
-			

SPECIFICATION

	May 16, 2024			
1			2. Locker Width: 15 inches.	
2		C.	Hardware:	
3			 Plastic Coat Hook: Top-mounted, two-pronged plastic coat hook. 	
4			2. Padlock hasp.	
5			3. Horizontal venting.	
6			4. Continuous hinge.	
7			5. Top: Slope top.	
8			6. End panel.	
9	2.4	CONS	STRUCTION	
10		Α.	Locker Doors and Frames: Made from high impact, high density polyethylene (HDPE)	
11			formed under high pressure into solid plastic components 1/2 inch thick with homogeneous	
12			color throughout.	
13		В.	Sides, Tops, Bottoms, Backs, and Shelves: Made from high impact, high density,	
14			polyethylene (HDPE) formed under pressure into solid plastic components 3/8 inch thick with	
15			homogenous natural color throughout. Vertical back, outside, inside panels, horizontal top,	
16		~	bottom, divider, and shelves will be grey in color.	
1/		C.	End Panels: Filler panels of plastic material in color of locker unless noted otherwise as an	
18		-	accent color.	
19		D.	Continuous Latch: Made from high impact HDPE plastic and capable of accepting various	
20			locking mechanisms. The spring-loaded latch shall be securely fastened to the entire length	
21		-	of the door providing a quiet positive latching function.	
22		E.	Door Hinge: Heavy duty zinc-plated steel. Full length, assembled onto door and locker front.	
23		F.	Assembly Profile: To be full height of lockers. Profile to be tongue-and-groove joint	
24		C	Construction using 3/8 inch thick HDPE.	
20		G.	Coal Hooks. Two-prong, high impact plastic, mounted to bottom of shell of divider. One	
20	25			
21	2.5		INIALO	
20		А.	1 Posing compounded under high pressure forming a single component which is	
29			waterproof, popalsorbent and has a self-lubricating surface that resists marks from	
31			nens, nencils, markers and other writing instruments	
32		B	HDPE Components: To have a smooth "orange neel" finish Locker doors and door frames	
33		D.	to be the same color	
34			Color: As selected from manufacturer's standard colors	
35	26	FARR	ICATION	
36	2.0	Δ	Locker Components: Eabricated square and rigid with a finish free of scratches and chins	
37		В.	Solid Plastic Locker Components: Snap together at profile connections or slide together at	
38			dovetail connections for easy assembly and provide a solid and secure anti-racking	
39			bookcase component construction for clean lines and precise reveals. Adjacent lockers	
40			share a common side panel. Locker units are manufactured for assembly in a group of no	
41			more than three adjacent lockers.	
42	PART	3 EX	(ECUTION	
43	3.1	EXAM	MINATION	
44		A.	Do not begin installation until substrates have been properly prepared.	
45		В.	If substrate preparation is the responsibility of another installer, notify Architect of	
46			unsatisfactory preparation before proceeding.	
47	3.2	PREF	PARATION	
48		Α.	Clean surfaces thoroughly prior to installation.	
49		В.	Prepare surfaces using the methods recommended by the manufacturer for achieving the	
50			best result for the substrate under the project conditions.	
51		C.	Report discrepancies to the Architect.	
52	3.3	INST/	ALLATION	
53		A.	Install in accordance with manufacturer's instructions.	
54		В.	Install lockers at the location shown in accordance with the manufacturers' instructions for	
55			plumb, level, rigid and flush installations.	

CITY OF MADISON

SPECIFICATION

	May	May 16, 2024			
1		C.	Anchor the units to the wall studs through the locker back and to the floor using 1-1/2 inch		
2			Tapcon screws.		
3		D.	Lockers shall be installed on a concrete base as indicated in drawings.		
4	3.4	ADJ	USTING		
5		Α.	Hardware Adjustment: Adjust and lubricate hardware in accordance with manufacturer's		
6			written instructions for proper operation.		
7	3.5	PROTECTION			
8		Α.	Protect installed products until completion of project.		
9		В.	Touch-up, repair or replace damaged products before Substantial Completion.		
10					
11			END OF SECTION		

SECTION 11 66 23

1		SECTION 11 66 23
2		GYMNASIUM EQUIPMENT
3	PART 1 - (GENERAL
4	1.1	SUMMARY
5	Α.	Section Includes:
6		1. Basketball equipment.
7		2. Volleyball equipment.
8		3. Safety pads.
9	1.2	PREINSTALLATION MEETINGS
10	Α.	Preinstallation Conference: Conduct conference at Project site.
11	1.3	ACTION SUBMITTALS
12	Α.	Product Data: For each type of product.
13	В.	Shop Drawings: For gymnasium equipment.
14		1. Include plans, elevations, sections, and attachment details.
15		2. Include diagrams for power, signal, and control wiring.
16	С.	Samples: For each exposed product and for each item and color specified.
17	1.4	INFORMATIONAL SUBMITTALS
18	Α.	Coordination Drawings: Court layout plans, reflected ceiling plans, and other details, drawn to scale, and
19		coordinated with ceiling-suspended gymnasium equipment, floor inserts, game lines, and markers applied to
20		finished flooring, and coordinated with each other, using input from installers of the items involved.
21	В.	Product Certificates: For each type of gymnasium equipment.
22	C.	Sample warranty.
23	1.5	CLOSEOUT SUBMITTALS
24	Α.	Operation and maintenance data.
25	1.6	QUALITY ASSURANCE
26	Α.	Installer Qualifications: An entity that employs installers and supervisors who are trained and approved by
27		manufacturer.
28	1.7	WARRANTY
29	Α.	Special Warranty: Manufacturer agrees to repair or replace components of gymnasium equipment that fail in
30		materials or workmanship within specified warranty period.
31		1. Warranty Period: Five years from date of Substantial Completion.
32	PART 2 -	PRODUCTS
33	2.1	BASKETBALL EQUIPMENT
34	Α.	Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be
35		incorporated into the Work include, but are not limited to the following:
36		1. Porter – basis-of-design.
37	В.	Standard Rules: Provide equipment according to the requirements of NFHS's "Basketball Rules Book."
38	С.	Connections: Manufacturer's standard connections or connections recommended in writing by manufacturer and
39		complying with Section 05 50 00 "Metal Fabrications" of size and type required to transfer loads to building
40		structure.
41	D.	Overhead-Supported Backstops:
42		1. Folding Type: Manufacturer's standard assembly for side-folding backstop, with hardware and fittings to
43		permit folding.
44		2. Goal Height Adjuster: Adjustable from 8 to 10 feet to top of ring with gear-drive mechanism, locking in any
45		position within adjustment range, with visible height scale attached to side of framing.
46		a. Operation:
47		1) Electrical: Electric operation with integral gear-drive motor, with limit switches preset to
48		goal heights.
49	Ε.	Winch: Hoist consisting of heavy-duty, fully enclosed worm-gear; brake; cable drum; cable; and fittings, for
50		mounting on wall with equipment-mounting board; designed to move and hold backboard in any raised or lowered
51		position.
52	F.	Backstop Electric Operator: Provide operating machine of size and capacity recommended in writing by
53		manufacturer for equipment specified, with electric motor and factory-prewired motor controls, starter, gear-
54		reduction unit, and remote controls. Coordinate wiring requirements and electrical characteristics with building
55		electrical system.
56		1. Electrical Components, Devices, and Accessories: Listed and labeled according to NFPA 70, by a qualified
57		testing agency, and marked for intended location and application.
58		2. Motor Electrical Characteristics:

May 16	5, 2024		
	a. Horsepower: 3/4 hp.		
	b. Voltage: 115 V ac, single phase, 60 hertz.		
	3. Remote-Control Station(s): NEMA ICS 6, Type 1 enclosure for recessed or flush mounting and momenta		
	contact, three-position, switch-operated control with up, down, and off functions.		
	a. Keys: Provide [one key] [two keys] [one set of dual keys] [two sets of dual keys] [dual keys, o		
	Key for up and one for down) per station.		
	4. Limit Switches. Adjustable Switches at each backstop, interfocked with motor controls and set		
G	Baskethall Backhoards:		
0.	1 Basis-of-Design: Bison		
	a Model: BA42XI – See Drawings for location(s)		
	 Model: BA48XI – See Drawings for location(s). Model: BA48XI – See Drawings for location(s). 		
	c. Other manufacturers will be considered if they substantially match the basis-of-design models.		
Н.	Goal-Mounting Assembly: Compatible with goal, backboard, and backstop.		
I.	Basketball Goals: Basket ring complete with flanges, braces, attachment plate, and evenly spaced loops weld		
	around underside of ring.		
	1. Single-rim basket ring competition goal.		
	2. Double-rim basket ring.		
	3. Туре:		
	a. Fixed: Nonmovable.		
	b. Movable: pressure-release design with manufacturer's standard breakaway mechanism a		
	rebound characteristics identical to those of fixed, nonmovable ring.		
	4. Finish: Manufacturer's standard finish.		
J.	Basketball Nets: 12-loop-mesh net, between 15 and 18 inches long, sized to fit ring diameter, and as follows:		
	1. Cord: Made from white cotton.		
К.	Backboard Safety Pads: Designed for backboard thickness and extending continuously along bottom and up sides		
	backboard and over backstop according to manufacturer's standard design.		
	1. Color: As selected by Architect from manufacturer's full range.		
L.	Basketball Shot Clock:		
	1. Dasis-01-Design. Disoli.		
	a. Model. In-time whereas with an necessary mounting hardware and accessories to be a fun-		
	h Other manufacturers will be considered if they substantially match the basis-of-design model		
22	VOLLEYBALL FOLLIPMENT		
Д .	Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may		
73.	incorporated into the Work include, but are not limited to the following:		
	1. Porter – basis-of-design.		
В.	Standard Rules: Provide equipment according to the requirements of NFHS's "Volleyball Rules Book".		
C.	Floor Insert: Solid-brass floor plate and steel pipe sleeve, concealed by floor plate, with capped bottom end, 3 in		
	diameter, minimum 12 inches long, to securely anchor pipe sleeve as indicated on Drawings; with anchors design		
	for securing floor insert to floor substrate indicated; quantity as indicated on Drawings.		
	1. Flush Floor Plate: Manufacturer's standard hinged access cover, designed to be flush with adjacent flooring		
	Provide one tool(s) for unlocking access covers.		
	2. Floor Plate: 5 inch diameter brass.		
2.3	SAFETY PADS		
Α.	Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may		
	incorporated into the Work include, but are not limited to the following:		
_	1. Porter Supersafe.		
В.	Surface-Burning Characteristics: Comply with ASTM E84; testing by a qualified testing agency. Identify products w		
	appropriate markings of applicable testing agency.		
	1. Flame-Spread Index: 25 or less.		
c	2. Sinoke-Developed Index: 50 or less.		
C.	rad coverings. Provide safety pad fabric covering that is fabricated from puncture- and tear-resistant, PVC-coal		
	with surface-burning characteristics indicated and lined with fire-retardant liner		
	With surface-building characteristics multated, and med with meretal dalit miler. Wall Safety Pads: Padded wall wainscot nanels designed to be attached in a continuous row: each nanel section		
υ.	consisting of fill laminated to backer board with visible surfaces fully covered by seamless fabric covering free		
	sag and wrinkles and firmly attached to back of backer board		
	and thinked the first actualized to back of backet both at		
	May 16,	2024	
----	------------	--	--
1		1. Backer Board: Minimum 7/16-inch-thick wood board.	
2		2. Fire-Resistive Fill: Multiple-impact-resistant foam minimum 2-inch-thick, fire-resistive neoprene, 6.0-lb/cu.	
3		ft. density.	
4		3. Size: Each panel section 24 inches wide by minimum 72 inches long.	
5		4. Number of Modular Panel Sections: As indicated on Drawings.	
6		5. Installation Method: Manufacturer's standard.	
7	E.	Fabric Covering Color(s): As selected by Architect from manufacturer's full range for one color(s).	
8	F.	Cutout Trim: Manufacturer's standard flanged cutout trim kits for fitting pads around switches, receptacles, and	
9		other obstructions.	
10		1. Color: Gray.	
11	2.4	MATERIALS	
12	Α.	Support Cable: Manufacturer's standard galvanized-stranded-steel wire rope. Provide fittings according to the wire	
13		rope manufacturer's written instructions for size, number, and installation method.	
14	В.	Support Chain and Fittings: For chains used for overhead lifting, provide Grade 80 heat-treated alloy-steel chains,	
15		according to ASTM A391/A391M, with commercial-quality, hot-dip galvanized steel connectors and hangars.	
16	С.	General-Purpose Chain: For chains not used for overhead lifting, provide carbon steel chain, according to	
17		ASTM A413/A413M (Grade 30 proof coil chain or higher grade recommended by gymnasium equipment	
18		manufacturer). Provide coating type, chain size, number, and installation method according to manufacturer's	
19		written instructions.	
20	D.	Castings and Hangers: Malleable iron, according to ASTM A47/A47M; grade as required for structural loading.	
21	Ε.	Softwood Plywood: DOC PS 1, exterior.	
22	F.	Particleboard: ANSI A208.1.	
23	G.	Equipment-Mounting Board: Wood, transparent or neutral-color-painted finish; size and quantity as required to	
24		mount gymnasium equipment according to manufacturer's written instructions.	
25	Н.	Anchors, Fasteners, Fittings, and Hardware: Gymnasium equipment manufacturer's standard corrosion-resistant c	
26		noncorrodible units; concealed.	
27	I.	Grout: Nonshrink, nonmetallic, premixed, factory-packaged, nonstaining, noncorrosive, nongaseous grout,	
28		according to ASTM C1107/C1107M, with minimum strength recommended in writing by gymnasium-equipment	
29		manufacturer.	
30	PART 3 - I	EXECUTION	
31	3.1	INSTALLATION, GENERAL	
32	Α.	Comply with manufacturer's written installation instructions.	
33	В.	Permanently Placed Gymnasium Equipment and Components: Install rigid, level, plumb, square, and true; anchored	
34		securely to supporting structure; positioned at locations and elevations indicated; in proper relationship to adjacent	
35		construction; and aligned with court layout.	
36	С.	Connections: Connect electric operators to building electrical system.	
37	D.	Removable Gymnasium-Equipment Components: Assemble in place to verify that equipment and components are	
38		complete and in proper working order. Disassemble removable gymnasium equipment after assembled	
39		configuration is approved by Owner, and store units in location indicated on Drawings.	
40	Ε.	Adjust movable components of gymnasium equipment to operate safely, smoothly, easily, and quietly; free from	
41		binding, warp, distortion, nonalignment, misplacement, disruption, or malfunction, throughout entire operational	
42		range; and lubricate as recommended in writing by manufacturer.	
43	3.2	DEMONSTRATION	
44	Α.	Train Owner's maintenance personnel to adjust, operate, and maintain gymnasium equipment.	
45		END OF SECTION	

Page Intentionally Left Blank

1

SECTION 11 66 53

2		GYMNASIUM DIVIDERS	
3	PART 1 -	GENERAL	
4	1.1	SUMMARY	
5	Α.	Section Includes:	
6		1. Electric operators.	
7		2. Divider curtains.	
8		3. Divider system accessories.	
9	1.2	PREINSTALLATION MEETINGS	
10	Α.	Preinstallation Conference: Conduct conference at Project site.	
11	1.3	ACTION SUBMITTALS	
12	Α.	Product Data: For each type of product.	
13	В.	Shop Drawings: For gymnasium dividers.	
14		1. Include diagrams for power, signal, and control wiring.	
15	С.	Samples: For each exposed product and for each item and color specified.	
16	1.4	INFORMATIONAL SUBMITTALS	
17	Α.	Coordination Drawings: Reflected ceiling plans with divider-curtain layouts, drawn to scale, on which the following	
18		items are shown and coordinated with each other, based on input from installers of the items involved:	
19	В.	Product Certificates: For each type of gymnasium divider.	
20	1.5	CLOSEOUT SUBMITTALS	
21	Α.	Operation and maintenance data.	
22	1.6	QUALITY ASSURANCE	
23	Α.	Installer Qualifications: An entity that employs installers and supervisors who are trained and approved by	
24		manufacturer.	
25	PART 2 -	PRODUCTS	
26	2.1	TOP-ROLL DIVIDER SYSTEMS	
27	Α.	Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be	
28		incorporated into the Work include, but are not limited to the following:	
29		1. Porter.	
30		a. Basis-of-Design: 92085000 2	
31		b. Other manufacturers will be considered if they substantially match the basis-of-design model.	
32	В.	Divider-Curtain System: Electrically operated, belt-guided, top/center-roll drive pipe, and as follows:	
33		1. Outer Edge Hems: Double turned and welded.	
34		Supports and Fittings: Corrosion-resistant steel clamps and hangers.	
35		3. Drive Pipe and Curtain Battens: Steel pipe or tubing.	
36	2.2		
37	2.3	ELECTRIC OPERATORS	
38	Α.	Provide factory-assembled electric operation system of size and capacity recommended in writing and provided by	
39		gymnasium divider manufacturer for gymnasium dividers specified, with electric motors and factory-prewired	
40		motor controls, control devices, and accessories required for proper operation.	
41		1. Include wiring from control stations to motors and between synchronizer and dual motors for long curtains.	
42		Coordinate operator wiring requirements and electrical characteristics with building electrical system.	
43	В.	Electrical Components, Devices, and Accessories: Listed and labeled according to NFPA 70, by a qualified testing	
44		agency, and marked for intended location and application.	
45	С.	Motor Electrical Characteristics:	
46		1. Horsepower: 1 hp.	
47		2. Voltage: 115 V ac, single phase, 60 hertz.	
48	D.	Limit Switches: Adjustable switches at each divider curtain, interlocked with motor controls and set to automatically	
49		stop divider curtain at fully extended and fully retracted positions.	
50	Ε.	Control System:	
51		1. Key-Switch Operation: NEMA ICS 6, Type 1 enclosure, momentary-contact, three-position switch-operated	
52		control with up, down, and off functions.	
53		a. Keys: Provide key(s) per station.	
54	2.4	DIVIDER CURTAINS	
55	Α.	Upper Curtain, Mesh: Woven mesh of polyester yarn double-coated with vinyl, weighing not less than 19 oz./sq.	
56		yd	
57		1. Mesh Color: As selected by Architect from full range of industry colors and color densities.	
58	В.	Lower Curtain, Solid: Woven polyester fabric double-coated with vinyl, 19 oz./sq. yd., 8-foot height above floor.	

	May 16, 2024				
1		1. Fabric Color(s): One color(s), as selected by Architect from full range of industry colors and color densities.			
2	С.	Hems: Folded and electronically welded.			
3	D.	Seams: Electronically welded.			
4	Ε.	Overall Curtain Height: 20 feet.			
5	F.	Bottom of Curtain: Approximately 2 inches above finished floor.			
6	G.	Divider-Curtain Flame-Resistance Rating: Passes NFPA 701 Test 2.			
7	2.5	DIVIDER SYSTEM ACCESSORIES			
8	Α.	Safety Lock: Locks drive system when speed exceeds manufacturer's recommended speed.			
9	В.	Audible Motion Alarm: Provide alarm with intermittent warning tone when curtain is raised or lowered.			
10	2.6	SUPPORT MATERIALS AND FASTENERS			
11	Α.	Support Chain and Fittings: For chains used for overhead lifting, provide Grade 80, heat-treated alloy-steel chains,			
12		according to ASTM A391/A391M, with commercial-quality, hot-dip galvanized or zinc-plated steel connectors and			
13		hangers.			
14	В.	General-Purpose Chain: For chains not used for overhead lifting, provide carbon steel chain, according to			
15		ASTM A413/A413M, Grade 30 proof coil chain or higher grade recommended by gymnasium divider manufacturer.			
16		Provide coating type, chain size, number, and installation method according to manufacturer's written instructions.			
17	С.	Anchors, Fasteners, Fittings, and Hardware: Manufacturer's standard corrosion-resistant or noncorrodible units;			
18		concealed.			
19	PART 3 -	PART 3 - EXECUTION			
20	3.1	INSTALLATION, GENERAL			
21	Α.	Comply with manufacturer's written installation instructions.			
22	В.	Install gymnasium dividers level, plumb, square, and true; anchored securely to supporting structure; positioned at			
23		locations and elevations indicated; in proper relation to adjacent construction; and aligned with sport-court layout.			
24	С.	Connections: Connect electric operators to building electrical system.			
25	D.	Adjust movable components of gymnasium dividers to operate safely, smoothly, easily, and quietly, free from			
26		binding, warp, distortion, uneven tension, nonalignment, misplacement, disruption, or malfunction, throughout			
27		entire operational range; and lubricate as recommended in writing by manufacturer.			
28	Ε.	Limit Switch Adjustment: Set and adjust upper and lower limit controls.			
29	3.2	DEMONSTRATION			
30	Α.	I rain Owner's maintenance personnel to adjust, operate, and maintain gymnasium dividers.			
31		END OF SECTION			

SECTION 12 36 61.16 ~~

		SECTION 12 36 61.16	
		SOLID SURFACING COUNTERTOPS	
PAF	RT 1 - (GENERAL	
1.1		SUMMARY	
	Α.	Section Includes:	
		1. Solid surface material countertops.	
		2. Solid surface material backsplashes – if identified in drawings.	
		 Solid surface material end splashes – if identified in drawings. Solid surface material ender factors if identified in drawings. 	
1 2		4. Solid surface material apron fronts – if identified in drawings.	
1.2		ACTION SUBMITTALS	
	A.	Product Data: For countertop materials and sinks.	
	в.	shop brawings: For countertops, show materials, infishes, edge and backsplash promes, methods of joining, and	
	c	Culouis for plumbing fixibles.	
DAD	с. этэг		
2 1	1 2 - 1		
2.1	Δ	Solid Surface Material: Homogeneous-filled plastic resin complying with ISEA 2-01	
	л.	1 Manufacturers: Colors and Patterns: As identified in the finish schedule and key in the drawings	
		 Type: Provide Standard type unless Special Purpose type is indicated 	
		3 Integral Sink Bowls: Comply with CSA R45 5/IAPMO 7124	
	в	Particleboard: ANSI A208 1 Grade M-2-Exterior Glue	
	С.	Plywood: Exterior softwood plywood complying with DOC PS 1. Grade C-C Plugged, touch sanded.	
2.2	0.	FABRICATION	
	A.	Fabricate countertops according to solid surface material manufacturer's written instructions and to the	
		AWI/AWMAC/WI's "Architectural Woodwork Standards."	
		1. Grade: Premium.	
	В.	Configuration:	
		1. Front: See cabinet details in drawings.	
		2. Backsplash: Straight, slightly eased at corner.	
		3. End Splash: Matching backsplash.	
	C.	Countertops:	
		1. 3/4-inch-thick, solid surface material with front edge built up with same material.	
	D.	Backsplashes: 3/4-inch-thick, solid surface material.	
	E.	Joints:	
		1. Fabricate countertops without joints to greatest extent possible.	
	F.	Cutouts and Holes:	
		1. Undercounter Plumbing Fixtures: Make cutouts for fixtures in shop using template or pattern furnished by	
		fixture manufacturer. Form cutouts to smooth, even curves.	
2.3		INSTALLATION MATERIALS	
	A.	Adhesive: Product recommended by solid surface material manufacturer.	
	В.	Sealant for Countertops: Comply with applicable requirements in Section 07 92 00 "Joint Sealants."	
PAF	RT 3 - E	EXECUTION	
3.1		INSTALLATION	
	Α.	Fasten countertops by screwing through corner blocks of base units into underside of countertop. Predrill holes for	
		screws as recommended by manufacturer.	
	В.	Fasten subtops to cabinets by screwing through subtops into cornerblocks of base cabinets. Shim as needed to align	
		subtops in a level plane.	
	C.	Secure countertops to subtops with adhesive according to solid surface material manufacturer's written	
		instructions.	
	D.	Bond joints with adhesive and draw tight as countertops are set. Mask areas of countertops adjacent to joints to	
		prevent adhesive smears.	
	Ε.	Install backsplashes and end splashes by adhering to wall and countertops with adhesive.	
	F.	Install aprons to backing and countertops with adhesive.	
	G.	Complete cutouts not finished in shop. Mask areas of countertops adjacent to cutouts to prevent damage while	
		cutting. Make cutouts to accurately fit items to be installed, and at right angles to finished surfaces unless beveling	
		is required for clearance. Ease edges slightly to prevent snipping.	
	Н.	Apply sealant to gaps at walls; comply with Section 07 92 00 "Joint Sealants."	
		END OF SECTION	

Page Intentionally Left Blank

1	SECTION 21 05 00		
2	COMMON WORK RESULTS FOR FIRE-SUPPRESSION		
3 4			
5	PART 1 - GENERAL		
6			
7	SCOPE		
8	This section includes information common to two or more technical fire protection specification sections or items that are of a		
9	general nature, not conveniently fitting into other technical sections. Included are the following topics:		
10			
12	Scope		
13	Related Work		
14	Related Documents		
15	Regulatory Requirements		
16	Reference Standards		
1/	Quality Assurance		
19	Definitions		
20	Coordination		
21	Continuity of Existing Services		
22	Protection of Finished Surfaces		
23	Sleeves and Openings		
24 25	Sealing and Firestopping		
26	Provisions for Future		
27	Off Site Storage		
28	Submittals		
29	Operating and Maintenance Instructions		
30	Record Drawings		
31	Training of Owner Personnel		
32 33	Cleaning		
34	Warranty		
35	Certified Startup Reports		
36			
37	PART 2 - PRODUCTS		
38	Access Panels and Doors Ding Departmentions		
39 40	Sealing and Firestonning		
41			
42	PART 3 - EXECUTION		
43	Demolition		
44	Concrete Work		
45	Openings, Cutting and Patching		
40 17	Building Access		
48	Coordination of Work		
49	Pipe Penetrations		
50	Identification		
51	Lubrication and Maintenance		
52	Sleeves		
53			
55	Provisions of Division 01 shall govern work under this Section.		
56			
57	This section applies to all Division 21 Sections of Fire Suppression.		
58			
59	REGULATORY REQUIREMENTS		
бU 61	Kerer to Division 01 of the Project Manual.		
62	Codes and Standards:		
63	Fire Protection work shall conform to the requirements of Wisconsin Building Code (COMM). NFPA Standards, and local		
64	regulations regarding design, materials and installation.		

1					
2	Materials a	and work	manship shall comply with applicable Codes, local ordinances, industry standards and utility regulations. In		
3	case of differences between Codes, and the Contract Documents, the most stringent shall govern.				
4					
5	Non-Compliance:				
6	Should the Contractor perform any work that does not comply with the above requirements, he shall bear all costs necessary to				
7	correct the	e deficien	cies.		
8					
9	Permits, In	nspection	is, and Fees:		
10	Request ar	nd obtain	permits and inspection appointments.		
11					
12	Provide fee	es and ch	arges for approvals, reviews, or other inspections.		
13					
14	Include cop	pies of th	e certificates in the Operating and Maintenance Instructions.		
15	_				
16	Fees and c	harges as	seessed by local utilities for water or other services shall be included in the bid.		
1/					
18	REFERENC	ESIAND	ARDS		
19	Abbreviatio	ons of sta	andards organizations referenced in this and other sections are as follows:		
20	A.C.A				
21			American Gas Association		
22			American National Standards Institute		
25			American Society of Plumbing Engineers		
24			American Society of realising and Materials		
25			American Society for results and Materials		
20			American Welding Society		
28	CGA		Compressed Gas Association		
29	CS		Commercial Standards, Products Standards Sections, Office of Engineering Standards Service, NBS		
30	DSPS		State of Wisconsin Department of Professional Services		
31	EPA		Environmental Protection Agency		
32	FM		Factory Mutual System		
33	FS		Federal Specifications, Superintendent of Documents, U.S.Government Printing Office		
34	IAPMO		International Association of Plumbing & Mechanical Officials		
35	IEEE		Institute of Electrical and Electronics Engineers		
36	ISA		Instrument Society of America		
37	MCA		Mechanical Contractors Association		
38	MICA		Midwest Insulation Contractors Association		
39	MSS		Manufacturer's Standardization Society of the Valve & Fitting Industry, Inc.		
40	NBS		National Bureau of Standards		
41	NEC		National Electric Code		
42	NEMA		National Electrical Manufacturers Association		
43	NFPA		National Fire Protection Association		
44	STI		Steel Tank Institute		
45	UL		Underwriters Laboratories Inc.		
46	_				
47	QUALITY A	SSURAN	CE		
48	Substitutio	on of Mat	erials: Refer to Division 01 of the Project Manual.		
49					
50	All product	ts and ma	sterials used are to be new, undamaged, clean and in good condition. Existing products and materials are not		
51	to be reuse	ed unless	specifically indicated.		
52			the second state of the se		
53	where equ	uipment	or accessories are used which differ in arrangement, comparation, dimensions, ratings, or engineering		
54	parameter		use indicated on the contract documents, the contractor is responsible for all costs involved in integrating the		
55	are placed	L OF ALLES	solies into the system and for obtaining the interface performance from the system into which these items		
57	are placed.	•			
58			ND SYMBOLS		
59	Key to ahh	reviation	s and symbols shall be on the Drawings		
60		. c viation	s and symbols shall be on the browings.		
61	The follow	ing are a	dditional abbreviations used in the Specifications:		
62			·····		
63	A	4/Е	Architect/Engineer		
64	0	GC	General Contractor		

1	PC	Plumbing Contractor
2	FPC	Fire Protection Contractor
3	HC	Heating Ventilating and Air Conditioning Contractor
4	EC	Electrical Contractor
5	KEC	Kitchen Equipment Contractor
6	LEC	Lab Equipment Contractor

LEC Lab Equipment Contractor

DEFINITIONS 8

9 Furnish:

10 Supply and deliver to Project site ready for unpacking, assembly and installation

11 12 Install:

13 Operations at Site including unpacking, assembling, erecting, placing, anchoring, applying, finishing, cleaning, and connecting 14 related devices required for product fully functional for intended use after installation.

15

7

16 Provide:

Furnish and install, such that product is fully functional for intended use. 17 18

19 COORDINATION

The Drawings show the general arrangement of piping and equipment and shall be followed as closely as actual building 20 construction and the work of other trades permits. Architectural and Structural Drawings shall take precedence. Because of the 21 scale of the Drawings, it is not possible to indicate all offsets, fittings, and accessories which may be required. Investigate conditions 22 affecting the Work and arrange accordingly, providing offsets, fittings and accessories as may be required to meet conditions. 23 24

25 CONTINUITY OF EXISTING SERVICES

26 Refer to Division 01 of the Project Manual.

27

31

28 Do not interrupt or change existing services without prior written approval from the Owner's Project Representative. When interruption is required, coordinate scheduling of down-time with the Owner to minimize disruption to his activities. Unless 29 30 specifically stated, all work involved in interrupting or changing existing services is to be done during normal working hours.

PROTECTION OF FINISHED SURFACES 32

33 Refer to Division 01, of the Project Manual.

34

35 SEALING AND FIRESTOPPING

Sealing and firestopping of sleeves/openings between piping, etc. and the sleeve or structural opening shall be the responsibility 36 37 of the contractor whose work penetrates the opening. The contractor responsible shall hire individuals skilled in such work to do 38 the sealing and fireproofing. These individuals hired shall normally and routinely be employed in the sealing and fireproofing 39 occupation. 40

EOUIPMENT FURNISHED BY OTHERS 41

42 None.

PROVISIONS FOR FUTURE 44

45 None.

46

43

47 OFF SITE STORAGE

48 Refer to Division 01 of the Project Manual.

49 50 SUBMITTALS

51 Refer to Division 01, of the Project Manual.

52

- Submit shop drawings with space for approval stamps of GC and A/E. 53
- 54 55 Refer to Division 01, of the Project Manual.

56

57 Not more than two weeks after award of contract but before any shop drawings are submitted, contractor to submit the following fire protection system data sheet. List piping material types, ASTM number, schedule or pressure class, joint type, manufacturer 58

and model number where appropriate. List valves, specialties and equipment with manufacturer and model number. The 59 approved fire protection system data sheet(s) will be made available to the Owners Project Representative for their use on this 60 61 project.

- 62
- 63

1 FIRE PROTECTION SYSTEM DATA SHEET 2 Pipe Service/Sizes Manufacturer/Model No. Remarks Item 3 Pipe 4 Fittings Hangers & Supports 5 Sprinkler Heads 6 7 Valves 8 Specialty Valves 9 Pipe Specialties 10 **Fire Protection Specialties Fire Protection Equipment** 11 12 13 Shop drawing submittals are to be bound in a three ring binder, labeled, contain the project manual cover page and a material 14 index list page showing item designation, manufacturer and additional items supplied with the installation. Submit for all equipment and systems as indicated in the respective specification sections, marking each submittal with that specification 15 16 section number. Mark general catalog sheets and drawings to indicate specific items being submitted and proper identification of equipment by name and/or number, as indicated in the contract documents. Include wiring diagrams of electrically powered 17 18 equipment. 19 Submittals shall be sent to the local Fire Chief or Fire Marshal for review prior to the Architect/Engineer. Include copy of approval 20 21 letter in submission to Architect/Engineer. 22 23 Submit plans indicating water supply location and size, piping layout and size, sprinkler locations and type, hanger locations and 24 type, equipment locations and type, valve locations and type, occupancy classes, hydraulic reference points, design areas and 25 discharge densities. 26 27 Submit hydraulic calculations for water supply and sprinkler systems. Include summary sheet and detailed work sheets. Describe 28 characteristics of water supply and location of effective point used in calculations. Include graph illustration of water supply, hose demand, sprinkler demand. 29 30 Submit sufficient quantities of data sheets and shop drawings to allow the following distribution: 31 32 Operating and Maintenance Manuals 2 copies 33 • Architect/Engineer 2 copies 34 • Local Fire Chief or Marshal 1 copy 35 36 Firestop Systems: 37 Contractor shall submit product data for each firestop system. Submittals shall include product characteristics, performance and 38 limitation criteria, test data, MSDS sheets, installation details and procedures for each method of installation applicable to this project. For non-standard conditions where no UL tested system exists, submit manufacturer's drawings for UL system with 39 40 known performance for which an engineering judgement can be based upon. 41 42 **OPERATING AND MAINTENANCE INSTRUCTIONS** 43 Refer to Division 01 of the Project Manual. 44 Assemble material in three-ring or post binders, using an index at the front of each volume and tabs for each system or type of 45 46 equipment. In addition to the data indicated in the General Requirements, include the following information: 47 Copies of all approved submittals along with approval letters. Manufacturer's wiring diagrams for electrically powered equipment. 48 • 49 • Records of tests performed to certify compliance with system requirements. Certificates of inspection by regulatory agencies. 50 . • Parts lists for equipment and specialties. 51 52 • Manufacturer's installation, operation and maintenance recommendations for equipment and specialties. Valve schedules 53 • 54 • Lubrication instructions, including list/frequency of lubrication 55 Warranties . Additional information as indicated in the technical specification sections 56 57 **RECORD DRAWINGS** 58 59 Refer to Division 01 of the Project Manual. 60 61 In addition to the data indicated in the General Requirements, maintain fire protection layout record drawings and hydraulic calculations on originals prepared by the installing contractor/subcontractor. Include copies of these record drawings and 62 63 calculations with the Operating and Maintenance manuals. 64

1 TRAINING OF OWNER PERSONNEL

Instruct Owner's personnel in the proper operation, maintenance and testing of systems and equipment provided as part of this
 project. Include not less than 2 hours of instruction, using the Operating and Maintenance manuals and record drawings during
 this instruction.

5 6 TESTING

7 Equipment, material and labor required for testing, shall be provided by the Contractor.

9 Contractor shall notify Inspector(s) one day prior to the time when the test is ready to be performed. Contractor shall notify the
 10 A/E of date and time for tests.

11

8

12 After the test, indicate in writing the time, date, name and title of the person approving the test. This shall also include the 13 description and what portion of the system has been tested. The person approving the test shall sign the certification.

14 15 16

18

26

28

30

33

15 Records shall be maintained of testing that has been completed, and shall be made available at the job site to authorities.

17 Upon completion of the work, records and certifications approving testing requirements shall be submitted.

19 Defective work or material shall be replaced or repaired, and the test repeated. Repairs shall be made with new materials.

20 21 **CLEANING**

Contractor shall keep the premises broom clean and free of all surplus materials, rubbish and debris which is caused by his
 employees or resulting from his work.

25 Foreign matter shall be blown out, or flushed out, of pipes, tanks, pumps, strainers, motors, devices, switches, and panels.

27 Identification plates on equipment shall be free of paint and dirt.

29 The Contractor shall leave his portion of the work ready for operation.

31 WARRANTY

32 Warrant that work functions for one year following acceptance of the system(s).

The Contractor shall keep the system in good working order at no expense, unless defects are clearly the result of improper or abnormal usage.

The Contractor shall submit to the A/E upon request for acceptance of the work, written certification that the entire system has
 been installed and adjusted for operation in accordance with the Contract Documents.

PART 2 - PRODUCTS

39 40

41

42

43 ELECTRICAL REQUIREMENTS

44 General:

45 Work shall conform to requirements of Division 26.

4647 Provide wiring diagrams.

4849 ACCESS PANELS AND DOORS

Provide access panels at locations requiring access to mechanical equipment. Locations include, but are not limited to areas above
 drywall ceilings, shaft enclosures and other furred-in spaces concealing valves, ducts or equipment. Provide UL listed, fire rated
 access panels when penetrating fire rated chase or shaft areas.

53

58

Access panels shall be of size required to provide adequate access to equipment. Minimum size shall be 12 inch by 12 inch for
 hand access and 24 inch by 24 inch for body access.

57 Panels shall be Milcor brand or equivalent.

Panels shall include concealed hinges, cam type locking devices, and have frame/border type necessary for particular wall or ceiling construction they are installed. Access panels shall be flush mounted, recessed frame type units. Access panels shall be prime coated steel, able to accept field painting for general applications and stainless steel for use in toilet rooms, shower rooms and similar wet areas.

63

64 Refer to Architectural Room Finish Schedule for wall and ceiling surfaces and finishes.

1 2 For non-security applications, panel construction shall utilize 16 gauge frame with not less than 18 gauge hinged door panel. Door locks shall be screwdriver operated for panels in general location applications and shall be key locked for public area applications. 3 4 5 **PIPE PENETRATIONS** Refer to Division 01 requirements as well as the following. 6 7 Fire, Smoke And Fire/Smoke Rated Surfaces: 8 3M CP 25N/S or CP 25S/L caulk, 3M FS 195 wrap/strip with restricting collar, 3M CS 195 composite sheet, Pipe Shields Inc. Series 9 10 F fire barrier kits, Proset Systems fire rated floor and wall penetrations, Insta-Foam Products Insta-Fire Seal Firestop Foam or Dow 11 Corning Fire Stop System. 12 13 All fire stopping systems shall be provided by the same manufacturer. 14 15 UL listed or tested by independent testing laboratory, approved by State and Local Code jurisdictions. 16 Use product that has a rating not less than rating of wall or floor being penetrated. Reference architectural drawings for 17 18 identification of fire and/or smoke rated walls and floors. 19 Sleeves in concrete to be Schedule 40 steel pipe with integral water stop unless fire stop material used includes a sleeve that is 20 21 an integral part of rated assembly. 22 23 Use firestop putty, caulk sealant, intumescent wrapstrips, intumescent firestop collars, firestop blocks, firestop mortar or a 24 combination of these products to provide a UL listed system for each application required for this project. Provide mineral wool 25 backing where specified in manufacturer's application detail. 26 27 **Non-Rated Surfaces:** 28 Stamped steel, chrome plated, hinged, split ring escutcheons or floor/ceiling plates for covering openings in occupied spaces. 29 30 In exterior wall openings below grade, use modular mechanical type seal consisting of interlocking synthetic rubber links shaped to continuously fill the annular space between the un-insulated pipe and cored opening or a water-stop type wall sleeve. 31 32 33 At interior partitions where pipe penetrations are sealed, use Tremco Dymonic, Sika Corp. Sikaflex 1a, Sonneborn Sonolastic NPI, 34 or Mameco Vulken 116 urethane caulk to effectively seal. Use galvanized sheet metal sleeves in hollow wall penetrations. 35 PART 3 - EXECUTION 36 37 GENERAL 38 39 Coordination Of Work: 40 Review the complete set of Drawings and Specifications and report discrepancies to the A/E. Obtain written instructions for changes necessary. Coordinate with each trade prior to beginning installation and make provisions to avoid interferences. 41 42 Changes required caused by neglect to coordinate shall be made without expense to the project. 43 44 Piping shall not be located above electrical panels. 45 46 Anchor Bolts, Sleeves, and Supports: 47 These items required for the Work shall be furnished by the FPC for proper installation of his work. They shall be installed (except as otherwise specified) by the trade furnishing and installing the material in which they are to be located. Location of anchor 48 49 bolts, sleeves, inserts and supports shall be directed by the trade requiring them. Expense resulting from the improper location or installation of anchor bolts, sleeves, inserts and supports shall be paid for by the Contractor for the trade with responsibility 50 for directing their proper location. 51 52 53 Adjustments In Locations: 54 Locations of pipes and equipment, shall be adjusted to accommodate the work interferences anticipated and encountered. Prior 55 to fabrication determine the exact route and location of each pipe (subject to A/E's approval). 56 57 **Right Of Wav:**

New lines which pitch shall have the right-of-way over those which do not pitch. For example: Gravity drains shall normally have
 right-of-way. Lines whose elevations cannot be changed shall have the right-of-way over lines whose elevations can be changed.
 Notify A/E and other trades of conflicts.

- 61
- 62 Offsets, transitions and changes in direction of electrical raceways, pipes, and ducts shall be made to maintain proper room and
- 63 pitch of sloping lines whether or not indicated on the Drawings.

64

1 DEMOLITION

Perform all demolition as indicated on the drawings to accomplish new work. Where demolition work is to be performed adjacent to existing work that remains in an occupied area, construct temporary dust partition to minimize the amount of contamination of the occupied space. Where pipe is removed and not reconnected with new work, cap ends of existing services as if they were new work. Coordinate work with the Owner to minimize disruption to the existing building occupants.

6

All pipe, sprinklers, equipment, wiring, associated conduit and similar items demolished, abandoned, or deactivated are to be
 removed from the site by the Contractor except as specifically noted otherwise. All designated equipment is to be turned over to
 the Owner for his use at a place and time he so designates. Maintain the condition of material and/or equipment that is indicated
 to be reused equal to that existing before work began.

11 12 CONCRETE V

12 CONCRETE WORK

Provide all layout drawings, anchor bolts, metal shapes, and/or templates required to be cast into concrete or used to form
 concrete for support or installation of plumbing piping, fixtures, specialties and equipment. Coordinate locations of equipment,
 pipe penetrations in wet areas, etc. with the Division 03 Contractor.

16

20

Plumbing related cast-in-place concrete on the exterior of the building to be provided by this Contractor in conformance with
 requirements of Division 03. This includes piping thrust restraints, pipe supports, hydrant supports, manholes, catch basins,
 grease traps, septic tanks, distribution boxes, valve pits, meter pits, cleanout cover pads, yard hydrant pads, etc.

21 OPENINGS, CUTTING AND PATCHING

22 Refer to Division 01 requirements.

23

Provisions for openings including chases, holes and clearances through walls, floors, and roof, ceilings and partitions shall be made in advance of construction of each part of the building. Openings shall (except for pipe sleeves) be provided by the GC for the respective materials in which openings occur, during the construction of the building with the exception of pipe sleeves. Furnish required opening dimensions and locations.

28

If the FPC neglects to inform the GC of his opening requirements before that portion of the building is complete, the FPC shall cut the openings, provide framing and lintels. In the event holes must be cut through reinforced concrete, drill so as to avoid spalling and unnecessary damage or weakening of structural members. No chopping or breaking out is permitted. Before cutting or drilling, the Contractor shall obtain permission from the A/E. Patch adjacent materials and repair damage resulting from the cutting.

34

The FPC may perform core drilling for openings in existing walls and floors at the direction of the A/E. Framed openings shall be by the GC.

37

38 BUILDING ACCESS

Arrange for the necessary openings in the building to allow for admittance of all apparatus. When the building access was not
 previously arranged and must be provided by this contractor, restore any opening to its original condition after the apparatus
 has been brought into the building.

42

43 EQUIPMENT ACCESS

Install all piping, valves, and accessories to permit access to equipment for maintenance. Coordinate the exact location of wall and ceiling access panels and doors with the General Contractor, making sure that access is available for all equipment and specialties. Where access is required in plaster walls or ceilings, furnish the access doors to the General Contractor.

47

Accessible ceilings, (i.e. lay-in ceilings) do not require access panels. Provide color coded thumb tacks or screws, depending on
 surface, for use in accessible ceilings.

50

51 COORDINATION OF WORK

52 Install systems, equipment and piping in cooperation with other trades. Locations of pipes, equipment, fixtures, etc., shall be adjusted to accommodate the work interferences anticipated and encountered. Prior to fabrication determine the exact route and location of each pipe (subject to A/E's approval).

55

Any work that is not coordinated and that interferes with other contractor's work shall be removed or relocated at the installing
 contractor's expense.

58

59 Verify that all devices are compatible for the type of construction and surfaces on which they will be used.

60

61 Offsets, transitions and changes in direction of electrical raceways, pipes and ducts shall be made as required whether or not 62 indicated on the Drawings.

62 Indicated on the Drawings.

1

Provide appropriate sections of work with required wall, roof and floor opening locations and dimensions. If Contractor neglects
 to coordinate information, openings shall be the responsibility of Contractor.

4

5 PIPING INSTALLATION

6 Installation Arrangement:

Install work to permit removal (without damage to other parts) of parts requiring replacement or maintenance. Arrange pipes
 and equipment to permit ready access to valves, cocks, traps, starters, motors, and control components and to clear the openings
 of swinging and overhead doors and of access panels.

11 Connections Different From Those Shown:

Where equipment requiring different arrangement or connections from those shown is used, install the equipment to operate properly and in harmony with the intent of the Drawings and Specifications. When requested by the A/E, submit drawings showing the proposed installation.

15

16 Upon approval of the revisions, make changes in piping, ductwork, supports, insulation, wiring, and panelboards. Provide 17 additional valves, fittings and other additional equipment required for the proper operation of the system resulting from the 18 selection of equipment, including required changes in affected trades. The Contractor shall be responsible for the proper location 19 of rough-in and connections by other trades.

20

21 Changes shall be made at no increase in the Contract amount or additional cost to the other trades.

23 LUBRICATION AND MAINTENANCE

Lubricate bearings with lubricant as recommended by manufacturer before equipment is operated for any reason. Once equipment has been run, maintain lubrication in accordance with the manufacturer's instructions until the work is accepted by the Owner. Maintain log of all lubricants used and frequency of lubrication. Include information in Operating and Maintenance Manuals at completion of project.

27 Manuals at completion of project.28

29 SLEEVES

Provide galvanized sheet metal sleeves for fire rated pipe penetrations through interior and exterior walls to provide a backing for sealant or firestopping. Patch wall around sleeve to match adjacent wall construction and finish. Grout area around sleeve in masonry construction. In finished spaces where pipe penetration through wall is exposed to view, sheet metal sleeve shall be installed flush with face of wall. In existing poured concrete walls where penetration is core drilled, pipe sleeve is not required. Grout holes directly around steel pipe.

35

46

In all piping floor penetrations, fire rated and non-fire rated, top of sleeve shall extend 3/4 inch above the adjacent finished floor. In existing floor penetrations, core drill sleeve opening large enough to insert schedule 40 sleeve and grout area around sleeve with hydraulic setting, non-shrink grout. If the pipe penetrating the sleeve is supported by a pipe clamp resting on the sleeve, weld a collar or struts to the sleeve that will transfer weight to existing floor structure.

40 41 **PIPE PENETRATIONS**

42 General:

Coordinate location of building surface penetrations with appropriate contractors. Furnish sleeves, inserts, and devices to be built
 into structure to contractor performing Work. Prepare Shop Drawings for approval for penetrations of structural elements,
 including floor slabs, shear walls, and bearing walls. Do not allow penetrations to be made until Shop Drawings are approved.

47 Fire Rated Surfaces:

Install products in accordance with the manufacturer's instructions where pipe penetrates a fire rated surface. When pipe is insulated, use product that maintains integrity of insulation and vapor barrier. Where sleeve must be installed in existing floor, grout area around sleeve to restore floor integrity. In wet area floor penetration, top surface of penetration to be 2 inches above adjacent floor with additional height obtained by means of concrete pad poured integral with floor.

53 Non-Rated Surfaces:

Install escutcheons or floor/ceiling plates where pipe penetrates non-fire rated surfaces in occupied spaces. Size units to accommodate insulation, where applicable. Escutcheons are not required when insulation completely covers wall opening and insulation end is trimmed in a neat manner. Occupied spaces for this Paragraph include only those rooms with finished ceilings and penetration occurs below ceiling.

58

In exterior wall openings below grade, place water-stop type wall sleeve before concrete pour or core drill opening after pour.
 Assemble rubber links to proper size for pipe and tighten in place in accordance with manufacturer's instructions.

- 61
- 62 Install galvanized sheet metal sleeve in hollow wall penetrations to provide backing for sealant. Apply sealant to both sides of
- 63 penetration in a manner that annular space between pipe sleeve and pipe or insulation is completely blocked.

64

1 Completely seal (or caulk) around pipe penetrations through non-rated, smoke tight corridor walls in healthcare facilities. Refer 2 to architectural drawings for additional information.

3 4

ESCUTCHEON PLATES

Provide plates on pipes passing through finished floors, walls and ceilings, with outside diameter to cover sleeve opening and 5 6 inside diameter to fit snugly around pipe. Set tight to building surface. Escutcheon plates shall be chromium plated metal.

PAINTING

8 9 Refer to Division 09.

10

7

IDENTIFICATION 11

Identify equipment in mechanical equipment rooms by stenciling equipment number and service with one coat of black enamel 12 13 against a light background or white enamel against a dark background. Use a primer where necessary for proper paint adhesion.

14

15 Where stenciling is not appropriate for equipment identification, engraved name plates may be used. 16

Identify interior piping mains not less than once every 25 feet, not less than once in each room, adjacent to each access door or 17 18 panel, and on both sides of the partition where exposed piping passes through walls or floors. Place flow directional arrows at 19 each pipe identification location. Use one coat of black enamel against a light background or white enamel against a dark 20 background or approved pipe marking label systems.

21 22 Identify valves with signs per NFPA rulings.

23

- 24 Provide hydraulic design information sign of permanently marked weatherproof metal or engraved nameplate material. Secure 25 to main fire risers/valves with brass chain. Information to include location of the design areas, discharge densities, required flow
- and residual pressure at the base of riser, hose stream demand and sprinkler demand. 26
- 27
- 28
- 29

END OF SECTION

Page Intentionally Left Blank

1	SECTION 21 05 29
2	HANGERS AND SUPPORTS FOR FIRE-SUPPRESSION PIPING AND EQUIPMENT
3	
4	
5	PART 1 - GENERAL
6	
7	SCOPE
8 9	This section includes specifications for supports of all fire protection equipment and materials as well as piping system anchors. Included are the following topics:
10	
11	PART 1 - GENERAL
12	Scope
13	Related Work
14	Reference Standards
15	Quality Assurance
16	Description
1/	
10	Submittais
20	
20	
21	Structural Supports
23	Pine Hangers and Sunnorts
24	Beam Clamos
25	Riser Clamps
26	Concrete Inserts
27	Anchors
28	Equipment Stands
29	Corrosive Atmosphere Coatings
30	
31	PART 3 - EXECUTION
32	Installation
33	Hanger and Support Spacing
34	Riser Clamps
35	Concrete Inserts and Continuous Insert Channels
30	Anchors Page Mounted Dising Supports
3/ 20	Kool Mounted Piping Supports
30 20	
40	Provisions of Division 01 shall govern work under this Section
41	
42	Division 03 - Concrete
43	
44	Section 21 05 00 – Common Work Results for Fire-Suppression
45	Section 21 10 00 – Water-Based Fire-Suppression Systems
46	
47	REFERENCE STANDARDS
48	MSS SP-58
49	MSS SP-69
50	NFPA 13 Installation of Sprinkler Systems (Latest prevailing addition).
51	NFPA 14 Installation of Standpipe and Hose Systems (Latest prevailing addition).
52	NFPA 20 Installation of Centrifugal Fire Pumps (Latest prevailing addition).
53	UL Underwriters Laboratories Listed.
54 55	εινι εαιτοι γινιατασί Αμμιονεα
55	
57	Substitution of Materials: Refer to Division 01 of the Project Manual
58	
59	DESCRIPTION
60	Provide all supporting devices as required for the installation of mechanical equipment and materials. All supports and
61	installation procedures are to conform to the latest requirements of the ANSI Code for building piping.

- 62
- Do not hang any mechanical item directly from a metal deck or run piping so its rests on the bottom chord of any truss or joist.

5

6

Fasteners depending on soft lead for holding power or requiring explosive powder actuation will not be accepted.

Support apparatus and material under all conditions of operation, variations in installed and operating weight of equipment and piping, to prevent excess stress, and allow for proper expansion and contraction.

7 DESIGN CRITERIA

8 Materials and application of pipe hangers and supports shall be in accordance with MSS Standard Practice SP-58 and SP-69 unless
 9 noted otherwise.

10

13

16

20 21

22

11 Materials and application of pipe hangers and supports shall be in accordance with NFPA rulings and be UL/FM listed and 12 approved.

14 SUBMITTALS

15 Submit data in accordance with Section 21 05 00 and Division 01 of the Project Manual.

Schedule of all hanger and support devices indicating attachment methods and type of device for each pipe size and type of service. Provide details on the working drawings submitted for approval with all pertinent information listed.

PART 2 - PRODUCTS

23 MANUFACTURERS

B-Line, Fee and Mason, Grinnell, Hilti, Michigan Hanger, Pate, PHD Manufacturing, Piping Technology, Powers/Rawl, Proset, Roof
 Products & Systems, Unistrut, or Victaulic.

26 27 STRUCTURAL SUPPORTS

Provide all supporting steel required for the installation of mechanical equipment and materials, including angles, channels, beams, etc. to suspended or floor supported tanks and equipment. All of this steel may not be specifically indicated on the drawings.

31

32 PIPE HANGERS AND SUPPORTS

33 Hangers for Pipe Sizes 1/2" through 4":

Carbon steel, adjustable swivel ring with 3/8" min. UL/FM approved hanger rods. B-Line B3170NF, Grinnell 69 or 70.

Carbon steel, adjustable clevis, standard, with UL/FM approved size hanger rods. B-Line B3100, Grinnell 260.

38 Hangers for Pipe Sizes 4" Through 8":

Carbon steel adjustable swivel ring with ½" min. UL/FM approved hanger rods. B-Line B3170NF, Grinnell 69 or 70.

40

37

41 Carbon steel, adjustable clevis, standard with UL/FM approved size hanger rods. B-Line B3100, Grinnell 260.

4243 Multiple or Trapeze Hangers:

44 Manufactured steel channel system with manufacturers slotted interlocking pipe clamps with screw/nut securing and threaded 45 hanger rods or steel channels with welded spacers and threaded hanger rods.

46

Steel channel, 12-gauge thickness, Dura-Green epoxy coating, B-Line B11. Restrain individual pipes with B-Line B2000 series or
 Vibraclamp series strut clamps.

49

50 Wall Support:

51 Carbon steel welded bracket with hanger. B-Line 3060 Series, Grinnell 190 Series.

52 Steel channels with pipe clamps.

53

54 Vertical Support:

55 Carbon steel riser clamp. B-Line B3373, Grinnell 261 for above floor use. Grinnell 40 with bolts and concrete anchors for 56 attachment to underside of concrete floor deck.

57

58 Floor Support:

- 59 Carbon steel pipe saddle, stand and bolted floor flange. B-Line B3088T/B3093.
- 60

61 **Copper Pipe Supports**:

62 All supports, fasteners, clamps, etc. directly connected to copper piping shall be copper plated or polyvinylchloride coated. Where

and avoid contact with the channel or clamp, equal to B-Line B1999 Vibra Cushion or provide manufacturers clamp and cushion 1 2 assemblies, B-Line BVT series, Grinnell PS 1400 series.

З

4 PIPE HANGER RODS

5 **Steel Hanger Rods:**

6 Threaded both ends, threaded one end, or continuous threaded, complete with adjusting and lock nuts. Steel, electro-plated, threads on both ends, B-Line B3205

- 7
- 8 9

Size rods for individual hangers and trapeze support as indicated in the following schedule:

10

Pipe Size:	Diam. Of Rod:
Up to and Including 4"	3/8" or 9.5mm min.
5",6" and 8"	½" or 12.7mm min.
10" and 12"	5/8" or 15.9mm min.

11

12 **BEAM CLAMPS**

MSS SP-69 Types 19 & 23 malleable black iron clamp for attachment to beam flange to 0.62 inches thick with a retaining ring and 13 threaded rod of 3/8, 1/2, and 5/8 inch diameter. Furnish with a hardened steel cup point set screw. B-Line B3036L/B3034, Grinnell 14 15 86/92.

16

MSS SP-69 Type 28 or Type 29 forged steel jaw type clamp with a tie rod to lock clamp in place, suitable for rod sizes to 1-1/2 17 18 inch diameter. B-Line B3054, Grinnell 228.

19 20 CONCRETE INSERTS

21 Poured in Place:

MSS SP-69 Type 18 wedge type to be constructed of a black carbon steel body with a removable malleable iron nut that accepts 22 threaded rod to 7/8 inch diameter. Wedge design to allow the insert to be held by concrete in compression to maximize the load 23 24 carrying capacity. B-Line B2505, Grinnell 281.

25

MSS SP-69 Type 18 universal type to be constructed of black malleable iron body with a removable malleable iron nut that accepts 26 27 threaded rod to 7/8 inch diameter. B-Line B3014N, Grinnell 282.

28

29 **Drilled Fasteners:**

30 Carbon steel expansion anchors, vibration resistant, with ASTM B633 zinc plating. Use drill bit of same manufacturer as anchor. 31 Hilti, Powers/Rawl, Redhead.

32

Mechanical-Expansion Anchors: 33

Insert-wedge-type anchors, for use in hardened portland cement concrete, with pull-out, tension, and shear capacities 34 35 appropriate for supported loads and building materials where used. Zinc-coated or stainless steel expansion anchors. Use drill 36 bit of same manufacturer as anchor.

37 38 Manufactured By:

39 Hilti, ITW/Read Head, or equal.

40

CONTINUOUS CONCRETE INSERT CHANNELS 41

42 Steel inserts with an industry standard pre-galvanized finish, nominally 1-5/8 inch by 1-3/8 inch deep by length to suit the application, designed to be nailed to concrete forms and provide a linear slot for attaching other support devices. Installed 43 44 channels to provide a load rating of 2000 pounds per foot in concrete. Manufacturer's standard brackets, inserts, and accessories 45 designed to be used with channel inserts may be used. Select insert length to accommodate all pipe sizes in the area.

46 ANCHORS 47

48 Use welding steel shapes, plates, and bars to secure piping to the structure.

49

EQUIPMENT SUPPORT 50

51 Support equipment plumb, rigid, and true to line. Examine Drawings, and manufacturer's data to determine how equipment and piping are to be supported, mounted, or suspended. Provide rods, bolts, inserts, pipe stands, brackets and accessories for proper 52 support. 53

54 55 **Equipment Stands:**

Use structural steel members welded to and supported by pipe supports. Clean, prime and coat with three coat rust inhibiting 56 57 alkyd paint or one coat epoxy mastic. Where exposed to weather, treat with corrosive atmosphere coatings.

58

1	CORROSIVE ATMOSPHERE COATINGS				
2 3 4	Factory coat supports and anchors used in corrosive atmospheres with hot dip galvanizing after fabrication, ASTM A123, 1.5 ounces/square foot of surface each side. Mechanical galvanize threaded products, ASTM B695 Class 50, 2.0 mil coating. Field cuts and damaged finishes to be field covered with zinc rich paint of comparable thickness to factory coating.				
5	-				
6	Corrosive atmospheres include the following locations:				
7	 Locker roc 	oms			
8					
9			DADT 2 EVECUTION		
1			PART 3 - EXECUTION		
2	INSTALLATION				
3	Size, apply and insta	ll supports and anchors in comp	liance with manufacturers reco	mmendations.	
5	Secure pipe in place	to prevent vibration, maintain p	proper slope and provide for exp	pansion and contraction.	
.7 .8 .9	Design supports of s Where support is fro and hangers to build	trength and rigidity to suit loadir om concrete construction, take ing steel framing wherever prac	ng, service, and manner which d care not to weaken concrete o tical. Do not use another pipe fo	o not unduly stress the building co or penetrate waterproofing. Fast or support. Do not use perforated	onstruction. en supports d iron, chain
0	or wire as hangers.				
2	Use inserts for susp	ending hangers from reinforced	concrete slabs wherever practi	ical. Where inserts are not practi	cal, provide
3	channels or angles fi	rom which to suspend hangers/s	supports. Fasten structural stee	el to concrete with expansion bolt	IS.
5	Provide expansion a	nchors in concrete slabs for inst	allation of threaded support roo	ds.	
7	Provide hangers cap	able of vertical adjustment afte	r piping is erected. Do not pier	ce ductwork with hanger rods. C	On threaded
8	support rods and bo	olts, weld nuts to rods, peen thr	eads, or provide double set of i	nuts with lock washers to preven	t loosening.
9	Use beam clamps fo	r attaching hangers to structura	l steel.		-
1	Coordinate hanger a	nd support installation to prope	rly group piping of all trades.		
3	Where piping can be conveniently grouped to allow the use of trapeze type supports, use standard structural shapes or				
4	continuous insert channels for the supporting steel. Where continuous insert channels are used, pipe supporting devices made				
5	specifically for use w	ith the channels may be substitu	uted for the specified supporting	g devices provided that similar typ	pes are used
6 7	and all data is submi	tted for prior approval.			
8 9	Perform welding in accordance with standards of the American Welding Society.				
0	HANGER AND SUPP	ORT SPACING			
1 2	Support horizontal piping per NFPA 13.				
3	Provide vertical supp	port at each floor level as the pi	be passes through the floor. Fo	r piping that does not pass throug	gh the floor,
4 5	provide adequate su	pport to stabilize the vertical po	ortion of the piping.		
5 7	Provide galvanized s	teel supports for steel piping.			
8 3	Provide CPVC dipped	d hangers or provide Unistrut "U	Ini-Cushion" vinyl strip at galvar	nized hangers for copper lines.	
)	Where several pipes	can be installed in parallel and	at the same elevation, provide	multiple or trapeze hangers.	
2	Support riser piping	independently of connected ho	rizontal piping.		
3 4	Adjust hangers to ob	otain the slope specified in the p	iping section of these specificat	tions.	
5	Space hangers for pi	pe as follows:	· - ·		
7	Dino Matarial	Dino Cizo:	May Haris Creation	May Vort Casaina	
	Pipe Material:	Pipe Size:	IVIAX. HORIZ. Spacing:	IVIAX. VERT. Spacing:	
	Copper	3/4" through 1"	8'-U'' 10' 0''	<u>10'-0"</u>	
	Copper	2" through 3"	10-0	10'-0"	
	Conner	3-1/2" through 8"	15'-0"	10'-0"	

1" through 1-1/4"

Steel

12'-0"

15'-0"

Steel	1-1/2" through 8"	15'-0"	15'-0"
Steel	8" through 12"	15'-0"	20'-0"

1 2

3

Unsupported length from the last hanger and an end sprinkler shall be as follows:

Pipe Size:	Length:
1" piping	Not greater than 36"
1-1/4" piping	Not greater than 48"
1-1/2" piping	Not greater than 60" or larger

4 5 **RISER CLAMPS**

6 Support vertical piping with clamps secured to the piping and resting on the building structure or secured to the building structure

7 below at each floor. Use method of securing the vertical risers to the building structure below in stairwell locations.

8

9 ANCHORS

10 Install where indicated on the drawings and details. Where not specifically indicated, install anchors at ends of principal pipe

- runs and at intermediate points in pipe runs. Make provisions for preset of anchors as required to accommodate both expansion and contraction of piping.
- 12 an
- 14
- 15

END OF SECTION

Page Intentionally Left Blank

1 2 3	SECTION 21 10 00 WATER-BASED FIRE-SUPPRESSION SYSTEMS
4	
5	PART 1 - GENERAL
6	
/ Q	SCOPE This section contains specifications for an Automatic Fire Sprinkler System for this project. Included are the following topics:
9	This section contains specifications for an Automatic fire spinicle system for this project. Included are the following topics.
10	PART 1 – GENERAL
11	Scope
12	Related Work
13	Reference Standards
14	System Description
16	Design Standards
17	Quality Assurance
18	Submittals
19	
20	PART 2 – PRODUCTS
21	Pipe
22	Fillings
23	Valves
25	Flow Switches
26	Tamper Switches
27	Sprinklers
28	Flexible Sprinkler Drop Fittings
29	Fire Department Connection
30	Local Alarm Bell
31 22	Pipe Covering System
32 33	Miscenaneous equipment
34	PART 3 – EXECUTION
35	Installation
36	General
37	Valves
38	Gauges
39	Switches
40	Sprinklers Electida Sectodas Deces Fittions
41	Flexible Sprinkler Drop Hittings
42 //3	Dry System Specialities Pre-Action System
44	Fire Department Connection
45	Pipe Covering System
46	Testing
47	
48	RELATED WORK
49	Applicable provisions of Division 01 shall govern work under this Section.
50 F1	Section 21.05.00 Common Work Decults for Fire Suppression
52	Section 21.05.00 – Common work Results for Fire-Suppression Section 21.05.29 – Hangers and Supports for Fire-Suppression Pining and Equipment
53	Section 21 05 25 Trangers and Supports for the Suppression riping and Equipment
54	REFERENCE STANDARDS
55	Applicable provisions of Division 01 shall govern work under this section.
56	
57	Local and State Codes and Regulations.
58	
59	National Fire Codes (NFC) published by NFPA; latest edition of standards listed:
6U 61	NFPA 13 - Sprinkler Systems
62	Local Fire Department requirements
63	
64	All items to be UL listed or FM approved for intended usage.

2 DESCRIPTION

1

3 4

5

6 7

8

9 10

11 12

15

20

Fire Protection Contractor shall furnish all calculations, design, drawings, material, equipment, labor and related items required to complete the work indicated on drawings and specifications.

The work under this Section includes, but is not limited to the following:

- Provide all components for a complete wet pipe automatic sprinkler system including shutoff valves with supervisory switch, fire department connection, main drain valve, test valve(s), alarms, piping, and all necessary components to make a complete, operational, and approved system.
 - Provide complete, approved automatic sprinkler system(s) to give fire suppression coverage to all areas/rooms, including electrical rooms, elevator shafts, and elevator equipment rooms.

13 This portion of the project is design build. The contractor shall follow the specifications for type of systems, materials and 14 equipment to use.

The contractor shall prepare, seal and submit drawings and calculations as required to obtain approval and building permit from
 State, Insurance Company, and local authority. Submit drawings and calculations to all authorities as required.

19 These documents, along with local regulations and codes, will be the basis for the Fire Protection design and construction.

The contractor shall calculate, size and select all systems as defined by the documents. This shall include coordination with other trade contractors including wiring of flow switch(es) and supervisory switch(es). All calculations, sizes, and system layouts shall include provisions for future additions.

25 SYSTEM DESCRIPTION

26 Connect to the existing fire protection system and extend into the new addition. Provide a wet pipe automatic sprinkler riser, 27 cross main, and branch piping to connect to sprinkler heads in all spaces of the addition. Provide a new fire department 28 connection (coordinate location with local Fire Marshall). See Fire Protection Drawings for location of main, riser, and Fire 29 Department Connection.

30

Provide pre-fabricated modular pipe covering system to conceal pipes which would otherwise be exposed along cell fronts, in dormitory, and other areas where exposed piping is required to be concealed for security.

33 34 DESIGN STANDARDS

Sprinkler system shall be designed and hydraulically calculated by the Contractor to provide densities as listed on the drawings.
 Hydraulically calculate the system based on Light Hazard Occupancy in general areas.

38 Available water supply data for system design is as follows:

37 38 39

Performed By: City of Madison

40 41

Water Supply	Outlet	Flow	Static	Residual
Hydrant Location	Elevation	<u>(GPM)</u>	<u>(PSI)</u>	<u>(PSI)</u>
Northport Drive	867.00 ft.	1,500	90	86

42

Water test data is preliminary for bidding purposes. Contractor shall perform a field flow and pressure test on municipal water supply main to verify existing conditions, as well as conditions of any new municipal main installation, in the adjacent street, and obtain any additional test data required for design. Tests to be representative of high water use periods.

46

47 Contractor shall submit seven (7) copies of hydraulic calculations with shop drawings on standard form specified in NFPA No. 13,
 48 Chapter 7, Sections 7-2 through 7-3.5 inclusive and Figures A-7-3.3 and A-7-3.4.

50 QUALITY ASSURANCE

51 Substitution of Materials: Refer to Section 21 05 00 and Division 01 of the Project Manual.

52

61

49

53 Fire protection system components shall be rated for a minimum operating pressure of 175 psig. 54

To assure uniformity and compatibility of piping components in grooved piping systems, all grooved products utilized shall be supplied by a single manufacturer. Grooving tools shall be supplied from the same manufacturer as the grooved components.

57 58 SUBMITTALS

59 Shop Drawings:

60 Submit shop drawings of all fire sprinkler system components.

1 Plans:

3

8

2 Submit contractor-prepared plans/drawings.

Submit per NFPA 13; installation plans, working plans, shop drawings, hydraulic calculations, and manufacturer's data on devices,
 etc., indicating by model and number to be used for review and approval. Contractor shall obtain the necessary insurance
 underwriters, State and Local Fire Department approvals prior to submitting shop drawings. Include copy of approval letter in
 submission to Architect/Engineer.

Prepare drawings at minimum scale of 1/8" per foot for plans and 1/4" per foot or larger for details. Show all piping, lighting,
 equipment, ductwork, sprinklers, hangers, roof construction and occupancy of each area, including ceiling and roof heights.

12 Installation shall be coordinated with the latest architectural, structural, mechanical, plumbing and electrical drawings.

Contractor shall submit drawings to Engineer which have been reviewed and stamped "approved" by the authority having
 jurisdiction. No work shall commence until all approvals have been obtained. Allow sufficient time in the construction schedule
 for the approvals.

18 As-Built Drawings:

Maintain at the site an up-to-date marked set of as-built drawings which shall be corrected and delivered to the Architect upon completion of the work.

22 Furnish the Architect one (1) reproducible print of corrected shop drawings, including plans, revised to show "as built" conditions.

23 24 25

30

32

34 35

21

PART 2 - PRODUCTS

26 27 **PIPE**

28 Wet Systems:

29 Carbon steel pipe, black, thickness per NFPA 13, conforming to ASTM A53, A135, A795.

31 Sprinkler piping shall be schedule 40 threaded up to and including 2" in size.

33 Schedule 10 threaded light wall not allowed (2" and under).

FITTINGS

36 Malleable iron, Class 150, threaded, ANSI B16.3. 37

38 Ductile iron, grooved end, 300 lb/in2 working pressure rating, UL listed or FM approved for automatic sprinkler.

39

44

50

Ductile or malleable iron, plain end with EPDM gasket, carbon steel bolts or locking lugs UL listed or FM approved for automatic
 sprinkler, Grinnell "Sock-it".

- 41 sprinkler, Grinnen 5
- 43 Carbon steel, butt-welded, class 150, ASTM A234.
- 45 Carbon steel, Class 150, flanged, ASTM A105.

46 47 **JOINTS**

48 Iron Pipe:

49 Tapered pipe threads, with Teflon tape, ANSI B2.1.

- 51 Mechanical coupling, EPDM gasket, UL listed or FM approved for automatic sprinkler.
- 52
- 53 Rigid Type:

Housings shall be cast with offsetting, angle-pattern bolt pads to provide system rigidity and support and hanging in accordance with NFPA 13. Tongue and recess rigid type couplings shall only be permitted if the contractor uses a torque wrench for installation. Required torque shall be in accordance with the manufacturer's latest recommendations. Victaulic FireLock® EZ Style 009H (2" thru 4") and Victaulic Style 107H QuickVicTM (2" thru 8") shall be installation ready stab-on design, for direct 'stab' installation onto grooved end pipe without prior field disassembly and no loose parts. 10" and larger sizes shall be Victaulic Style 07 Zero-Flex standard rigid coupling.

- 60
- 61 Flexible Type:

62 Use in seismic areas and where required by NFPA 13. Victaulic Style 177 QuickVicTM (2" thru 8") shall be installation ready stab-

63 on design, for direct 'stab' installation onto grooved end pipe without prior field disassembly and no loose parts. 10" and larger 64 sizes shall be Victaulic Style 75 or 77 standard flexible coupling.

1

2 VALVES

3 Manufacturers:

4 Grinnell, Nibco, TYCO, Victaulic, or Wilkins.

5

6 Shutoff Valve:

7 Butterfly Valve:

8 Ductile iron body, epoxy coated, EPDM encapsulated ductile iron disc, 300 psi maximum working pressure, indicating type, with 9 tamper switch in actuator, grooved end connections, UL Listed or FM approved, Victaulic Figure 705-W.

10

11 Check Valve:

12 Ductile iron body, rubber-encapsulated disc, 250 psi maximum working pressure, grooved end connections. Victaulic style 717.

13

14 Test Drain Valve:

15 Ball valve type, bronze, combination test and drain, with site glass, Sure-Test by G/J Innovations.

16

19

17 If design flow cannot be reached through the inspector's test drain, then the FPC shall install forward flow by-pass around the 18 fire department connection check valve.

20 FLOW SWITCHES

21 UL listed and FM approved vane type waterflow switch with metal enclosure, adjustable pneumatic retard and electrical 22 characteristics compatible with alarm system. Equal to Potter Model VSR-F.

23 24 TAMPER SWITCHES

For O S & Y valve or post indicator installations, UL listed, FM approved, to monitor position of valve, tamper resistant cover screws, single or double SPDT switch contacts, corrosion resistant, for indoor or outdoor use, NEMA 4 & 6P enclosures. Equal to Potter Model PCVS-1, -2 and OSYSU-1, 2.

28 29 SPRINKLERS

30 Manufacturer:

Products of the following manufacturers determined to be equal by the Architect/Engineer will be accepted: Grinnell, Reliable,
 TYCO, Victaulic and Viking.

33

34 General:

Fusible link or glass bulb type, cast brass or bronze construction. Provide heads with nominal 1/2" discharge orifice except where greater than normal density requires large orifice.

37

Select fusible link or glass bulb temperature rating to not exceed maximum ambient temperature rating allowed under normal conditions at installed location. Provide ordinary temperature (165 degree) fusible link or glass bulb type except at skylights, sealed display windows, unventilated attics and roof spaces, over cooking equipment, adjacent to diffusers, unit heaters, uninsulated heating pipes or ducts, mechanical rooms, storage rooms, or where otherwise indicated.

42

Provide quantity of spare heads as noted below and 1 wrench for each type of head and each temperature range installed. Provide
 6 spare heads per 300 or less installed heads, 12 per 1000 or less and 24 for more than 1000. Provide steel cabinet for storage of
 heads and wrenches.

47 Types:

Refer to Sprinkler Schedule on plans for sprinkler head types and finishes in each area. Provide sprinkler guards in areas where
 sprinklers may be subject to damage (i.e. mechanical rooms).

- 50
- 51 Finished Areas:

52 Chrome plated bronze body quick response pendent, concealed, or side-wall sprinklers with glass bulb heat sensor. Semi-recessed 53 and sidewall sprinklers shall have adjustable recessed escutcheon. Concealed sprinklers shall have adjustable cover plates. Cover 54 plates shall match ceiling color. Design Basis: Victaulic Model V27.

55

63

56 Unfinished Areas:

Plain bronze body, upright or pendent, quick response sprinklers, with solder link or glass bulb for wet system. Plain bronze,
 upright or pendent open sprinkler for dry system. Design Basis: Victaulic Model V27 or V36.

59 60 Ratings:

51 See sprinkler ratings indicated on Sprinkler Schedule on plans. Use higher temperature-rated sprinkler heads in areas near heat 52 sources, elevator equipment rooms, and elevator shafts.

64 FLEXIBLE SPRINKLER DROP FITTINGS

Manufacturers: FlexHead Industries, Victaulic or Viking.

1

2

Corrugated Type 304 stainless steel hose with braided Type 304 stainless steel exterior cover, welded stainless steel or zinc plated 3 4 steel inlet and outlet threaded fittings with EPDM seals. 175 PSI pressure rating. 225 oF temperature rating, 1" minimum internal 5 hose diameter. 40" maximum hose length, straight or angle outlet configuration. Galvanized steel ceiling support bar and brackets selected to match project ceiling support system requirements. UL Listed and FM approved. 6 7 8 Flexible drops are only allowed for use above fully accessible ACT ceilings. 9 10 FIRE DEPARTMENT CONNECTION Manufacturer: 11 12 Badger-Powhatan, Croker, Elkhart Brass, J.W. Moon, Potter-Roemer, and W.D. Allen. 13 14 Exposed: Wall mounted, exposed, two-way inlet, cast brass construction with pin lug swivel plugs, labeled backplate, with drop clappers. 15 16 threads to match local fire department and finish to match existing fire department connection color. Potter-Roemer 5751. 17 18 Flush: 19 Wall mounted, flush, two-way inlet, cast brass construction with pin lug swivel plugs, labeled backplate, with drop clappers, 20 threads to match local fire department and finish to match existing fire department connection color to match building color 21 scheme. Potter-Roemer 5021. 22 23 Storz: 24 Wall mounted, 4 inch storz inlet, cast brass construction, labeled backplate, with threads to match local fire department and hard 25 coated aluminum finish. Potter-Roemer 5795/5799 Series. 26 27 General: 28 Mounting height, distance from curb, etc., shall be as specified by local fire department. 29 30 LOCAL ALARM BELL 31 UL listed and FM approved weatherproof electric alarm bell with red painted metal housing, mounting base and gong, solenoid operator, weatherproofing O-ring seal and electrical characteristics compatible with alarm system. Equal to Potter Model PBA. 32 33 34 **Support Devices:** 35 All pipe support devices shall be tested and listed by an NFPA recognized independent laboratory for suitability with the intended 36 application. 37 38 UL Listed low profile steel hanger for securement of steel and listed CPVC sprinkler pipe. 39 40 UL Listed copper coated hanger for securement of copper tubing. 41 42 Zinc plated spring steel shield clips of the size recommended by manufacturer, for securement of the cover. 43 44 **Piping Cover:** The piping cover shall be 18-gauge cold rolled steel with paintgrip galvanized finish (treated for painting), ASTM A527, G90-U lock 45 46 forming. The piping cover shall have a snap-lock interfacing with the shield clips such that it is irremovable with the use of ordinary 47 tools. 48 49 Provide 5'-0" section shield covers with L-Design or U-Design shapes for sidewall or pendent installations respectively. 50 51 Cover joints shall be overlapping/interlocking integral joints with provision for securement utilizing steel rivets. Rivet spacing 52 shall be no greater than 1-1/4" intervals along the joint and positioned at a distance no greater than 1/2" from the end of the 53 overlapping section. 54 55 Cover design shall include a groove at the interfacing of the cover and the construction surface to facilitate the application of sealant/adhesive compounds and enhance the security of such compounds from dislodging. 56 57 58 Accessories: 59 The system shall include tamper-resistant end caps, pre-fabricated corners, wall flanges, couplings, and other items necessary to 60 complete the system. 61 62 The system shall include a spacer ring at the point of each sprinkler outlet to preclude intentional gapping between the sprinkler 63 escutcheon and the cover. 64

1 MISCELLANEOUS EQUIPMENT

Provide other equipment and accessories, not listed, but required for a complete sprinkler system in accordance with NFPA and
 FM requirements.

4 5 6

7

PART 3 - EXECUTION

8 INSTALLATION

9 Install sprinkler system in accordance with requirements of NFPA 13 and local regulations of the fire marshal. 10

Grooved joint piping systems shall be installed in accordance with the manufacturer's guidelines and recommendations. The gasket style and elastomeric material (grade) shall be verified as suitable for the intended service as specified. Gaskets shall be molded and produced by Victaulic. Grooved end shall be clean and free from indentations, projections, and roll marks in the area from pipe end to groove for proper gasket sealing. A Victaulic factory-trained field representative shall provide on-site training for contractor's field personnel in the proper use of grooving tools and installation of grooved piping products. Factory-trained representative shall periodically review the product installation. Contractor shall remove and replace any improperly installed products.

18

19 The sprinkler bulb protector must remain in place until the sprinkler is completely installed and before the system is placed in 20 service. Remove bulb protectors carefully by hand after installation. Do not use any tools to remove bulb protectors.

21 22 **GENERAL**

Install all piping parallel to building walls and ceilings and at heights which do not obstruct any portion of window, doorway, stairway or passageway. Where interferences develop in the field, offset or reroute piping as required to clear such interferences. Coordinate locations of fire protection piping with piping, ductwork, conduit and equipment of other trades to allow sufficient clearances. In all cases, consult drawings for exact location of pipe spaces, ceiling heights, ceiling grid layout, light fixtures and grilles before installing piping. All exposed overhead piping shall be installed above the bottom chord of roof joists.

28

30

29 Maintain piping in clean condition internally during construction.

31 Provide clearance for access to valves and piping specialties.

32

Install piping so that system can be drained. Where possible, slope to main drain valve. Piping may be installed level (WET SYSTEMS ONLY). Where piping cannot be fully drained, install nipple and cap for drainage of less than 5 gallons or valve/nipple/cap for drainage over 5 gallons.

37 Do not install piping within exterior walls.

38

46

36

Do not route piping above transformers, panelboards, or switchboards, including the required service space for this equipment,
 unless the piping is serving this equipment.

41 42 **VALVES**

Properly align piping before installation of valves. Do not support weight of piping system on valve ends. Mount valves in locations which allow access for operation, servicing and replacement. Install all valves with the stem in the upright or horizontal position. Valves installed with the stems down will not be accepted. All system shut-off valves shall have a supervisory switch.

47 GAUGES

48 Provide a valved pressure gauge in main sprinkler risers.

49 50 **SWITCHES**

51 Provide valved test connection for flow switch adjacent to flow switch. Test flow switch to verify proper operation.

53 SPRINKLERS

54 Locate sprinklers maintaining clearances from obstructions, ceilings and walls. Install sprinklers level in locations not subject to 55 spray pattern interference.

56

52

57 Sprinklers shall be centered in all ceiling panels and tiles. A 1" tolerance for sprinkler placement is acceptable. 58

59 FLEXIBLE SPRINKLER DROP FITTINGS

60 Install in accordance with manufacturer's installation instructions following minimum bend radii, maximum number of bends and

61 bend distance from end requirements.

62

1 FIRE DEPARTMENT CONNECTION

Support from structure independent of piping. Locate 24" above grade or as directed by local Fire Marshall. Fill wall penetration
 with insulation and caulk exterior and interior face of wall opening weather tight.

5 PIPE COVERING SYSTEM

Indicate on sprinkler design drawings locations where covering system will be used. Installation shall be per the manufacturer's
 guidelines.

9 The cover shall be linear and snug-fitting. Support devices shall be anchored squarely and firmly against the structure in a straight 10 line.

11

4

8

12 The installation shall be free of voids between the interfacing of the cover and the construction surface. Voids shall be sealed 13 with "Sikadur" 51 two part epoxy, "Sikaflux" TR two part epoxy. Manufacturer shall supply on-site installation instruction for the 14 project start-up.

15

16 TESTING

- 17 Refer to Section 21 05 00 Common Work Results for Fire Suppression.
- Hydro-statically pressure test the fire sprinkler system piping as required in NFPA 13. Keep records of all testing for submission
 in Operation and Maintenance Manuals.
- 21
- 22
- 23

END OF SECTION

Page Intentionally Left Blank

1	SECTION 22 05 00
2	COMMON WORK RESULTS FOR PLUMBING
3	
4	
5	PART 1 - GENERAL
6	
7	SCOPE
8	This section includes information common to two or more technical plumbing specification sections or items that are of a
9	general nature, not conveniently fitting into other technical sections. Included are the following topics:
10	
11	PART 1 – GENERAL
12	Scope
13	Related Work
14	Regulatory Requirements
15	Reference Standards
16	LEED Certification
17	Quality Assurance
18	Abbreviations and Symbols
19	Definitions
20	Coordination
21	Electronic Drawings
22	Continuity of Existing Services
23	Protection of Finished Surfaces
24	Sealing and Firestopping
25	Equipment Furnished by Others
26	Provisions for Future
27	
20	Sublituidis
29	Specified Materials and Equipment
21	Characterization and Maintenance Manuals
32	Operating and Maintenance Manuals Record Drawings
32	Training of Owner Personnel
32	Testing
35	Cleaning
36	Warranty
37	Certified Startun Reports
38	
39	PART 2 - PRODUCTS
40	Electrical Requirements
41	Access Panels and Doors
42	Pipe Penetrations
43	Equipment, Piping, and Valve Identification
44	Equipment Accessories
45	Thermometers and Gauges
46	Bedding and Backfill
47	-
48	PART 3 - EXECUTION
49	General
50	Asbestos Abatement
51	Demolition
52	Excavation and Backfill
53	Dewatering
54	Rock Excavation
55	Surface Restoration
56	Concrete Work
57	Openings, Cutting and Patching
58	Building Access
59	Equipment Access
60	Coordination of Work
6J	Piping Installation
62 62	Lubrication and Maintenance
03 64	Dieteves Dies Departmetions
04	ripe relieu autoris

1	Escut	cheon Plates
2	Flashi	ng of Roof and Wall Penetrations
3	Painti	ng Generation and the second s
4	Identi	fication
5		
0	Applicable provisi	one of Division 01 govern work under this Costion
/	Applicable provisi	ons of Division 01 govern work under this section.
8 9	This section applie	es to all Division 22 sections of plumbing.
10		
11	REGULATORY REC	QUIREMENTS
12	Codes and Standa	irds:
13	All plumbing wor	k shall conform to the requirements of Wisconsin Administrative Code SPS 382 and SPS 384, Wisconsin
14	Uniform Plumbing	g Code.
15		
16	All materials and	workmanship shall comply with applicable Codes, local ordinances, industry standards and utility regulations.
1/	In case of differen	ices between such Codes, and the Contract Documents, the most stringent shall govern. Promptly notify the
18	A/E in writing of a	ny such difference.
20	Non Compliance:	
20	Should the Contra	$r_{\rm constraint}$
22	he shall hear all co	bets performing work that does not comply with the above requirements, without having notified the A/E,
23	ne shan bear an ee	
24	Permits. Inspectio	ons and Fees:
25	All required, perm	its, and inspections shall be requested and obtained by the Contractor.
26	1 /1	
27	All fees and charge	es for approvals, reviews, or other inspections shall be paid by the Contractor.
28	-	
29	All fees and charg	ges assessed by local utilities for water, sewer, gas or other services shall be included in the bid and shall be
30	paid by the Contra	actor(s).
31		
32	REFERENCE STAN	DARDS
33	Standards cited in	the Specifications shall be the most recent editions.
34		the deader and the first of the debies and attended to the second state of the second state of the second state
35	Abbreviations of s	tandards organizations referenced in this and other sections are as follows:
30	ABIMA	American Boller Manufacturers Association
3/ 20		American Concrete Pipe Association
30 20		Air Novement and Control Association
39 40		American National Standards Institute
40		Air Conditioning and Befrigeration Institute
41		American Society of Mechanical Engineers
42	ASIVIL	American society of Plumbing Engineers
43	ASSE	American Society of Sanitary Engineering
45	ASTM	American Society for Testing and Materials
46	AWWA	American Water Works Association
47	AWS	American Welding Society
48	CISPI	Cast Iron Soil Pipe Institute
49	CGA	Compressed Gas Association
50	CS	Commercial Standards, Products Standards Sections, Office of Eng. Standards Service, NBS
51	EPA	Environmental Protection Agency
52	FS	Federal Specifications, Superintendent of Documents, U.S. Government Printing Office
53	GAMA	Gas Appliance Manufacturers Association
54	IAPMO	International Association of Plumbing & Mechanical Officials
55	IEEE	Institute of Electrical and Electronics Engineers
56	ISA	Instrument Society of America
5/		Niechanical Contractors Association
20 50		Windwest Insulation Contractors Association
22		National Rureau of Standards
61	NEC	National Electric Code
62	NEMA	National Electrical Manufacturers Association
63	NFPA	National Fire Protection Association

64 NSF National Sanitation Foundation

- 1 PDI Plumbing and Drainage Institute
- 2 SMACNA Sheet Metal and Air Conditioning Contractors' National Association. Inc.
- 3 STI Steel Tank Institute
- 4 UL Underwriters Laboratories Inc. 5
- 6 Standards referenced in this section:
- 7 ACI 614 Recommended Practice for Measuring, Mixing and Placing of Concrete
- 8 ASTM D1557 Standard Test Method for Moisture-Density Relations of Soils
- 9 ASTM E814 Standard Test Method for Fire Tests of Through-Penetration Fire Stops
- 10 ASTM E84 Standard Test Method for Surface Burning Characteristics of Building Materials
- 11 D.O.T. Standard Specifications for Road and Bridge Construction, State of Wisconsin, Dept. of Transportation
- 12 UL1479 Fire Tests of Through-Penetration Firestops
- 13 UL723 Surface Burning Characteristics of Building Materials

15 LEED CERTIFICATION

The project will be LEED Certified thru the United States Green Building Council's (USGBC) Leadership in Energy and Environmental Design (LEED) program. Refer to Section 01 81 13 – Sustainable Design Requirements for additional requirements.

19

14

In addition to complying with Division 22 drawings and specifications, equipment and material shall also comply with Section 01
 81 13 and LEED requirements.

The Division 22 contractor will be expected to provide all required documentation, submittals, etc. in accordance with prerequisites and credits associated with Division 22 work and LEED Certification.

26 QUALITY ASSURANCE

27 Substitution of Materials: Refer to Division 01 of the Project Manual.28

All products and materials used are to be new, undamaged, clean and in good condition. Existing products and materials are not to be reused unless specifically indicated.

31

25

Where equipment or accessories are used which differ in arrangement, configuration, dimensions, ratings, or engineering parameters from those indicated on the contract documents, the contractor is responsible for all costs involved in integrating the equipment or accessories into the system and for obtaining the intended performance from the system into which these items are placed.

37 ABBREVIATIONS AND SYMBOLS

38 Key to abbreviations and symbols shall be on the Drawings.

39

41

42

43

44

45

50

36

40 The following are additional abbreviations used in the Specifications:

- A/E Architect/Engineer
 - GC General Contractor
 - PC Plumbing Contractor
- FPC Fire Protection Contractor
- HC Heating Ventilating and Air Conditioning Contractor
- 46 EC Electrical Contractor

47 48 **DEFINITIONS**

49 Furnish:

Supply and deliver to Project site ready for unpacking, assembly and installation.

51 52 Install:

- 53 Operations at Site including unpacking, assembling, erecting, placing, anchoring, applying, finishing, cleaning, and connecting 54 related devices required for product fully functional for intended use after installation.
- 55
- 56 Provide:
- 57 Furnish and install, such that product is fully functional for intended use. 58

59 COORDINATION

- 60 The Drawings show the general arrangement of piping and equipment and shall be followed as closely as actual building
- 61 construction and the work of other trades permits. Architectural and Structural Drawings shall take precedence. Because of the
- 62 scale of the Drawings, it is not possible to indicate all offsets, fittings, and accessories which may be required. Investigate conditions
- 63 affecting the Work and arrange accordingly, providing offsets, fittings and accessories as may be required to meet conditions.

64

1 ELECTRONIC DRAWINGS

2 Drawings show general arrangements of piping, equipment and appurtenances and shall be followed as closely as actual 3 building construction and work of other trades permits. Work shall conform to requirements shown on Drawings. General and 4 structural drawings shall take precedence. Because of the scale of Drawings, it is not possible to indicate all offsets, fittings and 5 accessories required. Investigate structural and finish conditions affecting work and arrange work accordingly, providing offsets, 6 fittings and accessories required to meet constructed conditions.

8 Plumbing equipment and systems, including piping and ductwork shall be installed as high as possible unless otherwise noted 9 on Drawings. Equipment and systems shall also be installed to maintain required operation and maintenance clearances.

10 11 CONTINUITY OF EXISTING SERVICES

- 12 Refer to Division 01 of the Project Manual.
- 13

7

Do not interrupt or change existing services without prior approval from Owner, Architect, Engineer or Construction Manager. When interruption is required, coordinate down-time with Owner to reduce disruption to activities. Scope of Work is indicated on Contract Documents or described herein. Unless specifically stated, any work involved in interrupting or changing existing services is to be done during normal working hours.

18 19 PROTECTION OF FINISHED SURFACES

20 Refer to Division 01 of the Project Manual.

21

Furnish one can of touch-up paint for each different color factory finish to be finished surface of product. Deliver touch-up paint
 with other "loose and detachable parts" as covered in General Requirements.

25 SEALING AND FIRESTOPPING

Sealing and firestopping of sleeves/openings between piping, etc. and the sleeve or structural opening shall be the responsibility of the contractor whose work penetrates the opening. The contractor responsible shall hire individuals skilled in such work to do the sealing and fireproofing. These individuals hired shall normally and routinely be employed in the sealing and fireproofing occupation.

31 EQUIPMENT FURNISHED BY OTHERS

32 None.

30

3334 PROVISIONS FOR FUTURE

35 None.

36

37 OFF SITE STORAGE

38 Refer to Division 01 of the Project Manual.

3940 SUBMITTALS

41 Refer to Division 01, of the Project Manual.

43 Submit shop drawings with space for approval stamps of GC and A/E.

44

42

Submit the following plumbing system data sheet for approval by the GC and A/E. List piping material type for each piping service on the project, ASTM number, schedule or pressure class, joint type, manufacturer and model number where appropriate. List valves and specialties for each piping service, fixture and equipment with manufacturer and model number.

- 48
- 49 PLUMBING SYSTEM DATA SHEET

50	ltem	Pipe Service/Sizes	Manufacturer/Model No.	Remarks
51	Pipe			
52	Fittings			
53	Unions			
54	Valves:			
55		Ball		
56		Butterfly		
57		Balancing		
58		Check		
59		Other		
60	Pipe Spe	ecialties:		
61		Building Penetrations		
62	Hangers	& Supports		
63	Insulatio	on		

- Plbg. Specialties:
 Floor/Roof Draw
- 2 Floor/Roof Drains3 Cleanouts
 - Cleanouts Water Hammer Arrestors
- 5 Wall/Yard Hydrants

Urinals

- 6 Hose Bibbs 7 Plbg. Fixtures:
- 7 Plbg. Fixtures: 8 Lavatory
- 9 Faucet
- 10 Flush Valves
- 11 Stop/Supplies
- 12 Showers
- 13 Waste/Trap 14 Water Closets
- 14
- 16

4

17 Submit manufacturer's color charts where finish color is specified to be selected by Architect/Engineer.

Shop drawing submittals are to be bound, labeled, contain the project manual cover page and a material index list page showing item designation, manufacturer and additional items supplied with the installation. Submit for all equipment and systems as indicated in the respective specification sections, marking each submittal with that specification section number. Mark general catalog sheets and drawings to indicate specific items being submitted and proper identification of equipment by name and/or number, as indicated in the contract documents. Include wiring diagrams of electrically powered equipment.

25 Firestop Systems:

26 Contractor shall submit product data for each firestop system. Submittals shall include product characteristics, performance 27 and limitation criteria, test data, MSDS sheets, installation details and procedures for each method of installation applicable to 28 this project. For non-standard conditions where no UL tested system exists, submit manufacturer's drawings for UL system 29 with known performance for which an engineering judgement can be based upon.

30 31 SPECIFIED MATERIALS AND EQUIPMENT

Design is based on equipment specified by manufacturer and model number as specified on Drawing Schedules. Where certain items are specified by manufacturer or trade name, Contractor's bid shall be based on use of named item. Where one (1) make is described and other makes are listed, comparable models of other named equipment may also be used, provided they meet requirements of Specifications.

36

When equipment or accessories used differ in arrangement, configuration, dimensions, ratings, or engineering parameters from
 those on Drawing schedules, Contractor shall be responsible for costs involved in integrating equipment or accessories into system.
 Contractor shall be responsible for obtaining original design performance from system into which items are placed, regardless
 of whether manufacturer/model is specified equivalent or substitute.

41

If Contractor wishes to use items other than those named in Specifications in base bid, request for approval of substitution must be made in writing to A/E at least 14 days prior to opening of bids. Include complete technical and descriptive data with request. If approved, an Addendum will be issued notifying bidders of approval. Request for approval will be considered only if requested by prime bidding Contractor.

46 47 EQUIPMENT INSTALLATION

Drawings show general arrangement and location of equipment and appurtenances. It is Contractor's responsibility to install equipment in a location and manner that allows for proper service and maintenance access to equipment. Work shall generally conform to requirements shown on Drawings. However, location of equipment may require field adjustments to obtain required service space. DO NOT SCALE OFF PLANS to determine proper location of equipment. Because of scale of Drawings, it is not possible to indicate exact routing of piping, and offsets, fittings and accessories required to provide proper service access to equipment. Contractor shall route and install ductwork and piping to provide required service access to equipment.

54

If, during construction phase of Project, contractor feels inadequate space exists, or equipment locations must be substantially modified to provide proper service and maintenance access, prior to installing equipment, contractor shall notify engineer in writing, outlining general concerns and proposed modifications. Equipment installed without providing manufacturer's required maintenance and service clearance shall be considered defective. Contractor shall remove and relocate piping, ductwork and equipment, to provide required service clearances at contractor's expense.

60

61 **OPERATING AND MAINTENANCE INSTRUCTIONS**

- 62 Refer to Division 01 of the Project Manual.
- 63

1 2 3	Assemble material in three-ring or post binders, using an index at the front of each volume and tabs for each system or type of equipment. In addition to the data indicated in the General Requirements, include the following information: • Copies of all approved shop drawings.
4	 Manufacturer's wiring diagrams for electrically powered equipment
5	 Records of tests performed to certify compliance with system requirements
6	Certificates of inspection by regulatory agencies
7	 Parts lists for fixtures, equipment, valves and specialties.
8	 Manufacturer's installation, operation and maintenance recommendations for fixtures, equipment, valves and specialties
9 10	 Valve schedules
11	 Lubrication instructions, including list/frequency of lubrication
12	Warranties
13	 Additional information as indicated in the technical specification sections
14 15	
15 16	Refer to Division 01 of the Project Manual.
17	Maintain Depart Drawings on daily basis to be turned over at completion of Draiget
18 19	Maintain Record Drawings on daily basis to be turned over at completion of Project.
20	TRAINING OF OWNER PERSONNEL
21	Instruct Owner's personnel in proper operation and maintenance of systems and equipment provided as part of Project, using
22 22	Uperating and Maintenance manuals during instruction. Demonstrate startup and shutdown procedures for equipment.
23 24	
25	TESTING
26	Provide materials, labor, and equipment required for testing.
27	Notify Inspectar(a) and day prior to the time when the test is ready to be performed
28 29	Notify inspector(s) one day prior to the time when the test is ready to be performed.
30	After testing, submit in writing the time, date, name and title of the person approving the test. This shall also include the
31	description and what portion of the system has been tested. The person approving the test shall sign the submittal.
32	
33 34	Records shall be maintained of testing that has been completed, and shall be made available at the job site.
35 36	Upon completion of the work, records and certifications approving testing requirements shall be submitted.
37 38	Defective work or material shall be replaced or repaired, and the test repeated. Repairs shall be made with new materials.
39	CLEANING
40 41	Keep the premises broom clean and free of surplus materials, rubbish and debris.
41 42 42	After fixtures and equipment have been installed, remove stickers, rust stains, labels, and temporary covers.
43 44	Foreign matter shall be blown out, or flushed out, of pipes, tanks, pumps, strainers, motors, devices, switches, fixtures, and
45	panels.
46	
47 19	Identification plates on equipment shall be free of paint and dirt.
40 49	Leave the work in a condition ready for operation.
50	
51	WARRANTY
52 52	Warrant that work shall function for one year immediately following acceptance of the system(s).
53 54	Keen the system in good working order at no expense junless defects are clearly the result of improper or abnormal usage
55	Reep the system in good working order actio expense, ameas derects are clearly the result or improper or abhormal usage.
56	Submit for acceptance of the work, written certification that the entire system has been installed and adjusted for operation in
57	accordance with the Contract Documents.
58 59	PART 2 = PRODUCTS
60	
61	ELECTRICAL REQUIREMENTS
62	General:
63 64	Work shall conform to requirements of Division 26.
. .	
1 Power wiring shall be provided by the EC. Control wiring shall be provided by the PC. Plumbing Contractor shall provide wiring 2 diagrams for use by the Electrical Contractor.

4 ACCESS PANELS AND DOORS

5 Provide access panels at locations requiring access to mechanical equipment. Locations include, but are not limited to areas above drywall ceilings, shaft enclosures and other furred-in spaces concealing valves, ducts or equipment. Provide UL listed, fire 6 7 rated access panels when penetrating fire rated chase or shaft areas.

8

3

9 Access panels shall be of size required to provide adequate access to equipment. Minimum size shall be 12 inch by 12 inch for 10 hand access and 24 inch by 24 inch for body access.

- 11 12
- 13

Panels shall be Milcor brand or equivalent.

14 Panels shall include concealed hinges, cam type locking devices, and have frame/border type necessary for particular wall or ceiling construction they are installed. Access panels shall be flush mounted, recessed frame type units. Access panels shall be 15 16 prime coated steel, able to accept field painting for general applications and stainless steel for use in toilet rooms, shower 17 rooms and similar wet areas.

18

20

19 Refer to Architectural Room Finish Schedule for wall and ceiling surfaces and finishes.

21 For non-security applications, panel construction shall utilize 16 gauge frame with not less than 18 gauge hinged door panel. 22 Door locks shall be screwdriver operated for panels in general location applications and shall be key locked for public area 23 applications. 24

25 PIPE PENETRATIONS

26 Refer to Division 01 requirements as well as the following.

27 28 Fire, Smoke And Fire/Smoke Rated Surfaces:

3M CP 25N/S or CP 25S/L caulk, 3M FS 195 wrap/strip with restricting collar, 3M CS 195 composite sheet, Pipe Shields Inc. 29 30 Series F fire barrier kits, Proset Systems fire rated floor and wall penetrations, Insta-Foam Products Insta-Fire Seal Firestop 31 Foam or Dow Corning Fire Stop System.

33 All fire stopping systems shall be provided by the same manufacturer. 34

35 UL listed or tested by independent testing laboratory, approved by State and Local Code jurisdictions.

37 Use product that has a rating not less than rating of wall or floor being penetrated. Reference architectural drawings for 38 identification of fire and/or smoke rated walls and floors.

39

32

36

40 Sleeves in concrete to be Schedule 40 steel pipe with integral water stop unless fire stop material used includes a sleeve that is 41 an integral part of rated assembly.

42

43 Use firestop putty, caulk sealant, intumescent wrapstrips, intumescent firestop collars, firestop blocks, firestop mortar or a 44 combination of these products to provide a UL listed system for each application required for this project. Provide mineral wool backing where specified in manufacturer's application detail. 45 46

47 Non-Rated Surfaces:

Stamped steel, chrome plated, hinged, split ring escutcheons or floor/ceiling plates for covering openings in occupied spaces. 48

49

50 In exterior wall openings below grade, use modular mechanical type seal consisting of interlocking synthetic rubber links 51 shaped to continuously fill the annular space between the un-insulated pipe and cored opening or a water-stop type wall 52 sleeve.

53

At interior partitions where pipe penetrations are sealed, use Tremco Dymonic, Sika Corp. Sikaflex 1a, Sonneborn Sonolastic 54 55 NPI, or Mameco Vulken 116 urethane caulk to effect seal. Use galvanized sheet metal sleeves in hollow wall penetrations. 56

57 EQUIPMENT, PIPING AND VALVE IDENTIFICATION

58 Equipment Labels:

59 After painting and covering, identify equipment, including pumps, tanks, compressors, and control panels. Locate identification 60 conspicuously.

- 61
- 62 Identification of equipment shall be by engraved white letters on a black 1/16 inch thick plastic laminate panel, beveled edges,
- 63 screw mounting, permanently attached to the equipment.

Minimum size: 3/4" x 2 1/2" with 3/8" letters.

1 2 3

6 7

9

-, ,

4 Manufacturers:5 Setonply [®] Style

Setonply [®] Style 2060 by Seton Name Plate Company or Emedolite Style EIP by EMED Co., or equal by W. H. Brady.

Pipe Identification:

8 Pipe identification shall conform to ANSI A13.1 "Scheme for Identification of Piping Systems".

Printed labels identifying the fluid conveyed and direction of flow shall be attached to pipes in accessible locations, at intervals

not to exceed 20 feet, not less than once in each room, at each branch, adjacent to each access door or panel, at each valve and where exposed piping passes through walls and floors.

13

Outside Diameter of Pipe	Minimum Size of Letters
Covering	
up to 1¼"	1/2"
1½" to 2"	3⁄4"
2½" to 6"	1½"
8" to 10"	21/2"
10" and larger	3½"

14

15 Manufacturers:

16 EMED Co., Seton Name Plate Company, or W. H. Brady.

17

18 Stencils:

19 Not less than 1 inch high letters/numbers for marking pipe and equipment.20

21 Valve Tags:

Identify each valve by means of 1½" diameter brass tag fastened to body of valve with copper or brass chain. Identification number shall be stamped thereon with letters a minimum of ½" high. System identification abbreviation shall be stamped with letters a minimum of ¼" high.

25 26 27

28

The following prefixes shall be used:

PLBG - Plumbing

29 Manufacturers:

30 EMED Co., Seton Name Plate Company, or W. H. Brady.

31

32 Valve Charts:

Furnish three charts listing each valve. Two charts shall be delivered to A/E. An additional chart shall be framed behind glass
 and hung in location selected by Owner. Charts shall show the following:

Valve number	Size
Manufacturer	Type of valve
Type of service	Location

Furnish a typewritten chart indicating equipment or areas served by each numbered valve and incorporate in Operating and
 Maintenance Manuals.

42

43 EQUIPMENT ACCESSORIES

44 Provide equipment accessories, connections, and incidental items.

45 Install piping connecting to pumps and other equipment without strain at the piping connection. If requested by the A/E,

46 remove the bolts in these flanged connections, or disconnect piping, to demonstrate that piping has been properly connected.

47

48 BEDDING AND BACKFILL

Bedding up to a point 12-inches above the top of the pipe shall be thoroughly compacted sand or crushed stone chips meetingthe following gradations:

Gradation for Bedding Sand	
% Passing (by Wt)	Sieve Size
100	1/2 inch
45 - 80	No. 4
	% Passing (by Wt) 100 45 - 80

Gradation for Crushed Stone Chip Bedding	
Sieve Size	% Passing (by Wt)
1/2 inch	100
No. 4	75 - 100

No. 200	2 - 10	No. 1	.00	10 - 25	
Backfill above t perishable, and	he bedding in lawn areas frozen materials.	shall be thoroughly	compacted	excavated material free of large	stones, organic,
Backfill above tl materials, pit ru	ne bedding under existing n sand, gravel, or crushed s	and future utilities, I tone, free from large	oaving, sidew stones, organ	valks, curbs, roads and buildings s nic, perishable, and frozen materia	hall be granular ls.
		PART 3 – E	XECUTION		
GENERAL					
Coordination of	Work:				
Review the com changes necessa Changes require	plete set of Drawings and ary. Coordinate with each d caused by neglect to coordinate d caused by neglect by the set d caused by neglect by the set d caused by neglect by neglect to coordinate d caused by neglect by the set d caused by neglect by the set d caused by neglect by the set d caused by neglect by neglect by the set d caused by the set d c	Specifications and re trade prior to begin rdinate shall be made	port discrep nning installa without exp	ancies to the A/E. Obtain written tion and make provisions to avoi ense to the project.	instructions for id interferences.
Piping shall not l	be located above electrical	panels.			
Anchor Bolts, Sl	eeves, and Supports:				
These items req (except as other anchor bolts, sle location or insta	uired for the Work shall b wise specified) by the trad eves, inserts and supports allation of anchor bolts, sh	e furnished by the F e furnishing and inst shall be directed by t eeves, inserts and su	PC for prope alling the ma he trade req pports shall	r installation of his work. They s terial in which they are to be loca uiring them. Expense resulting fro be paid for by the Contractor for	hall be installed ated. Location of om the improper r the trade with
responsibility to	r directing their proper loca	ition.			
Adjustments In	Locations:				
Locations of pip Prior to fabricati	es and equipment, shall b on determine the exact rou	e adjusted to accom ite and location of ea	modate the ch pipe (subj	work interferences anticipated ar ect to A/E's approval).	nd encountered.
Diaht Of Mary					
New lines which	pitch shall have the right	-of-way over those w	/hich do not	pitch. For example: Gravity drair	s shall normally
have right-of-wa changed. Notify	ay. Lines whose elevations A/E and other trades of co	cannot be changed nflicts.	shall have th	e right-of-way over lines whose e	levations can be
Offsets, transitic pitch of sloping	ons and changes in direction ines whether or not indicat	n of electrical racewar ed on the Drawings.	ys, pipes, and	l ducts shall be made to maintain p	proper room and
DEMOLITION					
Perform all dem adjacent to exis contamination of services as if the	nolition as indicated on the ting work that remains in of the occupied space. W y were new work. Coordin	e drawings to accom an occupied area, c here pipe is remove ate work with the Ow	plish new work onstruct tem and not r ner to minin	ork. Where demolition work is t porary dust partition to minimize econnected with new work, cap nize disruption to the existing build	o be performed the amount of ends of existing ling occupants.
All pipe, fixtures removed from t to the Owner fo indicated to be r	s, equipment, wiring, assoc he site by the Contractor e or his use at a place and tin eused equal to that existing	ciated conduit and si xcept as specifically r me he so designates. g before work began.	milar items o noted otherw Maintain tl	demolished, abandoned, or deacti ise. All designated equipment is to ne condition of material and/or eo	ivated are to be be turned over quipment that is
Install lines pass soil.	ing under foundations with	n minimum of 1-1/2 in	nch clearance	e to concrete and insure no disture	pance of bearing
Before burying Representative.	piping, mark up Record	Drawings and dim	nensionally I	ocate piping. Deliver informatio	n to A/E Field
Unless otherwis depths of cover	e specifically indicated on E from existing grade or from	prawings, trenches for indicated finish grad	r utilities sha e, whichever	l be of depth that provides the foll is lower:	owing minimum
Storm	and sanitary sewers:				
As des	cribed in DPS 382.30 (11) (I	o). Provide insulation	for sewers in	stalled at less than minimum dept	:h.
Water The to	service and/or fire service p of pipe shall be installed i	piping: not less than six (6) fe	et below gra	de.	

1

5

8

Existing utility lines to be retained shown on Drawings or locations of which are made known to Contractor prior to excavation,
 as well as utility lines uncovered during excavation operations, shall be protected from damage during excavation and
 backfilling and if damaged, shall be repaired by Contractor at his expense.

6 Perform excavation and backfill work to accomplish indicated mechanical systems installation in accordance with Section 7 312000 – Earth Moving.

9 DEWATERING

Provide, operate and maintain all pumps and other equipment necessary to drain and keep all excavation pits, trenches and the entire subgrade area free from water under all circumstances. Obtain general permit from the Wisconsin Department of Natural Resources district office for discharge of construction dewatering effluent. Obtain well permit from the Wisconsin Department of Natural Resources district office for dewatering wells discharging more than 70 GPM. Comply with permit requirements.

15

16 ROCK EXCAVATION

17 Remove rock encountered in the excavation to a minimum dimension of six (6) inches outside the pipe. Rock excavation 18 includes all hard, solid rock in ledges, bedded deposits and unstratified masses, all natural conglomerate deposits so firmly 19 cemented as to present all the characteristics of solid rock; which material is so hard or so firmly cemented that in the opinion 20 of the Engineer it is not practical to excavate and remove same with a power shovel except after thorough and continuous 21 drilling and blasting. Rock excavation includes rock boulders of 1/2 cubic yard or more in volume.

22

Rock excavation will be computed on the basis of the depth of rock removed and a trench width two (2) feet larger than the outside diameter of the pipe where one (1) pipe is laid in the trench and three (3) feet larger than the combined outside diameter where two (2) pipes are laid in the trench. Include 6 inch pipe and structure bedding in rock excavation. Include rock excavation shown on the plans in the Base Bid.

28 SURFACE RESTORATION

29 Completely restore the surface of all disturbed areas to a like condition of the surface prior to the work. Level off all waste 30 disposal areas and clean up all areas used for the storage of materials or the temporary deposit of excavated earth. Remove all 31 surplus material, tools and equipment.

32

33 CONCRETE WORK

Cast-in-place concrete within the building will be performed by the Division 03 Contractor unless otherwise noted. Provide all layout drawings, anchor bolts, metal shapes, and/or templates required to be cast into concrete or used to form concrete for support or installation of plumbing piping, fixtures, specialties and equipment. Coordinate locations of equipment, pipe penetrations in wet areas, etc. with the Division 03 Contractor.

38

Plumbing related cast-in-place concrete on the exterior of the building to be provided by this Contractor in conformance with
 requirements of Division 03. This includes piping thrust restraints, pipe supports, hydrant supports, manholes, catch basins,
 grease traps, septic tanks, distribution boxes, valve pits, meter pits, cleanout cover pads, yard hydrant pads, etc.

43 **OPENINGS, CUTTING AND PATCHING**

44 Refer to Division 01 of the Project Manual.

45

Provisions for openings including chases, holes and clearances through walls, floors, and roof, ceilings and partitions shall be
 made in advance of construction of each part of the building. Openings shall be provided by the GC for the respective materials
 in which openings occur, during the construction of the building with the exception of pipe sleeves. The PC shall furnish to the
 GC opening dimensions and locations.

50

If the PC neglects to inform the GC of his opening requirements before that portion of the building construction is complete, the PC shall cut the openings and provide framing and lintels. In the event holes must be cut through reinforced concrete, avoid spalling and unnecessary damage or weakening of structural members. No chopping or breaking out is permitted. Before cutting or drilling, obtain permission from the A/E. Patch adjacent materials and repair damage resulting from the cutting.

55

The PC may perform core drilling for openings in existing walls and floors at the direction of the A/E. Framed openings shall be by the GC.

58

59 Patch interior trench excavation to match existing slab-on-grade with concrete: 3500 PSI at 28 days, 3" slump, 3/4" maximum 60 aggregate size, 5.5 bags of cement per cubic yard.

1 **BUILDING ACCESS**

2 Arrange for necessary openings in building to allow for admittance of all apparatus. When building access was not previously arranged and must be provided by Contractor, restore opening to original condition after the apparatus has been brought into 3 4 building. Coordinate with Architect/Engineer.

EQUIPMENT ACCESS 6

7 Install piping, conduit, fixtures, and accessories to permit access to equipment for maintenance. Coordinate exact location of wall and ceiling access panels and doors with General Contractor, making sure access is available for equipment and specialties. 8 9 Where access is required in plaster walls or ceilings, furnish and install access doors required. Coordinate for installation of 10 access doors utilizing General Contractor and other appropriate on-site subcontractor for access door installation.

11

14

5

Accessible ceilings, (i.e. lay-in ceilings) do not require access panels. Provide color coded thumb tacks or screws, depending on 12 13 surface, for use in accessible ceilings.

15 COORDINATION OF WORK

16 Install systems, equipment and piping in cooperation with other trades. Locations of pipes, equipment, fixtures, etc., shall be 17 adjusted to accommodate the work interferences anticipated and encountered. Prior to fabrication determine the exact route 18 and location of each pipe (subject to A/E's approval).

19

20 Any work that is not coordinated and that interferes with other contractor's work shall be removed or relocated at the installing 21 contractor's expense. 22

23 Verify that all devices are compatible for the type of construction and surfaces on which they will be used.

25 Offsets, transitions and changes in direction of electrical raceways, pipes and ducts shall be made as required to maintain proper room and pitch of sloping lines whether or not indicated on the Drawings. Furnish and install all traps, air vents, 26 27 sanitary vents, etc., as required to effect the offsets, transitions and changes in direction.

28

24

29 New lines which pitch shall have the right-of-way over those which do not pitch. For example: Gravity drains shall normally 30 have right-of-way. Lines whose elevations cannot be changed shall have the right-of-way over lines whose elevations can be 31 changed. Notify A/E and other trades of any conflicts. 32

33 Provide appropriate sections of work with required wall, roof and floor opening locations and dimensions. If Contractor 34 neglects to coordinate information, openings shall be the responsibility of Contractor.

35

PIPING INSTALLATION 36

37 General:

38 Expansion and contraction of piping shall be provided for by expansion loops, bends, swing joints, or expansion joints to 39 prevent damage to connections, piping, equipment of the building.

40

41 Unions or flanges shall be installed on all by-passes, ahead of all traps, adjacent to screw connection valves, and at all 42 connections to equipment, whether or not shown on drawings.

43

44 Installation Arrangement:

45 Install all Work to permit removal (without damage to other parts) of all parts requiring periodic replacement or maintenance. 46 Arrange pipes and equipment to permit ready access to valves, cocks, traps, starters, motors, control components and to clear 47 the openings of swinging and overhead doors and of access panels.

48

49 **Connections Different From Those Shown:**

50 Where equipment requiring different arrangement or connections from those shown is used, install the equipment to operate 51 properly and in harmony with the intent of the Drawings and Specifications. When requested by the A/E, submit drawings 52 showing the proposed installation.

53

If the proposed installation is approved, make all incidental changes in piping, ductwork, supports, insulation, wiring, 54 55 panelboards, etc. Provide any additional motors, controllers, valves, fittings and other additional equipment required for the proper operation of the system resulting from the selection of equipment, including all required changes in affected trades. 56 57 The Contractor shall be responsible for the proper location of rough-in and connections by other trades.

- 58

59 All changes shall be made at no increase in the Contract amount or additional cost to the other trades. 60

61 LUBRICATION AND MAINTENANCE

Lubricate all bearings with lubricant as recommended by the manufacturer before the equipment is operated for any reason. 62 63 Once the equipment has been run, maintain lubrication in accordance with the manufacturer's instructions until the work is

WARNER PARK COMM AND REC CENTER CONTRACT #: 9502 MUNIS #: 17196

accepted by the Owner. Maintain a log of all lubricants used and frequency of lubrication; include this information in the
 Operating and Maintenance Manuals at the completion of the project.

SLEEVES

5 Provide galvanized sheet metal sleeves for pipe penetrations through interior and exterior walls to provide a backing for sealant or firestopping. Patch wall around sleeve to match adjacent wall construction and finish. Grout area around sleeve in masonry construction. In finished spaces where pipe penetration through wall is exposed to view, sheet metal sleeve shall be installed flush with face of wall. In existing poured concrete walls where penetration is core drilled, pipe sleeve is not required.

10 Pipe sleeves are not required in existing poured concrete walls where penetrations are core drilled.

11

9

3 4

Pipe sleeves in new poured concrete construction shall be schedule 40 steel pipe (sized to allow insulated pipe to run throughsleeve), cast in place.

14

In all piping floor penetrations, fire rated and non-fire rated, top of sleeve shall extend 1 inch above the adjacent finished floor. In existing floor penetrations, core drill sleeve opening large enough to insert schedule 40 sleeve and grout area around sleeve with hydraulic setting, non-shrink grout. If the pipe penetrating the sleeve is supported by a pipe clamp resting on the sleeve, weld a collar or struts to the sleeve that will transfer weight to existing floor structure.

19

For floor penetrations through existing floors in mechanical and wet locations listed below, core drill opening and provide 1-1/2" x 1-1/2" x 1/8" galvanized steel angles fastened to floor surrounding the penetration or group of penetrations to prevent water from entering the penetration. Provide urethane caulk between angles and floor and fasten angles to floor a minimum of 8" on center. Seal corners watertight with urethane caulk. Or, core drill sleeve openings large enough to insert schedule 40 sleeve and grout area around sleeve with hydraulic setting non-shrink grout/cement.

25

For pipe penetrations through existing floors in food service areas, core drill sleeve opening large enough to insert schedule 40 sleeve and grout area around sleeve with hydraulic setting non-shrink grout/cement. Size sleeve to allow insulated pipe to pass through sleeve and paint the sleeve.

29

Pipe sleeves are not required in cored floor pipe penetrations through existing floors that are not located in mechanical rooms,
 food service areas or wet locations listed above.

33 **PIPE PENETRATIONS**

34 General:

Coordinate location of building surface penetrations with appropriate contractors. Furnish sleeves, inserts, and devices to be
 built into structure to contractor performing Work. Prepare Shop Drawings for approval for penetrations of structural elements,
 including floor slabs, shear walls, and bearing walls. Do not allow penetrations to be made until Shop Drawings are approved.

38

39 Fire Rated Surfaces:

Install products in accordance with the manufacturer's instructions where pipe penetrates a fire rated surface. When pipe is insulated, use product that maintains integrity of insulation and vapor barrier. Where sleeve must be installed in existing floor, grout area around sleeve to restore floor integrity. In wet area floor penetration, top surface of penetration to be 2 inches above adjacent floor with additional height obtained by means of concrete pad poured integral with floor.

45 Non-Rated Surfaces:

Install escutcheons or floor/ceiling plates where pipe penetrates non-fire rated surfaces in occupied spaces. Size units to accommodate insulation, where applicable. Escutcheons are not required when insulation completely covers wall opening and insulation end is trimmed in a neat manner. Occupied spaces for this Paragraph include only those rooms with finished ceilings and penetration occurs below ceiling.

50

In exterior wall openings below grade, place water-stop type wall sleeve before concrete pour or core drill opening after pour.
 Assemble rubber links to proper size for pipe and tighten in place in accordance with manufacturer's instructions.

53

Install galvanized sheet metal sleeve in hollow wall penetrations to provide backing for sealant. Apply sealant to both sides of
 penetration in a manner that annular space between pipe sleeve and pipe or insulation is completely blocked.

56

57 Completely seal (or caulk) around pipe penetrations through non-rated, smoke tight corridor walls in healthcare facilities. Refer 58 to architectural drawings for additional information.

5960 ESCUTCHEON PLATES

Provide plates on pipes passing through finished floors, walls and ceilings, with outside diameter to cover sleeve opening and inside diameter to fit snugly around pipe. Set tight to building surface. Escutcheon plates shall be chromium plated metal.

1 PAINTING

2 Refer to Division 09.

4 All exposed steel support structures (all metal surfaces located both inside and outside the building) shall be painted after 5 installation with one coat of a compatible metal primer coat and two coats of a finish coat of paint for the application. Color 6 shall be gray unless otherwise specified.

Exposed piping (uninsulated piping located inside the building) shall be painted after installation with 1 coat of compatible
 metal primer coat and 2 coats of finish coat of paint for application.

- 11 Coordinate paint color with architect.
- 12

3

7

Piping systems shall be clearly identified after painting with pipe markings specified under the Paragraph titled Identificationunder this Section.

15

16 IDENTIFICATION

17 Identify equipment in mechanical equipment rooms by stenciling equipment number and service with one coat of black enamel 18 against a light background or white enamel against a dark background. Use a primer where necessary for proper paint 19 adhesion.

20

21 Where stenciling is not appropriate for equipment identification, engraved name plates may be used. 22

Identify interior piping not less than once every 30 feet, not less than once in each room, adjacent to each access door or panel, and on both side of the partition where accessible piping passes through walls or floors. Place flow directional arrows at each pipe identification location. Use one coat of black enamel against a light background or white enamel against a dark background.

20 50

Identify all exterior buried piping for entire length with underground warning tape except for sewer piping which is routed in straight lines between manholes or cleanouts. Place tape 6"-12" below finished grade along entire length of pipe. Extend tape to surface at building entrances, meters, hydrants and valves. Where existing underground warning tape is broken during excavation, replace with new tape identifying appropriate service and securely spliced to ends of existing tape.

Identify valves with brass tags bearing a system identification and a valve sequence number. Identify medical gas and vacuum valves with brass tags and wall or cabinet mounted color coded engraved nameplate with the following "(Type of Gas) Shutoff Valve for (Location or Zone)". Valve tags are not required at a terminal device unless the valves are greater than ten feet from the device, located in another room or not visible from device. Provide a typewritten valve schedule and pipe identification schedule indicating the valve number and the equipment or areas supplied by each valve and the symbols used for pipe identification; locate schedules in mechanical room and in each Operating and Maintenance manual. Schedule in mechanical room to be framed under clear plastic.

- 40
- 41
- 42

END OF SECTION

WARNER PARK COMM AND REC CENTER CONTRACT #: 9502 MUNIS #: 17196

1	SECTION 22 05 14
2	PLOMINING SPECIALITES
2 2	
5	PART 1 - GENERAL
6	
7	SCOPE
8	This section includes specifications for backflow preventers, hose bibs, water hammer arrestors and other miscellaneous
9	plumbing specialties. Included are the following topics:
10	
11	PART 1 - GENERAL
12	Scope
13	Related Work
14	Reference Standards
15	Quality Assurance
17	Submittais
18	PART 2 - PRODUCTS
19	General
20	Hose Bibbs/Wall Hydrants
21	
22	PART 2 - EXECUTION
23	Installation
24	
25	RELATED WORK
26	Requirements of Division 01 shall govern work under this Section.
27	Section 22.0F.00 Common Work Pocults for Dlumbing
20	Section 22.05.00 – Common Work Results for Plumbing Pining and Equipment
30	Section 22 05 25 – Hangers and Supports for Frambing Fighing and Equipment
31	Section 22 11 00 – Facility Water Distribution
32	Section 22 13 00 – Facility Sanitary Sewerage
33	Section 22 14 00 – Facility Storm Drainage
34	Section 22 30 00 – Plumbing Equipment
35	Section 22 40 00 – Plumbing Fixtures
36	
37	REFERENCE STANDARDS
38	ANSI A112.14.1 - Backwater Valves
39	ANSI A112.26.1/PDI WH-201 - Water Hammer Arrestors.
40	ASSE 1001 - Pipe Applieu Almospheric Type Vacuum Breakers.
41	ASSE 1010 - Water Hammer Arrestors. ASSE 1011 - Hose Connection Vacuum Breakers
43	ASSE 1011 - Hose connection vacuum breakers.
44	ASSE 1013 - Reduced Pressure Principle Backflow Preventers.
45	ASSE 1018 - Trap Seal Primer Valves.
46	ASSE 1019 - Wall Hydrants, Frost Proof Automatic Draining, Anti-Backflow Type.
47	
48	QUALITY ASSURANCE
49	Substitution of Materials: Refer to Section 22 05 00 and Division 01 of the Project Manual.
50	
51	Plumbing products requiring approval by the State of Wisconsin Dept. of Safety and Professional Services must be approved or
52	nave pending approval at the time of shop drawing submission.
53	
55	Submit product data sheets in accordance with Division 01 and Section 22.05.00
56	
57	
58	
59	Submit and pay all fees to State of Wisconsin for reduced pressure zone backflow prevention device review. Submit State
60	approval of reduced pressure zone backflow prevention device with product data sheets in accordance with Division 01 and
61	Section 22 05 00.
62	
63	

2 GENERAL Refer to Plumbing Equipment Schedule for specific model numbers and sizing information regarding the plumbing fittings and specialties specified herein. 6 BACKFLOW PREVENTION DEVICES 8 Acceptable Manufacturers: 9 Cash-Acme, Chicago, Cla-Val, Conbraco, Febco, Nidel, Watts, Wilkins, or Woodford. 11 Vacuum Breakers: 12 For use in finished areas with concealed piping, brass construction, chromium plated, Chicago Faucet No. 892-G (½"). 13 For exposed piping in unfinished areas, brass construction, Watts series 288A. 16 Hose thread inlet and outlet, non-removable hose connection, vacuum breaker for use on service sink faucets, Chicago Faucet No. E27, % inch. 18 HOSE BIBBS/WALL HYDRANTS 19 HOSE BIBBS/WALL HYDRANTS 20 PART 3 - EXECUTION 21 INSTALLATION 22 PART 3 - EXECUTION 23 END OF SECTION	1	PART 2 - PRODUCTS
3 GENERAL 4 Refer to Plumbing Equipment Schedule for specific model numbers and sizing information regarding the plumbing fittings and specialties specified herein. 6 BACKFLOW PREVENTION DEVICES 7 BACKFLOW PREVENTION DEVICES 8 Acceptable Manufacturers: 9 Cash-Acme, Chicago, Cla-Val, Conbraco, Febco, Nidel, Watts, Wilkins, or Woodford. 10 Vacuum Breakers: 11 Vacuum Breakers: 12 For use in finished areas with concealed piping, brass construction, chromium plated, Chicago Faucet No. 892-G (½"). 13 For exposed piping in unfinished areas, brass construction, Watts series 288A. 14 Hose thread inlet and outlet, non-removable hose connection, vacuum breaker for use on service sink faucets, Chicago Faucet 16 Hose BIBBS/WALL HYDRANTS 17 Hose bibbs and wall hydrants shall be manufactured by Chicago Faucet, MIFAB, Woodford, or Zurn. 12 PART 3 - EXECUTION 24 PART 3 - EXECUTION 25 INSTALLATION 26 INSTALLATION 27 Install 24 inches above finished grade or floor. 28 END OF SECTION	2	
 Refer to Plumbing Equipment Schedule for specific model numbers and sizing information regarding the plumbing fittings and specialties specified herein. BACKFLOW PREVENTION DEVICES Acceptable Manufacturers: Cash-Acme, Chicago, Cla-Val, Conbraco, Febco, Nidel, Watts, Wilkins, or Woodford. Vacuum Breakers: For use in finished areas with concealed piping, brass construction, chromium plated, Chicago Faucet No. 892-G (%"). For exposed piping in unfinished areas, brass construction, Watts series 288A. Hose thread inlet and outlet, non-removable hose connection, vacuum breaker for use on service sink faucets, Chicago Faucet No. E27, ¼ inch. HOSE BIBBS/WALL HYDRANTS Hose bibbs and wall hydrants shall be manufactured by Chicago Faucet, MIFAB, Woodford, or Zurn. INSTALLATION Hose Bibbs/Wall Hydrants: Install 24 inches above finished grade or floor. END OF SECTION 	3	GENERAL
 specialties specified herein. BACKFLOW PREVENTION DEVICES Acceptable Manufacturers: Cash-Acme, Chicago, Cla-Val, Conbraco, Febco, Nidel, Watts, Wilkins, or Woodford. Vacuum Breakers: For use in finished areas with concealed piping, brass construction, chromium plated, Chicago Faucet No. 892-G (½"). For exposed piping in unfinished areas, brass construction, Watts series 288A. Hose thread inlet and outlet, non-removable hose connection, vacuum breaker for use on service sink faucets, Chicago Faucet No. E27, ¼ inch. HOSE BIBBS/WALL HYDRANTS Hose bibbs and wall hydrants shall be manufactured by Chicago Faucet, MIFAB, Woodford, or Zurn. INSTALLATION Hose Bibbs/Wall Hydrants: Install 24 inches above finished grade or floor. END OF SECTION 	4	Refer to Plumbing Equipment Schedule for specific model numbers and sizing information regarding the plumbing fittings and
6 ACKFLOW PREVENTION DEVICES 8 Acceptable Manufacturers: 9 Cash-Acme, Chicago, Cla-Val, Conbraco, Febco, Nidel, Watts, Wilkins, or Woodford. 10 Vacuum Breakers: 11 For use in finished areas with concealed piping, brass construction, chromium plated, Chicago Faucet No. 892-G (½"). 12 For exposed piping in unfinished areas, brass construction, Watts series 288A. 15 Hose thread inlet and outlet, non-removable hose connection, vacuum breaker for use on service sink faucets, Chicago Faucet No. E27, ¾ inch. 18 HOSE BIBBS/WALL HYDRANTS 19 HOSE BIBBS/WALL HYDRANTS 20 PART 3 - EXECUTION 21 PART 3 - EXECUTION 22 PART 3 - EXECUTION 23 PART 3 - EXECUTION 24 Install 24 inches above finished grade or floor. 25 INSTALLATION 26 Hose Bibbs/Wall Hydrants: 27 Install 24 inches above finished grade or floor. 28 END OF SECTION	5	specialties specified herein.
 BACKFLOW PREVENTION DEVICES Acceptable Manufacturers: Cash-Acme, Chicago, Cla-Val, Conbraco, Febco, Nidel, Watts, Wilkins, or Woodford. Vacuum Breakers: For use in finished areas with concealed piping, brass construction, chromium plated, Chicago Faucet No. 892-G (½"). For exposed piping in unfinished areas, brass construction, Watts series 288A. Hose thread inlet and outlet, non-removable hose connection, vacuum breaker for use on service sink faucets, Chicago Faucet No. E27, ¼ inch. HOSE BIBBS/WALL HYDRANTS Hose bibbs and wall hydrants shall be manufactured by Chicago Faucet, MIFAB, Woodford, or Zurn. INSTALLATION Hose Bibbs/Wall Hydrants: Install 24 inches above finished grade or floor. END OF SECTION 	6	
 Acceptable Manufacturers: Cash-Acme, Chicago, Cla-Val, Conbraco, Febco, Nidel, Watts, Wilkins, or Woodford. Vacuum Breakers: For use in finished areas with concealed piping, brass construction, chromium plated, Chicago Faucet No. 892-G (½"). For exposed piping in unfinished areas, brass construction, Watts series 288A. Hose thread inlet and outlet, non-removable hose connection, vacuum breaker for use on service sink faucets, Chicago Faucet No. E27, ¾ inch. HOSE BIBBS/WALL HYDRANTS Hose bibbs and wall hydrants shall be manufactured by Chicago Faucet, MIFAB, Woodford, or Zurn. PART 3 - EXECUTION Hose Bibbs/Wall Hydrants: Install 24 inches above finished grade or floor. END OF SECTION 	7	BACKFLOW PREVENTION DEVICES
 Cash-Acme, Chicago, Cla-Val, Conbraco, Febco, Nidel, Watts, Wilkins, or Woodford. Vacuum Breakers: For use in finished areas with concealed piping, brass construction, chromium plated, Chicago Faucet No. 892-G (½"). For exposed piping in unfinished areas, brass construction, Watts series 288A. Hose thread inlet and outlet, non-removable hose connection, vacuum breaker for use on service sink faucets, Chicago Faucet No. E27, ¾ inch. HOSE BIBBS/WALL HYDRANTS Hose bibbs and wall hydrants shall be manufactured by Chicago Faucet, MIFAB, Woodford, or Zurn. INSTALLATION Hose Bibbs/Wall Hydrants: Install 24 inches above finished grade or floor. END OF SECTION 	8	Acceptable Manufacturers:
 Vacuum Breakers: For use in finished areas with concealed piping, brass construction, chromium plated, Chicago Faucet No. 892-G (½"). For exposed piping in unfinished areas, brass construction, Watts series 288A. Hose thread inlet and outlet, non-removable hose connection, vacuum breaker for use on service sink faucets, Chicago Faucet No. E27, ¾ inch. HOSE BIBBS/WALL HYDRANTS Hose bibbs and wall hydrants shall be manufactured by Chicago Faucet, MIFAB, Woodford, or Zurn. PART 3 - EXECUTION Hose Bibbs/Wall Hydrants: Install 24 inches above finished grade or floor. END OF SECTION 	9	Cash-Acme, Chicago, Cla-Val, Conbraco, Febco, Nidel, Watts, Wilkins, or Woodford.
 Vacuum Breakers: For use in finished areas with concealed piping, brass construction, chromium plated, Chicago Faucet No. 892-G (½"). For exposed piping in unfinished areas, brass construction, Watts series 288A. Hose thread inlet and outlet, non-removable hose connection, vacuum breaker for use on service sink faucets, Chicago Faucet No. E27, % inch. HOSE BIBBS/WALL HYDRANTS Hose bibbs and wall hydrants shall be manufactured by Chicago Faucet, MIFAB, Woodford, or Zurn. INSTALLATION Hose Bibbs/Wall Hydrants: Install 24 inches above finished grade or floor. END OF SECTION 	10	
 For use in finished areas with concealed piping, brass construction, chromium plated, Chicago Faucet No. 892-G (½"). For exposed piping in unfinished areas, brass construction, Watts series 288A. Hose thread inlet and outlet, non-removable hose connection, vacuum breaker for use on service sink faucets, Chicago Faucet No. E27, ¾ inch. HOSE BIBBS/WALL HYDRANTS Hose bibbs and wall hydrants shall be manufactured by Chicago Faucet, MIFAB, Woodford, or Zurn. INSTALLATION Hose Bibbs/Wall Hydrants: Install 24 inches above finished grade or floor. END OF SECTION 	11	Vacuum Breakers:
 For exposed piping in unfinished areas, brass construction, Watts series 288A. Hose thread inlet and outlet, non-removable hose connection, vacuum breaker for use on service sink faucets, Chicago Faucet No. E27, ¼ inch. HOSE BIBBS/WALL HYDRANTS Hose bibbs and wall hydrants shall be manufactured by Chicago Faucet, MIFAB, Woodford, or Zurn. PART 3 - EXECUTION Hose Bibbs/Wall Hydrants: Install 24 inches above finished grade or floor. END OF SECTION 	12 13	For use in finished areas with concealed piping, brass construction, chromium plated, Chicago Faucet No. 892-G (½").
 Hor exposed piping in diministed areas, brass construction, watts series 2004. Hose thread inlet and outlet, non-removable hose connection, vacuum breaker for use on service sink faucets, Chicago Faucet No. E27, ¾ inch. HOSE BIBBS/WALL HYDRANTS Hose bibbs and wall hydrants shall be manufactured by Chicago Faucet, MIFAB, Woodford, or Zurn. PART 3 - EXECUTION Hose Bibbs/Wall Hydrants: Install 24 inches above finished grade or floor. END OF SECTION 	1/	For exposed nining in unfinished areas, brass construction. Watts series 2884
 Hose thread inlet and outlet, non-removable hose connection, vacuum breaker for use on service sink faucets, Chicago Faucet No. E27, ¾ inch. HOSE BIBBS/WALL HYDRANTS Hose bibbs and wall hydrants shall be manufactured by Chicago Faucet, MIFAB, Woodford, or Zurn. PART 3 - EXECUTION INSTALLATION Hose Bibbs/Wall Hydrants: Install 24 inches above finished grade or floor. END OF SECTION 	15	To exposed piping in driminine areas, brass construction, wates series 200A.
 No. E27, ¾ inch. HOSE BIBBS/WALL HYDRANTS Hose bibbs and wall hydrants shall be manufactured by Chicago Faucet, MIFAB, Woodford, or Zurn. PART 3 - EXECUTION INSTALLATION Hose Bibbs/Wall Hydrants: Install 24 inches above finished grade or floor. END OF SECTION 	16	Hose thread inlet and outlet, non-removable hose connection, vacuum breaker for use on service sink faucets. Chicago Faucet
 HOSE 227, 75 meth HOSE BIBBS/WALL HYDRANTS Hose bibbs and wall hydrants shall be manufactured by Chicago Faucet, MIFAB, Woodford, or Zurn. PART 3 - EXECUTION INSTALLATION Hose Bibbs/Wall Hydrants: Install 24 inches above finished grade or floor. END OF SECTION 	17	No E27 % inch
 HOSE BIBBS/WALL HYDRANTS Hose bibbs and wall hydrants shall be manufactured by Chicago Faucet, MIFAB, Woodford, or Zurn. PART 3 - EXECUTION INSTALLATION Hose Bibbs/Wall Hydrants: Install 24 inches above finished grade or floor. END OF SECTION 	18	
 Hose bibbs and wall hydrants shall be manufactured by Chicago Faucet, MIFAB, Woodford, or Zurn. PART 3 - EXECUTION INSTALLATION Hose Bibbs/Wall Hydrants: Install 24 inches above finished grade or floor. END OF SECTION 	19	HOSE BIBBS/WALL HYDRANTS
21 22 22 23 23 PART 3 - EXECUTION 24 25 25 INSTALLATION 26 Hose Bibbs/Wall Hydrants: 27 Install 24 inches above finished grade or floor. 28 29 30 END OF SECTION	20	Hose bibbs and wall hydrants shall be manufactured by Chicago Faucet, MIFAB, Woodford, or Zurn.
 PART 3 - EXECUTION INSTALLATION Hose Bibbs/Wall Hydrants: Install 24 inches above finished grade or floor. END OF SECTION 	21	
 PART 3 - EXECUTION INSTALLATION Hose Bibbs/Wall Hydrants: Install 24 inches above finished grade or floor. END OF SECTION 	22	
 24 25 INSTALLATION 26 Hose Bibbs/Wall Hydrants: 27 Install 24 inches above finished grade or floor. 28 29 30 END OF SECTION 	23	PART 3 - EXECUTION
 INSTALLATION Hose Bibbs/Wall Hydrants: Install 24 inches above finished grade or floor. 8 9 30 END OF SECTION 	24	
 Hose Bibbs/Wall Hydrants: Install 24 inches above finished grade or floor. END OF SECTION 	25	INSTALLATION
 Install 24 inches above finished grade or floor. END OF SECTION 	26	Hose Bibbs/Wall Hydrants:
28 29 30 END OF SECTION	27	Install 24 inches above finished grade or floor.
29 30 END OF SECTION	28	-
30 END OF SECTION	29	
	30	END OF SECTION

1	SECTION 22 05 29
2	HANGERS AND SUPPORTS FOR PLUMBING PIPING AND EQUIPMENT
з	
4	
5	DART 1 - GENERAL
5	
0	
/	
8	SCOPE
9	This section includes specifications for supports of all plumbing equipment and materials as well as piping system anchors.
10	Included are the following topics:
11	
12	PART 1 - GENERAL
13	Scope
14	Related Work
15	Reference Standards
16	Quality Assurance
17	Design Criteria
18	Submittals
19	
20	PART 2 - PRODUCTS
20	Manufacturers
21	Dine Hangers and Supports
22	Pipe Hangers and Supports
23	Poer Clampa
24	Bion Clamps
25	Riser Clamps
26	Concrete Inserts
27	Anchors
28	Equipment Support
29	
30	PART 3 - EXECUTION
31	Installation
32	Structural Supports
33	Hanger and Support Spacing
34	Riser Clamps
35	Concrete Inserts
36	Anchors
37	
38	RELATED WORK
39	Applicable provisions of Division 01 shall govern work under this section.
40	
41	Section 22 05 00 – Common Work Results for Plumbing
42	Section 22 07 00 – Plumbing Insulation
43	Section 22 11 00 – Facility Water Distribution
44	Section 22 13 00 – Facility Sanitary Sewerage
45	Section 22 14 00 – Facility Storm Drainage
46	Section 22 40 00 – Plumbing Fixtures
47	
48	REFERENCE STANDARDS
49	MSS SP-58
50	MSS SP-69
51	
52	
52	Refer to Division 01 of the Project Manual
55	
54	
55	Design Chinekia
סכ	iviatenais and application of pipe nangers and supports shall be in accordance with MISS standard Practice SP-58 and SP-69 UNIESS
5/	noted otherwise.
58	
59	Piping connected to pumps, compressors, or other rotating or reciprocating equipment is to have vibration isolation supports for
60	a distance of one hundred pipe diameters or three supports away from the equipment, whichever is greater. Standard pipe
61	hangers/supports as specified in this section are required beyond the 100 pipe diameter/3 support distance.
62	
63	Do not hang any mechanical item directly from a metal deck or run piping so its rests on the bottom chord of any truss or joist.
64	

1 General:

3

8

13

17

20 21

22

23

25

29

32

34

36

2 Secure pipe in place to prevent vibration, maintain proper slope and provide for expansion and contraction.

Design supports of strength and rigidity to suit loading, service, and manner which do not unduly stress the building construction.
Where support is from concrete construction, take care not to weaken concrete or penetrate waterproofing. Fasten supports
and hangers to building steel framing wherever practical. Do not use another pipe for support. Do not use perforated iron, chain
or wire as hangers.

Use inserts for suspending hangers from reinforced concrete slabs wherever practical. Where inserts are not practical, provide
 channels or angles from which to suspend hangers/supports. Fasten structural steel to concrete with expansion bolts.

12 Provide expansion anchors in concrete slabs for installation of threaded support rods.

Provide hangers capable of vertical adjustment after piping is erected. Do not pierce ductwork with hanger rods. On threaded
 support rods and bolts, weld nuts to rods, peen threads, or provide double set of nuts with lock washers to prevent loosening.
 Use beam clamps for attaching hangers to structural steel.

On piping insulated with vapor barrier covering, use protection shield to cover bottom one-half of insulated pipe. Shield to be a
 minimum of 12" long and of 16 gauge galvanized steel.

Exception:

For insulated drain pipe, the pipe may rest on the hanger and the insulation to wrap around the hanger and pipe.

24 Submit anchor drawings for approval upon request.

Hangers, supports, and support methods other than those specified shall not be used without obtaining approval on method of support by the Structural Engineer prior to installing piping systems. Submit support method arrangement, pipe weight and spacing scheme for approval.

30 Hanger and Support Spacing:

31 Install hangers to provide minimum 1/2 inch space between finished covering and adjacent work.

33 Place a hanger within 12 inches of each horizontal elbow, valve, strainer, or similar piping specialty item.

35 Use hangers with 1-1/2 inch minimum vertical adjustment.

37 Where several pipes can be installed in parallel and at the same elevation, provide multiple or trapeze hangers.

38 39 40

39 Support riser piping independently of connected horizontal piping.

41 Adjust hangers to obtain the slope specified in the piping section of these specifications.

43 Space hangers for pipe as follows:

42
43
44

Pipe Material	Pipe Size	Max. Horiz. Spacing	Max. Vert. Spacing
Cast Iron	2" and larger	5'-0"	15'-0"
Copper	1/2" through 3/4"	5'-0"	10'-0"
Copper	1" through 1-1/4"	6'-0"	10'-0"
Copper	1-1/2" through 2-1/2"	8'-0"	10'-0"
Copper	3"	10'-0"	10'-0"
Copper	4" and larger	12'-0"	10'-0"
Ductile Iron	All	10'-0"	20'-0"
Glass	Per Pipe Mfr.	8'-0''	12'-0"
Steel	1/2" through 1-1/4"	7'-0"	15'-0"
Steel	1-1/2" through 6"	10'-0"	15'-0"
Steel	8" through 12"	14'-0"	20'-0"
Steel	14" and over	20'-0"	20'-0"
Plastic	Drain and Vent	4'-0''	10'-0"
Plastic	1" or less	32"	4'-0"
Plastic	1-1/4" and over	4'-0"	6'-0"
Plastic	Pure Water 1-1/2" or less	Continuous	5'-0"

1 2	SUBMITTALS Submit data in accordance with Section 22 05 00 and Division 01 of the Project Manual.		
3 4 5 6	Schedule of all hanger and support devices indicating attachment methods and type of device for each pipe size and type of service.		
7 8	Submit anchor drawings to the A/E for approval upon request.		
9 10	PART 2 - PRODUCTS		
11	MANUFACTURED		
12 13 14	B-Line, Fee and Mason, Grinnell, Michigan Hanger, Pate, PHD Manufacturing, Piping Technology, Powers/Rawl, Proset, Roof Products & Systems, Unistrut, or Victaulic.		
15			
16	PIPE HANGERS AND SUPPORTS		
17 18 10	Overhead Supports: Adjustable clevis hanger, steel, Dura-Green epoxy coating or electro-plated, B-Line Figure B3100.		
20 21	Adjustable J hook hanger, steel, Dura-Green epoxy coating or electro-plated, B-Line figure B3690.		
22 22 23	Adjustable band hanger, steel, Dura-Green epoxy coating or electro-plated, B-Line Figure B3172.		
24 25 26	Where several pipes are running parallel and pitching in the same direction, strut style support may be used. Steel channel, 12- gauge thickness, Dura-Green epoxy coating or electro-plated, B-Line B11. Restrain individual pipes with B-Line B2000 series or Vibraclamp series strut clamps		
27	vioraciamp series strat claimps.		
28	Wall Support:		
29 30	Carbon steel welded bracket with hanger. B-Line 3068 Series, Grinnell 194 Series.		
31 32 33	Perforated, epoxy painted finish, 16-12 gauge, min., steel channels securely anchored to wall structure, with interlocking, split- type, bolt secured, galvanized pipe/tubing clamps. B-Line type S channel with B-2000 series clamps, Grinnell type PS 200 H with PS 1200 clamps.		
35 36 37	When copper piping is being supported, provide flexible elastomeric/thermoplastic isolation cushion material to completely encircle the piping and avoid contact with the channel or clamp, equal to B-Line B1999 Vibra Cushion or provide manufacturers clamp and cushion assemblies, B-Line BVT series, Grinnell PS 1400 series.		
39 40	Vertical Support: Riser clamp, steel, Dura-Green epoxy coating or electro-plated, B-Line Figure B3373.		
41			
42 43	Riser clamp, flexible sleeve with stainless steel band, Proset PS #33.		
44	Floor Support:		
45 46	Carbon steel pipe saddle, stand and bolted floor flange. B-Line B3088T/B3093.		
47	Copper Pipe Supports:		
48 49	All supports, fasteners, clamps, etc. directly connected to copper piping shall be copper plated or polyvinylchloride coated. Where steel channels are used, provide isolation collar between supports/clamps/fasteners and copper piping.		
5U E 1			
52	FIFE HANGER RODS Steel Hanger Rods:		
52 53 54	Steel, electro-plated, threaded both ends, threaded one end, or continuous threaded, complete with adjusting and lock nuts. B- Line 83205		
55 56	Size rods for individual hangers and trapeze support as indicated in the following schedule:		
57 58	Total weight of equipment including valves fittings nine nine content and insulation are not to exceed the limits indicated		
59	יסנמי שכוקות סי בקמוחוורות, ווכוממוווד עמועבז, ותנוודד, חוף לטותבות, מום ווזמומנוסוו, מוב ווסרנס בתכבים נוול ווווונז וומוכמנפט.		
	Maximum Load (Lbs.) Rod Diameter		
	(650°F Maximum Temp.) (inches)		

610

1130

3/8

1/2

1810	5/8
2710	3/4
3770	7/8
4960	1
8000	1-1/4

BEAM CLAMPS

MSS SP-69 Types 19 & 23 malleable black iron clamp for attachment to beam flange to 0.62 inches thick with a retaining ring and threaded rod of 3/8, 1/2, and 5/8 inch diameter. Furnish with a hardened steel cup point set screw. B-Line B3036L/B3034, Grinnell 86/92.

5 6 7

1 2

3

Δ

8 9 MSS SP-69 Type 28 or Type 29 forged steel jaw type clamp with a tie rod to lock clamp in place, suitable for rod sizes to 1-1/2 inch diameter. B-Line B3054, Grinnell 228.

10 CONCRETE INSERTS

11 Poured in Place:

MSS SP-69 Type 18 wedge type to be constructed of a black carbon steel body with a removable malleable iron nut that accepts threaded rod to 7/8 inch diameter. Wedge design to allow the insert to be held by concrete in compression to maximize the load carrying capacity. B-Line B2505, Grinnell 281.

15

18

16 MSS SP-69 Type 18 universal type to be constructed of black malleable iron body with a removeable malleable iron nut that 17 accepts threaded rod to 7/8 inch diameter. B-Line B3014N, Grinnell 282.

19 Mechanical-Expansion Anchors:

Insert-wedge-type anchors, for use in hardened portland cement concrete, with pull-out, tension, and shear capacities appropriate for supported loads and building materials where used. Zinc-coated or stainless steel expansion anchors. Use drill bit of same manufacturer as anchor.

- 23
- 24 Manufactured By:

25 Hilti, ITW/Read Head, or equal.

26

27 ANCHORS

Use welding steel shapes, plates, and bars to secure piping to the structure.

30 EQUIPMENT SUPPORT

Examine Drawings, and manufacturer's data to determine how equipment, fixtures, and piping are to be supported, mounted or suspended. Support all equipment plumb, rigid, and true to line. Provide rods, bolts, inserts, pipe stands, brackets and accessories for proper support.

34

35 Equipment Stands:

Use structural steel members welded to and supported by pipe supports. Clean, prime and coat with three coat rust inhibiting
 alkyd paint or one coat epoxy mastic. Where exposed to weather, treat with corrosive atmosphere coatings.

39 CORROSIVE ATMOSPHERE COATINGS

Factory coat supports and anchors used in corrosive atmospheres with hot dip galvanizing after fabrication, ASTM A123, 1.5
 ounces/square foot of surface each side. Mechanical galvanize threaded products, ASTM B695 Class 50, 2.0 mil coating. Field cuts
 and damaged finishes to be field covered with zinc rich paint of comparable thickness to factory coating.

- 44 Corrosive atmospheres include the following locations:
- 45 Locker rooms
- 46
- 47
- 48
- 49 50

52

.....

50 INSTALLATION

51 Size, apply and install supports and anchors in compliance with manufacturers recommendations.

Install supports to provide for free expansion of the piping system. Support all piping from the structure using concrete inserts, beam clamps, ceiling plates, wall brackets, or floor stands. Fasten ceiling plates and wall brackets securely to the structure and test to demonstrate the adequacy of the fastening.

PART 3 - EXECUTION

56

57 Coordinate hanger and support installation to properly group piping of all trades.

1 Where piping can be conveniently grouped to allow the use of trapeze type supports, use standard structural shapes or 2 continuous insert channels for the supporting steel. Where continuous insert channels are used, pipe supporting devices made

continuous insert channels for the supporting steel. Where continuous insert channels are used, pipe supporting devices made
 specifically for use with the channels may be substituted for the specified supporting devices provided that similar types are used
 and all data is submitted for prior approval.

5

8

- Size and install hangers and supports, except for riser clamps, for installation on the exterior of piping insulation. Where a vapor
 barrier is not required, hangers may be installed either on the exterior of pipe insulation or directly on piping.
- 9 Perform welding in accordance with standards of the American Welding Society.

10 11 STRUCTURAL SUPPORTS

Provide all supporting steel required for the installation of mechanical equipment and materials, including angles, channels, beams, etc. to suspended or floor supported tanks and equipment. All of this steel may not be specifically indicated on the drawings.

15

16 RISER CLAMPS

Support vertical piping with clamps secured to the piping and resting on the building structure or secured to the building structurebelow at each floor.

1920 CONCRETE INSERTS

Select size based on the manufacturer's stated load capacity and weight of material that will be supported. Use inserts for suspending hangers from reinforced concrete slabs and sides of reinforced concrete beams. Provide hooked rod to concrete reinforcement section for inserts carrying pipe over 4 inch size. Where concrete slabs form finished ceiling, provide inserts that are flush with the slab surface.

25

26 ANCHORS

Install where indicated on the drawings and details. Where not specifically indicated, install anchors at ends of principal pipe runs and at intermediate points in pipe runs between expansion loops. Make provisions for preset of anchors as required to accommodate both expansion and contraction of piping.

- 30
- 31
- 32

1 2 3	SECTION 22 07 00 PLUMBING INSULATION
4 5	PART 1 - GENERAL
6	
/ 8	SCOPE This Section includes insulation specifications for plumbing systems. Included are the following requirements:
9 10	
11	Scope
12	Related Work
13	Description
14	Quality Assurance
15	Definitions
16	Submittals
1/ 18	
10	Accentable Manufacturers
20	Insulation and Jackets
21	Plenum Wrap
22	
23	PART 3 - EXECUTION
24	General
25	Installation Bine Insulation Schoolule
20	Pipe insulation Schedule
27	RELATED WORK
29	Requirements of Division 01 shall govern work under this Section.
30	
31	Section 22 05 00 - Common Work Results for Plumbing
32	Section 22 05 29 - Hangers and Supports for Plumbing Piping and Equipment
33	Section 22 11 00 - Facility Water Distribution
34	Section 22 13 00 - Facility Sanitary Sewerage
35	Section 22 14 00 - Facility Storm Drainage
30 37	DESCRIPTION
38	Furnish and install insulating materials, fittings, finishes, and accessories specified for piping and related equipment. The following
39	types of insulation are specified in this Section:
40	Pipe insulation
41	
42	Install insulation materials in accordance with the latest edition of MICA (Midwest Insulation Contractors Association) Standard
43	and manufacturer's installation instructions. Exceptions to these standards will only be accepted where specifically modified in
44 4F	these Specifications, or where prior written approval has been obtained from Engineer.
45 46	ΟΠΑΠΤΥ ASSURANCE
47	Substitution of Materials: Refer to Section 22.05.00 and Division 01 of the Project Manual.
48	
49	Label insulating products delivered to construction site with the manufacturer's name and description of materials.
50	
51	DEFINITIONS
52	Concealed:
53	shafts, furred spaces, space above finished ceilings, utility tunnels and crawl spaces. Other areas, including walk-through tunnels,
54 55	shall be considered as exposed.
56	Exposed to weather:
57	Located outdoors, either on grade, on a wall, or on a roof, in location where sun, wind, rain, snow and other elements will come
58 59	in contact with it.
60	Unconditioned spaces:
61	Unheated or non-cooled attics, utility tunnels and crawl spaces where ambient temperatures may rise above 90 degrees F or
62 63	drop below 50 Degrees F. Ducts in these instances are considered to be located outside of building thermal envelope.

1	SUBMITTALS				
2	Submit data in accordance with Section 22 05 00 and Division 01 of the Project Manual				
3					
4	Include manufacturer's data for the following:				
5	Pipe insulation				
6					
7	Submittal shall include the following information:				
8					
9	Manufacturer's technical data sheets for each product with the following information:				
10	Density				
11	Thermal characteristics				
12	Temperature limitations				
13	Jacket type				
14	Materials of composition				
15	Material safety data sheets				
16					
17	Schedule of all insulating materials to be used including:				
18	 Application / intended use of each insulation type 				
19	Insulation type and thickness				
20	Jacket type				
21	 Fastening methods and adhesive type 				
22					
23					
24	PART 2 - PRODUCTS				
25					
26	ACCEPTABLE MANUFACTURERS				
27	Armstrong, Halstead, Johns-Manville, Knauf, or Owens-Corning.				
28					
29	INSULATION AND JACKETS				
30	Glass Fiber:				
31	Manville Micro-Lok meeting ASI M C547; rigid molded, non-combustible, "K" Value: 0.23 at 75°F, maximum service temperature:				
32	850°F, with vapor Retarder Jacket: AP-I Plus White Kraft paper reinforced with glass fiber yarn and bonded to aluminum foil,				
33	secure with self-sealing longitudinal laps and butt strips or AP Jacket with outward clinch expanding staples or vapor barrier				
34	mastic as needed.				
35					
36	PVC Fitting Covers and Jackets:				
3/	white PVC nim, gloss finish one side, semi-gloss other side, FS LP-335D, Composition A, Type II, Grade GO. Ottraviolet innibited				
38	indoor/outdoor grade to be used where exposed to high numidity, ultraviolet radiation. Jacket thickness to be .02 inch (20 mil).				
39					
40	DADT 2 EVECUTION				
41	PART 5 - EAECOTION				
42	GENERAL				
43	Application of insulation to nining equipment shall be done in accordance with the manufacturer's installation recommendations				
44	Application of instalation to piping equipment shall be done in accordance with the manufacturer's installation recommendations.				
45	where thickness of insulation is not specified, use thickness recommended by manufacture of required by applicable codes.				
40	Insulation shall be applied in as warm an environment as possible, and in no instance below 25° E				
47 //8	insulation shall be applied in as warm an environment as possible, and in to instance below 25 T.				
40	No nine shall be covered until after it has been installed inspected tested and approved				
50	No pipe shan be covered until arter it has been instance, inspected, tested, and approved.				
50	ΙΝΣΤΑΙΙΑΤΙΩΝ				
52	All nice insulation shall be installed with joints butted firmly together. All valves and fittings shall be insulated with mitered				
52	sections of insulation equal in density and thickness to the adioining insulation, or with insulating compared equal in thickness to				
54	the adjoining insulation or with "Zeston" type premoled PVC fittings installed in accordance with the manufacturer's				
55	instructions Fittings are to be finished with 8 or glass mesh and mastic (use breather mastic on systems operating above 50°E				
56	except where Zeston PVC covers are used). Jackets on pipe insulation may be stapled using outward clinch staples spaced 3"				
57	apart at least X'' in from the lan edge on systems operating at 60°F and above: below 60°F the lans are to be varies operating at 60°F and above: below 60°F the lans are to be varies operating at 60°F and above.				
58	self-sealing lan, lan-seal tane gun or adhesive such as Armstrong 520. All insulation ends are to be taneed and sealed regardless				
59	of service.				
60					
61	On all piping insulated with vapor barrier covering, use protection shield to over bottom one-half of insulated pipe. Shield to be				
62	minimum of 12" long and 16 gauge galvanized steel. Provide half-round. 12" long, hanger block at the bottom half of the pipe in				
63	place of the fiberglass pipe insulation. The hanger blocks shall be molded cork or calcium silicate pipe insulation of the same				
64	thickness as the adjoining fiberglass pipe insulation. The vapor barrier jacket shall be continuous through the hanger location.				

1				
2	Vapor barrier jackets shall be applied with a continuous, unbroken vapor seal. Pipe hangers shall be sized large enough to be			
3	installed over the outer surfaces of the insulation.			
4				
5	Exception:			
6	For insulated drain pipe, the pipe may rest directly on the hanger and the insulation to wrap around the hanger and			
7	pipe.			
8				
9	Provide a complete vapor barrier for insulation on the following systems:			
10	Cold hard water			
11	Cold soft water			
12	Non-potable cold soft water			
13	Storm drainage			
14	Clearwater waste drainage			
15				
16	Omit insulation for:			
17	Unions and flanges.			
18	 Vents to atmosphere, discharges from safety and relief valves and drain pipes. 			
19				
20	Provide finished edges at all access doors and end.			
21				
22	PIPE INSULATION SCHEDULE			
23	Provide insulation on new and remodeled piping.			
24				
25	Minimum Insulation Thickness:			

26

	<u>PIPE SIZE</u>			
SYSTEMS	1" or less	1-1/4" to 2"	2-1/2" to 4"	5" and up
Storm Drain*			1"	1″
Clearwater Waste*		1″	1″	1″
Domestic Cold Water	1/2"	1/2″	1″	1″
Domestic Hot Water	1″	1″	1-1/2"	1-1/2"
Domestic Hot Water Return	1″	1″	1-1/2″	
Non-Potable Cold Water	1/2"	1/2"	1″	
Non-Potable Hot Water	1″	1″	1-1/2"	

27

* Provide pipe insulation on above ground horizontal storm and clearwater drain piping, underside of roof drain, and initial 5 feet
 of vertical conductors.

- 30
- 31
- 32

1	SECTION 22 11 00
2 3	FACILITY WATER DISTRIBUTION
4	
5 6	PART 1 - GENERAL
7	SCOPE
8 9	This section contains specifications for plumbing pipe and pipe fittings for this project. Included are the following topics:
10	PART 1 – GENERAL
11	Scope
12	Related Work
13	Description
14 15	Submittals
16	Submittais
17	PART 2 – PRODUCTS
18	Water Distribution Pipe and Fittings
19	Valves
20	Unions and Flanges
21	Dielectric Couplings
22	water nammer suppressors
25 74	ΡΔΡΤ 3 - ΕΧΕΓΙΙΤΙΩΝ
25	Trenching, Backfilling and Compacting
26	Water Piping System
27	Testing
28	
29	RELATED WORK
30 21	Requirements of Division 01 shall govern work under this Section.
32	22 05 00 – Common Work Results for Plumbing
33	22 05 14 – Plumbing Specialties
34	22 05 29 – Hangers and Supports for Plumbing Piping and Equipment
35	
36	DESCRIPTION
37	Provide a domestic water distribution system including hot and cold water supply piping, hot water return piping, tempered
38 20	water piping, pure water piping, valves, fittings, hardware, and specialties. Connect to plumbing fixtures, specialties, and
39 40	equipment.
41	QUALITY ASSURANCE
42	Substitution of Materials: Refer to Section 22 05 00 and Division 01 of the Project Manual.
43	
44	Order all pipe with each length marked with the name or trademark of the manufacturer and type of pipe; with each shipping
45	unit marked with the purchase order number, metal or alloy designation, temper, size, and name of supplier.
46	Any installed material not masting the specification requirements must be realized with material that masts these specifications
47 18	without additional cost to the Owner
49	
50	To assure uniformity and compatibility of piping components in grooved piping systems, all grooved products utilized shall be
51	supplied by a single manufacturer. Grooving tools shall be supplied from the same manufacturer as the grooved components.
52	
53	SUBMITTALS
54 55	Submit valve product data sheets in accordance with Section 22 05 00 and Division 01 of the Project Manual.
55 56	Include materials of construction dimensional data ratings/capacities/ranges approvals test data and identification as
57	referenced in this section and/or on the drawings.
58	······································
59	

1	PART 2 - PRODUCTS
2 3 4	WATER DISTRIBUTION PIPE AND FITTINGS
5	2" and Smaller:
6 7	Copper tube, type K, soft temper, ASTM B88, with wrought copper fittings. ANSI B16.22. Join using lead free flux and solder, ASTM B32, flux ASTM B813.
8 9 10 11 12	Plastic, 80% Engel minimum Cross-linked Polyethylene ("PEX") pipe, type PEX-a only, ASTM F876, ASTM F877, 100 psi pressure rating at 180°F, ASTM F1960 cold expansion polymer or lead-free brass/bronze fittings from same manufacturer as pipe, all rated for potable water NSF/ANSI 61. Bronze/brass fittings shall be used when adapting from PEX to metallic piping. Approved manufacturers: Rehau, Sioux Chief, and Uponor.
13	Above Ground:
15 16 17	Copper tube, Type L, hard temper, ASTM B88; with wrought copper fittings, ANSI B16.22. Join using lead free flux, ASTM B813, and solder, ASTM B32.
18 19 20 21	Wrought copper, ANSI B16.22 or cast bronze, ANSI B16.18 fittings, copper tube dimensioned grooved ends (flaring of tube and fitting ends to IPS dimensions is not permitted), joined with mechanical couplings, synthetic rubber gasket seal, Victaulic style 607 QuickVic TM Installation Ready stab-on design, for direct 'stab' installation onto roll grooved copper tube without prior field disassembly and no loose parts.
22 22	VALVES
23	Manufacturer:
25	Valves throughout the project shall be by one manufacturer, unless otherwise specified.
26	
27 28	Standard valves are based on Nibco models. Equivalent style valves as manufactured by Apollo, Crane, DeZurik, Gustin-Bacon, Grinnell, Hammond, Jenkins, Lunkenheimer, Milwaukee Valve, Stockham, Victaulic, or Watts are acceptable. Valves shall be of
29 30	standard dimensions, comparable to the number specified.
31 32 33	Balancing valves are based on Bell & Gossett models. Equivalent style valves by Armstrong, Flowset, Nibco, Taco, or Victaulic/TA Hydronics are acceptable.
34	Shutoff Valves:
35 36 27	Except as otherwise specified, all shutoff valves 2-1/2 inch and smaller shall be ball valves and shutoff valves 3 inch and larger shall be butterfly valves, unless required otherwise by local Water Utility specifications.
3/	Dell Malvae
38	Ball Valves: Bronze, two niese full next hall valves with bronze hady, calder or threaded ands, shremium plated bross or staipless steel hall.
39 40 41	reinforced Teflon seats and seals, blowout proof stem design, rated at 600 PSI non-shock WOG, Nibco model T/S-585-70. Include handle extension for insulated piping, NIB-SEAL by Nibco.
42	
43 44	Bronze, two piece full port ball valves with bronze body, solder or threaded ends, stainless steel ball, reinforced Teflon seats and seals, blowout proof stem design, rated at 600 PSI non-shock WOG, Nibco model T/S-585-70-66. Include handle extension for insulated
45 46	piping, NIB-SEAL by Nibco.
40 17	Bronze three niece full nort ball valves with bronze body, solder or threaded ands, stainless, steal ball, reinforced Teflon seats and seals
47 78	blowquit proof stem design instead at 600 PSI non-shork WOG. Nibco model T(C-SSS-66) Include handle extension for insulated
40 49 50	piping, NIB-SEAL by Nibco.
51	Butterfly Valves:
52	Ductile iron butterfly valve, polymid coated, EPDM elastomer coated disc, extended neck, grooved ends, 300 psi WOG pressure
53 54	rated, Nibco GD 4765. Include lever handle through 6-inch size and gear operator for 8 inch and larger size.
55	Cast bronze butterfly valve, EPDM elastomer coated ductile iron disc, copper tube dimensioned grooved ends, 300 psi maximum
56 57	pressure rated, Victaulic Series 608. Include lever handle through 6-inch size.
58	Check Valves:
59	3" and Smaller:
60 61	Bronze body, Class 125, Y-pattern, swing type, check valve with solder ends, all bronze internal components and renewable seat and disc, Nibco model S-413-B.
62	

- 63 2" and Smaller:
- 64 Bronze body, ASTM B62, in-line lift type, spring, Buna-N disc, 250 psig WOG rating. Nibco 480

1	
2	Balancing Valves:
3	½" thru 2":
4	Bronze body balancing valve with sweat or threaded ends, calibrated brass orifice, integral adjustment knob with calibrated scale,
5	memory stop indicator, drain tapping and differential pressure metering connections, Bell & Gossett "Circuit Setter".
0 7	LINIONS AND ELANGES
, 8	Unions:
9	Bronze, solder connection, Nibco figure 733.
10	
11	Flanges:
12	Cast copper alloy, class 125, MSS SP-106, Nibco figure 741.
13	
14 15	DIELECTRIC COUPLINGS Stoel casing zine electroplated with inert thermonlastic lining various and types. Clearflow, style 47 by Victoulic
15 16	Steel casing, zinc electropiated, with mert thermopiastic inning, various end types, clearnow, style 47 by victadiic.
17	Dielectric flanges 2" and larger: with iron female pipe thread to copper solder joint or brass female pipe thread end connections.
18	non-asbestos gaskets and pressure rating of not less than 175 psig at 180 degrees Fahrenheit. Watts Regulator Company,
19	Lochinvar, Wilkins, Epco Sales, Inc.
20	
21	WATER HAMMER SUPPRESSORS
22 22	Acceptable manufacturers are MIFAB, PPP, Sloux Chief, and Watts.
25 74	Piston compressed air column type, with sealed air chamber
25	risten tempicisted dir column type, with scaled dir chamber.
26	Water supply piping serving fixtures, appliances, equipment and devices with quick closing and/or solenoid-actuated valves shall
27	be provided with water hammer arrestors. Also provide where indicated on the water supply piping as shown on the water
28	supply isometrics. Devices shall be mechanical arrestors installed in accordance with PDI Standard WH201. Air chambers are not
29	considered to be equal.
30 21	Shop drawings are required. Submit to Λ/F for approval prior to installation
32	
33	Water hammer arrestors must be accessible for inspection and replacement. Provide access panel.
34	
35	
36 27	PART 3 - EXECUTION
38	TRENCHING, BACKFILLING AND COMPACTING
39	See Section 22 05 00.
40	
41	WATER PIPING SYSTEM
42	Piping shall be pitched to drain entire system; install drain valves at low points. Provide unions at equipment and valves. Provide
43	offsets and transition fittings as required. Avoid dips or depressions in pipe runs.
44 15	No water nining shall be installed in exterior walls unless adequately protected from freezing. Two inch insulation shall be
46	installed on back and sides of chase, front shall be open to room heat, covered only by finished wall material.
47	······································
48	Install unions, couplings, or flanges at all final equipment connections and as required to facilitate removal of equipment.
49	
50	Install dielectric couplings at every connection between copper pipe and other metals. Use dielectric unions for connecting
51 52	copper and steel piping.
52 53	Provide backflow devices as required by Code on water connections to HVAC equipment and other equipment
54	
55	Extend hot water piping from water heater and connect to all fixtures and equipment as required.
56	
57	Hot water and cold water lines shall be kept at least 6 inches apart whenever possible.
58 50	Hat Watar Pa Circulating System:
59 60	not water re-circulating system. Install return system including check values halancing values and numns. Pitch and grade all lines as required to ensure
61	satisfactory circulation.

1 Adjust each balancing valve and set position stop. Balance system to minimum flow in return piping branches needed to maintain 2 even supply water temperature and to provide continuous circulation throughout building. Provide balancing report along with 3 O&M manual submittals. Test and demonstrate to A/F upon request.

3 O&M manual submittals. Test and demonstrate to A/E upon request.

Valve Installation:

5

8

16 17

18

19

31 32

33

34

35

36 37

Install shutoff valves with stem vertical. Exception; the stem may be horizontal if a vertical installation would not allow access to
 the valve handle

9 All valves with screwed ends shall be installed using "Teflon" tape applied on male portion of piping fitting. 10

11 Each individual fixture or piece of equipment shall have an independent shut-off valve adjacent to fixture in addition to the 12 required branch shut-off. Where valves are installed in walls an access panel shall be provided.

13 14 Branches:

15 Valve shut-off full size of branch for each branch take-off to supply stack or fixture group.

Drains:

Provide valved drains at low points of systems as required or directed. All piping shall be arranged to drain through valved drains.

20 Flushing Mains and Branch Piping:

Upon completion of the water distribution system, test all valves to insure their full opening and flush out the system progressively by opening drain valves and building outlets and permitting the flow to continue from each until the water runs clear.

2425 Pipe Insulation:

26 Provide pipe insulation for all domestic water piping per Section 22 07 00.

2728 Sterilization of Water Distribution System:

As soon as the water distribution system has been flushed out as above specified, it shall be sterilized in accordance with the requirements of the local Health Department/Water Utility or in the absence of such, by the following method:

Introduce chlorine or a solution of calcium or sodium hypochlorite, filling the lines slowly and applying the sterilizing agent at a rate of 50 parts per million of chlorine, as determined by residual chlorine tests at the ends of the lines. Open and close all valves and hydrants while the system is being chlorinated.

- After the sterilizing agent has been applied for 24 hours, test for residual chlorine at the ends of the lines. If less than 5 PPM as indicated, repeat the sterilization process.
- When tests show at least 5 PPM of residual chlorine flush out the system until all traces of the chemical used are
 removed.

41 42 Samples

After disinfecting the water distribution system, take water samples to check for bacteria. Take 5 water samples from remote faucets, plus the main entrance. Send the samples to the Wisconsin Department of Health Lab to sample for a safe water supply system.

47 TESTING

48 Refer to Division 01, "Starting of Systems" and Section 22 05 00.

Hydro-statically pressure test water piping to 150 psig for 4 hours. No decrease in pressure is allowed. Provide pressure gauge
 with shutoff and a bleeder valve at the highest point of the system tested. Inspect joints in system under test. No leaks allowed.

- 53 Do not conceal pipe until satisfactorily tested.
- 54

46

49

- 55 Testing with air will not be allowed.
- 56 57
- 58

1	SECTION 22 13 00			
2	FACILITY SANITARY SEWERAGE			
4				
5	PART 1 - GENERAL			
6 7	SCORE			
8	This section contains specifications for plumbing pipe and pipe fittings for this project. Included are the following topics:			
10	PART 1 – GENERAL			
11	Scope			
12	Related Work			
13	Description			
14	Quality Assurance			
15 16	Submittais			
17	PART 2 – PRODUCTS			
18	Underground Pipe Fittings			
19	Above Ground Pipe and Fittings			
20	Drains and Cleanouts			
21				
22	PART 3 - EXECUTION			
23	Drain and Vent Piping System			
24 25	Acid Drain and Vent System			
26	Vent Flashing			
27	Cleanouts			
28	Traps			
29	Testing			
30				
31	RELATED WORK Paquirements of Division 01 shall govern work under this Section			
52 33	Requirements of Division of shall govern work under this section.			
34	22 05 00 – Common Work Results for Plumbing			
35	22 05 14 – Plumbing Specialties			
36	22 05 29 – Hangers and Supports for Plumbing Piping and Equipment			
37				
38	DESCRIPTION			
39 40	Interior sanitary waste and vent and acid drain and vent piping systems including branches, drains, cleanouts, stacks, fittings and hardware.			
41	Work under this section shall commence from 5 feet outside the building wall with connections to sanitary building sewer			
43	lateral(s).			
44				
45	QUALITY ASSURANCE			
46	Substitution of Materials: Refer to Section 22 05 00 and Division 01 of the Project Manual.			
4/	Order all nine with each length marked with the name or trademark of the manufacturer and tune of nine, with each chinning			
48 ⊿9	unit marked with the nurchase order number metal or allow designation temper size, and name of supplier			
- 50	and marked with the parenase order number, metal or andy designation, temper, size, and name of supplier.			
51	Any installed material not meeting the specification requirements must be replaced with material that meets these specifications			
52	without additional cost to the Owner.			
53				
54	SUBMITTALS			
55	Submit data in accordance with Section 22.05.00 and Division 01 of the Project Manual.			
50 57	Schedule from the contractor indicating the ASTM or CISPI specification number of the nine being proposed along with its type			
58 59	and grade, and sufficient information to indicate the type and rating of fittings for each service.			
60 61 62 63	Include materials of construction, dimensional data, ratings/capacities/ranges, approvals, test data, and identification as referenced in this section and/or on the drawings.			

110	
Cas	st iron, no-hub, service weight, ASTM A888, CISPI 301, with rubber gasket couplings, ASTM C564, and stainless steel clamp, PI 310. Pipe and fittings shall be marked with the collective trademark of the Cast Iron Soil Pipe Institute or receive prior proval of the engineer. Biging and fittings shall be manufactured by AB81. Charlotte, or Tyler.
ahl	brovar of the engineer. Piping and fittings shall be manufactured by Abar, chanotte, of Tyler.
Cas ma fitt	st iron soil pipe, bell and spigot, service weight, coated, ASTM A74, with rubber gaskets, ASTM C564. Pipe and fittings shall be rked with the collective trademark of the Cast Iron Soil Pipe Institute or receive prior approval of the engineer. Piping and ings shall be manufactured by AB&I, Charlotte, or Tyler.
PV Sol	C, Schedule 40, Type I, ASTM D-1785, and PVC drain-waste-vent fittings, ASTM D-2665, with solvent weld joints, ASTM D2855. id wall PVC only.
Cas CIS app	st iron, no-hub, service weight, ASTM A888, CISPI 301, with rubber gasket couplings, ASTM C564, and stainless steel clamp, PI 310. Pipe and fittings shall be marked with the collective trademark of the Cast Iron Soil Pipe Institute or receive prior proval of the engineer. Piping and fittings shall be manufactured by AB&I, Charlotte, or Tyler.
PV Sol	C, Schedule 40, Type I, ASTM D-1785, and PVC drain-waste-vent fittings, ASTM D-2665, with solvent weld joints, ASTM D2855. id wall PVC only.
DR	AINS AND CLEANOUTS
Dra	ains and cleanouts manufactured by J.R. Smith, Josam, Wade, Watts, or Zurn.
Ins	tall trap-seal barrier type on floor drains in mechanical rooms.
Ref	fer to Plumbing Drain and Cleanout Schedule.
Flo	or Drains:
See	e Plumbing Drain and Cleanout Schedule for type, manufacturer model required for this project.
	PART 3 - EXECUTION
	PART 3 - EXECUTION
DR Cor Pro	PART 3 - EXECUTION AIN AND VENT PIPING SYSTEM nnect all drain and vent piping to each fixture and piece of equipment and install all required piping as shown on drawings. ovide all necessary fittings and hardware to make required offsets and transitions.
DR Cor Pro	PART 3 - EXECUTION AIN AND VENT PIPING SYSTEM nnect all drain and vent piping to each fixture and piece of equipment and install all required piping as shown on drawings. ovide all necessary fittings and hardware to make required offsets and transitions.
DR Cor Pro Cha 1/6	PART 3 - EXECUTION AIN AND VENT PIPING SYSTEM nnect all drain and vent piping to each fixture and piece of equipment and install all required piping as shown on drawings. ovide all necessary fittings and hardware to make required offsets and transitions. anges in direction of drainage piping shall be made by the appropriate use of 45 degree wyes, long or short sweep 1/4 bends, 5, 1/8, 1/16 bends or combination.
DR Con Pro Cha 1/6 Fitt	PART 3 - EXECUTION AIN AND VENT PIPING SYSTEM nnect all drain and vent piping to each fixture and piece of equipment and install all required piping as shown on drawings. ovide all necessary fittings and hardware to make required offsets and transitions. anges in direction of drainage piping shall be made by the appropriate use of 45 degree wyes, long or short sweep 1/4 bends, 5, 1/8, 1/16 bends or combination. tings to be installed to make for the least possibility of stoppage. All horizontal drainage piping less than 3 inches shall be ched a minimum of 1/4 inch per foot of run. Pitch drainage piping 3 inch and larger a minimum of 1/8" per foot of run.
DR Con Prc Cha 1/6 Fitt pitt	AIN AND VENT PIPING SYSTEM nnect all drain and vent piping to each fixture and piece of equipment and install all required piping as shown on drawings. wide all necessary fittings and hardware to make required offsets and transitions. anges in direction of drainage piping shall be made by the appropriate use of 45 degree wyes, long or short sweep 1/4 bends, 6, 1/8, 1/16 bends or combination. tings to be installed to make for the least possibility of stoppage. All horizontal drainage piping less than 3 inches shall be ched a minimum of 1/4 inch per foot of run. Pitch drainage piping 3 inch and larger a minimum of 1/8" per foot of run. Then running drain piping below a footing and parallel to it, piping shall be in all cases be at least one foot greater in distance ave from footing than below is bottom. Where possible, run sewers at centerpoint between two parallel footings and maintain
DR Col Pro Cha 1/6 Fitt pitu Wh aw abo as	AIN AND VENT PIPING SYSTEM nnect all drain and vent piping to each fixture and piece of equipment and install all required piping as shown on drawings. wide all necessary fittings and hardware to make required offsets and transitions. anges in direction of drainage piping shall be made by the appropriate use of 45 degree wyes, long or short sweep 1/4 bends, 5, 1/8, 1/16 bends or combination. tings to be installed to make for the least possibility of stoppage. All horizontal drainage piping less than 3 inches shall be ched a minimum of 1/4 inch per foot of run. Pitch drainage piping 3 inch and larger a minimum of 1/8" per foot of run. hen running drain piping below a footing and parallel to it, piping shall be in all cases be at least one foot greater in distance ay from footing than below its bottom. Where possible, run sewers at centerpoint between two parallel footings and maintain pove-mentioned distances at a minimum. When running drain piping under a footing, disturb as little of the soil under footing possible. Provide concrete fill under all footings where excavations wider than 18" are required.
DR Col Pro Cha 1/6 Fitt pite WP aw abo as	PART 3 - EXECUTION AIN AND VENT PIPING SYSTEM nect all drain and vent piping to each fixture and piece of equipment and install all required piping as shown on drawings. voide all necessary fittings and hardware to make required offsets and transitions. anges in direction of drainage piping shall be made by the appropriate use of 45 degree wyes, long or short sweep 1/4 bends, 5, 1/8, 1/16 bends or combination. tings to be installed to make for the least possibility of stoppage. All horizontal drainage piping less than 3 inches shall be ched a minimum of 1/4 inch per foot of run. Pitch drainage piping 3 inch and larger a minimum of 1/8" per foot of run. ten running drain piping below a footing and parallel to it, piping shall be in all cases be at least one foot greater in distance ay from footing than below its bottom. Where possible, run sewers at centerpoint between two parallel footings and maintain ove-mentioned distances at a minimum. When running drain piping under a footing, disturb as little of the soil under footing possible. Provide concrete fill under all footings where excavations wider than 18" are required. The running drain piping through a footing, provide a steel pipe sleeve with 2" thick minimum compressible wrap.
DR Col Pro Cha 1/6 Fitt pito aw abo as WP Col	PART 3 - EXECUTION AIN AND VENT PIPING SYSTEM meet all drain and vent piping to each fixture and piece of equipment and install all required piping as shown on drawings. ovide all necessary fittings and hardware to make required offsets and transitions. anges in direction of drainage piping shall be made by the appropriate use of 45 degree wyes, long or short sweep 1/4 bends, 5, 1/8, 1/16 bends or combination. tings to be installed to make for the least possibility of stoppage. All horizontal drainage piping less than 3 inches shall be ched a minimum of 1/4 inch per foot of run. Pitch drainage piping 3 inch and larger a minimum of 1/8" per foot of run. en running drain piping below a footing and parallel to it, piping shall be in all cases be at least one foot greater in distance ay from footing than below its bottom. Where possible, run sewers at centerpoint between two parallel footings and maintain ove-mentioned distances at a minimum. When running drain piping under a footing, disturb as little of the soil under footing possible. Provide concrete fill under all footings where excavations wider than 18" are required. nen running drain piping through a footing, provide a steel pipe sleeve with 2" thick minimum compressible wrap. nect to all drains, fixtures and equipment as required.
DR Col Pro Cha 1/đ Fitti pitu aw abo as Wh Col PIE	PART 3 - EXECUTION AIN AND VENT PIPING SYSTEM Innect all drain and vent piping to each fixture and piece of equipment and install all required piping as shown on drawings. ovide all necessary fittings and hardware to make required offsets and transitions. anges in direction of drainage piping shall be made by the appropriate use of 45 degree wyes, long or short sweep 1/4 bends, 5, 1/8, 1/16 bends or combination. tings to be installed to make for the least possibility of stoppage. All horizontal drainage piping less than 3 inches shall be ched a minimum of 1/4 inch per foot of run. Pitch drainage piping 3 inch and larger a minimum of 1/8" per foot of run. ener running drain piping below a footing and parallel to it, piping shall be in all cases be at least one foot greater in distance ay from footing than below its bottom. Where possible, run sewers at centerpoint between two parallel footings and maintain pove-mentioned distances at a minimum. When running drain piping under a footing, disturb as little of the soil under footing possible. Provide concrete fill under all footings where excavations wider than 18" are required. The running drain piping through a footing, provide a steel pipe sleeve with 2" thick minimum compressible wrap. The running drain piping through a footing, provide a steel pipe sleeve with 2" thick minimum compressible wrap.
DR Coi Pro Cha 1/e Fitt pitr Wr aw abo as Wr Coi PIP Ins "In	PART 3 - EXECUTION AIN AND VENT PIPING SYSTEM nnect all drain and vent piping to each fixture and piece of equipment and install all required piping as shown on drawings. voide all necessary fittings and hardware to make required offsets and transitions. anges in direction of drainage piping shall be made by the appropriate use of 45 degree wyes, long or short sweep 1/4 bends, 5, 1/8, 1/16 bends or combination. tings to be installed to make for the least possibility of stoppage. All horizontal drainage piping less than 3 inches shall be ched a minimum of 1/4 inch per foot of run. Pitch drainage piping 3 inch and larger a minimum of 1/8" per foot of run. then running drain piping below a footing and parallel to it, piping shall be in all cases be at least one foot greater in distance ay from footing than below its bottom. Where possible, run sewers at centerpoint between two parallel footings and maintain ove-mentioned distances at a minimum. When running drain piping under a footing, disturb as little of the soil under footing possible. Provide concrete fill under all footings where excavations wider than 18" are required. The running drain piping through a footing, provide a steel pipe sleeve with 2" thick minimum compressible wrap. Innect to all drains, fixtures and equipment as required. The points tall cast iron pipe and fittings, hubless pattern, as recommended by CISPI standards 301, 310, and in their publication stallation Suggestions for Cast Iron No-Hub Pipe and Fittings".
DR Col Pro Cha 1/6 Fitt pitu Wh awa abo as Wh Col PIP Ins "In Pre app	AIN AND VENT PIPING SYSTEM neet all drain and vent piping to each fixture and piece of equipment and install all required piping as shown on drawings. wide all necessary fittings and hardware to make required offsets and transitions. anges in direction of drainage piping shall be made by the appropriate use of 45 degree wyes, long or short sweep 1/4 bends, 5, 1/8, 1/16 bends or combination. tings to be installed to make for the least possibility of stoppage. All horizontal drainage piping less than 3 inches shall be ched a minimum of 1/4 inch per foot of run. Pitch drainage piping 3 inch and larger a minimum of 1/8" per foot of run. Then running drain piping below a footing and parallel to it, piping shall be in all cases be at least one foot greater in distance ay from footing than below its bottom. Where possible, run sewers at centerpoint between two parallel footings and maintain ove-mentioned distances at a minimum. When running drain piping under a footing, disturb as little of the soil under footing possible. Provide concrete fill under all footings where excavations wider than 18" are required. The running drain piping through a footing, provide a steel pipe sleeve with 2" thick minimum compressible wrap. Innect to all drains, fixtures and equipment as required. The JOINTS The pipe and fittings, hubless pattern, as recommended by CISPI standards 301, 310, and in their publication stallation Suggestions for Cast Iron No-Hub Pipe and Fittings". The pipe PVC pipe ends as recommended by manufacturer. Use a P-70 type primer (for PVC) and a PVC solvent cement propriate to the pipe size and temperature range.

1 VENT FLASHING

All vent pipes passing through roof shall be covered with sheet lead weighing not less than 4 pounds per square foot. Sheet lead shall be well flashed onto the roof, 12" around pipe. Vent pipes shall extend a minimum of 12" above roof.

5 CLEANOUTS

6 Provide and install cleanouts as shown on plans and as required by Code.

7 8 **TRAPS**

9 Trap all fixtures and equipment. Trap seals shall be standard depth, except when deep seals are required by Code. Traps shall 10 be set true and level and located within the limits of the Code requirements. A trap shall not be used as a separator, interceptor 11 or other type of device to retain solids. All traps above grade shall be provided with approved screw-type cleanout plugs.

12

4

13 Traps shall be protected during construction and sealed to prevent foreign matter from entering. Provide adjustable expansion 14 plug, plastic cap, or approved equivalent.

15

16 Install trap-seal protection barrier type on floor drains in mechanical rooms during trim out stage of floor drain installation.

17 18 **TESTING**

19 Refer to Testing paragraph of Section 22 05 00.

Hydro-statically pressure test all piping to 10 feet of water column pressure for 2 hours. No leaks allowed. Provide mint test of

- 22 entire system as required by local inspector.
- 23
- 24 25

1 2 3	SECTION 22 14 00 FACILITY STORM DRAINAGE
4 5	PART 1 - GENERAL
6 7	SCOPE
8	This section contains specifications for plumbing pipe and pipe fittings for this project. Included are the following topics:
9 10	PART 1 – GENERAL
11	Scope
12	Related Work
13	Description
14	Quality Assurance
15 16	Submittais
17	PART 2 – PRODUCTS
18	Underground Pipe Fittings
19	Above Ground Pipe and Fittings
20	Drains and Cleanouts
21 22	ΡΔΡΤ 3 - ΕΧΕΛΙΤΙΟΝ
23	Drain and Vent Piping System
24	Pipe Joints
25	Cleanouts
26	Traps
27	Testing
28 29	
30	Requirements of Division 01 shall govern work under this Section.
31	
32	22 05 00 – Common Work Results for Plumbing
33 34 35	22 05 29 - Hangers and Supports for Plumbing Piping and Equipment 22 05 14 - Plumbing Specialties
36	DESCRIPTION
37	Interior storm drainage, clear-water waste and vent piping systems including branches, drains, cleanouts, stacks, fittings and
38 39	hardware.
40 41	Work under this section shall commence from 5 feet outside the building wall with connections to storm building sewer lateral(s).
42	QUALITY ASSURANCE
43 11	Substitution of Materials: Refer to Section 22 05 00 and Division 01 of the Project Manual.
44 45 46	Order all pipe with each length marked with the name or trademark of the manufacturer and type of pipe; with each shipping unit marked with the purchase order number, metal or alloy designation, temper, size, and name of supplier.
47	
48	Any installed material not meeting the specification requirements must be replaced with material that meets these specifications
49	without additional cost to the Owner.
50 51	SUBMITTALS
52	Submit data in accordance with Section 22 05 00 and Division 01 of the Project Manual.
53	
54 55	Schedule from the contractor indicating the ASTM, or CISPI specification number of the pipe being proposed along with its type and grade, and sufficient information to indicate the type and rating of fittings for each service.
56 57 58	Include materials of construction, dimensional data, ratings/capacities/ranges, approvals, test data, and identification as referenced in this section and/or on the drawings.
59 60 61	PART 2 - PRODUCTS
62	

1 UNDERGROUND PIPE AND FITTINGS 2 Cast iron, no-hub, service weight, ASTM A888, CISPI 301, with rubber gasket couplings, ASTM C564, and stainless steel clamp, CISPI 310. Pipe and fittings shall be marked with the collective trademark of the Cast Iron Soil Pipe Institute or receive prior 3 4 approval of the engineer. Piping and fittings shall be manufactured by AB&I, Charlotte, or Tyler. 5 Cast iron soil pipe, bell and spigot, service weight, coated, ASTM A74, with rubber gaskets, ASTM C564. Pipe and fittings shall be 6 7 marked with the collective trademark of the Cast Iron Soil Pipe Institute or receive prior approval of the engineer. Piping and fittings shall be manufactured by AB&I, Charlotte, or Tyler. 8 9 10 PVC, Schedule 40, Type I, ASTM D-1785, and PVC drain-waste-vent fittings, ASTM D-2665, with solvent weld joints, ASTM D2855. Solid wall PVC only. 11 12 13 ABOVE GROUND PIPE AND FITTINGS 14 Cast iron, no-hub, service weight, ASTM A888, CISPI 301, with rubber gasket couplings, ASTM C564, and stainless steel clamp, CISPI 310. Pipe and fittings shall be marked with the collective trademark of the Cast Iron Soil Pipe Institute or receive prior 15 16 approval of the engineer. Piping and fittings shall be manufactured by AB&I, Charlotte, or Tyler. 17 18 PVC, Schedule 40, Type I, ASTM D-1785, and PVC drain-waste-vent fittings, ASTM D-2665, with solvent weld joints, ASTM D2855. 19 Solid wall PVC only. 20 21 DRAINS AND CLEANOUTS 22 Drains and cleanouts manufactured by J.R. Smith, Josam, Wade, Watts, or Zurn. 23 24 Refer to Plumbing Drain and Cleanout Schedule. 25 26 27 PART 3 - EXECUTION 28 DRAIN AND VENT PIPING SYSTEM 29 30 Connect all drain and vent piping to each fixture and piece of equipment and install all required piping as shown on drawings. 31 Provide all necessary fittings and hardware to make required offsets and transitions. 32 33 Changes in direction of drainage piping shall be made by the appropriate use of 45 degree wyes, long or short sweep 1/4 bends, 34 1/6, 1/8, 1/16 bends or combination. 35 Fittings to be installed to make for the least possibility of stoppage. All horizontal drainage piping less than 3 inches shall be 36 37 pitched a minimum of 1/4 inch per foot of run. Pitch drainage piping 3 inch and larger a minimum of 1/8" per foot of run. 38 39 When running drain piping below a footing and parallel to it, piping shall be in all cases be at least one foot greater in distance 40 away from footing than below its bottom. Where possible, run sewers at centerpoint between two parallel footings and maintain 41 above-mentioned distances at a minimum. When running drain piping under a footing, disturb as little of the soil under footing 42 as possible. Provide concrete fill under all footings where excavations wider than 18" are required. 43 44 When running drain piping through a footing, provide a steel pipe sleeve with 2" thick minimum compressible wrap. 45 46 Connect to all drains, fixtures and equipment as required. 47 PIPE JOINTS 48 49 Install cast iron pipe and fittings, hubless pattern, as recommended by CISPI standards 301, 310, and in their publication 50 "Installation Suggestions for Cast Iron No-Hub Pipe and Fittings". 51 52 Prepare PVC pipe ends as recommended by manufacturer. Use a P-70 type primer (for PVC) and a PVC solvent cement 53 appropriate to the pipe size and temperature range. 54 55 Soldered joints shall be as described in Section 22 05 00. 56 57 VENT FLASHING 58 All vent pipes passing through roof shall be covered with sheet lead weighing not less than 4 pounds per square foot. Sheet lead 59 shall be well flashed onto the roof, 12" around pipe. Vent pipes shall extend a minimum of 12" above roof. 60 61 CLEANOUTS 62 Provide and install cleanouts as shown on plans and as required by Code. 63

1 TRAPS

2 Trap all fixtures and equipment. Trap seals shall be standard depth, except when deep seals are required by Code. Traps shall 3 be set true and level and located within the limits of the Code requirements. A trap shall not be used as a separator, interceptor 4 or other type of device to retain solids. All traps above grade shall be provided with approved screw-type cleanout plugs.

5
6 Traps shall be protected during construction and sealed to prevent foreign matter from entering. Provide adjustable expansion
7 plug, plastic cap, or approved equivalent.

8 9 **TESTING**

- 10 Refer to Testing paragraph of Section 22 05 00.
- 11

Hydro-statically pressure test all piping to 10 feet of water column pressure for 2 hours. No leaks allowed. Provide mint test of entire system as required by local inspector.

- 14
- 15
- 16

SECTION 22 14 05 FOUNDATION DRAINAGE
PART 1 - GENERAL
This section includes specifications for plumbing pipe and pipe fittings for this project. Included are the following topics:
PART 1 - GENERAL
Scope
Related Work
Related Documents
Description
Product Handling
Quality Assurance
Submittals
PART 2 - PRODUCTS
Pipe and Fittings
Joints
Filter Fabric
Stone
Cleanouts
PART 2 - EXECUTION
Installation
RELATED WORK
Requirements of Division 01 shall govern work under this Section.
Section 22 05 00 – Common Work Results for Plumbing
Section 22 05 14 – Plumbing Specialties
Section 22 14 00 – Facility Storm Drainage
Section 22 30 00 – Plumbing Equipment
Provide subsoil/foundation drainage system where shown on the plans in a configuration for proper installation.
QUALITY ASSURANCE
Use adequate numbers of skilled workmen who are trained and experienced in the necessary crafts and who are familiar with
the specified requirements and the methods needed for proper performance of the work of this Section.
PRODUCT HANDLING
Comply with provisions of Division 01 - General Requirements.
SUBMITTALS
Submit shop drawings on items specified herein.
PART 2 - PRODUCTS
PIPE AND FITTINGS
PVC, perforated, ASTM Specification D2729.
PE, plastic corrugated tubing, drainage and footing type, similar to product by Advanced Drainage System (ADS).
Minimum size of 4" inside diameter.
JOINTS
Plastic PVC pipe, solvent cement ASTM Specification D2564.
Plastic corrugated pipe, plastic snap coupling.

1 FILTER FABRIC

Non-woven polypropylene fiber bonded at crossing points. Equivalent opening size (EOS) shall be 70 to 100 U.S. Standard sieve
 size.

Fabric to be Typar, Style 3401 by DuPont, Mirafi 140 by Celanese, GSI Systems.

STONE

Drainage stone shall be washed rock or gravel evenly graded with stone smaller than 2 inch size and larger than 3/4-inch size.

9 10 CLEANOUTS

11 Refer to Section 22 14 00.

12 13

5 6

7

8

14

15

18

19

21

27

28 29

30

31

32

PART 3 - EXECUTION

16 INSTALLATION17 The subsoil dra

The subsoil drainage system shall be installed as shown on drawings and details. Placement of the drainage stone shall be done to prevent tearing of the filter fabric. The pipe shall be pitched at a grade of 2 inches per 100 feet.

20 The jointing systems for pipe shall be installed to accomplish proper pipe alignment and pitch.

Filter fabric shall be handled and stored according to recommendations of the manufacturer.

24 Only pipe in the stone bedding shall be perforated. Extensions to cleanouts shall be solid pipe material. 25

26 Install filter fabric as an envelope around the pipe and stone as follows:

- Lay fabric in the trench.
- Place stone in the filter fabric.
- Level stone to proper grade and set perforated pipe on stone.
- Place remaining stone on side and top of pipe.
- Wrap filter fabric around and close with a minimum 6-inch lap.

33 The components (fabric, stone, pipe) from the subsoil drain conduit.

34

35 36

1 2	SECTION 22 40 00 PLUMBING FIXTURES			
3 4 5	PART 1 - GENERAL			
6	SCOPE			
7 8	This section includes specification	ons for plumbing fixtures, faucets and trim for this project. Included are the following topics:		
9	PART 1 – GENERAL			
10	Scope			
11	Related Work			
12	Description			
13	Reference Standard	S		
14	Quality Assurance			
15	Submittals			
16				
10 10	General			
10	Manufacturers			
20	Wandlacturers			
20	PART 2 - EXECUTION			
22	Installation			
23				
24	RELATED WORK			
25	Requirements of Division 01 sha	Il govern work under this Section.		
26				
27	Division 11 - Foodservice Equipr	nent		
28	Section 22 05 00 – Common Wo	rk Results for Plumbing		
29	Section 22 05 14 - Plumbing Spe	cialties		
30	Section 22 05 29 – Hangers and	Supports for Plumbing Piping and Equipment		
31	Section 22 11 00 - Facility Water	Distribution		
32	Section 22 13 00 - Facility Sanita	ry Sewerage		
33 21	DESCRIPTION			
25	Furnish and install nlumbing fixt	ures with trans drains stons faucets flush valves carriers and hardware		
36		ares with traps, drams, stops, radeets, nash valves, earners and nardware.		
37	REFERENCE STANDARDS			
38	ANSI A112.6.1M-88	Supports for Off-the Floor Plumbing Fixtures for Public Use.		
39	ANSI A112.18.1-94	Finished and Rough Brass Plumbing Fixture Fittings.		
40	ANSI A112.19.1-90	Enameled Cast Iron Plumbing Fixtures.		
41	ANSI A112.19.2M-82	Vitreous China Plumbing Fixtures.		
42	ANSI A112.19.5-79(R1990)	Trim for Water Closet Bowls, Tanks and Urinals.		
43	ANSI Z124.1-87	Plastic Bathtub Units.		
44	ANSI Z124.2-87	Plastic Shower Receptors and Shower Stalls.		
45	ARI-1010-94	Drinking Fountains and Self-Contained Mechanically Refrigerated Drinking Water Coolers.		
46	ASSE 1011-93	Hose Connection Vacuum Breakers.		
4/	ASSE 1014-90	Handheid Showers.		
48 40	ASSE 1035-93	Laboratory Faucet Backflow Preventers.		
49 50	OUALITY ASSURANCE			
51	Substitution of Materials: Refer	to 22.05.00 and Division 01 of the Project Manual		
52				
53	Plumbing products requiring app	proval by the State of Wisconsin Dept. of Commerce must be approved or have pending approval		
54 55	at the time of shop drawing submission.			
56	SUBMITTALS			
57 58	Submit product data sheets in a	ccordance with Division 01 and Section 22 05 00.		
59 60 61 62 63	Include data concerning sizes, finishes, manufacturer's installa	utility sizes, rough in-dimensions, capacities, materials of construction, ratings, weights, trim, tion requirements, manufacturer's performance limitations, and appropriate identification.		

1	PART 2 - PRODUCTS
2	
3 4 5	GENERAL Fixtures must conform to general requirements given below and to specified requirements for each type.
6 7	Vitreous china fixtures shall conform to ANSI A112.19.2M.
, 8 9	Enameled cast iron fixtures shall conform to ANSI A112.19.1M.
10 11	Stainless steel fixtures shall conform to ANSI A112.19.3.
12 13 14	Fixtures shall be installed so that parts are accessible for repairs when fixtures are in place. Manufacturer's trademark or name shall be visible on fixtures.
15 16 17	Faucets, traps, exposed fittings and trim shall be polished chrome plated unless otherwise specified. Provide polished chrome plated nipples at all lavatories.
18 19 20	Exposed piping penetrating walls, floors or ceilings shall have chrome plated escutcheons, or flanges of sufficient depth to seal the opening.
21 22 23	Fixture stops shall be heavy duty commercial grade, slow compression angle valves with 1/2" inlet and 3/8" or 1/2" chrome plated flexible riser.
24 25	Traps shall be semi-cast 17-gauge brass, chrome plated, with cleanout and escutcheon. Sink traps shall be 1-1/2" minimum.
26	MANUFACTURERS
27 28 29	Vitreous china and enameled cast iron fixtures shall be manufactured by Kohler, Sloan, Toto, or Zurn. Fixture color shall be white unless specified otherwise.
30 31	Flush valves shall be manufactured by Sloan ("G2" series), or Zurn ("Aquasense" series).
32 33 34	Solid plastic toilet seats shall be manufactured by Bemis, Benneke, Centoco, Church, Olsonite, Kohler, or Zurn. Seat color shall match fixture unless specified otherwise.
35 36	Carriers for wall-mounted fixtures shall be manufactured by J.R. Smith, Josam, MIFAB, Wade, Watts, or Zurn.
37 38	Drinking fountains and electric water coolers shall be manufactured by Acorn Aqua, Elkay, Filtrine, Halsey Taylor, Haws, Oasis, or Sunroc.
39 40	Cast terrazzo and molded stone products shall be manufactured by Crane/Fiat, Mustee, or Stern-Williams.
41 42	Stainless steel sinks shall be manufactured by Advance-Tabco, Elkay, or Just.
43 44	Electronic sensor operated faucets shall be manufactured by Sloan, or Zurn.
45 46	Stainless steel shower units shall be manufactured by Acorn, Bradley, Leonard, or Willoughby.
47 48 49	Shower mixing valves and accessories shall be manufactured by American Standard, Chicago Faucet, Kohler, Leonard, Powers, Speakman, Symmons, or Zurn.
50 51	Heavy duty stops and supplies shall be manufactured by Chicago Faucet, Dearborn, EBC, Kohler, McGuire, T&S Brass, or Zurn.
52 53 54	Traps shall be semi-cast 17 gauge brass, chrome plated, with cleanout and escutcheon as manufactured by Dearborn, EBC, Keeney, Kohler, McGuire, or Zurn.
55 56	Supply, drain and trap insulating kits shall be manufactured by Brocar, EBC, McGuire, Plumberex, or Truebro.
57 58	Special traps and solids interceptors shall be manufactured by J.R. Smith, Josam, Wade, Watts, or Zurn.
59 60 61 62 63	Fixtures: See Plumbing Fixture Schedule on drawings for type, manufacturer, and model for fixtures.
PART 3 - EXECUTION

3 INSTALLATION

1

2

4

5

6 7

16

19

25

27

29

Install plumbing fixtures in accordance with manufacturer's instructions. Set level and plumb. Secure in place to counters, floors and walls providing solid bearing and secure mounting. Bolt fixture carriers to floor and wall. Secure rough-in fixture piping to prevent movement of exposed piping.

Install each fixture with trap easily removable for servicing and cleaning. Install fixture stops in readily accessible location for
 servicing. Individual supplies to fixtures shall be provided with support to prevent movement.

Install barrier free fixtures in compliance with COMM 52, 69 and Federal ADA Accessibility Guidelines. Install barrier free lavatory
 traps parallel and adjacent to wall and supplies and stops elevated to avoid contact by wheelchair users.

Seal joints between countertop, wall, floor and fixtures with G.E. Silicone caulk; white, clear or color to match fixture with colored fixture manufacturer.

17 Each fixture shall have a stop valve installation to control the fixture. Stop valves shall be heavy duty type with brass stems and 18 screwed or sweat inlet connections. Compression type inlets are not acceptable.

Cover pipe penetrations with escutcheons. Exposed traps, stops, piping and escutcheons to be chrome plated brass, same items
 in concealed locations may be of rough brass finish.

23 Set floor mounted water closets, floor mounted service sinks; counter mounted lavs and sinks; lav and sink faucets and drains 24 with full setting bed of flexible non-staining plumber's putty. Cover exposed water closet bolts with bolt covers.

- 26 Set mop basins to floor and wall with grout or silicone sealant.
- 28 After installation, fixtures shall be protected to prevent scratching or other damage during construction.

30 Prior to acceptance, fixtures shall be cleaned with compounds recommended by the respective manufacturer.

31

32 33

Page Intentionally Left Blank

1		SECTION 22 42 23			
2		COMMERCIAL SHOWERS			
3	PART 1 - 0	RT 1 - GENERAL			
4	1.1	SUMMARY			
5	Α.	Section Includes:			
6		1. Shower basins.			
7	1.2	ACTION SUBMITTALS			
8	Α.	Product Data: For each type of product.			
9		1. Include construction details, material descriptions, dimensions of individual components and profiles, and			
10		finishes for shower basins.			
11		2. Include rated capacities, operating characteristics, and furnished specialties and accessories.			
12	1.3	CLOSEOUT SUBMITTALS			
13	А.	Maintenance Data: For cleaning shower basins.			
14	PART 2 - F	PRODUCTS			
15	2.1	PRECAST-TERRAZZO SHOWER BASINS			
16	А.	Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be			
17		incorporated into the Work include, but are not limited to the following:			
18		a. Acorn Engineering			
19		b. Florestone Products Co., Inc.			
20		c. Stern-Williams Co., Inc.			
21		d. Fiat Products.			
22		 Source Limitations: Obtain shower basins from single source from single manufacturer. 			
23		 Description: Precast-terrazzo base for built-up-type shower fixture. 			
24		4 Standard			
25		a. CSA B45.8/IAPMO 7403.			
26		b. IAPMO PS 99.			
27		c. 2010 ADA Standards for Accessible Design.			
28		5 Threshold Type: Standard commercial and/or Handicanned/accessible – see Drawings			
29		6 Shane: See Drawings			
30		7 Nominal Size: See Drawings			
31		8 Color: Color selected by Architect from manufacturer's standard color selection			
32		9 Outlet: Removable stainless steel grid strainer. NPS 2 no-caulk drain			
32		10 Tiling Flange: Integral stainless steel			
34		a Square or rectangular fixture: Provide for all walls – See Drawings			
35		h Corner Fixture: Left and right			
36		11 Rahbeted Joint Shoulders: 1 in high			
37		$12 \qquad \Delta cressories:$			
38		a Rubber drain gasket reducer for NPS 2 waste nine			
30		h Vandal resistant strainer			
40		c Double drain			
40 //1	2.2	GROUIT			
41 //2	Δ	Standard: ASTM C1107/C1107M Grade B post-bardening and volume-adjusting dry hydraulic-cement grout			
42 //3	R.	Characteristics: Nonshrink: recommended for interior and exterior annlications			
45 AA	D. C	Design Mix: 5000 pci - 28-day compressive strength			
44	с. D	Design Ivila. Soud psi, 28-day compressive strength. Rackaging: Bromized and factory nackaged			
45	DART 2 - F	D. Packagnig, Premixeu anu lactory packageu.			
40	2 1	FYAMINATION			
47	J.I ^	Examine rough-in of water-supply and capitary drainage and yent piping systems to verify actual locations of piping			
4 0 ДО	Π.	connections before shower installation			
50	R	Examine walls and floors for suitable conditions where showers will be installed			
51	р. С	Proceed with installation only after unsatisfactory conditions have been corrected			
52	32				
52	J.2 ^	INFIGURATION A Assemble shower basin components in accordance with manufacturers' written instructions			
54	д. В	Install shower basin components in accordance with manufacturers, written instructions.			
55	ь. С	Set shower basins in leveling hed of coment grout			
55	с. р	Seel joints between showers and floors and walls using capitary type, one part, mildow resistant silicone scalant			
50	D. Jean joints between showers and hours and wans using samilary-type, Une-part, innuew-resistant Sinconte Sediant. Match sealant color to fixture color. Comply with scalant requirements specified in Section 07.02.00 "Joint				
57		Solante"			
20		Jealants.			

- / - / -		
3.3	PIPING CONNECTIONS	
Α.	Connect fixtures with water supplies, stops, and risers, and with traps, soil, waste, and vent piping. Use size fittings	
	required to match fixtures.	
3.4	CLEANING AND PROTECTION	
Α.	After completing installation of shower basins, inspect and repair damaged finishes.	
В.	Clean shower basins and fittings with manufacturers' recommended cleaning methods and materials.	
С.	Provide protective covering for installed fixtures and fittings.	
D.	Do not allow use of showers for temporary facilities unless approved in writing by Owner.	
	END OF SECTION	
	3.3 A. 3.4 A. B. C. D.	

1	SECTION 230500			
2	COMMON WORK RESULTS FOR HVAC			
3				
4				
5	PART 1 - GENERAL			
6				
7	SCOPE			
8	This section includes information common to two or more technical specification sections or items that are of a general nature,			
9	not conveniently fitting into other technical sections. Included are the following topics:			
10				
11	PART 1 - GENERAL			
12	Scope			
13	Related Work			
14	Reference Standards			
15	LEED Certification			
16	Commissioning			
17	Quality Assurance			
18	Abbreviations			
19	Definitions			
20	Drawings			
21	Codes and Standards			
22	Continuity of Existing Services			
23	Protection of Finished Surfaces			
24	Sleeves and Openings			
25	Sealing and Firestopping			
26	Equipment Furnished By Others			
27	Provisions for Future			
28	Add Alternate Bid			
29	Submittals			
30	Specified Materials and Equipment			
31	Equipment installation			
32	Off Site Storage			
33	Certificates and inspections			
34 25	Operating and Maintenance instructions			
35				
30 27				
57 20	Netoru Drawings			
38				
39 40	Access Bands and Doors			
40	Access Failers and Doors			
41 42				
42 13	Sealing and Filestopping			
43				
44 15	Demolition			
45	Concrete Work			
40	Cutting and Patching			
47 //8	Building Across			
40 49	Faultiment Access			
50	Coordination			
50	Identification			
52				
53				
54	Sealing and Eirestopping			
55				
56	RELATED WORK			
57	Section 23 05 13 - Common Motor Requirements for HVAC.			
58	Section 23 33 00 - Air Duct Accessories.			
59				
60	Applicable provisions of Division 1 govern work under this section.			
61	rr b			
62	REFERENCE STANDARDS			
63	Abbreviations of standards organizations referenced in other sections are as follows:			

T		
2	AABC	Associated Air Balance Council
3	ADC	Air Diffusion Council
4	AGA	American Gas Association
5	AMCA	Air Movement and Control Association
6	ANSI	American National Standards Institute
7	ARI	Air-Conditioning and Refrigeration Institute
8	ASHRAE	American Society of Heating, Refrigerating and Air Conditioning Engineers
9	ASME	American Society of Mechanical Engineers
10	ASTM	American Society for Testing and Materials
11	CGA	Compressed Gas Association
12	IEEE	Institute of Electrical and Electronics Engineers
13	ISA	Instrument Society of America
14	MCA	Mechanical Contractors Association
15	MICA	Midwest Insulation Contractors Association
16	MSS	Manufacturer's Standardization Society of the Valve & Fitting Industry, Inc.
17	NBS	National Bureau of Standards
18	NEBB	National Environmental Balancing Bureau
19	NEC	National Electric Code
20	NEMA	National Electrical Manufacturers Association
21	NFPA	National Fire Protection Association
22	SMACNA	Sheet Metal and Air Conditioning Contractors' National Association. Inc.
23	UL	Underwriters Laboratories Inc.
24	ASTM E814	Standard Test Method for Fire Tests of Through-Penetration Fire Stops
25	ASTM E84	Standard Test Method for Surface Burning Characteristics of Building Materials
26	UL1479	Fire Tests of Through-Penetration Firestops
27	UL723	Surface Burning Characteristics of Building Materials
28		
29	LEED CERTIFICAT	ION
30	The project will b	e LEED Certified thru the United States Green Building Council's (USGBC) Leadership in Energy and
31	Environmental D	esign (LEED) program. Refer to Section 01 81 13 – Sustainable Design Requirements for additional
32	requirements.	
22		
3/	In addition to co	mnlying with Division 23 drawings and specifications, equipment and material shall also comply with Section 01
25	81 13 and I FED r	aniframente
33	OT TO GIIU LEED I	equirents.
30		

The Division 23 contractor will be expected to provide all required documentation, submittals, etc. in accordance with prerequisites and credits associated with Division 23 work and LEED Certification.

40 COMMISSIONING

39

44

The systems will be commissioned by an independent third party in accordance with USGBC LEED Energy and Atmosphere
 Credit C3 – Enhanced Commissioning. Refer to Sections 01 91 00 – Commissioning and 01 95 01 – Monitoring-Based
 Commissioning for additional requirements.

45 The Division 23 contractor shall work with the independent third party commissioning agent to complete the required 46 construction verification checklists and assist in the functional performance testing as specified in Section 01 91 00. 47

48 QUALITY ASSURANCE

49 Refer to Division 1, General Conditions, Equals and Substitutions.50

51 Where equipment or accessories are used which differ in arrangement, configuration, dimensions, ratings, or engineering 52 parameters from those indicated on the contract documents, the contractor is responsible for all costs involved in integrating 53 the equipment or accessories into the system and for obtaining the performance from the system into which these items are 54 placed. This may include changes found necessary during the testing, adjusting, and balancing phase of the project. 55

56 ABBREVIATIONS

- 57 A/E Architect/Engineer
- 58 GC General Contractor
- 59 FPC Fire Protection Contractor
- 60 PC Plumbing Contractor
- 61 HC Heating Contractor

- 1 EC Electrical Contractor
- 2 TCC Temperature Contractor
- 3 DDC Direct Digital Controls
- 4 BAS Building Automation System
- 5 TCS Temperature Control System 6

7 DEFINITIONS

- 8 Furnish:
- 9 Supply and deliver to Project site ready for unpacking, assembly and installation

10 11 Install:

- 12 Operations at Site including unpacking, assembling, erecting, placing, anchoring, applying, finishing, cleaning, and connecting 13 related devices required for product fully functional for intended use after installation.
- 14

17

- 15 Provide:
- 16 Furnish and install, such that product is fully functional for intended use.

18 DRAWINGS

Drawings show general arrangements of piping, equipment and appurtenances and shall be followed as closely as actual building construction and work of other trades permits. Work shall conform to requirements shown on Drawings. General and structural drawings shall take precedence. Because of the scale of Drawings, it is not possible to indicate all offsets, fittings and accessories required. Investigate structural and finish conditions affecting work and arrange work accordingly, providing offsets, fittings and accessories required to meet constructed conditions.

24

27

45

HVAC equipment and systems, including piping and ductwork shall be installed as high as possible unless otherwise noted on
 Drawings. Equipment and systems shall also be installed to maintain required operation and maintenance clearances.

28 CODES AND STANDARDS

29 Materials and workmanship shall comply with applicable codes, specifications, local ordinances, industry standards and utility 30 company regulations. In case of differences between building codes, specifications, state laws, local ordinances, industry 31 standards and utility company regulations and contract documents, the most stringent shall govern. Promptly notify A/E in 32 writing of differences.

3334 Non-Compliance:

If Contractor installs materials or performs Work that does not comply with above requirements, he shall correct Work and shall bear costs arising from correcting deficiencies.

38 CONTINUITY OF EXISTING SERVICES

Do not interrupt or change existing services without prior written approval from County Facilities Personnel. When interruption
 is required, coordinate the down-time with Facilities to minimize disruption to their activities. Unless specifically stated, all
 work involved in interrupting or changing existing services is to be done during normal working hours.

43 PROTECTION OF FINISHED SURFACES

44 Refer to Division 1, General Requirements, Protection of Installed Construction

Furnish one can of touch-up paint for each different color factory finish which is to be the final finished surface of the product.
 Deliver touch-up paint with other "loose and detachable parts" as covered in the General Requirements.

4849 SLEEVES AND OPENINGS

50 Refer to Division 1, General Requirements, Cutting and Patching.

5152 SEALING AND FIRESTOPPING

53 Sealing and firestopping of sleeves/openings between ductwork, piping, etc. and the sleeve, structural or partition opening shall 54 be the responsibility of the contractor whose work penetrates the opening. The contractor responsible shall hire individuals 55 skilled in such work to do the sealing and fireproofing. These individuals hired shall normally and routinely be employed in the 56 sealing and fireproofing occupation.

57 58 EQUIPMENT FURNISHED BY OTHERS

- 59 None.
- 60

61 PROVISIONS FOR FUTURE

- 62 (2) future 4" connections shall be provided for future geothermal borefield expansion.
- 63

ALTERNATE BID #2

1

2

3

4 5

6

7 8

9

11

12

13 14

15

16

17

18

19 20

21

25

31

42

51

57

59

60

Alternate Bid #2 shall be provided to evaluate the costs associated with providing the building heating load entirely through the heat recovery chiller and relying on the existing boiler as emergency backup.

Remove (1) existing boiler complete, convert the boiler piping from variable primary to a primary / secondary configuration, replace existing boiler isolation valve with inline circulating pump.

Refer to Drawing M502 for additional information regarding Alternate Bid #2.

10 SUBMITTALS

Refer to Division 1, General Conditions, Submittals.

Submit for all equipment and systems as indicated in the respective specification sections, marking each submittal with that specification section number. Mark general catalog sheets and drawings to indicate specific items being submitted and proper identification of equipment by name and/or number, as indicated in the contract documents.

Before submitting electrically powered equipment, verify that the electrical power and control requirements for the equipment are in agreement with the motor starter schedule on the electrical drawings. Include a statement on the shop drawing transmittal to the architect/engineer that the equipment submitted and the motor starter schedule are in agreement or indicate any discrepancies.

Include wiring diagrams of electrically powered equipment.

24 Provide electronic (PDF) copies of shop drawings for electronic distribution.

26 SPECIFIED MATERIALS AND EQUIPMENT

Design is based on equipment specified by manufacturer and model number as specified on Drawing schedules. Where certain items are specified by manufacturer or trade name, Contractor's bid shall be based on use of named item. Where one (1) manufacturer/model is described and other makes are listed, comparable models of other named equipment may also be used, provided they meet requirements of Specifications.

When equipment or accessories used differ in arrangement, configuration, dimensions, ratings, or engineering parameters from those on Drawing schedules, Contractor shall be responsible for costs involved in integrating equipment or accessories into system. Contractor shall be responsible for obtaining original design performance from system into which items are placed, regardless of whether manufacturer/model is specified equivalent or substitute. This may include changes found necessary during testing, adjusting, and balancing phase of Project.

If Contractor wishes to use items other than those named in Specifications in base bid, request for approval of substitution must be made in writing to A/E at least 14 days prior to opening of bids. Include complete technical and descriptive data with request. If approved, an Addendum will be issued notifying bidders of approval. Request for approval will be considered only if requested by prime bidding Contractor.

43 EQUIPMENT INSTALLATION

Drawings show general arrangement and location of equipment and appurtenances. It is Contractor's responsibility to install equipment in a location and manner that allows for proper service and maintenance access to equipment. Work shall generally conform to requirements shown on Drawings. However, location of equipment may require field adjustments to obtain required service space. DO NOT SCALE OFF PLANS to determine proper location of equipment. Because of scale of Drawings, it is not possible to indicate exact routing of ductwork and piping, and offsets, fittings and accessories required to provide proper service access to equipment. Contractor shall route and install ductwork and piping to provide required service access to equipment.

If, during construction phase of Project, contractor feels inadequate space exists, or equipment locations must be substantially modified to provide proper service and maintenance access, prior to installing equipment, contractor shall notify engineer in writing, outlining general concerns and proposed modifications. Equipment installed without providing manufacturer's required maintenance and service clearance shall be considered defective. Contractor shall remove and relocate piping, ductwork and equipment, to provide required service clearances at contractor's expense.

58 OFF SITE STORAGE

Ductwork, metal for making ductwork, duct lining, sleeves, pipe/pipe fittings and similar rough-in material will not be accepted for off site storage. For material that can be stored off site, no material will be accepted for offsite storage unless shop drawings for that material have been approved.

1	CERTIFICATES AND INSPECTIONS		
2	Refer also to Division 1, General Requirements, Regulatory Requirements.		
3	Obtain and new few all very ived City of Madican installation inspatience support these previded by the Ausbitast/Engineer in		
4	Obtain and pay for an required City of Madison installation inspections except those provided by the Architect/reigneer in		
5	actionalities with wis Auth code section LERK 30.12. Deliver ofiginals of these certificates to the Owner and A/E. Include copies		
7	of the certificates in the Operating and Maintenance Instructions.		
8	OPERATING AND MAINTENANCE INSTRUCTIONS		
9	All operations and maintenance data shall comply with the submission and content requirements specified under this section		
10	and under Division 1, General Requirements, Closeout Procedures.		
11	Assemble material in three-ring or post binders, using an index at the front of each volume and tabs for each system or type of		
12	equipment. In addition to the data indicated in the General Requirements, include the following information:		
13			
14	 Copies of all approved shop drawings. 		
15	 Manufacturer's wiring diagrams for electrically powered equipment 		
16	 Records of tests performed to certify compliance with system requirements 		
17	Certificates of inspection by regulatory agencies		
18	Temperature control record drawings and control sequences		
19	Parts lists for manufactured equipment		
20	Valve schedules		
21	 Lubrication instructions, including list/frequency of lubrication done during construction 		
22	 Warranties Additional information as indicated in the technical specification sections 		
23	Additional mormation as indicated in the technical specification sections		
24	Provide three (3) hardronies of the Operation and Maintenance Manual. Manuals shall be organized in three ring hinders with		
25	dividers and reference take. Manuals shall be delivered as follows:		
20	One convict to the building engineer		
27	 One copy to the tenant (to be kent on site) 		
29	One copy to the Owners Representative		
30			
31	Provide (3) electronic (Adobe PDF) copies of the Operation and Maintenance Manual.		
32	 Provide each copy on a separate portable USB flash drive. 		
33	 Deliver each portable USB flash drive with hard copy manuals to parties listed above. 		
34			
35	CLOSEOUT PROCEDURES		
36	Refer to this section and under Division 1, General Requirements, Closeout Procedures.		
37			
38	The Contractor shall complete and provide items and materials, training and start-up associated with project closeout as specified		
39	under Division 1 of the Project Manual. In addition to these items, the Contractor shall provide the following items prior to		
40	acceptance of the installation as specified in accordance with with 2009 IMC 403.7 and 2009 IECC 503.2.9.1 through 503.2.9.3:		
41			
42	 Final air and water system balancing, completed in accordance with the requirements of Section 23 05 93 		
43	and code, including the submission of testing, adjusting and balancing reports. Reports shall indicate the		
44	amount of total supply air, return air and outside ventilation air being provided to the spaces and to the air		
45	handling system(s).		
46			
47	• Submission of Operating and Maintenance instructions in accordance with the requirements of Division 1,		
48	this Section, and code. Operation and Maintenance Manuals shall include a copy of completed testing.		
49	adjusting and balancing report for Owner's records.		
50			
51	Submission of start-up report for temperature control system, signed by technician in responsible charge of		
52	control system indicating system has been adjusted, calibrated and put into operation in accordance with		
52	control system, indicating system has been adjusted, campiated and put into operation in accordance with		
55 57	ובקטוופווופוונג טו שבנוטווג בש טש ד4, בש טש בש, בש טש שש מווע נטעפ.		
54 55			
55 56	Individed of Owner Penjoinnel Instruct County Eacility Desconded in the proper operation and maintenance of systems and equipment provided as part of this		
50 57	notice county recipient in the proper operation and maintenance of systems and equipment provided as part of time		
58	project, video tape an inaming sessions. Include not less than 4 hours of instruction, using the operating and Maintenance manuals during this instruction. Demonstrate startun and shutdown procedures for all equipment		
59	המוזממום ממוזהה נוום ווזכו מכנוסה. שבוויסווזכו מנב זכמו נמף מות המונמטשון פוטכבממובה וסו מו בקמוףוויבוונ.		
60	All training to be during normal working hours.		
61			

RECORD DRAWINGS

1

2

3 4

5

6

7 8

9

10

15 16

17

22

28

37

Refer to Division 1, General Requirements, Record Drawings.

In addition to the data indicated in the General Requirements, maintain temperature control record drawings on originals prepared by the installing contractor/subcontractor. Include copies of these record drawings with the Operating and Maintenance manuals.

Maintain accurate as-built or record drawings throughout the duration of the project. As-built drawings shall be available on site at all times for review by the A/E, owner or owner's representative.

If, during project closeout, the A/E or owner observes installations that are not accurately recorded on the as-built or record drawings, the record drawings will not be accepted and the contractor will be required, at their own expense, to provide updated and accurate record drawings.

PART2 - PRODUCTS

18 ACCESS PANELS AND DOORS

19 LAY-IN CEILINGS:

Removable lay-in ceiling tiles in 2 X 2 foot or 2 X 4 foot configuration provided under Section 09500 are sufficient; no additional access provisions are required unless specifically indicated.

23 PLASTER WALLS AND CEILINGS:

16 gauge frame with not less than a 20 gauge hinged door panel, prime coated steel for general applications, stainless steel for use in toilets, showers, and similar wet areas, concealed hinges, screwdriver operated cam latch for general applications, key lock for use in public areas, UL listed for use in fire rated partitions if required by the application. Use the largest size access opening possible, consistent with the space and the equipment needing service; minimum size is 12" by 12".

29 **IDENTIFICATION**

30 STENCILS:

31 Not less than 1 inch high letters/numbers for marking pipe and equipment.

3233 SNAP-ON PIPE MARKERS:

Cylindrical self-coiling plastic sheet that snaps over piping insulation and is held tightly in place without the use of adhesive,
 tape or straps. Not less than 1 inch high letters/numbers and flow direction arrows for piping marking. W. H. Brady, Seton,
 Marking Services, or equal.

38 ENGRAVED NAME PLATES:

39 White letters on a black background, 1/16 inch thick plastic laminate, beveled edges, screw mounting, Setonply Style 2060 by 40 Seton Name Plate Company or Emedolite- Style EIP by EMED Co., or equal by Marking Services, or W. H. Brady.

41 42 VALVE TAGS:

Round brass tags with 1/2 inch numbers, 1/4 inch system identification abbreviation, 1-1/4 inch minimum diameter, with brass
 jack chains or brass "S" hooks around the valve stem, available from EMED Co., Seton Name Plate Company, Marking Services,
 or W. H. Brady.

46 47 SEALING AND FIRESTOPPING

48 FIRE AND/OR SMOKE RATED PENETRATIONS:

49 Manufacturers:

50 3M, Hilti, Rectorseal, STI/SpecSeal, Tremco, or approved equal.

- 52 All firestopping systems shall be provided by the same manufacturer.
- 53 54 Submittals:

Contractor shall submit product data for each firestop system. Submittals shall include product characteristics, performance and limitation criteria, test data, MSDS sheets, installation details and procedures for each method of installation applicable to this project. For non-standard conditions where no UL tested system exists, submit manufacturer's drawings for UL system with known performance for which an engineering judgement can be based upon.

60 Product:

Fire stop systems shall be UL listed or tested by an independent testing laboratory approved by the Department of Commerce.

1 Use a product that has a rating not less than the rating of the wall or floor being penetrated. Reference architectural drawings 2 for identification of fire and/or smoke rated walls and floors.

3

Contractor shall use firestop putty, caulk sealant, intumescent wrapstrips, intumescent firestop collars, firestop blocks, firestop
 mortar or a combination of these products to provide a UL listed system for each application required for this project. Provide
 mineral wool backing where specified in manufacturer's application detail.

78 NON-RATED PENETRATIONS:

9 Pipe Penetrations Through Below Grade Walls:

10 In exterior wall openings below grade, use a modular mechanical type seal consisting of interlocking synthetic rubber links 11 shaped to continuously fill the annular space between the uninsulated pipe and the cored opening or a water-stop type wall 12 sleeve.

- 13
- 14 Pipe Penetrations:

At pipe penetrations of non-rated interior partitions, floors and exterior walls above grade, use urethane caulk in annular space between pipe insulation and sleeve. For non-rated drywall, plaster or wood partitions where sleeve is not required use urethane caulk in annular space between pipe insulation and wall material.

1819 Duct Penetrations:

Pack annular space with fiberglass batt insulation or mineral wool insulation. Provide 4" sheet metal escutcheon around duct
 on both sides of partition or floor to cover annular space.

PART 3 - EXECUTION

25 DEMOLITION

Perform all demolition as indicated on the drawings to accomplish new work. Where demolition work is to be performed adjacent to existing work that remains in an occupied area, construct temporary dust partition to minimize the amount of contamination of the occupied space. Where pipe or duct is removed and not reconnected with new work, cap ends of existing services as if they were new work. Coordinate work with the user agency to minimize disruption to the existing building occupants.

31

23

24

All pipe, wiring and associated conduit, insulation, ductwork, and similar items demolished, abandoned, or deactivated are to be removed from the site by the Contractor. All piping and ductwork specialties are to be removed from the site by the Contractor unless they are dismantled and removed or stored by the user agency. All designated equipment is to be turned over to the user agency for their use at a place and time so designated. Maintain the condition of material and/or equipment that is indicated to be reused equal to that existing before work began.

37

38 CONCRETE WORK

All cast-in-place concrete will be performed by the General Contractor unless otherwise noted. Provide all layout drawings, anchor bolts, metal shapes, and/or templates required to be cast into concrete or used to form concrete for support of mechanical equipment.

42

43 CUTTING AND PATCHING

44 Refer to Division 1, General Requirements, Cutting and Patching.

45

46 BUILDING ACCESS

Arrange for the necessary openings in the building to allow for admittance of all apparatus. When the building access was not
 previously arranged and must be provided by this contractor, restore any opening to its original condition after the apparatus
 has been brought into the building.

50

51 EQUIPMENT ACCESS

Install all piping, conduit, ductwork, and accessories to permit access to equipment for maintenance and service. Coordinate the exact location of wall and ceiling access panels and doors with the General Contractor, making sure that access is available for all equipment and specialties. Access doors in general construction are to be furnished by the Mechanical Contractor and installed by the General Contractor.

56

Provide color coded thumb tacks or screws, depending on the surface, for use in accessible ceilings which do not require accesspanels.

59

60 COORDINATION

61 Verify that all devices are compatible for the surfaces on which they will be used. This includes, but is not limited to, diffusers,

62 register, grilles, and recessed or semi-recessed heating and/or cooling terminal units installed in/on architectural surfaces.

2

3 4

5

6

7

8

9

10

11

12 13

14

15

16 17

18

19 20 21

22 23

24 25 26

27

28

29

35

37

Coordinate all work with other contractors prior to installation. Any installed work that is not coordinated and that interferes with other contractor's work shall be removed or relocated at the installing contractor's expense.

Cooperate with the test and balance agency in ensuring Section 23 05 93 specification compliance.

Verify system completion to the test and balance agency (flushing, pressure testing, chemical treatment, filling of liquid systems, proper pressurization and air venting of hydronic systems, clean filters, clean strainers, duct and pipe systems cleaned, controls adjusted and calibrated, controls cycled through their sequences, etc.), ready for testing, adjusting and balancing work. Install dampers, shutoff and balancing valves, flow measuring devices, gauges, temperature controls, etc., required for functional and balanced systems. Demonstrate the starting, interlocking and control features of each system so the test and balance agency can perform its work.

IDENTIFICATION

Identify equipment in mechanical equipment rooms by stenciling equipment number and service with one coat of black enamel against a light background or white enamel against a dark background. Use a primer where necessary for proper paint adhesion. Do not label equipment such as cabinet heaters and ceiling fans in occupied spaces.

• Air terminal units to be labeled on bottom and side of each unit.

Where stenciling is not appropriate for equipment identification, engraved name plates may be used.

Identify piping not less than once every 20 feet, not less than once in each room, adjacent to each access door or panel, and on both side of the partition where exposed piping passes through walls, floors or roofs. Use one coat of black enamel against a light background or white enamel against a dark background for stenciling, or provide snap-on pipe markers as specified in Part 2 – Products. At each location, pipe identification shall include:

- Pipe content (HWS, HWR, CWS, CWR, CCC, CS, CR etc.).
- Flow directional arrows.
- Pipe size.

Identify valves with brass tags bearing a system identification and a valve sequence number. Valve tags are not required at a terminal device unless the valves are greater than ten feet from the device or located in another room not visible from the terminal unit. Provide a typewritten valve schedule indicating the valve number and the equipment or areas supplied by each valve; locate schedules in each mechanical room and in each Operating and Maintenance manual. Schedules in mechanical rooms to be framed under clear plastic.

36 Use engraved name plates to identify control equipment.

Place color coded thumbtack (or equivalent) on ceiling grid at locations of above ceiling air terminals, BAS DDC control equipment and control valves.

40 41 LUBRICATION

Lubricate all bearings with lubricant as recommended by the manufacturer before the equipment is operated for any reason. Once the equipment has been run, maintain lubrication in accordance with the manufacturer's instructions until the work is accepted by DFD. Maintain a log of all lubricants used and frequency of lubrication; include this information in the Operating and Maintenance Manuals at the completion of the project.

46 47 **SLEEVES**

48 PIPE SLEEVES:

49 Provide galvanized sheet metal sleeves for pipe penetrations through interior and exterior walls to provide a backing for sealant 50 or firestopping. Patch wall around sleeve to match adjacent wall construction and finish. Grout area around sleeve in masonry 51 construction. In finished spaces where pipe penetration through wall is exposed to view, sheet metal sleeve shall be installed 52 flush with face of wall. 53

Pipe sleeves are not required in interior non-rated drywall, plaster or wood partitions and sleeves are not required in existing poured concrete walls where penetrations are core drilled.

Pipe sleeves are not required in cored floor pipe penetrations through existing floors that are not located in mechanical rooms, food service areas or wet locations listed above.

Pipe sleeves in new poured concrete construction shall be schedule 40 steel pipe (sized to allow insulated pipe to run through sleeve), cast in place.

54

55

56 57

58

- Extend the top of sleeve 1 inch above the adjacent floor in piping floor penetrations located in the upper floor mechanical room
 and any first floor penetration. In finished areas sleeves shall be flush with rough floor.
- In exterior wall openings below grade, place water-stop type wall sleeve before concrete pour or core drill opening after pour.
 Assemble rubber links to proper size for pipe and tighten in place in accordance with manufacturer's instructions.
- 6 7 DUCT SLEEVES:

13

14 15

- 8 Duct sleeves are not required in non-rated partitions or floors.
- 10 PENETRATIONS SUBJECT TO WATER INTRUSION
- For penetrations (both rated and non-rated) in floors subject to water intrusion or in rooms housing electrical equipment (but not within walls) provide one of the following:
 - Pipe penetration where steel pipe sleeve is used extend steel sleeve 2" above the floor.
 - Pipe penetration where cast in place fire stopping device/sleeve is used, extend device/sleeve 2" above the floor (provided it meets the device's UL listing).
- Pipe penetration where there is no steel sleeve or cast in place fire stopping device/sleeve, provide 2"x 2" x 1/8"
 galvanized steel angles fastened to floor surrounding the penetration or group of penetrations to prevent water from
 getting to penetration. Provide urethane caulk between angles and floor and fasten angles to floor minimum 8" on
 center. Seal corners water tight with urethane caulk.
- Duct penetrations. Provide 2"x 2" x 1/8" galvanized steel angles fastened to floor surrounding the penetration or group of penetrations to prevent water from getting to penetration. Provide urethane caulk between angles and floor and fasten angles to floor minimum 8" on center. Seal corners water tight with urethane caulk.

24 Floors subject to water intrusion or rooms housing electrical equipment include the following locations:

- Mechanical/Plumbing Equipment Rooms
- Restrooms
 - Locker/Shower Rooms
 - Janitor Rooms w/ Sinks
- 28 29

23

25

26

27

30 SEALING AND FIRESTOPPING

31 NON-RATED PARTITIONS:

At all interior partitions and exterior walls, pipe penetrations are required to be sealed. Apply sealant to both sides of the penetration in such a manner that the annular space between the pipe sleeve or cored opening and the pipe or insulation is completely blocked.

35

Duct penetrations through non-rated partitions shall require sheet metal escutcheons with fiberglass or mineral wool insulation
 fill for spaces that include janitor closets, toilet rooms, mechanical rooms, conference rooms, private offices, and where noted
 on drawings elsewhere.

- 39
- 40

41

Page Intentionally Left Blank

1		SECTION 23 05 13		
2	COMMON MOTOR REQUIREMENTS FOR HVAC EQUIPMENT			
3				
4				
5		PART1-GENERAL		
6				
7	SCOPE			
8	This section includes rec	juirements for single and three phase motors that are used with equipment specified in other sections.		
9	Included are the following	ıg topics:		
10				
11	PART 1 - GENERAL			
12	Scope			
13	Related Work			
14	Reference			
15	Reference Star	ndards		
16	Quality Assura	nce		
17	Shop Drawings	i		
18	Operating and	Maintenance Data		
19	Electrical Coor	dination		
20	Product Criteri	a		
21				
22	PART 2 - PRODUCTS			
23	Three Phase, S	ingle Speed Motors		
24	Single Phase, S	ingle Speed Motors		
25	Motors Used o	n Variable Frequency Drives		
26				
27	PART 3 - EXECUTION			
28	Installation			
29				
30	RELATED WORK			
31	Section 23 05 14 - Variat	Jie Frequency Drives		
32	Section 23 09 14 - Electri	ic Instrumentation and Control Devices for HVAC		
33	Section 23 09 23 - Direct	Digital Control System for HVAC		
34 25	Section 23 21 23 – Hydro	Shic Pumps		
35	Section 23 34 00 – HVAC	rans		
30	Section 23 64 23 – Modular Heat Recovery Chillers			
3/	Section 23 73 13 – Wool	ilar indoor Central station Air Handling Units		
38 20	Section 23 82 00 – Heati	ng and Cooling Terminal Units		
39	Division 26.00.00 Flast	tion l		
40	Division 26 00 00 - Electr			
41 42	DEFEDENCE			
4Z 42	Applicable provisions of	Division 1 source work under this section		
43	Applicable provisions of	Division 1 govern work under this section.		
44 45				
45 46	ANSI/JEEE 112) Test Breedure for Debusiness Induction Meters and Consisters		
40 47		Notes and Consistence		
47		Notors and Generators		
48	ANSI/NFPA 70	National Electrical Code		
49 FO	OLIALITY ASSUDANCE			
50 F 1	QUALITY ASSURANCE			
51 51	Also refer to division 1.			
52 E 2				
55 E /	Befer to Division 1 Cons	scal Conditions Submittals		
54	Kelel to Division 1, Gele			
55	Include with the equip	ment which the motor drives the following motor information: motor manufacturer, bersonower		
50	voltage phase bortz rp	full load officional include the following motor monitation. In the contractor constitution for the		
52	work	in, ian load enciency. Include project withing diagrams prepared by the contractor specifically for this		
50	WOIN.			
59		ΓΕΝΑΝζΕ ΠΑΤΑ		
61	All operations and main	ntenance data shall comply with the submission and contant requirements specified under section		
62		remance data shan comply with the submission and content requirements specified dider section		
63				
55				

2

3

4

5 6

7

8

9

10

11 12 13

14 15

16 17

19 20

21 22

23

24 25

26 27

28 29

31 32 33

34 35

36

37 38

39 40

41 42

43 44

45

46

ELECTRICAL COORDINATION

All starters, overload relay heater coils, disconnect switches and fuses, relays, wire, conduit, pushbuttons, pilot lights, and other devices required for the control of motors or electrical equipment are furnished and installed by the Electrical Contractor. except as specifically noted elsewhere in this division of specifications or on the mechanical drawings.

Electrical drawings and/or specifications show number and horsepower rating of all motors furnished by this Contractor, together with their actuating devices if these devices are furnished by the Electrical Contractor. Should any discrepancy in size, horsepower rating, electrical characteristics or means of control be found for any motor or other electrical equipment after contracts are awarded, Contractor is to immediately notify the architect/engineer of such discrepancy. Costs involved in any changes required due to equipment substitutions initiated by this contractor will be the responsibility of this contractor. See related comments in Section 23 05 00 - Common Work Results for HVAC, under Shop Drawings.

Electrical Contractor will provide all power wiring and control wiring, except temperature control wiring.

Furnish project specific wiring diagrams to Electrical Contractor for all equipment and devices furnished by this Contractor and indicated to be wired by the Electrical Contractor.

18 PRODUCT CRITERIA

Motors to conform to all applicable requirements of NEMA, IEEE, ANSI, and NEC standards and shall be listed by U.L. for the service specified.

Select motors for conditions in which they will be required to perform; i.e., general purpose, splashproof, explosion proof, standard duty, high torque or any other special type as required by the equipment or motor manufacturer's recommendations.

Furnish motors for starting in accordance with utility requirements and compatible with starters as specified.

PART 2 - PRODUCTS

THREE PHASE. SINGLE SPEED MOTORS 30

Use NEMA rated 460 volt, three phase, 60 hertz motors for all motors 1/2 HP and larger unless specifically indicated.

Use NEMA general purpose, continuous duty, Design B, normal starting torque, T-frame or U-frame motors with Class B or better insulation unless the manufacturer of the equipment on which the motor is being used has different requirements. Use open drip-proof motors unless totally enclosed fan-cooled, totally enclosed non-ventilated, explosion-proof, or encapsulated motors are specified in the equipment sections.

Use grease lubricated anti-friction ball bearings with housings equipped with plugged/capped provision for relubrication, rated for minimum AFBMA 9, L-10 life of 20,000 hours. Calculate bearing load with NEMA minimum V-belt pulley with belt center line at the end of NEMA standard shaft extension. Stamp bearing sizes on nameplate.

All open drip-proof motors to have a 1.15 service factor. Other motor types may have minimum 1.0 service factors.

All motors 1 HP and larger, except specially wound motors and inline pump motors 56 frame and smaller, to be high efficiency design with full load efficiencies which meet or exceed the values listed below when tested in accordance with NEMA MG 1.

FULL LOAD NOMINAL MOTOR FEELCIENCY BY MOTOR SIZE AND SPEED

47	FULL LOAD	NOMINAL MOTOR EFF	ICIENCY BY MOTOR	SIZE AND SPEED
48		Open Drip-	Proof Motors	
49	MOTOR	Nominal	Motor Speed	
50	HP	1200 rpm	1800 rpm	3600 rpm
51				
52	1	82.5	85.5	77.0
53	1-1/2	86.5	86.5	84.0
54	2	87.5	86.5	85.5
55				
56	3	88.5	89.5	85.5
57	5	89.5	89.5	86.5
58	7-1/2	90.2	91.0	88.5
59				
60	10	91.7	91.7	89.5
61	15	91.7	93.0	90.2
62	20	92.4	93.0	91.0
63	25	93.0	93.6	91.7

1		Totally	Enclosed Fan-Coole	ed
2	MOTOR	Nominal	Motor Speed	
3	HP	1200 rpm	1800 rpm	3600 rpm
4				
5	1	82.5	85.5	77.0
6	1-1/2	87.5	86.5	84.0
7	2	88.5	86.5	85.5
8				
9	3	89.5	89.5	86.5
10	5	89.5	89.5	88.5
11	7-1/2	91.0	91.7	89.5
12				
13	10	91.0	91.7	90.2
14	15	91.7	92.4	91.0
15	20	91.7	93.0	91.0
16	25	93.0	93.6	91.7
17				

1819 SINGLE PHASE, SINGLE SPEED MOTORS

Use NEMA rated 115 volt, single phase, 60 hertz motors for all motors 1/3 HP and smaller.

Use permanent split capacitor or capacitor start, induction run motors equipped with permanently lubricated and sealed ball or sleeve bearings and Class A insulation. Service factor to be not less than 1.35.

25 MOTORS USED ON VARIABLE FREQUENCY DRIVES

In addition to the requirements specified above, the motor must be suitable for use with the drive specified in Section 23 05 14,
 including but not limited to motor cooling. Motor shall comply with NEMA MG1 Part 31 to provide windings capable to
 withstand up to 1600 peak Volts with a rise time of 0.1 μs.

Provide bearing protection grounding rings to bleed current from the motor shaft to the motor casing. Manufacturers: Aegis
 SGR, Inpro/Seal CDR, or equal.

PART 3 - EXECUTION

36 INSTALLATION

Mount motors on a rigid base designed to accept a motor, using shims if required under each mounting foot to get a secure
 installation.

When motor will be flexible coupled to the driven device, mount coupling to the shafts in accordance with the coupling manufacturer's recommendations. Using a dial indicator, check angular misalignment of the two shafts; adjust motor position as necessary so that the angular misalignment of the shafts does not exceed 0.002 inches per inch diameter of the coupling hub. Again using the dial indicator, check the shaft for run-out to assure concentricity of the shafts; adjust as necessary so that run-out does not exceed 0.002 inch.

45

20

21

24

33 34

35

46 When motor will be connected to the driven device by means of a belt drive, mount sheaves on the appropriate shafts in 47 accordance with the manufacturer's instructions. Use a straight edge to check alignment of the sheaves; reposition sheaves as 48 necessary so that the straight edge contacts both sheave faces squarely.

After sheaves are aligned, loosen the adjustable motor base so that the belt(s) can be added and tighten the base so that the belt tension is in accordance with the drive manufacturer's recommendations. Frequently recheck belt tension and adjust if necessary during the first day of operation and again after 80 hours of operation.

52

53 Verify the proper rotation of each three-phase motor as it is being wired or before the motor is energized for any reason. 54

Lubricate all motors requiring lubrication. Record lubrication material used and the frequency of use. Include this information in the maintenance manuals.

- 57 58
- 59

Page Intentionally Left Blank

1 2	SECTION 23 05 14 VARIABLE FREQUENCY DRIVES
3 4	
5	PART 1 - GENERAL
0 7	Applicable provisions of Division 1 shall govern all work under this Section
8 9	SCOPE
10 11	This section includes variable frequency drives, bypass starters, and line reactors. Included are the following topics:
12	PART 1 - GENERAL
13	Scope
14	Related Work
15	Reference
16	Reference Standards
1/ 10	Submittais
10 19	Equipment Startun
20	Warranty
21	than and y
22	PART 2 - PRODUCTS
23	Manufacturers
24	Design and Construction
25	Performance Requirements
26	Control Features
27	Protection Features
28	Diagnostics
29	Quality Assurance Tests
30	Bypass Equipment
31	AC Input Line Reactors
32	Output Line Filters
33	
34 25	PART 3 - EXECUTION
35 26	Construction Variation
50 27	Euler Line References Testing
20 20	
30	i annig
40	RELATED WORK
41	Section 23 34 00 - HVAC Fans
42	Section 23 73 13 – Modular Indoor Central Station Air Handling Units
43	Section 23 73 23 – Factory Fabricated Custom Air-Handling Units
44	Section 26 05 26 - Grounding and Bonding for Electrical Systems
45	Section 26 05 29 - Hangers and Supports for Electrical Systems
46	Section 26 05 53 - Identification for Electrical Systems
47	Section 26 27 02 – Equipment Wiring Systems
48	
49	REFERENCE
50	Applicable provisions of Division 1 govern work under this section.
51	
52	REFERENCE STANDARDS
53	ANSI/IEEE 519 Guide for Harmonic Control and Reactive Compensation of Static Power Converters
54 55	
55	Submit shon drawings and product data under provisions of Division 1. General Conditions of the Contract
57	Include physical, electrical, and performance characteristics of each variable frequency drive and associated components
58	including dimensions; weight: input and output performance: voltage, phase, current and overcurrent characteristics.
59	installation instructions; protective features; wiring and block diagrams indicating specified options: electrical noise attenuation
60	equipment where required to meet the criteria specified; line side voltage notch wave form and line side current harmonics;
61	certified efficiency versus load and speed curves; and required operating environment.

OPERATION AND MAINTENANCE DATA

All operations and maintenance data shall comply with the submission and content requirements specified under section GENERAL REQUIREMENTS.

EQUIPMENT STARTUP AND AGENCY TRAINING

Provide the services of a factory trained and certified technician to approve the installation; start-up, test, and adjust for proper operation of the unit(s). Upon completion of the equipment startup, submit a complete manufacturer's field report, including startup and test log, signed by the factory trained technician. Coordinate with the Temperature Control Contractor and the Balancing Contractor. The startup shall be coordinated with Division 26. Electrical and shall be completed within ten (10) working days from the startup date.

WARRANTY

The warranty shall be for a period of twenty-four (24) months from the date of project Substantial Completion. Further, the warranty shall include all parts, labor, travel time, administrative costs, overhead, travel expenses, technical support and any and all other costs to provide the warranty service.

PART2 - PRODUCTS

MANUFACTURERS

Danfoss. No other manufacturers will be allowed.

DESIGN AND CONSTRUCTION

The unit shall be variable torque, modular design for control of the motors as specified in Division 15 and rated at the motor full load nameplate amps.

The unit shall be U.L. listed, solid state, microprocessor-based with a pulse width modulated (PWM) output wave form (none others are acceptable).

The VFD shall employ a full wave bridge rectifier and capacitors to minimize the ripple of the rectified voltage to maintain near constant DC voltage. Insulated gate bipolar transistors (IGBT's) shall be employed as the output switching device.

The VFD package shall contain the equivalent of 5% impedance to reduce harmonic distortion. The 5% equivalent impedance shall be provided in the form of a DC bus choke, an input AC line reactor in each phase, or a combination of the two methods.

Control circuitry shall be plug-in, plug-out modular basis with a corrosion resistant coating on printed circuit boards.

Units to be suitable for an operating environment from 0°C to 40°C temperature and humidity up to 90% non-condensing.

Electrically and physically isolate control circuitry and conductors from power circuitry and power conductors. Control conductors and power conductors shall not be run in the same pathway.

The unit enclosure shall be NEMA 1 or 12 as required for the specific application minimum and all components shall be fully factory assembled and tested prior to leaving the manufacturing facility.

Include the following operating and monitoring devices mounted on the front cover:

A disconnect switch or circuit breaker to de-energize both the drive and bypass circuit with door interlocked handle and lock-open padlocking provisions.

Operating mode selector switch marked "hand-off-auto".

- Manual speed adjustment via keypad, mounted on the door.
- Manual bypass selector switch to select power through drive or bypass.

Provide a manual bypass circuit and bypass starter to transfer from variable frequency drive operation to bypass operation.

PERFORMANCE REQUIREMENTS

Units shall be suitable for input power of electrical system as scheduled on the drawings ±10%, 3 phase, 60 Hertz nominal.

Use a current limiting control device to limit output current to 110% continuous for one minute; also refer to Protection Features in this section. Full load output current available from drive shall not be less than motor nameplate amperage. The full load amp rating of the VFD shall not be less than the values indicated in the NEC Table 430-150.

Output power shall be suitable for driving standard NEMA B design, three phase alternating current induction motors at full rated speed with capability of 6:1 turndown.

1	
2	Additional performance capabilities to include the following:
2	Additional performance capabilities to include the following.
3	Nide till ough a momentary power outage of 15 cycles,
4	Statt into a rotating load without damage to drive components of motor,
5	Capable of automatic restart into a rotating load after a preset, adjustable time delay following a power outage
6	Input power factor: Min 0.95 throughout the speed range
/	Minimum efficiency: 95% at 100% speed, 85% at 50% speed
8	
9	CONTROL FEATURES
10	Use control circuits compatible with input signal from temperature control system in the automatic mode and from manual
11	speed control in the manual mode. Vary motor speed in response to the input control signal. Include components necessary to
12	accept the signal from the temperature control system in the form that it is sent. Refer to Division 23 00 00.
13	
14	Include the following additional control features:
15	 Hand-Off-Automatic (HOA) selector switch to select local or remote start/stop and speed control
16	 Analog input, selectable 0-10v or 4-20 mA, for automatic control from the temperature control system
17	Local speed control at the VFD
18	• Adjustable acceleration and deceleration rate so that the time period from start to full speed and from full speed to stop
19	can be field adjusted
20	 Adjustable minimum and maximum speed settings for both automatic and manual modes of operation
21	Manual transfer bypass circuit
22	Field adjustment of minimum and maximum output frequency
23	 Two (2) sets of programmable form "C" contacts for remote indication of variable frequency drive condition. Note: default
24	programming to be set for "Drive Run & Fault"
25	Illuminated display keynald
25	Evidence Capital Region
20	 External Fault indication One (1) input for a NO day contact type input for a 2 wire remote start/stop
27	• One (1) input for a N.C. dry contact type input for a z-wire remote star/stop
20	• One (1) input for a N.C. dry contact type input for external radits: (neezestats, ine alarm, smokes, etc). This input shall be
29	ractory wired to prevent both the VFD and bypass starter operation when external ratit is present.
30	• One (1) N.O. dry contact output for proving motor status. This output shall be programmed to detect beit or coupling
31	break that would remove the load from the motor. The dry contact will open on loss of load or VFD being off.
32	 PID control loop capable of VFD control from an external device connected to a VFD analog input.
33	 When specified in the 23 09 93 sequence of operations, provide a VFD input and output for shutoff damper control that
34	shall operate as follows: When the fan is remotely or locally commanded to start, VFD contact shall energize the shutoff
35	damper to open the damper. The damper position end switch shall be wired to a run permissive input on the VFD and
36	enable the VFD to start when the damper end switch provides the damper is open. This feature shall be provided for both
37	inverter and bypass operation (if bypass option is provided).
38	
39	The VFD controller shall convert VFD information into the BAS DDC protocol that will be compatible with the building direct
40	digital energy management system (EMS) supplied on the project. This output shall be through a serial interface port capable
41	of two-way communication with the building EMS provided on this project. Final connection shall not require any additional
42	intermediate gateway devices to provide throughput of data. The following data shall be provided at a minimum:
43	Fault condition
44	Speed
45	Amperage
46	• Frequency
47	Voltage
48	Bynass status (if supplied)
49	- /
50	PROTECTION FEATURES
51	Use electronic protection circuitry in the power circuits to provide an orderly shutdown of the drive without blowing fuses or
52	triping circuit brokers and prevent component loss under the following abnormal conditions:
52	Activation of any cafety device:
55	Activation of any safety device,
54	Bower line overednient and/or over voldage of output,
22	Phase loss
50	Phase loss; Single and three phase phase short singuiting:
5/	Single and three phase short circuiting;
58	Ground faults;
59	Control circuit malfunction;
60	Overtemperature; and
61	Output current over limit.
62	
63	Provide the following additional protective features:

- DC bus fusing or other electronic controls which limit the rate of rise of the DC bus current and de-energizes the drive at a predetermined current level;
- Fusing for the control circuit transformer;
- Grounded control chassis; and
- Devices and/or control circuitry to ensure that the variable frequency drive and bypass starter are not both energized and driving motor simultaneously.
- Drives shall have a 100kA SCCR rating.

DIAGNOSTICS

Provide an English character display (no error codes) with indicators for the following:

- Phase loss Ground fault
- Overcurrent
- Overvoltage
- Undervoltage
 - Over temperature
 - Overload
 - DC bus status

QUALITY ASSURANCE TESTS

Use a factory heat stress test to verify proper operation of all functions and components under full load.

Field performance test of variable frequency drives to determine compliance with this specification will be performed at the DFD's discretion and may include any specified feature, including operation of protective devices through a simulated fault. Contractor will pay for initial testing. Should drive be found deficient by this testing, drive manufacturer will be required to make any and all changes necessary to bring unit(s) into compliance with the specified performance and demonstrate this performance by retesting. Cost of changes and retest will be by this contractor.

BYPASS EQUIPMENT

Bypass Starters:

The bypass starters for 208 volt motors, 20 HP and less; and 480 volt motors, 40 HP and less, shall be across-the-line magnetic starter type.

Bypass Configuration:

Provide one main disconnect switch or circuit breaker to de-energize both the drive and bypass circuit. Provide a drive input disconnect switch or fuse block to allow the drive to be isolated while the bypass circuit is energized. Provide one output drive contactor and one output bypass contactor. The two output contactors shall be electrically interlocked to allow only one contactor to be closed at any one time.

Provide motor overload protection in the bypass circuit.

Provide bypass equipment in a common enclosure with the VFD or, if not available, in a separate enclosure.

AC INPUT LINE REACTORS

When needed to comply with the requirement for 5% equivalent impedance, furnish and factory install AC input line reactors.

Line reactors shall be installed in each phase of the AC input side of the VFD and mounted within a common enclosure with the VFD.

Line reactor shall be a three phase inductor, iron core, 600V, Class H insulation, 115 degree C rise, copper windings with screw type terminal blocks.

PART 3 - EXECUTION

VARIABLE FREQUENCY DRIVES

Install where indicated on drawings and in accordance with approved submittals and manufacturer's published recommendations. Installation to be by the Division 26 00 00 - Electrical contractor.

Input power wiring shall be installed in a separate conduit, output power wiring shall be installed in a separate conduit and control wiring shall be installed in a separate conduit. Do not mix input power, output power, or control wiring in a common conduit. Separate conduits for input and output power wiring shall be provided for each motor. Input and output power wiring for more than one motor shall not share a common conduit. Power wiring shall be furnished and installed by the Div. 26 contractor. If provided, do not mount output line filter above the drive.

Control signal for drive will be provided under Division 23.

Temperature Control Contractor will furnish and install the required temperature control wiring in metal conduit and in accordance with Division 26 00 00 - Electrical of this specification.

CONSTRUCTION VERIFICATION ITEMS

Contractor is responsible for utilizing the construction verification checklists supplied by the third party commissioning agent in accordance with the procedures defined for construction verification in Section 01 91 00.

11 FUNCTIONAL PERFORMANCE TESTING

12 Contractor is responsible for utilizing the functional performance test forms supplied by the third party commissioning agent in 13 accordance with the procedures defined for functional performance testing in Section 01 91 00.

14 15 **TRAINING**

16 Contractor to provide factory authorized representative and/or field personnel knowledgeable with the operations, 17 maintenance and troubleshooting of the system and/or components defined within this section for a minimum period of 4 18 hours.

19

20

1 2

3 4

5

6 7

8

9 10

21

Page Intentionally Left Blank

1	SECTION 23 05 15
2	PIPING SPECIALTIES
3	
4	
6	
7	SCOPE
8 9	This section contains specifications for HVAC piping specialties for all piping systems. Included are the following topics:
10	PART 1 - GENERAL
11	Scope
12	Related Work
13	Reference
14	Quality Assurance
15	Shop Drawings
10	Operation and Maintenance Data
17 18	Design Criteria
10	PART 2 - PRODUCTS
20	Thermometers
21	Thermometer Sockets
22	Test Wells
23	P/T (Pressure/Temperature) Test Plugs
24	Hose Connection Caps
25	Pressure Gauges
20	Strainers Expansion Tanks
27	Buffer Tanks
29	Coalescing Air and Dirt Separators
30	Air Vents
31	Suction Diffusers
32	Flow Sensing Devices
33	
34 25	PART 3 - EXECUTION
36	Thermometer Sockets
37	Test Wells
38	P/T (Pressure/Temperature) Test Plugs
39	Pressure Gauges
40	Strainers
41	Expansion Tanks
42	Buffer Lanks Coolessing Air and Dirt Senerators
45 11	Codiescing Air and Dirt Separators Air Vents
45	Suction Diffusers
46	Flow Sensing Devices
47	Construction Verification Items
48	
49	RELATED WORK
50	Section 23 05 23 - General-Duty Valves for HVAC Piping
51 52	Section 23 05 29 - Hangers and Supports for HVAC Piping and Equipment p
53	Section 23 21 13 - Hydronic Pining
54	Section 23 21 26 – Ground Source Vertical Heat Exchanger Field
55	Section 23 57 00 – Heat Exchangers for HVAC
56	Section 23 73 12 – Air Handling Unit Coils
57	
58	KEFEKENCE Applicable provisions of Division 1 govern work under this section
59 60	Applicable provisions of Division 1 govern work under this section.
61	OUALITY ASSURANCE
62	Refer to Division 1.
63	

SHOP DRAWINGS

1 2

3 4

5

6 7

8

9

10 11

12

13

14 15

16 17 18

19 20

23

33

38

42

45

48

Refer to Division 1.

Required for all items in this section. Include materials of construction, dimensional data, ratings/capacities/ranges, pressure drop data where appropriate, and identification as referenced in this section and/or on the drawings.

OPERATION AND MAINTENANCE DATA

All operations and maintenance data shall comply with the submission and content requirements specified under section GENERAL REQUIREMENTS.

DESIGN CRITERIA

All piping specialties are to be rated for the highest pressures and temperatures in the respective system in accordance with ANSI B31, but not less than 125 psig unless specifically indicated otherwise.

PART2 - PRODUCTS

THERMOMETERS

Manufacturers: Ashcroft, Marsh, Taylor, H. O. Trerice, U. S. Gauge, Weiss, Weksler.

Stem Type, cast aluminum case, nine inch scale, clear acrylic window. adjustable angle brass stem with stem of sufficient length 21 so the end of the stem is near the middle of a pipe without reducing the thickness of any insulation, red indicating fluid, black 22 lettering against a white background, with scale ranges as follows:

Service	Scale Range, °F	Min. Increment, °F
Hot Water	30 - 240	2
Chilled Water	0 - 100	1
Geothermal Water	0-120	1

30 THERMOMETER SOCKETS

31 Brass with threaded connections suitable for thermometer stems and temperature control sensing elements in pipeline. 32 Furnish with extension necks for insulated piping systems.

34 TEST WELLS

35 Similar to thermometer sockets except with a brass cap that thread into the inside of the test well to prevent dirt from 36 accumulating. Secure cap to body with a short chain. Furnish with extension necks, where appropriate, to accommodate the 37 pipeline insulation.

P/T (PRESSURE/TEMPERATURE) TEST PLUGS 39

40 Brass plug with 1/4" NPT threads, EPDM or neoprene valve core, knurled cap with cap strap. Use extended length plugs to clear insulated piping. Adaptors shall have 1/4" FPT connection for standard pressure gauges. 41

43 HOSE CONNECTON CAPS

44 Hose connection caps shall be pressure rated for 150 psig at 180 deg F.

46 PRESSURE GAUGES

47 Manufacturers: Ametek/U. S. Gauge Division, Ashcroft, Marsh, Taylor, H. O. Trerice, Weiss, Weksler.

Cast aluminum case of not less than 4.5 inches in diameter, double strength glass window, black lettering on a white 49 background, phosphor bronze bourdon tube with bronze bushings, recalibration from the front of the dial, 99% accuracy over 50 the middle half of the scale, 98.5% accuracy over the remainder of the scale, with scale range as follows: 51 52

Service	Scale Range, psig	Min. Increment, psig
Hot Water	0-100	1.0
Chilled Water	0-100	1.0
Geothermal Water	0-100	1.0
PRESSURE SNUBBERS:		

Bronze construction, suitable for system working pressure, 1/4" size.

59 COIL SYPHONS: 60

Bronze or steel construction, suitable for system working pressure, 1/4" size.

61 62 63

- 1 GAUGE VALVES:
- 2 Use valves as specified in Section 23 05 23 General-Duty Valves for HVAC Piping. For water systems, use 1/4" ball valves. 3

STRAINERS

5 Manufacturers: Armstrong, Hoffman, Illinois, Keckley, Metraflex, Mueller Steam, or Sarco.

Y type; cast iron body; stainless steel screens; bolted or threaded screen retainer tapped for a blowoff valve; threaded body in sizes through 2 inch and rated at not less than 175 psi WOG; flanged body in sizes over 2 inch and rated at not less than 125 psi
WOG at 240°F. Screen to be 20 mesh for line sizes 2 inch and less, 0.125 inch perforations for line sizes 2-1/2 inch through 4 inch, and 0.25 inch perforations for line sizes 5 inch and larger.

11

4

12 EXPANSION TANKS

13 Manufacturers:

- 14 Amtrol/Thrush, Armstrong Pumps, Bell and Gossett, Taco, or Wessels.
- 15

Steel construction with finish coat of rust resistant paint, tested and stamped in accordance with Section 8D of ANSI/ASME Code and furnished with the National Board Form U-1, rated for working pressure of not less than 125 PSIG. Tank shall be precharged with air to the initial fill pressure indicated on Drawings and have heavy-duty butyl rubber bladder rated for fluid temperatures up to 2402/degrees2F. Tank shall include tank drain connection, system connection, and charging valve connection (standard tire valve), and include integral lifting rings, and floor mounting skirt for vertical installation. Size and capacity of tank shall be as specified on Drawings. Tank and bladder construction must allow field replacement of bladder in event of failure.

24 BUFFER TANKS

25 Steel construction with prime coat finish, tested and stamped in accordance with Section 8D of ANSI/ASME Code and furnished

- 26 with National Board Form U-1, rated for working pressure of not less than 125 PSI. Tank shall have hand hole for inspection of
- 27 tank, drain valve tapping, system supply and return tappings, and mounting saddles for horizontal tank installations, or ring
- base or angle legs for vertical installations. Tank shall have internal baffles as specified on Drawings. Size and capacity as
- 29 specified on Drawings. Refer to Drawings for quantity and locations of tank tappings.
- Refer to Section 23 07 00 for tank insulation requirements.

32 COALESCING AIR AND DIRT SEPARATORS

33 Manufacturers: Bell & Gossett, Taco, Spirotherm, Inc., Wessels

2 inch and larger: Welded steel construction, ASME Section VIII, Division 1 constructed and stamped for a working pressure not less than 150 psig at 250°F, threaded or flanged connections for 2 inch size, flanged or grooved connections if grooved piping is allowed for all sizes over 2 inch. Provide unit with factory installed automatic air vent on the top, blow down connection at the bottom of the unit, and a skimming connection to remove floating dirt. Unit size shall be suitable for the system flow rates as indicated on the drawings.

39

40 Internal coalescing material shall be constructed of copper or 304 stainless steel. Unit shall include internal structured elements 41 filling the entire vessel to suppress turbulence and provide air elimination efficiency of 100% entrained air, and 99% dissolved

air at the installed location. Dirt separation efficiency shall be a minimum of 80% of all particles 30 micron and larger within 100
 passes.

44

45 Provide removable head to access material for inspection and cleaning.

46 47 **AIR VENTS**

- 48 MANUAL KEY TYPE VENTS:
- 49 Bell and Gossett Model 4V; Eaton/Dole Model 9, 9B, or 14A.
- 50

51 Bronze body with nonferrous internal parts, screwdriver operated, designed to relieve air from the system when vent is 52 opened, rated at not less than 125 psig at 220°F.

- 53
- 54 MANUAL BALL VALVE VENTS:
- Provide 1/4" ball valves for manual venting of air handling unit coils and where indicated elsewhere on drawings and details.
 Reference specifications section 23 05 23.
- 57 58 AUTOMATIC VENTS:
- 59 Thrush Model 720, Bell and Gossett Model 107, Watson McDaniel Model AV813W
- 60

61 Cast iron body with nonferrous internal parts, designed to vent air automatically with float principle without allowing air to 62 enter the system, rated at not less than 125 psig at 220°F.

SUCTION DIFFUSERS 1

2 Manufacturers:

Amtrol, Armstrong Pumps, Bell and Gossett, Mueller or Taco.

Body constructed of cast iron, ductile iron or carbon steel; cast iron, steel or stainless steel straightening vanes; steel or stainless steel strainer; brass or bronze fine mesh startup strainer, strainer blowdown connection, inlet pressure gage connection, provisions for field supplied support foot, and bolted flange for strainer removal and cleaning; rated at not less than 150 PSI working pressure at not less than 250[®] degrees[®]F.

FLOW SENSING DEVICES 10

11 For water flow sensing devices 2 inch and smaller, use balance valves as specified in Section 23 05 23 - General-Duty Valves for 12 HVAC Piping.

13

31 32

33 34

38

43

45

49 50

3

4 5

6

7

8

9

14 PITOT TUBE FLOW SENSORS:

Dieterich Standard/Annubar, Preso, or approved equal. 15 16

Multi-port averaging type flow sensor designed to sense the velocity of a fluid flowing in a pipe and produce a pressure output 17 18 that is proportional to the fluid velocity. Sensor to consist of a type 316 stainless steel probe with a diamond or elliptical shape of sufficient length to sense flow completely across the pipe section and to accommodate the insulation specified for the 19 20 pipeline; brass body gate, needle, or ball instrument connection valves with appropriate fitting for connection to a meter; single forged steel weld type installation fitting for pipe sizes through 6 inch, double forged steel weld type installation fittings for use 21 on opposite ends of the sensor for larger pipe sizes if recommended by the manufacturer for the application; accurate within 22 23 2% of the actual flow with a turndown ratio of 10:1 or better; permanently stamped nameplate attached to the sensor indicating the flow/differential pressure characteristics of the sensor; suitable for use on systems to 150 psig at 250°F. 24 25

26 Include one differential pressure meter kit that includes a six inch diameter gauge having an accuracy of 3% of full scale or 27 better and suitable for the differential pressures of the valves supplied for this project, color coded hoses not less than ten feet in length with brass connectors suitable for connection to the low and high pressure connections on the balance valves, 28 29 instrument valving so meter can be vented and drained, pressure and temperature rating at least equal to that of the valves. Provide meter and all accessories in a durable case with carrying handle. 30

PART 3 - EXECUTION

35 THERMOMETERS

36 STEM TYPE:

37 Install in piping systems as indicated on the drawings and/or details using a separable socket in each location.

DIAL TYPE FOR AIR TEMPERATURE MEASUREMENT: 39

40 Install in ductwork where detailed or specified. Support capillary inside duct so it measures a uniform sample of air. Mount readout so it is readily visible on a portion of ductwork that is not externally insulated or on a sheet metal angle support 41 42 secured to a nearby structural element.

44 THERMOMETER SOCKETS

Install at each point where a thermometer or temperature control sensing element is located in a pipeline.

46 47 TEST WELLS

48 Install in piping systems as indicated on the drawings and/or details wherever provisions are needed for inserting a thermometer at a later date.

P/T (PRESSURE/TEMPERATURE) TEST PLUGS 51

Install in piping systems as indicated on the drawings and/or details. Do not insulate over test plugs. 52

53 PRESSURE GAUGES 54

55 Install in locations where indicated on the drawings and/or details, including any gauge piping, with scale range appropriate to 56 the system operating pressures.

57 58 PRESSURE SNUBBERS:

59 Install in gauge piping for all gauges used on water services.

60

61 GAUGE VALVES

62 Install at each gauge location as close to the main as possible and at each location where a gauge tapping is indicated. 63

1 STRAINERS

Install all strainers where indicated on the project details, allowing sufficient space for the screens to be removed. Rotate
 screen retainer where required by the installation so blowdown can remove accumulated dirt from the strainer body.

5 WATER SYSTEMS:

6 Install a ball valve for blowdown in the tapped screen retainer; valve to be the same size as the tapping.

8 EXPANSION TANKS

9 Install tanks where indicated on Drawings, coordinating concrete base installation with General Contractor or fabricate steel 10 supports to suit application. Install specified tank accessories.

11

15

4

7

Verify proper air charge and recharge if necessary. Install isolation valve in piping connecting tank to system. In piping between tank and isolation valve, install pressure gage and drain valve with hose adapter. Install a drain valve with hose adapter in drain connection of tank. Make sure all drains are accessible and hose can be attached.

16 BUFFER TANKS

Install tanks where indicated on Drawings, coordinating concrete base installation with General Contractor. Provide structural
 supports required, including hanger rods for suspended installations and steel supports for wall mounted installations.
 Fabricate supports to suit application. Install specified tank accessories.

21 COALESCING AIR AND DIRT SEPARATORS

Mount in hot and/or chilled water lines as indicated on the drawings/details. Install ball valve with hose adapter in bottom blowdown connection and skimming connection.

24

20

25 Open the drain/blowdown valve after system cleaning and again after 30 days of operation.

26 27 **AIR VENTS**

28 MANUAL KEY TYPE VENTS:

Install at all high points where air may collect and not be carried by the system fluid. Use a soft Type L copper "pigtail" so the vent can be positioned for venting and collecting any water that might escape.

3132 MANUAL BALL VALVE VENTS:

33 Install on air handling coils and where indicated elsewhere as shown on drawings and details.

- 34
- 35 AUTOMATIC VENTS:

Install on the top of air separators on systems using bladder type expansion tanks. Install at other locations as indicated on the drawings or details. All locations to have a ball valve installed upstream of the vent for maintenance purposes.

39 SUCTION DIFFUSERS

Install at each pump suction connection for end suction pumps where shown on Drawings. Provide space for removal of strainer. Install drain valve in the blowdown connection. Install support below the suction diffuser so the weight of suction piping does not rest on pump suction connection.

43

49

38

Install pressure gage across the suction diffuser, valved so single gage can be used to read pump discharge pressure, inlet pressure at the suction diffuser and the outlet pressure at the suction diffuser. Use gage valves as specified with gages.

47 Open drain valve and blowdown strainer after system cleaning. If unit is furnished with fine mesh startup strainer, remove strainer 48 after the system has been flushed and cleaned just prior to testing and balancing.

50 FLOW SENSING DEVICES

51 Give portable meter to Owner at the completion of all balancing work.

- 52
- 53 PITOT TUBE FLOW SENSORS:

Install where indicated on the drawings and details for flow sensing in hydronic and/or steam piping systems. Butterfly valves installed at the location of a flow sensing device are to have a memory stop.

57 **CONSTRUCTION VERIFICATION ITEMS**

58 Contractor is responsible for utilizing the construction verification checklists supplied by the third party commissioning agent in 59 accordance with the procedures defined for construction verification in Section 01 91 00.

60

61 62

Page Intentionally Left Blank

1	SECTION 23 05 23
2	GENERAL-DUTY VALVES FOR HVAC PIPING
3	
4	
5	PART 1 - GENERAL
6	
7	SCOPE
8	This section includes valve specifications for all HVAC systems except where indicated under Related Work. Included are the
9	following topics:
10	
11	PART 1 - GENERAL
12	Scope
13	Related Work
14	Reference
15	Quality Assurance
16	Submittais
1/	Operation and Maintenance Data
18	Design Criteria
19	
20	PART 2 - PRODUCTS
21	Manufacturers
22 วว	water system valves
25 24	Built Valves
24 25	Glober Valves
25	Balance Valves
20	
27	Specialty Valves and Valve Accessories
29	Check Valves
30	Triple Duty Valves
31	Water Relief Valves
32	Gauge Valves
33	Chain Wheel Operators
34	Stem Extensions
35	
36	PART 3 - EXECUTION
37	General
38	Shut-off Valves
39	Balancing Valves
40	Calibrated Balancing Valves
41	Drain Valves
42	
43	RELATED WORK
44	Section 23 05 15 - Piping Specialties
45	Section 23 09 14 - Electric Instrumentation and Control Devices for HVAC
46	
47	REFERENCE
48	Applicable provisions of Division 1 govern work under this section.
49	
50	QUALITY ASSURANCE
51	Refer to Division 1, General Conditions, Equals and Substitutions.
52	
53	SUBMITTALS
54	Refer to Division 1, General Conditions, Submittals.
55	Contractors shall submit a schedule of all valves indicating type of service, dimensions, materials of construction, and
56	pressure/temperature ratings for all valves to be used on the project. Temperature ratings specified are for continuous
5/	operation.
58	
59	UPERATION AND MAINTENANCE DATA
0U 61	All operations and maintenance data shall comply with the submission and content requirements specified under section
67 01	UEINERAL REQUIREIVIEINIS.
62 62	
05	

DESIGN CRITERIA

1 2

3

4

5 6 7

8 9

10

11 12

13

14 15

16 17

19 20

21

22

27

29

32

40

42

45

47

50

Where valves are specified for individual mechanical services (i.e. hot water heating, steam, etc.) all valves shall be of the same manufacturer unless prior written approval is obtained from DFD.

PART 2 - PRODUCTS

MANUFACTURERS

Anvil, Apollo, Armstrong, Bell & Gossett, Cash-Acme, Dresser Consolidated, Conval, Crane, Anderson Greenwood and Crosby, Danfoss-Flomatic, DeZurik, Durco, Fisher, Grinnell, Griswold, Hammond, Hancock, Hoffman, Jamesbury, Keystone, Kunkle, Leslie, Lunkenheimer/Cincinnati, Metraflex, Milwaukee, Mueller, Newco, Nexus, Nibco, Powell, RP&C, Sarco, Spence, Stockham, Taco, Tasco, Thrush-Amtrol, Vogt, Watts, or approved equal.

WATER SYSTEM VALVES

All water system valves to be rated at not less than 125 psig water working pressure at 240°F unless noted otherwise.

18 BALL VALVES:

2" and smaller: Two piece bronze body; threaded or soldered ends, as appropriate to the pipe material; stainless steel or chrome plated brass/bronze ball; conventional port; glass filled teflon seat; threaded packing gland follower; blowout-proof stem; 600 psig WOG.

Valve stems shall allow operators to clear insulation without interference. Provide stem extensions when valve operators
 interfere with pipe insulation.

26 Apollo 70-100/200 series, Hammond 8301/8311, Milwaukee BA100/150, Nibco T/S 585-70, Stockham S206/216.

28 2-1/2" and over: Ball valves will not be accepted in sizes over 2 inch.

30 BUTTERFLY VALVES

2" and smaller: Use ball valves; butterfly valves will not be accepted in sizes 2 inch and smaller.

2-1/2" and larger: Cast iron body; stainless steel shaft; Teflon, nylatron, or acetal bearings; EPDM resilient seat. Disk to be bronze, aluminum-bronze, nickel plated ductile iron, cast iron with welded nickel edge, or 316 - stainless steel. Pressure rated to 150 psig. Valve assembly to be bi-directionally bubble tight to 150 psig with no downstream flange/pipe attached. Nylon coated ductile iron discs are not acceptable. Polymid or polyamide coated valves are not acceptable.

Valve stems shall allow operators to clear insulation without interference. Provide stem extensions when valve operators interfere with pipe insulation.

41 Use threaded lug type valves for installation with class 125/150 flanges.

43 Centerline series 200, DeZurik BOS-CL, Keystone Fig. 222, Nibco LD2000 (2-1/2"-12")/LD1000 (14" and above), Bray Series 31H,
 44 Victaulic 300 series (2-1/2"-12")/709 series (14"-24").

46 Provide ten-position lever actuators for valves 6" and smaller.

48 Where butterfly valves are indicated or specified to be installed at the location of a flow sensing device, provide the butterfly 49 valves with a memory stop.

- 51 GLOBE VALVES:
- 52 Do not use globe valves for water service, except in temperature control applications. 53
- 54 BALANCE VALVES:

2" and smaller: Bronze or copper alloy body with calibrated ball, globe or venturi/valve arrangement, integral pointer and calibrated scale to register degree of valve opening, memory stop, drain tapping, threaded or soldered ends, with or without integral unions, P/T or Shraeder pressure taps with integral check valves and seals, adjustable memory stop, suitable for 200 psig water working pressure at 250°F.

Armstrong CBV, Bell & Gossett Circuit Setter Plus, Griswold Quickset, Nexus Orturi, Nibco 1710 Series, Taco Accu-Flo, Tour &
 Anderson STAS/STAD, Victaulic series 786/787.

1 2 3	2-1/2" and larger: Use butterfly valves as specified in this section along with a flow sensing device as specified in Section 23 05 15.
4	DRAIN VALVES
5	Use 3/4 inch hall valve with threaded hose adapter excent strainer blowdown valves to be the same size as the blowdown
6 7	connection.
8	SPECIALTY VALVES AND VALVE ACCESSORIES
9	
10	
11	Swing Check Valves
12	2" And Smaller:
13	Ninco T413Y: bronze body threaded ends regrindable seat renewable disc teflon seat 200 psig WOG accentable for
14	installation in horizontal or vertical line with flow unward
15	
16	2%" And Larger:
17	Nibco F9188: cast iron body, flanged ends, bronze trim, bolted cap, renewable bronze seat and disc, non-asbestos gasket, 200
18 19	psig WOG, acceptable for installation in horizontal or vertical line with flow upward.
20	Spring Loaded Check Valves
21	2" and Smaller.
22	Nibco T480: bronze, threaded, bronze trim, stainless steel spring, teflon seat unless only bronze available, 250 psig WOG.
23	
24	2%" and Larger:
25	Nibco W910 Series: cast iron or semi-steel body, wafer style disk, bronze trim, bronze or EPDM seat, stainless steel spring.
26	stainless steel stem, and 125 psig WOG. Valves with ductile iron in contact with working fluid will not be accepted.
27	
28	TRIPLE DUTY VALVES:
29	2 Inch and Larger:
30	Bell & Gossett Triple Duty Valve; Cast or ductile iron body, threaded or flanged or grooved end connections, stainless steel
31	spring, bronze disc with EPDM seat, calibrated memory stop, backseating valve stem, inlet and outlet pressure tappings,
32	capable of being repacked under full line pressure, rated for working pressure of 125 PSIG at 2402 degrees P when used in hot
33	water heating systems.
34	
35	WATER RELIEF VALVES:
36	Iron or bronze body, direct pressure actuated, teflon seat, stainless steel stem and spring, rated for 125 psig water working
37	pressure at 240 [®] degrees [®] F and ASME stamped, with rated BTU capacity at required relief pressure. Provide required relief
38	valve pressure as specified on Drawings.
39	
40	GAUGE VALVES:
41	water Service: Use 1/4" ball valves.
42	
43	CHAIN WHEEL OPERATORS.
44 15	construct of cast of ductile from a continuous loop of chain a cash operator.
45	closure links to form a continuous loop of chain at each operator.
40	
47	Provide stam extensions, when value operators interfere with nine inculation
40	riovide stelli extensions when valve operators interfere with pipe insulation.
50	
51	ΡΑΚΤ 3 - ΕΧΕΓΠΤΙΟΝ
52	
53	GENERAL
54	Properly align piping before installation of valves in an upright position; operators installed below the valves will not be
55	accepted.
56	
57	Install valves in strict accordance with valve manufacturer's installation recommendations. Do not support weight of piping
58	system on valve ends.
59	
60	Install all temperature control valves.
61	

2

3

4 5

6

7

8 9

10 11

12

13 14

15 16

17

24

28

31

34

41

43

45

48

50

52 53 54 Install all valves with the stem in the upright position. Valves may be installed with the stem in the horizontal position only where space limitations do not allow installation in an upright position. Valves installed with the stems down, will not be accepted.

Where valves 2-1/2" and larger are located more than 12'-0" above mechanical room floors, install valve with stem in the horizontal position and provide a chain wheel operator.

Install stem extensions when shipped loose from valve.

Prior to flushing of piping systems, place all valves in the full-open position.

SHUT-OFF VALVES

Install shut-off valves at all equipment, at each branch take-off from mains, and at each automatic valve for isolation or repair.

WATER SYSTEM:

Butterfly valves installed at the location of a flow sensing device are to have a memory stop.

BALANCING VALVES

Provide balancing valves for all major equipment and at each major branch takeoff and at the discharge of each pump as indicated on drawings and details.

21 CALIBRATED BALANCE VALVES:

Install where indicated on the drawings and details for balancing of hydronic systems. Retain the shipping container for use as removable insulation.

25 DRAIN VALVES

Provide drain valves for complete drainage of all systems. Locations of drain valves include low points of piping systems, equipment locations specified or detailed including reheat coils, other locations required for drainage of systems.

29 SWING CHECK VALVES

30 Provide swing check valves where specified and detailed on Drawings.

32 SPRING LOADED CHECK VALVES

33 Install spring loaded check valve in each pump discharge line.

35 TRIPLE DUTY VALVES

Contractor may use triple duty combination shut-off, check, and balancing valves where separate shut-off valve, check valve, and balancing valve are specified or detailed in pump discharge piping.

39 WATER RELIEF VALVES

40 Provide relief valves for each water system.

42 Setpoint for water relief valves is indicated on Drawings.

44 Use air pressure to clean piping prior to installation of safety relief valves.

Install relief valves in locations indicated on Drawings. When not indicated on Drawings, locate relief valves at chiller on chilled
 water systems, on water side of heat exchangers, and on=boilers.

49 Inlet and outlet piping connecting to valves must be same size as valve connections or larger.

51 Pipe discharge from water system relief valves full size to nearest glycol fill tank.

1	SECTION 23 05 29
2	HANGERS AND SUPPORTS FOR HVAC PIPING AND EQUIPMENT
3	
4	
5	PART1-GENERAL
6	
7	SCOPE
8	In is section includes specifications for supports of all HVAC equipment and materials as well as piping system anchors.
9 10	Included are the following topics:
10	
12	Scone
13	Related Work
14	Reference
15	Reference Standards
16	Quality Assurance
17	Description
18	Shop Drawings
19	Design Criteria
20	
21	PART 2 - PRODUCTS
22	Pipe Hanger and Support Manufacturers
23	Structural Supports
24	Materials Bine Hangers and Supports
25	Pipe Hangers and Supports
20	Concrete Inserts
27	Post Installed Concrete Anchors
29	Corrosive Atmosphere Coatings
30	
31	PART 3 - EXECUTION
32	Installation
33	Hanger and Support Spacing
34	Vertical Riser Clamps
35	
36	RELATED WORK
37	Section 23 05 48 - Vibration and Seismic Controls for HVAC Piping and Equipment
38	Section 23 07 00 - HVAC Insulation
39	
40 41	REFERENCE Applicable provisions of Division 1 shall govern work under this section
41 12	Applicable provisions of Division 1 shall govern work under this section.
42 43	REFERENCE STANDARDS
44	MSS SP-58 Materials, Design, Manufacture, Selection, Application, and Installation
45	
46	QUALITY ASSURANCE
47	Refer to Division 1, General Conditions, Equals and Substitutions.
48	
49	DESCRIPTION
50	Provide all supporting devices as required for the installation of mechanical equipment and materials. All supports and
51	installation procedures are to conform to the latest requirements of the ANSI Code for pressure piping.
52	
53	Do not hang any mechanical item directly from a metal deck or run piping so it rests on the bottom chord of any truss or joist.
54	Current energy and material under all conditions of energian variations in installed and energian weight of equipment
55	support apparatus and material under all conditions of operation, variations in installed and operating weight of equipment
57	מווע פופוווא, גע פוביצווג פגנפגג גוופגג, מווע מווטא וטו פוטפו פגפמוגוטון מווע גטווגומנגוטוו.
58	Protect insulation at all hanger points: see Related Work above
59	
60	SHOP DRAWINGS
61	Refer to Division 1, General Conditions, Submittals.
62	

Schedule of all hanger and support devices indicating shields, attachment methods, and type of device for each pipe size and type of service. Reference section 23 05 00.

DESIGN CRITERIA

1 2

3 4

5

6

7 8

9

10 11

12 13

14

15

16

17 18

19 20

21 22

23 24

25 26 27

28 29

30

31

32

34 35

36 37

38

39 40

41 42

43 44

45

47

51

58

60

62

63

Materials and application of pipe hangers and supports shall be in accordance with MSS Standard Practice SP-58 unless noted otherwise.

Piping connected to base mounted pumps, compressors, or other rotating or reciprocating equipment is to have vibration isolation supports for a distance of one hundred pipe diameters or three supports away from the equipment, whichever is greater. Standard pipe hangers/supports as specified in this section are required beyond the 100 pipe diameter/3 support distance.

Piping flexible connections and vibration isolation supports are required for piping connected to coils that are in a fan assembly where the entire assembly is mounted on vibration supports; the vibration isolation supports are required for a distance of one hundred pipe diameters or three supports away from the equipment, whichever is greater. Piping flexible connection and vibration isolation supports are not required when the fan section is separately and independently isolated by means of vibration supports and duct flexible connections. Standard pipe hangers/supports as specified in this section are required when there are no vibration isolation devices in the piping and beyond the 100 pipe diameter/3 support distance.

Piping supported by laying on the bottom chord of joists or trusses will not be accepted.

Fasteners depending on soft lead for holding power or requiring powder actuation will not be accepted.

Allow sufficient space between adjacent pipes and ducts for insulation, valve operation, routine maintenance, etc.

PART 2 - PRODUCTS

PIPE HANGER AND SUPPORT MANUFACTURERS

Anvil, B-Line, Fee and Mason, Kindorf, Michigan Hanger, Unistrut, or approved equal. Anvil figure numbers are listed below; equivalent material by other manufacturers is acceptable.

33 STRUCTURAL SUPPORTS

Provide all supporting steel required for the installation of mechanical equipment and materials, whether or not it is specifically indicated or sized, including angles, channels, beams, etc. to suspend or floor support tanks and equipment.

Contractor shall be responsible for the design of trapeze pipe hangers, structural pipe supports, and equipment supports.

MATERIALS

Aluminum: ASTM B221.

Carbon Steel: ASTM A1011/A1011M.

Structural Steel: ASTM A36/A36M, carbon-steel plates, shapes, and bars; black and galvanized.

46 Stainless Steel: ASTM A240/A240M.

Grout: ASTM C1107/C1107M, factory-mixed and -packaged, dry, hydraulic-cement, nonshrink and nonmetallic grout; suitable
 for interior and exterior applications. Nonstaining, noncorrosive, and nongaseous. Design mix shall be 5000-psi, 28-day
 compressive strength.

52 PIPE HANGERS AND SUPPORTS

- 53 HANGERS FOR STEEL PIPE SIZES 1/2" THROUGH 2":
- 54 Carbon steel, adjustable, clevis, black finish. Anvil figure 65 or 260.
- 55 56 HANGERS FOR STEEL PIPE SIZES 2-1/2" AND OVER:
- 57 Carbon steel, adjustable, clevis, black finish. Anvil figure 260.
- 59 Adjustable steel yoke, cast iron roll, double hanger. Anvil figure 181.

61 MULTIPLE OR TRAPEZE HANGERS:

Steel channels with welded spacers and hanger rods if calculations are submitted.
	et with hanger. B-Line 3068 Series, Anvil 1	4 Series.	
Perforated epoxy p type, bolt secured, p 1200 clamps. Whe completely encircle manufacturers clam	ainted finish, 16-12 gauge min., steel ch galvanized pipe/tubing clamps. B-Line typ in copper piping is being supported, provi the piping and avoid contact with the c ip and cushion assemblies, B-Line BVT ser	nnels securely anchored to wall structure S channel with B-2000 series clamps, Anvi e flexible elastomeric/thermoplastic isolati annel or clamp, equal to B-Line B1999 Vik s, Anvil cushion clamp assembly.	with interlocking, spl I type AS200 H with A ion cushion material t ora Cushion or provic
VERTICAL RISER SUF	PPORT:		
Carbon steel riser c pipe.	lamp, copper plated when used with cop	per pipe. Anvil figure 261 for steel pipe, fi	gure CT121 for copp
)R PIPE SIZES THROUGH 4".		
Cast iron adjustable	pipe saddle, locknut nipple, floor flange,	nd concrete pier or steel support.	
FLOOR SUPPORT FC	OR PIPE SIZES 5" AND OVER:		
Adjustable cast iron	roll and stand, steel screws, and concrete	pier or steel support.	
COPPER PIPE SUPPO	DRT:		
Carbon steel ring, a	djustable, copper plated or polyvinylchlor	le coated.	
INSULATION PROTE	CTION SHIELDS:		
Galvanized carbon s	steel of not less than 18 gauge for use on	nsulated pipe 2-1/2 inch and larger. Minin	num shield length is 1
inches. Equal to An	vii figure 167.		
Refer to Section 23	07 00 for additional requirements includio	g required insulation inserts.	
STEEL HANGER ROD	DS:		
Threaded both ends	s, threaded one end, or continuous thread	ed, black finish.	
Sizo rods for individ	ual hangers and transite support as indica	ad in the following schedule	
5120 1003 101 11101010	ual hangers and trapeze support as indica	ed in the following schedule.	
Total weight of equi	ipment, including valves, fittings, pipe, pip	e content, and insulation, are not to exceed	the limits indicated.
Total weight of equ	ipment, including valves, fittings, pipe, pip Maximum Load (Lbs.)	content, and insulation, are not to exceed Rod Diameter	the limits indicated.
Total weight of equ	ipment, including valves, fittings, pipe, pip Maximum Load (Lbs.) <u>(650°F Maximum Temp.)</u>	e content, and insulation, are not to exceed Rod Diameter (inches)	the limits indicated.
Total weight of equ	ipment, including valves, fittings, pipe, pip Maximum Load (Lbs.) <u>(650°F Maximum Temp.)</u> 610	e content, and insulation, are not to exceed Rod Diameter (inches) 3/8	the limits indicated.
Total weight of equ	ipment, including valves, fittings, pipe, pip Maximum Load (Lbs.) <u>(650°F Maximum Temp.)</u> 610 1130	e content, and insulation, are not to exceed Rod Diameter (inches) 3/8 1/2	the limits indicated.
Total weight of equ	ipment, including valves, fittings, pipe, pip Maximum Load (Lbs.) <u>(650°F Maximum Temp.)</u> 610 1130 1810	e content, and insulation, are not to exceed Rod Diameter (inches) 3/8 1/2 5/8	the limits indicated.
Total weight of equ	ipment, including valves, fittings, pipe, pip Maximum Load (Lbs.) (650°F Maximum Temp.) 610 1130 1810 2710	e content, and insulation, are not to exceed Rod Diameter (inches) 3/8 1/2 5/8 3/4	the limits indicated.
Total weight of equ	ipment, including valves, fittings, pipe, pip Maximum Load (Lbs.) (650°F Maximum Temp.) 610 1130 1810 2710 3770	e content, and insulation, are not to exceed Rod Diameter (inches) 3/8 1/2 5/8 3/4 7/8	the limits indicated.
Total weight of equ	ipment, including valves, fittings, pipe, pip Maximum Load (Lbs.) (650°F Maximum Temp.) 610 1130 1810 2710 3770 4960	e content, and insulation, are not to exceed Rod Diameter (inches) 3/8 1/2 5/8 3/4 7/8 1	the limits indicated.
Total weight of equ	ipment, including valves, fittings, pipe, pip Maximum Load (Lbs.) (650°F Maximum Temp.) 610 1130 1810 2710 3770 4960 8000	e content, and insulation, are not to exceed Rod Diameter (inches) 3/8 1/2 5/8 3/4 7/8 1 1-1/4	the limits indicated.
Total weight of equ Provide rods comple	ipment, including valves, fittings, pipe, pip Maximum Load (Lbs.) (650°F Maximum Temp.) 610 1130 1810 2710 3770 4960 8000 ete with adjusting and lock nuts.	e content, and insulation, are not to exceed Rod Diameter (inches) 3/8 1/2 5/8 3/4 7/8 1 1-1/4	the limits indicated.
Total weight of equ Provide rods comple	ipment, including valves, fittings, pipe, pip Maximum Load (Lbs.) (650°F Maximum Temp.) 610 1130 1810 2710 3770 4960 8000 ete with adjusting and lock nuts.	e content, and insulation, are not to exceed Rod Diameter (inches) 3/8 1/2 5/8 3/4 7/8 1 1-1/4	the limits indicated.
Total weight of equ Provide rods comple BEAM CLAMPS	ipment, including valves, fittings, pipe, pip Maximum Load (Lbs.) (650°F Maximum Temp.) 610 1130 1810 2710 3770 4960 8000 ete with adjusting and lock nuts.	e content, and insulation, are not to exceed Rod Diameter (inches) 3/8 1/2 5/8 3/4 7/8 1 1-1/4	the limits indicated.
Total weight of equ Provide rods comple BEAM CLAMPS MSS SP-58 Type 23 3/8, 1/2, and 5/8 in figure 86.	ipment, including valves, fittings, pipe, pip Maximum Load (Lbs.) (650°F Maximum Temp.) 610 1130 2710 3770 4960 8000 ete with adjusting and lock nuts. malleable black iron clamp for attachme ch diameter, for use with pipe sizes 4 inch	e content, and insulation, are not to exceed Rod Diameter (inches) 3/8 1/2 5/8 3/4 7/8 1 1-1/4 t to beam flange to 0.62 inches thick for s and less. Furnish with a hardened steel cup	the limits indicated. single threaded rods point set screw. An
Total weight of equ Provide rods comple BEAM CLAMPS MSS SP-58 Type 23 3/8, 1/2, and 5/8 int figure 86.	ipment, including valves, fittings, pipe, pip Maximum Load (Lbs.) (650°F Maximum Temp.) 610 1130 1810 2710 3770 4960 8000 ete with adjusting and lock nuts. malleable black iron clamp for attachme ch diameter, for use with pipe sizes 4 inch	e content, and insulation, are not to exceed Rod Diameter (inches) 3/8 1/2 5/8 3/4 7/8 1 1-1/4 t to beam flange to 0.62 inches thick for s and less. Furnish with a hardened steel cup	the limits indicated. single threaded rods point set screw. An
Total weight of equivalent of	ipment, including valves, fittings, pipe, pip Maximum Load (Lbs.) (650°F Maximum Temp.) 610 1130 1810 2710 3770 4960 8000 ete with adjusting and lock nuts. malleable black iron clamp for attachme ch diameter, for use with pipe sizes 4 inch or Type 29 forged steel jaw type clamp v mited in application to pipe sizes 8 inch at	e content, and insulation, are not to exceed Rod Diameter (inches) 3/8 1/2 5/8 3/4 7/8 1 1-1/4 t to beam flange to 0.62 inches thick for s and less. Furnish with a hardened steel cup ith a tie rod to lock clamp in place, suitable d less without prior approval. Anvil figure 2	the limits indicated. single threaded rods point set screw. An e for rod sizes to 1-1 228.
Total weight of equ Provide rods complet BEAM CLAMPS MSS SP-58 Type 23 3/8, 1/2, and 5/8 inter figure 86. MSS SP-58 Type 28 inch diameter but li	ipment, including valves, fittings, pipe, pip Maximum Load (Lbs.) (650°F Maximum Temp.) 610 1130 1810 2710 3770 4960 8000 ete with adjusting and lock nuts. malleable black iron clamp for attachme ch diameter, for use with pipe sizes 4 inch or Type 29 forged steel jaw type clamp v mited in application to pipe sizes 8 inch au	e content, and insulation, are not to exceed Rod Diameter (inches) 3/8 1/2 5/8 3/4 7/8 1 1-1/4 t to beam flange to 0.62 inches thick for s and less. Furnish with a hardened steel cup ith a tie rod to lock clamp in place, suitable d less without prior approval. Anvil figure 2	the limits indicated. single threaded rods of point set screw. Any e for rod sizes to 1-1, 228.
Total weight of equi Provide rods complet BEAM CLAMPS MSS SP-58 Type 23 3/8, 1/2, and 5/8 interfigure 86. MSS SP-58 Type 28 inch diameter but li CONCRETE INSERTS Carbon steel over	ipment, including valves, fittings, pipe, pip Maximum Load (Lbs.) (650°F Maximum Temp.) 610 1130 1810 2710 3770 4960 8000 ete with adjusting and lock nuts. malleable black iron clamp for attachme ch diameter, for use with pipe sizes 4 inch or Type 29 forged steel jaw type clamp v mited in application to pipe sizes 8 inch an	e content, and insulation, are not to exceed Rod Diameter (inches) 3/8 1/2 5/8 3/4 7/8 1 1-1/4 t to beam flange to 0.62 inches thick for s and less. Furnish with a hardened steel cup ith a tie rod to lock clamp in place, suitable d less without prior approval. Anvil figure 2 4 B633 zinc plating. Use drill bit of come m	the limits indicated. single threaded rods point set screw. Any e for rod sizes to 1-1, 228.

59 POST INSTALLED CONCRETE ANCHORS

60 Hilti, Mason Industries, Powers Fasteners, Simpson Strong-Tie Co. Cooper B-Line.

61

Drilled-in and stud-wedge or female-wedge type in zinc-coated steel for interior applications and stainless steel for exterior applications. Select anchor bolts with strength for anchor and as tested according to ASTM E488/E488M.

Expansion-type anchor bolts are not permitted for equipment more than 10 hp that is not vibration isolated. Undercut expansion anchors are permitted.

CORROSIVE ATMOSPHERE COATINGS

Factory coat supports and anchors used in corrosive atmospheres with hot dip galvanizing after fabrication, ASTM A123, 1.5 ounces/square foot of surface, each side. Mechanical galvanized threaded products, ASTM B695 Class 150, 2.0 mil coating. Field cuts and damaged finishes to be field covered with zinc rich paint of comparable thickness to factory coating.

Corrosive atmospheres include the following locations:

- Exterior locations
- Hazardous storage rooms
- Chemical waste/storage rooms

PART 3 - EXECUTION

INSTALLATION

Install supports to provide for free expansion of the piping and duct system. Support all piping from the structure using concrete inserts, beam clamps, ceiling plates, wall brackets, or floor stands. Fasten ceiling plates and wall brackets securely to the structure and test to demonstrate the adequacy of the fastening.

Install pipe hangers in accordance with MSS SP-58.

Piping shall be supported independently from ductwork and all other trades.

Where piping can be conveniently grouped to allow the use of trapeze type supports, use standard structural shapes for the supporting steel.

Perform all welding in accordance with standards of the American Welding Society. Clean surfaces of loose scale, rust, paint or other foreign matter and properly align before welding. Use wire brush on welds after welding. Welds shall show uniform section, smoothness of weld metal and freedom from porosity and clinkers. Where necessary to achieve smooth connections, joints shall be dressed smooth.

HANGER AND SUPPORT SPACING

Place a hanger within 12 inches of each horizontal elbow, valve, strainer, or similar piping specialty item.

Where several pipes can be installed in parallel and at the same elevation, provide multiple or trapeze hangers.

Support riser piping independently of connected horizontal piping.

Adjust hangers to obtain the slope specified in the piping section of this specification

Space hangers for pipe as follows:

Pipe Material	Pipe Size Max. Spacing	
Steel	1/2" through 1-1/4"	6'-6"
Steel	1-1/2" through 6"	10'-0"
Steel	8" through 12"	14'-0"
Copper	1/2" through 1-1/4"	5'-0"
Copper	1-1/2" and larger	8'-0"

VERTICAL RISER CLAMPS

Support vertical piping with clamps secured to the piping and resting on the building structure or secured to the building structure below at each floor.

END OF SECTION

1 2	SECTION 23 05 48 VIBRATION AND SEISMIC CONTROLS FOR HVAC PIPING AND EQUIPMENT
3 4	
5	PART1-GENERAL
7	SCOPE
8 9	This section includes specifications for vibration isolation material for equipment, piping systems, and duct systems. Included are the following topics:
10	
11	PART 1 - GENERAL
12	Scope
13	Related Work
14	Reference
15	Quality Assurance
16	Design Criteria
17	Shop Drawings
18	
70	PART 2 - PRODUCTS
20	Waterials
21	
22	Type 1. Neopletie rau
23	
24	Type 4. Housed Spring with Neoprene
25	Type 5. Spring range: with Neoprene
20	Type 0. Frecompressed spring with Neoprene
28	Flexible Pining Connections
20	Performance
30	Blower Minimum Deflection Guide
31	
32	PART 3 - FXECUTION
33	Installation
34	Packaged Air Handling Units and Centrifugal Fans
35	
36	RELATED WORK
37	Section 23 05 29 - Hangers and Supports for HVAC Piping and Equipment
38	Section 23 21 23 - Hydronic Pumps
39	Section 23 33 00 - Air Duct Accessories
40	Section 23 34 00 - HVAC Fans
41	Section 23 64 23 – Modular Heat Recovery Chillers
42	Section 23 73 13 - Modular Indoor Central-Station Air-Handling Units
43	
44	REFERENCE
45	Applicable provisions of Division 1 govern work under this section.
46	
47	QUALITY ASSURANCE
48	Refer to Division 1, General Conditions, Equals and Substitutions.
49	
50	DESIGN CRITERIA
51	solate all motor driven mechanical equipment from the building structure and nom the systems which they serve to prevent
52	equipment violations from being transmitted to the structure. Consider equipment weight distribution to provide dimoni-
57	
55	For equipment with variable speed capability, select vibration isolation devices based on the lowest speed
56	
57	Provide flexible nining connections for all nining to rotating or reciprocating equipment mounted on vibration isolators excent
58	do not use flexible nining connectors on any type of gas nining or with inline numps. Dining connected to a coil which is in an
59	assembly mounted on vibration isolators is to have flexible nining connections and nining vibration hangers as specified below
60	Piping connected to a coil which is in an assembly where the fan is senarately isolated by means of vibration isolators and duct
61	flexible connections does not require flexible piping connectors or piping vibration hangers.
62	

SHOP DRAWINGS

1 2 3

4

5 6

7

8 9 10

11

12

13 14

15

16 17

18 19

20

21 22

28

33

41

Refer to Division 1, General Conditions, Submittals.

Include isolator type, materials of construction, isolator free and operating heights, and isolation efficiency based on the lowest operating speed of the equipment supported.

PART 2 - PRODUCTS

MATERIALS

Use materials that will retain their isolation characteristics for the life of the equipment served. Use industrial grade neoprene for elastomeric materials.

Treat all isolators to resist corrosion. For isolation devices exposed to the weather or used in high humidity areas, hot dip galvanize steel parts, apply a neoprene coating on all steel parts, or use stainless steel parts; include limit stops to resist wind.

Provide pairs of neoprene side snubbers or restraining springs where side torque or thrust may develop.

Use isolators with a ratio of lateral to vertical stiffness not less than 1.0 or greater than 2.0.

23 VIBRATION ISOLATOR MANUFACTURERS

24 Mason Industries, Amber/Booth Co., Vibration Mounting & Controls, Peabody Noise Control, or approved equal. 25

26 **TYPE 1: NEOPRENE PAD:**

27 Neoprene waffle pad, 40 durometer with 16 gauge shims between layers.

29 TYPE 3: UNHOUSED SPRING WITH NEOPRENE

Combination freestanding, unhoused spring and neoprene with rib molded antifriction base. Include leveling bolts for securing
 to the equipment. Springs to be laterally stable under load and selected so they have an additional travel to solid equal to 50%
 of the rated deflection. Use height saving brackets when appropriate to the application.

34 **TYPE 4: RESTRAINED SPRING WITH NEOPRENE**

Combination spring and neoprene with rib molded base similar to Type 3 mount above, but with a housing that includes vertical limit stops to prevent spring extension when weight is removed such that the installed and operating heights are the same. Maintain a minimum clearance of 1/2" around restraining bolts, and between the housing and the spring, so as not to interfere with the spring action.

Design isolator so limit stops are out of contact during normal operation. Use height saving brackets when appropriate to the application.

42 TYPE 5: SPRING HANGER WITH NEOPRENE

Vibration hanger with a steel spring and 0.3" deflection neoprene element in series. Use neoprene element molded with a rod isolation bushing that passes through the hanger box. Select spring diameters and size hanger box lower holes large enough to permit the hanger rod to swing through a 30 degree arc before contacting the hole and short circuiting the spring. Select springs so they have a minimum additional travel to solid equal to 50% of the rated deflection.

48 TYPE 6: PRECOMPRESSED SPRING HANGER WITH NEOPRENE

Vibration hanger similar to Type 5 but precompressed to the rated deflection to keep the piping or equipment at a fixed elevation during installation. Design hanger with a release mechanism to free the spring after the installation is complete and the hanger is subjected to its full load.

53 **TYPE IB: INERTIA BASE:**

54 Rectangular structural beam or channel form, designed to accommodate poured concrete fill for use as floating foundation. Include support for suction and discharge base ells for split case pump bases. Use perimeter steel members with minimum 55 depth equal to 1/12 of longest dimension of base but not less than 6 inches; base depth need not exceed 12 inches unless 56 57 specifically recommended by base manufacturer for mass or rigidity. Include concrete reinforcements consisting of steel angles or ½ inch bars welded in place on 6 inch centers running in 2 layers perpendicular to each other and 1-1/2 inch above bottom; 58 59 provide additional steel if required by structural conditions. Furnish form with steel bolting templates and anchor bolt sleeves 60 to receive equipment anchor bolts where anchor bolts fall in concrete locations. Use height saving brackets to maintain base clearance of at least 1 inch above floor or housekeeping pad. 61

62

1 FLEXIBLE PIPING CONNECTIONS

2 Suitable for pressure, temperature, and fluid involved; minimum pressure rating for any system is 125 psig at the design 3 temperature of the fluid. Use 12 inch minimum line length of flexible hose or length required to absorb 3/4" lateral movement,

- 4 whichever is greater.
- 5

6 MANUFACTURERS:

7 Flexonics, Mason, Mercer Rubber, Metraflex, or approved equal.

8 9 WATER:

Multiple plies of nylon tire cord fabric reinforced with an EPDM cover and liner. Do not use steel wire or rings as pressure reinforcement. Use threaded or soldered connections for sizes 2" and smaller and floating steel or ductile iron flanges for sizes 2-1/2" and larger; design the steel flange end so the steel flange is recessed to lock a steel wire bead ring in the raised face of the EPDM flange. Construct straight-through connections with twin spheres. Use control rods when recommended by the manufacturer.

1516 **PERFORMANCE**

Select vibration isolation devices as indicated below or to provide not less than 95% isolation efficiency, whichever is greater.

19									
20				Floor Spa	n or Colun	nn Spacin	g		
21		On 6	Grade	20 I	eet	30 F	eet	40 F	eet
22			Min.		Min.		Min.		Min.
23			Static		Static		Static		Static
24		lso.	Defl.	lso.	Defl.	lso.	Defl.	lso.	Defl.
25	TYPE OF EQUIPMENT	Type	In.	Type	In.	Type	ln.	Type	ln.
	REFRIGERATION MACHINES								
		1	0.10	4	0.75	4	0.75	4	1.50
		Crout							
	PUMPS, CLOSE COUPLED:	Grout		2 10	0.75	2 10	0.75	2 10	0.75
	Thru 5 HP	to Pad		3-IB	0.75	3-IB	0.75	3-IB	0.75
		Grout							
	7 ½ HP and Over	to Pad		3-IB	0.75	3-IB	1.5	3-IB	1.5

26

30

31

- 27 SQUARE IN-LINE,
- 28 MIXED FLOW:
- 29 Suspended

Use Type 4 mount with deflection from blower minimum deflection guide up to 0.75 inch deflection. Over .75 inch deflection, use Type 4-S mount.

- 32 33
- PIPING CONNECTED TOEQUIPMENT:
- 34 35 36

BLOWER MINIMUM DEFLECTION GUIDE

	Required Deflection (Inches)					
		On	20'	30'	40'	
Fan Speed (RPM)	Grade		Floor Span	Floor Span	Floor Span	
175-224		0.35	3.50	4.50	4.50	
225-299		0.35	3.50	3.50	3.50	
300-374		0.35	2.50	2.50	3.50	
375-499		0.35	1.50	2.50	3.50	
500 and over		0.35	0.75	1.50	2.50	

Flexible piping connections.

PART3-EXECUTION

4950 INSTALLATION

51 Install vibration isolation devices for motor driven equipment in accordance with the manufacturer's installation instructions. 52

53 Set steel and inertia bases for one inch clearance between the concrete floor or housekeeping pad and the base.

Do not allow installation practices to short circuit any isolation device.

Install flexible piping connections on the equipment side of shut-off valves.

PACKAGED AIR HANDLING UNITS AND CENTRIFUGAL FANS

Attach horizontal thrust restraints at the centerline of thrust and symmetrically on either side of the unit. Thrust restraints are not required when the fan section in not isolated from the remainder of the air handling unit by means of duct flexible connections.

8 9 10

1

2 3

4 5

6 7

END OF SECTION

1 2 3	SECTION 23 05 93 TESTING, ADJUSTING, AND BALANCING FOR HVAC
4 5	PART1-GENERAL
6 7 8	SCOPE This section includes air and water testing, adjusting and balancing for the entire project. Included are the following topics:
9 10	PART 1 - GENERAL
11	Scope
12	Related Work
13	Reference
14	Reference Standards
15	Description
16	Quality Assurance
1/	Pre-Installation Meeting and Scheduling
18	Pre-Balance Conference
20	Sublitudis
20	PART 2 - PRODUCTS
22	Instrumentation
23	
24	PART 3 - EXECUTION
25	Preliminary Procedures
26	Performing Testing, Adjusting and Balancing
27	VAV Supply Duct System Static Pressure Set Point
28	Hydronic System Differential Pressure Control Set Point
29	Deficiencies
3U 21	Functional Performance Testing
32	RELATED WORK
33	Section 23 05 00 Common Work Results for HVAC
34	Section 23 07 00 HVAC Insulation
35	Section 23 09 14 Electric Instrumentation and Control Devices for HVAC
36	Section 23 09 24 Direct Digital Control System for HVAC
37	
38	REFERENCE
39	Applicable provisions of the General Conditions, Supplementary General Conditions and General Requirements in Division 1
40 41	govern work under this section.
41 //2	REFERENCE STANDARDS
42 43	AABC National Standards for Total System Balance Sixth Edition 2002
44	ASHRAE ASHRAE Handbook. 2007 HVAC Applications. Chapter 37. Testing Adjusting and Balancing.
45	NEBB Procedural Standards for Testing Adjusting Balancing of Environmental Systems, Seventh Edition, 2005.
46	TABB Tab Procedural Guide, First Edition, 2003.
47	
48	DESCRIPTION
49	The Contractor will separately contract with an independent test and balance agency to perform all testing, adjusting, and
50	balancing of air and hydronic systems required for this project. Work related to the testing, adjusting, and balancing that must
51 51	be performed by the installing mechanical contractor is specified in other section of these specifications.
52 52	distribution adjustment of new and existing systems and equinment to provide design requirements indicated on the drawings
55 54	electrical measurement and verification of performance of all mechanical equipment, all in accordance with standards
55	published by AABC, NEBB, or TABB.
56	
57	Test, adjust and balance all air and hydronic systems so that each room, piece of equipment or terminal device meets the
58	design requirements indicated on the drawings and in the specifications.
59	A DEFENSION DE LA COMPANIA DE LA COMP
6U 61	Accomplish testing, adjusting and balancing work in a timely manner that allows partial occupancy of major buildings,
62	the Instruction to Bidders and in accordance with the completion schedule established for this project.

Verify that provisions are being made to accomplish the specified testing, adjusting and balancing work. If problems are found, handle as specified in Part 3 under Deficiencies.

QUALITY ASSURANCE

Qualifications

1 2

3 4

5

6 7

8

9 10

11 12

13 14 15

16 17

18

30

33

41

42

43

44 45

46 47

51

56

59

An independent Firm specializing in the Testing and Balancing of HVAC systems for a minimum of 3 years. A Firm not engaged in the commerce of furnishing or providing equipment or material generally related to HVAC work other then that specifically related to installing Testing and Balancing components necessary for work in this section such as, but not limited to sheaves, pulleys, and balancing dampers.

A certified member of AABC or certified by NEBB or TABB in the specific area of work performed. Maintain certification for the entire duration of the project. If certification of firm or any staff performing work is terminated or expires during the duration of the project, contact DFD immediately.

Technicians on this project must have satisfactorily completed work on a minimum of (3) three projects of at least 50% in size, and of similar complexity. Size is defined as the quantity of each specific individual item requiring testing and balancing such as, but not limited to, equipment, devices, terminal devices, and grilles and diffusers.

19 PRE-INSTALLATION MEETING AND SCHEDULING

The test and balance agency is required to attend a pre-installation meeting with all other project contractors before the construction process is started. The test and balance agency shall give the Lead Contractor a detailed schedule of testing and balancing tasks for incorporation into the project schedule.

24 **PRE-BALANCE CONFERENCE**

90 days prior to beginning testing, adjusting and balancing, schedule and conduct a meeting with the HVAC contractor, General Contractor, Temperature Control Contractor and Architect/Engineer. Provide a complete copy of the TAB plan for the project. The objective is final coordination and verification of system operation and readiness for testing, adjusting and balancing procedures and scheduling procedures with the above mentioned parties. Indicate work required to be completed prior to testing, adjusting, and balancing and identify the party responsible for completion of that work.

31 SUBMITTALS

32 See also Related Work in this section.

Submit testing, adjusting and balancing reports bearing the seal and signature of the NEBB, AABC or TABB Certified Test and Balance Supervisor. The reports certify that the systems have been tested, adjusted and balanced in accordance with the referenced standards; are an accurate representation of how the systems have been installed and are operating; and are an accurate record of all final quantities measured to establish normal operating values of the systems.

39 <u>Format</u>: Cover page identifying project name, project number and descriptive title of contents. Divide the contents of the 40 report into the below listed divisions:

- General Information
- Summary
- Air Systems
- Hydronic Systems

<u>Contents</u>: Provide the following minimum information, forms and data:

General Information: Inside cover sheet identifying Test and Balance Agency, Contractor, Architect, Engineer, Project Name and
 Project Number. Include addresses, contact names and telephone numbers. Also include a certification sheet containing the
 seal and signature of the Test and Balance Supervisor.

52 Summary: Provide summary sheet describing mechanical system deficiencies. Describe objectionable noise or drafts found 53 during testing, adjusting and balancing. Provide recommendations for correcting unsatisfactory performances and indicate 54 whether modifications required are within the scope of the contract, are design related or installation related. List 55 instrumentation used during testing, adjusting and balancing procedures.

57 The remainder of the report to contain the appropriate standard NEBB, AABC, or TABB forms for each respective item and 58 system. Fill out forms completely. Where information cannot be obtained or is not applicable indicate same.

PART 2 - PRODUCTS

INSTRUMENTATION 1

Provide all required instrumentation to obtain proper measurements. Application of instruments and accuracy of instruments 2 and measurements to be in accordance with the requirements of NEBB, AABC, or TABB Standards and instrument 3 manufacturer's specifications. 4

All instruments used for measurements shall be accurate, and calibration histories for each instrument to be available for 6 7 examination by the owner or AE upon request. Calibration and maintenance of all instruments to be in accordance with the 8 requirements of NEBB, AABC, or TABB Standards

PART 3 - EXECUTION

PRELIMINARY PROCEDURES 13

14 Review preconstruction meeting report, applicable construction bulletins, applicable change orders and approved shop 15 drawings of equipment, outlets/inlets and temperature controls.

17 Check filters for cleanliness, dampers and valves for correct positioning, equipment for proper rotation and belt tension, 18 temperature controls for completion of installation and hydronic systems for proper charge and purging of air.

19

27

28

29

30

31

32 33

34

35

36

37

38 39

40

41

42

44 45

16

5

9 10

11 12

Notify General Contractor on a daily basis during balancing. Identify deficiencies preventing completion of testing, adjusting and 20 21 balancing procedures. Do not proceed until systems are fully operational with all components necessary for complete testing, adjusting and balancing. Installing Contractors are required to provide personnel to check and verify system completion, 22 23 readiness for balancing and assist Balancing Agency in providing specified system performance. 24

25 EXISTING EQUIPMENT

26 The following shall be tested and balanced:

Existing Exhaust Fans EF-7

0

- Rebalance each exhaust fan airflow. 0
- AHU-1 (E)
 - Rebalance supply fan airflow. 0
 - Rebalance minimum and maximum outside airflow.
 - Coordinate outside air dampers positions with temperature controls contractor.
 - Rebalance chilled water coil. 0
 - Rebalance all grilles, registers, and diffusers associated with this air handling system as noted on the 0 Drawings.
 - BC-1 (E) & BC-2 (E)
 - Rebalance hot water coil. 0
- RF-1 (E)
 - Rebalance return fan airflow. 0
- AHU-2 (E)
 - 0 Rebalance chilled water coil.
- AHU-3 (E) 43
 - Rebalance chilled water coil. 0

PERFORMING TESTING, ADJUSTING AND BALANCING

Perform testing, adjusting and balancing procedures on each system identified, in accordance with the detailed procedures 46 outlined in the referenced standards except as may be modified below. 47

48 Unless specifically instructed in writing, all work in this specification section is to be performed during the normal workday.

49

In areas containing ceilings, remove ceiling tile to accomplish balancing work; replace tile when work is complete and provide 50 new tile for any tile that are damaged by this procedure. If the ceiling construction is such that access panels are required for 51 52 the work of this section and the panels have not been provided, inform the owner's project representative.

53

54 Cut insulation, ductwork and piping for installation of test probes to the minimum extent necessary for adequate performance 55 of procedures. Patch using materials identical to those removed, maintaining vapor barrier integrity and pressure rating of 56 systems.

57

In air systems employing filters, blank off sufficient filter area to simulate a pressure drop that is midway between that of a 58 59 clean filter and that of a dirty filter.

60

61 Measure and record system measurements at the fan and/or pump to determine total flow. Adjust equipment as required to yield specified total flow at terminals. Proceed taking measurements in mains and branches as required for final terminal 62 63 balancing. Perform terminal balancing to specified flows balancing branch dampers, deflectors, extractors and valves prior to

64 adjustment of terminals. Measure and record static air pressure conditions across fans, coils and filters. Indicate in report if cooling coil measurements were made on a wet or dry coil and if filter measurements were made on a clean or dirty filter. Spot check static air pressure conditions directly ahead of terminal units.

Adjust outside air, return air and relief air dampers for design conditions at both the minimum and maximum settings and record both sets of data. Balance modulating dampers at extreme conditions and record both sets of data. Balance variable air volume systems at maximum air flow rate, full cooling, and minimum flow rate, full heating; record all data.

Adjust register, grille and diffuser vanes and accessories to achieve proper air distribution patterns and uniform space temperatures free from objectionable noise and drafts within the capabilities of the installed system.

Provide fan and motor drive sheave adjustments necessary to obtain design performance. Provide drive changes specifically noted on drawings, if any. If work of this section indicates that any drive or motor is inadequate for the application, advise the owner's project representative by giving the representative properly sized motor/drive information (in accordance with manufacturers original service factor and installed motor horsepower requirements); Confirm any change will keep the duct/piping system within its design limitations with respect to speed of the device and pressure classification of the distribution system.

Required motor/drive changes not specifically noted on drawings or in specifications will be considered an extra cost and will require an itemized cost breakdown submitted to owner's project representative. Prior authorization is needed before this work is started.

Areas or rooms designed to maintain positive, negative or balanced air pressures with respect to adjacent spaces, as indicated by the design air quantities, require special attention. Adjust fan drives, distribution dampers, terminals and controls to maintain indicated pressure relationship.

Final air system measurements to be within the following range of specified cfm:

Fans	0% to +10%
Supply grilles, registers, diffusers	0% to +10%
Return/exhaust grilles, registers	0% to -10%

Final water system measurements must be within the following range of specified gpm:

Heating flow rates	0% to -10%
Cooling flow rates	-5% to +5%

Contact the temperature control Contractor for assistance in operation and adjustment of controls during testing, adjusting and balancing procedures. Cycle controls and verify proper operation and setpoints. Include in report description of temperature control operation and any deficiencies found.

Permanently mark equipment settings, including damper and valve positions, control settings, and similar devices allowing settings to be restored. Set and lock memory stops. 43

Leave systems in proper working order, replacing belt guards, closing access doors and electrical boxes, and restoring 44 temperature controls to normal operating settings.

Verify and record, in the T&B Report, "K" factors for all VAV air terminal devices and air flow stations.

Coordinate air handling unit minimum outside air set points with the Temperature Control Contractor.

VAV SUPPLY DUCT SYSTEM STATIC PRESSURE SET POINT

52 For VAV supply systems with VAV air terminal devices, determine the minimum required duct static pressure at the DDC static pressure sensor location(s) needed to ensure that all VAV air terminals are operating at their design airflows with the most 53 demanding VAV terminal wide open. Provide these static pressure numbers to the DDC temperature controls contractor and 54 record them in the T&B report for each system. 55

57 HYDRONIC SYSTEM DIFFERENTIAL PRESSURE CONTROL SET POINT

For hydronic systems with variable speed pumping, determine the minimum required system differential pressure set point 58 59 needed to insure that all terminal devices are operating at their design water flows with the most demanding terminals device control valve wide open. Provide the differential control setting set point to the DDC temperature control contractor and record 60 them in the T&B report for each system. 61 62

For HVAC pumps 10 horsepower or less, valve throttling alone may be used for hydronic system balancing.

45

46 47

48 49

50

51

56

63

64

1 Throttling of triple-duty valves shall not exceed 50% closed. Where additional throttling would be necessary to achieve the 2 system design flow the impellor shall be trimmed.

Verify Triple duty valve utilized on systems with Variable Frequency Drives are 100% open when balancing work is complete. 4

The pressure drop across triple duty valves shall not exceed 25 ft. w.g. Where additional throttling would be necessary to 6 7 achieve the system design flow the impellor shall be trimmed.

9 For HVAC pumps greater than 10 horsepower through 60 horsepower, trim the impellor where valve throttling will result in a 10 draw that exceeds 3 horsepower. 11

12 Future fouling of an open piping system may be considered when determining impellor trim requirements. 13

14 Verify butterfly valves utilized for hydronic system balancing are provided with position-lock operators (memory stops) in accordance with Section 23 05 23. The adjustment and marking of lever-lock operators that use throttling notches will not be 15 16 accepted. Lock all memory stops so the valves can be reopened to their balanced positions if they are used for isolation 17 purposes.

18 19 DEFICIENCIES

20 Division 23 00 00 contractor to correct any installation deficiencies found by the test and balance agency that were specified 21 and/or shown on the Contract Documents to be performed as part of that division of work. All corrective work shall be done at 22 no additional cost to the Owner. 23

24 Retest mechanical systems, equipment, and devices once corrective work is complete as specified.

FUNCTIONAL PERFORMANCE TESTING 26

Contractor is responsible for utilizing the functional performance test forms supplied by the third party commissioning agent in 27 28 accordance with the procedures defined for functional performance testing in Section 01 91 00.

29

25

3

5

8

30 31

END OF SECTION

Page Intentionally Left Blank

1	SECTION 23 07 00						
2	HVAC INSULATION						
3							
4							
6	FARTISOENERAL						
7	SCOPE						
8 9	This section includes insulation specifications for heating, ventilating and air conditioning piping, ductwork and equipment. Included are the following topics:						
10							
11	PART 1 - GENERAL						
12	Scope Related Work						
13	Reference Standards						
15	Quality Assurance						
16	Description						
17	Definitions						
18	Shop Drawings						
19	Operation and Maintenance Data						
20							
22	PART 2 - PRODUCTS						
23	Materials						
24	Insulation Types						
25	Jackets						
26	Insulation inserts and Pipe Shields						
28							
29	PART 3 - EXECUTION						
30	Examination						
31	Installation						
32	Protective Jacket Installation Bining, Valve and Eitting Insulation						
33	Pining Protective lackets						
35	Pipe Insulation Schedule						
36	Duct Insulation						
37	Ductwork Protective Coverings						
38	Duct Insulation Schedule						
39 40	Equipment Insulation						
40	Construction Verification Items						
42							
43	RELATED WORK						
44	Section 23 05 00 - Common Work Results for HVAC						
45	Section 23 05 29 - Hangers and Supports for HVAC Piping and Equipment						
40 47	Section 23 31 00 - HVAC Ducts and Casings						
48							
49	REFERENCE						
50	Applicable provisions of Division 1 govern work under this section.						
51							
52 53	ASTM B209 Aluminum and Aluminum Alloy Sheet and Plate						
55	ASTM C165 Test Method for Compressive Properties of Thermal Insulations						
55	ASTM C177 Heat Flux and Thermal Transmission Properties						
56	ASTM C302 Density of Preformed Pipe Insulation						
57	ASTM C303 Density of Preformed Block Insulation						
58 50	ASTIN C4355 LEST METHODS FOR LEST FOR WATER VAPOR TRANSMISSION OF THICK MATERIALS						
60	ASTM C518 Heat Flux and Thermal Transmission Properties						
61	ASTM C534 Preformed Flexible Elastomeric Thermal Insulation						
62	ASTM C578 Preformed, Block Type Cellular Polystyrene Thermal Insulation						
63	ASTM C591 Unfaced Preformed Rigid Cellular Polyisocyanurate Thermal Insulation						
64	ASIM C921 Properties of Jacketing Materials for Thermal Insulation						

ASTM C1136

1

2	ASTM D412	Standard Test Methods for Vulcanized Rubber and Thermoplastic Elastomers-Tension Matheds for Prossure Sonsitive Adhesive Coated Tapes Used for Electrical and Electronic Applications
3	ASTIVI D1000	Methods for Pressure-sensitive Aunesive-coaled rapes used for Electrical and Electronic Applications
4	ASTIVI D1621	Standard Test Method for Compressive Properties of Right Cellular Plastics
5	ASTM D1622	Standard Test Method for Apparent Density of Rigid Cellular Plastics
6	ASTM D1940	Method of Lest for Porosity of Rigid Cellular Plastics
/	ASTM D2126	Method for Response of Rigid Cellular Plastics to Thermal and Humid Aging
8	ASTM D2240	Standard Test Method for Rubber Property—Durometer Hardness
9	ASTM E84	Surface Burning Characteristics of Building Materials
10	ASTM E814	Standard Test Method for Fire Tests of Penetration Firestop Systems
11	MICA	National Commercial & Industrial Insulation Standards
12	NFPA 225	Surface Burning Characteristics of Building Materials
13	UL 723	Surface Burning Characteristics of Building Materials
14		
15	QUALITY ASSUR	ANCE
16	Refer to Divisior	1, General Conditions, Equals and Substitutions
17		
18 19	Label all insulati	ng products delivered to the construction site with the manufacturer's name and description of materials.
20	Insulation syste	ms shall be applied by experienced contractors. Within the past five (5) years, the contractor shall be able to
21	document the s	uccessful completion of a minimum of three (3) projects of at least 50% of the size and similar scope of the work
22	specified in this	section
22	specifica in this	Section.
23		
24	Eurnich and inst	all all inculating materials and accessories as specified or as required for a complete installation. The following
25	types of inculati	an an insulating matchais and accessories as spectred of as required for a complete installation. The following
20		in a resultion
27	• P	ipe insulation
28	• 0	luct insulation
29	• E	quipment Insulation
30		
31	Install all insula	tion in accordance with the latest edition of MICA (Midwest Insulation Contractors Association) Standard and
32	manufacturer's	installation instructions.
33		
34	DEFINITIONS	
35 36	Concealed: shaf through tunnels	ts, furred spaces, space above finished ceilings, utility tunnels and crawl spaces. All other areas, including walk- , shall be considered as exposed.
37		
38	SHOP DRAWING	SS Contraction of the second
39	Refer to Divisior	1, General Conditions, Submittals.
40		
41	Submit a schedu	Ile of all insulating materials to be used on the project, including adhesives, fastening methods, fitting materials
42	along with mat	erial safety data sheets and intended use of each material. Include manufacturer's technical data sheets
43	indicating densit	ty, thermal characteristics, jacket type, and manufacturer's installation instructions.
44		
45	OPERATION AN	D MAINTENANCE DATA
46	All operations a	and maintenance data shall comply with the submission and content requirements specified under section
47	GENERAL REQU	IREMENTS.
48		
49	ENVIRONMENT	
50	Do not store in	sulation materials on grade or where they are at risk of becoming wet. Do not install insulation products that
51	have been expo	send to water
52	nave been expo	
52	Protoct installos	l insulation work with plastic shooting to provent water damage
52		insulation work with plastic sheeting to prevent water damage.
54 FF		
55		
50		PARIZ - PRODUCIS
5/		

Flexible Low Permeance Vapor Retarders for Thermal Insulation

58 MATERIALS

Manufacturers: Armacell, Certainteed, Manson, Childers, Dow, Extol, Fibrex, Halstead, H.B. Fuller, Imcoa, Johns Manville,
 Knauf, Owens-Corning, Partek, Pittsburgh Corning, Rubatex, VentureTape or approved equal.

Materials or accessories containing asbestos will not be accepted.

- 1 Use composite insulation systems (insulation, jackets, sealants, mastics, and adhesives) that have a flame spread rating of 25 or 2 less and smoke developed rating of 50 or less, with the following exceptions:
- 2 3

Pipe insulation which is not located in an air plenum may have a flame spread rating not over 25 and a smoke developed rating
 no higher than 450 when tested in accordance with UL 723 and ASTM E84.

7 INSULATION TYPES

8 Insulating materials shall be fire retardant, moisture and mildew resistant, and vermin proof. Insulation shall be suitable to 9 receive jackets, adhesives and coatings as indicated.

10 11 FLEXIBLE FIBERGLASS INSULATION:

12 Minimum nominal density of 0.75 lbs. per cu. ft., and thermal conductivity of not more than 0.3 at 75 degrees F, rated for 13 service to 250 degrees F.

14

15 RIGID FIBERGLASS INSULATION:

16 Minimum nominal density of 3 lbs. per cu. ft., and thermal conductivity of not more than 0.23 at 75 degrees F, minimum 17 compressive strength of 25 PSF at 10% deformation, rated for service to 450 degrees F.

18 19 BLOCK INSULATION:

Owens-Corning Kaylo 10, Schuller Thermo-12 hydrous calcium silicate, ASTM C533, Type I, acceptable for use to a maximum operating temperature of 1,200²/₂degrees²/₂F. Material to be visually coded or marked to indicate it is asbestos free. Thermal conductivity shall not be more than 0.44 at 300²/₂degrees²/₂F, with dry density 12.5 lbs. per cu. ft. minimum.

Provide jacket of 6 oz. per sq. yd. fiberglass cloth embedded in 2 coats of weatherproof mastic. Field or factory applied stainless steel or aluminum jacket may be substituted for fiberglass cloth jacket at contractors option.

25

26 ELASTOMERIC INSULATION:

Flexible closed cell, minimum nominal density of 5.5 lbs. per cu. ft., thermal conductivity of not more than 0.27 at 75 degrees F, minimum compressive strength of 4.5 psi at 25% deformation, maximum water vapor permeability of 0.17 perm inch, maximum water absorption of 6% by weight, rated for service range of -20 degrees F to 220 degrees F on piping and 180 degrees F where adhered to equipment.

31 32 EXTRUDED POLYSTYRENE INSULATION:

Rigid closed cell, minimum nominal density of 1.6 lbs. per cu. ft., thermal conductivity of not more than 0.285 at 75 degrees F,
 minimum compressive strength of 20 psi, maximum water vapor permeability of 1.5 perm inch, maximum water absorption of
 .5 % by volume, rated for service range of -290 degrees F to 165 degrees F.

3637 POLYISOCYANURATE INSULATION:

Rigid closed cell polyisocyanurate, minimum nominal density of 2.0 lbs. per cu. ft., thermal conductivity of not more than 0.19
 at 75 degrees F aged 180 days, minimum compressive strength of 24 psi parallel and 13 psi perpendicular, maximum water
 vapor permeability of 4 perm inch, maximum water absorption of 2% by volume, rated for service range of -290 degrees F to
 300 degrees F.

- 42 43 **JACKETS**
- 44 PVC FITTING COVERS AND JACKETS (PFJ):

White PVC film, gloss finish one side, semi-gloss other side, FS LP-535D, Composition A, Type II, Grade GU. Ultraviolet inhibited indoor/outdoor grade to be used where exposed to high humidity, ultraviolet radiation, in kitchens or food processing areas or installed outdoors. Jacket thickness to be minimum .02" indoors/.03"outdoors for piping 12" and smaller, .03" indoors/.04"

48 outdoors for piping 15" and larger.

49 50 ALL SERVICE JACKETS (ASJ):

Heavy duty, fire retardant material with white kraft reinforced foil vapor barrier, factory applied to insulation with a self-sealing pressure sensitive adhesive lap, maximum permeance of .02 perms and minimum beach puncture resistance of 50 units.

- 53
- 54 FOIL SCRIM ALL SERVICE JACKETS (FSK):

55 Glass fiber reinforced foil kraft laminate, factory applied to insulation. Maximum permeance of .02 perms and minimum beach 56 puncture resistance of 25 units. 57

- 58 SELF-ADHERING JACKETS (SAJ):
- 59 5-ply, self-adhering multiple laminated waterproofing material with reflective aluminum foil, high density polymer films and
- cold weather acrylic adhesive providing zero (0.0) permeability. Minimum 6 mils material thickness, 35lb puncture resistance
- cold weather acrylic adhesive providing zero (0.0) permeability. Minimum 6 mils material thickness, 35lb puncture resistance
 when tested in accordance with ASTM D1000 and flame spread/smoke developed rating of 10/20 when tested in accordance
 with UL 723.
- 63

2

3

4 5

6

7

8

9 10

11 12

15

27

28 29

30

31 32 33

34

35

36

37 38 39

40 41 42

43 44

45 46

47 48

49 50

51 52

53

54 55 56

57 58 59

60

61

62 63

64

Vapor retarding tape shall be specifically designed and manufactured for use with the self-adhering jacket specified above. Tape shall be provided by the same manufacturer that provides jacketing. Vapor retarding tapes used with self-adhering jackets shall have a maximum permeance of 0.0 perms.

VAPOR RETARDING JACKETS (VRJ):

Polyvinylidene chloride (PVDC) vapor retarding jacket material with minimum 6 mils material thickness and maximum permeance of 0.01 perms. Material shall not support the growth of mold or mildew. Dow Saran or equivalent.

Vapor retarding tape shall be specifically designed and manufactured for use with the vapor retarding jacket specified above. Tape shall be provided by the same manufacturer that provides jacketing. Vapor retarding tapes used with vapor retarding jackets shall have a maximum permeance of 0.01 perms.

13 INSULATION INSERTS AND PIPE SHIELDS

14 Manufacturers: B-Line, Pipe Shields, Value Engineered Products

Construct inserts with polyisocyanurate (service temperatures below 300 degrees F only), minimum 140 psi compressive strength. Piping 12" and larger, supplement with high density 600 psi structural calcium silicate insert. Provide galvanized steel shield. Insert and shield to be minimum 180 degree coverage on bottom supported piping and full 360 degree coverage on clamped piping. On roller mounted piping and piping designed to slide on support, provide additional load distribution steel plate.

Where contractor proposes shop/site fabricated inserts and shields, submit schedule of materials, thicknesses, gauges and
 lengths for each pipe size to demonstrate equivalency to pre-engineered/premanufactured product described above.
 On low temperature systems, high density rigid polyisocyanurate may be substituted for calcium silicate provided insert and

shield length and shield gauge are increased to compensate for lower insulation compressive strength.

Precompressed 20# density molded fiberglass blocks, Hamfab or equal, of the same thickness as adjacent insulation may be substituted for calcium silicate inserts with one 1"x6" block for piping through 2-1/2" and three 1"x6" blocks for piping through 4". Submit shield schedule to demonstrate equivalency to pre-engineered/premanufactured product described above.

Wood blocks will not be accepted.

ACCESSORIES

All products shall be compatible with surfaces and materials on which they are applied, and be suitable for use at operating temperatures of the systems to which they are applied.

Adhesives, sealants, and protective finishes shall be as recommended by insulation manufacturer for applications specified.

Insulation bands to be 3/4 inch wide, constructed of aluminum or stainless steel. Minimum thickness to be .015 inch for aluminum and .010 inch for stainless steel.

Tack fasteners to be stainless steel ring grooved shank tacks.

Staples to be clinch style.

Insulating cement to be ANSI/ASTM C195, hydraulic setting mineral wool.

Finishing cement to be ASTM C449.

Bedding compounds to be non-shrinking and permanently flexible.

Vapor barrier coatings to have maximum applied water vapor permeance of .05 perms.

Fungicidal water base coating (Foster 40-20 or equal) to be compatible with vapor barrier coating.

PART 3 - EXECUTION

EXAMINATION

Verify that all piping, equipment, and ductwork are tested and approved prior to installing insulation. Do not insulate systems until testing and inspection procedures are completed.

Verify that all surfaces are clean, dry and without foreign material before applying insulation materials.

1 INSTALLATION

All materials shall be installed by skilled labor regularly engaged in this type of work. All materials shall be installed in strict accordance with manufacturer's recommendations, building codes, and industry standards. Do not install products when the ambient temperature or conditions are not consistent with the manufacturer's recommendations. Surfaces to be insulated must be clean and dry.

6

9

12

14

17

23

25

26

27

28

29

31

Locate insulation and cover seams in the least visible location. All surface finishes shall be extended in such a manner as to
 protect all raw edges, ends and surfaces of insulation.

Install insulation with smooth and even surfaces. Poorly fitted joints or use of filler in voids will not be accepted. Provide neatly
 beveled and coated terminations at all nameplates, uninsulated fittings, or at other locations where insulation terminates.

13 Install fabric reinforcing without wrinkles. Overlap seams a minimum of 2 inches.

Use full length material (as delivered from manufacturer) wherever possible. Scrap piecing of insulation or pieces cut undersize
 and stretched to fit will not be accepted.

All pipe and duct insulation shall be continuous through walls, ceiling or floor openings and through sleeves except where
 firestop or firesafing materials are required. Vapor barriers shall be maintained continuous through all penetrations.

Provide a continuous unbroken moisture vapor barrier on insulation applied to systems noted below. Attachments to cold surfaces shall be insulated and vapor sealed to prevent condensation.

24 Provide a complete vapor barrier for insulation on the following systems:

- Cold Water Make-Up
 - Chilled Water
 - Insulated Duct
 - Equipment, ductwork or piping with a surface temperature below 65 degrees F

30 **PROTECTIVE JACKET INSTALLATION**

32 SELF-ADHERING JACKETS (SAJ):

Install according to manufacturer's recommendations. Cut allowing minimum 4" overlap on ends and 6" on longitudinal joints.
 Align parallel to surface. Remove release paper and press flat to surface to avoid wrinkles. Rub entire surface for full adhesion and sealing at joint overlaps. On exterior applications, provide a bead of compatible caulk along exposed edges.

36

Piping with self-adhering (SAJ) jackets shall have elbows, fittings, valves and butt joints wrapped with 2 layers of vapor retarding tape. Piping with a PVC jacket (PFJ) installed over the self-adhering (SAJ) jacket may be provided with a single, lapped layer of vapor retarding tape for elbows, fittings and valves under the PVC jacket. Vapor retarding tape shall be compatible with the jacket material used.

41

42 VAPOR RETARDING JACKETS (VRJ):

Piping with vapor retarding (VRJ) jackets shall have elbows, fittings, valves and butt joints wrapped with 2 layers of vapor retarding tape. Piping with a PVC jacket (PFJ) installed over the vapor retarding (VRJ) jackets may be provided with a single, lapped layer of vapor retarding tape for elbows, fittings and valves under the PVC jacket. Vapor retarding tape shall be compatible with the jacket material used.

47

48 PVC FITTING COVERS AND JACKETS (PFJ):

Lap seams and joints a minimum of 2 inches and continuously seal PVC with welding solvent recommended by jacket manufacturer. Lap slip joint ends 4" without fasteners where required to absorb expansion and contraction. For sections where vapor barrier is not required and jacket requires routine removal, tack fasteners may be used. Secure PVC fitting covers with tack fasteners. For systems requiring a vapor barrier, apply a 1-1/2" band of mastic over ends, throat, seams and penetrations.

53

54 **PIPING, VALVE, AND FITTING INSULATION**

55 56 GENERAL:

57 Install insulation with butt joints and longitudinal seams closed tightly. Provide minimum 2" lap on jacket seams and 2" tape on

58 butt joints, firmly cemented with lap adhesive unless otherwise noted. Additionally secure with staples along seams and butt 59 joints. Coat staples, longitudinal and transverse seams with vapor barrier mastic on systems requiring vapor barrier.

60

61 Install insulation continuous through pipe hangers and supports with hangers and supports on the exterior of insulation. Where

62 a vapor barrier is not required or where roller hangers are not being used, hangers and supports may be attached directly to

63 piping with insulation completely covering hanger or support and jacket sealed at support rod penetration.

3 4

5

6

7

8

9

10 11

12

Where riser clamps are required to be attached directly to piping requiring vapor barrier, extend insulation and vapor barrier jacketing/coating around riser clamp.

Where insulated piping is installed on hangers and supports, the insulation shall be installed continuous through the hangers and supports. High density inserts shall be provided as required to prevent the weight of the piping from crushing the insulation. Pipe shields are required at all support locations. The insulation shall not be notched or cut to accommodate the supporting channels.

Fully insulate all reheat coil piping, fittings and valves (with the exception of unions) up to coil connection to prevent condensation when coil is inactive during cooling season. Provide a vapor proof seal between the pipe insulation and the insulated coil casing.

13 INSULATION INSERTS AND PIPE SHIELDS:

Provide pipe shields at all hanger and support locations. Rigid insulation inserts shall be installed between the pipe and the insulation shields. Quantity and placement of inserts shall be according to the manufacturer's installation instructions, however the inserts shall be no less than 12" in length. Inserts shall be of equal thickness to the adjacent insulation and shall be vapor sealed as required for system.

Provide insulation inserts and pipe shields at all hanger and support locations. Inserts may be omitted on 3/4" and smaller
 copper piping provided 12" long 22 gauge pipe shields are used.

2 FITTINGS AND VALVES:

Fittings, valves, unions, flanges, couplings and specialties may be insulated with factory molded or built up insulation of the same thickness as adjoining insulation. Where the ambient temperature exceeds 150 degrees F, cover insulation with fabric reinforcing and mastic. Where the ambient temperatures do not exceed 150 degrees, furnish and install PVC fitting covers.

7 ELASTOMERIC:

8 Where practical, slip insulation on piping during pipe installation when pipe ends are open. Miter cut fittings allowing sufficient 9 length to prevent stretching. Completely seal seams and joints for vapor tight installation. For elastomeric insulation, apply full 9 bed of adhesive to both surfaces. For polyeolefin, seal factory preglued seams with roller and field seams and joints with full 9 bed of hot melt polyolefin glue to both surfaces. Cover elastomeric insulation on systems operating below 40 degrees F with 9 vapor barrier mastic.

4 EXTRUDED POLYSTYRENE AND POLYISOCYANURATE:

5 Fittings, valves, unions, flanges, couplings and specialties shall be insulated with factory molded insulation of the same 6 thickness as adjoining insulation. Secure insulation sections with two wraps of nylon filament tape 9"-12" on center.

On single insulation layer systems and on the outer layer of double insulation layer systems, apply a thin coat of elastomeric joint sealant rated for system operating temperatures to all longitudinal and butt insulation joints covering entire face of joint. Allow sealant to fully cure before applying protective covering. For piping service below 0°F, use two layers of insulation with inner and outer butt and longitudinal joints staggered and offset 90 degrees. Where two layers of insulation are used, do not use sealant on the inner layer or adhere the inner layer to the outer layer. Apply vapor stop bead of joint sealant between pipe and insulation on both sides of valves, expansion/contraction joints, flanges, thermometers/gauges, attached vent and drain lines. Insulate attached non-circulated lines, control lines, vents, etc. for a minimum distance of 6" from pipe. Cover insulation with a protective jacket as specified below. Do not penetrate protective covering or insulation with mechanical fasteners.

PIPING PROTECTIVE JACKETS

In addition to the jackets specified in the pipe insulation schedule below the following protective jackets are required:

- Provide a protective PVC jacket (PFJ) for the following insulated piping:
 - All exposed piping in finished areas within 8' of the floor.
 - Provide a protective PVC (PFJ) jacket for the following insulated piping:
 - All piping within mechanical rooms or mechanical spaces.

PIPE INSULATION SCHEDULE:

Provide insulation on new and existing remodeled piping as indicated in the following schedule:

		JACKET	INSULATION THICKNESS BY PIPE SIZE					
SERVICE	INSULATION		< 1"	1" to < 1-1/2"	1-1/2" to < 4"	4" to < 8"	8" and Larger	
Heating Hot Water (1)	Rigid Fiberglass	ASJ	1.5″	1.5″	2″	2″	2″	
Chilled Water (1)	Polyisocyanurate	VRJ or SAJ	1.5″	1.5″	1.5″	1.5″	1.5″	
Cooling Coil Condensate Drain	Rigid Fiberglass	ASJ	0.5″	0.5″	1"	1″	1"	
Geothermal Condenser Water	Polyisocyanurate	VRJ or SAJ	1.5″	1.5″	1.5″	1.5″	1.5″	

4

5 6

7

(1) Exposed insulation jacket shall be paintable and field painted by GC.

The following piping and fittings are not to be insulated:

- Hot water piping inside radiation, convector, or cabinet heater enclosures
- Piping unions for systems not requiring a vapor barrier

For systems with fluid temperatures 65° F or less, furnish and install removable elastomeric insulation covers, plugs or caps for
all mechanical equipment and devices that require access by balancing contractors or service and maintenance
personnel. Examples include but are not limited to: flow sensing devices, circuit setters, manual ball valve air vents, drain
valves, blowdown valves, pressure/temperature test plugs, grease fittings, pump bearing caps, equipment labels, etc. Covers
shall be tight fitting to ensure a complete vapor barrier.

13 14 DUCT INSULATION

15 GENERAL:

Secure flexible duct insulation on sides and bottom of ductwork over 24" wide and all rigid duct insulation with weld pins. Space
 fasteners 18" on center or less as required to prevent sagging.

18

Secure rigid board insulation to ductwork with weld pins. Apply insulation with joints firmly butted as close as possible to the equipment surface. Pins shall be located a maximum of 3" from each edge and spaced no greater than 12" on center.

Install weld pins without damage to the interior galvanized surface of the duct. Clip pins back to washer and cover penetrations with tape of same material as jacket. Firmly butt seams and joints and cover with 4" tape of same material as jacket. Seal tape with plastic applicator and secure with staples. All joints, seams, edges and penetrations to be fully vapor sealed.

25

Stop and point insulation around access doors and damper operators to allow operation without disturbing insulation or jacket
 material.

External supply duct insulation is not required where ductwork contains continuous 1" acoustical liner. Provide 4" overlap of external insulation over ends of acoustically lined sections.

31

Where insulated ductwork is supported by trapeze hangers, the insulation shall be installed continuous through the hangers. Drop the supporting channels required to facilitate the installation of the insulation. Where rigid board or flexible insulation is specified, install high density inserts to prevent the weight of the ductwork from crushing the insulation.

Where insulated low temperature (below 45°F) ductwork is supported by steel metal straps or wire ropes that are secured directly to the duct, the straps or ropes shall be completely covered with insulation and sealed to provide a complete vapor barrier.

39

Where insulated duct risers are supported by steel channels secured directly to the duct, extend the insulation and vapor
 barrier jacketing to encapsulate the support channels.

43 **DUCT INSULATION SCHEDULE:**

- 44 Provide duct insulation on new and existing remodeled ductwork in the following schedule:
- 45
- 46 47
- 47 48

SERVICE	INSULATION TYPE	JACKET	THICKNESS
Outside air ducts	Rigid Fiberglass	FSK	2"
Mixed air ducts	Rigid Fiberglass	FSK	2"
Exposed supply ducts (1)	Rigid Fiberglass	FSK	1.5"
Concealed supply ducts	Flexible Fiberglass	FSK	2.25"
Exhaust and relief ducts downstream of motorized backdraft dampers	Rigid Fiberglass	FSK	2"
Louver blank-off panels	Ext. Polystyrene	SAJ	4"

12

1

(1) Exposed supply ducts located in the space they are serving do not require insulation. Exposed supply ducts running through spaces they do not serve shall be insulated as exposed supply ducts scheduled above.

EQUIPMENT INSULATION

GENERAL:

Do not insulate over equipment access manholes, fittings, nameplates or ASME stamps. Bevel and seal insulation at these locations.

EQUIPMENT INSULATION SCHEDULE:

Provide equipment insulation as follows:

EQUIPMENT	INSULATION TYPE	JACKET	THICKNESS
Reheat coil casing in exposed supply ducts	Rigid Fiberglass	FSK	2″
Pumps & Strainers (Chilled & Geothermal Water)	Elastomeric	NA	1″
Air Handling Unit Casings or attached component sections not factory insulated*	Rigid Fiberglass	ASJ	2″
Air Separators	Elastomeric	NA	2″
Buffer Tanks	Elastomeric	NA	1"

* The thickness and type of insulation provided for non-factory fabricated transitions or component sections shall be consistent with the sections constructed at the factory.

17 CONSTRUCTION VERIFICATION ITEMS

18 Contractor is responsible for utilizing the construction verification checklists supplied by the third party commissioning agent in 19 accordance with the procedures defined for construction verification in Section 01 91 00.

20

21 22

END OF SECTION

1	SECTION 23 09 14
2	PNEUMATIC AND ELECTRIC INSTRUMENTATION AND CONTROL DEVICES FOR HVAC
3	
4 5	PART 1 - GENERAL
6	
7	SCOPE
8	The work associated with this section be bid as part of the Division 23 scope of work.
9 10	This section includes proving the control system specifications for all HVAC work. Included are the following tenics:
10	This section includes pheumatic control system specifications for an trvAc work. Included are the following topics.
12	PART 1 - GENERAL
13	Scope
14	Point List
15	Related Work
10	Work Not Included
18	Quality Assurance
19	Reference Standards
20	System Description
21	Submittals
22	Design Criteria Operation and Maintonance Data
25 24	Material Delivery and Storage
25	watchar benvery and storage
26	PART 2 - PRODUCTS
27	Air Piping
28	Control Dampers
29	Control Valves
30 31	Control System Instrumentation Thermostat Guards
32	Electric/Electronic Thermostats
33	Duct Smoke Detector and Fire Alarm Interface Modules
34	Temperature Control Panels
35	Temperature Sensors
30 27	Humiaity Sensors Proseuro Transducors (Air)
38	Pressure Transducers (Air) Pressure Transducers (Liquid/Steam)
39	Differential Pressure Switches
40	Current Status Switches
41	Power Supplies
42	
43 11	PART 3 - EXECUTION
45	Wire and Air Piping Conduit and Tubing Installation Schedule
46	Control and Smoke Dampers
47	Control Valves
48	Room Thermostats and Temperature Sensors
49	Low Limit Thermostats (Freezestats)
50 51	Pressure Transducers
52	Differential Pressure Switches
53	Air Pressure Safety Switches
54	Current Status Switches
55	Construction Verification Items
56	DOINT LIST (Costion Drowings)
5/ 58	PUINT LIST (Section Drawings)
59	RELATED WORK
60	Section 23 05 93 - Testing, Adjusting, and Balancing for HVAC - Coordination
61	Section 23 09 23 - Direct Digital Control System for HVAC
62	Section 23 09 93 - Sequence of Operation
63	Section 23 33 00 - Ductwork Accessories (For control damper installation)
b 4	Division 23 - HVAC - Equipment provided to be controlled or monitored

REFERENCE 1 2

3 4

5

6

8

17

30

37

38

50

55

58

62

Applicable provisions of Division 1 govern work under this section.

WORK NOT INCLUDED

Control dampers as specified in Section 23 33 00.

7 **QUALITY ASSURANCE**

MANUFACTURER:

9 Installing contractor must be a manufacturer's branch office or an authorized representative of a Direct Digital Control (DDC) equipment manufacturer that provides engineering and commissioning of the DDC equipment. Submit written confirmation of 10 such authorization from the manufacturer. Indicate in letter of authorization that installing contractor has successfully 11 completed all necessary training required for engineering, installation, and commissioning of equipment and systems and that 12 such authorization has been in effect for a period of not less than five years. DDC equipment may or may not be required to be 13 14 installed by this contractor as part of the project, but the intent of this quality assurance specification is to ensure that the installing contractor has the capabilities to engineer, install, and commission the field devices supplied under this section for 15 temperature control. 16

18 Contractor is limited to firms regularly employing a minimum of five full-time temperature control technicians within 50 miles 19 of the job site.

20 **REFERENCE STANDARDS** 21

22	ANSI B16.22	Wrought Copper and Wrought Copper Alloy Solder Joint Pressure Fittings
23	ANSI/ASTM B32	Specification for Solder Metal
24	ASTM B75	Seamless Copper Tube
25	ASTM D1693	Environmental Stress-Cracking of Ethylene Plastics
26	ASTM D 635	Standard Test Method for Rate of Burning and/or Extent and Time of Burning of Plastics in a Horizontal
27		Position
28	UL 94	Tests for Flammability of Plastic Materials for Parts in Devices and Appliances
29	AMCA 500-D	Laboratory Method of Testing Dampers for Rating

31 SYSTEM DESCRIPTION

System is to use direct digital control with electric actuation for air handling units; direct digital control with electric actuation 32 33 for room temperature, room humidity, and terminal airflow control; and direct digital control with electric control for other 34 terminal units. 35

All pneumatic tubing and electrical wiring are to be permanently tagged or labeled (within one inch of terminal strip) with a 36 numbering system to correspond with the "Record Drawings". Tags or labels shall be printed not hand written.

39 SUBMITTALS

40 Include the following information: 41

42 Manufacturer's data sheets indicating model number, pressure/temperature ratings, capacity, methods and materials of construction, installation instructions, and recommended maintenance. General catalog sheets showing a series of the same 43 device is not acceptable unless the specific model is clearly marked. 44 45

Schematic flow diagrams of systems showing fans, pumps, coils, dampers, valves, and other control devices. Label each device 46 47 with setting or adjustable range of control. Indicate all wiring, clearly, differentiating between factory and field installed wiring. Wiring should be shown in schematics that detail contact states, relay references, etc. Diagrammatic representations of 48 devices alone are not acceptable. 49

Details of construction, layout, and location of each temperature control panel within the building, including instruments 51 52 location in panel and labeling. Also include on drawings location of mechanical equipment controlled (room number), 53 horsepower and flow of motorized equipment (when this data is available on plans), locations of all remote sensors and control 54 devices (either by room number or column lines).

56 Schedule of control dampers indicating size, leakage rating, arrangement, pressure drop at design airflow, and number and size 57 of operators required.

59 Schedule of control valves indicating system in which the device is to be used, rated capacity, flow coefficient, flow required by device served, actual pressure drop at design flow, size of operator required, close-off pressure, and locations where valves are 60 to be installed. 61

63 A complete description of each control sequence for equipment that is not controlled by direct digital controls. Direct digital 64 controlled equipment control sequences will be provided by the DDC control contractor.

1 Calculations completed to determine size of control air compressor(s) and dryer (s).

Prior to request for final payment, submit record documents which accurately record actual location of control components
 including panels, thermostats, wiring, and sensors. Incorporate changes required during installation and start-up.

6 **DESIGN CRITERIA**

7 Size all control apparatus to properly supply and/or operate and control the apparatus served.

9 Provide control devices subject to corrosive environments with corrosion protection or construct them so they are suitable for
 10 use in such an environment.

Provide devices exposed to outside ambient conditions with weather protection or construct them so they are suitable for outdoor installation.

14

5

8

11

Use only UL labeled products that comply with NEMA Standards. Electrical components and installation to meet all requirements of the electrical sections (Division 26) of project specifications.

18 OPERATION AND MAINTENANCE DATA

19 All operations and maintenance data shall comply with the submission and content requirements specified under section 20 GENERAL REQUIREMENTS.

22 MATERIAL DELIVERY AND STORAGE

Provide factory shipping cartons for each piece of equipment and control device. This contractor is responsible for storage of equipment and materials inside and protected from the weather.

25 26 27

28

30

31

32

21

PART 2 - PRODUCTS

29 AIR PIPING

ASTM B75 seamless, hard drawn or annealed copper tubing with ANSI B16.22 wrought copper fittings, except final connections to apparatus may be made with brass compression-type fittings. Use ANSI/ASTM B32, 95/5 tin antimony solder.

33 Virgin polyethylene plastic tubing classified as flame retardant under UL 94 and conforming to ASTM D1693 stress-crack test.

3435 CONTROL DAMPERS

36 Control dampers are specified in section 23 33 00.

37

38 Acceptable actuator manufacturers: Belimo or Honeywell.

39

Provide weatherproof NEMA 4 enclosures that have removable covers that have clasps or machine screws (no sheet metal screws) and that do not require removing fasteners from the ductwork to prevent actuator failure or freeze-up when mounting in locations exposed to harsh environments or outdoor locations. Actuators to have V bolt clamp with matching V toothed cradle (single bolt setscrew fasteners are not acceptable).

44

45 Size operators for smooth and positive operation of devices served, and with sufficient torque capacity to provide tight shutoff against system temperatures and pressure encountered. For electric modulating actuation, use fully proportional actuators 46 47 with zero and span adjustments. For terminal unit actuators, stepping motors may be used and zero and span is not required. 48 For two-position electric actuation use 24 VAC for DDC controlled actuators, 120 VAC actuators may be used for hardwire 49 interlocking. See 23 09 15 for specific type of input signal required. Actuator stroke times shall match the requirements of the 50 DDC controllers provided under 23 09 23 and/or the specific system requirements for proper operation. All electric actuators 51 will be provided with overload protection to prevent motor from damage when stall condition is encountered. Equip operators 52 with spring return or stored energy fail-safe return for applications involving fire, freeze protection, moisture protection or 53 specified normally open/closed operation. Provide damper end switches with form "C" contacts where control sequences 54 require damper position indication. End switches shall not contain mercury.

55

All power required for electric actuation shall be provided by this contractor if it is not able to be directly provided from the
 DDC controller.

58

59 Provide operators with linkages and brackets for mounting on device served.

60

61 CONTROL VALVES

- 62 Provide all control valves as shown on the plans/details and as required to perform functions specified. Spring ranges must be
- 63 selected to prevent overlap of operation and simultaneous heating and cooling.

2 3

4

5 6

7

8

9

10

11

12 13

15 16 17

18

19

20

21 22

23 24

25 26

27

Acceptable actuator manufacturers: Belimo or Honeywell.

Provide NEMA rated housings for applicable environment, with enclosures that have removable covers. Actuators to have V bolt clamp with matching V toothed cradle (single bolt setscrew fasteners are not acceptable).

Size operators to allow smooth and positive operation of devices served and to provide sufficient torque capacity for tight shutoff against system temperatures and pressure encountered. For electric modulating actuation, use fully proportional actuators with 0-10VDC inputs and zero and span adjustments unless specified otherwise in the chart below. If TriState with feedback is specified, valve position shall be fed back to the controller and controller shall position valve based on this feedback. For two-position electric actuators, for applications other than terminal units, shall be provided with a manual override capability. All electric actuators shall be provided with a visible position indicator.

14 All power required for electric actuation shall be provided by this contractor if it is not able to be directly provided from the 15 DDC controller.

Provide operators that are full proportioning or two-position, as required for specified sequence of operation. Provide springreturn for applications involving fire, freeze protection, moisture protection or specified normally open/closed operation. Valves shall move to their fail positions on loss of electrical power or air pressure to the actuator.

Two-position shut-off valves shall be sized for a maximum pressure drop of 2 PSI at design flow and shall be a minimum of line size.

Provide operators with linkages and brackets for mounting on device served.

All valves unless specifically noted on the plans or indicated below shall be globe style valves.

VALVE SERVING	TYPE	SIGNAL	SPRING	FAIL
	Globe	0-10 VDC	RETURN	POSITION
	Butterfly (BF)	TriState (24VAC)	REQUIRED	Open (thru Coil)
	Ball	2-Position Elect	Yes	Closed (bypass Coil)
	Press Independent	Pneumatic (Pneu)	No	Last Position
	Ball (PI Ball)			
Reheat Coil	Ball	0-10 VDC	No	Last Position
CUH	Ball	2-Pos Elect	Yes	Open
AHU Htg Coil	Ball	0-10 VDC	Yes	Open
AHU Clg Coil	Ball	0-10 VDC	Yes	Closed

28 29 30

31

32

33

34

37

See plan details, notes, and schedules for where two-way and three-way valves should be used.

WATER SYSTEMS:

Use equal percentage valves for two-way control valves; size for a pressure drop not less than 4 psi or more than 6 psi. Note: For low flows, the required minimum Cv size will result in lower pressure drop than 4 psi.

Use three-way valves sized for a maximum pressure drop of 5 psi and that have linear characteristics so that the valve pressure drop remains constant regardless of the valve position.

Ball valves 2" and smaller: Cast bronze or forged brass body, chrome plated brass or stainless steel full port ball, stainless steel
 stem, PTFE or RTFE seats and seals, screwed ends, suitable for use on water systems at 150 psig and 240° F. Seat leakage with
 actuator supplied will meet ANSI class IV leakage (0.01%).

Ball valves 2 1/2" and larger: Iron body, stainless steel full port ball, stainless steel stem, PTFE or RTFE seats and seals, flanged
 ends, suitable for use on water systems at 150 psig and 240° F.

44 45 CONTROL SYSTEM INSTRUMENTATION

46 Manufacturers: Averaging Type - Johnson Controls, or equal; Bulb Type - Johnson Controls, Ashcroft, Marshall, Weksler

4748 PIPE THERMOMETERS:

9 inch stem type with an adjustable swivel mount. Scale graduations of 2°F and mid-range accuracy of ±1°F. Install
 thermometers in separable brass wells filled with conductive fluid. Thermometer temperature range shall not be more than
 twice the expected temperature range at installed location.

1 REMOTE BULB THERMOMETERS:

2 3 inch or larger dial type with recalibration screw on face. Accuracy within 1% of scale range. Thermometers with sensing

elements in air ducts with an area of above 6 square feet to have averaging liquid or gas filled capillary sensing elements.
 Provide separable wells for all pipeline applications. Thermometer temperature range shall not be more than twice the

5 expected temperature range at installed location.

6 7

THERMOSTAT GUARDS

Provide clear plastic locking covers keyed the same. For locations that are subject to physical abuse, provide cast aluminum
 guard, Johnson Controls GRD10A-601 or equal.

10

11 ELECTRIC/ELECTRONIC THERMOSTATS

12 LOW LIMIT THERMOSTATS (freezestats):

Electric two-position type with temperature sensing element and manual reset for all applications except integral face and bypass steam heating coils which shall have auto-reset freezestats and latching relays (see execution section for details). Unit to be capable of opening control circuit if any one-foot length of sensing element is subject to a temperature below the setpoint. Length of sensing element to be not less than one lineal foot per square foot of coil surface areas. Unless otherwise indicated, set low limit controls at 36°F.

18

19 REMOTE BULB THERMOSTATS:

Line voltage type with single pole, double throw switch of adequate rating for the applied load. Thermostat to have adjustable
 setpoint suitable for controlled load.

22

23 IMMERSION TYPE THERMOSTAT SENSORS:

Rod and tube type with linear output. Provide separable wells with heat conductive fluid for installation in pipeline. Units shall be factory calibrated.

26

27 DUCT SMOKE DETECTOR AND FIRE ALARM INTERFACE MODULES

Duct smoke detectors and fire alarm control modules shall be provided by others. Provide wiring, conduit, and necessary interface with fire alarm system to perform specified sequence of operation.

31 TEMPERATURE CONTROL PANELS

Constructed of steel or extruded aluminum, with hinged door, keyed lock, and baked enamel finish. Install controls, relays, transducers and automatic switches inside panels. Label devices with permanent printed labels and provide asbuilt wiring/piping diagram within enclosure. Provide raceways for wiring and poly within panel for neat appearance and to separate high and low voltage wiring. Provide termination blocks and resettable circuit breaker for 120VAC power wiring. Provide label within the panel indicating circuit number of 120VAC serving panel. Label outside of panel with panel number corresponding to plan tags and asbuilt control drawings as well as building system(s) served.

38

49

Provide a service shutdown toggle switch for each air handling unit system located inside the temperature control panel that will initiate a logical shutdown of the air handling unit system. Label the switch so it is clear which position is shutdown and which is auto.

43 **TEMPERATURE SENSORS**

Use thermistor or RTD type temperature sensing elements constructed so accuracy and life expectancy is not affected by
 moisture, physical vibration, or other conditions that exist in each application.

47 RTD's shall be of nickel or platinum construction and have a base resistance of 1000Ω at 70°F and 77°F respectively. 100Ω 48 platinum RTD's are acceptable if used with temperature transmitters.

50 The temperature sensing device used must be compatible with the DDC controllers used on the project. 51

52 Acceptable Manufacturers: Room Temperature Sensor: Honeywell TR71

54	RTI	ר

7	RID	
55	Accuracy (Room Sensor Only)	minimum <u>+</u> 1.0°F
56	Accuracy (Averaging)	minimum <u>+</u> 1.2°F
57	Accuracy (Other than Room Sensor or Averaging)	minimum <u>+</u> 0.65°F
58	Range	minimum -40 - 220°F
59		
60	Thermistor	
61	Accuracy (All)	minimum <u>+</u> 0.36°F
62	Range	minimum -30 - 230°F
63	Heat Dissipation Constant	minimum 2.7 mW/°C
64		

1	Temperature Transmitter		
2	Accuracy	minimum <u>+</u> 0.1°F or <u>+</u> 0.2% of span	
3	Output	4-20 mA	
4			
5	Provide limited range or extended range s	ensors if required to sense the range expected for a respective point. Use RTD type	
6	sensors for extended ranges beyond -30 to 230°F. If RTD's are incompatible with DDC controller direct temperature input use		
7	temperature transmitters in conjunction with RTD's.		
8			
9	Use wire size appropriate to limit tempera	ure offset due to wire resistance to 1.0°F. If offset is greater than 1.0°F due to wire	
10	resistance, use temperature transmitter.	f feature is available in DDC controller, compensate for wire resistance in software	
11	input definition.		
12			
13	Provide sensors in occupied spaces with	brushed aluminum or brushed nickel covers unless otherwise noted or features	
14	specified will not allow for this. Terminal	unit sensors with setpoint adjustments and digital displays may use plastic covers.	
15	Provide information to the AE on sensor	colors offered by the manufacturer and obtain approval on what color should be	
16	provided on the project.		
17			
18	Terminal unit sensors shall be provided w	th digital displays that indicate room temperature and setpoint and have a manual	
19	occupancy override and indication of occ	upancy status. Provide setpoint adjustment as specified in the DDC Input/Output	

nt and have a manual the DDC Input/Output 20 Summary Table and sequence of operation. 21

22 Use averaging elements on duct sensors when the ductwork is ten square feet or larger. All mixed air and heating coil discharge 23 sensors shall have averaging elements regardless of duct size.

In piping systems use temperature sensors with separable wells designed to be used with temperature element. 24

25 26 HUMIDITY SENSORS

30

33

36

37

38

39

55

56 57

58

59

60

61

Use capacitive thin-film polymer sensor types with a range of 0-100% RH. Accuracy to be no less than $[\pm 2\%]$ in the range of 27 28 20% RH to 80% RH with a response time of 120 seconds or less. Provide covers for room humidity sensors as specified for 29 temperature sensors.

For outside air applications, use sensor designed for outside air use along with weather enclosure. Provide sensor equal to 31 Vaisala Model HMD60UO w/ DTR503B enclosure and weather resistant mounting hardware. 32

PRESSURE TRANSDUCERS (AIR) 34 35

Provide pressure transducers specified below for the following applications:

- Duct static pressure applications where setpoints are specified to control at greater than 0.1" w.c.
- Terminal unit air flow measurement regardless of the minimum velocity pressure unless otherwise noted in the contract documents.

Provide a transmitter that operates on the capacitance principle and is capable of sensing low positive, negative or differential 40 41 pressures. Transmitter shall have a minimum of three pressure ranges adjustable by an onboard switch or jumper. Size the transmitter where the middle or high range is suitable for the application. Use a bi-directional transmitter for applications that 42 may have both positive and negative pressure excursions. Transmitter shall be provided with an integral four-digit display of the 43 44 pressure sensed. 45

<u>+</u> 0.25% FS
35°-130° F
0-1"wc Range .09% FS/°F;
>1"wc Range .02% FS/°F
4-20 MA
800 Ω max.
0°-160° F

Provide pressure transducers specified below for the following applications:

- Duct static pressure applications where setpoints are specified to control at 0.1" w.c. or lower.
- All duct mounted pitot type air flow stations.
- Space/building static control or monitoring.

The airflow transducer shall provide noise filtration and automatic auto-zeroing. The automatic zeroing circuit shall be capable of maintaining the transducer output to within $\pm 0.25\%$ of operating span. The transducer output shall be locked and maintained at the last given output value during the automatic zeroing period so as not to interrupt the automatic control process.

- 1 Use a bi-directional transmitter for applications that may have both positive and negative pressure excursions. Transmitter shall 2 be provided with an integral four-digit display of the pressure sensed. 3 4 Transducer Span: <2 times the design velocity pressure at maximum flow, single range Accuracy: ±0.25% of full scale, including non-linearity, hysteresis, deadband, and non-repeatability 5 Temperature Effect: ±0.15% of full scale/°F 6 Response: 0.5 sec. for 98% of full span change 7 8 Overpressure: 15 PSIG Proof 9 Power: 24VAC/VDC Analog Output: 0-5VDC, 0-10VDC, or 4-20mA field adjustable 10 Auto Zero Frequency: every 1 to 24 hours on 1 hour intervals 11 12 PRESSURE TRANSDUCERS (LIQUID/STEAM) 13 14 Provide a transmitter that utilizes capacitive or thin film strain gauge sensing. Provide for an analog gauge piped in parallel with the transducer. Gauge shall meet specifications as specified in Section 23 05 15. Coordinate with mechanical contractor to 15 provide and install this gauge. For differential pressure applications provide with bypass valve manifold assembly with valved 16 17 venting capability. 18 19 Accuracy (including non-linearity and hysteresis) <u>+</u> 0.5% FS 20 **Compensated Temperature Range** 32°-150° F Temperature Effect (over compensated range) 0.03%/°F: 21 22 Output 4-20 MA 23 Load Impedance (smallest maximum acceptable) $600 \ \Omega$ Minimum 24 **Operating Temperature** 0°-175° F 25 Hysteresis 0.75% of span 26 27 DIFFERENTIAL PRESSURE SWITCHES 28 Differential pressure switches shall sense both inlet and outlet of fans and pumps. Device shall be rated for 150% of maximum 29 system pressures that may be encountered. Provide with pressure differential that will be required to meet specified operation 30 and/or to prevent nuisance "toggling" of the device in the system served. 31 32 **AIR PRESSURE SAFETY SWITCHES** Air pressure safety switches shall be a differential pressure switch that will sense differential, negative, or positive pressure as 33 34 required by the sequence of operation specification. Device shall be rated for a minimum of 150% of maximum system 35 pressures that may be encountered. Provide with pressure range that will be required to meet specified operation in the 36 system served. 37 Provide with a normally closed contact that will open above setpoint and will not close until the manual reset button is 38 39 depressed. Setpoint shall be manually adjustable. 40 41 **CURRENT STATUS SWITCHES** 42 Provide a current sensor with adjustable threshold and digital output with LED display, equal to a Veris model H-308. Threshold adjustment must be by a multi-turn potentiometer or set by multiprocessor that will automatically compensate for frequency 43 44 and amperage changes associated with variable frequency drives. When used on variable speed motor applications, use a 45 current sensor that will not change state due to varying speeds. 46 47 POWER SUPPLIES 48 Provide all required power supplies for transducers, sensors, transmitters and relays. All low voltage transformers shall have a 49 resettable secondary circuit breaker and be listed as class 2 power supplies. 50 51 PART 3 - EXECUTION 52 53 54 INSTALLATION 55 Install system with trained mechanics and electricians employed by the control equipment manufacturer or an authorized 56 representative of the manufacturer. Where installing contractor is an authorized representative of the control manufacturer, 57 such authorization shall have been in effect for a period of no less than three years. 58 59 Install all control equipment, accessories, wiring, and piping in a neat and workmanlike manner. All control devices must be 60 installed in accessible locations. This contractor shall verify that all control devices furnished under this Section are functional 61 and operating the mechanical equipment as specified in Section 23 09 93. 62
- 63 Label all control devices with the exception of dampers, valves, and terminal unit devices with permanent printed labels that 64 correspond to control drawings.

3 4

5

6

7

8 9

10

11 12

22

38

44

Temperature control junction and pullboxes shall be identified utilizing spray painted green covers. Other electrical system identification shall follow the 26 05 53 specification.

All control devices and electrical boxes mounted on insulated ductwork shall be mounted over the insulation. Provide mounting stand-offs where necessary for adequate support. Cutting and removal of insulation to mount devices directly on ductwork is not acceptable. This contractor shall coordinate with the insulation contractor to provide for continuous insulation of ductwork.

Mounting of electrical or electronic devices shall be protected from weather if the building is not completely enclosed. This Contractor shall be solely responsible for replacing any equipment that is damaged by water that infiltrates the building if equipment is installed prior to the building being enclosed.

Provide all electrical relays and wiring, line and low voltage, for control systems, devices and components. Install all high voltage and low voltage wiring (includes low voltage cable) in metal conduit, Electrical Non-metallic Tubing (ENT), or Electrical Metallic Tubing (EMT), as scheduled below and hereafter referred to generically as conduit. See Wire and Air Piping Conduit Installation Schedule below for specific conduit or tubing to be used. All conduit must be installed in accordance with electrical sections (Division 26) of this specification and the National Electrical code.

Conduit shall be a minimum of 1/2 " for low voltage control provided the pipe fill does not exceed 40%.

21 Minimum low voltage wiring gauge to be 18 AWG for outputs and 20 AWG for inputs. All low voltage wiring to be stranded.

Low voltage wiring can be run without conduit above accessible lay-in tile ceilings. All wiring in mechanical rooms, above inaccessible hard ceilings, exterior locations, and in any exposed areas, and in all other locations should be in conduit. Wire for wall sensors must be run in conduit. Wiring for radiation valves shall be run in conduit where routed through walls.

27 Where wiring is installed free-air, installation shall consider the following:

- Wiring shall utilize the cable tray wherever possible.
- Wiring shall run at right angles and be kept clear of other trades work.
- Wiring shall be supported utilizing "J" or "Bridal-type" steel mounting rings anchored to ceiling concrete, piping supports,
 walls above ceiling or structural steel beams. Mounting rings shall be of open design (not a closed loop) to allow additional
 wire to be strung without being threaded through the ring. For mounting rings that do not completely surround the wire,
 attach the wire to the mounting ring with a strap.
- Supports shall be spaced at a maximum 4-foot interval unless limited by building construction. If wiring "sag" at mid-span
 exceeds 6-inches; another support shall be used.
- Wiring shall never be laid directly on the ceiling grid or attached in any manner to the ceiling grid wires.
- Wall penetrations shall be sleeved.

Wiring shall not be attached to existing cabling, existing tubing, plumbing or steam piping, ductwork, ceiling supports or electrical or communications conduit.

This contractor shall be responsible for all 120VAC power, not provided in the Division 26 specifications, required for equipment
 provided under this section. Power shown for temperature control panels on plans may be utilized by the 23 09 24 and/or 23
 09 23, and 23 09 14 contractors.

Provide communication trunk wiring to integrated devices (i.e. VFD's, Flow Meters, Chillers, Lighting Panels, Electrical Meters, etc.) that are specified to be connected to the building automation system. Communication trunk wiring shall be as required by the equipment specified under the 23 09 24 or 23 09 23 Sections and shall be routed to the DDC panel designated for that equipment as shown on the plans or the closest DDC panel if not designated. If communication trunks required daisy chained style wiring, provide two communication cables to the DDC panel so that the communication trunk is not dead ended.

Install "hand/off/auto" selector switches on systems where automatic interlock controls are specified and "hand/off/auto" selector switches are not supplied with the equipment controlled. Control panel power will not be required for "hand" switch to operate. When switch is in "hand" position, allow manual operation of the selected device without operating the interlocked motors but allowing all unit safety devices to stay in the circuit.

All wiring in control panels shall be terminated on a terminal strip. Wire nuts are not acceptable. A maximum of two wires
 shall be terminated under any one terminal.

58 All pneumatic tubing and electrical wiring are to be permanently tagged or labeled within one inch of terminal strip with a 59 numbering system to correspond with the "Record Drawings".

61 After completion of installation, test and adjust control equipment. Submit data showing set points and final adjustments of 62 controls.

63

1 WIRE AND AIR PIPING CONDUIT AND TUBING INSTALLATION SCHEDULE

2 The following conduit schedule shall apply to both polyethylene tubing and wire in conduit where conduit is specified for air 3 tubing or wiring. Conduit and tubing referenced below shall meet specifications in Section 26 05 33 and as defined below.

- 5 Conduit other than that specified below for specific applications shall not be used.
- 6 7 8

10

12

4

Concealed in Concrete and Block Walls: Rigid steel conduit. Schedule 40 PVC conduit. Electrical Nonmetallic Tubing (ENT).

9 Concealed Dry Interior Locations: Rigid steel conduit. Intermediate metal conduit. Electrical metallic tubing.

11 Exposed Dry Interior Locations: Rigid steel conduit. Intermediate metal conduit. Electrical metallic tubing.

13 CONTROL DAMPERS

All control dampers are furnished and installed by the Mechanical Contractor under the coordinating control and supervision of
 the Control Contractor in locations shown on plans or where required to provide specified sequence of control.

16

17 Coordinate installation with the sheetmetal installer to obtain smooth duct transitions where damper size is different than duct 18 size. Blank off plates will not be accepted.

19

Each operator shall serve a maximum damper area of 36 square feet. Where larger dampers are used, provide multiple
 operators.

23 CONTROL VALVES

All temperature control valves furnished by the control manufacturer are to be installed by the Mechanical Contractor under the coordinating control and supervision of the Control Contractor in locations shown on plans or where required to provide specified sequence of control.

28 **ROOM THERMOSTATS AND TEMPERATURE SENSORS**

29 Check and verify location of thermostats, humidistats, and other exposed control sensors with plans and room details before 30 installation. Locate room thermostats and sensors 44 inches above floor. Align with light switches and humidistats. For 31 drywall installations, thermostat mounting shall use a back-box attached to a wall stud, drywall anchors are not acceptable.

32

Any room thermostats or sensors mounted on an exterior wall shall be mounted on a thermally insulated sub-base. Subbase to
 provide a minimum of one half inch of insulation.

36 Where thermostats or sensors are mounted on exterior walls or in any location where air transfer will affect the measured 37 temperature or humidity seal the conduit and any other opening that will effect the measurement.

38

Provide guards on thermostats and sensors in entrance hallways, other public areas, or in locations where thermostat is subject
 to physical damage.

41

42 LOW LIMIT THERMOSTATS (Freezestats)

Install low limit controls where indicated on the drawings or as specified. Unless otherwise indicated, install sensing element
 on the downstream side of heating coils.

Mount units using flanges and element holders. Provide duct collars or bushings where sensing capillary passes through
 sheetmetal housings or ductwork; seal this penetration to eliminate air leakage. Mount the units in an accessible location as to
 allow for resetting after low limit trips while still meeting manufacturer's installation requirements for proper function.

49

50 Distribute (serpentine) sensing element horizontally across the coil to cover every square foot of coil; on larger coils this may 51 require more than one instrument. Install controls at accessible location with mounting brackets and element duct collars 52 where required.

53

54 **PRESSURE TRANSDUCERS AND HIGH LIMIT PRESSURE SWITCHES**

Install capped tees in air piping at air pressure transducers for connection of calibration equipment. Capped tee shall consist of two inch poly tubing capped with a brass plug. Rubber caps are not acceptable. Install Petes Plugs fittings at each take-off from main piping for liquid pressure transducers for connection of calibration equipment. Install differential pressure transducers for filter monitoring at the filter section of the air handling unless otherwise specified. All other differential or static pressure transducers and differential or static pressure high limit switches for air applications should be mounted in the temperature control panel serving the equipment being controlled or monitored.

61

All devices mounted on equipment shall be mounted in a location that is at a maximum of five feet above the floor. For steam and liquid applications, provide shutoff valves at piping takeoff points.

TEMPERATURE CONTROL PANELS Mount control panels adjacent to a

Mount control panels adjacent to associated equipment on vibration-free walls or freestanding angle iron supports. All control panel openings shall be plugged. Conduits and other penetrations on the top of the cabinets shall be sealed on the exterior of the cabinet with silicone caulk to resist water penetration.

One cabinet may accommodate more than one system in same equipment room. Provide permanent printed labeling for instruments and controls inside cabinet and engraved plastic nameplates on cabinet face.

9 Provide as-built control drawings of all systems served by each local panel in a location adjacent to or inside of panel cover.
 10 Provide a protective cover or envelope for drawings.

12 DIFFERENTIAL PRESSURE SWITCHES

Provide for each fan or pump specified, or shown on point list. Provide shutoff valves at piping takeoff points. Readjust pressure and/or differential setpoints for proper operation after final balancing is completed.

16 CURRENT STATUS SWITCHES

Provide for each fan or pump specified, or shown on point list. Set threshold adjustment to indicate belt or coupling loss. Readjust threshold for proper operation after final balancing is completed. Use the variable frequency drive (VFD) integrated relay output for motor status, if provided on the VFD, in lieu of a discrete current switch. A separate current switch provided under this Section shall be wired in parallel with the VFD motor status relay when a bypass starter is provided on the VFD to prove motor status in the bypass mode.

23 CONSTRUCTION VERIFICATION ITEMS

24 Contractor is responsible for utilizing the construction verification checklists supplied by the third party commissioning agent in 25 accordance with the procedures defined for construction verification in Section 01 91 00.

26 27 28

3

4

5 6

7

8

11

15

END OF SECTION

1	SECTION 23 09 24
2	DIRECT DIGITAL CONTROL SYSTEM FOR HVAC
3	
4	
5	PARII-GENERAL
7	SCODE
8	The work associated with this section be bid as part of the Division 23 scope of work.
9	
10	Work in this section includes Direct Digital Control (DDC) panels, main communication trunk, software programming, and other
11	equipment and accessories necessary to constitute a complete Direct Digital Control (DDC) system. This system interfaced with
12	pneumatic/electric controls (Section 23 09 14) utilizing Direct Digital Control signals to operate actuated control devices will
13	meet, in every respect, all operational and quality standards specified herein.
14	
15	PART 1 - GENERAL
16	Scope Delated Work
1/	Related Work
10	Reference Standards
20	Quality Assurance
20	Warranty
22	Submittals
23	LEED Requirements
24	Software License Agreement
25	Operation and Maintenance Data
26	Material Delivery and Storage
27	
28	PART 2 - PRODUCTS
29	General Lecal Control Danois
30 21	Local Control Panels Direct Digital Controls (DDC)
37	Networking/Communications
32	BAChet Requirements
34	Supervisory Controllers
35	System Software Features
36	Programmable Controllers
37	Application Specific Controllers - HVAC
38	Operator Interface Requirements
39	Web Based HTML Browser Interface
40	Training
41	
42	PARI 3 - EXECUTION
43 44	
44 45	
46	Commissioning, Verification and Closeout
47	
48	RELATED WORK
49	Applicable provisions of Division 1 govern work under this Section.
50	
51	Section 23 05 93 - Testing, Adjusting, and Balancing for HVAC – Coordination
52	Section 23 09 14 - Electric Instrumentation and Control Devices for HVAC
53	Section 23 09 93 – Sequence of Operations for HVAC Controls
54	Division 23 - HVAC - Equipment provided to be controlled or monitored
55	Division 20. Electrical Environment and ideal to be controlled on mentioned
20 57	Division 20 - Electrical - Equipment provided to be controlled or monitored
57	RFFFRENCE
59	Applicable provisions of Division 1 govern work under this section.
60	Application provident a Borent work and of this section.
61	REFERENCE STANDARDS
62	FCC Part 15, Subpart J, Class A - Digital Electronic Equipment to Radio Communication Interference
63	

1 QUALITY ASSURANCE

2 <u>MANUFACTURER:</u> 3 Acceptable manufa

Acceptable manufacturer shall be Honeywell WEBs-AX.

5 INSTALLER:

6 A firm specializing and experienced in DDC control system installation for no less than 5 years. All engineering and 7 commissioning work shall be done by qualified employees of this manufacturer, or qualified employees of an Authorized 8 Representative of that manufacturer that provides engineering and commissioning of the manufacturer's control equipment. Where installing contractor is an authorized representative of the control equipment manufacturer, submit written 9 confirmation of such authorization. Indicate in letter of authorization that the installing contractor has successfully completed 10 all necessary training required for the engineering, installation, and commissioning of equipment and systems to be provided 11 for the project and that such authorization has been in effect for a period of not less than three years. The letter of 12 authorization should also indicate that the installing contractor is authorized to install the manufacturer's DDC equipment at 13 14 the project location at the time the project is bid. Installation of the equipment shall be done by qualified mechanics and/or electricians in the direct employ or be directly subcontracted and under the supervision of the manufacturer or Authorized 15 Representative. The contractor providing and installing the equipment under this specification section shall be the same 16 contractor providing and installing equipment under the 23 09 14 specification section. 17

18

21

32

37

42

50

54

59

4

19 Contractor is limited to firms regularly employing a minimum of five full-time temperature control technicians within 50 miles 20 of the job site.

22 RESPONSE TIME:

23 During warrantee period, four (4) hours or less, 24-hours/day, 7 days/week.

24 25 ELECTRICAL STANDARDS:

Provide electrical products, which have been tested, listed and labeled by Underwriters' Laboratories (UL) and comply with
 NEMA standards.

29 <u>DDC Standards</u>: DDC manufacturer shall provide written proof with shop drawings that the equipment being provided is in 30 compliance with F.C.C. rules governing the control of interference caused by Digital Electronic Equipment to Radio 31 Communications (Part 15, Subpart J, Class A).

33 WARRANTY

Provide all services, materials and equipment necessary for the successful operation of the entire system for a period of one year after acceptance. The adjustment, required testing, and repair of the system includes all computer equipment, transmission equipment and all sensors and control devices.

The on-line support services shall allow the system supplier to dial out to monitor and control the facility's BAS. This remote connection to the facility shall be within 2 hours of the time that the problem is reported. This coverall shall be extended to include normal business hours, after business hours, weekends and holidays.

SUBMITTALS

43 Include the following information:44

Details of construction, layout, and location of each temperature control panel within the building, including instruments location in panel and labeling. Indicate which piece of mechanical equipment is associated with each controller and what area within the building is being served by that equipment. For terminal unit control, provide a room schedule that would list mechanical equipment tag, room number of space served, address of DDC controller, and any other pertinent information required for service.

51 EQUIPMENT COORDINATION

52 The Controls Contractor shall obtain approved equipment submittals from other contractors to determine equipment wiring 53 connections, to choose appropriate controllers, and to provide programming.

55 Control valve selections shall be based on flow rates shown in approved shop drawings.

56
57 Coordinate the control interface of all equipment with the equipment manufacturers prior to
58 submittal submission.

60 SHOP DRAWINGS

Submit shop drawings per Section 23 05 00. In addition, submit an electronic copy of the shop drawings in Adobe Acrobat
 (.pdf) format to the Owner for review.

1 2 3	Cross-reference all control components and point names in a single table located at the beginning of the submittal with the identical nomenclature used in this section.
4 5 6	Submittal shall also include a trunk cable schematic diagram depicting operator workstations, control panel locations and a description of the communication type, media and protocol.
7 8 9 10 11	System Architecture: Provide riser diagrams of wiring between central control unit and all control panels. This shall include specific protocols associated with each level within the architecture. Identify all interface equipment between CPU and control panels. The architecture shall include interface requirements with other systems including, but not limited to, security systems, lighting control, fire alarm, elevator status, and power monitoring system.
12 13 14	Diagrams shall include: Wiring diagrams and layouts for each control panel showing all termination numbers.
15 16 17	Schematic diagrams for all control, communication, and power wiring.
18 10	Provide a schematic drawing of the central system installation.
20	Label all cables and ports with computer manufacturers' model numbers and functions.
21	Show all interface wiring to the control system.
23	Identification of all control components connected to emergency power.
24	Schematic diagrams for all field sensors and controllers.
25 26	A schematic diagram of each controlled system. The schematics shall have all control points labeled. The schematics shall graphically show the location of all control elements in the system.
27 28 29	A schematic wiring diagram for each controlled system. Each schematic shall have all elements labeled. Where a control element is the same as that shown on the control system schematic, label it with the same name. Label all terminals.
30 31	A tabular instrumentation list for each controlled system. The table shall show element name, type of device, manufacturer, model number and product data sheet number.
32	All installation details and any other details required to demonstrate that the system will function properly.
33 34	All interface requirements with other systems.
35 36 37 38	The network infrastructure shall conform to the published guidelines for wire type, length, number of nodes per channel, termination, and other relevant wiring and infrastructure criteria as published. The number of nodes per channel shall be no more than 80% of the defined segment (logical or physical) limit in order to provide future system enhancement with minimal infrastructure modifications.
40 41 42 43	Sequences: Submit a complete description of the operation of the control system, including sequences of operation. The description shall include and reference a schematic diagram of the controlled system. The wording of the control sequences in the submittal shall match verbatim that included in the construction documents to ensure there are no sequence deviations from that intended by the Architect/Engineer. Clearly highlight any deviations from the specified sequences on the submittals.
44 45 46 47 48 49 50 51 51	Points List Schedule: Submit a complete points list of all points to be connected to the TCS and FMCS. The points list for each system controller shall include both inputs and outputs (I/O), point number, the controlled device associated with the I/O point, the location of the I/O device, and reference drawings. Where a control point is the same as that shown on the control system schematic, label it with the same name. Points list shall specifically identify alarms, trends, event history, archive, totalization, graphic points, and all mapped points from other systems (security systems, lighting control, fire alarm, etc.). Provide points lists, point naming convention, and factory support information for systems provided and integrated into the FMCS.
52 53 54	Damper Schedule: Schedule shall include a separate line for each damper and a column for each of the damper attributes:
55 56	Damper Identification Tag. Location.
57 58	Damper Type. Damper Size.

- Duct Size. 1 2 Arrangement. 3 Blade Type. 4 Velocity. 5 Pressure Drop. 6 Fail Position. 7 Actuator Identification Tag. 8 Actuator Type. 9 Mounting. 10 11 Valve Schedule: Valve manufacturer shall size valves and create a valve schedule. Schedule shall include a separate line for each valve and a column for each of the valve attributes: 12 13 Valve Identification Tag. 14 15 Location 16 Valve Type. 17 Valve Size. Pipe Size. 18 Configuration. 19 Flow Characteristics. 20 21 Capacity. Valve C_v. 22 23 Design Pressure Drop. Pressure Drop at Design Flow. 24 25 Fail Position. 26 Close-off Pressure. 27 Valve and Actuator Model Number and Type. 28 29 30 Product Data Sheets: Required for each component that includes: unique identification tag that is consistent throughout the 31 submittal, manufacturer's description, technical data, performance curves, installation/maintenance instructions, and other 32 relevant items. When manufacturer's literature applies to a product series rather than a specific product, the data specifically applicable to the project shall be highlighted or clearly indicated by other means. Each submitted piece of literature and 33 34 drawings shall clearly reference the specification and/or drawing that the submittal is to cover. General catalogs shall not be 35 accepted as cutsheets to fulfill submittal requirements. 36 37 Provide PICS files indicating the BACnet® functionality and configuration of each device. 38 39 Provide documentation of submitted products that have been tested and listed by the BACnet Testing Laboratory (BTL), or 40 provide a letter on the manufacturer's company letterhead indicating the anticipated date by which testing is expected to be completed. If, for any reason, BTL testing and listing has not been completed, a written commitment to upgrade installed 41 42 controls to a version that meets BTL testing and listing requirements in the event that problems are found during BTL testing 43 is required. 44 45 Graphic Display: Include a sample graphic of each system and component identified in the points list with a flowchart (site map) indicating how the graphics are to be linked to each other for system navigation. 46 47 48 Software: A list of operating system software, operator interface software, color graphic software, and third-party software. 49 50 Control System Demonstration and Acceptance: Provide a description of the proposed process, along with all reports and checklists to be used. 51 52 53 Clearly identify work by others in the submittal. 54 55 Quantities of items submitted may be reviewed but are the responsibility of the Contractor to 56 verify. 57 58 PRODUCT DATA 59 Submit manufacturer's specifications for each control device furnished, including installation instructions and startup
- 59 Submit manufacturer's specifications for each control device furnished, including installation instructions and startu 60 instructions.

- 1 General catalog sheets showing a series of the same device is not acceptable unless the specific model is clearly marked.
- Annotated software program documentation shall be submitted for system sequences, along with descriptive narratives of the sequence of operation of the entire system involved. Submit wiring diagram for each electrical control device along with other
- details required to demonstrate that the system has been coordinated and will function as a system.

6 MAINTENANCE DATA

7 Submit maintenance data and spare parts lists for each control device. Include this data in maintenance manual.

8 9 RECORD DRAWINGS

Prior to request for final payment provide complete composite record drawings to incorporate the DDC and Pneumatic/Electric field work. All software addressing for device communication shall be noted for all devices provided under this section and the communication addressing required for devices provided by others that are integrated into the direct digital control system provided under this section. Coordinate with the supplier of the equipment specified to be interfaced through digital communications for communication addressing. Provide circuit number of 120VAC panel power circuit(s) feeding each control panel on record drawings. Label circuit number(s) inside the panel served.

17 LEED REQUIREMENTS

16

24 25

26 27

28

29 30

31 32

34

This project shall meet the requirements of the U.S. GREEN BUILDING COUNCIL LEADERSHIP IN ENERGY AND ENVIRONMENTAL
 DESIGN (LEED) program.

- The project will attempt to achieve the U.S. Green Building Council's LEED Version 4.0 certification level: Silver.
- 23 This Contractor shall carefully examine the LEED portion of this specification for full compliance with the following LEED points:
 - "Energy & Atmosphere": Prerequisite 1, "Fundamental Building Systems Commissioning," Prerequisite 2 "Minimum Energy Performance," Credit 3 - "Additional Commissioning," and Credit 5 - "Measurement and Verification," as described by LEED.
 - "Indoor Environmental Quality": Prerequisite 1 "Minimum IAQ Performance," Credit 1 "Outdoor Air Delivery Monitoring," Credit 2 "Increased Ventilation," Credit 6 "Controllability of Systems," Credit 6.1 "Lighting Control," and Credit 6.2 "Thermal Comfort."
- All labor and materials required for these and any other LEED initiatives shall be provided without additional cost to the Owner.

35 SOFTWARE LICENSE AGREEMENT

36 The Owner shall be the named license holder of all software associated with any and all incremental work on the project(s). In 37 addition, the Owner shall receive ownership of all job-specific configuration documentation, data files, configuration tools, and 38 application-level software developed for the project. This shall include, but is not limited to, all custom, job-specific software code and documentation for all configuration and programming that is generated for a given project and/or configured for use 39 40 with the NAC, FMCS Server(s), and any related LAN/WAN/intranet and/or Internet connected routers and devices. Provide the Owner with all required IDs and passwords for access to any component or software program. The Owner shall determine 41 42 which organizations shall be named in the SI organization ID ("orgid") of all software licenses. Owner shall be free to direct the modification of the "orgid" in any software license, regardless of supplier. 43 44

45 JOB CONDITIONS

Cooperation with Other Trades: Coordinate the Work of this section with that of other sections to ensure that the Work will be carried out in an orderly fashion. It is this Contractor's responsibility to check the Contract Documents for possible conflicts between the Work of this section and that of other crafts in equipment location; pipe, duct and conduit runs; electrical outlets and fixtures; air diffusers; and structural and architectural features.

50

54

51 OPERATION AND MAINTENANCE DATA

All operations and maintenance data shall comply with the submission and content requirements specified under section GENERAL REQUIREMENTS.

55 MATERIAL DELIVERY AND STORAGE

For Provide factory shipping cartons for each piece of equipment and control device. This contractor is responsible for storage of equipment and materials inside and protected from the weather.

58 59 60 **PART 2 - PRODUCTS** 61 62 63

GENERAL

1 2

3

4

5 6

7

8 9

10

11 12

13 14 15

16 17

18

19 20

21

22 23 24

25

29

32

38

43

46 47

48

49

51 52

53

54

55 56 57

58 59

60

61 62

63

Provide DDC control products in sizes and of capacities as required, conforming to manufacturer's standard materials and components as published in their product information, designed and constructed as recommended by the manufacturer and as required for application indicate.

System shall be capable of operating with 120 VAC power supply, fully protected with a shutdown-restart circuit, and associated hardware and software.

LOCAL CONTROL PANELS

Use control panels with suitable mounting brackets for each supply fan system. Locate panel adjacent to system served.

Fabricate panels of 14 gauge furniture grade steel or 6063-T5 extruded aluminum alloy, totally enclosed on six sides, hinged door and keyed lock, with manufacturer's standard shop painted finish and color.

Provide UL listed cabinets for use with line voltage devices.

Plastic control enclosures will be approved provided all conduits are bonded and grounded.

Provide control panels for all DDC Controllers, ASC's and associated function modules. All controls to be in panels except for terminal unit controllers mounted within the terminal unit equipment enclosure or VAV box controllers designed to be directly mounted on air terminals. Provide terminal unit equipment enclosures with removable cover for all terminal units located in exposed ceilings that completely enclose the DDC controller and allow for conduit terminations.

Permanently label all controls; tag all control wiring.

26 DIRECT DIGITAL CONTROLS

System to be capable of integrating multiple building functions, including equipment supervision and control, alarmmanagement, energy management, and trend data collection.

DDC to consist of Supervisory Controllers, Programmable Controllers, stand-alone Application Specific Controllers (ASC's),
 Operators Terminals, Operator Workstations, DDC system servers, and other operator interface devices.

The system shall be modular in nature, and shall permit expansion of both capacity and functionality through the addition of sensors, actuators, ASC's, and operator devices.

The failure of any single component or network connection shall not interrupt the execution of control strategies at other operational devices.

39 NETWORKING/COMMUNICATIONS

The design of the DDC shall be networked. The highest level networking shall use Ethernet and the sub-level networking shall use serial communications. Inherent in the system's design shall be the ability to expand or modify the highest network either via a local area network (LAN), wide area network (WAN), or a combination of the two schemes.

The highest-level DDC communications network shall be capable of direct connection to and communication with a high-speed LAN or WAN utilizing an Ethernet connection. Communication protocol used shall be BACnet/IP.

The supervisory controller shall directly oversee a local network such that communications may be executed directly to and between programmable controllers and ASC's. All operator devices, either network resident or connected via dial-up modems, shall have the ability to access all points and application reports on the network.

50 Provide serial communication ports on all ASC's for operator's terminal communications with the DDC Controller.

Access to system data shall not be restricted by the hardware configuration of the DDC system.

Global data sharing or global point broadcasting shall allow point data to be shared between programmable controllers and ASC's when it would be impractical to locate multiple sensors.

Network design shall include the following provisions:

- Data transfer rates for alarm reporting and quick point status from multiple programmable controllers and ASC's. The minimum baud rate shall be 9600 baud.
- Support of any combination of programmable controllers and ASC's. A minimum of 32 programmable controllers and ASC's shall be supported on a single local network. The buss shall be addressable for up to 32 ASC's.
 - Detection of single or multiple failures of ASC's or the network media.
| 1 | • | Error detection, correction, and re-transmission to guarantee data integrity. |
|----------|-----------------|---|
| 2 | • | Use commonly available, multiple-sourced, networking components. |
| 3 | • | Use of an industry standard communication transport, such as, ARCNET, Ethernet, and IEEE RS-485 |
| 4 | | communications interface. |
| 5 | | |
| 6
7 | BACNET REQU | UIKEINIENIS
abort lovel network communications will utilize BACnet/ID over Ethernet and field lovel communications shall |
| 8 | utilize BACnet | t MSTP. No other communication protocol is accentable |
| 9 | | |
| 10 | All controller | s shall provide a Protocol Implementation Conformance Statement (PICS) and BACnet Interoperability Building |
| 11 | Blocks (BIBB" | S) as required by the American National Standards Institute/American Society of Heating, Refrigerating, and Air- |
| 12 | Conditioning | Engineers (ANSI/ASHRAE) Standard 135-2001, BACnet protocol. |
| 13 | | |
| 14 | In general all | devices shall support the following: |
| 15 | | |
| 16 | Segmentation | n Capability |
| 1/ | Segmentation | n requests supported |
| 18 | Segmentation | n responses supported |
| 19 | Standard Obi | iact Tunas Sunnartad |
| 20 | Stanuaru Obj | Analog input |
| 21 | • | Analog nutnut |
| 23 | • | Analog value |
| 24 | • | Binary input |
| 25 | • | Binary output |
| 26 | • | Binary value |
| 27 | • | Calendar |
| 28 | • | Device |
| 29 | • | Event enrollment |
| 30 | • | Group |
| 31 | • | Multistate input |
| 32 | • | Multistate output |
| 33
24 | • | Multistate value |
| 25 | • | Schedule |
| 36 | - | |
| 37 | Character Set | s supported |
| 38 | • | ANSI X3.4 |
| 39 | • | ISO 10646 Universal Character Set-2 |
| 40 | | |
| 41 | All highest lev | el networked supervisory devices shall support the following: |
| 42 | | |
| 43 | Data Link Lay | er Option |
| 44 | • | BACnet Internet Protocol (IP) (Annex J) |
| 45
46 | Notworking C | ntions |
| 40 | Networking C | ptions |
| 47 | BACnet/IP Bro | nadcast Management Device (BBDM) |
| 49 | BACILCY II DI | |
| 50 | SUPERVISOR | Y CONTROLLERS |
| 51 | Supervisory c | ontrollers shall be microprocessor-based, multi-tasking, multi-user and digital control processors. |
| 52 | | |
| 53 | Each supervis | ory controller shall have sufficient memory to support its own operating system and databases including: |
| 54 | • | Control processes |
| 55 | • | Energy management application |
| 56 | • | Alarm management |
| 5/ | • | Henu udid
Maintananca support applications |
| 50 | - | Onerator I/O |
| 60 | • | Dial-up communications |
| 61 | • | Manual override monitoring |
| 62 | | ~
~ |

2	actuators.
3	Supervisory controllers shall provide at least two DC 222C or LICD social communication parts or Ethernat parts for simultaneous
4	supervisory controllers shall provide at least two RS-232C or USB serial communication ports or Ethernet ports for simultaneous
6	operation of multiple operator i/o devices, such as laptop computers, personal computers, and video display terminals.
7	Supervisory controllers shall monitor the status of all overrides and include this information in the logs and summaries to
8	inform the operator that automatic control has been inhibited.
9	·
10	Each supervisory controller shall continuously perform self-diagnostics, communications diagnostics, and diagnostics of all
11	subsidiary equipment. Supervisory controllers shall provide both local and remote annunciation of any detected component
12	failures, or repeated failure to establish communication. Indication of the diagnostic results shall be provided at each
13	supervisory controller.
14	the least of which has more than the stand shows to attend the second second state the state the second s
15 16	isolation shall be provided at all network terminations, as well as all field point terminations, to suppress induced voltage
10	in the same conduit as high voltage wiring accentable by electrical code
18	in the same conduit as high voltage winnig acceptable by cleating code.
19	In the event of the loss of normal power, there shall be an orderly shutdown of the supervisory controller to prevent the loss of
20	data base or operating system software. Non-volatile memory shall be incorporated for all critical controller configuration data,
21	and battery backup shall be provided to support the real-time clock and all volatile memory for a minimum of 72 hours.
22	
23	Upon restoration of normal power, the supervisory controller shall automatically resume full operation without manual
24	intervention.
25	
26	Should supervisory controller memory be lost for any reason, the supervisory controller shall have the capability of reloading
27	It's programming via high speed local area network from the control system archive workstation or server, the local RS-232C
20	port, or telephone line dial-lin.
30	SYSTEM SOFTWARE FEATURES
31	All necessary software to form a complete operating system, as described in this specification, shall be provided as an integral
32	part of the supervisory controller, and shall not be dependent upon higher level computer for execution.
33	Control software shall include a provision for limiting the number of times that each piece of equipment may be cycled within
34	any one-hour period.
35	
36	The system shall provide protection against excessive demand situations during start-up periods by automatically introducing
37	time delays between successive start commands to heavy electrical loads.
38	
39	Supervisory controllers shall have the ability to perform any or all of the following energy management routines:
40 41	 Calendar based scheduling
41 42	Calciular based scheduling Holiday scheduling
43	Ontimal start
44	Optimal stop
45	Demand limiting
46	Load rolling
47	Heating/cooling interlock
48	
49	All programs to be executed automatically without the need for operator intervention, and be flexible enough to allow user
50	customization. Programs shall be applied to building equipment described in Section 23 09 93 of this specification.
51	
52	Supervisory controllers shall be able to execute configured processes defined by the user to automatically perform calculations
53	and control routines.
54 55	It shall be possible to use any of the following in a configured process:
56	Any system-measured noint data or status
57	Any calculated data
58	Any results from other processes
59	Boolean logic operators (and, or)
60	
61	Configured processes may be triggered based on any combination of the following:
62	Time of day
63	Calendar date

The system shall be modular in nature, and shall permit easy expansion through the addition of field controllers, sensors, and

- 1 Other processes . 2 • Events (e.g., point alarms) 3 4 A single process shall be able to incorporate measured or calculated data from any and all other ASC's. 5 6 A single process shall be able to issue commands to points in any and all other programmable controllers and ASC's on the local 7 network. 8 9 Alarm management shall be provided to monitor, buffer, and direct alarm reports to operator devices and memory files. Each supervisory controller shall perform distributed; independent alarm analysis and filtering to minimize network traffic and 10 prevent alarms from being lost. At no time shall the ability of supervisory controllers to report alarms be affected by either 11 12 operator activity at the local I/O device or communications with other ASC's on the network. 13 14 All alarm or point change reports shall include the English language description of each point and the time and date of the 15 occurrence. 16 17 The user shall be able to define the specific system reaction for each point. Alarms shall be prioritized to minimize nuisance 18 reporting and to speed operator response to critical alarms. A minimum of three priority levels shall be provided. Users shall have the ability to manually inhibit alarm reporting for each point. 19 20 The user shall also be able to define conditions under which point changes need to be acknowledged by an operator and/or 21 22 logged for analysis at a later date. 23 24 Alarms reports and messages shall be directed to an operator device. 25 26 In addition to the point's descriptor and the time and date, the user shall be able to print, display or store a 60-character alarm 27 message to more fully describe the alarm condition or direct operator response. 28 29 Each supervisory controller shall be capable of storing a library of at least 100 messages. Each message may be assignable to 30 any number of points in the panel. 31 32 A data collection utility shall be provided to automatically sample, store, and display system data. 33 34 Measured and calculated analog and binary data shall be assignable to user definable trends for the purpose of collecting 35 operator specified performance data over extended periods of time. Sample intervals of 1 minute to 24 hours, in one minute or 36 one hour intervals, shall be provided. Each supervisory controller shall have a dedicated buffer for trend data and shall be 37 capable of storing 16 trend logs. Each trend log shall have up to four points trended at 48 data samples each. Data shall be 38 stored at the supervisory controller and up-loaded to the DDC system server when archiving is desired. 39 40 Supervisory controllers shall automatically accumulate and store runtime hours for binary input and output points specified in 41 Section 23 09 14 of this specification. 42 43 Supervisory controllers shall automatically sample, calculate and store consumption totals on a daily, weekly, or monthly basis, 44 user defined, for user-selected analog and binary pulse input type points. 45 46 Totalization shall provide calculation and storage accumulations of up to 9,999,999 units (e.g., KWH, gallons KBTU, tons, etc.). 47 48 The totalization routine shall have a sampling resolution of one minute. 49 50 The user shall have the ability to define a warning limit. Unique, user specified messages shall be generated when the limit is 51 reached. 52 53 The information available from pulse totalization shall include, but not be limited to, the following: 54 Peak demand, with date and time stamp • 55 24-hour demand log • 56 • Accumulated KWH for day 57 Sunday through Saturday KWH usage • Demand KW annual history for past 12 periods 58 • 59 KWH annual history for past periods 60 61 Supervisory controllers shall have the ability to count events, such as the number of times a pump or fan system is cycled on 62 and off.
- 63

1 The event totalization feature shall be able to store the records associated with a minimum of 9,999,999 events before reset. 2 3 **PROGRAMMABLE CONTROLLERS** 4 Programmable controllers shall be provided with a software program that shall allow the user to design flexible software 5 algorithms for the control sequences as described in Sections 23 09 14 and 23 09 93 portions of this specification. 6 7 Programmable controllers shall support all necessary point inputs and outputs to perform the specified control sequence in a 8 totally stand-alone fashion. 9 Each programmable controller shall perform its own limit and status monitoring and analysis to maximize network performance 10 11 by reducing unnecessary communications. 12 Each programmable controller shall support the use of a locally mounted status and adjust panel interface to allow for the local 13 14 adjustment of all setpoints, temporary override of any input or output points and status of all points directly at the controller. The capabilities of the locally mounted status and adjust panel shall include, but not be limited to, the following information for 15 the programmable controllers to which: 16 17 **Display temperatures** 18 **Display status** 19 • 20 • **Display setpoints** Display control parameters 21 • Override binary output control 22 23 • Override analog output control Override analog setpoints 24 • Modification of gain and offset constants 25 26 All system setpoints, proportional bands, control algorithms, and any other programmable parameters shall be stored such that 27 28 a power failure of any duration does not necessitate reprogramming the programmable controller. Programmable controllers shall support, but not be limited to, the following configurations of systems to address current 29 30 requirements as described in Sections 23 09 14 and 23 09 93 portions of this specification, and for future expansion of air handling units: 31 32 Mixed air handling units 100 percent outside air handling units 33 • Boiler or chiller plants with pump logic 34 • Zone pressurization of labs 35 • 36 • Smoke control systems 37 Generic system interlocking through hardware • 38 **APPLICATION SPECIFIC CONTROLLERS - HVAC APPLICATIONS** 39 40 Each supervisory controller shall be able to extend its monitoring and control through the use of stand-alone application specific controllers (ASC's). 41 42 Each ASC shall operate as a stand-alone controller capable of performing its specified control responsibilities independently of 43 other controllers in the network. Each ASC shall be a microprocessor based, multi-tasking, real-time digital control processor. 44 45 Each ASC shall have sufficient memory to support its own operating system and databases including: 46 47 • **Control Processes Energy Management Applications** 48 . Operator I/O (Portable Service Terminal) 49 • 50 The operator interface to any ASC point or program shall be through the supervisory controller connection to any ASC on the 51 52 network. 53 ASC's shall directly support the temporary use of a portable service terminal that can be connected to the ASC via zone 54 temperature or directly at the controller. The capabilities of the portable service terminal shall include, but not be limited to, 55 56 the following information for the: **Display temperatures** 57 • **Display status** 58 • 59 Display setpoints Display control parameters 60 Override binary output control 61 62 Override analog output control Override analog setpoints 63 •

1	 Modification of gain and offset constants
2	
3	All system setpoints, proportional bands, control algorithms, and any other programmable parameters shall be stored such that
4	a power failure of any duration does not necessitate reprogramming the ASC.
5	
6	ASC's shall support, but not be limited to, the following configurations of systems to address current requirements as described
7	in Sections 23 09 14 and 23 09 93 portions of this specification, and for future expansion of air handling units:
8	Variable Air Volume Terminals
9	Reheat Terminals
10	Fan Coils
11	Packaged Air Handling Units
12	
13	For butterfly type Variable Air Volume (VAV) Terminals, provide differential pressure transducers and damper actuators for flow
14	measurement and actuation of the VAV terminal damper. Pressure transducers shall be Ashcroft - Model CXLdp High Accuracy
15	Differential Pressure Transmitter.
16	
17	Provide filter on high side of flow pickups if flow measurement device requires airflow through the device.
18	
19	Terminal unit temperature sensors shall be furnished under this Section. Provide with digital displays with setpoint
20	adjustments and/or manual occupancy override and indication of occupancy status. Provide information to the Owner on
21	sensor colors offered by the manufacturer and obtain approval on what color should be provided on the project. Provide
22	setpoint adjustment as specified in the DDC input/Output Summary Table and sequence of operation.
23	Describe a method to view and write a community of community for the stars for flaw, as we have the NAV to main all the surply the DDC
24 25	Provide a method to view and print a summary of current k-factors for how correction for each VAV terminal through the DDC
25	system. The summary shall have a minimum of 50 k-factors per group of vAv terminals.
20	
27	OF ENATOR INTERFACE REQUIREMENTS
20	COMMAND ENTRY/MENILI SELECTION PROCESS:
30	Operator interface software shall minimize operator training through the use of English language promoting and English
31	language noint identification.
32	
33	TEXT-BASED DISPLAYS:
34	The operator interface shall provide consistent text-based displays of all system point and application data described in this
35	specification. Point identification, engineering units, status indication, and application naming conventions shall be the same at
36	all operator devices.
37	
38	<u>GRAPHIC-BASED DISPLAYS</u> :
39	The operator interface shall provide graphic based displays of each system. The point data associated with each system shall
40	dynamically update at a minimum of every 30 seconds. Graphic displays shall be linked to each other to provide a "drill down"
41	capability from main graphic displays to more specific system based displays. Provide a building level graphic display that links
42	to system graphics. For systems that have ASC controlled terminal unit controls, provide a building floor plan with dynamic
43	temperatures shown on the graphic that can be drilled into for more specific terminal information. Points provided in the
44	graphic shall have the override and adjust capability specified under operator commands.
45	
46	Update existing graphics for existing systems with new floor plans, air terminal locations and other changes as a result of this
47	project.
48	
49	Provide new graphics for all new systems.
50	
51	PASSWORD PROTECTION:
52	Multiple-level password access protection shall be provided to allow the user/manager to limit control, display, and data base
53	manipulation capabilities as ne deems appropriate for each user, based upon an assigned password.
54	Descuerde shall be evently the same for all energies
55	Passwords shall be exactly the same for all operator devices.
50	A minimum of three lough of access shall be supported.
57	A minimum of three levels of access shall be supported:
50	 Level 1. Data dicess and unspidy Level 2 - Level 1 + enerator everyides and commands
59	 Level 2 - Level 1 + Operator overhues and continualities Level 3 - Level 2 + database generation and modification
60 61	- Level 5 - Level 2 + ualabase generation and mounilation
62	A minimum of A passwords shall be supported at each supervisory controller
02	A minimum of + passwords shall be supported at each supervisory controller.

1 2	Operators will be able to perform only those commands available for their respective passwords. Menu selections displayed at any operator device shall be limited to only those items defined for the access level of the password used to log-on.		
3 4 5	Provide user devices on-lin	definable, automatic log-off timers of from 1 to 60 minutes to prevent operators from inadvertently leaving ne.	
6 7	OPERATOR CO	OMMANDS:	
, 8	The operator	interface shall allow the operator to perform commands including but not limited to the following:	
9	•	Start-up or shutdown selected equipment	
n	•	Adjust setnoints	
1	•	Override analog and binary outputs	
2	•	Add/modify/delete time programming	
.3	•	Enable/disable process execution	
4	•	Lock/unlock alarm reporting for each point	
.5	•	Enable/disable totalization for each point	
6	•	Enable/disable trending	
7	•	Enter temporary override schedules	
8	•	Define holiday schedules	
9	•	Change time/date	
20	•	Enter/modify analog alarm limits	
21	•	Enable/disable analog alarm limits	
22	•	Enable/disable demand limiting	
23	•	Enable/disable duty cycle	
24			
25	LOGS AND SU	IMMARIES:	
26	Reports shall	be generated manually, and directed to the displays. As a minimum, the system shall allow the user to easily	
27	obtain the fol	lowing general listing of all points in the system that shall include, but not be limited to:	
28	•	Points currently in alarm	
<u>19</u>	•		
50 1	•	Points currently in override status	
51 51	•	Points in weekly schedules	
5Z	•	Holiday programming	
5 7 /	Summarios ch	all be provided for specific points, for a logical point group, for a user selected group of groups, or for the entire	
25	facility without	tail be provided for specific points, for a logical point group, for a user-selected group of groups, or for the efficient due to the hardware configuration on the facility management system. Under no conditions shall the	
35 86	operator need	to specify the address of hardware controller to obtain system information	
,0 87	operator need	to spearly the address of hardware controller to obtain system mormation.	
88	SYSTEM CON	FIGURATION AND DEFINITION:	
39	All temperatu	ire and equipment control strategies and energy management routines shall be definable by the operator. System	
0 1	definition and	d modification procedures shall not interfere with normal system operation and control.	
2	The system s	hall be provided complete with all equipment, software, and documentation necessary to allow an operator to	
3	independently	y perform the following functions:	
4	•	Add/delete/modify application specific controllers	
5	•	Add/delete/modify points of any type, and all associated point parameters, and tuning constants	
-6	•	Add/delete/modify alarm reporting definition for each point	
7	•	Add/delete/modify energy management applications	
8	•	Add/delete/modify time and calendar-based programming	
9	•	Add/delete/modify totalization for every point	
50	•	Add/delete/modify historical data trending for every point	
51	•	Add/delete/modify configured control processes	
o2	•	Add/delete/modify dial-up telecommunication definition	
5	•		
94 15	•	Aud/delete/moully alarm messages	
56	PROCRANAN		
7	Definition of	operator device characteristics. ASC's individual points and shall be performed through fill-in-the-blank	
8	templates		
9	templates.		
0	NETWORK-W	IDE STRATEGY DEVELOPMENT:	
1	Inputs and ou	utputs for any process shall not be restricted to a single ASC, but shall be able to include data from any and all	
2	other ASC's to	allow the development of network-wide control strategies.	
53			

1 SYSTEM DEFINITION/CONTROL SEQUENCE DOCUMENTATION:

2 All portions of system definition shall be self-documenting and be capable of providing hardcopy printouts of all configuration 3 and application data.

5 WEB BASED HTML BROWSER INTERFACE

6 Provide a HTML based browser interface (Web Server) for accessing the DDC system. This shall include all hardware and 7 software to provide an Ethernet twisted pair connection to the owners local or wide area network (LAN or WAN) that can be 8 used to access the DDC system through a standard internet browser.

- 10 All information shall be provided to the owners IT staff to facilitate connection through the owners LAN/WAN.
- 11

9

4

At a minimum, this interface shall be capable of all functions described under the Operator Interface section, Password
 Protection, Operator Commands, and Logs and Summary subsections of this specification.

PART 3 - EXECUTION

14 15

16

16

18 GENERAL

All electronic work required as an integral part of the Direct Digital Control system work is the responsibility of this section
 unless specifically indicated otherwise in this section, Section 23 09 14, or in Division 26.

This contractor shall provide all labor, materials, engineering, software, permits, tools, checkout and certificates required to install a complete Direct Digital Control system as herein specified.

24

Any and all points added with this project shall be grouped for display purposes into the system such that all points associated with a new or existing DDC system can appear together on the CRT display or printed log. Assignment of points to a group shall not be restricted by hardware configuration of the points of direct digital control. It shall be possible to assign a point to appear in more than one system. An English descriptor and an alpha/numeric identifier shall identify each system.

29

This Direct Digital Control system as herein specified shall be fully integrated and completely installed by this section. It shall include all required computer CPU software and hardware. Include the engineering, installation, supervision, calibration, software programming, and checkout necessary for a fully operational system.

33 34

41

43

35 INSTALLATION

All work and materials are to conform in every detail to the rules and requirements of the National Electrical Code and present manufacturing standards. All wiring and cable installation shall conform with the wiring installation as specified in the installation section of Section 23 09 14. All material shall be UL approved.

40 Install system and materials in accordance with manufacturer's instructions, rough-in drawings and details on drawings.

42 Line voltage wiring to power the DDC Controllers, not provided by the Division 26 contractor, to be by this contractor.

Provide uninterruptable power supplies where necessary to provide proper startup of equipment or to accomplish power
 restart control sequences specified.

47 Mount control panels adjacent to associated equipment on vibration-free walls or free-standing angle iron supports. One 48 cabinet may accommodate more than one system in same equipment room. Provide engraved plastic nameplates for 49 instruments and controls inside cabinet and on cabinet face.

50

Provide as-built control drawings of all systems served by each local panel in a location adjacent to or inside of panel cover.
 Provide a protective cover or envelope for drawings.

- 53
- 54 Cable tray routing of the communication trunks is acceptable. 55

56 Provide all necessary routers and or repeaters to accomplish connection to the BAN via the panel-mounted port provided.

57 58 Provide two data jacks in control panels housing supervisory controllers and allocate 6"x6" for each data jack in the panel. The 59 first jack will be used for connecting the supervisory controller to the BAN. The second jack will be used as a spare for 60 connecting to the BAN by service personnel.

61

Provide an input for a service shutdown toggle switch for each air handling unit system provided inside the temperature control
 panel that will initiate a logical shutdown of the air handling unit system.

All tubing, cable and individual wiring is to be permanently tagged, with numbers corresponding with "Record Drawings", spares are to be labelled as "Spare".

Provide technician to work with air balancing contractor and/or provide balancing contractor with necessary hardware to override DDC controllers for air balancing.

TRAINING

1

2

3 4

5

6 7

8

9

10

11 12

13

14

15 16

17

24

26 27

28

29

30

34

Contractor to provide factory authorized representative and/or field personnel knowledgeable with the operations, maintenance and troubleshooting of the system and/or components defined within this section for a minimum period of 20 hours.

Provide two follow-up visits for troubleshooting and instruction, one six months after substantial completion and the other at the end of the warranty period. Length of each visit to be not less than 3 hours or the time necessary to provide required information and complete troubleshooting and inspection activity for all controls installed under 23 09 23, 23 09 14, and 23 09 93. Coordinate the visit with the owner and provide an inspection report to the owner of any deficiencies found.

COMMISSIONING, VERIFICATION, AND CLOSEOUT

18 Provide technicians to work with mechanical contractor, balancing contractor and owner to assist with balancing, air pressurization set-up, project set-up and closeout. 19 20

Prior to owner involvement, provide documentation to demonstrate that all points, input and output, have been checked out 21 22 and verified operational, note any points not operating properly with notation of reason. All points to be operational prior to 23 owner involvement with project set-up and commissioning.

25 Prior to owner involvement, the temperature control contractor shall self-commission all systems to verify (at minimum):

All systems are calibrated, under control and functioning as specified and designed.

Verification that all points, alarms and equipment are integrated into the BAS and are graphically represented (accurately).

- Air terminals operate and are under control.
- Dampers operate and are under control. 31 32
 - Temperature control valves operate and are under control.
- Setpoints and deadbands. 33
 - Sensor integration and calibration
- 35 Equipment interlocks 36

37 Contractor to provide written documentation that all systems have been self-commissioned, verified to be in working order and 38 are suitable for owner commissioning and system set-up.

39 40 41

1	SECTION 23 09 93
2	SEQUENCE OF OPERATION FOR HVAC CONTROLS
3	
4	
5	PARI 1-GENERAL
7	SCOPE
, 8 9	The work associated with this section will be bid as part of the Division 23 scope of work.
10 11 12	This section includes control sequences for HVAC equipment as well as equipment furnished by others that may need monitoring or control. Included are the following topics:
13	PART 1 - GENERAL
14	Scope
15	Related Work
16	Reference
17	Description of Work
18	Submittals
19	Operation and Maintenance Data
20	Design cinteria
22	PART 2 - PRODUCTS
23	Not Applicable
24	
25	PART 3 - EXECUTION
26	General Control Sequences
27	Project Specific Control Sequences
28	
29	RELATED WORK Applicable provisions of Division 1 govern work under this Section
30 31	Applicable provisions of Division 1 govern work under this section.
32	Section 23 05 93 - Testing, Adjusting, and Balancing for HVAC – Coordination
33	Section 23 09 14 - Pneumatic and Electric Instrumentation for HVAC Control
34	Section 23 09 24 – Direct Digital Control System for HVAC
35	Division 23 - HVAC - Equipment provided to be controlled or monitored
36 37	Division 26 - Electrical - Equipment provided to be controlled or monitored
38	REFERENCE
39	Section 23 09 14 work includes furnishing and installing all field devices, including electronic sensors for the DDC of this section,
40	equipment, and all related field wiring, interlocking control wiring between equipment, pneumatic tubing, sensor mounting,
41	etc., that is covered in that section.
42 43	Motorized control dampers and actuators, thermowells (temperature sensing wells), automatic control valves and their
44	actuators are also covered in Section 23 09 14.
45	
46	DESCRIPTION OF WORK
47	Control sequences are hereby defined as the manner and method by which automatic controls function. Requirements for
48	each type of operation are specified in this section.
49 50	Operation equipment, devices and system components required for automatic central systems are specified in other Division
50	23 control sections of these specifications
52	
53	All temperature, humidity, and pressure sensing, and all other control signal transportation for the control sequences shall be
54	furnished under Section 23 09 14. All pneumatic, electronic, and electric input/output signals shall be extended under Section
55	23 09 14, with adequate lead length for termination within the appropriate control panel being provided under Section 23 09
56	23.
5/	Sequences for equipment controlled by Direct Digital Controls (DDC) as specified are accomplished by bardware and software
50 59	provided under Section 23.09.23. Sequences for equipment controlled by pneumatic or electric self-contained controls are
60	accomplished by hardware provided under Section 23 09 14.
61	
62	SUBMITTALS
63	Refer to Division 1, General Conditions, Submittals, Section 23 05 00, Sections 23 09 14, and 23 09 23 for descriptions of what
64	should be included in the submittals.

3

4

5

6

7 8

9

10

11

13 14 15

16 17 18

19 20

21

30

37

Shop drawings shall be provided by contractor(s) providing equipment under Sections 23 09 24 and 23 09 14. The contractor providing the DDC equipment shall provide a complete narrative of the sequence of operations for equipment that is controlled through the DDC system. The contractor providing the 23 09 14 equipment shall provide a complete narrative of the sequence of operation for equipment that is controlled directly from that equipment (without control logic through the DDC system). The narrative of the sequence of operation shall not be a verbatim copy of the sequences contained herein, but shall reflect the actual operation as applied by the contractor.

OPERATION AND MAINTENANCE DATA

All operations and maintenance data shall comply with the submission and content requirements specified under section GENERAL REQUIREMENTS.

12 DESIGN CRITERIA

Reference Section 23 09 14.

PART2-PRODUCTS

Not applicable to this Section – reference Sections 23 09 23 and 23 09 14 for product descriptions.

PART 3 - EXECUTION

22 23 GENERAL CONTROL SEQUENCES

24 BACNET OBJECTS:

All hardwired points listed and any setpoints, timers, or other control elements that are specified to be adjustable (adj.) in the following control sequences shall be mapped as BACnet objects and be available on the user interface to be adjusted. Consult with the user agency HVAC and/or DDC personnel prior to programming to determine if there are any items that they do not want to have mapped as BACnet objects. This is especially important for DDC controlled items that are duplicative, i.e., air terminal units.

31 BACNET ADDRESSING:

BACnet instance ID's shall be coordinated with the agencies established BACnet instance ID addressing scheme. If there is not such a scheme in place, the contractor(s) providing BACnet DDC controllers shall work with the agency to establish such a scheme and document this in the asbuilt control drawings. BACnet/IP addressing shall be coordinated with the agency prior to installation. BACnet MSTP addressing shall be addressed to provide for consecutive addressing to provide for the best speed of response. Max Master address shall be set appropriately for speed of response.

38 USER INTERFACE/FEATURE SOFTWARE:

Consult with the user agency HVAC and/or DDC personnel prior to programming to determine BACnet object naming conventions, user views, graphic layout, security matrix, alarming, trending, and scheduling preferences desired by the agency. Failure to consult and come to agreement prior to programming shall require the DDC contractor to make changes in the above listed items as desired by the user agency to the system at no cost.

44 SETPOINTS:

All setpoints indicated in the control specification are to be adjustable. The setpoints shall be readily available to be modified in the mechanical system software system summary (either textual or graphic based) and under the same software level as hardware points. Some less used setpoints may be provided on a lower software level, if requested by the user Agency for clarity. The setpoints indicated herein are only specified as a calculated starting point (or initial system operation).

It is expected that setpoint adjustments and control loop tuning shall be required to provide optimum system operation based
 on requirements of the building.

The control contractor shall work with the balancing contractor and the user Agency to provide the final system setpoint adjustments and control loop tuning after the system is in operation and building is in use. Document all final setpoints on the as-built control drawings. Any questions regarding the intended operation of the HVAC equipment and control systems shall be referred to the HVAC design engineer through the appropriate construction communication process. The following setpoints should be used as initial setpoints unless otherwise specified in the individual control sequences:

58

- 59 Occupied Room Space Terminal Unit Heating: 68º F
- 60 Occupied Room Space Terminal Unit Cooling: 75º F
- 61 Entry Way and Stair Heating: 60° F
- 62 Mechanical or Unoccupied Space Ventilation: 82º F
- 63 Mechanical or Unoccupied Space Heating: 60º F
- 64

1 ANTI-CYCLING:

When HVAC equipment or a sequence is specified to be started and stopped by a temperature, humidity, pressure setpoint or 2 3 any other controlled variable, there shall be an adjustable differential setpoint that shall be set to prevent short cycling of the systems and equipment due to minor changes in the controlled variable. Temperature differential setpoints shall be set at 2° F 4 and non-temperature setpoints shall be set at 10% of the controlled range unless otherwise specified. Setpoints shall indicate 5 at when the process should be turned on. Heating and cooling differentials shall be set for above setpoint and will be used to 6 7 turn the process off. For example, an economizer sequence called to switch at 68° F, would turn on at 68° F and off at 70° F 8 since it is a cooling function. A heating lockout setpoint of 50° F would turn on heating control at 50° F and off at 52° F Nontemperature differentials shall be set above setpoint if the setpoint is indicating a minimum value or below setpoint if the 9 setpoint is indicating a maximum value. Provide minimum runtime timers for loads that are cycled to prevent over-cycling. 10 Timers shall be set as specified or as needed to prevent damage or excessive wear to the equipment. Unless otherwise 11 12 specified in the individual control sequences, fans and pumps shall have a minimum runtime on timers of 15 minutes (adj.) and off timers of 5 minutes (adj.). Safeties shall override runtime timers. 13 14

15 DEADBANDS:

Provide deadbands for all DDC control loops to prevent constant hunting of output signals to controlled devices. Deadbands shall be set to provide adequate control around setpoint as follows unless otherwise specified in the individual control sequences:

19

20 ALARMS:

Provide all alarmed points with adjustable time delays to prevent nuisance tripping under normal operation and on equipment start-up. For all commanded outputs that have status feedback, provide an alarm that will indicate the commanded output is not in its commanded state. Provide alarms on all points as indicated on point charts. For existing campus automations systems, add/delete what is called on the point charts for after consultation with user Agency to provide consistent alarming throughout the automation system.

26 27 TREND DATA:

Trends shall be provided for all hardware I/O points and integrated points listed as having trending in the DDC point charts and 28 29 for analog and binary data points mapped to the user interface as specified below. Interval trending with sample intervals of 10 30 minutes shall be provided on analog process variables (this includes both analog inputs and calculated process variables) and 31 process outputs. In addition, provide Change of Value (CoV) trending for all binary input and output points, binary data points 32 mapped to the user interface, and for all analog inputs and process variables. Analog inputs and process variables and setpoints shall be set at 5% CoV of setpoint. Analog process outputs shall have CoV set at 5% of the output range. Other analog 33 34 data points mapped to the user interface shall have CoV trends of 5% of their range. Consult with the user agency specific 35 standard values for interval and CoV trends for different points and control types. Data shall be stored at the supervisory 36 controller or in the field controller and up-loaded to the DDC system server when archiving is desired. Consult with the user agency to determine which trends should be archived. Trending shall be in place for a minimum of 24 hours prior to functional 37 38 testing by the commissioning provider.

39

40 EQUIPMENT START/STOP FAILURE STATES:

41 All start/stop points for equipment shall utilize normally open contacts unless called out specifically in the individual control 42 sequences.

43

44 LEAD/STANDBY SEQUENCING:

For sequences that call for lead/lag of equipment connected to building automation systems, the lead device shall be able to be chosen through a selectable day of the week and time of day through the building automation system. Coordinate with the user Agency for scheduling switchover and frequency. Unless otherwise directed, switchover shall occur at 10AM Tuesday and shall rotate the lead device on a weekly cycle rotating through all devices sequentially. For standalone lead/lag sequence controllers (non-DDC), the lead device shall be selected by a switch on the panel face.

50

51 VARIABLE FREQUENCY DRIVE (VFD) MOTOR RUN STATUS:

52 Use the VFD programmable relay dry contact output specified to be provided with the VFD under Section 23 05 14 to prove 53 motor run status and detect belt loss or coupling break. If a bypass contactor is provided with the VFD, provide an adjustable 54 current switch and wire it in parallel with the VFD output for proving motor status.

54 current switch and wire it in parallel with the VFD output for proving motor 55

56 VFD MINIMUM SPEED & RAMP TIMERS:

57 The VFD start-up technician shall work with the DDC Temperature Control Contractor determine the minimum speed required 58 for the motor controlled by the VFD to provide cooling of the motor as installed to prevent heat related problems. This 59 minimum speed shall be set in the VFD controller. The VFD start-up technician shall work with the DDC Temperature Control 60 Contractor to set the acceleration and deceleration timers in the VFD controller at 30 seconds for motors less than 40 HP and 61 60 seconds for motors 40 HP and greater.

- 62
- 63
- 64

CURRENT SWITCH SETUP: 1 2

When current switches are used for proving fan or pump status, they shall be set up so that they will detect belt or coupling loss by the reduction in current draw on loss of coupled load. The current switch set up shall be redone by the 23 09 14 contractor after the balancer is complete.

4 5

6 FAN INTERLOCKING:

3

16

7 Provide interlocks between supply and return or exhaust fan systems as scheduled on the plans or called out in individual control sequences. If DDC controlled, interlocks shall be done through DDC start/stop points unless otherwise specified in 8 individual control sequences. If not DDC controlled, interlocks shall be accomplished via hardwire interlocks between fan 9 10 starters or VFD's. 11

12 THERMOSTATS AND SENSORS:

All devices and equipment including terminal units, specified to be controlled in a control sequence by a thermostat or sensor, 13 shall be provided with a thermostat or sensor, whether or not the device is indicated on the plans. Consult the HVAC design 14 15 engineer for the thermostat or sensor location.

WEEKLY SCHEDULING 17

Provide scheduling of DDC terminal units in groups based on occupancy. Work with the user Agency to determine how many 18 19 groups are required and which zones should be included. Individual terminal units shall be able to receive temporary schedules that shall override the group schedules. Temporary override buttons at the zone sensor (where specified on point charts) shall 20 override the scheduling to occupied. When groups that consist of more than 20% of terminal units are indexed to occupied, the 21 22 associated air handling unit shall start if not already running. 23

24 CONTROLLED VARIABLE REQUIREMENTS

25 All controlled variables, i.e., static pressure, differential pressure, temperature, humidity, etc., shall be wired directly to the DDC 26 controller in which the software PID loop or other similar software loop resides unless the control sequence specifically allows the controlled variable to be routed over the network. Where a controlled variable is used for reset of a PID loop, the 27 28 controlled variable shall be allowed to be shared over the network unless specified to be directly wired to the DDC controller. 29

30 ORIGINAL EQUIPMENT MANUFACTURER (OEM) CONTROLLER DDC INTEGRATION:

31 Provide DDC programming to define all equipment integral input/output points, setpoints, data points, calculations, etc. that are available through the manufacturers communication interface. Consult with the Agency DDC operations personnel to 32 determine if some of the points should be omitted (for clarity or lack of value). The following equipment shall be integrated 33 34 into the DDC system: 35

- Variable Frequency Drives
 - Heat Recovery Chiller

38 PROJECT SPECIFIC CONTROL SEQUENCES

39 Refer to Airflow and Control Diagram Drawings for Sequences of Operation.

40 41 42

36 37

1 2 3	SECTION 23 21 13 HYDRONIC PIPING						
4 5	PART 1 - GENERAL						
6 7 8 9	SCOPE This section contains specifications for all HVAC hydronic pipe and pipe fittings for this project. Included are the following topics:						
10 11	PART 1 - GENERAL						
12	Scope						
13 14	Related WORK						
15	Reference Standards						
16	Shop Drawings						
17	Quality Assurance						
18	Delivery, Storage, and Handling						
19	Design Criteria						
20	Welder Qualifications						
21							
22	Heating Hot Water						
24	Chilled Water						
25	Geothermal Water (Aboveground)						
26	Makeup Water						
27	Chemical Treatment						
28	Vents and Relief Valves						
29	Looling Coll Condensate						
30 21	Gaskats						
32	Unions and Flanges						
33	Mechanical Grooved Pipe Connections						
34							
35	PART 3 - EXECUTION						
36	Preparation						
3/	Erection Wolded Bine Joints						
20	Threaded Pipe Joints						
40	Mechanical Grooved Pine Connections						
41	Copper Pipe Joints						
42	Water Systems						
43	Makeup Water						
44	Chemical Treatment						
45	Vents and Relief Valves						
46 47	Linions and Flanges						
48	Gaskets						
49	Piping System Leak Tests						
50	Hydronic Piping System Flushing						
51	Construction Verification Items						
52	Piping System Test Report						
53							
54	KELATED WORK Section 22.05.22 Conneral Duty Valves for HVAC Dining						
56	Section 23 05 15 - Piping Specialties						
57	Section 23 05 29 - Hangers and Supports for HVAC Piping and Equipment						
58	Section 23 07 00 - HVAC Insulation						
59	Section 23 25 00 - HVAC Water Treatment.						
60							
61 62	KEFEKENCE Applicable provisions of Division 1 govern work under this section						
62 63	Applicable provisions of Division 1 govern work under this section.						
64							

1 **REFERENCE STANDARDS** 2 ANSI B16.3 Malleable

- 2 Malleable Iron Threaded Fittings 3 **Cast Iron Threaded Fittings** ANSI B16.4 **Pipe Flanges and Flanged Fittings** 4 ANSI B16.5 5 ANSI B16.22 Wrought Copper and Wrought Copper Alloy Solder Joint Pressure Fittings Pipe, Steel, Black and Hot-Dipped, Zinc Coated Welded and Seamless 6 ASTM A53 7 ASTM A74 Cast Iron Soil Pipe and Fittings Forgings, Carbon Steel, for Piping Components 8 ASTM A105
- 9 ASTM A126 Gray Cast Iron Castings for Valves, Flanges, and Pipe Fittings
- 10 ASTM A181 Forgings, Carbon Steel for General Purpose Piping
- 11 ASTM A197 Cupola Malleable Iron
- 12 ASTM A234 Pipe Fittings of Wrought Carbon Steel and Alloy Steel for Moderate and Elevated Temperatures
- 13 ASTM A380 Practice for Cleaning and Descaling Stainless Steel Parts, Equipment, and Systems
- 14 ASTM B75 Seamless Copper Tube
- 15 ASTM B88 Seamless Copper Water Tube 16

SHOP DRAWINGS

17 18

19

20

21

25

30

33

37

40

49 50 51

56

60

61

Refer to division 1, General Conditions, Submittals.

Contractor shall submit schedule indicating the ASTM specification number of the pipe being proposed along with its type and grade and sufficient information to indicate the type and rating of fittings for each service.

2223 TYPE F STEEL PIPE:

24 Statement from manufacturer on his letterhead that the pipe furnished meets the ASTM specification contained in this section.

26 TYPE E OR S STEEL PIPE:

Mill certification papers, also known as material test reports, for the pipe furnished for this project, in English. Heat numbers on these papers to match the heat numbers stencilled on the pipe. Chemical analysis indicated on the mill certification papers to meet or exceed the requirements of the referenced ASTM specification.

31 COPPER TUBE:

32 Statement from manufacturer on his letterhead that the pipe furnished meets the ASTM specification contained in this section.

34 QUALITY ASSURANCE

Order all Type E and Type S steel pipe with heat numbers rolled, stamped, or stenciled to each length or each bundle, depending on the size of the pipe, and in accordance with the appropriate ASTM specification.

Any installed material not meeting the specification requirements must be replaced with material that meets these specifications without additional cost to the Owner.

41 DELIVERY, STORAGE, AND HANDLING

42 Promptly inspect shipments to insure that the material is undamaged and complies with specifications.43

Cover pipe to eliminate rust and corrosion while allowing sufficient ventilation to avoid condensation. Do not store materials directly on grade. Protect pipe, tube, and fitting ends so they are not damaged. Where end caps are provided or specified, take precautions so the caps remain in place. Protect fittings, flanges, and unions by storage inside or by durable, waterproof, above ground packaging.

Offsite storage agreements will not relieve the contractor from using proper storage techniques.

Storage and protection methods must allow inspection to verify products.

52 53 **DESIGN CRITERIA**

54 Use only new material, free of defects, rust and scale, and meeting the latest revision of ASTM specifications as listed in this 55 specification.

Construct all piping for the highest pressures and temperatures in the respective system in accordance with ANSI B31, but not
 less than 125 psig unless specifically indicated otherwise.

Where weld fittings are used, use only long radius elbows having a centerline radius of 1.5 pipe diameters.

Where ASTM A53 type F pipe is specified, ASTM A53 grade A type E or S, or ASTM A53 grade B type E or S may be substituted at Contractor's option. Where ASTM A53 grade A pipe is specified, ASTM A53 grade B pipe may be substituted at Contractor's option. Where the grade or type is not specified, Contractor may choose from those commercially available. 1 Where ASTM B88, type L hard temper copper tubing is specified, ASTM B88, type K hard temper copper tubing may be 2 substituted at Contractor's option.

4 WELDER QUALIFICATIONS

5 Before any metallic welding is performed, the Contractor shall submit his Standard Welding Procedure Specifications, 6 Procedure Qualification Records and Qualification Test Records for each Welder along with associated continuity records to 7 demonstrate compliance with ASME Section IX, paragraph QW-322.

9 The Contractor shall maintain a complete set of welder qualification documents at the jobsite, including Test Records and 10 Continunity Records for each welder.

11

16 17

18 19

8

3

The A/E or owner reserves the right to test the work of any welder employed on the project, at the Contractor's expense.
Testing will include a visual examination of the pipe and weld and may include radiography of any suspect welds. If the work of the welder is found to be unsatisfactory, the welder shall be prevented from doing further welding on the project. Any welds deemed unacceptable will be repaired at the contractor's expense.

PART 2 - PRODUCTS

20 HEATING HOT WATER

21 2" and Smaller: ASTM A53, type F, standard weight (schedule 40) black steel pipe with ASTM A126/ANSI B16.4, class 125, 22 standard weight cast iron threaded fittings.

23

2-1/2" and Larger: ASTM A53, standard weight (schedule 40) black steel pipe with ASTM A234 grade WPB/ANSI B16.9 standard
 weight, seamless, carbon steel weld fittings.

26

Contractor may use ASTM B88 seamless, type L, hard temper copper tube with ANSI B16.22 wrought copper solder-joint fittings
 in lieu of steel pipe for all sizes. Mechanically formed tee fittings may be used in lieu of wrought copper solder-joint tee fittings
 for branch takeoff up to one-half (1/2) the diameter of the main.

30 31 CHILLED WATER

2" and Smaller: ASTM A53, type F, standard weight (schedule 40) black steel pipe with ASTM A126/ANSI B16.4, class 125,
 standard weight cast iron threaded fittings.

34

2-1/2" and Larger: ASTM A53, standard weight (schedule 40) black steel pipe with ASTM A234 grade WPB/ANSI B16.9 standard
 weight, seamless, carbon steel weld fittings.

37

Contractor may use ASTM B88 seamless, type L, hard temper copper tube with ANSI B16.22 wrought copper solder-joint fittings
 in lieu of steel pipe for all sizes. Mechanically formed tee fittings may be used in lieu of wrought copper solder-joint tee fittings
 for branch takeoff up to one-half (1/2) the diameter of the main.

41

42 GEOTHERMAL WATER (ABOVE GROUND)

43 2" and Smaller: ASTM A53, type F, standard weight (schedule 40) black steel pipe with ASTM A126/ANSI B16.4, class 125,
 44 standard weight cast iron threaded fittings.

45

2-1/2" and Larger: ASTM A53, standard weight (schedule 40) black steel pipe with ASTM A234 grade WPB/ANSI B16.9 standard
 weight, seamless, carbon steel weld fittings.

48

Contractor may use ASTM B88 seamless, type L, hard temper copper tube with ANSI B16.22 wrought copper solder-joint fittings
 in lieu of steel pipe for all sizes. Mechanically formed tee fittings may be used in lieu of wrought copper solder-joint tee fittings
 for branch takeoff up to one-half (1/2) the diameter of the main.

52 53 **MAKEUP WATER**

54 Extend from where left by the Plumbing Contractor with the same materials.

55 56 CHEMICAL TREATMENT

57 Use pipe and pipe fittings as specified for the system to which the chemical treatment piping is connected. Plastic pipe 58 furnished with the chemical treatment materials may also be used if its pressure/temperature rating is acceptable for the 59 service.

60 s

61 VENTS AND RELIEF VALVES

- 62 Use pipe and pipe fittings as specified for the system to which the relief valve or vent is connected.
- 63

3

4 5

6 7

8

9

10

11 12

13

14 15

16 17

18

19

20 21

22

23 24

25 26

28 29

30

31

33 34 35

36 37 38

39

40

46

51

55

56 57

58 59

60 61

COOLING COIL CONDENSATE

ASTM B88, type L hard temper copper tubing with ASTM B145/ANSI B16.23 cast red bronze or ASTM B75/ANSI B16.29 wrought solder-type drainage fittings.

UNIONS AND FLANGES

2" and Smaller: ASTM A197/ANSI B16.3 malleable iron unions with brass seats. Use black malleable iron on black steel piping and galvanized malleable iron on galvanized steel piping. Use ANSI B16.18 cast copper alloy unions on copper piping. Use unions of a pressure class equal to or higher than that specified for the fittings of the respective piping service but not less than 250 psi.

2-1/2" and Larger: ASTM A181 or A105, grade 1 hot forged steel flanges of threaded, welding and of a pressure class compatible with that specified for valves, piping specialties and fittings of the respective piping service. Flanges smaller than 2-1/2" may be used as needed for connecting to equipment and piping specialties. Use raised face flanges ANSI B16.5 for mating with other raised face flanges on equipment with flat ring or full face gaskets. Use ANSI B16.1 flat face flanges with full face gaskets for mating with other flat face flanges on equipment.

GASKETS

Water and Glycol Systems: Branded, compressed, non-asbestos sheet gaskets. Klingersil C4401, Garlock 3000, JM Clipper 978 or approved equal.

MECHANICAL GROOVED PIPE CONNECTIONS

The use of mechanical grooved pipe connections will not be accepted.

PART 3 - EXECUTION

27 ERECTION

Carefully inspect all pipe, fittings, valves, equipment and accessories before installation. Any items that are unsuitable, cracked or otherwise defective shall be rejected and removed from the job site immediately. Excluding minor surface rust, piping that exhibits significant oxidation or corrosion will be rejected.

32 Exercise care at every stage of storage, handling, laying and erecting to prevent entry of foreign matter into piping, fittings, 33 valves, equipment and accessories. Do not erect or install any item that is not clean.

Remove all lose dirt, scale, oil, chips, burrs and other foreign material from the internal and external surfaces of all pipe and piping components prior to assembly, including debris associated with cutting, threading and welding.

During fabrication and assembly, remove slag and weld spatter from internal pipe surfaces at all joints by peening, chipping and wire brushing.

During construction, until system is fully operational, keep all openings in piping and equipment closed except when actual
 work is being performed on that item of the system. Use plugs, caps, blind flanges or other items designed for this purpose.

Furnish and install all flanges, caps, bypasses, drains, valves, etc. required to facilitate flushing and draining all heating and cooling system piping.

Install all piping parallel to building walls and ceilings and at heights which do not obstruct any portion of a window, doorway, stairway, or passageway. Where interferences develop in the field, offset or reroute piping as required to clear such interferences. In all cases, consult drawings for exact location of pipe spaces, ceiling heights, door and window openings, or other architectural details before installing piping.

Provide anchors, expansion joints, swing joints and/or expansion loops so that piping may expand and contract without damage
 to itself, equipment, or building.

Mitered ells, notched tees, and orange peel reducers are not acceptable. On threaded piping, bushings are not acceptable.

"Weldolets" and "Threadolets" will not be accepted.

Install drains throughout the systems to permit complete drainage.

Do not route piping through transformer vaults or above transformers, panelboards, or switchboards, including the required service space for this equipment, unless the piping is serving this equipment.

1 Install all valves, control valves, and piping specialties, including items furnished by others, as specified and/or detailed. Make 2 connections to all equipment installed by others where that equipment requires the piping services indicated in this section.

WELDED PIPE JOINTS

- Make all welded joints by fusion welding in accordance with ASME Codes, ANSI B31, and State Codes where applicable.
- All pipe welding shall be completed by Qualified Welders in accordance with the Contractor's Procedure Specifications.

11

12

13

14 15

3

4 5

6

- Contractor will ensure that these steps are followed where pipe sections will be joined by welding: 1. Cleaning – Welding surfaces will be clean and free of defects.
 - Alignment Inside diameter of piping components will be aligned as accurately as possible. Internal misalignment shall not exceed 1/16".
 - 3. Spacing Pipe sections will be spaced to allow deposition of weld filler material through the entire weld joint thickness.
 - 4. Girth Butt Welds:
 - a. Girth butt welds shall be complete penetration welds.
- 16 17 18

19

20

21

28

31

- b. Concavity will not exceed 1/32"c. Under cuts will not exceed 1/32"
- d. As welded surfaces are permitted however surfaces will be free from coarse ripples, grooves, abrupt ridges and valleys.

Electrodes shall be Lincoln, or approved equal, with coating and diameter as recommended by the manufacturer for the type and thickness of work being done.

25 THREADED PIPE JOINTS

26 Use a Teflon based thread lubricant or Teflon tape when making joints; no hard setting pipe thread cement or caulking will be 27 allowed.

29 MECHANICAL GROOVED PIPE CONNECTIONS

30 The use of mechanical grooved pipe connections will not be accepted.

32 COPPER PIPE JOINTS

Remove all slivers and burrs remaining from the cutting operation by reaming and filing both pipe surfaces. Clean fitting and tube with emery cloth or sandpaper. Remove residue from the cleaning operation, apply flux, and assemble joint. Use 95-5 solder or brazing to secure joint as specified for the specific piping service.

36

Where mechanically formed tee fittings are allowed, form mechanically extracted collars in a continuous operation, consisting of drilling a pilot hole and drawing out the tube surface to form a collar having a height of not less than three times the thickness of the tube wall. Use an adjustable collaring device. Notch and dimple the branch tube. Braze the joint, applying heat properly so that pipe and tee do not distort; remove distorted connections.

4142 WATER SYSTEM

Run water mains level or pitch horizontal mains up 1 inch in 40 feet in the direction of flow. Install manual air vents at all high points where air may collect. If vent is not in an accessible location, extend air vent piping to the nearest code acceptable drain location with vent valve located at the drain.

46

47 Main branches and runouts to terminal equipment may be made at the top, top 45 degree, side, and/or bottom 45 degree of 48 the main provided that there are drain valves suitably located for complete system drainage and manual air vents are located at 49 all top and top 45 degree connections. Bottom connections are not acceptable unless approved by the AE or owner. 50

51 Use top or top 45 degree connection to main for upfeed risers and bottom 45 degree connection to main for downfeed risers. 52 Bottom connections are not acceptable unless approved by the AE or owner.

53

54 Use a minimum of two elbows in each pipe line to a piece of terminal equipment to provide flexibility for expansion and 55 contraction of the piping systems. Offset pipe connections at equipment to allow for service, such as removal of the terminal 56 device.

57

58 Use eccentric fittings for changes in horizontal pipe sizes with the fittings installed for proper air venting. Concentric fittings 59 may be used for changes in vertical pipe sizes.

60 61 **MAKEUP WATER**

62 Install where indicated and/or specified, including all valves, piping specialties and dielectric unions required for a functional 63 system.

3

4 5

6

7

8

13

20

21

22 23

25 26 27

28

29

30

36

38

45

56 57

58 59

60

CHEMICAL TREATMENT

Install chemical treatment piping as indicated on the drawings, as detailed, and as recommended by the supplier of the chemical treatment equipment.

VENTS AND RELIEF VALVES

Install vent and relief valve discharge lines as indicated on the drawings, as detailed, and as specified for each specific valve or piping specialty item. In no event is a termination to occur less than six feet above a roof line.

9 COOLING COIL CONDENSATE

Trap each cooling coil drain pan connection with a trap seal of sufficient depth to prevent conditioned air from moving through the piping. Extend drain piping to nearest code approved drain location. Construct trap with plugged tee for cleanout purposes as detailed.

14 UNIONS AND FLANGES

15 Install a union or flange, as required, at each automatic control valve and at each piping specialty or piece of equipment which 16 may require removal for maintenance, repair, or replacement.

Where a valve is located at a piece of equipment, locate the flange or union connection on the equipment side of the valve.
Concealed unions or flanges are not acceptable.

GASKETS

Store horizontally in cool, dry location and protect from sunlight, water and chemicals. Inspect flange surfaces for warping, radial scoring or heavy tool marks. Inspect fasteners, nuts and washers for burrs or cracks. Replace defective materials.

Align flanges parallel and perpendicular with bolt holes centered without using excessive force. Center gasket in opening.
 Lubricate fastener threads, nuts and washers with lubricant formulated for application.

Draw flanges together evenly to avoid pinching gasket. Tighten fasteners in cross pattern sequence (12 - 6 o'clock, 3 - 9 o'clock, etc.), one pass by hand and four passes by torque wrench at 30% full torque, 60% full torque and two passes at full torque per ASME B16.5.

31 PIPING SYSTEM LEAK TESTS

Verify that the piping system being tested is fully connected to all components and that all equipment is properly installed, wired, and ready for operation. If required for the additional pressure load under test, provide temporary restraints at expansion joints or isolate them during the test. Verify that hangers can withstand any additional weight load that may be imposed by the test.

37 Provide all piping, fittings, blind flanges, and equipment to perform the testing.

Conduct pressure test with test medium of air or water unless specifically indicated. Minimum test time is indicated in the table below; additional time may be necessary to conduct an examination for leakage. Each test must be witnessed by the Division's representative. If leaks are found, repair the area with new materials and repeat the test; caulking will not be acceptable.

44 Do not insulate pipe until it has been successfully tested.

For air tests, gradually increase the pressure to not more than one half of the test pressure; then increase the pressure in steps
 of approximately one-tenth of the test pressure until the required test pressure is reached. Examine all joints and connections
 with a soap bubble solution or equivalent method. The piping system exclusive of possible localized instances at pump or valve
 packing shall show no evidence of leaking. After testing is complete, slowly release the pressure in a safe manner.

51	System	Pressure	Medium	Duration
52	Heating hot water	100 psig	Water	8 hr
53	Chilled water	100 psig	Water	8 hr
54	Geothermal water (Above grade)	100 psig	Water	8 hr
55				

All pressure tests are to be documented.

Refer to Section 23 21 26 for below grade geothermal piping testing requirements.

1 HYDRONIC PIPING SYSTEM FLUSHING

All new chilled water and heating hot water system piping shall be flushed thoroughly before the systems are put in to operation. Subsequent to executing the chemical cleaning processes specified in Section 23 25 00 – HVAC WATER TREATMENT, and prior to adding scale and corrosion inhibitors, flush all piping and components with a clean source of water until the discharge from the system is clean. Discharge shall be from drains provided at all low points in the piping, ends of headers and as otherwise necessary to flush and drain the entire system.

A clean water source shall be tapped into the system downstream of the main circulation pump(s). Provide minimum 2"
 connection between water source and hot water/chilled water systems including taps with ball valves (or line size tap and ball
 valve for piping systems smaller than 2").

11

7

Provide minimum 2" taps (or line size if mains are smaller than 2") at the ends of headers, the low pint of each of the mains on each floor and as otherwise necessary to flush and drain the entire system. Contractor shall identify proposed clean water source along with the method/location of drain discharge and review with the A/E and owner prior to installing flushing connections to water source and drain outlets. Provide code required temporary backflow prevention for the clean water source if needed. Provide all temporary taps, valves, piping, bypasses and hoses as needed to accomplish flushing procedures.

17

After flushing operations are complete, drain and/or blow out any residual water, clean and replace all strainers, and add scale and corrosion inhibitors as specified in Section 23 25 00. Leave flushing connections/valves in place and cap.

20

All flushing procedures shall be documented by completing and submitting the report form included at the end of this Section.

22 23 INITIAL FILL AND VENT

Fill hydronic systems with appropriate working fluids as specified. All system fluids shall be chemically treated as specified in Section 23 25 00 – HVAC WATER TREATMENT.

26

For closed piping systems, all air trapped at high points shall be relieved through the manual air vents

29 CONSTRUCTION VERIFICATION ITEMS

30 Contractor is responsible for utilizing the construction verification checklists supplied by the third party commissioning agent in 31 accordance with the procedures defined for construction verification in Section 01 91 00.

32

33 34

Page Intentionally Left Blank

PIPING SYSTEM LEAKAGE TEST REPORT

Date Submitted:			
Project Name:			
Contractor:			
System:			
Test Medium:	🗆 Air	Water Other	
Specified Test Duration	Hours	Specified Test Pressure	PSIG
System Identification:			
Describe Location / Section	of Piping:		
			<u> </u>
	Test Date:		
Start Test Time:		Initial Pressure:	PSIG
Stop Test Time:		Final Pressure:	PSIG
Tested By:		Witnessed By:	
Title:		Title:	
Signed:		Signed:	
Date:		Date:	
Comments:			

PIPING SYSTEM FLUSHING REPORT (Page 1)

Date Submitted:			
Project Name:			
Contractor:			
System Identification (check on	e):		
Chilled Water			
Heating Hot Water			
Geothermal Water			
Describe procedure:			
Flush Date:	Start Time: _	Stop Time:	
Pressure of Water Source:	PSIG	Describe water source and method of conn	ection to source :
Flushed By:		Witnessed By:	
Title:		Title:	
Company:		Agency:	
Signed:		Signed:	
Date:		Date:	
Describe results:			

1 2	SECTION 23 21 23 HYDRONIC PUMPS
3	
4	DADT 1 CENEDAL
6	PART I - GENERAL
7	SCOPE
8	This section includes specifications for water pumps used for HVAC applications. Included are the following topics:
9	
10	PART 1 - GENERAL Scope
12	Related Work
13	Reference
14	Quality Assurance
15	Shop Drawings
16 17	Operation and Maintenance Data
17 18	Design Criteria
19	PART 2 - PRODUCTS
20	Base Mounted Centrifugal Pumps
21	In-Line Centrifugal Pumps
22	In Line Circulation Pumps
23	
25	Installation
26	Construction Verification Items
27	Functional Performance Testing
28	
29	KELATED WORK Section 22.05.12 Common Motor Paguiroments for HVAC Equipment
31	Section 25 05 15 - Common Motor Requirements for HVAC Equipment
32	REFERENCE
33	Applicable provisions of Division 1 shall govern work under this section.
34	
35	QUALITY ASSURANCE Defer to division 1. Constal Conditions, Equals and Substitutions
37	Refer to division 1, General conditions, Equals and Substitutions.
38	SHOP DRAWINGS
39	Also refer to division 1.
40	
41	Include data concerning dimensions, capacities, materials of construction, ratings, weights, pump curves with net positive suction
42 43	identification
44	
45	Pump curves shall identify design point of operation.
46	
47	OPERATION AND MAINTENANCE DATA
48 49	All operations and maintenance data shall comply with the submission and content requirements specified under section GENERAL REQUIREMENTS
50	GENERAE REQUIREMENTS.
51	DESIGN CRITERIA
52	Pump sizes, capacities, pressures and operating characteristics shall be as scheduled.
53	
54 55	Provide all pumps with motors, impellers, drive assemblies, bearings, coupling guard, and other accessories specified. Statically and dynamically balance all rotating parts. Provide flanged connections on all numps unless specified otherwise. Service or repair
56	of base mounted pumps shall not require breaking piping connections or removal of motor.
57	
58	Furnish each pump and motor with a nameplate giving the manufacturer's name, serial number of pump, capacity in GPM and
59	head in feet at design condition, horsepower, voltage, frequency, speed and full load current.
60 61	Test all numps clean and naint before shinment. The manufacturer shall certify all nump ratings
62	rest an pumps, clean and paint before simplifient. The manufacturer sildir tertiny dir pump ratings.
63	All pumps to operate without excessive noise or vibration.
64	

1 2	PART 2 - PRODUCTS		
3			
4 5 6 7	BASE MOUNTED CENTRIFUGAL PUMPS MANUFACTURERS: Bell and Gossett Series 1510, Taco Series FI		
8 9 10 11	Description: Type: Horizontal shaft, single stage, single or double suction, split casing, 175 PSIG working pressure at operating temperature of 225 degrees F continuous, 250 degrees F intermittent. Pump shall be back pull-out design to allow for servicing without disturbing piping, motor or requiring shaft realignment.		
13 14 15	Casing: Cast iron with suction and discharge gage ports, renewable bronze wear rings, vent and drain plugs, flanged suction and discharge connections.		
15 16 17	Impeller: Bronze, hydraulically and dynamically balanced, keyed and locked to pump shaft, and protected by replaceable bronze shaft sleeve.		
19 20	Bearings: Oil or grease lubricated ball or roller bearings.		
21 22	Shaft: Alloy steel with copper, bronze, or stainless steel shaft sleeve.		
23 24	Seal: Carbon rotating against stationary ceramic seat, 225 degrees F maximum continuous operating temperature.		
25 26 27	Drive: Flexible spacer type coupling or coupling with extended hub to allow for pump service. Provide and OSHA approved guard for shaft/coupling assembly.		
28 29	Baseplate: Cast iron or fabricated steel with provisions for grouting.		
30 31 32	IN-LINE CENTRIFUGAL PUMPS MANUFACTURERS: Bell and Gossett Series 80, Taco Series KV		
33	Description		
34 35 36	Factory assembled and tested, centrifugal, overhung impeller, close-coupled, inline pump designed for installation with pump and motor shafts mounted horizontally or vertically.		
38 39 40	Casing: Radially split cast iron; flanged suction and discharge connection; with plugged taps for vent, drain, suction and discharge gages, and replaceable bronze wear rings.		
40 41 42	Impeller: Cast bronze, keyed to shaft, secured with a stainless steel locking cap screw, hydraulically and dynamically balanced.		
43 44	Pump shaft sleeve shall be bronze or stainless steel with stainless steel stub shaft.		
45 46	Bearings: 2 oil lubricated bronze sleeves or ball bearings capable of being greased.		
47 48 49	Seal: Mechanical type, carbon rotating against a stationary ceramic seat, 225 degrees F maximum continuous operating temperature.		
50 51	IN-LINE CIRCULATION PUMPS		
52 53	Bell and Gossett, Armstrong, Thrush, Taco, Grundfos, Aurora, or approved equal.		
54 55	Permanently lubricated circulation pump.		
56 57	Pump shall be rated to operate at maximum 225°F and 150psi pressure.		
58 59 60	Pump shall have iron body, stainless steel shaft, glass filled PPS impellor and permanently lubricated and sealed ball bearings. Mechanical seal faces shall be carbon on silicon carbide.		
61 62 63	Pump motor shall be a drip proof split capacitor with thermal overload protection.		

PART 3 - EXECUTION

3 INSTALLATION

1

2

12

14

Install all pumps in strict accordance with manufacturer's instructions. Access/service space around pumps shall not be less than
 minimum space recommended by pump manufacturer.

- For base mounted pumps located on 2nd floor and within new mechanical room, set pumps on concrete pads and concrete inertia
 bases as specified in Section 23 05 48. Fill entire base with non-shrinking grout when required by manufacturer's installation
 instructions. Level and bolt down pump prior to grouting of pump base. Base mounted pumps not mounted on inertia bases shall
 be mounted on concrete pads. Fill entire base with non-shrinking grout when required by manufacturer's installation instructions.
 Level and bolt down pump prior to grouting of pump base.
- 13 Support piping adjacent to pump such that no weight is carried on pump casings.
- 15 All valves and piping specialties must be full line size as indicated on the drawings.
- 1617 Install per plan details.

18 19 CONSTRUCTION VERIFICATION ITEMS

20 Contractor is responsible for utilizing the construction verification checklists supplied by the third party commissioning agency 21 in accordance with the procedures defined for construction verification in Section 01 91 00.

23 FUNCTIONAL PERFORMANCE TESTING

24 Contractor is responsible for utilizing the functional performance test forms supplied by the third party commissioning agent in 25 accordance with the procedures defined for functional performance testing in Section 01 91 00.

- 26
- 27

22

28

Page Intentionally Left Blank

1 2 3		SECTION 23 21 26 GROUND SOURCE CLOSED CIRCUIT VERTICAL EXCHANGER FIELD		
4				
5		PART 1 – GENERAL		
6				
/ g	SCOPE This design	has been prepared in accordance with the materials standards and accepted installation practices of the (IGSHDA)		
9 10	The loopfie	eld contractor shall comply with these standards and practices as well as all State and local regulations pertaining to tion.		
11				
12 13 14	The loopfic materials, o	eld contractor is responsible for all aspects involved with the complete geothermal loopfield installation. All drilling, excavation, hauling of backfill, pumping, soil compaction and labor required shall be included in the bid price.		
15	The loonfie	eld contractor shall take note: there is no guarantee to the loonfield contractor that the location of any existing		
16 17 18	utilities are contractor may be dar	e exactly as indicated on the plans. Some areas may require hand digging to locate that utility. The loopfield must include in the bid price, the repair of any domestic water, electrical, communication or any service line that maged during the construction of this project. Any offsets required to route over or under existing lines shall also be		
20	included in	the bid price of the project.		
20	PART 1 - GE	ENERAL		
22	S	cope		
23	R	telated Work		
24	C	Quality Assurance		
25	D	Design Criteria		
26 27	S	hop Drawings		
28	PART 2 -	PRODUCTS		
29	Р	lipe		
30	F	ittings		
31	U	J-Bend Pipe Separators		
32	В	Sentonite Grout		
33	L	ocation Tape		
34	F	ield Location		
35				
36	PART 3 - EX	KECUTION		
37	D)rilling		
38	U	J-Bend Pipe Assembly		
39	G	routing Procedure		
40	H	leat Fusion Pipe Joining		
41	E			
4Z 42	P	ape installation		
43	T T			
44		esting and cleaning		
46	C			
47 49	Division 23	work as specified shall be provided by HVAC Contractor unless otherwise specified on Bid Form.		
40	RELATED W	NORK		
50 51	Applicable	provisions of Division 01 govern work under this section.		
52 53	Section 00	31 32 Geotechnical Data & Geothermal Testing Boring Information		
54	QUALITY A	SSURANCE		
55 56	Substitutio	n of Materials: Refer to Division 01 and the General Conditions of the Contract.		
57	The loopfie	eld contractor must have on this project a certified International Ground Source Heat Pump Association (IGSHPA)		

installer. The loopfield contractor performing this work must have a minimum of two years experience in performing
 underground closed circuit vertical heat exchanger work of this project's size or larger.

Vertical heat exchanger fabricators must be heat fusion certified by an authorized high density polyethylene (HDPE) pipe manufacturer's representative of the brand of pipe used.

Certification must include: successful completion of a written heat fusion exam as well as demonstrating proper heat fusion techniques under the direct supervision of the authorized HDPE pipe manufacturer's representative.

Loopfield contractor shall review his own work for compliance with Construction Documents. Prior to punch list activity by A/E, contractor shall provide documentation to A/E that a review has taken place and shall issue a letter indicating that Work has been performed in compliance with Construction Documents.

In the event Contractor does not satisfactorily review his own work and results in additional site visits by A/E, Contractor shall reimburse A/E for additional time required to close out Project.

DESIGN CRITERIA

Below is a summary of the geothermal loopfield design:

- 500 foot vertical heat exchanger depth with minimum 6 foot trench depth
- 54 total vertical heat exchangers split into (3) 18 vertical heat exchanger fields
 - (3) 4" supply / return circuits to the building
- 20 foot nominal spacing between vertical bores
- 25% propylene glycol percentage in water
 - 40F and 85F shall be the minimum and maximum design supply water termperatures
- Geological Conditions:
 - Soil Formation Thermal Conductivity: 2.15 Btu/hr-ft-°F
 - Soil Formation Thermal Diffusivity: 1.45 ft² / day
 - Soil Undisturbed Formation Temperature: 51.2 52.2°F
 - Borehole Thermal Resistance: 0.248 hr-ft-°F/Btu
- Loopfield designed for the building block heating load at 20% outside airflow on all air handling units at 10F.
- The intent of the heat recovery chiller and borefield is to provide cooling for the entire building and the building heating load down to 25F.

SHOP DRAWINGS

Refer to Division 01 and General Conditions of the Contract.

PART 2 - PRODUCTS

PIPE

The pipe shall be PE3408 HDPE with a minimum cell classification of 45434C per ASTM D3035-93 and a DR11 rating for u-bends.

Header pipe 2 inches or smaller and a minimum of DR11 for header pipe greater than 2 inch in diameter. This pipe will carry a warranty of no less than 25 years.

Each pipe shall be permanently indent marked with the manufacturer's name, nominal size, pressure rating, relevant ASTM standards, cell classification number and date of manufacture.

All piping used will have factory hot-stamped lengths impressed on the side of the piping indicating the length at that point. The length stamp shall read zero on one end and the actual total length on the other end.

Only factory fused u-bend with pipe lengths long enough to reach grade from the bottom of the bore will be allowed. No field fusions are required below header level.

53 FITTINGS

61

54 Pipe fittings shall meet the requirements of ASTM D2683 (for socket fusion fittings) or ASTM D3261 (for butt/saddle fusion 55 fittings). Each fitting shall be identified with the manufacturer's name, nominal size, pressure rating, relevant ASTM standards 56 and date of manufacturer. 57

58 U-BEND PIPE SEPARATORS

The u-bend pipe separators used to position the u-bend pipes against the borehole wall directly across from one another. These separators will be positioned every ten feet on the u-bend pipe.

62 BENTONITE GROUT

The thermally enhanced bentonite grout used to seal the vertical heat exchanger shall have a minimum thermal conductivity of 1.0 Btu/hr-ft°F and a minimum of 32% solids. This grout will also have a permeability rate of less than 1X10-7cm/sec.

1 LOCATING TAPE

2 Locating tape must be foil backed, two inches wide or greater, with a continuous message printed every 36 inches or less 3 reading: "CAUTION GEOTHERMAL PIPELINE BURIED BELOW". The tape shall be highly resistant to alkalis, acids, and other 4 destructive agents found in the ground.

5 6 FIELD LOCATION

The four outside grid bores of each circuit should be surveyed after drilling is complete, but before horizontal trenching is done and provided as a record drawing of the installation.

PART 3 - EXECUTION

- 8 9
- 10

7

11 12

13 DRILLING

14 The vertical boreholes will be drilled to a depth that allows complete insertion of the u-tube assembly to its specified depth. 15 The maximum borehole diameter will be six inches. If a larger diameter is required, it must be approved by the design 16 engineer.

17

18 The wells will be cased during drilling and the casing removed during grouting operations.

1920 U-BEND PIPE ASSEMBLY

The u-bend pipe shall be filled with water and pressured to 100 psi to check for leaks before insertion. If necessary, an iron (sinker) bar can be attached at the base of each u-bend to overcome bouncy. This iron bar will have all sharp edges adequately taped to avoid scarring and/or cutting of the polyethylene pipe. No driving rod that is pulled out after u-bend insertion will be allowed. U-bend pipe separators (GeoClips) will be attached at ten feet intervals to the u-bend along with the grouting tremie as per manufacturer's recommendations. The entire u-bend pipe assembly is inserted to the specified depth in the borehole.

27 GROUTING PROCEDURE

Each bore shall be grouted from the bottom up, in a continuous fashion, using a one inch HDPE tremie pipe. The tremie pipe will be pulled out during the grouting procedure maintaining the pipe's end just below grout level within the borehole. Bore casing shall be removed during the grouting process. All State regulations will be met for borehole grouting.

32 HEAT FUSION PIPE JOINING

All underground pipe joining will be heat fused by socket, butt or saddle (sidewall) fusion in accordance to ASTM D2610, ASTM
 D2683 and the manufacturer's heat fusion specifications. The operator shall be heat fusion certified and experienced in
 executing quality fusion joints.

37 EXCAVATION AND BACKFILL

Perform all excavation and backfill work to accomplish indicated mechanical systems installation in accordance with Division 31
 Earthwork. Blasting will not be allowed without written permission of the Architect/Engineer and the user agency.

40

31

Install lines passing under foundations with minimum of 1-1/2 inch clearance to concrete and insure no disturbance of bearing
 soil.

43

44 Before burying piping, mark up Record Drawings and dimensionally locate piping. Deliver information to A/E Field 45 Representative.

46 Unless otherwise specifically indicated on Drawings, trenches for utilities shall be of depth that provides the following minimum
 47 depths of cover from existing grade or from indicated finish grade, whichever is lower:

48 49

50

• Geothermal (direct buried): 6'-0" foot minimum cover

51 Existing utility lines to be retained shown on Drawings or locations of which are made known to Contractor prior to excavation, 52 as well as utility lines uncovered during excavation operations, shall be protected from damage during excavation and 53 backfilling and if damaged, shall be repaired by Contractor at his expense. 54

55 General:

The loopfield contractor shall do all excavating, backfilling, shoring, bailing and pumping for the installation of his work and perform necessary grading to prevent surface water from flowing into trenches or other excavations. Sewer lines shall not be used for draining trenches.

59

All pipe and conduit ends shall be kept sealed and lines left clean and unobstructed during construction. Only material suitable for backfilling shall be piled a sufficient distance from banks of trenches to avoid overloading. Unsuitable backfill material shall

- 62 be removed and replaced.
- 63

3

4 5

6

7 8

9

10

11 12

13 14

15

16 17

18

19 20

21

28

32

38

46

Excavation shall comply with OSHA regulations. Review soil boring information for soil information. During excavation, material for backfilling shall be piled in an orderly manner a safe distance from banks of trench to avoid overloading and to prevent slides or cave ins. Excavated materials not required or acceptable for backfill shall be removed and wasted as requested by A/E.

Grading shall be done as necessary to prevent surface water from flowing into trenches or excavations. Water accumulating shall be removed by pumping or by other acceptable methods.

Sheeting or shoring shall be done as necessary for protection of Work and for safety of personnel. Comply with OSHA regulations.

Comply with Department of Labor (OSHA) 29 CFR, Part 1926 Occupational Safety and Health Standards. Specific mention of this Section however shall in no way imply, suggest or infer that compliance with other sections or regulations is not required.

Excavation shall be by open cut except short sections of a trench may be tunneled if, in the opinion of Soils Engineer, pipe or duct can be safely and properly installed and backfill can be properly tamped in tunnel sections per OSHA regulations.

Each excavation shall comprise required materials and shall include clay, silt, sand, rock, muck, gravel, hardpan, loose shale and loose stone.

Open trench ahead of pipe laying to reveal obstructions.

Provide trench crossings to accommodate public travel.

24 Contractor shall file written "Notification Of Excavation" with utility companies and Digger's Hotline at least 3 days prior to 25 excavating. 26

27 The loopfield contractor shall install geothermal locating tape 18 inches above all horizontal/header piping.

Prior to drilling or trenching, the loopfield contractor shall be responsible for reviewing with the general contractor the location of underground utilities. Existing utility lines uncovered during excavation shall be protected from damage during excavation and backfilling.

33 Trench Excavating:

Trenches shall be of necessary width or proper laying of pipe but not more than 16 inches wider than pipe diameter at base, and banks shall be provided per OSHA regulations. In particular, comply with the Department of Labor (OSHA) 29 CFR, Part 1926 Occupational Safety and Health Standards. Specific mention of this Section however, shall in no way imply, suggest or infer that compliance with other sections or regulations is not required.

Bottom of the trenches shall be accurately graded to provide uniform bearing and support for each section of pipe on undisturbed soil at points along its entire length. Except where rock is encountered, care shall be taken not to excavate below the depths indicated.

43 Where rock excavations are required, rock shall be excavated to a minimum overdepth of 4 inches below trench depths 44 indicated on Drawings or specified. Overdepths in rock excavations and unauthorized overdepths shall be backfilled with loose 45 granular material properly compacted.

Whenever wet or otherwise unstable soil incapable of properly supporting pipe, as determined by Soils Engineer, is encountered in bottom of trench, soil shall be removed to depth required and trench backfilled to proper grade with coarse sand, fine gravel or other material as specified.

51 Keep trenches free from water while construction is in progress. Pipe or appurtenances shall not be laid in water. Pump or bail 52 water from bell holes to permit proper jointing of pipes. Pipe discharge from trench dewatering to drains or natural drainage 53 channels. 54

55 Grading Trench Bottom:

Shape bottom of trench for Class C bedding except as otherwise noted. Perform final grading of trench bottoms by hand tools.
 Grade bottom of trenches evenly to insure bearing for pipes. Cut holes for joints and joint making.

5859 Backfilling of Trenches:

60 Backfill trenches only after piping has been inspected, tested and locations of pipe lines have been recorded. Comply with OSHA

61 regulations for work. In particular, comply with Department of Labor (OSHA) 29 CFR, Part 1926 Occupational Safety and Health

- 62 Standards.
- 63

1 2 2	Specific mention of this Section however, shall in no way imply, suggest or infer compliance with other Sections or regulations is not required.
3 4 5	For depth of 12 inches above top of pipe, backfill by hand with material specified for compacted backfill. Tamp backfill thoroughly in layers not exceeding 6 inches in thickness, taking care not to disturb the pipe.
6	
/ 8	For the remaining trench backfill and compact with material as specified in the following paragraphs.
9 10	Jetting the backfill with water will not be permitted unless approved in writing by the A/E.
11	Normal Backfill:
12 13 14	Where compacted backfill is not specified, trenches shall be carefully backfilled with excavated materials acceptable for backfilling, consisting of earth, loam, sandy clay, sand and gravel, soft shale or other acceptable materials, free from large clods of earth or stones over $2-1/2$ inch maximum dimension, denosited in 12 inch layers and compacted
15	
16 17	Surface shall be graded to reasonable uniformity and mounding over trenches left in uniform and neat condition.
18	Compacted Backfill:
19	Compacted backfill shall be used under slab on grade, slabs within building structure, concrete paving and asphaltic concrete
20	paving. Soils used in fill shall be granular in nature and not contain roots, sod, rubbish or stones over 2-1/2 inch maximum
21	dimension. A/E may reject on site or borrowed materials he considers unsuitable for intended use of fill.
22	
23	Compaction Density For Backfill
24	Fills shall be compacted to dry density equal to at least 95percent of maximum density determined in accordance with Procter
25 26	Test, ASTM D698 66T or modified D1557 66T. Maximum density and optimum moisture content shall be determined by A/E on basis of laboratory test conducted on materials used in fill.
27	Control Toots
28	Control rest.
29	accordance with standard recognized procedures for making required tests
21	accontaince with standard recognized procedures for maning required tests.
32	field and lab testing A minimum of 1 test per 50 lineal feet for each 2 foot lift. Contractor shall nay for density tests and submit
32	reports
34	
35	Adequacy of compaction shall be determined on basis of in-place density determinations to be conducted while fills are being
36 37	placed. Results of tests shall be basis on which satisfactory completion of work is judged. If fills fail to meet specified densities, Contractor shall remove and recompact soils until specified densities are achieved.
38	
39	PIPE INSTALLATION
40	The u-bend pipe ends will be sealed with fusion caps or tape prior to insertion into the borehole. Reasonable care shall be
41	taken to ensure that the geothermal loopfield pipe is not crushed, kinked, or cut. Should any pipe be damaged, the damaged
42	section shall be cut out and the pipe reconnected by heat fusion.
43	Dising must be connected as indicated as the plane. The booder desire consume for belowed flow or well as fluction and
44 4 E	Piping must be connected as indicated on the plans. The neader design accounts for balanced flow as well as flushing and provide the size of the plans are been as the size of the area of the area of the size of
45 46 47	the design engineer.
+/ 12	The minimum hend radius for each nine size shall be 25 times the nominal nine diameter or the nine manufacturer's
49	recommendations, whichever is greater. The depth of all headers and supply and return piper diameter of the pipe manufacturers
50	must be maintained.
51	
52 53	Circuits will be pressure tested before any backfilling of the header trenches is executed. The individual circuits will be pressure tested with water at 120 psi, however, not to exceed DR 11 pipe working pressure at bottom of the u-bend pipe.
54	
55	Coordinate work with other contractors prior to installation. Installed work not coordinated and that interferes with other
57	contractor's work shall be removed or relocated at installing contractor's expense.

5859 TESTING AND CLEANING

60 Cleaning

61 During installation, all debris, and small animals shall be kept out of the pipe. Ends of the HDPE pipe shall be sealed until the 62 pipe is joined to the circuits.

Flushing and Purging 1 2

Each supply and return circuit shall be flushed and purged with a water velocity of two feet per second. The lines shall be left filled with clean water and then pressure tested. If connection to the manifold is not immediate, piping must be capped.

5 The loopfield contractor must coordinate with the mechanical contractor on propylene glycol antifreeze installation. The mechanical contractor is responsible for the propylene glycol antifreeze. See mechanical specifications for antifreeze. 6

7 8 CONSTRUCTION VERIFICATION ITEMS

Contractor is responsible for utilizing the construction verification checklists supplied by the third party commissioning agent in 9 accordance with the procedures defined for construction verification in Section 01 91 00. 10

11 12

3

4

13

1	SECTION 23 25 00
2	HVAC WATER TREATMENT FOR CLOSED LOOP HYDRONIC SYSTEMS
3	
4	
5	P A R T 1 - G E N E R A L
6	SCORE .
/ 0	SLUPE This section includes specifications for chemical treatment of all water steam, and condensate systems. Included are the
9	following tonics:
10	
11	PART 1 - GENERAL
12	Scope
13	Reference
14	Related Work
15	Quality Assurance
16	Shop Drawings
10	Operation and Maintenance Data
10	Design Chiena Maintenance Service
20	
21	PART 2 - PRODUCTS
22	Manufacturers
23	System Cleaner
24	System Inhibitor
25	Closed Water System Treatment
26	Glycol Solution
27	Giycol Fill Equipment
20 29	PART 3 - EXECUTION
30	Preparation
31	Cleaning Sequence
32	Closed Water System Treament
33	Glycol Water Systems
34	Test Equipment
35	Construction Verification Items
36	Pipe Cleaning and Treatment Report
3/	
20	Annlicable provisions of Division 1 shall govern work under this Section
40	Applicable provisions of Division 1 shall govern work under this section.
41	RELATED WORK
42	Section 23 05 15 - Piping Specialties
43	
44	QUALITY ASSURANCE
45	Refer to division 1, General Conditions, Equals and Substitutions.
46	
4/	SHOP DRAWINGS
48 70	Refer to division 1, General Conditions, Submittals.
49 50	Required for all equipment and chemicals specified including data concerning dimensions, canacities, materials of construction
51	weights, operating sequence, composite wiring diagrams and appropriate identification. Chemical data to include the
52	description of the chemical, its composition, its function, and the associated material safety data sheet.
53	
54	OPERATION AND MAINTENANCE DATA
55	Provide for the services of the manufacturer's trained representative to approve the installation and instruct the user agency in
56	the operation of each system.
57	Include data on chemical feed pumps, agitators, and other equipment including spare parts lists, procedures, and
58 50	treatment programs. Include step by step instructions on test procedures including target concentrations.
60	DESIGN CRITERIA
61	Recommend and implement a periodic test procedure and chemical treatment program for each of the following systems:
62	
63	Chilled water
64	Hot water

Glycol water

All (3) water systems will utilize a 25% propylene glycol solution.

Existing hot water terminal equipment connected to the existing boiler plant shall be 100% water.

Provide initial chemical treatment for systems based on complete system fluid analysis prior to equipment installation and operation. Initial chemical treatment supply of chemicals for each system shall be adequate for start-up and testing period, for time systems are being operated by Contractor for temporary heating and cooling, and for 1 year after start-up of system.

Provide electrical devices, motors, wiring and conduit in accordance with applicable Sections of Electrical Specifications.

Electric devices shall be UL listed and bear UL label indicating approval.

MAINTENANCE SERVICE

Service and maintenance of steam, chilled water and hot water will be via the agency and their existing maintenance program.

PART 2-PRODUCTS

MANUFACTURERS

Betz Entac, Dearborn Div. - W. R. Grace & Co., Fremont Industries, Mitco Water Labs, Mogul Corporation, Nalco Chemical Co., Western Water Management, or approved equal.

SYSTEM CLEANER

Blend of organic alkaline penetrants, emulsifiers, surfactants and corrosion inhibitors that remove grease and petroleum products from the interior of piping systems. Cleaners that contain trisodium phosphate are specifically not acceptable.

Cleaner needs to be compatible with propylene glycol solution.

SYSTEM INHIBITOR

Scale and corrosion inhibitor consisting of boron nitrite, benzol thiazol, benzotriazole, mercapto-benzo-thiazole, and tolyltrizole silicates.

CLOSED WATER SYSTEM TREATMENT

Sequestering agent to reduce deposits and adjust pH: polyphosphate.

Corrosion inhibitors: boron-nitrite, sodium nitrite and borax, sodium totyltriazole, low molecular weight polymers, phosphonates, sodium molybdate, or sulphites.

Conductivity enhancers: phosphates or phosphonates.

GLYCOL SOLUTION

25% Propylene glycol based material specifically designed for use in closed heat transfer systems. Dow Chemical Dow Frost or equal.

TREATMENT EQUIPMENT FOR CLOSED WATER SYSTEMS

Bypass Feeder: 5 gallon minimum capacity, 125 PSIG working pressure, either screw type cover or a valved funnel opening to feed chemical into system, prime coat of paint.

GLYCOL FILL EQUIPMENT

Provide glycol make-up unit with integral pump and reservoir.

Refer to Drawings Schedules for manufacturer and model.

PART 3 - EXECUTION

58 CLEANING SEQUENCE

59 GENERAL:

55

56 57

Systems are to be cleaned before they are used for any purpose except conduct pressure test before cleaning. Add cleaner to
 closed systems at concentrations as recommended by the manufacturer. Remove water filter elements from the system before
 starting circulation. For steam systems, fill boilers only, using the water and cleaner solution.

64 Use neutralizer agents on recommendation of the system cleaner supplier and approval of the Architect/Engineer.

Remove, clean, and replace strainer screens.

4 Inspect, remove sludge, and flush low points with clean water after cleaning process is completed. Include disassembly of 5 components as required.

6 7

8

1 2

3

Use form within this section to document system cleaning, flushing, and proper startup.

9 HOT WATER HEATING SYSTEMS:

Add cleaner to the system water until the M alkalinity value is 250 above that of the initial fill water. Verify the M alkalinity 10 level before and after the addition of the cleaner by means of chemical tests that are observed by the Owner's construction 11 12 representative; include results of all tests in the Operating and Maintenance manuals. Apply heat while circulating, slowly raising temperature to 160°F and maintain for 12 hours minimum; vent all high points to assure 100% system circulation. 13 Remove heat and circulate to 100°F or less; drain system as quickly as possible and refill with clean water. Circulate for 6 hours 14 at design temperature, vent air at all high points, then drain. Refill with clean water and repeat until the system cleaner is 15 removed and the M alkalinity level returns to normal. Remove and clean all strainers. Re-vent the system and install clean 16 17 filter elements in water filters. Treat with scale and corrosion inhibitors before using the system for building heating or cooling.

18

19 CHILLED WATER SYSTEMS:

Add cleaner to the system water until the M alkalinity value is 250 above that of the initial fill water. Verify the M alkalinity level before and after the addition of the cleaner by means of chemical tests that are observed by the Owner's construction representative; include results of all tests in the Operating and Maintenance manuals. Circulate for 48 hours, then drain system as quickly as possible. Refill with clean water, circulate for 24 hours, then drain. Refill with clean water and repeat until system cleaner is removed and the M alkalinity level returns to normal. Remove and clean all strainers. Re-vent the system and install clean filter elements in water filters. Treat with scale and corrosion inhibitors before using the system for building heating or cooling.

2728 GLYCOL WATER SYSTEMS:

Clean and flush as above for water systems. Verify complete drainage by measuring amount of water used for initial fill versus amount actually drained to verify complete removal of cleaning solution. Remove traces of chloride from the system; test to verify removal and submit test results.

32 33 CLOSED WATER SYSTEM TREATMENT

Install separate bypass type feeder at pumps for each closed water system. Provide separate set of supply and return lines from each pump in system and install isolation and balancing valves in each line. Locate system connection that supplies feeder upstream of discharge shutoff valve for pump. Locate the system connection that returns treatment back to the system at convenient point downstream of pump suction shutoff valve. Provide drain valve at bottom of the feeder. Refer to Drawings and Plan Details for additional piping arrangement information.

3940 GLYCOL WATER SYSTEMS

- 41 All water systems utilize glycol water. Fill and charge this system as specified herein.
- 42

43 Completely flush traces of cleaning chemicals before adding glycol water mixture to system. Verify by chemical test.

44

Premix glycol water solution in 50 gallon polyethylene drum to concentration of 30 percent by volume. Use city water to make
 solution. Use hand pump to fill system from mixing tank. Circulate fluid for 24 hours, vent high points where air may collect, add
 more solution to system if needed.

48

After 48 hours, using sample test kit provided by glycol manufacturer, obtain sample and submit sample to manufacturer to test for proper concentration of glycol, and level of inhibitors (if applicable). Submit a copy of test report to Engineer and include a copy of the test report in Operating and Maintenance Manuals.

53 TEST EQUIPMENT

54 Locate test cabinet in convenient location.

56 CONSTRUCTION VERIFICATION ITEMS

57 Contractor is responsible for utilizing the construction verification checklists supplied by the third party commissioning agent in 58 accordance with the procedures defined for construction verification in Section 01 91 00.

END OF SECTION

59

55

Date Submitted:					
Project Name:					
Contractor:					
System Tested: Hot Water Hot Water Steam	leat Pump Loop Condens	Chilled Water_ sate			
System Volume:					
Materials Used (Provide MSDS for each)				
Cleaner:		Quantity Used:			
Inhibitor:		Quantity Used:			
Sequestering Agent:		Quantity Use	ed:		
Neutralizer:		Quantity Use	ed:		
Glycol:		Quantity Use	ed:		
Glycol Solution Water Source:	Percen	Percent glycol by volume:			
M Alkalinity					
Prior to Cleaning:	During Cleaning:	After F	lushing:		
System Temperature					
Prior to Cleaning:	During Cleaning:				
	Date/Time			Date/Time	
Duration	Start		Stop		
Initial Circulation					
Draindown					
System Refill					
Final Circulation					
Heating system Warmup					
Component Checklist (Describe procedu	ires performed at each)				
Strainers:					
Filters:					
Vents:					
Drains:					
Branch Lines:					
TerminalUnits:					
Boilers:					
Chillers:					
Traps:					
Comments:					
1 2 3		SECTION 23 31 00 HVAC DUCTS and CASINGS			
-------------	---	--	--	--	--
4 5		PART 1 - GENERAL			
6					
7 8	SCOPE This section includes specific	cations for all duct systems used on this project. Included are the following topics:			
9 10	PART 1 - GENERAL				
10	Scope				
12	Related Work				
13	Reference				
14	Reference Standar	ds			
15	Quality Assurance				
16	Shop Drawings				
1/	Design Criteria				
10					
20	General				
21	Materials				
22	High Temperature	Flexible Duct			
23	High Pressure Duc	twork (Pressure class 3 inch and over)			
24	Low Pressure Duct	work (Maximum 2 inch pressure class)			
25	Exhaust Duct (Moi	sture laden air)			
20 27	Duct Sediant				
28	Guskets				
29	PART 3 - EXECUTION				
30	Installation				
31	High Pressure Duc	t (Pressure class 3 inch and over)			
32	Low Pressure Duct	(Maximum 2 inch pressure class)			
33	Exhaust Duct (Moi	sture laden air)			
34 25	Cleaning Construction Marification Itoms				
35					
37	Leakage Test				
38	APPENDIX				
39	Duct Leakage Test	Report			
40					
41	RELATED WORK				
42	Section 23 05 93 - Testing, A	djusting, and Balancing for HVAC			
43	Section 23 33 00 – Air Duct A	Accessories			
44 45	REEFRENCE				
46	Applicable provisions of Divi	sion 1 govern work under this Section.			
47	· • • • • • • • • • • • • • • • • • • •				
48	REFERENCE STANDARDS				
49					
50	ANSI SS-EN 485-2	Aluminum and Aluminum Alloys-Sheet, Strip and Plate-Part 2: Mechanical Properties			
51	ASTM B209	Specification for Aluminum and Aluminum-Alloy Sheet and Plate			
52 53		Test Method for Weight of Coating on Zinc-Coated (Galvanized) fron or Steel Afficies			
53 54	ASTM A023	Specification for General Requirements for Steel Sheet, Zinc-Coated (Galvanized) by the Hot-Din			
55	,0111,02,	Process. Lock-Forming Quality			
56	ASTM 924	Standard Specification for General Requirements for Sheet Steel, Metallic-coated by the Hot-dip			
57		Method			
58	ASTM C 1071	Specification for Fibrous Glass Duct Lining Insulation			
59	ASTM C 411	Test Method for Hot Surface Performance of High Temperature Thermal			
60 61		Insulation			
01 62	ASTIVIE 04 ASTM (1228	rest internou for Surface Burning Characteristics of Bullding Materials Test Method for Determining Fungal Resistance of Insulation Materials and Facings			
63	ASTM G 21	Standard Practice for Determining Resistance of Synthetic Polymeric Materials			
64		to Fungi			

NFPA 90A	Standard Specification for Adnesives for Duct Thermal Insulation Standard for the Installation of Air Conditioning and Ventilating Systems
UL 181 NAIMA	Standard for Safety for Factory Made Air Ducts and Air Connectors. Fibrous Glass Duct Liner Standard
QUALITY ASSUR Refer to Divisior	ANCE 1, General Conditions, Equals and Substitutions.
SHOP DRAWING Refer to Divisior	i S 1, General Conditions, Submittals.
Include menufe	sturer's data and lar Contractor data for the following
S	chedule of duct systems including material of construction, gauge, pressure class, system class, metho einforcement, joint construction, fitting construction, and support methods, all with details as appropriate.
• D • D a	uct sealant and gasket material. uct liner including data on thermal conductivity, air friction correction factor, and limitation on tempera nd velocity.
DESIGN CRITERI	Δ
Construct all du conditions.	ictwork to be free from vibration, chatter, objectionable pulsations and leakage under specified oper-
Use material, v publications, un	weight, thickness, gauge, construction, and installation methods as outlined in the following SMA less noted otherwise:
• H	VAC Duct Construction Standards, Metal and Flexible, 3rd Edition, 2005
• H	VAC Air Duct Leakage Test Manual, 1 st Edition, 1985
• H • R	ectangular Industrial Duct Construction Standard, 2nd Edition, 2004
• R	ound Industrial Duct Construction Standards, 2 nd Edition, 1999
Use products wh than 50.	nich conform to NFPA 90A, possess a flame spread rating of not over 25 and a smoke developed rating no hi
DELIVERY, STOR	AGE AND HANDLING
Promptly inspec	t snipments to ensure that Ductwork is undamaged and complies with the specification.
Protect Ductwo	k against damage.
Protect Ductwo	rk by storing inside or by durable, waterproof, above ground packaging. Do not store material on grade. Pro
Ductwork from so caps/packaging	dirt, dust, construction debris and foreign material. Where end caps/packaging are provided, take precaung remain in place and free from damage.
Offsite storage a	greements do not relieve the contractor from using proper storage techniques.
Storage and pro	tection methods must allow inspection to verify products
Storage and pro	lection methods must allow inspection to verify products.
	PART 2 - PRODUCTS
GENERAL	used for construction of duct shall be 24 gauge or bequier except for round and spiral ductivery and spiral
take-offs 12" an 3rd Edition, 200	d below may be 26 gauge where allowed in SMACNA HVAC Duct Construction Standards, Metal and Flex 5.
Duct sizes indica	ted on plans are net inside dimensions; where duct liner is specified, dimensions are net, inside of liner.
All exposed duct	work shall be furnished in a "paint grip" galvanized finish suitable for final finish painting.
Minimum accer	table duct pressure class, for all ductwork except transfer ductwork, is 2 inch W.G. positive or nega
depending on th	ne application. Transfer ductwork minimum acceptable duct pressure class is 1 inch W.G. positive or nega

1	(AHU-1 (E)):	
2	Supply duct upstream of VAV boxes	3 in. pressure class
3	Supply duct downstream of VAV terminals	2 in. pressure class
4	Outside Air ducts	3 in. pressure class
5	Mixed Air ducts	3 in. pressure class
6	Return ducts	3 in. pressure class
7	Relief Air ducts	2 in. pressure class
8		
9	(AHU-4):	
10	Supply air ducts	3 in pressure class
11	Outside Air ducts	3 in pressure class
12	Mixed Air ducts	3 in pressure class
12	Poturn ducts	2 in prossure class
10	Relian ducts	2 in prossure class
14 1E	Relief All ducts	2 III. pressure class
15	(Concernel All Others Durature al.)	
16	(General – All Other Ductwork):	
1/	Supply duct	2 In. pressure class
18	Return duct	2 in. pressure class
19	Exhaust duct	2 in. pressure class
20	Transfer duct	2 in. pressure class
21		
22	MATERIALS	
23	GALVANIZED STEEL SHEET:	
24	Use ASTM A 653 galvanized steel sheet of lock forming of	juality. Galvanized coating to be 1.25 ounces per square foot, both
25	sides of sheet, G90 in accordance with ASTM A90. Provid	le "Paint Grip" finish or galvanneal sheetmetal for ductwork that will
26	be painted.	
27	•	
28	ALUMINUM SHEET (EXHAUST OUT OF RESTROOMS, LOCKE	R ROOMS, AND SHOWERS):
29	Use ANSI/ASTM B209 aluminum sheet, alloy 3003H-14, ca	pable of double seaming without fracture.
30		
30	HIGH PRESSURE DUCTWORK (Pressure class 3 inch and ov	ver)
32	Manufacturers: Aiay Semco United Sheet Metal Sheet M	latal Connectors or approved equal
22	Wandracturers. Ajax, Senico, Onited Sheet Wetal, Sheet W	icial connectors of approved equal.
33		at a sector stand of each and a stand
34	wachine formed round and/or flat oval spiral lock seam du	ict constructed of galvanized steel.
35	5 • • • • • • • • • • • • • • • • • • •	
36	Rectangular high pressure duct using a transverse joint sy	stem as manufactured by Ductmate, Nexus, TDC, TDF, or approved
37	equal, may be used at contractor's option. Duct to be flan	ged, gasketed and sealed.
38		
39	Contractor fabricated ductwork meeting specified	construction standards is acceptable with prior approval of
40	Architect/Engineer. Submit construction details, a description	tion of materials to be used, type of service, reinforcing methods, and
41	sealing procedures.	
42		
43	Use a perforated inner liner on double wall high-pressure	duct. Annular space between inner liner and outer duct to be filled
44	with 1 inch glass fiber insulation.	
45	^c	
46	Use cemented slip joints with 2 inch minimum overlap.	flanged connections, or welded/brazed connections, unless noted
47	otherwise for special applications. Prime coat welded join	ts.
48		
49	Provide standard 90 degree conical tee takeoffs excent	for exhaust at velocities over 2000 feet per minute use 45° lateral
50	connections: straight tans or hullhead tees are not accenta	
50	connections, straight taps of builteau tees are not accepta	bic.
51		
52	Internal bracing will not be accepted on ductwork below 4	8 inches.
53		
54	Use turning vanes as specified in Section 23 33 00.	
55		
56	Provide bellmouth fittings or expanded fittings at each due	t connection to air plenums.
57		
58	Provide pressure relief fittings as indicated on the plans an	d/or details.
59		
60	Transform duct sizes gradually, not exceeding 15 degrees of	livergence and 30 degrees convergence.
61	5 · · · · // · · · · · · · · · · · · · ·	
62	LOW PRESSURE DUCTWORK (Maximum 2 inch pressure c	lass)
63	Fabricate and install ductwork in sizes indicated on the dra	awings and in accordance with SMACNA recommendations, except as
64	modified below.	
-		

1 2

3

4

5

6 7

8

9 10 Construct so that all interior surfaces are smooth. Use slip and drive or flanged and bolted construction when fabricating rectangular ductwork. Use spiral lock seam construction when fabricating round spiral ductwork. Sheet metal screws may be used on duct hangers, transverse joints and other SMACNA approved locations if the screw does not extend more than 1/2 inch into the duct.

Use elbows and tees with a center line radius to width or diameter ratio of 1.5 wherever space permits. When a shorter radius must be used due to limited space, install single wall sheet metal splitter vanes in accordance with SMACNA publications, Type RE 3. Where space will not allow and the C value of the radius elbow, as given in SMACNA publications, exceeds 0.31, use rectangular elbows with turning vanes as specified in Section 23 33 00. Square throat-radius heel elbows will not be acceptable. Straight taps or bullhead tees are not acceptable.

Where rectangular elbows are used, provide turning vanes in accordance with Section 23 33 00.

Provide expanded take-offs or 45 degree entry fittings for branch duct connections with branch ductwork airflow velocities greater than 700 fpm. Square edge 90-degree take-off fittings or straight taps will not be accepted.

Button punch snaplock construction will not be accepted on any ducts.

Round ducts may be substituted for rectangular ducts if sized in accordance with ASHRAE table of equivalent rectangular and round ducts. No variation of duct configuration or sizes permitted except by written permission of the Architect/Engineer.

Increase duct sizes gradually, not exceeding 15 degrees divergence wherever possible. Divergence upstream of equipment shall not exceed 30 degrees; convergence downstream shall not exceed 45 degrees.

DUCT SEALANT

Manufacturer: Tremco Dymonic, 3M 800, 3M 900, H.B. Fuller/Foster, Hardcast, Hardcast Peal & Seal, Lockformer cold sealant, Mon-Eco Industries, United Sheet Metal, or approved equal.

Duct sealant shall be equal to Tremco Dymonic FC, high-performance, fast curing, single-component, hybrid sealant.

Silicone sealants are not allowed in any type of ductwork installation.

Install sealants in strict accordance with manufacturer's recommendations, paying special attention to temperature limitations.

Allow sealant to fully cure before pressure testing of ductwork, or before startup of air handling systems.

GASKETS

2 INCH PRESSURE CLASS AND LOWER:

Soft neoprene or butyl gaskets in combination with duct sealant for flanged joints.

3 INCH PRESSURE CLASS AND HIGHER:

Butyl gaskets.

44

45

60

61 62 63

64

INSTALLATION Verify dimensions at the site, making field measurements and drawings necessary for fabrication and erection. Check plans showing work of other trades and consult with Architect in the event of any interference.

PART 3 - EXECUTION

Make allowances for beams, pipes or other obstructions in building construction and for work of other contractors. Transform, divide or offset ducts as required, in accordance with SMACNA HVAC Duct Construction Standards, Figure 4-7, except do not reduce duct to less than six inches in any dimension and do not exceed an 8:1 aspect ratio. Where it is necessary to take pipes or similar obstructions through ducts, construct easement as indicated in SMACNA HVAC Duct Construction Standards, Figure 4-8, Fig. E. In all cases, seal to prevent air leakage. Pipes or similar obstructions may not pass through high pressure or fume exhaust ductwork.

Test openings for test and balance work will be provided under Section 23 05 93.

Provide frames constructed of angles or channels for coils, filters, dampers or other devices installed in duct systems, and make all connections to such equipment including equipment furnished by others. Secure frames with gaskets and screws or nut, bolts and washers.

Install duct to pitch toward outside air intakes and drain to outside of building. Solder or seal seams to form watertight joints.

_	
1 2 3	Where two different metal ducts meet, the joint shall be installed in such a manner that metal ducts do not contact each other by using proper seal or compound.
4 5 6	Install all motor operated dampers and connect to or install all equipment furnished by others. Blank off all unused portions of louvers, as indicated on the drawings, with 1-1/2 inch board insulation with galvanized sheet metal backing on both sides.
7	Do not install ductwork through dedicated electrical rooms or spaces unless the ductwork is serving this room or space.
9 10	Locate ducts with sufficient space around equipment to allow normal operating and maintenance activities.
10 11 12	Provide adequate access to ductwork for cleaning purposes.
12 13 14	Provide temporary capping of ductwork openings to prevent entry of dirt, dust and foreign material.
15 16	Protect diffusers, registers and grilles with plastic wrap or some other approved form of protection to maintain dirt and dust free and to prevent entry of dirt, dust and foreign material into the Ductwork.
17 18 19	During construction provide temporary closures of metal or taped polyethylene on open ductwork to prevent construction dust from entering ductwork system.
20	
21	DUCTWORK SUPPORT
22 23 24	with secure wire method is not allowed.
24	Support with 2/22 inch. 7 v.7, staipless steplain craft cable, with matching fastener rated for E00/ of actual load, will be allowed
25 26 27	on round ductwork under 12 inches if installed as detailed, with cable double looped on duct and at point of support.
28	HIGH PRESSURE DUICT (Pressure class 3 inch and over)
29 30	Seal all duct in accordance with SMACNA seal class "A"; all seams, joints, and penetrations shall be sealed.
31	LOW PRESSURE DUCT (Maximum 2 inch pressure class)
32	Seal all duct, with the exception of transfer ducts, in accordance with SMACNA seal class "A"; all seams, joints, and penetrations
33	snall be sealed.
34	
35 36 27	grille face dampers will not be accepted for balancing dampers.
37	Hangers must be wrapped around bottom edge of duct and securely fastened to duct with sheetmetal screws or non rivets
39 40	Trapeze hangers may be used at contractor's option.
41	CLEANING
42	Remove all dirt and foreign matter from the entire duct system and clean diffusers, registers, grilles and the inside of air-
43 44	handling units before operating fans.
44 45	Clean duct systems with high power vacuum machines where systems have been used for temporary heat air-conditioning, or
45	vantilation purpose during construction. Protect aquinment that may be barned by excessive dirt with filters, or bynass during
40	cleaning
48	
49	I FAKAGE TEST
50	Test all ductwork in accordance with test methods described in Section 5 of SMACNA HVAC Air Duct Leakage Test Manual. Do
51 52	not insulate ductwork until it has been successfully tested. Test pressure shall be equal to the duct pressure class.
53 54	If excessive air leakage is found locate leaks, repair the duct in the area of the leak, seal the duct, and retest.
55	Leakage rate shall not exceed more than 5% of the system air quantity for low and high pressure ductwork, determined in
56 57	accordance with Appendix C of the SMACNA <u>HVAC Air Duct Leakage Test Manual</u> .
58	Leakage testing method used shall be of the more stringent of two standards (% of airflow or cfm/100 square-feet). by
59 60	discretion of third party testing agent and/or the Owners preference.
61	Submit a signed report to the A/E and Owner, indicating test apparatus used. results of the leakage test. and any remedial work
62 63	required to bring duct systems into compliance with specified leakage rates.

CONSTRUCTION VERIFICATION ITEMS Contractor is responsible for utilizing the construction verification checklists supplied by the third party commissioning agent in accordance with the procedures defined for construction verification in Section 01 91 00.

DUCT LEAKAGE TEST REPORT

Project	Name:	
	Location:	
	Contractor:	
<u>Date</u>		
<u>System</u>	Fan ID:	Leakage Class (CL):
<u>Data</u>	Fan Design CFM:	Duct Pressure Class (P _c):
		Test Pressure (P _T):
<u>Test</u>		
<u>Equipment</u>	Manufacturer:	Model No: Serial No:

For large systems, use the reverse side for a simple sketch of the entire duct system. Then use letter designations to indicate the various duct sections being tested at one time. Also use the reverse side for test comments.

Design Data				Field Test Data								
			Allow	able	Dia	meter	Pres	sure				
			Leaka	ige			(in. v	wc.)				
		Duct	Leakage	CFM			In	Across				
Duct	Duct	Surface	Factor	for	Tube	Orifice	Duct	Orifice		Performed	Observed	Actual
Section	Shape	(Ft²)	(P ^{.65} C _L)	Section	(D ₁)	(D ₂)	(P)	(P _{drop})	Date	Ву	Ву	CFM
TOTAL												

1 2	SECTION 23 33 00 AIR DUCT ACCESSORIES
3	
4 5 6	PART 1 - GENERAL
7 8	SCOPE This section includes accessories used in the installation of duct systems. Included are the following topics:
9 10	
10	Related Work
12	Reference
13	Reference Standards
14	Quality Assurance
15	Shop Drawings
16 17	Operation and Maintenance Data
18	PART 2 - PRODUCTS
19	Manual Volume Dampers
20	Turning Vanes
21	Control Dampers
22	Smoke Detectors
23	Access Doors
24	Flexible Duct
25 26	Duct Lining
20	Duct Elexible Connections
28	Louvers
29	
30	PART 3 - EXECUTION
31	Manual Volume Dampers
32	Turning Vanes
33	Control Dampers
34 25	
36	Flashings
37	Duct Flexible Connections
38	Sound Attenuators
39	Louvers
40	Construction Verification Items
41	
42 42	KELATED WORK Section 22.05.20 – Hanger and Supports for HV/AC Bining and Equipment
45 44	Section 23 05 48 – Vibration and Seismic Controls for HVAC Piping and Equipment
45	Section 23 31 00 – HVAC Ducts and Casings
46	
47	REFERENCE
48	Applicable provisions of Division 1 govern work under this Section.
49	
50 E1	KEFEKENCE STANDARDS
52	SMACNA HVAC Duct Construction Standards - Metal and Elevible 2nd Edition 1995
53	UL 214
54	
55	QUALITY ASSURANCE
56	Also refer to Division 1.
57	
58 50	SHUP UKAWINGS Refer to Division 1. Constal Conditions, Submittals
59 60	
61	Submit for all accessories and include dimensions, capacities, ratings, installation instructions, and appropriate identification.
62	
63	Include certified test data on dynamic insertion loss, self-noise power levels, and aerodynamic performance of sound attenuators
64	

1

2 3

4

5

6 7 8

9 10

11

12 13

14 15

16 17

18 19

20

21 22

23

24 25

26

27 28

31

38

40

42

54 55

56 57

58 59

60

Submit manufacturer's color charts where finish color is specified to be selected by the Architect/Engineer.

OPERATION AND MAINTENANCE DATA

All operations and maintenance data shall comply with the submission and content requirements specified under section GENERAL REQUIREMENTS.

PART 2 - PRODUCTS

MANUAL VOLUME DAMPERS

Manufacturers: Ruskin, Vent Products, Air Balance, or approved equal.

Dampers must be constructed in accordance with SMACNA Fig. 2-12, Fig. 2-13, and notes relating to these figures, except as modified below.

Reinforce all blades to prevent vibration, flutter, or other noise. Construct dampers in multiple sections with mullions where width is over 48 inches. Use rivets or tack welds to secure individual components; sheet metal screws will not be accepted. Provide operators with locking devices and damper position indicators for each damper; use an elevated platform on insulated ducts. Provide end bearings or bushings for all volume damper rods penetrating ductwork constructed to a 3" w.c. pressure class or above.

TURNING VANES

Manufacturers: Aero Dyne, Anemostat, Barber-Colman, Hart & Cooley, or approved equal.

Construct turning vanes and runners for square elbows in accordance with SMACNA Fig. 2-3 and Fig. 2-4 except use only airfoil type vanes. Construct turning vanes for short radius elbows and elbows where one dimension changes in the turn in accordance with SMACNA Fig. 2-5 and Fig. 2-6.

29 CONTROL DAMPERS

30 Control damper actuators are specified in section 23 09 14.

Provide control dampers shown on the plans and schedules and as required to perform the specified functions. Dampers shall be rated for velocities that will be encountered at maximum system design and rated for pressure equal or greater than the ductwork pressure class as specified in Section 23 31 00 of the ductwork where the damper is installed.

36 Use only factory fabricated dampers with mechanically captured replaceable resilient blade seals, stainless steel jamb seals and 37 with entire assembly suitable for the maximum temperature and air velocities encountered in the system.

39 All dampers in aluminum ductwork shall be constructed of stainless steel or aluminum.

41 Dampers in galvanized ductwork shall be constructed of galvanized steel and/or aluminum.

All dampers, unless otherwise specified, to be rated at a minimum of 180° F working temperature. Leakage testing shall be
 certified to be based on latest edition of AMCA Standard 500-D and all dampers, unless otherwise specified, shall have leakage
 ratings as follows:

Damper Class	Differential Pressure	Leakage
Class IA	1″ w.g.	≤3 CFM/ft ²
Class I	4" w.g.	≤8 CFM/ft ²
Class I	8″ w.g.	≤11 CFM/ft ²
Class I	12″ w.g.	≤14 CFM/ft ²

Leakage rate dampers for differential pressures that they will encounter at maximum system design pressures.

Steel framed dampers: Nailor models 2010 & 2020; Greenheck models VCD-33 & VCD-42; Johnson Controls model V-1330; Ruskin Models CD60 & CD40; other approved equal.

Aluminum frame and blade dampers: Nailor models 2010EAF & 202EAF; Greenheck model VCD-43; Ruskin model CD50; Arrow model AFD-20; other approved equal.

Dampers used for directed mixing of airstreams, i.e. outside air and return air, to be parallel blade type and sized for an air velocity of 1800 to 2000 fpm with the damper blades shall be arranged so that the air streams are directed at one another to facilitate mixing. Dampers used for throttling or modulating applications other than air stream mixing to be opposed blade type. Two position dampers may be parallel or opposed blade type. 1 Dampers used for isolation on the discharge of centrifugal fans shall have damper blades perpendicular to the fan shaft to 2 minimize system effect. Dampers mounted with blades vertically shall be designed for vertical blade orientation.

Dampers to have frames of not less than 16 gauge galvanized steel or 12 gauge extruded aluminum. Blades to be two-ply steel
 airfoil of not less than 2 x 20 gauge galvanized steel (14 gauge equivalent) or extruded aluminum airfoil, with stainless steel,
 acetal, Celcon, bronze, or nylon bearings. Maximum allowable blade width is 8 inches. Use plated steel linkage hardware.

Maximum damper width is 48 inches; where required width exceeds 48 inches, use multiple damper sections. Inside frame free
 area shall be a minimum of 90% of total inside duct area.

Multiple width damper sections shall utilize jack shaft linkages unless noted below. Sections over 144 inches wide shall be actuated from two locations on the jack shaft. Double width damper sections for two-position operation may be actuated without jack shafts if each damper section is actuated separately.

14

3

7

10

Dampers that have multiple width and multiple vertical sections shall have a jackshaft for each vertically stacked set of dampers
 and be provided with crossover linkages between jack shafts to transfer uneven loading.

17

Jack shafts shall be extended outside of the ductwork for external actuator mounting. Provide bearings on the point of exit for support of damper shafts to prevent wear on the shaft and the ductwork. If locating actuators out of the air stream is impossible, obtain mounting location approval from the designer unless the contract documents indicate in air stream mounting is acceptable. In no cases shall damper actuators for fume exhaust systems be located in the air stream or require entering the air stream to service an actuator.

2324 SMOKE DETECTORS

25 Smoke detectors are furnished and installed by the Electrical Contractor.

26 27 ACCESS DOORS

Access door to be designed and constructed for the pressure class of the duct in which the door is to be installed. Doors in exposed areas shall be hinged type with cam sash lock. Hinges shall be steel full length continuous piano type. Doors in concealed spaces may be secured in place with cam sash latches. For both hinged and non hinged doors provide sufficient number of camp sash latches to provide air tight seal when door is closed. Do not use hinged doors in concealed spaces if this will restrict access. Use minimum 1" deep 24 gauge galvanized steel double wall access doors with minimum 24 gauge galvanized steel frames.

33

For non-galvanized ductwork, use minimum 1" deep double wall access door with frame that shall use materials of construction identical to adjacent ductwork. Provide double neoprene gasket that shall provide seals from the frame to the door and frame to the duct. When access doors are installed in insulated ductwork or equipment provide insulated doors with insulation equivalent to what is provided for adjacent ductwork or equipment. Access doors constructed with sheet metal screw fasteners will not be accepted.

40 FLEXIBLE DUCT

- 41 Manufacturers: Anco Products, Clevaflex, Thermaflex, Flexmaster or approved equal.
- 42

Factory fabricated , UL 181 listed as a class 1 duct, and having a flame spread of 25 or less and a smoke developed rating of 50 or under in accordance with NFPA 90A.

45

Suitable for pressures and temperatures involved but not less than a 180°F service temperature and ±2 inch pressure class,
 depending on the application.

49 Duct to be composed of polyester film, aluminum laminate or woven and coated fiberglass fabric bonded permanently to 50 corrosion resistant coated steel wire helix. Two-ply, laminated, and corrugated aluminum construction may also be used.

51

52 Where duct is specified to be insulated, provide a minimum 1 inch fiberglass insulation blanket with maximum thermal 53 conductance of 0.23 K (75 degrees F.) and vapor barrier jacket of polyethylene or metalized reinforced film laminate. Maximum 54 perm rating of vapor barrier jacket to be 0.1 perm. 55

56 DUCT LINING

57 Manufacturer: Manville, Owens-Corning, Knauf, or approved equal. 58

1 inch thick, flexible, mat faced insulation made from inorganic glass fibers bonded with a thermosetting resin with thermal conductivity of .25 Btu inch / hour sq.ft. deg F.

61

62 Meet erosion testing per UL 181 or ASTM C 1071 for 5000 fpm maximum air velocity. ASTM C 411 maximum operating

Meet requirements of ASTM C 1338 and ASTM G21 for fungi resistance.

Install liner using adhesive conforming to ASTM C 916.

FLASHINGS

1 2 3

4 5

6

7

8 9

10

11 12

13 14

15

16

22

26

31

39 40 41

42

44 45

46

47 48

49

50

51 52

53

54

55

56 57

58

60

61

Provide flashing to completely weatherproof connection of ductwork to louvers. Flashing to be constructed of material similar to louver material.

Flashing and curbs for duct and pipe penetrations of roof assemblies to be in accordance with details.

DUCT FLEXIBLE CONNECTIONS

Material to be fire retardant, be UL 214 listed, and meet the requirements of NFPA 90A.

Connections to be a minimum of 3 inches wide, crimped into metal edging strip, and air tight. Connections to have adequate flexibility and width to allow for thermal expansion/contraction, vibration of connected equipment, and other movement.

Use coated glass fiber fabric for all applications. Material for inside applications other than corrosive environments, fume exhaust, or kitchen exhaust to be double coated with neoprene, air and water tight, suitable for temperatures between -10°F and 200°F, and have a nominal weight of 30 ounces per square yard. Material used for outdoor applications other than corrosive environments, fume exhaust, or kitchen exhaust to be double coated with Hypalona, air and water tight, suitable for temperatures between -10°F and 250°F, and have a nominal wight of 26 ounces per square yard.

23 LOUVERS

24 Manufacturers: Airolite K6776, Industrial Louvers 658, American Warming and Ventilating LE-31, Construction Specialties 6177, 25 Ruskin ELF6375DX or approved equal.

Similar to Airolite Type K6776, extruded aluminum alloy not less than 12 gauge (.081" thick), 6063 series frame and blades, all welded assembly, 35 degree or 45 degree blades with water baffle, 6 inches thick.

Provide with bird screen of ½" x ½" mesh aluminum in 12 gauge aluminum frame and an aluminum sill. Locate the bird screen inside of the louver unless noted otherwise.

Louver to bear the AMCA certified ratings seal for both air performance and water penetration, having a free area not less than
 50% based on a 48" x 48" section, a water penetration less than 0.1 oz/square foot under AMCA test at 1000 feet per minute,
 and an intake pressure drop less than 0.20 inches of water at 1000 feet per minute.

Finish to be anodized or Kynar 500 in a custom color to be selected by the Architect. Furnish sufficient paint in the same color as the louver to paint the outer surface of panels over unused portions of louvers and to paint the interior portion of ductwork visible through the louvers.

PART 3 - EXECUTION

43 MANUAL VOLUME DAMPERS

Install manual volume dampers in each branch duct and for each grille, register, or diffuser as far away from the outlet as possible while still maintaining accessibility to the damper. Install so there is no flutter or vibration of the damper blade(s).

TURNING VANES

Install turning vanes in all rectangular, mitered elbows in accordance with SMACNA standards and/or manufacturer's recommendations.

Install double wall, airfoil, 2 inch radius vanes in ducts with vane runner length 18" or greater and air velocity less than 2000 fpm. Install double wall, airfoil, 4-1/2 inch radius vanes in ducts with vane runner length 18" or greater and air velocity 2000 fpm or greater.

If duct size changes in a mitered elbow, use single wall type vanes with a trailing edge extension. If duct size changes in a radius elbow or if short radius elbows must be used, install sheetmetal turning vanes in accordance with SMACNA Figure 2-5 and Figure 2-6.

59 CONTROL DAMPERS

Install dampers in locations indicated on the drawings, as detailed, and according to the manufacturer's instructions. Install blank-off plates or transitions where required for proper mixing of airstreams in mixing plenums. Provide adequate operating clearance and access to the operator. Install an access door adjacent to each control damper for inspection and maintenance.

62 63 64

SMOKE DETECTORS 1

Installation and wiring of detectors will be by the Electrical Contractor. Install an access door at each detector location. 2

3 ACCESS DOORS 4

Install access doors where specified, indicated on the drawings, and in locations where maintenance, service, cleaning or 5 6 inspection is required. Examples include, but are not limited to motorized dampers, fire and smoke dampers, smoke detectors, 7 fan bearings, heating and cooling coils, filters, valves, and control devices needing periodic maintenance.

8

Size and numbers of duct access doors to be sufficient to perform the intended service. Minimum access door size shall be 8 x 8 9 inch size for hand access. 18 x 18 inch size for shoulder access, or other size as indicated. Install access doors on both inlet and 10 outlet sides of reheat coils as well as other duct mounted coils. 11

12

FLEXIBLE DUCT 13

Flexible duct may only be used for final connections of air inlets and outlets at diffuser, register, and grille locations. Where 14 15 flexible duct is used, it shall be the minimum length required to make the final connections, but no greater than 5 feet in length, and have no more than one (1) 90 degree bend. 16

Secure inner jacket of flexible duct in place with stainless steel metal band clamp. Secure insulation vapor barrier jacket in place 17 18 with steel or nylon draw band. Sheetmetal screws and/or duct tape will not be accepted.

19 21

26

30

31 32

33

34

35

Flexible duct used to compensate for misalignment of main duct or branch duct will not be accepted. 20

22 Individual sections of flexible ductwork shall be of one piece construction. Splicing of short sections will not be accepted. 23

24 Flexible ductwork used as transfer duct shall be sized for a maximum velocity of 300 fpm.

25 Penetration of any partition, wall, or floor with flexible duct will not be accepted.

27 DUCT LINING

28 Apply lining to the following ductwork or as noted on the plans. 29

- Transfer air ductwork. .
- Return air ductwork where noted on plans.

Do not apply lining to the following ductwork:

- Outside air ductwork.
- Supply, return and exhaust ductwork associated with all air handling units.

Install liner in compliance with the latest edition of NAIMA's Fibrous Glass Duct Liner Standard.

36 37

42

38 Locate longitudinal joints at the corners of duct only. Cut and fit to assure lapped, compressed joints. Coat all transverse and 39 longitudinal joints and edges with adhesive. Provide metal nosing on leading edge where lined duct is preceded by unlined duct. 40 Adhere liner to duct with full coverage area of adhesive. Additionally secure liner to duct using mechanical fasteners spaced as 41 recommended by the liner manufacturer without compressing liner more than 1/8" with the fasteners.

43 FLASHINGS

Flashing for roof curbs, equipment supports or rails located on roof, will be installed by others. 44 45

DUCT FLEXIBLE CONNECTIONS 46

47 Install at all duct connections to rotating or vibrating equipment, including air handling units (unless unit is internally isolated), 48 fans, or other motorized equipment in accordance with SMACNA Figure 2-19. Install thrust restraints to prevent excess strain on 49 duct flexible connections at fan inlets and outlets; see Related Work.

50

51 For applications in corrosive environments or fume exhaust systems, use a double layer of the Teflon¿ coated fabric when making 52 the connector.

53 54 LOUVERS

- 55 Louvers are furnished and installed by others.
- 56

57 Furnish louvers to the General Contractor for mounting in exterior walls. Connect outside air intake duct to the louver, sealing 58 all connections air and water tight. 59

- 60 Provide bird screen on inside of active louver area where none is provided with louvers. Where louvers are equipped with
- inside birdscreen, remove screen at all locations where duct connections are not made. 61

62

Install insulated metal panel on unused portion of louver. Panels must be sealed weathertight to louver assembly with flashing as required for proper drainage to outside of building. Paint outside surface of panel to match louver prior to installation. Where ductwork is visible through louver when viewed from outside the building, paint inside of duct to match louver color.

CONSTRUCTION VERIFICATION ITEMS

Contractor is responsible for utilizing the construction verification checklists supplied by the third party commissioning agent in accordance with the procedures defined for construction verification in Section 01 91 00.

1

2 3

4 5

6

	HVAC FANS
	PART 1 - GENERAL
SCOPE This Sect requiren	tion includes specifications for fans that are not an integral part of a manufactured device. Included are the follo nents:
PART 1	- GENERAL
	Scope
	Related Work
	Reference Standards
	Quality Assurance
	Design Criteria
PART 2	- PRODUCTS
	General Jolina Contribural Fans
	mine centrilugai Fans
PART 3	- EXECUTION
	Installation – General
	Construction Verification Items
	Functional Performance Testing
) WORK
Provision	ns of Division 01 govern work under this Section.
Section	23.05.29 – Hangers and Supports for HVAC Piping and Equipment
Section 2 Section 2 Section 2	23 05 29 – Hangers and Supports for HVAC Piping and Equipment 23 05 13 – Common Motor Requirements for HVAC Equipment 23 05 48.10 – Vibration and Seismic Controls for HVAC Piping and Equipment
Section 2 Section 2 Section 2 Section 2	23 05 29 – Hangers and Supports for HVAC Piping and Equipment 23 05 13 – Common Motor Requirements for HVAC Equipment 23 05 48.10 – Vibration and Seismic Controls for HVAC Piping and Equipment 23 33 00 – Air Duct Accessories
Section 2 Section 2 Section 2 Section 2	23 05 29 – Hangers and Supports for HVAC Piping and Equipment 23 05 13 – Common Motor Requirements for HVAC Equipment 23 05 48.10 – Vibration and Seismic Controls for HVAC Piping and Equipment 23 33 00 – Air Duct Accessories NCE STANDARDS
Section 2 Section 2 Section 2 Section 2 REFEREN AMCA 20	23 05 29 – Hangers and Supports for HVAC Piping and Equipment 23 05 13 – Common Motor Requirements for HVAC Equipment 23 05 48.10 – Vibration and Seismic Controls for HVAC Piping and Equipment 23 33 00 – Air Duct Accessories NCE STANDARDS 03 AMCA Fan Application Manual - Troubleshooting
Section 2 Section 2 Section 2 REFEREN AMCA 20 AMCA 20	 23 05 29 – Hangers and Supports for HVAC Piping and Equipment 23 05 13 – Common Motor Requirements for HVAC Equipment 23 05 48.10 – Vibration and Seismic Controls for HVAC Piping and Equipment 23 33 00 – Air Duct Accessories NCE STANDARDS 03 AMCA Fan Application Manual - Troubleshooting 04 Balance Quality and Vibration Levels for Fans
Section 2 Section 2 Section 2 REFEREN AMCA 20 AMCA 20 AMCA 22	 23 05 29 – Hangers and Supports for HVAC Piping and Equipment 23 05 13 – Common Motor Requirements for HVAC Equipment 23 05 48.10 – Vibration and Seismic Controls for HVAC Piping and Equipment 23 33 00 – Air Duct Accessories VCE STANDARDS 03 AMCA Fan Application Manual - Troubleshooting 04 Balance Quality and Vibration Levels for Fans 10 Laboratory Method of Testing Fans for Rating
Section 2 Section 2 Section 2 Section 2 REFEREN AMCA 20 AMCA 20 AMCA 20 AMCA 30	 23 05 29 – Hangers and Supports for HVAC Piping and Equipment 23 05 13 – Common Motor Requirements for HVAC Equipment 23 05 48.10 – Vibration and Seismic Controls for HVAC Piping and Equipment 23 33 00 – Air Duct Accessories VCE STANDARDS 03 AMCA Fan Application Manual - Troubleshooting 04 Balance Quality and Vibration Levels for Fans 10 Laboratory Method of Testing Fans for Rating 00 Reverberant Room Method for Sound Testing of Fans
Section 2 Section 2 Section 2 Section 2 REFEREN AMCA 20 AMCA 20 AMCA 20 AMCA 30 NFPA 90	 23 05 29 – Hangers and Supports for HVAC Piping and Equipment 23 05 13 – Common Motor Requirements for HVAC Equipment 23 05 48.10 – Vibration and Seismic Controls for HVAC Piping and Equipment 23 33 00 – Air Duct Accessories VCE STANDARDS 03 AMCA Fan Application Manual - Troubleshooting 04 Balance Quality and Vibration Levels for Fans 10 Laboratory Method of Testing Fans for Rating 00 Reverberant Room Method for Sound Testing of Fans 04 Standard for the Installation of Air Conditioning and Ventilating Systems
Section 2 Section 2 Section 2 Section 2 REFEREN AMCA 20 AMCA 20 AMCA 30 NFPA 90 UL	 23 05 29 – Hangers and Supports for HVAC Piping and Equipment 23 05 13 – Common Motor Requirements for HVAC Equipment 23 05 48.10 – Vibration and Seismic Controls for HVAC Piping and Equipment 23 33 00 – Air Duct Accessories VCE STANDARDS 03 AMCA Fan Application Manual - Troubleshooting 04 Balance Quality and Vibration Levels for Fans 10 Laboratory Method of Testing Fans for Rating 00 Reverberant Room Method for Sound Testing of Fans 1A Standard for the Installation of Air Conditioning and Ventilating Systems Underwriters Laboratory for fan electrical components and wiring.
Section 2 Section 2 Section 2 Section 2 AMCA 20 AMCA 20 AMCA 20 AMCA 30 NFPA 90 UL UL 705	 23 05 29 - Hangers and Supports for HVAC Piping and Equipment 23 05 13 - Common Motor Requirements for HVAC Equipment 23 05 48.10 - Vibration and Seismic Controls for HVAC Piping and Equipment 23 33 00 - Air Duct Accessories NCE STANDARDS 03 AMCA Fan Application Manual - Troubleshooting 04 Balance Quality and Vibration Levels for Fans 10 Laboratory Method of Testing Fans for Rating 00 Reverberant Room Method for Sound Testing of Fans 14 Standard for the Installation of Air Conditioning and Ventilating Systems Underwriters Laboratory for fan electrical components and wiring. Underwriters Laboratory for emergency smoke control systems
Section 2 Section 2 Section 2 REFEREN AMCA 20 AMCA 20 AMCA 20 AMCA 20 AMCA 30 NFPA 90 UL UL 705	 23 05 29 - Hangers and Supports for HVAC Piping and Equipment 23 05 13 - Common Motor Requirements for HVAC Equipment 23 05 48.10 - Vibration and Seismic Controls for HVAC Piping and Equipment 23 33 00 - Air Duct Accessories NCE STANDARDS 03 AMCA Fan Application Manual - Troubleshooting 04 Balance Quality and Vibration Levels for Fans 10 Laboratory Method of Testing Fans for Rating 00 Reverberant Room Method for Sound Testing of Fans 10 Standard for the Installation of Air Conditioning and Ventilating Systems Underwriters Laboratory for fan electrical components and wiring. Underwriters Laboratory for emergency smoke control systems
Section 2 Section 2 Section 2 REFEREN AMCA 20 AMCA 30 AMCA 20 AMCA 30 AMCA 30	 23 05 29 - Hangers and Supports for HVAC Piping and Equipment 23 05 13 - Common Motor Requirements for HVAC Equipment 23 05 48.10 - Vibration and Seismic Controls for HVAC Piping and Equipment 23 33 00 - Air Duct Accessories VCE STANDARDS 03 AMCA Fan Application Manual - Troubleshooting 04 Balance Quality and Vibration Levels for Fans 10 Laboratory Method of Testing Fans for Rating 00 Reverberant Room Method for Sound Testing of Fans 1A Standard for the Installation of Air Conditioning and Ventilating Systems Underwriters Laboratory for fan electrical components and wiring. Underwriters Laboratory for emergency smoke control systems Y ASSURANCE tion of Materials: Refer to Division 01 and General Conditions of the Contract, Article 3.
Section 2 Section 2 Section 2 Section 2 REFEREN AMCA 20 AMCA 2	 23 05 29 - Hangers and Supports for HVAC Piping and Equipment 23 05 13 - Common Motor Requirements for HVAC Equipment 23 05 48.10 - Vibration and Seismic Controls for HVAC Piping and Equipment 23 33 00 - Air Duct Accessories NCE STANDARDS 03 AMCA Fan Application Manual - Troubleshooting 04 Balance Quality and Vibration Levels for Fans 10 Laboratory Method of Testing Fans for Rating 10 Reverberant Room Method for Sound Testing of Fans 10 Numeriters Laboratory for fan electrical components and wiring. 10 Underwriters Laboratory for emergency smoke control systems 10 Y ASSURANCE tion of Materials: Refer to Division 01 and General Conditions of the Contract, Article 3.
Section 2 Section 2 Section 2 Section 2 AMCA 20 AMCA 20 AMCA 20 AMCA 20 AMCA 30 NFPA 90 UL UL 705 QUALITY Substitut	 23 05 29 - Hangers and Supports for HVAC Piping and Equipment 23 05 13 - Common Motor Requirements for HVAC Equipment 23 05 48.10 - Vibration and Seismic Controls for HVAC Piping and Equipment 23 33 00 - Air Duct Accessories NCE STANDARDS 03 AMCA Fan Application Manual - Troubleshooting 04 Balance Quality and Vibration Levels for Fans 10 Laboratory Method of Testing Fans for Rating 00 Reverberant Room Method for Sound Testing of Fans 10 Laboratory Method for Sound Testing of Fans 10 Numeriviters Laboratory for fan electrical components and wiring. Underwriters Laboratory for emergency smoke control systems Y ASSURANCE tion of Materials: Refer to Division 01 and General Conditions of the Contract, Article 3.
Section 2 Section 2 Section 2 Section 2 AMCA 20 AMCA 20 AMCA 20 AMCA 20 AMCA 30 NFPA 90 UL UL 705 QUALITY Substitut	 23 05 29 – Hangers and Supports for HVAC Piping and Equipment 23 05 13 – Common Motor Requirements for HVAC Equipment 23 05 48.10 – Vibration and Seismic Controls for HVAC Piping and Equipment 23 33 00 – Air Duct Accessories NCE STANDARDS 03 AMCA Fan Application Manual - Troubleshooting 04 Balance Quality and Vibration Levels for Fans 10 Laboratory Method of Testing Fans for Rating 00 Reverberant Room Method for Sound Testing of Fans 10 Laboratory for the Installation of Air Conditioning and Ventilating Systems Underwriters Laboratory for fan electrical components and wiring. Underwriters Laboratory for emergency smoke control systems Y ASSURANCE tion of Materials: Refer to Division 01 and General Conditions of the Contract, Article 3. RAWINGS
Section 2 Section 2 Section 2 Section 2 REFEREN AMCA 20 AMCA 20 AMCA 20 AMCA 20 AMCA 20 AMCA 20 AMCA 20 UL UL 705 QUALITY Substitut SHOP DF Include of sound poperformed	 23 05 29 - Hangers and Supports for HVAC Piping and Equipment 23 05 13 - Common Motor Requirements for HVAC Equipment 23 05 48.10 - Vibration and Seismic Controls for HVAC Piping and Equipment 23 33 00 - Air Duct Accessories VCE STANDARDS 03 AMCA Fan Application Manual - Troubleshooting 04 Balance Quality and Vibration Levels for Fans 10 Laboratory Method of Testing Fans for Rating 10 Reverberant Room Method for Sound Testing of Fans 10 Standard for the Installation of Air Conditioning and Ventilating Systems 11 Underwriters Laboratory for fan electrical components and wiring. 12 Underwriters Laboratory for emergency smoke control systems 13 ASURANCE 14 ASSURANCE 15 tion of Materials: Refer to Division 01 and General Conditions of the Contract, Article 3. RAWINGS dimensions, capacities, fan curves, materials of construction, ratings, weights, motors and drives, motor efficier ower levels, appropriate identification and vibration isolation for equipment. Sound power levels to be based on ed in accordance with AMCA Standard 300.
Section 2 Section 2 Section 2 Section 2 Section 2 AMCA 20 AMCA	 23 05 29 – Hangers and Supports for HVAC Piping and Equipment 23 05 13 – Common Motor Requirements for HVAC Equipment 23 05 48.10 – Vibration and Seismic Controls for HVAC Piping and Equipment 23 33 00 – Air Duct Accessories VCE STANDARDS 03 AMCA Fan Application Manual - Troubleshooting 04 Balance Quality and Vibration Levels for Fans 10 Laboratory Method of Testing Fans for Rating 00 Reverberant Room Method for Sound Testing of Fans 10 Underwriters Laboratory for fan electrical components and wiring. 10 Underwriters Laboratory for emergency smoke control systems Y ASSURANCE tion of Materials: Refer to Division 01 and General Conditions of the Contract, Article 3. RAWINGS dimensions, capacities, fan curves, materials of construction, ratings, weights, motors and drives, motor efficient over levels, appropriate identification and vibration isolation for equipment. Sound power levels to be based on ed in accordance with AMCA Standard 300. color selection charts for equipment where applicable.
Section 2 Section 2 Section 2 Section 2 Section 2 AMCA 20 AMCA	 23 05 29 – Hangers and Supports for HVAC Piping and Equipment 23 05 13 – Common Motor Requirements for HVAC Equipment 23 05 48.10 – Vibration and Seismic Controls for HVAC Piping and Equipment 23 33 00 – Air Duct Accessories VCE STANDARDS 03 AMCA Fan Application Manual - Troubleshooting 04 Balance Quality and Vibration Levels for Fans 10 Laboratory Method of Testing Fans for Rating 10 Reverberant Room Method for Sound Testing of Fans 14 Standard for the Installation of Air Conditioning and Ventilating Systems 15 Underwriters Laboratory for fan electrical components and wiring. 16 Underwriters Laboratory for emergency smoke control systems 17 ASSURANCE 18 dimensions, capacities, fan curves, materials of construction, ratings, weights, motors and drives, motor efficier over levels, appropriate identification and vibration isolation for equipment. Sound power levels to be based on ed in accordance with AMCA Standard 300. 19 color selection charts for equipment where applicable. 10 ves shall indicate relationship of CFM to static or total pressure for various fan speeds. Brake horsepor ended selection range, and limits of operation are to be indicated on curves. Indicate operating point on fan curves
Section 2 Section 2 Section 2 Section 2 Section 2 AMCA 20 AMCA	 22 30 5 29 - Hangers and Supports for HVAC Piping and Equipment 23 05 13 - Common Motor Requirements for HVAC Equipment 23 05 48.10 - Vibration and Seismic Controls for HVAC Piping and Equipment 23 33 00 - Air Duct Accessories VCE STANDARDS 3 AMCA Fan Application Manual - Troubleshooting 04 Balance Quality and Vibration Levels for Fans 10 Laboratory Method of Testing Fans for Rating 00 Reverberant Room Method for Sound Testing of Fans 10 A Standard for the Installation of Air Conditioning and Ventilating Systems 10 Underwriters Laboratory for fan electrical components and wiring. 10 Underwriters Laboratory for emergency smoke control systems 11 A STANDARCE 12 tion of Materials: Refer to Division 01 and General Conditions of the Contract, Article 3. RAWINGS dimensions, capacities, fan curves, materials of construction, ratings, weights, motors and drives, motor efficien ower levels, appropriate identification and vibration isolation for equipment. Sound power levels to be based on ed in accordance with AMCA Standard 300. color selection charts for equipment where applicable. ves shall indicate relationship of CFM to static or total pressure for various fan speeds. Brake horseptieneded selection range, and limits of operation are to be indicated on curves. Indicate operating point on fan curva ir quantity and indicate manufacturer's recommended drive loss factor for specific application. Tabular ance data is not acceptable.
Section 2 Section 2 Section 2 Section 2 Section 2 AMCA 20 AMCA 20 AMCA 20 AMCA 20 AMCA 20 AMCA 20 AMCA 20 UL 705 QUALITY Substitut SHOP DF Include 0 sound po perform Submit c Fan cur recomm design a perform For varia	 23 05 29 - Hangers and Supports for HVAC Piping and Equipment 23 05 13 - Common Motor Requirements for HVAC Equipment 23 05 48.10 - Vibration and Seismic Controls for HVAC Piping and Equipment 23 33 00 - Air Duct Accessories VCE STANDARDS 03 AMCA Fan Application Manual - Troubleshooting 04 Balance Quality and Vibration Levels for Fans 10 Laboratory Method of Testing Fans for Rating 00 Reverberant Room Method for Sound Testing of Fans 14 Standard for the Installation of Air Conditioning and Ventilating Systems 15 Underwriters Laboratory for fan electrical components and wiring. 16 Underwriters Laboratory for emergency smoke control systems 17 ASSURANCE 18 to Materials: Refer to Division 01 and General Conditions of the Contract, Article 3. RAWINGS 19 dimensions, capacities, fan curves, materials of construction, ratings, weights, motors and drives, motor efficier over levels, appropriate identification and vibration isolation for equipment. Sound power levels to be based on ed in accordance with AMCA Standard 300. 10 color selection charts for equipment where applicable. 10 ves shall indicate relationship of CFM to static or total pressure for various fan speeds. Brake horsepp ended selection range, and limits of operation are to be indicated on curves. Indicate operating point on fan curva are quarity and indicate manufacturer's recommended drive loss factor for specific application. Tabular ance data is not acceptable.

Include vibration isolation equipment Shop Drawing information as part of fan submittal. Reference Section 23 05 48 for additional requirements.

DESIGN CRITERIA

1

2

3 4

5

6 7

8

9 10

11 12

13

14 15

16 17

18 19

20

21 22

23 24 25

26

27 28

29

30 31

32

38

40

43

45

55

57

59 60

61 62 63 Tested and certify fans are in accordance with applicable AMCA test code.

Each fan and motor combination shall be capable of delivering 110 percent of air quantity scheduled at scheduled static pressure. Motor furnished with fan shall not operate into motor service factor when operating under these conditions.

Consider drive efficiency in motor selection according to manufacturer's published recommendation or according to AMCA Publication 203, Appendix L.

Where inlet and outlet ductwork is field modified at fan location from that shown on Drawings, all required motor, drive and wiring changes required due to increased static pressure or baffling necessary to prevent uneven airflow or improve mixing.

Internal insulation and other components exposed to air stream to meet flame spread and smoke ratings contained in NFPA 90A.

Roof mounted equipment shall be provided with curbs or equipment stands in accordance with specification in Section 23 05 29.

PART 2 - PRODUCTS

GENERAL

Use fan size, class, type, arrangement, and capacity as scheduled on the drawings.

Furnish complete with motors, wheels, drive assemblies, bearings, vibration isolation devices, and accessories required for specified performance and proper operation.

Single-phase motors shall have inherent thermal overload protection.

Provide adjustable pitch sheaves for fans with V-belt drives. V-belt drives shall be sized for 150 percent of motor rating. Provide
 fixed sheaves for fans using variable frequency drives.

36 Use OSHA approved belt guards that totally enclose entire drive. Construct guards of expanded metal to allow for ventilation; 37 provide tachometer openings at shaft locations.

39 Statically and dynamically balance fans so they operate without objectionable noise or vibration.

41 IN-LINE CENTRIFUGAL FANS

42 Manufacturers: Carnes, Cook, or Greenheck.

44 Specification is applicable to fans similar in style to Greenheck SQ or a Cook SQ.

46 Construct housing of steel with reinforcing to prevent distortion. Fan housings shall include bolted and gasketed access panel 47 for inspection of fan drive and fan wheel. Use non-overloading, aluminum, backward inclined type fan wheels. For belt drive 48 units, isolate belt drives from air stream with belt tube and bearing cover.

For direct drive units, provide internal motor/bearing cover. Externally mount motors on an adjustable base. Bearings to be
 grease lubricated, self-aligning ball bearing type with grease seal and external grease fitting. Fan housing shall include square
 duct mounting collars.

53 Fan shall be configured for either horizontal or vertical airflow configuration and include integral support points for suspended 54 or base mounting as specified on Drawings.

56 Design vertically mounted fans to withstand vertical thrust loads.

58 Unit shall include factory mounted and prewired NEMA 1 fan motor disconnect switch.

Fan motor shall be NEMA approved, open ball bearing type with resilient mounting. Motors shall be totally enclosed, explosion proof or 2 speed type.

Provide 1/2 inch square galvanized wire mesh inlet screens for fans without inlet ductwork.

- For direct drive units, furnish solid state, variable speed controller for each fan. Speed controller to be used for final air
 balancing of fan.
 3
- 4 Provide each fan with neoprene or spring isolation hangers as specified in Section 23 05 48.10.

6

7

8

15

17

INSTALLATION - GENERAL

Install units as shown on Drawings, as detailed, and according to manufacturer's installation instructions. On units provided
 with drain connection, install drain valve and cap discharge of drain.

PART 3 - EXECUTION

Install thrust restraints in accordance with requirements of Section 23 05 48.10.

14 Fan drive sheaves on belt driven units shall be adjusted or replaced by Contractor to provide design air volumes.

16 Furnish wall and roof opening locations and dimensions to other sections of work requiring opening information.

For direct drive fans furnished with variable fan speed controllers, Electrical Contractor will wire and install fan speed controllers on fans.

21 Suspend inline fans from structure with vibration isolation devices as specified herein.

22 23 CONSTRUCTION VERIFICATION ITEMS

24 Contractor is responsible for utilizing the construction verification checklists supplied by the third party commissioning agent in 25 accordance with the procedures defined for construction verification in Section 01 91 00.

27 FUNCTIONAL PERFORMANCE TESTING

28 Contractor is responsible for utilizing the functional performance test forms supplied by the third party commissioning agent in 29 accordance with the procedures defined for functional performance testing in Section 01 91 00.

30 31

26

32

1 2 2	SECTION 23 36 00 AIR TERMINAL UNITS
4	
5	PART 1 - GENERAL
7	SCOPE
8 9	This section includes specifications for air terminal equipment. Included are the following topics:
10	PART 1 - GENERAL
11	Scope
12	Related Work
13	Reference
14	Reference Standards
15	Quality Assurance
16	Shop Drawings
1/	Operation and Maintenance Data
18	Design Criteria
19	
20	
21	Access Doors
22	ALLESS DUDIS
23	PART 3 - FXECUTION
25	Installation
26	Access Doors
27	Adjusting
28	Construction Verification Items
29	Functional Performance Testing
30	
31	RELATED WORK
32	Section 23 09 14 - Electric Instrumentation and Control Devices for HVAC
33	Section 23 09 93 – Sequence of Operations for HVAC Controls
34	Section 23 31 00 - HVAC Ducts and Casings
35	Section 23 33 00 - Air Duct Accessories
36	
37	REFERENCE
38	Applicable provisions of Division 1 govern work under this section.
39	
40	REFERENCE STANDARDS
41	NFPA 90A - Installation of Air Conditioning and Ventilation Systems.
42	UL 181 - Factory-Made Air Ducts and Connectors.
43	ARI-ADC Standard 880
44	ASIW E84 – Surface Burning Characteristics of Building Materials
45	UL 723 – Surface Burning Characteristics of Bunding Materials
40	
47 70	QUALITY ASSOCIATIVE
40 10	Refer to Division 1, General Conditions, Equals and Substitutions.
50	SHOP DRAWINGS
51	Refer to Division 1. General Conditions. Submittals
52	
53	Contractor shall submit air terminal unit data including materials of construction, dimensions, scheduled flow rates, pressure
54	drops, radiated and discharge sound power levels, reset volume controller data, actuator spring range and torque data.
55	
56	OPERATION AND MAINTENANCE DATA
57	All operations and maintenance data shall comply with the submission and content requirements specified under section
58	GENERAL REQUIREMENTS.
59	
60	DESIGN CRITERIA

- 61 Select sizes, capacities, configuration, and operating characteristics as shown on the plans and/or as scheduled.
- 62 63

1 PART 2 - PRODUCTS 2 3 SUPPLY AIR TERMINAL BOXES 4 Units shall be single duct and pressure independent. 5 6 MANUFACTURER: 7 Titus 8 9 CONSTRUCTION: 10 As scheduled, unit casing shall be minimum 22 gauge steel and internally insulated with 1" thick matte faced insulation with solid 22 gauge metal liner cover over all insulation including cut edges. Provide low leakage casing. Construction to meet UL 11 12 181 and NFPA 90A. 13 Casing outlet shall have slip and drive joint for connection to discharge ductwork. 14 15 Metal damper blade shall be mounted to shaft having self-lubricated bearings. Shaft end shall be marked to indicate damper 16 position and shall have a built-in stop to prevent overstroking. Damper blade shall close off against gasket to limit leakage to 17 10 cfm at 6.0 inches of differential static pressure. Damper linkage shall be sized to accept at least 40 inch-pounds of torque to 18 19 the damper shaft. Damper shaft shall be provided with a marking indicating damper position. 20 Round inlet collar shall be equipped with a multi-point flow sensor that shall amplify the measured velocity pressure. 21 Pneumatic tubing from flow sensor to differential pressure transducer shall be UL listed, fire retardant (FR) type. 22 23 24 HOT WATER REHEAT COIL: 25 Reference section 23 82 00 for hot water reheat coil specifications. 26 27 **TERMINAL AIR BOX CONTROLS** 28 ACCESS DOORS 29 STANDARD ACCESS DOORS 30 31 Access door to be designed and constructed for the pressure class of the duct in which the door is to be installed. Doors in exposed areas shall be hinged type with cam sash lock. 32 33 34 Hinges shall be steel full length continuous piano type. Doors in concealed spaces may be secured in place with cam sash 35 latches. For both hinged and non hinged doors provide sufficient number of camp sash latches to provide air tight seal when door is closed. Do not use hinged doors in concealed spaces if this will restrict access. Use minimum 1" deep 24 gauge 36 galvanized steel double wall access doors with minimum 24 gauge galvanized steel frames. 37 38 For non-galvanized ductwork, use minimum 1" deep double wall access door with frame that shall use materials of construction 39 40 identical to adjacent ductwork. Provide double neoprene gasket that shall provide seals from the frame to the door and frame to the duct. When access doors are installed in insulated ductwork or equipment provide insulated doors with insulation 41 equivalent to what is provided for adjacent ductwork or equipment. Access doors constructed with sheet metal screw 42 43 fasteners will not be accepted. 44 45 ROUND DUCT ACCESS DOORS: For duct pressure class positive or negative up to 6 in. wg. Access doors shall be constructed from 16 gauge stainless steel for 46 fume exhaust ducts and 16 gauge galvanized steel for general exhaust or return ducts. Hinges shall be continuous piano style 47 constructed from the same material as the access door. 48 49 Access doors shall be sealed with ¼" closed cell butyl gasketing permanently bonded on all four sides and no fewer than two 50 draw latches with strike plates. The strike plates shall match the duct/access door material. 51 52 53 INSULATION 54 Refer to Section 23 07 00 for insulation requirements for ductwork and reheat coils. 55 56 57 PART 3 - EXECUTION 58 59 INSTALLATION Install air terminal units as indicated on project drawings and in accordance with the manufacturer's installation instructions. 60 61

62 Mount air terminal boxes with a minimum 3 feet of straight ductwork upstream of inlet flow sensor for sizes 12" diameter and 63 below. Provide a minimum of 3X the inlet diameter of straight duct upstream of the inlet flow sensor for inlet sizes above 12" diameter.

64

1 Field mount coil separate from box with a 12-18" section of duct between the air terminal box and reheat coil. The reheat coil

- and 12-18" section of duct shall be wrapped with external insulation as indicated in specification section 23 07 00 HVAC
 Insulation.
- 4

7

5 Provide at least 24" of clearance on controller side of the air terminal unit. The clearance area shall extend the full length of the 6 supply air terminal unit and the full length (including the access door) of the exhaust/return air terminal unit

Support air terminal units from building structure using sheet metal straps or trapeze hanger with rods. Do not mount air
 terminal units off of adjacent ductwork or piping.

10

11 ACCESS DOORS

12 DUCT ACCESS DOORS – SQUARE DUCT:

Provide duct access doors in duct or extended supply air terminal unit upstream and downstream of the reheat coil. Duct access doors shall be as large as duct allows with a maximum size of 18"x18". Install heating coils in accordance with Section 23 73 12 - Air Handling Unit Coils.

16

17 DUCT ACCESS DOORS – ROUND DUCT:

Install round duct access doors on the side of the duct upstream of the return/exhaust terminal unit. At no time shall the access door be installed in the bottom of the duct. Piano hinged style access doors shall be installed with the piano hinges located ½ above the bottom of the duct to allow the access door to swing down toward the floor.

2122 ADJUSTING

Coordinate adjustment of air terminal units with section 23 05 93 - Testing, Adjusting and Balancing.

25 **CONSTRUCTION VERIFICATION ITEMS**

Contractor is responsible for utilizing the construction verification checklists supplied by the third party commissioning agent in
 accordance with the procedures defined for construction verification in Section 01 91 00.

29 FUNCTIONAL PERFORMANCE TESTING

Contractor is responsible for utilizing the functional performance test forms supplied by the third party commissioning agent in accordance with the procedures defined for functional performance testing in Section 01 91 00.

32

28

33 34

1	SECTION 23 37 13
2	DIFFUSERS, REGISTERS & GRILLES
3	
4 5	PART 1 - GENERAL
6	
7	SCOPE
8	This section includes specifications for air terminal equipment. Included are the following topics:
10	ΡΔΡΤ 1 - GENERΔΙ
11	Scope
12	Related Work
13	Reference
14	Reference Standards
15	Quality Assurance
16	Submittals
17	Design Criteria
18	PART 2 - PRODUCTS
19	Manufacturers Bound Coiling Diffusore - Plague
20	Side-Wall Registers and Grilles
21	Spiral Duct Grilles
23	High-Capacity Drum Louver Diffusers
24	Heavy Duty Return Grilles
25	
26	PART 3 - EXECUTION
27	Installation
28	Construction Verification Items
29	
30 21	RELATED WORK Section 22.05.02 Tecting Adjusting and Palancing for HVAC
32	Section 23 31 00 - HVAC Ducts and Casings
33	Section 23 33 00 - Air Duct Accessories
34	
35	REFERENCE
36	Applicable provisions of Division 1 govern work under this section.
37	
38	REFERENCE STANDARDS
39	NFPA 90A - Installation of Air Conditioning and Ventilation Systems.
40	UL 181 - Factory-Made Air Ducts and Connectors.
41 //2	ARI-ADC Stalluaru 880
43	OLIALITY ASSURANCE
44	Refer to division 1, General Conditions, Equals and Substitutions.
45	
46	SUBMITTALS
47	Refer to division 1, General Conditions, Submittals.
48	
49	Furnish submittal information including, but not limited to, the following:
50	Manufacturar's name and model number
51	Manufacturer's name and model number
52	Canacities/ratings
54	Materials of construction
55	Sound ratings
56	Dimensions
57	Finish
58	Color selection charts where applicable
59	Manufacturer's installation instructions
60	All other appropriate data
61 62	
02 63	All performance data shall be based on tests conducted in accordance with Air Diffusion Council (ADC) Test Code 1062 GPD 94
64	

1 2	PART 2 - PRODUCTS
3 4 5	MANUFACTURERS Manufacturers: Carnes, Krueger, Titus, Metal-Aire, and E.H. Price, and Nailor.
6 7	Acceptable manufacturers for specific products are listed under each item.
8 9 10	ROUNE CEILING DIFFUSERS – Three Cone Carnes SSRA, Krueger RM2PLQ, Metalaire R5750, Titus Omni R, or Price RPD
11 12	Devices shall be specifically designed for variable-air-volume flows.
13 14	Material: Steel unless otherwise indicated on Drawings.
15 16	Finish: Baked enamel, white unless otherwise indicated on Drawings.
17 18 10	Face Style: One cone.
19 20 21	Mounting, pattern, and damper requirements shall be indicated on Drawings.
22 23 24	SIDE-WALL REGISTERS AND GRILLES Price model 520 (Supply) or 630 (return/exhaust) or equal.
25 26	Aluminum (Steel) unless otherwise indicated, with frame type appropriate to installation.
27 28	Single deflection type blade supply registers and supply grilles allow deflection adjustment in all direction.
29 30	Opposed blade volume control damper supply registers, operable from face.
31 32	Fixed blade (45 degree) core return and exhaust registers and grilles.
33 34	Register and grille sizes as shown on drawings and/or as scheduled.
35 36	White, baked enamel finish or powder coat finish, unless otherwise indicated.
37	SPIRAL DUCT GRILLES
38 39	Price model SDGE or equal.
40 41	Aluminum unless otherwise indicated, with frame type appropriate to installation.
42 43	Double deflection type blade supply registers and supply grilles allow deflection adjustment in all direction.
44 45	Grille shall be provided with countersunk holes and closed cell foam insulation gasket around grille neck.
46 47	Opposed blade volume control damper supply registers, operable from face.
48 49	Register and grille sizes as shown on drawings and/or as scheduled.
50 51	Coordinate with architect for final finish of grille. Grilles shall have either baked on powder coat finish or clear anodized finish.
52 53 54 55	HIGH-CAPACITY DRUM LOUVERS Carnes R series, Krueger DPL, Metalaire RL, Titus DL, or Price HCD. Airflow Principle: Extended distance for high airflow rates.
56 57	Material: Aluminum, heavy gage extruded.
58 59	Finish: White baked acrylic.
60 61	Border: 1-1/4-inch width with countersunk screw holes.
62 63	Gasket between drum and border.
64	Body: Drum shaped; adjustable vertically.

1 2	Blades: Individually adjustable horizontally, minimum 60 degrees.
3	Mounting: Surface to duct or wall
5 6	Accessories: Opposed-blade steel damper, duct-mounting collars with countersunk screw holes.
7	HEAVY DUTY RETURN GRILLES
8	Price 90 Series or equal.
9	
10 11	Steel, 14 gauge extruded, with frame type appropriate to installation.
12 13	Fixed blade (45 degree) core return and exhaust registers and grilles.
14 15	Register and grille sizes as shown on drawings and/or as scheduled.
16 17	White, baked enamel finish or powder coat finish, unless otherwise indicated.
18 19 20	Border to be 1-1/4-inch width with countersunk screw holes.
20	
21	FARTS - EAECOHON
22	ΙΝΣΤΑΙΙΑΤΙΟΝ
24	Install grilles, registers and diffusers as shown on drawings and according to manufacturer's instructions.
25	
26	Furnish diffusers with equalizing grids where it is not possible to maintain minimum 2 duct diameter straight duct into diffuser.
27	Equalizing grids shall consist of individually adjustable vanes designed for equalizing airflow into diffuser neck and providing
28 29	directional control of airflow.
30 31	Unless otherwise indicated, size ductwork drops to diffusers or grilles to match unit collar size.
32 33	Seal connections between ductwork drops and diffusers/grilles airtight.
34 35	Blank off unused portion of linear slot diffusers and linear bar diffusers and grilles.
36	Where diffusers, registers and grilles cannot be installed to avoid seeing inside duct, paint inside of duct with flat black paint to
37 38	reduce visibility.
39	CONSTRUCTION VERIFICATION ITEMS
40	Contractor is responsible for utilizing the construction verification checklists supplied by the third party commissioning agent in
41 42	accordance with the procedures defined for construction verification in Section 01 91 00.
43	
44	END OF SECTION

1 2 3	SECTION 23 41 00 PARTICULATE AIR FILTRATION
4 5	PART 1 - GENERAL
6 7	SCOPE
8	This section includes specifications for air system filters. Included are the following topics:
9 10	PART 1 - GENERAL
11	Scope
12	Related Work
13 14	Reference Reference Standards
15	Quality Assurance
16	Shop Drawings
17	Operation and Maintenance Data
18	Design Criteria
20	PART 2 - PRODUCTS
21	Manufacturers
22	Panel Filters
23	MERV 8 Filters
24	MERV 13 Filters
25 26	Housings for MERV 8 Filters
27	Side Access Filter Housings
28	
29	PART 3 - EXECUTION
30	Installation
31	Construction Verification Items
32 22	
34	Section 23.07.00 – HVAC Insulation
35	Section 23 73 13 – Modular Indoor Central-Station Air-Handling Units
36	Section 23 73 23 – Factory Fabricated Custom AHU
37	Section 23 82 00 – Heating and Cooling Air Terminal Units
38	DEFENSE
39 40	REFERENCE Applicable provisions of Division 1 govern work under this section
40	Applicable provisions of Division 1 govern work under this section.
42	REFERENCE STANDARDS
43	ASHRAE Standard 52
44 45	UL 181 – Standard for Factory-Made Air Ducts and Air Connectors
45 46	UL 586 – Standard for Air Eilter Units
47	
48	QUALITY ASSURANCE
49	Refer to Division 1, General Conditions, Equals and Substitutions.
50	
51	SHOP DRAWINGS Refer to Division 1. Constal Conditions, Submittals
52 53	Include data concerning dimensions, submittais.
54	
55	OPERATION AND MAINTENANCE DATA
56	All operations and maintenance data shall comply with the submission and content requirements specified under section
57	GENERAL REQUIREMENTS.
58 50	
60	Use UL Class 1 or Class 2 filters unless noted otherwise. (Reference applicable UL standard referenced)
61	
62	Efficiencies indicated in this section are based on ASHRAE Standard 52.
63	For motors have been colored to encrote against the resistance of districtives or excelled in this continu
04	Fail motors have been selected to operate against the resistance of dirty fliters as specified in this section.

PART 2 - PRODUCTS

MANUFACTURERS

American Air Filter, Barnebey-Cheney, Cambridge, Continental, Flanders, Camil-Farr, Mine Safety Appliances, Research Products, BLC Industries or approved equal.

PANEL FILTERS

Use 1" (or as scheduled) thick fiberglass blanket enclosed in a cardboard frame and reinforced with a perforated metal retainer on the air leaving side, Coat media with flameproof, non-volatile adhesive.

Media nominal rating to be 500 FPM face velocity, 0.15 inch WG initial resistance, 0.50 inches WG recommended final resistance. Average arrestance of filter media shall be 80%.

Provide filter holding frame.

MERV 8 FILTERS

Use 2" thick, pleated panels, 100% synthetic, self supported media fully bonded and sealed in cardboard frame.

Media nominal rating to be 500 FPM face velocity, 0.20 inch WG initial resistance, 1.0 inches WG recommended final resistance., Average arrestance of filter media shall be 90-92%

Furnish a side access housing or holding frame as scheduled.

Filter tracks shall be constructed to provide a minimum clearance of 2 inches between the pre-filter and final-filter media to facilitate the installation of static pressure tips.

MERV 13 FILTERS

Use 4" thick, pleated panels, 100% synthetic, self supported media fully bonded and sealed in cardboard frame.

Media nominal rating to be 500 FPM face velocity, 0.30 inch WG initial resistance, 1.0 inches WG recommended final resistance., Average arrestance of filter media shall be 90-92%

Furnish a side access housing or holding frame as scheduled.

Filter tracks shall be constructed to provide a minimum clearance of 2 inches between the pre-filter and final-filter media to facilitate the installation of static pressure tips.

9 HOUSINGS FOR PANEL FILTERS

Manufactured by air handling unit manufacturer, filter media manufacturer, or contractor fabricated. Casing and tracks constructed of galvanized or enameled steel or aluminum. Provide access to the media tracks from outside the casing so media and be readily changed.

4 HOUSINGS FOR MERV 8 FILTERS

Housing or holding frame to be of the same manufacturer as filter media or provided by the air handling unit manufacturer. Contractor fabricated housings or filter racks will not be accepted. Casing and tracks constructed of galvanized or enameled steel or aluminum. Provide access to the media tracks from outside the casing so media and be readily changed. Filter tracks shall be constructed to provide a minimum clearance of 2 inches between the pre-filter and final-filter media to facilitate the installation of static pressure tips.

1 SIDE ACCESS FILTER HOUSINGS

2 Galvanized steel housing with aluminum or galvanized steel filter mounting tracks. Mounting tracks and access doors to have 3 gaskets to minimize air bypass around the filters. Housing assembly to be suitable for use in modular air handling system.

Standard filter sections provided by air handling unit manufacturers may be used for MERV 11 and MERV 14 filters but will not be accepted for HEPA filters or activated carbon filters.

Insulate housings where adjacent duct or air handling apparatus is insulated. Insulation to be contained within a 2" thick, double wall steel panel and meet the requirements specified for adjacent duct or apparatus.

Furnish a door on each end of the housing to facilitate filter changing. Doors to be hinged and provided with lever handle latches to secure the door. Doors shall not be secured with nuts, bolts, wing nuts, or sheet metal screws.

Furnish housings for MERV 13 filters with a lever action sealing mechanism to secure media in tracks.

Include an integral prefilter track for installation of MERV 7 prefilters. Filter tracks shall be constructed to provide a minimum clearance of 2 inches between the pre-filter and final-filter media to facilitate the installation of static pressure tips.

PART 3 - EXECUTION

INSTALLATION

9 Where air handling equipment is to be used for temporary heating or ventilation of a facility, do not operate the equipment until 10 specified filter media has been installed. Contractor shall be responsible for maintaining the cleanliness of air handling apparatus 11 and air distribution systems during construction through regular inspection and changing of filter media throughout the 12 construction period. 13

14 Where air handling apparatus is used during the construction period, install new filter media prior to start of air balancing. 15 Additionally, deliver one new set of media to the owner prior to substantial completion. 16

17 Install units as shown on drawings and details according to manufacturer's instructions. 18

19 Reinforce filter holding frames per manufacturer's instructions. 20

21 Maintain necessary clearance for changing filters.

22 23 CONSTRUCTION VERIFICATION ITEMS

24 Contractor is responsible for utilizing the construction verification checklists supplied by the third party commissioning agent in 25 accordance with the procedures defined for construction verification in Section 01 91 00.

26

1 2

3

4 5 6

7 8

27 28

1	SECTION 23 57 00
2	HEAT EXCHANGERS FOR HVAC
3 1	
5	PART 1-GENERAL
6	
7	SCOPE
8	This section includes specifications for shell and tube heat exchangers and plate heat exchangers. Included are the following
9 10	topics:
10	PART 1 - GENERAL
12	Scope
13	Related Work
14	Reference
15	Reference Standards
16	Quality Assurance
12	Submitted Submitted States
19	PART 2 - PRODUCTS
20	Brazed Plate Heat Exchangers
21	PART 3 - EXECUTION
22	Installation
23	Brazed Plate Heat Exchangers
24 25	Construction Verification Items
25	Agency Training
27	
28	RELATED WORK
29	Section 23 21 13 - Hydronic Piping
30	
31 22	REFERENCE Applicable provisions of Division 1 govern work under this section
32	Applicable provisions of Division 1 govern work under this section.
34	REFERENCE STANDARDS
35	ASME Boiler and Pressure Vessel Code VIII - Rules for Construction of Pressure Vessels-Latest Edition.
36	
37	QUALITY ASSURANCE
38	Refer to division 1, General Conditions, Equals and Substitutions
39 40	SURMITTALS
41	Refer to division 1. General Conditions. Submittals.
42	
43	Include data concerning dimensions, capacities, and material of construction.
44	
45	OPERATION AND MAINTENANCE DATA
40 //7	All operations and maintenance data shall comply with the submission and content requirements specified under section
48	GENERAL REQUIREMENTS.
49	PART 2 - PRODUCTS
50	
51	
52	BRAZED PLATE HEAT EXCHANGERS
53	Manufactures: Alfa Laval, Bell & Gossett, DHT, TTT, Grundfos, AlC, Standard or approved equal.
55	Brazed plate type with 316 stainless steel corrugated channel plates, 316 stainless steel cover plates and threaded 316 stainless
56	steel pipe connections. Copper or nickel brazing material shall be used. Design pressure of 435 psig at 437 degrees F.
57	
58	Provide heat exchangers with capacities and operating characteristics indicated on drawings.
59	
0U 61	PART 3 - FYECHTION
62	FARTS - LALCOTION
63	
64	

1 **INSTALLATION** 2 Install units as s

Install units as shown on plans, as detailed, and according to manufacturer's installation instructions. Provide clearance around units as shown on the drawings and as recommended by the manufacturer for service access. Provide elbows, flanges and unions on piping to allow for servicing heat exchangers.

6 BRAZED PLATE HEAT EXCHANGERS

7 Mount as shown on the drawings. Maintain clearance around unit so it can be easily removed and replaced.

9 CONSTRUCTION VERIFICATION ITEMS

10 Contractor is responsible for utilizing the construction verification checklists supplied by the third party commissioning agent in 11 accordance with the procedures defined for construction verification in Section 01 91 00.

13 FUNCTIONAL PERFORMANCE TESTING

14 Contractor is responsible for utilizing the functional performance test forms supplied by the third party commissioning agent in 15 accordance with the procedures defined for functional performance testing in Section 01 91 00.

17 AGENCY TRAINING

18 All training provided for agency shall comply with the format, general content requirements and submission guidelines specified 19 under Section 01 91 00.

20

21

3

4

5

8

12

16

22

1		SECTION 23 64 23
2		WODOLAR HEAT RECOVERY CHILLERS
4		
5		PART 1 - GENERAL
7	SCOPE	
8	This section includes specification	s for modular heat recovery scroll water chillers. Included are the following topics:
9 10	PART ONE - GENERAL	
11	Scope	
12	Related Work	
13	Reference	
14	Reference Standards	
15 16	Quality Assurance Performance Requirement	ante
17	Operating Sound Pressu	re l evel
18	Submittals	
19	Operation and Mainten	ance Data
20	Delivery, Storage and Ha	andling
21	Warranty	
22		
23 24	Manufacturers	
25	General	
26	Manufactured Units	
27	Compressors and Refrig	eration
28	Evaporator and Conden	ser Heat Exchanger
29	VITUAL MOVEABLE ENDCA	p II (VIVIE II)
30	Controls	
32	Electrical	
33	Vibration Isolation	
34		
35	PART THREE - EXECUTION	
30 27	Installation	
38	Training	
39	Construction Verificatio	n Items
40	Functional Performance	Testing
41		
42	RELATED WORK	
43 44	Section 23 05 00 - Common Work	Results for HVAC
44	Section 23 05 48 - Vibration and S	eismic Controls for HVAC Pining and Equinment
46	Section 23 21 13 – Hydronic Pipin	g
47	Section 23 09 23 – Direct Digital C	ontrol for HVAC
48	Section 23 09 93 - Sequence of Op	perations for HVAC Controls
49		
50	REFERENCE	chall govern work under this section
51	Applicable provisions of Division	i shall govern work under this section.
53	REFERENCE STANDARDS	
54	ARI 550/590-2003	Centrifugal or Rotary Screw Water-Chilling Packages
55	ARI 575	Method of Measuring Machinery Sound Within an Equipment Space
56	ASHRAE 15	Safety Code for Mechanical Refrigeration
5/	ASHRAE 90.1	Energy Standard for Building except Low Rise Residential Buildings
20 59	ASIVIE SEU O NEMA MG1	Doner and Pressure vesser Coue Motors and Generators
60	UL 1995	Central Cooling Air Conditioners
61	UL 984	Safety Standards for Hermetic Compressors
62	COMM 45	Wisconsin Department of Commerce Mechanical Refrigeration Code
63		
64		

1 QUALITY ASSURANCE

2 Refer to Division 1, General Conditions, Equals and Substitutions.

4 Construct, test and rate chiller performance in accordance with ARI 550 with exceptions as noted in this specification.

Construct, install and operate chillers in accordance with ANSI/ASHRAE 15- Safety Code for Mechanical Refrigeration and
 COMM 45 Wisconsin Mechanical Refrigeration Code.

- 9 Construct and test chillers in accordance with ASME SEC 8.
- 11 Construct and label chillers in accordance with UL 1995.

12 13 **PERFORMANCE REQUIREMENTS**

14 Include the following performance documentation submitted along with the submittals.

16 **OPERATING SOUND PRESSURE LEVEL**

The unit shall operate at full load and all part load conditions without exceeding 85-dBA sound pressure level in the equipment room. If units do not meet the 85-dBA requirements, as measured in accordance with latest version ARI Standard 575, furnish all attenuation devices necessary to meet this requirement. The sound pressure levels in all octave bands must be met as scheduled for full load and part load conditions.

22 SUBMITTALS

23 Refer to Division 1, General Conditions, Submittals

24

21

3

5

8

10

15

Submit chiller shop drawings including the following information: specific manufacturer and model numbers, dimensional and weight data, required clearances, materials of construction, capacities and ratings, minimum load achievable without hot gas bypass, pressure ratings, refrigerant charge, pumpout refrigerant storage capacity, component information, assembly information, size and location of piping connections, electrical connections, wiring diagrams, motor information (ref. 23 05 13), surfaces requiring insulation, SqFt of surface insulation, sound pressure levels in all octave bands at 25%/50%/75%/100% load, information for all specialties and accessories.

- Include an ARI approved chiller selection method for the specified refrigerants. Verification of date and version of computer program selection or catalog is available through the Vice-President, Engineering, ARI (703) 524-8800.
- Indicate ASME construction and stamping of pressure vessels or unit physical characteristics and ASME code section and paragraph references that allow non-compliance with this construction and stamping requirement.

37

31

Submit manufacturer's installation and start-up instructions, maintenance data, troubleshooting guide, parts lists, controls and accessories.

- 40
- 41 At substantial completion, submit warranty certificate and copy of start-up report. 42

43 OPERATION AND MAINTENANCE DATA

44 All operations and maintenance data shall comply with the submission and content requirements specified under section 45 GENERAL REQUIREMENTS.

46 47 DELIVERY, STORAGE AND HANDLING

48 Comply with manufacturer's installation instructions for rigging, unloading, and transporting units.

50 Protect units from physical damage. Leave factory-shipping covers in place until installation.

51 52 WARRANTY

- 53 Provide a one year all-inclusive warranty to begin upon acceptance of project by owner.
- Provide an additional four (4) year material and labor warranty extension for compressor motor, compressor assembly and unit
 controls.
- 57

49

58 59

60

PART 2 - PRODUCTS

61 MANUFACTURERS

GENERAL

- 62 Multistack.
- 63 64

with int	ler shall be a Multistack VME II patent pending geothermal heat recovery, heat recovery chiller. It shall be equipped egral valving that allows the assembly to serve the following functions:		
•	Simultaneous Heating and Cooling Mode – Chiller/Heater assembly must be capable of varying the flow rate on the evaporator and condenser sides of the modules to maintain heating and cooling water set points simultaneously. Simultaneous loads must be satisfied with a single compression cycle and cannot use the source/sink solution as the means of energy transfer. Systems that require double compression to satisfy simultaneous loads are not acceptable.		
•	Cooling Dominant Mode – Chiller/Heater must be able to reject cooling dominant load to the source/sink. Cooling dominant modules must be capable of running at a lower head pressure than simultaneous modules to minimize power consumption.		
•	Heating Dominant Mode – Chiller/Heater must be able to satisfy heating dominant load by extracting heat from the source/sink solution. Heating dominant modules must be capable of running at optimal suction pressure to minimize power consumption.		
•	Packaged System Shall Be Reversing Valve Free Design – Chiller/Heater must be reversing valve free and optimize heat transfer in all control modes.		
•	Source/Sink Water Connections – Chiller/Heater must allow geothermal loop water to enter both the evaporator and condenser side of the machine.		
System heating efficiend	shall be configured to allow modules to run in simultaneous DHRC™ mode, dominant cooling mode, and dominant mode. The Chiller/Heater must be capable of allowing modules to run in multiple modes at the same time to optimize cy.		
Provide	factory assembled and tested, packaged, heat recovery liquid chiller consisting of modular dual scroll compressor(s),		
compressor motor, condenser, evaporator, refrigeration accessories, instrument and control panel, gages and indicating lights,			
auxilialy			
Acceptable refrigerant is R-410A.			
Firmly a compread	attach metal nameplates to major components indicating the name of the manufacturer, unit model number, ssor/condenser/cooler type, refrigerant used, pounds of refrigerant needed for normal operation, operating pressures, : serial number.		
Module	s shall be provided with acoustical panels.		
COMPR	ESSORS AND REFRIGERATION		
Compre compre	ssor assemblies shall be run tested at the factory. Vibration shall not exceed 1.0 mil peak to peak. Over-speed test ssor impeller(s) to not less than 20% above operating conditions.		
Chiller r valve, li and two	nust be supplied with dual refrigeration circuits and each circuit complete with externally equalized thermal expansion quid line solenoid to prevent liquid migration during the off cycle, liquid line filter dryer, sight glass moisture indicator (2) hermetic scroll compressors.		
Each mo	odule have (2) stages of capacity at a minimum.		
Scroll co compre	ompressors must each be supplied with crankcase heaters, internal thermal protection, rubber isolation pads between ssor and frame, have two steps of capacity control by cycling of compressors, 100%, 50% and off.		
EVAPOF	RATOR AND CONDENSER HEAT EXCHANGERS		
Evapora standar	tor and condenser shall be dual circuit brazed plate heat exchanger designed and constructed to ASME and UL ds.		
Host or	changers shall be constructed of conner brazed 216 stainless steel plates. Heat exchangers shall be mounted below the		
compre up.	ssor, to eliminate the effect of migration of refrigerant to the cold evaporator with consequent liquid slugging on start-		
чp.			

1 The motorized actuators shall be NEMA 4X rated with easily visible position indicator and internal thermal motor overload 2 protection. Valves shall be fast acting type with a maximum stroke time (full closed to full open) of 15 seconds. Load side 3 valves shall modulate to maintain modular leaving load temperatures.

4 5

6 7

8

When heat exchangers are using sink/source due to unequal heating/cooling duty, master controller shall modulate valve to provide minimum required head pressure control in order to maximize efficiency of those Chiller/Heater modules and to provide equipment protection.

All valves must be installed such that proper piping practices are observed, including proper distances before and after elbows.
 Contractor shall provide required butterfly valves and actuators.

11

Each inlet water header shall incorporate a built-in 30 mesh inline strainer system to prevent heat exchanger fouling and
 accommodate 100% flow filtration with a minimum surface area of 475 square inches per module. This filtration is in addition
 to a field mounted Y-strainer in the chilled water loop piping.

15

21

Each module shall include supply and return mains for both load and source-sink water. Grooved end connections are provided
 for interconnection with Victaulic type couplings. Water Mains shall be installed such that they are beneath any power or
 control wiring so as to insure for safe operation in the event of any condensation or piping leaks.

20 Provide thermometer and pressure wells for temperature controller and low temperature cutout.

22 VIRTUAL MOVEABLE ENDCAP II (VME II)

The heat recovery chiller shall be designed for simultaneous variable heating and cooling capacity. The VME II valve module shall contain fast-acting motorized butterfly valves that open/close on a command from the central control system (ball valves are not acceptable). The motorized actuators shall be NEMA 4X rated with easily visible position indicators and internal thermal motor overload protection. Valves shall be fast acting type with a maximum stroke time (full closed to full open) of 30 seconds. Valve modules shall be built into pre-engineered headers and powered by the heat recovery chiller bussbar. VME II valves shall be Victaulic grooved connections.

30 INSULATION

3/4" thick, flexible closed cell elastomeric foam insulation; minimum nominal density of 5.5 lbs. per cu. ft., thermal conductivity 32 of not more than 0.27 at 75 °F, minimum compressive strength of 4.5 psi at 25% deformation, maximum water vapor 33 transmission of 0.17 perm inch, maximum water absorption of 6% by weight, rated for service range of -20 °F to 180 °F. 34

35 Factory insulate the following:

- Evaporator and condenser heat exchangers
- Motor housing (hermetic compressors)
- Motor cooling lines (hermetic compressors)
- All lines and surfaces 65°F or colder

40 41 **CONTROLS**

Provide fully automatic microprocessor controller in a lockable steel control panel containing solid state chiller operating and safety controls for each chiller. Factory mount, wire and test controls on chiller. Operating setpoints and diagnostic procedures to be programmed through a color-coded, tactile-feel keypad. Provide an alphanumeric display showing all system parameters, safety and cycle shutdowns in the English language with numeric data in English units. Safety and cycle shutdown display to consist of date, time, cause of shutdown, and type of restart required.

47

53

36

37

38

39

Provide master chiller controller, of same manufacturer of chillers, for sequencing, controlling and managing multiple modular
 chillers in a single plant.

51 The master controller shall sequence the operation of the of the various compressors, VME Isolating Valves, and Heat 52 Exchanger Variable Flow Valves to maximize efficiency and minimize system energy usage.

Provide the following safety controls arranged so that activation of any one will shut down the respective chiller or plant and require a manual reset:

- 56 57 Low refrigerant pressure
- 58 Loss of flow through the source/sink heat exchanger
- 59 Loss of flow through the load (cooling and/or heating) heat exchanger
- 60 High refrigerant pressure
- 61 High compressor motor temperature
- 62 Low suction gas temperature
- 63 Low leaving water temperature
- 64
| 1
2 | The Master Controller to be powered from the main Chiller/Heater power. Master controller will monitor and report the following on each refrigeration system: |
|----------|---|
| 3 | |
| 4 | Discharge Pressure Fault |
| 5 | Suction Pressure Fault |
| 6 | Suction Temperature |
| 7 | Load Leaving Water Temperature |
| 8 | Source-Sink Leaving Water Temperature |
| 9
10 | The Master Centreller shall monitor and report the following system parameters: |
| 11 | The Master controller shall monitor and report the following system parameters. |
| 12 | Cooling Load Water Entering and Leaving Temperature |
| 13 | Heating Load Water Entering and Leaving Temperature |
| 14 | Source-Sink Water Entering and Leaving Temperature |
| 15 | Load Water (both heating and cooling) and Source-Sink Water Flow |
| 16 | |
| 17 | Individual monitoring of leaving water temperatures from each refrigeration system shall be programmed to protect against |
| 18 | heat exchanger freeze-up. |
| 19 | The central system shall evaluate the water temperatures of the besting and cooling systems to access the required conscitutef |
| 20 | and cycle comproscers of the Chiller/Heater Medules, onen/close VME Isolation Values, & medulate Heat Exchanger |
| 21 | Variable Flow Valves to meet load requirements ontimize efficiency, minimize system energy usage and equalize compressor |
| 23 | run times. |
| 24 | |
| 25 | Chiller/Heater shall have a single point power connection and external inputs and outputs to be compatible with the building |
| 26 | management system. Hardwire Inputs/Outputs include: |
| 27 | |
| 28 | Remote Start/Stop |
| 29 | General Alarm |
| 30 | Cooling Load Limit |
| 31
27 | Heating Load Limit |
| 32 | Heating Load Reset |
| 33 | |
| 35 | Failure of DHRC Chiller/Heater/Heater to start or Chiller/Heater shutdown due to any of the above safety cutouts shall be |
| 36 | enunciated by display of the appropriate diagnostic description at the unit control panel. This annunciation will be in plain |
| 37 | English. Alphanumeric codes shall be unacceptable. |
| 38 | |
| 39 | The DHRC Chiller/Heater/Heater shall be furnished with a Master Controller as an integral portion of the Chiller/Heater control |
| 40 | circuitry to provide the following functions: |
| 41 | Dury ide substratic Chilles/Heater shutdering during pariods when the land level degrades heles, the assured exception |
| 42 | Provide automatic Chiller/Heater shutdown during periods when the load level decreases below the normal operating |
| 45
// | requirements of the chiller/Heater. Opon an increase in load, the chiller/Heater shall automatically restart. |
| 45 | Provisions for connection to automatically enable the Chiller/Heater from a remote energy management system. |
| 46 | |
| 47 | The control panel shall provide alphanumeric display showing all system parameters in plain English language with |
| 48 | numeric data in English units. |
| 49 | |
| 50 | Normal Chiller/Heater Operation |
| 51 | |
| 52 | When DHRC Chiller/Heater/Heater is enabled, the factory supplied Master Controller modulates the Chiller/Heater |
| 55 | nearing and cooling capacity from minimum to maximum as required by building load. |
| 55 | The DHRC Chiller/Heater/Heater control system shall respond to Entering Water Temperature and will have an integral |
| 56 | reset based on entering water temperature to provide for efficient operation at part-load conditions. |
| 57 | |
| 58 | Power Phase Monitor |
| 59 | |
| 60 | Provide a Power Phase Monitor on the incoming power supply to the Chiller/Heater. This device shall prevent the |
| 61 | Chiller/Heater from operating during periods when the incoming power is unsuitable for proper operation. |
| 62 | The Dewer Dhace Meniter shall provide protection against the following and ditions: |
| 03
64 | The Fower Phase Monitor shall provide protection against the following conditions: |
| 04 | Low voltage (blown-out) |

1 2	Phase Rotation Loss of Phase
3 4	Phase Imbalance
5	The Chiller/Heater shall be capable of communicating the above points with the Building Automation System via an Interoperability Web Portal. BacNet, Lonmark, or Modbus available.
7 8 9	Additional points shall include:
10	Chiller/Heater leaving chilled water temperature
11	Chiller/Heater leaving hot water temperature
12	Chiller/Heater percent cooling capacity
13	Chiller/Heater percent heating capacity Medule level leaving condenser temperature
15	Module level leaving condenser temperature
16	Individual Compressor Status On/Off
17	Condenser VME valves Open/Close status
18	Evaporator VME valve Open/Close status
19	The master chiller centraller shall convert information into the BACnet/IB protocol, or format compatible with the building
20	direct digital control system (DDC) as specified in Section 23.09.23. This output shall be through the appropriate interface port
22	capable of two-way communication with the building DDC system. Coordinate with the DDC contractor so that the data port
23	connection provided at the chiller shall not require any additional intermediate gateway or media conversion devices to provide
24	throughput of data. No additional labor by the DDC contractor to integrate the chiller data points to the DDC system shall be
25	required other than to make the communication trunk connection and program the points at the DDC workstation.
26	The shills reside a setuple shall allow the residues to have ID addresses assigned within a DUCD. Manufacturer compliad static
27 28	In exhibit master controller shall allow the modules to have in addresses assigned utilizing DHCP. Manufacturer supplied static
29	
30	ELECTRICAL
31	Modular chiller section groups shall have a single point electrical connection. Entire VME II package shall be equipped with a
32	bussbar electrical distribution system to allow for single point power.
33	Defense Durwing askedulas for skiller conting and
34 35	Refer to Drawing schedules for chiller section groups.
36	The pre-engineered system shall also incorporate individual module isolation circuit breakers for full redundancy and ability of a
37 38	module to be taken off-line for repair while the rest of the modules continue to operate.
39 40	Future chiller modules and VME II shall be provided with their own electrical power connection.
41	VIBRATION ISOLATION
42 43	The chiller supplier shall furnish refrigeration machine vibration isolation in accordance with 23.05.48 for the installation by the mechanical contractor.
45	
46	PART 3 - EXECUTION
47	
48	INSTALLATION
49	Install chillers in accordance with manufacturer's installation instructions.
50 51	Chillers shall be factory assembled, tested, and shinned to the job site. The shiller manufacturer is responsible for unleading at
51 52 53	the job site and the Mechanical Contractor is responsible for final setting and installation.
54	STARTUP
55	Include the service of a factory-trained technician/mechanic employed by the chiller manufacturer for the initial startup, 6
56	month maintenance review, and 1 year maintenance review. Accomplish initial startup before State acceptance of the
57	installation.
58 50	Eurnish a startup log to the Owner's operating personnel with a copy to the state construction representative for this project
60	Document each subsequent startup or shutdown procedure and send report to Owner's operating personnel. Demonstrate the
61	following items have been accomplished:
62	
63 64	1. Inspect/clean evaporator and condenser heat exchangers.
5-	

1 2 3		a. <u>6 Month & 1 Year Review</u> : Visually inspect for corrosion, pitting, erosion and general appearance. Chemical cleaning will not be allowed.
5 4 5	2.	Perform leak test on fabricated compressor, vessel and piping joints after the system has been serviced and closed.
6 7 8 9 10		 Initial startup: Follow manufacturer's instructions with respect to evacuation, charging, positive pressure and/or vacuum testing. Pressure/vacuum testing to be in accordance with manufacturer's instructions. Perform any repairs necessary to obtain a successful pressure test. Do not operate chiller until it is successfully pressure tested. Use nitrogen and suitable refrigerant for pressure test unless manufacturer's instructions require otherwise.
12 13		b. <u>6 Month & 1 Year Review</u> : Electronically leak test all fabricated compressor and vessel joints in accordance with manufacturer's instructions and perform any repairs necessary to obtain a successful test.
15 16	3.	Lubrication system
17 18 19		a. <u>Initial startup</u> : Charge unit with oil in accordance with manufacturer's instructions. Energize oil sump heater and verify thermostat setting per manufacturer's specification.
20 21 22 23 24 25		b. <u>6 Month & 1 Year Review</u> : Remove lubrication charge from modules. Visually inspect for color, dirt, sludge, and burnt or acidic condition. Test oil in the chiller manufacturer's laboratory or a laboratory approved by the manufacturer to determine condition of oil and notify owner of the results. Clean lubrication system if visual and/or laboratory results indicate necessary. Recharge with new oil per manufacturer's specifications and replace oil filter elements.
26 27	4.	Filters and strainers.
28 29 30		a. <u>6 Month & 1 Year Review</u> : Remove and replace all oil and refrigerant filters, strainers and filter-drier cores. Use acid/moisture type replacement filter-drier cores.
31 32	5.	Energize oil sump heater and verify thermostat setting is per manufacturer's instructions at each inspection.
33	6.	Electrical
34 35 36		Tighten all starter electrical power connections and all control terminations at each inspection.
37 38 39 40		Check all contactors at each inspection for proper mechanical linkage, freedom of operation and contact surfaces for pitting, corrosion and spring tension. Clean all contact surfaces as required; notify owner if replacement is recommended.
41 42 43 44		Megger test and record all compressor and oil pump motor insulation readings at initial startup and each succeeding inspection. Compare findings to previous readings and make recommendations on any preventative maintenance required.
45 46 47		Visually inspect and clean all components including resistor banks, disconnects, fuse holders, arc chutes, ammeters, voltmeters, watt-hour meters, dash-pots, etc.
48 49 50	8.	At initial startup and whenever refrigerant is transferred from a storage device to the chiller, record date and pounds of refrigerant in machine.
51 52	9.	Clean and touchup paint unit as required for protection.
53 54	10.	Repair or replace damaged insulation caused by service/repair/maintenance work.
55 56	11.	Give any used compressor oil to owner; owner will make arrangements for proper disposal.
57 58 59	12.	At the initial startup and each maintenance review, check all safety and operating controls, log all pertinent parameters of the unit, including but not limited to the following: a. Refrigerant pressure in cooler
60 61		b. Saturated refrigerant temperature in cooler
62		 water inject and outlet temperatures in cooler and condenser Water side pressure drop in cooler and condenser
63		e. Flow rate in gallons per minute in cooler and condenser
C 4		f. Bearing temperatures

1	g.	Oil sump temperature
2	ĥ.	Oil pressure
3	i.	Motor voltage and amperage in each phase
4	j.	Purge count and purge unit operating hours, if applicable
5	k.	Purge condensing pressure, if applicable
6	I.	Starter transition time, actual as measured
7		
8	TRAINING	

9 Contractor to provide factory authorized representative and/or field personnel knowledgeable with the operations, 10 maintenance and troubleshooting of the system and/or components defined within this section for a minimum period of 8 11 hours.

12

13 CONSTRUCTION VERIFICATION ITEMS

Contractor is responsible for utilizing the construction verification checklists supplied by the third party commissioning agent in
 accordance with the procedures defined for construction verification in Section 01 91 00.

17 FUNCTIONAL PERFORMANCE TESTING

18 Contractor is responsible for utilizing the functional performance test forms supplied by the third party commissioning agent in 19 accordance with the procedures defined for functional performance testing in Section 01 91 00.

20

21 22

1 2 3	SECTION 23 73 12 AIR HANDLING UNIT COILS
4 5 6	PART1 - GENERAL
6 7 8 9	SCOPE This section contains specifications for coils used in all indoor central station air handling units. Included are the following topics:
10 11	PART 1 - GENERAL
12	Scope
13	Related Work
14	Reference
15	Reference Standards
16	Quality Assurance
17 18	Submittais
10	Design Criteria
20	
21	PART 2 - PRODUCTS
22	Manufacturers
23	Hot Water Coils
24	Chilled Water Coils
25	
20 27	PART 3 - EXECUTION Hot Water Coils
27 28	Chilled Water Coils
29	Construction Verification Items
30	
31	RELATED WORK
32	Section 23 73 13 - Modular Indoor Central-Station Air-Handling Units
33	
34 25	REFERENCE Applicable provisions of Division 1 govern work under this section
35 36	Applicable provisions of Division 1 govern work under this section.
37	REFERENCE STANDARDS
38	ARI 410 Forced Circulation Air-Cooling and Air-Heating Coils
39	
40	QUALITY ASSURANCE
41	Refer to division 1, General Requirements, Equals and Substitutions
42	
43 44	SUBMITTALS Defer to division 1. Concercl Requirements, Submittals
44 15	Refer to division 1, General Requirements, Submittais.
45 46	Including data concerning dimensions capacities flow rate pressure drop materials of construction ratings weights and
47	appropriate identification at the same time that the air handling equipment in which the coils will be located are submitted.
48	
49	OPERATION AND MAINTENANCE DATA
50	All operations and maintenance data shall comply with the submission and content requirements specified under section
51	GENERAL REQUIREMENTS.
52	
53 54	DESIGN CRITERIA Select coil sizes canacities configuration and operating characteristics as shown on the plans and/or as scheduled. Coil
55 56	capacity ratings shall be ARI 410 certified.
57 50	
50 59	PART 2 - PRODUCTS
60	MANUFACTURERS
61	Aerofin or pre-approved equal
62	

HOT WATER COILS

Use galvanized steel casing, end supports, top channel, and bottom channel to produce a rigid frame with allowance for expansion and contraction of the finned tube section.

Construct coils of 0.025 inch tube wall seamless copper tubes of 5/8 inch maximum outside diameter with 0.008" thick aluminum fins and suitable for working pressures to 125 psig and temperatures to 250°F. Coil fins may be the continuous serpentine or plate fin type.

Coil headers may be constructed of cast iron, steel, or seamless copper. Where cast iron headers are used, expand tubes into the headers. Where steel or copper headers are used braze tubes to header.

Provide coils with bronze spring turbulators where required to provide the capacities indicated.

CHILLED WATER COILS

Use galvanized steel casing, end supports, top channel, and bottom channel to produce a rigid frame with allowance for expansion and contraction of the finned tube section.

Construct coils of 0.025 inch tube wall seamless copper tubes of 5/8 inch maximum outside diameter with 0.008" thick aluminum fins suitable for working pressures to 200 psig. Coil fins may be the continuous serpentine or plate fin type.

Coil headers may be constructed of cast iron, steel, or seamless copper. Where cast iron headers are used, expand tubes into the headers. Where steel or copper headers are used braze tubes to header.

Coils shall be drainable type with drain and vent plugs for each header.

PART 3 - EXECUTION

HOT WATER COILS

Install in central station air handling unit casings or on structural support frames for field erected units, making allowance for pitching as recommended by the manufacturer. Mount coils in field erected units to allow for individual removal.

Comb bent or crushed fins after installation. Clean dust and debris from each coil to ensure its cleanliness.

Install a separate air vent and drain valve for each coil header in such a manner that the vent and drain valves are located outside of air handling unit casing. Provide offsets in piping to facilitate coil removal.

Unless otherwise specified, pipe coils for counter flow arrangement.

CHILLED WATER COILS

Install in central station air handling unit casings or on structural support frames for field erected units, making allowance for pitching as recommended by the manufacturer.

Mount coils in field erected units to allow individual removal.

Comb bent or crushed fins after installation. Clean dust and debris from each coil to ensure its cleanliness.

Install a separate air vent and drain valve for each coil header in such a manner that the vent and drain valves are located outside of air handling unit casing. Provide offsets in piping to facilitate coil removal. Unless otherwise specified, pipe coils for counter flow arrangement.

Where coils are installed in ductwork or field erected air handling units, provide a 1-1/2" deep 18 gauge welded stainless steel drain pan as an integral part of the duct or at coil support.

Install condensate drain trap with proper depth from each cooling coil condensate drain to the nearest drain location.

CONSTRUCTION VERIFICATION ITEMS

Contractor is responsible for utilizing the construction verification checklists supplied by the third party commissioning agent in accordance with the procedures defined for construction verification in Section 01 91 00.

1	SECTION 23 73 13
2	MODULAR INDOOR CENTRAL-STATION AIR-HANDLING UNITS
3	
4	
5	PART 1 - GENERAL
6	
7	SCOPE
8 9	This section includes specifications for indoor central station package air handling units. Included are the following topics:
10	PART 1 - GENERAL
11	Scope
12	Related Work
13	Reference
14	Reference Standards
15	Quality Assurance
16	Submittals
17	Operation and Maintenance Data
18	Decign Criteria
10	Design circente
20	
20	Manufacturers
21	
22	
23	Easting
24	Fan Joculius
25	Coll Sections
20	
27	Access Sactions
20	Access Sections
20	
31	Installation
32	Construction Verification Items
32	Functional Performance Testing
34	
35	RELATED WORK
36	Section 23.05.13 - Common Motor Requirements for HVAC Equipment
37	Section 23 05 10 - Variable Frequency Drives
38	Section 23 05 28 - Vihitation and Seismic Controls for HVAC Pining and Equipment
30	Section 23 31 00 - HVAC Ducts and Casings
40	Section 23 41 00 - Particulate Air Eiltration
40 //1	Section 23 41 00 - Air Durt Accessories
41	Section 23 53 00 - Air Buck Accessiones
42	
43 AA	REEFRENCE
44 45	Applicable provisions of Division 1 govern work under this section
45	Applicable provisions of Division 1 govern work under this section.
40	REFERENCE STANDARDS
47 //8	ARI (20) (Jatest edition) Standard for Central Station Air Handling Units
40	NEPA 90A Standard for Installation of Air Conditioning and Ventilation Systems
50	
51	
52	Refer to Division 1. General Conditions. Equals and Substitutions
52	Refer to Division 1, General conditions, Equals and Substitutions.
57	SLIBMITTALS
55	Pater to Division 1. General Conditions. Submittals
56	Submit shon drawings including the following information: specific manufacturer and model numbers, submittal equipment
50	identification corresponding to project drawing sound schedules unit disposicional and weight data, materials of construction
58	canacities and ratings fan curves fan tyne drive and motor information vibration isolation coil performance data sound nowe
50	levels filter information information for all accessories
55	הביכוש, וווכר ווויסרוומנוסוו, ווויסרוומנוסוו וסר מו מכנכשטווכש.
61	ΟΡΕΡΑΤΙΩΝ ΑΝΟ ΜΑΙΝΤΕΝΑΝCE ΠΑΤΑ
62	All operations and maintenance data shall comply with the submission and content requirements specified under section
63	GENERAL REGULIREMENTS
64	
v T	

DESIGN CRITERIA

1 2

3

4 5

6 7

8 9

14

16 17 18

19

21

29

39

Furnish factory fabricated modular indoor central-station air handling units complete meeting the configuration shown on drawings and/or as scheduled.

Units to be tested, rated and certified in accordance with ARI Standard 430 and bear ARI certification label.

All material shall meet NFPA 90A flame spread and smoke develop rating requirements.

Any revisions made by the Contractor to the inlet and outlet ductwork conditions from that shown on the drawings shall not increase system effect and/or static pressure and shall not decrease mixing efficiencies.

PART 2 - PRODUCTS

15 MANUFACTURERS

Daikin, Carrier, or Trane,

Manufacturers must fit within the mechanical rooms while providing the required unit sections and unit maintenance clearances. Units with dimensions that do not fit within the mechanical rooms, and do not maintain required maintenance accesses will not 20 be accepted.

CASING 22

23 WALL/ROOF CONSTRUCTION

Construct walls and roof from 2"thick double wall panel assemblies. Panels shall be injected with polyurethane foam insulation 24 25 and shall have a minimum thermal conductivity (R) of at least 12.5. The outer shell shall be constructed of solid G90 galvanized steel with baked enamel or mill galvanized finish or G40 galvanized steel with gardobond finish. The inner liner shall be 26 27 constructed of solid G90 galvanized steel or G40 galvanized steel with gardobond finish. Panels shall be gasketed with permanently applied bulb-type gaskets and able to be removed without affecting the integrity of casing structure. 28

30 Under 55°F supply air temperature and design conditions on the exterior of the unit of 91°F dry bulb and 74°F wet bulb, condensation shall not form on the casing exterior. The AHU manufacturer shall provide tested casing thermal performance for 31 32 the scheduled supply air temperature plotted on a psychrometric chart. The design condition on the exterior of the unit shall also be plotted on the chart. If tested casing thermal data is not available, AHU manufacturer shall provide, in writing, a guarantee 33 against condensation forming on the unit exterior at the stated design conditions above. The guarantee shall note that the AHU 34 35 manufacturer will cover all expenses associated with modifying or replacing units should external condensate form on them. 36

37 Wall/Roof panel deflection shall not exceed L/240 ratio at a maximum +/- 5 inches of static pressure. Deflection shall be measured 38 at the midpoint of the panel.

40 FLOOR CONSTRUCTION

Construct floors from 2"thick double wall panel assemblies. Panels shall be injected with polyurethane foam insulation and shall 41 42 have a minimum thermal conductivity (R) of at least 12.5. The outer shell shall be constructed of solid G90 galvanized steel with baked enamel or mill galvanized finish or G40 galvanized steel with gardobond finish. The inner liner shall be constructed of solid 43 44 G90 galvanized steel or G40 galvanized steel with gardobond finish. Panels shall be gasketed with permanently applied bulb-type 45 gaskets. 46

47 Under 55°F supply air temperature and design conditions on the exterior of the unit of 91°F dry bulb and 74°F wet bulb, condensation shall not form on the casing exterior. The AHU manufacturer shall provide tested casing thermal performance for 48 the scheduled supply air temperature plotted on a psychrometric chart. The design condition on the exterior of the unit shall also 49 be plotted on the chart. If tested casing thermal data is not available, AHU manufacturer shall provide, in writing, a guarantee 50 against condensation forming on the unit exterior at the stated design conditions above. The guarantee shall note that the AHU 51 manufacturer will cover all expenses associated with modifying or replacing units should external condensate form on them. 52

Floor panel deflection shall not exceed L/240 ratio based upon a 300 lb concentrated load at the mid-span of the panel.

56 A full perimeter base rail shall be installed at each air handling unit. The base rail shall be constructed from a minimum of 16 gauge G90 galvanized steel and have a height required for proper condensate trapping (air handlers will also be placed on a 4" 57 concrete equipment pad). Panels shall be able to be removed without affecting the integrity of casing structure. 58 59

LEAKAGE RATE

Leakage rate shall not exceed 1% of the total system air quantity when subjected to +/- 5" static pressure.

60

53 54

55

WARNER PARK COMM AND REC CENTER CONTRACT #: 9502 MUNIS #: 17196

1 CASING PENETRATIONS

Install sealing collars to the interior and exterior of each penetration to prevent air leakage where coil piping, humidifier piping,
 air vents, drain piping, and electrical conduits penetrate air handling unit casing. Silicone sealants and duct sealants are not

acceptable to seal pipe penetrations of the air handling unit casing.

5

6 Duct sealant and/or gaskets as indicated in specification section 23 31 00 may be utilized to seal duct connections to the air 7 handling unit casing. Silicone sealants are not acceptable.

8 9 ACCESS DOORS

Access doors shall be double wall, of same construction and thickness as casing, hinged, continuously gasketed with bulb type gaskets and reinforced nylon handles with cam type latches. Provide access doors on both sides of casing for fan sections, access sections, air to air energy recovery sections, filter sections, damper sections, air blender sections and upstream and downstream of every coil and humidifier.

14

19

26

29

32

15 FAN SECTIONS

16 Double width, double inlet, housed centrifugal type or single width single inlet plenum type, statically and dynamically balanced 17 fans. For variable speed applications, fan shall be dynamically balanced through entire range of operation. Fan wheels shall be 18 backward inclined, forward curved or airfoil type as specified or required by performance characteristics.

Each fan and motor combination shall be capable of delivering 110% of air quantity scheduled at scheduled static pressure. The motor furnished with the fan shall not operate into the motor service factor when operating under these conditions.

23 Fans to be fastened to hollow or solid steel shafts and designed for continuous operation at maximum rated static pressure.

Fan bearings shall be self-aligning, pillow block, regreasable ball type selected for a minimum average L-50 life of 200,000 hours.

Furnish extended grease lines from bearings to allow servicing without entering the unit. Grease lines can be terminated within the unit as long as they are able to be easily serviced by opening the access door.

Furnish variable pitch sheaves for drives 3 hp and smaller, fixed pitch sheaves for drives 5 hp and larger. Drives shall be designed for 150% of motor rating. Furnish OSHA approved belt guards for all fans.

Consider drive efficiency in motor selection according to manufacturer's published recommendation or according to AMCA
 Publication 203, Appendix L.

Furnish a metal access guard at the access door of all plenum fan sections. A wheel guard may be substituted if a metal access guard is not available from the manufacturer.

38

42

Fan, drive and motor assembly shall be mounted inside fan casing section and integrally isolated within unit. Vibration isolation
 shall be in compliance with section 23 05 48. Provide flexible connection and thrust restraints at fan discharge connection to
 casing.

Furnish a label inside the fan section that identifies the specifications of the v-belt drive kit. Include motor sheave, drive sheave
 and belt data.

- 45
- 46 Fan motors shall be provided in accordance with section 23 05 13.

47 48 COIL SECTIONS

- 49 Coils shall be provided in accordance with section 23 73 12.
- 50

Air handling unit coils mounted in casing shall be accessible for removal from either side of unit casing without disturbing adjacent
 sections.

54 Entire coil frame, headers and U-bends shall be enclosed within air handling unit casing. Extend coil piping connections, air vent 55 and drain connections to exterior of casing.

- Support coils along entire length within casing and pitch coil for proper drainage.
- 59 Blank off space between coil frames and air handling unit casing.

60

58

Fabricate cooling coil drain pans from type 304 stainless steel. Install a drain pan under each cooling coil. Extend drain pans the entire width of each coil, including the header, and from the upstream face of each coil to a distance ½ of the vertical coil height of the bottom coil or 6", whichever is greater, downstream from the downstream face. Pitch drain pans in two directions towards the outlet. Pipe drain pans individually down to the drain pan below using a minimum 1" type 304 stainless steel piping. The bottom drain pan shall be piped to the exterior of the unit base using a minimum of 1.25" type 304 stainless steel piping.

FILTER SECTIONS

1 2

3

4

5

6 7

8

9

10 11 12

13

14 15

16 17 Filter box section may be furnished by air handling unit manufacturer in accordance with specification requirements of section 23 41 00. Provide static pressure tips that are arranged to prevent damage to the filter elements during replacement. Provide minimum 2" gap between final and prefilters for static pressure probes.

ACCESS SECTIONS

Provide access sections where shown on drawings.

PART 3 – EXECUTION

18 INSTALLATION

19 Install all air handling units and accessories as indicated on drawings and/or as scheduled and according to manufacturer's 20 installation instructions.

21 Mount units at appropriate height above floor to insure proper condensate trap depth and condensate drainage. 22

Install air-handling unit to provide for adequate service access. Coordinate with other trades to assure air handling unit does not
 infringe upon access or service clearances of other equipment.

Lubricate fan bearings. Verify fan isolators have proper deflection.

Upon completion of installation of air handling units, start-up and operate equipment to demonstrate capability and compliance
 with requirements. Field correct malfunctioning components, then retest to demonstrate compliance.

31 Furnish one spare set of fan drive belts and three reinforced nylon access door handles.

33 CONSTRUCTION VERIFICATION ITEMS

Contractor is responsible for utilizing the construction verification checklists supplied by the third party commissioning agent in accordance with the procedures defined for construction verification in Section 01 91 00.

37 FUNCTIONAL PERFORMANCE TESTING

Contractor is responsible for utilizing the functional performance test forms supplied by the third party commissioning agent in accordance with the procedures defined for functional performance testing in Section 01 91 00.

40 41 42

30

32

36

1	SECTION 23 82 00 HEATING AND COOLING TERMINAL LINITS
3	
4 5	PART 1 - GENERAL
6 7	SCOPE
8 9	This section includes specification for heating and cooling terminal equipment using water and/or steam as the source. Included are the following topics:
10 11	PART 1 - GENERAL
12	Scope
13 14	Related Work
15	Reference Standards
16	Quality Assurance
17	Shop Drawings
18 19 20	Operation and Maintenance Data Design Criteria
21	PART 2 - PRODUCTS
22	Cabinet Unit Heaters
23	
24 25	PART 3 - EXECUTION Installation
26	Cabinet Unit Heaters
27	Construction Verification Items
28	Functional Performance Testing
29	
30	RELATED WORK
31	Section 23 05 23 - Common Motor Requirements for HVAC Equipment
33	Section 23 36 00 - Air Duct Accessories
34	Section 23 41 00 - Particulate Air Filtration
35	
36	REFERENCE
37	Applicable provisions of Division 1 govern work under this Section.
38	
40	ARI 210 Standard for Unitary Air-Conditioning Equipment
41	ARI 410 Standard for Forced-Circulation Air-Cooling and Air-Heating Coils
42	CS 140
43	
44 45	Refer to division 1 General Conditions Equals and Substitutions
46	
47	SHOP DRAWINGS
48	Refer to division 1, General Conditions, Submittals.
49	
50 51 52	all equipment in this section. Include color selection chart where applicable.
53	OPERATION AND MAINTENANCE DATA
54	All operations and maintenance data shall comply with the submission and content requirements specified under section
55	GENERAL REQUIREMENTS.
56	
57 58	DESIGN CRITERIA Forced Circulation Coils: Ratings certified in accordance with ARI 410
59	י סיככת כויכמומנוסוו כסווז. המנוווצז נכו נוווכם זוז מננסו עמוונל שונוו אתו 410.
60	Electrical Equipment and heaters shall be UL listed for the service specified.
61	The states have a second s
62 63	Electrical components and work must be in accordance with National Electrical Code.

1

2 3

4

5 6

7

8 9

10 11

12

13 14

15 16 17

18

19 20

21 22 23

24 25 26

27 28 29

30

31

32 33

34 35 36

37

38 39

40

41

42 43

PART 2 - PRODUCTS

CABINET UNIT HEATERS

Manufacturers: Modine, McQuay, Trane, Airtherm, Sterling, Vulcan, Zehnder Rittling or approved equal.

Unit types and arrangements as specified on Drawing Schedule.

Vertical unit casings shall have 16 gauge steel front panels and minimum 18 gauge steel end and side panels. Horizontal units located in concealed spaces or mounted in ceiling shall have minimum 18-gauge front, end, and side panels.

Furnish exposed cabinets in baked enamel finish in one of manufacturer's standard colors, selected by Architect.

Furnish ceiling mounted units with hinged front panel to allow access to internal components.

Heating elements constructed of copper tubes with aluminum fins, tested at 200 psig.

Use centrifugal type fans, statically and dynamically balanced to operate without objectionable noise and vibration.

Motors to be 120 volt, single phase, permanently lubricated, with thermal overload protection and disconnect switch at unit.

Furnish each unit with filter rack and 1 inch panel filters as specified in 23 41 00.

PART 3 - EXECUTION

INSTALLATION

Install units in accordance with manufacturer's installation instructions.

Install branch water or steam/condensate piping to each unit with a minimum of three elbows to allow for expansion and contraction of the piping system.

Coordinate location of units with other trades to assure correct recess size for recessed units.

After installation, provide protective covers to prevent accumulation of dirt on units during balance of construction.

CABINET UNIT HEATERS

Mount units in locations indicated on Drawings and as detailed. Install drain valve on coil side of shutoff valves for each hot water cabinet heater.

CONSTRUCTION VERIFICATION ITEMS

Contractor is responsible for utilizing the construction verification checklists supplied by the third party commissioning agent in accordance with the procedures defined for construction verification in Section 01 91 00.

44 FUNCTIONAL PERFORMANCE TESTING

45 Contractor is responsible for utilizing the functional performance test forms supplied by the third party commissioning agent in 46 accordance with the procedures defined for functional performance testing in Section 01 91 00.

47 48

1	SECTION 26 05 00
2	COMMON WORK RESULTS FOR ELECTRICAL
3	
4	
5	PART 1 - GENERAL
6	
7	The electrical work included in all other divisions is the responsibility of the contractor performing the division 26
8	work unless noted otherwise.
9	
10	PROJECT OVERVIEW
11	The project consists of an addition and minor renovation of the existing Warner Park Community Recreation Center
12	facility on Northport Drive in Madison, Wisconsin. As part of the addition, new electrical distribution will be extended
13	and a new PV array will be provided on the roof of the addition.
14	
15	SCOPE
16	The work under this section includes basic electrical requirements, which are applicable to all Division 26 sections.
17	This section includes information common to two or more technical specification sections or items that are of a
18	general nature, not conveniently fitting into other technical sections. Included are the following topics:
19	
20	PART 1 - GENERAL
21	Project Overview
22	Scope
23	Related Work
24	Reference
25	Reference Standards
26	
27	Commissioning Describer on Description and the
28	Regulatory Requirements
29	Quality Assurance
30	Continuity of Existing Services and Systems
31	Protection of Finished Surfaces
32	Approved Electrical Testing Laboratories
33	Sieeves and Openings
34 25	Sealing and Fire Stopping
35	Omissions
30 27	Offilissions Submittale
57 20	Sublitudis Project /Site Conditions
20	Work Sequence and Scheduling
39 40	Work by Other Trades
40 //1	Salvage Materials
41	Certificates and Inspections
42 //3	Operating and Maintenance Data
43	Record Drawings
45	Accord Drawings
46	ΡΔΡΤ 2 - ΡΡΟΠΙΙΟΤS
40	Access Panels and Doors
48	Identification
49	Sealing and Fire Stopping
50	
51	PART 3 - EXECUTION
52	Paintability
53	Concrete Work
54	Cutting and Patching
55	Building Access
56	Equipment Access
57	Coordination
58	Sleeves and Openings
20	Siceves and Openings

1	Sealing and Fire Stopping
2	Housekeeping and Clean Up

- Housekeeping and Clean Up
- Agency Training

RELATED WORK

3

4 5

10

6 Applicable provisions of Division 1 govern work under this Section. 7

8 **REFERENCE STANDARDS**

9 Abbreviations of standards organizations referenced in this and other sections are as follows:

- 11 ANSI American National Standards Institute
- 12 ASTM American Society for Testing and Materials
- **Environmental Protection Agency** 13 EPA
- ETL 14 Electrical Testing Laboratories, Inc.
- 15 IEEE Institute of Electrical and Electronics Engineers
- 16 IES Illuminating Engineering Society
- 17 ISA Instrument Society of America
- 18 NBS National Bureau of Standards
- 19 NEC National Electric Code
- 20 NEMA National Electrical Manufacturers Association
- 21 NESC National Electrical Safety Code
- 22 NFPA National Fire Protection Association
- 23 UL Underwriters Laboratories Inc.
- 24 DSPS Wisconsin Department of Safety and Professional Services 25

26 LEED CERTIFICATION

- 27 The project will be LEED Certified thru the United States Green Building Council's (USGBC) Leadership in Energy and 28 Environmental Design (LEED) program. Refer to Section 01 81 13 – Sustainable Design Requirements for additional
- 29 requirements.
- 30

31 In addition to complying with Division 26 drawings and specifications, equipment and material shall also comply with 32 Section 01 81 13 and LEED requirements.

33

36

34 The Division 26 contractor will be expected to provide all required documentation, submittals, etc. in accordance with 35 prerequisites and credits associated with Division 26 work and LEED Certification.

37 COMMISSIONING

38 The systems will be commissioned by an independent third party in accordance with USGBC LEED Energy and

- 39 Atmosphere Credit C3 – Enhanced Commissioning. Refer to Sections 01 91 00 – Commissioning and 01 95 01 –
- 40 Monitoring-Based Commissioning for additional requirements.
- 41

42 **REGULATORY REQUIREMENTS**

43 All work and materials are to conform in every detail to applicable rules and requirements of the Wisconsin State 44 Electrical Code (SPS 316), the National Electrical Code (NFPA 70), other applicable National Fire Protection Association 45 codes, the National Electrical Safety Code, and present manufacturing standards (including NEMA).

46

47 All Division 26 work shall be done under the direction of a currently licensed State of Wisconsin Master Electrician. 48

49 QUALITY ASSURANCE

50 Where equipment or accessories are used which differ in arrangement, configuration, dimensions, ratings, or 51 engineering parameters from those indicated on the contract documents, the contractor is responsible for all costs 52 involved in integrating the equipment or accessories into the system and the assigned space, and for obtaining the 53 performance from the system into which these items are placed.

54

55 Manufacturer references used herein are intended to establish a level of quality and performance requirements 56 unless more explicit restrictions are stated to apply.

57

All materials shall be listed by and shall bear the label of an approved electrical testing laboratory. If none of the approved electrical testing laboratories has published standards for a particular item, then other national independent testing standards, if available, applicable, and approved by City of Madison, shall apply and such items shall bear those labels. Where one of the approved electrical testing laboratories has an applicable system listing and

5 label, the entire system shall be so labeled.

6

7 CONTINUITY OF EXISTING SERVICES AND SYSTEMS

No outages shall be permitted on existing systems except at the time and during the interval specified by the user agency and by the City of Madison Project Representative. The institution may require written approval. Any outage must be scheduled when the interruption causes the least interference with normal institutional schedules and business routines. No extra costs will be paid to the Contractor for such outages which must occur outside of regular weekly working hours.

13

This Contractor shall restore any circuit interrupted as a result of this work to proper operation as soon as possible.
 Note that institutional operations are on a seven-day week schedule.

16

17 **PROTECTION OF FINISHED SURFACES**

Furnish one can of touch-up paint for each different color factory finish furnished by the Contractor. Deliver touch-up
 paint with other "loose and detachable parts" as covered in the General Requirements.

20

21 APPROVED ELECTRICAL TESTING LABORATORIES

The following laboratories are approved for providing electrical product safety testing and listing services as required in these specifications:

- 24
- 25 Underwriters Laboratories Inc.

26 Electrical Testing Laboratories, Inc.

27

28 SLEEVES AND OPENINGS

29 Refer to Division 1, General Requirements, Sleeves and Openings.

30

31 SEALING AND FIRE STOPPING

Sealing and fire stopping of sleeves/openings between conduits, cable trays, wireways, troughs, cablebus, busduct,
 etc. and the sleeve, structural or partition opening shall be the responsibility of the contractor whose work penetrates
 the opening. Provide all fire stopping of fire rated penetrations and sealing of smoke rated penetrations in compliance
 with section 07 84 00 Fire Stopping.

36 37 INTENT

The Contractor shall furnish and install all the necessary materials, apparatus, and devices to complete the electrical equipment and systems installation herein specified, except such parts as are specifically exempted herein.

40

If an item is either called for in the specifications or shown on the plans, it shall be considered sufficient for the inclusion of said item in this contract. If a conflict exists within the Specifications or exists within the Drawings, the Contractor shall furnish the item, system, or workmanship, which is the highest quality, largest, or most closely fits the City's intent (as determined by the City of Madison Project Manager). Refer to the General Conditions of the Contract for further clarification.

46

It must be understood that the details and drawings are diagrammatic. The Contractor shall verify all dimensions atthe site and be responsible for their accuracy.

49

50 All sizes as given are minimum except as noted.

51

52 Materials and labor shall be new (unless noted or stated otherwise), first class, and workmanlike, and shall be subject 53 at all times to the City of Madison's and/or A/E's inspections, tests and approval from the commencement until the 54 acceptance of the completed work.

55

56 Whenever a particular manufacturer's product is named, it is intended to establish a level of quality and performance

- 57 requirements unless more explicit restrictions are stated to apply.
- 58

1 OMISSIONS

No later than ten (10) days before bid opening, the Contractor shall call the attention of the City of Madison to any
 materials or apparatus the Contractor believes to be inadequate and to any necessary items of work omitted.

5 SUBMITTALS

4

15

28

31

33

37

Submit for all equipment and systems as indicated in the respective specification sections, marking each submittal
with that specification section number. Mark general catalog sheets and drawings to indicate specific items being
submitted and proper identification of equipment by name and/or number, as indicated in the contract documents.
Failure to do this may result in the submittal(s) being returned to the Contractor for correction and resubmission.
Failing to follow these instructions does not relieve the Contractor from the requirement of meeting the project
schedule.

On request from the City of Madison, the successful bidder shall furnish additional drawings, illustrations, catalog
 data, performance characteristics, etc.

Submittals shall be grouped to include complete submittals of related systems, products, and accessories in a single submittal. Mark dimensions and values in units to match those specified. Include wiring diagrams of electrically powered equipment.

The submittals must be approved before fabrication is authorized.

21 Submit sufficient quantities of submittals to allow the following distribution:

Operating and Maintenance Manuals	2 copies
User agency	1 copy
A/E	1 copy

26 **PROJECT/SITE CONDITIONS**

27 Install Work in locations shown on drawings, unless prevented by project conditions.

Prepare drawings showing proposed rearrangement of work to meet project conditions, including changes to work
 specified in other sections. Obtain permission of City of Madison before proceeding.

Tools, materials and equipment shall be confined to areas designated by the City of Madison.

34 WORK SEQUENCE AND SCHEDULING

Install work in phases to accommodate user agency's occupancy requirements. During the construction period coordinate electrical schedule and operations with City of Madison's Construction Representative.

38 WORK BY OTHER TRADES

Every attempt has been made to indicate in this trade's specifications and drawings all work required of this Contractor. However, there may be additional specific paragraphs in other trade specifications and addenda, and additional notes on drawings for other trades which pertain to this trade's work, and thus those additional requirements are hereby made a part of these specifications and drawings.

43

Electrical details on drawings for equipment to be provided by others are based on preliminary design data only. This Contractor shall lay out the electrical work and shall be responsible for its correctness to match equipment actually provided by others.

47 48 SALVAGE MATERIALS

No materials removed from this project shall be reused unless specifically noted otherwise. All materials removed
 shall become the property of and shall be disposed of by the Contractor.

51

52 CERTIFICATES AND INSPECTIONS

53 Obtain and pay for all required installation inspections.

54

55 **OPERATION AND MAINTENANCE DATA**

All operations and maintenance data shall comply with the submission and content requirements specified under

- 57 section GENERAL REQUIREMENTS.
- 58

In addition to the general content specified under GENERAL REQUIREMENTS supply the following additional 1 2 documentation: 3 1. Manufacturer's wiring diagrams for electrically powered equipment. 4 5 **RECORD DRAWINGS** 6 The Contractor shall maintain at least one copy each of the specifications and drawings on the job site at all times. 7 8 The City of Madison will provide the Contractor with a suitable set of contract drawings on which daily records of 9 changes and deviations from contract shall be recorded. Dimensions and elevations on the record drawings shall 10 locate all buried or concealed piping, conduit, or similar items. 11 12 The daily record of changes shall be the responsibility of Contractor's field superintendent. No arbitrary mark-ups will 13 be permitted. 14 At completion of the project, the Contractor shall submit the marked-up record drawings to the Architect/Engineer 15 16 prior to final payment. 17 18 19 PART 2 - PRODUCTS 20 21 ACCESS PANELS AND DOORS 22 Lay-in Ceilings: 23 Removable lay-in ceiling tiles in 2 x 2 foot or 2 x 4 foot configuration provided under other divisions are sufficient; no 24 additional access provisions are required unless specifically indicated. 25 26 **IDENTIFICATION** 27 See Electrical section 26 05 53 – Identification for Electrical Systems. 28 29 SEALING AND FIRE STOPPING 30 FIRE AND/OR SMOKE RATED PENETRATIONS: 31 Provide all fire stopping of fire rated penetrations and sealing of smoke rated penetrations in compliance with section 32 07 84 00 "Fire Stopping". 33 34 NON-RATED PENETRATIONS: 35 Conduit and Cable Tray Penetrations: 36 At conduit and cable tray penetrations of non-rated interior partitions, floors and exterior walls above grade, use 37 urethane caulk in annular space between conduit and sleeve, or the core drilled opening. 38 PART 3 - EXECUTION 39 40 41 PAINTABILITY Any/all electrical equipment, conduit, wiring, boxes, etc. that is to be exposed shall be painted to match the 42 43 architectural colors throughout the construction limits. Provide the appropriate finish on all electrical equipment, 44 conduit, wiring, boxes, etc. such that painting is possible. Coordinate all finish requirements with architectural 45 documents. 46 47 CONCRETE WORK 48 The Division 3 Contractor will perform all cast-in-place concrete unless noted otherwise elsewhere. Provide all layout 49 drawings, anchor bolts, metal shapes, and/or templates required to be cast into concrete or used to form concrete for 50 the support of electrical equipment. 51 52 **CUTTING AND PATCHING** 53 Refer to Division 1, General Requirements, Cutting and Patching. 54 55 **BUILDING ACCESS** Arrange for the necessary openings in the building to allow for admittance of all apparatus. When the building access 56 57 was not previously arranged and must be provided by this contractor, restore any opening to its original condition 58 after the apparatus has been brought into the building.

1 EQUIPMENT ACCESS

Install all piping, conduit, ductwork, and accessories to permit access to equipment for maintenance. Coordinate the exact location of wall and ceiling access panels and doors with the General Contractor, making sure that access is available for all equipment and specialties. Where access is required in plaster or drywall walls or ceilings, furnish the cancers to the Contract to the Contract

access doors to the General Contractor and reimburse the General Contractor for installation of those access doors.

7 COORDINATION

8 The Contractor shall cooperate with other trades and City of Madison in locating work in a proper manner. Should it 9 be necessary to raise or lower or move longitudinally any part of the electrical work to better fit the general 10 installation, such work shall be done at no extra cost to the City of Madison, provided such decision is reached prior to 11 actual installation. The Contractor shall check location of electrical outlets with respect to other installations before 12 installing.

13

The Contractor shall verify that all devices are compatible for the surfaces on which they will be used. This includes, but is not limited to light fixtures, panelboards, devices, etc. and recessed or semi-recessed heating units installed in/on architectural surfaces.

17

20

18 Coordinate all work with other contractors prior to installation. Any installed work that is not coordinated and that 19 interferes with other contractor's work shall be removed or relocated at the installing contractor's expense.

21 SLEEVES AND OPENINGS

22 Conduit penetrations in existing concrete floors: Core drill openings.

- Where penetrating conduit weight is supported by floor, provide manufactured product or structural bearing collar
 designed to carry load.
- 26

27 SEALING AND FIRE STOPPING

28 FIRE AND/OR SMOKE RATED PENETRATIONS:

Provide all fire stopping of fire rated penetrations and sealing of smoke rated penetrations in compliance with section
 07 84 00 Fire Stopping.

31

36

40

41

42

43

44

45

46

47

49

50

51

52

53

54

32 NON-RATED PENETRATIONS:

At all interior walls and exterior walls, conduit penetrations are required to be sealed. Apply sealant to both sides of the penetration in such a manner that the annular space between the sleeve or cored opening and the conduit is completely blocked.

- 37 PENETRATIONS SUBJECT TO WATER INTRUSION:
- For penetrations (both rated and non-rated) in floors subject to water intrusion or in rooms housing electrical equipment (but not within walls) provide one of the following:
 - Conduit penetration where steel pipe sleeve is used extend steel sleeve 2" above the floor.
 - Conduit penetration where cast in place fire stopping device/sleeve is used, extend device/sleeve 2" above the floor (provided it meets the device's UL listing).
 - Conduit penetration where there is no steel sleeve or cast in place fire stopping device/sleeve, provide 2"x 2" x 1/8" galvanized steel angles fastened to floor surrounding the penetration or group of penetrations to prevent water from getting to penetration. Provide urethane caulk between angles and floor and fasten angles to floor minimum 8" on center. Seal corners water tight with urethane caulk.
- 48 Floors subject to water intrusion or rooms housing electrical equipment include the following locations:
 - Restrooms
 - Janitor Rooms w/ Sinks
 - Mechanical/Plumbing Equipment Rooms
 - Data/Telecommunications Rooms
 - Electrical Equipment Rooms

55 Provide waterproof caulk sealant top coating on fire stopping system (or other approved means to protect the fire 56 stopping system from water) in areas subject to wash down such as Food Service and Dish Washing Areas.

57 58

1 HOUSEKEEPING AND CLEAN UP

- 2 The Contractor shall clean up and remove from the premises, on a daily basis, all debris and rubbish resulting from its
- 3 work and shall repair all damage to new and existing equipment resulting from its work. When job is complete, this
- 4 Contractor shall remove all tools, excess material and equipment, etc., from the site. 5

6 AGENCY TRAINING

- 7 Contractor to provide factory authorized representative and/or field personnel knowledgeable with the operations,
- 8 maintenance and troubleshooting of the system and/or components defined within this section for a minimum period
- 9 of 4 hours.
- 10
- 11

Page Intentionally Left Blank

1 2	SECTION 26 05 02 ELECTRICAL DEMOLITION FOR REMODELING
3	
4 5 6	PART 1 - GENERAL
7	SCOPE
8	The work under this section includes the demolition associated with the addition and minor renovation of the existing
9	Warner Park Community Recreation Center facility on Northport Drive in Madison, Wisconsin. Included are the
10	following topics:
11	
12	PART 1 - GENERAL
13 17	Scope Related Work
14	
16	PART 2 - PRODUCTS
17	Materials and Equipment
18	
19	PART 3 - EXECUTION
20	Examination
21	Preparation
22	Demolition and Extension of the Existing Electrical Work
23	
24	KELATED WORK Applicable provisions of Division 1 govern work under this Section
25	Applicable provisions of Division 1 govern work under this section.
27	PART 2 - PRODUCTS
28	
29	MATERIALS AND EQUIPMENT
30	Materials and equipment for patching and extending work as specified in the individual Sections.
31	
32	PART 3 - EXECUTION
33	
34 25	EXAMINATION Verify field measurements and circuiting arrangements as shown on Drawings
36	verity field frieddie freddeling difungerienes as shown on Drawings.
37	Verify that abandoned wiring and equipment serve only abandoned facilities.
38	, , ,
39	Demolition Drawings are based on casual field observation and/or existing record documents. Report discrepancies to
40	the User Agency, Architect/Engineer, and City of Madison Field Representative before disturbing existing installation.
41	
42	Beginning of demolition means installer accepts existing conditions.
43	
44 45	PREPARATION
45 46	Disconnect electrical systems in wails, noors, and centings scheduled for removal.
40	Coordinate utility service outages with the User Agency. City of Madison Field Representative and
48	Architect/Engineer. Also, if applicable, coordinate utility service outages with the local Utility Company.
49	
50	Provide temporary wiring and connections to maintain existing systems in service during construction. When work
51	must be performed on energized equipment or circuits, use personnel experienced in such operations and follow the
52	safe working practice requirements of NFPA 70E.
53	
54	Existing Electrical Service: Maintain existing system in service until new system is complete and ready for service.
55	Disable system only to make switchovers and connections. Obtain permission from the City of Madison Field
סכ 57	Representative at least 48 nours perore partially or completely disabiling system. Minimize outage duration. If
58	required, make temporary connections to maintain service in dreds dujdcent to work dred.
55	

Existing Fire Alarm System: Maintain existing system in service until new system is accepted. Disable system only to make switchovers and connections. Obtain permission from the City of Madison Field Representative and local Authority Having Jurisdiction at least 48 hours before partially or completely disabling system. Minimize outage duration. If required, make temporary connections to maintain service in areas adjacent to work area.

6 DEMOLITION AND EXTENSION OF EXISTING ELECTRICAL WORK

Remove, relocate, and extend existing installations as necessary, to accommodate new construction and to meet all
 requirements of these specifications. Extend existing installations using materials and methods compatible with
 existing electrical installations, or as specified.

10 11

12

16

19

21

23

25

27

5

Remove abandoned wiring to source of supply.

Remove exposed abandoned conduit and abandoned conduit above accessible ceiling finishes, unless noted otherwise on drawings. Cut conduit flush with walls and floors, and patch surfaces. If certain conduits and boxes are abandoned but not scheduled for removal, they shall be shown on the "As Built Drawings".

Disconnect abandoned outlets and remove devices. Remove abandoned outlets if conduit and wiring servicing them is abandoned and removed. Provide blank cover for abandoned outlets which are not removed.

20 Disconnect and remove electrical devices and equipment serving utilization equipment that has been removed.

22 Disconnect and remove abandoned luminaires. Remove brackets, stems, hangers, and other accessories.

24 Provide revised typed circuit directory in panelboards that have circuits removed.

26 Repair adjacent construction and finishes damaged during demolition and extension work.

28 Maintain access to existing electrical installations which remain active. Modify installation or provide access panel as 29 appropriate.

30

Provide supplemental support for conduits that are routed through demolition area and are to remain. Supplemental support shall be added so that the conduit meets the support requirements of electrical specification section 26 05

32 support sha33 33.

34 35

1 2	SECTION 26 05 04 CLEANING, INSPECTION, AND TESTING OF ELECTRICAL EQUIPMENT
3	
4 5 6	PART 1 - GENERAL
7	SCOPE
, 8 9	The work under this section includes the required cleaning, inspection, adjustment, maintenance and testing of electrical equipment, as specified herein. This applies only to new electrical and existing electrical equipment being
10 11	furnished, modified, worked on or serviced by this contractor for this project. Included are the following topics:
12	PART 1 - GENERAL
13	Scope
14 15	Related Work
16	PART 2 - PRODUCTS
17 18	Not Used
19	PART 3 - EXECUTION
20	General Inspection and Cleaning of All Electrical Equipment
21	Grounding Systems
22	Light Fixtures
23	Occupancy Sensors
24	
25 26	RELATED WORK
20	Applicable provisions of Division 1 govern work under this section.
28	PART 2 - PRODUCTS
29	
30	Not Used.
32 33	PART 3 - EXECUTION
34	GENERAL INSPECTION AND CLEANING OF ALL ELECTRICAL EQUIPMENT
35 36	Inspect for physical damage and abnormal mechanical and electrical conditions.
37 38	Any item found to be out of tolerance, or in any other way defective as a result of the required inspection or testing, shall be reported to the City of Madison. Procedure for repair and/or replacement will be outlined. After appropriate
39 40	corrective action is completed the item shall be re-tested.
41 42	Compare equipment nameplate information with the latest single line diagram and report any discrepancies.
43 44	Verify proper auxiliary device operation and indicators.
45 46	Check tightness of accessible bolted electrical joints. Use torque wrench method.
47	Make a close examination of equipment and remove any shipping brackets, insulation, packing, etc. that may not
48 49	have been removed during original installation.
50 51 52	Make a close examination of equipment and remove any dirt or other forms of debris that may have collected in existing equipment or in new equipment during installation.
53	Clean All Equipment:
54	Vacuum inside of panelboards, switchboards, switchgear, transformer core and coils, bus ducts, MCC's, fire
55	alarm panels, communication/data panels, security panels, etc.
56	Loosen attached particles and vacuum them away.
57 58	Wipe all insulators with a clean, dry, lint free rag. Clean insulator grooves.
	-

- Inspect equipment anchorage.
 2
- 3 Inspect equipment and bus alignment.
- 5 Check all heater elements for operation and control.
- 7 Lubricate nonelectrical equipment per manufacturer's recommendations.

9 **GROUNDING SYSTEMS**

10 Inspect the ground system for adequate termination at all devices.

12 LIGHT FIXTURES

4

6

8

11

- 13 Check the bonding and proper lamping. Verify that recessed fixtures are installed with hold down clips. Confirm
- operation of the fixture with the proper switch or sensor.

16 OCCUPANCY SENSORS

- 17 Confirm operation of the sensor per the manufacturer's specification.
- 18 19 END OF SECTION

1 2	SECTION 260519 LOW-VOLTAGE ELECTRICAL POWER CONDUCTORS AND CABLES
3	
4	
5	PART 1 - GENERAL
6	
/ 0	SCOPE The work under this section includes furniching and installing required wiring and cabling systems including nulling
٥ ۵	the work under this section includes furnishing and installing required wring and cabling systems including pulling,
10	terminating and spitcing. Included are the following topics.
11	PART 1 - GENERAL
12	Scope
13	Related Work
14	References
15	Submittals
16 17	Project Conditions
18	PART 2 - PRODUCTS
19	General
20	Building Wire
21	Wiring Connectors
22	
23	PART 3 - EXECUTION
24	Wiring Installation in Receivance
25	Wiring Connections and Terminations
27	Field Quality Control
28	Wire Color
29	Branch Circuits
30	
31	RELATED WORK
32	Applicable provisions of Division 1 govern work under this Section.
33	
34	Section 26 05 33 – Raceway and Boxes for Electrical Systems.
35 26	Section 26 05 53 – Identification for Electrical Systems.
30	REERENCES
38	SPS 316- Electrical
39	
40	SUBMITTALS
41	Submit product data: Provide for each cable assembly type.
42	
43	Submit factory test reports: Indicate procedures and values obtained.
44	
45	Submit shop drawings for modular wiring system including layout of distribution devices, branch circuit conduit and
46	cables, circuiting arrangement, and outlet devices.
47 10	Submit manufacturar's installation instructions. Indicate application conditions and limitations of use stipulated by
40 70	submit manufacturer's installation instructions. Indicate application conditions and initiations of use stipulated by
50	product testing agency specified under Regulatory Requirements.
51	PROJECT CONDITIONS
52	Verify that field measurements are as shown on Drawings.
53	
54	Conductor sizes are based on copper.
55	
56	Wire and cable routing shown on Drawings is approximate unless dimensioned. Route wire and cable as required for
5/	project conditions.
20	

1 2 2	Where wire and cable routing is not shown, and destination only is indicated, determine exact routing and lengths required.
3 4 5	PART 2 - PRODUCTS
5	GENERAL
7 8 9	All wire shall be new, delivered to the site in unbroken cartons and shall be less than one year old out of manufacturer's stock.
10 11	All conductors shall be copper. All ground conductors shall be copper.
12 13	Insulation shall have a 600 volt rating.
14 15	All conductors shall be stranded.
16 17 18 19	Stranded conductors may only be terminated with UL OR ETL Listed type terminations or methods: e.g. stranded conductors may not be wrapped around a terminal screw but must be terminated with a crimp type device or must be terminated in an approved back wired method.
20	BUILDING WIRE
21 22	Description: Single conductor insulated wire 90 degree C.
23 24	Insulation: Type THHN/THWN-2, XHHW-2 insulation.
25	WIRING CONNECTORS
26 27	Split Bolt Connectors: Not acceptable.
28 29 30	solderless Pressure Connectors: High copper alloy terminal. May be used only for cable termination to equipment terminals. Not approved for splicing.
31 32 33 34	Twist Type Wire Connectors: Solderless twist type spring connector (wire-nut) with insulating cover for copper wire splices and taps. Use for conductor sizes 10 AWG and smaller. The manufacturer's wire fill capacity must be followed. Use Silicone filled twist type spring connectors in all wet location areas.
35 36 37	Mechanical Connectors: Bolted type tin-plated; high conductivity copper alloy; spacer between conductors; beveled cable entrances.
38 39 40 41 42	Compression (crimp) Connectors: Long barrel; seamless, tin-plated electrolytic copper tubing; internally beveled barrel ends. Connector shall be clearly marked with the wire size and type and proper number and location of crimps. Connector must be installed with a crimper tool listed for use with the manufacturer and type of compression connector.
43 44 45	Insulation Piercing Connectors: Molded insulated body, copper teeth, wrench tightened, UL 486B Listed. May be used only for connection of a tap conductor in run and tap type applications when main conductor is 8 AWG and larger.
46 47	PART 3 - EXECUTION
48	GENERAL WIRING METHODS
49 50	All wire and cable shall be installed in conduit.
51 52	Do not use wire smaller than 12 AWG for power and lighting circuits.
53 54 55 56	All phase, neutral and ground conductors shall be sized to prevent excessive voltage drop at rated circuit ampacity. As a minimum use 10 AWG conductors for 20 ampere, 120 volt branch circuit home runs longer than 100 feet (30 m), and for 20 ampere, 277 volt branch circuit home runs longer than 200 feet (61 m).
57 58	Ground conductor size shall be increased per NEC 250.122(B) when phase and phase/neutral conductors are increased in size.

Make conductor lengths for parallel conductors equal. 1 2 3 Splice only in junction or outlet boxes. 4 5 No conductor less than 10 AWG shall be installed in exterior underground conduit. 6 7 Identify ALL low voltage wire, 600V and lower, per section 26 05 53. 8 9 Neatly train and lace wiring inside boxes, equipment, and panelboards. 10 WIRING INSTALLATION IN RACEWAYS 11 12 Pull all conductors into a raceway at the same time. Use Listed water or silicone based wire pulling lubricant for 13 pulling 4 AWG and larger wires and for other conditions when necessary. Wax based lubricants are not allowed. 14 Pulling lubricant is not required for low friction type products where the cable manufacturer recommends that cables 15 be pulled without lube. 16 17 Install wire in raceway after interior of building has been physically protected from the weather and all mechanical 18 work likely to injure conductors has been completed. 19 20 Completely and thoroughly swab raceway system before installing conductors. 21 22 Place all conductors of a given circuit (this includes phase wires, neutral (if any), and ground conductor) in the same 23 raceway. If parallel phase and/or neutral wires are used, then place an equal number of phase and neutral 24 conductors in same raceway or cable. 25 26 In high ambient spaces, mechanical rooms, utility rooms and exterior exposed conduit, 90 degree C, XHHW-2 27 conductors shall be utilized. 28 29 WIRING CONNECTIONS AND TERMINATIONS 30 Splice only in accessible junction boxes. 31 32 Wire splices and taps shall be made firm, and adequate to carry the full current rating of the respective wire without 33 soldering and without perceptible temperature rise. 34 35 All splices shall be so made that they have an electrical resistance not in excess of two feet (600 mm) of the 36 conductor. 37 38 Use solderless twist type spring connectors (wire nuts) with insulating covers for wire splices and taps, 10 AWG and 39 smaller. 40 41 Use mechanical or compression connectors for wire splices and taps, 8 AWG and larger. Tape uninsulated conductors and connectors with electrical tape to 150 percent of the insulation value of the wiring. 42 43 44 Thoroughly clean wires before installing lugs and connectors. 45 46 At all splices and terminations, leave tails long enough to cut splice out and completely re-splice. 47 48 FIELD QUALITY CONTROL 49 Field inspection and testing will be performed under provisions of Section 26 05 04. 50 WIRE COLOR 51 52 General: 53 Solid colored insulation is required for all THHN/THWN-2 wire. For other wire types use colored wire or 54 identify wire with colored tape at all terminals, splices and boxes. Wire shall be colored as indicated below. 55 56 In existing facilities, use existing color scheme. 57

Switch legs shall be the same color as their associated circuit, except for the second switch leg used for dual-level switching. The second switch leg shall be the next phase color, e.g. if the first switch leg is brown (277/480V phase A), the second switch leg shall be orange (277/480V phase B).

Traveler conductors run between 3 and 4 way switches shall be colored pink or purple.

Neutral Conductors: White for 120/208V and 120/240V systems, Gray for 277/480V systems. Where there are two or
 more neutrals in one conduit, each shall be individually identified with a different stripe.

10 Branch Circuit Conductors: Three or four wire home runs shall have each phase uniquely color coded.

12 Feeder Circuit Conductors: Each phase shall be uniquely color coded.

Ground Conductors: Green colored insulation for THHN/THWN-2 wire. For other wire types use green colored wire or identify wire with green tape at both ends and at all access points, such as panelboards, motor starters, disconnects and junction boxes. When isolated grounds are required, contractor shall provide green with yellow tracer.

18 BRANCH CIRCUITS

19 The use of single-phase, multi-wire branch circuits with a common neutral is not permitted. All single-phase branch

- circuits shall be furnished and installed with an individual accompanying neutral, sized the same as the phase conductors.
- 22 23

1

2

3

4 5

6

11

13

17

1 2	SECTION 26 05 23 CONTROL-VOLTAGE ELECTRICAL POWER CABLES
3	
4 5 6	PART 1 - GENERAL
6 7	SCOPE
8	The work under this section includes furnishing and installing cabling for remote-control, signaling and power-limited
9	circuits. Included are the following topics:
10	
11	PART 1 - GENERAL
12	Scope
13	Related Work
14 15	References
16	Project Conditions
17	
18	PART 2 - PRODUCTS
19	General
20	Remote-Control and Signaling Cable
21	Wiring Connectors
22	
23	PART 3 - EXECUTION
24	General Wiring Installation In Pacoways
25	Wiring Connections and Terminations
27	Field Quality Control
28	
29	RELATED WORK
30 31	Applicable provisions of Division 1 govern work under this Section.
32	Section 26 05 33 – Raceway and Boxes for Electrical Systems.
33	Section 26 05 53 – Identification for Electrical Systems.
34	
35	KEFEKENCES
30 37	NFPA 70 - National Electrical Code.
38	SUBMITTALS
39 40	Submit product data: Provide for each cable assembly type.
41 42	Submit manufacturer's installation instructions. Indicate application conditions and limitations of use stipulated by product testing agency.
43	
44	PROJECT CONDITIONS
45	Verify that field measurements are as shown on Drawings.
40 17	Wire and cable routing shown on Drawings is approximate unless dimensioned. Route wire and cable as required to
48	meet Project Conditions.
49	
50 51	Where wire and cable routing is not shown, and destination only is indicated, determine exact routing and lengths required.
52	
53	PART 2 - PRODUCTS
54	
55 56	UEINEMAL All wire shall be new delivered to the site in unbroken cartons and shall be loss than one year old out of
50 57 58	manufacturer's stock.
50	

1	All conductors shall be copper.
3	Insulation shall have a 600 volt rating.
4	All conductors shall be suitable for the application intended. Conductors #12 and smaller may be solid or stranded
6	with the following requirements or exceptions:
8	All conductors terminated with crimp type devices shall be stranded.
9 10	Stranded conductors shall be terminated with LU OP ETL Listed type terminations or methods: e.g. stranded
10	conductors shall not be wrapped around a terminal screw but shall be terminated with a crimp type device
12	or in an approved back wired method.
13 17	
15	Refer to Section 28.46.21 for requirements for cable to be used on fire alarm systems
16	
17 18	Refer to Sections 27 00 05 and 27 21 33 for requirements for cable to be used on communication systems.
19 20	All other systems cabling shall meet the requirements of NEC Article 725 and the following:
20	Cable for Class 1 Remote-Control. Signaling and Power-Limited Circuits: 600 volt insulation, individual
22	conductors twisted together, shielded, and covered with an overall PVC jacket. Cable shall be Listed,
23	temperature rated, and suitable Type (general purpose, riser or plenum) for the application as required in
24	the National Electrical Code.
25	
26	Cable for Class 2 or Class 3 Remote-Control, Signaling and Power-Limited Circuits shall be Listed,
27	temperature rated, and suitable Type (general purpose, riser or plenum) for the application as required in
28	the National Electrical Code.
30	WIRING CONNECTORS
31	Split Bolt Connectors: Not acceptable.
32	
33	Spring Wire Connectors: Solderless spring type pressure connector with insulating covers for copper wire splices and
34	taps. Use for conductor sizes 10 AWG and smaller.
35	
36 37	All wire connectors used in underground or exterior pull boxes shall be get filled twist connectors or a connector designed for damp and wet locations.
38	
39	PART 3 - EXECUTION
40 41	GENERAL WIRING METHODS
42	Control-voltage cables shall be installed in conduit. However, they may be installed free-air (without conduit) above
43	accessible ceilings if the cable meets NEC requirements for the application, unless specified to be in conduit in other
44	sections of the specifications. See requirements for free-air cable installation below.
45	
46	Do not use wire smaller than 14 AWG for control wiring greater than 60 volts, or 18 AWG for voltages less than 60
47	volts, all sizes subject to NEC 725 requirements.
48	
49	Splice only in junction boxes.
50 E1	Identify wire per section 26.05.52
51	identify wire per section 26 05 53.
52 52	Neatly train and lace wiring inside boxes, and equipment
54	
55	WIRING INSTALLATION IN RACEWAYS
56	Pull all conductors into a raceway at the same time. Use Listed wire pulling lubricant for pulling conditions when
57	necessary.
58	

Install wire in raceway after interior of building has been physically protected from the weather and all mechanical
 work likely to injure conductors has been completed.

3

4 WIRING CONNECTIONS AND TERMINATIONS

Splice only in accessible junction boxes (except splices to low voltage occupancy sensor power packs and terminations
 to temperature control devices).

7

8 All splices shall be so made that they have an electrical resistance not in excess of two feet (600 mm) of the 9 conductor.

10

11 Use solderless spring type pressure connectors with insulating covers for wire splices and taps, 10 AWG and smaller.

12

13 Thoroughly clean wires before installing lugs and connectors.

15 At all splices and terminations, leave tails long enough to cut splice out and completely re-splice.

16

14

17 FIELD QUALITY CONTROL

- 18 Field inspection and testing will be performed under provisions of Section 26 05 04.
- 19
- 20

Page Intentionally Left Blank

1 2 3	SECTION 26 05 26 GROUNDING AND BONDING FOR ELECTRICAL SYSTEMS
4 5	PART 1 - GENERAL
6	
/	SCOPE
o Q	Communications systems. Included are the following tonics:
10	communications systems, included are the following topics.
11	PART 1 - GENERAL
12	Scope
13	Related Work
14	References
15	Regulatory Requirements
16	
17	PART 2 - PRODUCTS
18	Mechanical Connectors
19	Compression Connectors
20	Conductors
21	Bus/Busbar
22 22	
25 24	PART 5 - EXECUTION General
25	Less Than 600 Volt System Grounding
26	Communication System Grounding
27	Field Quality Control
28	Identification and Labeling
29	Construction Verification Items
30	Warranty
31	
32	All hardware, cables and related termination and support hardware shall be furnished, installed, wired, tested,
33	labeled, and documented by the Contractor, as detailed in this and related sections.
34	
35	RELATED WORK
36	Applicable provisions of Division 1 govern work under this Section.
3/ 20	
38 20	REFERENCES
39 40	ANSI/IEEE 142 (Latest edition) - Recommended Fractice for Grounding of Industrial and Commercial Power Systems
40 41	IEEE 837 - IEEE Standard for Qualifying Permanent Connections Used in Substation Grounding
42	TIA-607-C - Commercial Building Grounding (Earthing) and Bonding Requirements for Telecommunications
43	
44	REGULATORY REQUIREMENTS
45	Conform to requirements of NFPA 70.
46	
47	Furnish products listed and classified by Underwriters Laboratories, Inc. or testing firm acceptable to authority having
48	jurisdiction as suitable for purpose specified and shown.
49	
50	PART 2 - PRODUCTS
51	
52 52	MECHANICAL CONNECTORS
53 51	methanical connector bodies shall be manufactured from high strength, high conductivity cast copper alloy
54 55	connector body and shall be of the two bolt type
56	
57	Split bolt connector types are NOT allowed. Exception: the use of split bolts is acceptable for grounding of wire-
58	basket type cable tray, and for cable shields/straps of medium voltage cable.

4 C 5 T	המחטומנוטרפו.
4 C 5 T 6 b	
5 i 6 h	COMPRESSION CONNECTORS
n 11	The compression connectors shall be manuactured from pure wrought copper. The conductivity of this material shall
7 5	de no less than 99% by IACS standards.
/ о г	Task connector shall be fastery filled with an evide inhibiting compound
8 E	cach connector shall be factory filled with an oxide-inhibiting compound.
9 10 T	
10 I	The connectors shall meet or exceed the performance requirements of IEEE 837, latest revision.
11	The second second with the standard with the second frations, while second and second the second second second
12 1	ine connectors shall be clearly marked with the manufacturer, catalog number, conductor size and the required
13 C	compression tool settings.
14	
15 I	The installation of the connectors shall be made with a compression tool and die system, as recommended by the
16 m	nanufacturer of the connectors, and shall be irreversible.
17	
18 P	Pre-crimping of the ground rod is required for all irreversible compression connections to a ground rod.
19	
20 T	Ferminal lug for communication system grounding shall be compression type and conform to the following:
21	Material: Tin Plated Copper (aluminum not permitted).
22	Wire Size: to match conductor
23	Number of Stud Holes: 2
24	Stud Hole Size: 3/8"
25	Bolt Hole Spacing: per TIA-607-C
26	Tongue Angle: Straight
27	
28 C	CONDUCTORS
29 N	Material: Stranded copper (aluminum not permitted).
30	
31 F	eeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70,
31 F 32 w	Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the
31 F 32 w 33 si	Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility.
31 F 32 w 33 si 34	Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility.
31 F 32 w 33 si 34 35 B	Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors
31 F 32 w 33 si 34 35 B 36 ir	Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size.
31 F 32 w 33 si 34 35 B 36 ir 37	Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size.
31 F 32 w 33 si 34 35 B 36 ir 37 38 C	Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size.
31 F 32 w 33 si 34 35 36 ir 37 38 C 39 S	 Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size. Conductors for Telecommunications shall be as follows: Telecommunications Bonding Conductor (TMGB to Service Ground): No. 3/0 minimum or as shown on
31 F 32 w 33 si 34 35 36 ir 37 38 39 40	 Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size. Conductors for Telecommunications shall be as follows: Telecommunications Bonding Conductor (TMGB to Service Ground): No. 3/0 minimum or as shown on drawings.
31 F 32 w 33 si 34 35 36 ir 37 38 39 40 41 41	 Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size. Conductors for Telecommunications shall be as follows: Telecommunications Bonding Conductor (TMGB to Service Ground): No. 3/0 minimum or as shown on drawings. Telecommunications Bonding Backbone (TBB; TMGB to TGB): No. 3/0 minimum or as shown on drawings.
31 F 32 w 33 si 34 35 36 ir 37 38 39 40 41 42	 Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size. Conductors for Telecommunications shall be as follows: Telecommunications Bonding Conductor (TMGB to Service Ground): No. 3/0 minimum or as shown on drawings. Telecommunications Bonding Backbone (TBB; TMGB to TGB): No. 3/0 minimum or as shown on drawings. Telecommunications Grounding Equalizer (GE): No. 3/0 minimum or as shown on drawings.
31 F 32 w 33 si 34 35 36 ir 37 38 39 40 41 42 43 43	 Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size. Conductors for Telecommunications shall be as follows: Telecommunications Bonding Conductor (TMGB to Service Ground): No. 3/0 minimum or as shown on drawings. Telecommunications Bonding Backbone (TBB; TMGB to TGB): No. 3/0 minimum or as shown on drawings. Telecommunications Grounding Equalizer (GE): No. 3/0 minimum or as shown on drawings.
31 F 32 w 33 si 34 35 36 ir 37 38 39 40 41 42 43 44	 Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size. Conductors for Telecommunications shall be as follows: Telecommunications Bonding Conductor (TMGB to Service Ground): No. 3/0 minimum or as shown on drawings. Telecommunications Bonding Backbone (TBB; TMGB to TGB): No. 3/0 minimum or as shown on drawings. Telecommunications Grounding Equalizer (GE): No. 3/0 minimum or as shown on drawings. Decommunications Grounding Equalizer (GE): No. 3/0 minimum or as shown on drawings. Bonding Conductors shall be insulated with a Green Jacket or jacket marked with Green Tape or labeled per NEC Guidelines.
31 F 32 w 33 si 34 3 35 B 36 ir 37 3 38 C 39 40 41 42 43 44 45 5	 Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size. Conductors for Telecommunications shall be as follows: Telecommunications Bonding Conductor (TMGB to Service Ground): No. 3/0 minimum or as shown on drawings. Telecommunications Bonding Backbone (TBB; TMGB to TGB): No. 3/0 minimum or as shown on drawings. Telecommunications Grounding Equalizer (GE): No. 3/0 minimum or as shown on drawings. Bonding Conductors shall be insulated with a Green Jacket or jacket marked with Green Tape or labeled per NEC Guidelines.
31 F 32 w 33 si 34 3 35 B 36 ir 37 3 38 C 39 40 41 42 43 44 45 46 B	 Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size. Conductors for Telecommunications shall be as follows: Telecommunications Bonding Conductor (TMGB to Service Ground): No. 3/0 minimum or as shown on drawings. Telecommunications Bonding Backbone (TBB; TMGB to TGB): No. 3/0 minimum or as shown on drawings. Telecommunications Grounding Equalizer (GE): No. 3/0 minimum or as shown on drawings. Bonding Conductors shall be insulated with a Green Jacket or jacket marked with Green Tape or labeled per NEC Guidelines.
31 F 32 w 33 si 34 35 35 B 36 ir 37 38 39 40 41 42 43 44 45 46 B 47 N	 Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size. Conductors for Telecommunications shall be as follows: Telecommunications Bonding Conductor (TMGB to Service Ground): No. 3/0 minimum or as shown on drawings. Telecommunications Bonding Backbone (TBB; TMGB to TGB): No. 3/0 minimum or as shown on drawings. Telecommunications Grounding Equalizer (GE): No. 3/0 minimum or as shown on drawings. Bonding Conductors shall be insulated with a Green Jacket or jacket marked with Green Tape or labeled per NEC Guidelines. BUS/BUSBAR
31 F 32 w 33 si 34 35 35 B 36 ir 37 38 39 40 41 42 43 44 45 46 B 47 N 48 48	 Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size. Conductors for Telecommunications shall be as follows: Telecommunications Bonding Conductor (TMGB to Service Ground): No. 3/0 minimum or as shown on drawings. Telecommunications Bonding Backbone (TBB; TMGB to TGB): No. 3/0 minimum or as shown on drawings. Telecommunications Grounding Equalizer (GE): No. 3/0 minimum or as shown on drawings. Bonding Conductors shall be insulated with a Green Jacket or jacket marked with Green Tape or labeled per NEC Guidelines. BUS/BUSBAR Material: Copper (aluminum not permitted).
31 F 32 w 33 si 34 35 35 B 36 ir 37 38 39 40 41 42 43 44 45 44 45 46 B 47 N 48 49 S	 Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors noreased in size. Conductors for Telecommunications shall be as follows: Telecommunications Bonding Conductor (TMGB to Service Ground): No. 3/0 minimum or as shown on drawings. Telecommunications Bonding Backbone (TBB; TMGB to TGB): No. 3/0 minimum or as shown on drawings. Telecommunications Grounding Equalizer (GE): No. 3/0 minimum or as shown on drawings. Bonding Conductors shall be insulated with a Green Jacket or jacket marked with Green Tape or labeled per NEC Guidelines. BUS/BUSBAR Material: Copper (aluminum not permitted).
31 F 32 w 33 si 34 35 35 B 36 ir 37 38 39 40 41 42 43 44 45 B 47 N 48 49 S 50 S0	 Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size. Conductors for Telecommunications shall be as follows: Telecommunications Bonding Conductor (TMGB to Service Ground): No. 3/0 minimum or as shown on drawings. Telecommunications Bonding Backbone (TBB; TMGB to TGB): No. 3/0 minimum or as shown on drawings. Telecommunications Grounding Equalizer (GE): No. 3/0 minimum or as shown on drawings. Bonding Conductors shall be insulated with a Green Jacket or jacket marked with Green Tape or labeled per NEC Guidelines. BUS/BUSBAR Material: Copper (aluminum not permitted). Size:
31 F 32 w 33 si 34 35 35 B 36 ir 37 38 39 40 41 42 43 44 45 B 47 N 48 49 S 50 51	 Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size. Conductors for Telecommunications shall be as follows: Telecommunications Bonding Conductor (TMGB to Service Ground): No. 3/0 minimum or as shown on drawings. Telecommunications Bonding Backbone (TBB; TMGB to TGB): No. 3/0 minimum or as shown on drawings. Telecommunications Grounding Equalizer (GE): No. 3/0 minimum or as shown on drawings. Bonding Conductors shall be insulated with a Green Jacket or jacket marked with Green Tape or labeled per NEC Guidelines. BUS/BUSBAR Material: Copper (aluminum not permitted). Size: All Power systems: 1/4" X 2", length as needed (24" minimum). Telecommunications Main Ground Busbar (TMGB): 1/4" x 4" x 20" long (minimum).
31 F 32 w 33 si 34 35 35 B 36 ir 37 38 38 C 39 40 41 42 43 44 45 B 47 N 48 50 51 52	 Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size. Conductors for Telecommunications shall be as follows: Telecommunications Bonding Conductor (TMGB to Service Ground): No. 3/0 minimum or as shown on drawings. Telecommunications Bonding Backbone (TBB; TMGB to TGB): No. 3/0 minimum or as shown on drawings. Telecommunications Grounding Equalizer (GE): No. 3/0 minimum or as shown on drawings. Bonding Conductors shall be insulated with a Green Jacket or jacket marked with Green Tape or labeled per NEC Guidelines. SUS/BUSBAR Material: Copper (aluminum not permitted). Size: All Power systems: 1/4" X 2", length as needed (24" minimum). Telecommunications Main Ground Busbar (TMGB): 1/4" x 4" x 20" long (minimum). Telecommunications Main Ground Busbar (TGB): 1/4" x 2" x 12" long (minimum). Telecommunications Main Ground Busbar (TGB): 1/4" x 2" x 12" long (minimum). Telecommunications Main Ground Busbar (TGB): 1/4" x 2" x 12" long (minimum). Telecommunications Grounding Busbar (TGB): 1/4" x 2" x 12" long (minimum). Telecommunications Main Ground Busbar (TGB): 1/4" x 2" x 12" long (minimum). Telecommunications Grounding Busbar (TGB): 1/4" x 2" x 12" long (minimum). Telecommunications Grounding Busbar (TGB): 1/4" x 2" x 12" long
31 F 32 w 33 si 34 si 35 B 36 ir 37 si 38 C 39 40 41 42 43 44 45 B 47 N 48 50 51 52 53 53	 Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size. Conductors for Telecommunications shall be as follows: Telecommunications Bonding Conductor (TMGB to Service Ground): No. 3/0 minimum or as shown on drawings. Telecommunications Bonding Backbone (TBB; TMGB to TGB): No. 3/0 minimum or as shown on drawings. Telecommunications Grounding Equalizer (GE): No. 3/0 minimum or as shown on drawings. Bonding Conductors shall be insulated with a Green Jacket or jacket marked with Green Tape or labeled per NEC Guidelines. BUS/BUSBAR Material: Copper (aluminum not permitted). Size: All Power systems: 1/4" X 2", length as needed (24" minimum). Telecommunications Main Ground Busbar (TMGB): 1/4" x 4" x 20" long (minimum). Telecommunications Grounding Busbar (TGB): 1/4" x 2" x 12" long (minimum).
31 F 32 w 33 si 34 si 35 B 36 ir 37 si 38 C 39 40 41 42 43 44 45 B 47 N 48 49 50 51 52 53 53 54	 Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size. Conductors for Telecommunications shall be as follows: Telecommunications Bonding Conductor (TMGB to Service Ground): No. 3/0 minimum or as shown on drawings. Telecommunications Bonding Backbone (TBB; TMGB to TGB): No. 3/0 minimum or as shown on drawings. Telecommunications Grounding Equalizer (GE): No. 3/0 minimum or as shown on drawings. Bonding Conductors shall be insulated with a Green Jacket or jacket marked with Green Tape or labeled per NEC Guidelines. BUS/BUSBAR Material: Copper (aluminum not permitted). Size:
31 F 32 w 33 si 34 si 35 B 36 ir 37 si 38 C 39 40 41 42 43 44 45 B 47 N 48 49 50 51 51 52 53 54 55 54	 Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size. Conductors for Telecommunications shall be as follows: Telecommunications Bonding Conductor (TMGB to Service Ground): No. 3/0 minimum or as shown on drawings. Telecommunications Bonding Backbone (TBB; TMGB to TGB): No. 3/0 minimum or as shown on drawings. Telecommunications Grounding Equalizer (GE): No. 3/0 minimum or as shown on drawings. Bonding Conductors shall be insulated with a Green Jacket or jacket marked with Green Tape or labeled per NEC Guidelines. BUS/BUSBAR Material: Copper (aluminum not permitted). Size: All Power systems: 1/4" X 2", length as needed (24" minimum). Telecommunications Main Ground Busbar (TMGB): 1/4" x 4" x 20" long (minimum). Telecommunications Grounding Busbar (TGB): 1/4" x 2" x 12" long (minimum).
31 F 32 w 33 si 34 si 35 B 36 ir 37 38 38 C 39 40 41 42 43 44 45 B 47 N 48 49 50 51 51 52 53 54 B 55 56	 Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size. Conductors for Telecommunications shall be as follows: Telecommunications Bonding Conductor (TMGB to Service Ground): No. 3/0 minimum or as shown on drawings. Telecommunications Bonding Backbone (TBB; TMGB to TGB): No. 3/0 minimum or as shown on drawings. Telecommunications Grounding Equalizer (GE): No. 3/0 minimum or as shown on drawings. Bonding Conductors shall be insulated with a Green Jacket or jacket marked with Green Tape or labeled per NEC Guidelines. SUS/BUSBAR Material: Copper (aluminum not permitted). Size: All Power systems: 1/4" X 2", length as needed (24" minimum). Telecommunications Main Ground Busbar (TGB): 1/4" x 4" x 20" long (minimum). Telecommunications Grounding Busbar (TGB): 1/4" x 2" x 12" long (minimum).
31 F 32 w 33 si 34 si 35 B 36 ir 37 38 38 C 39 40 41 42 43 44 45 B 47 N 48 49 50 51 52 53 54 B 55 56 56 57	 Feeder and Branch Circuit Equipment Ground: Size as shown on drawings, specifications or as required by NFPA 70, whichever is larger. Differentiate between the normal ground and the isolated ground when both are used at the same facility. Branch Circuit Equipment Ground shall be proportionately increased in size when routed with phase conductors ncreased in size. Conductors for Telecommunications shall be as follows: Telecommunications Bonding Conductor (TMGB to Service Ground): No. 3/0 minimum or as shown on drawings. Telecommunications Bonding Equalizer (GE): No. 3/0 minimum or as shown on drawings. Bonding Conductors shall be insulated with a Green Jacket or jacket marked with Green Tape or labeled per NEC Guidelines. SUS/BUSBAR Material: Copper (aluminum not permitted). Size: All Power systems: 1/4" X 2", length as needed (24" minimum). Telecommunications Ground Busbar (TMGB): 1/4" x 4" x 20" long (minimum). Telecommunications Grounding Busbar (TGB): 1/4" x 4" x 20" long (minimum). Telecommunications Grounding Busbar (TGB): 1/4" x 4" x 20" long (minimum). Telecommunications Grounding Busbar (TGB): 1/4" x 4" x 20" long (minimum). Telecommunications Grounding Busbar (TGB): 1/4" x 4" x 20" long (minimum). Telecommunications Grounding Busbar (TGB): 1/4" x 4" x 20" long (minimum). Telecommunications Grounding Busbar (TGB): 1/4" x 4" x 20" long (minimum). Telecommunications Grounding Busbar (TGB): 1/4" x 4" x 20" long (minimum). Telecommunications Grounding Busbar (TGB): 1/4" x 4" x 20" long (minimum). Telecommunications Grounding Busbar (TGB): 1/4" x 4" x 20" long (minimum). Telecommunications Grounding Busbar (TGB): 1/4" x 4" x 20" long (minimum). Telecommunications Grounding Busbar (TGB): 1/4" x 4" x 20" long (minimum). Telecommunications Grounding Busbar (TGB): 1/4" x 4" x 20" long (minimum). Telecom

1 2 3	Provide main ground busbar located adjacent to main electrical service equipment to terminate all ground conductors. Refer to grounding detail on plans.
4	
5	PART 3 - EXECUTION
6	
7 8 9	Install Products in accordance with manufacturer's instructions.
10 11 12	Mechanical connections shall be accessible for inspection and checking. No insulation shall be installed over mechanical ground connections.
 13 14	Ground connection surfaces shall be cleaned and all connections shall be made so that it is impossible to move them.
15 16	Attach grounds permanently before permanent building service is energized.
17 18 19	Terminate each grounding conductor on its own terminal lug. Sharing a single lug by multiple conductors is not allowed.
20 21 22	All grounding electrode conductors and individual grounding conductors shall be installed in PVC conduit, in exposed locations.
23 24 25 26 27	LESS THAN 600 VOLT ELECTRICAL SYSTEM GROUNDING Equipment Grounding Conductor: Provide separate, insulated equipment grounding conductor within each raceway. Terminate each end on suitable lug, bus, enclosure or bushing. Provide a ground wire from each device to the respective enclosure.
28 29 30	Bond together system neutrals, service equipment enclosures, exposed non-current carrying metal parts of electrical equipment, metal raceway systems, grounding conductor in raceways and cables, receptacle ground connectors, and plumbing systems.
32 33 34 35 36	COMMUNICATION SYSTEM GROUNDING Grounding and Bonding System for Communications shall be an isolated grounding system with a single ground point. That ground point is to be the common grounding electrode system at the building electrical service entrance (main ground bar located in electrical room).
37 38 39	The system shall be compliant with ANSI J-STD-607-B with the exception that the ground cable shall not be bonded to building steel except at the electrical service entrance.
40 41 42 43	Provide Grounding Busbar for Telecommunications at each Telecommunications Room, the Main Equipment Room and at the electrical service entrance per project drawings. Coordinate Busbar location(s) and conductor routing per drawings with Division 27 contractor.
44 45 46 47	Provide Telecommunications Bonding Conductor from Telecommunications Main Grounding Busbar (TMGB) at the Communications Entrance Facility to building common grounding electrode system. Attach grounding conductor to building steel as allowed only at the main electrical service entrance. Provide physical protection as required.
48 49 50	Provide Telecommunications Bonding Backbone (TBB) conductor from the TMGB to Telecommunications Grounding Busbar (TGB) at each Telecommunication Room, Telecommunications Equipment Room and Telecommunications Enclosure
51 52 53 54	TBB shall be continuous and not connected through Telecommunications Grounding Busbars (TGBs). Bond TGBs to TBB via tap off of TBB. Gauge of conductor to be same at TBB. Leave 10 feet slack in conductor from TBB to TGB at TGB location(s). Do not bond TBB or TGB to building steel at TGB location(s).
55 56 57	FIELD QUALITY CONTROL Inspect grounding and bonding system conductors and connections for tightness and proper installation.

- 1 Provide resistance test at each electrical and telecommunications Busbar to ground.
- 2 3 IDENTIFICATION AND LABELING
- 4 Label Grounds at point of termination.
- 5 6 **CONSTRUCTION VERIFICATION**
- 7 Record locations of all electrical and telecommunications grounding electrodes, busbars and grounding conductors as
- 8 installed including recorded ground resistance test results.
- 9
- 10 WARRANTY
- 11 See Division 1, General Conditions, and General Requirements.
- 12 13
| 1
2
3 | SECTION 26 05 29
HANGERS AND SUPPORTS FOR ELECTRICAL SYSTEMS |
|--------------|---|
| 4
5 | PART 1 - GENERAL |
| 6 | |
| 7 | SCOPE |
| 8
9
10 | The work under this section includes conduit and equipment supports, straps, clamps, steel channel, etc., and fastening hardware for supporting electrical work. Included are the following topics: |
| 10 | |
| 12 | FART 1 - GENERAL
Scope |
| 13 | Scope
Belated Work |
| 14 | Submittals |
| 15 | Quality Assurance |
| 16 | |
| 17 | PART 2 - PRODUCTS |
| 18 | Material |
| 19 | |
| 20 | PART 3 - EXECUTION |
| 21 | Installation |
| 22 | |
| 23 | RELATED WORK |
| 24 | Applicable provisions of Division 1 govern work under this Section. |
| 25 | |
| 26
27 | Section 26 05 53 – Identification for Electrical Systems |
| 28 | SUBMITTALS |
| 29 | Product Data: Provide data for support channel. |
| 30 | |
| 30 | ΟΠΑΓΙΤΧ ΔΩΣΙΙΒΑΝCE |
| 32
33 | Support systems shall be adequate for weight of equipment and conduit, including wiring, which they carry. |
| 34 | PART 2 - PRODUCTS |
| 35 | |
| 36 | MATERIAL |
| 37 | Support Channel |
| 38 | • Epoxy Painted: |
| 39 | • Strut shall be made from steel meeting the minimum mechanical properties of ASTM A1011 SS |
| 40
41 | Grade 33, then painted with water born epoxy applied by a cathodic electro-deposition process. |
| 42 | • All fittings and hardware shall be zinc plated in accordance with ASTM B633 (SC3 for fittings. SC1 |
| 43 | for threaded hardware). |
| 44 | |
| 45 | Hot-dip Galvanized Steel |
| 46 | Strut shall be made from steel meeting the minimum mechanical properties of ASTM A1011 SS |
| 47 | Grade 33 and shall be bot-din galvanized after fabrication in accordance with ASTM A123 |
| 18 | |
| 40
/0 | Eithings shall be manufactured from steel meeting the minimum requirements of ASTM AQA SS |
| 49
50 | Grade 22 and hot-dingalyanized after fabrication in accordance with ASTM A122 |
| 50 | Grade 55, and not-dip gaivanized after fabrication in accordance with ASTM A125. |
| 21 | All bardware shall be staipless step! Type 204 or shremium size ACTM 51126 Cr. 2 |
| 52
52 | • All hardware shall be stallless steel type 304 or chromium zinc ASTM F1136 Gr. 3. |
| 53 | |
| 54 | • All hot-dip galvanized after fabrication products must be returned to point of manufacture after |
| 55 | coating for inspection and removal of all sharp burrs. |
| 56 | |
| 57 | Stainless Steel: |

1 All strut, fittings and hardware shall be made of AISI Type 304 or Type 316 stainless steel as 0 2 indicated. 3 4 **Conduit Supports** 5 Conduit clamps, straps, supports, etc., shall be steel or malleable iron. ٠ 6 7 One-hole straps shall be heavy duty type. All straps shall have steel or malleable backing plates when rigid • 8 steel conduit is installed on the interior or exterior surface of any exterior building wall. 9 10 Above suspended ceilings, bar joist conduit hangers: Spring Steel Clips with Snap-Close Clamps (Conduit Supports): Conduit clamps shall pivot a full 360 degrees and shall snap close around the conduit. Push-in 11 12 type conduit clamps are not allowed. Spring clips shall require a hammer to install onto supporting surface. 13 14 Stud wall applications: Spring Steel Clips with Push-in or Snap-Close Conduit Clamps (Conduit Supports): ٠ 15 Conduit clamps shall pivot a full 360 degrees. Spring clips shall require a fastener to install onto stud. 16 17 Box/conduit hanger with rod/wire clip (a.k.a. antlers): One assembly provides support for electrical box and 18 conduit from drop wire or rod. Conduit clamps shall snap close around the conduit. 19 20 • Spring Steel Clip products shall be provided with corrosion resistance and be warranted against failure from 21 corrosion for a period of ten (10) years from date of manufacture. 22 Threaded Rod: Minimum sized threaded rod for supports shall be 3/8" for trapezes and single conduits 1-1/4" and 23 24 larger, and ¼" for single conduits 1" and smaller. 25 26 Hardware: Corrosion resistant, or as noted for each product above. 27 28 **PART 3 - EXECUTION** 29 30 INSTALLATION 31 Fasten hanger rods, conduit clamps, and outlet-, junction-, and pull-boxes to building structure using pre-cast insert 32 system, preset inserts, beam clamps, or expansion anchors. 33 34 Use toggle bolts or hollow wall fasteners in hollow masonry, plaster, or gypsum board partitions and walls; expansion 35 anchors or preset inserts in solid masonry walls; self-drilling anchors or expansion anchors on concrete surfaces; sheet 36 metal screws in sheet metal studs and wood screws in wood construction. If nail-in anchors are used, they must be 37 removable type anchors. 38 39 Powder-actuated fasteners are not permitted. 40 41 Do not fasten supports to piping, ductwork, mechanical equipment, cable tray or conduit. Do not fasten to suspended 42 ceiling grid system. 43 44 Do not drill structural steel members unless approved by City of Madison. 45 46 Furnish and install all supports as required to fasten all electrical components required for the project, including free 47 standing supports required for those items remotely mounted from the building structure, catwalks, walkways etc. 48 49 Fabricate supports from galvanized structural steel or steel channel, rigidly welded or bolted to present a neat 50 appearance. Use hexagon head bolts with spring lock washers under all nuts. 51 52 Support Channel 53 Use one of the following types of support channel as appropriate for the installed environment: 54 Indoor: Epoxy Painted Steel, Hot-dipped Galvanized Steel, or as noted on the drawings. • 55 56 Exterior and wet locations: Hot-dipped Galvanized Steel or Stainless Steel, as appropriate for the • 57 environment or as noted on the drawings. Type 316 stainless steel shall be used for Food Service type 58 environments. Epoxy painted support channel shall not be used for exterior installations.

1		
2	•	Field cuts: File and de-bur cut ends of support channel and paint to prevent rusting. For epoxy-painted
3		support channel, paint cut ends to match the original color. For hot-dipped galvanized support channel,
4		spray cut ends with cold galvanized paint.
5		
6	Support	Wires
7	•	Support wires that are installed in addition to the ceiling grid support wires to provide secure support for
8		raceways, cables assemblies, boxes, cabinets, and fittings shall be secured at both ends (e.g. the ceiling
9		structure at the top and the ceiling grid at the bottom) per NEC 300.11(A).
10		
11	•	Compressed-air power-actuated fasteners may ONLY be used for the installation of separate ceiling wires
12		required for support of conduits and aircraft cable hung light fixtures.
13		
14	•	Support wires shall be identified per specification section 26 05 53.
15		
16	Spring St	eel Clip Conduit Supports
17	•	Above suspended ceilings: Spring steel clips with snap-close clamps may be used to support conduit from
18		bar joist (steel truss) systems above suspended ceilings.
-0 19		
20	•	Stud wall applications: Spring steel clips with push-in or snap-close conduit clamps may be used to support
21		conduit in interior metal stud wall applications. Use screw fasteners to install conduit clamp onto stud
22		
22	•	Boy/conduit hanger with rod/wire clin (a k a antlers). These may only be used in limited applications with
23 24	·	the pre-approval of the State of Wisconsin Electrical Inspector
25		the pre-approval of the state of wisconsin Electrical inspector.
26		END OF SECTION
20		

Page Intentionally Left Blank

1	SECTION 26 05 33
2	RACEWAY AND BOXES FOR ELECTRICAL SYSTEMS
3	
4	
5	PART 1 - GENERAL
6	
/	SCOPE
8	Inis section describes the products and execution requirements relating to furnishing and installing raceways and
9	for the previous line location of a factoway system for electrical, communications, and other low-voltage systems
10	for the project. Included are the following topics.
12	PART 1 - GENERAL
13	Scope
14	Related Work
15	References
16	Submittals
17	
18	PART 2 - PRODUCTS
19	General
20	Rigid Metal Conduit (RMC) and Fittings
21	PVC Coated Rigid Metal Conduit
22	Intermediate Metal Conduit (IMC) and Fittings
23	Electrical Metallic Tubing (EMT) and Fittings
24	Liquidtight Hexible Metal Conduit (LFMC) and Fittings
25	Conduit Supports
20	Pull and Junction Boxes
27	Ouliel Boxes Roxes for Audio Video Equipment
20	Boxes for Fire Alarm Audio-Visual Notification Appliances
30	boxes for the Alarm Audio Visual Notification Appliances
31	PART 3 - EXECUTION
32	Conduit Sizing, Arrangement, and Support
33	Conduit Installation
34	Conduit Installation Schedule
35	PVC Coated Rigid Metal Conduit Installation
36	Coordination of Box Locations
37	Pull and Junction Box Installation
38	Outlet Box Installation
39	Audio-Video System Box and Conduit Installation
40	
41	RELATED WORK
42	Applicable provisions of Division 1 govern work under this section.
43	Section 26 0E 26 Crounding and Bonding for Electrical Systems
44 15	Section 26.05.20 – Grounding and Supports for Electrical Systems
45	Section 26 27 02 – Faugers and Supports for Electrical Systems
47	Section 26 27 26 – Wiring Devices.
48	
49	REFERENCES
50	ANSI/TIA-569-C-Telecommunications Pathways and Spaces
51	
52	SUBMITTALS
53	Boxes - provide product data showing configurations, finishes, dimensions, and manufacturer's instructions.
54	

1 **PART 2 - PRODUCTS** 2 3 GENERAL 4 All steel fittings and conduit bodies shall be galvanized. 5 6 No cast metal or split-gland type fittings permitted. 7 8 Mogul-type condulets larger than 2 inch (50 mm) not permitted except as approved or detailed. 9 10 All condulet covers must be fastened to the condulet body with screws and be of the same manufacture. 11 12 C-condulets shall not be used in lieu of pull boxes. 13 14 All boxes shall be of sufficient size to provide free space for all conductors enclosed in the box and shall comply with 15 NEC requirements. 16 17 **RIGID METAL CONDUIT (RMC) AND FITTINGS** 18 Conduit: Heavy wall threaded, galvanized steel, schedule 40. 19 20 Fittings and Conduit Bodies: Use all steel threaded fittings and conduit bodies. 21 22 Expansion Fittings/Expansion Joints: Expansion Fittings shall be Internal Grounding type and shall not rely on external 23 bonding jumpers to maintain grounding continuity between raceway components. 24 25 PVC COATED RIGID METAL CONDUIT 26 PVC Externally Coated Conduit: Rigid heavy wall, schedule 40, steel conduit with external 40 mil (0.1 mm) PVC 27 coating. Conduit must be hot dipped galvanized inside and out including threads. The PVC coating bond to the 28 galvanized steel conduit shall be stronger than the tensile strength of the coating itself. 29 30 Fittings and Conduit Bodies: Threaded type, material to match conduit. PVC coated fittings and couplings shall have 31 specially formed sleeves to tightly seal to conduit PVC coating. The sleeves shall extend beyond the fitting or coupling 32 a distance equal to the pipe outside steel diameter or two inches (50 mm) whichever is greater. 33 34 INTERMEDIATE METAL CONDUIT (IMC) AND FITTINGS 35 Conduit: Galvanized steel, threaded. 36 37 Fittings and Conduit Bodies: Use all steel threaded fittings and conduit bodies. 38 39 Expansion Fittings/Expansion Joints: Expansion Fittings shall be Internal Grounding type and shall not rely on external 40 bonding jumpers to maintain grounding continuity between raceway components. 41 42 **ELECTRICAL METALLIC TUBING (EMT) AND FITTINGS** 43 Conduit: Steel, galvanized tubing. 44 45 Fittings: All steel, set screw type. No push-on or indenter types permitted. 46 Conduit Bodies: All steel threaded conduit bodies. 47 48 LIQUIDTIGHT FLEXIBLE METAL CONDUIT (LFMC) AND FITTINGS 49 Conduit: flexible, steel, galvanized, spiral strip with an outer Liquidtight, nonmetallic, sunlight-resistant jacket. 50 51 Fittings and Conduit Bodies: ANSI/NEMA FB 1, compression type. There shall be a metallic cover/insert on the end of 52 the conduit inside the connector housing to seal the cut conduit end. 53 54 **RIGID POLYVINYL CHLORIDE CONDUIT (PVC) AND FITTINGS** 55 Conduit: Rigid non-metallic conduit, Schedule 40 PVC minimum, Listed, sunlight resistant, rated for 90° C conductors. 56 Schedule 80 for locations exposed to physical damage or as required. 57 58 Fittings and Conduit Bodies: NEMA TC 2, Listed.

1	
2	CONDUIT SUPPORTS
3	See section 26 05 29.
4	
5	PULL AND JUNCTION BOXES
6 7	Interior Sheet Metal Boxes: code gauge galvanized steel, screw covers, flanged and spot welded joints and corners.
8	Interior Sheet Metal Boxes larger than 12 inches (300 mm) in any dimension shall have a hinged cover or a chain
9	installed between box and cover.
10	
11	Boyes 9 square feet and larger shall have hinged covers. Single covers shall not exceed 10 square feet
12	
12	Interior Sheet Metal Boxes connected to an exterior underground raceway shall have a drain hole located in the
14	bottom of the box.
15	
16	Exterior Boxes and Wet Location Installations: Type 4 and Type 6, flat-flanged, surface-mounted junction box, UL
17	listed as rain-tight. PVC box and cover with ground flange, neoprene gasket, and stainless steel cover screws.
18	
19	Box extensions and adjacent boxes within 48 inches of each other are not allowed for the purpose of creating more
20	wire capacity.
21	· · · · · · · · · · · · · · · · · ·
22	Junction boxes 6 inch-by-6 inch or larger size shall be without stamped knock-outs.
23	
24	Wireways shall not be used in lieu of junction boxes.
25	
26	OUTLET BOXES
27	Sheet Metal Outlet Boxes: galvanized steel, with stamped knockouts.
28	
29	Luminaire and Equipment Supporting Boxes: Rated for weight of equipment supported; include 3/8 inch male fixture
30	studs where required.
31	
32	Concrete Ceiling Boxes: Concrete type.
33	
34	Cast Boxes: Cast ferroalloy or aluminum, deep type, gasketed cover, threaded hubs.
35	
36	BOXES FOR AUDIO-VIDEO EQUIPMENT
37	Provide floor, wall, and/or ceiling boxes for Audio-Video (AV) Equipment as indicated on the Electrical and/or Audio-
38	Video drawings.
39	
40	FLAT SCREEN MONITOR BOXES
41	Provide a recessed wall box for mounting behind flat screen monitors, allowing the screens to sit flush against the
42	wall. These boxes shall provide a neat and secure environment for the audio, video, control and power connections.
43	
44	The recessed wall box shall install easily between any two standard studs in the wall. Connections and cable entry can
45	be on the top or the bottom depending on installation preference.
46	
47	The recessed wall box shall be provided with one low-voltage conduit entry box and Nationally Recognized Testing
48	Laboratory (NRTL) listed single gang box for AC power.
49	The recessed wall box cover shall be provided in white or black and shall be suitable for painting. The cover shall have
50	a cable exit slot for the display connections and the excess cable can easily be hidden inside of the box making the
51	entire installation as clean as possible. The cover screws onto the front of the box once all connections are in place.
52	
53	The recessed wall box shall be designed for new or existing construction. Brackets shall be included for mounting to
54	studs in new construction as well as surface mount clips for mounting to sheet rock or plywood in existing
55	construction.
56	

57 BOXES FOR FIRE ALARM AUDIO-VISUAL NOTIFICATION APPLIANCES

1 2 3	Recessed boxes for Fire Alarm audio, visual, and audio-visual notification appliances shall be galvanized steel sheet metal with stamped knockouts. Boxes shall be painted red.
5 4 5 6	For surface mounting, use manufacturer supplied <u>back boxes</u> and <u>trim plates</u> , painted red or off white to match device color, and shall contain no visible conduit knock-outs. Mark each device with its circuit number.
7	PART 3 - EXECUTION
8	
9	CONDUIT SIZING, ARRANGEMENT, AND SUPPORT
10 11 12	EMT is permitted to be used in sizes 4 inch (100 mm) and smaller for power and low-voltage systems. See CONDULT INSTALLATION SCHEDULE below for other limitations for EMT and other types of conduit.
12	Size nower conductor receivers for conductor type installed. Conduit size shall be 1/2 inch (16 mm) minimum excent
14 15 16	all homerun conduits shall be 3/4 inch (21 mm), or as specified elsewhere. Caution: Per the NEC, the allowable conductor ampacity is reduced when more than three current-carrying conductors are installed in a raceway. Contractor must take the NEC ampacity adjustment factors into account when sizing the raceway and wiring
17 18	system.
19	Size communications and other low-voltage systems raceways as follows:
20 21 22	Communications, including Outlet Box: 1 inch minimum. Conduit used for single device locations (e.g. Wireless Access Point, Video Surveillance Camera, and Wall mounted telephone) may be 3/4 inch minimum.
23 24	Control, security, signal, video, and other low-voltage applications: 3/4 inch minimum.
25	Fire Alarm: 1/2 inch minimum.
20 27 28	Provide one raceway from each communications outlet box to above accessible ceiling.
29 30	Arrange conduit to maintain 6'-8" clear headroom and present a neat appearance.
31 32	Route exposed conduit and conduit above accessible ceilings parallel and perpendicular to walls and adjacent piping.
33 34 35	Maintain minimum 6 inch (150 mm) clearance between conduit and piping. Maintain 12 inch (300 mm) clearance between conduit and heat sources such as flues, steam pipes, and heating appliances.
36 37 38 39	Arrange conduit supports to prevent distortion of alignment by wire pulling operations. Fasten conduit using galvanized pipe straps, conduit racks (lay-in adjustable hangers), clevis hangers, or bolted split stamped galvanized hangers.
40 41 42	Group conduit in parallel runs where practical and use conduit rack (lay-in adjustable hangers) constructed of steel channel with conduit straps or clamps. Provide space for 25 percent additional conduit.
43 44	Do not fasten conduit with wire or perforated pipe straps. Before conductors are pulled, remove all wire used for temporary conduit support during construction.
45 46	Support and fasten metal conduit at a maximum of 8 feet (2.4 m) on center.
47 48 49	Supports shall be independent of the installations of other trades, e.g. ceiling support wires, HVAC pipes, other conduits, etc., unless so approved or detailed.
50 51 52	Conceal all conduits except where noted on the drawings or approved by the Architect/Engineer. Contractor shall verify with Architect/Engineer all surface conduit installations except in mechanical rooms.
53 54 55	Changes in direction shall be made with symmetrical bends, cast steel boxes, stamped metal boxes or cast steel conduit bodies.
56 57	For indoor conduits, no continuous conduit run shall exceed 100 feet (30 meters) without a junction box.
58	All conduits installed in exposed areas shall be installed with a box offset before entering box.

1	
2	CONDUIT INSTALLATION
3	Cut conduit square; de-burr cut ends.
4	
5 6	Conduit shall not be fastened to the corrugated metal roof deck.
7 8	Bring conduit to the shoulder of fittings and couplings and fasten securely.
9	Use conduit hubs for fastening conduit to cast boxes. Use sealing locknuts or conduit hubs for fastening conduit to
10	sheet metal boxes in damp or wet locations.
11	
12	Threads cut in the field, and factory threads of conduit and nipples not coated with corrosion protection, shall be
13	coated with an approved electrically conductive corrosion compound per NEC 300.6.
14	Corrosion inhibitor, when used in a food service environment, shall be approved for Food Service locations.
15	
16 17 18	Terminate all conduit (except for terminations into conduit bodies) using conduit hubs, or connectors with one locknut, or utilize double locknuts (one each side of box wall).
19	Provide bushings for the ends of all conduit not terminated in a box. Refer to Section 26 05 26 – Grounding and
20	Bonding for Electrical Systems for grounding bushing requirements.
21	
22	Provide insulated bushings where raceways contain 4 AWG or larger conductors.
23	
24	Communication and Low Voltage systems conduits shall terminate in horizontal plane.
25	
26	Install no more than the equivalent of:
2/	Inree 90 degree bends between boxes for electrical systems.
28 20	Two 90 degree bands between boxes for communications and other low voltage systems. Note: Offsets
30	shall be considered 90 degrees
31	
32	No single bend may exceed 90 degrees.
33	
34	Use hydraulic one-shot conduit bender or factory elbows for bends in conduit larger than 2 inch (50 mm) size unless
35	sweep elbows are required.
36	
37	Bend conduit according to manufacturer's recommendations. Torches or open flame shall not be used to aid in
38	bending of PVC conduit.
39	
40 //1	Use suitable conduit caps of other approved seals to protect installed conduit against entrance of dirt and moisture.
42	Provide 1/8 inch (3 mm) pylon pull string in empty conduit, except sleeves and pipples.
43	
44	Install listed expansion-deflection fitting or other approved means shall be used where a raceway crosses a structural
45	joint for expansion, contraction or deflection, used in buildings, bridges, parking garages or other structurers.
46	
47	Route conduit through roof openings for piping and ductwork where possible.
48	
49	Where communication cabling is to be installed in conduit to the wiring hub (e.g. Telecom Room), multiple conduits
50	may be consolidated into fewer, larger conduits. Capacity of shared conduits shall equal the capacity of the individual
51	conduits unless otherwise noted.
52	Lies NDTL listed westelling strugging along a contracting conduit to achieve and
ンジ 54	Use INKI L listed metallic grounding clamps when terminating conduit to cable tray.
54 55	Ground and hand conduit under provisions of Section 26.05.26
56	
57	Conduit is not permitted in any slab topping of two inches (50 mm) or less.
58	· · · · · · · · ·

1 2 2	PVC conduit concrete pol	in concrete pole bases shall transition to galvanized rigid metal conduit 12 inches before it enters a e base. Inside the pole base, the elbow shall be galvanized rigid metal conduit. From the elbow, the
3 4	conduit shall	transition back to PVC as it continues up and out the top of the concrete pole base.
5 6	PVC conduit concrete floc	shall transition to galvanized rigid metal conduit before it enters a foundation wall or up through a pr.
7 0	Identify cond	luit under provisions of Section 26.05.52
9	identity cond	
10 11 12	Clean PVC co match weath	onduit with solvent, and dry before application of glue. The temperature rating of glue/cement shall ner conditions. Apply full even coat of cement/glue to entire area that will be inserted into fitting. The
13	chui e histain	
14	CONDUIT IN	STALLATION SCHEDULE
15	Conduit othe	er than that specified below for specific applications shall not be used.
16 17 18	•	Wet Interior Locations: Exposed: Rigid Metal conduit, Schedule 80 PVC conduit, PVC coated Rigid Metal conduit.
19 20 21	•	Concealed Dry Interior Locations: Rigid Metal conduit, Intermediate Metal conduit, Electrical Metallic
22		labile.
23	•	Interior Building Grounding Electrode Conductor: Schedule 80 PVC.
24		
25 26 27	•	Exposed Dry Interior Locations: Rigid Metal conduit, Intermediate Metal conduit, Electrical Metallic Tubing.
27 28	•	Motor and equipment connections: Liquidtight flexible metal conduit (LEMC) (all locations). Minimum
29 30		length shall be one foot (300 mm); maximum length shall be three feet (900 mm). Conduit must be installed perpendicular to direction of equipment vibration to allow conduit to freely flex.
31		
32 33	•	Light fixtures: Refer to specification section 26 51 13.
34	PVC COATED	RIGID METAL CONDUIT INSTALLATION
35 36 37	Installers of I for this type upon request	PVC Coated Rigid Metal Conduit shall be factory trained and certified in the proper installation methods of conduit. Proof of such certification shall be kept on the project site at all times and shall be produced t.
38 30	COORDINAT	
40 41	Provide elect and code cor	trical boxes as shown on Drawings, and as required for splices, taps, wire pulling, equipment connections, npliance.
42		
43 44 45	boxes and ou	itlets in offices and work areas prior to rough-in.
45 46	No outlet, iu	unction, or pull boxes shall be located where it will be obstructed by other equipment, piping, lockers,
47 48	benches, cou	inters, etc.
49	Conduit and	boxes shall not be fastened to the metal roof deck. If conduit and boxes are required to be located and
50 51	installed on i the metal roo	roof decks, the conduit and boxes are required to be spaced minimum 1-5/8 inch off the lowest part of of decking material, per NEC 300.4 (E).
52 53 54	It shall be th with workme	e Contractor's responsibility to study drawings pertaining to other trades, to discuss location of outlets en installing other piping and equipment and to fit all electrical outlets to job conditions.
55 56 57 58	In case of an Architect/Eng	ny question or argument over the location of an outlet, the Contractor shall refer the matter to the gineer and install outlet as instructed by the Architect/Engineer.

The proper location of each outlet is considered a part of this contract and no additional compensation will be paid to
 the Contractor for moving outlets which were improperly located.

3 4

Locate and install boxes to allow access to them. Where installation is inaccessible, coordinate locations and provide 18 inch (450 mm) by 24 inch (600 mm) access doors. Boxes must be installed within 12" from edge of the access door.

- 5 6 7
 - Locate and install to maintain headroom and to present a neat appearance.
- 8

9 Install boxes to preserve fire resistance rating of partitions and other elements, using approved materials and 10 methods.

11

12 PULL AND JUNCTION BOX INSTALLATION

Boxes shall be minimum 4 inches square (100 mm) by 2 1/8 inches (54 mm) deep for use with 1 inch (25 mm) conduit and smaller. On conduit systems using 1 1/4 inch (31.75 mm) conduit, minimum box size shall be 4 11/16 inches square by 2 1/8 inches deep.

16

17 Where used with raceway(s) containing conductors of 4 AWG or larger, box shall be sized as required unless 18 otherwise noted on the drawings.

19

Where used with raceway(s) containing conductors on systems over 600V, size box per NEC 314 Part IV unless otherwise noted as larger on the drawings.

22

23 Size boxes for communications per ANSI/TIA-568-C.

24

Locate boxes above accessible ceilings, in unfinished areas or furnish and install approved access panels in nonaccessible ceilings where boxes are installed. All boxes are to be readily-accessible.

27

Provide boxes for communications and other low voltage applications (a) in any section of conduit longer than 100 feet, (b) where there are bends totaling more than 180 degrees between pull points or pull boxes and (c) wherever there is a reverse bend in run. Locate boxes on straight section of raceway (e.g. do not use boxes in place of raceway bends).

32

33 Support boxes independent of conduit.

34

35 OUTLET BOX INSTALLATION

Do not install boxes back-to-back in walls. Provide minimum 6 inch (150 mm) separation, except provide minimum 24
 inch (600 mm) separation in acoustic-rated walls.

- 3839 Power:
- 40Recessed (1/4 inch maximum) outlet boxes in masonry, concrete, tile construction, or drywall shall be41minimum 4 inch square, with device rings. Device covers shall be square-cut except rounded corner plaster42rings are allowed in drywall applications. Angle cut plaster rings are not permitted. Coordinate masonry43cutting to achieve neat openings for boxes. A single gang box can be used in drywall and masonry, for a44single device location, when a single conduit enters box.45
- Shallow 4 inch square by 1 1/2 inch deep boxes can be used as device boxes for power provided the box and
 plaster ring is sized for installed device and conductors.
- 48
- 49 Low Voltage:
- 50Recessed (1/4 inch maximum) outlet boxes in masonry, concrete, tile construction or drywall shall be51minimum 4 11/16 inch square by 2 1/8 inch deep with single gang device ring (unless noted otherwise on52drawings). Device covers shall be square-cut except rounded corner plaster rings are allowed in drywall53applications. Angle cut plaster rings are not permitted. Coordinate masonry cutting to achieve neat54openings for boxes.55
- 56 Provide one conduit from each communications outlet box. Conduit runs between outlet boxes for 57 communications are not allowed. Terminate conduit above accessible ceiling in corridor.
- 58

2

5

8

10

12

15

18

22

24

26

29

Provide knockout closures for unused openings.

Support boxes independently of conduit except for cast boxes that are connected to two rigid metal conduits, both
 supported within 12 inches (300 mm) of box.

Use multiple-gang boxes where more than one device are mounted together; do not use sectional boxes. Provide
 non-metallic barriers to separate wiring of different voltage systems.

9 Install boxes in walls without damaging wall insulation.

11 Coordinate mounting heights and locations of outlets mounted above counters, benches, and backsplashes.

Ceiling outlets shall be 4 inch square, minimum 2 1/8 inch (54 mm) deep except that concrete boxes and plates will be approved where applicable. Position outlets to locate luminaires as shown on reflected ceiling plans.

16 In inaccessible ceiling areas, position outlets and junction boxes within 6 inches (150 mm) of recessed luminaire, to be 17 accessible through luminaire ceiling opening.

Provide recessed outlet boxes in finished areas; secure boxes to interior wall and partition studs, accurately positioning to allow for surface finish thickness. Use stamped steel stud bridges for flush outlets in hollow stud wall, and adjustable steel channel fasteners for flush ceiling outlet boxes.

23 Align wall-mounted outlet boxes for switches, thermostats, and similar devices.

25 Provide cast ferroalloy or aluminum outlet boxes in exterior and wet locations.

Surface wall outlets shall be 4 inch (100 mm) square with raised covers for one and two gang requirements. For three
 gang or larger requirements, use gang boxes with non-overlapping covers.

30 AUDIO-VIDEO SYSTEM BOX AND CONDUIT INSTALLATION

Conduit requirements for AV systems cabling may differ from those of other trades. It is important that the electrical contractor become familiar with these specialized requirements. AV systems cabling must be enclosed within continuously grounded ferrous metallic conduit or raceway. PVC conduit is not acceptable. Conduit and raceway is to be furnished and installed by electrical contractor. Conduits containing different wiring classes must maintain minimum separations to minimize interferences from electrical noise. Conduits sizes and quantities shown on bid documents are minimums. Separate conduit runs specified in bid documents may not be combined for any purpose.

Conduit runs entering or exiting the audio equipment racks shall be electrically isolated from the racks. PVC or other non-conductive fittings shall be used to isolate the conduit from the audio equipment racks.

Provide AV boxes as shown on the Electrical and/or Audio-Video drawings. Install boxes at heights and locations as
 indicated on the drawings. Coordinate all box installations with the AV equipment provider.

Flat screen monitor boxes shall be installed so that all cabling is concealed behind the monitor. Coordinate box
location with flat screen mounting brackets so that the box cover and cables are not blocked by the brackets.

46 47

40

1 2	SECTION 26 05 53 IDENTIFICATION FOR ELECTRICAL SYSTEMS
3	
4	
5	PART 1 - GENERAL
6	
/	SCOPE
8	The work under this section includes the products and execution requirements relating to labeling of power, control,
9	signaling and fire alarm wiring. Further, this section includes the installation of labels, nameplates, and directories for
10	electrical boxes, wiring devices, and equipment. Included are the following topics:
12	PART 1 - GENERAL
13	Scone
14	Related Work
15	Submittals
16	
17	PART 2 - PRODUCTS
18	Materials
19	
20	PART 3 - EXECUTION
21	General
22	Box Identification
23	Communication Conduit Labeling
24	Power, Control and Signal Wire Identification
25	Wiring Device Identification
26	Support wire identification
27	Namepiate Engraving for Electrical Equipment
20	raneiboard Directories
30	RELATED WORK
31	Applicable provisions of Division 1 shall govern work under this section.
32	
33	Section 26 05 19 – Low-Voltage Electrical Power Conductors and Cables
34	Section 26 05 23 – Control-Voltage Electrical Power Cables
35	
36	SUBMITTALS
37	Include schedule for nameplates.
38	
39	Prior to installation, the Contractor shall provide samples of all label types planned for the project. These samples
40	shall include examples of the lettering to be used. Samples shall be mounted on 8 1/2" x 11" sheets annotated,
41	explaining their purposed use.
4Z 42	
45 ЛЛ	PARTZ - PRODUCTS
44 45	ΜΔΤΕΡΙΔΙ S
46	Labels: All labels shall be permanent, and machine generated NO HANDWRITTEN OR NON-PERMANENT LABELS ARE
47	ALLOWED.
48	
49	Wire Labels: All wiring labels shall be white/transparent vinyl or vinyl-cloth, self-laminating, wraparound type. Flag
50	type labels are not allowed. The labels shall be of adequate size to accommodate the circumference of the cable
51	being labeled and properly self-laminate over the full extent of the printed area of the label.
52	
53	Tape (wiring phase identification only): Scotch #35 tape in appropriate colors for system voltage and phase.
54	
55	Nameplates: Engraved three-layer laminated plastic. Normal system shall use nameplates with black letters on white
56	background, emergency system (NEC 700) shall use white letters on red background.
57	Cas Devidentification and Minim Device Identification costs of the table of the second
58	See Box identification and wiring Device identification sections for allowed usage of permanent marker.

2	PART 3 - EXECUTION	
3 1	GENERAL	
5 6 7	Where mixed voltages are used in one building (e.g. 4160 volt, 480 vo and pull box, equipment, etc., on each system shall be labeled for vo herein.	olt, 208 volt) each switch, switchboard, junction oltage in addition to other requirements listed
9 10 11	All branch circuit and power panels shall be identified with the sa distribution center.	ame symbol used in circuit directory in main
12 13 14	Clean all surfaces before attaching labels with the label manufactur labels firmly as recommended by the label manufacturer. Labels shall l	rer's recommended cleaning agent. Install all be installed plumb and neatly on all equipment.
15 16 17	Install nameplates parallel to equipment lines. Secure nameplates manufacturer approved adhesive or cement.	s to equipment fronts using screws, rivets or
18 19	Embossed tape shall not be permitted for any application.	
20 21 22	Provide all warning labels to electrical equipment as required per NE current labeling to service equipment as required per NEC 110.24.	EC 110.16 and 110.21. Provide available fault
23 24	Fire pump disconnecting means shall be marked as "Fire Pump Disconn	necting Means", per NEC 695.4(B)(3)(c).
25	BOX IDENTIFICATION	
26	The following junction and pull boxes shall be identified utilizing spray	painted covers:
27		
28	System	Color(s)
29	Secondary Power – 480Y/277V	Brown
30	Secondary Power – 208Y/120V, 240/120V	White
31	Fire Alarm (see below)	Red
32	Temperature Control	Green
33	Door Access Control	Orange
34	Communications	Blue
35		
36 37 38 39	 All boxes with power wiring shall be further identified with circuit num All outlet and device boxes shall use machine-generated adhe marker. 	bers and source panel designation as follows: esive labels, or neatly hand-written permanent
40 41 42 43	 All exposed junction and pull boxes larger than 8" square sha minimum letter height. All exposed junction and pull boxes 8 generated adhesive labels. 	all utilize engraved nameplates with ½" 8" square or smaller shall utilize machine-
44 45 46	 All junction and pull boxes located above an accessible ceiling labels, or neatly hand-written permanent marker. 	g shall utilize machine-generated adhesive
47 48 49 50	All fire alarm boxes (covers and outer sides) shall be painted red and labeled "Fire Alarm" or "FA". When red conduit is used for the alarm system installation, there is no need to paint the box sides, - paint the covers only. Non-factory device boxes shall also be painted red.	
51 52	Other system boxes shall be further identified as shown on drawing de	tails or approved shop drawings.
53	COMMUNICATIONS CONDUIT LABELING	
54	Provide label on all conduits installed between Telecommunication I	Equipment Rooms. Both ends of the conduits
55 56	shall be labeled. All labels shall be mechanical, no hand-written labels.	
57 58	The label shall indicate the location of the far end of the conduit run a Room #216 – 01). Refer to agency standards where applicable.	and a unique conduit number. (i.e. TR-1A-01 or

2 POWER, CONTROL AND SIGNALING WIRE IDENTIFICATION

3 Provide wire labels on each conductor in panelboard gutters, all boxes, and at load connection. Identify with branch 4 circuit or feeder number for power and lighting circuits, and with wire number as indicated on schematic and 5 interconnection diagrams or equipment manufacturer's shop drawings for control and signaling wires.

All wiring shall be labeled within 2 to 4 inches of terminations. Each end of a wire or cable shall be labeled as soon as 6 7 it is terminated, including wiring used for temporary purposes. 8

WIRING DEVICE IDENTIFICATION 9

10 Wall switches, receptacles, occupancy sensors, photocells, poke-through fittings, access floor boxes, and time clocks shall be identified with circuit numbers and source (ex. Panel ABC-3). In exposed areas, identifications should be 11 made inside of device covers, unless directed otherwise. Use machine-generated adhesive labels, or neatly hand-12 13 written permanent marker.

14

SUPPORT WIRE IDENTIFICATION 15

16 Support wires that are installed in addition to the ceiling grid support wires to provide secure support for raceways, 17 cables assemblies, boxes, cabinets, and fittings shall be distinguishable from the ceiling grid support wires per NEC 18 300.11(A). This identification shall be either approximately 6 inches of fluorescent orange paint, or orange tape flags 19 3/4 inches high-by-2 inches wide (minimum) within 12 inches of the bottom of the support wires.

20

21 NAMEPLATE ENGRAVING FOR ELECTRICAL EQUIPMENT

Provide nameplates of minimum letter height as scheduled below. 22

23

24 All Panelboards (Distribution, Branch, Sub-feed, and Feed-Through), Switchboards and Motor Control Centers: 1 inch (25 mm); identify equipment designation. 1/2 inch (13 mm); identify voltage rating, source and room location of the 25 source. Panelboards serving NEC 700, 701 or 702 loads shall identify which branch they serve. Both panels in a 26 27 double tub application shall be labeled.

28

29 Circuit Breakers, Switches, and Motor Starters in Distribution Panelboards, Switchboards and Motor Control Centers: 30 1/2 inch (13 mm); identify circuit number and load served, including location.

31

32 Individual Circuit Breakers, Disconnect Switches, Enclosed Switches, and Motor Starters: ½ inch (13 mm); identify source and load served. 33

34

35 PANELBOARD DIRECTORIES

- Update existing directories with typed directories. 36
- 37

Typed directories for panelboards shall be covered with clear plastic, and have a metal frame. Room number on 38 directories shall be Owner's numbers, not Plan numbers unless Owner so specifies. 39

40 41

Page Intentionally Left Blank

SECTION 26 24 16 PANELBOARDS
PART 1 - GENERAL
SCOPE The work under this section includes branch circuit panelboards. Included are the following topics:
PART 1 - GENERAL
Scope
Related Work
References
Submittals Operation and Maintenance Data
Spare Parts
PART 2 - PRODUCTS
Branch Circuit Panelboards
Coordination of Overcurrent Protective Devices
PART 3 - EXECUTION
Installation
Field Quality Control
Applicable provisions of Division 1 govern work under this Section.
REFERENCES
NEMA AB 1 - Molded Case Circuit Breakers
NEMA KS 1 - Enclosed Switches
SUBMITTALS
Include outline and support point dimensions, voltage, main bus ampacity, circuit breaker arrangement and sizes, and interrupting ratings confirming a fully-rated system for all equipment and components.
OPERATION AND MAINTENANCE DATA
All operations and maintenance data shall comply with the submission and content requirements specified under section GENERAL REQUIREMENTS.
SDADE DADTS
Keys: Furnish 2 keys for each panelboard to Owner.
Handle lock-off: Furnish (2) 20/1P circuit breaker handle lock-off devices for each panelboard to Owner.
One set of three spare fuses of each size and type utilized.
PART 2 - PRODUCTS
BRANCH CIRCUIT PANELBOARDS
Lighting and Appliance Branch Circuit Panelboards: Circuit breaker type.
The panelboard and overcurrent devices contained within shall be fully-rated.
Enclosure: Type 1. Minimum cabinet size: 5-3/4 inches (144 mm) deep; 20 inches (508 mm) wide with 5" minimum gutter space top and bottom. Constructed of galvanized code gauge steel. Panel enclosure (back box) shall be of non-stamped type (without KO's) to avoid concentric break out problem.
Provide flush or surface cabinet front with concealed trim clamps, concealed hinge and flush cylinder lock all keyed alike. Front cover shall be hinged to allow access to wiring gutters without removal of panel trim. Hinged trim shall be held in place with screw fasteners. Finish in manufacturer's standard gray enamel.
Provide metal directory holders with clear plastic covers. Holder to be factory mounted.

2

3

4 5

6 7

8 9

10

11 12

13 14 15

16 17

18 19

20 21

22 23

24

27

35

36 37

38

39 40

41 42

43 44

45

46 47

48

49 50

54

- Provide panelboards with copper bus (phase buses, bus fingers, etc.), ratings as scheduled on Drawings.
- Provide ground bars in all panelboards. Phase, neutral and ground bar terminations can be dual rated ALCU9. All spaces shall have bus fully extended and drilled for the future installation of breakers.

Incoming conductors shall terminate at lug landing pads rated for the panelboard.

Provide compression type lugs to accommodate the conductor shown on drawings.

Minimum System (i.e. individual component) Short Circuit Rating: As required by short circuit/ coordination study.

Molded Case Circuit Breakers: Bolt-on type thermal magnetic trip circuit breakers. Provide UL Class A ground fault interrupter circuit breakers where shown on Drawings. Provide circuit breakers UL listed as Type HACR for air conditioning equipment branch circuits.

Do not use tandem circuit breakers.

Circuit breakers shall be bolt-on type with common trip handle for all poles. No handle ties of any sort will be approved.

Provide a minimum of 10% spare circuit breakers in branch panelboards.

All of the panelboards provided under this section shall be by the same manufacturer.

All panelboards installed side by side (double tub) shall utilize same enclosure height.

Double tub panelboard installations shall identify type of feed to adjacent panelboard- sub-feed or feed-thru. Identification shall be integral with panel label.

28 COORDINATION OF OVERCURRENT PROTECTIVE DEVICES

Provide a coordination study of the electrical system and recommend set points for all of the overcurrent and ground fault trip adjustments on the equipment provided. The coordination study and set point recommendations shall be submitted to the consulting engineer for approval as part of the 26 24 16 package. Submittal shall be on or before date of switchboard and panelboard equipment submittal. Failure to include coordination study will result in immediate rejection of the submittal package.

PART 3 - EXECUTION

INSTALLATION

See section 26 05 29 for support requirements.

- Install panelboards plumb with wall finishes.
- Height: Branch panelboards: 6'-0" to top of panelboard.

Install a crimp type stud termination to stranded conductor when terminating on circuit breakers without a captive assembly rated for terminating stranded conductors.

Provide filler plates for unused spaces in panelboards.

See section 26 05 53 for identification requirements. Provide typed circuit directory for each panelboard per NEC 408.4(A). Revise directory to reflect circuiting changes required to balance phase loads.

51
52 Stub three (3) empty ¾" conduits to accessible location above ceiling or below floor out of each recessed panelboard.
53 Cap these conduits to prevent material from entering them.

55 FIELD QUALITY CONTROL

The Contractor shall circuit the panelboards as shown on the drawings. Measure steady state load currents at each panelboard feeder. Should the difference at any panelboard between phases exceed 10 percent, rearrange circuits in the panelboard to balance the phase loads within 10 percent.

Visual and Mechanical Inspection: Inspect for physical damage, proper alignment, anchorage, and grounding. Check
 proper installation and tightness of connections.

62 63

1 2	SECTION 26 27 02 EQUIPMENT WIRING SYSTEMS
3	
4	
5	PART 1 - GENERAL
6	
7	SCOPE
8	The work under this section includes electrical connections to equipment specified under other Divisions and/or
9 10	Sections, of furnished by Owner, including, but not limited to:
10 11	-Misc. Equipment
11 12	-rivac and Fighting motors, VPDs, and panels
13	Included are the following topics:
14	
15	PART 1 - GENERAL
16	Scope
17	Related Work
18	Submittals
19	Coordination
20	
21	PART 2 - PRODUCTS
22 22	Coros and Caps
25 74	
25	PART 3 - EXECUTION
26	Inspection
27	Preparation
28	Installation
29	Miscellaneous Connections
30	HVAC and Plumbing Connections
31	Equipment Connection Schedule
32	
33 21	RELATED WORK
34	Applicable provisions of Division 1 govern work under this section.
36	Section 26 05 19 – Low-Voltage Electrical Power Conductors and Cables
37	Section 26 05 33 – Raceway and Boxes for Electrical Systems
38	
39	SUBMITTALS
40	Product Data: Provide data for cord and wiring devices.
41	
42	COORDINATION
43 44	coordinate all equipment requirements with the various contractors and the Owner. Review the complete set of drawings and specifications to determine the extent of wiring starters devices etc. required
44 15	drawings and specifications to determine the extent of wining, starters, devices, etc., required.
46	PART 2 - PRODUCTS
47	
48	CORDS AND CAPS
49	Straight-blade Attachment Plug: NEMA WD 1.
50	
51	Locking-blade Attachment Plug: NEMA WD 5.
52	
53	Attachment Plug Configuration: Match receptacle configuration at outlet provided for equipment.
54 55	Cord Construction: Ail-resistant thermoset insulated multi-conductor flexible cord with identified equipment
56	grounding conductor, suitable for bard usage in damp locations
57	
58	Cord Size: Suitable for connected load of equipment and rating of branch circuit overcurrent protection.

1	
2	OTHER PRODUCTS
3	Refer to related sections for other product requirements.
4	
5	PART 3 - EXECUTION
6	
7	INSPECTION
8	Verify that equipment is ready for electrical connection, wiring, and energizing.
9	
10	PREPARATION
11	Review equipment submittals prior to installation and electrical rough-in. Verify location, size, and type of
12	connections. Coordinate details of equipment connections with supplier and installer.
13	
14	INSTALLATION
15	Use wire and cable with insulation suitable for temperatures encountered in heat-producing equipment.
16	
17	Provide a green equipment ground conductor for all installed equipment wiring.
18	
19	Make conduit connections to equipment using flexible PVC-coated metal conduit.
20	
21	Install pre-finished cord set where connection with attachment plug is indicated or specified, or use attachment plug
22	with suitable strain-relief clamps.
23	
24	Provide suitable strain-relief clamps for cord connections to outlet boxes and equipment connection boxes.
25	
26	Make wiring connections in control panel or in wiring compartment of pre-wired equipment in accordance with
27	manufacturer's instructions. Provide interconnecting wiring where indicated.
28	
29	Install disconnect switches, controllers, control stations, and control devices such as limit switches and temperature
30	switches as indicated. Connect with conduit and wiring as indicated.
31	
32	MISCELLANEOUS CONNECTIONS
33	Hand Dryers: Provide handle lock on source circuit breaker to serve as required lock open disconnect.
34	
35	Drinking Fountains and Bottle Fill Fountains: Provide GFCI source circuit breaker to serve receptacle at fountain.
36	
37	HVAC AND PLUMBING CONNECTIONS
38	Provide all power wiring including all circuitry carrying electrical energy from panelboard or other source through
39	starters, variable frequency drives (VFDs), and disconnects to motors or to packaged control panels. Packaged control
40	panels may include disconnects and starters and overcurrent protection. Provide all wiring between packaged control
41	panels and motors.
42	
43	Contractor shall verify with mechanical contractor the electrical requirements including voltages, horsepower,
44	disconnecting means, starters and variable frequency drives for motors and equipment prior to ordering circuit
45	breakers, disconnects and starters.
46	
47	Provide 120 volts to each temperature control panel. Coordinate quantity and exact locations with HVAC/DDC
48	contractors.
49	
50	Unless otherwise specified, all electrical control devices such as aqua-stats, float and pressure switches, fan powered
51	VAV boxes, switches, electro-pneumatic switches, solenoid valves and damper motors requiring mechanical
52	connections shall be furnished and installed and wired by the Contractor supplying the devices.
53	
54	Provide 120V, single phase 15 ampere circuit and switching means to serve factory installed interior lighting within
55	each HVAC unit.
56	

- 1 Each motor terminal box shall be connected with a minimum 12", maximum 36" piece of flexible PVC-coated metal
- conduit to a fixed junction box. Conduit must be installed perpendicular to direction of equipment vibration to allow
 conduit to freely flex.
- 4
- 5 Check for proper rotation of each motor.
- 6

7 EQUIPMENT CONNECTION SCHEDULE

- 8 As indicated on the drawings.
- 9 10

Page Intentionally Left Blank

1	SECTION 26 27 26
2 3	WIRING DEVICES
4 5	PART 1 - GENERAL
6 7	SCODE
8 9 10	This section describes the products and execution requirements relating to furnishing and installing wiring devices and related systems for the project. Included are the following topics:
10	PART 1 - GENERAL
12	Scope
13	Related Work
14 1	Submittals
15 16	Operation and Maintenance Data
17	PART 2 - PRODUCTS
18	Modularly Connected (Modular) Devices
19	Wall Switches
20	Receptacles
21	Occupancy Sensors
22	Wall Dimmers
25 24	Device Plates and Box Covers
25	PART 3 - EXECUTION
26	Installation
27	Field Quality Control
28	Occupancy Sensors
29	Adjusting
30	
31 22	RELATED WORK
32 33	Applicable provisions of Division 1 govern work under this section.
34	SUBMITTALS
35	Provide product data showing model numbers, configurations, finishes, dimensions, and manufacturer's instructions.
36	
37	For occupancy sensor shop drawings, the manufacturer's actual layout of occupancy sensors and the wiring diagrams
38	shall be provided.
39	
40 //1	OPERATION AND MAINTENANCE DATA All operations and maintenance data shall comply with the submission and content requirements specified under
42	section GENERAL REOUIREMENTS.
43	
44	
45	PART 2 - PRODUCTS
46	
47	MODULARLY CONNECTED (MODULAR) DEVICES:
48 70	Modularly connected devices are allowed, but not required.
49 50	Modular Pigtailed Connector: Polarized connector with minimum six-inch stranded conner wire leads polycarbonate
51	right-angle housing, UL498 listed, with finger-safe connector housing which provides insulation from conductive
52	surfaces. Contacts shall be brass. Connector shall be manufactured so that it provides a secure connection such that it
53	will maintain contact with the device until the device is removed for replacement. Modular connectors shall be
54	provided with covers which protect the contacts from paint, drywall mud, and construction dust and debris.
55	Connectors shall be Hubbell SNAPConnect, Leviton Lev-Lok, Pass & Seymour PlugTail, or approved equal.
סט 57	
58	

1	WALL SWITCHES						
2	General: Heavy duty use toggle switch, rated 20 amperes and 120 volts AC. Switches shall be UL20 Listed and meet						
3	Enderal Specification WS-896 All switches shall be beauty duty Specification Grade						
4	reactal specification wis 050. All switches shall be neavy daty specification of dat.						
5	Handle: Ivory made of nylon or high impact resistant material						
6	Handle. Nory made of hytori of high impact resistant material.						
7	All quitches on amorganey circuits shall have a red handle with matching red sover plate						
0	All switches on emergency circuits shall have a reu handle with matching reu cover plate.						
0	Market Control of the State of Annual Annual Annual Annual (2010). All we take a shall be back and side of a second						
9	wall switches for Lighting Lircuits and wotor Loads Under 1/2 HP: All switches shall be back and side wired, screw						
10	clamp type, suitable for solid or stranded wire up to #10 AWG, with separate green ground screw. Switches shall be as						
11	tollows:						
12	Hubbell 1221*,						
13	Leviton 1221-S*,						
14	Pass & Seymour CSB20AC1-*,						
15	or approved equal. (* indicates color selection).						
16							
17	Modular Wall Switches for Lighting Circuits and Motor Loads Under 1/2 HP: Switches shall be as follows:						
18	Hubbell SNAP1221*NA,						
19	Leviton M1221-*,						
20	Pass & Seymour PT20AC1-*.						
21	or approved equal. (* indicates color selection).						
22							
23	RECEDENCIES						
20	General Beguirements: NEMA Type 5-20P, ivony pylop or high impact resistant face. Recentacles shall be 11/498						
25	Listed and most Endersit Specification WC SEG All duplay regentables chall be beau duty Specification Grade 20 amp						
25	Ested and meet rederal specification we-sso. An duplex receptacies shall be neavy duty specification Grade, 20 amp						
20							
27							
28	Generally, all receptacies shall be duplex convenience type unless otherwise noted.						
29							
30	All receptacles on emergency circuits shall have a red face with matching red cover plate.						
31							
32	All receptacles installed in bathrooms, kitchens, and within 6 feet of the outside edge of sinks shall be GFCI type.						
33							
34	All receptacles installed in outdoor locations, garages, rooftops, and in other damp or wet locations shall be GFCI type						
35	with a weather-resistant (WR) rating.						
36							
37	Convenience and Straight-blade Receptacles: All receptacles shall be back and side wired, screw clamp type, suitable						
38	for solid or stranded wire up to #10 AWG, with a separate green ground screw. Receptacles shall be as follows:						
39	Hubbell 5362*,						
40	Leviton 5362-*						
41	Pass & Seymour PS5362*						
42	or approved equal (* indicates color selection)						
43							
4J AA	GECL Recenterlas: Dupley convenience recenterle with integral ground fault current interruptor meeting the						
44	are incompted in the standard 0.42 Close A including soft start functionality and reverse line load miffred function						
45	requirements of OC standard 945 class A, including sentest functionality and reverse line-load missile function						
40	repeatability. GFCI receptacies snall be as follows:						
47	Hubbell GFK53625G*,						
48	Leviton GFN12-*,						
49	Pass & Seymour 2097*,						
50	or approved equal. (* indicates color selection).						
51							
52	GFCI Receptacles with a weather-resistant (WR) rating: Weather-Resistant duplex convenience receptacle with						
53	integral ground fault current interrupter meeting the requirements of UL standard 943 Class-A, including self-test						
54	functionality and reverse line-load misfire function repeatability. WR GFCI receptacles shall be as follows:						
55	Hubbell GFR5362SG*,						
56	Leviton GFWR2-*,						
57	Pass & Seymour 2097TRWR*,						
58	or approved equal. (* indicates color selection).						

1	
2	USB Charger Receptacles: Do not use combination duplex receptacles with USB chargers. Use duplex receptacles as
3	required for the application and as specified herein. Use separate 4-port USB charging devices.
4	
5	USB Charging Devices: Single-gang 4-port USB charging station. USB ports shall meet UL94 for 5V flammability rating,
6	and shall comply with battery charging specification USB BC1.2. USB ports shall be compatible with USB 1.1/2.0/3.0
/	devices, including Apple products. USB ports shall be rated 5VDC, 4.2A minimum. Devices shall be as follows:
o o	
10	Pass & Seymour TM&USBA*CC6
11	or approved equal. (* indicates color selection).
12	
13	Locking-Blade Receptacles: As indicated on drawings.
14	
15	Specific-use Receptacle Configuration: As indicated on drawings.
16	
17	Modular Convenience and Straight-blade Receptacles: Receptacles shall be as follows:
18	Hubbell SNAP5362*A,
19	Leviton M5362-*,
20	rass & seymour rissoz , or approved equal (* indicates color selection)
22	
23	Modular GFCI Receptacles: Duplex convenience receptacle with integral ground fault current interrupter meeting the
24	requirements of UL standard 943 Class A, including self-test functionality and reverse line-load misfire function
25	repeatability. GFCI receptacles shall be as follows:
26	Hubbell GFRST83SNAP*,
27	Leviton MGFN2-*,
28	Pass & Seymour PT2097*,
29	or approved equal. (* indicates color selection).
30 21	Medular CECI Decenteries with a weather resistant (MD) rating like hack and side wired devices in lieu of medular
37	weather-resistant rated GECI recentacles
32	
34	OCCUPANCY SENSORS
35	General Requirements: All occupancy sensors shall be hardwired type; battery type shall not be permitted.
36	
37	Sensors shall use either passive infrared, or if dual technology, passive infrared and passive acoustic sensing or
38	passive infrared and ultrasonic sensing for detecting room occupancy.
39	
40	Sensitivity shall be user adjustable or self-adjusting type.
41 42	The delay times shall be adjusted within a range of 6 to 20 minutes by the contractor in the field. The soncer shall
42 43	have a test mode for performance testing
44	
45	The test LED shall indicate motion.
46	
47	Line voltage sensors are acceptable, especially in exposed ceiling areas where all wiring shall be installed in conduit,
48	including low voltage cabling if power packs are used. Provide power pack as required for low voltage sensors.
49	
50	See drawings for actual types of sensors.
51	
52 52	occupancy sensors and power packs shall have five year warranties.
55 54	Wall Mounted (Wall Switch Type): The unit shall fit in/on a standard single gang switch how
55	wan mounted (wan switch Type). The ant shan it infor a standard single gang switch box.
56	Rated capacity: 600 watts minimum at 120 volts, 60 Hz.
57	The sensor shall have two switches where dual-level lighting is required. The switch shall have manual override for
58	positive OFF and automatic ON.

2 3

4

5 6

7

8 9

10

11

14

16

18

20

29 30

31 32

33

34

35

36

37

38

39

40

41 42

43

44 45

46

47

48

The area of coverage shall be approximately 180 degrees by 35-40 feet.

Ceiling Mounted: The unit shall fit in/on a standard octagon box. All ceiling mounted sensors shall be installed to a box with ring and box support.

The coverage area shall be 360 degrees by approximately 15 feet radius when mounted at 9 foot height. The sensor shall have provisions, such as masking, to block out problem areas.

Ceiling/Corner Mounted: The unit shall fit in/on a standard octagon box. All ceiling mounted sensors shall be installed to a box with ring and box support.

12 The coverage area shall be 90 degrees or greater by approximately 40 feet radius when mounted at 9 foot height. The 13 sensor shall have provisions, such as masking, to block out problem areas.

15 **Power Packs:** Provide power packs as required for low voltage sensors. Rated capacity shall be 20 amps at 120 volts.

17 The unit shall fit on a standard octagon box. All power packs shall be installed onto a supported box.

19 Low voltage cabling shall be plenum rated or installed in conduit in plenum-rated areas.

Auxiliary Contacts for HVAC Interlock: Provide auxiliary dry contacts for HVAC BAS interlock when required. Refer to the "Occ Sensor Interlock" column in the Air Terminal Schedule(s) on the HVAC drawings. When required, provide auxiliary contacts regardless if the occupancy sensors are line or low voltage.

The occupancy sensors and auxiliary contacts shall be wired such that the sensor still detects occupancy and controls the auxiliary contacts regardless if the light switch(es) are in the OFF position (e.g. the occupant has turned the lights OFF because there is enough daylight, but the occupant is still occupying the space, and the occupancy sensor senses the occupant and closes the auxiliary contacts for BAS input).

The BAS wiring to the auxiliary contacts shall be by the Division 23 contractor.

WALL DIMMERS

General:

- 1. Compatible with the voltage of the circuit being controlled: 120V;
- 2. Compatible with the load being dimmed;
- 3. Linear full-range slide control;
- 4. Separate ON/OFF switch: single-pole, 3-way, or multiple-location operation as indicated on the drawings;
- 5. No derating required in multi-gang applications;
- 6. Polycarbonate construction;
- 7. Color to match receptacles and/or standard toggle switches.

Line-voltage LED Dimmer:

1. Forward or reverse phase dimming control as required for the application;

0-10 V Dimmers:

- 1. Ratings: 30 mA sink current;
- 2. Adjustable dial allows users to trim the low-end dimming range;

49 DEVICE PLATES AND BOX COVERS

50 Decorative Cover Plate: Stainless steel.

51

Weatherproof Cover: All receptacles installed in wet locations shall have an enclosure that is weatherproof whether or not the attachment plug is inserted. Covers shall be gasketed metal with hinged "in-use" device covers, powder coat painted. Non-metallic covers are not allowed. Covers shall be latching type and shall be lockable. Covers shall be identified as "extra-duty" type per NEC 406.9(B)(1).

Damp Location Cover: All receptacles installed outdoors in a location protected from the weather or in other damp
 locations shall have an enclosure that is weatherproof when the receptacle is covered (attachment plug not inserted

1 2 2	and receptacle covers closed). Covers shall be gasketed metal with hinged device covers, powder coat painted. Non- metallic covers are not allowed.
3 4 5	Surface Cover Plate: Raised galvanized steel.
5 6 7	PART 3 - EXECUTION
8	INSTALLATION
9	See plans for device mounting heights.
10	
11	Install wall switches with OFF position down.
12	
13 17	wail dimmers: de-rate ganged dimmers as instructed by manufacturer; do not use common neutral.
15	Install convenience receptacles with grounding pole on bottom.
16	
17	Install box for information outlet at the same height as adjacent convenience receptacles. Locate boxes for
18	information outlet as close as practical to duplex power outlet, approximately 2-inches apart.
19	
20 21	Install box for telephone jack for wall telephone at 46-inches to center above finished floor.
21	Install specific-use receptacles at heights shown on Contract Drawings
23	Install decorative plates on switch, receptacle, and blank outlets in finished areas.
24	
25	Install galvanized steel plates on outlet boxes and junction boxes in unfinished areas, above accessible ceilings, and on
26	surface-mounted outlets.
27	Install devices and wall plates fluch and level
20 29	
30	Receptacles shall have a bonding conductor from grounding terminal to the metal conduit system. Self-grounding
31	receptacles using mounting screws as bonding means are not approved.
32	
33	FIELD QUALITY CONTROL
34 25	inspect each wiring device for defects.
36	Operate each wall switch and sensor with circuit energized, and verify proper operation.
37	
38	Verify that each receptacle device is energized.
39	
40 41	Test each receptacle device for proper polarity.
41 42	Test each GECI recentacle device for proper operation
43	
44	The City of Madison personnel reserve the right to be present at all tests.
45	
46	OCCUPANCY SENSORS
47	Power packs used in return air plenum ceiling areas shall be installed in an approved enclosure or UL listed for return
48 49	ar pienum.
50	Provide a minimum of 4' of coiled cable for ceiling-mounted sensors.
51	
52	Occupancy sensors shall be installed at locations indicated on the manufacturer's submittal layout drawings. Sensors
53	shall be located to prevent false "ON" tripping of the lights.
54	Constitute Test. After the concer has been energized for at least 15 minutes well to the middle of the second (f
55 56	sensitivity rest. After the sensor has been energized for at least 15 minutes, walk to the middle of the room (if
57	and down slowly. The test LED should blink.
58	

- Time Delay Test: Set the time delay for 10 minutes. Walk into the room to activate the sensor then leave room.
 Sensor must turn lights off at approximately 10 minutes. Walk into the room again to reactivate the lights. Lights
- 2 Sensor must turn lights off at approximately 10 minutes. Walk into the room again to reactivate the lights. Lights3 should activate within 1 second.
- 45 Provide controls suitable for receptacle/plugload control as required.6

7 ADJUSTING

- 8 Adjust devices and wall plates to be flush and level.9
- 10 Mark all conductors with the panel and circuit number serving the device with a machine generated label, at the 11 device, and on the back of the device cover.
- 12 13

1	SECTION 26 27 28
2 3	DISCONIECT SWITCHES
4 5	PART 1 - GENERAL
6	
/ 8 9	SCOPE The work under this section includes disconnect switches, fuses, and enclosures. Included are the following topics:
10	PART 1 - GENERAL
11	Scope
12	Related Work
13	References
14	Submittals
15	Operation and Maintenance Data
16	General
1/ 10	
10	Disconnect Switches
20	Filses
21	
22	PART 3 - EXECUTION
23	Installation
24	
25	RELATED WORK
26 27	Applicable provisions of Division 1 govern work under this Section.
28 29	Section 26 27 02 - Equipment Wiring Systems
30	REFERENCES
31	NECA (National Electrical Contractors Association) "Standard of Installation"
32	NEMA ICS 2 – Industrial Control Devices, Controllers, and Assemblies
33	NEMA KS 1 – Enclosed Switches
34	UL 50 – Enclosures for Electrical Equipment
35	UL 98 – Enclosed and Dead-front Switches
30 27	
38	Include outline drawings with dimensions and equipment ratings for voltage ampacity horsenower and short
39 40	circuit.
41	OPERATION AND MAINTENANCE DATA
42	All operations and maintenance data shall comply with the submission and content requirements specified under
43 44	section GENERAL REQUIREMENTS.
45	GENERAL
46	Provide disconnect switches for loads required by code. Review HVAC and Plumbing specifications to determine what
47	equipment is furnished with disconnect switches. Install disconnect switches whether furnished under this contract or
48	not. It is the Electrical Contractors responsibility to determine the need for a disconnect switch for each load. The
49	contractors shall include in their bid the code required disconnect switches whether indicated on the drawings or not.
50	
51	PART 2 - PRODUCTS
52	
53	DISCUNNELI SWITCHES
55	rusible switch Assemblies (use only when overcurrent protection is required): NEMA Type Heavy Duty; quick-make, quick-break load interrupter, enclosed knife switch with externally energible bandle interlocked to provent energies
56	front cover with switch in ON position. Handle lockable in OFF position. Fuse Clips: designed to accommodate Class R
57	Class J or Class CC (motors) cartridge type fuses.
58	

1 2 3 4	Nonfusible Switch Assemblies: NEMA Type Heavy Duty; quick-make, quick-break, load interrupter, enclosed knife switch with externally operable handle interlocked to prevent opening front cover with switch in ON position. Handle lockable in OFF position.
5	Enclosure:
6	Indoor: NEMA 1 code gauge steel with rust inhibiting primer and baked enamel finish.
7	
8	Outdoors: NEMA 3R code gauge zinc coated steel with baked enamel finish or NEMA 4 when indicated on
9	drawings.
10	
11	Provide manufacturer's equipment ground kit in all disconnect switches.
12	
13	In applications where the switch serves as the service entrance disconnect, provide service ground kit, label as service
14	disconnect and provide UL listing for service disconnect.
15	
16	FUSES
17	Fuses 600 Amperes and Less: Dual element, time delay, 600 volt, UL Class RK 5. Interrupting Rating: 200,000 rms
18	amperes.
19	•
20	Fuses 30 Amperes and less: Time-Delay, 600 volt, UL Class CC. Interrupting rating: 200,000 rms amperes.
21	
22	Provide three (3) spares of each size and type fuse.
23	
24	PART 3 - EXECUTION
25	
26	INSTALLATION
27	Install disconnect switches where indicated on Drawings or required by NEC.
28	
29	Provide identification as specified in Section 26 05 53.
30	
31	Provide label on inside of disconnect cover identifying the type and size of fuse to be utilized.
32	
33	END OF SECTION

1	SECTION 26 31 00					
2	PHOTOVOLTAIC SYSTEM PERFORMANCE REQUIREMENTS					
3	DADT	4 65				
4	PARI	1 - GEľ	JEKAL	.1		
5	1	⊥ ว		.⊥. 1		
0 7	1.2			.⊥ 1		
0	1			ידי. ר		
0	1	4 5		. ۲ ۲		
10	1		2008DINATION	. ۲ ۲		
11		 2 - PR(אראואראיז איז איז איז איז איז איז איז איז איז	· 2 2		
12	2	2 1 1 (SOLAR MODILIES	· 2 2		
12	2	2	INIVERTERS	. ב ג		
14	2	3	PV WIRING	. 5		
15	2		RACKING & ROOF ΔΤΤΔCHMENT & ROOF PENETRATIONS	 Δ		
16	2	5	INTERNET BASED MONITORING	5		
17	PART	3 F X F C		2		
18	3	1	FXAMINATION	5		
19	3	2	ARRAY REQUIREMENTS	5		
20	3	.3		.5		
21	3	.4	IDENTIEICATION	.5		
22	3	.5	ΕΙΕΙ Ο ΟΙ ΑΙ ΙΤΥ CONTROI	.5		
23						
24	PART	1 - GEI	NERAL			
25						
26	1.1	DESC	RIPTION			
27		Α.	This section includes general performance requirements that apply to installing a solar electric (PV) system for			
28			this project			
29		В.	Contractor is the Designer of Record for this system. Contractor is required to provide a Structural PE			
30			(Professional Engineer) Stamp for the structural design and an Electrical PE Stamp for the overall system design			
31		C.	Both the structural and electrical stamps are to be provided from experienced PV designers with at least 5 simil	ar		
32			completed projects.			
33		D.	Contractor is required to have experience with at least 5 similar completed PV projects.			
34		E.	Product specifications included in this section are the Basis for Design. Design substitutions shall meet the			
35			minimum performance requirements defined in this section. Contractor shall select number of inverters and			
36			perform string sizing.			
37		F.	Related Work and Requirements:			
38			1. Drawings and general provisions of the Contract, including General and Supplementary Conditions and			
39			Division 01 Specification Sections, apply to this Section.			
40		G.	Incentive Paperwork:			
41			1. Contractor to provide support with Owner's application for Focus on Energy incentives.			
42	1.2	DEFI	NITIONS			
43		Α.	MPPT: Maximum power point tracking.			
44		В.	STC: Standard test conditions, 1000 W/m2, 1.5 air mass, and 25°C cell temperature.			
45		C.	NABCEP: North American Board of Certified Energy Practitioners			
46		D.	PTC: PV USA Test Conditions, 1000 W/m2, 1.5 air mass, 20°C air temperature, and 1 meter/sec. wind speed.			
47		E.	Voc: Open circuit voltage			
48		F.	lsc: Short circuit current.			
49	1.3	SUBI	/IITTALS			
50		Α.	Experience: Submit resumes for individuals involved with the design and construction of the PV System. Submi			
51			references and summaries of five similar projects that these individuals have completed.			
52		В.	Product Data: For each type of component indicated below. Include rated capacities, operating characteristics	,		
53			and furnished specialties and accessories. All product data submittals shall be submitted for review by Owner			
54			prior to purchasing any materials or equipment.			
55			1. Solar modules			
56			2. Grid tied inverters, including efficiency data.			
57			3. Racking system, including rail, clamps, brackets, and/or roof attachments.			

1 2		C.	Shop Drawings: Detail equipment assemblies and indicate dimensions, weights, loads, required clearances, method of field assembly, components, and location and size of each field connection. All shop drawings shall be
3 4			 Dimensioned AutoCAD plan drawings of equipment including solar module array, inverters, disconnects,
5 6			 metering, and electrical conduit routing. Provide AutoCAD drafted one-line wiring diagram of solar PV system indicating ratings of all modules and
/		D	inverters, wire and conduit types and sizes, and disconnects.
8		D.	Design Calculations
9 10			1. The following design calculations shall be performed by Contractor and submitted for review by Owner prior to purchasing any materials or equipment
10			a Electrical calculations, including string sizing inverter selection, and voltage losses
12			 b. Structural calculations, including string sizing, inverter screetion, and voltage losses.
13			roof strength calculations.
14		E.	Permitting and Agreements
15			1. The following permits and agreements shall be prepared by Contractor on behalf of the Owner. All
16			approved permits and agreements shall be submitted for review by Owner prior to purchasing any
17			materials or equipment.
18			a. Utility interconnection agreement
19			b. Building permit
20		_	c. Electrical permit
21		F.	As built drawings:
22			1. Dimensioned AutoCAD plan drawings of equipment including solar module array, inverters, disconnects,
23			metering, and electrical routing.
24			2. Provide AutoCAD drafted one-line diagram of solar PV system indicating ratings of an modules and invorters, wire and conduit types and sizes, and disconnects.
25		G	Field quality-control test reports
27		0.	1. Include voltages and power output for each string. Measure and record solar intensity during testing.
28			Include time, date, and weather conditions of test.
29		Н.	Warranty: Copies of all manufacturer's and installer's warranties.
30	1.4	QUAL	ITY ASSURANCE
31		Α.	Installer Qualifications:
32			1. Maintenance Proximity: Not more than four hours' normal travel time from Installer's place of business
33			to Project site.
34			2. Installer must have PV Installer certification through NABCEP or applying for certification.
35		C.	Electrical Components, Devices, and Accessories: Listed and labeled as defined in NFPA 70, Article 100, by a
30		D	testing agency acceptable to authorities having jurisdiction, and marked for intended use.
37 20	1 5		
30	1.5		Coordinate metering and interconnection agreement with electric utility. Contractor shall nav all
40		73.	interconnection fees including the application review fee, engineering review fee, and distribution system study
41			fee. Contractor shall submit all required forms to utility.
42		В.	Coordinate all work affecting building's roof with roofing manufacturer to ensure the roof's warranty is
43			maintained.
44	1.6	WAR	RANTY
45		A.	Installer must provide a two year installation warranty covering any defects of the installation.
46		В.	Module Warranty Period:
47			1. 5 years workmanship warranty.
48			2. Module warranty provided by manufacturer, not through a third party
49			3. I year 9/% power output warranty, then:
50			a. 10 year 90% linear power output warranty.
52		C	Inverter Warranty Period: 15 year warranty
53		D.	Racking Warranty Period: 10 year warranty.
54	PART	2 - PRO	DUCTS
55	2.1	SOLA	R MODULES
56		A.	Preapproved Manufacturers: Subject to compliance with performance requirements, manufacturers offering
57			products that may be incorporated into the Work include:
58			1. Canadian Solar

1			2	
1 2			2.	Haliana
2			S. ⊿	Reliefle
<u>э</u>			4. c	Tring Solar
4 5			э. с	First Solar
5			0. 7	Filst Solar
0			7. o	
0			o. 0	Silfdu CrossBoads Salar
0		Р	J. If an al	Clossrodus Solal
9 10		в.		remate product is proposed, bid is to document now the proposed solution is more cost effective to the
10		c	Canaci	. Follow substitution request procedure per 01 25 15.
12		C.		All modules shall be from a single manufacturer
12			1. ว	All modules shall be norm a single manufacturer.
17			Ζ.	or 144 coll
15			2	OF 144 Cell.
16			⊿	Demonstrates $00/1+5\%$
17			4. 5	Namenlates: To identify electrical characteristics manufacturer's name and address and model and
10			5.	sorial number of component
10			6	Modulo officiency: minimum 10 50% at STC
20			0. 7	60. 72 (or 120. 144 half colls)
20		П	7. Matari	ials and construction
21		D.	1	Monocrystalling or Polycrystalling
22			1. 2	Monofacial
23			2.	Junction box with hypass diades
24			2.	Staubli MCA connectors
25			J. ∥	Anodized aluminum frame with drainage holes and grounding holes
20			4. 5	Anothized aluminatin name with dramage noises and grounding noises. Onerating temperature range of -10° C to $+85^{\circ}$ C
27			5. 6	Withstand 1" diameter hail at 50 mph without damage
20			0. 7	Load rated at 5400 Pa (113 nsf) when used with two rail system
23				
30	2.2	INVER	TFRS	
30 31	2.2	INVER A.	RTERS Preapo	proved Manufacturers: Subject to compliance with requirements, manufacturers offering products that
30 31 32	2.2	INVER A.	RTERS Preapp may be	proved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include:
30 31 32 33	2.2	INVER A.	RTERS Preapp may be 1.	proved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius
30 31 32 33 34	2.2	INVER A.	Preapp may be 1. 2.	proved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA
30 31 32 33 34 35	2.2	INVER A.	Preapp may be 1. 2. 3.	proved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge
30 31 32 33 34 35 36	2.2	INVER A.	TERS Preapp may be 1. 2. 3. 4.	proved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase
30 31 32 33 34 35 36 37	2.2	INVER A.	RTERS Preapp may be 1. 2. 3. 4. 5.	proved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon
30 31 32 33 34 35 36 37 38	2.2	INVER A. B.	TERS Preapp may be 1. 2. 3. 4. 5. If an al	proved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the
 30 31 32 33 34 35 36 37 38 39 	2.2	INVER A. B.	TERS Preapp may be 1. 2. 3. 4. 5. If an al owner	broved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the 5. Follow substitution request procedure per 01 25 13.
 30 31 32 33 34 35 36 37 38 39 40 	2.2	INVER A. B.	RTERS Preapp may be 1. 2. 3. 4. 5. If an al owner Standa	broved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the Follow substitution request procedure per 01 25 13.
 30 31 32 33 34 35 36 37 38 39 40 41 	2.2	INVER A. B. C.	TERS Preapp may be 1. 2. 3. 4. 5. If an al owner Standa 1.	broved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the Follow substitution request procedure per 01 25 13.
 30 31 32 33 34 35 36 37 38 39 40 41 42 	2.2	INVER A. B. C.	TERS Preapp may be 1. 2. 3. 4. 5. If an al owner Standa 1. 2.	broved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the Follow substitution request procedure per 01 25 13. ards IEEE 1547 UL 1741 – anti-islanding.
 30 31 32 33 34 35 36 37 38 39 40 41 42 43 	2.2	INVER A. B. C.	rters Preapp may be 1. 2. 3. 4. 5. If an al owner Standa 1. 2. Electric	broved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the Follow substitution request procedure per 01 25 13. Follow substitution request procedure per 01 25 13. IEEE 1547 UL 1741 – anti-islanding. cal characteristics
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44	2.2	A. B. C. D.	TERS Preapp may be 1. 2. 3. 4. 5. If an al owner Standa 1. 2. Electric 1.	proved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the Follow substitution request procedure per 01 25 13. Follow substitution per
 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 	2.2	INVER A. B. C. D.	TERS Preapp may be 1. 2. 3. 4. 5. If an al owner Standa 1. 2. Electric 1. 2.	proved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the . Follow substitution request procedure per 01 25 13. ards IEEE 1547 UL 1741 – anti-islanding. cal characteristics AC kW rating: Minimum DC-to-AC ratio of 1.2 Output voltage: 208 VAC 3 phase
 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 	2.2	INVER A. B. C. D.	TERS Preapp may be 1. 2. 3. 4. 5. If an al owner Standa 1. 2. Electric 1. 2. 3.	broved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the . Follow substitution request procedure per 01 25 13. ards IEEE 1547 UL 1741 – anti-islanding. cal characteristics AC kW rating: Minimum DC-to-AC ratio of 1.2 Output voltage: 208 VAC 3 phase Frequency: 60 Hz sine wave
 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 	2.2	INVER A. B. C. D.	TERS Preapp may be 1. 2. 3. 4. 5. If an allowner Standa 1. 2. Electric 1. 2. 3. 4.	proved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the . Follow substitution request procedure per 01 25 13. ards IEEE 1547 UL 1741 – anti-islanding. cal characteristics AC kW rating: Minimum DC-to-AC ratio of 1.2 Output voltage: 208 VAC 3 phase Frequency: 60 Hz sine wave Input voltage: Coordinated with solar array.
 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 	2.2	INVER A. B. C. D.	RTERS Preapp may be 1. 2. 3. 4. 5. If an al owner Standa 1. 2. Electric 1. 2. 3. 4. 5.	proved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the . Follow substitution request procedure per 01 25 13. ards IEEE 1547 UL 1741 – anti-islanding. cal characteristics AC kW rating: Minimum DC-to-AC ratio of 1.2 Output voltage: 208 VAC 3 phase Frequency: 60 Hz sine wave Input voltage: Coordinated with solar array. Max Voc: Coordinated with solar array.
 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 	2.2	A. B. C. D.	TERS Preapp may be 1. 2. 3. 4. 5. If an al owner Standa 1. 2. Electric 1. 2. 3. 4. 5. 6.	proved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the . Follow substitution request procedure per 01 25 13. ards IEEE 1547 UL 1741 – anti-islanding. cal characteristics AC kW rating: Minimum DC-to-AC ratio of 1.2 Output voltage: 208 VAC 3 phase Frequency: 60 Hz sine wave Input voltage: Coordinated with solar array. Max Voc: Coordinated with solar array.
 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 	2.2	INVER A. B. C. D.	TERS Preapp may be 1. 2. 3. 4. 5. If an al owner Standa 1. 2. Electric 1. 2. Electric 3. 4. 5. 6. 7.	proved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the . Follow substitution request procedure per 01 25 13. ards IEEE 1547 UL 1741 – anti-islanding. cal characteristics AC kW rating: Minimum DC-to-AC ratio of 1.2 Output voltage: 208 VAC 3 phase Frequency: 60 Hz sine wave Input voltage: Coordinated with solar array. Max DC current: Coordinated with solar array. Startup voltage: Coordinated with solar array.
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51	2.2	INVER A. B. C. D.	TERS Preapp may be 1. 2. 3. 4. 5. If an al owner Standa 1. 2. Electric 1. 2. 3. 4. 5. 6. 7. 8.	proved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the . Follow substitution request procedure per 01 25 13. ards IEEE 1547 UL 1741 – anti-islanding. cal characteristics AC kW rating: Minimum DC-to-AC ratio of 1.2 Output voltage: 208 VAC 3 phase Frequency: 60 Hz sine wave Input voltage: Coordinated with solar array. Max DC current: Coordinated with solar array. Max DC current: Coordinated with solar array. Startup voltage: Coordinated with solar array. Output power factor: Unity
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52	2.2	A. B. C. D.	TERS Preapp may be 1. 2. 3. 4. 5. If an al owner Standa 1. 2. Electric 1. 2. 3. 4. 5. 6. 7. 8. 9.	broved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the . Follow substitution request procedure per 01 25 13. ards IEEE 1547 UL 1741 – anti-islanding. cal characteristics AC kW rating: Minimum DC-to-AC ratio of 1.2 Output voltage: 208 VAC 3 phase Frequency: 60 Hz sine wave Input voltage: Coordinated with solar array. Max Voc: Coordinated with solar array. Startup voltage: Coordinated with solar array. Startup voltage: Coordinated with solar array. Output power factor: Unity DC to AC conversion efficiency:
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53	2.2	A. B. C. D.	Preapp may be 1. 2. 3. 4. 5. If an al owner Standa 1. 2. Bectric 1. 2. 3. 4. 5. 6. 7. 8. 9.	broved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the . Follow substitution request procedure per 01 25 13. ards IEEE 1547 UL 1741 – anti-islanding. cal characteristics AC kW rating: Minimum DC-to-AC ratio of 1.2 Output voltage: 208 VAC 3 phase Frequency: 60 Hz sine wave Input voltage: Coordinated with solar array. Max VOC: Coordinated with solar array. Startup voltage: Coordinated with solar array. Startup voltage: Coordinated with solar array. Startup voltage: Coordinated with solar array.
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54	2.2	A. B. C. D.	TERS Preapp may be 1. 2. 3. 4. 5. If an al owner Standa 1. 2. Electric 1. 2. 3. 4. 5. 6. 7. 8. 9.	broved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the Follow substitution request procedure per 01 25 13. Ards IEEE 1547 UL 1741 – anti-islanding. cal characteristics AC kW rating: Minimum DC-to-AC ratio of 1.2 Output voltage: 208 VAC 3 phase Frequency: 60 Hz sine wave Input voltage: Coordinated with solar array. Max DC current: Coordinated with solar array. Max DC current: Coordinated with solar array. Startup voltage: Coordinated with solar array. Output power factor: Unity DC to AC conversion efficiency: a. 97.5% CEC rated efficiency AC and DC rapid shutdown compliant with NEC 2017
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55	2.2	E.	TERS Preapp may be 1. 2. 3. 4. 5. If an al owner Standa 1. 2. Standa 1. 2. 3. 4. 5. 6. 7. 8. 9. 10. Feature	broved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the Follow substitution request procedure per 01 25 13. Ards IEEE 1547 UL 1741 – anti-islanding. cal characteristics AC kW rating: Minimum DC-to-AC ratio of 1.2 Output voltage: 208 VAC 3 phase Frequency: 60 Hz sine wave Input voltage: Coordinated with solar array. Max Voc: Coordinated with solar array. Max DC current: Coordinated with solar array. Startup voltage: Coordinated with solar array. Output power factor: Unity DC to AC conversion efficiency: a. 97.5% CEC rated efficiency AC and DC rapid shutdown compliant with NEC 2017
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55 56	2.2	A. B. C. D.	TERS Preapp may be 1. 2. 3. 4. 5. If an al owner Standa 1. 2. Electric 1. 2. 3. 4. 5. 6. 7. 8. 9. 10. Featur 1.	broved Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the . Follow substitution request procedure per 01 25 13. ards IEEE 1547 UL 1741 – anti-islanding. cal characteristics AC kW rating: Minimum DC-to-AC ratio of 1.2 Output voltage: 208 VAC 3 phase Frequency: 60 Hz sine wave Input voltage: Coordinated with solar array. Max Voc: Coordinated with solar array. Max Dcc current: Coordinated with solar array. Startup voltage: Coordinated with solar array. Output power factor: Unity DC to AC conversion efficiency: a. 97.5% CEC rated efficiency AC and DC rapid shutdown compliant with NEC 2017 es Transformerless design.
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55 56 57	2.2	A. B. C. D.	TERS Preapp may be 1. 2. 3. 4. 5. If an all owner Standa 1. 2. Bectric 1. 2. Standa 1. 2. 3. 4. 5. 6. 7. 8. 9. 10. Featur 1. 2.	browed Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the . Follow substitution request procedure per 01 25 13. ards IEEE 1547 UL 1741 – anti-islanding. cal characteristics AC kW rating: Minimum DC-to-AC ratio of 1.2 Output voltage: 208 VAC 3 phase Frequency: 60 Hz sine wave Input voltage: Coordinated with solar array. Max Voc: Coordinated with solar array. Max DC current: Coordinated with solar array. Startup voltage: Coordinated with solar array. Output power factor: Unity DC to AC conversion efficiency: a. 97.5% CEC rated efficiency AC and DC rapid shutdown compliant with NEC 2017 es Transformerless design. Forward facing DC disconnect
30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55 56 57 58	2.2	A. B. C. D.	TERS Preapp may be 1. 2. 3. 4. 5. If an all owner Standa 1. 2. Bectric 1. 2. Standa 1. 2. 3. 4. 5. 6. 7. 8. 9. 10. Featur 1. 2. 3.	browed Manufacturers: Subject to compliance with requirements, manufacturers offering products that e incorporated into the Work include: Fronius SMA Solar Edge Enphase Chilicon Iternate product is proposed, bid is to document how the proposed solution is more cost effective to the . Follow substitution request procedure per 01 25 13. ards IEEE 1547 UL 1741 – anti-islanding. cal characteristics AC kW rating: Minimum DC-to-AC ratio of 1.2 Output voltage: 208 VAC 3 phase Frequency: 60 Hz sine wave Input voltage: Coordinated with solar array. Max Dc current: Coordinated with solar array. Startup voltage: Coordinated with solar array. Startup voltage: Coordinated with solar array. Startup voltage: Coordinated with solar array. Output power factor: Unity DC to AC conversion efficiency: a. 97.5% CEC rated efficiency AC and DC rapid shutdown compliant with NEC 2017 es Transformerless design. Forward facing DC disconnect DC side ground fault protection.

1 2			4.	Inverter must limit power output to nameplate value. If connected to an array capable of producing more than the inverter's capacity, the inverter must limit the power without damage
3			5.	Maximum power point tracking over the range of voltages of the array, at the ambient temperatures of
4				the site.
5			6.	User navigable display.
6			7.	LED status lights on enclosure.
7			8.	Communication port for diagnostics and communication port for communication with multiple inverters
8				and internet interface device.
9			9.	NEMA 3R enclosure
10	2.3	PV W	/IRING	
11		Α.	Type P	V-WIRE, #10AWG, from array to combiner box, and where used as a jumper for connection between
12			modul	es.
13		В.	UV-Sta	abilized Cable Ties:
14			1.	Fungus inert, designed for continuous exposure to exterior sunlight, self extinguishing, one piece, self
15				locking, Type 6/6 nylon.
16			2.	Minimum Width: 3/16 inch (5 mm).
17			3.	Tensile Strength at 73 °F (23 °C), According to ASTM D 638: 12,000 psi (82.7 MPa).
18			4.	Temperature Range: -40 to +185 °F (-40 to +85 °C).
19			5.	Color: Black.
20		C.	Ampac	city of PV source circuits shall be a minimum of 156% of the sum of parallel strings short circuit currents.
21			1.	Shall be sized to limit voltage drop to 1.5% from array to inverter during full production at MPPT voltage
22				at maximum ambient temperature.
23			2.	Shall be in metallic conduit from combiner box, if installed, to inverter.
24	2.4	RACK	(ING & R	OOF ATTACHMENT & ROOF PENETRATIONS
25		А	Preapp	proved Manufacturers: Subject to compliance with requirements, manufacturers offering products that
26			may be	e incorporated into the Work include:
27			1.	Products for attached systems on flat roofs:
28				a. Roof attachment
29				i. Anchor Products U-Anchor
30				ii. Iron Ridge Flat Roof Attachment
31				iii. OMG Roofing Products Power Grip Plus
32				b. Racking
33				i. Iron Ridge XR
34				ii. Unirac SM
35			2.	Products for ballasted systems on flat roofs:
36				a. Unirac RM10
37				b. Ecolibrium Solar Ecotoot
38			3.	Products for pitched roofs:
39				a. Roof attachment
40				I. Anchor Products U-Anchor
41				II. Iron Ridge Flat Roof Attachment
42				III. OMG Roofing Products Power Grip Plus
43				iv. S-5 Clamps (for standing seam installations)
44				A.) Use S-S-U, S-S-S, or the required clamp for the specific rooting product.
45				B.) S-5 mini clamps are not acceptable.
46				v. EcoFasten GreenFasten or QuickFoot (for composite shingle installations)
47				D. Racking
48				I. Iron Ridge XR
49 50			1	II. UIII du SIVI Droducte for polo mount arraye
50			4.	Products for pole mount arrays
51 52				i. ivi i Solar Top of Pole Iviounts ii. Droformed Line Droducts Top of Dele Mounts
52 E2			F	II. Freiormea Line Products rop of Pole Mounts
53 F/			5.	Frounds for ground mount arrays
54 FF				I. IVI I SOIAF GFOUND MOUNTS
55				ii. Preformed Line Products Power Peak
טכ רק				iii. Iron kidge XK Ground Wount
37				

1	2.5	INTER	TERNET BASED MONITORING		
2		Α.	Provide standard package from inverter manufacturer and connect to the City Network. Coordinate with Owner.		
3			Contractor is required to test monitoring to confirm it is functioning.		
4	PART 3	EXECU	TION		
5	3.1	EXAMI	NATION		
6		Α.	Examine roughing-in of electrical connections. Verify actual locations of connections before module installation.		
7		В.	Proceed with installation only after unsatisfactory conditions have been corrected.		
8	3.2	ARRAY	/ REQUIREMENTS		
9		Α.	Install modules on racking designed for solar (PV) modules.		
10		В.	Structural Performance: Installation shall withstand all local wind and snow loads, and all local building		
11		-	department requirements.		
12		C.	If applicable, slip sheet is to be used between ballasted racking and root membrane		
13		D.	All fastening hardware must be stainless steel.		
14		E.	All materials must be metallurgically compatible where different materials are in contact with each other.		
15		F.	Root penetrations shall be made watertight using methods that are standard to the rooting industry, are		
10		C	approved by the rooming manufacturer, and that protect the warranty of the roof.		
10		G.	The modules shall be connected in analys with the area autimed in Exhibit A		
10			The modules shall be installed only in the area outlined in Exhibit A. Proposed alternate layout shall be submitted to CDM and approved prior to installation begins		
20			2. Proposed alternate ayout shall be provided with a combiner box		
20			 In needed, each array shall be provided with a combiner box. DV module cables may be installed exposed where routed directly behind modules, but all cables shall be 		
21			4. If v module cables may be installed exposed where routed directly behind modules, but all cables shall be installed in a section of conduit where crossing part of the roof pat under a module. Conduit running		
22			across roof shall be supported on roof using Cooper B-Line Dura-Blok or equivalent		
23			5. All PV module cables shall be installed in a neat and workmanship like manner. Excess wire shall be		
25			coiled and bundled neatly and supported securely in an area where they are not subject to		
26			environmental degradation, such as from wind, sun, and animals. Attach PV module cables to racking		
27			with zip-ties listed for use in direct sunlight.		
28			6. Modules shall be connected in series and parallel to match voltage and current ratings of inverter, across		
29			all ambient temperatures common to site (-25°C to 40°C).		
30			a. Open circuit voltage of array on coldest day of year in full sunlight shall not exceed maximum		
31			operating voltage rating of inverter, modules, or any other equipment.		
32			b. Open circuit voltage on warmest day of year in morning sunlight conditions (200W/m2 irradiance)		
33			shall exceed inverter startup voltage. Voltage under operating MPPT conditions, minus any		
34			voltage drop over conductors, shall exceed minimum inverter input voltage.		
35			c. Available short circuit current multiplied by 1.25 shall not exceed ratings for the inverter or any		
36			modules.		
37			d. All series strings of modules shall have same performance characteristics.		
38	3.3	ELECTE	RICAL INSTALLATION		
39		Α.	Ground equipment according to Division 26		
40			1. Size grounding conductors per NEC articles 250 and 690.		
41			2. All conductive equipment enclosures must be grounded.		
42			3. All module frames must be grounded.		
43			a. The removal of any module shall not interrupt a grounded conductor to another photovoltaic		
44			source circuit.		
45		В.	Install wiring, combiner boxes, conduit, disconnects, inverter, web based monitoring hardware, sensors and		
46			other equipment according to Division 26.		
47		c	Exception – Il Division 26 specifies otherwise, All Solar Electric Conduit Indienal is to be metallic.		
48	2.4	U.	Connect winning according to Division 26.		
49 50	5.4		Incarion		
50		А.	1 Provide a unique label for each inverter, BV output circuit, combiner box, BV Source circuit, and module		
52			Labeling shall match labeling shown on as-built diagram and plan provided by contractor		
52		в	Provide all labeling required by NFC article 690 including but not limited to:		
54		5.	1 Jahel disconnects canable of being energized from both directions as such		
55			 Provide plaque at utility service disconnect per article 690 56R Field verify exact location 		
56			3. Label each photovoltaic disconnecting means per NFC article 690 53		
57	3.5	FIELD (QUALITY CONTROL		
58		A.	Perform tests and inspections as indicated below and prepare test reports. Correct any deficiencies.		

1			1.	Visually inspect all connections.
2			2.	Visually inspect all supports.
3			3.	Measure Voc of each individual string of modules under full sunlight.
4				a. Verify Voc of all strings are balanced.
5				b. Verify measured Voc against calculated Voc for the ambient temperature. Extrapolate Voc to
6				temperatures expected at site, and verify they are within inverters ratings.
7			4.	Measure Isc of each string of modules.
8			5.	Verify correct operation of inverter.
9			6.	Verify correct operation of complete system.
10			7.	Replace any defective modules. Modules shall be replaced at contractor's expense.
11	3.6	DEM	ONSTR	ATION
12		Α.	Simu	late power outage by interrupting normal source, and demonstrate that system disconnects from utility.
13		В.	Prov	ide owner's maintenance personnel with minimum two hour training session and in compliance with Div 1
14			Trair	ning Requirements.
15			1.	Provide training on function of each piece of equipment.
16			2.	Provide training on maintaining the system.
17			3.	Explain means of disconnecting the system, and principals of operation and safety.
18				END OF SECTION
19				
1	SECTION 26 51 13			
------------	--			
2	INTERIOR LIGHTING FIXTURES			
3				
4 5				
5				
7	The work under this section includes interior luminaires and accessories exit signs, and building-mounted exterior			
8 9	lighting. Included are the following topics:			
10	PART 1 - GENERAL			
11	Scope			
12	Related Work			
13	Reference Standards			
14	Submittals			
15	Operation and Maintenance Data			
16	Extra Materials			
17	Definitions			
18				
19	PART 2 - PRODUCTS			
20 21	Interior Luminaires and Accessories			
21 22				
22				
24	PART 3 - EXECUTION			
25	Installation			
26	Adjusting and Cleaning			
27	Interface with Other Products			
28	Zero-to-10V Dimming Control Wiring Installation			
29	Field Quality Control			
30	Luminaire Connections			
31				
32	RELATED WORK			
33 21	Applicable provisions of Division 1 govern work under this Section.			
35	Section 26.27.26 – Wiring Devices			
36				
37	REFERENCE STANDARDS			
38	RoHS – Restriction of Hazardous Substances. Council of the European Union (EC) Directive 2002/95/EC on the			
39	restriction of the use of certain hazardous substances in electrical and electronic equipment.			
40	LM-79-08 (or latest) – IES Approved Method for the Electrical and Photometric Measurements of Solid-State Lighting			
41	Products.			
42	LM-80-08 (or latest) – IES Approved Method for Measuring Lumen Maintenance of LED Light Sources.			
43	TM-21-11 (or latest) – IES Technical Memorandum on Projecting Long Term Lumen Maintenance of LED Light Sources.			
44	NEMA SSL 1-2010 (or latest) – Electronic Drivers for LED Devices, Arrays, or Systems.			
45				
40 47	SUBINITIALS			
47 // R	data for each luminaire type			
40 49	data for each furninaire type.			
50	For each luminaire type, submit luminaire information including catalog cuts with highlighted catalog numbers and			
51	required accessories:			
52				
53	Luminaire:			
54	 Manufacturer and catalog number, 			
55	 Type (identification) as indicated on the plans and schedule, 			
56	 Delivered lumens, 			
57	 Input watts, 			
58	o Etticacy,			

1	• Color rendering index.
2	• Driver:
3	 Manufacturer and catalog number,
4	 Type (Non-Dimming, Step-dimming, Continuous dimming, etc.),
5	 Power Factor, Crest Factor, THD, etc.
6	
7	
8	All operations and maintenance data shall comply with the submission and content requirements specified under
9	SECTION GENERAL REQUIREMENTS.
10	
12	EXTRA WATERIALS
12	Fronde timee (5) percent of each lamp type, but not less than one (1) of each type.
14	Provide one (1) of each type of LED module light har or array (if applicable). If the LED's are integrated into the
15	luminaire and are not separate components, provide one (1) of each of these types of luminaires.
16	
17	Provide one (1) LED driver or ballast of each type.
18	
19	DEFINITIONS
20	Driver: The power supply used to power LED luminaires, modules, or arrays.
21	
22	L70, L70, or L70%: The reported life of an LED component or system to reach 70% lumen maintenance, or 70% of the
23	LED's original light output. This test is being developed by the IES and is currently described by TM-21-11.
24	
25	LED's: Broadly defined as complete luminaire with light emitting diode (LED) packages, modules, light bars or arrays,
20	complete with driver.
27	LED luminaire failure: Negligible light output from more than 10 percent of the LED's constitutes luminaire failure
29	
29 30	PART 2 - PRODUCTS
29 30 31	PART 2 - PRODUCTS
29 30 31 32	PART 2 - PRODUCTS INTERIOR LUMINAIRES AND ACCESSORIES
29 30 31 32 33	PART 2 - PRODUCTS INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by
29 30 31 32 33 34	PART 2 - PRODUCTS INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intert of the design and examples of the table.
29 30 31 32 33 34 35 26	PART 2 - PRODUCTS INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intent of the design, and are approved by the A/E prior to bid.
29 30 31 32 33 34 35 36 37	INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intent of the design, and are approved by the A/E prior to bid. Luminaire shall be certified by a Nationally Becognized Testing Laboratory (LIL_ETL_or JEC).
29 30 31 32 33 34 35 36 37 38	INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intent of the design, and are approved by the A/E prior to bid. Luminaire shall be certified by a Nationally Recognized Testing Laboratory (UL, ETL, or IEC).
29 30 31 32 33 34 35 36 37 38 39	PART 2 - PRODUCTS INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intent of the design, and are approved by the A/E prior to bid. Luminaire shall be certified by a Nationally Recognized Testing Laboratory (UL, ETL, or IEC). Provide luminaires with quick-connect disconnecting means, similar to Thomas & Betts Sta-Kon.
29 30 31 32 33 34 35 36 37 38 39 40	PART 2 - PRODUCTS INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intent of the design, and are approved by the A/E prior to bid. Luminaire shall be certified by a Nationally Recognized Testing Laboratory (UL, ETL, or IEC). Provide luminaires with quick-connect disconnecting means, similar to Thomas & Betts Sta-Kon.
29 30 31 32 33 34 35 36 37 38 39 40 41	PART 2 - PRODUCTS INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intent of the design, and are approved by the A/E prior to bid. Luminaire shall be certified by a Nationally Recognized Testing Laboratory (UL, ETL, or IEC). Provide luminaires with quick-connect disconnecting means, similar to Thomas & Betts Sta-Kon. LED LUMINAIRES
29 30 31 32 33 34 35 36 37 38 39 40 41 42	INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intent of the design, and are approved by the A/E prior to bid. Luminaire shall be certified by a Nationally Recognized Testing Laboratory (UL, ETL, or IEC). Provide luminaires with quick-connect disconnecting means, similar to Thomas & Betts Sta-Kon. LED LUMINAIRES • LED Luminaires shall meet all DesignLights Consortium® (DesignLights.org) Product Qualification Criteria.
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43	 PART 2 - PRODUCTS INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intent of the design, and are approved by the A/E prior to bid. Luminaire shall be certified by a Nationally Recognized Testing Laboratory (UL, ETL, or IEC). Provide luminaires with quick-connect disconnecting means, similar to Thomas & Betts Sta-Kon. LED LUMINAIRES LED Luminaires shall meet all DesignLights Consortium® (DesignLights.org) Product Qualification Criteria. This does not require that the luminaire be listed on the DesignLights Consortium's® Qualified Products List,
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44	PART 2 - PRODUCTS INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intent of the design, and are approved by the A/E prior to bid. Luminaire shall be certified by a Nationally Recognized Testing Laboratory (UL, ETL, or IEC). Provide luminaires with quick-connect disconnecting means, similar to Thomas & Betts Sta-Kon. LED LUMINAIRES • LED Luminaires shall meet all DesignLights Consortium® (DesignLights.org) Product Qualification Criteria. This does not require that the luminaire be listed on the DesignLights Consortium's® Qualified Products List, but they must meet the Product Qualification Criteria. The technical requirements that the luminaire shall
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45	PART 2 - PRODUCTS INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intent of the design, and are approved by the A/E prior to bid. Luminaire shall be certified by a Nationally Recognized Testing Laboratory (UL, ETL, or IEC). Provide luminaires with quick-connect disconnecting means, similar to Thomas & Betts Sta-Kon. LED LUMINAIRES • LED Luminaires shall meet all DesignLights Consortium® (DesignLights.org) Product Qualification Criteria. This does not require that the luminaire be listed on the DesignLights Consortium's® Qualified Products List, but they must meet the Product Qualification Criteria. The technical requirements that the luminaire shall meet for each Application Category are:
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46	PART 2 - PRODUCTS INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intent of the design, and are approved by the A/E prior to bid. Luminaire shall be certified by a Nationally Recognized Testing Laboratory (UL, ETL, or IEC). Provide luminaires with quick-connect disconnecting means, similar to Thomas & Betts Sta-Kon. LED LUMINAIRES LED Luminaires shall meet all DesignLights Consortium® (DesignLights.org) Product Qualification Criteria. This does not require that the luminaire be listed on the DesignLights Consortium's® Qualified Products List, but they must meet the Product Qualification Criteria. The technical requirements that the luminaire shall meet for each Application Category are: Minimum Light Output.
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47	PART 2 - PRODUCTS INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intent of the design, and are approved by the A/E prior to bid. Luminaire shall be certified by a Nationally Recognized Testing Laboratory (UL, ETL, or IEC). Provide luminaires with quick-connect disconnecting means, similar to Thomas & Betts Sta-Kon. LED LUMINAIRES • LED Luminaires shall meet all DesignLights Consortium® (DesignLights.org) Product Qualification Criteria. This does not require that the luminaire be listed on the DesignLights Consortium's® Qualified Products List, but they must meet the Product Qualification Criteria. The technical requirements that the luminaire shall meet for each Application Category are: • Minimum Light Output. • Zonal Lumen Requirements.
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 40	PART 2 - PRODUCTS INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intent of the design, and are approved by the A/E prior to bid. Luminaire shall be certified by a Nationally Recognized Testing Laboratory (UL, ETL, or IEC). Provide luminaires with quick-connect disconnecting means, similar to Thomas & Betts Sta-Kon. LED LUMINAIRES • LED Luminaires shall meet all DesignLights Consortium® (DesignLights.org) Product Qualification Criteria. This does not require that the luminaire be listed on the DesignLights Consortium's® Qualified Products List, but they must meet the Product Qualification Criteria. The technical requirements that the luminaire shall meet for each Application Category are: • Minimum Light Output. • Zonal Lumen Requirements. • Minimum Luminaire Efficacy. • Minimum Light Output.
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50	PART 2 - PRODUCTS INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intent of the design, and are approved by the A/E prior to bid. Luminaire shall be certified by a Nationally Recognized Testing Laboratory (UL, ETL, or IEC). Provide luminaires with quick-connect disconnecting means, similar to Thomas & Betts Sta-Kon. LED LUMINAIRES • LED Luminaires shall meet all DesignLights Consortium® (DesignLights.org) Product Qualification Criteria. This does not require that the luminaire be listed on the DesignLights Consortium's® Qualified Products List, but they must meet the Product Qualification Criteria. The technical requirements that the luminaire shall meet for each Application Category are: • Minimum Light Output. • Zonal Lumen Requirements. • Minimum CRI. • It To Lumen Maitenanarea
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 50 51	PART 2 - PRODUCTS INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intent of the design, and are approved by the A/E prior to bid. Luminaire shall be certified by a Nationally Recognized Testing Laboratory (UL, ETL, or IEC). Provide luminaires with quick-connect disconnecting means, similar to Thomas & Betts Sta-Kon. LED LUMINAIRES • LED Luminaires shall meet all DesignLights Consortium® (DesignLights.org) Product Qualification Criteria. This does not require that the luminaire be listed on the DesignLights Consortium's® Qualified Products List, but they must meet the Product Qualification Criteria. The technical requirements that the luminaire shall meet for each Application Category are: • Minimum Light Output. Zonal Lumen Requirements. • Minimum CRI. Uniminaire Efficacy. • Minimum CRI. Uniminaire Warranty of 5 years (not pro-rated) to include LED driver and all LED
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 50 51 52	PART 2 - PRODUCTS INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intent of the design, and are approved by the A/E prior to bid. Luminaire shall be certified by a Nationally Recognized Testing Laboratory (UL, ETL, or IEC). Provide luminaires with quick-connect disconnecting means, similar to Thomas & Betts Sta-Kon. LED LUMINAIRES • LED Luminaires shall meet all DesignLights Consortium® (DesignLights.org) Product Qualification Criteria. This does not require that the luminaire be listed on the DesignLights Consortium's® Qualified Products List, but they must meet the Product Qualification Criteria. The technical requirements that the luminaire shall meet for each Application Category are: • Minimum Light Output. • Zonal Lumen Requirements. • Minimum Luminaire Efficacy. • Minimum CRI. • L70 Lumen Maintenance. • Minimum Luminaire Warranty of 5 years (not pro-rated) to include LED driver and all LED components.
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 50 51 52 53	INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intent of the design, and are approved by the A/E prior to bid. Luminaire shall be certified by a Nationally Recognized Testing Laboratory (UL, ETL, or IEC). Provide luminaires with quick-connect disconnecting means, similar to Thomas & Betts Sta-Kon. LED LUMINAIRES • LED Luminaires shall meet all DesignLights Consortium® (DesignLights.org) Product Qualification Criteria. This does not require that the luminaire be listed on the DesignLights Consortium's® Qualified Products List, but they must meet the Product Qualification Criteria. The technical requirements that the luminaire shall meet for each Application Category are: Minimum Luminaire Efficacy. Minimum CRI. L70 Lumen Maintenance. Minimum Luminaire Warranty of 5 years (not pro-rated) to include LED driver and all LED components.
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 50 51 52 53 54	INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intent of the design, and are approved by the A/E prior to bid. Luminaire shall be certified by a Nationally Recognized Testing Laboratory (UL, ETL, or IEC). Provide luminaires with quick-connect disconnecting means, similar to Thomas & Betts Sta-Kon. LED LUMINAIRES • Mision require that the luminaire be listed on the DesignLights.org) Product Qualification Criteria. The technical requirements that the luminaire shall meet for each Application Category are: • Minimu Luminaire Efficacy. • Minimu Luminaire Efficacy. • Minimu Luminaire Warranty of 5 years (not pro-rated) to include LED dr
29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 50 51 52 53 54 55	PART 2 - PRODUCTS INTERIOR LUMINAIRES AND ACCESSORIES See the Luminaire Schedule on the drawings for type of luminaires and catalog numbers. Luminaires manufactured by others are equally acceptable provided they meet or exceed the performance of the indicated luminaires, meet the intent of the design, and are approved by the A/E prior to bid. Luminaire shall be certified by a Nationally Recognized Testing Laboratory (UL, ETL, or IEC). Provide luminaires with quick-connect disconnecting means, similar to Thomas & Betts Sta-Kon. LED LUMINAIRES • Colstant the luminaire the luminaire be listed on the DesignLights.org) Product Qualification Criteria. The technical requirements that the luminaire shall meet for each Application Category are:

1 2	•	Color Consistency: LED manufacturer shall use a maximum 3-step MacAdam Ellipse binning process to achieve consistent luminaire-to-luminaire color for interior luminaires. Exterior luminaires shall use a
3		maximum 5-step MacAdam Ellipse binning process.
4	•	Glare Control: Exterior luminaires shall meet DesignLights Consortium's® criteria for Zonal Lumen
5		Distribution requirements or Backlight-Uplight-Glare (BUG) standards for exterior luminaires.
6	•	Luminaire shall be mercury-free, lead-free, and RoHS compliant.
7	•	Luminaire shall comply with FCC 47 CFR part 15 non-consumer RFI/EMI standards.
8	•	Light output of the LED system shall be measured using the absolute photometry method following IES LM-
9		79 and IES LM-80 requirements and guidelines.
10	•	Luminaire shall maintain 70% lumen output (L70) for a minimum of 50,000 hours.
11	•	Lumen output shall not depreciate more than 20% after 10,000 hours of use.
12	•	Luminaire and driver shall be furnished from a single manufacturer to ensure compatibility.
13	•	Luminaire Color Rendering Index (CRI) shall be a minimum of 80 for interior luminaires, and a minimum of
14		70 for exterior luminaires.
15	•	LED luminaire shall be thermally designed as to not exceed the maximum junction temperature of the LED
16		for the ambient temperature of the location the luminaire is to be installed. Rated case temperature shall
17		be suitable for operation in the ambient temperatures typically found for the intended installation. Exterior
18		luminaires to operate in ambient temperatures of -20°F to 122°F (-29°C to 50°C).
19	•	Luminaire shall operate normally for input voltage fluctuations of plus or minus 10 percent.
20	•	Luminaire shall have a maximum Total Harmonic Distortion (THD) of 20% at full input power and across
21		specified voltage range.
22	•	All connections to luminaires shall be reverse polarity protected and provide high voltage protection in the
23		event connections are reversed or shorted during the installation process.
24	•	All luminaires shall be provided with knockouts for conduit connections.
25	•	The LED luminaire shall carry a limited 5-year warranty minimum for LED light engine(s)/board array, and
26		driver(s).
27	•	Provide all of the following data on submittals:
28		• Delivered lumens
29		 Input watts
30		o Efficacy
31		 Color rendering index.
32		
33		LED Luminaires used for Emergency Egress Lighting:
34	•	The failure of one LED shall not affect the operation of the remaining LEDs.
35		
36		Emergency LED Luminaire Compatibility with Inverters:
37	•	Emergency Inverters shall be sine-wave type, or have written confirmation from the luminaire manufacturer
38		that the luminaire will function with a square-wave inverter.
39		
40	LED DRIV	/ERS
41	General:	
42	•	Provide driver type (non-dimmed, step-dimmed, continuous-dimming, etc.) as indicated on the luminaire
43		schedule on the drawings.
44	•	Minimum Warranty of 5 years (not pro-rated) to include LED driver and all LED components.
45	•	Driver shall have a rated life of 50,000 hours, minimum.
46	•	Driver and LEDs shall be furnished from a single manufacturer to ensure compatibility.
47	•	Driver shall have a minimum power factor (pf) of 0.9 and a maximum crest factor (cf) of 1.5 at full input
48		power and across specified voltage range.
49	•	Driver shall operate normally for input voltage fluctuations of plus or minus 10 percent.
50	•	Driver shall have a maximum Total Harmonic Distortion (THD) of 20% at full input power and across
51		specified voltage range.
52	•	Wiring connections to LED drivers shall utilize polarized quick-disconnects for field maintenance.
53	•	Fuse Protections: All luminaires shall have built-in fuse protection. All power supply outputs shall be either
54		fuse protected or be Polymeric Positive Temperature Coefficient (PTC)-protected as per Class 2 UL listing.
55	•	Provide all of the following data on submittals:
56		 Input watts
		 Power Factor (nf)

1 2	 Crest Factor (cf) at full input power Total Harmonic Distortion (THD).
3	Dimming Drivers
5 6	 LED driver shall be compatible with dimming controls where dimming is indicated on the plans. Dimmable drivers shall use Dimming Constant Current (DCC) or Pulse Width Modulation (PWM) operation.
7 8 9 10 11	• Continuous Dimming Drivers: LED luminaires shall dim to (20%, 15%, 10%, 5%, or 0.1%) as specified in the Luminaire Schedule on the plans without visible flicker or "popcorn effect". "Popcorn effect" is defined as the luminaire being on a pre-set dimmed level (less than 100%), and going to 100% prior to returning to the pre-set level when power is returned to the luminaire. Continuous Dimming Drivers shall use 0-10V control.
12 14	PART 3 - EXECUTION
15	
16	INSTALLATION
17 18 19	Verify ceiling types with Architectural plans or with existing ceilings. Verify specified luminaires are compatible with specified ceiling type(s) prior to ordering luminaires.
20	Install in accordance with manufacturer's instructions.
22 23 24 25 26	Install suspended luminaires using aircraft cable, or pendants supported from swivel hangers. Heavy duty chain supports may be used where indicated on the luminaire schedule. Provide aircraft cable, pendants, or chain lengths required to suspend luminaire at indicated height. All aircraft cables or pendant supported luminaires shall have an independent support to structure at all cable or pendant support locations. When chain is used, tie-wrap the luminaire whip to the chain.
27 28 29	Support luminaires larger than 2 x 4 foot (600 x 1200 mm) size independent of ceiling framing.
30 21	Provide independent support for all luminaires over 50 lbs.
31 32 33	Locate ceiling luminaires as indicated on reflected ceiling plan.
34 35 36	Install surface mounted luminaires and exit signs plumb and adjust to align with building lines and with each other. Secure to prohibit movement.
37 38 39 40 41	The Contractor shall install luminaire supports as required. Luminaire installations with luminaires supported only by insecure boxes will be rejected. It shall be the Contractor's responsibility to support all luminaires adequately, providing extra steel work for the support of luminaires if required. Any components necessary for mounting luminaires shall be provided by the Contractor. No plastic, composition or wood type anchors shall be used.
42 43	Install recessed luminaires to permit removal from below.
44 45 46	Install recessed luminaires using accessories and firestopping materials to meet regulatory requirements for fire rating.
40 47 48	Install code required hardware to secure recessed grid-supported luminaires in place.
49 50 51	Install wall mounted luminaires and exit signs at height as scheduled. Use pendants supported from swivel hangers in exposed ceiling/structure locations where necessary to mount exit signs at the specified height.
52 53	Install accessories furnished with each luminaire.
54 55 56	Make wiring connections to branch circuit using building wire with insulation suitable for temperature conditions within luminaire.
57 58	Bond luminaires and metal accessories to branch circuit equipment grounding conductor.
59	Install specified lamps in each luminaire and exit sign.

- 1 Dimmed luminaire circuits shall have separate neutrals. 2 3 Dimmed LED luminaires shall have a positive OFF, which requires turning off the circuit to the luminaire so that the 4 luminaires don't "glow" at the lowest dimmed setting. This shall be accomplished using a switch, relay, or some other 5 means acceptable to A/E. 6 7 All lamps shall be delivered to the job in sealed cartons and protected from dirt and dust during storage on the 8 project. Lamps shall be taken directly from the cartons and installed in the luminaire with special care so that they do 9 not become dusty and are not soiled in the operation. 10 11 All new lamps shall be operational at the Substantial Completion of the project. 12 13 ADJUSTING AND CLEANING 14 Align luminaires and clean lenses and diffusers at completion of Work. Clean paint splatters, dirt, and debris from 15 installed luminaires. 16 17 Aim and adjust luminaires as indicated on Drawings or as directed by the A/E. 18 19 Touch up luminaire finish at completion of work. 20 21 INTERFACE WITH OTHER PRODUCTS Interface with air handling accessories furnished and installed under Division 23. 22 23 24 Provide controls as indicated on the plans. Refer to section 26 27 26 - Wiring Devices. Controls shall be compatible 25 with the luminaires/ballasts/drivers being installed. 26 27 ZERO-TO-10V DIMMING CONTROL WIRING INSTALLATION 28 Zero-to-10V dimming control conductors are classified by the NEC as Class 2 conductors and shall be kept separate 29 from line-voltage conductors per NEC 725.136(A). Matching the insulation rating of Conductors of Different Systems 30 does not apply to Class 2 conductors per NEC 300.3(C)(1), Informational Note No.1. 31 32 Wall box dimmers will typically have two conduits: One conduit for line-voltage power, and one conduit or conduit 33 stub for the 0-10V control wiring. 34 35 The 0-10V wiring may be routed in free air if: 36 The room is approximately 900 sq.ft. or less, ٠ 37 The 0-10V wiring stays within the room, • 38 ٠ The ceiling space is a non-plenum space, and 39 ٠ All splices of 0-10V wiring are spliced in a box. 40 The 0-10V wiring may be tie-wrapped to the outside of the luminaire fixture whip per NEC 300.11(B)(2). Tie-• 41 wraps shall be UL listed for UV resistance. 42 43 At each luminaire, separate openings (either manufactured knock-outs or punched openings) shall be used for the 44 line-voltage power and the 0-10V wiring. The EC shall use an NM cable connector at the opening for the 0-10V wiring. 45 Zero-to-10V conductors entering and within a luminaire enclosure shall maintain a minimum separation of 6 mm 46 (0.25 in.) per NEC 725.136(D). 47 48 METAL-CLAD (MC) CABLE 49 Metal-Clad (MC) type cable that combines power and Class 2 circuits into a single cable may be used for the luminaire
- wiring within a single room. Examples of such products are Encore Wire[®] MC-LED[™] or Southwire[®] MC-PCS Duo[™].
 Manufacturer's names and catalog numbers are used for quality and performance only. MC Cables manufactured by
 others shall be equally acceptable provided they meet or exceed in performance and quality as specified.
- 53

54 FIELD QUALITY CONTROL

- 55 Operate each luminaire after installation and connection. Inspect for proper connection and operation.
- 56

57 LUMINAIRE CONNECTIONS

58 Recessed, including Master-Satellite connections:

1	• Use a luminaire fixture whip from a J-box for recessed lay-in luminaires. Luminaire fixture whips shall be
2	aluminum of steel AC Cable (Armored Cable) of Flexible Metal Conduit (FMC).
3	 Cable/Conduit whips shall be 3/8" (10 mm) minimum diameter, six feet (1.8 m) maximum length.
4	 Flexible whips or pre-wired systems between master and satellite luminaires may be supported by the
5	ceiling grid wires.
6	• The flexible connectors shall be steel, galvanized, clamp type with locknut, snap-in type with locknut, or
7	snap-in connector type, including those used on the master-satellite units.
8	
9	Chain or Cable Hung (unfinished spaces):
10 11	 Use manufacturer's SO cord or a luminaire fixture whip from a J-box. Luminaire fixture whips shall be aluminum or steel AC Cable (Armored Cable) or Flexible Metal Conduit (FMC).
12	Conduit whins shall be 3/8" (10 mm) minimum diameter. Conduit whin or SO cord shall be cut to length (six
13	feet (1.8 m) maximum) and shall allow movement of the chain/cable/luminaire but shall not be long
14	enough to "loop" and shall present a neat and workmanlike appearance
15	Luminare field wired flexible cord installations shall be connected ner NEC 410.62
16	The florible connectors shall be stoll calvanized, sharp time with locknut, sharp in time with locknut, or
17	• The next connectors shall be seed, gavanized, claim type with locking, shap-in type with locking, or
10	Shap-in connector type, including those used on the master-sateline diffes.
10 10	• Conduit whip slack shall be tie-whapped to the chain supports. The whaps shall be of listed for ov resistance.
20	Cable Hung (finished snaces)
20	Cable finding finding spaces).
21 22	Ose manufacturer's SO cord from furninaire to a J-box.
22	 So cord shall be been each of (is refer (1.8 m) maximum) and shall allow movement of the cable/idminiare, but believe to be been each of (is refer (1.8 m) maximum) and shall allow movement of the cable/idminiare,
23	but shall not be long enough to "loop" and shall present a neat and workmanlike appearance.
24	 SO cord slack may be tie-wrapped to the cable supports. Tie-wraps shall be UL listed for UV resistance.
25	 Luminaire field wired flexible cord installations shall be connected per NEC 410.62.
26	
27	Surface Mounted (unfinished spaces):
28	Provide direct conduit and box connection.
29	
30	Surface Mounted (finished spaces):
31	• Provide direct conduit and box connection. Use surface metal raceway where indicated on drawings.
32	Conceal box and conduit where appropriate. Flexible metal conduit shall not be used where it is exposed.
33	
34	END OF SECTION

1		SECTION 27 00 05
2		STRUCTURED COMMUNICATIONS CABLING
3		
4	PARI 1 -	GENERAL
5	1.1.	
6 7	1.2.	
/	1.3.	
8	1.3.	
9 10	1.5.	APPROVED CONTRACTOR QUALIFICATIONS
11	1.0.	
11	1.7.	
12	1.8.	QUALITY ASSURANCE
17	1.9.	
14 15	1.10	APPLICABLE STANDARDS, CODES, AND REGULATIONS
10	1.11	
10	1.12	. DOCOMENTATION
10	1.13 DADT 2	
10	7 ANT 2 -	
20	2.1.	FACE DI ATES
20	2.2.	
21	2.5.	
22	2.4.	PATCH PANELS - CATEGORY 6
20	2.5.	CARINET – WALL MOLINT
25	2.0.	CABITE MANAGEMENT -VERTICAL CABLE MANAGEMENT 9
26	2.7.	CABLE MANAGEMENT - VERTICAE CABLE MANAGEMENT 10
27	2.0.	INNER-DUICT
28	PART 3 -	EXECUTION 11
29	31	APPROVED CONTRACTOR RESPONSIBILITIES 11
30	3.2.	DELIVERY, STORAGE AND HANDLING LOGISTICS
31	3.3.	PREPARATION 11
32	3.4.	INSTALLATION 12
33	3.5.	LABELING
34	3.6.	TESTING
35		
36	PART 1 -	GENERAL
37		
38	1.1. S	COPE OF WORK
39	A	This Document specifies the City of Madison for product design, performance, and quality assurance, and
40		contractor responsibilities for execution of work to install a complete Category 6 structured cabling system.
41		Execution of work includes delivery and storage of materials, preparation, installation, field-testing, and project
42		completion tasks. System certification and warranty submittal requirements for completed work and future
43		moves, additions and changes (MAC's) are also specified in this document. Compliance to applicable codes,
44		standards and regulations is required for all construction work performed.
45		
46	1.2. S	UMMARY
47	A	Section includes products and execution requirements pertaining to Division 27 systems. Copper and fiber
48		backbone and horizontal cabling along with support systems are covered under this document.
49	В	Product specifications, general design considerations, and installation guidelines are provided in this document.
50		Quantities for all cabling products shall be provided as required to complete cabling to all work areas as shown
51		on floor plans.
52	С	The Approved Contractor shall furnish, supply and install a complete Category 6 cabling infrastructure specified
53		in the contract documents.
54	D	. The Approved Contractor shall furnish, supply and install a complete Category 6 cabling infrastructure specified in
55		the contract documents.
56	E	Work shall include all detailed execution requirements, such as preparation, installation, system certification.
57	_	and project closeout activities according to the contract.

1 2		F.	Substitutions: No substituted products shall be installed except with written approval by Owner.
3	1.3.	DATA	AND VOICE COMMUNICATIONS CONTRACT WORK
4		A.	General
5			1. Furnish all labor, materials, tools, equipment and services for the installation in accordance with general
6			provisions of specifications and the Contract Drawings.
7			2. Report percentage of work completed on a monthly basis.
8			3. Completely coordinate with work of all other trades.
9			4. Provide all supplementary or miscellaneous items, appurtenances and devices incidental to or necessary
10			for a sound, secure and complete installation, whether or not specifically indicated in the Contract
11			Documents.
12			Provide labor for testing horizontal and backbone cabling.
13			6. Provide Firestopping.
14			 Provide Telecommunications grounding and bonding.
15		В.	Provide complete installation for Structured Telecommunications Cabling System including but not limited to:
16			1. Category 6 and 6A UTP horizontal cables.
17			2. Singlemode optical fiber backbone cables.
18			3. Work area telecommunication outlets.
19			4. Wall mounted voice outlets.
20			5. Equipment mounting racks and rack enclosures.
21			6. Category 6 and 6A modular patch panels.
22			7. Optical fiber patch panels.
23			8. Optical fiber LC connectors.
24			9. Wire management panels.
25			10. Field testing.
26			11. Firestopping.
27			
28	1.3.	SUBM	IITTALS
29		A.	Submittals shall be complete and at one time. Partial submittals will not be considered.
30		В.	Material lists, schedule of values, lists of subcontractors, and proof of Contractor qualifications shall be provided
31			to Engineer upon request and shall follow the guidelines as stated in the General Requirements (Division 1 of the
32			specification).
33		C.	Shop drawings shall be submitted. All communication system shop drawings shall include:
34		•.	1. Manufacturer's data (specifications, "cut sheets").
35			2. Wiring diagrams for all installed cabling.
36			3. Equipment rack/cabinet lavouts.
37			Pronosed labeling schemes and labeling method
38			5 List of cabling distances (typical and maximum) for all structured cabling
39			5. Submit conject of certifications for all technicians and the project manager who will support this project
40			The certifications shall include:
40			a Structured Cabling and termination equinment installation certifications for conner and ontical
41 12			fiber connectivity and cabling
42			h Approved manufacturer classes satisfactorily completed
43 44			 Approved manufacture classes satisfactoring completed. Contractor shall submit a test plan that defines the tests required to appure that the system meets
 15			tochnical operational and performance experiences of days prior to proposed test data
45			Work shall not proceed without the Owner's approval of the submitted items
40		П	Drawings & Inspection of Site:
47		D.	Drawings a inspection of site.
40			1. Communications not plan unawings are to scale and typically are not uniteristicities for environment and elements are elements are elements and elements are
49			tots scale unawings for equipment platement and clearances. Dimensions given on unawings shall always
50			care precedence over scaled unawilles.
51			2. Any existing wires, utilities, or equipment snown on the drawings are snown for general information and
52 52			to the best knowledge of the Engineer. The Contractor shall field verify all existing wires, utilities, or
53			equipment.
54			3. In e contractor shall field verify distances and equipment placements coordinating locations with other
55			trades, construction managers, and general Contractor prior to installation.
56			4. Ine Contractor shall review all site conditions prior to submitting a bid on this project. Any obvious
57			discrepancies between the site conditions and bidding documents shall be brought to the attention of
58			the Engineer at the time of bidding so clarification can be made by addendum.

1			5. Change order requests for additional costs related to the contractors misunderstanding related to the
2			amount of work involved and lack of knowledge related to the site conditions will not be allowed.
3		Ε.	Test Reports: Submit copies of complete reports of all testing performed to the General Contractor, with copies
4			to the Architect's Electrical Engineer upon completion of job.
5			
6	1.5.	APPRO	OVED CONTRACTOR QUALIFICATIONS
7		А.	The Contractor shall have experience in the installation and testing of similar systems as specified herein and
8			shall have completed at least two projects of similar size and scope within the last 24 months. The Contractor
9			shall provide references upon request (including the project name, address, date of implementation, client
10			name, title, telephone number, and project description."
11		В.	All members of the installation team must be certified by the manufacturer as having completed the necessary
12			training to complete their part of the installation. All personnel shall be adequately trained in the used of such
13			tools and equipment as required.
14		C.	The Contractor bidding on communication systems specified herein shall be certified by the connectivity
15			Manufacturer to install, service, and warranty the specified product prior to the time of bid and throughout the
16			duration of the installation. Manufacturer certifications shall not be project specific and should be valid for any
17			and all projects completed by Contractor.
18		D.	The Contractor shall own and maintain tools, installation equipment, and test equipment necessary for
19			successful installation and testing of optical and Category 6 & 6A premise distribution systems.
20		Ε.	The Owner reserves the right to require the Contractor to remove from the project any such employee the
21			Owner deems to be incompetent, careless or insubordinate.
22		F.	The Contractor must maintain a state Contractor's license as required by the state.
23			
24	1.6.	APPRO	OVED PRODUCT MANUFACTURERS
25		Α.	The manufacturer of the Connectivity products specified in this document, as required for construction of the
26			cabling Infrastructure per contract documents shall be:
27			1. Hubbell Premise Wiring
28		В.	The manufacturer of the Cabling products specified in this document, as required for construction of the copper
29			cable Infrastructure per contract documents shall be:
30			1. Mohawk cable
31		С.	The manufacturer of the fiber optic cabling products specified in this document, as required for construction of
32			the Fiber Optic cable per contract documents shall be:
33			1. Mohawk Cable or Equal
34		D.	Product substitutions are permitted under the conditions stated below. (1.7 A).
35			
36	1.7.	PROD	UCT SUBSTITUTIONS
37		А.	Product substitutions from other manufacturers shall require the approval of the owner or owner's
38			representative.
39			
40	1.8.	QUAL	ITY ASSURANCE
41		А.	Installed category 6 balanced UTP and fiber cabling systems, pathways and distribution facilities shall adhere to
42			manufacturer's instructions, contract drawings and specifications, and applicable codes, standards and
43			regulations.
44		В.	Installed category 6 balanced UTP cabling systems and field test results shall strictly adhere to requirements of
45			ANSI/TIA/EIA-568-C.0 and ANSI/TIA/EIA-568-C.2.
46		С.	Installed optical fiber cabling systems and field test results shall strictly adhere to requirements of ANSI/TIA/EIA-
47			568-C.0 and ANSI/TIA/EIA-568C.3.
48		D.	Where applicable, all equipment, components, accessories and hardware shall be UL listed for the intended
49			purpose of the installation.
50		Ε.	Installed products shall be manufactured by an ISO 9001 certified facility.
51		F.	Installed products shall be free from defects in material or workmanship from the manufacturer and shall be of
52			the quality indicated.
53		G.	All methods of construction that are not specified in the contract documents shall be subject to control and
54			approval by the Owner or Owner's Representative.
55		Н.	Installed products shall be lot-traceable by date code.
56		I.	All critical internal manufacturing operations for installed products shall have documented in-process inspection
57			and testing according to ISO9001.
58			

1	1.9.	DRAW	INGS	
2		Α.	Approv	ved or preliminary contract drawings furnished at the time of bid solicitation shall serve as the basis for
3			produc	ct selection, creation of bills of material, and determination of labor content.
4		В.	Chang	es, additions, or deletions to contract drawings prior to awarding of the contract, shall require an
5			amenc	lment to the original bid.
6		C.	Prior t	o submitting the bid, in reviewing the contract drawings, the Approved Contractor shall:
7			1.	Request the attention of the Engineer, Owner, or Design Agency to clarify any materials, apparatus or
8				work believed to be incorrect, inadequate, omitted, or in violation of applicable codes, standards or
9				regulations.
10			2.	Note any contingencies related to unknown aspects of any drawings or specifications.
11		D.	Contra	act drawings, prior to execution of the project shall be formally approved and released by the Engineer or
12			Design	Agency and shall be approved by the Owner or Owner's Representative.
13		Ε.	Execut	ion of work shall be according to approved drawings, in addition to applicable specifications and
14			contra	ctual obligations.
15				
16	1.10.	APPLIC	CABLE S	ITANDARDS, CODES, AND REGULATIONS
17		Α.	Installa	ation Standards: Cable installation shall comply with the following:
18			1.	American National Standards Institute, (ANSI)
19			2.	ANSI/TIA-568-C.0, "Generic Telecommunications Cabling for Customer Premises", published 2009
20			3.	ANSI/TIA-568-C.1, "Commercial Building Telecommunications Cabling Standard", published 2009
21			4.	ANSI/TIA-568-C.2, "Balanced Twisted-Pair Telecommunication Cabling and Components Standard",
22				published 2009
23			5.	ANSI/TIA-568-C.3, "Optical Fiber Cabling Components Standard", published 2008, errata issued in
24				October, 2008
25			6.	ANSI/TIA-568-C.4, "Coaxial Cabling Component Standard" Published 2010
26			7.	ANSITIA/EIA-569-B, Commercial Building Standards for Telecommunications Pathways and Spaces, 2003.
27			8.	ANSI/TIA-607-B, Commercial Building Grounding and Bonding Requirements for Telecommunications,
28				2010.
29			9.	ANSI/TIA/EIA-942, Telecommunications Infrastructure for Data Centers, 2004.
30			10.	ANSI/ICEA S-83-596, Fiber Optic Premises Distribution Cable, 2001.
31			11.	ANSI/TIA/EIA-598, Color Coding of Optical Fiber Cables, 2001
32			12.	ANSI/ICEA S-87-640, Fiber Optic Outside Plant Distribution Cable, 1999.
33			13.	ANSI/TIA/EIA-492AAAC, Detail Specification for 850nm Laser-Optimized 50um Core Diameter/125 um
34				Cladding Diameter Class 1A Graded Index Multimode Optical Fibers, 2003.
35			14.	ANSI/TIA/EIA-492CAAA, Detail Specification for Class Iva Dispersion-Unshifted Singlemode Optical fibers,
36			4-	
3/			15.	ANSI/TIA/EIA-758: Customer-Owned Outside Plant Telecommunications Cabling Standard, 2004.
38			16.	ANSI/TIA/EIA-526-7, Optical Power Loss Measurements of Installed Singlemode Fiber Plant: OFSTP-7,
39			47	2002. ANGU/TIA (FIA FOC 44 A Doubled Double Loss Masserson and a fillestalled Maltine de Filles Diagte OFGTD
40			17.	ANSI/ HA/EIA-526-14-A, Optical Power Loss Measurements of Installed Multimode Fiber Plant: OFSTP-
41			10	14A, 2003. ANGU/TIA /FIA TCR 12F. Cuidelines for Mainteirine Optical Fiber Relative Through Revenue Rain Residence
42			18.	ANSI/ HA/EIA-TSB-125, Guidelines for Maintaining Optical Fiber Polarity Inrough Reverse-Pair Positioning,
43			10	2001. ANSL/TIA/EIA TCR 140. Additional Guidalings for Eigld Tasting Longth Lass, and Balarity of Ontical Eiber
44 4E			19.	Colling Systems, 2004
45			20	Cability Systems, 2004. ANSI/TIA/EIA/EOA/A Administration Standard for Commercial Telecommunications Infrastructure 2002
40			20.	ANSI/FIA/EIA-000-A, Administration Standard for Commercial Telecommunications initiastructure, 2002.
47 10			21.	ANSI/EIA-510-D, Cabillets, Racks, Pallets, and Associated Equipment, 1992.
48			22.	ANSI/ HA/EIA-004 (Series), FOCIS FIDER OPTIC Connector Internated Dinty Standard, 2000-2003.
49 E0			25.	National Flie Protection Association, Inc., NFPA 70
50			24. 25	National Electric Code (NEC), 2005.
51			23. 26	NEC Article 200. Glouiluilig
52 52			20. 27	NEC Article 200. Sulface Non-Motallic Pacoways
55			21. ۲0	NEC Article 200: Communications Circuits
54			20. 20	NEC Article 770: Optical Eiker Cables and Paceway
55			29. 20	INCO ALIGUE 770. Optical FIDEL Cables allu Raceway
50			50. 21	Ulles A: Standard for Non-Metallic Receivers and Eittings
50			31. 22	UL-DA. Standard for Surface Motal Decouver and Eittings
50			JZ.	or of standard for ourface metal naceways and fittings

1			33. UL-5C: Standard for Surface Raceways and Fittings for Use with Data, Signal, and Control Circuits
2			34. UL-50: Standard for Enclosures for Electrical Equipment
3			35. UL-94-V0: Tests for Flammability of Plastic Materials
4			36. UL-498: Attachment Plugs and Receptacles
5			37. UL-1479: Fire Tests of Through-penetration Firestops (in Accordance with ASTM E814).
6			38 UII-1863: Standard for Safety of Communications Circuit Accessories
7			A National Electrical Manufacturer's Association (NEMA)
, 0			And (Anthe And Anthe Anth
0			40. Alsolveria wD-6-2002. Withig Devices – Dimensional Requirements
9			41. NEMA 250-2003: Enclosures for Electrical Equipment
10			42. ISO/IEC 11801, Ed. 2:2002, Information Technology – Generic Cabling for Customer Premises, 2002.
11			43. ISO/IEC 18010, Information Technology – Pathways and Spaces for Customer Premises Cabling, 2005.
12			44. ISO/IEC 14763-1, Information Technology – Implementation and Operation of Customer Premises Cabling
13			 – Part 1: Administration, 2004.
14			45. CSA C22.1-06, Canadian Electric Code (CEC), 2006
15			46 Federal Communications Commission (ECC) Title 47 Code of Federal Regulations, Part 68: Connection of
16			Terminal Equipment to the Telephone Network 1998
17			47 U.S. Dublic Law 256 101tt Congress ADA Americans with Disabilities Act of 1002
17			47. U.S. Public Law SSO. TUTSI COBJESS, ADA. Americans with Disabilities Act of 1992.
18			48. IEEE 802.3at, Data Terminal Equipment (DTE) Power Over Media Dependent Interface (MDI), 2003.
19			49. IEEE 802.3at (current draft), Data Terminal Equipment (DTE) Enhanced Power Over Media Dependent
20			Interface (MDI).
21			50. IEEE 802.3ae, Specification for 10 Gbit/s Ethernet Operation over Optical Fiber.
22			51. Telecommunications Distribution Methods Manual, 11th Ed., Building Industry Consulting Services
23			International (BICSI), 2006.
24			52. Information Transport Systems Installation Manual, 4th Ed., Building Industry Consulting Services
25			International (BICSI) 2004
26		R	This document is not a substitute for any code standard or regulation. The Approved Contractor must be aware
20		Б.	in a document is not a substitute for any coupling and the project. The current revision of any
27			on local codes that may impact the bit submitted of execution of the project. The current revision of any
28			applicable code, standard, or regulation shall take precedence at the point or project execution, unless otherwise
29			recognized by local authorities. Applicable standards or codes that affect construction, which are listed as
30			normative references within any governing document, are also the responsibility of the Approved Contractor for
31			compliance.
32		C.	Materials
33			1. All materials shall be UL or ETL listed and verified and shall be marked as such.
34			2. Products shall be regularly catalogued items of the manufacturer and shall be supplied as a complete unit
35			in accordance with the manufacturer's standard specifications with any optional items required for
36			proper installation unless otherwise noted
37			Material shall be delivered to the site in the original packing
20			5. Waterial shall be delivered to the site in the original packing.
20			
39	1.11.		
40		А.	All materials used on this project shall be new. Used and returbished equipment is not permitted unless
41			approved by CITY OF MADISON. Provide equipment to site in original packaging whenever practical.
42		В.	The Contractor is responsible for scheduling all deliveries and providing proper receipt, handling, and storage of
43			all materials. Protect all equipment from physical damages (dents, scratches, dust, water, paint, chemicals, and
44			temperature extremes) and vandalism, or theft. The Contractor shall replace any damaged or stolen equipment.
45			The Contractor is responsible for all equipment until final project acceptance by Owner.
46		С.	Maintenance of the cabling infrastructure is to be done by authorized personnel only or void of manufacturer's
47		-	warranty may result. It is the responsibility of the owner or end user to utilize a certified installer to maintain
18			warranty coverage on existing or new cabling infractructure
10		П	The telecommunications contractor shall furnish a quotation for time and material to perform maintenance and
49		υ.	rine telecommunications contractor shan furnish a quotation for time and material to perform maintenance and repeated to extract a subscript of refused of extracting subscript of extracting subscrip
50			repairs. The owner has the first right of refusal of selecting suitable contractor or qualified internal personnel to
51			perform maintenance and repairs on structured cabling.
52		Ε.	Additions of new cabling, either horizontal or backbone, shall be completed, tested, and documented into
53			permanent building records. New cabling installations intended to be covered by the manufacturer's warranty
54			shall adhere to the documentation submittal and system certification provisions stated above.
55		F.	The Contractor is responsible for cleaning the worksite every business day and remove debris from the facility.
56			, , , , , , ,
57	1.12.	DOCU	MENTATION
58		Α.	TEST RESULTS

1			1.	All test results are to be saved electronically on CD. Test documentation submitted on disk shall be
2				clearly marked on the cover with the words "Project Test Documentation", the project name, and the
3				date of completion (month and year). For multiple buildings, the building name, including floor or wing
4				I.D. should also be included on the test results disk.
5			2.	File names of the test results recorded for each link shall match the official identification. Test results
6				shall include a complete record for each link, including type of test, cable type, cable/port I.D.,
7				measurement direction, reference setup, date, and technician's name(s).
8			3.	The test equipment name, manufacturer, model number, serial number, software version and last
9				calibration date shall also be provided in the test results documentation.
10			4.	When repairs and re-tests are performed, the problem cause and corrective action taken shall be noted,
11				and both the failed and passed test data shall be documented.
12			5.	The owner, engineer, lead project manager, or owner's representative reserve the right to request
13				verification of test results with a re-test of installed cables, on a sampling basis. Re-testing shall be at the
14				expense of the installer unless otherwise noted in the contract documents.
15		В.	AS BUI	LT DRAWINGS
16			1.	Deviations from the approved drawings, whether or not a change order is submitted, shall be clearly
17				denoted as built on the working hard copy drawing by the telecommunications contractor. As-built
18				drawings shall be returned promotive to the owner or design agent for completion of drafting revisions to
19				the original design. See "Documentation – Change Orders" helow. Manufacturer's warranty
20				registrations may also require achuilt drawings
20			2	Floor plan drawings shall at minimum include detailed cable and pathway layouts, eyact locations of
21			۷.	workstation outlets, and cable distribution bardware locations. Workstation outlets shall have
22				alphanumeric identifiers on the drawings as specified by the end user or owner
23		C		
24		C.	1	Any deviation from the approved contract drawings or specifications shall be submitted as a written
25			1.	change order
20			C	Cildinge Order.
27			Ζ.	execution of work, to perform changes, shall not proceed without prior written approval. Any changes
20				done without written approval will be at no cost to CTY OF MADISON. If the work is shown to be
29			2	Incorrect the contractor will have to correct the problem at no cost to CITY OF MADSON.
30			3.	Significant changes may require a written quotation of additional labor and materials from the
31				telecommunications contractor.
32			4.	It is the responsibility of the owner or owner's representative to bear the added cost of any substantial
33				cabling system design changes. The contractor will not proceed with any change orders without written
34				approval by the owner's representative. Any changes not approved by the owner's representative will be
35				responsibility of the contractor and at no cost to CITY OF MADISON.
36			5.	Field changes that are completed without issuance of revised drawings shall be clearly denoted on the
37				working as-built drawing. Refer to "As-Built Drawings" above.
38		D.	PUNCH	I LISTS AND CORRECTIVE ACTION
39			1.	As required in the contract documents, the telecommunications contractor shall correct punch-lists items
40				determined to be in violation of drawings, specifications, codes, standards or regulations.
41			2.	The contractor shall be responsible for timely re-work of faulty cabling or hardware installations.
42			3.	The owner reserves the right to withhold final payment until punch list items are resolved satisfactorily.
43				
44	1.13.	WAR	RANTY	
45		Α.	THE CI	TY OF MADISON requires a Permanent Link warranty for the project. Manufacturer requires Permanent
46			Link Te	est.
47		В.	The ler	ngth of the extended warranty shall be a minimum of twenty-five (25) years.
48		C.	Warrai	nty covering all components, equipment and workmanship shall be submitted in writing with system
49			docum	ientation.
50		D.	The wa	arranty period shall begin on the system's first use by the owner.
51		Ε.	Should	the cabling system fail to perform its expected operation within this warranty period due to inferior or
52			faulty i	material and/or workmanship, the contractor shall promptly make all required corrections without cost to
53			the Ow	vner.
54		F.	Upon (Completion of the project the Telecommunication Contractor shall forward the signed Warranty
55			Registr	ration Form and warranty certificate to the Owner.
56		G.	The ma	anufacturer warrants category 6 cabling, category 6A cabling, optical fiber cabling and connecting
57			compo	nents free of defects in material or workmanship.

1 2		Н.	Category 6, category 6A and optical fiber cabling and components are warranted to perform the intended application upon completion of proper installation and testing.
3		Ι.	Warranty coverage includes application assurance and compliance to applicable performance specifications.
4		J.	Installed category 6 and 6A cabling systems may be granted a full Channel warranty under the conditions stated
5			below.
6 7			 A certified installer registered who has completed a Manufacturer's training program performs the construction
, 8			2 Contractors performing the certified installation are properly registered in the Manufacturer's warranty
0			2. contractors performing the certifical instantation are property registered in the inantateurer's warranty
10			The channel components are supplied entirely by one Manufacturer including natch cords
11			 The channel components are supplied entirely by one manufacturer, including patch cords. Cable used in the installation is qualified and recognized by Connectivity Manufacturer
12			4. Cable used in the installation is qualified and recognized by connectivity manufacturer.
12			5. Installed link systems are property documented and tested with a PASS Tesuit. The county requires a
10			Field test equipment used for category 6 cabling is minimum level III classification and complies with
14			5. Field test equipment used for category 6 cabling is minimum level in classification and complies with
15			TIA/EIA-568-8.2 requirements.
16			7. Required test results, stored on a CD, and project documentation including as-built drawings, are to be
1/			submitted to the Manufacturer by the registered contractor.
18			
19	1.14.	MOVE	S, ADDS AND CHANGES
20		Α.	Moves, additions and changes initiated by the owner, end user, project manager, or design agent, which are
21			beyond the scope of work in the original contract, shall require a revised quotation by the telecommunications
22			contractor.
23		В.	It is the responsibility of the owner or owner's representative to bear the added cost of any substantial cabling
24			system design changes.
25		C.	Moves, additions and changes shall either be issued in revised drawings, or otherwise shall be clearly denoted on
26			as-built drawings.
27		D.	Moves, additions and changes that affect installations covered in a manufacturer's warranty shall be performed
28			by a certified contractor that is properly registered in the manufacturer's warranty program.
29			
30	1.15.	CLEAN	
	1.10.		
31	1.10	Α.	The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job
31 32		Α.	The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition.
31 32 33		А . В.	The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the
31 32 33 34		А . В.	The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense.
31 32 33 34 35		А . В.	The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense.
31 32 33 34 35 36	PART 2	A. B. 2 - PRO	The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense.
31 32 33 34 35 36 37	PART 2	A. B. 2 - PRO	The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense.
31 32 33 34 35 36 37 38	<u>PART 2</u> 2.1.	A. B. <u>2 - PROI</u> WORK	The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS
31 32 33 34 35 36 37 38 39	<u>PART 2</u> 2.1.	A. B. <u>2 - PROI</u> WORK A.	The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks
31 32 33 34 35 36 37 38 39 40	<u>PART 2</u> 2.1.	A. B. <u>2 - PROI</u> WORK A.	The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks 1. Jacks shall be standard 8-position, RJ-45 style, un-keyed, FCC compliant.
31 32 33 34 35 36 37 38 39 40 41	<u>PART 2</u> 2.1.	A. B. <u>2 - PRO</u> WORK A.	The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks 1. Jacks shall be standard 8-position, RJ-45 style, un-keyed, FCC compliant. 2. Jacks shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable.
31 32 33 34 35 36 37 38 39 40 41 42	<u>PART 2</u> 2.1.	A. B. <u>2 - PROI</u> WORK A.	The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks 1. Jacks shall be standard 8-position, RJ-45 style, un-keyed, FCC compliant. 2. Jacks shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable. 3. Jacks shall terminate 26-22 AWG solid or stranded conductors.
31 32 33 34 35 36 37 38 39 40 41 42 43	<u>PART 2</u> 2.1.	А. В. <u>2 - PROI</u> WORK А.	The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks 1. Jacks shall be standard 8-position, RJ-45 style, un-keyed, FCC compliant. 2. Jacks shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable. 3. Jacks shall terminate 26-22 AWG solid or stranded conductors. 4. Jacks shall include a dust cap for wire retention.
31 32 33 34 35 36 37 38 39 40 41 42 43 44	<u>PART 2</u> 2.1.	А. В. <u>2 - PROI</u> WORK А.	 The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks 1. Jacks shall be standard 8-position, RJ-45 style, un-keyed, FCC compliant. 2. Jacks shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable. 3. Jacks shall terminate 26-22 AWG solid or stranded conductors. 4. Jacks shall include a dust cap for wire retention. 5. Jacks shall accept FCC compliant 6 position plugs.
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45	<u>PART 2</u> 2.1.	А. В. <u>2 - PROI</u> WORK А.	The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks 1. Jacks shall be standard 8-position, RJ-45 style, un-keyed, FCC compliant. 2. Jacks shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable. 3. Jacks shall terminate 26-22 AWG solid or stranded conductors. 4. Jacks shall include a dust cap for wire retention. 5. Jacks shall accept FCC compliant 6 position plugs. 6. Jacks shall have attached wiring instruction labels to permit either T568A or T568B wiring configurations
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46	<u>PART 2</u> 2.1.	A. B. <u>2 - PROI</u> WORK A.	 The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks Jacks shall be standard 8-position, RJ-45 style, un-keyed, FCC compliant. Jacks shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable. Jacks shall terminate 26-22 AWG solid or stranded conductors. Jacks shall include a dust cap for wire retention. Jacks shall accept FCC compliant 6 position plugs. Jacks shall have attached wiring instruction labels to permit either T568A or T568B wiring configurations.
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47	<u>PART 2</u> 2.1.	A. B. <u>2 - PROI</u> WORK A.	The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks 1. Jacks shall be standard 8-position, RJ-45 style, un-keyed, FCC compliant. 2. Jacks shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable. 3. Jacks shall terminate 26-22 AWG solid or stranded conductors. 4. Jacks shall include a dust cap for wire retention. 5. Jacks shall accept FCC compliant 6 position plugs. 6. Jacks shall have attached wiring instruction labels to permit either T568A or T568B wiring configurations. 7. Category 6 jacks shall be backward compatible with existing Category 3, 5, and 5e cabling systems for fit, form and function
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48	<u>PART 2</u> 2.1.	A. B. <u>2 - PROI</u> WORK A.	 The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks 1. Jacks shall be standard 8-position, RJ-45 style, un-keyed, FCC compliant. 2. Jacks shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable. 3. Jacks shall terminate 26-22 AWG solid or stranded conductors. 4. Jacks shall include a dust cap for wire retention. 5. Jacks shall accept FCC compliant 6 position plugs. 6. Jacks shall have attached wiring instruction labels to permit either T568A or T568B wiring configurations. 7. Category 6 jacks shall be backward compatible with existing Category 3, 5, and 5e cabling systems for fit, form, and function. 8 lacks shall be manufactured in the LISA
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49	<u>PART 2</u> 2.1.	A. B. <u>2 - PROI</u> WORK A.	 The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks Jacks shall be standard 8-position, RJ-45 style, un-keyed, FCC compliant. Jacks shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable. Jacks shall terminate 26-22 AWG solid or stranded conductors. Jacks shall include a dust cap for wire retention. Jacks shall accept FCC compliant 6 position plugs. Jacks shall have attached wiring instruction labels to permit either T568A or T568B wiring configurations. Category 6 jacks shall be backward compatible with existing Category 3, 5, and 5e cabling systems for fit, form, and function. Jacks shall be manufactured in the USA. Category 6 jacks shall meet or exceed Category 6 transmission requirements for connecting bardware as
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50	<u>PART 2</u> 2.1.	A. B. <u>2 - PROI</u> WORK A.	 The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks Jacks shall be standard 8-position, RJ-45 style, un-keyed, FCC compliant. Jacks shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable. Jacks shall terminate 26-22 AWG solid or stranded conductors. Jacks shall include a dust cap for wire retention. Jacks shall accept FCC compliant 6 position plugs. Jacks shall have attached wiring instruction labels to permit either T568A or T568B wiring configurations. Category 6 jacks shall be backward compatible with existing Category 3, 5, and 5e cabling systems for fit, form, and function. Jacks shall be manufactured in the USA. Category 6 jacks shall meet or exceed Category 6 transmission requirements for connecting hardware, as specified in ANSI/TIA/EIA-568-C 2. Transmission Performance Specifications for 4-Pair 100 ohm
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51	<u>PART 2</u> 2.1.	A. B. <u>2 - PROI</u> WORK A.	 The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks Jacks shall be standard 8-position, RJ-45 style, un-keyed, FCC compliant. Jacks shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable. Jacks shall terminate 26-22 AWG solid or stranded conductors. Jacks shall include a dust cap for wire retention. Jacks shall accept FCC compliant 6 position plugs. Jacks shall have attached wiring instruction labels to permit either T568A or T568B wiring configurations. Category 6 jacks shall be backward compatible with existing Category 3, 5, and 5e cabling systems for fit, form, and function. Jacks shall be manufactured in the USA. Category 6 jacks shall meet or exceed Category 6 transmission requirements for connecting hardware, as specified in ANSI/TIA/EIA-568-C.2, Transmission Performance Specifications for 4-Pair 100 ohm.
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52	<u>PART 2</u> 2.1.	A. B. <u>2 - PROI</u> WORK A.	 The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks Jacks shall be standard 8-position, RJ-45 style, un-keyed, FCC compliant. Jacks shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable. Jacks shall terminate 26-22 AWG solid or stranded conductors. Jacks shall include a dust cap for wire retention. Jacks shall accept FCC compliant 6 position plugs. Jacks shall have attached wiring instruction labels to permit either T568A or T568B wiring configurations. Category 6 jacks shall be backward compatible with existing Category 3, 5, and 5e cabling systems for fit, form, and function. Jacks shall be manufactured in the USA. Category 6 jacks shall meet or exceed Category 6 transmission requirements for connecting hardware, as specified in ANSI/TIA/EIA-568-C.2, Transmission Performance Specifications for 4-Pair 100 ohm. Jacks shall be UL LISTED and CSA certified.
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 52	<u>PART 2</u> 2.1.	A. B. <u>2 - PROI</u> WORK A.	 The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks 1. Jacks shall be standard 8-position, RJ-45 style, un-keyed, FCC compliant. 2. Jacks shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable. 3. Jacks shall terminate 26-22 AWG solid or stranded conductors. 4. Jacks shall include a dust cap for wire retention. 5. Jacks shall accept FCC compliant 6 position plugs. 6. Jacks shall accept FCC compliant 6 position plugs. 7. Category 6 jacks shall be backward compatible with existing Category 3, 5, and 5e cabling systems for fit, form, and function. 8. Jacks shall be manufactured in the USA. 9. Category 6 jacks shall meet or exceed Category 6 transmission requirements for connecting hardware, as specified in ANSI/TIA/EIA-568-C.2, Transmission Performance Specifications for 4-Pair 100 ohm. 10. Jacks shall be UL LISTED and CSA certified. 11. Colors to specified by end user
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54	<u>PART 2</u> 2.1.	A. B. <u>2 - PRO</u> WORK A.	 The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks Jacks shall be standard 8-position, RJ-45 style, un-keyed, FCC compliant. Jacks shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable. Jacks shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable. Jacks shall terminate 26-22 AWG solid or stranded conductors. Jacks shall include a dust cap for wire retention. Jacks shall accept FCC compliant 6 position plugs. Jacks shall have attached wiring instruction labels to permit either T568A or T568B wiring configurations. Category 6 jacks shall be backward compatible with existing Category 3, 5, and 5e cabling systems for fit, form, and function. Jacks shall be manufactured in the USA. Category 6 jacks shall meet or exceed Category 6 transmission requirements for connecting hardware, as specified in ANSI/TIA/EIA-568-C.2, Transmission Performance Specifications for 4-Pair 100 ohm. Jacks shall be UL LISTED and CSA certified. Colors to specified by end user Category 6 modular jacks, as specified in the Contract Documents, shall be: Whend
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55	<u>PART 2</u> 2.1.	A. B. <u>2 - PRO</u> WORK A.	 The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks Jacks shall be standard 8-position, RJ-45 style, un-keyed, FCC compliant. Jacks shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable. Jacks shall terminate 26-22 AWG solid or stranded conductors. Jacks shall include a dust cap for wire retention. Jacks shall accept FCC compliant 6 position plugs. Jacks shall have attached wiring instruction labels to permit either T568A or T568B wiring configurations. Category 6 jacks shall be backward compatible with existing Category 3, 5, and 5e cabling systems for fit, form, and function. Jacks shall be manufactured in the USA. Category 6 jacks shall meet or exceed Category 6 transmission requirements for connecting hardware, as specified in ANSI/TIA/EIA-568-C.2, Transmission Performance Specifications for 4-Pair 100 ohm. Jacks shall be UL LISTED and CSA certified. Colors to specified by end user Category 6 modular jacks, as specified in the Contract Documents, shall be: Hubbell Hubbell Hubbell
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55 55 56	<u>PART 2</u> 2.1.	A. B. <u>2 - PRO</u> WORK A.	 The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks Jacks shall be standard 8-position, RJ-45 style, un-keyed, FCC compliant. Jacks shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable. Jacks shall terminate 26-22 AWG solid or stranded conductors. Jacks shall include a dust cap for wire retention. Jacks shall accept FCC compliant 6 position plugs. Jacks shall accept FCC compliant 6 position plugs. Jacks shall accept FCC compliant 6 position plugs. Jacks shall be manufactured in the USA. Stacks shall be manufactured in the USA. Category 6 jacks shall be manufactured in the USA. Jacks shall be UL LISTED and CSA certified. Colors to specified by end user Category 6 modular jacks, as specified in the Contract Documents, shall be: Hubbell HJ06El (Category 6 – Ivory)
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55 56 57	<u>PART 2</u> 2.1.	A. B. <u>2 - PROP</u> WORK A.	 The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks Jacks shall be standard 8-position, RJ-45 style, un-keyed, FCC compliant. Jacks shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable. Jacks shall iterminate 26-22 AWG solid or stranded conductors. Jacks shall include a dust cap for wire retention. Jacks shall accept FCC compliant 6 position plugs. Jacks shall have attached wiring instruction labels to permit either T568A or T568B wiring configurations. Category 6 jacks shall be manufactured in the USA. Category 6 jacks shall be manufactured in the USA. Category 6 jacks shall be user or exceed Category 6 transmission requirements for connecting hardware, as specified in ANSI/TIA/EIA-568-C.2, Transmission Performance Specifications for 4-Pair 100 ohm. Jacks shall be UL LISTED and CSA certified. Colors to specified by end user Category 6 modular jacks, as specified in the Contract Documents, shall be: a. Hubbell HXI6EI (Category 6 – Ivory)
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55 56 57 57	<u>PART 2</u> 2.1. 2.2.	А. В. 2 - PROI WORK А.	 The communications Contractor shall clean up all debris related to this work on a regular basis leaving the job site in a clean, safe condition. Protect all equipment from damage during construction. Equipment not protected shall be replaced at the Contractor's expense. DUCTS CAREA CONNECTORS Category 6 Jacks Jacks shall be standard 8-position, RJ-45 style, un-keyed, FCC compliant. Jacks shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable. Jacks shall terminate 26-22 AWG solid or stranded conductors. Jacks shall include a dust cap for wire retention. Jacks shall accept FCC compliant 6 position plugs. Jacks shall have attached wiring instruction labels to permit either T568A or T568B wiring configurations. Category 6 jacks shall be backward compatible with existing Category 3, 5, and 5e cabling systems for fit, form, and function. Jacks shall meet or exceed Category 6 transmission requirements for connecting hardware, as specified in ANSI/TIA/EIA-568-C.2, Transmission Performance Specifications for 4-Pair 100 ohm. Jacks shall be UL LISTED and CSA certified. Colors to specified by end user Category 6 modular jacks, as specified in the Contract Documents, shall be: Hubbell Hubbell Hubbell Hubbell

	CITY SPEC May	TY OF MADISON PECIFICATION May 16, 2024			
	Ividy	10, 2024	r		
1			1.	Faceplates shall be constructed of high impact, UL94 V-0 rated thermoplastic.	
2			2.	Faceplates shall be compatible with standard NEMA openings and boxes.	
3			3.	Faceplates shall be 2.75" W x 4.5" H (69.8 mm x 114.3 mm) for single gang and 4.5" X 4.5" (114.3 X 114.3	
4				mm) for double gang.	
5			4.	Port size in each faceplate shall fit the Category 6 Modular Jack or Shap-Fit fiber optic, audio, and video	
0 7			5	Faceplates shall be provided with clear plastic and color-matched label field covers. Faceplates shall	
, 8			5.	provide for ANSI/TIA/EIA-606-A compliant workstation outlet labeling.	
9			6.	#6-32 pan head Phillips/slotted mounting screws shall be included with each faceplate.	
10			7.	Faceplates shall be UL LISTED and CSA certified.	
11			8.	Work area faceplates, as specified in the Contract Documents, shall be:	
12			ä	a. Hubbell (IFP Series)	
13				p. IFP14ei (4-Port Ivory)	
14	2.3.	CABLE	E		
15 16		А.	Categ	JORY 6 UTP Denum Cable construction shall be four twisted pairs of 22 AWC insulated solid conductors, with a	
17			1.	rincord surrounded by a tight outer jacket	
18			2.	Non-plenum - Cable construction shall be four twisted pairs of 24 AWG insulated solid conductors, with a	
19				ripcord, surrounded by a tight outer jacket.	
20			3.	NO minimum compliant cable will be accepted. The facility requires additional bandwidth.	
21			4.	Ripcord shall be directly underneath the outer jacket.	
22			5.	Cable shall be marked with Manufacturer and pertinent information. UL, ETL, or CSA agency certification	
23				or verification markings shall be marked on the cable jacket according to the certifying agency's	
24 25			6	requirements.	
25 26			0.	a Pair 1: White/Blue: Blue	
27				b. Pair 2: White/Orange: Orange	
28				c. Pair 3: White/Green/Green	
29				d. Pair 4: White/Brown/Brown	
30			7.	Plenum or Riser rated jackets	
31			8.	Cable shall be supplied in 1000 ft spools or 1000 ft Reelex boxes.	
32			9.	Cable shall exceed Category 6 transmission requirements specified in ANSI/TIA/EIA-568-C.2.	
33 24			10.	Cable shall be UL and C (UL) listed.	
34			11.	Length	
36			12.	Category 6 UTP horizontal distribution cable, as specified in the Contract Documents, shall be:	
37				a. Mohawk AdvanceNet Cable	
38				b. Plenum M57193	
39				C. Riser M57202	
40		В.	Backk	oone distribution cable – Fiber Optic	
41			1.	Singlemode fiber backbone distribution cable shall be available in multi-strand constructions for intra-	
4Z 13			2	Duilding applications. OENR or OENR will be determined at each site. The contractor will be responsible to assure that the	
43			۷.	proper type of jacketing is being used. Failure to meet the local code will be cause for replacement of	
45				cable at no expense to CITY OF MADISON.	
46			3.	Singlemode fiber shall be dispersion un-shifted fiber in compliance with ANSI/TIA/EIA-492CAAA.	
47			4.	Intra-building fiber distribution cable design shall be according to ANSI/ICEA S-83-596.	
48			5.	Singlemode backbone fiber distribution cable, when installed, shall exceed the performance	
49			<i>c</i>	requirements of ANSI/TIA/EIA-568-C.3.	
50			6.	Singlemode optical fiber Backbone Fiber distribution cable, as specified in the Contract Documents, shall	
51 52				ue: a Mohawk Cable or equal	
52 53				 a. Worldwic Cable of Equal b. Singlemode Riser M9W042 (12 Strand) unless otherwise specified by the City of Madison 	
54				C. Singlemode Plenum M9W048 (12 Strand) unless otherwise specified by the City of Madison.	
55					
56	2.4.	CONN	IECTOR	S – FIBER OPTIC	
57		Α.	Pre-p	olished fiber connector basic design shall be a factory pre-polished Ic-style optical fiber connector with a	
58			zircor	nium ceramic ferrule.	

1		В.	Index-matching gel is factory-injected into the cleaved fiber stub splice to minimize connector insertion loss.
2		C.	LC Singlemode factory pre-polished connectors shall HAVE pre-installed fibers.
3 4		D.	Connector materials shall be designed with thermal stability to comply with environmental requirements of ANSI/TIA/FIA-568-B 3 and Telcordia GB-1081-CORE
5		F	Pre-polished ic connectors shall require no field polishing AND REOLURE NO ADHESIVES FOR TERMINATION
6		F.	Connector design and termination technique shall be independent of cable type or manufacturer, and shall be
7		••	compatible for either 900 micron huffer or 250 micron huffer distribution cables
8		G	Pre-polished IC fiber connectors, when properly installed onto gualified cable, shall meet the 10 Gb/s Ethernet
q		О.	nerformance requirements of IEEE802.3
10		ц	IC fiber connectors, properly installed onto qualified cable, shall exceed the mechanical and environmental
11			performance requirements of ANSI/TIA/FIA-568-C 3
12			Ontical fiber horizontal distribution cable, as specified in the Contract Documents, shall be:
12		1.	1 Hubball ProClick
14			2 Singlemode I C $=$ ECI C900KSM12
15			3 AFL (Fast)
16			4 Singlemode I C- fast-I C-SM
17			
18	2.5.	PATC	TH PANELS – CATEGORY 6
19		Α.	Category 6 patch panels shall be standard 8-position, RJ-45 style, un-keyed, FCC-compliant receptacle, in 24- and
20			48-port configurations.
21		В.	Panel frames shall be black powder coated 14-gage steel with rolled edges top and bottom for proper stiffness.
22		C.	Panels shall accommodate a minimum of 24 ports for each rack mount unit (1 RMU = 1.75 in.). 48 ports are
23			recommended.
24		D.	Panels shall be designed for 4-pair, 100 ohm balanced unshielded twisted pair (UTP) cable.
25		Ε.	Panels shall terminate 26-22 AWG solid conductors.
26		F.	Panels shall have individual port identification numbers on the front and rear of the panel. Panels shall have the
27			Category 6 designation, visible from the front when installed.
28		G.	Printed circuit boards shall be fully enclosed front and rear for physical protection.
29		Н.	Panel contacts shall accept a minimum of 2000 mating cycles without degradation of electrical or mechanical
30			performance.
31		١.	Panel termination method shall follow the industry standard 110 IDC punch-down, using a standard 110 impact
32			termination tool.
33		J.	Category 6 panels shall be backward compatible with existing Category 3, 5, and 5e cabling systems for fit, form,
34			and function.
35		К.	Category 6 patch panels, when installed, shall exceed the link or channel performance requirements of
36			ANSI/TIA/EIA-568-C.2.
37		L.	Category 6 patch panels shall be able to accommodate 10G in a 37 meter channel per TSB-155.
38		M.	Category 6 patch panels, as specified in the Contract Documents, shall be:
39			1. Hubbell (NEXTSPEED 6 Series)
40			2. 24 Port - P6E24U
41			3. 48 Port - P6E48U
42			
43	2.6.	CABI	NET – WALL MOUNT Declarate sight belief a subscription of the advector of the second second second finish
44		A.	Rack material shall be welded steel construction with a durable black polyurethane powder coat finish.
45		в.	Installed racks shall have a static load capacity of 300 Lbs.
40		С. Р	Cabinet shall be 48-inch high, 23-inch wide and 30-inch depth.
47		D.	rapped holes in the vertical rails for mounting of panels shall be #12-24 thread size. Coating shall not interfere
48 40		-	with thread fit.
49 E0		с. с	Cabinet shall lidve a Capacity OF 20 KIVIO.
50 E1		г.	cabinet consist of. Cabinet & mounts to secure the cabinet, a vertical Electrical 20 amp Outlet strip (winning m
57 57		G	Nell mount cabinet and accessories as specified in the Contract Documents, shall be:
52 52		О.	wan mount cabinet and accessories, as specified in the confidet Documents, shall be.
50			2 Hubbell HSQ1836
55			2. 110000111304030
56	27	CARI	F MANAGEMENT -VERTICAL CABLE MANAGEMENT - NOT LISED ON THIS PROJECT
57	_ ./.	A.	7-channel design offers:
58			1. Airflow

1			2.	Minimizes weight
2			3.	Maximum cable capacity with unobstructed access to cable
3		В.	Snap i	in Spools with ability to put them where they will do the most good.
4		C.	Rear	cable management allows cable to be run on both left and right sides, while leaving the area behind the
5			electr	onics and patch panels open for increased airflow.
6		D.	Const	ruction:
7			1.	Cold Rolled steel z-channels
8			2.	Cold rolled steel covers
9		E.	Moun	its to 84" Equipment racks
10		F.	Chanr	nel width: 6"W
11		G.	Vertic	cal Cable Management and accessories, as specified in the Contract Documents, shall be:
12			1.	Hubbell (NEXTFRAME series)
13			2.	V\$76
14				
15	2.8.	CABL	E MANA	AGEMENT –HORIZONTAL CABLE MANAGEMENT
16		Α.	Horizo	ontal management will be constructed of 14 ga cold-rolled steel (CRS)
17		В.	Finish	shall be a Durable, black powder coat.
18		С.	Size:	2RU
19		D.	Front	Ring Depth: 3.5"
20		E.	All ste	eel construction - rugged, non-flammable, no fasteners to wear or break, no fingers to fuss with.
21		F.	Modu	ilar components easily configured in field to adapt to demanding applications.
22		G.	Hinge	d Front Cover - Locks in place when completely open to prevent cover from being removed or lost.
23		н.	HORIZO	ontal Cable Management and accessories, as specified in the Contract Documents, shall be:
24			1. ว	
25			Z. Enclos	HU219UE3N surgefiber rock mount
20		1.	1	Pack mounted, nowder coated formed cold relied steel enclosure
27			1. 2	Swing-out or pull-out inner tray shall provide access to inner cables and connections, and maintain
20			۷.	proper cable hand radius throughout the range of motion
30			3	Fiber rack-mount enclosures shall be a 19-inch formed/welded and nowder coated modular design sized
31			5.	according to the cable installation.
32			4.	Fiber rack-mount enclosures may serve as a main, horizontal, or intermediate cross connect facility.
33			5.	Panel mounting brackets shall be configurable to either 19" or 23" racks per ANSI/EIA-310-D.
34			6.	Enclosure chassis shall have two mounting bracket locations for either flush mount or center mount on
35			-	the rack.
36			7.	Inner tray shall have a threaded mounting boss to accept a mounting stud for splice trays. Splice tray
37				capacity shall be (2) 10" splice trays, each with 24-splice capacities (48 splices total). Splice tray mounting
38				boss shall also accept a stud for mounting 1-RMU blown fiber adapter brackets.
39			8.	Inner tray mounting posts for modular panels shall also accept 12-fiber MTP-style cassettes for "plug &
40				play" installations.
41			9.	Inner tray shall have rear cable tie-down features to accept various diameter backbone cables entering
42				the enclosure.
43			10.	Enclosures shall be constructed of 16 gage cold rolled steel (CRS)
44			11.	Fiber rack-mount enclosures and accessories, as specified in the Contract Documents, shall be:
45				a. Clearfield – Fieldsmart fiber crossover distribution system.
46				
47		J.	Adapt	ter panels – optical fiber
48			1.	Optical fiber Adapter panels shall be a modular design powder coated stamped metal construction.
49			2.	ADAPTER PANELS SHALL BE LC.
50			3.	High or low-density versions.
51			4.	Adapter panels shall have quick-release snap fasteners to fit directly into fiber enclosures.
52			5.	Fiber patch panels, as specified in the Contract Documents, shall be:
53				a. Clearneid – Clearview class patch only cassette.
54 FF	2.0			
22 E C	2.9.			Ontic Cable shall be installed with Innordust for protection of fiber schlas in a shared pathway
50 57		А. Р	Their	Optic Cable shall be installed with innerouct for protection of fiber cables in a shared pathway.
57		ь. С	Throa	nier duct win de rateu for the environment that it is denig instanted in. Menuin diff fiser falled.
20		U.	intee	anner Ducts win be run between closets. One for current instanation, two spare for future applications.

	D.	Size: 1	1" CORRUGATED
	E.	Flexible	e & Lightweight for ease of handling
	F.	Pre-thr	readed with pull line
PART	Г 3 - ЕХЕ	CUTION	
2 1			
5.1.			annoved Contractor shall assume the following responsibilities:
	д.	1 110 AP	Execute construction in accordance with contract drawings and specifications
		1. 2	Adhere to project schedules and job site rules
		2.	Adhere to the quality, regulatory, legistics, and decumentation requirements
		⊃. ⊿	Adhere to the graduat requirements outlined in DADT 2 shous
		4. r	Adhere to the product requirements outlined in PART 2 above.
		5. 6	Autere to the execution guidelines outlined below.
		0.	Furnish the cabing system certification and warranty provisions outlined in this specification section.
3.2.	DELI	VERY. STO	ORAGE AND HANDLING LOGISTICS
	A.	Materi	ials delivered to the construction site shall be stored in a dry, secure area, preferably indoors. Storage
		temner	rature of materials shall adhere to manufacturer's recommendations. Movement of nackaged materials
		shall he	e in a manner to avoid damage of contents. On-site storage, either indoors or trailer, shall have
		nermis	ssion by the owner, and shall not interfere with other construction activity
	в	Installa	ation of category 6 cable shall be within the recommended temperature range specified by the
	5.	manuf	acturer. Cable installation temperature above 50E is recommended
		manan	
3.3.	PREF	ARATION	V
	Α.	Cable p	pathways and Firestops
		1.	Cable pathways, including conduit, cable tray, ladder rack, raceway, slots, sleeves, etc. shall be located
			and mounted according to contract drawings and manufacturer's instructions. Pathways shall not be
			installed in wet areas.
		2.	Cable pathway fill ratio, bend radius, run length, number of bends, and proximity to EMI sources shall be
			in accordance with ANSI/TIA/EIA-569-B. Maximum cable count of the initial installation shall not exceed
			40% fill ratio in any pathway.
		3.	In accordance with NEC 2005, power wiring and communications cabling shall not share the same
			pathway or outlet unless separated by a physical barrier.
		4.	Cable pathways shall be secured to a structural member of the building, or permanent wall studs. Wall
			surfaces for raceway mounting should be finished complete.
		5.	Metallic pathways shall be electrically continuous, free of sharp edges, and properly bonded to an
			approved ground. EMI sources such as ballasts, motors, and bus conductors shall be avoided by using
			proper separation distances.
		6.	Pathways that penetrate fire-rated barriers shall be fire stopped according to local codes and recognized
			practices. Fire stop materials or devices shall be qualified to UL-1479, in accordance with ASTM E814.
			Fire stop method shall have P.E. approval.
		7.	Core drilling of holes for fire-rated poke-through outlet devices shall have approval by a structural
			engineer or P.E. on the contract drawings prior to start of work.
		8.	Pathways for vertical cable runs, such as slots and sleeves, shall be installed in the proper location in
		-	accordance with applicable codes and standards.
			. P.P
	В.	Teleco	mmunications rooms and equipment rooms
		1.	Telecommunications room (TR) layout, location and design shall be in accordance with the guidelines of
			ANSI/TIA/EIA-569-B. TR's on each floor of the building should be centrally located and vertically aligned
			to simplify backbone cable and pathway routing. TR's shall not be installed in wet areas, or near EMI
			sources or caustic chemicals.
		2.	Layout of rack, cabinet or enclosure locations shall be according to contract drawings.
		3.	Racks and cabinets shall be secured to the floor using proper anchors and fasteners.
		4.	Mount and assemble racks, cabinets, brackets and enclosures per manufacturer's instructions. Mount
			patch panels and cable management accessories in the specified locations.
		5.	Adjoining pathways (ladder rack, cable tray, etc.) shall be properly secured and positioned to allow
		-	adequate bend radius of cables entering the rack or cabinet.
	C.	Wall or	utlets and recessed wall boxes

1 2			1.	Wall outlet and cable drop pathway location shall be according to contract drawings. Guidelines from ANSI/TIA/EIA-569-B should be followed for location with electrical outlets and outlet height above
3 4			2.	finished floor. Outlet boxes shall be fastened securely to a wall stud or structural element, in a manner to permit flush
5			2.	mounting of the faceplate with the finished wall.
6			3.	Multi-connect boxes shall be installed in a manner to comply with separation rules for power and
7				communications wiring in close proximity.
8			4.	Refer to specific manufacturer's recommendations for wall outlet selection, cable deployment, and
9 10		П	Surface	e housings and MUTOA outlets
10		D.	1	Received or conduit should be deployed to the surface housing location. For through-wall cable entry
12			1.	cut the wall opening to match the opening in the housing hase
13			2.	Lay out mounting holes onto the desired wall location. For wallboard, concrete or cinder block walls, drill
14				to the proper depth and install anchors.
15			3.	Always use proper wall anchors. Installing mounting screws directly into wallboard without using
16 17				anchors can cause screw pullout and detachment of the surface housing. Mounting the base plate to
18			Л	Mount have plate of surface box or MUTOA to outlet location using proper fasteners. Note: furniture and
19			т .	wall outlet applications require mounting of base plate prior to cable pulling and connector termination.
20			5.	Install cover onto base plate.
21			6.	Refer to detailed manufacturer's guidelines for cable deployment and termination of Jacks into surface
22				Certain restrictions may apply when dressing category 6 cabling into surface bousings
23 24				Certain restrictions may apply when dressing category o cabling into surface nousings.
25	3.4.	INSTA	LLATIO	N
26		A.	Cable S	Support
27			1.	This Contractor shall install all supports for cables specified in this section. Traditional Ladder rack will be
28				used in each telecommunications room, basket tray and j-hooks will be used in the horizontal.
29			2.	Cable supports shall be spaced randomly, but no further than 5'-0" apart.
30			3.	Inner-ducts will be run between each closet or telecommunications room. One for current installation
31				with three multi cells for future installations or changes. In each telecommunications room the inner-
32				ducts entering the space will be combined, in a size appropriate metallic box that is mounted on the wall.
33				The combined inner ducts will then be routed to the rack and the fiber bay.
34			4.	Provide all additional cable management products, sleeves or conduit raceways as required to protect
35 26			E	exposed cabling and complete the installation of cables in a neat manner.
30 27			5.	A nonzonial conduit system consists of conduits radiating from the telecommunications room to the workstation outlots in the floor, walks collings, and columns of a building. When using a conduit
20				distribution system utilize the most direct route following the building lines
30			6	The size and number of conduits or sleeves used for backhone nathways depends on the usable floor
40			0.	space served by the backbone system, at least three 4 trade size sleeves are recommended.
41			7.	Conduit is only required if building codes or environmental conditions necessitate it. Rigid or EMT metal
42				conduits are deemed suitable for building installation. Adequate planning should allow for a minimum of
43				one 1-inch conduits to each workstation location if code requires conduit for voice and data cables.
44			8.	Conduit fill ratios shall not exceed 40%; contact your cable manufacturer to get recommendation on fill
45				rates.
46			9.	No conduit run should be designed with more than two (2), 90 degree bends between pull points or pull
47				boxes. If a run requires more than two 90 degree bends, install a pull box.
48				a. Exceptions:
49				i. The total run is not longer than 33 feet.
50				II. I ne conduit size is increased to next trade size. III. One of the bonds is located within 12 inches of the cells and
51			10	III. One of the bends is located within 12 inches of the cable end.
52 52			10.	An conducts will be equipped with a contiguous length of plastic of hylon pull string with a minimum rating of 200 lbs. (40 Kg)
54			11	A conduit run should not be designed with continuous closed sections longer than 100 ft without null
55			±±.	points or pull boxes installed.
56			12.	All conduits should terminate above or in the installed ladder racks and allow for proper cable racking.
57				Cable waterfalls should be considered in areas that have excessive distance between the conduit and
58				ladder rack.

1 2		13.	Trays and conduits located within the ceiling shall protrude into the room a distance of 1 to 2 in without a bend and above 8 ft high. Clear unobstructed access to the ladder rack and conduits shall be provided
3			within telecommunications rooms.
4		14.	Conduits entering through the floor shall terminate at least two (2) inches above the finished floor
5		15.	Locate slot/sleeve systems in places where pulling and termination will be easy.
6		16.	If possible, locate sleeves, slots, and/or conduits on the left side of the room: this placement enhances
7		20.	the use of wall space from left to right.
8		17.	When possible, entrance conduit and distribution conduit/cable tray should enter and exit on the same
9			wall: if this is not possible, ladder rack inside the room should be provided for distribution from wall to
10			wall.
11		18.	All floor penetrations shall be core drilled with a maximum 1/4 inch size greater than the exterior
12			dimension of the riser conduit.
13		19.	Conduits entering through a wall shall be reamed and bushed, and terminated as close as practicable to
14			the terminating rack or wall.
15		20.	Terminating above a suspended ceiling must terminate not less 3 inches above finished ceiling and
16			finished with bushing opening.
17		21.	All conduit will be labeled for easy identification.
18		22.	All floor penetrations shall be at columns, exterior walls or in equipment rooms.
19		23.	Cables shall be supported at height of bottom flange of structural beams using a rigid support method
20			(i.e. threaded rod, beam clamps, etc.).
21		24.	Do not support cables from ductwork, sprinkler piping, water piping, waste piping, conduit, ceiling wire,
22			or other system supports.
23		25.	The conduits or sleeve will be installed per TIA/EIA-569-B and seal all penetration with approved fire stop
24			product.
25		26.	Provide independent support system for each low voltage cabling system.
26	В.	Cable	
27		1.	Category 6 cable will be run for data. Category 6A will be run to all Wireless access points. Category 6
28			Gelled filled cable will be run in the backbone for all communications applications. Certain environments
29			may require the use of different cables and/or cable jackets.
30		2.	All Terminations will utilize T568B wiring in THE CITY OF MADISON facility. Any Contractor not
31			complying with this wiring requirement will fix the problem at no cost to CITY OF MADISON.
32		3.	Maximum cable lengths to be 295 feet (90 m) including service loop. Provide all necessary installation
33			materials, tools and equipment to perform insulation displacement type terminations at all
34			communications outlets, patch panels.
35		4.	All communications cabling that has become abandoned as part of new renovation projects, previous
36			renovation projects, or temporary communication cables used during the construction process shall be
37			completely removed.
38		5.	Refer to detailed manufacturer's guidelines for deployment of category 6 cable. Certain restrictions
39		-	apply, and specific techniques are recommended.
40		6.	All cabling shall be installed in accordance with manufacturers' written bend radius and pulling tensions.
41			General industry guidelines recommend the following bend radius and pulling tensions:
42		_	a. Tensile loading on a single 4-pair copper UTP cable shall not exceed 25 lbf.
43		7.	Bend radius of a single 4-pair copper UTP cable shall not exceed 4 times the diameter of the cable.
44		8.	Bend radius of multi-pair copper UTP and optical fiber cable shall not exceed 10 times the diameter of
45		0	the caple.
40		9.	All conduits and conduit sleeves shall have bushings or grommets shall be installed prior to the
47			Installation of communications capies to avoid damage and abrasions to capie sneathing and insulation.
48			furnish and install buchings prior to pulling communications cabling
49 50		10	Horizontal cable length for 4 pair conner UTP cables shall not exceed 205 feet. Prior to hidding and
50		10.	installation, the contactor shall review the drawings and verify no sable run exceed 205 feet. Phot to bloding and
52 52			the communications designer of cable runs that may exceed 205 foot
52		11	Solices are not permitted in any voice or data cable unless other specified or shown on drawings
57		12.	Avoid placing conner cables near sources of extreme beat (i.e. boilers radiators beat coils)
55		13	Maintain cable twists for all LITP cables. For terminations cable sheathing shall be strinning back no more
56		10.	than %" back from termination point for all Category 6 cables
57		14	All cables shall be supported by cable tray, cable runway, or I-hooks. When large quantities of cables
58			leave travs or runways, cables shall be supported by dron-outs or cable support hardware manufactured

1 2			specifically for the purpose of supporting cables. J-hooks shall be installed a minimum of every 5 feet and cabling shall maintain minimal deflection and strain (less than 12" deflection). Cables shall not be
3			supported from ceiling grid wires. Cables shall not run above iron joists.
4		15.	All cables shall be separated and bundled into like groups.
5		16.	Service loops shall be provided at both ends of installed norizontal and backbone cabling. A 12" service
о 7			soop shall be installed in the ceiling space hear workstation outlets (excessive cable shall not be colled in outlet haves). A 10' service loop shall be provided in sommunication rooms and shall be installed to allow
7 8			for future equipment rack/cabinet relocations without the need to re-terminate natch nanels: the 10'
0			service loop shall be neatly bundled and secured in ceiling space with large D-rings or place in cable travs
10			Cable slack and service coils shall be stored properly above the ceiling or under the access floor. A
10			"figure-eight" service loop is recommended for category 6 cabling to reduce FMI coupling Loose
12			random hundling is recommended
13		17.	Any cabling installing in equipment rooms shall be neatly placed in cabling travs, cabling runways, or
14			horizontal and vertical rack/cabinet cable managers.
15		18.	Velcro straps shall be utilized in the TR and inside TC enclosures for all cable bundling. Tie wraps shall be
16			prohibited in the telecommunication rooms.
17		19.	Separation: Maintain the following distances between cables, other system cables and other building
18			systems:
19			a. One (1) foot from Fluorescent Light.
20			b. One (1) foot from power cable in parallel.
21			c. One (1) foot from electrical conduits, other systems cables or other electrical equipment.
22			d. Four (4) feet from motors or transformers
23			e. Three (3) feet from hot water piping or other mechanical equipment.
24			f. Ten (10) Feet from Bus Conductors or high-current branch circuits.
25		20.	All low voltage cables shall be run parallel or at right angles to building structural framework. Do not run
26			cables diagonally across ceiling space without written authorization by the Architect's Electrical Engineer
27			or CITY OF MADISON Representative.
28		21.	Communications cabling that must cross power cables or conduit shall cross at a 90-degree angle, and
29			shall not make physical contact.
30		22.	Fire seal around all cables running through rated floors and walls. Firestop all cables and pathways that
31			penetrate fire-rated barriers using approved methods and according to local codes.
32		23.	Leave spare pull string with every outlet installed.
33		24.	Do not install cable in wet areas, or in proximity to hot water pipes or boilers.
34		25.	Cable ends for termination shall be clean and free from crush marks, cuts, or kinks left from pulling
35			operations. Installed cable jackets shall have no abrasions with exposed conductor insulation or bare
36			copper "shiners". The installer is responsible to replace damaged cables.
37		26.	Backbone cables shall be installed and bundled separately from horizontal distribution cables. Backbone
38			and horizontal cable bundles shall be loose and random.
39		27.	Backbone cables spanning more than three floors shall be supported at the top of the cable run with a
40			wire mesh grip and on alternating floors, unless otherwise specified by local codes or manufacturer's
41			guidelines.
42		28.	Vertical runs of backbone cables entering each TR shall be securely fastened along a properly prepared
43	6	C	wall in the TR on each floor. Use of cable ladder is recommended.
44	C.	Comm	unications infrastructure Maximum achieves the teste is 205 fact (00 m) including any inclusion. Devide all accessors installation
45		1.	Maximum cable lengths to be 295 feet (90 m) including service loop. Provide all necessary installation
40		2	materials, tools and equipment.
47		Ζ.	support and secure capies at patch panels using real capie management bracket, spools of management
48		2	Gevice.
49	D	3. Ontical	Cross-connects shall be completed as per construction schedule.
50	D.	1	Inner ducts of the proper rating will be run between each closet
51		1. ว	Cables for direct burial, agrial, or other outside applications shall be designed specifically for the intended
52		۷.	capies for an ectivatial, actial, or other outside applications shall be designed specifically for the interfaced
53		3	purpose. All ontical fiber installations shall be installed using onen cabling methods. Limit cable bending radius to
54		5.	An optical fiber installations shall be installed using open cabling methods. Limit cable-bending radius to 20 times the cable diameter during installation, and 10 times the diameter after installation. Browide all
55			required tools materials consumables and equipment peressary for field mounting of LC connectors
57		4	Do not exceed the maximum null tension specified by the cable manufacturer. Use appropriate lubricants
58		- T .	as required to reduce nulling friction. Avoid kinking and twisting of cables during installation
50			as required to reduce pulling metion. Avoid kinking and twisting of cables during installation.

1		5.	Label each end of each cable as to source and destination. Terminate optical fibers in consistent,
2			consecutive manner at each end. Place all material in inner-duct between Label Optical Fiber raceway
3			cable with yellow "Caution - Optical Fiber Cable" tags every 10 feet. Leave 10 feet of slack at each fiber
4			termination point. Neatly coil slack optical fiber cable on top of rack above optical fiber patch panel
5			enclosure at each rack location.
6		6.	Optical fiber cable terminations shall utilize enclosures and components in quantities consistent with the
7			required fiber counts at each end of each segment.
8		7	During optical fiber connector termination, visually inspect all terminations with a 200 or 400-nower
9			microscone
10		Q	Follow all of the connector manufacturer's recommendations
10		0. 0	Inaccontable flaws in the terminations will include but not limited to scratches full or partial cracks
12		9.	bubbles nits one residued distributed in maintaine, but not immediate dobris The secondale
12			bubbles, bits, epoxy residual, unt, dust, on moisture, grinning and saitung debits. The acceptable
13			ladding. All unseentable connectors shall be inserted after sources
14		10	Caduring. All unacceptable connectors snall be inspected after rework.
15		10.	During installation of optical fiber cable do not allow pulling tension to exceed cable manufacturer's
16			specification for the cable being installed. Unly the strength member of the cable shall be subjected to
1/			the pulling tension.
18		11.	Clean all optical fiber connector tips prior to inserting them into matting receptacles or bulkheads. Install
19			all dust covers.
20		12.	Using approved methods, pull cable into conduit, or place into raceway or cable tray as specified. A pull
21			cord (nylon; 1/8" minimum) shall be co-installed with all cable installed in any conduit.
22		13.	Where cables are installed in air return plenum, riser rated cable shall be installed in metallic conduit.
23		14.	Backbone and horizontal cables shall be installed and bundled separately in any pathway.
24		15.	Cables above ceilings or below access floors shall be installed in cable tray or open-top cable hangers.
25		16.	Cable slack and service coils shall be stored properly above the ceiling or under the access floor. Pathway
26			fill ratio in conduit, tray, raceway, etc. shall not exceed 40% of pathway cross-sectional area.
27		17.	A service coil of at least 1 meter is recommended within workstation outlets, and at least 2 meters is
28			recommended for telecommunications enclosures. Main trunk and OSP cables shall also have a large
29			diameter service coil in the specified location.
30		18.	Recommended maximum spacing of cable supports above the ceiling is 60 in.
31		19.	Backbone cables spanning more than three floors shall be securely attached at the top of the cable run
32			with a wire mesh grip and on alternating floors or as required by local codes.
33		20.	Vertical runs of cable shall be supported to messenger strand, cable ladder, or other approved structure
34			to support the weight of the cable. Do not exceed maximum cable vertical rise limits.
35		21.	Cables that are damaged during installation shall be replaced by the contractor.
36	E.	RACKS	AND ENCLOSURES:
37		1.	Freestanding equipment racks and enclosures shall be protected free of all dust, debris and other
38			environmental elements during construction until substantial completion walk-through.
39		2.	Each rack, enclosure shall have a dedicated #6 AWG ground wire to a grounding buss bar or building
40			ground as defined by NFC.
41		3	Secure racks and enclosures to floor using rack installation kit
42	F	CATEG	ORY 6 LACKS
42		1	Refer to specific manufacturer's guidelines for termination of jacks and dressing category 6 cables inside
44		±.	wall outlets and surface housings. Due to the larger size of category 6 cable service coils in outlet boyes
44			and curfese housings are not recommended
45		2	and surface housings are not recommended.
40		2.	All lock will be wired utilizing TEGOD
47		כ. ⊿	All jack will be willed utilizing 1506B.
48		4.	To assure to daster performance, maintain wring pair twists as close as possible to the point of
49			during installation
50		-	during installation.
51		5.	The length of wiring pair un-twist in each termination shall be less than 0.5 inches (13 mm).
52		ь.	Jacks shall be properly mounted in plates, frames, or housings with dust caps fully installed over IDC
53		-	
54		7.	Horizontal cables extending from mounted jacks shall maintain a minimum bend radius of at least 4 times
55			the cable diameter, unless space is restricted. Note: Refer to specific manufacturer's recommendations
56			for restricted cable bend radius.
5/		8.	Cable terminations shall minimize tensile or bending strain on IDC contacts after assembly of faceplate or
58			housing to the wall outlet.

1		G.	CATEG	GORY 6 PATCH PANELS
2		-	1.	Properly mount patch panels into the designated rack, cabinet, or bracket locations with the #12-24
3				screws provided.
4			2.	Terminate cables behind the patch panel according to manufacturer's instructions.
5			3.	To assure performance, maintain wiring pair twists as close as possible to the point of termination. Also
6				minimize the length of exposed pairs from the jacket to the ICD termination point during installation.
7			4.	The length of wiring pair un-twist in each termination shall be less than 0.5 inches (13 mm), and shall be
8				kept to a minimum.
9			5.	Each terminated and dressed cable shall be maintained perpendicular to the rear cover using the
10				recommended cable management hardware.
11			6.	Horizontal or backbone cables extending from the rear panel terminations shall maintain a minimum
12				bend radius of at least 4 times the cable diameter.
13			7.	Cable terminations shall have minimal tensile or bending strain on panel IDC contacts in each installed
14				location.
15			8.	Panels shall be properly labeled on the front and back with the cable number and port connections for
16				each port.
17		Н.	Harsh	Environment Housing and Connectivity
18			1.	Mount connector housing from front of device but Install Gasket or optional Protective Cap before
19			2	mounting connector housing into device.
20			2.	Secure connector housing to device using supplied plastic nut. Fighten nut with 6-7 inch/pounds of
21			2	torque.
22			3. ⊿	Ensure that mounting surface is clean and free of debris.
25			4. 5	Installing the Jack into the mounted connector housing.
24			Э.	latch in the connector enoning. Retate the lack securing the spring latch
25			6	Clean and remove any obstructions from the surface that the wall plate assembly will be installed against
20			0. 7	Place washers provided with HI Impact series plates onto screws. Align rubber gasket on back side of
28			7.	nlate prior to installing to hox/wall by placing screws through plate and rubber gasket
29			8.	Secure the wall plate assembly to box/wall by tightening screws with 5 inch/pounds of torque.
30			9.	Attach patch cords and field term plug assemblies (sold separately) to the mounted connector.
31		١.	OPTIC	AL FIBER CONNECTORS, HORIZONTAL AND BACKBONE
32			1.	Installed fiber connectors shall have proper cable support, routing and strain relief.
33			2.	Installed connectors shall be inspected 100% for polish quality, and contamination.
34			3.	Fusion splices for pigtail connections shall be protected in a suitable enclosure.
35		J.	GROU	INDING and BONDING SYSTEMS: Basic Guidelines
36			1.	Telecommunications grounding and bonding system shall be installed in accordance with NEC
37				requirements, and per the guidelines of ANSI J-STD-607-A.
38			2.	The Telecommunications Main Grounding Buss Bar (TMGB) shall be bonded to the building main
39				electrical service ground (Grounding Electrode Conductor or GEC), using approved lugs or exothermic
40				weld methods. Bonding to the GEC or TMGB with sheet metal screws is prohibited.
41			3.	The Telecommunications Bonding Backbone shall be a minimum 6 AWG copper wire conductor. A
42				Telecommunications Grounding Buss Bar (TGB) shall be installed in the TR on each floor, and shall be
43			_	bonded to the TBB. All metal racks, cabinets, pathway and enclosures shall be bonded to the TGB.
44			4.	Telecommunications equipment shall be grounded according to manufacturer's instructions and in
45			-	accordance with applicable codes.
46			5.	All metallic pathways, including conduit, raceway ladder or cable trays shall be electrically continuous and
47			C	shall be bonded to ground on each end.
48			6.	USP cable entering the building or backbone cables having metal sheaths shall have isolation protection.
49 50				
50	35	I ARFI	ING	
52	5.5.	A.	Gener	ral:
53			1.	All labels shall be permanent, machine generated labels produced by a labeling machine. Labels shall
54				be a permanent polyester material clear in color with label lettering black in color. No hand written
55				labels will be accepted.
56			2.	Labeling information will be reviewed at Pre-Install Meeting, and the Owner shall approve the labeling
57				scheme prior to the installation of any cabling.

	SPEC May	SPECIFICATION May 16, 2024					
1 2 3 4		-	3. 4.	Surfaces shall be cleaned before attaching labels. All labels shall be attached firmly and vertically plumb on equipment, faceplates, patch panels termination blocks, etc. All labeling of cables, equipment, and components shall be included in as-built documentation, floor plan drawings, and schematic deigns.			
5		В.	Cablir	ng			
6 7			1.	All structured cables (horizontal and backbone) shall be labeled at both ends within 6" of cable termination point. Where voice backbone cables extend behind termination blocks, cable labels shall be placed at a location on the cable where the labels are visible from the front of the termination blocks.			
8 9 10			2.	Labels shall have an adhesive backing and shall wrap completely around the circumference of the cable jacket. Label and lettering sizes shall be of appropriate size in regard to cable diameter.			
11		C	Fauin	ment Backs, Termination Hardware, and Facenlates			
12 13		С.	1.	LABELING SCHEME TO BE SPECIFIED BY OWNER.			
14	3.6.	TEST	ING				
15		Α.	Categ	ory 6 and 6A Cable Testing			
16 17 18			1.	Permanent Link Testing shall be completed on all horizontal (station) cables. The Contractor will be responsible to supply a Channel warranty, but CITY OF MADISON is requiring that the contractor supply all manufactures patch cords per the contract			
10			2	an manufacturer patch cousper the contract.			
19			Ζ.	Category 6 and 6A cabling systems shall be tested as an installed inforzonical permanent link computation.			
20 21				be terminated complete and fully dressed in proper cable management.			
22 23			3.	All wiring shall be certified to meet or exceed the specifications as set forth in TIA-568C for Category 6 requirements for permanent link. All test will be performed to 250MHz			
24			Л	Field Testing shall include the following parameters for each pair of each cable installed:			
25			ч.	Name of the person performing the test			
25				a. Test equipment manufacturer and model number			
20				c Cable I.D. The test sheets will be in numerical order by cable ID.			
27				d Date of test			
20				u. Date of test.			
29				f Longth (in foot)			
3U 21				r. Length (in reel)			
31 22				g. Insertion Loss.			
32				n. Near End Crosstalk (NEXT).			
33				I. Power sum near End Crosstalk (PSNEXT).			
34				K. Equal-Level Far End Crosstalk (ELFEXT).			
35				I. Power Sum Equal-Level Far End Crosstalk (PSELFEXT).			
36				m. Return Loss.			
37				n. Delay Skew.			
38			-	 Attenuation to Crosstaik ratio (ACR). A "DACG" is disation shall be abtained for each link using standard intervention with 			
39			5.	A PASS indication shall be obtained for each link, using at minimum a level ill tester that complies with			
40			c	TIA/EIA-308-B.2 field lest requirements.			
41			0.	Record test results for each cable and turn over to the General Contractor Opon completion of the job.			
4Z				correct manunctions when detected, and re-test to demonstrate compliance. Note: Test equipment			
43		Б	Ontio	shall be a Type III cable Tester.			
44		в.	Optic	al Fiber Testing:			
45			1.	Test procedures shall be as described by the TIA/EIA-568-B: Commercial Building Telecommunications			
40				Cabling Standard, Parts 2 and 3 and HA/EIA-526-14-A-1998 - Optical Power Loss Measurements of			
47			2	Installed Multimode Fiber Cable Plant-OFSTP-14A			
48			Ζ.	Preinstallation Testing:			
49 E0				 a. Test each conductor of every optical fiber cable on the reel with a light source and a power meter. b. Obtain the cable manufacturer neuron meter test require for each real used on the registration. 			
5U				b. Obtain the cable manufacturer power meter test results for each real used on the project. Using			
51				the attached Optical Fiber Test Form record the readings and the manufacturer's reel number.			
52				Prior to completion of project, turn over the completed optical fiber test form, optical fiber cable			
53			2	reei i u tags and optical fiber cable manufacturer's test results.			
54			3.	Acceptance resting:			
55				a. Each terminated fiber strand in the norizontal or backbone intrastructure shall be tested			

56

57

CITY OF MADISON

	1110, 2021	
1	b.	Testing for multimode shall be at 850 and 1300 nanometers. Total link insertion loss (dB) shall be
2		within the specified link loss budget.
3	с.	Tier 1 testing for each installed singlemode link shall be performed as an optical power insertion
4		loss measurement, as defined by ANSI/TIA/EIA-526-7. Testing for singlemode shall be at 1310 and
5		1550 nanometers. Total link insertion loss (dB) shall be within the specified link loss budget.
6	d.	Tier 2 testing, if required for each installed singlemode or multimode link, shall be performed as
7		an OTDR measurement, as defined in TIA-TSB-140. We require Tier 2 testing on all fibers installed
8		in the facility for future troubleshooting.
9	e.	Multimode optical fiber attenuation shall be tested on all individual fibers of each cable segment
10		using an LED light source and power meter to determine the actual loss. These tests shall be
11		performed at the 850nm and 1300nm windows in both directions. Test set up and performance
12		shall be in accordance with ANSI/TIA/EIA-526-14A, Method B.
13	f.	A reference power measurement shall be obtained by connecting one end of test jumper 1 to the
14		light source and the other end to the power meter. After recording the reference power
15		measurement, test jumper 1 shall be disconnected from the power meter without disturbing the
16		light source and attached to the cable plant. The power meter shall be moved to the far end of
17		the cable plant and attached to the cable plant with test 2.
18	g.	Readings must not be higher than the "Optimal Attenuation Loss." The OAL will be calculated
19		using the manufacturer's factory certified test results, (db/km) converted to the actual installed
20		lengths plus the manufacturer's best published attenuation losses for the connector and/or splice
21		installed on this project. (0.30+/-0.30 for Connectors and 0.10 for splices). The construction
22		manager shall use the OAL for comparison with the end to end power loss test results prior to
23		acceptance.
24	h.	Test Results: Must be completed and turned over to the General Contractor prior to active
25		equipment installation. Specific due dates for optical fiber will be established at pre-install
26		meeting.
27		
28		END OF SECTION

		SECTION 27 21 33 WIRELESS ACCESS POINTS (WAP)			
	1 C				
PARI	1-0 11				
-	1.1. 1.2				
-	1.Z. 1.2				
	1.3. כור בי				
PARI	2 - Pr				
DADT	2.1. 2.5	WIRELESS ACCESS POINT (WAP) DEVICES			
PARI	3 - EX				
:	3.1.				
	3.2. 				
	3.3.				
:	3.4.	WAKKAN I Y			
PART	1-G	ENERAL			
1.1.	sco	DPE			
	A.	The work under this section is for the installation of OWNER PROVIDED, CONTRACTOR INSTALLED Wireless Access Points (WAP).			
	В.	The WAPs shall be installed by the contractor providing and installing the Communications Cable and Equipment. All contractor qualifications and certifications for that section shall apply to this section.			
12	RFI	ATED SPECIFICATIONS			
1.2.		The Contractor shall be responsible for reviewing all other specifications for requirements associated with the			
	7	complete installation of WAP's. This includes but is not limited to the following:			
		1 01 31 23 Project Management Web Site			
		2 27 00 05 Communications Cable and Equipment			
1.3.	SUE	BMITTALS			
	Α.	Contractor licenses and qualifications are required as part of the complete Division 27 submittal package			
		as indicated under Specification 27 00 05.			
	B.	No submittals are required for the owner provided WAP.			
	C.	Submittals are required for installation/hanger equipment, connectors, and any other required			
	-	equipment/material required for a complete WAP installation.			
<u>PART</u>	2 - PI	RODUCTS			
2.1.	WI	RELESS ACCESS POINT (WAP) DEVICES			
	Α.	The City of Madison Information Technology Department (CoM-IT) will be providing the WAP devices for this project.			
	В.	The WAP device being used will be as manufactured by the Cisco and shall be used for all types of ceiling mounted			
		installations. (suspended, gyp board, open truss, etc.).			
PART	3 - E)	(ECUTION			
3.1.	ow	/NER RESPONSIBILITIES			
	Α.	The CoM-IT shall be responsible for ordering, making payment and configuring all WAP devices.			
	В.	The CoM-IT shall configure and test each WAP to CoM-IT specifications prior to providing the WAP to the			
		contractor for installation.			
	C.	The CoM-IT shall number each WAP and provide the contractor with a location map indicating where each WAP			
		shall be installed.			
	D.	The CoM-IT shall test each WAP after installation to verify configuration and signaling is correct prior to accepting			
		the final installation of the WAP system.			
	~~				
3.2.		NIKACIUKS KESPUNSIBILITIES The Contractor shall be callely recomposible for eccordination with CoNATE the select diamond reaction of all MAD			
	А.	The contractor shall be solely responsible for coordinating with COM-11 the scheduling and receipt of all WAP			
		devices with the installation schedule.			

1 2		В.	The Contractor shall inspect all WAP devices upon receipt for damage. CoM-IT shall be notified immediately of any damage.
3 4		C.	The Contractor shall provide all mounting hardware, blocking, and other items required for a complete installation that meets the manufacturer's installation requirements.
5		D.	The Contractor shall install all WAP devices per plans and specifications including cable connections.
6 7		E.	The Contractor shall be responsible to pick up WAP devices from City IT and delivery to the jobsite.
8	3.3.	FINAL	TESTING
9		Α.	Contractor shall provide final testing of all WAP devices after installation is complete.
10 11		В.	In the event any WAP device is not operating properly the contractor shall trouble shoot the Installation and work with the CoM-IT to determine if re-configuration of the device will be required.
12 13 14		C.	The CoM-IT shall be responsible for reconfiguring WAP's as needed after installation is complete. The contractor shall be responsible for verifying connections, cabling and connectivity of the installation is correct.
15	3.4.	WAR	RANTY
16 17		A.	The CoM-IT will be responsible for registering any warranty information associated with the purchase and ownership of all the WAP devices.
18 19		В.	The Contractor shall warrant the installation of the WAP device for one (1) year per the terms of this contract.
20 21			END OF SECTION

1 2		SECTION 28 46 21 ADDRESSABLE FIRE-ALARM SYSTEMS				
3 4	<u> PART 1 - (</u>	GENERAL				
5 6	1.1	SUMMARY				
7	Δ	Section Includes:				
8	7.0	1 Addressable fire-alarm system				
9		2. Fire-alarm control panel (EACP).				
10		3. Manual fire-alarm boxes.				
11		4. System smoke detectors.				
12		5. Duct smoke detectors.				
13		6. Heat detectors.				
14		7. Fire-alarm notification appliances.				
15		8. Fire-alarm remote annunciators.				
16		9. Fire-alarm addressable interface devices.				
17		10. Digital alarm communicator transmitters (DACTs).				
18	В.	Related Requirements:				
19		1. Section 08 71 00 "Door Hardware" for magnetic door holders that release in response to fire-alarm outputs.				
20		2. Section 26 05 19 "Low-Voltage Electrical Power Conductors and Cables" or Section 26 05 23 "Control Voltage				
21		Electrical Power Cables" for cables and conductors for fire-alarm systems.				
22						
23	1.2	DEFINITIONS				
24	Α.	DACT: Digital alarm communicator transmitter.				
25	В.	FACP: Fire-alarm control panel.				
26	С.	Voltage Class: For specified circuits and equipment, voltage classes are defined as follows:				
27		1. Control Voltage: Listed and labeled for use in remote-control, signaling, and power-limited circuits supplied				
28		by a Class 2 or Class 3 power supply having rated output not greater than 150 V and 5 A, allowing use of				
29		alternate wiring methods complying with NFPA 70, Article 725.				
30		2. Low Voltage: Listed and labeled for use in circuits supplied by a Class 1 or other power supply having rated				
31		output not greater than 1000 V, requiring use of wiring methods complying with NFPA 70, Article 300, Part I.				
32						
33	1.3	SEQUENCING AND SCHEDULING				
34	Α.	Existing Fire-Alarm Equipment: Maintain existing equipment fully operational until new equipment has been tested				
35		and accepted. When new equipment is installed, label it "NOT IN SERVICE" until it is accepted. Remove labels from				
36		new equipment when put into service, and label existing fire-alarm equipment "NOT IN SERVICE" until removed from				
37		building.				
38	В.	Equipment Removal: After acceptance of new fire-alarm system, remove existing disconnected fire-alarm equipment				
39		and wiring.				
40						
41	1.4	SUBMITTALS				
42	А.	Product Data: For each type of product, including furnished options and accessories.				
43		1. Include construction details, material descriptions, dimensions, profiles, and finishes.				
44		2. Include rated capacities, operating characteristics, and electrical characteristics.				
45	В.	Shop Drawings: For fire-alarm system.				
46		1. Comply with recommendations and requirements in "Documentation" section of "Fundamentals" chapter in				
47		NFPA /2.				
48 40		2. Include plans, elevations, sections, and details, including details of attachments to other work.				
49 50		5. Include details of equipment assemblies. Indicate dimensions, weights, loads, required clearances, method				
50 E1		of field assembly, components, and locations. Indicate conductor sizes, indicate termination locations and requirements, and distinguish between factory and field wiring.				
51 51		A Detail assembly and support requirements				
52 52		4. Detail assembly and support requirements.				
22		5. Include voltage utop calculations for normication-appliance circuits.				
54 55		 Include ballety-size calculations. Include input/output matrix 				
56		 Include written statement from manufacturer that equipment and components have been tested as a system. 				
57		and comply with requirements in this Section and in NFDA 72				
58		9 Include performance parameters and installation details for each detector				

1		10.	'erify that each duct detector is listed for complete range of air velocity, temperature, and humidity possible	
3		11.	rovide control wiring diagrams for fire-alarm interface to HVAC; coordinate location of duct smoke detectors	
4 5			nd access to them. . Show critical dimensions that relate to placement and support of sampling tubes, detector housing,	
6			and remote status and alarm indicators.	
7			. Show field wiring and equipment required for HVAC unit shutdown on alarm.	
8			. Locate detectors in accordance with manufacturer's written instructions.	
9		12.	nclude voice/alarm signaling-service equipment rack or console layout, grounding schematic, amplifier	
10			ower calculation, and single-line connection diagram.	
11		13.	nclude floor plans to indicate final outlet locations showing address of each addressable device. Show size	
12			nd route of cable and conduits and point-to-point wiring diagrams.	
13	C.	Delega	d Design Submittal: For notification appliances and smoke and heat detectors, in addition to submittals listed	
14 15		above	alcate compliance with performance requirements and design criteria, including analysis data signed and available professional engineer responsible for their proparation	
15		sealed	volumed professional engineer responsible for their preparation.	
10		1.	istallation details as needed to comply with listing conditions of device.	
18		2.	besign Calculations: Calculate requirements for selecting spacing and sensitivity of detection, complying with	
19			IFPA 72. Calculate spacing and intensities for strobe signals and sound-pressure levels for audible appliances.	
20		3.	ndicate audible appliances required to produce square wave signal per NFPA 72.	
21				
22	1.5	CLOSE	JT SUBMITTALS	
23	А.	Opera	n and Maintenance Data: For fire-alarm systems and components to include in emergency, operation, and	
24 25		mainte	Ince manuals.	
25		1.	allowing and deliver conjects authorities having jurisdiction:	
20			Comply with "Records" section of "Inspection Testing and Maintenance" chapter in NEPA 72	
27			Provide "Fire-Alarm and Emergency Communications System Record of Completion Documents" in	
29			accordance with "Completion Documents" Article in "Documentation" section of "Fundamentals"	
30			chapter in NFPA 72.	
31			. Complete wiring diagrams showing connections between devices and equipment. Each conductor	
32			must be numbered at every junction point with indication of origination and termination points.	
33			. Riser diagram.	
34			. Device addresses.	
35			Air-sampling system sample port locations and modeling program report showing layout meets	
36			performance criteria.	
37			. Record copy of site-specific software.	
38			. Provide "Inspection and Testing Form" in accordance with "Inspection, Testing and Maintenance"	
39			chapter in NFPA 72, and include the following:	
40			1. Equipment tested.	
41			2. Frequency of testing of installed components.	
42			 Frequency of inspection of installed components. A province state and account of the state of the second state of the state of the second st	
43			4. Requirements and recommendations related to results of maintenance.	
44 15			5. Widnuidclurer's user training manuals. Manufacturer's required maintenance related to system warranty requirements	
45			Abbreviated operating instructions for mounting at EACP and each appunciator unit	
40 47			Abbreviated operating instructions for mounting at PACP and each annunciator unit.	
48	1.6	MAIN	VANCE MATERIAL SUBMITTALS	
49	 A.	Extra	ck Material: Furnish extra materials that match products installed and that are packaged with protective	
50		coveri	for storage and identified with labels describing contents.	
51		1.	amps for Remote Indicating Lamp Units: Quantity equal to 10 percent of amount installed, but no fewer than	
52			ne unit.	
53		2.	amps for Strobe Units: Quantity equal to 10 percent of amount installed, but no fewer than one unit.	
54		3.	moke Detectors, Fire Detectors: Quantity equal to 10 percent of amount of each type installed, but no fewer	
55			nan one unit of each type.	
56		4.	retector Bases: Quantity equal to two percent of amount of each type installed, but no fewer than one unit	
57		_	t each type.	
58		5.	eys and Tools: One extra set for access to locked or tamper proofed components.	

l

	6. 7.	Audibl Fuses: types	le and Visual Notification Appliances: One of each type installed. Two of each type installed in system. Provide in box or cabinet with compartments marked with fuse and sizes.
1.7	WARR	RANTY	
Α.	Specia becau:	al Warra se of de	nty: Manufacturer agrees to repair or replace fire-alarm system equipment and components that fail fects in materials or workmanship within specified warranty period.
	1.	Warra	nty Period: Five years from date of Substantial Completion.
<u> PART 2 - I</u>	PRODUC	<u>TS</u>	
2.1	ADDR	ESSABLE	E FIRE-ALARM SYSTEM
А.	Descri	ption:	
	1.	Nonco notific	nded, UL-certified addressable system, with multiplexed signal transmission and voice -and-strobe cation for evacuation.
В.	Perfor	mance (Criteria:
	1.	Regula	atory Requirements:
		a.	Fire-Alarm Components, Devices, and Accessories: Listed and labeled by a NRTL in accordance with NEPA 70 for use with selected fire-alarm system and marked for intended location and application
	2.	Gener	al Characteristics:
		a.	Automatic sensitivity control of certain smoke detectors.
		b.	Fire-alarm signal initiation must be by one or more of the following devices and systems:
			1. Manual stations.
			2. Heat detectors.
			3. Smoke detectors.
			4. Duct smoke detectors.
			5. Automatic sprinkler system water flow.
			6. Fire-extinguishing system operation.
			7. Fire standpipe system.
		с.	Fire-alarm signal must initiate the following actions:
			 Continuously operate alarm notification appliances including voice evacuation notices. Identify alarm and specific initiating device at EACP and somete appliances.
			 Indentity diaminate specific initiating device at FACF and remote annunciators. Inlock electric door locks in designated egress naths
			 Onlock electric door locks in designated egress paths. Release fire and smoke doors held open by magnetic door holders
			5. Activate voice/alarm communication system.
			6. Switch HVAC equipment controls to fire-alarm mode.
			7. Close smoke dampers in air ducts of designated air-conditioning duct systems.
			8. Record events in system memory.
		d.	Supervisory signal initiation must be by one or more of the following devices and actions:
			1. Valve supervisory switch.
			2. Independent fire-detection and -suppression systems.
			3. Zones or individual devices have been disabled.
			4. FACP has lost communication with network.
		e.	System trouble signal initiation must be by one or more of the following devices and actions:
			 Open circuits, shorts, and grounds in designated circuits. Opening temperature with an energy include a large initiating and even without sized initiating devices.
			 Opening, tampering with, or removing alarm-initiating and supervisory signal-initiating devices. Loss of communication with addressable consort input module, relay, control module, remote
			3. Loss of communication with addressable sensor, input module, relay, control module, remote
			4 Loss of primary power at FΔCP
			5. Ground or single break in internal circuits of FACP.
			6. Abnormal ac voltage at FACP.
			7. Break in standby battery circuitry.
			8. Failure of battery charging.
			9. Abnormal position of switch at FACP or annunciator.
			10. Voice signal amplifier failure.
		f.	System Supervisory Signal Actions:
			1. Identify specific device initiating event at FACP and remote annunciators.
			2. Transmit system status to building management system.

1		 Display system status on graphic annunciator. 	
2		g. Device Gualus.	tor sons or
<u>з</u>		1. Description: welded wire mesh of size and shape for manual station, smoke detec	tor, going, or
5		a) Factory fabricated and furnished by device manufacturer	
6		h) Finish: Daint of color to match protected device	
7		b Document Storage Boy:	
, 8		1 Description: Enclosure to accommodate standard 8-1/2-by-11 inch manuals and loo	se document
9		records Legend sheet will be permanently attached to door for system required door	rumentation
10		key contacts and system information. Provide two key ring holders with location	on to mount
11		standard business cards for key contact personnel	
12		2. Material and Finish: 18-gauge cold-rolled steel: four mounting holes.	
13		3. Color: Red powder-coat epoxy finish.	
14		 Labeling: Permanently screened with 1-inch-high lettering "SYSTEM RECORD DOCUL 	MENTS" with
15		white indelible ink.	
16		5. Security: Locked with 3/4-inch barrel lock. Provide solid 12-inch stainless steel pian	o hinge.
17			0
18	2.2	FIRE-ALARM CONTROL PANEL (FACP)	
19	Α.	Manufacturers: Subject to compliance with requirements, available manufacturers offering products	that may be
20		incorporated into the Work include, but are not limited to, the following:	
21		1. Bosch Security Systems, Inc.	
22		2. Edwards; Carrier Global Corporation.	
23		Gamewell-FCI; Honeywell International, Inc.	
24		4. Notifier; Honeywell International, Inc.	
25		5. Potter Electric Signal Company, LLC.	
26		6. Siemens Industry, Inc., Building Technologies Division.	
27		7. Simplex; brand of Johnson Controls International plc, Building Solutions North America.	
28		8. Or approved equal	
29	В.	Description: Field-programmable, microprocessor-based, modular, power-limited design with electronic	c modules.
30	С.	Performance Criteria:	
31		1. Regulatory Requirements: Comply with NFPA 72 and UL 864.	
32		2. General Characteristics:	
33		a. System software and programs must be held in nonvolatile flash, electrically erasable, pro	ogrammable,
34		read-only memory, retaining information through failure of primary and secondary pow	er supplies.
35		b. Include real-time clock for time annotation of events on event recorder and printer.	
36		c. Provide communication between FACP and remote circuit interface panels, annun	iciators, and
3/		alsplays.	
38		 FACP must be listed for connection to central-station signaling system service. Dravide nervice static research for output detekage legic and exercise system and exercise static s	
39		e. Provide nonvolatile memory for system database, logic, and operating system and e	vent history.
40		System must require no manual input to initialize in the event of complete power dow	m condition.
41 12		f Addressable Initiation Device Circuits: EACP must indicate which communication zone	s have been
42		silenced and must provide selective silencing of alarm potification appliance	by building
43 11		communication zone	by building
45		1 Addressable Control Circuits for Operation of Notification Appliances and	Mechanical
46		Equipment: FACP must be listed for releasing service	meenamear
47		g. Fire-Alarm Annunciator: Arranged for interface between human operator at FACP and	addressable
48		system components including annunciation and supervision. Display alarm, supe	rvisory, and
49		component status messages and programming and control menu.	,,
50		1. Annunciator and Display: LCD, 80 characters, minimum.	
51		2. Keypad: Arranged to permit entry and execution of programming, display,	and control
52		commands.	
53		h. Initiating-Device, Notification-Appliance, and Signaling-Line Circuits:	
54		1. Pathway Class Designations: NFPA 72, Class B	
55		2. Pathway Survivability: Level 1.	
56		3. Install no more than 50 addressable devices on each signaling-line circuit.	
57		4. Install fault circuit isolators to comply with circuit performance requirements of NFF	PA 72 or with
58		manufacturer's written instructions, whichever is more conservative.	

1		i	i. S	Serial Interfaces:		
2			1	L. One dedicated RS 485 port for central-station operation using point ID DACT.		
3			2	2. One RS 485 port for remote annunciators, Ethernet module, or multi-interface module (printer		
4				port).		
5			. 3	3. One RS 232 port for voice evacuation interface.		
6		j	j. r	Notification-Appliance Circuit:		
7			1	L. Audible appliances must sound in three-pulse temporal pattern, as defined in NFPA 72.		
8			2	2. Where notification appliances provide signals to sleeping areas, alarm signal must be 520 Hz		
9				square wave with intensity 15 dB above average ambient sound level or 5 dB above maximum		
10				sound level, or at least 75 dB(A-weighted), whichever is greater, measured at pillow.		
11			3	3. Visual alarm appliances must flash in synchronization where multiple appliances are in same field		
12			_	of view, as defined in NFPA 72.		
13			k. [Door Controls: Door hold-open devices that are controlled by smoke detectors at doors in smoke-		
14			. t	parrier walls must be connected to fire-alarm system.		
15			I. T	ransmission to Remote Alarm Receiving Station: Automatically transmit alarm, supervisory, and		
16			t	rouble signals to remote alarm station.		
17		l	m. \	oice/Alarm Signaling Service: Central emergency communication system with redundant		
18			r	nicrophones, preamplifiers, amplifiers, and tone generators provided as special module that is part		
19			C	of FACP.		
20			n. I	ndicate number of alarm channels for automatic, simultaneous transmission of different		
21			a	announcements to different zones or for manual transmission of announcements by use of central-		
22			C	control microphone. Amplifiers must comply with UL 1/11.		
23			1	 Allow application of, and evacuation signal to, indicated number of zones and simultaneously allowurs is assigned to athen as a second signal to a si		
24			_	allow voice paging to other zones selectively or in combination.		
25			2	2. Programmable tone and message sequence selection.		
26			2	 Standard digitally recorded messages for "Evacuation" and "All Clear." Consists tapped to be accurated with evaluation and the recorded by NEDA 73 and 		
27			2	4. Generate tones to be sequenced with audio messages of type recommended by NFPA 72 and that are compatible with tone patterns of patification appliance singuite of FACP.		
28			~ ~ ~	that are compatible with tone patterns of notification-appliance circuits of FACP.		
29			0. 3	status Annunciator: inuicate status of various voice/alarm speaker zones and status of menginers		
3U 21			۱ س ۲	wo-way telephone communication zones.		
31 22			р. н	reampliners, ampliners, and tone generators must automatically transfer to backup units, on		
5Z 22			۲ م	miniary equipment failure.		
33 24			q. ғ	Anicout of Events: Of receipt of signal, print diarm, supervisory, and trouble events. Identify zone,		
24 25				sevice, and function. Include type of signal (diam, supervisory, of flotuble) and date and time of		
26			:	ncluding same information for dovice, location, date, and time. Commands initiate printing of list of		
30 27			1	including same information for device, location, date, and time. Commands initiate printing of its of existing alarm, supervision, and trouble conditions in system and historical log of events		
20			r D	Primary Power: 24 V/dc) obtained from 120 V/ac) service and nower-supply module initiating		
30			т. г с	having rower. 24 v(uc) obtained from 120 v(ac) service and power-supply module. Initiating		
10				nowered by 24 V(dr) source		
40 41			۲ ک د ل	Narm current draw of entire fire-alarm system must not exceed 80 percent of power-supply module		
4 <u>1</u> 42			5. / r	ating		
42			י t כ	Secondary Power: 24 V(dc) supply system with batteries, automatic battery charger, and automatic		
43			t. 5	ransfer switch		
45			u. F	Batteries: Sealed lead calcium.		
46	D	Accesso	ries [.]			
47	5.	1.	Instructio	ons: Computer printout or typewritten instruction card mounted behind plastic or glass cover in		
48			stainless	steel or aluminum frame. Include interpretation and describe appropriate response for displays and		
49			signals. E	Briefly describe functional operation of system under normal, alarm, and trouble conditions.		
50		2.	Preactio	n System Functionality:		
51			a. I	nitiate Presignal Alarm: This function must cause audible and visual alarm and indication to be		
52			r	provided at FACP. Activation of initiation device connected as part of preaction system must be		
53			a	annunciated at FACP only, without activation of general evacuation alarm.		
54				· •		
55	2.3	MANUAL FIRE-ALARM BOXES				
56	Α.	Manufa	cturers:	Subject to compliance with requirements, available manufacturers offering products that may be		
57		incorpoi	rated inte	o the Work include, but are not limited to, the following:		
58		1.	Bosch Se	curity Systems, Inc.		

1		2. Edwar	rds; Cai	rier Glob	al Corporation.
2		3. Game	well-FC	ी; Honey	well International, Inc.
3		4. Notifie	er; Hor	evwell Ir	iternational, Inc.
4		5. Potter	r Electr	, ic Signal (Company, LLC.
5		6. Sieme	ens Indi	ustry. Inc.	Building Technologies Division.
6		7 Simple	ex bra	nd of Ioh	nson Controls International nlc. Building Solutions North America
7		8 Or an	nroved	equal	
γ Q	в	General Requi	iromon	ts for Ma	anual Fire-Alarm Boyes: Comply with LIL 38, Boyes must be finished in red with molded
0	D.	raised letter	oporati	ng instru	indal file-Alarm boxes. Comply with or 56. boxes must be missive infred with model,
10		mounted on r	Sperati	d outlot l	ictions in contrasting color, must show visible indication of operation, and must be
10			ecesse	u outiet t	Jox. Il indicated as surface mounted, provide manufacturer's surface back box.
11		I. Single	-action	mechar	ism, pull-lever type; with integral addressable module arranged to communicate
12		manua	al-stati	on status	(normal, alarm, or trouble) to FACP.
13		2. Statio	n Reset	t: Key- or	wrench-operated switch.
14		3. Indoo	r Prote	ctive Shie	eld: Factory-fabricated, clear plastic enclosure hinged at top to permit lifting for access
15		to init	iate ala	arm. Liftir	ng cover actuates integral battery-powered audible horn intended to discourage false-
16		alarm	operat	ion.	
17		4. Able t	o perfo	orm at up	to 90 percent relative humidity at 90 deg F .
18		5. Able t	o be us	ed in ind	loor areas.
19					
20	2.4	SYSTEM SMO	KE DET	ECTORS	
21	Α.	Photoelectric	Smoke	Detector	rs:
22		1. Manu	facture	ers: Subje	ect to compliance with requirements, available manufacturers offering products that
23		may b	e incor	porated	into the Work include, but are not limited to, the following:
24		a.	Bosc	h Securit [,]	y Systems, Inc.
25		b.	Edwa	ards; Carr	rier Global Corporation.
26		С.	Gam	ewell-FCI	: Honeywell International. Inc.
27		d	Notif	ier Hone	vwell International Inc
28		а. е	Potte	er Flectric	c Signal Company LLC
20		c. f	Siem	ons Indu	stry Inc. Building Technologies Division
20		1. a	Simn	low bran	d of Johnson Controls International plc. Building Solutions North America
50 21		g. k	Simp	iex, biali	a of Johnson Controls International pic, Building Solutions North America.
31		11. 2 Doutes	Orap	Jproved e	equal
32		Z. Perfor	mance	Criteria:	
33		a.	Regu	latory Re	quirements:
34			1.	NFPA /2.	
35			2.	UL 268.	
36		b.	Gene	eral Chara	acteristics:
37			1.	Detector	's must be two-wire type.
38			2.	Integral	Addressable Module: Arranged to communicate detector status (normal, alarm, or
39				trouble)	to FACP.
40			3.	Base Mo	unting: Detector and associated electronic components must be mounted in twist-lock
41				module t	that connects to fixed base. Provide terminals in fixed base for connection to building
42				wiring.	
43			4.	Self-Rest	oring: Detectors do not require resetting or readjustment after actuation to restore
44				them to	normal operation.
45			5.	Integral	visual-Indicating Light: LED type, indicating detector has operated and power-on status.
46			6.	Detector	address must be accessible from FACP and must be able to identify detector's location
47			0.	within sv	istem and its sensitivity setting
18			7	Operator	r at EACP having designated access level must be able to manually access the following
70 //Q			/.	for each	detector.
50				یں۔ ما	Drimary status
50				aj b)	ninary status.
21				D)	Device type. Drosont average value
52				C)	Present average value.
53				d)	Present sensitivity selected.
54				e)	Sensor range (normal, dirty, etc.).
55			8.	Detector	must have functional humidity range within 10 to 90 percent relative humidity.
56			9.	Sensitivit	y levels based on time of day.
57					

1	2 5	DUCT SMOKE DETECTORS							
1	2.5	DUCT SMUKE DETECTORS							
2	А.	Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may b							
3		incorporated into the Work include, but are not limited to, the following:							
4		1. Bosch Security Systems, Inc.							
5		2. Edwards; Carrier Global Corporation.							
6		3. Gamewell-FCI; Honeywell International, Inc.							
7		4. Notifier; Honeywell International, Inc.							
8		5. Potter Electric Signal Company, LLC.							
9		6. Siemens Industry, Inc., Building Technologies Division.							
10		7. Simplex; brand of Johnson Controls International plc, Building Solutions North America.							
11		8. Or approved equal							
12	В.	Description: Photoelectric-type, duct-mounted smoke detector.							
13	C.	Performance Criteria:							
14		1. Regulatory Requirements:							
15		a. NFPA 72							
16									
17		2 Generatoristics							
10		2. General characteristics.							
10		a. Detectors must be rour -wire type.							
19		b. Integral Addressable Module: Arranged to communicate detector status (normal, alarm, or trouble)							
20									
21		c. Self-Restoring: Detectors do not require resetting or readjustment after actuation to restore them to							
22		normal operation.							
23		d. Integral Visual-Indicating Light: LED type, indicating detector has operated and power-on status.							
24		e. Detector address must be accessible from FACP and must be able to identify detector's location within							
25		system and its sensitivity setting.							
26		f. Operator at FACP, having designated access level, must be able to manually access the following for							
27		each detector:							
28		1. Primary status.							
29		2. Device type.							
30		3. Present average value.							
31		4. Present sensitivity selected.							
32		5 Sensor range (normal dirty etc.)							
22		g Weatherproof Duct Housing Enclosure: NEMA 250 Type 4X: NETL listed for use with supplied							
37		g. Weather proof back detaction in HVAC system ducts							
25 25		b Each concer must have multiple of detection constituity							
55 26		Lach sensor must have multiple levels of detection sensitivity. Sensitivity Compliant Tubers: Design and dimensions as recommended by							
30		i. Sampling rubes, besign and dimensions as recommended by manufacturer for specific duct size, an							
37		velocity, and installation conditions where applied.							
38		J. Relay Fan Shutdown: Fully programmable relay rated to interrupt fan motor-control circuit.							
39									
40	2.6	HEAT DETECTORS							
41	Α.	Combination-Type Heat Detectors:							
42		1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that							
43		may be incorporated into the Work include, but are not limited to, the following:							
44		a. Bosch Security Systems, Inc.							
45		b. Edwards; Carrier Global Corporation.							
46		c. Gamewell-FCI; Honeywell International, Inc.							
47		d. Potter Electric Signal Company, LLC.							
48		e. Siemens Industry, Inc., Building Technologies Division.							
49		f. Simplex; brand of Johnson Controls International plc, Building Solutions North America.							
50		g. Or approved equal							
51		2. Performance Criteria:							
52		a. Regulatory Requirements:							
53		1 NFPA 72							
57		2 111 521							
54		2. OL J21. h General Characteristics:							
55		U. UCHELDI CHDI DULCH SUICS.							
00 57		Temperature sensors must test for and communicate sensitivity range of device.							
5/ F0		c. Actuated by fixed temperature of 155 deg F or rate of rise that exceeds 15 deg F per minute unless							
50		otherwise multated.							

1		d	Mounting, Twist lock base interchangeable with smalle detector bases				
1		u.	Integral Addressable Medule: Arranged to communicate detector status (normal alarm, or trouble)				
2		e.	Integral Addressable Module: Arranged to communicate detector status (normal, alarm, or trouble)				
3		to FAUP.					
4	п	T. Detector must have functional number y range of 10 to 90 percent relative number.					
5	в.	rixed-remperature-Type Heat Detectors:					
0		I. IVIdriu	racturers. Subject to compliance with requirements, available manufacturers offering products that				
/		may b	Preseb Security Systems, Inc.				
8 0		d. h	Busch Security Systems, inc.				
9		D.	Edwards; Carrier Global Corporation.				
10		C.	Gameweil-FCI; Honeyweil International, Inc.				
11		d.	Notifier; Honeywell International, Inc.				
12		e.	Poller Electric Signal Company, LLC.				
13		Т.	Siemens Industry, Inc., Building Technologies Division.				
14		g.	Simplex; brand of Johnson Controls International pic, Building Solutions North America.				
15		n. Deufeu	Or approved equal				
16		2. Perfor	mance Criteria:				
17		a.	Regulatory Requirements:				
18			1. NFPA /2.				
19			2. UL 521.				
20		b.	General Characteristics:				
21			1. Actuated by temperature that exceeds fixed temperature of 190 deg F.				
22			2. Mounting: Twist-lock base interchangeable with smoke-detector bases.				
23			3. Integral Addressable Module: Arranged to communicate detector status (normal, alarm, or				
24			trouble) to FACP.				
25			4. Detector must have functional humidity range of 10 to 90 percent.				
26							
27	2.7	FIRE-ALARIVI N	NOTIFICATION APPLIANCES				
28	А.	Fire-Alarm Vol	ice/ Ione Notification Appliances:				
29		I. IVIAIIUI	Beach Coourity Systems the				
30		a.	Bosch Security Systems, Inc.				
31		D.	Edwards; Carrier Global Corporation.				
32		С.	Gameweil-FCI; Honeyweil International, Inc.				
33		a.	Notifier; Honeywell International, Inc.				
34		e.	Potter Electric Signal Company, LLC.				
35		Г.	Siemens Industry, Inc., Building Technologies Division.				
36		g.	Simplex; brand of Johnson Controls International pic, Building Solutions North America				
37		n. 2 Daari	Ur approved equal.				
38		2. Descri	ption: Notification appliances capable of outputting voice evacuation messages.				
39		3. Perfor	mance Criteria:				
40		a.	Regulatory Requirements:				
41			1. NFPA /2.				
42		6	2. UL 1480.				
43		D.	General Characteristics:				
44			1. Speakers for voice Notification: Locate speakers for voice notification to provide intelligibility				
45			requirements of Notification Appliances and Emergency Communications systems chapters				
46			IN NEPA 72.				
47			2. High-Kange Units: Rated 2 to 15 W.				
48			LOW-Range Units: Rateu 1 to 2 W. Mounting: Fluch or surface mounted and hidirectional ner the plane				
49			4. Mounting, riush of surface mounted and bidirectional per the plans.				
50			5. Matching Transformers: Tap range matched to acoustical environment of speaker location.				
51			6. Combination Devices: Factory-Integrated audible and visible devices in single-mounting				
52	п	assembly, equipped for mounting as indicated, and with screw terminals for system connections.					
55 E 4	В.	Fire-Alarm visible Notification Appliances:					
54 55		1. Intranutacturers: Subject to compliance with requirements, available manufacturers offering products that					
55		may be incorporated into the work include, but are not limited to, the following:					
50 57		а. к	Luwaius, Cattlet Global Colporational Inc.				
57 58		D.	Notifier, Honeywell International, Inc. Dotter Electric Signal Company, IIC				
20		ι.	Foller Lieunic Signal Company, LLC.				

1		Ь	Siemens Industry, Inc., Building Technologies Division					
2		и. е	Simplex: brand of Johnson Controls International plc. Building Solutions North America					
2		f Or approved equal						
4		2. Performance Criteria:						
5		2. renon	Regulatory Requirements:					
6		u.	1 NFPA 72					
7			2 1971					
, 8		h	General Characteristics:					
9		5.	1 Rated Light Output:					
10			a) $15/30/75/95/110$ cd selectable in field					
11			2 Clear or nominal white polycarbonate lens mounted on aluminum facenlate					
12			 Mounting: Wall mounted unless otherwise indicated 					
12			A For units with guards to prevent physical damage light output ratings must be determined with					
1/			guards in place					
15			5 Elashing must be in temporal pattern, synchronized with other units					
16			5. Hashing must be in temporal pattern, synchronized with other units.					
17			7 Mounting Eacentate: Eactory finished red					
18								
10	28	FIRE-ALARM R						
20	Δ	Manufacturers	: Subject to compliance with requirements, available manufacturers offering products that may be					
20	Π.	incorporated in	to the Work include but are not limited to the following:					
22		1 Bosch	Security Systems Inc					
22		2 Or ann	roved equal					
23	в	Performance C	riteria:					
25	υ.	1 Regula	tory Requirements:					
25		1. Negula	NEPA 72					
20		2 Genera	I Characteristics					
27		2. Ochere	Annunciator functions must match those of FACP for alarm supervisory and trouble indications					
29		u.	Manual switching functions must match those of FACP including acknowledging silencing resetting					
30			and testing					
31			1 Mounting: Surface cabinet NEMA 250 Type 1					
32		h	Display Type and Functional Performance: Alphanumeric display and LED indicating lights must match					
32		5.	those of EACP Provide controls to acknowledge silence reset and test functions for alarm					
34			supervisory and trouble signals					
35			supervisory, and trouble signals.					
36	2.9	FIRF-AI ARM A	DDRESSARI E INTEREACE DEVICES					
37	Δ	Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be						
38	7	incornorated into the Work include, but are not limited to the following:						
39		1 Bosch	Security Systems Inc					
40		2 Notifie	r: Honeywell International Inc					
40		3 Or ann	roved equal					
42	B.	Performance C	riteria:					
43	5.	1 Regula	tory Requirements:					
44		a a	NEPA 72					
45		2. Genera	I Characteristics:					
46		a	Include address-setting means on module					
47		b.	Store internal identifying code for control nanel use to identify module type					
48		с. С.	Listed for controlling HVAC fan motor controllers.					
49		d.	Monitor Module: Microelectronic module providing system address for alarm-initiating devices for					
50			wired applications with normally open contacts.					
51		P.	Integral Relay: Capable of providing direct signal to elevator controller to initiate elevator recall					
52			1. Allow control panel to switch relay contacts on command.					
53			2. Have minimum of two normally open and two normally closed contacts available for field wiring					
54		f.	Control Module:					
55		••	1. Operate notification devices.					
56			2. Operate solenoids for use in sprinkler service.					

PART 3 - EXECUTION

3.1 **EXAMINATION**

Examine areas and conditions for compliance with requirements for ventilation, temperature, humidity, and other Α. conditions affecting performance of the Work. 1 Verify that manufacturer's written instructions for environmental conditions have been permanently

- established in spaces where equipment and wiring are installed, before installation begins.
- В. Examine roughing-in for electrical connections to verify actual locations of connections before installation.
- Proceed with installation only after unsatisfactory conditions have been corrected. C.

3.2 PREPARATION 11

- A. Preinstallation Testing: Perform verification of functionality of installed components of existing system prior to starting work. Document equipment or components not functioning as designed.
- В. Interruption of Existing Fire-Alarm Service: Do not interrupt fire-alarm service to facilities occupied by Owner or others unless permitted under the following conditions and then only after arranging to provide temporary guard service in accordance with requirements indicated:
 - Notify Owner no fewer than seven days in advance of proposed interruption of fire-alarm service. 1.
 - 2. Do not proceed with interruption of fire-alarm service without Owner's written permission.
 - C. Protection of In-Place Conditions: Protect devices during construction unless devices are placed in service to protect facility during construction.
- 22 INSTALLATION OF EQUIPMENT 3.3
 - Comply with NECA 305, NFPA 72, NFPA 101, and requirements of authorities having jurisdiction for installation and Α. testing of fire-alarm equipment. Install electrical wiring to comply with requirements in NFPA 70 including, but not limited to, Article 760, "Fire Alarm Systems."
 - Devices placed in service before other trades have completed cleanup must be replaced. 1.
 - 2. Devices installed, but not yet placed, in service must be protected from construction dust, debris, dirt, moisture, and damage in accordance with manufacturer's written storage instructions.
 - Β. Install wall-mounted equipment, with tops of cabinets not more than 78 inch above finished floor.

C. Manual Fire-Alarm Boxes:

- Install manual fire-alarm box in normal path of egress within 60 inch of exit doorway. 1.
- 2. Mount manual fire-alarm box on background of contrasting color.
- Operable part of manual fire-alarm box must be between 42 and 48 inch above floor level. Devices must be 3. mounted at same height unless otherwise indicated.
- D. Smoke- and Heat-Detector Spacing:
 - Comply with "Smoke-Sensing Fire Detectors" section in "Initiating Devices" chapter in NFPA 72, for smoke-1. detector spacing.
 - 2. Comply with "Heat-Sensing Fire Detectors" section in "Initiating Devices" chapter in NFPA 72, for heatdetector spacing.
 - 3. Smooth ceiling spacing must not exceed 30 ft. .
 - Spacing of detectors for irregular areas, for irregular ceiling construction, and for high ceiling areas must be 4. determined in accordance with Annex A or Annex B in NFPA 72.
 - 5. HVAC: Locate detectors not closer than 36 inch from air-supply diffuser or return-air opening.
 - Lighting Fixtures: Locate detectors not closer than 12 inch from lighting fixture and not directly above pendant 6. mounted or indirect lighting.
- Ε. 46 Install cover on each smoke detector that is not placed in service during construction. Cover must remain in place 47 except during system testing. Remove cover prior to system turnover.
 - F. Duct Smoke Detectors: Comply with NFPA 72 and NFPA 90A. Install sampling tubes so they extend full width of duct. Tubes more than 36 inch long must be supported at both ends.
 - Do not install smoke detector in duct smoke-detector housing during construction. Install detector only during 1. system testing and prior to system turnover.
- 52 G. Remote Status and Alarm Indicators: Install in visible location near each smoke detector, sprinkler water-flow switch, 53 and valve-tamper switch that is not readily visible from normal viewing position.
- 54 Н. Audible Alarm-Indicating Devices: Install not less than 6 inch below ceiling. Install bells and horns on flush-mounted back boxes with device-operating mechanism concealed behind grille. Install devices at same height unless otherwise 55 56 indicated.
- 57 I. Visible Alarm-Indicating Devices: Install adjacent to each alarm bell or alarm horn and at least 6 inch below ceiling. 58 Install devices at same height unless otherwise indicated.

1

2 3

4

5

6

7

8

9

20

21

23

24

25

26

27

28

29

30

31

32

33

34

35

36

37

38 39

40

41

42

43

44

45

48

49

50

51
1	J.	Device Location-Indicating Lights: Locate in public space near device they monitor.						
2	3.4	FIECTRICAL CONNECTIONS						
4	Δ	Connect wiring in accordance with Section 26.05.19 "Low-Voltage Electrical Power Conductors and Cables."						
5	В.	Ground equipment in accordance with Section 26.05.26 "Grounding and Bonding for Electrical Systems."						
6	C.	Install electrical devices furnished by manufacturer, but not factory mounted, in accordance with NFPA 70 and						
7	С.	NECA 1.						
8	D.	Install nameplate for each electrical connection, indicating electrical equipment designation and circuit number						
10		The second connection.						
11		1. Namepiate must be lammated acrylic of melamme plastic signs with black background and engraved white						
11 12		letters at least 1/2 inch fligh.						
12	2 5							
17	5.5	Linstell control and electrical newer wiring to field mounted control devices						
14 15	А. В	Connect control wiring in accordance with Section 26 05 22 "Control Voltage Electrical Power Cables "						
10	ь. С	Connect control winning in accordance with Section 20 05 25 Control-voltage Electrical Power Cables.						
10	C.	install namepiate for each control connection, indicating field control panel designation and i/O control designation						
10		reeding connection.						
10	20	DATINALANC						
19	3.0	PAIHWAYS Any and all fire alarm cabling chall be routed in EMT conduit at a minimum. No free air cabling is allowed. Bathways						
20	А.	Any and all fire alarm cabling shall be routed in EMT conduit at a minimum. No free air cabling is allowed. Pathways						
21		above recessed ceilings and in inaccessible locations may be routed exposed.						
22		1. Exposed pathways located less than 96 inch above floor must be installed in EWT.						
23	В.	Exposed EMT must be painted red enamel.						
24	27	CONNECTIONS						
25	5.7	CONNECTIONS Make addressele as a set and with our any load interface doubles to the fallowing doubles and surtains. Install interface						
20	А.	Make addressable connections with supervised interface device to the following devices and systems. Install interface						
27		device less than 36 inch from device controlled. Make addressable confirmation connection when such reedback is						
28		available at device or system being controlled.						
29		1. Alarm-initiating connection to smoke-control system (smoke management) at hrenghters smoke-control						
30		system panel.						
31		2. Alarm-Initiating connection to stairwell and elevator-snart pressurization systems.						
32		3. Smoke dampers in air ducts of designated HVAC duct systems.						
33		4. Magnetically held-open doors.						
34 25		5. Electronically locked doors and access gates.						
35		 Supervisory connections at valve supervisory switches. Supervisory connections at law size supervisory switches. 						
36		7. Supervisory connections at low-air-pressure switch of each dry-pipe sprinkier system.						
3/		8. Supervisory connections at elevator snunt-trip breaker.						
38		9. Data communication circuits for connection to building management system.						
39 40		10. Supervisory connections at fire-extinguisher locations.						
40 //1	38	ΙΔΕΝΤΙΕΙΛΑΤΙΩΝ						
41 42	Δ	Identify system components wiring cabling and terminals. Comply with requirements for identification specified in						
42 //2	74.	Section 27.05.53 "Identification for Communications Systems."						
ч.5 ЛЛ	в	Install framed instructions in location visible from FACP						
44 15	Б.							
45 46	20	GROUNDING						
40 17	3.9 ^	Ground EACE and associated circuits in accordance with Section 26.05.26 "Grounding and Bonding for Electrical						
47 10	А.	Suctome "						
40 10	D	Systems. Ground chielded cables at control namel location only. Insulate chield at device location						
49 50	Б.	Ground shielded cables at control panel location only. Insulate shield at device location.						
50	3 10							
52	5.10	Field tests must be witnessed by authorities baying jurisdiction						
52	д. В	Administrant for Tests and Inspections:						
55	D.	Administrative resis and inspections.						
55	ſ	Tests and Inspections:						
56	U.	1 Visual Inspection: Conduct visual inspection prior to testing						
50		The state inspection, conduct visual inspection prior to testing.						

I

5

6

7

8

9

10

11

14 15

20

24

30

31

32

33

34

35

36

37

a.	Inspection must be based on completed record Drawings and system documentation that is required
	by "Completion Documents, Preparation" table in "Documentation" section of "Fundamentals"
	chapter in NFPA 72.

- Comply with "Visual Inspection Frequencies" table in "Inspection" section of "Inspection, Testing and Maintenance" chapter in NFPA 72; retain "Initial/Reacceptance" column and list only installed components.
- System Testing: Comply with "Test Methods" table in "Testing" section of "Inspection, Testing and Maintenance" chapter in NFPA 72.
- Factory-authorized service representative must prepare "Fire Alarm System Record of Completion" in "Documentation" section of "Fundamentals" chapter in NFPA 72 and "Inspection and Testing Form" in "Records" section of "Inspection, Testing and Maintenance" chapter in NFPA 72.
- 12D.Reacceptance Testing: Perform reacceptance testing to verify proper operation of added or replaced devices and13appliances.
 - E. Fire-alarm system will be considered defective if it does not pass tests and inspections.
 - F. Prepare test and inspection reports.
- 16G.Maintenance Test and Inspection: Perform tests and inspections listed for weekly, monthly, quarterly, and17semiannual periods. Use forms developed for initial tests and inspections.
- 18H.Annual Test and Inspection: One year after date of Substantial Completion, test fire-alarm system complying with19visual and testing inspection requirements in NFPA 72. Use forms developed for initial tests and inspections.

21 3.11 DEMONSTRATION

A. Train Owner's maintenance personnel to adjust, operate, and maintain fire-alarm system. Provide video recording
 of training to Owner.

25 **3.12 MAINTENANCE**

- A. Maintenance Service: Beginning at Substantial Completion, maintenance service must include 12 months' full
 maintenance by skilled employees of manufacturer's designated service organization. Include preventive
 maintenance, repair or replacement of worn or defective components, lubrication, cleaning, and adjusting as required
 for proper operation. Parts and supplies must be manufacturer's authorized replacement parts and supplies.
 - 1. Include visual inspections in accordance with "Visual Inspection Frequencies" table in "Testing" paragraph of "Inspection, Testing and Maintenance" chapter in NFPA 72.
 - 2. Perform tests in "Test Methods" table in "Testing" paragraph of "Inspection, Testing and Maintenance" chapter in NFPA 72.
 - 3. Perform tests per "Testing Frequencies" table in "Testing" paragraph of "Inspection, Testing and Maintenance" chapter in NFPA 72.

END OF SECTION 28 46 21

1				SECTION 31 05 00				
2		COMMON WORK RESULTS FOR EARTHWORK						
3 4	PART	1 - GEN	ERAL					
5			r.					
6 7 8 9	1.1	A.	This section prov that are of a gen applicable. Inclu	ides information common to two or more technical site work specification sections or items eral nature, and not included in other sections. This section applies to ALL site work, as ded are the following topics:				
10 11			PART 1 - GEI Scope	NERAL				
12			Related	Work				
13			Referen	ced Organizations				
14			Referen	ced Documents				
15			Quality	Assurance				
16			Safety					
17			Permits					
18			Constru	ction Limits				
19			Submitt	als				
20			Off-Site	Storage				
21			Codes	Manage and Taxana Manage				
22			Certifica	tions and inspections				
23			PART Z - IVIA	ALERIALS				
24 25			Barricad	es, signs, and warning Devices				
25								
20			Mainter	ance of Site Access/Europs				
27			Continui	ty of Evicting Traffic/Parking and Traffic Control				
20			Protecti	on and Continuity of Existing Hitilities				
30			Protecti	on of Existing Work and Facilities				
31			Stormw	ater/Excavation Water Management				
32			0101111					
33	1.2	RELA	TED WORK					
34		Α.	Applicable provis	ions of Division 1 govern work under this Section. Related sections include:				
35			1. Section 3	1 20 00 - Earthmoving				
36			2. Section 3	1 22 16.15 - Subgrade Preparation				
37			3. Section 3	2 91 19 - Topsoil-Select Fill Materials and Application				
38								
39	1.3	REFEF	RENCED ORGANIZA	ITIONS				
40		Α.	Applicable provis	ions of Division 1 shall govern all work under this section.				
41		В.	Abbreviations of	organizations referenced in these specifications are as follows:				
42			AASHTO	American Association of State Highway and Transportation Officials				
43			ACPA	American Concrete Pipe Association				
44			ANSI	American National Standards Institute				
45			ASCE	American Society of Civil Engineers				
46			ASME	American Society of Mechanical Engineers				
47			ASTM	American Society for Testing and Materials				
48			AWWA	American Water Works Association				
49			AWS	American Welding Society				
50			FHA	Federal Highway Administration				
51			EPA	Environmental Protection Agency				
52			NEC	National Electric Code				
53			NEMA	National Electrical Manufacturers Association				
54			NEPA	National Fire Protection Association				
55			NSF	National Sanitation Foundation				
50			USHA	Occupational Safety and Health Administration				
5/			511	Steel lank Institute				
58			UL	Underwriters Laboratories Inc.				

1 2			WDNR WISDOT	State of Wisconsin Department of Natural Resources State of Wisconsin Department of Transportation				
3								
4	1.4	REFERENCED DOCUMENTS						
5		Α.	Where reference is	made to the "Construction Standards," it shall be construed to mean the City of Madison's				
6			Construction Standa	ards, as they may pertain, except the method of measurement and basis of payment shall not				
7		_	apply.					
8		В.	Where reference is	made to the "Standard Specifications," it shall be construed to mean the pertinent section of				
9			the Standard Spec	incations for Sewer and Water Construction in Wisconsin, "current edition, and all				
10			supplemental and li	nterim supplemental specifications, as they may pertain, except the method of measurement				
11		C	Whore reference is	in Sildii not appiy.				
12		C.	"Standard Specifica	tions for Highway and Structure Construction " current edition and all supplemental and				
1/			interim supplement	al specifications as they may pertain except the method of measurement and basis of				
15			navment shall not a	nnly				
16		D	Where reference is	made to the "Geotechnical Report" it shall be construed to mean the geotechnical report				
17		υ.	provided in Section					
18			provided in Section					
19	1.5	QUALI	TY ASSURANCE					
20	-	A.	Provide materials a	nd products as required by individual specification sections. Refer to General Conditions of				
21			the Contract regard	ing substitutions.				
22		В.	Provide quality assu	irance testing and reporting as required by individual specification sections.				
23			. ,					
24	1.6	SAFET	Y					
25		Α.	Contractor is solely	responsible for worksite safety.				
26		В.	Perform all work in	accordance with applicable OSHA, state, and local safety standards.				
27		C.	Contact Diggers Ho	tline at 1-800-242-8511 in accordance with statutory requirements. Request that non-				
28			member utilities an	d private utilities be located by the appropriate parties.				
29								
30	1.7	PERM	ITS					
31		Α.	Unless otherwise no	oted in the Contract Documents, Owner shall be responsible for obtaining and paying for all				
32			permits necessary t	o complete the work.				
33								
34	1.8	CONST	RUCTION LIMITS					
35		Α.	Construction Limits	are defined by lines denoted as Construction Fencing/Limits of Disturbance as indicated on				
36			the Drawings. In th	e absence of such a designation on the Drawings, confine work to the minimum area				
37			reasonably necessa	ry to undertake the work as determined by the City's Construction Representative. In no				
38		-	case shall construct	ion activities extend beyond property lines or construction easements.				
39		в.	The Contractor shall	I restore all disturbed areas in accordance with the Drawings and Specifications. If the				
40			Drawings and Speci	incations do not address restoration of specific areas, these areas will be restored to pre-				
41		C	Construction condit	ions as approved by the City's Construction Representative.				
42		C.	coordinate work ur	werk on the site related to other contracts				
43			providing separate	work on the site related to other contracts.				
44 15	10	CLIRM	ΙΤΤΛΙ ς					
45	1.5		Referalso to the Ge	neral Conditions of the Contract section and Division 1				
40		R.	Submit manufactur	er's shop drawings product data samples substitutions and Operation and Maintenance				
48		D.	(0&M) data for ann	royal as required by individual specification sections				
49		C.	Submittals shall be	provided to the Owner's Construction Representative for review and approval unless				
50		с.	otherwise directed	Submittals shall be sent electronically by email in * odf format unless otherwise directed				
51			ethermise uncettu.	seasure and a server electronically by entail in the normal and so other wise an electronic				
52	1.10	OFF-SI	TE STORAGE					
53		A.	Refer to Division 1					
54		В.	In general, the payr	nents for materials stored off-site will only be considered in instances where there is limited				
55			space available for	storage on the site. Prior approval by the Owner's Construction Representative. together				
56			with the execution	of a Storage Agreement will be required.				

1	1.11	1 CODES							
2		Α.	Comply with the requirements of all applicable local, state, and federal codes.						
3									
4	1.12	CERTI	FICATIONS AND INSPECTIONS						
5		A. Refer to the General Conditions of the Contract.							
6		В.	Obtain and pay for all required sampling, testing, inspections, and certifications except those expressly listed as						
7			provided by the Owner or City in the Contract Documents. Deliver originals of certificates and documents to the						
8			City's Construction Representative within three (3) days; provide copies to the Owner's Construction						
9			Representative. Include copies of the certifications and documents in the O&M Manual.						
10 11 12	<u>PART</u>	2 - MA	TERIALS						
12 13	21	BARR	ICADES SIGNS AND WARNING DEVICES						
14	2.1	Δ	Traffic barricades traffic signs, and warning devices shall meet the requirements of current applicable OSHA						
15 16		,	standards and MUTCD.						
 17	2.2	ТЕМР	ORARY PLASTIC BARRIER FENCING						
18		Α.	UV stabilized high-density polyethylene barrier fence free of holes tears and other defects. Provide 5-foot tall						
19 20		R	fence in diamond or rectangular pattern. Fencing shall be a "safety orange" color, unless otherwise noted.						
21		Б.	rosts for temporary plastic barrier tenenis shan be s feet tail, miniman 12 gauge, paintea metai posts.						
22	PART	3 - EXE	CUTION						
23									
24	3.1	MAIN	ITENANCE OF SITE ACCESS/EGRESS						
25		Α.	Unless otherwise shown or directed, maintain existing access and egress to the facility throughout construction.						
26			Contact the City and Owner prior to any construction activities to obtain directives for preferred access to the						
27			site. Maintain ANSI A117 compliant access to the site for disabled persons, delivery access, emergency vehicle						
28			access, and emergency egress. Do not interrupt access and egress without prior written approval from the						
29			Owner's Construction Representative.						
30 21	2 2	CONT	IN UTV OF EVISTING TRAFFIC /DARKING AND TRAFFIC CONTROL						
3.J 2.T	5.2		Refer also to Section 02 20.00 - General Sitework Requirements						
32		R.	Do not interrupt or change existing traffic delivery or parking without prior written approval from the City or						
34		р.	Owner's Construction Representative When interruption is required coordinate schedule with the City and						
35			Owner's Construction Representative to minimize disruptions. When working in public right-of-way, obtain all						
36			necessary approvals and permits from the City if not provided by the Owner.						
37		C.	When Contractor's activities impede or obstruct traffic flow, Contractor shall provide traffic control devices,						
38			signs, and flaggers in accordance with other Contract Documents and current applicable OSHA standards and						
39			MUTCD. Contractor shall be responsible for all costs associated with temporary traffic control. All barricades,						
40			signs, and warning devices shall be included under the traffic control bid item.						
41		D.	Access to the future high school construction site must be maintained at all times. Two-way traffic must be						
42			maintained on all existing roadways at all times. Flagging will be permitted, as necessary, for work during						
43			daylight hours if approved by the City's Construction Representative.						
44									
45	3.3	PROT	ECTION AND CONTINUITY OF EXISTING UTILITIES						
46		А.	Verify the locations of any water, drainage, gas, sewer, electric, drainage, gas, sewer, electric,						
47			telephone/communication, fuel, steam lines or other utilities and site features which may be encountered in any						
48			excavations or other sitework. All lines shall be properly underpinned and supported to avoid disruption of						
49 50			service.						
50 F 1		в.	Do not interrupt or change existing utilities without prior written approval from the City's Construction						
51 51			representative, anected utilities, and users. Notify an users impacted by outages a minimum of 48 nours in advance of outage. Notification shall be provided in writing and describe the nature and duration of outages.						
52			advance of outage. Notification shall be provided in writing and describe the nature and duration of outages						
55		C	Any service connections encountered which are to be removed shall be cut off at the limits of the excavation and						
55 55		С.	capped in accordance with the requirements of annlicable codes and any specifications governing such						
56			removals.						

1	3.4	PROTE	ECTION OF EXISTING WORK AND FACILITIES
2		Α.	Verify the locations of, and protect, any signs, paved surfaces, buildings, structures, landscaping, streetlights,
3			utilities, and all other such facilities that may be encountered or interfered with during the progress of the work.
4			Take measures necessary to safeguard all existing work and facilities that are outside the limits of the work or
5			items that are within the construction limits but are intended to remain. Report any damage to existing facilities
6			to the City's Construction Representative immediately. Correct and pay for all damages.
7			
8	3.5	STORM	MWATER/EXCAVATION WATER MANAGEMENT
9		Α.	Control grading around structures, pitch ground to prevent water running into excavated areas.
10		В.	Pits, trenches within building line, and other excavations shall be maintained and free of water.
11		C.	Provide trenching, pumping, other facilities required.
12		D.	Notify Owner's Construction Representative in the event that springs or running water are encountered in
13			excavation; provide discharge by trenches, drains, pumping to point outside of excavation. Provide information
14			to Owner's Construction Representative of points and areas that water will be discharged. At the Owner and
15			City's Construction Representative's option, the Contractor shall drain the spring to the storm sewer system by
16			the use of field tile.
17		Ε.	Establish and maintain an on-site Erosion Control Maintenance Log. The log shall document erosion control
18			installation locations and date of establishment, rainfall event dates and amounts, erosion control failure
19			locations, corrective measures taken, and weekly inspection documentation. This log shall be available on-site
20			during the entire construction process and available to the Owner, Owner's Construction Representative,
21			Governing Municipality, and authorized WDNR staff.
22		F.	Be responsible for control measures to prevent damage from flooding, erosion, and sedimentation to on-site
23			and off-site areas.
24			
25			END OF SECTION

	SECTION 31 10 00					
	SITE CLEARING					
PART	1 - GEI	NERAL				
1.1	RELA	TED SEC	TIONS			
	A.	Sectio	n 02 20 00 - General Sitework Requirements			
	в. С.	Sectio	n 02 32 00 - Geotechnical Investigation			
	0.	00000				
1.2	REFE	RENCE S	TANDARDS			
	Α.	Where	e reference is made to the "Construction Standards," it shall be construed to mean the City of Madison's			
		"Stand	dard Specifications for Public Works Improvements."			
	в.	wnere	e reference is made to the "standard Specifications," It shall be construed to mean the pertinent section of			
		the S	tanuard specifications for sever and water construction in wisconsin, current edition, and an			
		supple	anental and interim supplemental specifications, as they may pertain, except the method of measurement			
	C	M/borg	asis of payment shall not apply.			
	С.	"Stope	dard Specifications for Highway and Structure Construction " current edition, and all supplemental and			
		interir	and specifications for highway and structure construction, current earlier, and an supperinental and as so of			
		navme	ent shall not annly			
	D	Where	e reference is made to the "Geotechnical Report" it shall be construed to mean the Geotechnical Report			
	υ.	provid	Jed in Section 02 32 00 - Geotechnical Investigation.			
		p				
1.3	DEFI	NITIONS				
	Α.	Clearii	ng:			
		1.	Clearing shall consist of the felling, trimming, and cutting of trees into sections and the satisfactory			
			disposal of the trees and other vegetation designated for removal, including down timber, snags, brush,			
			and rubbish occurring in the areas to be cleared.			
	В.	Grubb	ing:			
		1.	Grubbing shall consist of the removal and disposal of stumps, roots larger than 3 inches in diameter, and			
			matted roots from the designated grubbing areas.			
14	SUB	λιτταις				
1.4	Δ	The fo	llowing shall be submitted.			
	7	1.	Written permission to dispose of such products on private property shall be filed with the City's			
			Construction Representative.			
		2.	Submit documentation from the disposal facility to verify that it is licensed to accept the material.			
PART	2 - PRC	<u>DDUCTS</u>				
Not L	Jsed.					
PART	3 - EXE		-			
31		RING				
5.1	A.	Trees	stumps, roots, brush, and other vegetation in areas to be cleared shall be cut off flush with or below the			
		origina	al ground surface, except such trees and vegetation as may be indicated or directed to be left standing.			
	В.	Trees	designated to be left standing within the cleared areas shall be trimmed of dead branches 1½ inches or			
		more	in diameter and shall be trimmed of all branches below the heights indicated or directed.			
		1.	Limbs and branches to be trimmed shall be neatly cut close to the bole of the tree or main branches.			
		2.	Cuts more than 1½ inches in diameter shall be painted with an approved tree-wound paint.			
		3.	Trees and vegetation to be left standing shall be protected from damage incidental to clearing, grubbing,			
			and construction operations by the erection of barriers or by such other means as the circumstances			
			require.			
	C.	Clearii	ng shall also include the removal and disposal of structures that obtrude, encroach upon, or otherwise			
		obstru	uct the work.			
	D.	Dispos	sal of Elm and Ash trees, roots, or branches shall be in accordance with local and state regulations.			

1	3.2	GRUB	BING
2		Α.	Remove material to be grubbed, together with logs and other organic or metallic debris not suitable for roadway
3			construction in accordance with Section 201.3 of the State Specifications, except the minimum depths for
4			removal shall be as follows:
5			1. In cut areas, 18 inches below final subgrade.
6			2. In embankments areas, 18 inches below the existing grade.
7		в.	Depressions made by grubbing shall be filled with suitable material and compacted to make the surface conform
8			to the proposed surface of the ground.
9		C.	Burning or burying as a means of disposal is prohibited.
10			
11	3.3	TREE	REMOVAL
12		Α.	Where indicated or directed, individual trees and stumps that are designated shall be removed from areas
13			outside those areas designated for clearing and grubbing.
14 15		В.	This work shall include the felling of such trees and the removal of their stumps and roots as specified in paragraph GRUBBING.
16		C.	Dispose of materials as specified in paragraph CLEAN UP.
17			
18	3.4	TOPS	OIL
19		Α.	Topsoil: Topsoil is defined as friable clay loam surface soil found in a depth of not less than 4 inches.
20			Satisfactory topsoil is reasonably free of subsoil, clay lumps, stones, and other objects over 2 inches in diameter,
21			and without weeds, roots, and other objectionable material. Refer to Section 32 91 19 - Topsoil-Select Fill
22			Materials and Application for further information.
23			1. Strip topsoil to whatever depths encountered in a manner to prevent intermingling with underlying
24			subsoil or other objectionable material. Remove heavy growths of grass from areas before stripping.
25			2. Stockpile topsoil in storage piles in areas indicated or directed. Construct storage piles to provide free
26			drainage of surface water.
27	2 5		
28	3.5		AN UP
29		А.	Logs, stumps, roots, brush, rotten wood, and other refuse from the clearing and grubbing operations, shall be
21			in writing
22		D	III writing All waste and debris shall be dispessed of in compliance with state and local regulations within five (5) days of
22		ь.	All waste and debits shall be disposed of in compliance with state and local regulations within five (5) days of being cut or removed
21		C	Disposal of Film trees shall be in accordance with local regulations
25		с. D	Submit the location of any disposal facility located outside the limits of the project to the City's Construction
36		D.	Representative prior to removal from the project site
37		F	Submit documentation from the disposal facility to verify that it is licensed to accent the material
38		F.	No material shall be removed from the project site without prior approval from the City's Construction
39		••	Representative.
40		G.	Burning or burving as a means of disposal is prohibited.
41			
42			END OF SECTION

1 2					SECTION 31 20 00 EARTHMOVING			
3								
4	PART	1 - GEN	ERAL					
5	11	SUM						
7	1.1	A.	This S	ection includes the follow	ving:			
8			1.	Stripping of topsoil and	stockpiling.			
9			2.	Excavation, preparation	n, backfilling, and compaction of subgrades per the Geotechnical Report.			
10			3.	Cutting, filling, grading,	and compaction for drives, walks, roads, and parking subgrade.			
11			4.	Cutting, filling, grading,	and compaction for landscaping area subgrade ready for topsoil.			
12								
13	1.2	RELAT	ED SEC	TIONS include the follow	/ing:			
14		Α.	Sectio	on 02 20 00 - General Site	work Requirements			
15		В.	Sectio	on 31 10 00 - Site Clearing				
16		C.	Sectio	on 31 25 00 - Erosion Con	trol			
1/	1 2	CTAN		DECIFICATIONS				
10	1.3			PECIFICATIONS	a "Construction Standards" it shall be construed to mean the City of Madicon's			
20		А.	Const	ruction Standards as the	w may pertain except the method of measurement and basis of navment shall not			
20			annly	in action Standards, as the	y may pertain, except the method of measurement and basis of payment shall not			
22		В.	Wher	e reference is made to th	e "Standard Specifications." it shall be construed to mean the pertinent section of			
23			the St	tandard Specifications for	Sewer and Water Construction in Wisconsin, Current Edition, and all			
24			suppl	emental and interim supp	plemental specifications, as they may pertain, except the method of measurement			
25			and b	asis of payment shall not	apply.			
26		C.	Wher	e reference is made to th	e "State Specifications," it shall be construed to mean the pertinent section of the			
27			Stand	lard Specifications for Hig	hway and Structure Construction, Current Edition, and all supplemental and			
28			interi	m supplemental specifica	tions, as they may pertain, except the method of measurement and basis of			
29		_	paym	ent shall not apply.				
30		D.	Wher	re reference is made to th	e "Geotechnical Report," it shall be construed to mean the geotechnical report in			
31 22		F	Sectio	on UZ 32 UU. Astruction of public facilit	tios and/or work within public lands or rights of way shall conform to the			
32 33		с.	requi	rements and conditions of	f the Standard Specifications stated above with the most stringent applying			
34		F	The n	ublications listed below f	orm a part of this specification to the extent referenced Publications are			
35		••	refere	enced within the text by t	he basic designation only.			
36			1.	ASTM International (AS	TM):			
37				ASTM D422	Particle Size Analysis of Soil			
38				ASTM D698	Laboratory Compaction Characteristics of Soil Using Standard Effort (12,400			
39					ft/lbf/ft3 (600 kN.m/m3))			
40				ASTM D1557	Laboratory Compaction Characteristics of Soil Using Modified Effort (56,000 ft-			
41					lbf/ft3 (2,700 Kn.m/m3))			
42				ASTM D2487	Classification of Soils for Engineering Purposes (Unified Soil Classification			
43					System)			
44				ASTM D2488	Description and Identification of Soils (Visual-Manual Procedures)			
45				ASTM D4318	Liquid Limit, Plastic Limit, and Plasticity Index of Solis			
40				ASTIM D0938	Methods (Shallow Donth)			
47			2	American Association o	f State Highway and Transportation Officials (AASHTO):			
49			2.	AASHTO T 88	Particle Size Analysis of Soils			
50			3.	National Fire Protection	Association (NFPA):			
51				NFPA 70 Nation	nal Electrical Code			
52			4.	American Water Works	Association (AWWA):			
53				AWWA C200	Standard for Steel Water Pipe - 6 In. (150 mm) and Larger			
54				AWWA C206	Field Welding of Steel Water Pipe			
55								
56	1.4	SUBN	IITTALS					
57		Α.	Subm	it documentation of mate	erials meeting the required specifications.			

1		В.	Testing:							
2			1. The City shall provide testing for work performed for public improvements.							
3			2. The Contractor shall provide quality control testing as defined in the Contract Documents.							
4			3. The Contractor shall coordinate work and testing requirements with the Owner's Construction							
5			Representative and City's testing agency.							
6										
7	1.5	DEFIN	ITIONS							
8		Α.	Backfill: Soil materials used to fill an excavation.							
9		В.	Base Course: Course placed between the sub-grade and the hot-mix asphalt, concrete pavement, walks, or							
10			curbs.							
11		C.	Breaker Run Stone: Meet the requirements defined in Wisconsin Department of Transportation (WisDOT)							
12			Section 311.							
13		D.	Borrow Soil: Satisfactory soil imported from off-site for use as fill or backfill.							
14		Ε.	Excavation: Removal of material encountered above the subgrade elevations and to lines and dimensions							
15			indicated.							
16			1. Authorized Additional Excavation: Excavation below subgrade elevations or beyond indicated lines and							
17			dimensions as directed by Owner's Construction Representative. Authorized additional excavation and							
18			replacement material will be paid for according to Contract provisions.							
19			2. Unauthorized Excavation: Excavation below subgrade elevations or beyond indicated lines and							
20			dimensions without direction of Owner's Construction Representative. Unauthorized excavation, as well							
21			as remedial work directed by the Owner's Construction Representative, shall be without additional							
22			compensation.							
23		F.	Fill: Soil materials used to raise existing grades.							
24		G.	Subgrade: Surface or elevation remaining after completing excavation, or top surface of a fill or backfill							
25			immediately below subbase, drainage fill, or topsoil materials.							
26		Н.	Topsoil: Excavated on-site material, free of large tree roots, rocks, subsoil, debris, and weeds.							
27										
28	1.6	CONTO	OURS (GRADE ELEVATIONS)							
29		Α.	Contours indicated on the Drawings are the finished grade elevations. Contractor shall review all grade							
30			elevations before commencing to ensure that proper slopes for drainage, slope for drives, walks, paving, etc.,							
31			are maintained. If Contractor believes a deficiency is apparent, they shall notify the City or Owner's Construction							
32			Representative for clarification.							
33										
34	PART	2 - PROI	DUCTS							
35										
36	2.1	SOIL N	NATERIALS							
37		А.	General: All materials shall conform to requirements of the Geotechnical Report.							
38		В.	Materials:							
39			1. Fill and Backfill. Satisfactory materials excavated from the site.							
40			2. Imported Fill Material: Satisfactory material provided from off-site borrow areas when sufficient							
41			satisfactory materials are not available from required excavations.							
42			3. Irench Backfill: ASIM D2321 and the Standard Specifications, unless otherwise specified or shown on							
43			the Drawings.							
44			4. Subgrade Sub-base Material: As required by the Geotechnical Report and/or Section 31 22 16.15.							
45			5. Building Sub-base Material: Sub-base for building and appurtenances slabs on ground is specified in							
46			Section 03 30 00 or the Geotechnical Report, as applicable.							
47			6. Bedding: Aggregate type as indicated on the plans or naturally or artificially graded mixture of natural or							
48			crushed gravel, crushed stone, and natural or crushed sand; ASTM D2940; except with 100% passing a							
49			1-inch sieve and not more than 8% passing a No. 200 sieve.							
50			7. Drainage Fill: Washed, narrowly graded mixture of crushed stone, or crushed or uncrushed gravel; ASTM							
51			D448; coarse-aggregate grading Size 57; with 100% passing a 1½-inch sieve and 0% to 5% passing a No. 8							
52			SIEVE.							
53			 Filter initiaterial: Inarrowly graded mixture of natural or crushed gravel, or crushed stone and natural sand; ASTAD 2449, second approache and disc Size C7, with 4000 percises 4 line bridge and 000 to 500 percent. 							
54			AS I WE 448; coarse-aggregate grading Size 67; with 100% passing a 1-inch sieve and 0% to 5% passing a							
55			NO. 4 SIEVE.							
50			9. Topsoil: Topsoil shall consist of stripping material excavated from the site. Topsoil shall consist of							
5/			organic sufficial soil found in depth of not more than 6 inches. Topsoil shall be as further defined in							
~×			Section 32 91 19 - Topsoli-Select Fill Materials and Application.							

1 2		C.	Where conflicts between this specification, the Drawings, and the Geotechnical Report exist, requirements of the Geotechnical Report shall govern.
3		D.	Source Quality Control:
4			1. Laboratory testing of off-site materials proposed for use in the project shall be provided by the
5			Contractor's testing consultant. Test results shall be provided to the Owner's Construction
6			Representative for approval before incorporation into the work.
7			2 The following tests shall be performed on each type of imported soil material used as compacted fill:
, 8			 Molecular and Density Relationship: A STM D698 or ASTM D1557
0			a. Morehanical Analysis: AASENTO TS9 or ASTM D432
9 10			D. Michailta Alaysis. Astrio 188 01 Astrio 0422
10			c. Plasticity index: ASTW D4318
11			
12	PART :	3 - EXEC	
13			
14	3.1	GENER	RAL
15		Α.	General: All work shall be executed and conform to requirements of the Geotechnical Report.
16			1. Where conflicts between this specification and the Geotechnical Report exist, requirements of the
17			Geotechnical Report shall govern.
18			2. For any material provided by the Owner, the Contractor shall provide a minimum of five days' notice for
19			the material and shall include the quantity of material and delivery location requested for each day.
20			Delivered material shall be available Monday-Friday 7:00 a m to 3:30 n m unless otherwise agreed upon
21			by both the Owner and Contractor "
21			by both the owner and contractor.
22	• •		RATION.
23	3.2	PREPA	ika ilon Gubera das fillenstanisti and ens dies for assument remas side velle, and structures shall conform to the
24		А.	subgrades, hill material, and grading for pavement, ramps, sidewalks, and structures shall conform to the
25		_	recommendations in the Geotechnical Report.
26		В.	Protect structures, utilities, sidewalks, pavements, and other facilities from damage caused by settlement, lateral
27			movement, undermining, washout, and other hazards created by earthwork operations.
28		C.	Protect subgrades and foundation soils against freezing temperatures or frost. Provide protective insulating
29			materials as necessary.
30		D.	Provide erosion control measures to prevent erosion or displacement of soils and discharge of soil-bearing water
31			runoff or airborne dust to adjacent properties and walkways.
32			
33	3.3	CLEAR	ING AND GRUBBING
34		A.	Limits of clearing and grubbing shall be areas which are affected by excavation and grading.
35		B	Refer to Section 31 10 00 - Site Clearing
36		C.	Remove trees stumps roots brush other vegetation debris existing foundations navements fences and
27		с.	other three which interfore with new construction
20		D	Unier nems which meriere with new construction.
30		D.	Remove sumps, logs, roots, and other organic material including existing structure occurring outside the
39			structure excavation to depths below the following:
40			1. Walks: 18 inches
41			2. Roads and Drives: 18 inches
42			3. Parking Areas: 36 inches
43			4. Lawn Areas: 12 inches
44			5. Concrete Pads: 24 inches
45			6. Depressions within areas shall be filled and compacted as specified under Controlled Backfill.
46		Ε.	Removal of existing trees which are to remain will not be permitted. Notify Owner's Construction
47			Representative if existing trees create a difficulty when grades are raised or lowered in excess of 6 inches.
48			
49	3.4	DEWA	TERING
50	- 1	Α.	Prevent surface water and ground water from entering excavations, from ponding on prepared subgrades, and
51			from flooding Project site and surrounding area
52		в	All dewatering activities must meet all the requirements set forth in the WIDNR Construction Site Erosion &
52		ы.	Sodiment Control Technical Standard 1061. The Contractor shall obtain any necessary nermits for devictoring
22		c	Second of recipition recipition statutary 1001. The contractor shall obtain any necessary permits for dewatering.
54 FF		ι.	Protect subgrades from softening, undermining, washout, and damage by rain or water accumulation:
55			1. Reroute surface water runoff away from excavated areas.
56			a. Do not allow water to accumulate in excavations.
57			b. Do not use excavated trenches as temporary drainage ditches.

			2	to define the standard barrier to be an extension of a standard standard barrier to the standard barrier to the				
1			2.	Install a dewatering system to keep subgrades dry and convey groundwater away from excavations.				
2			Ducuial	Manitain until dewatering is no longer required.				
3		D.	Provid	e dewatering systems as required for excavations:				
4			1.	Design and provide dewatering system using accepted and professional methods consistent with current				
5				ridustry practice to enminate water entering the excavation under hydrostatic nead from the bottom of				
6				sides. Design system to prevent differential hydrostatic head, which would result in floating out soil				
/				particles in a manner, termed as a "quick" or "boiling" condition. System shall not be dependent solely				
8				upon sumps or pumping water from within the excavation where differential head would result in a quick				
9			2	condition, which would continue to worsen the integrity of the excavation's stability.				
10			2.	Provide dewatering system of sufficient size and capacity to prevent ground and surface water flow into				
11			2	the excavation and to allow Work to be installed in a dry condition.				
12			3.	Control, by acceptable means, all water regardless of source. The Contractor shall be responsible for				
13				disposal of the water.				
14			4.	Control groundwater in a manner that preserves strength of foundation soils, does not cause instability				
15				or raveling of excavation slopes, and does not result in damage to existing structures. Where necessary,				
16				lower water level in advance of excavation utilizing wells, well points, jet educators, or similar positive				
17				methods. The water level as measured by piezometers shall be maintained a minimum of 3 feet below				
18				prevailing excavation level.				
19			5.	Commence dewatering prior to any appearance of water in excavation and continue until Work is				
20				complete to the extent that no damage results from hydrostatic pressure, flotation, or other causes.				
21			6.	Open pumping with sumps and ditches will be allowed provided it does not result in boils, loss of fines,				
22				softening of the ground, or instability of slopes.				
23			7.	Install wells or well points, if required, with suitable screens and filters so that continuous pumping of				
24				fines does not occur. Arrange discharge to facilitate collection of samples by the Owner or municipal				
25				agencies. During normal pumping and upon development of wells, levels of fine sand or silt in the				
26				discharge water shall meet WDNR discharge standards.				
27			8.	Control grading around excavations to prevent surface water from flowing into excavation areas.				
28			9.	Designate and obtain the services of a qualified dewatering specialist to provide dewatering plan as may				
29				be necessary to complete the Work.				
30			10.	Contractor shall be responsible for the accuracy of the Drawings, design data, and operational records				
31				required.				
32			11.	Contractor shall be responsible for the design, installation, operation, maintenance, and any failure of				
33				any component of the system.				
34		Ε.	Mainta	aining Excavation in Dewatering Condition:				
35			1.	Dewatering shall be a continuous operation. Interruptions due to power outages or any other reason will				
36				not be permitted.				
37			2.	Continuously maintain excavation in a dry condition with positive dewatering methods during				
38				preparation of subgrade, installation of pipe, and construction of structures until the critical period of				
39				construction or backfill is completed to prevent damage of subgrade support, piping, structure, side				
40				slopes, or adjacent facilities from flotation or other hydrostatic pressure imbalance.				
41			3.	Provide standby equipment on site, installed, wired, and available for immediate operation if required to				
42				maintain dewatering on a continuous basis in the event any part of the system becomes inadequate or				
43				fails. If dewatering requirements are not satisfied due to inadequacy or failure of dewatering system,				
44				perform such work as may be required to restore damaged structures and foundation soils at no				
45				additional cost to the Owner.				
46			4.	System maintenance shall include but not be limited to 24-hour supervision by personnel skilled in the				
47				operation, maintenance, and replacement of system components and any other work required to				
48				maintain excavation in dewatered condition.				
49								
50	3.5	EXPLO	SIVES					
51		Α.	Blastin	g of materials classified as rock shall be permitted only when authorized by the Owner's Construction				
52			Repres	sentative and municipality. Contractor shall meet all federal, state, and local requirements				
53		В.	Blastin	g shall be done with explosives of quantity and power, and fired in such sequence and locations as to not				
54			injure	personnel, damage or crack rock against which concrete is to be placed, damage property, or damage				
55			existin	g work or other portions of new work. Contractor shall be responsible for damage caused by blasting				
56			operat	ions.				
57		C.	The Co	ntractor shall submit a Blasting Plan, prepared and sealed by a registered professional engineer that				
58			include	es calculations for overpressure and debris hazard. Blasting mats shall be provided and non-electric				

1 2 3			blasting caps shall be used. The Contractor shall obtain written approval prior to performing any blasting and shall notify the Owner's Construction Representative a minimum of 24 hours prior to blasting. The plan shall contain provisions for storing, handling, and transporting explosives as well as for the blasting operations.
4 5	26	стрірр	
5	3.0		String the site in conformance with the requirements of the Geotechnical Benert referenced in Section 02.22.00
7		д. В	Strip the site in comon soft the site which will accur walks roads drives parking areas and where grade changes
, 8		Б.	are to be made, by a minimum of denth of tonsoil indicated by the soils report plus additional soil as required to
9			reach soil free of roots or organic debris subject to rotting and settling
10		C	Stocknile reusable tonsoil for use in finish grading and restoration. Reusable tonsoil shall be fertile friable
11		с.	agricultural soil capable of sustaining vigorous plant growth and suitable for growth of grass, neither excessively
12			alkaline or acidic, free from subsoil, clay lumps, gravel, brush, objectionable weeds, litter, stones larger than
13			1 inch in diameter, and other material. Do not permit surplus topsoil to leave the project site until the finish
14			grading is nearing completion or unless otherwise approved in writing by the Owner's Construction
15			Representative.
16		D.	Do not excavate, grade, or work topsoil in frozen or muddy conditions.
17			
18	3.7	ROCK	IN EXCAVATIONS
19		Α.	When rock as defined above is encountered before the proper subgrade is reached, work shall proceed as
20			follows:
21			1. The excavation shall stop at this point and it shall be determined if such material is classified as rock.
22			2. Material classified as rock shall be removed to the lines and grades indicated to permit installation of
23			permanent construction without exceeding the following dimensions:
24			a. 24 inches outside of concrete forms.
25			b. 6 inches outside of minimum required dimensions of concrete cast against grade.
26			c. 6 incres beneath the bottom of concrete slabs on grade.
27	38		Υ ΡΟΓΚ ΕΧΓΑΛΑΤΙΟΝS
29	5.0	Δ	When rock as defined above is encountered when excavating for water main or storm sewer, work shall proceed
30		/	as follows:
31			1. The excavation shall stop at this point and it shall be determined if such material is classified as rock.
32			2. Material classified as rock shall be removed to 6 inches below the proposed utility location to permit
33			installation of the utility without exceeding the following dimensions:
34			a. 8 feet in width centered on the proposed utility.
35			b. 6 inches beneath the bottom of the proposed water main or storm sewer.
36			c. Utility rock excavations shall be completed in accordance with Section 608 of the State
37			Specifications.
38			
39	3.9	EXCAV	ATION FOR WALKS AND PAVEMENTS
40		Α.	Excavations shall be in conformance with the requirements of the Geotechnical Report referenced in Section
41		D	U2 32 00.
4Z 12		в.	Excavate surfaces under walks and pavements to indicated cross sections, elevations, and grades.
43	3.10	SUBGR	RADE INSPECTION
45	0.20	A.	Notify Owner's Construction Representative when excavations have reached subgrade.
46		В.	If Owner's Construction Representative determines that unsatisfactory soil is present, continue excavation and
47			replace with compacted backfill or fill material as directed.
48		C.	Proof-roll shall be completed in accordance with Geotechnical Report referenced in Section 02 32 00 and in
49			accordance with Section 31 22 16.15 - Subgrade Preparation.
50		D.	Authorized additional excavation and replacement material will be paid for according to the Contract provisions.
51		Ε.	Reconstruct subgrades damaged by freezing temperatures, frost, rain, accumulated water, or construction
52			activities, as directed by the Owner's Construction Representative, without additional compensation from the
53			Owner.
54			
55	3.11	STORA	IGE OF SOIL MATERIALS
56		A.	stockpile porrow materials and satisfactory excavated soil materials. Stockpile soil materials without
5/ E0			Intermixing. Place, grade, and snape stockpiles to drain surrace water. Cover to prevent windblown dust.
50			rowac necessary crosion control devices as shown on the Lrosion Control Figh.

1			1.	Stockpile soil materials away from edge of excavations. Do not s	tore within drip l	line of remaining trees.
2						
3	3.12	BACKF	ILL			
4 5		Α.	Backfi 02 32	II shall be in conformance with the requirements of the Geotechni 00.	cal Report refere	enced in Section
6		В.	Place	and compact backfill in excavations promptly, but not before com	pleting the follow	ving:
/ 0			1. ว	Removing trash and debris.		
0			Ζ.	Removing deletenous materials.		
9 10	3 13	FILL				
11	5.15	Δ	Prena	ration. Remove vegetation tonsoil debris unsatisfactory soil mat	terials obstructio	ons and deleterious
12		7	mater	ials from ground surface before placing fills.		
13		В.	Plow,	scarify, bench, or breakup sloped surfaces steeper than one vertic	al to four horizor	ntal so fill material will
14			bond	with existing material.		
15		C.	Place	and compact fill material in layers to required elevations per the G	eotechnical Rep	ort.
16		D.	Fill are	eas to contours and elevations shown on the Drawings with mater	ials deemed satis	sfactory.
17		E.	Place	fills in continuous lifts specified herein.		
18 19		F.	fill wi	thin proposed building subgrade, paving subgrade, and outparcel s er than 6 inches in any dimension.	subgrades shall n	iot contain rock or stone
20		G.	Unless	s otherwise specified for rock fill, rock or stone less than 12 inches	in largest dimen	sion may be used in fill
21			below	structures, paving, outparcels, and graded areas, up to 24 inches	below surface of	proposed subgrade of
22			hard s	urface paved areas or 24 inches below finish grade of landscape a	nd turf graded ar	reas when mixed with
23			satisfa	actory material. Rock or stone less than 4 inches in largest dimension	ion may be used	in fill within the upper
24			24 inc	hes of proposed subgrade or finish grade of graded areas when m	ixed with satisfac	ctory material.
25		Н.	Rocks	larger than 12 inches in diameter shall be separated and stockpile	d at an on-site lo	ocation determined by
26			the O	wner's Construction Representative.		
27		Ι.	Fill ma	aterials used in preparation of subgrade shall be placed in lifts or la	ayers not to exce	ed 12 inches.
28	~					
29	3.14		UKE CO	UNIKUL Imily maintan ar parata subgrada and each subsequent fill ar back	ill lover before e	magazian to within 20/
30 21		А.	of ont	imum moisten or aerate subgrade and each subsequent fill or backt	ill layer before co	ompaction to within 2%
27 21			1 01 0PL	Do not place backfill or fill material on surfaces that are muddy f	frozen or contai	n frost or ice
32			1. 2	Remove and replace, or scarify and air-dry, otherwise satisfactor	v soil material th	at exceeds ontimum
34			۷.	moisture content by 2% and is too wet to compact to specified d	ry unit weight	
35				mosture content by 2% and is too wet to compact to specifica a	ry unit weight.	
36	3.15	сомр	ΑΟΤΙΟΙ	N OF BACKFILLS AND FILLS		
37		A.	Place	backfill and fill materials in layers not more than 8 inches (200 mm	ı) in loose depth	for material compacted
38			by hea	avy compaction equipment, and not more than 4 inches (100 mm)	in loose depth fo	or material compacted
39			by har	nd-operated tampers.		
40		В.	Place	backfill and fill materials evenly on all sides of structures to require	ed elevations, an	d uniformly along the
41			full lei	ngth of each structure.		
42		C.	Comp	act as follows:		
43						
44				Percent (%) of Maximum Laboratory Density		
45						
46				<u>ocation</u>	<u>ASTIVI D698</u>	<u>ASTM D1557</u>
47 10			د د	nd upper 12 inches of area to be payed	08	05
40 10			a c	ubgrade and fill in all other areas	98	93
49 50			5		55	52
51		D.	Maint	ain moisture content of not less than 1% helow and not more than	n 2% ahove ontin	num moisture content of
52			fill ma	terials to attain required compaction density.		
53		E.	Exerci	se proper caution when compacting immediately over top of pipe	s or conduits. W	ater jetting or flooding is
54			not pe	ermitted as method of compaction.		, , , , , , , , , , , , , , , , , , , ,
55		F.	Correc	ctive Measures for Non-Complying Compaction: Remove and re-co	ompact deficient	areas until proper
56			compa	action is obtained.	-	
57						
58						

1	3.16	GRAD	VING
2		Α.	General: Uniformly grade areas to a smooth surface, free from irregular surface changes. Comply with
3			compaction requirements and grade to cross sections, lines, and elevations indicated.
4			1. Provide a smooth transition between adjacent existing grades and new grades.
5			2. Cut out soft spots, fill low spots, and trim high spots to comply with required surface tolerances.
6		В.	Site Grading: Slope grades to direct water away from buildings and to prevent ponding. Finish subgrades to
7			required elevations within the following tolerances:
8			1. Lawn or Unpaved Areas: ±1 inch (25 mm).
9			2. Walks: ±1 inch (25 mm).
10			3. Pavements: ±1/2 inch (13 mm).
11			
12	3.17	FINISH	H GRADING
13		Α.	Grade areas where finish grade elevations or contours are indicated on the Drawings, other than paved areas,
14			outparcels, and buildings, including excavated areas, filled and transition areas, and landscaped areas. Graded
15			areas shall be uniform and smooth, free from rock, debris, or irregular surface changes. Ground surfaces shall
16			vary uniformly between indicated elevations. Grade finished ditches to allow for proper drainage without
17			ponding and in manner that will minimize erosion potential. For topsoil, sodding, and seeding requirements
18			refer to Section 32 92 00 and Section 32 92 19.
19		В.	Correct settled and eroded areas within one year after date of completion at no additional expense to the City.
20			Bring grades to proper elevation.
21			
22	3.18	MAIN	ITENANCE OF SUBGRADE
23		Α.	Verify finished subgrades to ensure proper elevation and conditions prior to construction above subgrade.
24		В.	Protect subgrade from excessive wheel loading during construction, including concrete trucks, dump trucks, and
25			other construction equipment. If rutting or damage to the subgrade does occur, regrade and compact to project
26			specified tolerances.
27		C.	Construct temporary ditches and perform such grading as necessary to maintain positive drainage away from
28			subgrade at all times. Contractor shall be responsible for maintaining grades and subgrades throughout
29			construction from frost, moisture and excessive wheel loading. Contractor shall be responsible for choosing
30			means, methods and best management practices to protect the subgrade.
31			
32	3.19	BORR	OW AND SPOIL SITES
33		Α.	Comply with WPDES and local erosion control permitting requirements for any and all on-site and off-site,
34			disturbed spoil and borrow areas. Upon completion of spoil or borrow operations, clean up spoil or borrow
35			areas in a neat and reasonable manner to the satisfaction of the Developer or off-site property owner, if
36			applicable.
37		В.	Topsoil stripping and re-spread will be paid for at the quantity and unit price noted on the Bid Form. Excavation
38			shall be paid as part of the Lump Sum price for Excavation Common as noted on the Bid Form. Seeding and
39			mulching shall also be paid at the unit prices noted in the Bid Form.
40			
41	3.20	PROT	ECTION
42		Α.	Protecting Graded Areas: Protect newly graded areas from traffic, freezing, and erosion. Keep free of trash and
43			debris.
44		В.	Repair and re-establish grades to specified tolerances where completed or partially completed surfaces become
45			eroded, rutted, settled, or where they lose compaction due to subsequent construction operations or weather
46			conditions.
47			1. Scarify or remove and replace soil material to depth as directed by Owner's Construction Representative;
48			reshape and recompact.
49		C.	Where settling occurs before Project correction period elapses, remove finished surfacing, backfill with
50			additional soil material, compact, and reconstruct surfacing.
51			1. Restore appearance, quality, and condition of finished surfacing to match adjacent work, and eliminate
52			evidence of restoration to the greatest extent possible.
53			
54	3.21	DISPC	DSAL OF SURPLUS AND WASTE MATERIALS
55		A.	Disposal: Remove surplus satisfactory soil and waste material, including unsatisfactory soil, trash, and debris,
56			and legally dispose of it off Owner's property.
57			
58			

3

4

5

6 7 8

FIELD QUALITY CONTROL Field quality control shall be the responsibility of the Owner's Construction Representative. Except for specified mandatory testing, field quality control testing and inspection shall be at the discretion of the Contractor as necessary to assure compliance with Contract requirements. The Owner's testing agency will perform retesting and re-inspection as necessary until corrections are fully completed by the Contractor at the Contractor's expense.

END OF SECTION

WARNER PARK COMM AND REC CENTER CONTRACT #9502 MUNIS #17196

1 2 2					SECTION 31 22 16.15 SUBGRADE PREPARATION		
3 4 5	PART	1 - GEN	ERAL				
5 6	1.1	SCOP	E				
7 8 9		A.	The wo necessa as othe	ork under this ary to comple erwise deeme	section shall consist of providing all work, mate te roadway and parking lot grading, as required d necessary to complete the work.	rials, labor, equipment, and su in these specifications, on the	pervision Drawings and
11	1.2	RELAT		RK			
12 13 14 15 16 17 18		A.	Applica 1. 2. 3. 4. 5. 6.	ble provision Section 02 20 Section 02 32 Section 31 20 Section 31 23 Section 31 32	s of Division 1 govern work under this Section. F 00 - General Sitework Requirements 00 - Geotechnical Investigation 00 - Earthmoving 16.13 - Trenching 00 - Erosion Control 00 - Soil Stabilization	Related sections include:	
20 19	13	ΟΠΑΙ	ΙΖΖΑ ΥΤΙ	IRANCE			
21 22 23 24	1.0	<u>д</u> ел.	The Co analysi engine	ntractor shall s as required er shall meet	retain the services of a geotechnical consulting by this section and elsewhere in the Contract Do the requirements of ASTM E329-00b.	engineer to conduct sampling ocuments. The geotechnical co	testing and insulting
			Mater	rial	Test Required	Test/Sample Frequency	
			Granu	ılar Fill	D422-63 (1998) - Standard Test Method for Particle Size Analysis of Soils	1 test/500 cy placed	
			Granı	ılar Fill	ASTM D1557 - Optimum Moisture-Maximum Density Determination (Modified Proctor)	1 test per type of material	
25					Table 31 22 16.15 - 1		
20 27	1.4	REFEF	RENCES				
28		A.	Where	reference is r	nade to the "Construction Standards," it shall be	e construed to mean the City o	f Madison's
29 30 31 32		В.	Constru Where the "St suppler	uction Standa reference is r andard Specif mental and in	rds, except the method of measurement and ba nade to the "Standard Specifications," it shall be ications for Sewer and Water Construction in W terim supplemental specifications, as they may	sis of payment shall not apply. e construed to mean the pertin isconsin," current edition, and pertain, except the method of	ent section of all measurement
34 35 36 37		C.	and basis of payment shall not apply. Where reference is made to the "State Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Highway and Structure Construction," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply				
38 39		D.	Where Section	reference is r 02 32 00.	nade to the "Geotechnical Report," it shall be co	onstrued to mean the Geotech	nical Report in
40 41 42		E.	ASTM I 1.	nternational (ASTM D698	ASTM): Laboratory Compaction Characteristics of So (600 kN-m/m ³))	oil Using Standard Effort (12 40	0 ft-lbs/ft ³
43 44 45			2.	ASTM D1557	Laboratory Compaction Characteristics of So (2,700 kN-m/m ³))	bil Using Modified Effort (56,00	i0 ft-lbs/ft³
45 46 47			3.	ASTM D6938	(Shallow Depth)	and Soll-Aggregate by Nuclear I	vietnods
48	1.5	PERM	ITS/FEES	5			
49 50 51		A.	Contra provide not lim	ctor shall be s ed by the Owr ited to, perm	olely responsible for obtaining all permits neces ner. Contractor shall pay all fees associated with its for work within public right-of-way, and build	sary to complete the work that obtaining permits. These incl ing permits.	t are not ude, but are
52 53		В.	The Ow Wiscon	vner will obta nsin Departme	in and provide the WDNR WRAPP permit, City Ent ent of Safety and Professional Services (WDSPS)	rosion Control and Stormwater Exterior Plumbing permit.	permits, and

1 2 2	1.6	PROVISIONS FOR FUTURE WORK None.					
3 4	1.7	SURVEY AND STAKING					
5		Α.	Owner will provide benchmarks and control points for the project as defined in Division 1.				
6		B	Contractor shall be responsible for transferring benchmarks, control points, lines, and grades as necessary to				
7		Б.	complete his work.				
8 9	DART	MA	ATERIALS				
10	<u>r An</u>	2 - 1417					
11	2.1	AGG	REGATE MATERIALS - GENERAL				
12		Α.	Alternate crushed aggregate material blends that are locally available will be considered on a project by project				
13			basis for crushed aggregate base courses, and will be subject to the City's approval. The Owner's Construction				
14			Representative will require the Contractor to furnish a gradation report on the materials.				
15 16	2.2	SPEC					
17	2.2		In certain cases, special fill materials may be required for specific nurnoses, such as stabilizing subgrades				
18		7	backfilling, undercut excavations or filling behind retaining walls. Fill materials shall meet the requirements of				
20			Rackfill Section 205 for Dance Graded Pase Course, Section 213 for Select Crushed Material, and Section 211 for				
20			Breaker Run of the State Specifications				
21			bleaker fun of the state specifications.				
22	23	GEO	ΤΕΧΤΙΙ Ε ΕΔΒRIC				
24	2.0	A.	Refer to Section 31 32 00 - Soil Stabilization.				
25							
26	PART	3 - EXE	ECUTION				
27 20	2 1	DDEC					
20	5.1		Poview plans and propare work plan and schedule. Coordinate any possessary interruptions in site access with				
29		А.	Owner's Construction Representative in accordance with other specification sections				
30		в	Contact Diggers Hotline Locate and protect utilities structures payement trees landscaping benchmarks and				
32		υ.	other features in the work area				
32		C	Lavout work Establish and transfer line and grade as necessary to complete the work				
34		С. D	Remove tonsoil from work area. Sawcut and remove navement from work area				
35		F.	Grade roadways and parking areas to drain water away				
36		L.					
37	3.2	PREF	PARE FOUNDATION FOR ASPHALTIC PAVEMENT				
38		A.	Provide all labor, materials, and equipment necessary to prepare the foundation for to a condition suitable for				
39			constructing and supporting asphaltic pavement in accordance with these Specifications and Section 211 of the				
40			State Specifications.				
41							
42	3.3	EXCA	VATION				
43		Α.	Excavate to elevations and dimensions as shown on the Drawings and as necessary to complete construction.				
44			Excavations shall be sufficiently deep to provide for all proposed base course and pavement.				
45		В.	Notify the Owner's Construction Representative if correction of unauthorized excavation or over-excavation is				
46			necessary. Said excavations will be corrected based on recommendations of the Owner's Construction				
47			Representative or Owner's Geotechnical Consultant. Contractor will be responsible for all costs associated with				
48			correcting these excavations, including fees charged by City and/or Owner's Geotechnical Consultant.				
49		C.	Segregate the various materials excavated. Reserve material meeting the requirements of backfill for the				
50			location. Excavated material that does not meet the requirements of backfill, and excess excavated material,				
51			shall be removed from the site and disposed by the contractor, unless directed otherwise by other specification				
52			sections or the Owner's Construction Representative.				
53		D.	Locate spoil piles in accordance with OSHA requirements, and so that it does not interfere with public travel,				
54			adjacent landowners or other construction activities.				
55	c						
56	3.4	FILL	AND COMPACTION				

Excavation shall be reasonably free of water prior to beginning filling. Do not place material on frozen surfaces Α. 57 58 or use frozen material.

1	В.	Fill areas using the material specified on Table 31 22 16.15 - 2, or as shown on the Drawings.
2	С.	Place and compact material to minimize settlement and avoid damage to structures, pipes, utility lines and other
3		features. Hand-place and compact material as necessary.
4	D.	Place backfill simultaneously on both sides of structures.
5	Ε.	Moisture condition backfill material as necessary to achieve density required for given use.
6	F.	Compact fill material as required by Table 31 22 16.15 - 2 for the given use. Compaction requirements based on
7		Modified Proctor Dry Density (ASTM D1557).
8	G.	It is the responsibility of the Contractor to provide all necessary compaction equipment and other grading
9		equipment that may be required to obtain the specified density. Vibratory plate or tamping type walk behind
10		compactors will be required whenever backfill is placed adjacent to structures, pipes, utility lines confined
11		spaces (i.e., backfilling undercut areas) and other features.
12		

	Percent (%) Co	mpaction (1)
Area	Clay/Silt	Sand/Gravel
Within 10 ft. of building lines		
Footing bearing soils	95	95
Under floors, steps and walks		
 Lightly loaded floor slab 	95	95
 Heavily loaded floor slabs and thicker fill zones 	95	95
Beyond 10 ft. of building lines		
Under walks and pavements - Granular Fill		
 Less than 2 ft. below subgrade 	95	95
 Greater than 2 ft. below subgrade 	92	92
Table 31 22 16.15 -	2	

- 15 Н. Where additional filling or excavation is necessary, or placement of base course will be delayed, roll surface of proposed roadway with a smooth drum roller to provide relatively impervious surface and promote drainage. 16 17 Roll with a smooth single drum vibratory roller having a minimum operating static weight of 12,000 pounds and 18 a minimum centrifugal force of 22,000 pounds to provide relatively impervious surface and promote drainage. 19 In the event the material is deficient in moisture content for readily obtaining the necessary density, it shall be 20 moistened to the degree necessary by means of approved equipment. The compaction operation shall continue 21 until the Engineer observes no visible displacement of material laterally or longitudinally under the compaction 22 equipment or hauling equipment. 23
- 24 3.5 SUBGRADE APPROVAL/PROOF-ROLLING
- A. Prior to undercutting or excavating below subgrade (EBS) or placing any base course, contact the Owner's
 Construction Representative to schedule inspection of subgrade and proof-rolling. Provide minimum of 24 hours
 confirmed notice. All proof-rolling shall be completed in the presence of the Owner's Construction
 Representative or Owner's Geotechnical Consultant.
- 29B.To complete proof-rolling, entire roadway subgrade shall be provided with a relatively smooth surface, suitable30for observing soil reaction during proof-rolling.
- C. Contractor shall schedule and provide a fully loaded tri-axle dump truck for proof-rolling. Loaded truck shall
 have a minimum gross operating weight of 30 tons. Test shall be conducted with "tag" or "pusher" axles
 retracted from the ground.
- 34D.Test rolling shall be accomplished in a series of traverses parallel to the centerline of the street or parking area.35The truck shall traverse the length of the street or parking area once for each 12 feet of width. Additional passes36along the traverse shall be completed as directed by the Owner's Construction Representative, to further define37unsatisfactory subgrade.
- 38 E. Soft areas, yielding areas, cracked areas or areas where rolling or wave action is observed shall be considered
 39 indicative of an unsatisfactory subgrade. Such areas shall be undercut as outlined in subsequent subsections of
 40 this specification.
- 41F.Once the subgrade has been proof-rolled and approved, protect the soils from becoming saturated, frozen, or42adversely altered.

1	3.6	UNDE	RCUTTING/EXCAVATION BELOW SUBGRADE (EBS)
2		Α.	Undercutting/EBS shall be completed only when directed by the Owner's Construction Representative. The
3			Contractor shall not be compensated for any unauthorized undercutting/EBS. Measure and document undercut
4			areas and depths in consultation with Owner's Construction Representative.
5		В.	Payment for undercutting/EBS shall be made on a unit price (cubic yard) basis measured in place at the rate as
6			defined in the Contract. Payment will be made only for the measured quantity of undercutting/EBS directed by
7			the Owner's Construction Representative to be performed. The unit price shall include all costs for labor and
8			materials necessary to remove and replace undercut areas including providing backfill materials and disposal of
9			excavated materials off-site.
10		C.	Excavate undercut areas to the depth specified using equipment with smooth cutting edge. Excavated undercut
11			material that does not meet the specifications for fill needed elsewhere on site shall be removed from the site
12			and legally disposed.
13		D.	Undercut areas shall be backfilled with 3-inch dense graded base course, as directed by the Owner's
14			Construction Representative in maximum of 6-inch thick lifts (compacted) or as directed by the Owner's
15			Construction Representative. Three-inch dense graded base course shall be compacted in thin lifts with a
16			vibratory compactor until no further consolidation is evident.
17			
18	3.7	GEOT	EXTILE FABRIC
19		Α.	When required by the Owner's Construction Representative geotextile fabric shall be installed over the subgrade
20			layer and prior to installing base aggregates. The Owner's Construction Representative shall determine if
21			geotextile fabric installation is required at the time of subgrade proof-rolling.

END OF SECTION

1 2					SECTION 31 23 16.13 TRENCHING
3 4	PART	1 - GEI	NERAL		
5 6	1.1	SECT			
7 8		A.	Exca	vation of trenches, p	ipe bedding, backfilling, and compaction for storm sewer, culverts, and water service.
9	1.2	RELA	TED SE	CTIONS	
10		Α.	Appli	icable provisions of I	Division 1 govern work under this Section. Related sections include:
11			1.	Section 02 20 00 -	- General Sitework Requirements
12			2.	Section 02 32 00 -	- Geotechnical Investigation
13			3.	Section 31 20 00 -	- Earthmoving
14			5.	Section 33 40 00 -	- Storm Sewer Construction
15 16	1 3	RFFF	RENCE	ORGANIZATIONS	
17	1.5	A.	Ame	rican Society for Test	ting and Materials:
18			1.	ASTM C33-586	Specification for Concrete Aggregate
19			2.	ASTM C136-84a	Method for Sieve Analysis of Fine and Coarse Aggregate
20			3.	ASTM D698-78	Test Methods for Moisture-Density Relations of Soils and Soil-Aggregate Mixtures,
21					Using
22					5.5-lb (2.49-kg) Rammer and 12-in. (304.8 mm) Drop
23			4.	ASTM D1557-78	Test Methods for Moisture-Density Relations of Soil-Aggregate Mixtures Using 10-lb.
24					(4.54- kg) Rammer and 18-in. (457-mm) Drop
25			5.	ASTM D2487-85	Classification of Soils for Engineering Purposes
26			6.	ASTM D2922-81	Test Methods for Density of Soil and Soil-Aggregate in Place by Nuclear Methods
27					(Shallow Depth)
28			7.	ASTM D3017-78	Test Method for Moisture Content of Soil and Soil-Aggregate in Place by Nuclear
29					Methods (Shallow Depth)
30		В.	Whe	re reference is made	e to the "Construction Standards", it shall be construed to mean the City of Madison's
31			Cons	truction Standards, o	except the method of measurement and basis of payment shall not apply.
32		C.	Whe	re reference is made	e to the "Standard Specifications," it shall be construed to mean the pertinent section of
33			the "	Standard Specification	ons for Sewer and Water Construction in Wisconsin," current edition, and all
34			supp	lemental and interin	n supplemental specifications, as they may pertain, except the method of measurement
35			and b	pasis of payment sha	II not apply.
36		D.	Whe	re reference is made	e to the "State Specifications," it shall be construed to mean the pertinent section of the
37			"Star	ndard Specifications	for Highway and Structure Construction," current edition, and all supplemental and
38			inter	im supplemental spe	ecifications, as they may pertain, except the method of measurement and basis of
39		_	paym	nent shall not apply.	
40		E.	Whe	re reference is made	e to the "Geotechnical Report," it shall be construed to mean the geotechnical report in
41			Secti	on 02 32 00.	
42				-	
43	1.4	SOBL	VIIIIAL)	of Division 1
44		A.	Subri	Testing Departs	of Division 1.
45		в.	Field	Density and Maist	tura Tasta. Submit within 14 days of tast data
40			1.	Density and worst	ture rests. Submit within 14 days of test date.
47 //8	DART	2 - PR(סדיווחר	:	
40 //Q		2-110	500013	<u>-</u>	
50	21	GENI	FRAI		
51	2.1	A.	Conf	orm to the requirem	ents of the Standard Specifications:
52			1.	Where conflicts b	etween this specification, the Standard Specifications, and the Construction Standards
53			-	exist, the most str	ingent requirements shall apply.
54					0
55	2.2	BEDD		ND COVER MATERIA	LS
56		Α.	Wate	er Mains:	
57			1.	Bedding and cove	r material shall conform to the Construction Standards and Standard Specifications.

1	2.3	BASE	MATERIAL
2		Α.	Crushed Stone: Hard, durable particles of crushed stone or gravel substantially free from shale or lumps of clay
3			or loam. When crushed stone base is required under sewer, water main, or structures, gradation shall meet the
4			requirements of Type 1 below. When crushed stone for trench bottom stabilization is required to affect soil
5			stability or drainage, it shall meet the gradation requirements of Type 2 below.
6			
7			Type 1: 1%-Inch Crushed Stone
8			The rest of the reason of the rest of the
0			Porcent Passing
9 10			
11			2 Jack 100
11			
12			1½ Inch 90-100
13			1 Inch 20-55
14			3/4 Inch 0-15
15			1/2 Inch 0-5
16			
17			Type 2: 2-Inch Crushed Stone
18			
19			Percent Passing
20			Sieve Size by Weight
21			2½ Inch 100
22			2 Inch 90-100
23			1½ Inch 35-70
24			1 Inch 0-15
25			1/2 Inch 0-5
26			
27	2.4	ВАСК	FILL
28		Α.	Granular Backfill: Durable particles ranging from fine to coarse in a substantially uniform combination.
29			Sufficient fine material shall be present to fill all the voids of the coarse material. Some fine clay or loam
30			particles are desirable, but clay or loam lumps shall not be present. Conform to the following gradation:
31			h
32			Granular Backfill
22			
3/			Dercent Passing
25			Siove Size by Weight
26			
27			
37 20			
20			
39			Finer than No. 200 5-15
40			Everyted Meterial, Network selfs die ACTM D2407 as Crevels (CW, CD CM and CC). Cande (CW, CD CM
41		в.	excluded material: Natural solis classified in ASTM D2487 as Graves (GW, GP GM and GC), salids (SW, SP, SM
42			and SC), and Sits and Clays (ML and CL). Sits and Clays classified as OL, MH, CH, OH, and PT are not acceptable
43			unless specifically allowed by Engineer. Soil material shall be free from vegetable or other organic matter, trash,
44			debris, stones larger than three inches and frozen material.
45		C.	Use of excavated material for backfill of public utilities shall be subject to approval of the Owner's Construction
46			Representative and Owner's Geotechnical Consultant <u>prior</u> to its use.
47			
48	PART	3 EXEC	CUTION
49			
50	3.1	GENE	
51		А.	Conform to the requirements of the Standard Specifications.
52			1. Where conflicts between this specification, the Standard Specifications, and the Construction Standards
53			exist, the most stringent requirements shall apply.
54			
55	3.2	EXAN	/INATION
56		Α.	Verify fill materials to be used are acceptable.
57			
58			

1	3.3	PREPARATION				
2		A.	Identify required lines, levels, contours and datum.			
3		В.	Maintain and protect existing utilities remaining, which pass through work area.			
4		C.	Protect plant life, lawns, and other features remaining as a portion of the final landscaping.			
5		D.	Protect bench marks, existing structures, shore protection structures and base materials, sidewalks, paving and			
6			curbs from excavation equipment and vehicular traffic.			
7		Ε.	Protect above and below grade utilities which are to remain.			
8		F.	Strip topsoil and stockpile on-site for re-use.			
9		G.	When excavating across or within existing pavement, saw cut in neat straight lines.			
10						
11	3.4	DEWAT	TERING			
12		A.	Do not allow water to accumulate in the trench.			
13		В.	Provide all dewatering equipment needed to accomplish the Work. Unless indicated otherwise, no additional			
14			compensation will be made for dewatering.			
15		C.	No additional compensation will be made for crushed stone used for trench drainage.			
16		D.	Dispose of water in a suitable manner without damage to property.			
17		E.	Install, operate and abandon dewatering equipment in accordance with applicable state and local codes.			
18		F.	Contact the Wisconsin Department of Natural Resources (WDNR) for a permit if the quantity of water to be			
19			pumped from dewatering equipment is in excess of 70 GPM.			
20						
21			Wisconsin Department of Natural Resources			
22			Private Water Supply Section			
23			P.O. Box 7921			
24			Madison, WI 53707-7921			
25	<u>.</u>					
26	3.5	EXCAV/	ATION Execute subscilles as wired double and such			
27		A.	Excavate subsoil to required depth and grade.			
28		в.	Cut trenches sufficiently while to enable installation of the utilities and allow inspection. Normal trench width			
29		c	below the top of the pipe shall be the nominal pipe diameter plus 24 inches. Do not undercut trench walls.			
30		C.	Trench wails above the top of the pipe shall be as dictated by soil type and safety requirements. Provide shoring			
31		P	and bracing as required to maintain safe working conditions.			
32		D.	stockpile excavated material in area designated on-site.			
27	26					
25	5.0		No Disce hedding in trench hefere installing nine			
36		A. R	Support nine during placement and compaction of hedding			
30		в. С	Provide a minimum of 4 inches of hedding material under the nine harrel and under the hell			
38		с. D	Lightly consolidate the material so that it fills and supports the baunch area and encases the nine to the limits			
30		D.	shown on the Drawings			
40		F	If excavation is carried deeper than 6 inches below the nine barrel backfill the excess denth with 1½-inch			
40		с.	crushed stone meeting the requirements of paragraph 2 3-A of this section			
42		F.	After the pipe has been laid and jointed, place bedding materials by hand or equally careful means around the			
43			sides of the pipe and up to a level 12 inches above the pipe. Lightly consolidate the material.			
44						
45	3.7	BACKFI	ILING			
46	-	A.	Backfill trenches to contours and elevations with unfrozen materials.			
47		В.	Do not backfill over wet, frozen, or spongy subgrade surfaces.			
48		C.	Granular Backfill: Place and compact materials in continuous layers not exceeding 12 inches compacted depth.			
49		D.	Natural Soil Backfill: Place and compact material in continuous layers not exceeding 8 inches compacted depth.			
50		E.	Maintain optimum moisture content of backfill materials to attain required compaction density.			
51		F.	Utilize surplus backfill materials on project or remove surplus backfill material from site.			
52		G.	Leave fill material stockpile areas completely free of excess fill materials.			
53		Н.	At all manholes, 3/4-inch crusher run stone shall be installed from the top of the cone to the top of the casting.			
54						
55	3.8	TOLERA	ANCES			
56		Α.	Top Surface of Backfilling Under Paved Areas: Plus or minus (±) .05 feet from required elevations			
57		В.	Top Surface of General Backfilling: Plus or minus (±) 0.2 feet from required elevations			
58						

1 3.9 FIELD QUALITY CONTROL

- A. Field inspection and testing will be performed under the provisions of Division 1 by the Owner's testing agency.
 - B. Density/moisture relationship will be determined in accordance with ASTM D1557 (Modified Proctor).
 - C. Compaction testing will be performed by Owner's testing agency and will be in accordance with ASTM D2922 and ASTM D3017.
 - D. If tests indicate Work does not meet specified requirements, remove Work, replace, and retest at no cost to the Owner. Additional testing of the removed and replaced Work will be at the expense of the Contractor.
 - E. Frequency of Tests:
 - 1. For trenches under paved areas, one test per 100 linear feet of trench.
 - 2. For trenches under unpaved areas, one test per 250 linear feet of trench.

12 3.10 COMPACTION SCHEDULE

- A. For paved areas compact to at least 95% of optimum density in accordance with ASTM D1557.
- B. For unpaved areas compact to at least 92% of optimum density in accordance with ASTM D1557.
- 14 15 16

2

3

4

5

6

7

8

9

10

11

13

END OF SECTION

1 2			SECTION 31 25 00 FROSION CONTROL
3			
4	PART	<u>1 - GEN</u>	ERAL
5	1.1	SUMN	1ARY
7		A.	Related sections include:
8			1. Section 31 20 00 - Earthmoving
9			2. Section 31 32 00 - Soil Stabilization
10			3. Section 32 92 19 - Seeding and Sodding
11			
12	1.2	REFER	ENCE STANDARDS
13		Α.	Where reference is made to the "Construction Standards," it shall be construed to mean the City of Madison's
14			Construction Standards, except the method of measurement and basis of payment shall not apply.
15		В.	Where reference is made to the "Standard Specifications," it shall be construed to mean the pertinent section of
16			the Standard Specifications for Sewer and Water Construction in Wisconsin, Current Edition, and all
17			supplemental and interim supplemental specifications, as they may pertain, except the method of measurement
18			and basis of payment shall not apply.
19		C.	Where reference is made to the "State Specifications," it shall be construed to mean the pertinent section of the
20			Standard Specifications for Highway and Structure Construction, current Edition, and all supplemental and
21			interim supplemental specifications, as they may pertain, except the method of measurement and basis of
22			payment shall not apply. Where reference is mode to the "Contrological Depart" it shall be construed to mean the post-solution with
23		D.	Soction 02.22.00
24			Section 02 52 00.
25	13	SURM	ΙΤΤΔΙ S
27	1.5	A.	Provide a detailed Erosion Control and Sequencing Plan for approval by the Owner's Construction
28			Representative, if differing from the approved sequencing and erosion control plans issued as part of the
29			Contract Documents.
30		В.	Provide manufacturer's data and WisDOT Product Acceptability List verification for silt fence, temporary ditch
31			checks and erosion mat for review and approval by Owner's Construction Representative prior to procurement.
32		C.	Identify seed supplier and provide seed source, purity and germination specifications, for all seed mixes specified
33			for installation in this section, to Owner's Construction Representative for approval prior to procurement.
34		D.	Provide manufacturer's data for fertilizer for review and approval by Owner's Construction Representative prior
35			to procurement.
36			
37	1.4	QUAL	TY ASSURANCE
38		Α.	Contractor shall ensure that the General Provisions and Special Conditions of the following permits issued for the
39			project shall be complied with at all times:
40			1. City of Madison Stormwater Management and Erosion Control Permit.
41			2. WDNR General Permit to discharge under the Wisconsin Pollutant Discharge Elimination System, for land
4Z 42		в	disturbing construction activities.
43		в. С	The status of eracion control materials and supplies after delivery to verify that no damage has occurred.
44 15		C.	corrective actions required during construction meetings shall be accomplished within three working days of the
45			meeting date
40		D	Contractor shall provide weekly written reports on the erosion control system for the previous week to the
48		υ.	Owner's Construction Representative for the duration of construction in a format approved by the Engineer.
49			These reports shall be provided at each weekly construction meeting and shall be reported to the City's erosion
50			control reporting representative (electronic PDF preferred). The weekly erosion control report shall describe:
51			1. The extent of erosion control system installed.
52			2. The condition of erosion control measures for that week, based on field observations.
53			3. Any accidental release of sediment.
54			4. A summary of daily rainfall/snowmelt data for the week.
55			5. Any specific corrective action taken.
56			6. Corrective action that needs to be taken.
57			7. The person that conducted the observations shall sign the report.
58			

1	1.5	WARRANTY					
2		Α.	Work conducted under this section shall be subject to the one-year warranty provisions described in the General				
3			Conditions of contract.				
4							
5	1.6	SEQU	EQUENCING AND SCHEDULING				
6		Α.	The sequencing of project construction activities will be generally as described in the plans and Contract				
7			Documents. The specific sequence for construction within a particular area shall be agreed upon with Owner's				
8			Construction Representative prior to construction within that area.				
9		В.	All erosion control measures shall be completely installed for each construction area and approved by Owner's				
10			Construction Representative before any other construction activity takes place.				
11							
12	PART :	2 - PRO	<u>DUCTS</u>				
13							
14	2.1	MATE	RIALS				
15		Α.	Silt Fence:				
16			1. Silt fence shall be as specified in the WDNR Construction Site Erosion & Sediment Control Technical				
17			Standard 1056.				
18		В.	Erosion Mat - Class I, Type B and Class II, Type B:				
19			1. Erosion control mat shall be to the requirements of WDNR Construction Site Erosion & Sediment Control				
20			Technical Standard 1052.				
21			2. WisDOT Erosion Mat Class I, Type B erosion mat meeting the requirements of Section 628.2.2 of the State				
22			Specifications shall be used for all seeded areas within the public right-of-way unless noted otherwise on				
23			the plans.				
24		C.	Seeding Temporary:				
25			1. Temporary seed shall be 100% Annual Ryegrass, with purity and germination requirements as specified in				
26			Section 630.2.1.5.1.2 of the State Specifications or as indicated in the WDNR Construction Site Erosion &				
27			Sediment Control Technical Standard 1059. Temporary seeding will be incidental to the grading items in				
28			the contract.				
29		D.	Mulch:				
30			1. Mulch proposed for use shall be clean straw, with no weed material or seeds, and shall be approval				
31			Engineer before use.				
32			2. Mulch shall meet the standards set forth within the WDNR Construction Site Erosion & Sediment Control				
33			Technical Standard 1058.				
34		Ε.	Tracking Pads:				
35			1. Stone for use in temporary access pads shall range in size from 3 to 6 inches in diameter.				
36			2. Pad shall be a minimum of 50 feet long.				
37			3. Pad shall meet the requirements of WDNR Construction Site Erosion & Sediment Control Technical				
38		_	Standard 1057.				
39		F.	Fertilizer - Type A:				
40		-	1. Fertilizer shall be as specified in Section 629.2.1.2 of the State Specifications for Fertilizer, Type A.				
41		G.	Riprap:				
42			1. Provide riprap as specified in Section 606.2.1 of the State Specifications for the size and type indicated on				
43			the construction drawings or bid form. If the size is not specified, medium riprap shall be used.				
44		н.	Temporary Ditch Checks:				
45			1. Provide temporary ditch checks of material found on WisDOI's PAL list.				
46			2. Submit a written copy of the proposed material and manufacturer's specification for installing the				
47			product on slopes channels, and next to live traffic lanes as applicable to the project to the Owner's				
48			Construction Representative for approval prior to installation.				
49			3. Erosion Bales shall not be used on this project as a sole means of perimeter erosion control. Erosion				
50			bales may be used to reinforce or support other primary means of perimeter erosion control, like silt				
51			Tence.				
52		Ι.	Iniet Protection - Kigid Framed:				
53			1. Use Flexstorm Catch-It rigid framed temporary inlet protection complying with ASTM D8057. Contractor				
54			snall provide inlet protection to fit the existing drainage structures.				
55		J.	KOCK UNECK Dams:				
50			1. Provide rock check dams in accordance with the standard detail drawings at locations identified in the				
5/			plans and as directed by the Owner's Construction Representative.				
ъŏ							

1	PART 3 - EXECUTION						
2	3 1	GENERAL					
4	5.1		tablish all beights and grades to properly execute work from benchmark established by others				
5		B. Co	intractor shall provide all surveys to accurately locate the construction on the site.				
6		C. Pro	ovide temporary erosion control measures in accordance with the Contractor's approved erosion control and				
7		se	guencing plan. These measures may include temporary sedimentation basins, diversion berms and swales and				
8		ot	her measures constructed in accordance with the WDNR Technical Standards.				
9							
10	3.2	EROSION	CONTROL STRUCTURES				
11		A. Ru	noff diversion berms shall be constructed of clean topsoil 2 feet high with 3H:1V side slopes and seeded and				
12		m	ulched immediately after installation.				
13		B. Sil	t fence shall be placed according to the WDNR Construction Site Erosion & Sediment Control Technical				
14		Sta	andard 1056.				
15	3.3	SEEDING T	TEMPORARY AND MULCHING				
17	0.0	A. Te	emporary seeding shall be conducted as described in Section 630.3.3 of the State Specification, with sowing				
18		usi	ing either Method A or Method B. Temporary seeding areas shall receive fertilizer at the rate of				
19		10) pounds/1,000 square feet.				
20		B. Te	mporary seed shall receive mulch at the rate of 2,500 pounds/acre, and shall be crimped into the soil using				
21		W	isDOT Procedure specified in Section 627.3.2.3 of the State Specifications.				
22		C. Dis	sturbed areas within the construction site shall be graded, prepared for seeding, and seeded to conform to the				
23		fol	llowing requirement for the maximum duration of bare-ground conditions:				
24		1.	Areas within 100 feet of and draining directly to wetlands or watercourses, with slopes less than 5%:				
25		2	seven days				
20		Ζ.	Areas within 100 feet of and draining directly to wetlands or water courses, with slopes between 5% and				
27		з	Areas in the interior of the site that do not drain directly to wetlands and water courses: 30 days				
29		5.	Areas in the interior of the site that do not aran anceay to we lands and water courses. So days.				
30	3.4	EROSION	MAT CLASS I TYPE B and CLASS II TYPE B				
31		A. Ero	osion control mat shall be applied according to WDNR Technical Standards 1052 or 1053 as applicable and				
32		ma	anufacturer's requirements.				
33							
34	3.5	TRACKING	3 PAD				
35		A. Ins	stall tracking pads at the locations as shown in the plans or as directed by the Owner's Construction				
36		Re D T	presentative. The Bada data line installed and university discovered area with Creation C20 2 4C of the Chate Creatification				
3/ 20		B. Ira	acking Pads shall be installed and maintained in accordance with Section 628.3.16 of the State Specification.				
30 20		C. 110	liacent work area is stabilized				
40		au					
41	3.6	FERTILIZER TYPE A					
42		A. Fe	rtilizer applied to temporary seeding areas shall be applied as specified in Section 629.2.1.2 of the State				
43		Sp	ecification at locations where temporary seeding is required.				
44							
45	3.7	RIPRAP					
46		A. Pla	ace riprap of the specified size at locations as shown in the construction drawings.				
47		B. Pla	ace riprap in accordance with Section 606.3 of the State Specifications.				
48		C. Rip	prap at outfail locations shall be placed immediately after or concurrent with the placement of the apron				
49 50		en	awan. Riprap at the outraits is intended to be left in place as a permanent erosion control measure.				
50	3 8	TEMPORA					
52	5.5	A. Pla	ace and maintain temporary ditch checks at the locations shown on the construction drawing and as directed				
53		bv	the Owner's Construction Representative				
54		B. Pla	ace and maintain temporary ditch checks in accordance the manufacturer's instructions and Section 628.3.14				
55		of	the State Specifications, except erosion bales are not to be used as temporary ditch checks on this project.				
56		C. Re	move ditch checks after the slope ditches are stabilized in accordance with Section 628.3.14 of the State				
57		Sp	ecifications.				

1	3.9	INLET PROTECTION - RIGID FRAMED		
2		Α.	Furnish install and maintain ridged framed inlet protection in accordance with Flex-Storm Catch-it manufactures	
3			specifications, and City of Madison and Dane County requirements.	
4		В.	Inlet protection shall be maintained throughout construction, and removed once the area adjacent to the inlet	
5			has been stabilized and as directed by the Owner's Construction Representative.	
6				
7	3.10	MAIN	TENANCE AND CLEANUP	
8		Α.	The erosion control system shall be maintained throughout the duration of the construction project, in	
9			accordance with the procedures identified in Section 628.3.4.2 of the State Specifications.	
10		В.	The erosion control system shall be inspected immediately after each rainfall of more than 0.5 inch, and daily	
11			during prolonged rainfall. All inspections shall be reported to the Owner's Construction Representative in the	
12			weekly erosion control system report.	
13		C.	Accumulated sediment within the erosion control system shall be removed before one-half of the storage	
14			capacity of the erosion control measure is used, or as specified by the Owner's Construction Representative.	
15		D.	Accumulated sediment in riprap shall be removed as directed by the Owner's Construction Representative	
16			during the project, and as a final condition of acceptance if deficiencies are noted at final walk through.	
17				
18	3.11	ROCK	CHECK DAMS	
19		А.	Place and maintain rock check dams at the locations shown on the construction drawing and as directed by the	
20			Owner's Construction Representative	
21		В.	Remove sediment deposits when the build-up reaches approximately one-third of the height of the rock check	
22			dam, and as directed by the Owner's Construction Representative. Contractor may also remove and replace the	
23			stone check with sediment and replace with new stone at their discretion. Each location will be paid for initial	
24			placement only maintenance is incidental to this item.	
25		С.	Remove rock check dams after the slopes and ditches are stable and turf develops enough to make future	
26			erosion unlikely. The Owner's Construction Representative will determine when the contractor meets this	
27			criteria.	
28				
29			END OF SECTION	

1 2 2				SECTION 31 32 00 SOIL STABILIZATION		
5 4	<u>PART</u>	T 1 - GENERAL				
5	11	SUM	MARV			
7	1.1	A.	Secti	ion includes:		
8		7	1.	Geotextile fabric and geogrid for stabilization of subgrade.		
9		В.	Relat	ted requirements, refer to the following section:		
10			1.	Section 31 20 00 - Earthmoving		
11						
12	1.2	REFE	RENCES	5		
13		Α.	The p	publications listed below form a part of this specification to the extent referenced. Publications are		
14		_	refer	renced within the text by the basic designation only.		
15		В.	Whe	re reference is made to the "State Specifications," it shall be construed to mean the pertinent section of the		
15			Stand	dard Specifications for Highway and Structure Construction, current edition, and all supplemental and		
17 18			navn	and shall not apply		
19			payn	nent shan hot appry.		
20	1.3	SUBN		S		
21		A.	Subn	nit manufacturer's specifications for geotextile fabric and geotextile grid.		
22						
23	PART	2 - PRC	DUCTS	<u>S</u>		
24						
25	2.1	MAN	UFACT	URERS		
26		A.	Provi	ide products from one of the following manufacturers as specified in the Materials paragraph below:		
27			1.	Tencate Geosynthetics North America (Mirati), Pendergrass, GA - (706) 693-2226, <u>www.tencate.com</u>		
28			2. 2	Hanes Geo Components (WEBTEC), Winston Salem, NC - (336) 747-1600, <u>WWW.nanesgeo.com</u>		
29			5. 1	Thrace-LINO Inc. Summenville, SC - $(8/3)$ 873-5800, www.thraceling.com		
31			5.	DuPont (Typar), Summerville, SC - (843) 832-6860, www.typargeo.com		
32			6.	Synteen Technical Fabrics, Lancaster, SC - (800) 796-8336, www.synteen.com		
33						
34	2.2	MAT	ERIALS			
35		Α.	Aggr	egate:		
36			1.	Coarse Aggregate: Crushed carbonate, crushed gravel, crushed air-cooled slag, granulated slag, a mixture		
37				of crushed and granulated slag, or other types of suitable material meeting the gradation requirements		
38				of Section 305 of "State of Wisconsin Standard Specifications for Highway and Structure Construction,"		
39			2	latest edition.		
40 41			Ζ.	Fine Aggregate: Sand - Natural river of bank sand; washed; free of slit, clay, loam, friable of soluble		
41 42				Standard Specifications for Highway and Structure Construction " latest edition		
43			3.	Subsoil: Existing to be re-used.		
44						
45	2.3	ACCE	SSORIE	is a second s		
46		Α.	Geot	extile Fabric for Stabilization - provide one of the following:		
47			1.	Mirafi HP 370 or HP 570, by TenCate		
48			2.	SF40 or SF65, by DuPont		
49			3.	GTF-200 or 300, by Thrace-LINQ		
50			4.	TerraTex HD, by Hanes		
51 52		В.	Geog	grid for Stabilization - provide one of the following:		
52 E 2			1. ว	Biaxial Geogrid Type 1 (formerly BX 1100), by Tensar Biaxial Geogrid Type 2 (formerly BX 1200), by Tensar		
55 54			∠. २	Biaxiai Geogriu Type 2 (Tormeny DA 1200), by Tensai Mirafi RXG 11 hv TenCate		
55			3. 4.	Mirafi BXG 12, by TenCate		
56			5.	SF 11, by Synteen		
57			6.	SF 12, by Synteen		

1 PART 3 - EXECUTION 2

4

5

6

7

9

10

11

13

14

15

16

17 18

24

3 3.1 PREPARATION

- Start stabilization only when weather and soil conditions are favorable for successful application of proposed Α. material.
- Β. Proof-roll subgrade to identify areas in need of stabilization.

EQUIPMENT 8 3.2

Α. Perform operations using suitable, well maintained equipment capable of excavating subsoil, mixing and placing materials, wetting, consolidating, and compacting of material.

EXCAVATION 12 3.3

- Α. Excavate subsoil to depth sufficient to accommodate soil stabilization.
- Β. Remove lumped subsoil, boulders, and rock that interfere with achieving uniform subsoil conditions.
- Notify Owner's Construction Representative in writing of unexpected subsurface conditions. Discontinue C. affected work in area until notified to resume work.
- D. Remove excess excavated material from site.

3.4 **GEOTEXTILE FABRIC AND/OR GEOGRID** 19

20 Α. Place geotextile fabric and/or geogrid over subsoil surface, lap edges and ends in accordance with 21 manufacturer's recommendations in those areas that are shown on Construction Drawings or in those areas that 22 need additional stabilization prior to placement of base course. Place geotextile fabric and/or geogrid in 23 accordance with manufacturer's recommendations.

25 3.5 FIELD QUALITY CONTROL

- 26 Α. Field quality control shall be the responsibility of the Contractor in accordance. Except for specified mandatory 27 testing, field quality control testing and inspection shall be at the discretion of the Contractor as necessary to 28 assure compliance with Contract requirements. The Owner's Construction Representative specified below shall 29 not be considered a substitute for the Contractor's responsibility to perform similar routine, necessary, and 30 customary testing and inspection of the methods and frequency suitable for the type of work involved. 31 32 TESTING 3.6 Field Density: Field in-place density shall be determined as specified in Section 31 20 00 - Earthmoving. 33 Α. 34 35
 - END OF SECTION

1					SECTION 32 05 00	
2		COMMON WORK RESULTS FOR EXTERIOR IMPROVEMENTS				
3						
4	PAR	RT 1 - GENERAL				
5				_		
6	1.1	sco	PE			
7		Α.	This s	ection provides infor	mation common to two or more technical sitework specification sections or items that are	
8			of a g	eneral nature, and ne	ot included in other sections.	
9						
10	1.2	REL/	ATED W	/ORK		
11		Α.	Appli	cable provisions of D	ivision 1 govern work under this Section. Related sections include:	
12			1.	Section 31 23 16 1	L3 - Trenching	
13			2.	Section 31 25 00 -	Erosion Control	
14			3.	Section 32 91 19 -	Topsoil-Select Fill Materials and Application	
15						
16	1.3	REFI		D ORGANIZATIONS		
17		Α.	Abbre	eviations of organizat	tions referenced in these specifications are as follows:	
18				AASHTO	American Association of State Highway and Transportation Officials	
19				ACPA	American Concrete Pipe Association	
20				ANSI	American National Standards Institute	
21				ASCE	American Society of Civil Engineers	
22				ASME	American Society of Mechanical Engineers	
23				ASTM	American Society for Testing and Materials	
24				AWWA	American Water Works Association	
25				AWS	American Welding Society	
26				FHA	Federal Highway Administration	
27				EPA	Environmental Protection Agency	
28				NEC	National Electric Code	
29				NFMA	National Electrical Manufacturers Association	
30				NEPA	National Fire Protection Association	
31				NSF	National Sanitation Foundation	
32				OSHA	Occupational Safety and Health Administration	
32				STI	Steel Tank Institute	
34					Underwriters Laboratories Inc	
25					State of Wisconsin Department of Natural Resources	
36				WISDOT	State of Wisconsin Department of Transportation	
30				WISDOT	State of Wisconsin Department of Transportation	
20	1 /	DEEI				
30	1.4		Whor	e reference is made	to the "Construction Specifications" it shall be construed to mean the pertinent section of	
10		д.	the C	ity of Madison's "St	tandard Construction Specifications, " current edition, and all supplemental and interim	
40 //1			suppl	emental specification	and and construction specifications, current edition, and an supplemental and interna-	
41 12			annly	cinental specification	is, as they may pertain, except the method of measurement and basis of payment shall not	
42		D	appiy Whor	o roforonco is mado :	to the "Standard Specifications" it shall be construed to mean the portinent section of the	
43		Б.	"Stan	dard Specifications f	or Sewer and Water Construction in Wisconsin " current edition, and all supplemental and	
44			intori	m supplemental spec	rifications, as they may portain, except the method of measurement and basis of navment	
45 46			chally	ni supplemental spec	cincations, as they may pertain, except the method of measurement and basis of payment	
40		c	M/bor	o reference is made	to the "Ctate Specifications" it shall be construed to mean the participant section of the	
47 10		C.	WicD	OT "Standard Specify	cotions for Highway and Structure Construction " current edition, and all cumplemental and	
40			intori	m supplemental specific	rifications as they may portain, except the method of measurement and basic of navement	
49 E0			chally	ni supplemental spec	cincations, as they may pertain, except the method of measurement and basis of payment	
50		р	Siidli I	o roforonco is modo	to the "Contachnical Panart" it shall be construed to mean the gestachnical report in	
51		D.	Voner		to the Geolechnical Report, it shall be construed to mean the geolechnical report in	
52			Sectio	011 UZ 3Z UU.		
JJ ⊑∕	4 -	<u></u>				
54 57	1.5	ųυ# ^			ducto as required by individual englification costions. Defer to Constal Conditions of the	
55		А.	Contro	ae materiais and pro	required by individual specification sections. Refer to General Conditions of the	
טכ רק		Б	Contr		regarding substitutions.	
5/		в.	Provid	ae quality assurance	testing and reporting as required by individual specification sections.	
ЪQ						

1.6 SAFETY 1 2

3

4

5

6

7

8

9

10

12

13

14

15

16

17

18

20

21

22

23

24

25 26

27

28

29

30

31

33

34

36

37

38

39

40

41

43

45

46

47

- Contractor is solely responsible for worksite safety. Α.
- Perform all work in accordance with applicable OSHA, state, and local safety standards. Β.
 - C. Contact Diggers Hotline at 1-800-242-8511 in accordance with statutory requirements. Request that non-member utilities and private utilities be located by the appropriate parties.

PERMITS 1.7

Unless otherwise noted in the Contract Documents, Contractor shall be responsible for obtaining and paying for all Α. permits necessary to complete the work.

11 1.8 CONSTRUCTION LIMITS

- Construction Limits are indicated on the Drawings. In the absence of such a designation on the Drawings, confine Α. work to the minimum area reasonably necessary to undertake the work as determined by the Owner's Construction Representative. In no case shall construction activities extend beyond state property lines or construction easements.
 - Β. The Contractor shall restore all disturbed areas in accordance with the Drawings and Specifications. If plans and specifications do not address restoration of specific areas, these areas will be restored to pre-construction conditions as approved by the Owner's Construction Representative.

SUBMITTALS 19 1.9

- Refer also to General Conditions of the Contract for Construction and Division 1. Α.
- Β. Submit manufacturer's shop drawings, product data, samples, substitutions, and operation and maintenance (O&M) data for approval as required by individual specification sections.
- C. Submittals shall be provided to the Owner's Construction Representative for review and approval, unless otherwise directed. Submittals shall be submitted electronically by email in *.pdf format unless otherwise directed.

1.10 OFF-SITE STORAGE

- Α. Refer to Division 1.
- Β. In general, the payments for materials stored off site will only be considered in instances where there is limited space available for storage on the site. Prior approval by the Owner's Construction Representative, together with the execution of a Storage Agreement, will be required.

32 1.11 CODES

Α. Comply with the requirements of all applicable local, state, and federal codes.

35 1.12 CERTIFICATIONS AND INSPECTIONS

- Refer to General Conditions of the Contract for Construction. Α.
- Β. Obtain and pay for all required sampling, testing, inspections, and certifications except those expressly listed as provided by the Architect/Engineer (A/E) or other third-party in the Contract Documents. Deliver originals of certificates and documents to the Owner's Construction Representative within three (3) days; provide copies to the City Engineer. Include copies of the certifications and documents in the O&M Manual.

42 PART 2 - MATERIALS

44 2.1 BARRICADES, SIGNS, AND WARNING DEVICES

Α. Traffic barricades, traffic signs, and warning devices shall meet the requirements of applicable OSHA standards and the FHA Manual of Uniform Traffic Control Devices (MUTCD).

48 TEMPORARY PLASTIC BARRIER FENCING 2.2

- 49 50
- UV stabilized high-density polyethylene barrier fence free of holes tears and other defects. Provide 4-foot tall fence Α. in diamond or rectangular pattern. Fencing shall be a "safety orange" color, unless otherwise noted.
- Β. Posts for temporary plastic barrier fencing shall be 5-foot tall, minimum 12-gauge, painted metal posts.

53 PART 3 - EXECUTION

54 55

51

52

3.1 MAINTENANCE OF SITE AND BUILDING ACCESS/EGRESS

56 Unless otherwise shown or directed, maintain existing access and egress to the facility throughout construction. Α. 57 Maintain ANSI A117 compliant access for disabled persons, delivery access, emergency vehicle access, and emergency

1 egress. Do not interrupt access and egress without prior written approval from the Owner's Construction 2 Representative. 3 4 CONTINUITY OF EXISTING TRAFFIC/PARKING AND TRAFFIC CONTROL 3.2 5 Refer also to General Conditions of the Contract for Construction; or in the specifications, particularly in the General Α. 6 Requirements (Division 1). Β. Do not interrupt or change existing traffic, delivery, or parking outside the requirements of the staging plan without 7 8 prior written approval from the Owner's Construction Representative. When interruption is required, coordinate 9 schedule with the Owner's agency to minimize disruptions. When working in public right-of-way, obtain all necessary 10 approvals and permits from applicable municipalities and WISDOT. 11 C. When Contractor's activities impede or obstruct traffic flow, Contractor shall provide traffic control devices, signs and flaggers in accordance with other Contract Documents and the current version of the MUTCD, or as shown on the 12 13 Drawings. 14 15 3.3 PROTECTION AND CONTINUITY OF EXISTING UTILITIES 16 Verify the locations of any water, drainage, gas, sewer, electric, drainage, gas, sewer, electric, Α. 17 telephone/communication, fuel, steam lines or other utilities and site features which may be encountered in any 18 excavations or other sitework. All lines shall be properly underpinned and supported to avoid disruption of service. Β. Do not interrupt or change existing utilities without prior written approval from the Owner's Construction 19 20 Representative, affected utilities and users. Notify all users impacted by outages a minimum of 48 hours in advance 21 of outage. Notification shall be provided in writing and describe the nature and duration of outages and provide the 22 name and number of Contractor's foreman or other contact. 23 C. Any service connections encountered which are to be removed shall be cut off at the limits of the excavation and 24 capped in accordance with the requirements of applicable codes and any specifications governing such removals. 25 26 3.4 PROTECTION OF EXISTING WORK AND FACILITIES 27 Verify the locations of and protect any signs, paved surfaces, buildings, structures, landscaping, streetlights, utilities, 28 and all other such facilities that may be encountered or interfered with during the progress of the Work. Take 29 measures necessary to safeguard all existing Work and facilities that are outside the limits of the Work or items that are within the construction limits but are intended to remain. Report any damage to existing facilities to the Owner's 30 31 Construction Representative immediately. Correct and pay for all damages. 32 33 3.5 STORMWATER/EXCAVATION WATER MANAGEMENT 34 Control grading around structures; pitch ground to prevent water running into excavated areas. Α. Pits and other excavations shall be maintained free of water. 35 Β. 36 С. Provide trench pumping and other dewatering facilities required. 37 D. Notify A/E if springs or running water are encountered in excavation; provide discharge by trenches, drains, pumping 38 to point outside of excavation. Provide information to A/E of points and areas that water will be discharged. At the 39 Engineer's option, the Contractor shall drain the spring to the storm sewer system by the use of field tile. 40 Ε. Be responsible for control measures to prevent damage from flooding, erosion, and sedimentation to on-site and off-41 site areas. 42 43 END OF SECTION

Page Intentionally Left Blank

1 2	SECTION 32 11 23 AGGREGATE BASE COURSE						
3 4 5	<u>PART</u>	PART 1 - GENERAL					
6	1.1	SUMN	IARY				
7 8		A.	This Section includes prov paving, concrete curb and	isions for providing aggregate base course as the foundation for hot-mixed asphalt gutter, and concrete sidewalk.			
10	1.2	RELAT	ED SECTIONS				
11		A.	Drawings and general prov	visions of the Contract, including General and Supplementary Conditions and			
12			Division 1 - Specification S	ections, apply to this Section.			
13		В.	The following sections cor	tain requirements that relate to this Section:			
14			1. Section 02 20 00 -	General Sitework Requirements			
15			3. Section 31 20 00 -	Earthmoving			
16			4. Section 31 22 00 - 5	Soil Stabilization			
17			5. Section 32 91 19 -	Topsoil-Select Fill Materials and Application			
10	1 2	DEEED					
20	1.5		Where reference is made	to the "Construction Specifications" it shall be construed to mean the pertinent			
21		/	section of the City of Mad	ison's Standard Construction Specifications, current edition, and all supplemental and			
22			interim supplemental spec	cifications, as they may pertain, except the method of measurement and basis of			
23			payment shall not apply.				
24		В.	Where reference is made	to the "Standard Specifications," it shall be construed to mean the pertinent section of			
25			the "Standard Specificatio	ns for Sewer and Water Construction in Wisconsin," current edition, and all			
26			supplemental and interim	supplemental specifications, as they may pertain, except the method of measurement			
27			and basis of payment shal	I not apply.			
28		C.	Where reference is made	to the "State Specifications," it shall be construed to mean the pertinent section of the			
29			wisDOT "Standard Specific	cations for Highway and Structure Construction," current edition, and all supplemental			
30 21			and interim supplemental	specifications, as they may pertain, except the method of measurement and basis of			
32		П	Where reference is made	to the "Geotechnical Report" it shall be construed to mean the geotechnical report in			
33		υ.	Section 02 32 00.				
34		Ε.	American Society for Testi	ng and Materials (ASTM):			
35			1. ASTM D1557-78:	Test Methods for Moisture-Density relation of Soil and Soil-Aggregate Mixtures			
36				Using 10 lbs. (4.54-kg) Rammer and 18-in. (457 mm) Drop			
37			2. ASTM D698:	Laboratory Compaction Characteristics of Soil Using Standard Effort (12 400 ft-			
38				lbs./ft ³ (600 kN-m/m ³))			
39			3. ASTM D1557:	Laboratory Compaction Characteristics of Soil Using Modified Effort (56,000 ft-			
40				lbs./ft ³ (2,700 kN-m/m ³))			
41			4. ASTM D6938:	In-Place Density and Water Content of Soil and Soil-Aggregate by Nuclear Methods			
42				(Shallow Depth)			
43 44	1 /	CLIDM	ΙΤΤΛΙ ς				
44 45	1.4		General: Submit the follow	wing in accordance with Conditions of Contract and Division 1 Specification Sections			
46		74.	1. Submit 50-pound s	amples of each type of aggregate to testing laboratory for materials not obtained from			
47			on-site stockpiles a	ind for blended aggregate.			
48			2. Weight slips of eac	h load showing the net weight of the aggregate.			
49			0				
50	PART	2 - PRO	DUCTS				
51							
52	2.1	MATE	RIALS	· · · · · · · · · · · · · · · · · · ·			
53		A.	Provide materials meeting	the requirements of the Geotechnical Report and WisDOT Sections 301, 305, and 306			
54			and as shown in the Draw	Ings. Jod acceptite products will not be an acceptable alternative or equal to 11/ Deves Grade			
55 56			 Recidimed of recyc Rase material 	neu asphait products will not be an acceptable alternative or equal to 1% Dense Grade			
57		в	Hot-Mix Sand Asnhalt Rass	es. Asphalt Institute Type VI VII or VIII Mixes for Hot-Mix Sand Asphalt Rases. Hot-			
58		5.	mix base shall be used onl	y under asphaltic concrete surfaces.			

1	<u>PART</u>	ART 3 - EXECUTION						
2	2.1							
3	3.1	PREP						
4		А.	Prepare the subgrade in accordance with the Geotechnical Report and Section 31 20 00 - Earthmoving as					
5			necessary for undercut.					
5		50111						
/	3.2	EQUI	Wien I					
8 9		А.	Meet requirements of WisDOT Section 301.3.1.					
10	3.3	SPRE/	ADING AND SHAPING					
11		Α.	Meet requirements of WisDOT Section 305.3.4.					
12 13		В.	Construct to thickness indicated on Construction Drawings. The minimum base thickness as shown on Drawings shall be achieved throughout all pavement areas.					
14			 Aggregate Base: Apply in lifts or layers not exceeding 8 inches, measured loose. 					
15			2. Sand Base: Apply in lifts or layers not exceeding 6 inches, measured loose.					
16			3. Hot-Mix Sand Asphalt Bases: Apply in lifts or layers not exceeding 3 inches, measured loose.					
17								
18	3.4	сом	PACTION					
19		Α.	Meet requirements of WisDOT Section 305.3.2.2, except as modified below.					
20			1. Compact base material to not less than 98% of optimum density as determined by ASTM D698 or 95% of					
21			optimum density, as determined by ASTM D1557 unless otherwise indicated on the Drawings.					
22								
23	3.5	TOLE	RANCES					
24		Α.	Smoothness: Maximum variation of 3/8 inch when measured with a 10-foot straight edge					
25		В.	Compacted thickness: within 1/4 inch					
26								
27	3.6	FIELD	D QUALITY CONTROL					
28		Α.	Field inspection and testing will be done by the City/County in public right-of-way or the Owner's Construction					
29			Representative.					
30		В.	Repair or remove and replace unacceptable base course as directed by the Owner's Construction					
31			Representative.					
32								
33			END OF SECTION					
1					SECTION 32 12 16.13			
----------	--------------------------	------------	------------	------------------------	---	--	--	--
2	PLANT-MIX ASPHALT PAVING							
3								
4	PART 1 - GENERAL							
6	1.1	SUM	ЛАКУ					
7		A.	This	section includes hot	-mix asphalt paving, hot-mix asphalt patching, and pavement-marking paint.			
8		В.	Rela	ated sections include	:			
9			1.	Section 02 20 00 - G	General Sitework Requirements			
10			2.	Section 31 20 00 - E	arthmoving			
11			3.	Section 32 11 23 - A	ggregate Base Course			
12								
13	1.2	REFEF	RENCE	S				
14		Α.	AST	M International (AST	⁻ M):			
15			1.	ASTM D1188	Bulk Specific Gravity and Density of Compacted Bituminous Mixtures Using Coated			
16					Samples			
17			2.	ASTM D2041	Theoretical Maximum Specific Gravity and Density of Bituminous Paving Mixtures			
18			3.	ASTM D2950	Density of Bituminous Concrete in Place by the Nuclear Methods			
19			4.	ASTM D2726	Bulk Specific Gravity and Density of Non-Absorptive Compacted Bituminous Mixture			
20			5.	ASTM D5444	Mechanical Size Analysis of Extracted Aggregate			
21		В.	Ame	erican Association of	State Highway and Transportation Officials (AASHTO):			
22			1.	AASHTO M017	Mineral Filler for Bituminous Paving Mixtures			
23			2.	AASHTO M140	Emulsified Asphalt			
24			3.	AASHTO M208	Cationic Emulsified Asphalt			
25			4.	AASHTO M320	Performance-Graded Asphalt Binder			
26			5.	AASHTO M323	Superpave Volumetric Mix Design			
27			6.	AASHTO T164	Quantitative Extraction of Asphalt Binder from Hot-Mix Asphalt (HMA)			
28			7.	AASHTO T166	Bulk Specific Gravity of Compacted Hot-Mix Asphalt Mixtures Using Saturated			
29					Surface-Dry Specimens			
30			8.	AASHTO T209	Theoretical Maximum Specific Gravity and Density of Hot-Mix Asphalt (HMA)			
31			9.	AASHTO T245	Resistance to Plastic Flow of Bituminous Mixtures Using Marshall Apparatus			
32			10.	AASHTO T275	Bulk Specific Gravity of Compacted Hot-Mix Asphalt Mixtures Using Paraffin-Coated			
33					Specimens			
34			11.	AASHTO T308	Asphalt Content of Hot-Mix Asphalt (HMA) by the Ignition Method			
35			12.	AASHTOT312	Preparing and Determining the Density of Hot-Mix Asphalt (HMA) Specimens by			
36					Means of the Superpave Gyratory Compactor			
37			13.	AASHTO T331	Bulk Specific Gravity and Density of Compacted Bituminous Mixtures Using Automatic			
38					Vacuum Sealing Method			
39		C.	Nati	ional Asphalt Paveme	ent Association (NAPA):			
40			1.	IS 123 - Recycling	Hot-Mix Asphalt Pavements			
41			2.	IS 128 - HMA Pav	ement Mix Type Selection Guide			
42		~~ • • • •						
43	1.3	SIAN		SPECIFICATIONS				
44		А.	vvne	ere reference is mad	e to the Standard Specifications, it shall be construed to mean the pertinent section of			
45			the	vvisconsin Departme	dition and all supplemental and interim supplemental energifications for Road and Structure			
40			Con	struction, current e	union, and an supplemental and interim supplemental specifications, as they may			
47 10		р	peri Wh	an, except the metr	too of measurement and basis of payment shall not apply.			
40		ь.	Coot	tion 02 22 00	e to the dedictinical report, it shall be construed to mean the geotechnical report in			
49 E0		C	Seci	1011 UZ 32 UU.	a facilities and for work within public lands or rights of way shall conform to the			
50		C.		uiromonts and condit	tions of the City of Madican's "Standard Specifications for Public Works Construction"			
51			curr	cont odition				
52			curr	ent eution.				
55	1 /			I C				
55	T' 4			Mix Designs for each	a job mix proposed for the work			
56		R.	Mot	erial Certificates sig	ned by material producer and Contractor, certifying that each material item complies			
57		υ.	with	or exceeds snecifi	ed requirements			
58		C.	102	d Tickets shall he sub	pritted if payment is to be by tonnage.			
~~		. .	-00					

1	1.5	DEFIN	ITIONS				
2		A.	Plant-Mix Asphalt Paving Terminology: Refer to ASTM D8 for definitions of terms.				
5 4	16	SYSTEM DESCRIPTION					
5	1.0	Δ	Provide hot-mix asphalt paving according to materials, workmanshin, and other applicable requirements of				
6		7	standard specifications of WisDOT				
7							
8	1.7	QUALI	TY ASSURANCE				
9		Α.	Asphalt-Paving Publication: Comply with AI MS-22, "Construction of Hot-Mix Asphalt Pavements," unless more				
10			stringent requirements are indicated.				
11		В.	Pre installation Meeting: Convene a pre installation meeting at the site at least two (2) weeks prior to				
12			commencing work of this Section. Require attendance of parties directly affecting work of this Section,				
13			including, but not limited to, Developer, Developer's Engineer and Inspector, Contractor, paving sub-contractor,				
14			and job foreman.				
15			1. Contact Developer' Engineer and the City/County three (3) weeks prior to pre-installation conference to				
16			confirm schedule.				
17			2. Record discussions of meeting and decisions and agreements (or disagreements) reached, and furnish				
18			copy of record to each party attending. Review foreseeable methods and procedures related to paving				
19			work, including the following:				
20			a. Review preparation and installation procedures and coordinating and scheduling required with				
21			related work.				
22			b. Review proposed sources of paving materials, including capabilities and location of plant that will manufacture bet mix acabalt				
25			Tour inspect and discuss condition of subgrade, drainage structures, and other preparatory work				
24			d Review requirements for protecting paying work including restriction of traffic during installation				
26			neriod and for remainder of construction period				
27			e. Review and finalize construction schedule and verify availability of materials, installer's personnel.				
28			equipment, and facilities needed to make progress and avoid delays.				
29			f. Review paying requirements (drawings, specifications, and other contract documents).				
30			g. Review required submittals, both completed and yet to be completed.				
31			h. Review required inspections, testing procedures.				
32			i. Review weather and forecasted weather conditions, and procedures for coping with unfavorable				
33			conditions.				
34			j. Review safety precautions relating to placement of paving.				
35							
36	1.8	DELIVE	ERY, STORAGE, AND HANDLING				
37		Α.	Deliver pavement-marking materials to Project site in original packages with seals unbroken and bearing				
38			Manufacturer's labels containing brand name and type of material, date of manufacture, and directions for				
39		_	storage.				
40		В.	Store pavement-marking materials in a clean, dry, protected location within temperature range required by				
41			Manufacturer. Protect stored materials from direct sunlight.				
4Z 12	10						
45 11	1.9		Environmental Limitations: Do not apply asphalt materials if subgrade is wet or excessively damp				
45		л. В	Paving under conditions when surface temperatures approximately 3 feet above grade are at or below				
46		5.	40 degrees E (40°E) shall be in accordance with Cold Weather Paving provisions in the State Specifications—				
47			specifically Section 450.				
48		C.	Pavement-Marking Paint: Proceed with pavement marking only on clean, dry surfaces and at a minimum				
49			ambient or surface temperature of 40 degrees F (40°F) for oil-based materials, 50 degrees F (50°F) for water-				
50			based materials, and not exceeding 95 degrees F (95°F). Paint color shall be as specified on the Drawings.				
51							
52	PART	2 - PROI	DUCTS				
53							
54	2.1	GENEF	RAL				
55		Α.	All materials and methods for on-site pavement shall conform to WisDOT Standard Specifications. Where				
56			conflicts between this specification and the WisDOT Standard Specifications exist, requirements of the WisDOT				
57			shall govern.				

1 2 3		В.	All materials and methods for public roadways shall conform to City of Madison's "Standard Specifications for Public Works Construction," current edition.					
4	2.2	MATE	MATERIALS AND MIXES					
5 6 7		A.	Provide asphaltic pavement per WisDOT Standard Specifications, Sections 460.2 and 460.3, and the Pavement Design section of the Geotechnical Report, but excluding limitations in Section 460.3.2 restricting layer thickness by aggregate size.					
8 9		В.	Pavement thickness: See pavement sections in Drawings: 1. Bituminous Concrete: Refer to Section 460-3					
10			2. Base Course: Refer to Section 301.3.4.2 - Standard Compaction					
11		C.	Mixture Type: See Drawing sections; Table 460-2 of the WisDOT Standard Specifications.					
12		D.	Bituminous Material: Per WisDOT Standard Specifications—of suitable grade and consistency for application.					
13		Ε.	Tack Coat: Per WisDOT Standard Specifications—of suitable grade and consistency for application.					
14		F.	Water: Potable.					
15								
16	2.3	MARK	ING MATERIALS					
17		А.	Type S or Type N Traffic Paint in accordance with AASHTO Designation M248. Regular set drying time.					
18		В.	Waterborne Paint: Paints shall conform to FS TT-P-1952.					
19		С.	Solvent Borne Paint: Paint shall conform to FS A-A-2886 or AASHTO M248. Paint shall be non-bleeding, quick					
20			drying, and alkyd petroleum base paint suitable for traffic bearing surface and be mixed in accordance with					
21		_	manufacturer's instructions before application for colors White, Yellow, Blue, and Red.					
22		D.	Epoxy marking from the WisDOT approved products list and in accordance with Section 646.2.4.					
23		E.	Glass Beads: AASHTO M 247, Type 1 or FS TT-B-1325, Type 1, Gradation A					
24 25	PART 3	B - EXEC	CUTION					
26								
27	3.1	EXAM	INATION					
28		Α.	Verify that subgrade is dry and in suitable condition to support paving and imposed loads.					
29		В.	Proof-roll subbase using heavy, pneumatic-tired rollers to locate areas that are unstable or that require further					
30			compaction per Geotechnical Report.					
31 32		C.	Proceed with paving only after unsatisfactory conditions have been corrected.					
33	3.2	PATCH	ling					
34		Α.	Hot-Mix Asphalt Pavement: Sawcut perimeter of patch and excavate existing pavement section to sound base.					
35			Excavate rectangular or trapezoidal patches, extending 12 inches into adjacent sound pavement, unless					
36			otherwise indicated. Cut excavation faces vertically. Remove excavated material. Recompact existing unbound-					
37			aggregate base course to form new subgrade.					
38		В.	Tack Coat: Apply uniformly to vertical surfaces abutting or projecting into new, hot-mix asphalt paving at a rate					
39			of 0.05 to 0.15 gal./sq. yd.					
40			 Allow tack coat to cure undisturbed before applying hot-mix asphalt paving. 					
41			2. Avoid smearing or staining adjoining surfaces, appurtenances, and surroundings. Remove spillages and					
42			clean affected surfaces.					
43		С.	Patching: Partially fill excavated pavements with hot-mix asphalt base mix and, while still hot, compact. Cover					
44			asphalt base course with compacted, hot-mix surface layer finished flush with adjacent surfaces.					
45								
46	3.3	SURFA	ACE PREPARATION					
47		Α.	General: Immediately before placing asphalt materials, remove loose and deleterious material from substrate					
48			surfaces. Ensure that prepared subgrade is ready to receive paving.					
49			1. Sweep loose granular particles from surface of unbound-aggregate base course. Do not dislodge or					
50			disturb aggregate embedded in compacted surface of base course.					
51		в.	Tack coat: Apply uniformity to surfaces of existing pavement at a rate of 0.05 to 0.15 gal./sq. yd.					
52 52			Anow tack coat to cure undisturbed before applying not-mix aspnait paving. Avoid empering or steining edicining surfaces applying not-mix aspnait paving.					
55			 Avoid smearing or staming adjoining surfaces, appurtenances, and surroundings. Remove spillages and clean affected surfaces. 					
J4			נוכמון מווכנובע אנוומנבא.					

1	3.4	HOT-N	-MIX ASPHALT PLACING				
2 3		A.	Machine place hot-mix asphalt on prepared surface, spread uniformly, and strike off. Place asphalt mix by hand to areas inaccessible to equipment in a manner that prevents segregation of mix. Place each course to required				
4			grade, cross section, and thickness when compacted.				
5			 Place hot-mix asphalt binder course in number of lifts and thicknesses indicated. Caused mix at minimum temperature of 200 degrees 5 (200°5) 				
6 7			 Spread mix at minimum temperature of 250 degrees F (250 F). Bogin applying mix along contacting of grown for growned sections and on high side of one way slongs. 				
8			unless otherwise indicated				
9			4 Regulate naver machine speed to obtain smooth continuous surface free of nulls and tears in asnhalt-				
10			paving mat.				
11		В.	Place paving in consecutive strips not less than 10 feet wide unless in-fill edge strips of a lesser width are				
12			required.				
13			1. After first strip has been placed and rolled, place succeeding strips and extend rolling to overlap previous				
14			strips. Complete a section of asphalt binder course before placing asphalt surface course.				
15		С.	Promptly correct surface irregularities in paving course behind paver. Use suitable hand tools to remove excess				
16			material forming high spots. Fill depressions with hot-mix asphalt to prevent segregation of mix; use suitable				
17			hand tools to smooth surface.				
18	э г						
20	5.5		Construct joints to ensure a continuous bond between adjoining paying sections. Construct joints free of				
20		Λ.	depressions with same texture and smoothness as other sections of hot-mix asphalt course.				
22			1. Clean contact surfaces and apply tack coat to joints.				
23			 Offset longitudinal joints, in successive courses, a minimum of 6 inches. Coat longitudinal joints that are 				
24			not completed before the previously laid mixture has cooled to a temperature below 140 degrees F				
25			(140°F), with liquid asphalt just before paving is continued.				
26			3. Offset transverse joints, in successive courses, a minimum of 24 inches. If placing of material is				
27			discontinued or if material in place becomes cold, make a joint running perpendicular to the direction				
28			traveled by the paver. Before placement continues, trim the edge of the previously placed pavement to a				
29			straight line perpendicular to the paver and cut back to expose an even vertical surface for the full				
30			thickness of the course. When placement continues, position the paver on the transverse joint so that				
31			sufficient not mixture will be spread in order to create a joint after rolling that conforms to the required				
32 33			(140°E) before paying is resumed, give the exposed vertical face a thin cost of liquid asphalt just before				
34			naving is continued				
35			 Construct transverse joints as described in AI MS-22. "Construction of Hot Mix Asphalt Pavements." 				
36			5. Compact joints as soon as hot-mix asphalt will bear roller weight without excessive displacement.				
37			6. Compact asphalt at joints to a density within 2% of specified course density.				
38		В.	Construction joints shall have same texture, density, and smoothness as other sections of asphalt concrete				
39			course.				
40							
41	3.6	COMP					
42		А.	General: Begin compaction as soon as placed hot-mix paving will bear roller weight without excessive				
43			inaccossible to rollors				
44 15			1 Complete compaction before mix temperature cools to 185 degrees F (185°F)				
46		В.	Breakdown Rolling: Complete breakdown or initial rolling immediately after rolling joints and outside edge.				
47			Examine surface immediately after breakdown rolling for indicated crown, grade, and smoothness. Correct				
48			laydown and rolling operations to comply with requirements.				
49		C.	Intermediate Rolling: Begin intermediate rolling immediately after breakdown rolling while hot-mix asphalt is				
50			still hot enough to achieve specified density. Continue rolling until hot-mix asphalt course has been uniformly				
51			compacted to the following density:				
52			1. Average Density: 96% of reference laboratory density according to AASHTO T 245, but not less than 94%				
53			nor greater than 100%.				
54			2. Average Density: 92% of reference maximum theoretical density according to ASTM D2041, but not less				
55		D	than 92% hor greater than 97%.				
50 57		D. F	Finish Kolling: Finish foll paved surfaces to remove foller marks while not-mix asphalt is still warm. Edge Shaning: While surface is being compacted and finished, trim edges of navement to proper alignment.				
58		L.	Bevel edges while asphalt is still hot; compact thoroughly.				

1 2 3 4 5 6 7 8		F. G. H.	Protection: After final rolling, do not permit vehicular traffic on pavement until it has cooled and hardened. Erect barricades to protect paving from traffic until mixture has cooled enough not to become marked. Any masked or marred finish surfaces shall be repaired or smoothed. Compaction at Unsupported Edges of Pavements: Start the first roller pass 12 to 15 inches from the unsupported edge. Allow the uncompacted asphalt to act as a dike to hold the mat in place. The final pass over the uncompacted dike should not slough off if the roller is supported on the compacted mat. Repairs: Remove paved areas that are defective or contaminated with foreign materials and replace with fresh, hot-mix asphalt. Compact by rolling to specified density and surface smoothness.
9 10 11		ı. J.	Erect barricades to protect paving from traffic until mixture has cooled enough not to become marked.
12	3.7	INSTAL	LATION TOLERANCE
13	-	Α.	Thickness: Compact each course to produce the thickness indicated within the following tolerances:
14			1. Binder Course: Plus or minus (±) 1/4 inch.
15			2. Surface Course: Plus (+) 1/4 inch, no minus (-).
16		В.	Surface Smoothness: Compact each course to produce a surface smoothness within the following tolerances as
17			determined by using a 10-foot straightedge applied transversely or longitudinally to paved areas:
18			1. Binder Course: 1/4 inch
19			2. Surface Course: 1/8 inch
20			3. Crowned Surfaces: Test with crowned template centered and at right angle to crown. Maximum
21			allowable variance from template is 1/4 inch.
22			
23	3.8	FIELD	QUALITY CONTROL
24		А.	resting Agency: The Owner's construction Representative will engage a testing agency to perform field tests
25			and inspections and to prepare test reports.
20			with or deviates from specified requirements
28		в	Additional testing and inspecting at Contractor's expense will be performed to determine compliance of
29		5.	replaced or additional work with specified requirements.
30		C.	Thickness:
31		-	1. In-place compacted thickness of hot-mix asphalt courses will be determined according to ASTM D3549.
32			2. The CTL will measure thickness of each core sample taken. At each core location, the thickness of the
33			course shall meet or exceed the thickness shown. If the thickness of a lower course of asphalt is less than
34			the thickness shown, it shall be identified as a deviation and recorded. The Contractor shall either
35			remove and replace the deficient pavement or increase the thickness of the upper course so that the
36			total thickness of the pavement meets or exceeds the design thickness, provided that the specified
37			compaction of the lower lift is achieved.
38		D.	Surface Smoothness: Finished surface of each hot-mix asphalt course will be tested for compliance with
39			smoothness tolerances. In areas of obvious depressions or bumps, suspect areas of each lift shall be checked
40			with a 10'-0" straightedge both parallel with, and at right angles to, centerline of the paved area. The variation
41		E	of the surface between two (2) contact points shall not exceed 1/4 inch.
42 //3		с.	does not comply with specified requirements
43			ades not comply with specifica requirements.
45	3.9	SPLIT P	PAVING
46		A.	Where specified, paving of the binder course and surface course are to be performed in same construction
47			seasons.
48		В.	If paving of binder and surface course are done in successive construction season, the Contractor shall be
49			responsible for damage to the binder course and curbs until the surface course is placed and accepted by the
50			Owner as appropriate.
51			
52	3.10	DISPOS	SAL
53		A.	Except for material indicated to be recycled, remove excavated materials from Project site and legally dispose of
54		D	them in an EPA-approved landfill.
55		в.	Do not allow excavated materials to accumulate on-site.
טכ 57	2 1 1	MADV	ING ADDI ICATION DAINT
52	3.11		Devement markings shall be placed at locations and to dimensions shown on Drawings
50			. a content manine brance at recentlers and to amendors shown on pravillasi

1		В.	Applied lines shall have a uniform cross section.			
2		C.	Lines shall have sharp cut-off defined edges on both side and ends.			
3		D.	Pavement markings applied to new asphaltic pavement surfaces shall be applied to surface course within seven			
4			(7) days of placement.			
5		Ε.	Pavement markings applied to new concrete surfaces shall be applied prior to allowance of any traffic on			
6			surface.			
7		F.	Agitate paint for 5-10 minutes prior to application to ensure even distribution of paint pigment.			
8		G.	Paint shall be applied in accordance with manufacturer's recommendations and apply two coats of same color of			
9			paint as specified without addition of thinner, with maximum of 100 square feet per gallon or as required to			
10			provide a minimum wet film thickness of 15 mils and dry film thickness of 7½ mils per coat.			
11		Н.	Dispense paint at ambient degrees Fahrenheit to wet-film thickness of 15 mils.			
12		Ι.	Apply markings to indicated dimensions at indicated locations.			
13		J.	Prevent splattering and over spray when applying markings.			
14		К.	Apply glass beads at pedestrian crosswalk striping and at lane striping and arrows at driveways connecting to			
15			public streets. Broadcast glass beads uniformly into wet markings at a rate of 6 lbs./gal.			
16		L.	Unless material is track free at end of paint application convoy, use traffic cones to protect markings from traffic			
17			until track free. When vehicle crosses a marking and tracks it or when splattering or over spray occurs, eradicate			
18			affected marking and resultant tracking and apply new markings.			
19		M.	Collect and legally dispose of residues from painting operations.			
20						
21	3.12	MARKING APPLICATION - EPOXY				
22		Α.	Application shall be in accordance with Section 646.3.3.2 of the State Specifications.			
23						
24			END OF SECTION			

END OF S	ECTION
----------	--------

1 2					SECTION 32 16 00 CONCRETE PAVEMENT, CURB, AND SIDEWALKS
3 4	<u>PART</u>	1 - GEN	IERAL		
5	1 1	SUM			
7	1.1	A.	Sectio	on includes:	
8			1.	Portland cemei	nt concrete pavements, curbs, gutters, and sidewalks, except sidewalks on structural
9			Delet	footings and st	oops and/or specified by the Architectural or Structural Drawings.
10		в.	Relate	ed requirements:	10. Canaral Sitawark Dequirements
11			1. ว	Section 02 20 0	0 - General Sitework Requirements
12			2. 2	Section 21 20 0	0 - Geolecinical Investigation
15			5. ⊿	Section 31 22 0	6 15 - Subarade Prenaration
14			4. 5	Section 32 11 2	13 - Aggregate Base Course
16			Э.	50011 52 11 2	J - Aggregate base course
17	1.2	REFE	RENCES		
18		Α.	The p	ublications listed	below form a part of this specification to the extent referenced. Publications are
19			refere	enced within the	text by the basic designation only.
20		В.	Ameri	ican Concrete Ins	stitute (ACI):
21			1.	ACI 305R	Hot Weather Concreting
22			2.	ACI 306R	Cold Weather Concreting
23			3.	ACI 306.1	Cold Weather Concreting
24			4.	ACI 308	Curing Concrete
25		C.	ASTM	International (A	STM):
26			1.	ASTM A185	Steel Welded Wire Fabric, Plain, for Concrete Reinforcement
27			2.	ASTM A615	Deformed and Plain Billet-Steel for Concrete Reinforcement
28			3.	ASTM C31	Making and Curing Concrete Test Specimens in the Field
29			4.	ASTM C39	Comprehensive Strength of Cylindrical Concrete Specimens
30			5.	ASTM C42	Obtaining and Testing Drilled Cores and Sawed Beams of Concrete
31			6.	ASTM C94	Ready-Mixed Concrete
32			7.	ASTM C138	Test Method for Unit Weight, Yield, and Air Content (Gravemetric) of Concrete
33			8.	ASTM C143	Slump of Hydraulic Cement Concrete
34			9.	ASTM C231	Air-Content of Freshly Mixed Concrete by the Pressure Method
35			10.	ASTM C172	Sampling Freshly Mixed Concrete
36			11.	ASTM C173	Test Method for Air Content of Freshly Mixed Concrete by the Volumetric Method
37			12.	ASTM C260	Air-Entraining Admixtures for Concrete
38			13.	ASTM C309	Liquid Membrane-Forming Compounds for Curing Concrete
39			14.	ASTM C618	Fly Ash and Raw or Calcined Natural Pozzolan for use as a Mineral Admixture in Portland
40			45		Cement Concrete Commit Consults of Plant Function Class for they in Committee and Mantaux
41			15.	ASTM C989	Ground Granulated Blast-Furnace Slag for Use in Concrete and Mortars
42			10.	ASTIVI C1064	Temperature of Freshly Mixed Portland Concrete Cement
43			10		Coloium Chlorida
44 45			10.		Calcium Chionae Droformod Evpansion Joint Filler for Congrete (Pitumineus)
45			19. 20	ASTNI D334	Concrete Joint Scalar, Het Doured, Electic Type
40			20.	ASTM D1150	Performed Expansion Joint Fillers for Concrete Daving and Structural Construction (Non
47			21.	ASTIVI D1751	extruding and Resilient Rituminous Types)
40 49			22	ASTM D2628	Preformed Polychloronrene Flastomeric Joint Seals for Concrete Pavements
50		D	Eeder	al Specifications	(FS)·
50		υ.	1	FS HH-F-341	Fillers Expansion Joint: Bituminous (Asphalt and Tar)
52		F	T. Wisco	nsin Departmen	t of Transportation (WisDOT).
52		_ .	1	State of Wiscor	nsin Standard Specifications for Highway and Structure Construction latest edition
54					
55	1.3	SUBN	1ITTALS		
56	-	Α.	Mix D	esign:	
57			1.	Fill out and sub	mit attached Concrete Mix Design Submittal Form.
58			2.	Submit three (3	3) copies of each proposed mix.

CITY C SPECII <u>May 1</u>)F MADIS FICATION .6, 2024	SON I
		3 Submit separate mix design for concrete to be placed by numping in addition to the mix design for
		concrete to be placed directly from the truck chute
		4 Include applicable information shown on the Mix Design Submittal Form and the following:
		a Proportions of cementitious materials fine and coarse aggregate and water
		b. Water-cementitious material ratio. 28-day compressive design strength, slump, and air content.
		c. Type of cement, fly ash, slag and aggregate.
		d. Aggregate gradation.
		e. Type and dosage of admixtures.
		f. Special requirements for pumping.
		g. Range of ambient temperature and humidity for which design is valid.
		h. Special characteristics of mix which require precautions in mixing, placing, or finishing techniques
		to achieve finished product specified.
		i. Materials and methods for curing concrete.
	В.	Submit certified laboratory test data or manufacturer's certificates and data for the items listed below certifying
		that materials are in conformance requirements specified herein. Submit to the Engineering Consultant of
		Record and the Construction Testing Laboratory for review and approval and within seven calendar days after
		receipt of Notice to Proceed.
		1. Concrete mix design(s)
		2. Type and source of Portland cement, fly ash, and slag
		3. Aggregate gradations
		4. Preformed expansion joint filler
		5. Field molded/poured sealant
		6. Dowel bars
		7. Expansion sieeves
		8. The bars
		9. Reinforcing steel bars
		10. Welded will'e labitc
		12. Mater reducing set retarding and set accelerating admixtures (if used)
	C	Test Reports: Submit field quality control test reports
	с.	rest reports. Submit neu quanty control test reports.
1.4	PROJ	ECT CONDITIONS
	Α.	Maintain access for vehicular and pedestrian traffic as required for other construction activities. Utilize
		temporary striping, flagmen, barricades, warning signs, and warning lights as required.
PART	2 - PRC	DUCTS
2.1	MAT	ERIALS
	Α.	Forms shall be of wood or metal and shall be straight and of sufficient strength to resist springing, tipping or
		other displacement during the process of depositing and consolidating the concrete. If of wood, forms shall be
		surfaced plank of at least 2-inch nominal thickness stock except for sharply curved sections; and if of metal, they
		shall be of approved section. The forms shall be of the full depth of the required curb or curb and gutter,
		driveway or sidewalk sections and shall be designed to permit secure fastening. Face boards, if used, shall be so
		constructed and shaped that their lower edge conforms to the lines and radius indicated by the cross section for
		the pertinent structure as shown on the plans. Flexible or curved forms of proper radius shall be used for curves
		of 100-foot radius or less. All forms shall be cleaned thoroughly and oiled before the concrete is placed against
		them. Michael Million Marcha Michael a later and discuss standards folicity ACTNA AAOE and AAOUTO MOD Conds CO. Examini
	в.	Welded Wire Mesh: Welded plain cold-drawn steel wire fabric, ASTM A185 and AASHTO M31 Grade 60. Furnish
	6	In flat sneets.
	C.	Reinforcing Steel: Deformed steel bars, ASTM A615, Grade 60.
	U. E	Epuxy Coalling. ASTIVEA775 WHELE HOLED III the Didwings. Bartland Compart: Shall conform to ASTM C150. Typo Lunloss Typo III is specified
	с. F	For dana Cement. Shall conform to ASTIVECTOD, Type Fulliess Type III is specified. Exterior Devement Joint Materials:
	г.	LALEHOL FAVEHIEH JUHIL MIDLEHIDS. 1 Preformed Expansion Joint Filler shall be 3/4 inch in thickness and shall conform to AASUTO M 212
		Inint Rack-up Material: Polyethylene foam 100% closed cell
		3 Sealant
		a. Dow 888. by Dow Corning
		b. 301 NS by Pecora
		,

1			c. Spectrum 800 or 900 by Tremco
2		G.	Aggregate: ASTM C33
3		Н.	Water: Clean and potable
4		I.	Dowel Bars: ASTM A615, Grade 60, and plain steel bars
5		J.	Air Entrainment:
6			1. Air entrained concrete shall be used for all concrete construction. Slip formed concrete pavement shall
7			contain 7.0% air +1.5%. Other concrete shall contain 6.0% air +1.5%
, Q			2 ASTM (260)
0			2. Astimiczowa Alexandra (Company) and Alexandra (Comp
9			a. All-Ivitx of AEA-32, by Euclid Chemical Colp.
10			D. MB-VR MB-AE 90, OF MICTO-AIT, BY BASE
11			c. Daravair or Darex Series, by W.R. Grace
12			d. Equivalent approved products
13		К.	Liquid Membrane Curing and Sealing Compound: ASTM C1315, Type I, Class A or Class B, 25% minimum solids
14			content, clear non-yellowing with no styrene-butadiene. Specifications for Liquid Membrane-Forming
15			Compounds for Curing Concrete, AASHTO M-148, Type 2 shall also apply if more stringent.
16			1. Water Based, VOC less than 350 g/l:
17			a. Super Aqua Cure, by Euclid Chemical Corp.
18			b. Kure 1315 by BASF
19			2. Solvent Based:
20			a. Super Rez-Seal, by Euclid Chemical Corp.
21			b. Kure-N-Seal 30 by BASE
22		1	Polyethylene Sheeting: Polyethylene sheeting for curing concrete shall conform to the requirements for white
22		L.	on aque nolverbylene film of the Snecification for Sheet Materials for Curing Concrete, AASHTO M.171
23		N.4	Synthetic Pointerroment: Neverals 060 Marco Synthetic Elber Blord or aquivalant, complying with ACTM
24		101.	Sinterio cententent. Novomesi 550 Marco Sinterio Fiber Biend, or equivalent, complying with Astron
25			CIIIO/CIIIO/A, Type III, Fiber Reinforced Concrete.
26			
27	2.2	CONCI	
28		А.	Mix concrete and deliver in accordance with ASTM C94. Design mix shall produce normal weight concrete
29			consisting of Portland cement, supplementary cementitious materials, aggregates, admixtures and water to
30			produce the following:
31			1. Compressive Strength: 3,500 psi minimum at 28 days unless otherwise indicated on the Drawings.
32			2. Slump Range: 2 to 4 inches for hand placed concrete, 1¼ to 3 inches for machine placed (slip-form)
33			concrete.
34		В.	Supplementary Cementitious Materials (SCM):
35			1. Concrete mix shall contain SCM at the amounts specified unless other amounts are approved by the
36			Owner's Civil Engineer. Either fly ash or ground granulated blast furnace slag (GGBFS) may be used for
37			the SCM, but shall not be used together to form a ternary mix. Use of fly ash or GGBES in the concrete
38			mix is mandatory.
39			2 Fly Ach
10			Substitute fly ash for Portland cament at 15% of the total camentitious content ASTM C618
40			a. Substruct in astron Portand cement at 15% of the total cementations content. Astro-Color,
41			Liass C.
42			b. Use only one (1) type and source throughout project.
43			3. Ground Granulated Blast Furnace Slag (GGBFS): Substitute GGBFS for Portiand cement at 20% of the total
44			cementitious content.
45			a. If required to mitigate potential sulfate exposure or aggregate reactivity, up to 50% substitution
46			of Portland cement is allowed.
47			b. ASTM C989, Grade 100 or 120. Use only one (1) type and source throughout project.
48			4. Maintain air-entrainment at specified levels.
49		C.	Calcium chloride:
50			1. Not allowed.
51			
52	PART :	<u>3 - EXE</u> C	CUTION
53			
54	3.1	PREPA	RATION
55		Α.	Begin paving work only after unsuitable areas have been corrected and are ready to receive paving
56		В	Remove loose material from compacted base material surface to produce firm smooth surface immediately
57		5.	hefore placing concrete
58			
50			

1	3.2	INSTA	STALLATION			
2		Α.	Form Construction:			
3			 Set forms to required grades and lines, rigidly braced and secured. 			
4			2. Install sufficient quantity of forms to allow continuance of work and so that forms remain in place			
5			minimum of 24 hours after concrete placement.			
6			Check completed formwork for grade and alignment to following tolerances:			
7			a. Top of forms not more than 1/8 inch in 10'-0".			
8			b. Vertical face on longitudinal axis, not more than 1/4 inch in 10'-0".			
9			4. Clean forms after each use and coat with form release agent as often as required to ensure separation			
10			from concrete without damage.			
11		В.	Reinforcement:			
12			a. Fasten reinforcing bars or welded wire fabric (if required) accurately and securely in place with			
13			suitable supports and ties. Remove from reinforcement all dirt, oil, loose mill scale, rust, and			
14			other substances that will prevent proper bonding of the concrete to the reinforcement.			
15			b. Incorporate synthetic fiber reinforcement into the concrete mix design, as recommended by the			
16			manufacture's specifications and guidelines.			
17		C.	Concrete Placement:			
18			1. Concrete shall be mixed and placed when the air temperature in the shade and away from artificial heat			
19			is a minimum of 35 degrees F (35°F) and rising.			
20			2. Hot and cold weather concreting shall be in accordance with ACI 305R (hot weather) and 306.1 and 306R			
21			(cold weather). Do not place concrete until base material and forms have been checked for line and			
22			grade. Moisten base material if required to provide uniform dampened condition at time concrete is			
23			placed. Do not place concrete around manholes or other structures until set at required finish elevation			
24			and alignment.			
25			3. Place concrete using methods that prevent segregation of mix. Consolidate concrete along face of forms			
26			and adjacent to transverse joints with internal vibrator. Keep vibrator away from joint assemblies,			
27			reinforcement, or side forms. Consolidate with care to prevent dislocation of reinforcing, dowels, and			
28			joint devices.			
29			4. Deposit and spread concrete in continuous operation between transverse joints, as far as possible. If			
30			interrupted for more than a half hour, place construction joint. Automatic machine may be used for curb			
31			and gutter placement. Machine placement shall be at required cross section, line, grade, finish, and			
32			jointing as specified for formed concrete. If results are not acceptable, remove and replace with formed			
33			concrete as specified herein.			
34		D.	Joint Construction:			
35			1. Contraction Joints: Construct concrete curb or combination concrete curb and gutter, where specified on			
36			Construction Drawings, in uniform sections of length specified on Construction Drawings. Form joints			
37			between sections either by steel templates, 1/8 inch in thickness, of length equal to width of curb and			
38			gutter, and with depth which will penetrate at least 2 inches below surface of curb and gutter; or with			
39			3/4-inch thick performed expansion joint filler cut to exact cross section of curb and gutter; or by sawing			
40			to depth of at least 1/4 (one-quarter) of the poured thickness of concrete while concrete is between 4			
41			and 24 hours old. If steel templates are used, they shall be left in place until concrete has set enough to			
42			hold its shape, but shall be removed while forms are still in place. Contraction joints in concrete curb and			
43			gutter shall be at a maximum spacing of 10 feet. Contraction joints in concrete sidewalk or pavement			
44			shall be at spaced at a length to width ratio not exceeding 1.5:1, with no dimension greater than 15 feet			
45			unless approved by the Engineer.			
46			2. Longitudinal Construction Joints: Tie concrete curb or combination concrete curb and gutter, where			
47			specified on Construction Drawings, to concrete pavement with 1/2-inch round deformed reinforcement			
48			bars of length and spacing shown on Construction Drawings.			
49			3. Transverse Expansion Joints: Concrete curb, combination concrete curb and gutter, or concrete sidewalk			
50			shall have filler cut to exact cross section of curb, gutter, or sidewalk. Joints shall be similar to type of			
51			expansion joint used in adjacent pavement. Expansion joints to be located at high points, utility			
52			structures, curb returns, cold joints, or 100-foot maximum spacing.			
53		E.	Joint Fillers: Extend joint fillers full-width and depth of joint, and not less than 1/2 inch or more than 1 inch			
54			below finished surface where joint sealer is indicated. Furnish joint fillers in one piece lengths for full width			
55			being placed, wherever possible. Where more than one length is required, lace or clip joint filler sections			
56			together.			
57		F.	Joint Sealants: Install in accordance with manufacturer's recommendations.			
58						

1	3.3	CONC	CRETE FINISHING						
2		Α.	After striking off and consolidating concrete, smooth surface by screeding and floating. Adjust floating to						
3			compact surface and produce uniform texture. After floating, test surface for trueness with 10'-0" straightedge.						
4			Distribute concrete as required to remove surface irregularities, and refloat repaired areas to provide continuous						
5			smooth finish.						
6		В.	Work edges of sidewalks, gutters, back top edge of curb, and formed joints with edging tool, rounding edge to						
7			1/2-inch radius. Eliminate tool marks on concrete surface. After completion of floating and troweling excess						
8			moisture or surface sheen has disappeared, complete surface finishing as follows:						
9			1. Curbs, gutters, and sidewalks: Broom finish by drawing fine-hair broom across surface perpendicular to						
10			flow of traffic. Repeat operation as necessary to produce fine line texture.						
11			2. Concrete Pavement: Broom finish by drawing medium-hair broom across surface parallel to direction of						
12			vehicle traffic. Repeat operation as necessary to produce even textured finish.						
13		C.	Do not remove forms for 24 hours after concrete has been placed. After form removal, clean ends of joints and						
14			point up minor honeycombed areas. Remove and replace areas or sections with major defects as directed by						
15		_	Developer.						
16		D.	Check surface areas at intervals necessary to eliminate ponding areas. Remove and replace unacceptable work						
1/			as directed by Developer.						
18	2.4	NICI							
19	3.4	NIGH	Concreting energians shall be discontinued due to insufficient natural light jupless on adequate and energy of						
20 21		А.	concreting operations shall be discontinued due to insumclent natural light, unless an adequate and approved						
21 22			al tilicial lighting system is provided and operated.						
22 23	35	CURI	NG AND PROTECTION						
23 24	5.5		Protect and cure finished concrete paying using curing compound or with acceptable moist-curing methods in						
25		7	accordance with "water-curing" section of ACI 308. Cure for a period not less than seven days.						
26		В.	Use solvent based curing compound when compound is applied below 40 degrees F (40°F).						
27									
28	3.6	ВАСК	(FILL						
29		Α.	After concrete has set sufficiently, spaces on either side of concrete pavements, by curb, combination concrete						
30			curb and gutter, or concrete sidewalk shall be refilled to required elevation with suitable material compacted in						
31			accordance with WisDOT Standard Specifications Section 02300.						
32									
33	3.7	CLEA	NING AND PROTECTION						
34		Α.	Sweep concrete pavement and wash free of stains, discolorations, dirt, and other foreign material just prior to						
35			final inspection.						
36		В.	Protect concrete from damage until acceptance of work. Exclude traffic from pavement for at least 14 days after						
37			placement. When construction traffic is permitted, maintain pavement as clean as possible by removing surface						
38			stains and spillage of materials.						
39									
40	3.8	FIELD	QUALITY CONTROL						
41		Α.	Field quality control shall be the responsibility of the Owner's Construction Representative in accordance with						
4Z 42			Division 1 and this section. Uther field quality control testing and inspection shall be at the discretion of the						
43 11			Contractor as necessary to assure compliance with Contract requirements.						
44 15			END OF SECTION						
-5									

Page Intentionally Left Blank

1 2			SECTION 32 17 23 PAVEMENT MARKINGS			
3 4 5	PART 1 - GENERAL					
6	1.1	SECTIC	DN INCLUDES			
7		Α.	Traffic line paint and thermoplastic material for traffic striping and marking.			
8		В.	Application of traffic striping and control markings.			
9						
10	1.2	RELAT	ED SECTIONS			
11		А.	Asphalt pavement is specified in Section 31 22 16.15 - Subgrade Preparation and Section 32 11 23 - Aggregate			
12		Б	Base Course.			
13		в.	Concrete pavement is specified in Section 32 16 00 - Concrete Pavement, Curb, and Sidewalks.			
14 15	1 3	ΜΕΔ	UREMENT AND PAYMENT			
16	1.5	A.	Measurement: Pavement markings will be measured for payment by the lump sum method, acceptably			
17			performed and completed.			
18		В.	Payment: Pavement markings will be paid for at the indicated Contract lump sum price, as indicated in the Bid			
19			Schedule of the Bid Form.			
20						
21	1.4	REFER	ENCES			
22		А.	State of California, Department of Transportation (Caltrans), Standard Specifications, latest edition:			
23			1. Section 84: Traffic Stripes and Pavement Markings			
24 25		в.	California Air Resources Board (CARB):			
25			1. Latest regulations governing permissible content of volatile organic compounds (voc) in paints			
20	1.5	SUBM	ITTALS			
28		A.	General: Refer to Section 01 33 23 - Submittals for submittal requirements and procedures.			
29		В.	Shop Drawings: Submit drawings and diagrams, indicating stripe width of roadway divider stripes and parking			
30			stalls, configuration and dimensions of directional arrows, style and size of letters for "compact car" designation,			
31			configuration and dimensions of international handicapped symbol, and any other traffic control markings on			
32			pavement, such as "in" and "out" or "enter" and "exit" designations as indicated.			
33		C.	Certificate of Compliance: Submit evidence or affidavit that certifies that paint to be used complies with latest			
34 25			CARB/VOC regulations.			
36	DART		חורדא			
37						
38	2.1	MATE	RIALS			
39		Α.	Type S or Type N traffic paint in accordance with AASHTO Designation M248. Regular set drying time.			
40		В.	Waterborne Paint: Paints shall conform to FS TT-P-1952.			
41		С.	Solvent Borne Paint: Paint shall conform to FS A-A-2886 or AASHTO M248. Paint shall be non-bleeding, quick			
42			drying, and alkyd petroleum base paint suitable for traffic bearing surface and be mixed in accordance with			
43		_	manufacturer's instructions before application for colors of white, yellow, blue, and red.			
44		D.	Epoxy marking from the WisDUT approved products list and in accordance with Section 646.2.4.			
45 46		E.	Glass Beads: AASHTO M 247, Type 1 of FS TT-B-1325, Type 1, Gradation A.			
40 47	22	ΔΡΡΙΙά	CATION			
48	2.2	ALLER	Apply no paint or thermoplastic material until pavement has cured for at least three (3) days or for the number			
49			of days as recommended by the manufacturer, whichever is longer. Ensure that pavement has cured sufficiently			
50			to carry application equipment without damage.			
51		В.	Provide traffic striping and control markings on pavement, parking stalls, and curbs in accordance with the			
52			layout, configurations, and dimensions indicated on the Contract Drawings and approved Shop Drawings.			
53		C.	Application equipment and procedures shall conform to the applicable requirements of the Caltrans Standard			
54		_	Specifications. Keep paint thoroughly mixed throughout application.			
55		D.	Trattic control markings and parking stalls shall be applied with the use of substantial cutout patterns and			
50 57			templates, or with striping equipment that applies straight, uniform width, sharp lines. Coverage shall be			
57			recommendations and the Caltrans Standard Specifications			
50			הביטוווויבווענוטוס מות נווכ כמונימוס סנמועמות סףכנוונמנוטוס.			

1		1.	Provide three (3) coats for painted striping and pavement markings. Application rate per coat shall match that specified in the Caltrang Standard Specifications for each sect of two spect system.
2		2.	Apply thermoplastic material at a minimum thickness of 0.125 inch or at the manufacturer's
4			recommended minimum thickness, whichever is greater.
5	E.	. Tr	affic control markings and parking stalls shall be sharp and accurate, straight where required, without
6		fu	zziness at edges of lines.
7	F.	Ac	ccessible parking stalls shall include the International Symbol for Accessibility.
8	G	. At	completion, the Contractor shall check the work thoroughly and shall touchup traffic control markings and
9		pa	arking stalls that are not distinct or thorough in coverage, or are not uniform in color.
10			
11	2.3 T(OLERAN	CES AND APPEARANCE
12	A.	. In	addition to the tolerances and appearance requirements specified in the Caltrans Standard Specifications,
13		ec	lges shall be uniform with local variations not exceeding 1/8 inch per foot and surfaces shall be smooth and
14		ur	niform.
15	В.	. Le	tter sizes and patterns shall be as indicated on the Contract Drawings with variations of not more than ±15% in
16		di	mension.
17			
18	<u>PART 3 -</u>	EXECUT	ION
19	Not Used	1	
20		•	
21			END OF SECTION

1					SECTION 32 91 13.50	
2					STORMWATER RAIN GARDEN	
3 4	PART 1 - GENERAL					
5						
6	1.1	SCO	PE			
7 8 9 10 11 12 13 14 15		Α.	The w neces: all wo Garde P	ork under ti sary to cons rk, material n Device. Ir ART - GENE Scope Related Referen Submitt Quality	 is section shall consist of providing all work, materials, labor, equipment and supervision truct Stormwater Rain Garden Devices. The work under this section does not include providing s, labor, equipment, and supervision necessary to install plantings for the Stormwater Rain icluded are the following topics: RAL Work ce Standard als Assurance 	
L7			Р	ART 2 - MA	ſERIALS	
18				Geotext	ile Fabric	
19				Aggrega	tes	
20				Sand	+	
21				Enginee	l red Soil	
23				Frosion	Mat	
24			Р	ART 3 - EXE	CUTION	
25				Protecti	on Measures	
26				Tempor	ary Erosion and Sediment Controls	
27				Excavati	on	
28				Storage	Layer	
29				Enginee	red Soil	
30				Erosion	Mat	
31						
32	1.2	RELA	ATED WO	RK	en ef Division d'envenue des des las Continue. De late d'en stiene includes	
33		А.	Аррис	able provisi	ons of Division 1 govern work under this Section. Related sections include:	
94 85			1. 3	Section 32	2000 – Editimoving 91 19 – Topsoil-Select Fill Materials and Application	
36			3. 4	Section 32	92 00 – Trees Shrubs and Other Plantings	
37			5.	Section 32	92 19 – Seeding and Sodding	
88			-			
39	1.3	REFE	RENCE S	TANDARDS		
10		Α.	WISDOT	Γ PAL	Wisconsin Erosion Control Product Acceptability List (PAL)	
11		В.	WISDOT	r sshsc	Standard Specifications for Highway and Structure Construction	
12		C.	WDNR		Standard 1002 - Site Evaluation for Stormwater Infiltration	
13		D.	WDNR		Standard 1004 - Bioretention for Infiltration	
14		E.	WDNR		S100 Specification for Compost	
15 16	1 4	CLID				
+0 17	1.4	30B	Provic	le product d	ata for the following materials:	
18		л.	1	Geotextile	Fabrics	
19			2.	Pipe		
50			3.	Aggregate	5	
51			4.	Sand		
52			5.	Compost		
53			6.	Engineere	d Soil	
54			7.	Erosion M	ət	
55		В.	Provic	le product d	ata for engineered soil blend components: Sand and Compost in compliance with WDNR	
56			Standa	ard 1004 – E	sioretention for Infiltration for review and approval by the Owner's Construction	
57			Repre	sentative.		
8						

1.5	OUA	YASSURANCE
	Д .	Contractor shall submit, in writing to the Owner's Construction Representative, a certification from compost
		supplier that any compost used on the project is in compliance with the requirements outlined in WDNR
		Specification S100.
	В.	Contractor shall submit, in writing to the Owner's Construction Representative, a certification from engineered
		oil supplier that any engineered soil used on the project is in compliance with the requirements outlined in
		NDNR Standard 1004 Bioretention for Infiltration.
PA	RT 2 - MA	RIALS
2 1	CEO.	
2.1		Dine Sock
	7.	The openings of the geotextile fabric shall be small enough to prevent sand particles from entering the
		underdrain pipe. The fabric shall meet the requirements of the WisDOT SSSHC Section 612.2.8.
	В.	-ilter Fabric:
		The fabric shall meet the requirements of WisDOT SSSHC Section 645.2.4, Geotextile Fabric Type DF,
		Schedule B.
2.2		JAIES All aggregates used in the construction of Stormwater Rain Garden devices shall be double washed and free of
	А.	brganic material and fines.
		1. Storage Laver Aggregate:
		a. The aggregate used for the storage layer shall meet the following gradation requirements:
		Sieve Size Percent Passing by Weight
		2-inch 100
		1½-inch 90-100
		1-inch 20-55
		3/4-inch 0-15
		3/8-inch 0-5
		b. A layer of sand may also used in lieu of aggregates specified and shall be in accordance with the requirements outlined in WDNP Standard 1004 Bioretentian for Infiltration and cond specifications.
		outlined in this specification
	B.	Clear Stone Bedding
	2.	Washed angular stone or pea gravel shall be used to cover the underdrain pipe. Washed angular stone or
		pea gravel, graded from 3/8 inch to 1/4 inch.
2.3	SAN	
	Α.	The preferred sand component consists of mostly SiO ₂ , but sand consisting of dolomite or calcium carbonate
		nay also be used.
	В.	Manufactured sand or stone dust is not allowed.
	С.	The sand shall be washed and drained to remove clay and silt particles prior to mixing.
	D.	Sand shall meet one of the following gradation requirements:
		L. USDA Coarse Sand (0.02-0.04 inches)
		ASTMIC33 (Fine Aggregate Concrete Sand)
		wisdot SSHSC Section 501.2.5.3.4 (Fine Aggregate Sand)
24	COM	९ ग
2.4		Compost shall meet the requirements of WDNR Specification \$100 - Compost
	7.	
2.5	ENG	ERED SOIL
	Α.	Engineered Soil shall comply with WDNR Standard 1004. Engineered Soil hall be a blend of Sand and Compost.
	В.	Engineered Soil shall consist of a mixture of 70 to 85% Sand and 15 to 30% Compost. The percentages are based
		on volume.
	С.	Engineered soil mix shall be free of rocks, stumps, roots, brush or other material over 1 inch in diameter. No
		other materials shall be mixed with the planting soil that may be harmful to plant growth or prove a hindrance
		o planting or maintenance.

1		D.	Engineered soil mix shall have a pH between 5.5 and 8.0.
2		Ε.	Do not fertilize.
3		F.	Thoroughly blend engineered soil off-site before delivering to site and installing.
4		G.	Engineered soil shall be delivered to the site and stored on plastic sheeting.
5 6		Н.	The moisture content shall be low enough to prevent clumping and compaction during placement.
7	2.6	EROS	SION MAT
8 9		Α.	Erosion Mat shall comply with the PAL for Urban, Class 1, Type B, as defined by "Standard Specifications for Highway and Structure Construction" and the PAL.
10 11 12		В.	Erosion Mat shall be American Excelsior-Curlex Net-Free, Erosion Control Blanket-S32BD, Western Excelsior- Excel SS-2 All Natural, Ero-Guard EG-25 (NN), Erosion Tech ETRS2BN, or approved equal.
13	PART	3 - EXE	CUTION
14 15	2.1		
15	5.1		ECTION MEASURES
10		А.	Pre-installation Meeting.
18			Construction Representative, and the Contractor shall conduct a pre-installation meeting.
19		в.	Stabilization:
20 21 22			area has been stabilized with vegetation and/or hardscapes. Construction site runoff from disturbed areas
22		C	Weather:
23		с.	1 Construction shall be suspended during periods of rainfall or snowmelt. Construction shall remain
25			suspended of ponded water is present or if residual soil moisture contributes significantly to the potential
26			for soil smearing, clumping, or other forms of compaction.
27		P	2. Delays resultant from weather shall not serve as a basis for a Change Order.
28		D.	Compaction Avoidance.
29			1. Compaction and smearing of the soils beneath the hoor and side slopes of the stormwater Rain Garden
30 21			2 During construction, the area dedicated to the Stormwater Rain Garden Device shall be cordoned off to
32			2. During construction, the area dedicated to the stormwater hair darden bevice shall be condoned on to prevent access by heavy equipment
32			3 Accentable equipment for constructing the Stormwater Rain Garden Device includes excavation boes light
34			equipment with turf type tires marsh equipment or wide-track loaders
35		F	Compaction Remediation
36		L.	1 If compaction occurs at the base of the Stormwater Rain Garden Device, the soil shall be refractured to a
37			depth of at least 24 inches.
38			 If smearing occurs, the smeared areas shall be corrected by raking or roto-tilling. Compaction and smearing remediation shall be conducted by the Contractor of no additional cost to the
39 40			3. Compaction and smearing remediation shall be conducted by the contractor at no additional cost to the Owner.
41		F.	Field Infiltration Confirmation:
42			1. Immediately after rough grading of Stormwater Rain Garden Devices, the Geotechnical Representative or
43			Engineer shall confirm that correct subgrade is reached for adequate infiltration.
44			2. Confirmation shall be obtained prior to continuing work on the Rain Garden Device.
45			
46	3.2	TEM	PORARY EROSION AND SEDIMENT CONTROLS
47		Α.	The Contractor shall install temporary erosion and sediment controls prior to beginning construction of the
48			Stormwater Rain Garden Device. The temporary erosion and sediment controls shall divert stormwater runoff
49			away from the Stormwater Rain Garden Device until it is completed.
50			
51	3.3	EXCA	VATION
52		Α.	Excavation equipment shall work from the sides of the Stormwater Rain Garden Device to excavate the area to
53			the depths and dimensions as shown on the Drawings. Excavation equipment shall have adequate reach so that
54			they do not need to be located within the footprint of the Stormwater Rain Garden Device to excavate it.
55		В.	Any accidental compaction shall be remediated as prescribed above.
56		C.	Over-excavate clay soils down to well-draining soils as direct by Engineer.

Over-excavate clay soils down to well-draining soils as direct by Engineer. C.

1 3.4 STORAGE LAYER

2

3

6

7

8

14

15

19

21

22 23 A. Place the Storage Layer Aggregate to the depth as indicated in the Drawings.

4 **3.5 ENGINEERED SOIL** 5 A. Verify mois

- A. Verify moisture condition of Engineered Soil is low enough to prevent clumping and compaction during placement. Engineered Soil shall not be placed unless it meets these conditions.
- B. Place Engineered Soil in lifts not to exceed 12 inches in depth until the desired elevation of the Stormwater Rain Garden Device is achieved.
- 9C.Re-examine the surface within 48 to 72 hours following placement of Engineered Soil. Place additional10Engineered Soil until desired elevation of the Stormwater Rain Garden Device is achieved at no additional costs11to the Owner.
- 12D.Steps may be taken to induce mild settling of the Engineered Soil as needed to prepare a stable planting medium13and to stabilize the ponding depth.
 - E. Vibrating plate style compactors shall not be used to induce settling.
 - F. No equipment travel on or across placed Engineered Soil is permitted.
- 16G.Install silt fence or other means of erosion control around the perimeter of the engineered soil to protect from17siltation or contamination from adjacent landscape or paved surfaces and construction activities. Leave erosion18control in place until site landscape establishment and construction is complete.

20 3.6 EROSION MAT

A. Install Erosion Mat on top of surface prior to installation of vegetation.

END OF SECTION

1			SECTION 32 91 19			
2	TOPSOIL-SELECT FILL MATERIALS AND APPLICATION					
5 4 5	PART 1 - GENERAL					
6	1.1	SCOPE				
7 8		Α.	The work under this section shall consist of providing all topsoil, labor, material and equipment required to complete the work described herein in strict accordance with the drawings and/or terms of the contract.			
9 10		B.	All work on the public lands and/or public rights-of-way shall conform to the applicable City of Madison's Standard Construction Specifications stated below.			
11 12		C.	All work shall be in accordance with applicable manufacturer's instructions.			
13	1.2	RELAT	ED WORK AND PROVISIONS			
14		A.	Applicable provisions of Division 1 shall govern all work. Related sections include:			
15 16			 Section 02 20 00 - General Sitework Requirements Section 31 20 00 - Earthmoving 			
17			3. Section 31 25 00 - Erosion Control			
18			4. Section 32 92 19 - Seeding and Sodding			
19						
20	1.3	REFER	ENCES			
21 22 23		A.	Where reference is made to the "Construction Specifications," it shall be construed to mean the pertinent section of the City of Madison's "Standard Construction Specifications," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of			
24			payment shall not apply.			
25		В.	Where reference is made to the "Standard Specifications," it shall be construed to mean the pertinent section of			
26 27 28			the "Standard Specifications for Sewer and Water Construction in Wisconsin," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply			
29		C	Where reference is made to the "State Specifications" it shall be construed to mean the pertinent section of the			
30		0.	WisDOT "Standard Specifications for Highway and Structure Construction." current edition, and all supplemental			
31			and interim supplemental specifications, as they may pertain, except the method of measurement and basis of			
32			payment shall not apply.			
33		D.	Where reference is made to the "Geotechnical Report," it shall be construed to mean the geotechnical report in			
34			Section 02 32 00.			
35						
36	1.4	QUALI	ITY ASSURANCE			
37 38		A.	Pre-Work Meeting: Convene a pre-work meeting minimum 30 days prior to commencing work on this Section. Review conditions of operations, procedures, and coordination with related work. The pre-work meeting shall			
39			be setup as a conference call or virtual meeting with the Landscape Architect.			
40			1. Review planting schedule and maintenance.			
41 42		р	2. Review required inspections, schedule of topsoil testing, and environmental procedures.			
42		Б.	Multi-Residue Herbicide/Resticide Screen: A NELAC (National Environmental Laboratory Accreditation			
43 44			Conference) certified independent soil testing laboratory with the experience and canability to conduct			
45			the testing indicated based on local conditions.			
46			2. Topsoil Analysis: Independent soil testing laboratory employing a landscape or soil agronomist familiar			
47			with the final use of the material and construction practices for large earthwork sites.			
48						
49	PART 2 - PRODUCTS					
50						
51	2.1	MATE	RIALS			
52		Α.	Select fill shall be a loamy sand, sandy loam, clay loam, loam, silt loam, sandy clay loam or other soil approved by			
53			the Owner's construction Representative. It shall not have a mixture of subsoil and shall contain no slag,			
54 55			diameter. Solect fill must also be free of viable plants or plant parts of common Dermude gross swell areas			
55 56			unameter. Select initiations also be need of viable plants of plant parts of common Bermuda grass, quack grass,			
57			by a reputable laboratory for nH and soluble salts. If peeded nH correction material shall be applied at a rate			
58			sufficient to correct the pH to a range of 6.0 to 7.0. Soluble salts shall not be higher than 500 parts per million.			

1 2 3 4 5 6 7 8		В. С. D.	No turfgrass sod shall be placed on soil which has been chemically treated until sufficient time has elapsed to permit dissipation of all toxic materials. The general contractor shall assume full responsibility for any loss or damage to turfgrass sod arising from improper use of chemicals or due to his failure to allow sufficient time to permit dissipation of toxic residues, whether or not such materials are specified herein. Topsoil on the existing site may often be used; however, it should meet the same standards as set forth in these specifications. Refer to Drawings for Specifications on Engineered Soils and Sand Storage Layers, as specified by Wisconsin Department of Natural Resources (WDNR).			
9						
10	PART	<u>3 - EXEC</u>	UTION			
11		-				
12	3.1	GRADI	NG			
13 14		Α.	The select fill shall be uniformly distributed on the designated area(s) and it shall be a minimum of 6 inches deep after firming.			
15		В.	No grading shall be done beyond the limits specified within the Grading and Erosion Control Plan.			
16 17		C.	Spreading shall be performed in such a manner that sod installation can proceed with a minimum of additional coil proparation and tillago			
10		П	Soli preparation and tillage.			
18 19		D.	prevent the formation of depressions or water pockets.			
20 21 22		E.	Select fill shall not be placed while in a frozen or muddy condition, when the subgrade is excessively wet, or in a condition that may otherwise be detrimental to proper grading or proposed for seeding.			
23	3.2	CLEAN UP				
24 25		Α.	After the select fill has been spread and the final grade approved, it shall be cleared of all grade stakes, surface trash or other objects that would hinder seeding and other plantings.			
26		В.	Paved areas over which hauling operations are conducted shall be kept clean and any soil which may be brought			
27			upon the surfacing shall be promptly removed.			
28		C.	The wheels of all vehicles shall be kept clean to avoid tracking soil on the surfacing of roads, walks, or other			
29			paved areas.			
30						
31	3.3	ACCEP	TANCE			
32		Α.	Acceptance will be given by the Owner's Construction Representative, upon satisfactory completion of each			
33			section or area(s), as indicated on the Drawings or as otherwise specified.			
34						
35			END OF SECTION			

1 2 2			SECTION 32 92 00 TREES, SHRUBS, AND OTHER PLANTINGS
3 4 -	<u>PART</u>	1 - GEN	IERAL
5	11	SCOD	F
7 8 9 10	1.1	A.	These specifications, along with contract drawings and lists of plant materials, apply to those items necessary for and incidental to the preparation, execution, completion and maintenance of the landscape planting activities (excluding lawn areas) specified in the contract. The scope includes the planting of trees, shrubs, perennials, and grasses, and the maintenance activities of fertilizing, pruning, and watering.
11 12	1.2	RFI A	
13 14 15 16 17 18 19		A.	 Applicable provisions of Division 1 govern work under this Section. Related sections include: Section 02 20 00 - General Sitework Requirements Section 31 20 00 - Earthmoving Section 31 25 00 - Erosion Control Section 32 91 19 - Topsoil-Select Fill Materials and Application Section 32 92 19 - Seeding and Sodding
20	1.3	REFE	RENCE STANDARDS
21 22 23 24 25		А. В. С.	American Standards for Nursery Stock, ANSI Z60.1, current edition (American Association of Nurserymen, Inc.). Standardized Plant Names, Second Edition (1942). American Joint Committee on Horticulture Nomenclature, Horace McFarland Company, Harrisburg, PA. American National Standard for Tree Care Operations - Tree, Shrub and Other Woody Plant Maintenance - Standard Practices ANSI 4300. current edition
26 27 28		D.	Where reference is made to the "Geotechnical Report," it shall be construed to mean the geotechnical report in Section 02 32 00.
29	1.4	QUAL	LITY ASSURANCE
30 31 32 33 34		А. В.	All plant material shall conform to the American Standards for Nursery Stock, unless noted otherwise herein. All plant material shall be true to the species and variety/hybrid/cultivar specified, and nursery-grown in accordance with good horticultural practices, and under climatic conditions similar to those of the site location. Specimens' nursery-dug to be replanted shall have been freshly dug and properly prepared for planting. Trees and Shrubs:
35 36 37 38 39 40			 Shall be trained in development and appearance as to be superior in form, compactness and symmetry. Trees with multiple leaders, unless specified otherwise, and shrubs with damaged or cut mainstem(s), will be rejected. With a damaged, cut or crooked leader, abrasion of bark, sunscald, frost crack, disfiguring knots, insects (including eggs and larvae) or insect damage, cankers/cankerous lesions or fungal mats, mold, prematurely-opened buds, or cuts of limbs over 3/4 inches (1.9 cm) in diameter that are not completely
41 42			callused will be rejected.Shall have healthy, well-developed root systems, and be free from physical damage or other hindrances
43 44 45 46 47			 Balled and burlapped plants shall be dug with solid balls of a diameter not less than that recommended by the American Standards for Nursery Stock, and of sufficient depth to include both fibrous and feeding roots. Balls shall be securely wrapped with burlap, and tightly bound with rope or twine. No plant shall be bound with rope or wire in such manner as to damage bark or break branches. The root flare should
48 49 50 51 52			 be within the top 2 inches (5.1 cm) of the soil ball. Balled and burlapped plants will not be accepted if the ball is dry, cracked, or broken before or during planting. 5. Containerized plants are to be well-established within the container, with a root system sufficiently developed to retain its shape and hold together when removed from the container. Soil within the container should be held together by the roots, in form and whole. Plants shall not be pot-bound, nor
53 54		P	have kinked, circling, or bent roots. Herbaceous perennials and grasses shall only be supplied from pursaries certified by State Plant Increasters
54 55		D.	nerbaceous perenniais and grasses snan only be supplied from nurseries certified by State Plant Inspectors.
56	1.5	MEAS	SUREMENT
57 58		A.	Plants shall conform to the measurements specified within the Contract Documents. Specified height and spread dimensions will refer to the main body of the plant and not from branch tip to branch tip. Plants meeting

1			a specified measurement, but judged to lack the balance between height and spread characteristic of the
2		_	species, will be rejected.
3		В.	Plants shall be measured when branches are in their normal position.
4 5		C.	No plant shall be less than the minimum size specified, and no less than 50% of the plants shall be as large as the maximum size specified.
6		D.	Caliper measurements shall be taken 54 inches (1.4 m) above ground level
7		Ε.	Containerized shrubs shall be measured by height and width for conformity with the plant list
8		F.	Herbaceous perennials and grasses shall be measured by pot size, not by top growth
9		G.	All other measurements, such as number of canes, ball sizes, and quality designations, shall conform to
10			American Standards for Nursery Stock (ASNS).
11			
12	1.6	SUBST	ITUTIONS
13		А.	The substitution of plant materials is not permitted unless authorized, in writing, by the Owner's Construction
14			Representative. If written proof is submitted by the Contractor that a plant of specified species, variety, or size
15			is unavailable, consideration will be given towards the nearest available size or variety, or towards an alternate
16			species selection, with a corresponding adjustment of the contract price.
1/	4 7		
10	1.7		The Contractor is to arrange for the acceptance and unleading of plants at the project site
19		А. D	All plants are to be labeled by plant name and size. Labels shall be attached securely to all plants, bundles, and
20		ь.	All plants are to be labeled by plant hame and size. Labels shall be durable and logible, with information given in
21			weather registrant ink or embessed process lettering. The Owner should verify all plant labels: upon approval
22			the plant labels shall be removed by the Contractor
25		C	All plant raterials, shipmonts, and deliveries shall comply with surrent state and federal laws and regulations
24		С.	an plant materials, singments, and deliveres shall comply with current state and required by law or regulations
25			Cortificate of Inspection, or a copy thereof, for injurious insects, plant diseases, and other plant pasts shall
20			accompany each shipment or delivery of plant material. The certificate shall hear the name(s) and address(es) of
27			the source of the plant stock
20		П	During transport, no plant shall be bound with rope or wire in a manner that damages trunks or breaks
30		D.	branches. Plants shall also not be dragged lifted or nulled by the trunk branches or foliage in a damaging way
31			No plant shall be thrown off of a truck or loader to the ground
32		F	Prior to installation, all plants must be protected from sun and drying winds
33		F.	Containerized or balled and burlanned plants not being installed immediately must be kent in a shaded area
34			well-covered with wood chips, soil, or other approved material, and kept well-watered. Install all plants within
35			three (3) days of delivery.
36		G.	Fertilizer shall be delivered to the site in original, sealed containers, and stored in a waterproof space.
37		•	Containers shall bear the manufacturer's name, analysis, trademark, and guarantee as per standards of the
38			Wisconsin Department of Agriculture.
39		Н.	Contractor shall protect all plants, lawns, and grass from damage at all times. Damaged plants, lawns, or grass
40			areas shall be replaced or treated as required to conform to specifications herein for fresh stock. Damage
41			incurred as a result of replacement or installation operations shall be repaired by Contractor at no cost to
42			Owner.
43			
44	1.8	PLANT	ING SCHEDULE
45		Α.	Plants shall be installed as appropriate for that specific plant species to ensure healthy vigorous growth.
46		В.	All plants shall be guaranteed to be in healthy and flourishing condition for one (1) full year after installation and
47			acceptance by the Owner.
48		C.	Plants not thriving shall be replaced at no cost to the Owner. The Contractor may suggest substitutions for
49			replacement plants.
50		Ε.	At any time during the guarantee period the Contractor shall remove or replace, without cost to the Owner and
51			within a specified planting period, all plants not in healthy and flourish conditions as determined by the Owner.
52			
53	1.9	MAIN	TENANCE
54		Α.	The Contractor shall maintain plantings and lawn for at least a period of 60 days, or until final acceptance from
55			the Owner. The Contactor is responsible for adequately watering plants and lawn during this 60-day period or
56			after a satisfactory stand of grass has been established.
57		В.	Fertilizing: Any and all chemical applications are to be performed in accordance with current federal, state and
58			local laws, through EPA-registered materials and application techniques, and performed under the supervision of

1			a licensed certified applicator. Apply fertilizer to planted areas at the specified rate, and as per manufacturer's
2			recommendations.
3		C.	Watering: All plant materials installed under the contract shall be watered within the first 24 hours of initial
4			planting and not less than twice weekly until final acceptance by the Owner. Water used shall be of sufficient
5			quality for irrigation and free of materials harmful to plant growth
6		D.	Pesticide: Any use of pesticides during the contracted maintenance period, as determined by the Owner, shall
7			utilize the minimum amount of approved pesticide needed to control pests on plant materials installed under
8			the contract. Pesticide applications are to be performed in accordance with current federal, state, and local
9			laws, through EPA-registered materials and application techniques, and performed under the supervision of a
10			licensed certified applicator. Apply at the specified rate and as per manufacturer's recommendations.
11			
12	PART	2 - PROI	<u>DUCTS</u>
13			
14	2.1	MATE	RIALS
15		Α.	Plant Materials: A complete list of plant materials, including a schedule of quantities, sizes, and other
16			requirements, shall be included in the contract documents. If discrepancies occur between the printed plant list
17			and the contract drawings, the printed list will take precedent.
18		В.	Topsoil: Naturally fertile, agricultural soil, capable of sustaining vigorous growth, of uniform composition
19			throughout, without admixtures of subsoil, free of clay, stones larger than 1 inch (2.5 cm) in diameter, roots,
20			trash, and debris of any kind, supplied by Contractor at their expense, and subject to approval by the Owner's
21			Construction Representative.
22		C.	Planting Mixture: Material used in tamping around balls and roots during the planting operation shall be
23			prepared on-site by mixing two parts topsoil, one part sand, and one part compost. All mixing shall be done by
24			mechanical means subject to the approval of the Owner's Construction Representative.
25		D.	Fertilizer: Granular, non-burning product composed of not less than 50% organic slow-acting, guaranteed
26			analysis professional fertilizer. Commercial fertilizer shall conform to Wisconsin State Statutes. Section 94.64.
27			and meet the standards of the Wisconsin Department of Agriculture as to registration and labeling. Fertilizer
28			shall be specified in the Contract Documents as to composition, but is subject to revision to suit project site
29			conditions
30		F	Shredded Hardwood Bark Mulch: Shredded bardwood bark mulch free of material detrimental to bealthy plant
31			growth Mulch shall be finely shedded weed free and dye-free
32		F	Stone Mulch: Planted stone mulch areas shall follow the plan specifications and be spread to a minimum and
32			consistent denth of 3 inches. Stone mulch areas shall receive woven weed harrier fahric
31			consistent depth of 5 menes. Stone match aleas shall receive woven weed barrier fablic.
35	PART	3 - FXFC	ΊΠΟΝ
36		J-LALC	
37	3 1	INSPE	CTION
38	0.1	Δ	Tonsoil · Refer to Tonsoil-Select Fill Materials and Application technical specifications
39		R.	Verify that prepared soil hase is ready to receive the work of this Section
40		с.	Beginning of installation means accentance of existing site conditions
40 //1		С.	beginning of installation means acceptance of existing site conditions.
41 //2	32	DRFDA	
42	5.2		Prenare subsoil to eliminate uneven areas and low snots. Maintain lines levels profiles and contours. Make
43		Α.	changes in grade gradual. Blond clones into level areas
44 1E		р	Changes in grade gradual. Diend slopes into rever areas.
45		Б.	Nentove lot eight materials, weeks, and undesnable plants and their roots. Nentove any containinated subson.
40		c	Plants can be removed through application of glyphosphate. Follow manufacturer's instructions for proper use.
47		C.	scarify subsolit to a depth of 3 inches where topsolitis to be placed. Repeat cultivation in areas where equipment,
48			used for hauling and spreading topsoli, has compacted subsoli.
49	2 2		
5U F1	3.3	PLACI	NG IUPSUIL Defer to Tangail Calact Fill Materials and Application Castion and Afficiations
LJ 2T		А. Р	Refer to ropson-select riminatemats and Application Section Specifications.
52 52		в.	spread any needed amendments per soli test and till soli to a depth of 3 to 4 inches. Ideal seed bed will be a
53		c	combination of soil particles ranging from approximately a 1/4 inch to a full 1 inch in size.
54		L.	opsoil compaction should be below 250 psi. If topsoil compactions are greater, soil should be ripped, disced, or
55			otherwise loosened to a depth of at least 12 inches.

1	3.4	PREPA	RATION
2		Α.	Stake all planting areas and notify Diggers Hotline (1-800-242-8511 statewide) to verify location of all
3			underground utilities prior to excavation.
4		В.	Excavate planting areas as shown in included specifications.
5		C.	Adequately barricade with proper warning devices any planting pit left open when planting work is not in
6			progress, and that poses a hazard to vehicles and/or pedestrians.
7		D.	Maintain site housekeeping to provide for a safe and orderly project site. Collect and dispose of debris as they
8			accumulate.
9		Ε.	The planting pit for containerized and balled and burlapped plants shall be at least 2.5 to 3 times the diameter of
10			the soil ball, or to a dimension otherwise specified.
11		F.	The planting pit for a single shrub shall be 12 inches (30.5 cm) wider than the root ball.
12		G.	Loosen the soil beyond the edge of the planting pit. The soil at the base of the planting pit is to remain
13			undisturbed, the depth of which shall correspond to the distance from the bottom of the soil ball to the root
14			flare, or slightly less.
15		Н.	Fence and/or box in all trees and plant material which are to remain at the drip line before work is started. Do
16			not permit heavy equipment or stockpiles within branch spread. Remove interfering branches without injury to
17			trunks; cover scars with tree paint.
18		Ι.	For a shrub mass planting, the entire bed area is to be tilled to a depth of 4 to 6 inches (10.2 to 15.2 cm).
19			Excavate individual shrub pits to the proper depth.
20			
21	3.5	PLANT	ING OF TREES AND SHRUBS
22		А.	Remove plant containers by cutting or carefully inverting the container. For plants grown in plastic containers,
23			slash the edges of the root ball from top to bottom with vertical 1-inch (2.5 cm) cuts using a sharp blade.
24		В.	Root balled plants shall have rope, string, wire baskets, burlap and other wrapping material removed from the
25			top half of the ball after the plant has been set in the hole. Remaining wrappings, other than those made from
26			plastic or synthetic material, may be left around the bottom half of the ball.
27		C.	Shrubs grown using root containment material shall have the containment bag removed prior to setting.
28		D.	Set trees and shrubs straight and upright and in the center of the planting hole and on the unexcavated base of
29			the planting pit with the most desirable face towards the most prominent view.
30		Ε.	Root-balled shrubs are to be carried and set in the hole by the root ball.
31		F.	Backfilling: Backfill pits with excavated soil. No soil in frozen or muddy condition shall be used for backfilling.
32		G.	When pit is approximately two-thirds backfilled, tamp down and water to eliminate air pockets. After initial
33			watering, add remainder of the soil to the top of pit, water without puddling, and firmly tamp without over-
34			compacting. Form a 2- to 3-inch (5.1 to 7.6 cm) high saucer around the outer rim of the pit prior to mulching.
35		Н.	All trees shall be installed with 5-foot diameter tree ring with 3-inch mulch layer. Tree rings shall have shovel
36			edging.
37		Ι.	All parking islands shall receive a minimum of 18 inches of topsoil.
38			
39	3.6	PLANT	ING OF PERENNIALS, FORBS, AND GRASSES
40		А.	Preparation: Loosen soil of the planting bed to a depth of 4 to 6 inches (10.2 to 15.2 cm) by mechanical or hand
41			tilling while soil is dry. For bulbs, the depth of loosened soil will be determined by the type of bulb planted and
42		_	specified in the contract or landscape plan.
43		В.	After soil is loosened, till organic material into the soil across the planting bed to a uniform depth of 2 inches
44			(5.1 cm) for peat moss or 1 inch (2.5 cm) for compost.
45		C.	Fertilizer, at amounts determined by the soil test, shall be topdressed to the soil.
46		D.	Apply approved mulch uniformly across the entire planting bed to a depth of 1 to 2 inches (2.5 to 5.1 cm).
47		E.	Planting: Space as described in the landscape plan.
48		⊦.	Unless otherwise specified, install plants no closer than 12 inches (30.5 cm) to the trunks of trees or shrubs
49		~	within planting bed and to within 6 inches (15.2 cm) of the edge of the bed.
50		G.	Prior to planting, biodegradable plant containers shall be split and non-biodegradable containers removed. The
51			root systems of all such plants shall be split or crumbled by hand.
52		н.	All parking islands that contain perenniais (not including bio-infiltration area) shall have a minimum of 18 inches
53			ot topsoil. These areas shall also have 3 inches of mushroom compost spread uniformly over the parking island
54 FF			and the into the top 6 inches of the soil.
55	2 7	FINIS	
טכ רק	3./	FINISH	Linich grade planting excess to required elevations of the plants have fully estitled
5/		А.	Finish-grade planting areas to required elevations after plants have fully settled.

1 2		В.	No soil is to cover the top of the root ball. All plants shall be completely mulched over the root system with a 3-inch (7.6 cm) layer of specified mulching material immediately after planting. Pull back mulch no less than
3			3 inches (7.6 cm) and no more than 6 inches (15.2 cm) from the trunk.
4		C.	Thoroughly water plants immediately after planting and before mulching, primarily within and filling the saucer.
5		D.	Prune any dead or broken branches. Prune in accordance with NAA Guidelines conforming to the American
6			Standard for Tree Care Operations. Prune shrubs in accordance with standard horticultural practices. On cuts of
7			3/4 inches in diameter and bruises or scars on bark, trace the inured cambium layer back to living tissues and
8			remove. Smooth and shape wounds so as not to retain water and coat the treated area within approved
9			antiseptic tree paint.
10		Ε.	Remove all twine and rope after planting, along with any labels attached around trunks or branches.
11			
12	3.8	CLEAN	ling
13		Α.	Dispose of excess soil. Remove all cuttings and waste materials.
14		В.	Soil, branches, binding, and wrapping material, rejected plants, or other debris resulting from plant installation
15			shall be promptly cleaned up and removed. New landscape construction in and around the planting areas are to
16			be especially well cleaned.
17		C.	Under no condition/circumstance shall the accumulation of soil, branches, or other debris be allowed upon a
18			public property in such a manner as to result in a public hazard. Likewise, under no circumstances shall any
19			debris or incidental materials be allowed upon adjacent private property.
20			
21			END OF SECTION

Page Intentionally Left Blank

1 2	SECTION 32 92 19 SEEDING AND SODDING				
3	<u>PART :</u>	L - GENI	ERAL		
5 6 7	1.1	SECTION INCLUDES			
8		В.	Placing topsoil		
9		C.	Fertilizing		
10		D.	Seeding		
11		Ε.	Seed Protection		
12		F.	Mulching		
13					
14	1.2	RELAT	ED WORK		
15		Α.	Applicable provisions of Division 1 govern work under this Section. Related sections include:		
16			1. Section 02 20 00 - General Sitework Requirements		
17			2. Section 31 20 00 - Earthmoving		
18			3. Section 31 25 00 - Erosion Control		
19			4. Section 32 91 19 - Topsoil-Select Fill Materials and Application		
20			5. Section 32 92 00 - Trees, Shrubs, and Other Plantings		
21					
22	1.3	DELIVE	EKY, STORAGE, AND HANDLING Conducted the deliver of the two in its emission to many advects in a set of the led on the unitable set of the		
23		А.	seed shall be delivered to the site in its original, unopened container and labeled as to weight, analysis, and		
24			safe from damage from heat or any other deleterious weather conditions		
25			sale non damage non near of any other deletenous weather conditions.		
20	14	RFFFR	ENCE SPECIFICATIONS		
28		A.	Where reference is made to the "Construction Specifications." it shall be construed to mean the pertinent		
29			section of the City of Madison's "Standard Construction Specifications," current edition, and all supplemental		
30			and interim supplemental specifications, as they may pertain, except the method of measurement and basis of		
31			payment shall not apply.		
32		В.	Where reference is made to the "Standard Specifications," it shall be construed to mean the pertinent section of		
33			the "Standard Specifications for Sewer and Water Construction in Wisconsin," current edition, and all		
34			supplemental and interim supplemental specifications, as they may pertain, except the method of measurement		
35			and basis of payment shall not apply.		
36		C.	Where reference is made to the "State Specifications," it shall be construed to mean the pertinent section of the		
37			WisDOT "Standard Specifications for Highway and Structure Construction," current edition, and all supplemental		
38			and interim supplemental specifications, as they may pertain, except the method of measurement and basis of		
39			payment shall not apply.		
40		D.	Where reference is made to the "Geotechnical Report," it shall be construed to mean the geotechnical report in		
41			Section 02 32 00.		
42					
43	1.5	GUAR	ANTEE - Communities also the statistic for a second of 12 as earths following the Contestantial Communities. Data is according to		
44		А.	Guarantee plant material for a period of 12 months following the Substantial Completion Date in accordance		
45			With the Extended Maintenance/ warranty Chart included in Part 3 hereinafter.		
40			1. A limit of one (1) replacement of each plant shall be required, except for losses of replacements due to failure to comply with requirements		
47 78			2 Remove from site any plant that is dead or unsatisfactory to the Owner or Landscape Architect Replace		
40 19			2. Remove non-site any plant that is dead of disatisfactory to the Owner of Landscape Architect. Replace		
50					
51	1.6	SUBM	JBMITTALS		
52		A.	Submittals shall be available at all times to the Owner.		
53		В.	Grower/Nursery Information: Submit name, address, phone number .and contact person for each Grower/		
54			Nursery 30 days prior to plant material selection meeting.		
55		C.	Materials Test Reports: Submit topsoil borrow area test reports to the Owner a minimum six (6) weeks prior to		
56			delivery to site.		
57			1. Provide location of topsoil area tested.		
58			2. Provide name of independent soil testing laboratory.		

1 2		D.	 Provide date of sampling and testing. Product Data:
3			1. Submit certification tags from sod and seed verifying type and purity to the Owner
4		Ε.	Closeout Submittals:
5 6			Submit meeting and inspection logs prior to Final Completion of the Project. Certification of Conformance: Provide Certificate of Satisfactory Performance of Planting Operations
7 8			signed by the Contractor and Landscape Architect.
9	1.7	MAIN	TENANCE
10		A.	The Contractor shall maintain lawn for at least a period of 60 days, or until final acceptance from the Owner.
11			The Contractor is responsible for adequately watering the lawn during this 60-day period. Contractor is
12			responsible for establishing healthy vigorous lawn growth. Long-term maintenance is the responsibility of the
13			Owner.
14 15 16	PART	2 - PRO	DUCTS
10	2.1	SEED	MIXTURE
18		Α.	Grass Seed: All grass seed shall conform to the requirements of Wisconsin State Statutes, Chapter 94 (Seed
19			Law), and the Wisconsin Administrative Code Chapter ATCP 20, regarding noxious weed seed content and
20			labeling. Seed shall not be used later than one year following the test date labeled.
21		В.	Public Seed Mixture: Use State Specifications Mix 40 in the right-of-way.
22		C.	Grounds Seed Mixture:
23		D	1. Use seed mixtures as specified on Drawings.
24 25		D.	Determion Basin Seeding.
25			1. See drawings for rain garden seeding, plug plantings of native vegetative mat requirements.
27	2.2	SOIL	VIATERIALS
28		A.	Topsoil: Refer to Section 32 91 19 - Topsoil-Select Fill Materials and Application.
29			
30	2.3	SOD	
31		А.	Provide sod species suitable as lawn turf for the region. Sod shall be strongly rooted, weed, disease, and pest
32			free, and uniform in thickness.
33 21	24	ACCE	SCODIES
35	2.4	ACCL.	Mulching Material:
36			1. Oat or wheat straw, free from weeds, foreign matter detrimental to plant life, and dry. Hay or chopped
37			cornstalks are <u>not</u> acceptable
38			2. Where necessary to maintain erosion control, seed shall be applied using Method B, Hydroseeding from
39			the State Specifications.
40		В.	Fertilizer: Standard commercial packaged or bulk product in granular form conforming to the requirements of
41		<u> </u>	Dane County and the Wisconsin Statutes and of the Wisconsin Administrative Code Chapter Agriculture 17.
42 42		С.	Water: Clean, fresh, and free of substances or matter which could inhibit vigorous growth of grass.
45 44	ΡΔ RT	3 - FXF	CUTION
45	<u>1 AIU</u>		
46	3.1	INSPE	CTION
47		Α.	Verify that prepared soil base is ready to receive the work of this Section.
48		в.	Beginning of installation means acceptance of existing site conditions.
49			
50	3.2	PREP	ARATION OF SUBSOIL
51		Α.	Prepare subsoil to eliminate uneven areas and low spots. Maintain lines, levels, profiles, and contours. Make
52 E2		р	changes in grade gradually. Biend slopes into level areas.
55 54		ь. С	Remove foreign materials weeds and undesirable plants and their roots. Remove any contaminated subsoil
55		с.	Plants can be removed through application of glyphosate. Follow manufacturer's instructions for proper use
56		D.	Scarify subsoil to a depth of 3 inches where topsoil is to be placed. Repeat cultivation in areas where equipment,
57			used for hauling and spreading topsoil, has compacted subsoil.
58		Ε.	Unsuitable Subsoils: Locations containing unsuitable subsoil shall be treated by one or more of the following:

1 2 3			1. Where unsuitability is deemed by the Owner to be due to excessive compaction caused by heavy equipment and where natural subsoil is other than AASHTO classification of A6 or A7, loosen such areas with spikes, discing, or other means to loosen soil to condition acceptable to the Owner Loosen soil to
4			minimum depth of 12 inches with additional loosening as required to obtain adequate drainage.
5			Contractor may introduce peat moss, sand, or organic matter into the subsoil to obtain adequate
6			drainage. Such remedial measures shall be considered as incidental, without additional cost to the
7			Owner.
8			2. Where unsuitability is deemed by the Owner to be due to presence of boards, mortar, concrete, or other
9 10			construction materials in subgrade and where natural subsoil is other than AASH to classification of A6 or
10			A7, remove debris and objectionable material. Such remedial measures shall be considered as incidental, without additional cost to the Owner.
12			Without additional cost to the Owner. Where unsuitability is deemed by the Owner to be because natural subsoil falls into AASHTO
12			classification of A6 or A7 and contains moisture in excess of 30% then installation of sub-drainage system
14			or other means described elsewhere in these Specifications shall be used. Where such conditions have
15			not been known or revealed prior to planting time and they have not been recognized in preparation of
16			the Drawings and Specifications, then the Owner shall issue pricing order to install proper remedial
17			measures.
18			
19	3.3	PLACI	NG TOPSOIL
20		Α.	Refer to Section 32 91 19 - Topsoil-Select Fill Materials and Application.
21		В.	Spread any needed amendments per soil test and till soil to a depth of 3 to 4 inches. Ideal seed bed will be a
22			combination of soil particles ranging from approximately a 1/4 inch to a full 1 inch in size.
23			
24	3.4	FERTI	
25		A.	Apply seed starter fertilizer at the rate specified by the product manufacturer.
26		в.	Fertilizer must be phosphorus free and meet Dane County requirements.
27		С. р	Apply after smooth raking of topsoli.
20		D. E	Do not apply let mizer at same time of with same machine as will be used to apply seed.
30		F.	Lightly water to aid the dissination of fertilizer
31		••	
32	3.5	SEEDI	NG
33		Α.	Firm up soil with light irrigation—lightly dampen soil before seeding.
34		В.	Sow seed using either Method A or Method B as defined in Section 630.3.3 of the Standard Specifications for
35			Highway Construction.
36		C.	Protect seeded slopes of 4:1 or greater against erosion with erosion control materials specified on grading and
37			erosion control plan.
38		D.	Apply seed evenly in two directions at a rate specified by the product manufacturer. Rake in lightly. A
39			cultipacker or similar equipment shall be used to enhance soil/seed contact. Care shall be taken to avoid
40			damage to erosion mat in areas where erosion mat is specified. Do not seed areas in excess of that which can be
41		F	mulched on the same day.
42		E.	Do not sow immediately following rain, when ground is too dry, or during windy periods.
45 11		г. С	Do not broadcast of drop seed when wind velocity exceeds 5 mpn.
44 15		ы. Н	Sow seed at a rate of 1% pounds per 1 000 square feet. In addition to lawn seed, appual rive shall be applied to
46			all disturbed areas at a rate of 1% pounds per 1,000 square feet
40		1	Roll seeded area with 24-inch width roller not exceeding 112 nounds
48		J.	Immediately following seeding and compacting, apply mulch to a thickness of 1/8 inches.
49		K.	Apply water with a fine spray immediately after each area has been mulched keeping the top 1 to 2 inches of soil
50			moist but not soaking. Water adequately to achieve a healthy stand of weed free lawn. Do not let soil dry out.
51		L.	Apply a second application of seed starter fertilizer at the rate specified by the product manufacturer three (3)
52			weeks after seeding.
53		M.	Begin weekly mowing when first seedlings reach 2 inches. Do not mow right after watering. Raise mowing
54			height to 3 inches after six (6) weeks. Never remove more than 1/3 (one third) of the grass blade at a time.
55		Ν.	Begin standard fertilization and irrigation programs after eight (8) weeks. Do not apply any weed control
56			products until lawn has been mowed at least four times and a minimum of eight (8) weeks have passed. Follow
57			manufacturer's recommendations for new lawns.
58			

1	3.6	SEED PROTECTION						
2		Α.	Identify seeded areas with stakes around area periphery. Refer to Drawings for signage.					
3								
4	3.7	SODDING						
5		Α.	Cut and lay sod on same day. Only healthy vigorous growing sod shall be laid.					
6		В.	Lay sod across slope and tightly together to result in solid coverage free of gaps.					
7		C.	Roll or firmly but lightly tamp new sod with suitable wooden or metal tamper sufficiently to set or press sod into					
8			underlying soil.					
9		D.	All finished sodding shall be smooth and free of lumps and depressions.					
10		Ε.	After sodding has been completed, clean up and thoroughly water newly sodded areas.					
11								
12	3.8	MAIN	NTENANCE DURING CONSTRUCTION					
13		Α.	Begin maintenance operations immediately after each plant is planted and continue as required until					
14			acceptance. Water, mulch, weed, prune, spray, fertilize, cultivate, and otherwise maintain and protect plants.					
15			Reset settled plants to proper grade and position, restore planting saucers, and remove dead, diseased, or					
16			unhealthy plant material. Tighten and repair stakes and wires. Correct defective work as soon as possible after					
17			it becomes apparent and weather and season permit.					
18		В.	Upon completion of the planting operations, clean up landscaped areas to be free of stones, containers, trash,					
19			and other waste and debris to leave area in a neat and well-groomed appearance.					
20		C.	Supplement rainfall as required to provide an equivalent of 1 inch of water per week until the plants have rooted					
21			and are established.					
22		D.	Make weekly inspections to determine moisture content of soil and adjust watering schedule established by					
23			irrigation system installer to fit conditions.					
24		Ε.	After grass growth has started, reseed or sod areas that fail to show uniform stand of grass in accordance with					
25			the Drawings and as specified herein. Continue reseeding and sodding such areas repeatedly until areas are					
26			covered with satisfactory growth of grass. Perform removal and replacement or topsoil conditioning if required					
27			to facilitate establishment of grass.					
28		F.	Water in such manner and as frequently as is deemed necessary by the Owner to assure continued growth of					
29			healthy grass. Water areas of site in such a manner as to prevent erosion due to excessive quantities applied					
30			over small areas and to avoid damage to finished surface due to watering equipment.					
31		G.	Provide water for execution and maintenance at no expense to the Owner. Furnish portable tanks, pumps, hose,					
32			pipe, connections, nozzles, and any other equipment required to transport water from available outlets and					
33			apply it to seeded areas in an approved manner.					
34		Н.	Mowing:					
35			1. Initiate mowing of turf grass areas when grass has attained height of 3 inches and roots are firmly					
36			established. Maintain turf grass height at 2½ to 3 inches at subsequent cuttings depending on time of					
37			year. Remove no more than one-third (1/3) of grass leaf at any cutting and cutting shall not occur more					
38			than ten (10) days apart.					
39			2. Mow native grass areas no more than three times per year to a height of no less than 6 inches.					
40			3. Remove heavy cuttings to prevent destruction of underlying turf. If weeds or other undesirable					
41			vegetation threaten to smother planted species, such vegetation shall be mowed or, in case of rank					
42			growths, shall be uprooted, raked, and removed from area by methods approved by the Owner.					
43		Ι.	Remove weeds and other undesirable vegetation by applying herbicides as recommended by the manufacturer					
44			or by uprooting. Rake and remove uprooted vegetation from area by methods approved by the Owner.					
45		J.	Protect seeded area from pedestrian or vehicular trespassing while grass is germinating. Provide fences, signs,					
46			barriers, or other necessary temporary protective devices. Repair damage resulting from trespass, erosion,					
47			washout, settlement, or other causes.					
48		К.	Remove fences, signs, barriers, or other temporary protective devices after final acceptance.					
49		L.	Grassed areas damaged during process of work shall be restored or repaired to condition satisfactory to the					
50			Owner. Fill, grade, re-fertilize, replant, or mulch as required to restore to contract requirements.					
51								
52			END OF SECTION					

PART 1-GENERAL 1. SUMMARY A. The work under this section shall consist of providing all work, materials, labor, equipment, and supervision necessary to provide for the sanitary sever work beginning at a point five 5 feet outside of the building wall, unless otherwise specifications and an the Drawings. This specifications shall apply to all sanitary sever work beginning at a point five 5 feet outside of the building wall, unless otherwise specifications. 1. Section 312 00 - Earthmoving 2. Section 312 30 - Earthmoving 3. Section 312 30 - Earthmoving 3. Section 312 30 - Earthmoving 4. Where reference is made to the "Construction Specifications," it shall be construed to mean the pertinent section of the City of Madison's "Standard Construction Specifications," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 2. Where reference is made to the "Standard Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Sever and Water Construction in Wisconsin," current edition, and all supplemental and hasis of payment shall not apply. 2. Where reference is made to the "Standard Specifications," as they may pertain, except the method of measurement and basis of payment shall not apply. 2. SUBMITTALS A. Reference A. Reference	1 2	SECTION 33 30 00 SANITARY SEWER UTILITIES								
Summary Sector 6 1.1 SUMMARY A. The work under this section shall consist of providing all work, materials, labor, equipment, and supervision necessary to provide for the sanitary sever work beginning at a point five 5 feet outside of the building wall, unless otherwise specified. 10 B. Related sections include the following: 11. Section 31.20 00 - Earthmoving 12. Section 31.20 16.13 - Trenching 14 Section 31.23 16.13 - Trenching 15 STANDARD SPECIFICATIONS A. Where reference is made to the "Construction Specifications," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. B. Where reference is made to the "Standard Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Sever and Water Construction in Wisconsin," current edition, and all supplemental apply. C. Where reference is made to the "Standard Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Highway and Structure Construction in Wisconsin," current edition, and all supplemental apply. C. Where reference is made to the "Standard Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Highway and Structure Construction," current edition, and all supplemental and interim supplemental specifications of Highway and Structure Construction," current edition, and all	3	PART	PART 1 - GENERAL							
6 1.1 SUMMARY A. The work under this section shall consist of providing all work, materials, labor, equipment, and supervision necessary to provide for the sanitary sewer work required in these specifications and on the Drawings. This specification shall apply to all sanitary sewer work beginning at a point five 5 feet outside of the building wall, unless otherwise specified. 11 B. Related sections include the following: 12 1. Section 31 20 0.0 Farthmoving 13 2. Section 31 20 0.0 Farthmoving 14 A. Where reference is made to the "Construction Specifications," ti shall be construed to mean the pertinent section of the City of Madison's "Standard Construction Specifications," current edition, and all supplemental and interim supplemental specifications, are the method of measurement and basis of payment shall not apply. 20 B. Where reference is made to the "Standard Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Sever and Water Construction in Wisconsin," current edition, and all supplemental and hasis of payment shall not apply. 21 SubMitTALS C. 22 Where reference is made to the "State Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications or Highway and Structure. Omstructon," current edition, and all supplemental and hasis of payment shall not apply. 23 SUBMITTALS C. Where reference is made to the "	5									
A. The work under this section shall consist of providing all work, materials, labor, equipment, and supervision necessary to provide for the sanitary sever work required in these specifications and on the Drawings. This specification shall apply to all sanitary sever work beginning at a point five 5 feet outside of the building wall, unless otherwise specified. 11 B. Related sections include the following: 12 1. Section 31 20 0.0 rearthnowing 13 2. Section 31 20 0.0 rearthnowing 14 2. Section 31 20 0.0 rearthnowing 15 A. Where reference is made to the "Construction Specifications," it shall be construed to mean the pertinent section of the City of Madios's "Standard Specifications," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 20 B. Where reference is made to the "Standard Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Sever and Water Construction in Wisconsin", current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 24 C. Where reference is made to the "State Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Highway and Structure Construction." Current edition, and all supplemental specifications for Highway and Structure Construction." Current edition, and all supplemental spe	6	1.1	SUMI	MARY						
 necessary to provide for the sanitary sever work required in these specifications and on the Drawings. Inits of Specification shall apply to all sanitary sever work beginning at a point five 5 feet outside of the building wall, unless otherwise specified. Related sections includely to all sanitary sever work beginning at a point five 5 feet outside of the building wall, unless otherwise specified. Section 31 2000 - Earthmowing: Section 31 2000 - Earthmowing Section 31 2010 - Earthmowing Section 31 2010 - Earthmowing Section 31 2010 - Earthmowing Statuant Sections for the York and the York of the York of the Section Sections, "It shall be construed to mean the pertinent section of the "Standard Specifications," It shall be construed to mean the pertinent section of the "Standard Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications," it shall be construed to mean the pertinent section of measurement and basis of payment shall not apply. Where reference is made to the "Standard Specifications," it shall be construction for mean the pertinent section of payment shall not apply. SUBMITTALS Refer to Division 1. The Owner's testing agency. SUBMITTALS A refer to Division 1. Sub Reference Standards: A merican Society for Testing and Materials (ASTM): A merican Society for Testing and Materials (ASTM):	7		Α.	The w	ork under this section shall	consist of providing all work, materials, labor, equipment, and supervision				
 specification shall apply to all samiarly sever work beginning at a point rive 5 feet outside of the building wall, unless of there were specified. B. Related sections include the following: Section 31 200 - Earthmoving Section 31 200 - Earthmoving STANDARD SPECIFICATIONS A. Where reference is made to the "Construction Specifications," it shall be construed to mean the pertinent section of the City of Madison's "Standard Construction Specifications," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. B. Where reference is made to the "Standard Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Severe and Water Construction in Wisconsin," current edition, and all supplemental and interim supplemental specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Highway and Structure Construction my current edition, and all supplemental and interim supplemental specifications," at they may pertain, except the method of measurement and basis of payment shall not apply. C. Where reference is made to the "State Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Highway and Structure Construction, "current edition, and all supplemental and interim supplemental specifications," at they may pertain, except the method of measurement and basis of payment shall not apply. SUBMITTALS A. Refer to Division 1. B. The Owner shall provide testing agency do all testing. The Contractor shall coordinate the tests with the Owner's testing agency. Subard Specificat	8			neces	sary to provide for the san	tary sewer work required in these specifications and on the Drawings. This				
 Related sections include the following: Section 31 2000 - Farthmoving Section 31 2000 - Farthmoving Section 31 2000 - Farthmoving 12 TANDARD SPECIFICATIONS Where reference is made to the "Construction Specifications," trishall be construed to mean the pertinent section of the City of Madison's "Standard Construction Specifications," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. Where reference is made to the "Standard Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Sewer and Water Construction in Wisconsin," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. Where reference is made to the "State Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Highway and Structure Construction," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 13 SUBMITTALS	9 10			specifi	cation shall apply to all sar	ntary sewer work beginning at a point live 5 feet outside of the building wall,				
1. Section 31 20 00 - Earthmoving 13 2. Section 31 23 16.13 - Trenching 14 2. Section 31 23 16.13 - Trenching 15 1.2 STANDARD SPECIFICATIONS 16 A. Where reference is made to the "Construction Specifications," our rend edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 20 B. Where reference is made to the "Standard Specifications," it shall be construct on in Wisconsin," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 21 C. Where reference is made to the "Standard Specifications," it shall be construct on in Wisconsin," current edition, and all supplemental and interim supplemental specifications," it shall be construct on the method of measurement and basis of payment shall not apply. 23 SUBMITTALS 24 C. Where reference is made to the "State Specifications," it shall be construct on the method of measurement and basis of payment specifications for Highway and Structure Construction in Wisconsin," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 29 1.3 SUBMITTALS 30 A. Refererenc Standards: <td>11</td> <td></td> <td>в</td> <td>Relate</td> <td>ed sections include the follo</td> <td>wing.</td>	11		в	Relate	ed sections include the follo	wing.				
13 2. Section 31 23 16.13 - Trenching 14 12 STANDARD SPECIFICATIONS 16 A. Where reference is made to the "Construction Specifications," it shall be construed to mean the pertinent section of the City of Madison's "Standard Construction Specifications," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 19 B. Where reference is made to the "Standard Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Sewer and Water Construction in Wisconsin," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 24 C. Where reference is made to the "State Specifications," ti shall be construed to mean the pertinent section of the "Standard Specifications for Highway and Structure Construction," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 29 1.3 SUBMITTALS A. Refer to Division 1. B. The Owner shall provide testing agency do all testing. The Contractor shall coordinate the tests with the Owner's testing agency. 31 I. A REFERENCE A. Reference Standards: A. Dizebie provisions of Division 1 shall govern all work under this section. B. Reference Standards J. Dizebie-04 Standard Specification for Rigid P	12		υ.	1.	Section 31 20 00 - Farthm	oving				
14 Image: Construction Specifications of the Construction Specifications," or server deliton, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 20 B. Where reference is made to the "Standard Specifications," it shall be construct to mean the pertinent section of the City of Madison's Sewer and Water Construction in Wisconsin," current edition, and all supplemental and interim supplemental specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Highway and Structure Construction in Wisconsin," current edition, and all supplemental and interim supplemental specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Highway and Structure Construction," current edition, and all supplemental and interim supplemental specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Highway and Structure Construction," current edition, and all supplemental and interim supplemental specifications," it shall be constructed to measurement and basis of payment shall not apply. 29 1.3 SUBMITTALS A. Refer to Division 1. B. The Owner shall provide testing agency do all testing. The Contractor shall coordinate the tests with the Owner's testing agency. 20 A. Referer to Division 1 shall govern all work under this section. B. Reference Exandards: 1. Nemerica Society for Testing and Materials (ASTM): 0 D1784-03	13			2.	Section 31 23 16.13 - Trer	iching				
12 STANDARD SPECIFICATIONS 16 A. Where reference is made to the "Construction Specifications," it shall be construed to mean the pertinent section of the City of Madison's "Standard Construction Specifications," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 20 B. Where reference is made to the "Standard Specifications," it shall be construct on mean the pertinent section of the "Standard Specifications for Sever and Water Construction in Wisconsin," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 20 Where reference is made to the "State Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Highway and Structure Construction," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 20 Where reference is made to the "State Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Highway and Structure Construction," current edition, and all supplemental and interim supplemental specifications as they may pertain, except the method of measurement and basis of payment shall not apply. 20 Noter reference is made to the "State Specifications," it shall be construed to mean the pertinent section of the City not shall cordinate the tests with the Owner's testing agency. 21 Is the "Standard Specification for Acrylonitrile-Butadiene-St	14									
16 A. Where reference is made to the "Construction Specifications," it shall be construed to mean the pertinent section of the City of Madison's "Standard Construction Specifications," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 20 B. Where reference is made to the "Standard Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Sewer and Water Construction in Wisconsin," current edition, and all supplemental and interim supplemental applications, as they may pertain, except the method of measurement and basis of payment shall not apply. 24 C. Where reference is made to the "State Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Highway and Structure Construction," current edition, and all supplemental and interim supplemental specifications," as they may pertain, except the method of measurement and basis of payment shall not apply. 29 1.3 SUBMITTALS 30 A. Refer to Division 1. 31 B. The Owner shall provide testing agency do all testing. The Contractor shall coordinate the tests with the Owner's testing agency. 32 A. Applicable provisions of Division 1 shall govern all work under this section. 36 B. Reference Standards: 37 1. American Society for Testing and Materials (ASTM): 38	15	1.2	STAN	DARD S	PECIFICATIONS					
17 section of the City of Madison's "Standard Construction Specifications," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 20 B. Where reference is made to the "Standard Specifications," it shall be construct to mean the pertinent section of the "Standard Specifications for sever and Water Construction In Wisconsin," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 21 Where reference is made to the "State Specifications," it shall be construct to mean the pertinent section of the "Standard Specifications for Highway and Structure Construction," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 23 13 SUBMITTALS 24 C. Where reference is made to the "State Specifications," thall be construct of measurement and basis of payment shall not apply. 28 13 SUBMITTALS 30 A. Refer to Division 1. 31 B. The Owner shall provide testing agency do all testing. The Contractor shall coordinate the tests with the Owner's testing agency. 33 1. American Society for Testing and Materials (ASTM): 36 D1784-03 Standard Specification for Rigid Poly(Vinyl Chloride) (PVC) Compounds and Chlorinated Poly(Vinyl Chloride) (PVC) Compounds.	16		Α.	Where	e reference is made to the	"Construction Specifications," it shall be construed to mean the pertinent				
18 and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 20 B. Where reference is made to the "Standard Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Sewer and Water Construction in Wisconsin," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 24 C. Where reference is made to the "State Specifications," it shall be construct to mean the pertinent section of the "Standard Specifications for Highway and Structure Construction," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 29 1.3 SUBMITTALS 30 A. Refer to Division 1. 31 B. The Owner shall provide testing agency do all testing. The Contractor shall coordinate the tests with the Owner's testing agency. 31 A. Reference Standards: 37 1. American Society for Testing and Materials (ASTM): 38 D1784-03 Standard Specification for Rigid Poly(Vinyl Chloride) (PVC) Compounds and Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds. 39 D1784-03 Standard Specification for Acryonitrile-Butadiene-Styrene (ABS) and Poly(Vinyl Chloride) (PVC) Composite Sewer Piping <t< td=""><td>17</td><td></td><td></td><td>sectio</td><td>n of the City of Madison's "</td><td>'Standard Construction Specifications," current edition, and all supplemental</td></t<>	17			sectio	n of the City of Madison's "	'Standard Construction Specifications," current edition, and all supplemental				
19 payment shall not apply. 20 B. Where reference is made to the "Standard Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Sewer and Water Construction in Wisconsin," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 24 C. Where reference is made to the "State Specifications," it shall be construct onean the pertinent section of the "Standard Specifications for Highway and Structure Construction," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 26 "State To Division 1. 27 payment shall not apply. 28 "SubmitTALS 30 A. Refer to Division 1. 31 B. The Owner shall provide testing agency do all testing. The Contractor shall coordinate the tests with the Owner's testing agency. 33 A. Applicable provisions of Division 1 shall govern all work under this section. 36 B. Reference Standards: D1784-03 37 1. American Society for Testing and Materials (ASTM): D1784-03 38 D1784-03 Standard Specification for Solvent Cements for Poly(Vinyl Chloride) (PVC) Compounds D1784-03 40 D2564-04	18			and in	iterim supplemental specifi	cations, as they may pertain, except the method of measurement and basis of				
20 B. Where reference is made to the "Standard Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for State Specifications," it shall be construed to mean the pertinent section of the and basis of payment shall not apply. 21 Where reference is made to the "State Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for State Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications for Highway and Structure Construction," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 29 1.3 SUBMITTALS 30 A. Refer to Division 1. 31 B. The Owner shall provide testing agency do all testing. The Contractor shall coordinate the tests with the Owner's testing agency. 31 Marcina Specification for Testing and Materials (ASTM): B. 32 D1784-03 Standard Specification for Solvent Cempounds and Chlorinated Poly(Vinyl Chloride) (PVC) Compounds and Chlorinated Poly(Vinyl Chloride) (PVC) Compounds and Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds 33 D1784-03 Standard Specification for Arylonitrile-Butadiene-Styrene (ABS) and Poly(Vinyl Chloride) (PVC) Compounds 34 1.4 Reference Standards: Field Color (CPVC) Compounds 34 D1784-03 Standard Specification for	19			payme	ent shall not apply.					
21 supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 24 C. Where reference is made to the "State Specifications, it shall be construct to mean the pertinent section of the "Standard Specifications for Highway and Structure Construction," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 29 1.3 SUBMITTALS 30 A. Refer to Division 1. 31 B. The Owner shall provide testing agency do all testing. The Contractor shall coordinate the tests with the Owner's testing agency. 34 1.4 REFERENCE 36 A. Reference Standards: 37 1. American Society for Testing and Materials (ASTM): 38 D1784-03 Standard Specification for Rigid Poly(Vinyl Chloride) (PVC) Compounds and Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds and Specification for Solvent Cements for Poly(Vinyl Chloride) (PVC) 41 D2564-04 Standard Specification for Acrylonitrile-Butadiene-Styrene (ABS) and Poly(Vinyl Chloride) (PVC) Campounds 42 D2680-01 Standard Specification for Acrylonitrile-Butadiene-Styrene (ABS) and Poly(Vinyl Chloride) (PVC) Camposite Sewer Piping 44 D3034-04a Standard Specification for Joints for Drai	20		В.	the "S	e reference is made to the "	"standard Specifications," It shall be construed to mean the pertinent section of				
and basis of payment shall not apply. C. Where reference is made to the "State Specifications," it shall be construed to mean the pertinent section of the "Standard Specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. and basis of payment shall not apply. C. Where reference is made to the "State Specifications," it shall be construction, and all supplemental section of the "Standard Specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. and basis of payment shall not apply. Note: Specifications of the "State Specifications," current edition, and all supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. and basis of payment shall not apply. Note: Specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. and basis of payment shall not apply. Note: Specification specification, "current edition, and all supplemental specification for Specification for shall coordinate the tests with the Owner's testing agency. 33 A. Reference Standards: 34 1.4 REFERENCE 5 A. Applicable provisions of Division 1 shall govern all work under this section. 36 B. Reference Standards: 37 1. American Society for Testing and Materials (ASTM): 38 D1784-03 Sta	21			supple	amental and interim supple	mental specifications, as they may pertain, except the method of measurement				
24 C. Where reference is made to the "State Specifications," it shall be construed to mean the pertinent section of the 25 "Standard Specifications for Highway and Structure Construction," current edition, and all supplemental and 26 interim supplemental specifications, as they may pertain, except the method of measurement and basis of 27 payment shall not apply. 28 . 29 1.3 SUBMITALS 30 A. Refer to Division 1. 31 B. The Owner shall provide testing agency do all testing. The Contractor shall coordinate the tests with the 32 Owner's testing agency. . 34 1.4 REFERENCE 35 A. Applicable provisions of Division 1 shall govern all work under this section. 38 D1784-03 Standard Specification for Right Poly(Vinyl Chloride) (PVC) Compounds and Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds 39 D2564-04 Standard Specification for Solvent Cements for Poly(Vinyl Chloride) (PVC) 41 D2680-01 Standard Specification for Solvent Cements for Poly(Vinyl Chloride) (PVC) 42 D2680-01 Standard Specification for Solvent Sewer Piping 44 D3034-04a Standard Specification for Copposite Sewer Pipi	22			and h	asis of navment shall not ar	nnental specifications, as they may pertain, except the method of measurement				
 "Standard Specifications for Highway and Structure Construction," current edition, and all supplemental and interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 1.3 SUBMITTALS A. Refer to Division 1. B. The Owner shall provide testing agency do all testing. The Contractor shall coordinate the tests with the Owner's testing agency. 1.4 REFERENCE A. Applicable provisions of Division 1 shall govern all work under this section. B. Reference Standards: I. American Society for Testing and Materials (ASTM): D1784-03	24		C.	Where	e reference is made to the	"State Specifications." it shall be construed to mean the pertinent section of the				
 interim supplemental specifications, as they may pertain, except the method of measurement and basis of payment shall not apply. 30 A. Refer to Division 1. B. The Owner shall provide testing agency do all testing. The Contractor shall coordinate the tests with the Owner's testing agency. 1.4 REFERENCE A. Applicable provisions of Division 1 shall govern all work under this section. B. Reference Standards: 1.1 American Society for Testing and Materials (ASTM): D1784-03 Standard Specification for Rigid Poly(Vinyl Chloride) (PVC) Compounds and Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds D2564-04 Standard Specification for Solvent Cements for Poly(Vinyl Chloride) (PVC) D26680-01 Standard Specification for Type PSM Poly(Vinyl Chloride) (PVC) Sewer Pipe and Fittings D3034-04a Standard Specification for Joints for Drain and Sewer Plastic Piping D3034-04a Standard Specification for Joints for Drain and Sewer Plastic Pipes Using Fiexible Elastomeric Seals F477-02e1 Standard Specification for Flastomeric Seals (Gaskets) for Joining Elastic Pipe F679-03 Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Pipe and Fittings C104/ANSI A21.495 Standard Specification for Ductile-Iron Pipe and Fittings for Water C104/ANSI A21.495 Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids 	25			"Stand	dard Specifications for High	way and Structure Construction," current edition, and all supplemental and				
27 payment shall not apply. 28 I.3 SUBMITTALS 30 A. Refer to Division 1. 31 B. The Owner shall provide testing agency do all testing. The Contractor shall coordinate the tests with the Owner's testing agency. 34 I.4 REFERENCE 35 A. Applicable provisions of Division 1 shall govern all work under this section. 36 B. Reference Standards: 37 1. American Society for Testing and Materials (ASTM): 38 D1784-03 Standard Specification for Rigid Poly(Vinyl Chloride) (PVC) Compounds and Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds. 39 Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds 40 D2564-04 Standard Specification for Acrylonitrile-Butadiene-Styrene (ABS) and Poly(Vinyl Chloride) (PVC) Compounds 42 D2680-01 Standard Specification for Joints for Drain and Sewer Piping 44 D334-04a Standard Specification for Joints for Drain and Sewer Piping 45 B3122-96a(2003)e1 Standard Specification for Joints for Drain and Sewer Plastic Pipes Using Flexible Elastomeric Seals 46 D3212-96a(2003)e1 Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity Sewer Pipe and Fittings	26			interir	n supplemental specificatio	ons, as they may pertain, except the method of measurement and basis of				
 SUBMITTALS A. Refer to Division 1. B. The Owner shall provide testing agency do all testing. The Contractor shall coordinate the tests with the Owner's testing agency. I.4 REFERENCE A. Applicable provisions of Division 1 shall govern all work under this section. B. Reference Standards: I.4 American Society for Testing and Materials (ASTM): D1784-03 Standard Specification for Rigid Poly(Vinyl Chloride) (PVC) Compounds and Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds and Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds D2564-04 Standard Specification for Solvent Cements for Poly(Vinyl Chloride) (PVC) D2680-01 Standard Specification for Acrylonitrile-Butadiene-Styrene (ABS) and Poly(Vinyl Chloride) (PVC) Composite Sewer Piping D3034-04a Standard Specification for Type PSM Poly(Vinyl Chloride) (PVC) Sewer Pipe and Fittings D3212-96a(2003)e1 Standard Specification for Diain and Sewer Plastic Pipes Using Fiexible Elastomeric Seals F477-02e1 Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity Sewer Pipe and Fittings C104/ANSI A21.4-95 Standard Specification for Doly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity Sewer Pipe and Fittings C104/ANSI A21.4-95 Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity Sewer Pipe and Fittings C104/ANSI A21.4-95 Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids C104/ANSI A21.4-95 Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids	27			payme	ent shall not apply.					
9 1.3 SUBMITTALS 30 A. Refer to Division 1. 9 1.4 Reference shall provide testing agency do all testing. The Contractor shall coordinate the tests with the Owner's testing agency. 33 1.4 REFERENCE 35 A. Applicable provisions of Division 1 shall govern all work under this section. 6 B. Reference Standards: 37 1. American Society for Testing and Materials (ASTM): 9 D1784-03 Standard Specification for Rigid Poly(Vinyl Chloride) (PVC) Compounds and Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds 9 D2564-04 Standard Specification for Solvent Cements for Poly(Vinyl Chloride) (PVC) 41 D2680-01 Standard Specification for Acrylonitrile-Butadiene-Styrene (ABS) and Poly(Vinyl Chloride) (PVC) Composite Sewer Piping 42 D3034-04a Standard Specification for Type PSM Poly(Vinyl Chloride) (PVC) Sewer Pipe and Fittings 46 D3212-96a(2003)e1 Standard Specification for Dist for Drain and Sewer Plastic Pipes Using Flexible Elastomeric Seals 47 F477-02e1 Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity Sewer Pipe and Fittings 50 F679-03 Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity	28									
30 A. Refer to Division 1. 31 B. The Owner shall provide testing agency do all testing. The Contractor shall coordinate the tests with the Owner's testing agency. 33 1.4 REFERENCE 34 1.4 REFERENCE 35 A. Applicable provisions of Division 1 shall govern all work under this section. 36 B. Reference Standards: 37 1. American Society for Testing and Materials (ASTM): 38 D1784-03 Standard Specification for Rigid Poly(Vinyl Chloride) (PVC) Compounds and Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds 40 D2564-04 Standard Specification for Solvent Cements for Poly(Vinyl Chloride) (PVC) 41 D2680-01 Standard Specification for Acrylonitrile-Butadiene-Styrene (ABS) and Poly(Vinyl Chloride) (PVC) Sewer Pipe and Fittings 42 D2680-01 Standard Specification for Type PSM Poly(Vinyl Chloride) (PVC) Sewer Pipe and Fittings 43 D3034-04a Standard Specification for Joints for Drain and Sewer Plastic Pipes Using Flexible Elastomeric Seals 44 D3212-96a(2003)e1 Standard Specification for Poly(Vinyl Chloride) (PVC) Sewer Pipe and Fittings 50 F477-02e1 Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity Sewer Pipe and Fittings<	29	1.3	SUBN	1ITTALS						
31 B. The Owner shall provide testing agency do all testing. The Contractor shall coordinate the tests with the Owner's testing agency. 32 1.4 REFERENCE 35 A. Applicable provisions of Division 1 shall govern all work under this section. 36 B. Reference Standards: 37 1. American Society for Testing and Materials (ASTM): 38 D1784-03 Standard Specification for Rigid Poly(Vinyl Chloride) (PVC) Compounds and Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds 40 D2564-04 Standard Specification for Solvent Cements for Poly(Vinyl Chloride) (PVC) 41 D2680-01 Standard Specification for Acrylonitrile-Butadiene-Styrene (ABS) and Poly(Vinyl Chloride) (PVC) Composite Sewer Piping 44 D3034-04a Standard Specification for Type PSM Poly(Vinyl Chloride) (PVC) Sewer Pipe and Fittings 45 D3212-96a(2003)e1 Standard Specification for Joints for Drain and Sewer Plastic Pipes Using Flexible Elastomeric Seals 48 F477-02e1 Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity Sewer Pipe and Fittings 50 F679-03 Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity Sewer Pipe and Fittings 51 C104/ANSI A21.4-95 Standard for Cement-Mortar Lining for Ductile-Iron Pi	30		A.	Refer	to Division 1.					
33 1.4 REFERENCE 35 A. Applicable provisions of Division 1 shall govern all work under this section. 36 B. Reference Standards: 37 1. American Society for Testing and Materials (ASTM): 38 D1784-03 Standard Specification for Rigid Poly(Vinyl Chloride) (PVC) Compounds and Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds 40 D2564-04 Standard Specification for Solvent Cements for Poly(Vinyl Chloride) (PVC) 41 D2680-01 Standard Specification for Acrylonitrile-Butadiene-Styrene (ABS) and Poly(Vinyl Chloride) (PVC) Composite Sewer Piping 42 D2680-01 Standard Specification for Type PSM Poly(Vinyl Chloride) (PVC) Sewer Pipe and Fittings 44 D3034-04a Standard Specification for Type PSM Poly(Vinyl Chloride) (PVC) Sewer Pipe and Fittings 45 F477-02e1 Standard Specification for Joints for Drain and Sewer Plastic Pipes Using Flexible Elastomeric Seals 48 F477-02e1 Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity Sewer Pipe and Fittings 50 F679-03 Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity Sewer Pipe and Fittings 51 C104/ANSI A21.4-95 Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for Water	31		В.	The O	wher shall provide testing a	agency do all testing. The Contractor shall coordinate the tests with the				
334 1.4 REFERENCE 355 A. Applicable provisions of Division 1 shall govern all work under this section. 36 B. Reference Standards: 37 1. American Society for Testing and Materials (ASTM): 38 D1784-03 Standard Specification for Rigid Poly(Vinyl Chloride) (PVC) Compounds and Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds 39 D2564-04 Standard Specification for Solvent Cements for Poly(Vinyl Chloride) (PVC) 40 D2680-01 Standard Specification for Acrylonitrile-Butadiene-Styrene (ABS) and Poly(Vinyl Chloride) (PVC) Composite Sewer Piping 44 D3034-04a Standard Specification for Type PSM Poly(Vinyl Chloride) (PVC) Sewer Pipe and Fittings 46 D3212-96a(2003)e1 Standard Specification for Joints for Drain and Sewer Plastic Pipes Using Flexible Elastomeric Seals 48 F477-02e1 Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity Sewer Pipe and Fittings 50 F679-03 Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity Sewer Pipe and Fittings 52 2. American Water Works Association (AWWA): 53 C104/ANSI A21.4-95 Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for Water 55 C151/ANSIA21.	32 22			Owne	r's testing agency.					
A. Applicable provisions of Division 1 shall govern all work under this section. B. Reference Standards: I. American Society for Testing and Materials (ASTM): D1784-03 Standard Specification for Rigid Poly(Vinyl Chloride) (PVC) Compounds and Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds O D2564-04 Standard Specification for Solvent Cements for Poly(Vinyl Chloride) (PVC) Plastic Piping Systems D2680-01 Standard Specification for Acrylonitrile-Butadiene-Styrene (ABS) and Poly(Vinyl Chloride) (PVC) Composite Sewer Piping D4 D3034-04a Standard Specification for Type PSM Poly(Vinyl Chloride) (PVC) Sewer Pipe and Fittings 6 D3212-96a(2003)e1 Standard Specification for Joints for Drain and Sewer Plastic Pipes Using Flexible Elastomeric Seals 7 F477-02e1 Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Pipe 50 F679-03 Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity Sewer Pipe and Fittings 52 2. American Water Works Association (AWWA): 53 C104/ANSI A21.4-95 Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for Water 54 C151/ANSIA21.53-00 Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids	34	1.4	REFE	RENCE						
36 B. Reference Standards: 37 1. American Society for Testing and Materials (ASTM): 38 D1784-03 Standard Specification for Rigid Poly(Vinyl Chloride) (PVC) Compounds and Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds 40 D2564-04 Standard Specification for Solvent Cements for Poly(Vinyl Chloride) (PVC) 41 Plastic Piping Systems 42 D2680-01 Standard Specification for Acrylonitrile-Butadiene-Styrene (ABS) and Poly(Vinyl Chloride) (PVC) Composite Sewer Piping 44 D3034-04a Standard Specification for Type PSM Poly(Vinyl Chloride) (PVC) Sewer Pipe and Fittings 46 D3212-96a(2003)e1 Standard Specification for Lastomeric Seals (Gaskets) for Joining Elastic Pipe 50 F477-02e1 Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity Sewer Pipe and Fittings 52 2. American Water Works Association (AWWA): 53 C104/ANSI A21.4-95 Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for Water 54 C151/ANSIA21.53-00 Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids	35		A.	Applic	able provisions of Division	1 shall govern all work under this section.				
371.American Society for Testing and Materials (ASTM):38D1784-03Standard Specification for Rigid Poly(Vinyl Chloride) (PVC) Compounds and Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds39D2564-04Standard Specification for Solvent Cements for Poly(Vinyl Chloride) (PVC)40D2680-01Plastic Piping Systems42D2680-01Standard Specification for Acrylonitrile-Butadiene-Styrene (ABS) and Poly(Vinyl Chloride) (PVC) Composite Sewer Piping44D3034-04aStandard Specification for Type PSM Poly(Vinyl Chloride) (PVC) Sewer Pipe and Fittings46D3212-96a(2003)e1Standard Specification for Joints for Drain and Sewer Plastic Pipes Using Flexible Elastomeric Seals48F477-02e1Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Pipe50F679-03Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity Sewer Pipe and Fittings522.American Water Works Association (AWWA):53C104/ANSI A21.4-95Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for Water54C151/ANSIA21.53-00Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids	36		В.	Refere	ence Standards:	-				
38D1784-03Standard Specification for Rigid Poly(Vinyl Chloride) (PVC) Compounds and39Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds40D2564-04Standard Specification for Solvent Cements for Poly(Vinyl Chloride) (PVC)41Plastic Piping Systems42D2680-01Standard Specification for Acrylonitrile-Butadiene-Styrene (ABS) and43Poly(Vinyl Chloride) (PVC) Composite Sewer Piping44D3034-04aStandard Specification for Type PSM Poly(Vinyl Chloride) (PVC) Sewer Pipe45and Fittings46D3212-96a(2003)e1Standard Specification for Joints for Drain and Sewer Plastic Pipes Using47Flexible Elastomeric Seals48F477-02e1Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic50F679-03Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic522.American Water Works Association (AWWA):53C104/ANSI A21.4-95Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for Water55C151/ANSIA21.53-00Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids	37			1.	American Society for Test	ing and Materials (ASTM):				
39Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds40D2564-04Standard Specification for Solvent Cements for Poly(Vinyl Chloride) (PVC)41Plastic Piping Systems42D2680-01Standard Specification for Acrylonitrile-Butadiene-Styrene (ABS) and43Poly(Vinyl Chloride) (PVC) Composite Sewer Piping44D3034-04aStandard Specification for Type PSM Poly(Vinyl Chloride) (PVC) Sewer Pipe45and Fittings46D3212-96a(2003)e1Standard Specification for Joints for Drain and Sewer Plastic Pipes Using48F477-02e1Standard Specification for Elastomeric Seals (Gaskets) for Joining Elastic49Pipe50F679-03Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic51Gravity Sewer Pipe and Fittings522.American Water Works Association (AWWA):53C104/ANSI A21.4-95Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for Water55C151/ANSIA21.53-00Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids	38				D1784-03	Standard Specification for Rigid Poly(Vinyl Chloride) (PVC) Compounds and				
40D2564-04Standard Specification for Solvent Cements for Poly(Vinyl Chloride) (PVC)41Plastic Piping Systems42D2680-01Standard Specification for Acrylonitrile-Butadiene-Styrene (ABS) and43Poly(Vinyl Chloride) (PVC) Composite Sewer Piping44D3034-04aStandard Specification for Type PSM Poly(Vinyl Chloride) (PVC) Sewer Pipe45and Fittings46D3212-96a(2003)e1Standard Specification for Joints for Drain and Sewer Plastic Pipes Using47F477-02e1Standard Specification for Elastomeric Seals48F477-02e1Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic49OGravity Sewer Pipe and Fittings50F679-03Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic51Gravity Sewer Pipe and Fittings522.American Water Works Association (AWWA):53C151/ANSIA21.53-00Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for Water55C151/ANSIA21.53-00Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids56C153/021 53Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids	39					Chlorinated Poly(Vinyl Chloride) (CPVC) Compounds				
41Plastic Piping Systems42D2680-01Standard Specification for Acrylonitrile-Butadiene-Styrene (ABS) and43Poly(Vinyl Chloride) (PVC) Composite Sewer Piping44D3034-04aStandard Specification for Type PSM Poly(Vinyl Chloride) (PVC) Sewer Pipe45and Fittings46D3212-96a(2003)e1Standard Specification for Joints for Drain and Sewer Plastic Pipes Using47F477-02e1Standard Specification for Elastomeric Seals48F477-02e1Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic49Pipe50F679-03Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic51Gravity Sewer Pipe and Fittings522.American Water Works Association (AWWA):53C104/ANSI A21.4-95Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for54WaterStandard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids55C151/ANSIA21.53-00Standard for Ductile Iron Pipe, Centrifugally Cast for Water Service	40				D2564-04	Standard Specification for Solvent Cements for Poly(Vinyl Chloride) (PVC)				
42D2680-01Standard Specification for Acrylonitrile-Butadiene-Styrene (ABS) and Poly(Vinyl Chloride) (PVC) Composite Sewer Piping43Poly(Vinyl Chloride) (PVC) Composite Sewer Piping44D3034-04aStandard Specification for Type PSM Poly(Vinyl Chloride) (PVC) Sewer Pipe and Fittings46D3212-96a(2003)e1Standard Specification for Joints for Drain and Sewer Plastic Pipes Using Flexible Elastomeric Seals48F477-02e1Standard Specification for Elastomeric Seals (Gaskets) for Joining Elastic Pipe50F679-03Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity Sewer Pipe and Fittings522.American Water Works Association (AWWA):53C104/ANSI A21.4-95Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for Water55C151/ANSIA21.53-00Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids56C153/A21 53Standard for Ductile Iron Pipe, Centrifugally Cast for Water Sensice	41				D2600.04	Plastic Piping Systems				
43Poly(Min Chloride) (PVC) Composite Sewer Piping44D3034-04aStandard Specification for Type PSM Poly(Vinyl Chloride) (PVC) Sewer Pipe and Fittings46D3212-96a(2003)e1Standard Specification for Joints for Drain and Sewer Plastic Pipes Using Flexible Elastomeric Seals48F477-02e1Standard Specification for Elastomeric Seals (Gaskets) for Joining Elastic Pipe50F679-03Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity Sewer Pipe and Fittings522.American Water Works Association (AWWA):53C104/ANSI A21.4-95Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for Water55C151/ANSIA21.53-00Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids56C153/A21 53Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids	42 42				D2680-01	Standard Specification for Acrylonitrile-Butadiene-Styrene (ABS) and				
44b3034-04aStandard Specification for Type PSW Poly(Villy Chloride) (PVC) Sewer Pipe45and Fittings46D3212-96a(2003)e147Standard Specification for Joints for Drain and Sewer Plastic Pipes Using Flexible Elastomeric Seals48F477-02e149Pipe50F679-0351Gravity Sewer Pipe and Fittings522.2.American Water Works Association (AWWA):53C104/ANSI A21.4-9554Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids55C151/ANSIA21.53-0056Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids	43 11				D3034-045	Poly(Vinyi Chionae) (PVC) Composite Sewer Piping Standard Specification for Type PSM Poly(Vinyi Chloride) (DVC) Sewer Dine				
46D3212-96a(2003)e1Standard Specification for Joints for Drain and Sewer Plastic Pipes Using47Flexible Elastomeric Seals48F477-02e1Standard Specification for Elastomeric Seals (Gaskets) for Joining Elastic49Pipe50F679-03Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic51Gravity Sewer Pipe and Fittings522.American Water Works Association (AWWA):53C104/ANSI A21.4-95Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for54Water55C151/ANSIA21.53-00Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids56C153/A21 53Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids	45				03034 040	and Fittings				
47Flexible Elastomeric Seals48F477-02e1Standard Specification for Elastomeric Seals (Gaskets) for Joining Elastic49Pipe50F679-03Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic51Gravity Sewer Pipe and Fittings522.American Water Works Association (AWWA):53C104/ANSI A21.4-95Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for54Water55C151/ANSIA21.53-00Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids56C153/A21 53Standard for Ductile Iron Compact Fittings for Water Service	46				D3212-96a(2003)e1	Standard Specification for Joints for Drain and Sewer Plastic Pipes Using				
48F477-02e1Standard Specification for Elastomeric Seals (Gaskets) for Joining Elastic49Pipe50F679-03Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity Sewer Pipe and Fittings51Caracter Works Association (AWWA):522.American Water Works Association (AWWA):53C104/ANSI A21.4-95Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for Water55C151/ANSIA21.53-00Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids56C153/A21 53Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids	47					Flexible Elastomeric Seals				
49Pipe50F679-03Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic Gravity Sewer Pipe and Fittings51Gravity Sewer Pipe and Fittings522.American Water Works Association (AWWA):53C104/ANSI A21.4-95Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for Water54Vater55C151/ANSIA21.53-00Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids56C153/A21 53Standard for Ductile Iron Pipe, Centrifugally Cast for Water Service	48				F477-02e1	Standard Specification for Elastomeric Seals (Gaskets) for Joining Elastic				
50F679-03Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic51Gravity Sewer Pipe and Fittings522.American Water Works Association (AWWA):53C104/ANSI A21.4-95Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for54Water55C151/ANSIA21.53-00Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids56C153/A21 53Standard for Ductile Iron Pipe, Centrifugally Cast for Water Service	49					Pipe				
51 Gravity Sewer Pipe and Fittings 52 2. American Water Works Association (AWWA): 53 C104/ANSI A21.4-95 Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for Water 54 Water 55 C151/ANSIA21.53-00 Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids 56 C153/A21 53 Standard for Ductile Iron Compact Fittings for Water Service	50				F679-03	Standard Specification for Poly(Vinyl Chloride) (PVC) Large-Diameter Plastic				
52 2. American Water Works Association (AWWA): 53 C104/ANSI A21.4-95 Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for Water 54 Vater 55 C151/ANSIA21.53-00 Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids 56 C153/A21 53 Standard for Ductile Iron Compact Fittings for Water Service	51					Gravity Sewer Pipe and Fittings				
53 C104/ANSI A21.4-95 Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for 54 Water 55 C151/ANSIA21.53-00 Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids 56 C153/A21 53 Standard for Ductile Iron Compact Fittings for Water Service	52			2.	American Water Works A	ssociation (AWWA):				
54 water 55 C151/ANSIA21.53-00 Standard for Ductile Iron Pipe, Centrifugally Cast for Water or Other Liquids 56 C153/A21.53 Standard for Ductile Iron Compact Fittings for Water Service	53				C104/ANSI A21.4-95	Standard for Cement-Mortar Lining for Ductile-Iron Pipe and Fittings for				
SS CISI/ANSIAZI.SS-00 Standard for Ductile from Pipe, Centinugally Cast for Water or Other Liquids 56 C153/Δ21.53 Standard for Ductile from Compact Eittings for Water Service	54 55				C151/ANGIA21 E2 00	Water Standard for Ductile Iron Ding, Contrifugally Cast for Water or Other Ligwide				
	56				C153/A21.53	Standard for Ductile Iron Compact Fittings for Water Service				

1	1.5	SUBMITTALS			
2		Α.	Provide manufacturers product information (cut sheets), shop drawings and O&M information for sewer		
3			materials, including:		
4			1. Pipe		
5			2. Fittings		
6			3. Structures		
7			4. Castings		
8		В.	Provide reports documenting pressure testing, mandreling, and televising.		
9		C.	Provide copies of record drawings.		
10					
11	1.6	RECO	RD DRAWINGS		
12		A.	Maintain record drawings that show the actual locations, sizes and types of utilities and other features		
13			encountered.		
14		В.	Note any modifications to proposed sewer system size, location or elevation. Record any other deviations from		
15			the Drawings.		
16					
17	PART	2 - MAT	TERIALS		
18					
-0 19	2.1	GENE	RAL		
20		Δ	Conform all materials to the size and type shown on the plans or as called for in the specifications and to		
_0 21		7.0	applicable Laws, Codes, and Ordinances.		
22		B	All products and materials are to be new undamaged clean and in good condition. Existing products and		
22		υ.	materials are not to be reused unless specifically indicated		
23		C	Be responsible for the safe storage and handling of all materials utilized in the work. Store all materials in areas		
25		С.	designated by the Construction Representative in cooperation with the Owner		
25		П	Berformall work in accordance with any annicate manufacturer's instructions		
20 27		υ.	r chorn al work in accordance with any applicable manufacturer 5 instructions.		
27 28	22	DIDF			
20		Δ	Provide the size, type and class/schedule of nine as indicated on the Drawings		
30		R.	Use only nine supplied from the same manufacturer, and of the same type, unless otherwise specified or		
30 31		υ.	annroved in advance by the Engineer		
37		C	Only nine, joints, material and installation approved by Wisconsin Department of Natural Resources (W/DNR)		
32		С.	and/or the Wisconsin Department of Safety & Professional Services (WDSDS) for the intended use in the State of		
37			Wisconsin shall be used		
25			wisconsin shall be used.		
36	22				
30 27	2.5		Conform to ASTM D2024 with solvent weld or electomeric joints. Dine shall be SDR-35 or 26 unless otherwise		
28 28		А.	noted on the Drawings		
20 20		D	De net mix different manufacturer's products er fittings		
29 40		ь. С	Do not mix unreferring and fittings shall be electomeric. Fittings shall be of standard manufacture, injection		
40 11		C.	solitis in FVC sewer pipe and numers shall be elastometric. Fittings shall be of standard manufacture, injection		
41 40		D	nioueu, ano shan nave a maximum standaro umension ratio (2016) of 20.		
4Z 40		D.	the metavials to be further that the shall have an experience record substantiating acceptable performance of		
45 44		г	DVC nine shall not be discolored		
44 15		с.	PVC pipe shall not be discolored.		
45 4C	2.4				
40	2.4	BEDD	ING/INITIAL COVER		
47		А.	21.22 (12) Tranship		
48			31 23 16.13 - Frenching.		
49 50		в.	Sanitary sewer and sewer services shall be provided with 6 inches of bedding material and 12 inches of initial		
50			cover material (both measured at the bell of the pipe). Crushed stone bedding shall be used for both bedding		
51		C	and initial cover. Deskilling shall include 10 incluse of group long to the second black in the second state of the second state of		
52		C.	Backfilling shall include 12 inches of gravel under roadways, consolidation by mechanical means to 95 percent		
53			Standard Proctor density for imported granular material and 95 percent for excavated materials of existing		
54			material in the adjacent trench wall.		
55					
56	2.5	CONN	IECTIONS FOR DISSIMILAR PIPE MATERIALS		
57		А.	Where new sewer connects to and existing dissimilar pipe, the connection shall be made with a no hub type		
58			coupling meeting the requirements of CISPI 310. Couplings shall have neoprene gaskets with stainless steel		

1			shield	d, and multiple stainless steel clamps with worm gear tightening device. The couplings shall be made			
2			specif	fically for the type and size of pipe materials being connected. Couplings shall be Fernco Strongback RC			
3			series with stainless steel hardware or approved equal.				
4							
5	2.6	MANI	HOLES				
6		Α.	Gene	ral:			
7		7.0	1	Provide precast concrete manholes. Concrete block or cast-in-place manholes may only be used after			
, 8				receiving written approval by the City of Madison's Construction Representative and the Engineer for			
0				eusting which happend a bases			
9			2	customized mannote sizes and shapes.			
10			2.	Submit manufacturer's preproduction (snop) Drawings for approval prior to the start or manufacturing.			
11			3.	Contractor shall carefully locate all pipe locations, sizes, orientation and elevation prior to ordering new			
12				manholes. For sewer re-lays, verify if each pipe encountered is active. In- active pipe shall not be			
13				connected to the new sewer.			
14		В.	Preca	ist Manhole Sections:			
15			1.	Precast concrete manhole sections, including bottom and top shall meet the requirements of ASTM C478.			
16			2.	Unless otherwise noted, provide four, 4 foot diameter manholes. If field conditions require a larger			
17				structure, contact the Owner's Project Representative or Engineer.			
18			3.	For 4-foot diameter manholes, provide eccentric cone top sections with a minimum clear opening of			
19			0.	24 inches Elation slabs may be used on manholes greater than 4 feet in diameter			
20			л	Manhola wall thickness shall be minimum of 5 inches for 4-foot diameter manholas. 6 inches for 5-foot			
20			4.	diamote manholes and Zinches for C foot and Z foot diamote manholes			
21			-	dialiteter manifoles, and 7 inches for 6-100 and 7-100 dialiteter manifoles.			
22			5.	Mannole bottom section shall be pre-cast with integral base having a minimum thickness of 6 inches			
23		-		unless otherwise noted.			
24		C.	Joints	51 51			
25			1.	Provide manhole riser and barrel sections, cones, and flat tops, with standard pipe section tongue and			
26				groove joints.			
27			2.	Seal joints watertight with prefabricated rubber or plastic gaskets or formed in place butyl rubber seal.			
28			3.	Joint sealers: EZ-stick, Kent Seal, Ram-Nek, or Mas-Stik butyl rubber gaskets or butyl rubber ropel, or			
29				approved equal.			
30			4.	External Seals: The City of Madison's Construction Representative will determine which new manholes			
31				will receive an exterior joint wrap. If called for, the wrap shall be Mar-Mac Manufacturing Company			
32				MacWrap or an equal approved by the City prior in writing			
22		П	Conn	intervise of an equal approved by the exp pror, in writing.			
24		D.	1	Constructions for connections shall be set in place or cored and appropriately sized for the type and size of			
34 2E			1.	openings for connections shall be cast-in-place of core and appropriately sized for the type and size of			
35			2	pipe being connected.			
36			2.	Provide flexible, watertight, pipe-to-manhole connections (or "boots") for sanitary sewers; Kor-N-Seal,			
37		_		Interpace, A-Lok, or an approved equal.			
38		Ε.	Manh	nole Steps:			
39			1.	Provide steps at 16-inch O.C. and project approximately 6 inches from wall.			
40			2.	Unless otherwise indicated on the Drawings, locate manhole steps over the downstream pipe opening.			
41			3.	Manhole steps shall be steel reinforced polypropylene with 1/2-inch diameter deformed reinforcing bar.			
42				Steps shall be permanently secured in the manhole wall. Manhole steps shall be American Step			
43				Company, M.A. Industries, or approved equal.			
44		F.	Bench	h and Flowline:			
45		••	1	Provide either a pre-cast or cast-in-place bench or flowline			
46			2	Index otherwise indicated on the Drawings, banch beight shall be 3/4 the diameter of the downstream			
40			۷.	nings Slope based towards flowlings at a minimum 1/2 inch par foot. Provide light brane finish on			
47				hansh. The bench shall extend to the ten of the downstream rise			
40			2	bench. The bench shall extend to the top of the downstream pipe.			
49			3.	riowines shall be formed with gradual, uniform sweeps directed towards the downstream pipe. Provide			
50				smooth, troweled finish for flowlines.			
51			4.	When cast-in-place benches and flowline are used, lay the sewer pipe through the manhole.			
52		G.	Adjus	ting Rings:			
53			1.	All final grade adjustment of manhole covers and frame assemblies shall be completed utilizing precast			
54				concrete adjusting rings, high density polyethylene (HDPE) or expanded polypropylene (EPP) as			
55				manufactured by LADTECH, Cretex (Pro-Ring), or an approved equal.			
56			2.	Adjusting rings shall be mortared (for concrete rings) or sealed with butyl rubber sealant or gaskets(HDPE			
57				and EPP).			

1	2.7	CASTINGS				
2		Α.	Genera	al:		
3			1.	All manhole castings shall be heavy duty iron conforming to ASTM A48, Class 20 and rated for AASHTO H-		
4				20 loading. Provide water-tight, gasketed, self-sealing, non-rocking lids with two (2) concealed pickholes.		
5				If discrepancies exist between the specific castings in this specification and the plans, the plans shall		
6				govern.		
7			2.	Standard Manhole Frame and Casting:		
8				a. Neenah Foundry R-1550, with Type B lid; or approved equal.		
9			3.	Low Profile Manhole Frame and Casting:		
10				a. Neenah Foundry R-1689, with Type B lid; or approved equal.		
11			4.	Standard Security Manhole Frame and Casting (Solid Lid):		
12				a. Neenah Foundry Company R-1916-C with bolt down Type B lid; or approved equal. Lid shall be		
13				water tight, gasketed, self-sealing, with concealed pick-hole.		
14			5.	Low Profile Security Manhole Frame and Casting (Solid Lid):		
15				a. Neenah Foundry R-1689, with Type B lid having four (4) Type "E" countersunk flathead pent		
16				socket screws; or approved equal. Lid shall be water tight, gasketed, self-sealing, with concealed		
17				pick-hole.		
18			6.	Off-Street Castings:		
19				a. Off street castings shall have bolt down lids. Bolt down castings shall be Neenah 1916-C castings.		
20				The casting shall be bolted to the manhole.		
21	2.8	MANH	HOLE CH	IMNEY SEAL		
22		A.	Provide	e an external frame/cone seal meeting requirements of Sections 8.42.3-8.42.5 of the Standard		
23			Specifi	cations for Sewer and Water Construction in Wisconsin for all manholes. Provide Type I flexible		
24			watert	ight frame/chimney joint with Cretex internal chimney seals.		
25						
26	2.9	PIPE I	NSULATI	ION		
27		A.	Rigid, c	closed-cell extruded polystyrene insulation. Insulation shall be suitable for buried insulation.		
28		В.	Individ	ual boards shall have dimensions of 8'x4'x2".		
29		C.	Dow St	tyrofoam, or approved equal.		
30						
31	2.10	TRACE	TRACER WIRE			
32		A.	The tra	acer wire shall be installed on top of all PVC sanitary sewer laterals and taped at 6-foot intervals from the		
33			wye at the sewer main to the center of the terrace. A second wire shall be extended from the center of the			
34			terrace	e to the end of the lateral stubbed into the lot. A spool of excess wire shall be provided at the end of the		
35			lateral	equal to one-half the depth of the lot for use by the building plumber at the time of building connection.		
36			The co	nductor shall be brought to the surface to a test box located directly above the sewer lateral in the center		
37			of the	terrace between the back of the curb and the front of the sidewalk.		
38		В.	Below	grade splices are discouraged. If splices are necessary, the splice should be made with a 3M Scotchlok		
39			Self-Sti	ripping Connector, or approved equal.		
40		C.	A 12-ga	auge, solid copper, neoprene wrapped, green tracer wire shall be installed on top of all PVC sanitary sewer		
41			laterals	S.		
42		D.	The en	d of the tracer wire at the sewer wye shall be connected to a 2-foot long, 5/8-inch diameter ground rod.		
43		Ε.	Test Ac	ccess Box: Taylor P200NFG or a Valvco CP Test Mini-box with "SEWER" stamped lid and tracer wire		
44			connec	ction holes with stainless steel bolts. Lid shall have standard pentagonal head key.		
45						
46	2.11	MARK		rs		
47		A.	Point o	of termination shall be marked with a 6-foot steel fence post with 1 foot of the post below the end of the		
48			lateral.	. A 6-foot 4'x4' treated wood post will also be installed at the end of each lateral with 3 feet of the post		
49			being b	ouried below grade. The exposed portion of the post shall be painted green.		
50			- 0	J I I I I I I I I I I I I I I I I I I I		
51	2.12	MARK	ER BALL	S		
52		A.	Marke	r balls shall be used to mark location of sewer lateral. Marker balls shall be 4-inch diameter 3M EMS		
53			Extend	led Range 5-foot Ball Marker or approved equal.		
54						

1 PART 3 - EXECUTION

2

4

5

6

7

8

10

11

12 13

14

15

16

17

25

26

29

30

33

38

39

41

42

46

3 3.1 NOTIFICATION

A. Contractor, prior to excavation work, shall notify all utilities, governmental agencies, or entities, known to, or which can reasonably be assumed to, have above or below ground pipe, conduit cables, structures or similar items within limits of project, to locate and mark location of such items. The Contractor shall expose potential pipe conflicts prior to installation of sewers to allow for any field changes to the design to be made.

9 3.2 BYPASS PUMPING

A. Unless otherwise noted, all tributary buildings and services will remain occupied during construction. Wastewater will continue to be discharged to the sanitary sewers during construction. Contractor shall provide, operate and maintain all diversion and bypass pumping equipment necessary to carry out the work and allow wastewater to be discharged. Provide all necessary generators or other power source necessary to operate pumps on a continuous basis. Extra pumping and power equipment shall be staged onsite to maintain bypass pumping in the event of failure of the primary bypass pumping equipment. The Contractor is solely responsible for wastewater bypassing.

18 3.3 BYPASS PLAN

19A.Contractor shall provide a wastewater bypass pumping plan indicating the order and schedule for completion of20the work and associated bypassing provisions. The plan shall indicate the location of proposed bypassing,21discharge locations, and the type and size of pumping equipment to be used. The plan shall describe22contingencies to be used in the event of failure of the primary bypass pumps. Contractor's by-passing plan is23subject to Owner's approval prior to implementation.

24 3.4 LAYING PIPE

- A. Install all pipe in accordance with ASTM specifications which pertain to the specified type of pipe material and the installation situation.
- 27 B. Do not use any pipe or fittings cracked in cutting or handling or otherwise not free from defects.
- 28 C. Clean all pipe of any dirt and/or debris both inside and out prior to placing in the trench.

D. Cap all pipe ends to prevent material from entering pipe. Only use caps supplied by the same manufacturer, and of the same type, unless otherwise specified or approved in advance by the Engineer.

- E. Make joints in accordance with manufacturer's directions with due care to avoid damaging pipe and/or
 disturbing previously laid pipe.
 - F. Cut pipe only according to manufacturer's directions.
- 34G.Lay all sewer pipes to horizontal alignment and grade shown on the plans with bell ends up hill. Establish and35maintain horizontal alignment using total station, transit or theodolite. Use pipe laser or level to establish and36maintain grade of pipe. Discrepancies from the required horizontal alignment or grade at any location shall not37be greater than 0.10 foot or 0.03 foot, respectively.
 - H. Do not exceed specified trench widths.

40 **3.5 BEDDING/INITIAL COVER**

- Provide 3/4-inch clear stone bedding and initial cover in accordance with the applicable requirements of Section
 31 23 16.13 Trenching.
- 43B.Sanitary sewer and sewer services shall be provided with 6 inches of bedding material and 12 inches of initial44cover material (both measured at the bell of the pipe). Crushed stone bedding shall be used for both bedding45and initial cover.

47 3.6 MANHOLES

- A. Contractor shall determine the proper location, size, elevation, and orientation of all pipes entering new
 manholes before ordering. Do not connect abandoned pipes to new manholes. Manholes having improper
 location and/or orientation of the pipe connections will be rejected. Field repairs or adjustments of connection
 points are not permitted.
- 52B.Limit the excavation for manholes so as to provide only the necessary amount of space to sufficiently prepare53the subgrade, set the base, set the manhole or structure, and lay pipe. Provide a minimum of 1 foot of clearance54between structure and trench wall for adequate backfilling and compaction.
- 55C.Where excavation occurs below the bottom elevation of the structure's base, bring the excavation to the56required elevation by the use of compacted crushed stone bedding. A minimum of 8 inches of compacted57Crushed Stone Bedding shall be placed below manhole base.

1 2 3		D.	Set manhole base in accordance with elevation and location as indicated on the plans. Install base plumb and level. Install subsequent pre-cast manhole sections in accordance with shop drawing layout. Provide watertight gaskets between each manhole section.	
4 5		E.	Pour inverts with smooth surface draining to downstream pipe. Where two or more lines meet at an angle, provide curved channel. Slope manhole bench at 2 inches/foot towards flow channel.	
6 7 8		F.	Manholes shall be provided with between 4 inches and 8 inches of adjusting rings, with the top adjusting ring being 2 inches thick. Provide butyl sealant material between rings. Once rings are in place, tuck point the exterior joint and provide the entire exterior surface of the adjusting ring riser with a coating of mortar.	
9 10 11		G.	When indicated on the Drawings, the manhole frame shall be set with a Type I frame/chimney joint as specified in the Standard Specifications for Sewer and Water Construction in Wisconsin, latest edition. The frame and adjusting rings shall be sealed with an internal rubber sleeve as detailed in File 12A of the Standard	
12 13 14		Н. I.	Specifications. Drop manholes shall be constructed in accordance with File No. 19 of the Standard Specifications. At all manholes, 3/4-inch crusher run stone shall be installed from the top of the cone to the top of the casting.	
15				
16	3.7	CASTIN	NG INSTALLATION	
17		A.	Install casting type as indicated on the plans or in the specifications.	
18 19 20		В.	to match adjacent proposed grades.	
20	3 8	CONNE		
21	5.0	A.	Make all necessary openings into existing structures or sewers including the reconstruction of existing inverts or	
23 24			benches, as necessary. Patch all openings permanently watertight with concrete brick and mortar, or hydraulic cement and waterstops, or for sanitary sewers, hydraulic cement and flexible watertight boots.	
25 26	3 9	SEW/EB	ΖΙΔΤΕΡΔΙς	
20	3.5	A.	Connect existing sewer laterals in accordance with all of the requirements of the sewer mains, including	
28 29		<i>,</i>	bedding, backfill, compaction, and jointing of the pipe. Connect sewer laterals to the sewer main by means of an approved "wye" fitting. Connect the new pipe to the existing lateral material using a no-hub coupling or	
30 31 32		В.	approved transition fitting. Coupling/fitting shall be selected for the specific pipe material being connected. Based on local municipality requirements, cut-in type saddle wyes are not permitted on existing sanitary sewers where service laterals are to be connected to the sewer.	
33 34		C.	Sewer lateral coordinates will be obtained via GPS by the Owner's Construction representative during construction.	
35		D.	All laterals shall have tracer wires from the main to the end.	
36 27	2 10			
38	5.10		Provide insulation when indicated on the Drawings, or where denth of cover is less than 6 feet. Unless	
39		Λ.	otherwise noted, install 2-inch thick polystyrene boards for insulation over the pipe.	
40		В.	Install insulation on compacted initial cover material, 6 inches above the top of the pipe. Stagger joints where	
41			more than one layer of insulation is required. Provide insulation with a minimum of 1 foot of initial cover	
42			material. Place cover and backfill material in manner that does not damage insulation; replace any damaged	
43			insulation.	
44				
45	3.11	LOCAT	OR TAPE	
46		А.	The tracer wire shall be installed on top of all PVC sanitary sewer laterals and taped at 6-foot intervals from the	
47 10			wye at the sewer main to the center of the terrace. A second wire shall be extended from the center of the terrace to the and of the lateral stubbed into the lat. A speel of excess wire shall be provided at the and of the	
40 10			lateral equal to one-half the denth of the lot for use by the building plumber at the time of building connection	
50			The conductor shall be brought to the surface to a test box located directly above the sewer lateral in the center	
51			of the terrace between the back of the curb and the front of the sidewalk.	
52				
53	3.12	DEFLEC	CTION TESTING	
54		Α.	Test all PVC sewer pipe in the presence of the Owner's Construction Representative by a "go-no-go" deflection	
55			test mandrel furnished by the Contractor. Do not perform deflection testing any sooner than 30 days following	
56			the installation of the PVC pipe. Pull the mandrel by hand, or hand operated winch so as to avoid any damages	
57			to the pipe that may be caused by mechanized pulling equipment.	
1 2		В.	Size the to test the pipeline for a maximum allowable internal deflection of the pipe (in any direction) of not to exceed 5 percent (5%) of the original internal diameter for the pipelines tested, regardless of how long after	
--------	------	------------------	---	--
3			installation the testing takes place.	
4		C.	Deflection testing may be done concurrently with any necessary televising of the sewers. When done	
5			concurrently with sewer televising, the mandrel may be pulled by mechanized equipment, provided the mandrel	
6			is visible in the television picture during the testing and the operation of the mandrel can be quickly halted	
7			before damage to the pipe occurs.	
8		D.	Where poor trench soils conditions require the pipe excavation to be undercut and/or over excavated, the	
9			Construction Representative reserves the right to require an additional deflection test prior to the expiration of	
10			the Contractor's one year performance guarantee.	
11		E.	Remove and replace all pipe that fails to pass the five (5) percent vertical deflection testing until the pipe passes	
12			the deflection test.	
13				
14	3.13	LEAK	AGE TESTING	
15		A.	All new sanitary sewer lines shall be leakage tested in accordance with Chapter 3.7.0 of Standard Specifications	
16			for Sewer and Water Construction.	
17		В.	Air leak testing shall be used unless the ground water surface is greater than 2 feet above the top of the sewer	
18			pipe.	
19				
20	3.14	SEWER TELEVISING		
21		A.	Upon completion of the sewer construction all new sewers shall be televised to provide a record of the actual	
22			conditions inside the newly constructed sewers via closed circuit televising equipment. The City or Owner's	
23			Construction Representative may or may not be present during sewer inspections via this method.	
24		В.	Utilize televising equipment with a color camera specially designed and equipped for the conditions of the	
25			sewers to be televised, and with a monitor screen. Provide equipment capable of recovering digital tape, video,	
26			and pictures for future reference.	
27		C.	Transport the camera equipment through the sewers by means of mechanical or hand operated winches,	
28			coordinated to provide speed and directional control necessary to fully observe the sewer interior. Provide a	
29			light source for the necessary illumination.	
30		D.	Provide televising equipment equipped with an on-screen distance meter, capable of registering distances in the	
31			sewer from the starting manhole, and accurate to the nearest 0.5-foot station, so as to facilitate in the locating	
32			of sewer features and/or defects from the ground surface.	
33		E.	Provide televising equipment with an on-screen date and time clock, so as to permit the verification of the date	
34			and time of the television inspection.	
35		F.	Any video tapes of the sewer inspection shall contain audio notes describing the sewer location, direction of	
36			inspection, and a description of any pertinent features observed during the televised inspection (service	
37			locations, leaking or faulty joints, debris in the line, offset joints, etc.). In addition, record this information on a	
38			written log or record, in a format of the Contractor's choosing.	
39		G.	The Contractor shall provide to the Owner's Construction Representative with two (2) copies of the televising	
40			DVD. Electronic video files (MP4 or other format) are also acceptable.	
41				
42			END OF SECTION	

Page Intentionally Left Blank

1	SECTION 33 40 00				
2 3	STORM SEWER CONSTRUCTION				
4	PART 1 - GENERAL				
5		~			
6 7	1.1	SUMIN	ARY	les sterm souver nine and assessaries	
/ Q		A. B	The following sect	ions contain requirements that relate to this Section:	
9		Б.	1 Section 31.23	R 16 13 - Trenching	
10			2 Section 31 25	5 00 - Erosion Control	
11			2. 50000015120		
12	1.2	REFER	ENCES		
13		Α.	ASTM A615-89	Specification for Deformed and Plain Billet-Steel Bars for Concrete Reinforcement	
14		В.	ASTM C76-90	Reinforced Concrete Culvert, Storm Drain and Sewer Pipe	
15 16		C.	ASTM D698-91	Moisture-Density Relations of Soils and Soil-Aggregate Mixtures, Using 5.5-pound (2.5kg) Rammer and 12-inch (304.8-mm) Drop	
17		D.	AASHTO M-198	Joints for Circular Concrete Sewer and Culvert Pipe Using Flexible Water tight Gaskets	
18		Ε.	AASHTO M252	Corrugated HDPE Pipe Underdrain, 3-inch to 10-inch	
19		F.	AWWA C905	PVC Pressure Pipe and Fabricated Fittings, 14 inches through 48 inches for Water	
20				Transmission and Distribution	
21 22		G.	AWWA C906	PE Pressure Pipe and Fittings 4 inches through 63 inches for Water Transmission and Distribution	
23		Н.	NRCS 378	Anti-Seep Collar for Discharge Pipe Through Embankment	
24		١.	Where reference	is made to the "Construction Specifications," it shall be construed to mean the pertinent	
25			section of the City	y of Madison's Standard Construction Specifications, current edition, and all supplemental and	
26			interim suppleme	ental specifications, as they may pertain, except the method of measurement and basis of	
27			payment shall not	apply.	
28		J.	Where reference i	is made to the "Standard Specifications," it shall be construed to mean the pertinent section of	
29			the "Standard S	pecifications for Sewer and Water Construction in Wisconsin," current edition, and all	
30			supplemental and	Interim supplemental specifications, as they may pertain, except the method of measurement	
31		V	and basis of paym	ient snall not apply.	
32 22		к.	"Standard Specifi	is made to the State Specifications, it shall be constructed to mean the pertinent section of the	
27			interim suppleme	antal specifications as they may pertain except the method of measurement and basis of	
34			navment shall not	and specifications, as they may pertain, except the method of measurement and basis of	
36		1	Where reference	is made to the "Geotechnical Report" it shall be construed to mean the geotechnical report in	
37			Section 02 32 00.		
38					
39	1.3	REGUI	LATORY AGENCIES		
40		Α.	City of Madison		
41		В.	The Department of	of Natural Resources (WDNR)	
42					
43	1.4	SUBM	ITTALS		
44		Α.	Submit product da	ata under provisions of Division 1.	
45		В.	Submit product da	ata for pipe and pipe accessories.	
46					
47	1.5	PROJE	CT RECORD DOCU	MENTS	
48		A.	Submit document	s under provisions of Division 1.	
49 50		в.	Accurately record	location of pipe runs, connections, mannole rim elevations, and invert elevations.	
5U 51	16	MEAS		VMENT	
52	1.0	A	Measurement and	a navment for storm sewer construction and related work specified herein shall be by unit price.	
53		73.	as shown on the P	Bid Form. Work shall include all labor, equipment and materials related to storm sewer	
54			construction.		
55			1. Lengths of	storm sewer construction are from center of structure to center of structure.	
56			6		

1 PART 2 - PRODUCTS

2.1	STORM SEWER PIPE MATERIALS		
	A. Reinforced Concrete:		
		1. Pipe: Reinforced concrete pipe meeting requirements of ASTM C76 or ASTM C507. Provide Class III	
		unless indicated otherwise in the Specifications or on the Drawings.	
		2. Joints:	
		a. Circular Pipe: Tongue and groove meeting requirements of ASTM C443.	
		3. Joint ties shall be in accordance with Detail Drawings.	
		4. All Public Storm Sewer shall be concrete pipe	
	в	PVC Pine	
	Б.	Conform to ASTM D3034 with solvent weld or electomeric joints	
		2 Dine shall be SDR-35, unless otherwise noted. Dine over 15 inches in diameter shall meet the	
		2. I requirements of ASTM E679-03. Do not mix different manufacturer's products or fittings	
	C	HDDE Solid Wall Dipa-	
	С.	Conference of ACTION DOTATION FOR THE MATCHINE WITH a coll classification of 225424C or bottor. Bing shall be SDP	
		1. Comorni to ASTW 2550 for FL material with a cell classification of 555454C of better. Fipe shall be 5DK	
		II, unless other wise noted.	
	D	2. Joints shall be thermal but rusion in accordance with the manufacturer's recommendation.	
	D.	nore contugated wait ripe:	
		Corrugated pipe with an integrally formed smooth liner. Bines which are between 4 inch dispertenced 2C inch dispertence hell recet the requirements of AACUTO	
		2. Pipes which are between 4-inch diameter and 36-inch diameter shall meet the requirements of AASH10	
		M252 and M294, Type S.	
		3. Pipe and fittings shall be manufactured from virgin PE compounds conforming to the requirements of	
		ASTM D3350, cell class 324420C.	
		4. Joints for fittings and pipe shall be soil-tight bell and spigot, provided with rubber gasket. Rubber gasket	
	_	shall be installed by the pipe manufacturer.	
	E.	HDPE Corrugated Underdrain Pipe:	
		1. Corrugated underdrain pipe with perforations.	
		2. Pipes which are between 4-inch diameter to 10-inch diameter shall meet the requirements of AASHTO	
		M252, Type CP.	
		3. Joints and wyes for pipe shall be soil tight.	
1.7	STOR	M SEWER PRE-CAST MANHOLES AND INLETS	
	Α.	Frames, grates, and manhole lids shall be constructed and installed per City's Construction Specifications and are	
		considered incidental to the price of the structure.	
		 Manhole castings shall be Neenah R-1550 with non-rocking lids and Type D grates. 	
		2. Inlet castings shall be R-3067 with Type L grates at low points and where grades are less than 1 percent.	
		Vane grates shall not be used at low points.	
1.8	CONN	VECT TO EXISTING STORM MANHOLE OR STORM PIPE	
	Α.	All work and materials to comply with City's Construction Specifications.	
	В.	Contractor to supply all materials, equipment, labor and supervision necessary to construct the required	
		connections per the contract documents.	
1.9	PRE-C	CAST STORM SEWER PIPE END SECTIONS AND WELDED GRATES	
	Α.	Pre-cast Apron end Walls Shall be provided and installed with hinged metal grates in accordance with	
		Section 502 of the State Specifications. Apron end wall shall be provided with cut-off walls to prevent	
		undermining.	
	В.	Apron End Walls shall be tied to a minimum of two pipe segments upstream of the endwall (three joint ties).	
1.10	HDPE	STORM SEWER PIPE END SECTIONS AND WELDED GRATES	
	A.	HDPE apron endwalls shall be provided and installed with removable/hinged metal grates and cut-off walls to	
		prevent undermining.	
	В.	Apron endwalls shall be securely fastened to the HDPE storm pipes with a fastening system that has been	
		reviewed and approved by Engineer prior to installation.	
	2.1 1.7 1.8 1.9	 2.1 STOR A. B. C. D. E. 1.7 STOR A. I.8 CONN A. B. I.9 PRE-C B. I.10 HDPE A. B. 	

1	1.11	BEDDING AND COVER MATERIAL		
2		A. Provide 3/4-inch clear stone bedding and cover material in accordance with the Drawings and Section		
3		31 23 16.13 -Trenching.		
4		В.	Granular Backfill material shall conform to the requirements of the City's Standard Specifications. Granular	
5			backfill shall be used under public and private pavement/walks and where shown in the plans unless otherwise	
6			directed by the City's Construction Representative or Owner's Project Representative based on geotechnical	
7			evaluation of native materials.	
8				
9	1.12	CRUSH	IED STONE	
10		А.	Provide crushed stone base in accordance with the drawings and Section 31 23 16.13 - Trenching.	
11		В.	Stone trench backfill for storm sewer underdrain shall be 3/4-inch washed in accordance with AASHTO #57.	
12				
13	1.13	ANTI-S	SEEP COLLAR	
14		A.	Provide and install anti-seep collar(s) at the locations indicated on the plan, of type and size noted within the	
15			plan details and in compliance with NRCS 378.	
16				
1/	1.14	LOCAT		
18		A.	Detectable metal locator tape shall be specifically manufactured for marking utilities.	
19		в.	Tape shall be a minimum of 6 inches wide and shall be marked with "STORIVI".	
20	1 15			
21	1.15		BASINS AND INLETS	
22		А.	Delicidi.	
23			referenced within the contract specifications. The ductile iron grates for each of these fittings are to be	
24			considered an integral part of the surface drainage inlet and shall be furnished by the same	
25			manufacturor. The surface drainage inlets shall be as manufactured by Nylenlast a division of Advanced	
20			Drainage Systems Inc. or prior approved equal	
27		R	Materiale.	
20		Б.	The drain basins required for this contract shall be manufactured from PVC nine stock utilizing a	
30			thermoforming process to reform the nine stock to the specified configuration. The drainage nine	
31			connection stubs shall be manufactured from PVC nine stock and formed to provide a watertight	
32			connection with the specified nine system. This joint tightness shall conform to ASTM D3212 for joints	
33			for drain and sewer plastic pipe using flexible elastomeric seals. The flexible elastomeric seals shall	
34			conform to ASTM F477. The pipe bell spigot shall be joined to the main body of the drain basin or catch	
35			basin. The raw material used to manufacture the pipe stock that is used to manufacture the main body	
36			and pipe stubs of the surface drainage inlets shall conform to ASTM D1784. Cell Class 12454.	
37			2. The grates and frames furnished for all surface drainage inlets shall be ductile iron for sizes 8, 10, 12, 15,	
38			18. 24, and 30 inches and shall be made specifically for each basin so as to provide a round bottom flange	
39			that closely matches the diameter of the surface drainage inlet. Grates for drain basins shall be capable	
40			of supporting various wheel loads as specified by Nyloplast. The 12-inch and 15-inch square grates will	
41			be hinged to the frame using pins. Ductile iron used in the manufacture of the castings shall conform to	
42			ASTM A536 grade 70-50-05. Grates and covers shall be provided painted black.	
43			3. Where shown, the basins shall be provided with the sump specified.	
44				
45	1.16	STORM	/ FORCE MAIN PIPE	
46		Α.	Stormwater force main shall be Polyvinyl Chloride (PVC) meeting the requirements of AWWA C900 or C905 as	
47			applicable.	
48			1. Unless otherwise noted, PVC force main pipe shall have a dimension ratio (DR) of 25 or less.	
49			2. Retainer Glands for PVC Pipe:	
50			a. Wedge action retainer glands designed for use with PVC pipe.	
51			b. Glands shall be constructed of ductile iron. Restraint shall be provided by a minimum of four (4)	
52			wedges which are tightened onto the exterior of the pipe using a threaded, torque limiting	
53			mechanism.	
54			c. Glands shall be tested to provide restraint at 200 psi operating pressure.	
55			 Retainer glands shall be MEGA-LUG by EBBA Iron, or approved equal. 	
50			3. IVIAIN Clamp Hardware:	
5/			 a. Corrosion resistant steel nardware specifically fabricated for use in pipe restraint systems. 	
58			b. Astrai, or approved equal.	

1			4. Threaded Rod Joint Restraint:
2			a. 3/4-inch diameter threaded rod.
3			b. Rod shall be constructed of carbon steel having a minimum tensile strength of 30 ksi.
4			c. Rod shall be zinc plated.
5			5. Fittings shall be ductile iron meeting AWWA C151 or C153.26.
6			
7	1.17	DUCTI	LE IRON STORM FORCE MAIN
8		Α.	Ductile iron force main shall be centrifugally cast, cement mortar lined ductile iron force main meeting the
9			requirements of ANSI/AWWA C151/A21.51 and ANSI/AWWA C104/A21.4.
10		В.	Unless otherwise specified, ductile force main shall be Class 52 as defined by ANSI/AWWA C151/A21.51.
11			
12	PART	<u>2 - EXEC</u>	CUTION
13			
14	2.1	HAND	LING OF MATERIALS
15		Α.	Handle materials with care to avoid damage. Do not dump or drop materials. Remove all damaged or flawed
16			materials from the site.
17		В.	Arrange for suitable sites for material storage.
18			
19	2.2	LINES	AND GRADE
20		Α.	Benchmarks and Construction Layout:
21			1. Engineer will provide vertical and horizontal control.
22			2. Contractor shall provide construction layout.
23		В.	Contractor shall provide all materials, equipment, and labor to maintain line and grade.
24			1. The laser beam method is the preferred method for controlling line and grade. Equipment shall be
25			operated in accordance with the manufacturer's instructions. A person who is competent with the
26			operation of the laser equipment shall be present at the jobsite whenever it is being used.
27			2. Grade boards may be used. Use straight and even-edged 2X6 boards nailed or clamped to substantial
28			stakes on either side of the trench. Use stout twill line fastened at the center of the alignment, pulled
29			sufficiently tight to remove any noticeable or measurable sag. Measure down from the line to set the
30			alignment of the pipe. Maintain a minimum of three boards at all times.
31			3. Banjo strings may be used only when approved by the Engineer.
32			-
33	2.3	JOINT	S
34		Α.	Construct joints as described herein and in accordance with manufacturer's installation instructions. Provide
35			pipe joint type for soil tight, silt tight, or watertight only silt tight or watertight only watertight joint performance

- 36 37

in accordance with the following table. The table applies only to the extent as applicable to the pipe and joint type and the joint performance as shown or specified.

Pipe and Joint Type	Joint Performance			
	Watertight	Silt Tight	Soil Tight	
RCP:				
Rubber O-Ring Gasket	Х	х	Х	
Bitumen or Butyl Rubber Sealant			х	
HDPE:				
Rubber Gasket:				
Hancor BLUE SEAL	Х	х	Х	
ADS N-12 WT	Х	х	Х	
Hancor Sure-Lok		х	Х	
ADS N-12 ST		х	Х	
Corrugated Coupling Bands:				
Hancor Hi-Q			Х	
ADS N-12			Х	
PE Wrap			Х	
PVC:				
Restrained Gasket	Х	х	х	

1	2.4	UNSTABLE FOUNDATION		
2		Α.	Remove undesirable material below the trench bottom, such as organic soils, which cannot support the pipe.	
3			Replace the material with crushed stone meeting the requirements of Section 31 23 16.13 for 2-inch crushed	
4			stone base material.	
5		В.	Crushed stone base material will be paid for at the unit price bid or on the basis of a negotiated price if there is	
6			no bid price. Payment for crushed stone base will be made only if the Owner's Project Representative is notified	
7			prior to its placement. Payment will not be made for crushed stone base used for dewatering the trench.	
8				
9	2.5	LAYIN	G OF PIPE	
10		Α.	Lay pipe uniformly to line and grade so that the finished sewer presents a uniform bore. Noticeable variations	
11			from true alignment and grade will be sufficient cause for rejection of the work.	
12		В.	Commence at the lowest point and proceed to the upper end. Lay pipe with bell-end pointing up-grade.	
13		C.	Provide a minimum of 6 inches between the pipe or box wall and the trench wall.	
14		D.	Rest each pipe on the full length of its barrel. Place box culvert sections on 6 inches of bedding material.	
15		E.	Do not lay the next pipe until the previous pipe is back-filled sufficiently to prevent movement during joining.	
16		F.	Keep water out of the pipe. Do not let water rise into or around the pipe until the trench is filled at least 1 foot	
17			above the pipe.	
18		G.	When work is stopped for any reason, securely plug the end of the pipe.	
19		Н.	Pipe Jointing: Assemble joints in accordance with the pipe manufacturer's instructions.	
20				
21	2.6	BEDD	ING AND COVER	
22		Α.	Use the following bedding sections as indicated on the Drawings.	
23		В.	Class C:	
24			1. Provide a minimum of 6 inches of bedding material under the pipe barrel and 4 inches under the bell.	
25			Provide crushed stone bedding meeting requirements of Section 31 23 16.13. Spade or shovel-slice the	
26			material so that it fills and supports the haunch area and encases the pipe to the limits shown on the	
27			Drawing detail. If excavation is carried deeper than 6 inches below the pipe barrel, backfill the excess	
28			depth with 1½-inch crushed stone base material meeting requirements of Section 31 23 16.13.	
29			2. After the pipe has been laid and jointed, place cover material by hand or equally careful means around	
30			the sides of the pipe and up to a level twelve inches above the pipe. Provide cover material meeting the	
31			requirements of Section 31 23 16.13.	
32			3. For pipes 36 inches in diameter or larger, backfill material may be substituted for cover material. If	
33			backfill material is used, the bedding material shall extend to the spring line of the pipe.	
34				
35	2.7	SEPAR	RATION FROM WATER MAIN	
36		Α.	Provide a minimum horizontal separation of 10 feet when constructing parallel to the water main.	
37		В.	Vertical Separation	
38			1. When a sewer crosses under a water main, provide a minimum of 12 inches between the bottom of the	
39			water main and the top of the sewer.	
40			2. When a sewer crosses over a water main, provide a minimum of 36 inches between the bottom of the	
41			sewer and the top of the water main.	
42		C.	Excess excavated trench material shall be transported and disposed on site in a location approved by the	
43			Construction Coordinator or Engineer. Trench excavation material cannot be cast into piles within the roadway.	
44				
45	2.8	MANH	HOLES, CATCH BASINS, INLETS, AND JUNCTION BOXES	
46		Α.	Construct drainage structures in accordance with details shown on Drawings and in accordance with Section	
47			33 40 00, as applicable.	
48		В.	Precast Sections:	
49			1. Install precast section with bases in accordance with Sections 33 30 00 and 33 40 00 or as shown on	
50			Drawings.	
51			2. Align pipe openings to that of the pipe entering and leaving the manhole, etc. Properly pipe with	
52			connections to manholes, etc., as shown on the Drawings.	
53			a. Construct cast-in-place sections as shown on the Drawings and in accordance with Section	
54			03 30 00 - Cast-in-Place Concrete.	
55			3. Form bottom of excavation clean and smooth to correct elevation.	
56			4. Form and place cast-in-place concrete base pad, with provision for storm sewer pipe to be placed at	
57			proper elevation.	

1 2 3 4 5 6 7		C.	 Form and place cast-in-place concrete walls, sleeved at proper elevation to receive storm sewer pipe in accordance with details shown on Drawings. Invert channels shall be smooth and accurately shaped to a semicircular bottom conforming to the inside of the adjacent sewer section. Shape invert channels and structure bottoms with cement mortar. Changes in size and grade of invert shall be made gradually and evenly. Changes in direction of the sewer entering branch or branches shall have a true curve of as large a radius as the manhole will permit. 	
8	3.9	TELEV	ISE REPORT	
9 10 11 12	0.0	А. В.	All new storm sewer installation completed between structures shall be televise once backfilled, and a copy of the television report shall be submitted to the City's Construction Representative and the Engineer. The Contractor is responsible for conducting and reporting the results of testing. The televising of the storm sewer shall be done in the presence of the City's Construction Representative.	
13				
14	3.10	CONDUCTIVITY TESTING		
15 16		A.	Perform continuity testing on all newly installed storm sewer mains in accordance with the City's Construction Standards.	
17		В.	The Contractor is responsible for conducting and reporting the results of testing. Conductivity testing shall be	
18			done in the presence of the City's Construction Representative.	
19				
20			END OF SECTION	